

# Second Austrian Assessment Report on Climate Change | AAR2

**Full Report** 

## **Second Austrian Assessment Report** on Climate Change | AAR2

This publication was subject to an anonymous international peer review process guided by several review editors, the AAR2 co-chairs and the APCC Steering Committee.

The views or opinions expressed in this publication are not necessarily those of the institutions with whose support this work was carried out. The publisher and the supporting institutions do not guarantee the accuracy or continued availability of urls to external websites or third-party websites mentioned in this publication and accept no responsibility for the accuracy or appropriateness of any content contained therein.

#### **Recommended citation**

APCC (2025): "Second Austrian Assessment Report on Climate Change (AAR2) of the Austrian Panel on Climate Change (APCC)". [D. Huppmann, M. Keiler, K. Riahi, H. Rieder (eds.)]. Austrian Academy of Sciences Press, Vienna, Austria | doi: doi.org/10.1553/aar2

You can download the recommended citation in the formats Endnote (.ris) and BibTex (.bib) at aar2.ccca.ac.at/citation.

Vienna, June 2025 ISBN: 978-3-7001-5110-4 | austriaca.at/5110-4 (full report)

#### Production

Publisher: Austrian Academy of Sciences Press, Vienna Typesetting: Crossdesign Werbeagentur GmbH, Graz Cover: paulaidaanton | Agentur für digiloges Design Graphic design: CCBox 1 Figure 1, CCBox 1 Figure 2, CCBox 1 Figure 3, Table 2.2, Figure 3.2, Figure 4.1, CCBox 6 Figure 1, Figure 5.3, Figure 5.6, Table 6.2, Figure 7.3, Figure 8.4, Figure 8.5: paulaidaanton | Agentur für digiloges Design Al-based tools such as DeepL were used to support the linguistic revision.

Contact address for product safety: Austrian Academy of Sciences Press Dr. Ignaz Seipel-Platz 2 A-1010 Wien Tel.: +43-1-51581-3420 E-Mail: verlag@oeaw.ac.at

#### Funding acknowledgement

The AAR2 was funded by the Climate and Energy Fund under the 14th Austrian Climate Research Program (ACRP).

AUSTRIA

#### **Copyright and license**

© 2025 Austrian Panel on Climate Change (APCC) This report is published under a Creative Commons CC-BY 4.0-License. creativecommons.org/licenses/by/4.0



#### Coordination



OF SCIENCES Climate Change Centre PRESS

The AAR2 is a report by the Austrian Panel on Climate Change (APCC). https://aar2.ccca.ac.at/en



# Bundespräsident Alexander Van der Bellen

200 Wissenschaftler:innen. 100 externe Gutachter:innen. 3 Jahre intensive Arbeit.
Zahllose Daten und Analysen. Das Ergebnis davon halten wir nun in Händen: den Zweiten Österreichischen Sachstandsbericht zum Klimawandel.
Es ist ein Blick in den Spiegel, ein Realitycheck, der uns zeigt, wie sich die
Klimakrise in den letzten Jahren beschleunigt hat – global und hier in Österreich.
Hitzewellen, Dürren und Hochwässer sind Realität geworden, und das mit gravierenden Folgen für Gesundheit, Wohlstand und Wirtschaft.

Eine gute Nachricht gibt es, und zwar dass die heimischen Treibhausgas-Emissionen auf einem historischen Tiefstand sind. Großartig! Aber wie so oft bei guten Nachrichten hat auch diese einen Haken, nämlich sind wir – fast schon typisch gemütlich Österreichisch – ein bisschen spät dran, was unsere Emissionsreduktion betrifft. Wollen wir bis 2040 klimaneutral sein, müssen wir nicht nur vieles ändern, sondern vieles rasch ändern. In der Energieversorgung, Mobilität, Ernährung und mehr. Das ist natürlich nicht einfach. Und es kostet! Aber der Bericht macht klar: Nichts zu tun kostet uns noch viel mehr. Und damit meine ich gar nicht Geld, sondern etwas viel Wertvolleres: unsere Zukunft. Klimaschutz lohnt sich allemal – ökologisch, sozial und wirtschaftlich. Entscheidend ist, dass wir ihn nicht isoliert von allem rundherum betrachten, sondern Biodiversität, Energieunabhängigkeit und vor allem auch soziale Gerechtigkeit einbeziehen. Denn die Klimakrise trifft sozial Schwächere am stärksten.

Dieser Sachstandsbericht ist aber nicht nur ein Weckruf, der uns zeigt, wo wir stehen. Sondern zugleich auch eine Anregung, die uns den Weg in eine nachhaltige Zukunft zeigt. Die Klimakrise ist real, die allermeisten Menschen wissen das. Sie fordern mehr Mut, mehr Taten – und ich schließe mich da an. Ich bin sicher, dieser Bericht ist dazu ein geeigneter Anstoß.



A. C. Adlen

Bundesministerium Land- und Forstwirtschaft, Klima- und Umweltschutz, Regionen und Wasserwirtschaft



Bundesminister Mag. Norbert Totschnig, MSc

Mit dem Zweiten Österreichischen Sachstandsbericht zum Klimawandel liegt nach nun dreijähriger interdisziplinärer Forschungsarbeit eine der fundiertesten und umfassendsten Analysen der Klimawandelfolgen sowie der möglichen Handlungsoptionen für Österreich vor. Dafür möchte ich mich bei den rund 200 beteiligten Expertinnen und Experten herzlich bedanken.

Die Ergebnisse zeigen die Herausforderungen in aller Deutlichkeit. In Österreich sind die Folgen des Klimawandels besonders spürbar. Unser Land erhitzt sich stärker als der globale Durchschnitt. Die Folgen zeigen sich längst in allen Lebens- und Wirtschaftsbereichen – von Land- und Forstwirtschaft und Gesundheit bis hin zu Infrastruktur, Tourismus und Ökosystemen.

2024 war das wärmste Jahr der Messgeschichte mit belastenden Hitzeperioden und schweren Überflutungen. Mit fortschreitendem Klimawandel verstärken sich diese Extreme und deren Auswirkungen. Angesichts dessen braucht es ambitionierte Maßnahmen mit gezielten Investitionen und ein abgestimmtes Vorgehen beim Klimaschutz und der Anpassung an den Klimawandel – quer durch alle Sektoren. Wir müssen sicherstellen, dass unsere Energieversorgung, Mobilität, Wärmebereitstellung, Industrie sowie Land- und Forstwirtschaft zukunftsfähig ausgestaltet sind und dabei gleichzeitig unsere Wirtschaft und unser Standort gestärkt aus der Transformation hervorgehen.

Wenn wir es richtig angehen, können Investitionen in Klimaschutz und Anpassung positive Auswirkungen auf Beschäftigung, Einkommen und Wohlstand haben und die Abhängigkeit von Energieimporten sowie Folgekosten verringern. Das Ziel der Klimaneutralität ist im Regierungsprogramm verankert. Wir arbeiten auf Hochtouren an einem Klimagesetz, um damit den Rahmen für abgestimmtes Handeln zu setzen. Der Sachstandsbericht zeigt eindrücklich, dass wir rasch und entschlossen handeln müssen – als Gesellschaft, als Politik, als Einzelne. Gemeinsam können wir unsere Natur, Wirtschaft und unser Leben resilienter und nachhaltiger machen – darauf kommt es jetzt an. Ich bin davon überzeugt, dass wir unsere Klimaziele erreichen können, wenn wir gemeinsam an Lösungen arbeiten, die ökonomisch sinnvoll und umsetzbar sowie sozial tragfähig sind.

Nebert Johning

### **TABLE OF CONTENTS**

AAR2 Preface	VII
Summaries	Х

#### Chapter 1:

Phys	sical and ecological manifestation of climate	
char	nge in Austria	1
Exec	utive summary	5
1.1.	Chapter introduction	8
1.2.	Atmosphere	10
1.3.	Cryosphere	31
	Cross-Chapter Box 1	
	Evaluation of climate-related risks and	
	key risks	32
1.4.	Hydrosphere	41
1.5.	Pedosphere	48
1.6.	Biosphere	50
1.7.	Lithosphere	57
1.8.	Cross-cutting topic: Natural hazards	60
Refe	rences	62

#### Chapter 2:

Clim	ate change, land use, ecosystem services	
and	health	83
Exec	utive summary	88
2.1.	Chapter Introduction	91
2.2.	Terrestrial ecosystem services under the	
	influence of climate change	98
	Cross-Chapter Box 2	
	Health and climate change	110
2.3.	Transformations and measures in the land	
	use system to mitigate and adapt to climate	
	change	118
2.4.	Conflicts, lock-ins and enabling conditions –	
	focusing on the example of agricultural land	
	use	138
Refe	rences	143
Cha	-4 7.	
Chap	oler 5:	
Built	environments and mobility	171
Exec	utive summary	175
3.1.	Chapter introduction	177

3.4. Transport and infrastructure	198
Conclusion	214
References	215

#### Chapter 4:

Prov	rision of goods and services in a climate-	
resil	ient economy via materials, energy and work	231
Exec	utive summary	235
4.1.	Chapter introduction	239
4.2.	Provisioning of goods and services	241
	Cross-Chapter Box 3	
	Needs orientation in provisioning systems	248
	Cross-Chapter Box 4	
	Avoid – Shift – Improve: The ASI-framework	249
	Cross-Chapter Box 5	
	Circular economy	252
4.3.	The case of tourism and climate change	256
4.4.	Industry and manufacturing	259
4.5.	Energy supply systems	265
4.6.	Decarbonization of hard-to-abate industries	
	by carbon capture, geological storage, and	
	utilization	274
	Cross-Chapter Box 6	
	Carbon dioxide removal	275
4.7.	Provision by work	281
Refe	rences	286

#### Chapter 5:

Navi achie	gating demand-side transformations to eve net zero through human decisions and	
beha	avior	308
Exec	utive summary	312
5.1.	Chapter introduction	314
5.2.	Well-being and inequality related to climate	
	mitigation and adaptation	316
5.3.	Mitigation and adaptation decisions of citizens	
	and households: Lifestyles and consumption $\ldots$	325
5.4.	Organizations as agents of change	337
5.5.	Circular and sharing economy and the role	
	of digitalization	343
5.6.	Distributional justice, social tipping points and	
	social acceptance	351
Refe	rences	359

#### Chapter 6:

Clim	ate governance: Political, legal, economic	
and	societal aspects	383
Exec	utive summary	387
6.1.	Introduction	389
6.2.	The structural and dynamic context of climate	
	governance	389
	Cross-Chapter Box 7	
	Degrowth and other beyond growth	
	concepts: The desirability and feasibility	
	of (limiting) growth	391
6.3.	Regulatory framework of adaptation and	
	mitigation	396
6.4.	Institutions and actors of climate governance	
	in Austria	398
6.5.	Policy instruments for	
	climate change mitigation and adaptation	404
6.6.	Climate policy litigation	418
6.7.	Budget, green public finances, investments	
	and cost of (in)action	419
6.8.	Societal aspects	424
References		432
Chap	oter 7:	

#### . . . .

The Austrian Alps as multi-dimensional focal a	area	457
Executive summary		461
7.1. Chapter introduction		463

AAR2

7.2.	Delineation and differentiation of the	
	Austrian Alps	464
7.3.	Manifestation of climate change in the	
	Austrian Alps	466
7.4.	Impacts of climate change on the	
	Austrian Alps	469
7.5.	Needs for resilient mountain regions in Austria	493
Refe	rences	497
Chap	oter 8:	
Tran	sformation pathways	513
Exec	utive Summary	517
8.1.	Introduction	520
8.2.	Biophysical limits and carbon budgets	520
8.3.	Synthesis and assessment of mitigation	
	options	524
8.4.	Assessment of Austrian mitigation scenarios	529
8.5.	Politico-economic and socio-cultural feasibility	
	challenges for climate-friendly transformation	
	pathways	543

8.6. Climate-friendly transformation pathways ...... 556

Green Finance 565

Cross-Chapter Box 8

8.7. Synthesis of policy-relevant insights of

#### **AAR2 PREFACE**

Daniel Huppmann, Margreth Keiler, Keywan Riahi, Harald Rieder

The effects of climate change can already be observed in Austria – it is reshaping life and landscapes in Austria. Detectable changes include extended dry periods, increasing frequency of heatwaves and floods, and retreating glaciers, which are consistent with global trends in a warming world. These impacts, driven by anthropogenic greenhouse gas emissions, have become more frequent, intense, and widespread, with increasing risks for natural systems, the economy, public health, and infrastructure. The regional manifestation of these changes in alpine as well as lowland areas demands targeted responses from all parts of society.

At the same time, Austria has committed itself to ambitious climate goals at the national, European, and international level. These include the Austrian target of climate neutrality by 2040, the legally binding European Climate Law (greenhouse gas neutrality by 2050), and Austria's contribution to the Paris Agreement of limiting global warming to well below 2 °C. Emission reductions observed since 2022 indicate a shift of the long-term trend, but the country's per capita and consumption-based emissions remain high and significantly above the EU average.

#### Aim and scope

The Second Austrian Assessment Report on Climate Change (AAR2) provides a systematic and comprehensive assessment of the scientific literature on climate change for Austria across all relevant scientific domains. It builds upon the First Austrian Assessment Report on Climate Change published in 2014 and a series of subsequent Special Reports published by the Austrian Panel on Climate Change (APCC) under the auspices of the Climate Change Centre Austria (CCCA).

The objective of the AAR2 is to inform an evidence-based societal transformation by assessing the physical, ecological, economic, and social implications of climate change in Austria. More than 200 scientists collaborated to produce this report. It offers a science-based foundation for policymakers and public dialogue on the portfolio of options for emissions reduction and adaptation. The assessment does not advocate for specific policy instruments but highlights the risks, op-

portunities, barriers, trade-offs and synergies of alternative climate-friendly transformation pathways. This includes interactions with sustainable development and societal well-being, notably the Sustainable Development Goals (SDGs).

#### Structure of the report

The AAR2 combines insights from various scientific perspectives including natural sciences, engineering, social sciences, economics, law, and public health. This integration is essential for assessing the multi-faceted challenges related to climate change. It reflects Austria's diverse institutional, ecological, and socio-economic contexts and strengthens the report's systemic and interdisciplinary perspective.

The report consists of eight chapters.

Chapter 1 provides an overview of past, current, and projected changes in Austria's climate system. Key climate indicators are analyzed across the atmosphere, cryosphere, hydrosphere, biosphere, and lithosphere. A Cross-Chapter Box on Risk introduces the IPCC risk framework and identifying key climate-related risks and their evolution with rising global warming levels (GWLs).

Chapter 2 assesses the impacts of climate change on land use, ecosystem services, and human health with a focus on climate-sensitive sectors such as agriculture and forestry. The chapter explores both domestic and international supply chains and identifies co-benefits from nature-based solutions. Mitigation in land systems focuses on sustainable consumption and ecosystem-based carbon sinks.

Chapter 3 focuses on the systems that structure Austria's built environment, including settlements, spatial planning, buildings, and transport. It examines how urban form, infrastructure, and mobility systems influence emissions, energy demand, and climate risks. Mitigation strategies include compact development, electrification of heating and transport, and promotion of low-carbon mobility. Adaptation priorities include resilient infrastructure, urban heat mitigation, and inclusive spatial planning. **Chapter 4** focuses on the **provision of goods and services in a climate-resilient economy** with emphasis on material provisioning, energy systems, and labor. The chapter includes a Cross-Chapter Box on the Avoid–Shift–Improve (ASI) framework. Mitigation in provisioning systems relies on decarbonization of energy and industry, circular economy principles, and sufficiency-oriented approaches. Adaptation focuses on reducing systemic vulnerability through diversification of supply chains, strengthening infrastructure resilience, and building institutional capacity.

Chapter 5 investigates the demand-side transformations necessary to achieve net zero through human decisions and behavior, with particular emphasis on how household consumption, social norms, and lifestyles contribute to greenhouse gas emissions and climate vulnerability. It also assesses inequality, collective action, and the enabling role of institutions, education, and civil society.

Chapter 6 reviews the legal, political, economic, and societal frameworks for climate governance in Austria. The analysis covers policy instruments, just transition mechanisms and financing needs as well as governance challenges and capacity gaps.

**Chapter 7** focuses on the Alps as an area of particular relevance for Austria. The Alps are examined on three levels: As a natural or near-natural environment, as a provider of services, goods and livelihoods, and as a home to communities. The chapter highlights climate-related impacts and adaptation needs with an integrated environmental and socio-economic perspective.

Chapter 8 synthesizes knowledge on transformation pathways toward climate neutrality. It presents integrated scenario assessments of emissions, energy demand, carbon budgets, and adaptation to examine the feasibility of different pathways from a biophysical, economic, and socio-cultural perspective.

# Assessment approach and methodological frameworks

The assessment follows the procedures and methods established by the Intergovernmental Panel on Climate Change (IPCC). Key findings are formulated with a level of confidence as assessed by the author team and using the calibrated language of the IPCC to indicate the strength of the scientific evidence and level of agreement in the literature. In situations where available evidence and region-specific published literature was limited or insufficient for a reliable assessment, the authors assessed evidence from other regions if the insights were transferable and relevant for Austria.

The IPCC's further procedures and methods were adapted for the AAR2, and the various approaches and methodological frameworks applied in the AAR2 are discussed in cross-chapter and chapter boxes throughout the report.

This assessment evaluates climate-related **risks** related to climate change following the IPCC risk concept, which considers the interaction of **hazard**, **exposure**, **and vulnerabil-ity** {Cross-Chapter Box 1}. This enables a differentiated understanding of how and why risks emerge and escalate. The report identifies the most relevant key risks for Austria and discusses options to increase climate resilience, acknowl-edging limits of adaptation.

The risks related to climate change depend directly on the speed of global temperature increase and the implied changes on climatic conditions in Austria. Rather than describing the climatic conditions in a particular set of years (2030, 2040, 2050), the report evaluates impacts from climate change when certain thresholds of global temperatures are crossed (1.5°C, 2°C, 3°C, 4°C), the so-called **Global Warming Levels** (GWL). This approach reflects the physical basis of climate change and aligns with international policy targets. It enables the identification of climate tipping points, changes in the level of risk, and the effectiveness of adaptive responses at different warming levels. GWLs also allow integration across sectors and scales and ensure policy relevance under conditions of timing uncertainty {Box 1.1}.

The report explores both demand-side and supply-side options to reduce greenhouse gas emissions, including changes to the provisioning systems. It is generally preferable to prioritize (1) the avoidance of climate-harmful activities, (2) the shift toward sustainable alternatives, and (3) the improvement of efficiency in existing systems. This is known as the **Avoid-Shift-Improve (ASI) framework** {Cross-Chapter Box 4}. Wherever possible, this report presents an assessment of options that follows the avoid-shift-improve rationale. Not all sectors will be able to reach carbon-neutrality within the timeframes currently discussed by policymakers. As a complementary measure to ambitious mitigation, **Carbon Dioxide Removal** (CDR) can help balance residual emissions in hard-to-abate sectors. The portfolio comprises nature-based solutions (e.g., increasing natural sinks through afforestation) and technical approaches, but CDR is constrained by costs, land-use conflicts, and limited social acceptance {Cross-Chapter Box 6}.

The transformation of the Austrian society towards climate resilience will require substantial changes to the economy, reducing emissions and resource use while ensuring sufficient resources for well-being of every person living in Austria. Two related concepts to reduce resource-intensity of goods and services are the notion of a circular economy (maximize re-use of resources, minimize waste) {Cross-Chapter Box 5} and a sharing economy (moving from ownership of goods to use-as-a-service) {Section 5.5}. In addition, there is an ongoing debate whether continued economic growth is a prerequisite to support the transition or whether a deliberate focus on reducing material consumption is a more effective strategy to ensuring adequate living conditions for every person living in Austria {Cross-Chapter Box 7}.

Further overarching topics across the AAR2 are health and climate change {Cross-Chapter Box 2}, the needs orientation in provisioning systems {Cross-Chapter Box 3}, and green finance, i.e. the redirection of investments to facilitate transformations while ensuring growth and competitiveness and societal well-being {Cross-Chapter Box 8}.

The report draws on a range of **quantitative analysis and model-based scenarios** to explore plausible futures for Austria under different emissions trajectories. This allows to understand the feasibility, costs, and trade-offs of various climate policy options in the energy system, industry, transportation, and land use. It illustrates the broad range of policy choices towards a climate-resilient society that are available to policymakers and society while navigating the transformation to climate neutrality.

#### Acknowledgement

We express our deepest gratitude for the contribution and commitment of the entire author team. We appreciate the dedication and hard work of the Technical Support Unit and Chapter Scientists who supported the chapters and cross-chapter teams in the development of the report. We are grateful for the commitment and diligence of the Review Editors, which were critical in assisting the author teams and ensuring the integrity of the review process. We would also like to thank all the expert reviewers who submitted comments on the drafts to ensure that the final report meets the highest levels of scientific rigor and quality, and we are indebted to all the stakeholders that participated in multiple workshops to help developing a scientific assessment that is hopefully useful and usable for decisionmakers at all levels of society.

The Steering Committee of the Austrian Panel on Climate Change (APCC) and the Climate Change Centre Austria (CCCA) have been instrumental for guiding the AAR2 from the initial scoping phase to its publication over a period of more than five years. We acknowledge the funding and the support from the Climate and Energy Fund throughout the entire process that facilitated the development of the report. We thank the production team including translation, layout, graphic design, website development and public relations, which did an amazing job under tight deadlines. Last, but not least, we appreciate the support of all universities, institutes and research organizations that facilitated the contribution of the authors and thereby enabled this report as an accomplishment of the Austrian climate science community.

#### A call to action

The report highlights that achieving climate neutrality by 2040 requires urgent action across all parts of society – supported by science-based planning and governance, effective climate policy, and broad engagement with civil society. The AAR2 aims to support this transformation by providing a shared knowledge base for public debate, institutional reform, and cross-sectoral cooperation. We hope that this assessment report provides relevant input for effective climate action.

#### **SUMMARIES**

The *Summary for Policymakers* (SPM) is available at the following links: <u>https://aar2.ccca.ac.at/summary</u> (in English) and <u>https://aar2.ccca.ac.at/zusammenfassung</u> (in German).

The *Technical Summary* (TS) is available at the following links <u>https://aar2.ccca.ac.at/technical-summary</u> (in English) and <u>https://aar2.ccca.ac.at/wissenschaftliche-zusammenfassung</u> (in German).





**Chapter 1** 

# Physical and ecological manifestation of climate change in Austria

### Second Austrian Assessment Report on Climate Change | AAR2

### Chapter 1 Physical and ecological manifestation of climate change in Austria

This publication was subject to an anonymous international peer review process guided by several review editors, the AAR2 co-chairs and the APCC Steering Committee.

The views or opinions expressed in this publication are not necessarily those of the institutions with whose support this work was carried out. The publisher and the supporting institutions do not guarantee the accuracy or continued availability of urls to external websites or third-party websites mentioned in this publication and accept no responsibility for the accuracy or appropriateness of any content contained therein.

#### **Recommended citation**

H. Formayer, J. Parajka, J.S. Petermann, et al. (2025): Physical and ecological manifestation of climate change in Austria.
In "Second Austrian Assessment Report on Climate Change (AAR2) of the Austrian Panel on Climate Change (APCC)".
[D. Huppmann, M. Keiler, K. Riahi, H. Rieder (eds.)]. Austrian Academy of Sciences Press, Vienna, Austria | doi: doi.org/10.1553/aar2-ch1

You can download the recommended citation in the formats Endnote (.ris) and BibTex (.bib) at <u>aar2.ccca.ac.at/citation</u>.

Vienna, June 2025 ISBN: 978-3-7001-5110-4 | austriaca.at/5110-4 (full report)

#### Production

Publisher: Austrian Academy of Sciences Press, Vienna Typesetting: Crossdesign Werbeagentur GmbH, Graz Cover: paulaidaanton | Agentur für digiloges Design Graphic design CCBox 1 Figure 1, CCBox 1 Figure 2, CCBox 1 Figure 3: paulaidaanton | Agentur für digiloges Design Al-based tools such as DeepL were used to support the linguistic revision.

Contact address for product safety: Austrian Academy of Sciences Press Dr. Ignaz Seipel-Platz 2 A-1010 Wien Tel.: +43-1-51581-3420 E-Mail: verlag@oeaw.ac.at

#### Funding acknowledgement

The AAR2 was funded by the Climate and Energy Fund under the 14th Austrian Climate Research Program (ACRP).

#### **Copyright and license**

© 2025 Austrian Panel on Climate Change (APCC) This report is published under a Creative Commons CC-BY 4.0-License. <u>creativecommons.org/licenses/by/4.0</u>





The AAR2 is a report by the Austrian Panel on Climate Change (APCC). https://aar2.ccca.ac.at/en

#### **Coordinating Lead Authors**

Herbert Formayer Juraj Parajka Jana S. Petermann

#### **Lead Authors**

Andreas Baumgarten Benedikt Becsi Stefan Dullinger Stephan Glatzel Andreas Gobiet Martin Kainz Gregor Laaha Manfred Lexer Douglas Maraun Lindsey Nicholson Marc Olefs Christine Stumpp Christian Zangerl

#### **Contributing Authors**

**Birgit Bednar-Friedl Sven Fuchs** Manfred Ganekind Katharina Gangl Veronika Gaube Sarah Haider-Nash Susanne Hanger-Kopp **Klaus Haslinger** Johann Hiebl Martin Jury **Roland Kaitna** Margreth Keiler Stefan Kienberger Konrad Mayer Albert Osso Castillon Stefanie Peer Georg Pistotnik **Thomas Schinko** Jan Steinhauser Anna-Maria Tilg

### **Chapter Scientist**

Benedikt Becsi

### **Review Editors**

Georg Kaser Massimiliano Zappa BOKU University Technical University Vienna | TU Wien Paris Lodron University of Salzburg | PLUS

Austrian Agency for Health and Food Safety | AGES BOKU University University of Vienna University of Vienna GeoSphere Austria University for Continuing Education Krems BOKU University BOKU University University of Graz University of Innsbruck | UIBK GeoSphere Austria BOKU University BOKU University

University of Graz **BOKU University** GeoSphere Austria Institute for Advanced Studies | IHS **BOKU University** University for Continuing Education Krems International Institute for Applied Systems Analysis | IIASA GeoSphere Austria GeoSphere Austria City of Graz **BOKU University** University of Innsbruck | UIBK & Austrian Academy of Sciences | ÖAW GeoSphere Austria GeoSphere Austria University of Graz Vienna University of Economics and Business | WU **GeoSphere Austria** International Institute for Applied Systems Analysis | IIASA International Institute for Applied Systems Analysis | IIASA GeoSphere Austria

#### **BOKU University**

University of Innsbruck | UIBK Swiss Federal Institute for Forest, Snow and Landscape Research | WSL

#### Liaison from AAR2 Technical Support Unit

Monika Mayer

#### **TABLE OF CONTENTS**

Exec	utive s	summary	5
1.1.	Chapt	er introduction	8
	1.1.1	Setting the scene	8
	1.1.2	Physiographic setting of Austria	8
	1.1.3	Climatological conditions	9
1.2.	Atmos	sphere	10
	1.2.1.	Past changes	10
		Observed changes in global climate drivers	10
		Observed changes in large scale circulation, weather patterns and modes of variability	10
		Observed changes in selected essential climatic variables in Austria and Europe	10
	1.2.2.	Future changes	17
		Uncertainties in regional climate projections	17
		Large-scale context of Austrian climate change	18
		Temperature projections	20
		Precipitation projections	21
		Drought projections	23
		Wind projections	23
		Elevation dependencies	24
	Chapt Globa for Au	er Box 1.1   I warming levels and their implications Istria	24
	Chapt	er Box 1.2	
	Tippir	ng points in the climate system	28
	Chapt Case s	er Box 1.3   study: Lake Neusiedl as sentinel of	20
	climat	e change	30
1.3.	Cryos	phere	31
	1.3.1.	Snow	31
		Observed snow depth and snow cover duration	31
		Future snow depth and snow cover duration	31
		Related consequences	32

C	ross-	Chapter Box 1	
E	valua	ation of climate-related risks and	
k	ey ris	iks	32
1	.3.2.	Glaciers	39
		Observed trends	39
		Future changes	39
		Related consequences	39
1	.3.3.	Permafrost	40
		Observed trends	40
		Future changes	40
		Related consequences	41
1.4. H	lydro	sphere	41
1	.4.1.	Rivers and streams	41
		Mean flow	41
		High flows/floods	42
		Low flows	43
1	.4.2.	Lakes	45
		Lake temperatures	45
		Lake ice	46
		Nutrients and oxygen dynamics	46
1	.4.3.	Groundwater and soil water	47
		Observed trends	47
		Future changes	48
1.5. P	edos	phere	48
16 B	liosnł	Dere	50
1	61	Wetlands	50
1	62	Forests	51
1	63	Grasslands	52
1	.6.4.	Biodiversity	53
171	ithor	nhoro	57
1.7. L	7 1	Theoretical background	57
1	.7.1.	Climate change impact on landelides	57
1	.7.2.	Observed trends and future changes	58
10.0	./.J.		50
1.8. C	.ross-	cutting topic: Natural hazards	60
Refere	ences	·	62
Glossa	ary	aar2.ccca.ac.at/gloss	ary
	appr vibe	eviationsaar2.ccca.ac.at/abbreviationabbreviati	
Full re	port.	aar2.ccca.ac.at/appendi	ort
	•		

#### **EXECUTIVE SUMMARY**

Changes in Austria's thermal environment are detectable both for the mean and for the extremes, with a particularly pronounced acceleration of temperature increase since the 1980s (*high confidence*). The temperature increase is markedly stronger in Austria than in the global mean, mainly due to the regional thermodynamic and dynamical forcings in large parts of Europe (*high confidence*). {1.2.1}

- In Austria, the annual mean surface air temperature until 2024 has increased by 3.1°C since pre-industrial times. Since the 1980s, this warming has been particularly strong (+0.5°C per decade) in Austria and other parts of Europe. {1.2.1}
- Temperature extremes during the day and at night, such as hot days or tropical nights, and the associated heat stress, have increased significantly in the summer halfyear (April-September). This increase is particularly pronounced in urban areas. In Austrian federal province capitals, heat waves are now 50 % more frequent and 1–4 days longer than in previous decades (1991–2020 vs. 1961–1990). {1.2.1}
- The number of hot days in very hot years (with a statistical return period of 10 years) has increased on average by the factor of 3 in these periods. {1.2.1}

All climate projections show a further increase of the global mean temperature at least until mid-century. The development thereafter strongly depends on the amount of anthropogenic greenhouse gas emissions. Assuming a continuation of current policies, anthropogenic greenhouse gas emissions are projected to lead to a median global warming of  $2.7^{\circ}$ C [2.2 to  $3.4^{\circ}$ C] by 2100. The temperature increase in Austria continues to be higher than the global mean. Heat extremes will become more severe. The frequency and duration of heat waves will increase and the temperature within individual heat waves will strongly increase (*high confidence*). {1.2.2}

- Mean temperatures over Austria are clearly projected to increase for all seasons with increasing levels of global warming (*high confidence*). Compared to the past (climate period 1961–1990), annual mean temperatures are expected to increase by 2.0°C, 2.6°C, 3.6°C and 4.9°C at global warming level (GWL) 1.5°C, GWL 2.0°C, GWL 3.0°C and GWL 4.0°C, respectively. {1.2.2}
- The number of hot days in a very hot year occurring once in 10 years in the past (climate period 1961–1990) will likely multiply 3.5 times at GWL 1.5°C, 4.3 times at GWL 2.0°C, 5.7 times at GWL 3.0°C, and even 8.1 times

at GWL 4.0°C on average for Austria. In addition, the average temperature of heat waves will likely increase by 6.6°C at GWL 4.0°C compared to the climate period 1961–1990 (*robust evidence, low agreement*). {1.2.2}

• Projections of future temperature for Austria represent conservative estimates and need to be interpreted with caution given the cold bias (largely driven by deficiencies in aerosol forcing and atmospheric dynamical components) of the regional climate models backing the ÖKS15 ensemble. Current warming is already exceeding temperature projections at low GWLs. It is expected that Austrian mean air temperatures in the future exceed global mean temperatures by at least a factor of 1.3 (*high confidence*). {1.2.1, 1.2.2}

The observed and future temperature changes in Austria are directly linked to changes in several other climatological variables and have far-reaching consequences (*robust evidence, medium agreement*). {1.2.1, 1.2.2}

- Average evaporation in Austria has increased by about 110 mm/yr in total over 35 years (1978–2013) and will increase with further warming (*high confidence*). {1.4.1} Increased evaporation contributes significantly to low flows in rivers and streams, to the severity of droughts and to changes in the water level of Lake Neusiedl (*robust evidence, medium agreement*). {1.4}
- Since the 1960s, snow cover duration (SCD) has decreased by 5–10 days per decade in the entire Alpine region, including Austria. Until the year 2050, SCD in Austria will most likely continue to decrease by at least 5 days per decade on average (*high confidence*). {1.3.1}
- All glaciers in Austria are losing mass and have been shrinking at an accelerating rate in recent years. 40 % of glacier area in Austria were lost between 1969 and 2015. Actual deglaciation before the end of the century is virtually certain (*high confidence*). {1.3.2}
- The intensity of multi-day, daily and small-scale short-duration (hourly) heavy precipitation events has increased in the warm season (May-September), closely following the observed rise in temperature and associated increase in the water vapor content of the air (*robust evidence, medium agreement*). {1.2.1}
- Daily maximum precipitation with a 10-year return period will likely increase by 14.3 % at GWL 1.5°C, 17.2 % at GWL 2.0°C, 23.4 % at GWL 3.0°C and 30.3 % at GWL 4.0°C compared to the past (climate period 1961–1990) (*medium confidence*). {1.2.2}

The impact of anthropogenic climate change on precipitation in Austria is much more complex than on temperature. It shows regional and seasonal variations and differs for climatological sums (e.g., seasonal or annual sums) and for individual precipitation events (*robust evidence, medium agreement*). {1.2, 1.4}

- Seasonal mean precipitation in Austria is projected to increase in winter (*high confidence*) and decrease in summer (*robust evidence, medium agreement*). These changes are clearly evident under high emissions scenarios or GWL 3.0°C and higher. {1.2.2}
- There is a regional difference in long-term changes in seasonal precipitation over the last 200 years, especially in winter, of the order of 10–15 %, with an increase in western Austria and a decrease in south-eastern Austria (*high confidence*), probably due to decadal variability (*medium confidence*). {1.2.1}

Observed floods (fluvial and pluvial) show an increase in part of the catchments (*medium confidence*). For fluvial floods, the trend is more pronounced in smaller catchments (24 % show increasing trends) than in larger catchments. There is a clear seasonal shift from winter and spring floods to summer floods, so that summer is now also the dominant flood season in the eastern half of Austria. Climate projections indicate a potential increase and notable seasonal shift in flood occurrences in Austria, particularly an increased frequency of summer floods due to more intense and localized extreme precipitation events, especially in the Alps and smaller river basins (*robust evidence, medium agreement*). {1.4.1}

Low-flow trends in Austria (both drying and wetting) have become more pronounced in recent years compared to the previous assessment (*high confidence*). Low flow discharges have increased in alpine catchments (+23 % over 43 years (1977–2019)) and decreased in lowland catchments (-17 % over the same period). They are expected to continue to decrease in the lowlands of eastern and southern Austria. In alpine catchments, low flows will increase significantly due to decreasing frost. {1.4.1}

The surface water temperature of all monitored lakes has increased during the last 20 to 40 years (*high confidence*). The surface temperature of selected lakes in Austria has increased concurrently with increasing air temperature by an average of between 2.0°C (last 40 years) and 1.5°C (last 15 years). Since lake surface water temperature is closely cor-

related with changes in air temperature, lake surface water temperature is expected to increase in the future. Higher lake water temperatures lead to higher lake metabolism, which leads to higher respiration, and thus to higher emissions of greenhouse gases such as  $CO_2$ , and possibly N<sub>2</sub>O and  $CH_4$ . {1.4.2}

The water loss in Lake Neusiedl in the years 2021 and 2022 is primarily the result of evaporation due to high temperatures (surface water temperatures exceeded 30°C at times in summer) and a decrease in precipitation (*medium confidence*). Ongoing climate change will have a significant impact on the future of Lake Neusiedl and its surrounding ecosystems, with lower average water levels, especially in summer, and as yet unknown effects on the species community.  $\{1.4.2\}$ 

Currently and in future projections, there is a west-east gradient with generally higher groundwater recharge in the western part of Austria than in the eastern or south-eastern part (*high confidence*). Seasonal changes in groundwater recharge are expected in the future, but they depend strongly on climate scenarios (*limited evidence, medium agreement*). Available groundwater resources may decrease by up to 23 % by 2050, and regionally by as much as 30 % (*medium confidence*). {1.4.3}

Changes in water and temperature regimes will significantly increase soil erosion (*high confidence*). This occurs due to increased frequency of convective events and expected extreme precipitation, if no countermeasures are taken. {1.5}

High temperatures combined with sufficient water supply will lead to increased loss of soil organic carbon (*high confidence*). Mineralization of soil organic carbon will be enhanced by higher temperatures at sites with sufficient precipitation (e.g., at higher sea level). {1.5}

Land use and climate change are the most important drivers of wetland degradation (*high confidence*). Many wetlands will lose important ecosystem functions such as carbon storage, resulting in positive feedbacks with climate change (*high confidence*). Other commonly affected ecosystem functions include water retention (*medium evidence, high agreement*) and the support of biodiversity (*medium evidence, high agreement*). {1.6.1}

Limited water supply will be the most important direct impact factor in Austrian forests (*high confidence*). Disturbance regimes (storms, snow, insects, fungi) will intensify and strongly affect the provision of ecosystem services (*high confidence*). Forest management decisions will strongly determine the pathways and speed of adaptation in species composition (*medium confidence*). {1.6.2}

Climate change drives species ranges upslope in elevation and promotes the replacement of cold-adapted species by warm-adapted species in many taxonomic groups (*high confidence*) and a decline in some habitats such as alpine grasslands (*limited evidence*, *high agreement*). Under strong further warming, major population declines, and increased vulnerability are expected especially for cold-adapted species of higher altitudes (*high confidence*), including many endemic taxa, whose eventual extinction would reduce Austria's unique contribution to the global species pool (*high confidence*). {1.6.4}

Climate change mitigation via land-based carbon dioxide removal and energy decarbonization could have significant negative impacts on biodiversity (*medium evidence*, *high agreement*). However, carefully designed, e.g. via appropriate spatial planning or tree species selection for afforestation these measures can also generate double benefit for the intertwined climate and biodiversity crises. {1.6.4}

Climate change will lead to responses in Earth's surface processes, and thus to a change in the causes and triggers of landslides (*medium confidence*). Permafrost degradation, glacier retreat, and extreme precipitation conditions are the most obvious factors responsible for increasing landslide activity of various types, although data for the Austrian Alps need to be substantially improved. The frequency of shallow earth/debris slides and debris flows will increase in Austria. In the high mountains, affected by glacier retreat and permafrost degradation, a further increase in rock fall/avalanche and rock slide activity is expected. {1.7}

Depending on the topographic situation, climate change will have a significant impact on different types of natural hazard processes in Austria (*medium confidence*). Numerous cascading or compound processes have been identified for Austria in the past, some of which have resulted in major damage and high financial losses. These processes should be given special attention in the future. {1.8}

#### 1.1. Chapter introduction

#### 1.1.1 Setting the scene

The concept of Chapter 1 is to provide general information on the current situation as well as on historical and future developments of the climate system, its components and the direct impacts on the natural environment in Austria. Basic processes and drivers and their response to anthropogenic climate change are described. The chapter provides important reference data on past and future changes in climate indices, general trends and extremes for the subsequent chapters. The chapter is divided into subchapters according to the spheres involved in the processes.

- The introduction (Section 1.1) describes the physiographical and climatological characteristics of Austria.
- Section 1.2 focuses on the atmosphere and discusses processes relevant to weather and climate in Austria and their response to anthropogenic climate change in the past and the future.
- Section 1.3 covers the cryosphere, including regional/ temporal occurrence, drivers and magnitude of past and future changes, and interactions and impacts on other natural systems.
- The subchapter 'Hydrosphere' (Section 1.4) summarizes the understanding and assessment of the impacts of climate change on runoff generation processes and the magnitude, seasonality and frequency of water fluxes in streams, groundwater and lakes, including extremes and water quality.
- The subchapter 'Pedosphere' (Section 1.5) discusses the types and characteristics of soils in Austria, identifies the most important representatives and assesses the effects and feedbacks of climate change and other global change drivers on soils via erosion and the dynamics of organic matter and water.
- The subchapter 'Biosphere' (Section 1.6) focuses on selected important habitats and land use categories (according to the IPCC) that will be strongly affected by climate change and also have important feedback effects on climate change. The effects of a changing climate on biodiversity are summarized across habitats and taxonomic groups.
- The subchapter 'Lithosphere' (Section 1.7) focuses on the description of the Earth's geological and geomorphological processes in Austria, with emphasis on exogenous

processes, in particular landslides of various types relevant to the Alpine region of Austria. These processes are strongly controlled by climate.

• The last subchapter (Section 1.8) focuses on the cross-cutting issue of natural hazards, describing those occurring in Austria and their response to anthropogenic climate change. An overview of natural hazard processes and relevant hazard cascades in Austria is summarized in a table (Table 1.A.4).

#### 1.1.2 Physiographic setting of Austria

Austria is a small country in Central Europe (see Figure 1.1). It has an area of 84,000 km<sup>2</sup> and a population of about 9 million (Statistik Austria, 2023) with an urbanization rate of ~60 % (World Bank, 2023). The population increased by 11.3 % between 2000 and 2020, with the proportion of very old people (80+) increasing from 3.5 % to 5.5 % (Federal Ministry of Social Affairs, Health, Care and Consumer Protection, 2021). Almost two-thirds of the country is mountainous. Austria's high mountain regions are dominated by the Eastern Alps, but the granite and gneiss plateau in the north of the country also reaches heights of 1,379 m a.s.l.

The Alps in Austria extend from the western edge of the country to the far east. At the Großglockner, they reach 3,798 m a.s.l., which is the highest point in the country. Throughout western Austria, peaks exceed 3,000 m a.s.l. While Austria's western regions are entirely in the Alps, east of Salzburg, the granite and gneiss plateau extends to Austria's borders with Germany and the Czech Republic, north of the foothills of the Alps. The east of the country is occupied by lowlands: The Carpathian foothills to the north and the Pannonian Plain farther south. In the southeast, the southeastern foothills of the Alps border Hungary and Slovenia. The lowest point of Austria is located east of Lake Neusiedl, near the border with Hungary, at 114 m a.s.l. Austria's topographic situation has a strong influence on the country's settlement structure, land use, vegetation and climate, including the occurrence and relevance of natural hazards.

Austria's topographic and climatological conditions result in very heterogeneous and diverse landscapes. On the highest mountain peaks, a relevant part of the precipitation currently falls as snow, even in summer, and at altitudes above 3,000 m a.s.l., there are glaciers and permafrost. Thus, several climate zones, from the snow-dominated nival to the wine-growing lowlands, extend over a few tens of kilometers (Rubel et al., 2017). Elevation-dependent air temperature



Figure 1.1 Topographic map showing the terrain in Austria (Data source: European Environment Agency, 2016).

variability is influenced by windward and leeward modulated precipitation, leading to small-scale variability of ecosystems. Climate gradients themselves are superimposed by land use pressure and history to shape ecosystem variability. Due to the short distances between different climatic and ecosystem zones, climate change-induced migration of biota can occur immediately, unless there are local (natural or anthropogenic) barriers to migration (Holzinger et al., 2008).

#### 1.1.3 Climatological conditions

Due to Austria's location in the mid-latitudes (~48 °N), radiation and temperature show a pronounced seasonal cycle. In the warmest areas of Austria, the lowlands in the east and south, the annual mean temperature reaches 12°C and drops to -6°C on the highest mountain peaks. Extreme daily maximum temperatures reach 40°C in summer and -30°C in winter. These extremely low temperatures are not limited to the mountain peaks. Even in the northeastern lowlands, temperatures below -20°C occur during cold spells (Lhotka and Kyselý, 2015), when continental 'Siberian' air masses are advected to Central Europe.

The Austrian precipitation distribution is dominated by the advection of maritime air masses and the associated windward/downwind effects at mountain ridges. The main source of moisture is the Atlantic, but maritime air masses can also originate from the North, Baltic, Mediterranean or Black Sea, leading to very complex precipitation patterns. The driest area is in the northeast, with an average annual rainfall of nearly 500 mm. In the northern and southern 'Stau-regions' values up to 2500 mm occur (Stangl et al., 2021). Due to windward/downwind effects even nearby mountain valleys, such as the upper Inn valley, receive only 700 mm.

Convective precipitation is an important source of moisture in Austria during the summer half year. Especially areas in the transition zone between the lowlands and the foothills, such as the 'Steirische Randgebirge' or the Upper Austrian foothills, have a high probability of thunderstorms (Schulz et al., 2005) and associated high local short-term precipitation intensities.

Large-scale precipitation extremes are mainly associated with cut-off lows centered at the European Alps (Awan and Formayer, 2017), where warm and moist Mediterranean air is transported to the Alps over several days and precipitation totals of more than 200 mm can be reached.

The hydrological conditions show a similarly high diversity (Merz et al., 2008). Nival-dominated discharge patterns are common for alpine rivers, with low flows in winter and a pronounced discharge maximum during the snowmelt period. In foothill catchments in the east and south, snowmelt plays a minor role and low flows occur not in winter but in summer and fall. Steep terrain and localized heavy precipitation from thunderstorms often lead to pluvial floods and mass movements. Fluvial floods, especially those of the Danube, are caused by completely different processes, such as cut-off lows, but snowmelt and the elevation of the snow line during the precipitation event also contribute to these events.

The first Austrian Assessment Report (APCC, 2014) provides a detailed description of the Austrian climate. In the decade since this report was written, anthropogenic climate change has continued in Austria. Several hot and dry summers, such as during 2015 or 2018, highlighted the vulnerability of Austrian agriculture and especially forestry, as well as the importance of heat mitigation measures in urban areas. Local or temporary water shortages became a topic in the media, and other extreme events such as local flash floods, forest fires, or spring frost damage became more visible to the general public.

#### 1.2. Atmosphere

This subchapter discusses the drivers and processes relevant for weather and climate in Austria and their response to anthropogenic climate change in the past and in the future. It describes the current situation and past and future changes based on climatological indicators. Climate change projections are discussed in the context of climate change signal, depending on emission scenarios and general skills and limitations of models at global and regional scale.

Regional climate variability and change result from both natural and anthropogenic forcings external to the climate system, internal climate variability, and feedbacks between them (Doblas-Reyes et al., 2021). Anthropogenic forcings include changes in greenhouse gas concentrations, aerosols, and land use. The strength of the regional response to external forcings is first determined by climate sensitivity, which is the overall temperature response to radiative forcings such as greenhouse gases. Our understanding of this key property of the climate system has improved considerably in recent years. The best estimate of equilibrium climate sensitivity is a 3°C increase in global mean surface temperature (with likely and very likely ranges of 2.5°C to 4°C and 2°C to 5°C, respectively) for a doubling of CO<sub>2</sub> concentrations (IPCC, 2021b). A key determinant of regional climate change is changes in atmospheric circulation, i.e., how large-scale to regional-scale winds and weather systems respond to overall temperature changes. Regional feedbacks between the land-surface and the atmosphere further modulate the strength of regional climate changes.

#### 1.2.1. Past changes

The current IPCC report (AR6) assesses past global changes back to the deep past (65 million years before present) in Chapters 2 and 11 (observations, weather and climate extremes) (Gulev et al., 2021; Seneviratne et al., 2021). The historical investigations in this subchapter are limited to the period after 1850. This is consistent with the IPCC definition of the 'pre-industrial' period. The focus is on describing climatological changes within this period based on impact-relevant climatological indicators and their dependence on large-scale processes and drivers.

#### Observed changes in global climate drivers

Increases in greenhouse gas concentrations and trends in aerosol concentrations and properties due to anthropogenic activities lead to a net positive effective radiative forcing and an imbalance in the Earth's energy balance, resulting in a warming planet since the late 19th century and at an increasing rate since the 1970s. The change in effective radiative forcing due to natural factors since 1750 is negligible compared to anthropogenic forcings (Gulev et al., 2021).

# Observed changes in large scale circulation, weather patterns and modes of variability

Since the late 19th century, major modes of climate variability relevant to the European climate, such as the North Atlantic Oscillation (NAO) or the Atlantic Multidecadal Variability (AMV) show no persistent trends but fluctuate in frequency and magnitude on interdecadal time scales. Due to the expansion of the tropical circulation, a poleward shift of extratropical jets and cyclone tracks has been observed since the 1980s, with marked seasonality in the trends. Since the 1970s, there has been a global weakening of surface winds, especially over land in the northern hemisphere (Gulev et al., 2021).

# Observed changes in selected essential climatic variables in Austria and Europe

#### Air temperature

According to a recent analysis (Chimani et al., 2024), the mean annual air temperature in Austria has increased by 3.1°C since pre-industrial times, from 5.1°C (1850–1900) to 8.1°C in 2024 (Figure 1.2, Table 1.1). Primary drivers (con-



**Figure 1.2** Time series of annual mean air temperature deviations from the pre-industrial mean: Global mean, global land areas, European land areas (bounding box 25 °W – 40 °E, 34 °N – 72 °N) and Austria for the period 1841–2024 (2023 for European land areas). Note that Austrian data is based on 7 homogenized stations and that the global datasets only start in 1850. Thin lines show annual values (for clarity only shown for Austria and the global mean). Thick lines are smoothed with a 42-year LOESS filter. Horizontal lines show the mean values of the 30-year climate normals (1961–1990, solid; 1991–2020, dashed) (Data sources: Geosphere Austria; Auer et al., 2007; Morice et al., 2021; Chimani et al., 2024).

tribution of ~70 %) of this excessive warming compared to the global average are regional thermodynamic and dynamical forcings in large parts of Europe, as described below. Secondary drivers (contribution of ~30 %) are related to heat capacity (faster warming of air over land) and latent heat fluxes (evaporative cooling over oceans) as well as the different scales (IPCC, 2019a; Chimani et al., 2021).

As a primary driving component of the excessive warming, the regional thermodynamic forcing is dominated by increasing incoming near-surface solar radiation in large parts of Europe since the 1980s. This is due to a 50 % decrease of aerosol loads (Glantz et al., 2022; Philipona et al., 2023; Schumacher et al., 2024) and a decrease in cloudiness associated with the indirect aerosol effect and largescale atmospheric circulation changes (Sfîcă et al., 2021). Schumacher et al. (2024) estimate the relative contributions of these two effects to be two thirds (aerosols) and one third (dynamic circulation effect), respectively. From a temporal perspective, the aerosol effect is mainly restricted to the period 1983-2002, while the cloudiness effect unfolds in the period 2001-2020 (Schilliger et al., 2024). A reduced cooling effect due to less aerosols unmasks the thermal greenhouse warming effect (increasing thermal longwave downward radiation) due to steadily increasing anthropogenic greenhouse gas concentrations in the atmosphere (Philipona et al., 2023) and readjusts the shortwave clear-sky transmissivity of the atmosphere to pre-1950s values (Wild et al., 2021) and beyond (the latter due to circulation changes). In addition to the strong reduction of aerosol loads, positive regional feedbacks, especially the soil moisture (Haslinger et al., 2021; Glantz et al., 2022) and snow/ice albedo (Kiem et al., 2024) feedbacks, amplify the regional thermodynamic effect, leading to the fact that Europe has warmed faster than any other World Meteorological Organization region since the 1980s (Schumacher et al., 2024); in Austria with an enormous rate of almost 0.5°C per decade. Moreover, a large part of the warming since pre-industrial times has occurred in the last four decades since 1980, and the five-year period 2019-2023 was already 2.7°C warmer than in pre-industrial times (Table 1.1). The observed warming of global land masses (CRUTEM data) exceeds global average warming by a factor of at least around 1.3. Since this discrepancy is mainly attributed to different heat capacities, it is expected to hold as lower limit for future temperature development.

Within Austria, the long-term variations of the annual air temperature show a large spatial concordance. There are no significant regional or altitudinal differences. In mountainous regions, the atmosphere warmed by the same amount as in the lowlands (Auer et al., 2007; Kuhn and Olefs, 2020). Air temperature variations in the free atmosphere at about

Table 1.1	Mean air temperature deviations from the pre-industrial average (1850–1900) for different regions during the 30-year average periods
1961-1990	, 1991–2020 and the 2023 and 2024 values, respectively.

\* defined by the bounding box 25 °W – 40 °E, 34 °N – 72 °N;

\*\* derived from the 41-year smoothed value (LOESS filter).

Climate normals (30-year average periods)	Global mean (HadCRUT5 ensemble mean) [°C]	Global land areas (CRUTEM) [°C]	European land areas (CRUTEM subset*) [°C]	Austria (SOCRATES) [°C]			
2024**	1.4	1.8	n/a	3.1			
2023**	1.32	1.71	2.26	3.0			
1991–2020	0.89	1.21	1.47	1.98			
1961–1990	0.37	0.48	0.48	0.7			

3,000 m a.s.l. derived from radiosonde data are very similar to the time series recorded at summit stations (Haimberger et al., 2012). This lack of observed elevation dependence based on Austrian station data contrasts with future temperature projections (see Section 1.2.2 and Chapter 7) and analyses of gridded datasets in the greater Alpine region (Pepin et al., 2022), which show stronger warming at mid to high elevations. As many mechanisms for elevation dependence operate simultaneously, some independently, most of them interconnected by feedbacks, this discrepancy may be due to model simplification (e.g., too coarse spatial resolution, parameterizations) or to the fact that relevant processes have not yet been triggered (Kuhn and Olefs, 2020; Pepin et al., 2022).

From a seasonal perspective and based on homogenized long-term observations, the warming in the lowlands during the period 1991–2020 vs. 1850–1900 is strongest in spring and winter (+2.1°C and +2.3°C, respectively), followed by summer (+2.0°C), and weakest in autumn (+1.4°C). The only significant warming difference between lowland and summit regions is observed in winter (+2.3°C vs. +1.7°C) (Kuhn and Olefs, 2020). This may be due to the reduction of aerosols at their highest concentration near ground sources, which contributes to stronger warming at low elevations (Philipona et al., 2009; Philipona, 2013).

Smaller-scale phenomena such as temperature inversions are associated with the accumulation of pools of cold air near the surface. An analysis of daily gridded observational data in the period 1961–2017 shows that inversions have become less frequent and less intense, especially in the months of October, December, and January in southern Austria and near the main Alpine ridge (Hiebl and Schöner, 2018).

The marked temperature increase since 1980 has led to pronounced changes in temperature dependent extreme values. For example, the number of summer days and hot days (daily maximum temperature  $\geq 25^{\circ}$ C and  $\geq 30^{\circ}$ C, respectively) in the Austrian lowlands increased significantly by 40 % and 66 %, respectively (see Table 1.2). The number of hot days in very hot years (occurring with a statistical return period of 10 years) has tripled between the periods 1961-1990 and 1991-2020 on average over Austria (GeoSphere Austria, 2020). In addition, heatwaves have increased in intensity, duration and frequency in the Austrian federal province capitals: In the period 1991-2020 compared to 1961-1990, heatwaves became 50 % more frequent and 1 to 4 days longer (GeoSphere Austria, 2023a). In urban areas, heat stress is intensified during the day and at night (urban heat island effect) (Oke et al., 2017). In Austrian province capitals, the number of hot days increased by 6 to 14 days (factor of 1.5 to 4.3) in the 30-year period 1991-2020 compared to 1961-1990. In the same period, tropical nights (daily minimum temperature  $\geq 20^{\circ}$ C) increased by up to 5 days.

As of 2024, a record number of 29 to 52 hot days in a single year is recorded in these cities (GeoSphere Austria, 2023b). In addition to these increasing mean and extreme temperature values, the available literature shows no significant long-term change in temperature variability on synoptic time scales in Austria since the end of the 19th century (Hiebl and Hofstätter, 2012).

In general, annual and multi-decadal variations are superimposed on the long-term anthropogenic warming trend. Due to atmosphere-ocean feedbacks and the relative importance of circulation changes during the cold season, winter temperatures in Austria show the largest variability of all seasons. After a short cooling phase (between 1995–2005 in the lowlands and 1989–2012 in the summit regions) induced by natural variability and associated atmospheric circulation changes (Saffioti et al., 2016), winter temperatures started to increase again (Olefs et al., 2021).

Finally, the observed temperature increase in Austria is confirmed by independently measured air pressure time series at different elevations (Kuhn and Olefs, 2020).

Short	altitudinal range (m a.s.l.)		≤500 m				501–1,000 m				1,001–1,500 m					1,501–2	2,000 m		≥2,001 m			
name	Indicator name	Unit	1961– 1990	1991– 2020	diff.	diff. (%)	1961– 1990	1991– 2020	diff.	diff. (%)	1961– 1990	1991– 2020	diff.	diff. (%)	1961– 1990	1991– 2020	diff.	diff. (%)	1961– 1990	1991– 2020	diff.	diff. (%)
su25	summer days	d	45	64	+19	+42	25	39	+14	+56	7	15	+8	+114	1	2	+1	+100	0	0	0	
su30	hot days	d	6	16	+10	+167	2	6	+4	+200	0	1	+1		0	0	0		0	0	0	
tr20	tropical nights	d	0	1	+1		0	0	0		0	0	0		0	0	0		0	0	0	
kys	heat episodes (kysely days)	d	4	14	+10	+250	0	4	+4		0	0	0		0	0	0		0	0	0	
cool	cooling degree days (20°C tresh)	°C	43	107	+64	+149	9	34	+25	+278	1	5	+4	+400	0	0	0		0	0	0	
gsl	vegetation period (5°C)	d	224	235	+11	+5	196	208	+12	+6	166	180	+14	+8	123	142	+19	+15	52	75	+23	+44
fd0	freezing days	d	104	90	-14	-13	134	117	-17	-13	163	146	-17	-10	191	173	-18	-9	245	224	-21	-9
id0	ice days	d	29	22	-7	-24	41	32	-9	-22	56	44	-12	-21	82	67	-15	-18	149	128	-21	-14
id7	extreme ice days	d	17	11	-6	-35	27	18	-9	-33	40	30	-10	-25	61	48	-13	-21	115	95	-20	-17
hdd	heating degree days	°C	3662	3306	-356	-10	4384	3976	-408	-9	5232	4796	-436	-8	6196	5723	-473	-8	7647	7189	-458	-6
rr1	precipitation days (1 mm)	d	112	110	-2	-2	135	134	-1	-1	143	143	0	+0	149	149	0	+0	150	153	+3	+2
rr20	heavy precipitation days (20 mm)	d	6	7	+1	+17	10	12	+2	+20	15	16	+1	+7	17	18	+1	+6	17	18	+1	+6
sdii	precipitation intensity	mm	6.4	6.7	+0	+5	7.6	7.9	+0	+4	8.7	9	+0	+3	9.1	9.4	+0	+3	8.9	9.3	+0	+4

Table 1.2 Selected prime climatic indicators for the national territory of Austria. Definitions of the indicators used in Chapter 1 are given in Table 1.A.1.

Short name	altitudinal range (m a.s.l.)		≤500 m			501–1,000 m				1,001–1,500 m				1,501–2,000 m				≥2,001 m				
	Indicator name	Unit	1961– 1990	1991– 2020	diff.	diff. (%)	1961– 1990	1991– 2020	diff.	diff. (%)	1961– 1990	1991– 2020	diff.	diff. (%)	1961– 1990	1991– 2020	diff.	diff. (%)	1961– 1990	1991– 2020	diff.	diff. (%)
rx5day	max. 5-day precipitation	mm	70	77	+7	+10	91	97	+6	+7	108	111	+3	+3	116	119	+3	+3	115	121	+6	+5
cdd	longest dry period	d	24	24	0	+0	21	21	0	+0	20	20	0	+0	20	20	0	+0	19	19	0	0
hd1525	dry hiking days	d	100	86	-14	-14	83	82	-1	-1	67	71	4	+6	40	47	7	+18	10	16	6	60
cloudy	number of cloudy days	d	149	136	-13	-9	149	137	-12	-8	141	130	-11	-8	143	134	-9	-6	145	137	-8	-6
clear	number of clear days	d	69	86	+17	+25	76	88	+12	+16	88	93	+5	+6	93	96	+3	+3	98	98	0	0
SPEI	climatic water balance (normalized)	-	0.10	-0.10	-0.20		-0.01	0.00	0.01		-0.01	0.00	0.01		0.02	0.00	-0.02		-0.02	0.08	0.10	
CWB	climatic water balance (abs)	mm	-60	-90	-30	+50	+337	+339	+2	+1	+556	+558	+2	+0	+658	+650	-8	-1	+740	+752	+12	+2

#### Drought conditions

Meteorological droughts are prolonged periods of below-normal water availability due to a lack of precipitation and/or increased atmospheric evaporative demand (i.e., potential evapotranspiration). An important driver of droughts are anomalies in atmospheric circulation that result in frequent and often persistent anticyclonic weather patterns (Haslinger et al., 2019; Lhotka et al., 2020). In northern Europe, the North Atlantic Oscillation (NAO) is of great importance in this respect, but less so in the Alpine region. In Austria, the East Atlantic-Western Russia (EAWR) circulation pattern explains the occurrence of early spring drought (Ionita et al., 2020; Haslinger and Mayer, 2022). Spring droughts often lead to soil moisture-temperature feedbacks, which may contribute to elevated temperatures during summer droughts (Seneviratne et al., 2010; Haslinger and Blöschl, 2017). When meridional circulation features dominate during the summer season, large-scale moisture transport is reduced and local moisture recycling is enhanced. In already dry soils, insufficient moisture is available for local evapotranspiration processes and precipitation deficits occur (soil moisture-precipitation feedback) (Haslinger et al., 2021). The seasonality of drought events has shifted from predominantly winter and spring events in the 19th century to late summer/autumn events. This seasonal shift is accompanied by a spatial change from the northwest to the southeast of Austria (Haslinger and Blöschl, 2017).

There are no significant trends in the frequency, duration and intensity of precipitation deficits in Austria over the last two centuries for which instrumental records are available (Haslinger and Blöschl, 2017; Hanel et al., 2018). However, periods with distinct negative precipitation anomalies are observed, most notably the 'drought decades' of the 1860s and 1940s (Brunetti et al., 2009; Haslinger et al., 2019). The current sequence of drought years (e.g., 2015, 2018, 2021) may be indicative of the manifestation of another drought decade, as drought drivers similar to those of the 1860s and 1940s are at play more recently; namely very dry springs followed by atmospheric circulation characteristics that favor a positive soil moisture-precipitation feedback during the summer. Furthermore, longer drought reconstructions for the last millennium and beyond based on paleoclimate archives such as tree-ring widths (Ionita et al., 2021) and stable isotopes in tree-rings (Büntgen et al., 2021) for Central Europe are contradictory and do not allow a definitive conclusion. However, potential evapotranspiration increased during the period 1977-2014, mainly due to changes in global radiation, but also due to increases in air temperature (Duethmann and Blöschl, 2018), leading to drier soil conditions during the summer half year (Trnka et al., 2009). Higher temperatures also resulted in an earlier onset of the vegetation period and thus a longer evaporation season (Duethmann and Blöschl, 2018), leading to drier conditions during the warm season and, in particular, promoting heat waves (Haslinger and Blöschl, 2017).

The climatic water balance (CWB) is defined as the difference between precipitation and atmospheric evaporative demand and is often used in drought research and monitoring (here mostly as a standardized indicator, e.g., Standardized Precipitation Evapotranspiration Index – SPEI). Comparing two 30-year periods (1961–1990 vs. 1991–2020), the annual CWB in lowland areas (<500 m a.s.l.) decreased by 30 mm, mainly due to an increase in atmospheric evaporative demand. This change is rather small compared to the inter-annual variability of 107 mm (standard deviation of the annual CWB 1961–2020). In contrast, higher elevation areas show no clear change, although in the high alpine regions (>2,000 m a.s.l.) an increase of 12 mm is visible, which is still a rather small increase compared to the mean annual CWB from 1991–2020 of 752 mm (see Table 1.2).

Drought impact inventories in the Alpine region, as well as for Austria specifically, show an increasing trend of reported impacts within the last 45 years, especially after 2000. However, this increase could be influenced by reporting behavior, accessibility as well as awareness of the drought hazard (Stephan et al., 2021).

#### Precipitation, including convection and associated impacts

Opposing long-term precipitation trends are found in different regions, especially in winter: In western (Vorarlberg, northern Tyrol) and southeastern Austria (parts of Carinthia, western and eastern Styria, southern Burgenland), precipitation totals have increased and decreased by about 10 % and 15 %, respectively, over the last 200 years (Auer, 2014).

An analysis based on the Austrian-wide gridded observation-based precipitation dataset SPARTACUS (Hiebl and Frei, 2018) shows an increase in the annual number of days with heavy (95–98th percentile) to extreme (>98th percentile) precipitation from 1961 to 2014, especially in autumn and summer. This corresponds to a shift in intensities, as the number of days with weak to moderate precipitation decreases. This finding is supported by homogenized daily station data. Table 1.2 shows that the number of heavy precipitation days and the maximum 5-day precipitation sum have increased by 11 %

AAR2

and 5 %, respectively, in the period 1991-2020 compared to 1961-1990. Especially for large-scale heavy precipitation events caused by distinct low-pressure systems (e.g., 'Vb cyclones'), slightly increasing trends are found for the period 1961-2015 for the northern edge of the Alps (Hofstätter et al., 2018). In September 2024, such an event led to extreme precipitation totals in Central Europe. In Austria, new records for 5-day precipitation totals above 400 mm were set over Upper and Lower Austria and Vienna (Greilinger et al., 2024). In the Tulln area, the previous record of 143 mm was more than doubled. This led to very high water levels in the rivers, even in the Danube, where discharge values with a return period of 30 years were reached. Flood levels in the rivers of the Vienna Woods exceeded the 300-year return period (e.g., Traisen) (Krammer et al., 2024). According to the results of a preliminary attribution study, anthropogenic climate change doubled the probability of this event and made the rainfall 7 % more intense (Kimutai et al., 2024).

Small-scale heavy precipitation events are tied to convection and occur especially in the summer half year. Due to the low density of conventional meteorological measurement networks and the low temporal resolution of measurements before automation in the late 1980s, no reliable estimates of past trends in convective precipitation events and thunderstorms are currently possible in Austria (Pistotnik et al., 2020). This also applies to all phenomena associated with convective events such as hail and tornadoes.

An analysis of extreme hourly precipitation totals (99th percentile) at the Vienna Hohe Warte shows an increase of up to 10 % per °C temperature rise (Formayer and Fritz, 2016). Considering shorter time scales, Schroeer and Kirchengast (2018) found values of up to 14 % per °C based on extreme 10-minute precipitation sums (98th percentile) in southeastern Austria. Both rates clearly exceed the theoretical value given by the Clausius-Clapeyron law, which is confirmed by numerous studies of convective precipitation events (Ivancic and Shaw, 2016; Lochbihler et al., 2019). Increasing latent heat of condensation due to rising air humidity increases instability and thus intensifies convection and precipitation (Lenderink and Van Meijgaard, 2008, 2010).

From a European and global perspective, there is strong evidence that heavy precipitation has increased in recent decades due to anthropogenic forcing in many parts of Europe and the world, although regional differences are evident (Lehmann et al., 2015; Fischer and Knutti, 2016; Fowler et al., 2021; IPCC, 2021a; Robinson et al., 2021; De Vries et al., 2023). Results from a statistical model validated with event data and driven by gridded observational data in the period 1979 to 2016 (ERA-interim) show an increase in the probability of occurrence of thunderstorms, strong winds, and large hail events (Rädler et al., 2018). In addition, Maraun et al. (2024) set up a statistical model for the European domain that relates observed impacts of convective events (local flooding, hail, downbursts etc.) to atmospheric variables indicating convective potential (e.g., Lifted Index) to derive potential days with impactful convective events. The model is applied to ERA5 data from 1950 to 2022. A general increase in the number of days with convective events is found, independent of atmospheric circulation characteristics, due to increasing temperatures and potential for atmospheric instability and convection during the warm season. For example, convective events classified as 'severe' (associated with an orange weather warning (level 2 of 3)) increased by 15 % (1991-2020 vs. 1961-1990). Simon et al. (2023) found a corresponding increase in the annual and diurnal cycles of lightning in the high alpine region as well as on the northern and southern alpine rim.

Although no long-term change in annual precipitation totals can be detected for Austria, seasonal sums show clear regional trends. Multi-day, daily and short-duration (hourly) heavy precipitation events indicate clear increasing trends in recent decades with the latter being most strongly related to anthropogenic warming and associated with other potentially harmful small-scale but intense impacts (e.g., local flooding, strong wind gusts, hail and lightning).

#### Sunshine duration

Since the end of the 19th century, a two-stage increase in observed sunshine duration can be observed in Austria: The first until the middle of the 20th century, the second since the 1980s. A dimming phase between 1950 and 1980 interrupts these increases. The changes are more pronounced at low altitude stations (Auer et al., 2007; Kuhn and Olefs, 2020). The similarity of this trend with observed temperatures and solar surface radiation indicates an important role of solar radiation in explaining the larger temperature increase in Austria and parts of Europe compared to the global land areas (Scherrer and Begert, 2019; Philipona et al., 2023) (see also Figure 1.2 and Sections Air temperature as well as Radiation fluxes and cloudiness). Since 1961, observed sunshine duration in Austria has increased by 13.9 % to 14.8 %, according to a newly available gridded dataset and homogenized long-term time series (Hiebl et al., 2024). By definition, changes in sunshine duration are driven by changes in the direct component of surface solar radiation.

#### Radiation fluxes and cloudiness

There is strong observational evidence that the amount of solar radiation received at the Earth's surface undergoes significant multidecadal variations that are internal to the Earth's atmosphere and not externally forced by the Sun. Coherent periods and regions of dominant decreases ('dimming') and increases ('brightening') in surface solar radiation have been detected in the global observational networks (Wild, 2009; Wild et al., 2021), often coinciding with anthropogenic air pollution patterns (Nabat et al., 2014; Pfeifroth et al., 2018; Glantz et al., 2022). In Europe and Austria, three typical periods can be found in the observations: the 'early brightening' (1940s), the subsequent dimming (1950s to 1980s) and the still ongoing brightening phase (since the 1980s) (Manara et al., 2016; Kuhn and Olefs, 2020; Philipona et al., 2023).

In recent years, several studies have investigated and discussed the relative contributions of aerosol vs. large-scale circulation (i.e. cloudiness) changes in causing the current brightening phase in Central Europe (Sfică et al., 2021; Wild et al., 2021; Julsrud et al., 2022; Manara et al., 2023). Using satellite data, a recent study by Schilliger et al. (2024) attributes the increasing surface solar radiation in the periods 1983–2002 and 2001–2020 mainly to aerosol and cloudiness changes, respectively. The study by Ferreira Correa et al. (2024) points out that changes at low altitude stations are dominated by a strong clear sky forcing, in contrast to high altitude stations where changes are more related to cloud optical properties and surface albedo.

Based on data from twenty stations in Switzerland, the actual increase in net surface solar radiation is as much as 30 % greater than the observed increase in downward thermal longwave radiation due to increasing anthropogenic greenhouse gas concentrations in the atmosphere, which is the strongest regional-scale driver of climate change in central Europe since 1980 (Philipona et al., 2023). For Austria, surface solar radiation time series are currently being homogenized (Seitner et al., 2023) and downward thermal longwave time series have only been established since 2010 (Olefs et al., 2016).

#### Storminess

Existing studies show no long-term trend in past observed European storminess (extratropical storms with wind speeds of more than 75 km/h or 9 Beaufort) (Matulla et al., 2008, 2020; Feser et al., 2015).

#### 1.2.2. Future changes

In the following, we present an overview of the projected changes in key climate indicators for Austria. These changes are driven by the interplay between large-scale changes in temperature and atmospheric circulation and local processes and feedbacks (Section 1.2). The projected changes in these processes are affected by various sources of uncertainty. Therefore, a discussion of projection uncertainties and the large-scale context of climate change in Austria is presented first. The indicators presented are, unless otherwise indicated, derived from the Austrian climate scenarios ÖKS15 (Chimani et al., 2016; Truhetz and Leuprecht, 2018; Chimani et al., 2019; Maraun et al., 2021), which are a bias-adjusted variant of a series of EURO-CORDEX simulations. Figures for a selection of indicators are shown below, with additional figures provided in the Appendix (Figures 1.A.4, 1.A.5, 1.A.6). Definitions of the indicators used in Chapter 1 are given in Table 1.A.1.

#### Uncertainties in regional climate projections

Climate projections are affected by three main sources of uncertainty (IPCC, 2021a): Forcing uncertainty, climate response uncertainty and internal variability. Forcing uncertainty refers to the fact that future climate forcings are inherently unpredictable. In particular, greenhouse gas emissions depend on current and future political, economic and social decisions. Therefore, a range of plausible concentration pathways and shared socio-economic pathways has been constructed to explore how different socio-economic developments may affect future climate. Uncertainty in climate response refers to the fact that we do not fully understand the climate system and also have limited computational resources to model it. This uncertainty is particularly important at the regional scale and for extreme events, and is influenced by global-scale uncertainty in climate sensitivity, uncertainties in changes in large-scale atmospheric circulation, and regional processes and feedbacks (Chen et al., 2021). A major contributor to uncertainty in climate response is tipping elements (see Chapter Box 1.2). Multi-model ensembles are used to capture uncertainty in climate response. However, these ensembles cannot fully quantify the uncertainty arising from systemic model bias, where model projections do not cover the true range of plausible future climate states.

Also, internal variability is a major source of uncertainty at the regional scale (Figure 1.3). As a result, multidecadal trends at the local scale may be substantially different –



**Figure 1.3** Internal variability as a source of uncertainty in regional climate projections. The left panels show observed and projected summer precipitation, globally (top) and for Innsbruck (bottom). Gray lines indicate individual simulations with different random realizations of internal climate variability. The green and brown lines depict the strongest positive and negative trends over 50 years. The right panels show the corresponding box plots of the 50-year trends. From Maraun and Jury (2022). This figure is based on data from the MPI large ensemble (Maher et al., 2019).

stronger, weaker or even opposite – to those expected from increasing greenhouse gas concentrations (Doblas-Reyes et al., 2021). Initial condition ensembles have been used to sample uncertainties due to internal variability. In general, uncertainties are not fully sampled by climate models, i.e., they are larger than the simulated model spread (Doblas-Reyes et al., 2021). Global warming levels have been proposed to resolve uncertainties in global forcing and climate sensitivity by expressing changes relative to a given level of global mean temperature increase (Chapter Box 1.1).

#### Large-scale context of Austrian climate change

# Global and European temperature and precipitation projections

For all emission scenarios considered by the IPCC, the global surface temperature will continue to rise until at least mid-century (*high confidence*) (IPCC, 2021a). Global warming of 1.5°C and 2°C will be exceeded during the 21st century unless deep reductions in CO<sub>2</sub> and other greenhouse gas emissions occur in the coming decades (IPCC, 2021a). Assuming a continuation of current policies, anthropogenic greenhouse gas emissions are projected to lead to a median global warming of 2.7°C [2.2 to 3.4°C] by 2100 (Climate Action Tracker, 2024) (*high confidence*). In the worst-case scenario SSP5–8.5, the global mean surface temperature is projected to increase by 3.3 to 5.7°C (very likely range) in the period 2081–2100 compared to 1850–1900. Such increases would be associated with drastic changes in the climate system, including increased frequency and intensity of hot extremes, ocean heat waves, heavy precipitation, and, in some regions, agricultural and ecological droughts (IPCC, 2021a). Some tipping points are also likely to be crossed (see Chapter Box 1.2).

In Europe, mean temperatures are projected to increase more than the global average, as land surfaces warm faster than oceans (IPCC, 2021a). In winter, polar amplification causes the strongest changes towards the northeast, while in summer the strongest changes are expected over the Mediterranean. For the IPCC region Western Central Europe (WCS) to which Austria belongs, the annual mean changes from 1850-1900 to 2081-2100 under a worst-case scenario (SSP5-85) range from 5°C in the EURO-CORDEX ensemble to 5.8°C in the CMIP5 ensemble and 6.2°C in the CMIP6 ensemble. These changes are slightly higher in summer and lower in winter. Under a scenario compatible with the Paris Agreement (SSP1-2.6, with overshoot) these changes could be limited to values between 1.7°C (EURO-CORDEX), 1.9°C (CMIP5) and 2.0°C (CMIP6). Changes in EURO-CORDEX, and thus also in ÖKS15, are likely to be underestimated because several models may not have included changes in greenhouse gas concentrations in the European domain (Jerez et al., 2018).

Regional changes in precipitation are determined by the interplay of the Clausius-Clapeyron relationship (warmer air can hold about 7 % more water vapor) and regional changes in atmospheric circulation (Pfahl et al., 2017; Ritzhaupt and Maraun, 2024). As a result, a strong north-south gradient over Europe is expected: Mean precipitation is projected

to increase in northern Europe and decrease in southern Europe. Although the exact location of the transition zone is uncertain, it robustly shifts to the north in summer and to the south in winter (Ritzhaupt and Maraun, 2023). On an annual average, Austria is close to this transition zone.

# Weather patterns and storm projections over the Atlantic and Europe

As discussed in Section 1.2.1, regional climate and weather extremes in Austria are also controlled by large-scale weather patterns. In winter, the North Atlantic jet is projected to strengthen over the Atlantic and slightly expand into Europe (Harvey et al., 2020). These jet changes are associated with a strengthening of the North Atlantic storm track and its slight expansion into Europe. In the summer, the North Atlantic jet is expected to shift northward, associated with a weakening and slight northward shift of the storm track (Harvey et al., 2020).

In the winter season, climate models project a strong increase in the number of days with a positive North Atlantic Oscillation (NAO), a slight increase in the number of days with a negative NAO, and a significant decrease in the number of days with both Scandinavian blocking and the North Atlantic Ridge (Woollings et al., 2018; Davini and D'Andrea, 2020; Fabiano et al., 2021). The persistence of positive NAO regimes is projected to increase slightly, that of negative NAO to remain constant, and that of the Scandinavian blocking and the North Atlantic Ridge to decrease slightly (Fabiano et al., 2021; Dorrington et al., 2022). A slight decrease in the number of blocked days is also projected for summer (Woollings et al., 2018; Davini and D'Andrea, 2020). There are no studies on changes in the persistence of summer blocking. The frequency of cut-off lows over Europe is projected to increase slightly, with long-lasting events showing a substantial increase in spring and summer (Mishra et al., 2025).

All of the changes described above characterize the average changes across the CMIP climate model ensembles. Simulated changes for individual models may be much stronger, and for weak multi -model mean changes they may even be of opposite sign. Overall, there is only *medium confidence* in changes in atmospheric circulation, and they represent a large source of uncertainty in regional climate change (Shepherd, 2014; Doblas-Reyes et al., 2021).

Various regional feedbacks, including soil moisture-temperature coupling and snow-albedo feedbacks, can further amplify or dampen regional climate changes (Doblas-Reyes et al., 2021).



**Figure 1.4** Projected changes in seasonal mean temperature for (a) summer (July-August) and (b) winter (December–February) for different warming levels, based on ÖKS15 simulations. Observations are derived from SPARTACUS data. Maps depict ensemble mean changes, (c) boxplots show both inter-annual and ensemble range of changes for three regions in Austria. The regions associated with the labels 'West', 'North' and 'South' can be seen in Figure 1.A.3. The changes are calculated as averages (a, b) or annual values (c) of the periods at GWL 1.5°C, GWL 2.0°C, GWL 3.0°C and GWL 4.0°C. They relate to averages at GWL 1.0°C, which corresponds to the period 2001–2020 for the observations and different periods for each ÖKS15 model. The black bar in the boxplots (c) shows annual anomalies from the average of the period 2001–2020. Observations for the indicators shown in this figure are presented in Figure 1.A.7. Data sources: ÖKS15 and SPARTACUS (Truhetz and Leuprecht, 2018; GeoSphere Austria, 2020).

#### **Temperature projections**

Consistent with our physical understanding, seasonal mean temperatures over Austria are clearly projected to increase for all seasons with increasing levels of global warming (high confidence), based on simulations with CMIP3, 5 and 6 global climate models (Meehl et al., 2007; IPCC, 2014b; Lee et al., 2021), and the regional climate models PRU-DENCE, ENSEMBLES and EURO-CORDEX (Christensen and Christensen, 2007; van der Linden and Mitchell, 2009; Jacob et al., 2014). These changes are most pronounced over high terrain, as can be seen for the ÖKS15 simulations (see Figure 1.4, see also Section Elevation dependencies). These changes are also higher than the global mean, mainly due to the stronger warming over land.

In addition to the warming that happened in Austria up to GWL 1.0°C (which occurred in the period 2001–2020), annual mean temperatures are expected to rise by 0.54°C, 1.08°C, 2.19°C and 3.36°C (average of the ÖKS15 ensemble at GWL 1.5°C, GWL 2.0°C, GWL 3.0°C, and GWL 4.0°C, respectively). Mean temperatures are projected to further increase in all seasons with increasing levels of global warming. At GWL 4.0°C, the warming is strongest in summer (+3.4°C), fall (+3.4°C), and winter (+3.3°C), and slightly weaker in spring (+2.7°C) according to the ÖKS15 simulations (Truhetz and Leuprecht, 2018) (Figure 1.4 and Figure 1.A.4).

These changes are accompanied by an increase in heat stress, indicated by a higher number of hot days and tropical nights (Figure 1.5), and a reduction in cold stress, exemplified by a reduced number of ice days and extreme ice days.

As a result of rising mean temperatures, the length and severity of heat waves are also projected to increase. Changes in heat extremes tend to be stronger than changes in mean temperatures because of an increase in temperature variability over central Europe in response to a strengthening of the soil moisture-temperature coupling (Seneviratne et al., 2006; Fischer et al., 2012). Overall, there is high confidence that heat extremes in Austria will become more severe in a warming climate. For example, the number of hot days in a very hot year occurring once in 10 years in the past (climate period 1961-1990) will likely multiply by 3.5 GWL 1.5°C, by 4.3 at GWL 2.0°C, by 5.7 at GWL 3.0°C, and even by 8.1 at GWL 4.0°C on average for Austria. In addition, the average temperature of heat waves will likely increase by 6.6°C at GWL 4.0°C compared to the climate period 1961-1990 (robust evidence, low agreement).

In response to rising temperatures, the vegetation period is projected to start earlier and become longer. The relevance









Figure 1.5 Projected changes in annual mean number of (a) hot days and (b) tropical nights for different warming levels, based on ÖKS15 simulations. Observations are derived from SPARTACUS data. Maps depict ensemble mean changes, (c) boxplots show both inter-annual and ensemble range of changes for three regions in Austria. The regions associated with the labels 'West', 'North' and 'South' can be seen in Figure 1.A.3. The changes are calculated as averages (a, b) or annual values (c) of the periods at GWL 1.5°C, GWL 2.0°C, GWL 3.0°C and GWL 4.0°C. They relate to averages at GWL 1.0°C, which corresponds to the period 2001–2020 for the observations and different periods for each ÖKS15 model. The black bar in the boxplots (c) shows annual anomalies from the average of the period 2001–2020. Observations for the indicators shown in this figure are presented in Figure 1.A.8. Data sources: ÖKS15 and SPARTACUS (Truhetz and Leuprecht, 2018; GeoSphere Austria, 2020).

of temperature changes for glacier and permafrost retreat is discussed in Section 1.3.

#### Precipitation projections

As Austria is located in the transition region between the drying trends of the Mediterranean and the wetting trends of northern Europe, projections are affected by considerable uncertainties depending on the season.

For total annual precipitation, older ENSEMBLES models do not agree on the projected changes over Austria, nor are these changes significant (Jacob et al., 2014). However, the EURO-CORDEX ensemble projects an increase in total annual precipitation (Jacob et al., 2014; Haslinger et al., 2023).

For both winter and summer, projections of seasonal mean precipitation are robust across a range of different global and regional climate model ensembles (Jacob et al., 2014; Rajczak and Schär, 2017; Kotlarski et al., 2023; Ritzhaupt and Maraun, 2023). While most models simulate an increase in winter, a decrease is projected for summer (with some EURO-CORDEX models also showing significant increases). The decrease in summer is confirmed by ensemble projections with convection-permitting regional climate models (Pichelli et al., 2021). In both seasons, however, the contribution of internal variability is strong and may reverse the expected future trends for the next decades (Maraun, 2013). For spring and fall, there is no clear signal, and internal climate variability dominates (Maraun, 2013). Overall, there is high confidence for an increase in winter precipitation and medium confidence for a decrease in summer precipitation over Austria. These changes are clearly evident under high emission scenarios or global warming levels. Figure 1.6 illustrates these changes as projected by the ÖKS15 ensemble.

The number of wet days does not show a clear change signal in winter, spring and fall, but for summer many models agree on a reduced number of wet days under strong warming (Rajczak and Schär (2017): -10 % to -25 % based on EN-SEMBLES and EURO-CORDEX; Pichelli et al. (2021): -4 % to -10 % based on CORDEX-FPS convection-permitting regional climate models, multi-model means for RCP8.5).

Changes in mean precipitation are also reflected in changes in extreme precipitation. Multi-model means for RCP8.5 daily precipitation intensities are projected to increase in winter (up to 25 %), spring and fall (up to 20 %), and with a similar but weaker trend in summer (up to 15 %), based on ENSEMBLES and EURO-CORDEX (Rajczak and

#### Change in seasonal mean precipitation sum (JJA) relative to GWL1.0

(a)



Figure 1.6 Projected changes (in %) in seasonal mean precipitation sum for (a) summer (July-August) and (b) winter (December-February) for different warming levels, based on ÖKS15 simulations. Observations are derived from SPARTACUS data. Maps depict ensemble mean changes, (c) boxplots show both inter-annual and ensemble range of changes for three regions in Austria. The regions associated with the labels 'West', 'North' and 'South' can be seen in Figure 1.A.3. The changes are calculated as averages (a, b) or annual values (c) of the periods at GWL 1.5°C, GWL 2.0°C, GWL 3.0°C and GWL 4.0°C. They relate to averages at GWL 1.0°C, which corresponds to the period 2001–2020 for the observations and different periods for each ÖKS15 model. The black bar in the boxplots (c) shows annual anomalies from the average of the period 2001-2020. Observations for the indicators shown in this figure are presented in Figure 1.A.9. Data sources: ÖKS15 and SPARTACUS (Truhetz and Leuprecht, 2018; GeoSphere Austria, 2020).

Schär, 2017). Extremes of daily precipitation are projected to increase in all seasons, though the strength of the signal varies considerably across the considered ensemble (as shown in Figure 1.7) and indicator (Rajczak and Schär, 2017; Ritzhaupt and Maraun, 2023). For summer, coarse-resolution climate models simulate a decrease in extreme precipitation, but there is evidence that their resolution is too coarse to represent topographic influences on convection, making their simulated changes implausible (Giorgi et al., 2016; Ritzhaupt and Maraun, 2023). Overall, there is *medium to high confidence* for an increase in extreme daily precipitation for winter, spring, and fall, and *medium confidence* for summer.

For extreme rainfall over 5-day periods, there is a clear signal for strong warming in winter (up to 25 %), spring, and fall (up to 20 %) (*medium confidence*), and no signal for summer (RCP8.5, multi-model means from ENSEMBLES and EURO-CORDEX) (Rajczak and Schär, 2017).

Newly available convection-permitting climate models with kilometer-scale resolution (Coppola et al., 2020) were evaluated for changes in hourly precipitation extremes for summer and fall (Pichelli et al., 2021). These models project higher intensities and extremes for both seasons (with a slight decrease in summer extremes over the eastern lowlands). However, the number of rainfall hours is projected to decrease in summer and fall (with a slight increase in the northeast). These results are broadly consistent with changes simulated by the driving EURO-CORDEX regional climate models, but have not been assessed for significance or robustness. Pseudo-global warming studies (e.g., Doblas-Reyes et al., 2021) of a single event suggest the possibility of a super-Clausius Clapeyron dependence (relative to global temperature increase) of extreme short-term summer rainfall, but this may be counteracted by a possible stabilization of the atmosphere (Maraun et al., 2022). Given the limited evidence, the lack of full physical interpretation, and the absence of uncertainty assessment, confidence in the projected increase in hourly rainfall extremes in summer and fall is currently low.

There has been little research on severe convective storms over Europe such as mesoscale convective systems and supercells, which can, in turn, cause hailstorms and tornadoes. One reason is the limited availability, until recently, of climate simulations representing such storms. As a way out, Brooks (2013) and Allen (2018) suggest the analysis of largescale atmospheric environments conducive to such storm events, i.e., high instability and low-level moisture as well as strong deep-layer wind shear. EURO-CORDEX simulations

#### (a) Change in seasonal precipitation extremes (JJA) relative to GWL1.0





Figure 1.7 Projected changes (in %) in seasonal extremes (99.9 percentile) of daily precipitation sums for (a) summer (July-August) and (b) winter (December-February) for different warming levels, based on ÖKS15 simulations. Observations are derived from SPARTACUS data. Maps depict ensemble mean changes, (c) boxplots show both inter-annual and ensemble range of changes for three regions in Austria. The regions associated with the labels 'West', 'North' and 'South' can be seen in Figure 1.A.3. The changes are calculated as averages (a, b) or annual values (c) of the periods at GWL 1.5°C, GWL 2.0°C, GWL 3.0°C and GWL 4.0°C. They relate to averages at GWL 1.0°C, which corresponds to the period 2001–2020 for the observations and different periods for each ÖKS15 model. The black bar in the boxplots (c) shows annual anomalies from the average of the period 2001-2020. Observations for the indicators shown in this figure are presented in Figure 1.A.10. Data sources: ÖKS15 and SPARTACUS (Truhetz and Leuprecht, 2018; Geo-Sphere Austria, 2020).



## (a) Change in SPEI of dryest summer month (Apr-Sep) relative to GWL1.0

Figure 1.8 Projected changes in the water balance surplus or deficit, represented by the standardized precipitation-evapotranspiration index (SPEI) of the driest month in the summer half-year (Apr-Sep) for different warming levels, based on ÖKS15 simulations. Observations are derived from SPARTACUS data. (a) Maps depict ensemble mean changes, (b) boxplots show both inter-annual and ensemble range of changes for three regions in Austria. The regions associated with the labels 'West', 'North' and 'South' can be seen in Figure 1.A.3. The changes are calculated for events with (a) a 10-year return period (10th percentile) or (b) annual values of the SPEI for the periods at GWL 1.5°C, GWL 2.0°C, GWL 3.0°C and GWL 4.0°C. They relate to (a) the 10th percentiles or (b) averages of SPEI at GWL 1.0°C, which corresponds to the period 2001-2020 for the observations and different periods for each ÖKS15 model. The black bar in the boxplots (b) shows annual anomalies from the average of the period 2001-2020. Observations for the indicators shown in this figure are presented in Figure 1.A.11. Data sources: ÖKS15 and SPARTACUS (Truhetz and Leuprecht, 2018; GeoSphere Austria, 2020).

at 0.44° grid spacing project a robust increase in such environments under strong warming, except at high altitudes, due to an increase in low-level moisture (Púčik et al., 2017). Chan et al. (2023) analyzed projections of mesoscale convective systems over Europe in two convection-permitting models and found inconsistent changes in frequency, magnitude, and velocity, but a consistent increase in peak intensity, total precipitation volume, and temporal clustering.

#### **Drought projections**

A simple and easy-to-interpret indicator of meteorological drought is the number of consecutive dry days within a season or year. For winter, there is no change in this indicator, but for summer, a robust increase is found in EURO-COR-DEX simulations (Dosio, 2016).

Soil moisture or agricultural drought takes into account the drying of the soil caused by a lack of precipitation and strong evapotranspiration due to high temperatures. A study based on two hydrological and two land surface models driven by the CMIP5 global climate model ensemble shows an increase in the duration of extreme soil moisture drought events (from 30 to more than 100 months) and the average number of drought months (from 2-3 to 3-5 months per year) under 3°C warming (Samaniego et al., 2018). The latter changes are particularly large for eastern Austria. Similar results are found in other studies using the regional climate models ENSEMBLES and EURO-CORDEX (Heinrich and Gobiet, 2012; Haslinger et al., 2016; Spinoni et al., 2018; Haslinger et al., 2023). The driving factor for these robust changes is rising temperatures. Overall, there is high confidence for more severe agricultural droughts with global warming over Austria. Figure 1.8 shows the projected changes in the standardized precipitation-evapotranspiration index (SPEI) based on the ÖKS15 ensemble. The SPEI indicator shows negative values for all GWLs in the summer half-year, which corresponds to an increase in dry periods.

#### Wind projections

Wind is a variable that directly represents atmospheric dynamics, and projections are therefore affected by high uncertainties. These uncertainties are exemplified by a wide range of plausible changes simulated for winter wind speed over Austria, including no change or a significant increase depending on the actual large-scale circulation changes (Zappa and Shepherd, 2017).

For the annual mean, no robust signal in wind speed changes is found in the EURO-CORDEX projections (Wohland, 2022). Consistent with this, Tobin et al. (2016) find a wide range of possible changes in the annual mean wind energy yield. Similar results are found for wind energy output, where two models suggest no change (Hueging et al., 2013) and an ensemble of nine combinations of two RCMs and five GCMs projects a slight decrease (Moemken et al., 2018). The latter results vary over the seasons, with a consistent decrease in summer. Uncertainties are high for projections of sub-daily wind variability, although they tend to increase with strong warming. Monthly variability is projected to increase (Tobin et al., 2016).

For wind extremes, Outten and Sobolowski (2021) find a slight increase in 30-year return periods in the EURO-CORDEX simulations, but without explicit uncertainty assessment. An earlier study found a strong dependence of extreme wind gust projections on the driving GCM, with no robust signal emerging (Nikulin et al., 2011).

Only one study has analyzed climate projections for foehn winds in the Alps (Maier et al., 2025). In the EURO-COR-DEX simulations, the frequency of southern foehn events over Tyrol is projected to decrease slightly, while widespread events are projected to increase in both Tyrol and Vorarlberg. A seasonal shift from July to October into the spring months is projected.

#### **Elevation dependencies**

Climate change is known to be elevation dependent (Pepin et al., 2022) (see also Section 1.2.1 for observations). In both ENSEMBLES (Kotlarski et al., 2015) and EURO-CORDEX (Kotlarski et al., 2023), the warming rates have a maximum between about 1,500–2,000 m in spring and between about

2,000–2,500 m in summer. For winter, the warming rate saturates above 1,500 m, and for fall, the warming rate increases with elevation up to the resolved altitude. However, a comparison of global climate model simulations with resolutions varying from about 16–125 km found a strong but unsystematic dependence of the existence and strength of elevation-dependent warming on model resolution (Palazzi et al., 2019). These findings highlight the associated uncertainties.

Precipitation projections in the ENSEMBLES simulations also show a clear elevation dependence, but with strong regional and seasonal variations (Kotlarski et al., 2015). For the Eastern Alps, positive trends decrease with altitude and negative trends increase.

Not surprisingly, the number of snow days (Kotlarski et al., 2015) and snow water equivalent (Kotlarski et al., 2023) show clear elevation-dependent changes (*high confidence* based on simulations and physical understanding). Decreases in the number of snow days are greatest at intermediate elevations of about 1,000 m in winter, 1,500 m in spring, and 2,000 m in summer and fall. Decreases in snow water equivalent from September to May are higher at lower elevations. For low warming, almost no decrease is found above 2,500 m, while for high warming levels, significant decreases are found at all elevations.

Changes in snow are discussed in more detail in Section 1.3.1.

#### Chapter Box 1.1. Global warming levels and their implications for Austria

#### Concept and method in AAR2

Since the adoption of the Paris Agreement in 2015, the global and regional implications of a 1.5°C or a 2.0°C warmer world gained traction in the public, political and scientific debate. In its Special Report on Global Warming of 1.5°C (IPCC, 2018), the IPCC assessed a range of climate change impacts at these two levels of global mean surface temperature (GMST) increase. In the sixth Assessment Report cycle, global warming levels (GWLs) are used as a 'robust and useful dimension of integration' (Chen et al., 2021) for comparing climate studies across many scientific disciplines and beyond.

This integrative property of GWLs stems from a different way of looking at climate projections. Most model-based climate projections are derived from models that simulate the climate system and its interactions on a global scale. Conventionally, climate change and its impacts are categorized in terms of time periods (e.g., mid-century) and underlying greenhouse gas (GHG) emission scenarios outlined in the IPCC Special Report on Emissions Scenarios (SRES) (Nakićenović et al., 2000), representative concentration pathways (RCPs) (Van Vuuren et al., 2011), or shared socio-economic pathways (SSPs) (Riahi et al., 2017). GWLs, on the other hand, are defined as a point in time when GMST

exceeds a certain threshold relative to a (most often pre-industrial) reference period, as illustrated in Figure 1.A.1. All subsequent analyses refer to this period. The main difference to the conventional approach is that the same GWL can include model results from different time periods, emission or concentration scenarios. In a sense, the GWL perspective bypasses the uncertainties associated with the underlying forcing scenarios and model-internal climate sensitivity and looks directly at the resulting global temperature change.

The GWLs concept also involves some inherent assumptions and shortcomings. These are discussed extensively in the literature (e.g., James et al., 2017) but a few examples are presented here. First, while a number of regional climate impacts have been shown to be linearly related to GMST and therefore well represented by GWLs (Wartenburger et al., 2017; Seneviratne et al., 2018b; Lewis et al., 2019; Tebaldi et al., 2020; Iyakaremye et al., 2021), showing changes in terms of GWLs can mask substantial differences between variables with a strong dependence on local or regional aero-sol concentrations, such as mean precipitation or local precipitation extremes (McCoy et al., 2022; Persad et al., 2022). Second, the transient (warming) or equilibrium (stabilized) state of the climate at a given GWL can lead to substantially different regional and local climate conditions (Seneviratne et al., 2018a; King et al., 2020, 2021; Zhang and Zhou, 2021). However, because equilibration occurs on centennial timescales (Rugenstein et al., 2019), transient behavior is more relevant to impact studies, and the AAR2 therefore only considers the 20-year period in which a global temperature threshold is first crossed. This, in turn, limits the ability to represent slow-changing or delayed impacts, such as glacial retreat, purely in terms of GWLs. In a few cases throughout AAR2, such impacts are therefore additionally characterized by time periods or forcing scenarios.

Details of the method used to identify GWLs from global climate model output are presented in Becsi and Formayer (2024). The main GWLs throughout the AAR2 are 1.5°C, 2.0°C, 3.0°C and 4.0°C of GMST increase relative to 1850–1900. Where study results were available only for predetermined periods, the approach was reversed to calculate discrete GWLs for the respective periods. GWL periods for the most commonly used models are provided in Table 1.A.2.

#### The shift to CMIP6

While the ÖKS15 climate scenarios were based on regional climate models forced by global climate models of the fifth coupled model intercomparison project cycle (CMIP5) (Taylor et al., 2012), global models of the newer generation CMIP6 (Eyring et al., 2016) have now been released. It will be years before local climate scenarios downscaled from the CMIP6 GCM/RCM chain are available<sup>1</sup>. However, Becsi and Formayer (2024) have shown with the Austrian climate scenarios ÖKS15 that GWLs are well suited to represent temperature changes at the regional scale, and that generational shifts in global climate models can be accounted for using the GWLs concept. There are two main conclusions of the study:

- Warming trends for future periods are significantly higher in CMIP6 than in CMIP5, and therefore future GWLs are reached earlier in CMIP6 than in CMIP5. This acceleration amounts up to a decade between the CMIP5 and CMIP6 ensemble medians at GWL 4.0°C, and less at lower GWLs.
- Since models usually either over- or underestimate temperature trends, the related errors in absolute temperatures can accumulate over time. Because the GWL concept characterizes climate projections on the basis of a silimar climate instead of similar timing, it can adjust for this type of error in downstream scenario analysis.

<sup>&</sup>lt;sup>1</sup> The initiative 'Klimaszenarien.AT' is currently developing new Austrian climate scenarios. See <u>klimaszenarien.at</u> for more information.

Box 1.1 Figure 1 supports these findings. It shows the differences in timing between CMIP5 and CMIP6 in reaching a given GWL (a) and the differences in regional temperature signals at that GWL (b). The baseline for all changes is GWL 1.0°C rather than a fixed time period. As the figure shows, the GWL concept accounts for the different warming rates in the models and selects periods in such a way that regional temperature changes between GWLs are quite stable across model generations in Austria.

This approach of relating changes to GWLs instead of reference periods is adopted in the following sections, so that these changes can be assumed to be valid also in light of the transition from CMIP 5 to CMIP6. This underlines the integrative capacity of the GWLs concept to compare climate impacts across time periods, emissions scenarios and even model generations.



**Box 1.1 Figure 1** (a) Different timing of GWL periods and (b) associated temperature change at four global warming levels (GWLs) in comparison between selected CMIP5 and CMIP6 models in Austria. The boxes represent GWL 1.5°C, GWL 2.0°C, GWL 3.0°C and GWL 4.0°C for the CMIP5 and CMIP6 model ensembles (box pairs from left to right). All changes are relative to GWL 1.0°C, i.e., to a different 20-year period for each model. The numbers in parentheses after the legend entries indicate the number of models included. In the observational data, GLW 1.0°C corresponds to the period 2001 to 2020.

#### The cold bias in ÖKS15

Recent warming trends are underestimated in the ÖKS15 ensemble. Since the downscaling approach applied in the scenario development (Switanek et al., 2017) preserves the change signals of the EURO-CORDEX models in the background, this cold bias is rooted in the RCMs and has also been identified in the literature (Kjellström et al., 2018; Schumacher et al., 2024).

It is further exacerbated by the fact that the period chosen to adjust the bias in the raw EURO-CORDEX models with observational data is 1971–2000, which is 20 years behind the current climate normal period (World Meteorological Organization (WMO), 2017). The cold bias in the model trends has therefore accumulated over time to a significant gap compared to the current observed warming.

Box 1.1 Figure 2 illustrates the lagged warming of the ÖKS15 scenarios compared to observations (reference period 1971–2000). This period is relevant for all applications that use the absolute values of the ÖKS15 temperature data. All temperature values are averaged over the area of Austria and the respective year. The reference point for the observed warming is the year 2024. It is calculated from annual observations using a 41-year LOESS filter (Clarke and Richardson, 2021), since the common approach of using averages of climate normal periods lags at least 15 years (for 30-year periods) behind current climatic conditions. The LOESS filter approach is better suited to represent recent warming. The same filter function was applied to the ÖKS15 temperature data. The figure highlights the temporal perspective of the cold bias rooted in ÖKS15: Current (2024) temperature levels are reached in 2050 in the ensemble median of RCP8.5 and in 2065 in the ensemble median of RCP4.5. The delay is 27 years (RCP8.5) and 42 years (RCP4.5), respectively. Almost all models in RCP2.6 are below recent levels of temperature increase by 2100.




To mitigate errors accumulated over the past 40 years since the period used for bias-adustment, model trends can be compared against more recent observational data. Though even when relating changes to the current climate normal period (1991–2020), almost no model in the RCP2.6 ensemble reaches the observed temperature level of 2024 until 2100. By relating the changes to GWL 1.0°C instead of a fixed reference period, the resulting adjustment (see Section The shift to CMIP6) significantly reduces the delay to observed levels of warming.

As a guideline for interpreting ÖKS15-based climate studies in the light of this cold bias, Figure 1.A.2 quantifies the time lag of all ÖKS15 models for commonly used reference periods as well as for GWL 1.0°C. The corresponding data are shown in Table 1.A.3.

There are some general recommendations for handling the cold bias in ÖKS15 in future climate change studies. Foremost, it makes sense to minimise the period over which past errors accumulate by selecting an observational reference period as close to the current climate as possible, and only evaluating future model trends from that point on. In light of the recent warming rate of 0.6°C per decade in Austria, 30-year periods might be too long for averages to represent stationary conditions, and 20-year periods might be better suited for that purpose.

Where feasible, relating changes to GWLs instead of determined periods can reduce the timing-sensitive part of the error and has the benefit of being robust across model generations. However, there are cases where climatological averages do not provide the right scope. Some impact models require transient data on (sub)daily time scales as input. These cases could still benefit from the trend-adjusting properties of the GWLs concept, for example when time series are initialised at a given GWL rather than a fixed time period for each input model. Although the GWLs perspective provides a helpful way to maintain the usefulness of the ÖKS15 climate scenarios for the coming years, the issue of underestimated warming trends can ultimately only be addressed with the release of the next generation of Austrian climate scenarios.

### Chapter Box 1.2. Tipping points in the climate system

### What are tipping points?

The IPCC defines a tipping point as a critical threshold beyond which a system undergoes a significant and often abrupt or irreversible reorganization into a new state. A tipping element, in turn, refers to 'a component of the Earth system that is vulnerable to reaching such a tipping point' (Chen et al., 2021). Tipping points are characteristic of systems with amplifying feedbacks, where small disturbances can lead to disproportionately large responses. Moreover, once an irreversible tipping point is crossed, the system does not return to its previous state, even if the original forcing is reversed.

As early as the 2000s, the IPCC cautioned that tipping points could be reached within the coming centuries if greenhouse gas emissions continued to increase (IPCC, 2001). However, more recent evidence suggests that these thresholds could be crossed at lower levels of warming and much sooner – possibly within this century – raising the risk of cascading effects that could impact other critical components of the climate system (Möller et al., 2024).

Tipping points are identified through expert judgement that integrates observational data, paleoclimatic evidence, and Earth system model simulations. This evaluation is complemented by an assessment of the feedback mechanisms specific to each tipping element, the global temperature thresholds that could trigger them, their potential irreversibility, the timescales of their transitions, and their likely global and regional impacts (Lenton et al., 2008; Armstrong McKay et al., 2022). Based on these criteria, tipping points have been identified in the biosphere, cryosphere, and ocean-atmosphere systems (see Box 1.2 Table 1).

While tipping points have occurred in the past without human influence – such as the abrupt climate shift 11,700 years ago that marked the beginning of the Holocene epoch – this report focuses on tipping elements that are vulnerable to human activities, including greenhouse gas emissions and land-use change, that have the potential to influence the global climate system. Given Austria's integration into the global climate and societal system, any tipping point affecting the planets' climate is likely to have an impact on the country.

Below, we introduce two tipping elements that could greatly impact Austria. Box 1.2 Table 1 provides a summary of the major climate tipping points identified to date, along with their associated global temperature thresholds. These tipping elements have the capacity to cause profound disruptions to Earth's systems and human societies.

### Tipping elements with potentially high impacts on Austria

#### A collapse of the Atlantic Meridional Overturning Circulation (AMOC)

The AMOC is a critical ocean current system that regulates climate patterns globally. Its surface branch transports warm, salty water northward, while its deep branch transports cold water southward. This circulation is primarily driven by deep water formation, which occurs when surface water cools and sinks in the northern Atlantic.

The AMOC is vulnerable to climate change, as rising temperatures slow the cooling of surface water, weaken deep water formation, and reduce the strength of the current. Evidence suggests that the AMOC has weakened by about 15 % since the mid-20th century (Caesar et al., 2018; Thornalley et al., 2018), with further declines projected by Earth system models, including those from the CMIP projects (Weijer et al., 2020).

Paleoclimate records and theoretical and modeling studies suggest that the AMOC may be reaching a tipping point. Stommel (1961) proposed that the stability of the AMOC depends on the delicate balance between the effects of temperature and salinity on water density. Currently, the system is primarily temperature driven, with cold water in the northern latitudes maintaining a strong AMOC. However, a significant influx of freshwater – resulting from accelerated melting of the Greenland ice sheet or increased precipitation – could disrupt this balance, increasing the risk of AMOC collapse.

While most current climate models do not predict a complete AMOC shutdown during the 21st century, some projections suggest that a collapse could occur before 2300 if greenhouse gas emissions continue to increase (Bakker et al., 2016). A recent study, based on a single model, simulated an AMOC collapse and identified the processes that unfolded both near and during the tipping event. By comparing the tipping 'fingerprints' identified by the model with reanalysis data, the authors concluded that a tipping event could occur within this century (Van Westen et al., 2024). However, the temperature threshold for such a collapse remains highly uncertain, with considerable variability among model projections. An AMOC collapse would have profound consequences, including substantial cooling in the North Atlantic and Europe, disrupted rainfall patterns, and stronger winter storms, positioning the AMOC as a low-probability but high-impact tipping point in the near future.

#### Collapse of the Greenland Ice Sheet (GIS)

The GIS is shrinking at an accelerating rate due to increased surface melting and increased ice calving into the ocean, both driven by rising global and regional temperatures (IPCC, 2019b). Although complete melting of the GIS is unlikely to occur abruptly and would take at least the entire millennium, there is strong evidence that the GIS may be approaching – or may have already passed – a critical tipping point beyond which its melting could become irreversible (Höning et al., 2023). This process is strongly influenced by feedback mechanisms, with elevation feedback playing a key role. As the ice sheet melts and loses elevation, its surface becomes more exposed to the warmer temperatures at lower altitudes, further accelerating the rate of melting.

The complete melting of the GIS would result in a global sea level rise of about 7.2 meters, with far-reaching consequences, including potential disruptions to ocean and atmospheric circulation patterns that could significantly affect Europe (IPCC, 2019b).

**Box 1.2 Table 1** Overview of the changes in global mean surface temperature at which global tipping points may be triggered, the timescales associated with transitions between states, and the estimated impacts (global and local additional warming) of the global core tipping points identified by Armstrong MacKay et al. (2022). The leftmost categories represent Earth system domains: CR for cryosphere, BI for biosphere, and AO for atmosphere-ocean. In addition, colors assigned to element names and estimates indicate the subjective confidence levels. Table from Ossó and Roither (2024).

Earth system domain	Proposed climate tipping element and (tipping point)	Threshold (°C)			Timescale (years)			Maximum impact (°C)	
		Best Estimate	Min.	Max.	Best Estimate	Min.	Max.	Global	Regional
CR	Greenland Ice Sheet (collapse)	1.5	0.8	3.0	10k	1k	15k	0.13	0.5 to 3.0
CR	West Antarctic Ice Sheet (collapse)	1.5	1.0	3.0	2k	500	13k	0.05	1.0
CR	East Antarctic Subglacial Basins (collapse)	3.0	2.0	6.0	2k	500	10k	0.05	?
CR	Arctic Winter Sea Ice (collapse)	6.3	4.5	8.7	20	10	100	0.60	0.6 to 1.2
CR	East Antarctic Ice Sheet (collapse)	7.5	5.0	10.0	?	10k	?	0.60	2.0
BI	Labrador-Irminger Seas/ SPG* Convection (collapse)	1.8	1.1	3.8	10	5	50	-0.5	3.0
BI	Atlantic Meridional Overturning Circulation (collapse)	4.0	1.4	8.0	50	15	300	-0.50	-4 to -10
AO	Amazon rainforest (dieback)	3.5	2.0	6.0	100	50	200	Partial 0.1 Total 0.2	0.4 to 2.0
AO	Boreal Permafrost (collapse)	4.0	3.0	6.0	50	10	300	0.2 to 0.4	

\* SPG: Subpolar Gyre

```
High confidence
```

Medium confidence

Low confidence



Chapter Box 1.3. Case study: Lake Neusiedl as sentinel of climate change

Box 1.3 Figure 1 Mean annual changes in surface water temperatures of Lake Neusiedl. From 1976 to 2022, the water temperature has increased by 1.9°C (data source: Hydrographischer Dienst Burgenland, <u>wasser.bgld.gv.at/</u>).

Lake Neusiedl, Austria's largest lake, is exposed to wind, extremely shallow, and is currently losing substantial amounts of lake water due to rising air temperatures that increase evaporation. Large floods and several droughts have been recorded at Lake Neusiedl in the past, and it last dried up in 1865–1868 (Draganits et al., 2022). Unlike other lakes in Austria, Lake Neusiedl receives most of its water from direct precipitation and is disconnected from groundwater inflow (Tchaikovsky et al., 2019). Thus, the current loss of lake water is mainly the result of water evaporation due to high temperatures (Box 1.3 Figure 1) and decreasing precipitation. Compared to its average, long-term (since 1965) lake surface water level of ~115.5 m a.s.l., Lake Neusiedl in 2022 was ~10–50 cm below this long-term average water level<sup>2</sup> (*high confidence*).

Ongoing climate change will have a major impact on the future of Lake Neusiedl and its surrounding ecosystems. Current climate change models, such as the ÖKS15 for Burgenland, show that lower average water levels can be expected, especially during the summer months. This applies not only to Lake Neusiedl, but also to neighboring water bodies such as the soda pans. The maximum water temperatures of Lake Neusiedl have continued to increase and often exceed 30°C in summer. Therefore, higher evaporation rates are to be expected with as yet unknown effects on the biocoenosis.

<sup>2</sup> wasser.bgld.gv.at/hydrographie/die-seen/mittler-wasserstand-neusiedler-see

# 1.3. Cryosphere

This chapter focuses on the description of the Austrian cryosphere, including the regional/temporal occurrence, drivers, and magnitude of historical and future changes in response to climate forcing. Interactions and impacts on other natural and socio-economic systems are briefly mentioned, but as the perennial cryosphere in Austria is limited to high elevations, the broader context of the relevance of the cryosphere and the impacts of its change are discussed in Chapter 7.

# 1.3.1. Snow

Snow plays an important role in Austria, as its extent and duration are relevant for soil moisture, river flow characteristics, landscape appearance, and winter tourism. These factors in turn influence regional ecology, economy, agriculture and forestry (drought risk, forest fire risk), vegetation distribution, reservoir recharge, flood risk, snow avalanche and snow load risk, and hydropower yield.

#### Observed snow depth and snow cover duration

There is a growing body of literature on observed trends in snow-related indicators in the Alpine region, including Austria. For example, Klein et al. (2016) analyzed Swiss station data for the period 1970-2015 and found an average reduction in snow cover duration (SCD) of 8.9 days per decade. Schöner et al. (2019) found stronger negative trends in snow depth (SD; up to -12 cm per decade) in the southern parts of the Alps than in the northern parts between 1961 and 2012. This is also confirmed by Bozzoli et al. (2024), who evaluated snowfall depth over the European Alps between 1920 and 2020 and found reductions of about 4-5 % per decade in the southern Alps and about 2.3 % per decade in the northern Alpine region. Olefs et al. (2021) found a consistent decrease in SD across Austria of about 3 cm per decade on average and a decrease in SCD of about 7 days per decade in the period 1961-2020 (see also Figure 1.9: Gray bars, for the change within the last 30 years). In an alpine-wide analysis for the period 1971-2019, Matiu et al. (2021) found negative trends in all snow indicators below 2,000 m a.s.l., e.g., a decrease in SCD of about 5 days per decade. Marty et al. (2017b, 2023) analyzed long-term SWE time series in several Alpine countries, including Austria, and found, e.g., trends in the date of end of snow cover of about -3 days per decade in the period 1957-2022. Vorkauf et al. (2021) found earlier snowmelt dates of -2.8 days per decade between 1958 and 2019, which

is consistent with the SCD trends noted above. In general, most studies find stronger changes in spring (earlier melting) than in fall (later onset of snow cover). In summary, estimates for recent changes in snow cover duration (SCD) in Austria and the entire Alpine region range from about -5 to -10 days per decade.

### Future snow depth and snow cover duration

There are few studies on future snow conditions in Austria. Earlier studies for neighboring regions include Marty et al. (2017a, 2017b, 2023), who analyzed two catchments in the Swiss Alps and found that a decrease in SCD of about 10 days per decade can be expected at an elevation of about 1,500 m a.s.l. by the end of the 21st century, assuming the A2 scenario. In an early Austrian case study for a ski resort in Styria, Marke et al. (2014) found a decrease in SCD of 5 to 6 days per decade in the first half of the 21st century, assuming the A1B scenario. Frei et al. (2018) found an area-averaged decrease in snowfall over the Alps of about 25 % for the RCP4.5 scenario and 45 % for the RCP8.5 scenario. Matiu and Hanzer (2022) found an overall reduction in snow cover fraction of 14 % for RCP2.6 and 48 % for RCP8.5 for 2071-2100 compared to 2001-2020. Le Roux et al. (2023) projected snowfall extremes in the French Alps and concluded that annual maxima of daily snowfall are expected to decrease below 3,000 m a.s.l., but even increase above 3,600 m a.s.l. in the second half of the 21st century under the RCP8.5 scenario. Kotlarski et al. (2023) analyzed the loss of snow water equivalent (SWE) in the Alps and found, e.g., a reduction of about 20 % in the low-end scenario RCP2.6 and 80 to 90 % in the high-end scenario RCP8.5 at altitudes below 500 m a.s.l. by the end of the 21st century. Recently, a set of snow cover scenarios based on observations and the regional, bias-adjusted ÖKS15 climate simulations was published for Austria (Project FuSE-AT) (Koch, 2021b), used as a basis for climate change impact studies (e.g., Kiem et al., 2024) and expert opinions in the winter tourism sector.<sup>3</sup> Figure 1.9 shows the estimates of SCD averaged over Austria at different elevation levels based on the FuSE-AT results. From the previous (1961-1990) to the current climate normal period (1991-2020), SCD decreased significantly at all elevation levels, on average by about 7 days per decade (Olefs et al., 2021). By the year 2050, the SCD in Austria will most likely continue to decrease by at least 5 days per decade on average. With respect to the GWLs, we expect a further decrease of the SCD by 10 to 15 days at

Examples are given at fuse-at.ccca.ac.at



Figure 1.9 Snow cover duration (SCD) in Austria at different elevation levels. Left: Days with SD >1 cm; Middle: Days with SD >10 cm; Right: Days with SD >30 cm (Source: GeoSphere Austria).

GWL 2.0°C, by 35 to 45 days at GWL 3.0°C, and by 60 to 80 days at GWL 4.0°C. This estimate applies for all except for the lowest elevations, where the current SCD is already shorter than the projected changes for higher elevations.

### Related consequences

Feedback on radiative forcing: Kiem et al. (2024) showed that the additional radiative forcing due to decreasing albedo caused by snow cover retreat within 80 years (2021–2100) corresponds to 25–500 % (depending on emission scenario and climate model) of the radiative forcing effect of Austria's total annual  $CO_2$ eq emissions in 2021. Natural hazards: Snow is associated with a number of natural hazards, including droughts (Sections 1.2, 1.4.3), floods (Section 1.4.1), wildfires (Cross-Chapter Box 1), and snow avalanches (Sections 2.2.1, 7.4.1).

Winter tourism and recreation (see also Sections 2.2.1 and 4.7.2): The FuSE-AT project assessed the meteorological conditions necessary for the production of technical snow (Koch, 2021a). Over two consecutive 30-year periods (1991–2020 vs. mid-century 2021–2050), the number of hours in December (a critical period for base snowmaking) with favorable conditions for technical snow production is estimated to decrease by 10–15 % at 1,000 m and 5–10 % at 2,000 m a.s.l.

# Cross-Chapter Box 1. Evaluation of climate-related risks and key risks

Sven Fuchs; Birgit Bednar-Friedl; Herbert Formayer; Katharina Gangl; Veronika Gaube; Susanne Hanger-Kopp; Margreth Keiler; Stefan Kienberger; Sarah Haider-Nash; Stefanie Peer; Thomas Schinko; Jan Steinhauser

This box provides an overview of the concepts used to derive risks and key risks relevant to the AAR2 and builds on the concepts and terminology used in recent IPCC reports (Garschagen et al., 2019; IPCC, 2022b). Burning ember diagrams of key risks for Austria are provided. Key terms used for risk assessment in the AAR2 are listed in the glossary, which is based on IPCC (2022a). Further details on the used approach are provided in the 1.A.3.

# The nature of climate-related risk

Throughout the chapters of the AAR2, risk provides a framework for understanding the potentially severe impacts of climate change on ecosystems and the human system, and how adverse consequences can be reduced for the future. *Risk* – defined as the potential for adverse consequences – results from the interaction of *vulnerability* (of the affected system), its *exposure* over time (to the hazard), and the (climate-related) *hazard* and the likelihood of its occurrence (CCBox 1 Figure 1). *Impacts* are the consequences of realized (materialized) risks on natural and human systems, such as observed losses and damage related to hazard events.



**CCBox 1 Figure 1** Core components of risk and related terms. The figure is based on the initial IPCC risk figure and the conceptual framework of risk-adaptation relationships used in the SROCC (Garschagen et al., 2019), but has been slightly adapted for the AAR2 (for further discussion, see Fuchs et al., 2024). The white arrows illustrate actions affecting hazard, exposure, and vulnerability that shape risks over time (increase, decrease). The red line in the center illustrates hard limits to adaptation, and the outer dark gray solid line illustrates the status quo of risk. The dark gray dashed line illustrates the soft limits of adaptation. The area between the red and dark gray lines (light purple) is the solution space. The remaining residual risk is the inner core of the risk figure (white), defined by the red line. Illustrative examples for risk management actions addressing each of the three risk components are provided.

The AAR2 assessment recognizes the complex nature of climate-related risks by addressing changes in hazard, exposure and vulnerability, including feedback and possible cascading processes of overall ecosystem and anthroposphere impacts. Hazard, exposure and vulnerability may each be subject to uncertainty and may vary in time and space due to socio-economic dynamics and human decision-making. Thus, risk may increase or decrease over time in relation to the dynamic interactions between the three risk components, which in turn may increase or decrease over time. *Adaptation* and risk management play an important role in reducing the overall risk associated with climate change. However, there are also limits to adaptation due to hard limits (no adaptive action is possible to avoid intolerable risks) or soft limits (options may exist but are not currently available to avoid intolerable risks through adaptive action) that will result in increasing losses and damages. The remaining solution space (CCBox 1 Figure 1) includes options for strategies and measures that are available and appropriate to address climate risks.

# Climate-related risks as assessed in this report

Our understanding of climate-related risks is based on multiple and diverse sources of information, including climate change science, natural sciences, engineering, social sciences, and economics. The standardized workflow for climate-related risk assessment in Austria is based on the corresponding IPCC framework and workflow (IPCC, 2022b: Chapter 16). Following the structure of the AAR2, the assessment of climate-related risks and the identification of key risks are organized in a cross-chapter approach. Chapter 1 provided information on climate change, climate-related hazards and ecosystem impacts (see Table 1.A.4). Based on the information provided in Chapter 1, Chapter 2, Chapter 3, Chapter 4 and Chapter 7, we evaluated the observed impacts within their respective subsystems and assessed the future evolution of climate-related risks, shaped by changes in climate-related hazards as well as exposure and vulnerability. These risks

were further evaluated through the filters of *severity* and *urgency* to identify and describe Austria's key risks. While some key risks may already reflect dangerous disruptions today, they may typically become more severe over time due to changes in the nature of the hazards and/or the exposure/vulnerability of societies or ecosystems to these hazards. They may also become severe due to the adverse consequences of adaptation or mitigation responses to the risk. Key risks are identified as potentially urgent to determine the priority for action with different adaptation/risk management strategies according to (societal) risk preference/tolerance and soft/hard adaptation limits to be crossed. In the next step, these results were assessed in Chapter 5 and Chapter 6, taking into account the needs in the context of societal adaptation options and from a governance perspective, and fed into Chapter 8.

# Resulting climate-related key risks for Austria

Rising temperatures lead to health risks, especially in urban areas and to reduced labor productivity: Increasing frequency and duration of heat waves, including tropical nights, pose a risk to human health, especially for people living in urban areas and those exposed to heat in their working environment (*high confidence*). A GWL 4.0°C leads to an increase in the number of hot days (>30°C) from 25 to 60 per year in the hottest regions of Austria (see CCBox 1 Figure 2). Potential impacts include an increase in heat-related morbidity and mortality, a decrease in quality of life, out-migration, and a decrease in labor productivity, resulting in economic losses for various economic sectors and households. Heat exposure is greatest in densely populated urban areas because their natural cooling capacity is reduced by a lack of open and green spaces and often limited building quality. Work-related heat impacts are higher for those who perform physical work outdoors or work in non-air-conditioned indoor environments. Individuals in the low and high age groups, and especially those with lower socio-economic status or chronic illnesses, are particularly vulnerable to extremely high temperatures. Possible adaptation options to address heat exposure include increasing green and shaded areas or technical solutions for cooling. In the working environment, adaptation options include providing air conditioning at workplaces or, if this is not possible, shifting working hours to cooler times of the day, reducing working hours, increasing breaks and ensuring health recovery and relaxation during breaks (Cross-Chapter Box 2 and Sections 3.2 and 4.4.3).

The risk of drought will increase in Austria: Increasing evapotranspiration and precipitation variability will increase the risk of droughts seasonally and regionally. This will considerably increase the uncertainty of agricultural production (quantity and quality of certain profitable crops) and severely affect highly vulnerable ecosystems, leading to potentially irreversible losses (*medium evidence, high agreement*). As a result, agricultural irrigation needs will increase, especially in eastern Austria, which may further degrade fragile ecosystems, associated biodiversity, and other sectoral water needs. This may also lead to more frequent water shortages (*limited evidence, high agreement*). Households, businesses and services that depend on water availability in these regions will be more likely to experience water scarcity. Possible adaptation measures include adapted crop rotations (including drought-tolerant crops), sustainable soil management, a networked expansion of water supply, the use of rainwater for irrigation and water-saving production (see Sections 1.2.2, 1.6, 1.8 and 2.2.1).

**Risk of pluvial and river flooding will increase in Austria:** There is evidence that pluvial floods, and to some extent river floods, will become more frequent and more intense in Austria (*medium confidence*). This will affect communities by causing major economic losses, especially those that expand their built environment. In particular, areas with a high degree of surface sealing are vulnerable to pluvial flooding. In addition, floods affect transport infrastructure and services, and thus the mobility of the population and the accessibility of communities, and can interrupt pan-European transport routes. The elements at risk have different (physical) vulnerabilities to flood processes. Social and economic aspects of vulnerability can increase the overall susceptibility. A lack of (compulsory) building insurance is the bottle-neck of economic vulnerability. Strengthening and implementing spatial planning, unsealing urban and rural areas, efficient rainwater drainage in urban areas and upgrading drainage infrastructure are effective adaptation tools for both fluvial and pluvial floods. Further improvements in monitoring systems could improve warning and preparedness (see Sections 1.4, 1.8, 3.2.3 and 7.4.1).

The risk of torrential floods will increase in Austrian mountain valleys: In the Alpine region, torrential floods will become more frequent and more intense in the future (*robust evidence, medium agreement*), driven by an increase in intense precipitation events and in catchments with increased sediment availability. Sediment availability is expected to increase at high elevations where (i) glacier retreat, (ii) permafrost degradation and (iii) increased physical weathering occur. Torrential floods and related cascading events will affect alpine communities at local and regional levels, causing significant economic losses, especially as the built environment and land use expand. They will also affect transport infrastructure and services, thus affecting population mobility and the accessibility of mountain communities and economics, and may disrupt pan-European transport routes. The trend in torrential events is also strongly related to risk management, as financial resources for maintaining existing technical protection are limited. Therefore, current measures could be complemented by a portfolio of adaptation options to reduce the risk of torrential floods, such as enforcement of land use regulations, mandatory building insurance, pooled prevention funds to also encourage non-governmental investment in local (property) protection (see Sections 1.4, 1.8 and 7.4.1).

The risk of wildfire will increase in Austria: The effects of climate change are expected to lead to an increase in heat waves and changes in precipitation patterns and thus to an increase in fire weather days, which will ultimately increase the risk of wildfire in Austria (*high confidence*). This will affect local communities through economic losses, especially as the built environment and land use expand and the wildland-urban interface increases. It will also affect transportation infrastructure and services, thus affecting population mobility and accessibility of communities, and may disrupt pan-European transportation routes (especially railways). In addition, wildfires are a source of massive increases in health-threatening particulate matter, both locally and along wind paths. Wildfire-adapted assets and fire-resistant vegetation may be less vulnerable depending on the type of wildfire. Adaptation includes shifting to less fire-prone tree species (e.g., hardwoods) and organizational preparedness. Furthermore, information and awareness-raising of the population are important, as about 85 % of forest fires in Austria are directly or indirectly caused by human activities. With regard to the residual risk, the agricultural and forestry sector can be compensated by insurance (e.g., hail insurance/*Hagelversicherung*), and some damage to the built environment can be compensated by insurers (see Sections 1.7, 2.2.2 and 7.4.1).

**Risk of decreasing quantity and quality of protection forests:** Rising air temperatures and more frequent and severe droughts will lead to shifts in tree species distribution and forest composition, which may have negative consequences for the protective function of forests at elevations <1,000 m a.s.l. In parallel with the dynamics of natural forest disturbances, e.g., by wind, fire, pests and insects, their protective function will decrease at higher elevations, e.g., during the ongoing unprecedented bark beetle outbreak in Austria (*high confidence*). Monocultures are more susceptible to disturbances and thus more exposed to wind, fire, pests and insects than mixed stands. Tangible and intangible losses to infrastructure, buildings, and human lives can result from the increased hazard potential downslope of a degenerated protective forest. Furthermore, secondary hazards (such as snow avalanches and debris flows) may develop on affected areas, and soil loss and increased surface runoff may be observed. These secondary effects may eventually pose existential risks to exposed communities in the valleys below, requiring transformative risk management measures such as planned resettlement. Post-disturbance management decisions can have an important impact on protection. If left in the stand, deadwood could maintain the protective function of forests after windthrow and bark beetle disturbances, especially during the first 15 years. To better adapt to these threats, uneven and multi-layered stands with trees of all sizes and age classes, a minimum canopy cover of about 40 %, only small openings and a sufficient presence of natural regeneration should be considered (see Sections 1.6.2, 2.2.2 and 7.4.2).

**Risk of biodiversity loss:** There is evidence of upward migration of mountain species due to warmer conditions. Therefore, cold environments above the tree line will shrink and this will lead to biodiversity loss. Long-ranged, warm-demanding species will replace short-ranged, cold-adapted species (some of them endemic, restricted to the Austrian Alps), especially at elevations >2,600 m a.s.l. There is an increasing risk of mismatched interactions between organism groups (e.g., plants and pollinators) and physiological stress of high-elevation species due to warming. Furthermore, weakening of tree vitality and increased susceptibility to bark beetles can be expected, leading to a reduction

in the protective function of forests. The magnitude and likelihood of adverse consequences are high due to the pervasiveness of impacts across the Austrian Alps. There is a high potential for irreversible consequences due to the loss of intraspecific genetic diversity prior to eventual extinction. Although the likelihood is high, the effects will only become manifest in the coming decades because many alpine plants are long-lived species, and some may persist even when habitat conditions are no longer suitable. The lack of awareness of the many cascading effects of biodiversity loss leads to a lack of public attention. There is an increasing need for nature conservation in many alpine areas to avoid additional pressures from human land use (see Sections 1.6.4, 2.2.2 and 7.4.1).

The increasing risk to the infrastructure of the energy system limits Austria's energy security: Increasing electrification and thus grid-connected energy infrastructure threatened by climate-related extreme events pose an increasing risk to energy security (*high confidence*). Potential impacts include damage to energy generation (PV panels, wind turbines) and transmission infrastructure (overhead power lines, pipelines), resulting in energy supply disruptions to households, industry and infrastructure. Thermal generation units are will also be affected due to increased cooling demand and efficiency losses due to higher temperatures. Climate change impacts on energy demand, supply and infrastructure can lead to cascading effects and large-scale disruptions due to the high degree of interconnectedness between the energy system and all components of modern societies. Long investment cycles in energy infrastructure assets require early adaptation of planning, construction and operation to avoid lock-in effects. Possible adaptation measures to address this risk include designing energy infrastructure assets more climate-resilient, diversifying energy supply sources, increasing the degree of interconnection of European energy systems, decentralization, and climate-adapted network planning (see Section 4.5.1).

Increasing risk of power shortage due to compound weather events: Compound weather events, such as Europe-wide cold spells in winter combined with low river discharge, lead to high energy demand and low energy production in a highly electrified and renewables-based energy system, resulting in power shortages (*high confidence*). Fluctuations in demand and supply due to compound events may further threaten the stability of the electricity system. Coupled with an expected shift in electricity load towards summer months (increased cooling demand), power shortages will be exacerbated and economic losses and costs **may** occur due to the need for balancing energy and/or higher electricity prices. Possible adaptation measures include increasing the degree of interconnection of the European energy system, reducing energy and cooling water demand, increasing back-up capacities, spatial and technological diversification of energy sources and flexibility options (see Section 4.5.1).

Climate change leads to a loss of competitiveness in the Austrian tourism industry (snow and ice sports): Decreased snowfall and snow cover at lower altitudes due to rising temperatures, together with deteriorating conditions for technical snow production, will lead to a loss of competitiveness in snow- and ice-related tourism activities (*high confidence*). The natural snow season has decreased by an average of 7 days per decade (1961–1990 vs. 1991–2020) and will continue to decrease by 60–80 days (GWL 4.0°C). There is strong evidence that winter tourism based on snow and ice sports will be severely affected by shortening ski seasons and deteriorating snowmaking conditions. Households, businesses, and services in lower elevation ski resorts will be particularly affected. The high dependence of local economies on snow and ice sports and the lack of economic alternatives will increase the economic and social vulnerability of Austria's mountain regions. The associated loss of jobs in regions dependent on winter tourism will increase commuting and out-migration. Opportunities for adaptation and transition include the development of and investment in snow-independent tourism (see Sections 1.3, 4.3.3, 4.3.4 and 7.4.2).

Human responses to climate-related risks can amplify or reduce the expected impacts: Disaster and crisis management plans, contingency plans for energy and natural resource shortages, or extreme weather events can limit expected impacts. Behavior-based communication and compliance strategies for management plans are critical to their effectiveness (e.g., emergency notification standards, support from societal role models, etc.). These can be built on existing risk management structures (e.g., training courses on climate impact responses as part of work safety).



**CCBox 1 Figure 2** Heat and persons exposed in Austria: (a) Number of hot days ( $\geq$ 30°C) in Austria at present (1991–2020) and for the future according to different Global Warming Levels (GWLs) (Becsi and Formayer, 2024; Chimani et al., 2016; GeoSphere Austria, 2020), (b) number of people exposed to hot days at present and at different GWLs considering the population for year 2050 under Shared Socioeconomic Pathway 2 (Riahi et al., 2017; Wang et al., 2022; data for Austria provided by Marbler, 2024).

Austria is currently affected by climate-related damage and losses, and future climate change together with socio-economic development, will intensify these effects: Losses due to climate change (damage caused by extreme weather events as well as negative consequences of climate change) are currently estimated at an average of EUR 2 billion per year and are expected to increase to at least EUR 2.5 to 5.2 billion per year by 2030 and to EUR 4.3 to 10.8 billion per year by 2050. However, there is little quantitative evidence from Austrian studies. International studies provide some insights, but their relevance for subnational and sector-specific assessments in Austria is limited. Standardized documentation of losses is needed, and more research is needed to better quantify and provide robust figures on potential climate-related direct and indirect impacts on society, livelihoods, economic sectors and services.

### Resulting burning embers for Austria

At the current GWL 1.4°C [1.2–1.6°C] in the year 2024, corresponding to a mean temperature change in Austria of 3.1°C in 2024, moderate risk levels have already been identified and attributed to climate change for a number of key risks (CCBox 1 Figure 3): Mortality due to heat; agricultural crop yield losses due to drought; ecosystem disturbances (such as changes in species numbers and composition, ecosystem functions, or range shifts); and loss and damage due to riverine, torrential, and pluvial floods. For grassland yields, water scarcity and ski tourism, moderate levels of risk have been observed in some regions (drought risk and water scarcity in some regions and some years; ski tourism at lower altitudes); these risks are projected to intensify and spread to other regions and higher altitudes (*medium confidence*). Most of the key risks are characterized by high risk levels around GWL 3.0°C, due to the larger areas affected by the risk and the persistence and severity of the impacts. For some key risks, such as mortality, crop and forest yields, ecosystems, water supply, and ski tourism, very high risks cannot be excluded at current levels of adaptation; however, confidence in the GWL at which the transition from high to very high risk is projected to occur is low (*high agreement* that the transition cannot be excluded, but low number of publications).



**CCBox 1 Figure 3** Burning ember diagram of selected key risks for Austria, clustered by sector or system at risk. Diagrams show the change in the levels of impacts and risks assessed for global warming of 0–4.5°C global surface temperature change relative to the pre-industrial period (1850–1900). The global warming is also converted to mean Austrian surface temperature change. For a number of key risks, confidence levels are given as ranges to indicate that the risk transition will be at a lower global warming level in some regions or altitudes and at a higher global warming level in others. For some key risks, burning ember diagrams could not be generated due to insufficient quantitative data across global warming levels, too few studies for Austria, or insufficient process understanding. Confidence levels: High [•••], medium [••], low [•], very low [(•)]. In addition to the authors of the Cross-Chapter Box, the following authors contributed to the development of the Burning Embers: Maria Balas, Johanna Schauer-Berg, Katharina Brugger, Hermine Mitter, Manfred Lexer, Manfred Kleidorfer, Jana Petermann, Stefan Dullinger, Roman Neunteufel, Anna Burton, Judith Köberl, Nina Knittel, Anna Viktoria Rohrer, Valentin Schalk (data extraction), Johanna Wittholm (data extraction), Jonas Peisker (data extraction).

#### 1.3.2. Glaciers

Glaciers exist where snowfall can survive the summer and accumulate over successive years. Glaciers act as seasonal and multiannual water reservoirs, gaining mass and growing when climatic conditions are more favorable for snowfall and its survival. Glaciers and their changes are relevant to mountain hydrology, tourism, natural hazards, and mountain ecosystems.

# **Observed trends**

Glaciers of the European Alps have been retreating since the Little Ice Age (ca. 1850), and the rate of retreat has increased in recent decades (Sommer et al., 2020). Geomorphological mapping indicates that the maximum extent of Austrian glaciers during the Little Ice Age (ca. 1850) was 941 km<sup>2</sup> (Fischer et al., 2015). Subsequent Austrian national glacier inventories were mapped from satellite imagery centered around 1969, 1998, 2006, and 2015, respectively (Buckel et al., 2018). Some inconsistencies may arise from the specific delineation methods used, but the pattern is clear, with ice extent decreasing from 555 to 479 to 415 to 329 km<sup>2</sup> over the four successive inventories, representing a loss of 40 % of the 1969 glacier area by 2015, with the highest percentage losses in the most recent time interval (*high confidence*). Although changes in glacier mass or volume are better measures of climate-induced glacier change, changes in length of over 100 Austrian glaciers show that all measured glaciers have receded since 1970 (Lieb and Kellerer-Pirklbauer, 2023). Austrian glaciers were estimated to store 12.4 km<sup>3</sup> of freshwater in 2016 (Helfricht et al., 2019). Eight Austrian glaciers have mass balance records longer than 30 years (World Glacier Monitoring Service (WGMS), 2022). These glaciers lost mass at an average rate of 1092 kg/m<sup>2</sup> per year during 2011–2020, which is almost double the annual rate of loss during 1960-2000. 2022 was a particularly negative mass balance year, driven by low snow amounts and positive summer temperature anomalies. For example, the Hintereisferner in the Ötztal lost 5 % of its ice volume in one year (Voordendag et al., 2023), compared to a typical volume loss of 30 % for Austrian glaciers during 1969-2006 (Helfricht et al., 2019). Glacier recession can lead to the formation of new alpine lakes, which have increased in number over the last century and at an increasing rate in the last 30 years (Buckel et al., 2018). Glacier retreat in Austria is associated with glacier fragmentation, and more of the remaining ice is buried by rock debris and survives only in shadowed cirques (Fischer et al., 2021). In other parts of the Alps, observations of glacier temperatures indicate warming ice (e.g., Gilbert et al., 2014) and more meltwater production at high elevations, which are associated with triggering (Bondesan and Francese, 2023) and increasing (Chiarle et al., 2022) collapses of remnant hanging glaciers, but significant ice avalanches or glacier collapses are very rare in Austria (Kellerer-Pirklbauer et al., 2012b).

## Future changes

Part of future glacier mass loss is a delayed response to climate change that has already occurred. State-of-the-art modeling indicates that 35 % of European glacier volume will be lost by 2050 under current climate conditions (Huss and Hock, 2018), and in all future climate scenarios half of the glacier mass will be lost within the next two decades (Rounce et al., 2023). Thereafter, the GWL (see Chapter Box 1.1) determines how much glacier ice will remain by the end of the century: Glacier models show that for GWLs of 1.5, 2, 3 and 4°C in 2100 compared to pre-industrial levels, the proportion of glacier mass remaining at the end of the century in central Europe is about 20 %, 15 %, 5 % and none, respectively (Rounce et al., 2023) (Figure 1.10). As Austrian glaciers are typically smaller and have lower maximum elevations than glaciers in the western Alps (Sommer et al., 2020), glacier loss in Austria is more pronounced than the European Alpine average (high confidence), and at GWL 1.5°C a maximum of 6 % of Austrian glacier mass remains (Figure 1.10). These are likely to be conservative estimates, as the models used to make these projections do not include positive feedbacks on glacier retreat rates due to ongoing glacier fragmentation, and regional-scale assessments in Austria using higher-resolution local datasets indicate greater glacier retreat (Hartl et al., 2024). The formation of new alpine lakes and remaining ice is more likely to be debris-covered, so glacier meltwater runoff is expected to decrease in the coming decades. Glacier ice is expected to warm, and associated changes in basal conditions could affect the stability of the remaining ice (Gilbert et al., 2014).

# Related consequences

Continued deglaciation will (i) shift seasonal peak runoff to earlier in the year (*high confidence*), coinciding with peak snowmelt (Hanus et al., 2021) (see Sections 1.3.1 and 7.4.1); (ii) affect the infrastructure and operation of ski resorts (Mayer and Abegg, 2024) (Sections 2.2.1, 4.7.2 and



**Figure 1.10** Projections of total glacier mass change over time for all glaciers in Austria corresponding to four global warming levels at the end of the century. Note that due to the time lag in the adjustment of glacier mass to the forcing, the end-of-century mass does not include the full adjustment to the end-of-century GWL (see Chapter Box 1.1 for further explanation of GWLs). Figure created by Lilian Schuster (github. com/lilianschuster/glacier\_climate\_plots/tree/main/glacier\_projections\_for\_austria\_rounce\_et\_al\_2023) using openly available data from Rounce et al. (2023).

7.4.2); (iii) facilitate natural hazards such as landslides, glacier collapse, and glacier lake outburst floods (e.g., Kellerer-Pirklbauer et al., 2012a; Piroton et al., 2024) (see Sections 1.8, 7.4.1 and Table 1.A.4) and (iv) alter conditions for mountain ecosystems through changes in water temperature, nutrient and sediment supply (e.g., Brighenti et al., 2019) (Section 7.4.1). Together, these contribute to changes in societal services provided by mountain regions (see Section 7.4.2).

# 1.3.3. Permafrost

Rock, sediment, or soil that freezes and thaws seasonally is called the active layer, while permafrost is rock, sediment, or soil that does not undergo seasonal melting, and remains at freezing temperatures throughout the year. In Austria permafrost occurs only in discontinuous patches at high elevations under favorable conditions.

### **Observed trends**

Estimates of permafrost coverage are based on statistical models of potential permafrost distribution that have been optimized for a small, well-studied region (the Hohe Tauern range) and applied to a larger area. Different models yield estimates of the maximum area potentially suitable for permafrost in Austria of between 1,400 and 3,000 km<sup>2</sup>, with large uncertainties (Otto et al., 2020). For mountainous regions such as the Hohe Tauern National Park, 25 % of the area is potentially suitable for discontinuous permafrost (Schrott et al., 2012). Isolated patches of permafrost have been identi-

fied well below the regional lower limit of permafrost, suggesting that patches of permafrost persist outside the climate zones identified by statistical models (Stiegler et al., 2014). Rock glaciers have only recently been consistently inventoried at the national scale (Wagner et al., 2020), revealing 303 km<sup>2</sup> of rock glacier area, but 60 % by number are relict features that no longer contain ice. The volume of ice stored in rock glaciers within Austria was just over 1 km<sup>3</sup> in 2019, which is 1/12 of the volume of Austrian glaciers (Wagner et al., 2021) (see Section 1.3.2). Rock glacier volume and velocity have generally decreased at Austrian sites since 2000 (Fleischer et al., 2021; Kellerer-Pirklbauer et al., 2024). Peak velocities and periods of deceleration are synchronous with other Alpine regions (Kellerer-Pirklbauer et al., 2024). Rock glaciers can undergo destabilization processes that result in extreme acceleration and a shift in the dynamic regime from permafrost creep to landslide-like movement (e.g., Marcer et al., 2019). For example, the lower sector of the Äußeres Hochebenkar in the Ötztal has destabilized and shows very rapid movement of 20-30 m/yr in the fastest section (Hartl et al., 2023). The destabilization of this rock glacier leads to an increased rock fall hazard affecting a nearby access road. Borehole temperature data in Austria are available from Sonnblick (since 2008) and Kitzsteinhorn (since 2010) (Global Terrestrial Network for Permafrost, 2023) and show that the seasonal melt layer has tended to penetrate deeper in recent years, reaching 4 m at Kitzsteinhorn in 2022 (Hansche et al., 2023). Data from the extensive 'Swiss Permafrost Monitoring Network' (Swiss Permafrost Monitoring Network, 2022), which go back further in time, show increasing ground temperatures, active layer depths and rock glacier velocities associated with warming temperatures (Haberkorn et al., 2021), confirming the trends observed in the sparser observations in Austria.

### Future changes

Robust distributed projections of future permafrost distribution are not currently available, in part due to insufficient present-day permafrost mapping and borehole data, but also due to the strong heterogeneity of mountain permafrost (e.g., Stiegler et al., 2014). Future permafrost evolution has been simulated for Swiss sites, indicating continued warming and deepening of the active layer (Marmy et al., 2016), although the slow rates of change imply that permafrost patches will still exist at the end of the century (Beniston et al., 2018).

### Related consequences

Continued permafrost degradation is variously associated with surface drying, increased landslide formation, gravitational slope failure, and release of heavy metals in runoff (Beniston et al., 2018), all of which require further research to understand the processes and uncertainties involved, connected with changing natural hazards (Fuchs et al., 2022) (see Sections 1.8 and 7.4.1) and impacts on local infrastructure (Pläsken et al., 2020) (Section 7.4.2).

# 1.4. Hydrosphere

This subchapter summarizes the understanding and assessment of the impacts of climate change on runoff generation processes and on the magnitude, seasonality and frequency of water fluxes in streams, groundwater and lakes (including extremes and water quality).

# 1.4.1. Rivers and streams

This section discusses the observed and projected changes in river discharge in Austria and in the European context. The current state of knowledge is based on the first Austrian Assessment Report published in 2014 (Volume 2, Chapter 2; Nachtnebel et al., 2014), updated by the GeoSphere/TU Vienna follow-up study on climate change in water management (Blöschl et al., 2017) and recent studies from the scientific literature. The results are summarized for the main characteristics: Mean flow, high flow and low flow extremes, with impacts on stream temperature and aquatic biodiversity.

# Mean flow

The mean flow describes the water resources available in a catchment on average over the course of a year. It is closely related to the climatic water balance between precipitation and evaporation, with negligible changes in storage (Blöschl et al., 2017). On a pan-European scale, climates without a frost season show decreasing streamflow trends throughout the year, while climates with a frost season show decreasing trends in summer and increasing trends in winter (Peña-Angulo et al., 2022). The trends are regionally consistent but mostly insignificant, with significant winter trends in the Alps, northern and western Europe, and significant summer trends in southwestern Europe (see Figure 1.11).

#### Observed trends

For the Austrian domain, trends in water balance components for the period 1978–2013 were recently re-evaluated by Duethmann et al. (2020). Average precipitation has increased by a total of about 120 mm over 35 years. The increased precipitation compensates the evaporation trends of about +110 mm over 35 years. Thus, the average annual runoff has not changed. Although there are decreases and increases in CWB at low and high elevations (Section 1.2.1), these changes balance out at all elevations. This confirms



**Figure 1.11** Direction and significance of trends in the monthly Standardized Streamflow Index over the period 1962–2017 from a European dataset of climate-sensitive and anthropogenically influenced gauges. Climates without a frost season show decreasing streamflow trends throughout the year, climates with a frost season show decreasing trends in summer and increasing trends in winter (Peña-Angulo et al., 2022).

the results of previous assessments (Blöschl et al., 2017) and is also supported by the most recent data (Haslinger et al., 2023; Laaha et al., 2025). The observed increase in evaporation can be attributed to increases in temperature and solar radiation, as well as to changes in vegetation dynamics, such as the lengthening of the growing season (*high confidence*).

With respect to regional trends, the first Austrian Assessment Report (APCC, 2014) is slightly tempered due to some changes in trend patterns. Current observations indicate decreasing trends especially in the northwest (Bregenzerwald, Salzburg), and increasing trends in the main Alps (*medium confidence*). Apart from the northwest, trends are mostly insignificant, but there is *high confidence* in a pronounced seasonal behavior with mostly significant increasing winter trends in the Alps due to frost attenuation (APCC, 2014; Laaha et al., 2025).

#### Future changes

For the future, the water balance assessment of APCC (2014) remains essentially valid: According to Blöschl et al. (2017) and Haslinger et al. (2023), there is a trend towards slightly decreasing annual runoff in the south and southeast (low confidence), decreasing winter runoff in the south (low confidence), and increasing winter runoff in the Alps (medium confidence). However, the projected trends are of low confidence because hydrological model projections have been shown to overestimate discharge trends especially in central Austria, southern Carinthia and western Austria. This is mainly due to inhomogeneities such as a variable number of stations in the precipitation data used for calibration and a model structure that neglects evaporation-relevant changes in vegetation dynamics (Duethmann et al., 2020). Zhang et al. (2023) expect global streamflow to be lower than predicted by Earth System Models in the near future (2021–2050), corresponding to a GWL 1.4°C/2.0°C (RCP4.5/8.5), due to smaller contributions from precipitation and greater sensitivity of streamflow to changes in evapotranspiration. However, this effect tends to be less pronounced for the European continent.

# High flows/floods

Floods (fluvial flooding) are generated by combinations of rainfall and runoff generation processes (Blöschl et al., 2013). The magnitude of floods is initiated by precipitation, but determined by the interplay of its magnitude and seasonality with snowmelt and catchment pre-wetness (Blöschl et al., 2011). Among the dominant meteorological drivers, weather patterns relevant for large-scale heavy precipitation have not increased in frequency in recent decades, but have been observed to bring more intense precipitation in fall and winter (Hofstätter et al., 2018) (Section 1.2.1).

#### Observed trends

At the European scale, Blöschl et al. (2019) observed flood discharge trends (variable observation periods) range from an increase of about 11 % per decade to a decrease of 23 %. Three distinct regions with roughly similar trend behavior can be observed: Increasing floods in northwestern Europe due to increasing fall and winter precipitation; decreasing floods in medium and large catchments in southern Europe due to decreasing precipitation and increasing evaporation; and decreasing floods in eastern Europe due to decreasing and earlier snowmelt resulting from warmer spring temperatures (see Figure 1.A.12). Austria lies at the intersection of these regions with the Alpine topography, resulting in a complex interplay of flood-generating processes.

For the Austrian domain, the increasing trends in observed flood discharges for the period 1976-2007 (APCC, 2014) have slightly intensified in recent years (Laaha et al., 2025). This has also exacerbated the flood problem (robust evidence, medium agreement). Over 43 years (1978-2020), floods have increased significantly in about 22 % (vs. 16 % for 1976-2007) of the catchments. Overall, there is a clear seasonal shift from winter and spring floods to summer floods, so that summer is now also the dominant flood season in the eastern half of Austria. Overall, 24 % of the catchments <500 km<sup>2</sup> show an increasing trend. The trend is more pronounced in smaller catchments than in larger ones, and more pronounced in the (mostly non-dominant) fall and winter seasons. Especially in southern Lower Austria and Upper Styria, as well as in areas along the main Alpine ridge, there is a significant increase in annual flood events, often exceeding 20 % in 43 years (+5% per decade). Overall, the average trend is +11.8 % in 43 years (+2.7 % per decade), with a stronger average trend of +22.5 % in 43 years (+5.2 % per decade) observed in the winter season (previous studies: annual trend of 6.8 and 6.6 % increase per decade).

### Future changes

Projections of changes in floods are challenging with the current state of knowledge, because future trends in climate extremes, especially small-scale heavy precipitation, are not simulated with sufficient reliability in the currently available bias-corrected regional climate simulations for Austria. The most recent nationwide assessment of climate impacts on floods in Austria (Blöschl et al., 2017) therefore projects future changes based on a synopsis of the ÖKS15 and expert scenarios that focus on mechanisms and cause-effect relationships. It only allows statements in terms of time periods rather than GWLs (here the period 2021–2050 is used to make the statements comparable), as the attribution of GWLs is not directly possible for the combination of approaches.

Among the climate drivers, large-scale daily precipitation shows a strong dependence on mean air temperature (see Section 1.2.1). In the future, therefore, an increase in precipitation amounts of large-scale events is mostly expected (see Section 1.2.2). However, heavy precipitation-relevant weather trajectories will tend to become less frequent in the coming decades, but could produce heavier amounts of precipitation (see Section 1.2.2). To better quantify the impacts of heavy precipitation changes on floods, the representation of convection-relevant processes in climate models needs to be improved (Blöschl et al., 2017).

When the mechanisms for possible changes in flood generation are analyzed in the form of scenarios (see Figure 1.A.13), the following picture emerges (projected change in  $HQ_{100}$  floods in 2021–2050 compared to 1997–2007):

- Changes in seasonal precipitation hardly lead to any projected changes in design flood estimates such as the 100-year flood (HQ<sub>100</sub>): -2 to +2 % (vs. -3 to +2 % in APCC, 2014) (*medium evidence, low/medium agreement*).
- An increase in the convective fraction of precipitation results in increased flooding everywhere, especially in small catchments: +2 to +8 % (vs. +2 to +10 %) (*robust evidence, medium agreement*).
- An increase in the snow line has almost no effect on floods: 0 to +4 % (APCC, 2014: ditto) (*high confidence*).
- The influence of earlier snowmelt and thus lower runoff coefficients in summer, combined with higher evaporation, slightly reduces floods, especially in the Alps: -5 to +2 % (APCC, 2014: ditto) (*medium confidence*).

The combination of all mechanisms is expected to result in regionally varying changes in extreme flood discharges (HQ<sub>100</sub>), ranging from -5 to +8 % (*medium evidence, low/medium agreement*). The projections show hardly any decreases (previous study: small decreases) of floods along the northern rim of the Alps, because the projected shifts in seasonal precipitation are small and the rise of the snow line should have little effect. In the rest of Austria, the scenarios show increases in floods that are generally small, but slightly larger increases are expected in the Innviertel and Mühlviertel regions. A slight shift of flooding from summer to winter in these regions may be due to an increase in air temperature as a result of climate change, leading to less snow and more winter rain.

### Other types of floods

National flood assessments in Austria have so far focused on river floods, while other types of floods have rarely been analyzed. First studies for the contiguous United States suggest that they will become more prone to flash floods in a highend emissions scenario (Li et al., 2022), while Bates et al. (2021) found that the combined flood risk of different flood types (pluvial, fluvial and coastal flooding) for the USA will increase even under a stabilizing emission scenario (RCP4.5). In particular, changes in pluvial floods associated with projected increases in precipitation intensity need to be better understood. Some attempts have been evaluated for Upper Austria, where Nocker and Laaha (2019) found that during the observation period 2007-2013, the most severe damage to agriculture from pluvial floods was caused by intense rainfall events. Prolonged low-intensity precipitation events played a minor role. Although pluvial floods are triggered by high rainfall intensity, their magnitude is also strongly dependent on the antecedent soil wetness, as this influences the proportion of rainfall that exceeds the infiltration capacity of the soil and produces surface runoff. The impact of varying precipitation inputs and antecedent conditions were also analyzed by Humer et al. (2017). They found that rainfall above a certain amount can trigger a flash flood regardless of preconditions, although the influence of preconditions on the discharge of a flash flood is very significant.

Changes in temporal precipitation patterns are considered to be of key importance for assessing future changes in pluvial flood risk in Austria. However, they are simulated with low confidence in climate projections and still represent a knowledge gap in climate studies. Overall, however, heavy precipitation intensities are expected to increase due to global warming, as shown by recent events (see Section 1.2.2), which is expected to intensify pluvial floods.

#### Low flows

Low flow generation processes in Austria differ mainly by elevation: Summer low flows are triggered by precipitation deficits in the lowlands, winter low flows are caused by freezing and temporary snow storage in the Alps (Laaha and Blöschl, 2006). As these processes are fundamentally different, seasonality must be taken into account when interpreting the impacts of climate change on low flow events (Laaha et al., 2016; Tallaksen and Van Lanen, 2023). Summer and mixed regimes dominate below a mean catchment elevation of 900 m a.s.l., while winter regimes dominate in the higher catchments. Summer low flows generally lead to higher impacts on water-related sectors.

### Observed trends

For the Austrian domain, low flow trends were reassessed for the period 1976–2014 by Laaha et al. (2016) and Blöschl et al. (2017) and further updated for the period 1977–2019 (43 years) in Laaha et al. (2025). Compared to the previous assessment (APCC, 2014), the low flow trends in Austria have become more pronounced in recent years. This applies to both the magnitude and the number of stations affected. The effects are more pronounced in the alpine catchments than in the lowlands.

Drying trends can be observed below 900 m a.s.l., where low flow (measured by the 5th percentile of the daily runoff, Q95) has decreased in about 16 % (previously 10 %) of the catchments in Austria during the last 43 years (high confidence). This is expected to increase the incidence of zero flow and increased water stress in intermittent streams, with potential impacts on biota and biogeochemistry (Straka et al., 2021; Tramblay et al., 2021). In contrast, low flows have increased in 6 % (previously 5 %) of the catchments. Significant decreasing trends are observed especially in southeastern Austria and in the Innviertel region in the north. Wetting trends dominate in alpine catchments above 900 m a.s.l., where low flows have increased in about 41 % (previously 14 %) of the catchments in Austria over the last 43 years. In contrast, only 3 % of the catchments (unchanged) experienced a decrease in low flows. An increase is particularly evident in the Central Alps due to frost attenuation (high confidence). This applies to areas without significant influence from water management reservoirs.

In terms of the overall magnitude of change, the mean low flow trend of all stations in Austria has intensified: Below 900 m a.s.l. -17 % over 43 years (-3.9 % per decade). Above 900 m a.s.l. +23 % over 43 years (+5.3 % per decade). Similar but more pronounced patterns than for annual low flows are observed for seasonal low flows, with summer low flows showing decreasing trends at lower elevations and winter low flows showing increasing trends in the Alps. Overall, summer and winter low flows show consistent patterns of increase/decrease across Austria, with the exception of Lower Carinthia, where winter wetting trends exceed summer drying trends.

# Future changes

For future low flow scenarios in Austria (Laaha et al., 2016), the regional trend patterns concluded in ARR14 remain essentially valid (see Figure 1.12). However, the observed trends over 43 years already exceed the projected trends for the 45-year horizon, so the magnitude needs to be revised. In the Austrian Alps (high areas, winter low flow regime), the scenario calculations show a significant increase in low flows for the period 2021-2050 compared to 1976-2006, with an increase of about 20-30 % (previously 10-25 %; high confidence). In the Austrian lowlands as well as in the Alpine foreland, the scenario calculations show mostly decreasing trends (medium evidence, low/medium agreement). In some catchments a slight increase is expected (e.g., Mühlviertel). In other areas a decrease of low flows of about 15-20 % (previously 10-15 %) is expected (e.g., Weinviertel, northern Burgenland, parts of southern Styria, parts of the Lower Austrian Alpine foreland). In exceptional cases, the decrease will be somewhat higher. In the southeast, the scenarios show slightly larger decreases than in the previous study. The decrease in low flows in the east affects all seasons, while the increase in the west affects winter and spring.

#### Low flows and stream temperature

During heat waves, low flows are typically associated with high stream temperatures, leading to water quality degradation and potential threats to ecosystems and public health (Cross-Chapter Box 2). When flow and temperature regimes are sufficiently altered, biological indicators normally found under reference conditions may be endangered (Solheim et al., 2010), and biodiversity and ecological status may be affected (Pletterbauer et al., 2018). In warmer, drier regions, major biodiversity losses are expected due to increased droughts and hot weather. This is particularly true for small rivers (e.g., in southeastern Austria), where the type of water body may change from permanent to temporary (i.e. intermittent), drying-out in summer (Straka et al., 2021; Tramblay et al., 2021). Higher temperatures combined with increased nutrient inputs from more extreme precipitation events can lead to lower oxygen levels, which can cause

stress and habitat degradation, especially for coldwater species such as salmonids. River warming is driven by the energy input of air temperature and solar radiation, resulting in a characteristic diurnal variation of the temperature regime. Warming is greatest at low flow and low water levels and can be enhanced by surface water inflow. However, shading by trees and inflow from groundwater and other sources can have a cooling effect (Kalny et al., 2017).

Streamflow temperature has been assessed at the national scale for Switzerland (Michel et al., 2020). The study showed a general increase in annual water temperature of 0.33/0.37°C per decade over a long-term (40 years)/recent short-term (20 years) period. An increase was reported for all seasons, with trends being most pronounced in summer and less pronounced in winter. On an annual scale, air temperature was found to be the main driver of water temperature. Trends are more pronounced at lower elevations than in alpine streams, where snow and glacier melt temporarily compensate for warming trends in air temperature. For Austria, the assessment is still fragmented, but the results are generally consistent with those for Switzerland. For alpine rivers in western Austria a substantial warming effect of mountain rivers with significant month-specific warming



Figure 1.12 Generalized changes in annual maximum floods (left) and Q95 low flows (right) from scenario analyses (top); long-term trends in annual values for the 1955–2014 series (middle), and more recent trends for 1978–2020 (floods) and 1977–2019 (low flows) (bottom). Blue colors indicate regions with increasing discharges, red colors indicate regions with decreasing discharges, and gray colors indicate regions with no clear trend. Future scenarios are derived from a synopsis of ÖKS15 RCP4.5 and RCP8.5 and expert scenarios focusing on mechanisms and cause-effect relationships. Based on Figures 3.21 and 5.21 of Blöschl et al. (2017) and Laaha et al. (2025).

rates was shown (Niedrist, 2023), with river size dependent rates of +0.24 and +0.44°C per decade at the annual scale. Here, the warming trends were found for all seasons and for the whole temperature range including minimum and maximum temperatures.

# 1.4.2. Lakes

Lakes are particularly vulnerable to climate change, they have been warming worldwide since the 1980s (Adrian et al., 2009; O'Reilly et al., 2015) with concurrent impacts on greenhouse gas emissions (Walter et al., 2006), nutrient cycling (Trochine et al., 2011; Michelutti et al., 2015), and changing biodiversity (Heino et al., 2021). In Austria, there are about 25,000 lentic aquatic ecosystems, including lakes, ponds, gravel pit lakes, and reservoirs. Together they cover 613 km<sup>2</sup> or 0.7 % of the Austrian territory<sup>4</sup>. Although these are relatively small ecosystems, they contain high levels of biodiversity and provide many ecosystem services.

The largest and one of the shallowest lakes in Austria, Lake Neusiedl ( $320 \text{ km}^2$ ), is currently at risk of drying up (see Chapter Box 1.3), while the deepest lakes, Traunsee (191 m) and Attersee (171 m), also contain the largest volumes of water (2,189 million m<sup>3</sup> and 3,890 million m<sup>3</sup>, respectively). Lakes in Austria >0.5 km<sup>2</sup> range from alpine lakes, the highest being Vilsalpsee (1,165 m a.s.l.), to lowland lakes, the lowest being Lake Neusiedl (115 m a.s.l.).

## Lake temperatures

The surface temperature of monitored lakes with long-term data series in Austria has increased on average between 2.0°C (since the 1980s) and 1.5°C (since the beginning of this century) in parallel with increasing air temperature (*high confidence*). According to Kainz et al. (2017), the surface water temperature of the subalpine Lunzer See (608 m, 68 ha, 34 m depth) increased by 0.8°C in the period 1920–2015, with the strongest increase in spring and summer months (~1–2°C) and less in fall (~0.3°C). Since 1980, however, lake surface temperatures have increased more rapidly by ~1.5°C, with the lowest mean surface temperature recorded in 1980 (10.6°C) and the highest in 2015 (14.3°C, Figure 1.13a, Figure 1.13b; *high confidence*). Similarly, the lake surface temperature in the southernmost province, Carinthia, increased

<sup>&</sup>lt;sup>4</sup> info.bml.gv.at/themen/wasser/wasser-oesterreich/zahlen/ fluesse\_seen\_zahlen.html



**Figure 1.13** Annual mean lake surface water temperature (LSWT; °C) during the ice-free periods of (a) Lunzer See (1920–2015), (b) Lunzer See (1980–2015, Kainz et al., 2017), (c) Mondsee (Luger et al., 2021), (d) Piburgersee (Sommaruga et al., 2023).

between 1.2°C (Faaker See) and 2.5°C (Klopeiner See)<sup>5</sup> over a period of 16 years (2007–2022), and the current report on the environment in Styria<sup>6</sup> concludes that Grundlsee (732 m a.s.l.) also significantly increased its lake surface water temperature by an average of 0.5°C to 3.5°C (in 2018) over the period 2007–2021, compared to the mean lake surface temperatures of an earlier reference period (2000–2006). Mondsee (Upper Austria) also increased its surface water temperature consistently by ~2°C from 1975–2020, while the mean annual temperatures at the bottom (60–65 m depth) remained stable at ~4.5°C (Luger et al., 2021) (Figure 1.13c). Piburgersee (Tyrol, 913 m a.s.l.) also increased its lake surface water temperature by ~2°C, from 9 to 11°C, between 1966 and 2022 (Sommaruga et al., 2023) (*high confidence*) (Figure 1.13d).

Changes in water temperatures below the lake surface are lake-specific and cannot be generalized to all lakes. For example, there is *robust evidence* that the subsurface temperature (0–5 m below the water-air interface) of Grundlsee has increased from 0°C (2011) to 2°C (2013) relative to the mean water temperature profile of the reference period (2000–2006). The higher surface temperature of the lake in winter is a tipping point. If the surface temperature does not fall below 4°C (highest water density), convective mixing does not occur and the water column remains stratified, resulting in reduced oxygen supply to the hypolimnion.

# Lake ice

There is no consistent data record on the duration of lake ice on Austrian lakes. Since 1905, full lake ice cover at Lunzer See has decreased dramatically from 93 days (1905–1915) to 35 days in the last decade (2010–2020). In 2007 and 2014, Lunzer See had no ice cover at all (Kainz et al., 2017) (Figure 1.14a). While Piburger See (Tyrol) was fully ice-covered for 119 days in 1966, the duration of full ice cover decreased steadily to 84 days in 2023 (Sommaruga et al., 2023) (Figure 1.14b). In general, lake ice-on occurs later in the season and lake ice-off occurs earlier (*medium confidence*). The observed decrease in annual lake ice duration is comparable to other lakes in the Northern Hemisphere, where ice duration has become 28 days shorter on average over the past 150 years (Woolway et al., 2022).

# Nutrients and oxygen dynamics

Temporal nutrient dynamics, in contrast to consistently increasing lake water surface temperatures, are lake and probably catchment-specific (*high confidence*). Mondsee strongly decreased its total phosphorus (TP) concentrations above the sediment-water interface from ~350 µg L<sup>-1</sup> (1975) to ~20 µg L<sup>-1</sup> (1990s), after which [TP] increased again to ~100 µg L<sup>-1</sup> by 2023 (Luger et al., 2021). A similar pattern was observed for ammonium concentrations. This temporal nutrient pattern coincided with changes in dissolved oxygen (dissO<sub>2</sub>) concentrations. At Mondsee, dissO<sub>2</sub> concentrations increased from 1975 until the 1990s and then decreased until 2023. Low dissO<sub>2</sub> concentrations in the hypolimnion remobilize phosphorus from the sediment-water interface back into the lake water. The concurrent low dissO<sub>2</sub> and increasing

<sup>&</sup>lt;sup>5</sup> data.gv.at/katalog/dataset/8f554c0a-fd7b-4d66-bfdb-dc223731 8a52#resources

<sup>6</sup> app.luis.steiermark.at/berichte/Download/Umweltschutzberichte/Umweltbericht\_20-21\_Gesamt.pdf



Figure 1.14 Changes in full ice cover of (a) Lunzer See in days from 1921 to 2015 (Kainz et al., 2017), and (b) Piburgersee ('Eislegung': ice-on and 'Eisbruch': ice-off) from 1966 to 2023 (Sommaruga et al., 2023).

phosphorus concentrations result in restricted habitats for fish and other oxygen-demanding animals, but may result in higher algal biomass enabled by the re-mobilized phosphorus in the photic zone of the lakes, and may eventually lead to re-eutrophication of certain lakes. Similar patterns, but much lower [TP], have been reported for Piburgersee, where [TP] decreased from ~35  $\mu$ g L<sup>-1</sup> (1975) to ~20  $\mu$ g L<sup>-1</sup> by 2000, and then increased again to ~30  $\mu$ g L<sup>-1</sup> by 2023 (Sommaruga et al., 2023). The decreasing dissO<sub>2</sub> concentrations of Austrian lakes are very consistent with observed decreases in dissO<sub>2</sub> levels of lakes worldwide, which are reported to be 3–9 times greater than in the world's oceans (Jane et al., 2021). There is *robust evidence* that declining dissO<sub>2</sub> levels have the potential to dramatically shift species habitats and threaten lake ecosystem services (Schindler, 2017).

Overall, the most important changes relate to temperature, stratification, dissO<sub>2</sub> and TP content: all four parameters are very closely related or dependent on each other. Therefore, particularly careful management of avoidable nutrient loads remains important. Steadily increasing [TP], as currently measured in these lakes at the reduced dissO<sub>2</sub> hypolimnion, is a clear indication of internal fertilization.

#### 1.4.3. Groundwater and soil water

The vadose zone is the variably saturated zone between the ground surface and the permanent water table of groundwater (Stumpp and Kammerer, 2022). The vadose zone and groundwater are important parts of the water cycle (Aquilina et al., 2023), storing more than 10,000 times the amount of water held by rivers worldwide (Oki and Kanae, 2006). In Austria, groundwater, including spring water, is the most important water resource, accounting for 100 % of the drinking water supply (Vogel, 2001; BMLRT, 2021). For the sustainable management of water resources, and thus for the protection of groundwater resources in terms of quantity and quality, it is important to understand the impacts of climate change on subsurface water fluxes and storage.

### **Observed trends**

In the last Austrian Assessment Report (APCC, 2014), Nachtnebel et al. (2014) summarized that, based on different emission scenarios and regional climate models (AR-PEGE-ALADIN, ECHAM5-RegCM3, REMO-UBA), only small variations in groundwater recharge rates can be expected for the future; exceptions are individual, drier regions with decreasing precipitation rates and unfavorable hydrogeological situations. However, in the agricultural areas of eastern and southeastern Austria, more groundwater was expected to be needed for irrigation due to increasing crop water demand.

These past results are mainly confirmed by recent studies that provide additional information on quantities and uncertainties. An analysis of past groundwater recharge in Austria found a west-east gradient, with generally higher groundwater recharge in the western part of Austria compared to the eastern or southeastern part (BMLRT, 2021; Schübl et al., 2023) (high confidence). The ratio of groundwater recharge to precipitation (GWR/P) is strongly correlated with the amount of precipitation, and higher recharge rates and lower actual evapotranspiration are correlated with a higher percentage of sand in soils (Schübl et al., 2023). Groundwater recharge rates are seasonally more variable (71-265 %; quantified as the coefficient of variation of the standard deviation between monthly sums and annual means) compared to precipitation (52-76 %) and actual evapotranspiration (64-76 %) for selected grassland sites in Austria (Schübl et al., 2023). More detailed studies of groundwater recharge for forested or agricultural sites across Austria are lacking and would require a systematic analysis of monitoring data for representative sites.

Groundwater use and climate change can affect groundwater levels and thus subsurface water availability. Groundwater level trends for the period 1976–2000 show that 12 % and 18 % of the wells investigated showed a significantly increasing and decreasing groundwater level trend, respectively. 70 % showed no significant trend (Blaschke et al., 2011). Integrated results for Austria show overall decreasing trends in groundwater levels, with a pronounced decrease until 1985 and increasing trends until 2016; regional differences were found for the analysis from 1980–2018, with increasing trends in Lower Austria, Vienna, Vorarlberg, Burgenland and Upper Austria, mostly decreasing trends in Carinthia, and varying or no trends in Styria and Salzburg (Haas and Birk, 2019) (medium confidence). Changes in groundwater levels are directly related to climate change, especially during dry periods (Haas and Birk, 2017), but also indirectly due to increased water consumption and water management (see Sections 2.2.1 and 7.4.1) and proximity to surface water bodies. In general, the impact of hydrometeorological extreme events on groundwater depends on local hydrogeological conditions, particularly whether precipitation or stream-aquifer interaction is the primary driver of groundwater level changes (Haas and Birk, 2019). A nationwide analysis of groundwater level changes and groundwater droughts with more recent data is still lacking. It has to be emphasized that the concept of groundwater drought research is rather new and only limited index-based methods exist (Bloomfield and Marchant, 2013). Due to the lack of Austria-wide studies, no confident and reliable statement can be made for the current situation; such research should include the development of new methods based on time series analysis, similar to streamflow drought (Stahl et al., 2020).

It is not known whether declines in groundwater levels due to exploitation or reduced groundwater recharge can cause infrastructure-relevant subsidence in Austria, as has been observed in other – mainly densely populated – areas worldwide (Corti et al., 2009; Chaussard et al., 2021). However, it is expected to be highly uncertain given the groundwater level variability currently observed in Austria, as drastic water level declines in clay-rich sediments would have to persist over longer time scales (Collados-Lara et al., 2020).

Furthermore, little is known about the effects of climate change on future groundwater quality and ecology. For Austria, absolute groundwater temperature changes were observed that were even higher  $(+0.7\pm0.8^{\circ}C)$  compared to air temperature changes for the period 1994–2013  $(+0.5\pm0.3^{\circ}C)$  (Benz et al., 2018). Therefore, these temperature increases are also expected to affect biogeochemical processes, similar to the potential impacts of geothermal energy use on groundwater quality and ecology (Griebler et al., 2016). As with climate change impacts on groundwater levels, the impacts on groundwater quality will depend on whether the main source is precipitation or also indirect recharge from streams.

#### Future changes

Predictions of groundwater recharge rates (1981–2100) for grassland sites vary drastically between three ÖKS15 models

(MOHC-HadGEM2-ES, ICHEC-EC-EARTH, IPSL-CM5A-MR) and two RCPs (4.5 and 8.5), especially for the summer months (Schübl, 2023) (see Figure 1.A.14). An increase in recharge is simulated mainly for winter and for western sites, due to higher temperatures with less snow accumulation and/or higher amounts of winter precipitation, followed by decreasing recharge rates in spring. If the scenarios show decreasing trends in annual groundwater recharge, they are more pronounced at western sites and at higher elevations, with longer droughts lasting until later in the calendar year. Uncertainty in recharge prediction is largely dominated by the difference in climate projections; only at the dry sites in the east and for shorter time periods, do uncertainties in soil hydraulic parameters play a role in predicting water fluxes. Significant increasing trends in recharge are projected for IPSL (RCP8.5) and decreasing trends for MOHC (RCP4.5) until the end of the 21st century (limited evidence, medium agreement). However, an increase in potential evapotranspiration for MOHC (RCP4.5) does not consistently lead to an increase in actual evapotranspiration rates, as the predicted dry conditions would not provide enough water to meet the atmospheric evaporative demand (Schübl, 2023). Such behavior is also predicted for other dry regions (Ng et al., 2010). Integrated over Austria, the available groundwater resources may decrease by up to 23 % by 2050, regionally by as much as 30 %, whereas the decrease in the east of Austria is small and groundwater resources may even increase with increasing winter precipitation (BMLRT, 2021; Schübl, 2023) (medium confidence). The resulting water fluxes from the use of different climate models and emission scenarios differ, highlighting the need for more accurate future scenarios for predicting water fluxes and availability with lower uncertainty. In addition, changes in water fluxes should not only be compared between climate scenarios, but also in relation to GWLs. It would also be necessary to consider the consequences of land use and land cover changes on subsurface water fluxes due to climate change, such as longer growing seasons, changes in plant diversity or human water consumption.

#### 1.5. Pedosphere

The diversity of geology, landscape and climatic conditions in Austria results in a large variety of soils. Agricultural soils (cropland and grassland) are well documented.<sup>7</sup> For the ap-

See <u>bodenkarte.at</u>

proximately 50 % of the Austrian land area that is not used for agriculture, the documentation is either coarse or limited to certain areas. Improved documentation of forest soils is currently being carried out. However, only a part of this data is yet available to the public.

Land use has influenced Austria's soils for millennia, depending on its intensity and soil type. Therefore, Austria's current 'soilscape' is as much a result of natural processes as it is of human intervention. The current driving forces shaping Austria's soils are management practices: Land use (cropland, grassland, forest), tillage, fertilization, crop rotation, crop residue management, water management, cutting practices, etc., as well as micro- and macroclimatic conditions. In combination, land use and climatic conditions produce additional changes, often within relatively short periods of time. As a result of these drivers, important impacts on both the abiotic and biotic environment can be expected, in particular changes in organic matter dynamics and erosion (including water and wind erosion, see also Section 1.4). Countermeasures against soil degradation should be specifically linked to management practices capable of preserving soil functions. First and foremost, the long-term stabilization or increase of soil organic carbon is an important aspect in countering the consequences of climate change. Carbon-rich soils have higher aggregate stability, higher water infiltration rates and water retention capacity (Amlinger and Geszti, 2001; Erhart and Hartl, 2010; Kolbe and Zimmer, 2015), making them more resilient to extreme events such as heavy rainfall and drought (Fliessbach et al., 2008; Petersen and Weigel, 2015), reduce the risk of erosion (Erhart and Hartl, 2010) and, above all, produce higher yields in the medium to long term (see Chapter 2) (Johnston et al., 2009; Kolbe and Zimmer, 2015). In specific cases, management changes may also be an option to maintain key soil functions. While the overall effects of management measures are well documented, detailed mechanisms remain to be elucidated. Furthermore, research is needed to foster the implementation of respective measures in the practical land management.

In general, the carbon sequestration potential of most Austrian soils is not large, with the notable exception of histosols (peat soils, see Section 2.1.1 and Table 2.1). First estimates for arable soils (Baumgarten et al., 2022) show a sequestration potential of max. 2.2 tC/ha, where the results show clear differences in different regions (from losses to moderate sequestration). Acceleration is possible both with respect to climate change (e.g., enhanced mineralization) and specific measures related to carbon farming practices (for more details, see Table 2.3 and Section 2.3.1). The effect of measures may also differ regionally (e.g., soil tillage, waterlogging).

As soil erosion is an important mechanism of soil degradation and affects soil organic matter (SOM) and water dynamics, it is discussed in more detail: Precipitation-induced soil erosion is strongly dependent on precipitation intensity. Common model approaches to estimate so-called precipitation erosivity are often closely coupled to the maximum half-hourly precipitation intensity (I30). Currently, however, most climate models do not provide sufficient temporal resolution to accurately estimate peak intensities in the intermediate and distant future. Cluster studies (Vasquez et al., 2023) show that especially convective precipitation types in summer, often of short duration and high peak intensity, dominate the erosivity behavior in Austria - even though the total precipitation amounts of highly erosive events can often be relatively lower than those of long-lasting stratiform types.

The clear trends towards temperature increase are accompanied by an increased tendency to convective events, and thus extreme precipitation erosivity. In particular, the southeastern foothills of the Alps in Carinthia and Styria have an increased risk of highly erosive events (Johannsen et al., 2022; Vasquez et al., 2023). Ultimate soil erosion, however, depends on factors other than precipitation erosivity, such as terrain, land use, vegetation cover, and soil infiltration and erodibility. Erosion plot studies by, e.g., Strohmeier et al. (2016) or Rab et al. (2023), have shown that the local context of precipitation erosivity and soil erodibility is critical to properly assess erosion behavior. The study by Strohmeier et al. (2016) showed that in Mistelbach (Lower Austria), for example, the long-term average soil erosion is dominated by the extreme events (i.e., >20 years recurrence interval), whereas in Pixendorf near Tulln (Lower Austria), about 50 km away, with slightly higher average annual precipitation, the large number of less erosive but much more frequent events (i.e., <<20 years recurrence interval) have the largest share in the total erosion events. Climate research to determine future precipitation intensities will be of central importance to properly assess erosion behavior and thus the sustainability of, e.g., agricultural ventures in context (high confidence).

In addition, higher temperatures may lead to higher mineralization rates of SOM if the water supply is still sufficient (Schindlbacher et al., 2015; Wiesmeier et al., 2016). This can be expected especially in the western regions of Austria with higher elevations. This loss of SOM may lead to an increase in GHG emissions (Wiesmeier et al., 2016), but may also have negative effects on soil structure (Jensen et al., 2019) (*high confidence*).

## 1.6. Biosphere

#### 1.6.1. Wetlands

This section describes the characteristics of wetlands in Austria and assesses the threats that wetlands in Austria face from climate change and inappropriate land management, which ultimately lead to habitat degradation and changes in their specific animal and plant communities.

Wetlands can be categorized according to different guidelines. The IPCC classifies wetlands as a land use category (IPCC, 2014a), while other classifications define wetlands more broadly (Ramsar, 1971). Although this report takes a similar approach to the IPCC, it does not define wetlands according to the IPCC wetlands supplement (IPCC, 2014a). Here, both peat accumulating wetlands and wetlands that do not accumulate peat, such as some floodplains and soda pans, are included, but drained sites are not, as these are covered in Section 1.5.

Austria hosts a wide variety of mires. Due to the climatic and topographic diversity and varying degrees of land use intensity, many different types of mires still exist today. In the mountains, mires predominate, while in the lowlands, floodplains, riparian wetlands or soda pans occur.

The majority of wetlands in Austria are managed. Many mires and floodplains are dominated by grassland or forest. Management practices strongly influence the occurrence and abundance of species. For examples of management and its impact on biodiversity, see paragraph on anthropopogenic drivers of wetland degradation below and Section 1.6.4. Historically, and especially in the last 80 years, drainage to convert natural wetlands to agricultural or forest land has been the most important driver of wetland loss (high confidence). Although it is possible to restore wetlands, drainage can alter wetland characteristics to such an extent that they cannot be successfully restored in the medium term. Mires in particular experience severe changes in soil porosity and peat composition that may make restoration very difficult even in the long term (Loisel and Gallego-Sala, 2022) (high confidence).

Despite the often detrimental influence of land management on wetlands, wetland management is sometimes welcomed for the sake of species conservation. In Austria, many mires are extensively managed without strong drainage. Such sites are valued by conservationists as they host valuable biodiversity. Nevertheless, it must be emphasized that the survival of wetland species is impossible without intact wetlands, so wetland conservation is the highest management priority.

Wetlands have been lost to a greater extent than other ecosystem types (*high confidence*). 80 % of Europe's wetlands have been lost in the last 100 years (Verhoeven, 2014). They are highly threatened by climate change, as a positive water balance is a prerequisite for the existence of wetlands. The climatic drivers of wetland degradation are droughts, increased evapotranspiration, extreme precipitation and the resulting strong runoff, and a more incoherent water supply due to the disappearance of glaciers and reduced water storage in snow (*high confidence*) (see Section 1.2.2).

Of all ecosystems, peatlands store the most carbon per area. Covering 3 % of the Earth's land surface, they store one third of global soil carbon. When intact, they are persistent carbon sinks, but when drained, they become strong carbon sources. On a global scale, they are estimated to contribute 5 % of global greenhouse gas emissions (Leifeld and Menichetti, 2018). In Austria, this figure is unknown, but is likely to be around 1-2 % of national greenhouse gas emissions (BMLRT, 2022). The high carbon dioxide release associated with peatland drainage is itself a driver of climate change (see Section 2.2.2), creating a positive feedback loop by releasing greenhouse gases that in turn promote peatland drying-out in a warmer climate. Climate change has detrimental effects on most wetlands in Austria, although in some locations climate change is expected to foster the development of mires in the medium term (Essl et al., 2012). Overall, most wetlands - such as salt pans, floodplains and most mires - can be expected to be negatively affected by climate change (high confidence). In Austria, the effects of land use are immediate and more severe than those of climate change in the case of drainage, damming and diking of rivers, and the establishment of most agricultural crops or commercial trees (Gaube et al., 2024) (high confidence).

In addition to their value for biodiversity, the carbon cycle and greenhouse gas emissions, wetlands are important controls in the water cycle, slowing the runoff of excess water in rivers or on the soil surface. In the case of peatlands, they store water like a sponge and release it slowly under drier conditions (*high confidence*), alleviating water runoff problems in Austrian river valleys. Furthermore, wetlands provide local cooling due to their high evapotranspiration rates (Worrall et al., 2022). Thus, especially in Austria with its many valleys, rivers and small depressions, wetlands are an important insurance against extreme weather events and, more generally, against the effects of climate change. They are an important component of Austria's climate-resilient landscapes.

# 1.6.2. Forests

Forest ecosystems play an important role in global biogeochemical cycles, act as both sources and sinks of greenhouse gases, and thus have a significant influence on the Earth's climate. Here, we assess the current state and potential impacts of a changing climate on Austrian forests, as modulated by the tree population dynamics processes of growth, mortality and reproduction. How ecosystem services may be affected by a changing climate is discussed in Chapter 2.

According to the latest results of the Austrian National Forest Inventory (ÖWI 2016–21)<sup>8</sup>, the forest area in Austria is 4.02 million ha (47.9 % of the land area). Of this, 14.7 % is classified as non-productive forest (not in yield), representing steep, inaccessible and unproductive forest areas. During the last decades, the forest area in Austria increased by 3,000–4,000 ha per year. Almost the entire forest area has been subject to land use (wood utilization, grazing, litter raking) with varying intensity for centuries.

According to a comprehensive study by Grabherr et al. (1998), 25 % are natural and semi-natural, 41 % are moderately modified and 34 % are heavily modified. Due to ongoing changes in species composition and deadwood accumulation, the proportion of moderately modified forests may have increased moderately. At present, 58.6 % of the accessible forest area is coniferous, 24.1 % is deciduous, 17.3 % is cleared and currently not stocked with timber species. *Picea abies* is the most common coniferous species with 46.2 % of the total area, the most common deciduous species is *Fagus sylvatica* (10.4 %).

Ownership is a key feature for understanding the current state of Austrian forests and drawing conclusions for future development paths. Small private owners (<200 ha) account for 53.5 % of the forest area, large private owners and communities for 35.7 %, and the state-owned share of 14.8 % is managed by the Austrian Federal Forests (ÖBf AG). The current average standing stock is 303 m<sup>3</sup>/ha. The standing stock has been increasing for decades because the average annual harvest is lower than the annual gross growth. The relationship between growth and harvest varies between the different types of ownership. For small owners, it has been varying between 0.75 and 0.85, depending on timber prices and unplanned harvests due to disturbances. For large private owners and ÖBf AG, the balance between growth and harvest is approximately equal. The Environment Agency Austria (Umweltbundesamt, 2024) estimates that due to high damage caused by disturbances (storms, bark beetles) and reduced increment, timber removals exceeded growth in the years 2018 and 2019. This may occur more frequently in the future (*medium confidence*).

Climate change affects forest development by influencing the growth, reproduction and mortality processes of tree populations. These effects will vary depending on soil conditions, region and altitude. At higher elevations, temperature is currently a limiting factor for many tree species, especially broadleaf species. In a warmer climate, habitat for these species will increase in the mountains (Fagus sylvatica, Acer pseudoplatanus), while in the currently already warm and dry lowlands, drought-adapted tree species such as oaks will gain habitat and Fagus sylvatica will lose habitat (Augustine and Reinhardt, 2019; Buras et al., 2020; Kessler and Lexer, 2023). Productivity in mountain forests (>800 m a.s.l.) will increase by about 10-25 % by the end of the 21st century due to more favorable thermal conditions (Irauschek et al., 2017a; Albrich et al., 2020). At GWL 4.2°C, however, productivity in the lowlands (<800 m a.s.l.) will decrease by up to 50 %. Overall, temperature will be the dominant factor in the mountains and water limitation at low elevations (<800 m a.s.l.).

Disturbances lead to abrupt changes in forest structure and processes and are thus a key determinant in shaping the future composition and structure of forest ecosystems (Seidl et al., 2017; Patacca et al., 2023). The potential for natural regeneration (seed availability) and browsing pressure by ungulates (Capreolus capreolus, Cervus elaphus, Rupicapra rupicapra) will play an important role in shaping future species composition. In Austrian forests, the most relevant disturbance agents are wind, snow breakage, bark beetles and pathogenic fungi. Since about 1992, the amount of timber volume damaged by spruce bark beetles (mainly Ips typographus and Pityogenes chalcographus) has increased significantly from a mean annual damaged volume of about 250,000 m<sup>3</sup>/year (1950-1990) to a mean damage volume of about 1.9 million m<sup>3</sup>/year (2002-2012) to a mean damage volume of 2.8 million m3/year (2010-2023). The main drivers are warmer temperatures, which favor the development of insect populations, and the increasing vulnerability of

<sup>8</sup> waldinventur.at/#/



**Figure 1.15** Damaged timber volume due to bark beetle and storm disturbances in Austrian forests (for a description of the approach, see Steyrer and Tomiczek, 2002).

the host species, Norway spruce, to water shortages (Figure 1.15). In mountainous regions, the high proportion of Norway spruce and warmer temperatures will increase the risk of intense bark beetle outbreaks, with significant negative impacts on the provision of ecosystem services such as protection against gravitational hazards (see Section 2.2.2, 7.4.2). New invasive pests, such as the fungus *Hymenoscyphus fraxineus* which caused the ash dieback in the 2000s, are a significant risk for the future (Halmschlager and Kirisits, 2008; Lapin et al., 2019).

The frequency of forest fires does not show a clear trend and varies between 150–300 fires per year. However, there appears to be an increasing trend in the frequency of fire events with area burnt >30 ha (Müller et al., 2020a). Expected increases in the intensity of droughts and heat waves, combined with more recreational activities, are likely to increase forest fire activity in the Alpine region (Müller et al., 2020b).

The long natural lifespan of trees and the long production cycles in managed temperate forests do not allow for rapid adaptation to environmental changes. This gap between adaptation needs and long lead times can seriously hamper the sustainable provision of ecosystem services from forests (Irauschek et al., 2017b; Sotirov et al., 2024) (see Chapter 2).

# 1.6.3. Grasslands

Grasslands are widespread in Austria. They cover about half of the cultivated land (excluding forestry) (Umweltbundesamt, 2020) and are mostly managed as pastures or meadows. Without management, a large part of these grasslands would develop into forests (or in some cases shrublands). Natural grasslands are limited to regions above the tree line and small areas that are either too dry or too wet for forests (Leuschner and Ellenberg, 2017). Semi-natural and intensively used grasslands are found in most parts of Austria, but are rare in the Pannonian region, where crop cultivation dominates. Grasslands provide a number of ecosystem services (ecosystem services are discussed in detail in Section 2.2). They provide forage for livestock and are important for carbon storage, tourism (see Sections 4.4.3 and 7.4.2), filtration of drinking water (e.g., Vienna's drinking water catchments), reduction of runoff and erosion control, especially in mountainous areas (Section 7.4.2). In addition, many extensively managed grasslands in Austria are located on peat soils and thus have a very specific species composition and high importance for soil carbon storage (see Sections 1.6.1 and 2.2.2).

The area of cultivated grasslands below the tree line has been decreasing for several decades, mainly because less productive and/or more difficult to cultivate areas have been abandoned and/or afforested (Pils, 1994; Umweltbundesamt, 2020). The use of the remaining grasslands was massively intensified in the second half of the 20th century, followed by a slight tendency towards de-intensification in recent decades. Currently, 30 % of Austria's cultivated grasslands are under organic farming regimes (BMNT, 2019). However, eutrophication and the high number and early timing of cuttings remain serious problems for many biota adapted to grassland ecosystems. As a result of the double pressure of abandonment and intensification, grassland specialists among plants and animals are high on national Red Lists (Schratt-Ehrendorfer et al., 2022; Umweltbundesamt, 2023), and at the habitat level, traditionally used, nutrient-poor to moderately nutrient-rich hay meadows with one or two mowings per year have become habitats of conservation concern and are listed in Annex I of the European Union's Habitat Directive (Types 6510 and 6520: lowland and mountain hay meadows).

Thus, climate change is not yet a major threat to grassland ecosystems in Austria. However, increasing drought could reduce the economic value or make them dependent on irrigation, possibly triggering conversion to other types of use. Depending on the type, some grasslands are quite resistant to drought in terms of species composition and also productivity (Deléglise et al., 2015). On average, however, grassland productivity is reduced under drier or even drought conditions, especially in the already dry areas of Austria (*high confidence*). Prolonged droughts can also lead to a shift in species composition (Stampfli and Zeiter, 2004). The high biodiversity of natural and low-intensity grasslands may act as a buffer against the effects of drought and increase the resilience of grasslands (Kreyling et al., 2017), but research on these topics in Austrian grasslands is lacking.

In high-alpine areas, the current shrinkage of grasslands is a consequence of the rise of the tree line and the densification of forests, which is primarily caused by the regionally varying abandonment of summer pastures (Gehrig-Fasel et al., 2007). Climate impacts on high alpine grasslands are likely to be slow to manifest (Dullinger et al., 2004; Tasser et al., 2017), but potentially massive (*medium confidence*). Areas above the tree line could shrink by >50 % in the European Alps even under moderate warming (Nagy and Grabherr, 2009), with severe consequences for tourism (Section 7.4.2) and biodiversity (see Section 1.6.4).

Natural or semi-natural grasslands below the tree line are either wetlands or dry grasslands. While the former are likely to decrease under climate warming (Essl et al., 2012; Baatar et al., 2019) (*medium evidence, high agreement*) (see Section 1.6.1), the latter may even benefit, as many of their biota are adapted to warm and dry conditions (Baatar et al., 2019). However, actual expansion will depend on whether concurrent land use provides the necessary space. Where grasslands are managed, use may intensify in the future to keep productivity high despite drier conditions, offsetting potential positive effects of climate change on warm or drought-adapted species.

# 1.6.4. Biodiversity

In Austria, climate change affects flora and fauna that are already under severe pressure from other stressors, such as agricultural intensification, changes in the composition and structure of pristine forest tree species, abandonment of traditional low-intensity land use, land consumption, pollution, or changes in water flow regimes (high confidence). Depending on the species group, between 25-65 % of the species occurring in Austria are listed as critically endangered, endangered or vulnerable in groups such as mammals, birds, amphibians, reptiles, fish, a number of invertebrate species or vascular plants, with only a minority so far citing climate change as the main reason for the threat (Schratt-Ehrendorfer et al., 2022; Umweltbundesamt, 2023). Nevertheless, the effects of climate change have already been observed in numerous species and are increasingly contributing to the pressure on biodiversity. This occurs both directly - through changes in temperature or drought conditions that exceed tolerance limits (Dorts et al., 2012; Borgwardt et al., 2020; Reiner et al., 2021) - and indirectly through adverse effects on non-climatic environmental conditions (Thom et al., 2017) or biotic interactions (Gilgen et al., 2010; Temperli et al., 2013; Alexander et al., 2015; Jakoby et al., 2019; Rehnus et al., 2020). Depending on the level of future warming and the local or regional effects, predictive models suggest that the contribution of climate change to the pressure on biodiversity will increase and may even exceed that of further land-use changes in certain taxonomic groups, habitats and regions (Dullinger et al., 2020; Barras et al., 2021a; Neff et al., 2022) (*medium confidence*).

Austria is a mountainous country, and mountains are the environment where climate impacts on species have been most clearly documented. In particular, the warming of recent decades has begun to affect the spatial distribution of mountain biota (high confidence) (see Chapter 7 for more details). Reports on upslope species shifts in Austria or neighboring countries are available for different taxonomic groups, e.g., mammals (Schai-Braun et al., 2021), birds (Popy et al., 2009; Knaus, 2018; Bani et al., 2019; Schai-Braun et al., 2021; Teufelbauer et al., 2024), butterflies (Bonelli et al., 2021; Kerner et al., 2023), bees and bumblebees (Biella et al., 2017; Maihoff et al., 2023; Scharnhorst et al., 2023), grasshoppers (Illich and Zuna-Kratky, 2022), snails (Baur and Baur, 2013), millipedes (Gilgado et al., 2022), vascular plants (e.g., Küchler et al., 2015; Lamprecht et al., 2018; Rumpf et al., 2018) and bryophytes (Bergamini et al., 2009). A recent meta-analysis from Switzerland - with results likely applicable to Austria - documents rates of shift of elevational optima and upper limits of up to 36 m per decade since 1970, albeit with strong variation among taxonomic groups and species (Vitasse et al., 2021). In vascular plants, a parallel regression of range limits has been demonstrated, implying that local populations of many species have become extinct at former lower range limits (Rumpf et al., 2018) (see Figure 1.16). Similarly, between 1981-1985 and 2013-2018, Austrian mountain birds showed both a contraction of the breeding range and a significant upward shift in elevation (Teufelbauer et al., 2024). The relative contribution of climate and land use change to these range shifts could not be quantified. In any case, upslope shifts of most species lag behind warming trends (Chen et al., 2011; Rumpf et al., 2019; Vitasse et al., 2021) (high confidence). Variation in these lag times among species may disrupt biotic interactions, such as those between faster-moving insects (Maihoff et al., 2023) and slower-moving host plants (Kerner et al., 2023). As a special case, it has been reported that non-native species, which are currently still rare at higher elevations, move upward considerably faster than native species (Dainese et al., 2017), with potential risks for populations of competitively

inferior native species. In the Alps, continued upward shift will inevitably lead to shrinking populations of high-elevation specialists (Revermann et al., 2012; Lamprecht et al., 2018; Rehnus et al., 2018; Brambilla et al., 2022) (high confidence), as a consequence of conical mountain shapes, limited summit elevations, and concurrent tree line shifts induced by both summer pasture abandonment and climate warming (Dullinger et al., 2004; Gehrig-Fasel et al., 2007; Ferrarini et al., 2017; Bani et al., 2019). Microclimatic variation may mitigate extinction risks to some extent (Scherrer and Körner, 2011; Ohler et al., 2020), but declining populations will become increasingly vulnerable to all kinds of hazards, including habitat destruction from the expansion of infrastructure such as ski slopes (Imperio et al., 2013). The magnitude of population decline will depend strongly on the level of warming, and could be as high as  $\geq 80$  % for many alpine species under severe warming (Engler et al., 2011). The decline will be even more pronounced and/or reach comparable levels already at moderate levels of warming in the endemic flora and fauna of high altitudes in Austria, which includes many immobile species restricted to marginal alpine mountain ranges at rather low elevations (Dirnböck et al., 2011; Semenchuk et al., 2021) (medium evidence, high agreement).

Apart from elevational shifts, warm-demanding species, especially those with a (sub-)Mediterranean range center, have expanded their geographic distribution northward and increased their population sizes in Central Europe in recent decades, while cold habitat specialists have tended to decrease in population size, occupancy or population performance (high confidence). Corresponding reports, also from comparable neighboring regions such as northern Italy, Bavaria, Switzerland or the Czech Republic, exist for plants (Loacker et al., 2007; Lamprecht et al., 2018; Reich et al., 2018; Rumpf et al., 2018), various insect taxa (e.g., grasshoppers, dragonflies and butterflies; Kenyeres et al., 2019; Engelhardt et al., 2022; Neff et al., 2022), birds (Lemoine et al., 2007; Hušek and Adamík, 2008; Reif et al., 2008; Stiels et al., 2021), and some mammals such as alpine chamois (Chirichella et al., 2021), marmot (Tafani et al., 2013) and mountain hare (Rehnus et al., 2018). Species that are expanding in response to climate change include several forest and agricultural pests that have increased in distribution, population size and/or voltinism (= number of generations per year) - or are predicted to do so under further warming - such as the bark beetle (Ips typographus) (Seidl et al., 2009; Marini et al., 2012; Temperli et al., 2013), pine processionary moth (Thaumetopoea pityocampa) (Battisti et al., 2005), European corn borer (Ostrinia nubilalis) (Trnka et al., 2007) or codling moth (Cydia pomonella, (Stoeckli et al., 2012). Among the potential 'winners' of climate change are also a number of neobiota native to regions with warmer climates (Walther et al., 2007; Dullinger et al., 2017) (medium evidence, high agreement). Detailed studies of several invasive species of ecological, economic and/or health concern have confirmed that climate change has already triggered or facilitated their spread, or will do so in the future, e.g., Ambrosia artemisiifolia (Storkey et al., 2014; Mang et al., 2018), Robinia pseudacacia (Kleinbauer et al., 2010), Vespa velutina (Barbet-Massin et al., 2013), Aedes albopictus (Caminade et al., 2012; Kraemer et al., 2019; Bakran-Lebl et al., 2022). With continued warming, additional non-native pests and pathogens may establish and expand in Austria (Seidl et al., 2018).

The proximate causes of population declines under a changing climate are likely to vary among species and are only partially understood. Several studies suggest that an increased frequency and intensity of extreme events, such as periods of drought and heat or intense summer precipitation, may often be more important than warmer average temperatures (medium confidence). Corresponding results from Austria and neighboring countries are available for shrubs at the tree line (summer drought and heat: Francon et al., 2020), insects in Switzerland (high rainfall in summer: Neff et al., 2022), soil insects in acid oak forests (drought: Xu et al., 2012) or dry grasslands (prolonged drought: Flórián et al., 2019), truffles (hot and dry summers: Steidinger et al., 2022), and bats (high summer rainfall: Lučan et al., 2013). Intense summer droughts are also the main climatic driver of the widespread growth decline of spruce, the most important Austrian forest tree (Bosela et al., 2021), and may be involved in the population decline of high-elevation specialists among vascular plants (Lamprecht et al., 2018). In addition to extreme events, climate impacts on biota may also be mediated by the timing of snowmelt. Earlier melt-out advances phenology, with possible negative consequences due to increased frost risk (e.g., Rana temporaria) (Bison et al., 2021). Moreover, the changing climate affects species by decoupling biotic interactions in both space and time (medium evidence, high agreement). In space, different rates and directions of climate-driven migration between interaction partners may lead to reduced opportunities for range adaptation, e.g., of specialized herbivores such as certain butterfly species (Descombes et al., 2016; Kerner et al., 2023), or may intensify or relax pressure on hosts (e.g., spruce and bark beetle, larch and larch budmoth (Zeiraphera griseana) (Johnson et al., 2010; Temperli et al., 2013; Jakoby et al., 2019; Büntgen et al., 2020). Over time, different responses of partners to climate change may disrupt the evolved co-timing of phenophases, e.g., between predators and prey (dormice and burrowing birds: Adamík and Král (2008); birds and larval tipulids: Schano et al. (2021); ring ouzel feeding habitat: Barras et al. (2020, 2021a, 2021b)) or plants and herbivores (chamois breeding and vegetation development: Chirichella et al., 2021). The cuckoo is a prominent example of such a temporal mismatch resulting from differences between its own migratory phenology and that of other species it parasitizes (Saino et al., 2009). In general, a shift of spring phenology to earlier dates is a widespread effect of climate change on biota in seasonally cold climates such as Austria (high confidence). A comprehensive review for Switzerland (Vitasse et al., 2021) demonstrates such shifts for the activity of vertebrates (amphibians, reptiles, birds), invertebrates (e.g. butterflies) and plants, with average rates of shift varying between 1-6 days per decade over the last 40 years. The data also show the idiosyncratic nature of these shifts and thus the potential for further temporal mismatches of biotic interaction partners.

In freshwater ecosystems, climate-driven distribution shifts are less commonly reported (Vitasse et al., 2021). However, upward elevational shifts have been documented for some wetland bird species in Austria (Teufelbauer et al., 2024). Similarly, elevational shifts of fish along mountain streams have been observed (Comte and Grenouillet, 2013), and some detailed studies suggest negative effects of warmer water on the physiology and reproduction of several fish species, such as brown trout (Salmo trutta; Borgwardt et al., 2020), European grayling (Thymallus thymallus) (Wedekind and Küng, 2010) or European bullhead (Cottius gobio) (Dorts et al., 2012). Opposite shifts in population size and occupancy of warm- and cold-adapted species have also been reported for freshwater-bound invertebrates such as dragonflies in Switzerland and Bavaria (Engelhardt et al., 2022; Neff et al., 2022), and this turnover is predicted to accelerate under further climate warming and to also affect all pond-associated fauna (Rosset et al., 2010). In both streams and alpine lakes, warming water (see Sections 1.4.1 and 1.4.2) has also triggered changes in the composition of invertebrate (Haase et al., 2019) and phytoplankton (Weckström et al., 2017) communities. The warmer climate and longer droughts will also increase the number of intermittently dry streams, which may also lead to strong turnover of their fauna (Crabot et al., 2021). Lakes that are currently fed by glacial meltwater but will become hydrologically disconnected as glaciers continue to retreat are expected to be particularly affected. Such a change in hydrological regime is likely to result in possibly richer but profoundly different communities and food webs, and unique species and processes of glacier-fed lakes will be lost (Tiberti et al., 2020; Kleinteich et al., 2022) (*medium evidence, high agreement*).

At the local scale of individual communities, global meta-analyses have highlighted considerable turnover in species composition despite relatively stable (average) species richness over the last decades (Vellend et al., 2017; Blowes et al., 2019), with a general trend towards homogenization through the replacement of specialist by generalist species (Jandt et al., 2022; Staude et al., 2022). While these changes are often due to land use, climate change may have similar homogenizing effects in both terrestrial and freshwater ecosystems of Austria/Central Europe (vascular plats in alpine grasslands, Jurasinski and Kreyling, 2007; Liberati et al., 2019; ant communities in grasslands, Gallé, 2017; cyanobacterial communities in lakes, Monchamp et al., 2017; and butterflies in alpine valleys, Bonelli et al., 2021) (medium confidence). However, climate effects on the richness and composition of individual communities are and are likely to be highly idiosyncratic, depending on taxonomic group, habitat type, and geographic location (Rosbakh et al., 2014; Thom et al., 2017; Liberati et al., 2019; Geppert et al., 2021; Kleinteich et al., 2022) (high confidence). At the regional scale of all of Austria, calculations of climate-induced changes in species numbers have not yet been published, either for the climate of recent decades or for future scenarios. Given the vulnerability of many narrowly distributed and dispersal-limited endemics to a warming climate, Austria's contribution to the global species pool is likely to shrink (Dirnböck et al., 2011; Semenchuk et al., 2021) (medium evidence, high agreement).

Both the positive and negative impacts of climate change on species and habitats are modified by other pressures on biodiversity. Reduced land-use pressure may stabilize populations and facilitate climate-driven range shifts to some extent, at least under moderate levels of climate change (Thomas et al., 2012; Wessely et al., 2017) (*medium evidence*, *high agreement*). Similarly, bird species adapted to a warmer climate appear to have benefited from recent warming only when cropland is not the primary habitat, as persistent adverse farming practices hinder their population growth (Nemeth et al., 2016). In freshwater ecosystems, flow regulation (Bruno et al., 2019), eutrophication (Monchamp et al., 2017) and pollution exacerbate climate effects on biota. As a consequence, the rapid and thorough implementation of recent international agreements on protected areas and ecosystem restoration (such as the Kunming-Montreal Global biodiversity framework of the Convention on Biological Diversity (CBD/COP/DEC/15/4, UNEP (2022), or the European Green Deal and the associated EU's biodiversity strategy for 2030 (European Commission Directorate General for Environment, 2021)) in Austria would also contribute to the capacity of biota to adapt to climate change (*medium evidence, high agreement*).

The vulnerability of biodiversity to land-use changes also sets sustainability limits for climate mitigation measures such as land-based CO<sub>2</sub>-removal (CDR, see Section 2.3 and Cross-Chapter Box 6) or the expansion of renewable energy. In Austria, biodiversity conservation often implies the maintenance or restoration of traditional low-intensity land use, as many species of Austrian flora and fauna depend on the habitats created by this type of use (high confidence). Allocating such economically marginal areas to CDR measures such as afforestation or bioenergy capture and storage (BEECS, see Section 4.6 and Cross-Chapter Box 6 is therefore inconsistent with international commitments to halt biodiversity loss and with national biodiversity targets (Kerschbaumer, 2022). In contrast, 'nature-based' CDR, which includes the restoration of degraded ecosystems such as mires or the replacement of tree monocultures with near-natural mixed forests as well as other adaptive forest management strategies such as longer forest rotation cycles and associated higher deadwood volumes (Kropik et al., 2021; Jandl et al., 2024), could provide a double benefit for the intertwined biodiversity and climate crises (Deprez et al., 2024) (see Sections 2.3.2 and 2.3.3). Similar considerations apply to the expansion of renewable energy. The costs of irreversible biodiversity loss associated with energy decarbonization can be reduced or avoided if renewable energy infrastructure (see Section 4.5) does not overlap with protected areas or habitats important for biodiversity (Rehbein et al., 2020), if siting takes into account animal behavior (Gauld et al., 2022; Reusch et al., 2023; Salguero et al., 2023), and if their management thrives on harnessing synergies (Blaydes et al., 2021; Giuntoli et al., 2022). Thus, a comprehensive national strategy and careful spatial planning for land-based CDR that balances decarbonization benefits and biodiversity costs will be necessary to simultaneously meet international climate mitigation and biodiversity conservation commitments. A more detailed discussion of different CDR measures and their side-effects on biodiversity, among others, can be found in Sections 2.3 and 2.4.

In summary, climate warming has begun to reshuffle the distribution of biota in Austria (*high confidence*). The dom-



**Figure 1.16** Proportional changes in abundance and elevational amplitude of 183 mountain plants of the Eastern Alps between the period before 1970 and the years 2013/14, based on a sample of 1576 vegetation plots (complete plant species lists with cover-abundance data, upper panel). Upper blue triangles symbolize 'winners' (both abundance and elevational range size have increased), lower red triangles 'losers' (a loss in both dimensions). Species symbolized by dots combine gain in one dimension with loss in the other. Among these, the dashed diagonal separates those for which the gain in one feature is less than the loss in the other (to the right of the diagonal) from those for which the opposite is true (to the left of the diagonal). The lower panel relates the sum of changes in the two features per species to its optimum elevation prior to 1970 (figure adapted from Rumpf et al., 2018).

inant trend in many taxonomic groups and habitats is a decline of cold-adapted and drought-sensitive species and an expansion of warm-adapted and/or drought-resistant species. As many Austrian (sub-) endemics belong to the former group, climate warming threatens Austria's unique contribution to biodiversity. In contrast, species that are detrimental to society, such as various invasive pests and pathogens, appear to benefit from climate change. In general, however, climate effects on biodiversity in Austria are unsystematically and insufficiently documented, despite a considerable body of literature. To improve our knowledge, a systematic, nationwide, long-term monitoring program with a broad coverage of taxa and habitats, including both terrestrial and freshwater habitats, and a design that disentangles climate effects from those of other drivers would be foremost required. Inclusion of the understudied soil (micro-)biota in this scheme is mandatory, as they are crucial for ecosystem functioning. Ideally, monitoring should also capture effects

56

at the intra-specific level, as changes in genetic make-up and diversity are considered important for the adaptive capacity of biota, but are poorly documented and understood.

### 1.7. Lithosphere

This section presents an assessment of the impact of climate change on geological and geomorphological processes of the Earth's lithosphere. Since endogenous processes such as earthquake activity (Hainzl et al., 2006; Husen et al., 2007) or volcanism are not or only marginally affected by climate change, the focus is on exogenous processes, in particular on landslides of different types relevant for the Alpine region of Austria. Surface settlements caused by groundwater lowering or permafrost degradation are addressed, but phenomena related to natural caves in limestone, collapsible soils, peat, evaporites and mining are out of scope.

# 1.7.1. Theoretical background

Endogenic processes are the driving force for the mountain landscape of Austria on geological time scales, causing a diverse temporally and spatially variable distribution of different rocks. Exogenic processes, including weathering, erosion and all transport processes of geological material, shape the landscape and lead to the characteristic geological and topographic situation (Dietrich and Krautblatter, 2017). This results in the great topographic and climatic diversity in Austria, which is prone to different types of natural hazards. Surface processes are expected to be affected by climate change on different time scales. Weathering and denudation, transport and deposition of sediments are dominated by fluvial, glacial, gravitational and aeolian processes. Landslides are one of the most relevant natural hazards that frequently affect the Austrian terrain, and climate change will negatively influence the magnitude-frequency relationship (Huggel et al., 2012b; Gariano and Guzzetti, 2016) as well as the seasonality and spatial occurrence, especially in dependence on elevation (Stoffel et al., 2014a).

Landslides can result from a variety of causes, including geological, morphological, physical and human, but often have only one trigger (Cruden and Varnes, 1996). A trigger is an external stimulus, such as a rainfall or snowmelt event, that causes a landslide to form almost immediately (Wieczorek, 1996). Preconditioning factors such as high antecedent moisture, degradation of mechanical stability, or changes in land cover can modify the trigger magnitude required to initiate a landslide (e.g., Chiarle et al., 2021). In many cases, however, no obvious attributable trigger can be identified, making it difficult to establish a clear cause-effect relationship between climate change variables and landslide occurrence (Wieczorek, 1996). Identifying and quantifying the impact of climate change on landslides is therefore challenging due to (i) the lack of long-term and comprehensive event records for conclusive statistical analysis, (ii) the difficulty of identifying causes and triggers, as well as their combination, interaction and delimitation, and (iii) the occurrence of cascading or compound processes.

A globally accepted classification system for landslides, based on the material involved and the type of movement, has been proposed by Cruden and Varnes (1996) and applied here to consider the various climate change-related influencing factors in a process-oriented manner. This classification distinguishes between rock and fine- and coarsegrained soils, referred to here as earth and debris. While rock is a solid aggregate of minerals that form a hard and solid mass, soil is an aggregate of mineral and rock particles resulting from sedimentation processes. Earth is a finegrained soil in which 80 % or more of the particles or grains are less than 2 mm in size. Debris contains a substantial amount of coarse-grained soil in which 20–80 % of the particles are larger than 2 mm, which is above the upper limit of sand particle size.

To capture the natural variability of movement mechanisms, Cruden and Varnes (1996) defined five main types, including falling, toppling, sliding, flowing and spreading (Figure 1.17). Taking into account the different geological materials, (i) rock falls, (ii) rock slides, (iii) debris/earth slides, (iv) rock spreads, and (v) flow-like landslides such as debris/earth flows and rock flows (avalanches) are identified as primarily relevant for Austria and are assessed for the influence of climate change.

# 1.7.2. Climate change impact on landslides

Climate change will affect landslide causes and trigger factors in a variety of ways, and the response of landslides to climate change depends on movement type, volume, material, and whether there is first failure or reactivation/acceleration of pre-existing landslides (Crozier, 2010; Gariano and Guzzetti, 2016; IPCC, 2021a). The climate-sensitive causes and triggers relevant for landslide occurrence in Austria include (i) extreme precipitation in terms of intensity and duration (see Section 1.2.2), (ii) rapid and intense snowmelt, (iii) rising groundwater levels or rapid drawdown (see



**Figure 1.17** Landslide classification based on the main movement type: a) fall, b) topple, c) slide, d) flow and e) spread (modified after Cruden and Varnes, 1996; Zangerl et al., 2008).

Section 1.4.3), (iv) flooding accompanied by fluvial erosion of the slope base (see Section 1.4.1), (v) freeze-thaw cycles or shrink-swell weathering, (vi) permafrost thawing or degradation (see Section 1.3.3), (vii) glacier retreat and glacier discharge (see Section 1.3.2), (viii) glacier lake outburst floods (see Section 1.3.2), (ix) fluvioglacial processes, and (x) vegetation removal by forest fire or drought (Glade, 2020; Jacquemart et al., 2024) (see Section 1.6.2).

# 1.7.3. Observed trends and future changes

Recent case study reports of **rock fall** activity in glacial and periglacial regions in the Alps provide evidence that increased activity is associated with high temperatures, heavy precipitation, permafrost thawing, and/or glacier retreat (e.g., Ravanel et al., 2017; Paranunzio et al., 2019). A study of the rock fall database in Austria from 1900 to 2010, which primarily includes events below the permafrost line, does not support the hypothesis of increased activity during periods of higher temperatures (Sass and Oberlechner, 2012). However, case study specific data suggest an increase in rock fall activity in permafrost and glaciated regions of Austria (Hartmeyer et al., 2020a, 2020b). Due to limited data for larger regions, the evidence for the trend of climate change impacts on rock falls in high alpine regions can be considered medium, in other regions limited (Table 1.A.4).

Rock slides are usually deep-seated landslides that can reach volumes of several hundred million m<sup>3</sup> and velocities of a few mm/yr to several m/s. Fragmentation of the sliding mass into slabs with the potential for acceleration phases is often observed. Recent studies report first failures (mostly slabs) or reactivation/acceleration due to climate change-related causes and triggers, including (i) glacier retreat and permafrost degradation (Kellerer-Pirklbauer et al., 2012a; Zangerl et al., 2019; Rechberger and Zangerl, 2022), (ii) slope base erosion due to flooding in combination with heavy rainfall (Eder et al., 2006), (iii) intensive snowmelt (Hofmann and Sausgruber, 2017), and/or (iv) extreme precipitation that raises the water table (Brückl et al., 2013; Pfeiffer et al., 2021; Zieher et al., 2023). Most of the studies are based on single-case investigations, highlighting the need to identify the relevant causes and triggers on a case-by-case basis. In glacier retreat zones and permafrost-affected high mountains, a trend towards an increased occurrence of rock slides can be assumed. However, due to a lack of data and regional statistical analyses, there is medium evidence that the frequency has increased over the past half century. The assessment of the influence of climate change on rock slides at elevations below permafrost and glacier retreat areas is even less clear. Theoretical considerations suggest that initial failures or reactivation/acceleration phases will increase, but there is *limited evidence* for this trend.

Although there is a strong theoretical basis for an increase in **shallow earth and debris slides** activity due to climate change (e.g., Crozier, 2010), there is *limited* statistical *evidence* of changes in recent occurrence due to the lack of a homogeneous database, including the exact location and timing of the event. Case studies have been conducted for several locations worldwide (Gariano and Guzzetti, 2016), and the number of detailed studies for Austria has increased in recent years. Changes in initiation conditions for shallow earth/debris slides are expected in sensitive regions such as the Bregenzer Wald, the Karawanken and the Northern Alps (Lexer, 2019). A scenario-based predictive study showed an increase in shallow earth/debris slide susceptibility in two forested watersheds in Tyrol, with forest disturbances and forest management practices having the potential to partially compensate for negative effects of a future climate on slope stability (Scheidl et al., 2020). Analysis of the 2009 landslide events in the district of Feldbach, Styria, shows that about 10 % of the slides can be attributed to climate change (Mishra et al., 2023) and that the affected area could increase by up to 45 % in a future climate (Maraun et al., 2022). In particular, the uncertainty associated with landslide triggering models can be higher than the uncertainty of climate model variability (Knevels et al., 2023). As these initial results for Austria are supported by other studies (see review by Gariano and Guzzetti, 2016), a trend towards increased shallow slide activity in the future is assumed, but with *medium evidence*.

The response of deep-seated earth or debris slides to climate change is controlled by changes in seasonal or annual rainfall (Gariano and Guzzetti, 2016). Case studies in Italy and the USA, predict a slowdown by the end of the 21st century (Coe, 2012; Rianna et al., 2014), while increased activity may be observed in regions with increased annual rainfall (Gariano and Guzzetti, 2016). Based on a modeling study, Schmidt and Dikau (2004) find that local geomorphological and geological settings are more relevant for deep-seated slides in different soils than variations in groundwater in a changing climate. In contrast, a case study for a deep-seated earth/debris slide in Tyrol shows a temporal relationship between reactivation and intensive snowmelt (Dai et al., 2023), and another case study demonstrates the influence of increased pore water pressure in the subsoil on movement behavior (Jaritz and Soranzo, 2014). Thus, the influence of groundwater on slope movement can be demonstrated for selected case studies, although no general statements can be made. A constant or slightly increasing trend is assumed, but with limited evidence.

In Austria **rock spreading** is mainly located in the Northern Calcareous Alps and occurs when a large slab of limestone or dolomite overlies a ductile substratum (e.g., marls, limestone/marl interbeds, clays, evaporites) (Rohn et al., 2004, 2005). Lateral extension occurs due to plastic behavior and when a basal shear zone is absent or poorly defined. Secondary processes such as rock falls and rock flows originating from the margin zones are common. Recent studies focusing on the influence of climate change variables, in particular the role of groundwater on rock slope spreading, are not available and therefore projected changes are unreliable, resulting in *limited evidence* of the trend.

Past **debris flow** activity in the European Alps has been investigated on a case study basis (e.g., Huggel et al., 2012a; Kiefer et al., 2021) and on a regional basis (e.g., Jomelli et

al., 2004, 2007; Stoffel et al., 2014b). There is a general expectation of an increase in potential rainfall trigger events (Berg et al., 2013) and an increase in sediment availability in high alpine regions, mostly related to changes in the cryosphere and changes in weathering processes (Beniston et al., 2018; Jacquemart et al., 2024). This may not be the case in regions with limited sediment availability (Hirschberg et al., 2021). A database study for the Austrian Alps found an increasing trend in triggering rainfall indices in recent decades, but no corresponding signal for debris flow activity (Schlögl et al., 2021). A predictive study reports regional and seasonal changes in hydrometeorological trigger conditions for debris flows, including a trend toward critical conditions earlier in the year (Kaitna et al., 2023). Due to limited data on the complex interplay between short-term trigger conditions and long-term geomorphologic susceptibility, there is *medium evidence* for the proposed trend.

Earth flows characterized by low to moderate velocities occur on low-inclined slopes, typically in plastic, clay-rich soils. Reactivation or acceleration can occur when the earth flow is destabilized, usually by a temporary increase in pore pressure, often associated with undrained loading mechanisms (Sausgruber et al., 2004; Poisel et al., 2012; Hungr et al., 2014). Studies focusing on climate change variables affecting earth flows are not available for Austria (Glade, 2020). Based on the sensitivity of earth flows to water, it is likely that changing precipitation conditions will have an impact in the future, although there is *limited evidence* for this trend.

Rock flows or rock avalanches with volumes of up to several hundred million m<sup>3</sup> occurred continuously in the Austrian Alps during the Holocene, although two clusters, one at 10,000-9,000 and one at 4,200-3,000 cal. BP were found (Prager et al., 2008; Ostermann et al., 2017). In many cases, the triggers for these extreme events are not clearly verifiable, but for some events a temporal relationship between failure time and seismic activity can be demonstrated (e.g., Lenhardt, 2007; Oswald et al., 2021). Smaller events with volumes in the range of several 10,000 to 1,000,000 m<sup>3</sup> are more frequent in the Austrian Alps and are often, but not exclusively, related to the high alpine environment affected by permafrost degradation and glacier retreat (Krautblatter et al., 2024). Data from the European Alps show a trend of increasing frequency in recent decades (Huggel et al., 2010, 2012a), and this trend is expected to continue for the high alpine area in Austria, suggesting medium evidence for this trend. It is still difficult to assess the effects of climate change on steep rock slopes at lower elevations, where there

is no influence from glacier retreat and permafrost degradation, because data and studies are scarce. However, from a rock mechanics perspective, it is expected that (i) thermo-mechanical forcing due to larger temperature variations (Gischig et al., 2011a, 2011b; Luethi et al., 2015), (ii) increase in frost-thaw cycles, and (iii) extreme precipitation events will amplify and accelerate the rock strength weakening process (e.g., Atkinson and Meredith, 1987). It is assumed that the number of events will remain the same or increase only slightly, although the *evidence* for this trend is *limited*.

**Surface settlement** due to climate change is caused by permafrost thaw (Harris et al., 2001) and/or rock glacier creep, and by lowering of groundwater levels due to reduced recharge rates (see Sections 1.3.2 and 1.3.3). Studies on permafrost-driven surface settlement processes are related to infrastructure damage (Duvillard et al., 2019), creep phenomena of rock glaciers (Avian et al., 2009; Fey and Krainer, 2020), and in fractured rock masses (Klug et al., 2017). It is likely that permafrost-induced subsidence will become more common as temperatures continue to rise (see Section 1.3.3).

Surface settlements due to groundwater drawdown receive little attention in Austria because the adverse effects are generally small (Bedini, 2021). The assessment of projected settlement magnitudes is based on scenarios of natural groundwater drawdown due to reduced recharge rates (Glade, 2020). However, the expected magnitudes may be small compared to settlements caused by anthropogenic factors (e.g., excessive groundwater pumping, see Section 1.4.3). Due to the lack of data on both permafrost and groundwater drawdown related settlement, there is *limited evidence* for this trend.

#### 1.8. Cross-cutting topic: Natural hazards

According to IPCC (2021a), hazards define the potential for the occurrence of a natural or human-induced physical event or physical impact that may cause loss of life, injury or other health impacts, as well as damage and loss to property, infrastructure, livelihoods, services, ecosystems and environmental resources. In this section, the term hazard primarily refers to processes and events related to the different earth spheres, usually called natural hazards.

Table 1.A.4 lists and summarizes selected natural hazard processes that are susceptible to climate change in Austria, grouped into the categories atmosphere, cryosphere, hydrosphere, biosphere, and lithosphere (see Sections 1.2–1.7).

Only individual, i.e., single hazards and/or climate impact drivers (Ruane et al., 2022) are presented here in fine resolution, with a brief description of the type, the affected region, proposed indicators, and how climate change affects the natural hazard process in Austria (Fuchs et al., 2022). Information on future trends is included by providing an assessment of the evidence and agreement of trends (Mastrandrea et al., 2010). Irrespective of single hazards, there is often the potential for multi-hazards to occur, which can lead to much more severe damage to settlements and infrastructure and, in the worst case, cause fatalities (Kappes et al., 2012). A number of studies on this topic have been published in the last decade to clarify ambiguities in terminology, but most importantly to demonstrate the need to consider spatial, temporal, and causal dependencies of hazard processes, with a general distinction between cascading and compound processes (Cutter, 2018; Mani et al., 2023). In the context of this assessment, numerous cascading or compound processes can be identified for Austria in the past, some of which led to major damage and high financial losses. A selection of the hazard types identified in this assessment as particularly relevant for Austria and their expected trends is shown in Figure 1.18. The concept and identification of key risks for Austria is presented in Cross-Chapter Box 1.

Extreme rainfall over large areas has triggered debris flow events in headwater catchments, which deliver large amounts of sediment to mountain rivers in very short time scales (Hübl et al., 2005; Pfurtscheller, 2014), leading to river damming and subsequent flooding. Riverbank erosion triggered or reactivated deep-seated rock slides by undercutting the base of the slope during the 2005 flood event in Tyrol (Eder et al., 2006).

A recent temporally multi-hazard event in Austria occurred in 2021 in the Pinzgau region. Due to a succession of rainstorm events on August 14, 15, and 16, debris flows from several torrent catchments, including three from the same catchment, devastated settlements and infrastructure in the Salzach valley floor (Hübl et al., 2022).

A weather situation characterized by a warm and wet winter, such as in January 2018 and 2019, caused multiple hazard events over large areas, including snow avalanches at high elevations and a combination of local floods, debris flows, and earth/debris slides at lower elevations due to a sequence of rain-on-snow events (Stoffel and Corona, 2018).

Drought affects all water resources and can lead to extensive damage and high economic losses (Laimighofer and Laaha, 2022; Haslinger et al., 2023; Tallaksen and Van Lanen, 2023). Drought hazards in Austria typically occur in



summer, when a prolonged dry spell leads to a precipitation deficit (meteorological drought), which propagates through the hydrological cycle and can lead to droughts in the entire hydrological system (soil moisture, streamflow and groundwater drought). Socio-economic sectors are differently vulnerable to different types of droughts, so each sector is affected differently depending on the specific genesis of the event and the timing and region affected (Van Lanen et al., 2016; Kramsall, 2018). Meteorological droughts are often associated with heat waves that cause health hazards and mortality. They increase the risk of wildfires, which in turn can trigger deforestation and increase the risk of avalanches or rock falls. On the other hand, wildfires lead to soil degradation and erosion processes, triggering debris flows, landslides, rock falls and other natural hazard cascades.

Soil moisture droughts affect plant growth and, depending on the season, cause losses to agriculture (agricultural drought) or biological systems. Droughts and heat waves can lead to reduced tree vitality, which favors the growth of pathogens (fungi) and insects (bark beetles), thereby triggering snow avalanches, erosion and rock fall. Other hazard chains triggered by drought and heat waves include tree damage leading to bark beetle infestations and deforestation promoting snow avalanches.

Streamflow droughts manifest themselves as low-flow events and can, among other things, threaten water quality, restrict navigation, and reduce hydropower production (Laaha and Blöschl, 2007). In particular, the combination of low flows and high water temperatures can lead to fish mortality, plant damage, and ecosystem degradation. Finally, groundwater droughts can limit the supply of water for drinking and irrigation, increasing the risk of agricultural drought. The damage caused by droughts has been shown to exceed that caused by major floods, as the 2003 event demonstrated not only for Europe but also for water-rich Austria.

### REFERENCES

- Adamík, P., and Král, M. (2008). Climate- and resource-driven long-term changes in dormice populations negatively affect hole-nesting songbirds. *Journal of Zoology* 275, 209–215. <u>https://doi.org/10.1111/j.1469-7998.2008.00415.x</u>
- Adrian, R., O'Reilly, C. M., Zagarese, H., Baines, S. B., Hessen, D. O., Keller, W., et al. (2009). Lakes as sentinels of climate change. *Limnol. Oceanogr.* 54, 2283–2297. <u>https://doi.org/10.4319/lo.2009.54.6\_part\_2.2283</u>
- Albrich, K., Rammer, W., and Seidl, R. (2020). Climate change causes critical transitions and irreversible alterations of mountain forests. *Global Change Biology* 26, 4013–4027. <u>https://doi. org/10.1111/gcb.15118</u>
- Alexander, J. M., Diez, J. M., and Levine, J. M. (2015). Novel competitors shape species' responses to climate change. *Nature* 525, 515–518. https://doi.org/10.1038/nature14952
- Allen, J. T. (2018). "Climate Change and Severe Thunderstorms", in Oxford Research Encyclopedia of Climate Science. <u>https://doi.org/10.1093/acrefore/9780190228620.013.62</u>
- Amlinger, F., and Geszti, J. (2001). Evaluierung der nachhaltig positiven Wirkung von Kompost auf die Fruchtbarkeit und Produktivität von Böden., in *Fachgruppe Boden, Fachgruppe Düngemittel und verwertbare Abfallstoffe*, (Wolfpassing), 37–38. Available at: <a href="https://raumberg-gumpenstein.at/jdownloads/Tagungen/Alvatagung/Alvatagung\_2001/2a\_2001\_amlinge1.pdf">https://raumberg-gumpenstein.at/jdownloads/Tagungen/Alvatagung\_2001/2a\_2001\_amlinge1.pdf</a>
- APCC (2014). Österreichischer Sachstandsbericht Klimawandel 2014 (AAR14)., ed. Austrian Panel on Climate Change (APCC).
  Wien, Österreich: Verlag der Österreichischen Akademie der Wissenschaften. <u>https://doi.org/10.1553/aar14</u>
- Aquilina, L., Stumpp, C., Tonina, D., and Buffington, J. M. (2023). "Chapter 1 – Hydrodynamics and geomorphology of groundwater environments", in *Groundwater Ecology and Evolution* (Second Edition), eds. F. Malard, C. Griebler, and S. Rétaux (San Diego: Academic Press), 3–37. <u>https://doi.org/10.1016/B978-0-12-819119-4.00014-7</u>
- Armstrong McKay, D. I., Staal, A., Abrams, J. F., Winkelmann, R., Sakschewski, B., Loriani, S., et al. (2022). Exceeding 1.5°C global warming could trigger multiple climate tipping points. *Science* 377, eabn7950. <u>https://doi.org/10.1126/science.abn7950</u>
- Atkinson, B. K., and Meredith, P. G. (1987). "THE THEORY OF SUBCRITICAL CRACK GROWTH WITH APPLICATIONS TO MINERALS AND ROCKS", in *Fracture Mechanics of Rock*, (Elsevier), 111–166. <u>https://doi.org/10.1016/B978-0-12-066266-1.50009-0</u>
- Auer, I. (2014). "Kapitel 3: Vergangene Klimaänderung in Österreich", in Österreichischer Sachstandsbericht Klimawandel 2014 (AAR14). Austrian Panel on Climate Change (APCC), eds. H. Kromp-Kolb, N. Nakicenovic, K. Steininger, A. Gobiet, H. Formayer, A. Köppl, et al. (Wien: Verlag der Österreichischen Akademie der Wissenschaften), 227–300. <u>https://doi.org/10.1553/aar14s227</u>
- Auer, I., Böhm, R., Jurkovic, A., Lipa, W., Orlik, A., Potzmann, R., et al. (2007). HISTALP—historical instrumental climatological surface time series of the Greater Alpine Region. *International Journal of Climatology* 27, 17–46. <u>https://doi.org/10.1002/</u> joc.1377
- Augustine, S. P., and Reinhardt, K. (2019). Differences in morphological and physiological plasticity in two species of first-year conifer seedlings exposed to drought result in distinct survi-

vorship patterns. *Tree Physiology* 39, 1446–1460. <u>https://doi.org/10.1093/treephys/tpz048</u>

- Avian, M., Kellerer-Pirklbauer, A., and Bauer, A. (2009). LiDAR for monitoring mass movements in permafrost environments at the cirque Hinteres Langtal, Austria, between 2000 and 2008. *Nat. Hazards Earth Syst. Sci.* 9, 1087–1094. <u>https://doi.org/10.5194/</u> nhess-9-1087-2009
- Awan, N. K., and Formayer, H. (2017). Cutoff low systems and their relevance to large-scale extreme precipitation in the European Alps. *Theor Appl Climatol* 129, 149–158. <u>https://doi. org/10.1007/s00704-016-1767-0</u>
- Baatar, U.-O., Dirnböck, T., Essl, F., Moser, D., Wessely, J., Willner, W., et al. (2019). Evaluating climatic threats to habitat types based on co-occurrence patterns of characteristic species. *Basic and Applied Ecology* 38, 23–35. <u>https://doi.org/10.1016/j.</u> baae.2019.06.002
- Bakker, P., Schmittner, A., Lenaerts, J. T. M., Abe-Ouchi, A., Bi, D., Van Den Broeke, M. R., et al. (2016). Fate of the Atlantic Meridional Overturning Circulation: Strong decline under continued warming and Greenland melting. *Geophysical Research Letters* 43, 12252–12260. https://doi.org/10.1002/2016GL070457
- Bakran-Lebl, K., Pree, S., Brenner, T., Daroglou, E., Eigner, B., Griesbacher, A., et al. (2022). First Nationwide Monitoring Program for the Detection of Potentially Invasive Mosquito Species in Austria. *Insects* 13, 276. <u>https://doi.org/10.3390/insects13030276</u>
- Bani, L., Luppi, M., Rocchia, E., Dondina, O., and Orioli, V. (2019). Winners and losers: How the elevational range of breeding birds on Alps has varied over the past four decades due to climate and habitat changes. *Ecology and Evolution* 9, 1289–1305. <u>https:// doi.org/10.1002/ece3.4838</u>
- Barbet-Massin, M., Rome, Q., Muller, F., Perrard, A., Villemant, C., and Jiguet, F. (2013). Climate change increases the risk of invasion by the Yellow-legged hornet. *Biological Conservation* 157, 4–10. <u>https://doi.org/10.1016/j.biocon.2012.09.015</u>
- Barras, A. G., Braunisch, V., and Arlettaz, R. (2021a). Predictive models of distribution and abundance of a threatened mountain species show that impacts of climate change overrule those of land use change. *Diversity and Distributions* 27, 989–1004. <u>https://doi.org/10.1111/ddi.13247</u>
- Barras, A. G., Marti, S., Ettlin, S., Vignali, S., Resano-Mayor, J., Braunisch, V., et al. (2020). The importance of seasonal environmental factors in the foraging habitat selection of Alpine Ring Ouzels *Turdus torquatus alpestris*. *Ibis* 162, 505–519. <u>https://doi.org/10.1111/ibi.12764</u>
- Barras, A. G., Niffenegger, C. A., Candolfi, I., Hunziker, Y. A., and Arlettaz, R. (2021b). Nestling diet and parental food provisioning in a declining mountain passerine reveal high sensitivity to climate change. *Journal of Avian Biology* 52, jav.02649. <u>https:// doi.org/10.1111/jav.02649</u>
- Bates, P. D., Quinn, N., Sampson, C., Smith, A., Wing, O., Sosa, J., et al. (2021). Combined Modeling of US Fluvial, Pluvial, and Coastal Flood Hazard Under Current and Future Climates. *Water Resources Research* 57, e2020WR028673. <u>https://doi.org/10.</u> <u>1029/2020WR028673</u>
- Battisti, A., Stastny, M., Netherer, S., Robinet, C., Schopf, A., Roques, A., et al. (2005). EXPANSION OF GEOGRAPHIC RANGE IN THE PINE PROCESSIONARY MOTH CAUSED BY INCREASED WINTER TEMPERATURES. *Ecological Applications* 15, 2084–2096. https://doi.org/10.1890/04-1903
- Baumgarten, A., Haslmayr, H.-P., Schwarz, M., Spiegel, H., Stüger, H.-P., Schwarz, M., et al. (2022). ASOQseq – Austrian Soil Organic Carbon Sequestration Map. Wien: Österreichische Agentur für Gesundheit und Ernährungssicherheit. Available at: <u>https://dafne.at/content/report\_release/ac2c73ab-c767-4ad1-8679-aa101b81762f\_0.pdf</u> (Accessed September 21, 2023).
- Baur, B., and Baur, A. (2013). Snails keep the pace: shift in upper elevation limit on mountain slopes as a response to climate warming. *Can. J. Zool.* 91, 596–599. <u>https://doi.org/10.1139/cjz-2013-0036</u>
- Becsi, B., and Formayer, H. (2024). Linking local climate scenarios to global warming levels: applicability, prospects and uncertainties. *Environ. Res.: Climate* 3, 045003. <u>https://doi.org/10.1088/2752-5295/ad574e</u>
- Bedini, E. (2021). Land Subsidence Assessment by Using Persistent Scatterer Interferometry of Sentinel-1 Data: A Study of Vienna City, Austria. International Journal of Innovative Technology and Interdisciplinary Sciences Vol. 4 No. 1, 604–611 Pages. <u>https:// doi.org/10.15157/IJITIS.2021.4.1.604-611</u>
- Beniston, M., Farinotti, D., Stoffel, M., Andrassen, L. M., Coppola, E., Eckert, N., et al. (2018). The European mountain cryosphere: a review of its current state, trends, and future challenges. *The Cryosphere* 12, 759–794. <u>https://doi.org/10.5194/tc-12-759-2018</u>
- Benz, S. A., Bayer, P., Winkler, G., and Blum, P. (2018). Recent trends of groundwater temperatures in Austria. *Hydrol. Earth Syst. Sci.* 22, 3143–3154. <u>https://doi.org/10.5194/hess-22-3143-</u> 2018
- Berg, P., Moseley, C., and Haerter, J. O. (2013). Strong increase in convective precipitation in response to higher temperatures. *Nature Geosci* 6, 181–185. <u>https://doi.org/10.1038/ngeo1731</u>
- Bergamini, A., Ungricht, S., and Hofmann, H. (2009). An elevational shift of cryophilous bryophytes in the last century – an effect of climate warming? *Diversity and Distributions* 15, 871– 879. <u>https://doi.org/10.1111/j.1472-4642.2009.00595.x</u>
- Biella, P., Bogliani, G., Cornalba, M., Manino, A., Neumayer, J., Porporato, M., et al. (2017). Distribution patterns of the cold adapted bumblebee Bombus alpinus in the Alps and hints of an uphill shift (Insecta: Hymenoptera: Apidae). *J Insect Conserv* 21, 357–366. https://doi.org/10.1007/s10841-017-9983-1
- Bison, M., Yoccoz, N. G., Carlson, B. Z., Klein, G., Laigle, I., Van Reeth, C., et al. (2021). Earlier Snowmelt Advances Breeding Phenology of the Common Frog (Rana temporaria) but Increases the Risk of Frost Exposure and Wetland Drying. *Front. Ecol. Evol.* 9, 645585. https://doi.org/10.3389/fevo.2021.645585
- Blaschke, A. P., Merz, R., Parajka, J., Salinas, J., and Blöschl, G. (2011). Auswirkungen des Klimawandels auf das Wasserdargebot von Grund- und Oberflächenwasser. Österreichische Wasser- und Abfallwirtschaft 63, 31–41. <u>https://doi.org/10.1007/ s00506-010-0273-3</u>
- Blaydes, H., Potts, S. G., Whyatt, J. D., and Armstrong, A. (2021). Opportunities to enhance pollinator biodiversity in solar parks. *Renewable and Sustainable Energy Reviews* 145, 111065. <u>https://doi.org/10.1016/j.rser.2021.111065</u>
- Bloomfield, J. P., and Marchant, B. P. (2013). Analysis of groundwater drought building on the standardised precipitation index approach. *Hydrol. Earth Syst. Sci.* 17, 4769–4787. <u>https://doi. org/10.5194/hess-17-4769-2013</u>
- Blöschl, G., Bertola, M., Breinl, K., Haslinger, K., Höfler, A., Homann, M., et al. (2023). How do circulation patterns affect

changes and the spatial concurrence/superposition of floods? The Wetrax+ project.

- Blöschl, G., Hall, J., Viglione, A., Perdigão, R. A. P., Parajka, J., Merz, B., et al. (2019). Changing climate both increases and decreases European river floods. *Nature* 573, 108–111. <u>https://doi.org/10.1038/s41586-019-1495-6</u>
- Blöschl, G., Parajka, J., Blaschke, A. P., Hofstätter, M., Haslinger, K., and Schöner, W. (2017). Klimawandel in der Wasserwirtschaft – Follow-up zur ZAMG/TU-Wien Studie (2011) Anpassungsstrategien an den Klimawandel für Österreichs Wasserwirtschaft. Vienna: Bundesministerium für Land- und Forstwirtschaft, Umwelt und Wasserwirtschaft. Available at: <u>https:// info.bml.gv.at/dam/jcr:e137df9c-4431-458b-a84e-e97c313856</u> 48/Klimawandel\_2017\_final29112017fv4.pdf
- Blöschl, G., Schöner, W., Kroiß, H., P. Blaschke, A., Böhm, R., Haslinger, K., et al. (2011). Anpassungsstrategien an den Klimawandel für Österreichs Wasserwirtschaft – Ziele und Schlussfolgerungen der Studie für Bund und Länder. Österreichische Wasser- und Abfallwirtschaft 63, 1–10. <u>https://doi.org/10.1007/s00506-010-0274-2</u>
- Blöschl, G., Sivapalan, M., Wagener, T., Viglione, A., and Savenije, H. (2013). Runoff Prediction in Ungauged Basins: Synthesis across Processes, Places and Scales. Cambridge University Press. <u>https://doi.org/10.1017/CBO9781139235761</u>
- Blowes, S. A., Supp, S. R., Antão, L. H., Bates, A., Bruelheide, H., Chase, J. M., et al. (2019). The geography of biodiversity change in marine and terrestrial assemblages. *Science* 366, 339–345. <u>https://doi.org/10.1126/science.aaw1620</u>
- BMLRT (2021). Wasserschatz Österreichs. Grundlagen für nachhaltige Nutzungen des Grundwassers. Wien, Österreich: Bundesministerium für Landwirtschaft, Regionen und Tourismus. Available at: <u>https://info.bml.gv.at/dam/jcr:75a703dd-9c25-452a-ac06-5240abbd118a/Bericht\_Wasserschatz.pdf</u>
- BMLRT (2022). Moorstrategie Österreich 2030+. Wien: Bundesministerium für Landwirtschaft, Regionen und Tourismus. Available at: <u>https://info.bml.gv.at/service/publikationen/wasser/moorstrategie-oesterreich-2030.html</u> (Accessed September 21, 2023).
- BMNT (2019). Grüner Bericht 2019. Die Situation der österreichischen Land- und Forstwirtschaft. Wien: Bundesministerium für Nachhaltigkeit und Tourismus. Available at: <u>https://gruenerbericht.at/cm4/jdownload/download/2-gr-bericht-terreich/</u> 2007-gb2019
- Bondesan, A., and Francese, R. G. (2023). The climate-driven disaster of the Marmolada Glacier (Italy). *Geomorphology* 431, 108687. <u>https://doi.org/10.1016/j.geomorph.2023.108687</u>
- Bonelli, S., Cerrato, C., Barbero, F., Boiani, M. V., Buffa, G., Casacci, L. P., et al. (2021). Changes in Alpine Butterfly Communities during the Last 40 Years. *Insects* 13, 43. <u>https://doi.org/10.3390/</u> insects13010043
- Borgwardt, F., Unfer, G., Auer, S., Waldner, K., El-Matbouli, M., and Bechter, T. (2020). Direct and Indirect Climate Change Impacts on Brown Trout in Central Europe: How Thermal Regimes Reinforce Physiological Stress and Support the Emergence of Diseases. *Front. Environ. Sci.* 8, 59. <u>https://doi.org/10.3389/fenvs.</u> 2020.00059
- Bosela, M., Tumajer, J., Cienciala, E., Dobor, L., Kulla, L., Marčiš, P., et al. (2021). Climate warming induced synchronous growth decline in Norway spruce populations across biogeographical gradients since 2000. *Science of The Total Environment* 752, 141794. https://doi.org/10.1016/j.scitotenv.2020.141794

- Bozzoli, M., Crespi, A., Matiu, M., Majone, B., Giovannini, L., Zardi, D., et al. (2024). Long-term snowfall trends and variability in the Alps. *Intl Journal of Climatology* 44, 4571–4591. <u>https:// doi.org/10.1002/joc.8597</u>
- Brambilla, M., Rubolini, D., Appukuttan, O., Calvi, G., Karger, D. N., Kmecl, P., et al. (2022). Identifying climate refugia for high-elevation Alpine birds under current climate warming predictions. *Global Change Biology* 28, 4276–4291. <u>https://doi. org/10.1111/gcb.16187</u>
- Brighenti, S., Tolotti, M., Bruno, M. C., Wharton, G., Pusch, M. T., and Bertoldi, W. (2019). Ecosystem shifts in Alpine streams under glacier retreat and rock glacier thaw: A review. *Science of The Total Environment* 675, 542–559. <u>https://doi.org/10.1016/j. scitotenv.2019.04.221</u>
- Brooks, H. E. (2013). Severe thunderstorms and climate change. Atmospheric Research 123, 129–138. <u>https://doi.org/10.1016/j.atmosres.2012.04.002</u>
- Brückl, E., Brunner, F. K., Lang, E., Mertl, S., Müller, M., and Stary, U. (2013). The Gradenbach Observatory—monitoring deep-seated gravitational slope deformation by geodetic, hydrological, and seismological methods. *Landslides* 10, 815–829. <u>https://doi.org/10.1007/s10346-013-0417-1</u>
- Brunetti, M., Lentini, G., Maugeri, M., Nanni, T., Auer, I., Böhm, R., et al. (2009). Climate variability and change in the Greater Alpine Region over the last two centuries based on multi-variable analysis. *Int. J. Climatol.* 29, 2197–2225. <u>https://doi.org/10. 1002/joc.1857</u>
- Bruno, D., Belmar, O., Maire, A., Morel, A., Dumont, B., and Datry, T. (2019). Structural and functional responses of invertebrate communities to climate change and flow regulation in alpine catchments. *Global Change Biology* 25, 1612–1628. <u>https://doi. org/10.1111/gcb.14581</u>
- Buckel, J., Otto, J. C., Prasicek, G., and Keuschnig, M. (2018). Glacial lakes in Austria – Distribution and formation since the Little Ice Age. *Global and Planetary Change* 164, 39–51. <u>https://doi. org/10.1016/j.gloplacha.2018.03.003</u>
- Büntgen, U., Liebhold, A., Nievergelt, D., Wermelinger, B., Roques, A., Reinig, F., et al. (2020). Return of the moth: rethinking the effect of climate on insect outbreaks. *Oecologia* 192, 543–552. <u>https://doi.org/10.1007/s00442-019-04585-9</u>
- Büntgen, U., Urban, O., Krusic, P. J., Rybníček, M., Kolář, T., Kyncl, T., et al. (2021). Recent European drought extremes beyond Common Era background variability. *Nature Geoscience* 14, 190–196. <u>https://doi.org/10.1038/s41561-021-00698-0</u>
- Buras, A., Rammig, A., and Zang, C. S. (2020). Quantifying impacts of the 2018 drought on European ecosystems in comparison to 2003. *Biogeosciences* 17, 1655–1672. <u>https://doi.org/10.5194/bg-17-1655-2020</u>
- Caesar, L., Rahmstorf, S., Robinson, A., Feulner, G., and Saba, V. (2018). Observed fingerprint of a weakening Atlantic Ocean overturning circulation. *Nature* 556, 191–196. <u>https://doi.org/ 10.1038/s41586-018-0006-5</u>
- Caminade, C., Medlock, J. M., Ducheyne, E., McIntyre, K. M., Leach, S., Baylis, M., et al. (2012). Suitability of European climate for the Asian tiger mosquito *Aedes albopictus*: recent trends and future scenarios. *J. R. Soc. Interface*. 9, 2708–2717. https://doi.org/10.1098/rsif.2012.0138
- Chan, S. C., Kendon, E. J., Fowler, H. J., Kahraman, A., Crook, J., Ban, N., et al. (2023). Large-scale dynamics moderate impact-relevant changes to organised convective storms. *Commun Earth Environ* 4, 8. <u>https://doi.org/10.1038/s43247-022-00669-2</u>

- Chaussard, E., Havazli, E., Fattahi, H., Cabral-Cano, E., and Solano-Rojas, D. (2021). Over a Century of Sinking in Mexico City: No Hope for Significant Elevation and Storage Capacity Recovery. JGR Solid Earth 126, e2020JB020648. <u>https://doi.org/10.1029/2020JB020648</u>
- Chen, D., Rojas, M., Samset, B. H., Cobb, K., Diongue Niang, A., Edwards, P., et al. (2021). "Framing, Context, and Methods", in *Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*, eds. V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press), 147–286. <u>https://doi.org/10.1017/9781009157896.003</u>
- Chen, I.-C., Hill, J. K., Ohlemüller, R., Roy, D. B., and Thomas, C. D. (2011). Rapid Range Shifts of Species Associated with High Levels of Climate Warming. *Science* 333, 1024–1026. <u>https://doi.org/10.1126/science.1206432</u>
- Chiarle, M., Geertsema, M., Mortara, G., and Clague, J. J. (2021). Relations between climate change and mass movement: Perspectives from the Canadian Cordillera and the European Alps. *Global and Planetary Change* 202, 103499. <u>https://doi.org/ 10.1016/j.gloplacha.2021.103499</u>
- Chiarle, M., Viani, C., Mortara, G., Deline, P., Tamburini, A., and Nigrelli, G. (2022). Large glacier failures in the Italian Alps over the last 90 years. *Geografia Fisica e Dinamica Quaternaria* 45, 19–40. <u>https://doi.org/10.4461/GFDQ.2022.45.2</u>
- Chimani, B., Ganekind, M., and Olefs, M. (2021). Temperaturentwicklung in Österreich im globalen Kontext. CCCA Factsheet #35. Available at: <u>https://ccca.ac.at/fileadmin/00\_DokumenteHauptmenue/02\_Klimawissen/FactSheets/35\_temperaturentwicklung\_in\_oesterreich\_202110.pdf</u>
- Chimani, B., Heinrich, G., Hofstätter, M., Kerschbaumer, M., Kienberger, S., Leuprecht, A., et al. (2016). ÖKS15 – Klimaszenarien für Österreich. Daten, Methoden und Klimaanalyse. Wien. Availabe at: <u>https://www.bmk.gv.at/dam/jcr:7fd75e22-1b88-415f-a4a8-6ea8aa51d575/OEKS15\_Endbericht\_kleiner.pdf</u>
- Chimani, B., Koch, R., Tilg, A.-M., Ganekind, M., Günther, D., Rohrböck, A., et al. (2024). Bestimmung der aktuellen Klimaerwärmung für Österreich. CCCA Factsheet #50. Available at: <u>https://ccca.ac.at/fileadmin/00\_DokumenteHauptmenue/02\_ Klimawissen/FactSheets/50\_klimaaerwaermung\_oesterr\_ 202412.pdf</u>
- Chimani, B., Matulla, C., Eitzinger, J., Hiebl, J., Hofstätter, M., Kubu, G., et al. (2019). GUIDELINE zur Nutzung der ÖKS15-Klimawandelsimulationen sowie der entsprechenden gegitterten Beobachtungsdatensätze. Climate Change Centre Austria. Available at: <u>https://ccca.ac.at/fileadmin/00\_DokumenteHauptmenue/02\_Klimawissen/Guideline\_STARC\_Impact\_2018.pdf</u>
- Chirichella, R., Stephens, P. A., Mason, T. H. E., and Apollonio, M. (2021). Contrasting Effects of Climate Change on Alpine Chamois. J Wildl Manag 85, 109–120. <u>https://doi.org/10.1002/jwmg.21962</u>
- Christensen, J. H., and Christensen, O. B. (2007). A summary of the PRUDENCE model projections of changes in European climate by the end of this century. *Climatic Change* 81, 7–30. <u>https://doi.org/10.1007/s10584-006-9210-7</u>
- Clarke, D. C., and Richardson, M. (2021). The Benefits of Continuous Local Regression for Quantifying Global Warming. *Earth and Space Science* 8, e2020EA001082. <u>https://doi.org/ 10.1029/2020EA001082</u>

- Climate Action Tracker (2024). Warming Projections Global Update November 2024. Available at: <u>https://climateactiontracker.org/documents/1277/CAT\_2024-11-14\_GlobalUpdate\_</u> <u>COP29.pdf</u>
- Coe, J. A. (2012). Regional moisture balance control of landslide motion: Implications for landslide forecasting in a changing climate. *Geology* 40, 323–326. <u>https://doi.org/10.1130/G32897.1</u>
- Collados-Lara, A.-J., Pulido-Velazquez, D., Mateos, R. M., and Ezquerro, P. (2020). Potential Impacts of Future Climate Change Scenarios on Ground Subsidence. *Water* 12, 219. <u>https://doi.org/10.3390/w12010219</u>
- Comte, L., and Grenouillet, G. (2013). Do stream fish track climate change? Assessing distribution shifts in recent decades. *Ecography* 36, 1236–1246. <u>https://doi.org/10.1111/j.1600-0587.</u> 2013.00282.x
- Coppola, E., Sobolowski, S., Pichelli, E., Raffaele, F., Ahrens, B., Anders, I., et al. (2020). A first-of-its-kind multi-model convection permitting ensemble for investigating convective phenomena over Europe and the Mediterranean. *Climate Dynamics* 55, 3–34. <u>https://doi.org/10.1007/s00382-018-4521-8</u>
- Corti, T., Muccione, V., Köllner-Heck, P., Bresch, D., and Seneviratne, S. I. (2009). Simulating past droughts and associated building damages in France. *Hydrol. Earth Syst. Sci.* 13, 1739– 1747. <u>https://doi.org/10.5194/hess-13-1739-2009</u>
- Crabot, J., Polášek, M., Launay, B., Pařil, P., and Datry, T. (2021). Drying in newly intermittent rivers leads to higher variability of invertebrate communities. *Freshwater Biology* 66, 730–744. <u>https://doi.org/10.1111/fwb.13673</u>
- Crozier, M. J. (2010). Deciphering the effect of climate change on landslide activity: A review. *Geomorphology* 124, 260–267. <u>https://doi.org/10.1016/j.geomorph.2010.04.009</u>
- Cruden, D. M., and Varnes, D. J. (1996). Landslide Types and Processes. Special Report – National Research Council, Transportation Research Board 247, 65.
- Cutter, S. L. (2018). Compound, Cascading, or Complex Disasters: What's in a Name? *Environment: Science and Policy for Sustainable Development* 60, 16–25. <u>https://doi.org/10.1080/00139157.</u> 2018.1517518
- Dai, X., Schneider-Muntau, B., Krenn, J., Zangerl, C., and Fellin, W. (2023). Mechanisms for the Formation of an Exceptionally Gently Inclined Basal Shear Zone of a Landslide in Glacial Sediments—The Ludoialm Case Study. *Applied Sciences* 13, 6837. <u>https://doi.org/10.3390/app13116837</u>
- Dainese, M., Aikio, S., Hulme, P. E., Bertolli, A., Prosser, F., and Marini, L. (2017). Human disturbance and upward expansion of plants in a warming climate. *Nature Clim Change* 7, 577–580. <u>https://doi.org/10.1038/nclimate3337</u>
- Davini, P., and D'Andrea, F. (2020). From CMIP3 to CMIP6: Northern Hemisphere Atmospheric Blocking Simulation in Present and Future Climate. *Journal of Climate* 33, 10021–10038. https://doi.org/10.1175/JCLI-D-19-0862.1
- De Vries, I. E., Sippel, S., Pendergrass, A. G., and Knutti, R. (2023). Robust global detection of forced changes in mean and extreme precipitation despite observational disagreement on the magnitude of change. *Earth Syst. Dynam.* 14, 81–100. <u>https://doi.org/10.5194/esd-14-81-2023</u>
- Deléglise, C., Meisser, M., Mosimann, E., Spiegelberger, T., Signarbieux, C., Jeangros, B., et al. (2015). Drought-induced shifts in plants traits, yields and nutritive value under realistic grazing and mowing managements in a mountain grassland. Agri-

culture, Ecosystems & Environment 213, 94-104. https://doi. org/10.1016/j.agee.2015.07.020

- Deprez, A., Leadley, P., Dooley, K., Williamson, P., Cramer, W., Gattuso, J.-P., et al. (2024). Sustainability limits needed for CO<sub>2</sub> removal. *Science* 383, 484–486. <u>https://doi.org/10.1126/science.</u> adj6171
- Descombes, P., Pradervand, J.-N., Golay, J., Guisan, A., and Pellissier, L. (2016). Simulated shifts in trophic niche breadth modulate range loss of alpine butterflies under climate change. *Ecography* 39, 796–804. <u>https://doi.org/10.1111/ecog.01557</u>
- Dietrich, A., and Krautblatter, M. (2017). Evidence for enhanced debris-flow activity in the Northern Calcareous Alps since the 1980s (Plansee, Austria). *Geomorphology* 287, 144–158. <u>https:// doi.org/10.1016/j.geomorph.2016.01.013</u>
- Dirnböck, T., Essl, F., and Rabitsch, W. (2011). Disproportional risk for habitat loss of high-altitude endemic species under climate change. *Global Change Biology* 17, 990–996. <u>https://doi.org/10.1111/j.1365-2486.2010.02266.x</u>
- Doblas-Reyes, F. J., Sörensson, A. A., Almazroui, M., Dosio, A., Gutowski, W. J., Haarsma, R., et al. (2021). "Linking Global to Regional Climate Change", in *Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth* Assessment Report of the Intergovernmental Panel on Climate Change, eds. V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press), 1363–1512. https://doi.org/10.1017/9781009157896.012
- Dorrington, J., Strommen, K., Fabiano, F., and Molteni, F. (2022). CMIP6 Models Trend Toward Less Persistent European Blocking Regimes in a Warming Climate. *Geophysical Research Letters* 49, e2022GL100811. <u>https://doi.org/10.1029/2022GL100811</u>
- Dorts, J., Grenouillet, G., Douxfils, J., Mandiki, S. N. M., Milla, S., Silvestre, F., et al. (2012). Evidence that elevated water temperature affects the reproductive physiology of the European bullhead Cottus gobio. *Fish Physiol Biochem* 38, 389–399. <u>https:// doi.org/10.1007/s10695-011-9515-y</u>
- Dosio, A. (2016). Projections of climate change indices of temperature and precipitation from an ensemble of bias-adjusted high-resolution EURO-CORDEX regional climate models. *JGR Atmospheres* 121, 5488–5511. <u>https://doi.org/10.1002/2015</u> <u>JD024411</u>
- Draganits, E., Weißl, M., Zámolyi, A., and Doneus, M. (2022). "Lake Neusiedl Area: A Particular Lakescape at the Boundary Between Alps and Pannonian Basin", in *Landscapes and Landforms of Austria*, ed. C. Embleton-Hamann (Cham: Springer International Publishing), 207–222. <u>https://doi.org/10.1007/978-3-030-92815-5\_13</u>
- Duethmann, D., and Blöschl, G. (2018). Why has catchment evaporation increased in the past 40 years? A data-based study in Austria. *Hydrol. Earth Syst. Sci.* 22, 5143–5158. <u>https://doi.org/</u> 10.5194/hess-22-5143-2018
- Duethmann, D., Blöschl, G., and Parajka, J. (2020). Why does a conceptual hydrological model fail to correctly predict discharge changes in response to climate change? *Hydrology and Earth System Sciences* 24, 3493–3511. <u>https://doi.org/10.5194/hess-24-3493-2020</u>
- Dullinger, I., Gattringer, A., Wessely, J., Moser, D., Plutzar, C., Willner, W., et al. (2020). A socio-ecological model for predicting impacts of land-use and climate change on regional plant diversity in the Austrian Alps. *Global Change Biology* 26, 2336–2352. https://doi.org/10.1111/gcb.14977

- Dullinger, I., Wessely, J., Bossdorf, O., Dawson, W., Essl, F., Gattringer, A., et al. (2017). Climate change will increase the naturalization risk from garden plants in Europe: Naturalization risk from garden plants. *Global Ecol. Biogeogr.* 26, 43–53. <u>https://doi. org/10.1111/geb.12512</u>
- Dullinger, S., Dirnböck, T., and Grabherr, G. (2004). Modelling climate change-driven treeline shifts: relative effects of temperature increase, dispersal and invasibility. *Journal of Ecology* 92, 241–252. <u>https://doi.org/10.1111/j.0022-0477.2004.00872.x</u>
- Duvillard, P.-A., Ravanel, L., Marcer, M., and Schoeneich, P. (2019). Recent evolution of damage to infrastructure on permafrost in the French Alps. *Reg Environ Change* 19, 1281–1293. <u>https:// doi.org/10.1007/s10113-019-01465-z</u>
- Eder, S., Poscher, G., and Prager, C. (2006). Risk Analyses and Risk Management – Slope Instabilities in Alpine Environments., in *Geohazards*, (Lillehammer, Norway). Available at: <u>https://dc. engconfintl.org/geohazards/15/</u> (Accessed November 5, 2024).
- Engelhardt, E. K., Biber, M. F., Dolek, M., Fartmann, T., Hochkirch, A., Leidinger, J., et al. (2022). Consistent signals of a warming climate in occupancy changes of three insect taxa over 40 years in central Europe. *Global Change Biology* 28, 3998–4012. <u>https:// doi.org/10.1111/gcb.16200</u>
- Engler, R., Randin, C. F., Thuiller, W., Dullinger, S., Zimmermann, N. E., Araújo, M. B., et al. (2011). 21st century climate change threatens mountain flora unequally across Europe: CLIMATE CHANGE IMPACTS ON MOUNTAIN FLORAE. *Global Change Biology* 17, 2330–2341. <u>https://doi.org/10.1111/j.1365-2486.2010.02393.x</u>
- Erhart, E., and Hartl, W. (2010). "Compost Use in Organic Farming", in *Genetic Engineering, Biofertilisation, Soil Quality and Organic Farming*, ed. E. Lichtfouse (Dordrecht: Springer Netherlands), 311–345. <u>https://doi.org/10.1007/978-90-481-8741-6\_11</u>
- Essl, F., Dullinger, S., Moser, D., Rabitsch, W., and Kleinbauer, I. (2012). Vulnerability of mires under climate change: implications for nature conservation and climate change adaptation. *Biodivers Conserv* 21, 655–669. <u>https://doi.org/10.1007/s10531-011-0206-x</u>
- European Commission Directorate General for Environment (2021). *EU biodiversity strategy for 2030: bringing nature back into our lives*. LU: Publications Office. Available at: <u>https://data.europa.eu/doi/10.2779/677548</u> (Accessed May 8, 2024).
- European Environment Agency (2016). Elevation map of Europe (1 km grid). Available at: <u>https://sdi.eea.europa.eu/catalogue/</u> <u>datahub/api/records/cafa8ccf-d4d1-4b03-bc4e-681aae516</u> 730/formatters/xsl-view?output=pdf&language=eng&approved=true
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., et al. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geosci. Model Dev.* 9, 1937–1958. <u>https://doi. org/10.5194/gmd-9-1937-2016</u>
- Fabiano, F., Meccia, V. L., Davini, P., Ghinassi, P., and Corti, S. (2021). A regime view of future atmospheric circulation changes in northern mid-latitudes. *Weather Clim. Dynam.* 2, 163–180. <u>https://doi.org/10.5194/wcd-2-163-2021</u>
- Federal Ministry of Social Affairs, Health, Care and Consumer Protection (2021). Fourth review and appraisal of the implementation of the Madrid International Plan of Action on Ageing 2002 and the Regional Implementation Strategy (MIPAA / RIS) 2018 – 2022. Vienna, Austria: Federal Ministry of Social Affairs,

Health, Care and Consumer Protection. Available at: <u>https://unece.org/sites/default/files/2022-01/mipaa20-report-austria.pdf</u>

- Ferrarini, A., Alatalo, J. M., and Gustin, M. (2017). Climate change will seriously impact bird species dwelling above the treeline: A prospective study for the Italian Alps. Science of The Total Environment 590–591, 686–694. <u>https://doi.org/10.1016/j.scitotenv.2017.03.027</u>
- Ferreira Correa, L., Folini, D., Chtirkova, B., and Wild, M. (2024). Decadal trends in observed surface solar radiation and their causes in Brazil in the first two decades of the 21st century. <u>https://doi.org/10.5194/egusphere-2024-509</u>
- Feser, F., Barcikowska, M., Krueger, O., Schenk, F., Weisse, R., and Xia, L. (2015). Storminess over the North Atlantic and Northwestern Europe – A Review. Q. J. R. Meteorol. Soc. 141, 350–382. https://doi.org/10.1002/qj.2364
- Fey, C., and Krainer, K. (2020). Analyses of UAV and GNSS based flow velocity variations of the rock glacier Lazaun (Ötztal Alps, South Tyrol, Italy). *Geomorphology* 365, 107261. <u>https://doi. org/10.1016/j.geomorph.2020.107261</u>
- Fischer, A., Schwaizer, G., Seiser, B., Helfricht, K., and Stocker-Waldhuber, M. (2021). High-resolution inventory to capture glacier disintegration in the Austrian Silvretta. *The Cryosphere* 15, 4637–4654. <u>https://doi.org/10.5194/tc-15-4637-2021</u>
- Fischer, A., Seiser, B., Stocker Waldhuber, M., Mitterer, C., and Abermann, J. (2015). Tracing glacier changes in Austria from the Little Ice Age to the present using a lidar-based high-resolution glacier inventory in Austria. *The Cryosphere* 9, 753–766. <u>https://doi.org/10.5194/tc-9-753-2015</u>
- Fischer, E. M., and Knutti, R. (2016). Observed heavy precipitation increase confirms theory and early models. *Nature Clim Change* 6, 986–991. <u>https://doi.org/10.1038/nclimate3110</u>
- Fischer, E. M., Rajczak, J., and Schär, C. (2012). Changes in European summer temperature variability revisited. *Geophysical Research Letters* 39, 2012GL052730. <u>https://doi.org/10.1029/2012GL052730</u>
- Fleischer, F., Haas, F., Piermattei, L., Pfeiffer, M., Heckmann, T., Altmann, M., et al. (2021). Multi-decadal (1953–2017) rock glacier kinematics analysed by high-resolution topographic data in the upper Kaunertal, Austria. *The Cryosphere* 15, 5345–5369. <u>https://doi.org/10.5194/tc-15-5345-2021</u>
- Fliessbach, A., Schmid, H., and Niggli, U. (2008). Die Vorteile des Öko-Landbaus für das Klima. Ökologie & Landbau, 17–19.
- Flórián, N., Ladányi, M., Ittzés, A., Kröel-Dulay, G., Ónodi, G., Mucsi, M., et al. (2019). Effects of single and repeated drought on soil microarthropods in a semi-arid ecosystem depend more on timing and duration than drought severity. *PLoS ONE* 14, e0219975. <u>https://doi.org/10.1371/journal.pone.0219975</u>
- Formayer, H., and Fritz, A. (2016). Temperature dependency of hourly precipitation intensities – surface versus cloud layer temperature. *International Journal of Climatology* 37, 1–10. <u>https:// doi.org/10.1002/joc.4678</u>
- Fowler, H. J., Lenderink, G., Prein, A. F., Westra, S., Allan, R. P., Ban, N., et al. (2021). Anthropogenic intensification of short-duration rainfall extremes. *Nat Rev Earth Environ* 2, 107–122. <u>https://doi.org/10.1038/s43017-020-00128-6</u>
- Francon, L., Corona, C., Till-Bottraud, I., Carlson, B. Z., and Stoffel, M. (2020). Some (do not) like it hot: shrub growth is hampered by heat and drought at the alpine treeline in recent decades. *American J of Botany* 107, 607–617. <u>https://doi.org/10.1002/ ajb2.1459</u>

- Frei, P., Kotlarski, S., Liniger, M. A., and Schär, C. (2018). Future snowfall in the Alps: projections based on the EURO-CORDEX regional climate models. *The Cryosphere* 12, 1–24. <u>https://doi.org/10.5194/tc-12-1-2018</u>
- Fuchs, S., Karagiorgos, K., Keiler, M., Papathoma-Köhle, M., and Nyberg, L. (2024). The ambiguity in IPCC's risk diagram raises explanatory challenges. *Nat Hazards* 120, 12559–12564. <u>https:// doi.org/10.1007/s11069-024-06643-9</u>
- Fuchs, S., Wenk, M., and Keiler, M. (2022). "Geomorphic Hazards in Austria", in *Landscapes and Landforms of Austria*, ed. C. Embleton-Hamann (Cham: Springer International Publishing), 105–117. https://doi.org/10.1007/978-3-030-92815-5\_6
- Gallé, L. (2017). Climate change impoverishes and homogenizes ants' community structure: a long term study. *Community Ecology* 18, 128–136. <u>https://doi.org/10.1556/168.2017.18.2.2</u>
- Gariano, S. L., and Guzzetti, F. (2016). Landslides in a changing climate. *Earth-Science Reviews* 162, 227–252. <u>https://doi.org/ 10.1016/j.earscirev.2016.08.011</u>
- Garschagen, M., Adler, C., Crate, S., Jacot Des Combes, H., Glavovic, B. C., Harper, S., et al. (2019). "Cross-Chapter Box 2 | Key Concepts of Risk, Adaptation, Resilience and Transformation", in *IPCC Special Report on the Ocean and Cryosphere in a Changing Climate*, eds. H.-O. Pörtner, D. C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press), 87–90. Available at: https://doi.org/10.1017/9781009157964.003
- Gaube, V., Schneider, S., Thaler, T., Borsky, S., Kirchner, M., Loibl, W., et al. (2024). "Kapitel 3. Sozioökonomische und klimatische Treiber der Änderung der Landnutzung in Österreich", in APCC Special Report: Landnutzung und Klimawandel in Österreich, eds. R. Jandl, U. Tappeiner, C. B. Foldal, and K. Erb (Berlin, Heidelberg: Springer Spektrum), 163–216. <u>https://doi.org/10.1007/978-3-662-67864-0\_5</u>
- Gauld, J. G., Silva, J. P., Atkinson, P. W., Record, P., Acácio, M., Arkumarev, V., et al. (2022). Hotspots in the grid: Avian sensitivity and vulnerability to collision risk from energy infrastructure interactions in Europe and North Africa. *Journal of Applied Ecolo*gy 59, 1496–1512. <u>https://doi.org/10.1111/1365-2664.14160</u>
- Gehrig-Fasel, J., Guisan, A., and Zimmermann, N. E. (2007). Tree line shifts in the Swiss Alps: Climate change or land abandonment? *Journal of Vegetation Science* 18, 571–582. <u>https://doi.org/ 10.1111/j.1654-1103.2007.tb02571.x</u>
- GeoSphere Austria (2020). SPARTACUS v2.1 Tagesdaten. <u>https://</u> <u>doi.org/10.60669/DNSV-AY89</u>
- GeoSphere Austria (2023a). Hitzewellen: länger und häufiger. Available at: <u>https://www.zamg.ac.at/cms/de/klima/news/hitze-wellen-laenger-und-haeufiger</u>
- GeoSphere Austria (2023b). Siebentwärmster Sommer der Messgeschichte. Available at: <u>https://www.zamg.ac.at/cms/de/klima/</u> <u>news/siebentwaermster-sommer-der-messgeschichte</u>
- Geppert, C., La Bella, G., Boscutti, F., Sanna, F., Marangoni, F., and Marini, L. (2021). Effects of temperature and plant diversity on orthopterans and leafhoppers in calcareous dry grasslands. *J Insect Conserv* 25, 287–296. <u>https://doi.org/10.1007/s10841-021-</u> 00300-3
- Gilbert, A., Gagliardini, O., Vincent, C., and Wagnon, P. (2014). A 3-D thermal regime model suitable for cold accumulation zones of polythermal mountain glaciers. *JGR Earth Surface* 119, 1876–1893. <u>https://doi.org/10.1002/2014JF003199</u>
- Gilgado, J. D., Rusterholz, H., and Baur, B. (2022). Millipedes step up: species extend their upper elevational limit in the Alps in re-

sponse to climate warming. *Insect Conserv Diversity* 15, 61–72. https://doi.org/10.1111/icad.12535

- Gilgen, A. K., Signarbieux, C., Feller, U., and Buchmann, N. (2010). Competitive advantage of Rumex obtusifolius L. might increase in intensively managed temperate grasslands under drier climate. Agriculture, Ecosystems & Environment 135, 15–23. https://doi.org/10.1016/j.agee.2009.08.004
- Giorgi, F., Torma, C., Coppola, E., Ban, N., Schär, C., and Somot, S. (2016). Enhanced summer convective rainfall at alpine high elevations in response to climate warming. *Nature Geoscience* 9, 584–589. <u>https://doi.org/10.1038/ngeo2761</u>
- Gischig, V. S., Moore, J. R., Evans, K. F., Amann, F., and Loew, S. (2011a). Thermomechanical forcing of deep rock slope deformation: 1. Conceptual study of a simplified slope. *J. Geophys. Res.* 116, F04010. <u>https://doi.org/10.1029/2011JF002006</u>
- Gischig, V. S., Moore, J. R., Evans, K. F., Amann, F., and Loew, S. (2011b). Thermomechanical forcing of deep rock slope deformation: 2. The Randa rock slope instability. *J. Geophys. Res.* 116, F04011. <u>https://doi.org/10.1029/2011JF002007</u>
- Giuntoli, J., Barredo, J. I., Avitabile, V., Camia, A., Cazzaniga, N. E., Grassi, G., et al. (2022). The quest for sustainable forest bioenergy: win-win solutions for climate and biodiversity. *Renewable and Sustainable Energy Reviews* 159, 112180. <u>https://doi. org/10.1016/j.rser.2022.112180</u>
- Glade, T. ed. (2020). ExtremA 2019. Aktueller Wissensstand zu Extremereignissen alpiner Naturgefahren in Österreich. Göttingen: Vienna University Press. Available at: <u>https://library.oapen.org/ handle/20.500.12657/43512</u>
- Glantz, P., Fawole, O. G., Ström, J., Wild, M., and Noone, K. J. (2022). Unmasking the Effects of Aerosols on Greenhouse Warming Over Europe. JGR Atmospheres 127, e2021JD035889. <u>https://doi.org/10.1029/2021JD035889</u>
- Global Terrestrial Network for Permafrost (2023). GTN-P database. Available at: <u>http://gtnpdatabase.org/boreholes</u> (Accessed May 7, 2024).
- Grabherr, G., Koch, G., Kirchmeir, H., and Reiter, K. (1998). Hemerobie österreichischer Waldökosysteme. Veröffentlichungen des Österreichischen MaB-Programms 17.
- Greilinger, M., Orlik, A., Tilg, A., Schellander, H., Pistotnik, G., Lehner, S., et al. (2024). Klimatologische Analyse betreffend des Niederschlagereignisses vom 12. bis 16. September 2024 in Österreich. Available at: <u>https://infothek.bmimi.gv.at/septem ber-2024-brachte-rekordniederschlaege-hoechsttemperaturen/</u>
- Griebler, C., Brielmann, H., Haberer, C. M., Kaschuba, S., Kellermann, C., Stumpp, C., et al. (2016). Potential impacts of geothermal energy use and storage of heat on groundwater quality, biodiversity, and ecosystem processes. *Environmental Earth Sciences* 75, 1391. <u>https://doi.org/10.1007/s12665-016-6207-z</u>
- Gulev, S. K., Thorne, P. W., Ahn, J., Dentener, F. J., Domingues, C. M., Gerland, S., et al. (2021). "Changing State of the Climate System", in *Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*, eds. V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press), 287–422. <u>https://doi.org/10.1017/9781009157896.004</u>
- Haas, J. C., and Birk, S. (2017). Characterizing the spatiotemporal variability of groundwater levels of alluvial aquifers in different settings using drought indices. *Hydrol. Earth Syst. Sci.* 21, 2421–2448. https://doi.org/10.5194/hess-21-2421-2017

- Haas, J. C., and Birk, S. (2019). Trends in Austrian groundwater Climate or human impact? *Journal of Hydrology: Regional Studies* 22, 100597. <u>https://doi.org/10.1016/j.ejrh.2019.100597</u>
- Haase, P., Pilotto, F., Li, F., Sundermann, A., Lorenz, A. W., Tonkin, J. D., et al. (2019). Moderate warming over the past 25 years has already reorganized stream invertebrate communities. *Science of The Total Environment* 658, 1531–1538. <u>https://doi.org/</u> <u>10.1016/j.scitotenv.2018.12.234</u>
- Haberkorn, A., Kenner, R., Noetzli, J., and Phillips, M. (2021). Changes in ground temperature and dynamics in mountain permafrost in the Swiss Alps. *Frontiers in Earth Science* 9, 626686. <u>https://doi.org/10.3389/feart.2021.626686</u>
- Haimberger, L., Tavolato, C., and Sperka, S. (2012). Homogenization of the Global Radiosonde Temperature Dataset through Combined Comparison with Reanalysis Background Series and Neighboring Stations. *Journal of Climate* 25, 8108–8131. <u>https:// doi.org/10.1175/JCLI-D-11-00668.1</u>
- Hainzl, S., Kraft, T., Wassermann, J., Igel, H., and Schmedes, E. (2006). Evidence for rainfall-triggered earthquake activity. *Geophys. Res. Lett.* 33, L19303. <u>https://doi.org/10.1029/2006GL 027642</u>
- Halmschlager, E., and Kirisits, T. (2008). First report of the ash dieback pathogen *Chalara fraxinea* on *Fraxinus excelsior* in Austria. *Plant Pathology* 57, 1177–1177. <u>https://doi.org/10.1111/j.1365-3059.2008.01924.x</u>
- Hanel, M., Rakovec, O., Markonis, Y., Máca, P., Samaniego, L., Kyselý, J., et al. (2018). Revisiting the recent European droughts from a long-term perspective. *Sci Rep* 8, 9499. <u>https://doi.org/10.1038/s41598-018-27464-4</u>
- Hansche, I., Fischer, A., Greilinger, M., Hartl, L., Hartmeyer, I., Helfricht, K., et al. (2023). KryoMon.AT – Kryosphären Monitoring Österreich: 2021/22 Kryosphärenbericht Nr. 1. Graz: Bundesministerium Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie. <u>https://doi.org/10.25364/402.2023.1</u>
- Hanus, S., Hrachowitz, M., Zekollari, H., Schoups, G., Vizcaino, M., and Kaitna, R. (2021). Future changes in annual, seasonal and monthly runoff signatures in contrasting Alpine catchments in Austria. *Hydrol. Earth Syst. Sci.* 25, 3429–3453. <u>https:// doi.org/10.5194/hess-25-3429-2021</u>
- Harris, C., Davies, M. C. R., and Etzelmüller, B. (2001). The assessment of potential geotechnical hazards associated with mountain permafrost in a warming global climate. *Permafrost & Periglacial* 12, 145–156. <u>https://doi.org/10.1002/ppp.376</u>
- Hartl, L., Schmitt, P., Schuster, L., Helfricht, K., Abermann, J., and Maussion, F. (2024). Recent observations and glacier modeling point towards near complete glacier loss in western Austria (Ötztal and Stubai mountain range) if 1.5 °C is not met. <u>https:// doi.org/10.5194/egusphere-2024-3146</u>
- Hartl, L., Zieher, T., Bremer, M., Stocker-Waldhuber, M., Zahs, V., Höfle, B., et al. (2023). Multi-sensor monitoring and data integration reveal cyclical destabilization of the Äußeres Hochebenkar rock glacier. *Earth Surface Dynamics* 11, 117–147. <u>https:// doi.org/10.5194/esurf-11-117-2023</u>
- Hartmeyer, I., Delleske, R., Keuschnig, M., Krautblatter, M., Lang, A., Otto, J.-C., et al. (2020a). Current glacier recession causes significant rockfall increase: the immediate paraglacial response of deglaciating cirque walls. *Earth Surface Dynamics* 8, 729–751. <u>https://doi.org/10.5194/esurf-8-729-2020</u>
- Hartmeyer, I., Keuschnig, M., Delleske, R., Krautblatter, M., Lang, A., Schrott, L., et al. (2020b). A 6-year lidar survey reveals enhanced rockwall retreat and modified rockfall magnitudes/

frequencies in deglaciating cirques. *Earth Surf. Dynam.* 8, 753–768. https://doi.org/10.5194/esurf-8-753-2020

- Harvey, B. J., Cook, P., Shaffrey, L. C., and Schiemann, R. (2020). The Response of the Northern Hemisphere Storm Tracks and Jet Streams to Climate Change in the CMIP3, CMIP5, and CMIP6 Climate Models. *JGR Atmospheres* 125, e2020JD032701. https://doi.org/10.1029/2020JD032701
- Haslinger, K., and Blöschl, G. (2017). Space-Time Patterns of Meteorological Drought Events in the European Greater Alpine Region Over the Past 210 Years. Water Resources Research 53, 9807–9823. <u>https://doi.org/10.1002/2017WR020797</u>
- Haslinger, K., Hofstätter, M., Kroisleitner, C., Schöner, W., Laaha, G., Holawe, F., et al. (2019). Disentangling Drivers of Meteorological Droughts in the European Greater Alpine Region during the Last Two Centuries. *Journal of Geophysical Research: Atmospheres* 124, 12404–12425. <u>https://doi.org/10.1029/</u> 2018JD029527
- Haslinger, K., Hofstätter, M., Schöner, W., and Blöschl, G. (2021). Changing summer precipitation variability in the Alpine region: on the role of scale dependent atmospheric drivers. *Clim Dyn* 57, 1009–1021. <u>https://doi.org/10.1007/s00382-021-05753-5</u>
- Haslinger, K., and Mayer, K. (2022). Early spring droughts in Central Europe: Indications for atmospheric and oceanic drivers. *Atmospheric Science Letters* 24, e1136. <u>https://doi.org/10.1002/ asl.1136</u>
- Haslinger, K., Schöner, W., Abermann, J., Laaha, G., Andre, K., Olefs, M., et al. (2023). Apparent contradiction in the projected climatic water balance for Austria: wetter conditions on average versus higher probability of meteorological droughts. *Natural Hazards and Earth System Sciences* 23, 2749–2768. <u>https://doi. org/10.5194/nhess-23-2749-2023</u>
- Haslinger, K., Schöner, W., and Anders, I. (2016). Future drought probabilities in the Greater Alpine Region based on COSMO-CLM experiments – spatial patterns and driving forces. *metz* 25, 137–148. <u>https://doi.org/10.1127/metz/2015/0604</u>
- Heino, J., Alahuhta, J., Bini, L. M., Cai, Y., Heiskanen, A., Hellsten, S., et al. (2021). Lakes in the era of global change: moving beyond single-lake thinking in maintaining biodiversity and ecosystem services. *Biological Reviews* 96, 89–106. <u>https://doi. org/10.1111/brv.12647</u>
- Heinrich, G., and Gobiet, A. (2012). The future of dry and wet spells in Europe: a comprehensive study based on the ENSEM-BLES regional climate models. *Intl Journal of Climatology* 32, 1951–1970. <u>https://doi.org/10.1002/joc.2421</u>
- Helfricht, K., Huss, M., Fischer, A., and Otto, J.-C. (2019). Calibrated Ice Thickness Estimate for All Glaciers in Austria. *Frontiers* in Earth Science 7, 68. <u>https://doi.org/10.3389/feart.2019.00068</u>
- Hiebl, J., Bourgeois, Q., Tilg, A.-M., and Frei, C. (2024). Daily sunshine grids for Austria since 1961 – combining station and satellite observations for a multi-decadal climate-monitoring dataset. *Theor Appl Climatol* 155, 8337–8360. <u>https://doi. org/10.1007/s00704-024-05103-5</u>
- Hiebl, J., and Frei, C. (2018). Daily precipitation grids for Austria since 1961—development and evaluation of a spatial dataset for hydroclimatic monitoring and modelling. *Theor Appl Climatol* 132, 327–345. <u>https://doi.org/10.1007/s00704-017-2093-x</u>
- Hiebl, J., and Hofstätter, M. (2012). No increase in multi-day temperature variability in Austria following climate warming. *Climatic Change* 113, 733–750. <u>https://doi.org/10.1007/s10584-011-0389-x</u>

AAR2

- Hiebl, J., and Schöner, W. (2018). Temperature inversions in Austria in a warming climate – changes in space and time. *Meteorologische Zeitschrift* 27, 309–323. <u>https://doi.org/10.1127/metz/</u> 2018/0899
- Hirschberg, J., Fatichi, S., Bennett, G. L., McArdell, B. W., Peleg, N., Lane, S. N., et al. (2021). Climate Change Impacts on Sediment Yield and Debris-Flow Activity in an Alpine Catchment. *Journal of Geophysical Research: Earth Surface* 126, e2020JF005739. <u>https://doi.org/10.1029/2020JF005739</u>
- Hofmann, R., and Sausgruber, J. T. (2017). Creep behaviour and remediation concept for a deep-seated landslide, Navistal, Tyrol, Austria: Kriechverhalten und Sanierungskonzept einer Großhangbewegung, Navistal, Tirol, Österreich. Geomechanics and Tunnelling 10, 59–73. <u>https://doi.org/10.1002/geot.2016</u> 00066
- Hofstätter, M., Lexer, A., Homann, M., and Blöschl, G. (2018). Large-scale heavy precipitation over central Europe and the role of atmospheric cyclone track types. *International Journal of Climatology* 38, e497–e517. <u>https://doi.org/10.1002/joc.5386</u>
- Holzinger, B., Hülber, K., Camenisch, M., and Grabherr, G. (2008). Changes in plant species richness over the last century in the eastern Swiss Alps: elevational gradient, bedrock effects and migration rates. *Plant Ecol* 195, 179–196. <u>https://doi.org/10.1007/ s11258-007-9314-9</u>
- Höning, D., Willeit, M., Calov, R., Klemann, V., Bagge, M., and Ganopolski, A. (2023). Multistability and Transient Response of the Greenland Ice Sheet to Anthropogenic CO<sub>2</sub> Emissions. *Geophysical Research Letters* 50, e2022GL101827. <u>https://doi. org/10.1029/2022GL101827</u>
- Hübl, J., Ganahl, E., Bacher, M., Chiari, M., Holub, M., Kaitna, R., et al. (2005). Dokumentation der Wildbachereignisse vom 22./23. August 2005 in Tirol. Band 1: Generelle Aufnahme. Institute of Mountain Risk Engineering. Available at: <u>https://boku. ac.at/fileadmin/data/H03000/H87000/H87100/IAN\_Reports/ REP0109\_Band1.pdf</u>
- Hübl, J., Kuschel, E., Egger, M., and Schöffl, T. (2022). IAN RE-PORT 220 Band 2: Ereignisdokumentation August 2021 Wald im Pinzgau – Krimml. Institute of Mountain Risk Engineering. Available at: <u>https://forschung.boku.ac.at/en/publications/</u> <u>146986</u>
- Hueging, H., Haas, R., Born, K., Jacob, D., and Pinto, J. G. (2013). Regional Changes in Wind Energy Potential over Europe Using Regional Climate Model Ensemble Projections. *Journal of Applied Meteorology and Climatology* 52, 903–917. <u>https://doi.org/10.1175/JAMC-D-12-086.1</u>
- Huggel, C., Allen, S., Deline, P., Fischer, L., Noetzli, J., and Ravanel, L. (2012a). Ice thawing, mountains falling-are alpine rock slope failures increasing? *Geology Today* 28, 98–104. <u>https://doi.org/10.1111/j.1365-2451.2012.00836.x</u>
- Huggel, C., Clague, J. J., and Korup, O. (2012b). Is climate change responsible for changing landslide activity in high mountains?: CLIMATE CHANGE AND LANDSLIDES IN HIGH MOUN-TAINS. *Earth Surf. Process. Landforms* 37, 77–91. <u>https://doi. org/10.1002/esp.2223</u>
- Huggel, C., Salzmann, N., Allen, S., Caplan-Auerbach, J., Fischer, L., Haeberli, W., et al. (2010). Recent and future warm extreme events and high-mountain slope stability. *Phil. Trans. R. Soc. A.* 368, 2435–2459. <u>https://doi.org/10.1098/rsta.2010.0078</u>
- Humer, G., Achleitner, S., Lumassegger, S., and Reithofer, A. (2017). Climate Change and Flash Floods. Copernicus Climate Change Service. Available at: <u>https://climate.copernicus.eu/</u>

sites/default/files/2021-02/D2.1%20SWICCA\_Workflows\_Final\_Climate\_Change\_and\_Flash-Floods.pdf (Accessed September 14, 2023).

- Hungr, O., Leroueil, S., and Picarelli, L. (2014). The Varnes classification of landslide types, an update. *Landslides* 11, 167–194. <u>https://doi.org/10.1007/s10346-013-0436-y</u>
- Hušek, J., and Adamík, P. (2008). Long-term trends in the timing of breeding and brood size in the Red-Backed Shrike Lanius collurio in the Czech Republic, 1964–2004. J Ornithol 149, 97–103. https://doi.org/10.1007/s10336-007-0222-5
- Husen, S., Bachmann, C., and Giardini, D. (2007). Locally triggered seismicity in the central Swiss Alps following the large rainfall event of August 2005: Locally triggered seismicity in the central Swiss Alps. *Geophysical Journal International* 171, 1126–1134. https://doi.org/10.1111/j.1365-246X.2007.03561.x
- Huss, M., and Hock, R. (2018). Global-scale hydrological response to future glacier mass loss. *Nature Climate Change* 8, 135–140. <u>https://doi.org/10.1038/s41558-017-0049-x</u>
- Illich, I., and Zuna-Kratky, T. (2022). Population dynamics of an alpine grasshopper (Orthoptera) community over 30 years and the effects of climate warming and grazing. *J Insect Conserv* 26, 435–451. <u>https://doi.org/10.1007/s10841-022-00381-8</u>
- Imperio, S., Bionda, R., Viterbi, R., and Provenzale, A. (2013). Climate Change and Human Disturbance Can Lead to Local Extinction of Alpine Rock Ptarmigan: New Insight from the Western Italian Alps. *PLOS ONE* 8, e81598. <u>https://doi.org/10.1371/journal.pone.0081598</u>
- Ionita, M., Dima, M., Nagavciuc, V., Scholz, P., and Lohmann, G. (2021). Past megadroughts in central Europe were longer, more severe and less warm than modern droughts. *Commun Earth Environ* 2, 61. <u>https://doi.org/10.1038/s43247-021-00130-w</u>
- Ionita, M., Nagavciuc, V., Kumar, R., and Rakovec, O. (2020). On the curious case of the recent decade, mid-spring precipitation deficit in central Europe. *npj Clim Atmos Sci* 3, 49. <u>https://doi.org/10.1038/s41612-020-00153-8</u>
- IPCC (2001). Climate Change 2001: Synthesis Report. A Contribution of Working Groups I, II, and III to the Third Assessment Report of the Intergovernmental Panel on Climate Change, ed. R. T. Watson. Cambridge, United Kingdom, and New York, NY, USA: Cambridge University Press. <u>https://www.ipcc.ch/site/assets/uploads/2018/05/SYR\_TAR\_full\_report.pdf</u>
- IPCC (2014a). 2013 Supplement to the 2006 IPCC Guidelines for National Greenhouse Gas Inventories: Wetlands., eds. T. Hiraishi, T. Krug, K. Tanabe, N. Srivastava, B. Jamsranjav, M. Fukuda, et al. Switzerland: IPCC Task Force on National Greenhouse Gas Inventories Technical Support Unit. Available at: <u>https://www. ipcc-nggip.iges.or.jp/public/wetlands/index.html</u> (Accessed September 21, 2023).
- IPCC (2014b). "Long-term Climate Change: Projections, Commitments and Irreversibility", in *Climate Change 2013 – The Physical Science Basis*, (Cambridge University Press), 1029–1136. <u>https://doi.org/10.1017/CBO9781107415324.024</u>
- IPCC (2018). Global Warming of 1.5°C. An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty., eds. V. Masson-Delmotte, P. Zhai, H.-O. Pörtner, D. Roberts, J. Skea, P. R. Shukla, et al. Cambridge, UK and New York, NY, USA: Cambridge University Press. <u>https://doi. org/10.1017/9781009157940</u>

- IPCC (2019a). "Summary for Policymakers", in Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems, eds. P. R. Shukla, J. Skea, E. Calvo Buendia, V. Masson-Delmotte, H.-O. Pörtner, D. C. Roberts, et al. (Geneva: IPCC). Available at: https://doi.org/10.1017/9781009157988.001
- IPCC (2019b). Summary for Policymakers. In: IPCC Special Report on the Ocean and Cryosphere in a Changing Climate., 1st Edn, eds. H.-O. Pörtner, D. C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, et al. Cambridge, UK and New York, NY, USA: Cambridge University Press. <u>https://doi.org/10.1017/9781009157964</u>
- IPCC (2021a). Climate Change 2021 The Physical Science Basis: Working Group I Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change., 1st Edn. Cambridge University Press. <u>https://doi.org/10.1017/9781009157896</u>
- IPCC (2021b). "Summary for Policymakers", in Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, eds. V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press), 3–32. <u>https://doi.org/10.1017/9781009157896.001</u>
- IPCC (2022a). "Annex II: Glossary", in Climate Change 2022 Impacts, Adaptation and Vulnerability. Working Group II Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, (Cambridge University Press). <u>https://doi.org/10.1017/9781009325844</u>
- IPCC (2022b). Climate Change 2022 Impacts, Adaptation and Vulnerability: Working Group II Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change., 1st Edn. Cambridge University Press. <u>https://doi.org/10.1017/</u> 9781009325844
- Irauschek, F., Rammer, W., and Lexer, M. J. (2017a). Can current management maintain forest landscape multifunctionality in the Eastern Alps in Austria under climate change? *Regional Environmental Change* 17, 33–48. <u>https://doi.org/10.1007/s10113-015-0908-9</u>
- Irauschek, F., Rammer, W., and Lexer, M. J. (2017b). Evaluating multifunctionality and adaptive capacity of mountain forest management alternatives under climate change in the Eastern Alps. *European Journal of Forest Research* 136, 1051–1069. https://doi.org/10.1007/s10342-017-1051-6
- Ivancic, T. J., and Shaw, S. B. (2016). A U.S.-based analysis of the ability of the Clausius-Clapeyron relationship to explain changes in extreme rainfall with changing temperature. *JGR Atmospheres* 121, 3066–3078. <u>https://doi.org/10.1002/2015JD024288</u>
- Iyakaremye, V., Zeng, G., and Zhang, G. (2021). Changes in extreme temperature events over Africa under 1.5 and 2.0 global warming scenarios. *Intl Journal of Climatology* 41, 1506–1524. <u>https://doi.org/10.1002/joc.6868</u>
- Jacob, D., Petersen, J., Eggert, B., Alias, A., Christensen, O. B., Bouwer, L. M., et al. (2014). EURO-CORDEX: new high-resolution climate change projections for European impact research. *Reg Environ Change* 14, 563–578. <u>https://doi.org/10.1007/s10113-013-0499-2</u>
- Jacquemart, M., Weber, S., Chiarle, M., Chmiel, M., Cicoira, A., Corona, C., et al. (2024). Detecting the impact of climate change on alpine mass movements in observational records from the

European Alps. Earth-Science Reviews 258, 104886. https://doi. org/10.1016/j.earscirev.2024.104886

- Jakoby, O., Lischke, H., and Wermelinger, B. (2019). Climate change alters elevational phenology patterns of the European spruce bark beetle (*Ips typographus*). *Global Change Biology* 25, 4048–4063. <u>https://doi.org/10.1111/gcb.14766</u>
- James, R., Washington, R., Schleussner, C., Rogelj, J., and Conway, D. (2017). Characterizing half-a-degree difference: a review of methods for identifying regional climate responses to global warming targets. WIREs Clim Change 8, e457. <u>https://doi. org/10.1002/wcc.457</u>
- Jandl, R., Haeler, E., Kindermann, G., Lapin, K., Oettel, J., and Schüler, S. (2024). Management and biodiversity conservation in Central European forests. *Trees, Forests and People* 17, 100601. <u>https://doi.org/10.1016/j.tfp.2024.100601</u>
- Jandt, U., Bruelheide, H., Jansen, F., Bonn, A., Grescho, V., Klenke, R. A., et al. (2022). More losses than gains during one century of plant biodiversity change in Germany. *Nature* 611, 512–518. https://doi.org/10.1038/s41586-022-05320-w
- Jane, S. F., Hansen, G. J. A., Kraemer, B. M., Leavitt, P. R., Mincer, J. L., North, R. L., et al. (2021). Widespread deoxygenation of temperate lakes. *Nature* 594, 66–70. <u>https://doi.org/10.1038/</u> <u>s41586-021-03550-y</u>
- Jaritz, W., and Soranzo, E. (2014). Engineering geology of the large slope movement Sibratsgfäll and Rindberg / Ingenieurgeologie der Großhangbewegungen Sibratsgfäll und Rindberg. Geomechanics and Tunnelling 7, 306–316. <u>https://doi.org/10.1002/geot.</u> 201400021
- Jensen, J. L., Schjønning, P., Watts, C. W., Christensen, B. T., Peltre, C., and Munkholm, L. J. (2019). Relating soil C and organic matter fractions to soil structural stability. *Geoderma* 337, 834– 843. <u>https://doi.org/10.1016/j.geoderma.2018.10.034</u>
- Jerez, S., López-Romero, J. M., Turco, M., Jiménez-Guerrero, P., Vautard, R., and Montávez, J. P. (2018). Impact of evolving greenhouse gas forcing on the warming signal in regional climate model experiments. *Nat Commun* 9, 1304. <u>https://doi.org/10.1038/s41467-018-03527-y</u>
- Johannsen, L. L., Schmaltz, E. M., Mitrovits, O., Klik, A., Smoliner, W., Wang, S., et al. (2022). An update of the spatial and temporal variability of rainfall erosivity (R-factor) for the main agricultural production zones of Austria. *CATENA* 215, 106305. <u>https://doi.org/10.1016/j.catena.2022.106305</u>
- Johnson, D. M., Büntgen, U., Frank, D. C., Kausrud, K., Haynes, K. J., Liebhold, A. M., et al. (2010). Climatic warming disrupts recurrent Alpine insect outbreaks. *Proc. Natl. Acad. Sci. U.S.A.* 107, 20576–20581. <u>https://doi.org/10.1073/pnas.1010270107</u>
- Johnston, A. E., Poulton, P. R., and Coleman, K. (2009). "Chapter 1 Soil Organic Matter", in Advances in Agronomy, (Elsevier), 1–57. <u>https://doi.org/10.1016/S0065-2113(08)00801-8</u>
- Jomelli, V., Brunstein, D., Grancher, D., and Pech, P. (2007). Is the response of hill slope debris flows to recent climate change univocal? A case study in the Massif des Ecrins (French Alps). *Climatic Change* 85, 119–137. <u>https://doi.org/10.1007/s10584-</u>006-9209-0
- Jomelli, V., Pech, V. P., Chochillon, C., and Brunstein, D. (2004). Geomorphic Variations of Debris Flows and Recent Climatic Change in the French Alps. *Climatic Change* 64, 77–102. <u>https:// doi.org/10.1023/B:CLIM.0000024700.35154.44</u>
- Julsrud, I. R., Storelvmo, T., Schulz, M., Moseid, K. O., and Wild, M. (2022). Disentangling Aerosol and Cloud Effects on Dimming and Brightening in Observations and CMIP6. *Journal*

of Geophysical Research: Atmospheres 127, e2021JD035476. https://doi.org/10.1029/2021JD035476

- Jurasinski, G., and Kreyling, J. (2007). Upward shift of alpine plants increases floristic similarity of mountain summits. *J Vegetation Science* 18, 711–718. <u>https://doi.org/10.1111/j.1654-1103.2007.</u> <u>tb02585.x</u>
- Kainz, M. J., Ptacnik, R., Rasconi, S., and Hager, H. H. (2017). Irregular changes in lake surface water temperature and ice cover in subalpine Lake Lunz, Austria. *Inland Waters* 7, 27–33. <u>https:// doi.org/10.1080/20442041.2017.1294332</u>
- Kaitna, R., Prenner, D., Switanek, M., Maraun, D., Stoffel, M., and Hrachowitz, M. (2023). Changes of hydro-meteorological trigger conditions for debris flows in a future alpine climate. *Science* of The Total Environment 872, 162227. <u>https://doi.org/10.1016/j.</u> scitoteny.2023.162227
- Kalny, G., Laaha, G., Melcher, A., Trimmel, H., Weihs, P., and Rauch, H. P. (2017). The influence of riparian vegetation shading on water temperature during low flow conditions in a medium sized river. *Knowl. Manag. Aquat. Ecosyst.* 418, 5. <u>https:// doi.org/10.1051/kmae/2016037</u>
- Kappes, M. S., Keiler, M., Von Elverfeldt, K., and Glade, T. (2012). Challenges of analyzing multi-hazard risk: a review. Nat Hazards 64, 1925–1958. <u>https://doi.org/10.1007/s11069-012-0294-2</u>
- Kellerer-Pirklbauer, A., Bodin, X., Delaloye, R., Lambiel, C., Gärtner-Roer, I., Bonnefoy-Demongeot, M., et al. (2024). Acceleration and interannual variability of creep rates in mountain permafrost landforms (rock glacier velocities) in the European Alps in 1995–2022. *Environ. Res. Lett.* 19, 034022. <u>https://doi. org/10.1088/1748-9326/ad25a4</u>
- Kellerer-Pirklbauer, A., Lieb, G. K., Avian, M., and Carrivick, J. (2012a). Climate change and rock fall events in high mountain areas: numerous and extensive rock falls in 2007 at mittlerer burgstall, central austria. *Geografiska Annaler: Series A, Physical Geography* 94, 59–78. <u>https://doi.org/10.1111/j.1468-0459.</u> 2011.00449.x
- Kellerer-Pirklbauer, A., Slupetzky, H., and Avian, M. (2012b). Iceavalanche impact landforms: the event in 2003 at the glacier nördliches Bockkarkees, Hohe Tauern range, Austria. *Geografiska Annaler: Series A, Physical Geography* 94, 97–115. <u>https:// doi.org/10.1111/j.1468-0459.2011.00446.x</u>
- Kenyeres, Z., Takács, G., and Bauer, N. (2019). Response of orthopterans to macroclimate changes: A 15-year case study in Central European humid grasslands. *JOR* 28, 187–193. <u>https://doi. org/10.3897/jor.28.34102</u>
- Kerner, J. M., Krauss, J., Maihoff, F., Bofinger, L., and Classen, A. (2023). Alpine butterflies want to fly high: Species and communities shift upwards faster than their host plants. *Ecology* 104, e3848. <u>https://doi.org/10.1002/ecy.3848</u>
- Kerschbaumer (2022). Biodiversitäts-Strategie Österreich 2030+ Wien. Convention on Biological Diversity. Brazil. Available at: <u>https://policycommons.net/artifacts/12315313/biodiversitatsstrategie-osterreich-2030/13211656/</u> (Accessed May 13, 2024).
- Kessler, M., and Lexer, M. J. (2023). "Baumarteneignung", in Dynamische Waldtypisierung Steiermark, (Graz: Amt der Steiermärkischen Landesregierung, ABT10 Land- und Forstwirtschaft, Landesforstdirektion), 24–29. Available at: <u>https://www.agrar.</u> <u>steiermark.at/cms/dokumente/12733633\_151504582/daf</u> <u>513e5/FORSITE\_Band1\_030823.pdf</u> (Accessed September 21, 2021).
- Kiefer, C., Oswald, P., Moernaut, J., Fabbri, S. C., Mayr, C., Strasser, M., et al. (2021). A 4000-year debris flow record based on am-

phibious investigations of fan delta activity in Plansee (Austria, Eastern Alps). *Earth Surf. Dynam.* 9, 1481–1503. <u>https://doi.org/10.5194/esurf-9-1481-2021</u>

- Kiem, J., Hammerle, A., Montagnani, L., and Wohlfahrt, G. (2024). Future scenarios of albedo and radiative forcing resulting from changes in snow depth in Austria. <u>https://doi.org/10.5194/egusphere-2024-881</u>
- Kimutai, J., Vautard, R., Zachariah, M., Tolasz, R., Šustková, V., Cassou, C., et al. (2024). Climate change and high exposure increased costs and disruption to lives and livelihoods from flooding associated with exceptionally heavy rainfall in Central Europe. Imperial College London. <u>https://doi.org/10.25561/</u> 114694
- King, A. D., Borowiak, A. R., Brown, J. R., Frame, D. J., Harrington, L. J., Min, S., et al. (2021). Transient and Quasi-Equilibrium Climate States at 1.5°C and 2°C Global Warming. *Earth's Future* 9, e2021EF002274. https://doi.org/10.1029/2021EF002274
- King, A. D., Lane, T. P., Henley, B. J., and Brown, J. R. (2020). Global and regional impacts differ between transient and equilibrium warmer worlds. *Nat. Clim. Chang.* 10, 42–47. <u>https://doi. org/10.1038/s41558-019-0658-7</u>
- Kjellström, E., Nikulin, G., Strandberg, G., Christensen, O. B., Jacob, D., Keuler, K., et al. (2018). European climate change at global mean temperature increases of 1.5 and 2 °C above pre-industrial conditions as simulated by the EURO-CORDEX regional climate models. *Earth Syst. Dynam.* 9, 459–478. <u>https://</u> doi.org/10.5194/esd-9-459-2018
- Klein, G., Vitasse, Y., Rixen, C., Marty, C., and Rebetez, M. (2016). Shorter snow cover duration since 1970 in the Swiss Alps due to earlier snowmelt more than to later snow onset. *Climatic Change* 139, 637–649. <u>https://doi.org/10.1007/s10584-016-1806-y</u>
- Kleinbauer, I., Dullinger, S., Peterseil, J., and Essl, F. (2010). Climate change might drive the invasive tree Robinia pseudacacia into nature reserves and endangered habitats. *Biological Conservation* 143, 382–390. <u>https://doi.org/10.1016/j.biocon.2009.10.024</u>
- Kleinteich, J., Hanselmann, K., Hildebrand, F., Kappler, A., and Zarfl, C. (2022). Glacier melt-down changes habitat characteristics and unique microbial community composition and physiology in alpine lake sediments. *FEMS Microbiology Ecology* 98, fiac075. <u>https://doi.org/10.1093/femsec/fiac075</u>
- Klug, C., Rieg, L., Ott, P., Mössinger, M., Sailer, R., and Stötter, J. (2017). A Multi-Methodological Approach to Determine Permafrost Occurrence and Ground Surface Subsidence in Mountain Terrain, Tyrol, Austria: Analyses of Surface Subsidence in a Periglacial Environment. *Permafrost and Periglac. Process.* 28, 249–265. https://doi.org/10.1002/ppp.1896
- Knaus, P. (2018). Schweizer Brutvogelatlas 2013–2016 Verbreitung und Bestandsentwicklung der Vögel in der Schweiz und im Fürstentum Liechtenstein. Sempach: <u>https://www.vogelwarte.ch/</u> <u>de</u>
- Knevels, R., Petschko, H., Proske, H., Leopold, P., Mishra, A. N., Maraun, D., et al. (2023). Assessing uncertainties in landslide susceptibility predictions in a changing environment (Styrian Basin, Austria). *Nat. Hazards Earth Syst. Sci.* 23, 205–229. https://doi.org/10.5194/nhess-23-205-2023
- Koch, R. (2021a). FuSE-AT Daily snowmaking hours. <u>https://doi.org/10.60669/BABA-JW56</u>
- Koch, R. (2021b). FuSE-AT: Simulation with snow-cover model SNOWGRID-CL. <u>https://doi.org/10.60669/3B2F-QB11</u>
- Kolbe, H., and Zimmer, J. (2015). Leitfaden zur Humusversorgung – Informationen f
  ür Praxis, Beratung und Schulung. Landesamt

AAR2

für Umwelt, Landwirtschaft und Geologie. Available at: <u>https://</u>publikationen.sachsen.de/bdb/artikel/25484

- Kotlarski, S., Gobiet, A., Morin, S., Olefs, M., Rajczak, J., and Samacoïts, R. (2023). 21st Century alpine climate change. *Climate Dynamics* 60, 65–86. <u>https://doi.org/10.1007/s00382-022-06303-3</u>
- Kotlarski, S., Lüthi, D., and Schär, C. (2015). The elevation dependency of 21st century European climate change: an RCM ensemble perspective. *Intl Journal of Climatology* 35, 3902–3920. <u>https://doi.org/10.1002/joc.4254</u>
- Kraemer, M. U. G., Reiner, R. C., Brady, O. J., Messina, J. P., Gilbert, M., Pigott, D. M., et al. (2019). Past and future spread of the arbovirus vectors Aedes aegypti and Aedes albopictus. *Nat Microbiol* 4, 854–863. <u>https://doi.org/10.1038/s41564-019-0376-y</u>
- Krammer, C., Kahl, B., and Higer, F. (2024). Zusammenfassung aus hydrologischer Sicht zum Hochwasser vom 13.9.2024 bis zum 20.9.2024 in NÖ. Abteilung WA2 Wasserwirtschaft – Hydrographsicher Dienst und Hochwasserprognosen. Available at: <u>https://www.noe.gv.at/wasserstand/staticdata/Analysen/3/ Vorl%C3%A4ufige\_Analyse\_Hochwasser\_Sept2024\_Hydro\_ web.pdf</u>
- Kramsall, C. C. (2018). Drought impacts in Austria: index selection and modelling in forestry, wild fires and Danube transport. Vienna. Available at: <u>https://permalink.obvsg.at/bok/AC15185802</u>
- Krautblatter, M., Weber, S., Dietze, M., Keuschnig, M., Stockinger, G., Brückner, L., et al. (2024). The 2023 Fluchthorn massive permafrost rock slope failure analysed. <u>https://doi.org/10.5194/</u> egusphere-egu24-20989
- Kreyling, J., Dengler, J., Walter, J., Velev, N., Ugurlu, E., Sopotlieva, D., et al. (2017). Species richness effects on grassland recovery from drought depend on community productivity in a multisite experiment. *Ecol Lett* 20, 1405–1413. <u>https://doi.org/10.1111/ ele.12848</u>
- Kropik, M., Zechmeister, H. G., Moser, D., Bernhardt, K. G., and Dullinger, S. (2021). Deadwood volumes matter in epixylic bryophyte conservation, but precipitation limits the establishment of substrate-specific communities. *Forest Ecology and Management* 493, 119285. <u>https://doi.org/10.1016/j.foreco.2021.</u> <u>119285</u>
- Küchler, M., Küchler, H., Bedolla, A., and Wohlgemuth, T. (2015). Response of Swiss forests to management and climate change in the last 60 years. *Annals of Forest Science* 72, 311–320. <u>https:// doi.org/10.1007/s13595-014-0409-x</u>
- Kuhn, M., and Olefs, M. (2020). "Elevation-Dependent Climate Change in the European Alps", in Oxford Research Encyclopedia of Climate Science, (Oxford University Press). Available at: <u>https://oxfordre.com/climatescience/view/10.1093/acrefore/ 9780190228620.001.0001/acrefore-9780190228620-e-762</u> (Accessed January 8, 2021).
- Laaha, G., and Blöschl, G. (2006). Seasonality indices for regionalizing low flows. *Hydrological Processes* 20, 3851–3878. <u>https:// doi.org/10.1002/hyp.6161</u>
- Laaha, G., and Blöschl, G. (2007). A national low flow estimation procedure for Austria. *Hydrological Sciences Journal* 52, 625– 644. <u>https://doi.org/10.1623/hysj.52.4.625</u>
- Laaha, G., Laimighofer, J., Parajka, J., Bertola, M., and Blöschl, G. (2025). Abflusstrends in Österreichs Fließgewässern 1977– 2020. Österr Wasser- und Abfallw. <u>https://doi.org/10.1007/s00</u> <u>506-024-01106-8</u>
- Laaha, G., Parajka, J., Viglione, A., Koffler, D., Haslinger, K., Schöner, W., et al. (2016). A three-pillar approach to assessing

climate impacts on low flows. *Hydrol. Earth Syst. Sci.* 20, 3967–3985. https://doi.org/10.5194/hess-20-3967-2016

- Laimighofer, J., and Laaha, G. (2022). How standard are standardized drought indices? Uncertainty components for the SPI & SPEI case. *Journal of Hydrology* 613, 128385. <u>https://doi.org/10.1016/j.jhydrol.2022.128385</u>
- Lamprecht, A., Semenchuk, P. R., Steinbauer, K., Winkler, M., and Pauli, H. (2018). Climate change leads to accelerated transformation of high-elevation vegetation in the central Alps. *New Phytologist* 220, 447–459. https://doi.org/10.1111/nph.15290
- Lapin, K., Oettel, J., Steiner, H., Langmaier, M., Sustic, D., Starlinger, F., et al. (2019). Invasive alien plant species in unmanaged forest reserves, Austria. NB 48, 71–96. <u>https://doi.org/10.3897/</u> neobiota.48.34741
- Le Roux, E., Evin, G., Eckert, N., Blanchet, J., and Morin, S. (2023). Projection of snowfall extremes in the French Alps as a function of elevation and global warming level. Available at: <u>https://egusphere.copernicus.org/preprints/2023/egusphere-2023-335/</u> (Accessed March 11, 2023).
- Lee, J.-Y., Marotzke, J., Bala, G., Cao, L., Corti, S., Dunne, J. P., et al. (2021). "Future Global Climate: Scenario-Based Projections and Near-Term Information", in *Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*, eds. V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press), 553–672. https://doi.org/10.1017/9781009157896.006
- Lehmann, J., Coumou, D., and Frieler, K. (2015). Erratum to: increased record-breaking precipitation events under global warming. *Climatic Change* 132, 517–518. <u>https://doi.org/10. 1007/s10584-015-1466-3</u>
- Leifeld, J., and Menichetti, L. (2018). The underappreciated potential of peatlands in global climate change mitigation strategies. *Nat Commun* 9, 1071. <u>https://doi.org/10.1038/s41467-018-03406-6</u>
- Lemoine, N., Bauer, H.-G., Peintinger, M., and Böhning-Gaese, K. (2007). Effects of Climate and Land-Use Change on Species Abundance in a Central European Bird Community. *Conservation Biology* 21, 495–503. <u>https://doi.org/10.1111/j.1523-1739.</u> <u>2006.00633.x</u>
- Lenderink, G., and Van Meijgaard, E. (2008). Increase in hourly precipitation extremes beyond expectations from temperature changes. *Nature Geosci* 1, 511–514. <u>https://doi.org/10.1038/ ngeo262</u>
- Lenderink, G., and Van Meijgaard, E. (2010). Linking increases in hourly precipitation extremes to atmospheric temperature and moisture changes. *Environ. Res. Lett.* 5, 025208. <u>https://doi. org/10.1088/1748-9326/5/2/025208</u>
- Lenhardt, W. (2007). Earthquake-triggered landslides in Austria Dobratsch revisited. Jahrbuch der Geologischen Bundesanstalt 147, 193–199.
- Lenton, T. M., Held, H., Kriegler, E., Hall, J. W., Lucht, W., Rahmstorf, S., et al. (2008). Tipping elements in the Earth's climate system. *Proc. Natl. Acad. Sci. U.S.A.* 105, 1786–1793. <u>https://doi. org/10.1073/pnas.0705414105</u>
- Leuschner, C., and Ellenberg, H. (2017). Vegetation Ecology of Central Europe. Cham: Springer. Available at: <u>https://link.springer.</u> <u>com/book/9783319507101</u> (Accessed September 21, 2023).
- Lewis, S. C., King, A. D., Perkins-Kirkpatrick, S. E., and Mitchell, D. M. (2019). Regional hotspots of temperature extremes un-

der 1.5 °C and 2 °C of global mean warming. *Weather and Climate Extremes* 26, 100233. <u>https://doi.org/10.1016/j.wace.2019.</u> 100233

- Lexer, M. J. (2019). GravitationaL hazards Amplified by Degradation of protection forests and Extreme precipitation episodes. Wien: Universität für Bodenkultur Wien, Institut für Waldbau. Available at: <u>https://projektdatenbank.net/finalreport/6174/</u> (Accessed September 22, 2023).
- Lhotka, O., and Kyselý, J. (2015). Characterizing joint effects of spatial extent, temperature magnitude and duration of heat waves and cold spells over Central Europe: HEAT WAVES AND COLD SPELLS OVER CENTRAL EUROPE. *Int. J. Climatol.* 35, 1232–1244. <u>https://doi.org/10.1002/joc.4050</u>
- Lhotka, O., Trnka, M., Kyselý, J., Markonis, Y., Balek, J., and Možný, M. (2020). Atmospheric Circulation as a Factor Contributing to Increasing Drought Severity in Central Europe. JGR Atmospheres 125, e2019JD032269. <u>https://doi.org/10.1029/</u> 2019JD032269
- Li, Z., Gao, S., Chen, M., Gourley, J. J., Liu, C., Prein, A. F., et al. (2022). The conterminous United States are projected to become more prone to flash floods in a high-end emissions scenario. *Commun Earth Environ* 3, 86. <u>https://doi.org/10.1038/</u> <u>s43247-022-00409-6</u>
- Liberati, L., Messerli, S., Matteodo, M., and Vittoz, P. (2019). Contrasting impacts of climate change on the vegetation of windy ridges and snowbeds in the Swiss Alps. *Alp Botany* 129, 95–105. <u>https://doi.org/10.1007/s00035-019-00223-5</u>
- Lieb, G. K., and Kellerer-Pirklbauer, A. (2023). Gletscherbericht 2021/22. Österreichischer Alpenverein. Available at: <u>https://www.alpenverein.at/portal/service/presse/2023/2023\_03\_31\_gletscherbericht-2021-22.php</u> (Accessed September 26, 2023).
- Loacker, K., Kofler, W., Pagitz, K., and Oberhuber, W. (2007). Spread of walnut (Juglans regia L.) in an Alpine valley is correlated with climate warming. *Flora – Morphology, Distribution, Functional Ecology of Plants* 202, 70–78. <u>https://doi.org/10.1016/j.flo-</u> <u>ra.2006.03.003</u>
- Lochbihler, K., Lenderink, G., and Siebesma, A. P. (2019). Response of Extreme Precipitating Cell Structures to Atmospheric Warming. JGR Atmospheres 124, 6904–6918. <u>https://doi.org/10.1029/ 2018JD029954</u>
- Loisel, J., and Gallego-Sala, A. (2022). Ecological resilience of restored peatlands to climate change. *Commun Earth Environ* 3, 208. <u>https://doi.org/10.1038/s43247-022-00547-x</u>
- Lučan, R. K., Weiser, M., and Hanák, V. (2013). Contrasting effects of climate change on the timing of reproduction and reproductive success of a temperate insectivorous bat. J Zool 290, 151– 159. <u>https://doi.org/10.1111/jzo.12021</u>
- Luethi, R., Gruber, S., and Ravanel, L. (2015). Modelling transient ground surface temperatures of past rockfall events: towards a better understanding of failure mechanisms in changing periglacial environments. *Geografiska Annaler: Series A, Physical Geography* 97, 753–767. https://doi.org/10.1111/geoa.12114
- Luger, M., Kammerlander, B., Blatterer, H., and Gassner, H. (2021). Von der Eutrophierung in die Klimaerwärmung – 45 Jahre limnologisches Monitoring Mondsee. Österr Wasser- und Abfallw 73, 418–425. <u>https://doi.org/10.1007/s00506-021-00786-w</u>
- Maher, N., Milinski, S., Suarez-Gutierrez, L., Botzet, M., Dobrynin, M., Kornblueh, L., et al. (2019). The Max Planck Institute Grand Ensemble: Enabling the Exploration of Climate System Variability. J Adv Model Earth Syst 11, 2050–2069. <u>https://doi.org/10.1029/2019MS001639</u>

- Maier, P., Klisho, T., Formayer, H., Lehner F. (2025). Analysing the future trends of foehn-enabling synoptic patterns over two valleys in the Eastern Alps in CMIP5 EURO-CORDEX models. *Theor Appl Climatol* 156, 155. <u>https://doi.org/10.1007/s00704-025-05365-7</u>
- Maihoff, F., Friess, N., Hoiss, B., Schmid-Egger, C., Kerner, J., Neumayer, J., et al. (2023). Smaller, more diverse and on the way to the top: Rapid community shifts of montane wild bees within an extraordinary hot decade. *Diversity and Distributions* 29, 272–288. https://doi.org/10.1111/ddi.13658
- Manara, V., Brunetti, M., Celozzi, A., Maugeri, M., Sanchez-Lorenzo, A., and Wild, M. (2016). Detection of dimming/brightening in Italy from homogenized all-sky and clear-sky surface solar radiation records and underlying causes (1959–2013). *Atmos. Chem. Phys.* 16, 11145–11161. <u>https://doi.org/10.5194/acp-16-11145-2016</u>
- Manara, V., Brunetti, M., Wild, M., and Maugeri, M. (2023). Variability and trends of the total cloud cover over Italy (1951–2018). *Atmospheric Research* 285, 106625. <u>https://doi.org/10.1016/j.atmosres.2023.106625</u>
- Mang, T., Essl, F., Mosef, D., and Dullinger, S. (2018). Climate warming drives invasion history of Ambrosia artemisiifolia in central Europe. *Preslia* 90, 59–81. <u>https://doi.org/10.23855/ preslia.2018.059</u>
- Mani, P., Allen, S., Evans, S. G., Kargel, J. S., Mergili, M., Petrakov, D., et al. (2023). Geomorphic Process Chains in High-Mountain Regions—A Review and Classification Approach for Natural Hazards Assessment. *Reviews of Geophysics* 61, e2022RG000791. <u>https://doi.org/10.1029/2022RG000791</u>
- Maraun, D. (2013). When will trends in European mean and heavy daily precipitation emerge? *Environ. Res. Lett.* 8, 014004. <u>https:// doi.org/10.1088/1748-9326/8/1/014004</u>
- Maraun, D., and Jury, M. (2022). Der Klimawandel und Naturgefahren im Alpenraum. Wildbach- und Lawinenverbau 190, 38–49.
- Maraun, D., Knevels, R., Mishra, A. N., Truhetz, H., Bevacqua, E., Proske, H., et al. (2022). A severe landslide event in the Alpine foreland under possible future climate and land-use changes. *Commun Earth Environ* 3, 87. <u>https://doi.org/10.1038/s43247-022-00408-7</u>
- Maraun, D., Ritzhaupt, N., Gobiet, A., and Pistotnik, G. (2024). Supporting climate service Providers by dIstilling infoRmation about future precipitation exTremes. Vienna, Austria. Available at: <u>https://www.klimafonds.gv.at/wp-content/uploads/2024/11/</u> B960272-ACRP11-SPIRIT-KR18AC0K14748-EB.pdf
- Maraun, D., Truhetz, H., and Schaffer, A. (2021). Regional Climate Model Biases, Their Dependence on Synoptic Circulation Biases and the Potential for Bias Adjustment: A Process-Oriented Evaluation of the Austrian Regional Climate Projections. *JGR Atmospheres* 126, e2020JD032824. <u>https://doi.org/10.1029/2020</u> JD032824
- Marbler, A. (2024). Pop-AUT: Subnational SSP Population Projections for Austria. <u>https://doi.org/10.5281/ZENODO.10477870</u>
- Marcer, M., Serrano, C., Brenning, A., Bodin, X., Goetz, J., and Schoeneich, P. (2019). Evaluating the destabilization susceptibility of active rock glaciers in the French Alps. *The Cryosphere* 13, 141–155. <u>https://doi.org/10.5194/tc-13-141-2019</u>
- Marini, L., Ayres, M. P., Battisti, A., and Faccoli, M. (2012). Climate affects severity and altitudinal distribution of outbreaks in an eruptive bark beetle. *Climatic Change* 115, 327–341. <u>https://doi. org/10.1007/s10584-012-0463-z</u>

- Marke, T., Strasser, U., Hanzer, F., Stötter, J., Wilcke, R. A. I., and Gobiet, A. (2014). Scenarios of Future Snow Conditions in Styria (Austrian Alps). J. Hydrometeor. 16, 261–277. <u>https://doi.org/10.1175/JHM-D-14-0035.1</u>
- Marmy, A., Rajczak, J., Delaloye, R., Hilbich, C., Hoelzle, M., Kotlarski, S., et al. (2016). Semi-automated calibration method for modelling of mountain permafrost evolution in Switzerland. *The Cryosphere* 10, 2693–2719. <u>https://doi.org/10.5194/tc-10-2693-2016</u>
- Marty, C., Rohrer, M. B., Huss, M., and Stähli, M. (2023). Multidecadal observations in the Alps reveal less and wetter snow, with increasing variability. *Front. Earth Sci.* 11, 1165861. <u>https:// doi.org/10.3389/feart.2023.1165861</u>
- Marty, C., Schlögl, S., Bavay, M., and Lehning, M. (2017a). How much can we save? Impact of different emission scenarios on future snow cover in the Alps. *The Cryosphere* 11, 517–529. <u>https://doi.org/10.5194/tc-11-517-2017</u>
- Marty, C., Tilg, A.-M., and Jonas, T. (2017b). Recent Evidence of Large-Scale Receding Snow Water Equivalents in the European Alps. *Journal of Hydrometeorology* 18, 1021–1031. <u>https://doi.org/10.1175/JHM-D-16-0188.1</u>
- Mastrandrea, M. D., Field, C. B., Stocker, T. F., Edenhofer, O., Ebi, K. L., Frame, D. J., et al. (2010). Guidance Note for Lead Authors of the IPCC Fifth Assessment Report on Consistent Treatment of Uncertainties. Intergovernmental Panel on Climate Change (IPCC). Available at: <u>https://pure.mpg.de/rest/items/ item\_2147184/component/file\_2147185/content</u> (Accessed May 8, 2024).
- Matiu, M., Crespi, A., Bertoldi, G., Carmagnola, C. M., Marty, C., Morin, S., et al. (2021). Observed snow depth trends in the European Alps: 1971 to 2019. *The Cryosphere* 15, 1343–1382. https://doi.org/10.5194/tc-15-1343-2021
- Matiu, M., and Hanzer, F. (2022). Bias adjustment and downscaling of snow cover fraction projections from regional climate models using remote sensing for the European Alps. *Hydrol. Earth Syst. Sci.* 26, 3037–3054. https://doi.org/10.5194/hess-26-3037-2022
- Matulla, C., Feser, F., Tordai, J., Schlögl, M., Starke, H., Schöner, W., et al. (2020). "Stürme", in *ExtremA 2019. Aktueller Wissensstand zu Extremereignissen alpiner Naturgefahren in Österreich*, eds. T. Glade, M. Mergili, and K. Sattler (Vienna: Vienna University Press), 123–144. <u>https://library.oapen.org/handle/20.500.12657/43512</u>
- Matulla, C., Schöner, W., Alexandersson, H., von Storch, H., and Wang, X. L. (2008). European storminess: late nineteenth century to present. *Clim Dyn* 31, 125–130. <u>https://doi.org/10.1007/</u> <u>s00382-007-0333-y</u>
- Mayer, M., and Abegg, B. (2024). Development of summer skiing days in Austrian glacier ski areas in the first two decades of the twenty-first century. *International Journal of Biometeorology* 68, 547–564. <u>https://doi.org/10.1007/s00484-022-02371-6</u>
- McCoy, I. L., Vogt, M. A., and Wood, R. (2022). Absorbing Aerosol Choices Influences Precipitation Changes Across Future Scenarios. *Geophysical Research Letters* 49, e2022GL097717. <u>https://doi.org/10.1029/2022GL097717</u>
- Meehl, G. A., Stocker, T. F., Collins, W. D., Friedlingstein, P., Gaye, A. T., Gregory, J. M., et al. (2007). "Global climate projections. Chapter 10. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change", in *Climate Change 2007. The Physical Science Basis*, eds. S. Solomon, D. Quin, M. Manning, Z. Chen, M. Marquis,

K. B. Averyt, et al. (United Kingdom). Available at: <u>https://www.osti.gov/etdeweb/biblio/20962171</u> (Accessed May 3, 2024).

- Merz, R., Blöschl, G., and Humer, G. (2008). National flood discharge mapping in Austria. Nat Hazards 46, 53–72. <u>https://doi.org/10.1007/s11069-007-9181-7</u>
- Michel, A., Brauchli, T., Lehning, M., Schaefli, B., and Huwald, H. (2020). Stream temperature and discharge evolution in Switzerland over the last 50 years: annual and seasonal behaviour. *Hydrol. Earth Syst. Sci.* 24, 115–142. <u>https://doi.org/10.5194/ hess-24-115-2020</u>
- Michelutti, N., Wolfe, A. P., Cooke, C. A., Hobbs, W. O., Vuille, M., and Smol, J. P. (2015). Climate Change Forces New Ecological States in Tropical Andean Lakes. *PLoS ONE* 10, e0115338. https://doi.org/10.1371/journal.pone.0115338
- Mishra, A. N., Maraun, D., Knevels, R., Truhetz, H., Brenning, A., and Proske, H. (2023). Climate change amplified the 2009 extreme landslide event in Austria. *Climatic Change* 176, 124. https://doi.org/10.1007/s10584-023-03593-2
- Mishra, A. N., Maraun, D., Schiemann, R., Hodges, K., Zappa, G., and Ossó, A. (2025). Long-lasting intense cut-off lows to become more frequent in the Northern Hemisphere. *Commun Earth Environ* 6, 115. <u>https://doi.org/10.1038/s43247-025-02078-7</u>
- Moemken, J., Reyers, M., Feldmann, H., and Pinto, J. G. (2018). Future Changes of Wind Speed and Wind Energy Potentials in EURO-CORDEX Ensemble Simulations. *JGR Atmospheres* 123, 6373–6389. <u>https://doi.org/10.1029/2018JD028473</u>
- Möller, T., Högner, A. E., Schleussner, C.-F., Bien, S., Kitzmann, N. H., Lamboll, R. D., et al. (2024). Achieving net zero greenhouse gas emissions critical to limit climate tipping risks. *Nat Commun* 15, 6192. <u>https://doi.org/10.1038/s41467-024-49863-0</u>
- Monchamp, M.-E., Spaak, P., Domaizon, I., Dubois, N., Bouffard, D., and Pomati, F. (2017). Homogenization of lake cyanobacterial communities over a century of climate change and eutrophication. *Nat Ecol Evol* 2, 317–324. <u>https://doi.org/10.1038/</u> <u>s41559-017-0407-0</u>
- Morice, C. P., Kennedy, J. J., Rayner, N. A., Winn, J. P., Hogan, E., Killick, R. E., et al. (2021). An Updated Assessment of Near-Surface Temperature Change From 1850: The HadCRUT5 Data Set. *Geophys Res Atmos* 126. <u>https://doi.org/10.1029/2019JD032361</u>
- Müller, M. M., Vilardell, L. V., Vacik, H., Mayer, C., Mayr, S., Carrega, P., et al. (2020a). Forest fires in the Alps State of knowledge, future challenges and options for an integrated fire management White Paper for policy makers. <u>https://doi.org/10.13140/RG.</u> 2.2.15609.42081
- Müller, M. M., Vilà-Vilardell, L., and Vacik, H. (2020b). Towards an integrated forest fire danger assessment system for the European Alps. *Ecological Informatics* 60, 101151. <u>https://doi.org/</u> 10.1016/j.ecoinf.2020.101151
- Nabat, P., Somot, S., Mallet, M., Sanchez-Lorenzo, A., and Wild, M. (2014). Contribution of anthropogenic sulfate aerosols to the changing Euro-Mediterranean climate since 1980: NABAT ET AL. *Geophys. Res. Lett.* 41, 5605–5611. <u>https://doi.org/10. 1002/2014GL060798</u>
- Nachtnebel, H. P., Dokulil, M., Kuhn, M., Loiskandl, W., Sailer, R., and Schöner, W. (2014). "Der Einfluss des Klimawandels auf die Hydrosphäre", in Österreichischer Sachstandsbericht Klimawandel 2014 (AAR14), ed. Austrian Panel on Climate Change (APCC) (Vienna: Verlag der Österreichischen Akademie der Wissenschaften), 411–465. <u>https://doi.org/10.1553/aar14</u>
- Nagy, L., and Grabherr, G. (2009). *The biology of alpine habitats*. Oxford: Oxford University Press.

- Nakićenović, N., Alcamo, J., Davis, G., De Vries, B., Fenhann, J., Gaffin, S., et al. (2000). Special Report on Emissions Scenarios: a special report of Working Group III of the Intergovernmental Panel on Climate Change., ed. Intergovernmental Panel on Climate Change. Cambridge; New York: Cambridge University Press. https://www.ipcc.ch/site/assets/uploads/2018/03/sres-en.pdf
- Neff, F., Korner-Nievergelt, F., Rey, E., Albrecht, M., Bollmann, K., Cahenzli, F., et al. (2022). Different roles of concurring climate and regional land-use changes in past 40 years' insect trends. *Nat Commun* 13, 7611. <u>https://doi.org/10.1038/s41467-022-35223-3</u>
- Nemeth, E., Auer, I., Hollosi, B., and Teufelbauer, N. (2016). "Ein Klima-Einfluss-Index für die Brutvögel Österreichs. Endbericht von StartClim2015.C", in StartClim2015: Weitere Beiträge zur Umsetzung der österreichischen Anpassungsstrategie. Im Auftrag von BMLFUW, BMWF, ÖBf und Land Oberösterreich, (Vienna, Austria: BirdLife Österreich und Zentralanstalt für Meteorologie und Geodynamik, Wien). Available at: https://startclim.at/ fileadmin/user\_upload/StartClim2015/StCl2015C\_lang.pdf
- Ng, G.-H. C., McLaughlin, D., Entekhabi, D., and Scanlon, B. R. (2010). Probabilistic analysis of the effects of climate change on groundwater recharge. *Water Resources Research* 46. <u>https://doi. org/10.1029/2009WR007904</u>
- Niedrist, G. H. (2023). Substantial warming of Central European mountain rivers under climate change. *Reg Environ Change* 23, 43. <u>https://doi.org/10.1007/s10113-023-02037-y</u>
- Nikulin, G., Kjellström, E., Hansson, U., Strandberg, G., and Ullerstig, A. (2011). Evaluation and future projections of temperature, precipitation and wind extremes over Europe in an ensemble of regional climate simulations. *Tellus A: Dynamic Meteorology and Oceanography* 63, 41. <u>https://doi.org/10.1111/j.1600-0870.2010.00466.x</u>
- Nocker, C., and Laaha, G. (2019). Evaluation of statistical methods for pluvial flood risk assessment. Available at: <u>https://rainman-toolbox.eu/wp-content/uploads/2020/06/RAINMAN</u> Pluvio-Flood Final-Report.pdf (Accessed September 15, 2023).
- Ohler, L.-M., Lechleitner, M., and Junker, R. R. (2020). Microclimatic effects on alpine plant communities and flower-visitor interactions. *Sci Rep* 10, 1366. <u>https://doi.org/10.1038/s41598-020-58388-7</u>
- Oke, T. R., Mills, G., Christen, A., and Voogt, J. A. (2017). Urban Climates., 1st Edn. Cambridge University Press. Available at: <u>https://doi.org/10.1017/9781139016476</u> (Accessed September 1, 2023).
- Oki, T., and Kanae, S. (2006). Global hydrological cycles and world water resources. *Science* 313, 1068–1072. <u>https://doi.org/10.1126/science.1128845</u>
- Olefs, M., Baumgartner, D. J., Obleitner, F., Bichler, C., Foelsche, U., Pietsch, H., et al. (2016). The Austrian radiation monitoring network ARAD – best practice and added value. *Atmospheric Measurement Techniques* 9, 1513–1531. <u>https://doi.org/10.5194/</u> amt-9-1513-2016
- Olefs, M., Formayer, H., Gobiet, A., Marke, T., Schöner, W., and Revesz, M. (2021). Past and future changes of the Austrian climate – Importance for tourism. *Journal of Outdoor Recreation* and Tourism 34, 100395. <u>https://doi.org/10.1016/j.jort.2021.</u> <u>100395</u>
- O'Reilly, C. M., Sharma, S., Gray, D. K., Hampton, S. E., Read, J. S., Rowley, R. J., et al. (2015). Rapid and highly variable warming of lake surface waters around the globe. *Geophysical Research Letters* 42, 10,773-10,781. <u>https://doi.org/10.1002/2015GL066235</u>

- Ossó, A., and Roither, L. (2024). Kipppunkte im Klimasystem. CCCA Factsheet #47. Available at: <u>https://ccca.ac.at/fileadmin/00\_DokumenteHauptmenue/02\_Klimawissen/Fact-Sheets/47\_kipppunkte\_klimasystem\_202401.pdf</u>
- Ostermann, M., Ivy-Ochs, S., Sanders, D., and Prager, C. (2017). Multi-method (<sup>14</sup>C, <sup>36</sup>Cl, <sup>234</sup>U/ <sup>230</sup>Th) age bracketing of the Tschirgant rock avalanche (Eastern Alps): implications for absolute dating of catastrophic mass-wasting: Onto the age of Tschirgant rock avalanche. *Earth Surf. Process. Landforms* 42, 1110–1118. https://doi.org/10.1002/esp.4077
- Oswald, P., Strasser, M., Hammerl, C., and Moernaut, J. (2021). Seismic control of large prehistoric rockslides in the Eastern Alps. *Nat Commun* 12, 1059. <u>https://doi.org/10.1038/s41467-021-21327-9</u>
- Otto, J.-C., Krautblatter, M., and Sattler, K. (2020). "Permafrostgefahren.", in *ExtremA 2019. Aktueller Wissensstand zu Extremereignissen alpiner Naturgefahren in Österreich.*, (Vienna: Vienna University Press), 537–561.
- Outten, S., and Sobolowski, S. (2021). Extreme wind projections over Europe from the Euro-CORDEX regional climate models. Weather and Climate Extremes 33, 100363. <u>https://doi.org/10.1016/j.wace.2021.100363</u>
- Palazzi, E., Mortarini, L., Terzago, S., and Von Hardenberg, J. (2019). Elevation-dependent warming in global climate model simulations at high spatial resolution. *Clim Dyn* 52, 2685–2702. <u>https://doi.org/10.1007/s00382-018-4287-z</u>
- Paranunzio, R., Chiarle, M., Laio, F., Nigrelli, G., Turconi, L., and Luino, F. (2019). New insights in the relation between climate and slope failures at high-elevation sites. *Theor Appl Climatol* 137, 1765–1784. <u>https://doi.org/10.1007/s00704-018-2673-4</u>
- Patacca, M., Lindner, M., Lucas-Borja, M. E., Cordonnier, T., Fidej, G., Gardiner, B., et al. (2023). Significant increase in natural disturbance impacts on European forests since 1950. *Global Change Biology* 29, 1359–1376. <u>https://doi.org/10.1111/gcb.16531</u>
- Peña-Angulo, D., Vicente-Serrano, S. M., Domínguez-Castro, F., Lorenzo-Lacruz, J., Murphy, C., Hannaford, J., et al. (2022). The Complex and Spatially Diverse Patterns of Hydrological Droughts Across Europe. *Water Resources Research* 58, e2022WR031976. https://doi.org/10.1029/2022WR031976
- Pepin, N. C., Arnone, E., Gobiet, A., Haslinger, K., Kotlarski, S., Notarnicola, C., et al. (2022). Climate Changes and Their Elevational Patterns in the Mountains of the World. *Reviews of Geophysics* 60, e2020RG000730. <u>https://doi.org/10.1029/2020RG000730</u>
- Persad, G. G., Samset, B. H., and Wilcox, L. J. (2022). Aerosols must be included in climate risk assessments. *Nature* 611, 662–664. <u>https://doi.org/10.1038/d41586-022-03763-9</u>
- Petersen, U., and Weigel, H.-J. (2015). Klimaresilienz durch Agrobiodiversität? Literaturstudie zum Zusammenhang zwischen Elementen der Agrobiodiversität und der Empfindlichkeit von landwirtschaftlichen Produktionssystemen gegenüber dem Klimawandel. Braunschweig: Johann Heinrich von Thünen-Institut. Available at: <u>https://doi.org/10.3220/REP\_25\_2015</u> (Accessed September 20, 2023).
- Pfahl, S., O'Gorman, P. A., and Fischer, E. M. (2017). Understanding the regional pattern of projected future changes in extreme precipitation. *Nature Clim Change* 7, 423–427. <u>https://doi. org/10.1038/nclimate3287</u>
- Pfeiffer, J., Zieher, T., Schmieder, J., Rutzinger, M., and Strasser, U. (2021). Spatio-temporal assessment of the hydrological drivers of an active deep-seated gravitational slope deformation: The

Vögelsberg landslide in Tyrol (Austria). *Earth Surf Processes Landf* 46, 1865–1881. <u>https://doi.org/10.1002/esp.5129</u>

- Pfeifroth, U., Bojanowski, J. S., Clerbaux, N., Manara, V., Sanchez-Lorenzo, A., Trentmann, J., et al. (2018). Satellite-based trends of solar radiation and cloud parameters in Europe. *Advances in Science and Research* 15, 31–37. <u>https://doi. org/10.5194/asr-15-31-2018</u>
- Pfurtscheller, C. (2014). Regional economic impacts of natural hazards – the case of the 2005 Alpine flood event in Tyrol (Austria). *Natural Hazards and Earth System Sciences* 14, 359–378. <u>https:// doi.org/10.5194/nhess-14-359-2014</u>
- Philipona, R. (2013). Greenhouse warming and solar brightening in and around the Alps. *Int. J. Climatol.* 33, 1530–1537. <u>https:// doi.org/10.1002/joc.3531</u>
- Philipona, R., Behrens, K., and Ruckstuhl, C. (2009). How declining aerosols and rising greenhouse gases forced rapid warming in Europe since the 1980s: AEROSOL VERSUS GREEN-HOUSE GAS FORCING. *Geophys. Res. Lett.* 36. <u>https://doi.org/</u> 10.1029/2008GL036350
- Philipona, R., Marty, C., Duerr, B., and Ohmura, A. (2023). Rising solar and thermal greenhouse radiation drive rapid warming over continents. *metz* 32, 5–14. <u>https://doi.org/10.1127/metz/</u> 2022/1148
- Pichelli, E., Coppola, E., Sobolowski, S., Ban, N., Giorgi, F., Stocchi, P., et al. (2021). The first multi-model ensemble of regional climate simulations at kilometer-scale resolution part 2: historical and future simulations of precipitation. *Climate Dynamics* 56, 3581–3602. <u>https://doi.org/10.1007/s00382-021-05657-4</u>
- Pils, G. (1994). Die Wiesen Oberösterreichs: eine Naturgeschichte des oberösterreichischen Grünlandes unter besonderer Berücksichtigung von Naturschutzaspekten., 1. Aufl. Linz: Forschungsinstitut für Umweltinformatik.
- Piroton, V., Emmer, A., Schlögel, R., Hřebřina, J., Pummer, E., Mergili, M., et al. (2024). Geomorphological processes and landforms in the Alpine Sulzenau Valley (Tyrol, Austria): Glacier retreat, glacial lake evolution and the 2017 glacial lake outburst flood. *Earth Surf Processes Landf* 49, 4823–4841. <u>https://doi.org/</u> 10.1002/esp.5956
- Pistotnik, G., Hofstätter, M., and Lexer, A. (2020). "Starkniederschlag und Hagel", in *ExtremA 2019. Aktueller Wissensstand* zu Extremereignissen alpiner Naturgefahren in Österreich, eds. T. Glade, M. Mergili, and K. Sattler (Vienna: Vienna University Press), 141–172.
- Pläsken, R., Keuschnig, M., and Krautblatter, M. (2020). Permafrost rocks and high-alpine infrastructure: Interrelated, interconnected, interacting. *Geomechanics and Tunnelling* 13, 628– 633. <u>https://doi.org/10.1002/geot.202000028</u>
- Pletterbauer, F., Melcher, A., and Graf, W. (2018). "Climate change impacts in riverine ecosystems", in *Riverine Ecosystem Management. Aquatic Ecology Series*, eds. S. Schmutz and J. Sendzimir (Cham: Springer), 203–223. Available at: <u>https://doi.org/10. 1007/978-3-319-73250-3\_11</u> (Accessed November 19, 2024).
- Poisel, R., Hofmann, R., Preh, A., Sausgruber, T., and Schiffer, M. (2012). Gschliefgraben mudslide (Austria): hazard evaluation and risk mitigation. *Nat Hazards* 61, 267–279. <u>https://doi. org/10.1007/s11069-011-9865-x</u>
- Popy, S., Bordignon, L., and Prodon, R. (2009). A weak upward elevational shift in the distributions of breeding birds in the Italian Alps: Elevational shift in Alpine bird distributions. *Journal of Biogeography* 37, 57–67. <u>https://doi.org/10.1111/j.1365-2699.2009.02197.x</u>

- Prager, C., Zangerl, C., Patzelt, G., and Brandner, R. (2008). Age distribution of fossil landslides in the Tyrol (Austria) and its surrounding areas. *Nat. Hazards Earth Syst. Sci.* 8, 377–407. <u>https://doi.org/10.5194/nhess-8-377-2008</u>
- Púčik, T., Groenemeijer, P., Rädler, A. T., Tijssen, L., Nikulin, G., Prein, A. F., et al. (2017). Future Changes in European Severe Convection Environments in a Regional Climate Model Ensemble. J. Climate 30, 6771–6794. <u>https://doi.org/10.1175/JCLI--D-16-0777.1</u>
- Rab, G., Krammer, C., Brunner, T., Schmaltz, E., Széles, B., Blöschl, G., et al. (2023). Analysis of erosion drivers in the Hydrological Open Air Laboratory Petzenkirchen. <u>https://doi.org/10.5194/</u> egusphere-egu23-7340
- Rädler, A. T., Groenemeijer, P., Faust, E., and Sausen, R. (2018). Detecting Severe Weather Trends Using an Additive Regressive Convective Hazard Model (AR-CHaMo). *Journal of Applied Meteorology and Climatology* 57, 569–587. <u>https://doi.org/10.1175/</u> <u>JAMC-D-17-0132.1</u>
- Rajczak, J., and Schär, C. (2017). Projections of future precipitation extremes over Europe: A multimodel assessment of climate simulations. *Journal of Geophysical Research: Atmospheres* 122, 10773–10800. <u>https://doi.org/10.1002/2017JD027176</u>
- Ramsar (1971). Übereinkommen über Feuchtgebiete, insbesondere als Lebensraum für Wasser- und Watvögel, von internationaler Bedeutung. Available at: <u>https://www.ramsar.org/sites/</u> <u>default/files/documents/library/scan\_certified\_g.pdf</u> (Accessed September 21, 2023).
- Ravanel, L., Magnin, F., and Deline, P. (2017). Impacts of the 2003 and 2015 summer heatwaves on permafrost-affected rock-walls in the Mont Blanc massif. *Science of The Total Environment* 609, 132–143. https://doi.org/10.1016/j.scitotenv.2017.07.055
- Rechberger, C., and Zangerl, C. (2022). Rock mass characterisation and distinct element modelling of a deep-seated rock slide influenced by glacier retreat. *Engineering Geology* 300, 106584. <u>https://doi.org/10.1016/j.enggeo.2022.106584</u>
- Rehbein, J. A., Watson, J. E. M., Lane, J. L., Sonter, L. J., Venter, O., Atkinson, S. C., et al. (2020). Renewable energy development threatens many globally important biodiversity areas. *Global Change Biology* 26, 3040–3051. <u>https://doi.org/10.1111/ gcb.15067</u>
- Rehnus, M., Bollmann, K., Schmatz, D. R., Hackländer, K., and Braunisch, V. (2018). Alpine glacial relict species losing out to climate change: The case of the fragmented mountain hare population (Lepus timidus) in the Alps. *Global Change Biology* 24, 3236–3253. <u>https://doi.org/10.1111/gcb.14087</u>
- Rehnus, M., Peláez, M., and Bollmann, K. (2020). Advancing plant phenology causes an increasing trophic mismatch in an income breeder across a wide elevational range. *Ecosphere* 11, e03144. https://doi.org/10.1002/ecs2.3144
- Reich, D., Barta, T., Pils, P., and Sander, R. (2018). Beiträge Zur Kenntnis Der Gattung Vulpia (Poaceae) In Österreich Mit Besonderer Berücksichtigung Von Vulpia Ciliata, Neu Für Wien Und Niederösterreich. *Neilreichia* 9, 247–267. <u>https://doi.org/ 10.5281/ZENODO.1196421</u>
- Reif, J., Voříšek, P., Šťastný, K., Koschová, M., and Bejček, V. (2008). The impact of climate change on long-term population trends of birds in a central European country. *Animal Conservation* 11, 412–421. <u>https://doi.org/10.1111/j.1469-1795.2008.00200.x</u>
- Reiner, R., Zedrosser, A., Zeiler, H., Hackländer, K., and Corlatti, L. (2021). Forests buffer the climate-induced decline of body mass

in a mountain herbivore. *Global Change Biology* 27, 3741–3752. https://doi.org/10.1111/gcb.15711

- Reusch, C., Paul, A. A., Fritze, M., Kramer-Schadt, S., and Voigt, C. C. (2023). Wind energy production in forests conflicts with tree-roosting bats. *Current Biology* 33, 737-743.e3. <u>https://doi.org/10.1016/j.cub.2022.12.050</u>
- Revermann, R., Schmid, H., Zbinden, N., Spaar, R., and Schröder, B. (2012). Habitat at the mountain tops: how long can Rock Ptarmigan (Lagopus muta helvetica) survive rapid climate change in the Swiss Alps? A multi-scale approach. J Ornithol 153, 891–905. https://doi.org/10.1007/s10336-012-0819-1
- Riahi, K., Van Vuuren, D. P., Kriegler, E., Edmonds, J., O'Neill, B. C., Fujimori, S., et al. (2017). The Shared Socioeconomic Pathways and their energy, land use, and greenhouse gas emissions implications: An overview. *Global Environmental Change* 42, 153–168. https://doi.org/10.1016/j.gloenvcha.2016.05.009
- Rianna, G., Zollo, A., Tommasi, P., Paciucci, M., Comegna, L., and Mercogliano, P. (2014). Evaluation of the Effects of Climate Changes on Landslide Activity of Orvieto Clayey Slope. *Procedia Earth and Planetary Science* 9, 54–63. <u>https://doi.org/ 10.1016/j.proeps.2014.06.017</u>
- Ritzhaupt, N., and Maraun, D. (2023). Consistency of Seasonal Mean and Extreme Precipitation Projections Over Europe Across a Range of Climate Model Ensembles. *JGR Atmospheres* 128, e2022JD037845. <u>https://doi.org/10.1029/2022JD037845</u>
- Ritzhaupt, N., and Maraun, D. (2024). State-of-the-art climate models reduce dominant dynamical uncertainty in projections of extreme precipitation. *Environ. Res.: Climate* 3, 021001. <u>https://doi.org/10.1088/2752-5295/ad2eb2</u>
- Robinson, A., Lehmann, J., Barriopedro, D., Rahmstorf, S., and Coumou, D. (2021). Increasing heat and rainfall extremes now far outside the historical climate. *npj Clim Atmos Sci* 4, 45. <u>https://doi.org/10.1038/s41612-021-00202-w</u>
- Rohn, J., Ehret, D., Moser, M., and Czurda, K. (2005). Prehistoric and recent mass movements of the World Cultural Heritage Site Hallstatt, Austria. *Environ Geol* 47, 702–714. <u>https://doi.org/10.1007/s00254-004-1201-2</u>
- Rohn, J., Resch, M., Schneider, H., Fernandez-Steeger, T. M., and Czurda, K. (2004). Large-scale lateral spreading and related mass movements in the Northern Calcareous Alps. *Bulletin of Engineering Geology and the Environment* 63, 71–75. <u>https://doi.org/10.1007/s10064-003-0201-x</u>
- Rosbakh, S., Bernhardt-Römermann, M., and Poschlod, P. (2014). Elevation matters: contrasting effects of climate change on the vegetation development at different elevations in the Bavarian Alps. *Alp Botany* 124, 143–154. <u>https://doi.org/10.1007/s00035-014-0139-6</u>
- Rosset, V., Lehmann, A., and Oertli, B. (2010). Warmer and richer? Predicting the impact of climate warming on species richness in small temperate waterbodies: POND SPECIES RICHNESS UNDER CLIMATE WARMING. *Global Change Biology* 16, 2376–2387. <u>https://doi.org/10.1111/j.1365-2486.2010.02206.x</u>
- Rounce, D. R., Hock, R., Maussion, F., Hugonnet, R., Kochtitzky, W., Huss, M., et al. (2023). Global glacier change in the 21st century: Every increase in temperature matters. *Science* 379, 78–83. <u>https://doi.org/10.1126/science.abo1324</u>
- Ruane, A. C., Vautard, R., Ranasinghe, R., Sillmann, J., Coppola, E., Arnell, N., et al. (2022). The Climatic Impact-Driver Framework for Assessment of Risk-Relevant Climate Information. *Earth's Future* 10, e2022EF002803. <u>https://doi.org/10.1029/2022</u> EF002803

- Rubel, F., Brugger, K., Haslinger, K., and Auer, I. (2017). The climate of the European Alps: Shift of very high resolution Köppen-Geiger climate zones 1800–2100. *metz* 26, 115–125. <u>https://</u> doi.org/10.1127/metz/2016/0816
- Rugenstein, M., Bloch-Johnson, J., Abe-Ouchi, A., Andrews, T., Beyerle, U., Cao, L., et al. (2019). LongRunMIP: Motivation and Design for a Large Collection of Millennial-Length AOGCM Simulations. *Bulletin of the American Meteorological Society* 100, 2551–2570. <u>https://doi.org/10.1175/BAMS-D-19-0068.1</u>
- Rumpf, S. B., Hülber, K., Klonner, G., Moser, D., Schuetz, M., Wessely, J., et al. (2018). Range dynamics of mountain plants decrease with elevation. *Proceedings of the National Academy of Sciences of the USA* 115, 1848–1853. <u>https://doi.org/10.1073/pnas.1713936115</u>
- Rumpf, S. B., Hülber, K., Wessely, J., Willner, W., Moser, D., Gattringer, A., et al. (2019). Extinction debts and colonization credits of non-forest plants in the European Alps. *Nature Communications* 10, 4293. <u>https://doi.org/10.1038/s41467-019-12343-x</u>
- Saffioti, C., Fischer, E. M., Scherrer, S. C., and Knutti, R. (2016). Reconciling observed and modeled temperature and precipitation trends over Europe by adjusting for circulation variability: CIRCULATION AFFECTING TRENDS OVER EU-ROPE. Geophysical Research Letters 43, 8189–8198. <u>https://doi. org/10.1002/2016GL069802</u>
- Saino, N., Rubolini, D., Lehikoinen, E., Sokolov, L. V., Bonisoli-Alquati, A., Ambrosini, R., et al. (2009). Climate change effects on migration phenology may mismatch brood parasitic cuckoos and their hosts. *Biol. Lett.* 5, 539–541. <u>https://doi.org/10.1098/ rsbl.2009.0312</u>
- Salguero, M. D. M., De La Cruz, A., Muñoz, A.-R., and Muñoz Arroyo, G. (2023). Bat mortality in wind farms of southern Europe: temporal patterns and implications in the current context of climate change. *Biodivers Conserv* 32, 3953–3971. <u>https://doi.org/10.1007/s10531-023-02674-z</u>
- Samaniego, L., Thober, S., Kumar, R., Wanders, N., Rakovec, O., Pan, M., et al. (2018). Anthropogenic warming exacerbates European soil moisture droughts. *Nature Clim Change* 8, 421–426. <u>https://doi.org/10.1038/s41558-018-0138-5</u>
- Sass, O., and Oberlechner, M. (2012). Is climate change causing increased rockfall frequency in Austria? *Natural Hazards and Earth System Sciences* 12, 3209–3216. <u>https://doi.org/10.5194/</u> nhess-12-3209-2012
- Sausgruber, T., Marschallinger, R., and Scheuringer, E. (2004). The Fürwag landslide – Models, monitoring and measures. *Felsbau* 22, 51–59.
- Schai-Braun, S. C., Jenny, H., Ruf, T., and Hackländer, K. (2021). Temperature increase and frost decrease driving upslope elevational range shifts in Alpine grouse and hares. *Global Change Biology* 27, 6602–6614. <u>https://doi.org/10.1111/gcb.15909</u>
- Schano, C., Niffenegger, C., Jonas, T., and Korner-Nievergelt, F. (2021). Hatching phenology is lagging behind an advancing snowmelt pattern in a high-alpine bird. *Sci Rep* 11, 22191. <u>https://doi.org/10.1038/s41598-021-01497-8</u>
- Scharnhorst, V. S., Thierolf, K., Neumayer, J., Becsi, B., Formayer, H., Lanner, J., et al. (2023). Changes in Community Composition and Functional Traits of Bumblebees in an Alpine Ecosystem Relate to Climate Warming. *Biology* 12, 316. <u>https://doi. org/10.3390/biology12020316</u>
- Scheidl, C., Heiser, M., Kamper, S., Thaler, T., Klebinder, K., Nagl, F., et al. (2020). The influence of climate change and canopy disturbances on landslide susceptibility in headwater catch-

ments. Science of The Total Environment 742, 140588. <u>https://doi.org/10.1016/j.scitotenv.2020.140588</u>

- Scherrer, D., and Körner, C. (2011). Topographically controlled thermal-habitat differentiation buffers alpine plant diversity against climate warming: Topographical control of thermal-habitat differentiation buffers alpine plant diversity. *Journal of Biogeography* 38, 406–416. <u>https://doi.org/10.1111/j.1365-2699.</u> 2010.02407.x
- Scherrer, S. C., and Begert, M. (2019). Effects of large-scale atmospheric flow and sunshine duration on the evolution of minimum and maximum temperature in Switzerland. *Theoretical* and Applied Climatology 138, 227–235. <u>https://doi.org/10.1007/ s00704-019-02823-x</u>
- Schilliger, L., Tetzlaff, A., Bourgeois, Q., Correa, L. F., and Wild, M. (2024). An investigation on causes of the detected surface solar radiation brightening in Europe using satellite data. <u>https://doi. org/10.22541/essoar.171136948.88001430/v1</u>
- Schindlbacher, A., Schnecker, J., Takriti, M., Borken, W., and Wanek, W. (2015). Microbial physiology and soil CO<sub>2</sub> efflux after 9 years of soil warming in a temperate forest – no indications for thermal adaptations. *Global Change Biology* 21, 4265–4277. https://doi.org/10.1111/gcb.12996
- Schindler, D. E. (2017). Warmer climate squeezes aquatic predators out of their preferred habitat. *Proceedings of the National Academy of Sciences* 114, 9764–9765. <u>https://doi.org/10.1073/ pnas.1712818114</u>
- Schlögl, M., Fuchs, S., Scheidl, C., and Heiser, M. (2021). Trends in torrential flooding in the Austrian Alps: A combination of climate change, exposure dynamics, and mitigation measures. *Climate Risk Management* 32, 100294. <u>https://doi.org/10.1016/j. crm.2021.100294</u>
- Schmidt, J., and Dikau, R. (2004). Modeling historical climate variability and slope stability. *Geomorphology* 60, 433–447. <u>https:// doi.org/10.1016/j.geomorph.2003.11.001</u>
- Schöner, W., Koch, R., Matulla, C., Marty, C., and Tilg, A.-M. (2019). Spatiotemporal patterns of snow depth within the Swiss-Austrian Alps for the past half century (1961 to 2012) and linkages to climate change. *International Journal of Climatology* 39, 1589–1603. https://doi.org/10.1002/joc.5902
- Schratt-Ehrendorfer, L., Niklfeld, H., Schröck, C., and Stöhr, O. (2022). Rote Liste der Farn- und Blütenpflanzen Österreichs. Dritte, völlig neu bearbeitete Auflage. Available at: <u>https://www. zobodat.at/publikation\_articles.php?id=509508</u> (Accessed November 7, 2024).
- Schroeer, K., and Kirchengast, G. (2018). Sensitivity of extreme precipitation to temperature: the variability of scaling factors from a regional to local perspective. *Clim Dyn* 50, 3981–3994. https://doi.org/10.1007/s00382-017-3857-9
- Schrott, L., Otto, J. C., and Keller, F. (2012). Modelling alpine permafrost distribution in the Hohe Tauern region, Austria. *Austrian Journal of Earth Sciences* 105, 169–183. Available at: <u>http://www.parcs.at/npa/pdf\_public/2018/36019\_20180315\_101924\_NP\_Hohe\_Tauern\_Conference\_5\_0697-0702.pdf</u>
- Schübl, M. A. (2023). Groundwater recharge estimation and uncertainty analysis. Vienna. Available at: <u>https://permalink.obvsg.at/</u> <u>bok/AC16970259</u>
- Schübl, M., Brunetti, G., Fuchs, G., and Stumpp, C. (2023). Estimating vadose zone water fluxes from soil water monitoring data: a comprehensive field study in Austria. *Hydrol. Earth Syst. Sci.* 27, 1431–1455. https://doi.org/10.5194/hess-27-1431-2023

- Schulz, W., Cummins, K., Diendorfer, G., and Dorninger, M. (2005). Cloud-to-ground lightning in Austria: A 10-year study using data from a lightning location system. *J. Geophys. Res.* 110, D09101. <u>https://doi.org/10.1029/2004JD005332</u>
- Schumacher, D. L., Singh, J., Hauser, M., Fischer, E. M., Wild, M., and Seneviratne, S. I. (2024). Exacerbated summer European warming not captured by climate models neglecting long-term aerosol changes. *Commun Earth Environ* 5, 182. <u>https://doi. org/10.1038/s43247-024-01332-8</u>
- Seidl, R., Klonner, G., Rammer, W., Essl, F., Moreno, A., Neumann, M., et al. (2018). Invasive alien pests threaten the carbon stored in Europe's forests. *Nat Commun* 9, 1626. <u>https://doi.org/10.1038/s41467-018-04096-w</u>
- Seidl, R., Schelhaas, M.-J., Lindner, M., and Lexer, M. J. (2009). Modelling bark beetle disturbances in a large scale forest scenario model to assess climate change impacts and evaluate adaptive management strategies. *Reg Environ Change* 9, 101–119. <u>https:// doi.org/10.1007/s10113-008-0068-2</u>
- Seidl, R., Thom, D., Kautz, M., Martin-Benito, D., Peltoniemi, M., Vacchiano, G., et al. (2017). Forest disturbances under climate change. *Nature Climate Change* 7, 395–402. <u>https://doi. org/10.1038/nclimate3303</u>
- Seitner, V., Chimani, B., and Haimberger, L. (2023). Comparison of Break Point Detection Methods for global radiation in Austria. Available at: <u>https://www.met.hu/downloads.php?fn=/en/ omsz/rendezvenyek/homogenization\_and\_interpolation\_2023/ downloads/posters/Poster\_Seitner.pdf</u>
- Semenchuk, P., Moser, D., Essl, F., Schindler, S., Wessely, J., Gattringer, A., et al. (2021). Future Representation of Species' Climatic Niches in Protected Areas: A Case Study With Austrian Endemics. *Frontiers in Ecology and Evolution* 9, 685753. <u>https:// doi.org/10.3389/fevo.2021.685753</u>
- Seneviratne, S. I., Corti, T., Davin, E. L., Hirschi, M., Jaeger, E. B., Lehner, I., et al. (2010). Investigating soil moisture–climate interactions in a changing climate: A review. *Earth-Science Reviews* 99, 125–161. <u>https://doi.org/10.1016/j.earscirev.2010.02.004</u>
- Seneviratne, S. I., Lüthi, D., Litschi, M., and Schär, C. (2006). Landatmosphere coupling and climate change in Europe. *Nature* 443, 205–209. <u>https://doi.org/10.1038/nature05095</u>
- Seneviratne, S. I., Rogelj, J., Séférian, R., Wartenburger, R., Allen, M. R., Cain, M., et al. (2018a). The many possible climates from the Paris Agreement's aim of 1.5 °C warming. *Nature* 558, 41– 49. <u>https://doi.org/10.1038/s41586-018-0181-4</u>
- Seneviratne, S. I., Wartenburger, R., Guillod, B. P., Hirsch, A. L., Vogel, M. M., Brovkin, V., et al. (2018b). Climate extremes, land– climate feedbacks and land-use forcing at 1.5°C. *Phil. Trans. R. Soc. A.* 376, 20160450. <u>https://doi.org/10.1098/rsta.2016.0450</u>
- Seneviratne, S. I., Zhang, X., Adnan, M., Badi, W., Dereczynski, C., Di Luca, A., et al. (2021). "Weather and Climate Extreme Events in a Changing Climate", in *Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*, eds. V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press), 1513–1766. <u>https://doi.org/10.1017/9781009157896.013</u>
- Sfică, L., Beck, C., Nita, A., Voiculescu, M., Birsan, M., and Philipp, A. (2021). Cloud cover changes driven by atmospheric circulation in Europe during the last decades. *Int J Climatol* 41. <u>https:// doi.org/10.1002/joc.6841</u>

- Shepherd, T. G. (2014). Atmospheric circulation as a source of uncertainty in climate change projections. *Nature Geosci* 7, 703– 708. <u>https://doi.org/10.1038/ngeo2253</u>
- Simon, T., Mayr, G. J., Morgenstern, D., Umlauf, N., and Zeileis, A. (2023). Amplification of annual and diurnal cycles of alpine lightning. *Clim Dyn* 61, 4125–4137. <u>https://doi.org/10.1007/ s00382-023-06786-8</u>
- Solheim, A. L., Austnes, K., Eriksen, T. E., Seifert, I., and Holen, S. (2010). Climate change impacts on water quality and biodiversity. Background Report for EEA European Environment State and Outlook Report 2010. Prague: ETC Water. Available at: <u>https://citeseerx.ist.psu.edu/document?repid=rep1&type=pdf</u> <u>&doi=124ffa0d61a4661f33475bda55e1e82b435c365d</u> (Accessed November 19, 2024).
- Sommaruga, R., Dechent, E., Larsen, G., Gómez, S. M., and Weniger, E. (2023). Limnologische Untersuchungen am Piburger See
  Bericht für 2022. Innsbruck: Universität Innsbruck, Institut für Ökologie.
- Sommer, C., Malz, P., Seehaus, T. C., Lippl, S., Zemp, M., and Braun, M. H. (2020). Rapid glacier retreat and downwasting throughout the European Alps in the early 21st century. *Nature Communications* 11, 3209. <u>https://doi.org/10.1038/s41467-020-16818-0</u>
- Sotirov, M., Jonsson, R., Kleinschmit von Lengefeld, A. N., Krasovskiy, A., Kraxner, F., Lexer, M. J., et al. (2024). Europe's wood supply in disruptive times. An evidence-based synthesis report. Vienna, Austria: International Union of Forest Research Organizations (IUFRO). Available at: <u>https://pure.iiasa.ac.at/id/ eprint/19557/</u>
- Spinoni, J., Vogt, J. V., Naumann, G., Barbosa, P., and Dosio, A. (2018). Will drought events become more frequent and severe in Europe? *Intl Journal of Climatology* 38, 1718–1736. <u>https:// doi.org/10.1002/joc.5291</u>
- Stahl, K., Vidal, J.-P., Hannaford, J., Tijdeman, E., Laaha, G., Gauster, T., et al. (2020). The challenges of hydrological drought definition, quantification and communication: an interdisciplinary perspective. *Proc. IAHS* 383, 291–295. <u>https://doi.org/ 10.5194/piahs-383-291-2020</u>
- Stampfli, A., and Zeiter, M. (2004). Plant regeneration directs changes in grassland composition after extreme drought: a 13year study in southern Switzerland: *Plant regeneration directs changes. Journal of Ecology* 92, 568–576. <u>https://doi.org/10.1111/j.0022-0477.2004.00900.x</u>
- Stangl, M., Formayer, H., Hiebl, J., Orlik, A., Hinger, D., Bauer, C., et al. (2023). Klimastatusbericht Österreich 2022. Wien: CCCA. Available at: <u>https://ccca.ac.at/fileadmin/00\_Dokumente-Hauptmenue/02\_Klimawissen/Klimastatusbericht/KSB\_2022/ KBS\_2022\_BF/Klimastatusbericht\_OE\_2022\_BF.pdf</u> (Accessed September 29, 2023).
- Stangl, M., Formayer, H., Hiebl, J., Orlik, A., Höfler, A., Kalcher, M., et al. (2021). Klimastatusbericht Österreich 2020. Graz: CCCA. Available at: <u>https://ccca.ac.at/fileadmin/00\_DokumenteHauptmenue/02\_Klimawissen/Klimastatusbericht/Klimastatusbericht\_OEsterreich\_2020.pdf</u> (Accessed September 29, 2023).
- Statistik Austria (2023). Demographisches Jahrbuch 2021. Wien: STATISTIK AUSTRIA. Available at: <u>https://www.statistik.at/ser-vices/tools/services/publikationen/detail/1517</u> (Accessed September 29, 2023).
- Staude, I. R., Pereira, H. M., Daskalova, G. N., Bernhardt-Römermann, M., Diekmann, M., Pauli, H., et al. (2022). Directional

turnover towards larger-ranged plants over time and across habitats. *Ecology Letters* 25, 466–482. <u>https://doi.org/10.1111/</u>ele.13937

- Steidinger, B. S., Büntgen, U., Stobbe, U., Tegel, W., Sproll, L., Haeni, M., et al. (2022). The fall of the summer truffle: Recurring hot, dry summers result in declining fruitbody production of *Tuber aestivum* in Central Europe. *Global Change Biology* 28, 7376–7390. <u>https://doi.org/10.1111/gcb.16424</u>
- Stephan, R., Erfurt, M., Terzi, S., Žun, M., Kristan, B., Haslinger, K., et al. (2021). An inventory of Alpine drought impact reports to explore past droughts in a mountain region. *Nat. Hazards Earth Syst. Sci.* 21, 2485–2501. <u>https://doi.org/10.5194/nhess-21-2485-2021</u>
- Steyrer, G., and Tomiczek, C. (2002). Forstschutzsituation 2001 in Österreich – Erhebungen und Diagnosen des BFW und Dokumentation von Waldschädigungsfaktoren 2001. Forstschutz Aktuell 28, 66.
- Stiegler, C., Rode, M., Sass, O., and Otto, J. C. (2014). An Undercooled Scree Slope Detected by Geophysical Investigations in Sporadic Permafrost below 1000 M ASL, Central Austria. *Permafrost and Periglacial Processes* 25, 194–207. <u>https://doi.org/ 10.1002/ppp.1813</u>
- Stiels, D., Bastian, H.-V., Bastian, A., Schidelko, K., and Engler, J. O. (2021). An iconic messenger of climate change? Predicting the range dynamics of the European Bee-eater (Merops apiaster). *J Ornithol* 162, 631–644. <u>https://doi.org/10.1007/s10336-021-01867-z</u>
- Stoeckli, S., Hirschi, M., Spirig, C., Calanca, P., Rotach, M. W., and Samietz, J. (2012). Impact of Climate Change on Voltinism and Prospective Diapause Induction of a Global Pest Insect – Cydia pomonella (L.). *PLoS ONE* 7, e35723. <u>https://doi.org/10.1371/journal.pone.0035723</u>
- Stoffel, M., and Corona, C. (2018). Future winters glimpsed in the Alps. Nature Geoscience 11, 458–460. <u>https://doi.org/10.1038/ s41561-018-0177-6</u>
- Stoffel, M., Mendlik, T., Schneuwly-Bollschweiler, M., and Gobiet, A. (2014a). Possible impacts of climate change on debris flow activity in the Swiss Alps. *Climatic Change* 122, 141–155. <u>https://doi.org/10.1007/s10584-013-0993-z</u>
- Stoffel, M., Tiranti, D., and Huggel, C. (2014b). Climate change impacts on mass movements Case studies from the European Alps. Science of The Total Environment 493, 1255–1266. <u>https://doi.org/10.1016/j.scitotenv.2014.02.102</u>
- Stommel, H. (1961). Thermohaline Convection with Two Stable Regimes of Flow. *Tellus* 13, 224–230. <u>https://doi.org/10.3402/ tellusa.v13i2.9491</u>
- Storkey, J., Stratonovitch, P., Chapman, D. S., Vidotto, F., and Semenov, M. A. (2014). A Process-Based Approach to Predicting the Effect of Climate Change on the Distribution of an Invasive Allergenic Plant in Europe. *PLoS ONE* 9, e88156. <u>https://doi.org/10.1371/journal.pone.0088156</u>
- Straka, M., Polášek, M., Csabai, Z., Zweidick, O., Graf, W., Meyer, E. I., et al. (2021). Stream drying bioindication in Central Europe: A Biodrought Index accuracy assessment. *Ecological Indicators* 130, 108045. <u>https://doi.org/10.1016/j.ecolind.2021.108045</u>
- Strohmeier, S., Laaha, G., Holzmann, H., and Klik, A. (2016). Magnitude and Occurrence Probability of Soil Loss: A Risk Analytical Approach for the Plot Scale For Two Sites in Lower Austria. *Land Degrad Dev* 27, 43–51. <u>https://doi.org/10.1002/ldr.2354</u>
- Stumpp, C., and Kammerer, G. (2022). "The Vadose Zone—A Semi-Aquatic Ecosystem", in Encyclopedia of Inland Waters (Sec-

ond Edition), eds. T. Mehner and K. Tockner (Oxford: Elsevier), 331–338. https://doi.org/10.1016/B978-0-12-819166-8.00179-1

- Swiss Permafrost Monitoring Network (2022). Swiss Permafrost Bulletin 2022. Swiss Permafrost Monitoring Network. <u>https://doi.org/10.13093/PERMOS-BULL-2023</u>
- Switanek, M. B., Troch, P. A., Castro, C. L., Leuprecht, A., Chang, H.-I., Mukherjee, R., et al. (2017). Scaled distribution mapping: a bias correction method that preserves raw climate model projected changes. *Hydrol. Earth Syst. Sci.* 21, 2649–2666. <u>https:// doi.org/10.5194/hess-21-2649-2017</u>
- Tafani, M., Cohas, A., Bonenfant, C., Gaillard, J.-M., and Allainé, D. (2013). Decreasing litter size of marmots over time: a life history response to climate change? *Ecology* 94, 580–586. <u>https:// doi.org/10.1890/12-0833.1</u>
- Tallaksen, L. M., and Van Lanen, H. A. J. eds. (2023). *Hydrological drought: processes and estimation methods for streamflow and groundwater.*, Second edition. Amsterdam: Elsevier.
- Tasser, E., Leitinger, G., and Tappeiner, U. (2017). Climate change versus land-use change—What affects the mountain landscapes more? *Land Use Policy* 60, 60–72. <u>https://doi.org/10.1016/j.</u> <u>landusepol.2016.10.019</u>
- Taylor, K. E., Stouffer, R. J., and Meehl, G. A. (2012). An Overview of CMIP5 and the Experiment Design. *Bulletin of the American Meteorological Society* 93, 485–498. <u>https://doi.org/10.1175/</u> BAMS-D-11-00094.1
- Tchaikovsky, A., Häusler, H., Kralik, M., Zitek, A., Irrgeher, J., and Prohaska, T. (2019). Analysis of n (<sup>87</sup>Sr)/ n (<sup>86</sup>Sr), 8<sup>88</sup>Sr/<sup>86</sup>Sr<sub>SRM987</sub> and elemental pattern to characterise groundwater and recharge of saline ponds in a clastic aquifer in East Austria. *Isotopes in Environmental and Health Studies* 55, 179–198. <u>https://doi.org/ 10.1080/10256016.2019.1577832</u>
- Tebaldi, C., Armbruster, A., Engler, H. P., and Link, R. (2020). Emulating climate extreme indices. *Environ. Res. Lett.* 15, 074006. <u>https://doi.org/10.1088/1748-9326/ab8332</u>
- Temperli, C., Bugmann, H., and Elkin, C. (2013). Cross-scale interactions among bark beetles, climate change, and wind disturbances: a landscape modeling approach. *Ecological Monographs* 83, 383–402. <u>https://doi.org/10.1890/12-1503.1</u>
- Teufelbauer, N., Probst, R., and Nemeth, E. (2024). ÖSTERREI-CHISCHER BRUTVOGELATLAS 2013–2018. Wien: Naturhistorisches Museum.
- Thom, D., Rammer, W., Dirnböck, T., Müller, J., Kobler, J., Katzensteiner, K., et al. (2017). The impacts of climate change and disturbance on spatio-temporal trajectories of biodiversity in a temperate forest landscape. *J Appl Ecol* 54, 28–38. <u>https://doi.org/10.1111/1365-2664.12644</u>
- Thomas, C. D., Gillingham, P. K., Bradbury, R. B., Roy, D. B., Anderson, B. J., Baxter, J. M., et al. (2012). Protected areas facilitate species' range expansions. *Proc. Natl. Acad. Sci. U.S.A.* 109, 14063–14068. https://doi.org/10.1073/pnas.1210251109
- Thornalley, D. J. R., Oppo, D. W., Ortega, P., Robson, J. I., Brierley, C. M., Davis, R., et al. (2018). Anomalously weak Labrador Sea convection and Atlantic overturning during the past 150 years. *Nature* 556, 227–230. <u>https://doi.org/10.1038/s41586-018-0007-4</u>
- Tiberti, R., Buscaglia, F., Callieri, C., Rogora, M., Tartari, G., and Sommaruga, R. (2020). Food Web Complexity of High Mountain Lakes is Largely Affected by Glacial Retreat. *Ecosystems* 23, 1093–1106. <u>https://doi.org/10.1007/s10021-019-00457-8</u>
- Tobin, I., Jerez, S., Vautard, R., Thais, F., Van Meijgaard, E., Prein, A., et al. (2016). Climate change impacts on the power genera-

tion potential of a European mid-century wind farms scenario. *Environ. Res. Lett.* 11, 034013. <u>https://doi.org/10.1088/1748-9326/11/3/034013</u>

- Tramblay, Y., Rutkowska, A., Sauquet, E., Sefton, C., Laaha, G., Osuch, M., et al. (2021). Trends in flow intermittence for European rivers. *Hydrological Sciences Journal* 66, 37–49. <u>https://doi.org/10.1080/02626667.2020.1849708</u>
- Trnka, M., Kyselý, J., Možný, M., and Dubrovský, M. (2009). Changes in Central-European soil-moisture availability and circulation patterns in 1881–2005. *Int. J. Climatol.* 29, 655–672. <u>https://doi.org/10.1002/joc.1703</u>
- Trnka, M., Muška, F., Semerádová, D., Dubrovský, M., Kocmánková, E., and Žalud, Z. (2007). European Corn Borer life stage model: Regional estimates of pest development and spatial distribution under present and future climate. *Ecological Modelling* 207, 61– 84. <u>https://doi.org/10.1016/j.ecolmodel.2007.04.014</u>
- Trochine, C., Guerrieri, M., Liboriussen, L., Meerhoff, M., Lauridsen, T. L., Søndergaard, M., et al. (2011). Filamentous green algae inhibit phytoplankton with enhanced effects when lakes get warmer: Filamentous algae suppress phytoplankton. *Freshwater Biology* 56, 541–553. <u>https://doi.org/10.1111/j.1365-2427.</u> 2010.02521.x
- Truhetz, H., and Leuprecht, A. (2018). ÖKS15 Bias Corrected EU-RO-CORDEX Model Precipitation, Radiation, Temperature. <u>https://doi.org/10.60669/B37Q-JD39</u>
- Umweltbundesamt (2020). Austria's National Inventory Report 2020. Vienna: Umweltbundesamt GmbH. Available at: <u>https:// www.umweltbundesamt.at/fileadmin/site/publikationen/rep</u> 0724.pdf (Accessed September 21, 2023).
- Umweltbundesamt (2023). Gefährdungsanalyse ausgewählter Insektengruppen (Hummeln, Ameisen, Wanzen) Österreichs. Wien: Umweltbundesamt GmbH. Available at: <u>https://www. umweltbundesamt.at/fileadmin/site/publikationen/rep0846.</u> pdf
- Umweltbundesamt (2024). Austria's National Inventory Report 2024. Vienna: Umweltbundesamt. Available at: <u>https://www. umweltbundesamt.at/studien-reports/publikationsdetail?pub\_</u> id=2527&cHash=4f93d43917ca9a3ce6e2ba4c1191ca1a
- UNEP (2022). DECISION ADOPTED BY THE CONFERENCE OF THE PARTIES TO THE CONVENTION ON BIOLOGI-CAL DIVERSITY: 5/4. Kunming-Montreal Global Biodiversity Framework. Available at: <u>https://www.cbd.int/doc/decisions/</u> <u>cop-15/cop-15-dec-04-en.pdf</u>
- van der Linden, P., and Mitchell, J. (2009). ENSEMBLES: climate change and its impacts: summary of research and results from the ENSEMBLES project. *Met Office Hadley Centre FitzRoy Road Exeter EX1 3PB UK*, 160.
- Van Lanen, H. A. J., Laaha, G., Kingston, D. G., Gauster, T., Ionita, M., Vidal, J., et al. (2016). Hydrology needed to manage droughts: the 2015 European case. *Hydrological Processes* 30, 3097–3104. <u>https://doi.org/10.1002/hyp.10838</u>
- Van Vuuren, D. P., Edmonds, J., Kainuma, M., Riahi, K., Thomson, A., Hibbard, K., et al. (2011). The representative concentration pathways: an overview. *Climatic Change* 109, 5–31. <u>https://doi.org/10.1007/s10584-011-0148-z</u>
- Van Westen, R. M., Kliphuis, M., and Dijkstra, H. A. (2024). Physics-based early warning signal shows that AMOC is on tipping course. *Sci. Adv.* 10, eadk1189. <u>https://doi.org/10.1126/sciadv.</u> <u>adk1189</u>
- Vasquez, C., Klik, A., Stumpp, C., Strauß, P., Laaha, G., Özcelik, N. B., et al. (2023). Erosive rainfall clustering across Austria's agri-

cultural areas. Vienna, Austria: EGU General Assembly. <u>https://doi.org/10.5194/egusphere-egu23-2588</u>

- Vellend, M., Baeten, L., Becker-Scarpitta, A., Boucher-Lalonde, V., McCune, J. L., Messier, J., et al. (2017). Plant Biodiversity Change Across Scales During the Anthropocene. *Annu. Rev. Plant Biol.* 68, 563–586. <u>https://doi.org/10.1146/annurev-arplant-042916-040949</u>
- Verhoeven, J. T. A. (2014). Wetlands in Europe: Perspectives for restoration of a lost paradise. *Ecological Engineering* 66, 6–9. <u>https://doi.org/10.1016/j.ecoleng.2013.03.006</u>
- Vitasse, Y., Ursenbacher, S., Klein, G., Bohnenstengel, T., Chittaro, Y., Delestrade, A., et al. (2021). Phenological and elevational shifts of plants, animals and fungi under climate change in the European Alps. *Biological Reviews* 96, 1816–1835. <u>https://doi. org/10.1111/brv.12727</u>
- Vogel, W. (2001). Österreich Vorreiter im Grundwasserschutz. Ländlicher Raum 3, 1–7.
- Voordendag, A., Prinz, R., Schuster, L., and Kaser, G. (2023). Brief communication: The Glacier Loss Day as an indicator of a record-breaking negative glacier mass balance in 2022. *The Cryosphere* 17, 3661–3665. <u>https://doi.org/10.5194/tc-17-3661-2023</u>
- Vorkauf, M., Marty, C., Kahmen, A., and Hiltbrunner, E. (2021). Past and future snowmelt trends in the Swiss Alps: the role of temperature and snowpack. *Climatic Change* 165, 44. <u>https:// doi.org/10.1007/s10584-021-03027-x</u>
- Wagner, T., Pleschberger, R., Kainz, S., Ribis, M., Kellerer-Pirklbauer, A., Krainer, K., et al. (2020). The first consistent inventory of rock glaciers and their hydrological catchments of the Austrian Alps. *Austrian Journal of Earth Sciences* 113, 1–23. <u>https://doi.org/10.17738/ajes.2020.0001</u>
- Wagner, T., Seelig, S., Helfricht, K., Fischer, A., Avian, M., Krainer, K., et al. (2021). Assessment of liquid and solid water storage in rock glaciers versus glacier ice in the Austrian Alps. *Science* of *The Total Environment* 800, 149593. <u>https://doi.org/10.1016/j. scitotenv.2021.149593</u>
- Walter, K. M., Zimov, S. A., Chanton, J. P., Verbyla, D., and Chapin, F. S. (2006). Methane bubbling from Siberian thaw lakes as a positive feedback to climate warming. *Nature* 443, 71–75. <u>https://doi.org/10.1038/nature05040</u>
- Walther, G.-R., Gritti, E. S., Berger, S., Hickler, T., Tang, Z., and Sykes, M. T. (2007). Palms tracking climate change. *Global Ecol Biogeography* 16, 801–809. <u>https://doi.org/10.1111/j.1466-8238.2007.00328.x</u>
- Wang, X., Meng, X., and Long, Y. (2022). Projecting 1 km-grid population distributions from 2020 to 2100 globally under shared socioeconomic pathways. *Sci Data* 9, 563. <u>https://doi.org/10.1038/s41597-022-01675-x</u>
- Wartenburger, R., Hirschi, M., Donat, M. G., Greve, P., Pitman, A. J., and Seneviratne, S. I. (2017). Changes in regional climate extremes as a function of global mean temperature: an interactive plotting framework. *Geosci. Model Dev.* 10, 3609–3634. <u>https://doi.org/10.5194/gmd-10-3609-2017</u>
- Weckström, K., Lewis, J. P., Andrén, E., Ellegaard, M., Rasmussen, P., Ryves, D. B., et al. (2017). "Palaeoenvironmental History of the Baltic Sea: One of the Largest Brackish-Water Ecosystems in the World", in *Applications of Paleoenvironmental Techniques in Estuarine Studies. Developments in Paleoenvironmental Research, vol 20.*, eds. K. Weckström, K. M. Saunders, P. A. Gell, and C. G. Skilbeck (Dordrecht: Springer Netherlands), 615–662. https://doi.org/10.1007/978-94-024-0990-1\_24

- Wedekind, C., and Küng, C. (2010). Shift of Spawning Season and Effects of Climate Warming on Developmental Stages of a Grayling (Salmonidae): Climate Change and Spawning Season. *Conservation Biology* 24, 1418–1423. <u>https://doi.org/10.1111/j.1523-1739.2010.01534.x</u>
- Weijer, W., Cheng, W., Garuba, O. A., Hu, A., and Nadiga, B. T. (2020). CMIP6 Models Predict Significant 21st Century Decline of the Atlantic Meridional Overturning Circulation. *Geophysical Research Letters* 47, e2019GL086075. <u>https://doi.org/10. 1029/2019GL086075</u>
- Wessely, J., Hülber, K., Gattringer, A., Kuttner, M., Moser, D., Rabitsch, W., et al. (2017). Habitat-based conservation strategies cannot compensate for climate-change-induced range loss. *Nature Climate Change* 7, 823–827. <u>https://doi.org/10.1038/nclimate3414</u>
- Wieczorek, G. F. (1996). "LANDSLIDE TRIGGERING MECHA-NISMS", in LANDSLIDES: INVESTIGATION AND MITIGA-TION, (Transportation Research Board), 76–90. Available at: <u>https://trid.trb.org/view/462502</u>
- Wiesmeier, M., Poeplau, C., Sierra, C. A., Maier, H., Frühauf, C., Hübner, R., et al. (2016). Projected loss of soil organic carbon in temperate agricultural soils in the 21st century: effects of climate change and carbon input trends. *Sci Rep* 6, 32525. <u>https:// doi.org/10.1038/srep32525</u>
- Wild, M. (2009). Global dimming and brightening: A review. J. Geophys. Res. 114, D00D16. <u>https://doi.org/10.1029/2008JD011470</u>
- Wild, M., Wacker, S., Yang, S., and Sanchez-Lorenzo, A. (2021). Evidence for Clear-Sky Dimming and Brightening in Central Europe. *Geophysical Research Letters* 48, e2020GL092216. <u>https://doi.org/10.1029/2020GL092216</u>
- Wohland, J. (2022). Process-based climate change assessment for European winds using EURO-CORDEX and global models. *Environ. Res. Lett.* 17, 124047. <u>https://doi.org/10.1088/1748-9326/ aca77f</u>
- Woollings, T., Barriopedro, D., Methven, J., Son, S.-W., Martius, O., Harvey, B., et al. (2018). Blocking and its Response to Climate Change. *Curr Clim Change Rep* 4, 287–300. <u>https://doi. org/10.1007/s40641-018-0108-z</u>
- Woolway, R. I., Huang, L., Sharma, S., Lee, S.-S., Rodgers, K. B., and Timmermann, A. (2022). Lake Ice Will Be Less Safe for Recreation and Transportation Under Future Warming. *Earth's Future* 10, e2022EF002907. <u>https://doi.org/10.1029/2022EF002907</u>
- World Bank (2023). Urbanisierungsgrad in Österreich von 2012 bis 2022. Available at: <u>https://de.statista.com/statistik/daten/stu-die/217716/umfrage/urbanisierung-in-oesterreich/</u> (Accessed September 29, 2023).
- World Glacier Monitoring Service (WGMS) (2022). Fluctuations of Glaciers Database. 686 MB. <u>https://doi.org/10.5904/wgmsfog-2022-09</u>
- World Meteorological Organization (WMO) (2017). WMO Guidelines on the Calculation of Climate Normals., 2017 edition. Geneva: WMO.
- Worrall, F., Howden, N. J. K., Burt, T. P., Rico-Ramirez, M. A., and Kohler, T. (2022). Local climate impacts from ongoing restoration of a peatland. *Hydrological Processes* 36, e14496. <u>https:// doi.org/10.1002/hyp.14496</u>
- Xu, G.-L., Kuster, T. M., Günthardt-Goerg, M. S., Dobbertin, M., and Li, M.-H. (2012). Seasonal Exposure to Drought and Air Warming Affects Soil Collembola and Mites. *PLoS ONE* 7, e43102. https://doi.org/10.1371/journal.pone.0043102

- Zangerl, C., Fey, C., and Prager, C. (2019). Deformation characteristics and multi-slab formation of a deep-seated rock slide in a high alpine environment (Bliggspitze, Austria). *Bull Eng Geol Environ* 78, 6111–6130. <u>https://doi.org/10.1007/s10064-019-</u> 01516-z
- Zangerl, C., Prager, C., Brandner, R., Brückl, E., Eder, S., Fellin, W., et al. (2008). Methodischer Leitfaden zur prozessorientierten Bearbeitung von Massenbewegungen. *Geo. Alp* 5, 1–51. <u>https://opac.geologie.ac.at/ais312/dokumente/GeoALP\_05\_001\_051.</u> <u>pdf</u>
- Zappa, G., and Shepherd, T. G. (2017). Storylines of Atmospheric Circulation Change for European Regional Climate Impact Assessment. J. Climate 30, 6561–6577. <u>https://doi.org/10.1175/</u> JCLI-D-16-0807.1
- Zhang, W., and Zhou, T. (2021). The Effect of Modeling Strategies on Assessments of Differential Warming Impacts of 0.5°C. *Earth's Future* 9, e2020EF001640. <u>https://doi.org/10.1029/2020</u> <u>EF001640</u>
- Zhang, Y., Zheng, H., Zhang, X., Leung, L. R., Liu, C., Zheng, C., et al. (2023). Future global streamflow declines are probably more severe than previously estimated. *Nat Water* 1, 261–271. <u>https:// doi.org/10.1038/s44221-023-00030-7</u>
- Zieher, T., Gallotti, G., Rianna, G., Reder, A., and Pfeiffer, J. (2023). Exploring the effects of climate change on the water balance of a continuously moving deep-seated landslide. *Natural Hazards* 115, 357–387. <u>https://doi.org/10.1007/s11069-022-05558-7</u>





Chapter 2

# Climate change, land use, ecosystem services and health

# Second Austrian Assessment Report on Climate Change | AAR2

# Chapter 2 Climate change, land use, ecosystem services and health

This publication was subject to an anonymous international peer review process guided by several review editors, the AAR2 co-chairs and the APCC Steering Committee.

The views or opinions expressed in this publication are not necessarily those of the institutions with whose support this work was carried out. The publisher and the supporting institutions do not guarantee the accuracy or continued availability of urls to external websites or third-party websites mentioned in this publication and accept no responsibility for the accuracy or appropriateness of any content contained therein.

#### **Recommended citation**

K.-H. Erb, U. Tappeiner, W. Winiwarter, et al. (2025): Climate change, land use, ecosystem services and health. In "Second Austrian Assessment Report on Climate Change (AAR2) of the Austrian Panel on Climate Change (APCC)". [D. Huppmann, M. Keiler, K. Riahi, H. Rieder (eds.)]. Austrian Academy of Sciences Press, Vienna, Austria | doi: <u>doi.org/10.1553/aar2-ch2</u>

You can download the recommended citation in the formats Endnote (.ris) and BibTex (.bib) at <u>aar2.ccca.ac.at/citation</u>.

Vienna, June 2025 ISBN: 978-3-7001-5110-4 | <u>austriaca.at/5110-4</u> (full report)

#### Production

Publisher: Austrian Academy of Sciences Press, Vienna Typesetting: Crossdesign Werbeagentur GmbH, Graz Cover: paulaidaanton | Agentur für digiloges Design Graphic design Table 2.2: paulaidaanton | Agentur für digiloges Design Al-based tools such as DeepL were used to support the linguistic revision.

Contact address for product safety: Austrian Academy of Sciences Press Dr. Ignaz Seipel-Platz 2 A-1010 Wien Tel.: +43-1-51581-3420 E-Mail: verlag@oeaw.ac.at

#### Funding acknowledgement

The AAR2 was funded by the Climate and Energy Fund under the 14th Austrian Climate Research Program (ACRP).

#### **Copyright and license**

© 2025 Austrian Panel on Climate Change (APCC) This report is published under a Creative Commons CC-BY 4.0-License. <u>creativecommons.org/licenses/by/4.0</u>





The AAR2 is a report by the Austrian Panel on Climate Change (APCC). https://aar2.ccca.ac.at/en

#### **Coordinating Lead Authors**

Karlheinz Erb	BOKU University
Ulrike Tappeiner	University of Innsbruck   UIBK & Eurac Research
Wilfried Winiwarter	International Institute for Applied Systems Analysis   IIASA

GeoSphere Austria

**BOKU University** 

**BOKU University** 

**BOKU University** 

**BOKU University** 

Eurac Research

**BOKU University** 

**BOKU University** 

### **Lead Authors**

Christian Fritz	HBLFA Raumberg-Gumpenstein
Simone Gingrich	BOKU University
Susanne Hanger-Kopp	International Institute for Applied Systems Analysis   IIASA
Hans-Peter Hutter	Medical University of Vienna   MedUni Vienna
Jochen Kantelhardt	BOKU University
Hanns Moshammer	Medical University of Vienna   MedUni Vienna
Nicolas Roux	BOKU University
Lena Schaller	BOKU University
Günther Schauberger	University of Veterinary Medicine, Vienna   vetmeduni
Uta Schirpke	Eurac Research
Martin Schönhart	Federal Institute of Agricultural Economics, Rural and Mountain Research
Heide Spiegel	Austrian Agency for Health and Food Safety   AGES
Harald Vacik	BOKU University

Austrian National Public Health Institute | GÖG

Austrian Research Centre for Forests | BFW

Technical University Vienna | TU Wien

### **Contributing Authors**

Katharina Brugger Brigitta Hollosi Hans-Peter Kaul Georg Kindermann Christian Lauk Andreas Magerl Marianne Penker Erich Tasser Matthias Zessner Werner Zollitsch

## **Chapter Scientist**

Nicolas Roux

### **Review Editors**

Stefan Frank Carl Schleussner International Institute for Applied Systems Analysis | IIASA International Institute for Applied Systems Analysis | IIASA

### Liaison from AAR2 Technical Support Unit

Monika Mayer	BOKU University
--------------	-----------------

## **TABLE OF CONTENTS**

Exec	cutive	summary	88
2.1.	Chapt	er Introduction	91
	2.1.1.	Resource use and greenhouse gas emissions from Austria's land system	92
		Biomass flows from ecosystems to consumption	92
		Current ecosystem carbon stocks and greenhouse gas emissions from agriculture, forestry and other land uses .	94
		National upstream or downstream emissions (land-related)	96
		Across-border impacts of Austrian consumption of agricultural and forestry products	97
2.2.	Terres	trial ecosystem services under the	
	influe	nce of climate change	98
	2.2.1.	Impacts of climate change on provisioning ecosystem services and related benefits and limitations	100
		Impacts of climate change on the provision of food and feed from arable	100
		Impacts of climate change on the	100
		Impacts of climate change on the provision of livestock products	101
		Impacts of climate change on the provision of products from hunting and	
		wild animals Impacts of climate change on the	102
		provision of products from aquaculture	102
		Impacts of climate change on health aspects related to food provision	102
		Impacts of climate change on the provision of fiber	103
		Impacts of climate change on the provision of bio-based energy	104
		Impacts of climate change on the provision of water	105

Chapt	er Box 2.1		
Impac	ts of climate change on Austria's		
globa	l supply chains of food and		
other	biomass products	106	
2.2.2.	Impacts of climate change on regulating ecosystem services and related benefits .	106	
	Impacts of climate change on carbon sinks in ecosystems	106	
	Impacts of climate change on the regulation of urban micro-climate	107	
	Impacts of climate change on biochemical cycles	107	
	Impacts of climate change on mass movement and erosion control	107	
	Impacts of climate change on wind and fire protection	108	
	Impacts of climate change on the water	100	
	Impacts of climate change on	100	
	pollination	108	
	Impacts of climate change on pest control, invasive species, vectors, and		
	related health issues	109	
Cross-	Chapter Box 2		
Health	n and climate change	110	
2.2.3.	Impacts of climate change on cultural ecosystem services and related benefits .	115	
	Impacts of climate change on recreational activities	115	
	Impacts of climate change on cultural and heritage value	117	
	Impacts of climate change on aesthetic value	117	
	Impacts of climate change on symbolic value	117	
	related to cultural ecosystem services	118	

2.3.	Transformations and measures in the land			
	use system to mitigate and adapt to climate change			
	2.3.1.	'Improve' measures	124	
		Adaptive forest management –		
		'Improved' forest management to		
		mitigate and adapt to climate change	124	
		Agricultural management to mitigate		
		and adapt to climate change	125	
	2.3.2.	'Shift and avoid' measures	127	
		Protein transition – shift towards healthy,		
		plant-based diets	127	
		Non-protein related food measures –		
		'avoid' measures	129	
		A sufficiency-oriented bioeconomy –		
		'shift and avoid'	131	

2.5.5	Nature-based (climate) solutions	4
	Nature-based solutions to increase the carbon sink of managed ecosystems 13	4
	Nature-based solutions to adapt	
	to climate change and disaster risk	
	reduction 13	5
Chap	ter Box 2.2	
Carb	on sequestration potentials of forests 13	6
2.4. Conf	licts, lock-ins and enabling conditions –	
focu	sing on the example of agricultural land	
focu: use	sing on the example of agricultural land 13	8
focu: use 2.4.1	sing on the example of agricultural land 13 Barriers, conflicts, and lock-ins	<b>8</b>
focu: use 2.4.1 2.4.2	sing on the example of agricultural land 13 Barriers, conflicts, and lock-ins	8 8
focus use . 2.4.1 2.4.2 Reference	sing on the example of agricultural land 13 Barriers, conflicts, and lock-ins	8 8 0 .0
focus use . 2.4.1 2.4.2 Reference Glossary	sing on the example of agricultural land 13 Barriers, conflicts, and lock-ins	8 8 .0 .3 . <u>y</u>
focus use 2.4.1 2.4.2 Reference Glossary List of abb	sing on the example of agricultural land 13 13 13 14 14 14 14 14 14 14 14 14 14	8 8 0 3 7 7 15
focus use 2.4.1 2.4.2 Reference Glossary List of abb Appendix	sing on the example of agricultural land 13 13 13 14 14 14 14 14 14 14 14 14 14	8 8 0 3 7 15 2 2

#### **EXECUTIVE SUMMARY**

Land-related human activities that benefit from multiple ecosystem services constitute an important element of Austria's greenhouse gas (GHG) budget (*high confidence*). Agricultural activities, which are essential for food production, contributed 9.9 % to Austria's total GHG emissions during the most recent available 5-year average (2019–2023), dominated by CH<sub>4</sub> and N<sub>2</sub>O (*high confidence*). For the same period, the sector land use, land use change and forestry, contributed 2.3 % of Austria's GHG emissions (*medium confidence*). This sector represented in the past a strong carbon sink, which decreased since the early 2000's and turned into a source around 2020, characterized by large year-to-year variations.  $\{2.1.1\}$ 

The use of land for human livelihoods alters biogeochemical cycles and ecological patterns and processes, affecting the climate system, biodiversity and ecosystem services (*high confidence*). There are multiple trade-offs between different ecosystem services, mostly because the (global) land surface is limited. Rapid reductions in anthropogenic greenhouse gas (GHG) emissions that limit warming to 'well-below' 2°C would greatly alleviate, while warming above 2°C will exacerbate existing trade-offs. {Chapter 2}

The future provision of agricultural crops for feed, food, fiber and fuel will be compromised by climate change, particularly regarding water availability (*medium confidence*). Among the wide range of temperature, precipitation and adaptation scenarios, impacts of climate change on agricultural yields show a high variance from negative to positive (*high confidence*). Impacts of extreme weather events, other disturbance factors (e.g., pests), the role and effect of CO<sub>2</sub> fertilization, adaptation measures (e.g., new crop types, plant breeding) still represent knowledge gaps. {2.2.1}

Forest-related disturbance regimes (drought, fire, windthrow, snow and ice breakage, bark beetle) are projected to increase with climate change, negatively affecting some important ecosystem services (*high confidence*). However, the extent of the resulting damage to forests and impacts on ecosystem services cannot yet be reliably quantified and assessed. Impacts on the protection function of forests against gravitational natural hazards and on the protection of drinking water resources can be assessed with *high confidence*, while impacts on the provision of wood for timber and energy or on carbon storage are difficult to quantify. {2.2.1, 2.2.2, Chapter Box 2.2}

Climate change threatens the extent and quality of services provided by ecosystems (*medium confidence*). In addition to provisioning and regulating services in agriculture and forestry, cultural ecosystem services are also affected. This includes impacts on certain recreational activities through reduced snow fall, increased rock fall, and reduced water quality. However, warmer temperatures also create opportunities, e.g., for bathing activities or summer tourism. {2.2.2, 2.2.3}

In limited situations, climate change improves ecosystem productivity due to increasing temperatures extending growing seasons (*medium confidence*). In both forestry and agriculture, observations show that a warmer climate increases plant productivity. These opportunities are jeopardized by the speed of change – in forestry, challenges are related to the long rotation periods in forest management, which limit the rapid adaption of tree species to take advantage of a different climate, while in agriculture, infrastructure changes are needed to allow for handling extreme weather situations (droughts, cold spells, heat waves). {2.2}

Irrigation needs will require new perspectives on water availability (*medium evidence, high agreement*). Provisioning of drinking water will be impaired in certain regions of northern and eastern Austria, whereas the increased need for agricultural irrigation will impact water resource planning significantly across Austria during drought events. {2.2.1}

Ongoing climate change will have regionally diverse impacts on the provision of ecosystem services due to complex biophysical conditions (*high confidence*). Notably, (1) a general decrease of all ecosystem services, particularly provisioning ecosystem services, is expected in the northern Alpine foothills and highlands, in the southern Alps with the Klagenfurt Basin and in the central Alps (*medium confidence*); (2) a spatial shift of dominant ecosystem services towards higher elevations will occur for regulating ecosystem services and towards low and peripheral areas for provisioning ecosystem services, depending on water availability and plant stress profiles (*medium confidence*). {2.2} Some of Austria's international supply chains will become more vulnerable (*medium confidence*). Austria is a net importer of biomass, importing mainly raw materials and exporting mainly highly processed and manufactured final products. About 16 % of Austria's imports of food and feed and 5 % of Austria's imported wood products originate from regions that are moderately to highly vulnerable to climate change and do not yet appear to be ready to adapt, especially tropical and Mediterranean regions. {Chapter Box 2.1}

Technical and management measures in agriculture and forestry can reduce greenhouse gas emissions and increase sinks but their mitigation potential is limited (*high confidence*). Compared to a reference scenario, measures in agricultural management (e.g., increasing soil organic carbon, reducing emissions from agricultural inputs, optimizing livestock farming) show mitigation potentials of 0.4 to 2.5 MtCO<sub>2</sub>eq/yr (*medium confidence*). Adaptive forest management like changing tree species shows a mitigation potential up to 1.9 MtCO<sub>2</sub>eq/yr (*low confidence*), calculated over 2020–2150. This potential materializes only under a high global warming level (GWL; between GWL 4.0°C and GWL 5.0°C) and primarily beyond 2080. {2.3.1}

Tapping far-reaching mitigation potentials in agriculture and forestry requires structural measures reducing the use of biomass and changing current economic practices and consumption patterns (high confidence). Ambitious mitigation will require far-reaching measures with substantial costs, involving trade-offs and structural disruptions, such as loss of livelihoods for farmers or the need to deal with spillover effects (high confidence). Shifting towards healthy, plant-based diets shows mitigation potentials of 5.0 to 15.4 MtCO<sub>2</sub>eq/yr, when agricultural production can be restructured to reduce agricultural emissions and increase sinks through afforestation on freed agricultural land (medium confidence). Halving food waste may reduce emissions by 2.1 to 4.7 MtCO<sub>2</sub>eq/yr (medium confidence). Prioritizing long-lived harvested wood products over energy show mitigation potentials of 0.1 to 0.5 MtCO<sub>2</sub>eq/yr (low confi*dence*). In the currently only available set of model results for Austria, contrasting two scenarios shows that annual average wood harvest of 25 million Vfm/yr (15 million Efm/ yr) compared to wood harvest levels of 30 million Vfm/yr (25 million Efm/yr) will yield a mitigation potential of 7 to 8.4 MtCO<sub>2</sub>eq/yr until 2150. Current harvest levels (2016-2022) are 25 million Vfm/yr (20 million Efm/yr). To which extent the realization of all these potentials is constrained by climate change induced disturbances (e.g., increases in windthrows, insect infestations, vegetation fires) remains a research gap. {2.3.2, 2.3.3, Chapter Box 2.2}

Implementing mitigation measures along the entire value chain, including production, processing, trade and consumption, is crucial to avoid leakage or spillovers (*high confidence*). Shifting emissions and environmental pressures to other regions through increased imports and exports negates any climate benefit, but trade measures can mitigate leakage. Research gaps exist for mitigation strategies that consider the total food system, encompassing more than just emissions from agricultural production. {2.1.1, 2.3.3, 2.A.9, 2.A.11}

Various land users compete for the resources, services and benefits they derive from land, inducing conflicts of objectives and interests (*high confidence*). A major conflict exists between biomass extraction on the one hand and carbon storage, biodiversity and nature conservation (with their associated social benefits and costs) on the other hand. There are also potential conflicts between agricultural uses and renewable energy, e.g., in the case of agriphotovoltaics, wind and water energy. Ambitious mitigation measures entail important challenges and conflicts. Transparent and democratic negotiation of trade-offs between values and stakeholders is essential to overcome barriers to the transformations of the land use system, counteracting path dependencies and unequal power relations among actors, while supporting vulnerable land users. {2.4}

A fundamental conflict unfolds between economic growth and environmental conservation (*high confidence*). Economic growth, in its current form, drives increased production and consumption and thus land demand, making it a key driver of climate change and biodiversity loss, e.g., due to land-use intensification and land-conversions. Technological improvements that increase yields partly offset land demand, but rebound effects in the land system often annihilate the efficiency gains. {2.4}

Adverse effects on the health of the Austrian population due to climate change are visible already, and these effects are expected to increase, as climate change continues (*high confidence*). In particular, extreme weather events such as heat waves pose serious health threats including premature death, but also non-fatal disease, reduced well-being and productivity. These impacts affect different population groups in different ways, raising the issue of environmental justice. The poor, elderly and chronically ill, those with physically or mentally demanding jobs, and members of ethnic minorities are and will be most affected. Geographical differences must also be considered, both at the small (e.g., urban heat island, poor neighborhoods) and larger scale (e.g., mountain regions). {Cross-Chapter Box 2}

Local mitigation measures, while benefiting the climate globally, can also exert remarkable beneficial effects on public health locally, e.g., regarding healthy diets or sustainable (pollution-free) modes of transport (*high confidence*). Nevertheless, poor implementation of measures (both mitigation and adaptation) could also have negative effects on health. Evaluating measures from a health perspective (health in all policies concept) allows to avoid such implementation failures. {Cross-Chapter Box 2} The health care system contributes to climate change with its own carbon footprint (*limited evidence, high agreement*) and could also contribute substantially to the mitigation efforts. {Cross-Chapter Box 2}

Human health is highly dependent on functioning provisioning systems, which include global supply systems, local infrastructure and many ecosystem services. These are threatened by ongoing climate change (*medium evidence*, *high agreement*). Both ongoing long-term changes and acute extreme events can have long-lasting negative consequences for physical and mental health and social well-being. Pressures from climate change elsewhere in the world could affect health and the health care system in Austria. Chronic effects, especially on mental health, of extreme events and climate change have been documented elsewhere but are still heavily understudied in Austria. The interplay between ecosystems and human health, together with animal health, is an important aspect of One Health. {2.2, 2.3, Cross-Chapter Box 2}

#### 2.1. Chapter Introduction

The impact of climate change on life is far-reaching. Rising temperatures, altered precipitation patterns, and different wind exposure influence the way land is used by humans and covered by vegetation. Changes in land use and vegetation also influence animal husbandry as well as wildlife. Human livelihood and health are affected by climate, climate change and the ecosystems around them.

Land ecosystems provide a wide range of essential, partly irreplaceable services to human society (*high confidence*). They support production of crops and livestock serving as food to nourish our global population (IPCC, 2019a). Ecosystems also produce fibers and energy resources, contributing to various industries and ensuring our energy needs are met, and they are hosts to biodiversity (IPBES, 2019). Freshwater, an essential resource, is regulated by land ecosystems through processes such as filtration and groundwater recharge (Reid et al., 2005; IPCC, 2019a), securing a vital supply for drinking, agriculture, and industry. Moreover, these ecosystems actively participate in climate regulation by sequestering carbon, mitigating climate change, and influencing local weather patterns (IPCC, 2019b; Erb et al., 2023a).

This chapter assesses the current state of scientific knowledge for Austria on the interlinkages between climate change, ecosystems, and the services they provide to human society, including human health (see also Cross-Chapter Box 2). Here we use the concept of ecosystem services as a classification scheme that describes the benefits humans



Figure 2.1 Chapter structure and overview of topics.

derive from ecosystem functioning (see 2.A.1). These services are closely linked to human land use. As land is a finite resource, land use in many cases involves land competition, i.e., the use of land for only one of several possible purposes (Haberl, 2015; Meyfroidt et al., 2022; Erb et al., 2024). Certain land uses may specifically affect the adaptation to and mitigation of climate change, in particular by inducing greenhouse gas emissions, depleting carbon stocks, or altering the resilience of ecosystems to climate change. However, ecosystem functions can also be employed to remove and sequester carbon from the atmosphere by increasing carbon pools in ecosystems (Griscom et al., 2017; Bossio et al., 2020), an essential but not sufficient strategy to combat global warming (Boysen et al., 2017; IPCC, 2019a). Moreover, ecosystems can also contribute to mitigating the impacts of climate change, especially through their regulating capacity.

In agreement with the concept of ecosystem services, the approach taken here is deliberately anthropocentric. This does not mean that we ignore or undervalue the ethical intrinsic values of all living beings and ecosystems (Vilkka, 2021), but it is consistent with the basic objectives of the assessment report. These are (i) to identify societal strategies to adapt to unavoidable climate change, (ii) to contribute to the design and implementation of such strategies, and (iii) to devise pathways for human activities (including production and consumption) and their land use impacts that are consistent with the goal of keeping global warming well below 2°C above pre-industrial levels. Thus, we describe activities and changes in human practices to adapt to or reduce the impacts of climate change. Since natural or near-natural ecosystems cover a relatively small part of the Austrian landscape and have already been covered in Section 1.6, this chapter will focus mainly on human land use.

Key input to this chapter was taken from science performed in Austria and about Austria. Relevant compilations are available as the recently completed Special Reports to the Austrian Assessment Report (in German language), specifically on Land Use and Climate Change (Jandl et al., 2024c), on Health, Demography and Climate Change (APCC, 2018), and on Structures for a Climate-Friendly Living (Aigner et al., 2022). The discussions that these reports were able to provide in much detail inform the views arrived at in this chapter.

The chapter covers the following elements (see Figure 2.1): Section 2.1 summarizes the state of current land use in Austria, associated resources, greenhouse gas emissions and ecological carbon stocks. Section 2.2 assesses the impacts of climate change on a range of ecosystems services are discussed and if possible quantified. Here, the Common International Classification of Ecosystem Services (CICES) is used to classify impacts - see 2.A.2 (Haines-Young and Potschin, 2018). Where traceable, indirect impacts of climate change (as on human health) are also assessed. Given the high dependence of the Austrian supply systems of land-based products from global value chains, special attention is paid to the assessment of the linkages between Austrian land use and global ecosystems (Chapter Box 2.1). Based on various national (Allianz Nachhaltige Universitäten in Österreich, 2021; Aigner et al., 2022) and international classifications (Roe et al., 2019), Section 2.3 discusses possible land-based activities and transformations to achieve climate change adaptation and mitigation and assesses their potentials (including health impacts). The resulting challenges related to potential landuse conflicts and opportunities to overcome them conclude the analysis (Section 2.4). Direct land-use considerations for human health are discussed throughout the chapter, while other health-related issues are addressed separately in Cross-Chapter Box 2, with interrelated concepts described in 2.A.3. The notion of considering ecosystems and their services as a foundation of human well-being (Reid et al., 2005) serves as the overarching concept.

# 2.1.1. Resource use and greenhouse gas emissions from Austria's land system

#### Biomass flows from ecosystems to consumption

Most land use and associated greenhouse gas emissions are related to the provision of biomass. Biomass can be used for the provision of food, feed, fibers and fuel. Figure 2.2 shows an approximation of biophysical biomass flows through the Austrian economic system (Kalt, 2015; Eisenmenger et al., 2016; Jacobi et al., 2018; Strimitzer et al., 2022), but following simple approaches (see 2.A.14). In total, the biomass input to the economy is about 24.5 MtC/yr (year 2020). The Austrian economy is deeply interconnected with global markets and heavily dependent on both imports and exports (high confidence). In terms of carbon units, biomass imports represent more than one third of the biomass input to Austria's economy. Exports are about half the size of imports (note that due to data limitations, exports of paper and highly processed wood products are not included). Agriculture production and wood harvest contribute about equally to domestic biomass extraction.



**Figure 2.2** Approximation of biomass flows in Austria's socioeconomic system in MtC/yr and related GHG emissions. Biomass flows (year 2020) (Kalt, 2015; BMK, 2022; Erb et al., 2023a; Roux et al., 2023, updated as explained in 2.A.14). LULUCF and agriculture emissions are from Umweltbundesamt (2025a), on average for 2019–2023. These emissions are displayed in detail and for several years in Table 2.A.2 in 2.A.5. Consumption-based emissions are from Frey and Bruckner (2021) for the year 2013. Food system emissions are from Crippa et al. (2021) for the year 2015, and wood supply chain emissions are from Kühmaier et al. (2022) for the year 2018. Wood flows in  $[m^3]$  have been converted to [tC] using a crude factor of carbon density [500 t dry matter/m<sup>3</sup>], dry matter biomass flows were converted to carbon fluxes applying a carbon content factor of 50 %. Note that foreign trade of manufactured wood products (millwork, furniture, paper, etc.) is missing in this scheme. <sup>5</sup> Forests acted as carbon sinks in 2020 (grey arrows), but are a source of  $CO_2$  in the five-year average (colored stars). Color coding is used to indicate similarity of components (imports, domestic extraction, processing stages, exports, final consumption, etc.). The loop below animal husbandry represents animal products used as feed (especially milk and whey). Data are subject to various uncertainties, including statistical discrepancies (e.g., between forest increment, forest sink and wood harvest).

The largest processing nodes in Austria's economy are the wood processing industries and the energetic use of biomass (9 and 7.4 MtC/yr, respectively). The latter is associated with corresponding C-emissions to the atmosphere. At the production side, biomass provision is sustained by a flux of carbon from the atmosphere to the land ecosystems. For the quantification and balance of these flows (see Section 2.1.1). Also, an important node is the livestock sector, with feed input of about 5.8 MtC/yr and 1.6 MtC/yr of bedding, yielding 0.8 Mt of animal products and ca. 3 Mt of manure and bedding residues, while the rest is metabolized and exhaled by animals as  $CO_2$  and  $CH_4$ .

The wood sector is particularly important in biophysical terms with a total domestic material input (DMI; domestic extraction plus imports) of 11.8 MtC/yr in 2020. Imports of wood and wood products exceeded the production of wood from domestic forests by 11 %. Exports reported in the Austrian wood flow diagram (roundwood, sawn wood and sawmill-byproducts) are about half of the import volume, however, foreign trade with millwork and paper is not included in the wood flow diagram and Figure 2.2, resulting in an underestimation of wood exports (see below). The domestic wood harvest averaged 89 % of annual stem wood increment in the last inventory period (2016-2021) and showed an increasing trend over the last decade, leaving only little room for enhancing domestic wood provision under sustainability criteria and with current management practices (European Environment Agency, 2017; Beck-O'Brien et al., 2022; Kirchmeir et al., 2022; Erb et al., 2023a). The amount of woody biomass used for energy purposes, at 6.4 MtC/yr, is almost equal to the total amount of wood harvested domestically from forest and non-forest ecosystems with woody vegetation in Austria. Overall, the energy use of wood biomass covers about 17 % of the gross domestic consumption of energy (BMK, 2023). 38 % of the energy use of wood derives directly from domestic harvest, which is directly used for energy, and 12 % comes from direct imports (firewood, sawmill by-products and pellets). The remainder (50 %) are residual materials from processing, including black liquor from paper production, which are of mixed domestic and imported origin. On average, the average share of imports in the energy use of wood biomass is 39 % (Erb et al., 2023a).

The agricultural sector shows a DMI of 12.4 MtC/yr with a dominance of domestic extraction. 1.3 MtC/yr are consumed as food (11 % of the DMI), while material and energy uses of agricultural biomass are significantly smaller. Animal husbandry, including bedding straws represent 64 % of the agriculture DMI and leads to an output of 0.8 MtC/yr in the form of animal products (meat, milk, eggs, including hatching eggs and animal products fed to other animals). The remainder consists of manure, bedding residues, emissions and waste and represents a key component of AFOLU emissions (see below).

# Current ecosystem carbon stocks and greenhouse gas emissions from agriculture, forestry and other land uses

Available data on C stocks are essential for estimating stock changes and hence greenhouse gas emissions or sinks in ecosystems. Land use related C stocks occur as soil organic carbon (SOC) and biomass carbon stock (often separated into above and below ground biomass). Land use is a key determinant of these carbon pools, interacting with other factors (e.g., organic soils show considerably higher SOC contents than mineral soils). The total C-stock in vegetation and soils is 1273 (±341) MtC (Table 2.1), of which about one third is in biomass and two thirds in the form of soil organic carbon (SOC). Conceding regional differences due to biophysical conditions, including soil types and local climatic conditions, changing land use will lead to a gradual adjustment of SOC in soil over decades (Baumgarten et al., 2021). Forests contribute more than 60 % to the total ecosystem carbon stock of Austria, followed by unmanaged/ unsurveyed land, while all other categories play a smaller role.

On average over the last five years with data (2019–2023), the total sum of emissions from the UNFCCC-sectors 'Agriculture' and 'Land use, land-use change and forestry' (LULUCF) fluxes was a net source of greenhouse gases (flux from the biosphere to the atmosphere), of about 9.4 MtCO<sub>2</sub>/yr (Figure 2.3 and Table 2.A.2), or about 12.3 % of the total GHG emissions in Austria (19.8 % in the year 2023). In the five-year average, the total carbon flux in agriculture and LULUCF reached a maximum sink strength of >13 MtCO<sub>2</sub>eq/yr around 2000, which decreased continuously (by approximately 1 MtCO<sub>2</sub>/yr) to turn from a sink to a source after 2014 (Figure 2.3A). Here and elsewhere in this chapter, total GHG emissions include also those of LULUCF.

Agricultural emissions are dominated by  $CH_4$  emissions from enteric fermentation, i.e., ruminant livestock production, whereas soil N<sub>2</sub>O emissions (direct and indirect) play a much smaller role, followed by manure management (see also Lauk et al., 2024). In total and on average between 2019 and 2023, emissions from agriculture, excluding direct and

	Area	Average carbon stock in biomass per area (above and below ground)	Average soil organic carbon per area in the layer 0–30 cm	Range of carbon stocks§
Land use	[1,000 ha]	[tC/ha]	[tC/ha]	[MtC]
Cropland	1,283	6.7±4.1b	62.5±19.9*	63–115
Mountain grassland	12	8.3±2.3a	108.73±41.9	1–2
Extensively used grassland	323	7.9±2.2a	122.2±58.1	23–61
Intensively used grasslands	459	6.1±1.9a	95.3±15.0	40–53
Vineyards / orchards	40	25.4±6.9c	49.1±15.6*	2–4
ForestΩ	4,015	106.5±10.0	115.3±46.0	702–1,080
Peatland**	21	41.3±14.5b	220.0±70.0*	4–7
Settlements	568	2±0.6b	39.6±12.6*	16–31
Unmanaged land/unsurveyed land	1,667	9.6±2.5a	146.8±25.5	218–303
Total	8,389	54.9±4.9	107.5±22.9	1,165–1,559

 Table 2.1
 Area of land use and carbon stocks in Austria, 2020 (from Umweltbundesamt, 2023a; Baumgarten et al., 2021; <sup>a</sup> Seeber et al., 2022;

 <sup>b</sup> Tappeiner et al., 2008; <sup>c</sup> Tasser et al., 2020). Note that a fraction of grazing land is subsumed under unsurveyed land in this table.

\* Range estimate unavailable, the average of all ranges (±32 %) was taken.

\*\* Natural peatland area only

 $\Omega$  Per-area C-stock data from the year 2020 FAO/FRA. Data do not contain deadwood.

§ For ranges, error margins of individual components were combined assuming error propagation of a normal distribution for independent data. Error margins of total were derived using the same logics. Hence also the range of the total is smaller than the difference between sums of low and high estimates.

indirect technical emissions (e.g., machinery, mineral fertilizer production) amounted to 9.9 % of total GHG emissions (10 % for the year 2023) (*medium evidence, high agreement*).

On average between 2019-2023, the sector LULUCF contributed +2.3 % to Austria's total GHG emissions (in the year 2023: 9.9 %). The LULUCF inventory shows strong annual fluctuations, particularly in the forest compartment (stock increases and decreases in existing forests, not so much in area expansion), which makes the interpretation of individual time cuts intricate. Figure 2.3A shows the total GHG fluxes between 1990 and 2023 according to the 2025 GHG inventory (Umweltbundesamt, 2025a). The data represent a major update of the calculation methodology, see 2.A.4 for a comparison with the previous inventory version. Changes in forest (forest remaining forest and new forest land) account for 0.3 % of total emissions in the five-year average (7.1 % in the year 2023). In 2018, and especially in 2019 and 2023, forests decreased in carbon stocks so that the aggregated agriculture and LULUCF sectors acted as strong net source of GHG emissions (medium confidence). In all years before 2018, forests had acted as carbon sinks and compensated for a considerable share of Austria's GHG emissions, especially around the year 2000. Changes in other land-use types (i.e., excluding forests), account for about 3.6 % of the total emissions on average (3.7 % in 2023) and harvested wood products represent a sink of -1.6 % on the five-year average (-0.9 % in 2023).

While emissions from agriculture slightly decreased over the same period, mainly due to a decreasing livestock herd (FAOSTAT), the overall development is mainly driven by forest dynamics (Figure 2.3B). Increment increased from 2.8 m<sup>3</sup>/ha/yr in 1954 and 5.7 m<sup>3</sup>/ha/yr in 1961-70, to 9.4 m³/ha/yr for the 1981-1985 inventory, and decreased since then, reaching 9.3 m<sup>3</sup>/ha/yr in the period 1986–1990 and 8.2 m<sup>3</sup>/ha/yr in the period 2007-2009 to 2016 (Austrian National Forest Inventory, accessed in 2024). In the meantime, the inventory method has changed and might influence those findings. Schieler (1997) proved that the increments based on the 1961-1980 method are comparable with those since 1981. On the other hand, annual wood harvest increased from 3.4 m<sup>3</sup>/ha/yr in 1954 to 7.8 m<sup>3</sup>/ha/ yr in the period 2000-2002 to 2007-2009 and was 7.4 m<sup>3</sup>/ ha/yr in 2007-2009 to 2016-2021. Taking the total wood harvest from census statistics (BMLF, 2024b), the share of wood harvest triggered by wind, snow or bark beetle damage in Austria was on average 26 % between 1990 and 2021 (28 % in the last decade), and showed strong variations, with a minimum in 2001 (9 %) and a maximum in 2008 (55 %). Note that harvest estimates from the annual census statistics (HEM) are significantly lower than harvest estimates from



**Figure 2.3** Austrian GHG inventory 1990–2023, version 2025 (Umweltbundesamt, 2025a). Panel (A) Total GHG-fluxes of the two compartments Agriculture (A), and Land use, Land-use change and Forestry (LULUCF). The agricultural flux is split into  $CH_4$  from enteric fermentation (ruminant livestock; A-enteric fermentation) and other agricultural emissions (A-other), mainly N<sub>2</sub>O from soils and manure management. The LULUCF flux is separated in three components, Forest land (LULUCF-Forest land), harvested wood products (LULUCF-HWP) and the sum of all other fluxes (LULUCF-all other). The gray line in panel (A) indicates the resulting annual net-flow, the black line the 5-year moving average. Not all years prior to 2005 were available in tabular format in the 2025 version; missing values were extracted from a figure published by Umweltbundesamt (2025b). Panel (B) Forest and forestry dynamics 1990–2021: Areas refer to harvest volumes, disaggregated into salvage logging for storm and snow and bark beetle as well as regular and other wood harvests according to BFW (Steyrer, 2020) and Umweltbundesamt (2023a). The gray and black lines indicate the total wood harvest and wood fuel harvest, respectively, according to the 'Holzeinschlagsmeldung' (HEM) that reports annual 'harvestable volumes' (not solid volumes) without bark and without natural losses ('Erntefestmeter' – Efm). HEM results are thus significantly smaller than the harvest volumes according to the Forest Inventories (FIs), which are, however, only appraised in inventory periods. The annual proxy for increment is based on Forest Inventory data disaggregated to annual fluxes using HEM information (Umweltbundesamt, 2023a). Note that the difference between increment and harvest in panel (B) basically determines the net C flux of forests in panel (A).

the forest inventory, even when differences in methodology and definitions are taken into account (Gschwantner, 2019). The harvest of wood fuel (including primary wood, thinning wood and harvest residues) increased exceptionally strongly, by a factor of 1.9 between 1990 and 2019, from 16 % to 28 % of the total harvest (excluding bark and natural losses; FAOSTAT).

# National upstream or downstream emissions (land-related)

Land systems integrate various processes from production to consumption. In addition to the emissions subsumed in the emission accounts described above, these processes are associated with further emissions, that are accounted for in other sectors in the national GHG inventory (for details and further information, see 2.A.5). These emissions derive from on-farm energy use, from processing and retailing of agricultural products (downstream emissions), and the embodied emissions of materials used in farming such as fertilizers and pesticides (upstream emissions), excluding the provision of infrastructure and machinery.

Globally, Crippa et al. (2021) estimate that food systems account for 34 % of total GHG emissions. About 71 % of these are related to agriculture and land use/land use change with the remainder resulting from processing, transport, storage, or disposal. For Austria, Crippa et al. (2021) estimate total food system emissions at 21.5 MtCO<sub>2</sub>eq (based on GWP-100) in 2015, which is substantially higher than the agricultural emissions reported in the national GHG inventory (see above).

The current supply chain of wood biomass involves fossil energy inputs and thus generates emissions in addition to the land-use emissions that are included in the AFOLU sector. Life cycle assessments (LCA) have been conducted since the early 1990s to analyze the environmental impacts of various forest products and processes, but consistent results are lacking due to different boundary conditions (e.g., processes included in the accounts, assumptions about productivity rates, or fuel consumption of machineries). A meta-study showed large differences in greenhouse gas emission estimates of wood products, ranging from 2.4-59.6 kgCO<sub>2</sub>eq/ m<sup>3</sup> over bark calculated from site preparation to forest road, and from 6.3-67.1 kgCO<sub>2</sub>eq/m<sup>3</sup> over bark from site preparation to reaching a plant gate or consumer (Klein et al., 2015). In 2018, 492 ktCO<sub>2</sub>eq were emitted for harvesting and transporting of 19.2 million°m3 of timber, which corresponded to 25.63 kgCO<sub>2</sub>eq/m<sup>3</sup>. Transport accounted for the largest share of emissions (77 %) within the supply chain, extraction accounted for 14 % of emissions, felling and processing 5 %, and chipping 4 %. Emissions from felling, delimbing, and crosscutting were much lower when using a chainsaw compared to a harvester (Kühmaier et al., 2022). However, chainsaw operations are associated with higher work-related accident rates (SVS, 2024).

# Across-border impacts of Austrian consumption of agricultural and forestry products

There is *high confidence* that the consumption of forestry and agricultural products in Austria drives land use related pressures on ecosystems and greenhouse gas emissions in other countries (for an assessment, see Jandl et al., 2024c, pp. 147-152). Biomass use and processing in Austria is embedded in international trade networks, with imports exceeding exports. In terms of primary biomass equivalents, about half (45 % on average in 2010-2013) (Kalt et al., 2021) of the food, feed, fiber and biofuels consumed in Austria relies on imports (Schaffartzik et al., 2014; Kalt et al., 2021). Frey and Bruckner (2021) estimate that greenhouse gas emissions embodied in Austria's consumption of agricultural products reached 16.5 MtCO<sub>2</sub>eq in 2013, with food products accounting for 12.7 MtCO<sub>2</sub>eq. Almost two thirds of the greenhouse gas emissions embodied in the Austrian consumption of agricultural products occur outside of Austria, especially in other European countries (22 %), Asia (19 %) and Latin America (14 %) (Frey and Bruckner, 2021). This account includes emissions from agricultural activities and land use change, typically deforestation. Leather is the most important non-food agricultural product causing emissions (only beef, milk and dairy, and pork cause higher emissions), of which only 0.7 % occur domestically, while 25 % occur in India and 15 % in Australia.

Austria's dependence on bioenergy imports is higher than indicated by national energy statistics on direct wood energy imports, as part of the wood processing residues used for energy come from imported sources (see above), questioning claims that bioenergy represents a self-sufficient energy source (Anca-Couce et al., 2021). This is because a large share of imports - Austria is the world's second largest importer of roundwood after China (FAOSTAT) - is processed in Austria, while final products are exported, and residues are used for paper production and energy purposes. Total roundwood imports amounted to 4.6 million m<sup>3</sup> in 1990, 8.6 million m<sup>3</sup> in 2000, 8.7 million m<sup>3</sup> in 2010, 12.5 million m<sup>3</sup> in 2020 and 8.7 million m<sup>3</sup> in 2022 while the sawn wood exports show a similar pattern of 4.2 million m<sup>3</sup> (1990), 6.4 million m<sup>3</sup> (2000), 6.1 million m<sup>3</sup> (2010), 6.1 million m<sup>3</sup> (2020) and 5.9 million m<sup>3</sup> (2022) (FAOSTAT). An older study calculated that taking embodied imports in the Austrian use of bioenergy from wood and agricultural products into account, reduced the domestic share from 84 to 67 % (Kalt and Kranzl, 2012). An analysis of wood flows in Austria reveals that in 2020, on average, 39 % of all wood and wood products used for energetic purposes in Austria originated from imports (BMK, 2023; Erb et al., 2023a). There is *high confidence* that biomass imports are associated with considerable direct and indirect embodied emissions (Marques et al., 2019; Pendrill et al., 2019; Peng et al., 2023), which have not yet been quantified for Austria's trade of wood.

Such impacts are operationalized with, e.g., concepts such as 'imported deforestation' risk, which in the Austrian case has declined from 3,000-5,000 ha/yr to 1,000-4,000 ha/ yr, representing a 20-60 % reduction of deforestation related CO<sub>2</sub> emissions (Pendrill et al., 2022). However, in some countries, better governance of deforestation has shifted export driven agricultural expansion frontiers to non-forest ecosystems (Dou et al., 2018). Decreasing rates of deforestation embodied in Austrian consumption are laudable but need to be interpreted with caution as such spillover effects on non-forest ecosystems are often overlooked and may offset reductions in carbon emissions and biodiversity loss associated with decreasing rates of deforestation (Searchinger et al., 2015; Dou et al., 2018). The degradation of non-forest ecosystems embodied in Austrian imports remains largely unquantified. Agricultural imports have increased rapidly in recent decades. The share of imports in total agricultural area embodied in Austria's consumption increased by 130 % since the late 1980s (data from Roux et al., 2023). Depending on the specific indicator chosen, the methods used and system boundaries, studies find diverging trends associated to the environmental impacts of Austrian biomass imports (Gingrich et al., 2024). While pressures on ecosystems, mea-



**Figure 2.4** Human Appropriation of Net Primary Production (HANPP) embodied in Austria's production and consumption of agricultural and forestry products, by land use types (data from Roux et al., 2023, updated) (see 2.A.15). HANPP is an indicator for land use and pressures on ecosystems, accounting for the extent and intensity of land use (Haberl et al., 2014). Greenhouse gas emissions embodied in imports of agricultural products (dotted line) are from Frey and Bruckner (2021). Imports and exports are corrected for re-exports.

sured in Human Appropriation of Net Primary Production (HANPP) embodied in Austria's consumption of agriculture and forestry products decreased, the HANPP embodied in imports and exports increased since 1990 (Figure 2.4). HANPP embodied in biomass imports and exports are of similar magnitude.

# 2.2. Terrestrial ecosystem services under the influence of climate change

This section assesses the impacts of climate change on ecosystem services and their benefits, especially human health and economic income. Austria's ecosystems have been influenced by long-term management practices, affecting their resilience or vulnerability to climate change. Although it is difficult to disentangle the effect of management from those of climate effects, we attempt to do so in the following subchapters. The section is structured following the Common International Classification of Ecosystem Services (CICES), to ensure the comprehensiveness of the assessment. Where possible, the assessment is made under different Global Warming Levels (GWL; see Chapter 1), and also considers changes in precipitation. The quantification of impacts is sometimes done based on ÖKS15 Scenarios. Note that, as explained in Section 1.2.2, these scenarios are subject to a time lag or 'cold bias': The impacts of climate change appear to occur earlier than modeled by ÖKS15. There is *high confidence* that climate change has an overall negative impact on ecosystem services in Austria, but that the impacts differ by region and ecosystem service considered. Table 2.2 provides an overview of the magnitude of the impacts of temperature and precipitation changes on the ecosystem services assessed in this chapter. See 2.A.6 for detailed quantifications and Figure TS.7 for a cartographic representation.
**Table 2.2** Mean potential impacts of climate change on Ecosystem Services (E.S.) under GWL 3.0°C and GWL 5.0°C, differentiated by three major biogeographical regions and classified into qualitative impact levels (Schirpke and Tasser, 2024). Impacts were derived by identifying potential impacts on multiple indicators, used to quantify the potential ecosystem service supply, based on current literature. The indicators were multiplied by the potential impacts and spatially weighted with projections of temperature and precipitation under the two GWLs. For methodological details, see Schirpke and Tasser (2024). A strong negative impact is indicated by '-3', a medium negative by '-2', a low negative by '-1', and a low positive by '1'. Values are averaged across regions. The regions encompass several ecosystems. E.S. not covered by Schirpke and Tasser were estimated by expert judgement (blue font). If supported by literature levels of confidence are given (\*/\*\*/\*\*\* = low/medium/high).

Values from		c	€WL 3.0°C		Confi- dence		GWL 5.0°C		Confi- dence
and Tasser, 2024) attributed to ecosys- tem services from this chapter and reclassi- fied into qualitative impact levels	Chapter section	North- eastern plains, Alpine foothills and highlands	Northern and Central Alps	Southern Iowlands and Alps		North- eastern plains, Alpine foothills and highlands	Northern and Central Alps	Southern Iowlands and Alps	
Provisioning ecosystem s	ervices								
P1) Food and feed from arable crops	2.2.1	-1/+1	-1/+1	-1/+1	**	-1	-1	-1	*
P1) Food and feed from permanent crops	2.2.1				Resear	ch gap			
P2) Livestock products	2.2.1	1	1	1	**	1	1	-1/+1	**
P3) Products from hunting and wild animals	2.2.1	-1	-2	-1	**	-1*	-3*	-1*	*
P4) Products from aquaculture	2.2.1	1	1	1	**	1*	1*	1*	*
P5) Fiber, bio-based energy	2.2.1	-1	-1/+1	-1	**	-1	-2	-2	*
P6) Water	2.2.1	-1	-1	-1	**	-1/-2	-2	-2	**
Regulating ecosystem ser	vices								
R1) Climate regulation/ carbon sinks	2.2.2	-1	-1	-1	**	-1	-1	-2	*
R2) Biochemical cycles	2.2.2	-1	-1	-1	Research gap	-1	-1	-1	Research gap
R3) Mass movement and erosion control	2.2.2	-1	-1	-1	*	-1	-1	-1	**
Wind and fire protection	2.2.2				Resear	ch gap			
Water flow regulation and flood control	2.2.2				Resear	ch gap			
R6) Pollination	2.2.2	1	1	-1	**	-1	1	-1	**
R7) Pest control and invasive species	2.2.2	1	1	1	***	1	1	1	***
Cultural ecosystem servic	es								
C1) Recreation	2.2.3	-1	-1	-1	***	-2	-2	-2	**
C2) Cultural heritage	2.2.3	-1	-1	-1	**	-1	-1	-1	*
C3) Aesthetic value	2.2.3	-1	-1	-1	***	-2	-2	-2	**
C4) Symbolic value	2.2.3	-1	-1	-1	**	-1	-1	-1	*

# 2.2.1. Impacts of climate change on provisioning ecosystem services and related benefits and limitations

### Impacts of climate change on the provision of food and feed from arable crops

Austria's arable crop production for food and feedstock covers an area of about 1.3 million ha. The largest arable production areas are planted with wheat, maize/corn, barley and soybeans. Cereals, produced on about 58 % of arable land, are the most important crop group, followed by feedstocks (18 %) and oilseeds (12 %) (Statistik Austria, 2023a). On average, the production of arable crops incl. feedstocks generates 30–38 % of the production value of the entire agricultural sector (BMLF, 2023a).

There is *medium evidence* and *high agreement* that the effects of climate change described in Section 1.2.2, such as increased temperatures, altered vegetation periods and enhanced CO<sub>2</sub> fertilization will affect crop yields and product quality. Thereby, predicted yield deviations vary strongly. The variance depends on the GWL, the climatic region and the time period considered. In particular, the characterization of precipitation patterns as well as regional variables such as soil quality, soil water storage capacity, crop types and management practices (e.g., irrigation) play a role (Schönhart et al., 2014, 2016; Kirchner et al., 2015; Thaler, 2021). With medium confidence, for scenarios close to GWL 2.0°C, Austrian studies, all considering CO<sub>2</sub> fertilization effects and comparing to yields around the beginning of 2000, present yield changes spanning from -10 to +10 %, with regional disparities (Thaler et al., 2012; Eitzinger et al., 2013; Schönhart et al., 2014; Mitter et al., 2015a). Heterogeneity is observed between humid western and semi-arid eastern regions (Schönhart et al., 2014, 2018; Mitter et al., 2015a, 2015b; Kirchner et al., 2016; Thaler, 2021). With more pronounced climate change, one study predicts yield increases of 3-23 % for winter wheat and spring barley under close to GWL 4.0°C scenarios, but strong yield reductions of >-20 % for maize (Thaler, 2021). With high *confidence*, CO<sub>2</sub> effects on C<sub>3</sub> crops exceed those on C<sub>4</sub> crops (Ebrahimi et al., 2016; Ainsworth and Long, 2020; Thaler, 2021). However, the effects of CO<sub>2</sub> fertilization remain uncertain, and the effects of technical advancements (plant breeding, digitalization and other improvements in crop management) call for further research. In terms of product quality, wheat protein concentration could increase in scenarios with the largest yield reductions (Ebrahimi et al., 2016).

In addition to general yield trends, specific climate change phenomena as described in Section 1.2.2 - with medium confidence - will also increase the uncertainty of crop yields. Indirectly, trends such as elevated temperatures and altered precipitation patterns will affect the dynamics of weeds, pests and diseases (Eitzinger et al., 2013; Deutsch et al., 2018; Juroszek et al., 2020; Lehmann et al., 2020). Directly, the increasing occurrence of weather extremes such as thunderstorms, heavy rainfall (Mitter et al., 2015b; Naumann et al., 2015; Rajczak and Schär, 2017; Lehmann et al., 2018; Mäkinen et al., 2018), heat spells (Trnka et al., 2014; Mitter et al., 2015b; Mäkinen et al., 2018; Lorenz et al., 2019b; Bönecke et al., 2020), hail, and especially increased frequency and severity of droughts (Mitter et al., 2015a; Naumann et al., 2015; Hochrainer-Stigler et al., 2019) can lead to remarkable production losses. One study covering two Austrian drought scenarios up to the year 2040, models annual yield decreases of winter crops of up to 27 % on national average, causing annual production value losses of the most important crops of EUR<sub>2023</sub> 78-191 million, depending on the scenario (Mitter et al., 2015a). The highest drought-related yield vulnerabilities of >20 % (maize) are modeled for the Pannonian zone in Burgenland and Lower Austria (Hochrainer-Stigler et al., 2019).

The impacts of climate change on arable crop production vary between the different production areas in Austria. While adaptation of management (irrigation, soil management, adaptation of crop rotations and choice of crop species) will be necessary to counteract negative impacts, productive cropping areas may also emerge in the western part of the country (Kirchner et al., 2016).

### Impacts of climate change on the provision of food from permanent crops

In 2021, the two major branches of Austrian permanent crop production, namely wine and fruit production, covered an area of 46,634 ha (vineyards) and 15,760 ha (orchards). In 2022 wine and fruit production generated a value of  $EUR_{2023}$  740 million and  $EUR_{2023}$  447 million, respectively, representing 10 % of the agricultural sector (Statistik Austria, 2023b). The main fruits are apples and pears, apricots (and strawberries), and the major production areas are located in Styria (64 % of area), Lower Austria, Upper Austria, Burgenland (Statistik Austria, 2023b). Major wine production areas are located in Lower Austria, Vienna, Burgenland and Styria (BMLF, 2023a).

Literature quantifying yield reductions in permanent crops under different GWLs is not available for Austria and represents a major research and knowledge gap. However, there is *limited evidence* but *high agreement* that climate change will provoke changes in the risk of crop failure and product quality of permanent crops. The literature agrees that yields and product quality in Austrian wine and fruit production will be impacted mostly by elevated average temperatures and the related increase in the length of vegetation periods. In fruits, increasing spring temperatures lead to changes in phenology, and thereby especially to the earlier timing of bud break (Unterberger et al., 2018, for Styria; Haokip et al., 2020), potentially causing a phenological decoupling of plant-pollinator mutualism (Settele et al., 2016, for Europe) and strongly increasing the risk of severe damage up to total yield losses and quality losses due to the occurrence of late frosts (Unterberger et al., 2018; Dalhaus et al., 2020). For example, in 2023, late frost caused a total loss of apricot harvests in Wachau/Lower Austria (BMLF, 2024a), and in the years 2023, 2021 and 2020 production losses in all fruits of 10 to 14 % compared to the 10-year average (BMLF, 2023a, 2024a). In addition to late frosts, extreme weather events (heavy rain, thunderstorms, hail) directly endanger crops. Prolonged periods of heat and short-term extreme droughts can affect yields and product quality and require irrigation measures and investments in management systems such as cooling, shading, and smart irrigation (Hannah et al., 2013, international scale). In wine production, earlier ripening and higher temperatures during ripening alter quality parameters (Vršič and Vodovnik, 2012, for northeast Slovenia; Vršič et al., 2014). Production of high-quality wine grapes in regions at the margins of their climatic limits will become more difficult, while elevated temperatures could make some regions more favorable to produce current or new varieties (Moriondo et al., 2010, for Tuscany; Eitzinger et al., 2013; Santillán et al., 2020, for northern Italy). In Austria, the entire Waldviertel and Mühlviertel regions will potentially be suitable for viticulture by the end of the century (Formayer and Goler, 2013). Migration of production areas to higher altitudes is also possible (Eccel et al., 2016, for Trentino; Egarter Vigl et al., 2018, for South Tyrol; Alikadic et al., 2019, for Trento). Threats from climate change-induced changes in pest and disease dynamics are expected to increase (Salinari et al., 2006, for northwestern Italy; Caffarra et al., 2012, for northern Italy; Lehmann et al., 2020). For Austria, both fungal and insect dynamics are expected to be impacted by climatic parameters, with mixed results depending on climatic

AAR2

variations (Legler et al., 2012; Zaller et al., 2013; Redl et al., 2021; Möth et al., 2023).

### Impacts of climate change on the provision of livestock products

In 2021, livestock production in Austria produced a value of  $EUR_{2023}$  4,417 million, representing 44 % of the agricultural sector, using large areas of cropland and grassland (Table 2.1). Animal production is, in economic terms, dominated by dairy products (40 % of the commercial value), beef production including veal (23 %), pork (20 %) and poultry and eggs.

Climate change will impact Austrian livestock production through the effect of heat stress on animals (Hempel et al., 2019; Mikovits et al., 2019) on the one hand, and through impacts on feed supply chains (see Section 2.2.1 for more detail) (Haslmayr et al., 2018) on the other hand (*medium evidence, high agreement*). However, estimates for Austria indicate rather low heat stress related losses in milk sales revenue of 0.6 % (0.1–2.5 %) even under a GWL 3.6°C (Schönhart and Nadeem, 2015). In addition to productivity losses, impaired welfare due to the increased frequency and intensity of heat stress occurrence may increase skepticism among societal groups towards animal farming. This emphasizes the urgency of mitigation strategies from an animal welfare perspective (Shields and Orme-Evans, 2015).

There is medium evidence and high agreement that, depending on the severity of climate change, grassland productivity can be expected to be maintained or even enhanced in large parts of western and central Austria, which are dominated by grassland (Haslmayr et al., 2018) (see Section 1.6.3). Water availability and rainfall distribution are the main factors for the expected development of grassland yields, with an increasing probability of yield depression of individual cuts (high confidence), thereby putting at risk a constant feed supply for ruminant livestock (Finger et al., 2010; Neuwirth and Hofer, 2013; Bahn et al., 2023). In alpine grasslands, severe climate change, represented by a reduced water supply of the order of 40-50 %, together with droughts of several weeks and partially an increased ambient temperature, induces shifts in vegetation and soil microbes, leading to enhanced water use efficiency (under GWL 1°C: 2018; no increased temperature: Tello-García et al., 2020, 2023) (medium confidence). Regions focused on the production of important feed crops (maize, cereal and legume grains, oilseeds) for all livestock species will be stronger affected by climate change (see above). The impact

on feed supply will be considerably variable across regions (see above).

In addition, securing the water supply for livestock has become an increasingly important issue for livestock farmers in certain parts of Austria in recent years (*limited evidence, medium agreement*). The livestock farming sector currently requires about 55 million m<sup>3</sup> of water. An increase of 46 % is expected by 2050 (BMLRT, 2021b). Most likely, the situation will vary regionally, mainly depending on precipitation patterns and the type of water source (groundwater vs. spring discharge; for details see below). Given the high drinking water requirement of livestock, roughly three times the daily dry matter intake (Schlink et al., 2010), securing a continuous water supply is a critical issue for livestock farms, which will become even more challenging due to increasing conflicts of use (BAFU, 2014).

Animal health in a narrow sense (i.e., occurrence of health disorders as a consequence of climate change) is perceived as a potential challenge, but has not yet been researched in depth for the specific conditions in Austria. The infection pressure by certain vector-borne pathogens is expected to increase because of climate change (Gale et al., 2010). Bluetongue disease in ruminants, which has also appeared in Austria in the last one and a half decades (Pinior et al., 2018), is an example for vector-borne diseases, where in particular the interactions between vector and host are still poorly understood (Baylis, 2013). This poses a challenge for the development of specific preventive actions.

### Impacts of climate change on the provision of products from hunting and wild animals

There is high confidence that climate change leads to elevational shifts in the distribution of habitats for many plants and animals of about 140-180 m per degree of temperature increase (Zu et al., 2021; Rubenstein et al., 2023). Since the 1970s, there has been an elevation migration of 36 m/decade (see Section 1.6.4). Especially in the peripheral Alps, but also in the lower elevations of the Central Alps, the subalpine and lower alpine, forest-free habitats are colonized by forests and shrubs and have thus largely disappeared (Chamberlain et al., 2013; Tasser et al., 2017). The distribution of most forest and shrub game species (especially ungulates) has remained stable or increased in response to the expansion of suitable habitats (Reimoser and Reimoser, 2016; Tasser et al., 2023). However, there is high confidence that the distribution of alpine habitat specialists could decline sharply as open grassland areas disappear (Chamberlain et al., 2013; Imperio et al., 2013; Rehnus et al., 2018) (see also Section 1.6.4). These results confirm that the conservation of alpine game species such as the rock ptarmigan (Revermann et al., 2012) and the marmot (Tafani et al., 2013), but also chamois (Chirichella et al., 2021), will become problematic in the future, but this can be reduced by management strategies to preserve summer pasture areas (Chamberlain et al., 2013; Tasser et al., 2024).

## Impacts of climate change on the provision of products from aquaculture

There is *high confidence* that climate change increases the temperature of freshwater bodies (Niedrist, 2023), causing a deterioration of water quality, e.g., due to a reduced dissolved oxygen and increased algal blooms and toxicity of pollutants (Ficke et al., 2007; Orság et al., 2023) (see also Section 1.4.3 and below). For fisheries and aquaculture in Austria, this also means increased stress for fish and an increase in associated diseases and parasite infestations, as well as lower feeding and growth performance (Handisyde et al., 2006; Schindler, 2017; Casas-Mulet et al., 2021).

Austrian fish farming focuses mainly on salmonids, carp and African catfish (3,599 t salmonids, especially rainbow, brown and brook trout, 617 t carp and 494 t catfish) (Statistik Austria, 2022a). Furthermore, 32.9 million spawning fish and 25 million stocking fish are produced. Since salmonids are generally cold-water fish (average preferred temperature of 11-15°C in summer), while carp and catfish are cold- to warm-water fish (>21°C), climate warming will lead to a shift of breeding populations from salmonids to cyprinids, pike and catfish (Meisner et al., 1987; Reist et al., 2006) (high confidence). At the same time, salmonid species are expected to be displaced upstream in Alpine rivers, but warm-water species such as carp and some invasive species such as exotic rainbow trout, grass carp and sunfish will become established (Matulla et al., 2007; Rabitsch et al., 2013; Pletterbauer et al., 2015).

## Impacts of climate change on health aspects related to food provision

Looking at food-related health issues, climate change certainly affects food safety (Duchenne-Moutien and Neetoo, 2021). Although food safety issues are not of major concern in Austria compared to lower- and middle-income countries (Cissé, 2019; Seitner et al., 2022), Austria is dependent on global supply chains (Leidwein et al., 2014) (see Chapter Box 2.1) and also in Western European countries, an association between ambient temperature and number of salmonella infections has been demonstrated, even after controlling for seasonal effects (Kovats et al., 2004). Behavioral changes (e.g., more outdoor preparation of food) contribute to these increasing risks. However, it should be noted that climate effects on food safety and food-borne diseases in Austria are understudied. In addition to food quality and security, the effects of climate change on food quantity will also directly impact human health. Springmann et al. (2016) in their global modeling study also provide estimates for Austria under GWL 2.0°C-GWL 3.0°C (main scenarios) for 2050 accounting for (i) reduced calorie intake (mostly beneficial), (ii) reduced red meat intake (beneficial), and (iii) reduced fruit and vegetable intake (detrimental). All factors combined they estimate 220 (95 % confidence interval: 70-360) additional deaths annually due to these changes (compared to a 'no climate change' reference scenario).

#### Impacts of climate change on the provision of fiber

For Austria, fiber production from agriculture (fiber plants: Hemp, flax; animal fibers: Sheep wool) is minimal and hence not covered. In contrast, the use of wood fiber includes most wood products (timber, paper, cellulose) and therefore, the provision of fibers is mostly dependent on the forestry sector.

Forest ecosystems and forest management will be impacted not only by changes in climate trends, but also by increased variability and intensity of extreme weather events, such as drought, storms and floods (high confidence) (see also Chapter 1). In particular, mountain forests will experience higher temperatures (see also Chapter 7), which will lead to higher production rates and an expansion of the forest area at higher elevations in sites that are currently limited by temperature (Lindner et al., 2010). The disturbance regime will also be influenced by the changing climatic conditions, leading to more frequent and intense bark beetle infestations, wind throws and forest fires (Senf and Seidl, 2021). Climate change is and will continue to impact the tree species composition, productivity, and ultimately wood supply. At local warming levels above 2°C, especially conifer-dominated landscapes in mountain areas of Austria characterized by large trees, might shift to a landscape dominated by smaller, predominantly broadleaved trees (Albrich et al., 2020). Higher growth rates can be expected in higher elevation mountain areas due to longer vegetation periods and warmer temperatures, and generally, the provision of timber might be influenced by increased disturbances and droughts (Obojes et al., 2018, 2020, 2022). Businesses will therefore face a potential scarcity of wood availability for certain assortments or species as forest management adapts to the changing climate. The results of the Austrian Forest Inventory 2016/21 of the Federal Research Centre for Forests (BFW) show a clear trend towards more broadleaved forests. Pure conifer stands have decreased by 6 % over the last decade, while mixed hardwood stands have increased by the same percentage. Due to climate change, spruce has already lost parts of its distribution between 600 and 800 m a.s.l (BMLF, 2023b).

Despite Austria's forests having lost their property of being a carbon sink (see Figure 2.3) large quantities of CO<sub>2</sub> are taken up in forest ecosystems. The current increment of Austria's forests (i.e., before harvest and mortality, as opposed to stock changes) is 24.7 MtCO<sub>2</sub>/yr. However, scenarios show a decrease to 21.1 MtCO<sub>2</sub>/yr if local temperatures increase by 3°C. In addition, if precipitation decreases by 30 %, capacity drops further to 19.0 MtCO<sub>2</sub>/yr. The distribution of changes is heterogeneous (Figure 2.5), with higher altitudes experiencing increased productivity and lower altitudes experiencing decreased productivity, especially when accompanied by increased temperature. The probability of a single 'bad year' increases with increasing temperature and can lead to the destruction of large forest areas and the gradual release of previously stored CO<sub>2</sub>. Furthermore, Kindermann (2021) points out that the CO<sub>2</sub> uptake becomes smaller than the CO<sub>2</sub> release, turning a forest from a sink to a source. Increased mortality results in an increased mass of deadwood in the forest, which may either remain in the forest or be harvested and used for fiber production. However, lower carbon uptake by forests means that, to keep the carbon stock constant, harvest of deadwood must also be proportionally reduced. On the other hand, forest biodiversity benefits greatly from structural diversity and the abundance of deadwood in forests (Oettel and Lapin, 2021; Rigo et al., 2024).

The distribution of forests showed an accumulation of ages with high rates of productivity rates in 1990. 30 years later, forests have matured and the age peak has flattened, leading to a decrease in productivity from 9.3 m<sup>3</sup>/ha/yr in 1990 to 8.2 m<sup>3</sup>/ha/yr in 2020. This trend of decreasing productivity is estimated to continue by the forest growth models EFDM, Caldis (Kindermann and Ledermann, 2023), G4M and EFISCEN (Böttcher et al., 2011). In the coming decades, this trend can be reversed, by harvesting old and overmature forests with low increments and regenerating



**Figure 2.5** Estimated average productivity in tC/ha/yr of spruce (*Picea abies L.*) and beech (*Fagus sylvatica L.*) for (i) current climatic conditions; (ii) increased local temperature (+3°C) and reduced precipitation (70 % of current climatic conditions) (Kindermann, 2021). Effects on other tree species can be found in Figure 2.A.2 for (i) current climatic conditions; (ii) increased local temperature (+3°C); and (iii) increased local temperature (+3°C) and reduced precipitation (70 % of current climatic conditions) (Kindermann, 2021).

them to younger forest stands with higher growth rates (see Section 2.3.1 for related effects on climate change mitigation). Overall, the growth of the same species in the same location was higher in the following stand generation (Rössler, 2014). In addition to climate, other factors such as nitrogen deposition can influence forest growth. Spiecker et al. (1996) have shown this trend of increasing growth for Europe. A recent study by Chakraborty et al. (2024) expects that growth will decline in the future.

In any case, the expected high amount of salvage wood will affect the market price of timber and can cause losses of income for forest owners due to limited opportunities to sell high quality sawn timber, rather than industrial round wood with lower quality. Based on the analysis of a past storm event in Italy, a 40 % drop in timber prices was observed (Udali et al., 2021).

### Impacts of climate change on the provision of bio-based energy

Cropland-based biomass is insignificant for Austria's overall energy supply. Despite the negligible role of yield quality and therefore climate-induced changes in yield quality, the impacts of climate change on crops grown for energy uses are similar to the impacts on crops grown for food, feed, and other industrial purposes (see above). The two major cropland-based energy sources, bio-based liquid fuels and biogas, respectively account for only 4.4 % and 0.8 % of total renewable energy demand in 2020 (Biermayr, 2022). About 11 % of cereal consumption in Austria was used for bio-ethanol production (AMA, 2022). Biogas feedstock was produced on 15,800 ha cropland, 7,000 ha grassland and cropland with second harvests or crop residues. Livestock manure and residues from the food industry and sewage systems are also included.

About 26.9 million m<sup>3</sup> of the timber being harvested domestically, imported or coming from other sources (e.g., waste wood, groves) are used annually for energy supply in Austria, accounting for 55 % of the total energy supply from renewable resources (see Section 4.5). The material originates from harvested industrial roundwood or firewood, processes and by-products in the sawmill, pulp and paper industries (e.g., cross-cut ends, barks, products from further wood processing, wood chips, pellets), which has been argued to be CO<sub>2</sub> neutral by Jandl et al. (2024) but not in other sources (Norton et al., 2019; Erb et al., 2022; Peng et al., 2023) (for a discussion on carbon neutrality of biomass use, see 2.A.5 and Box 1.1 in Jandl et al, 2024c). The impacts of climate change on the provision of forest fuels are similar to those for fiber, although there is *limited evidence* and medium agreement that climate change will increase the amount of salvage wood harvested from the increasing number of disturbances (see above). This will probably increase the supply for renewable energy and other material uses in the future (Sohngen and Sedjo, 2005; Lantz et al., 2022).

#### Impacts of climate change on the provision of water

The provision of ecosystem services and related benefits of water bodies (surface- and groundwater) is manifold (Keeler et al., 2012; Hallouin et al., 2018; Lupi et al., 2023). In addition to supplying drinking water, healthy waters are rich in biodiversity, provide fishing grounds and recreational areas including bathing, and are essential landscape elements. They are also indispensable elements of water and food supply for neighboring ecosystems (e.g., wetlands). Reduced water availability and deteriorating water quality endanger these services.

There is *high confidence* that changes in rainfall patterns, drought duration and intensity can significantly impact pollutant transport, dilution and, together with increasing water temperature, related water quality. Therefore, climate change has the potential to counteract efforts to improve water quality and may lead to further deterioration. See 2.A.7 and Section 1.4 for a discussion of climate impacts on water fluxes and related substance transport, or indirect impacts due to climate change adaptation measures.

There is a *high confidence* that phosphorus pollution is responsible for 21.1 %, of Austrian surface water bodies failing good ecological status as defined in the EU Water Framework Directive (WFD). For conditions similar to GWL 2.0°C and increasing precipitation trends, Schönhart et al. (2018) estimate an increase in median total phosphorus river loads of about 20 % and, for decreasing precipitation trends, an increase in the of rivers orthophosphate environmental quality standards (EQS) by a factor of about 1.2 (medium confidence). Nitrate is responsible for 9 % of groundwater monitoring stations exceeding the WFD threshold (high confidence). Mehdi-Schulz et al. (2024) estimate a heterogeneous pattern across Austria with an increase of up to 10-15 % for N concentrations in some rivers in northeastern and southeastern Austria for a GWL 2.0°C and decreasing precipitation trends (medium confidence). Nitrate in groundwater can be assumed to follow a similar pattern. These impacts may affect current efforts to reduce nitrate levels below EQS by environmental programs in agriculture (e.g., Österreichisches Programm für umweltgerechte Landwirtschaft, ÖPUL) (high confidence).

In addition, pollution with priority substances deteriorates the chemical status of surface waters, with 100 % failing to achieve good status if either of the ubiquitous substances mercury and polybrominated diphenyl ethers (PBDE) are considered, while only 17 % are at risk of failing good status if these substances are neglected (BMLRT, 2022). While the vulnerability of water bodies with decreasing flow under climatic stress continues to increase, this could become even more critical if the European Commission adjusts/lowers the thresholds for priority substances (European Commission, 2022; Brielmann et al., 2023; Clara and Müller-Rechberger, 2023) (*high confidence*).

There is medium evidence and high agreement that impacts of climate change on the provision of water will manifest locally and seasonally, with areas in northeastern and eastern Austria being most at risk of facing water scarcity (Mitter and Schmid, 2019; ZAMG, 2020; BMLRT, 2021a; Hanger-Kopp et al., 2024). Drinking water demand across Austria will rise until 2050 under GWL 2.0°C, particularly in western and Alpine regions with a high share of tourism, where water demand in some hotspots may increase by 50 % under GWL 2.0-3.0°C (BMLRT, 2021a). In a dry scenario, agricultural water demand may double by 2050 (GWL 2.0-3.0°C), whereas other material water uses, such as for industrial purposes, are not predicted to change significantly (BMLRT, 2021a). Usage intensity across Austria will largely remain within sustainable boundaries; however, in a very dry scenario, regions with intense usage in Eastern Austria exceed the available groundwater resources (GWL 2.0-3.0°C). Demand for drinking water will rise by approximately 15 % by 2050 under a very dry scenario - GWL 2.0-3.0°C, with about 10 % of these 15 % being due to population growth, and 5 % due to climate change (BMLRT, 2021a).

Water demand for agricultural irrigation will rise, particularly in northern, eastern, and southeastern Austria (medium evidence, high agreement). Across Austria, the total irrigable land area has increased by 30,349 ha (from 91,998 to 122,347 ha) between 2010 and 2020 (Statistik Austria, 2022b), representing 4.7 % of the total used agricultural area (Statistik Austria, 2024). About 85 % of the irrigated areas in Austria are concentrated in Lower Austria and Burgenland (Statistik Austria, 2022b). Very dry scenarios (GWL 2.0°C) show an increase in irrigable land in these federal provinces, as well as in Upper Austria and Styria, by 2050 (BMLRT, 2021a). In the same regions, scenarios considering severe multi-annual dry spells show increases in irrigated areas by a factor of 5.6 compared to a reference scenario (GWL 1.5°C), where increases further depend on the level of risk aversion of farmers (Mitter and Schmid, 2019). Currently, there is no monitoring of actual water use for irrigation. Such data would significantly improve the accuracy of predictions and thus adaptation decisions (BMLRT, 2021b; Hanger-Kopp et al., 2024).

### Chapter Box 2.1. Impacts of climate change on Austria's global supply chains of food and other biomass products

There is medium evidence but high agreement that Austria's supply of food, feed, fiber, and bioenergy is increasingly dependent on global supply chains, and that imports from tropical and Mediterranean regions are more vulnerable to climate change than Austria's domestic supply. Due to the increasing reliance on imports, the supply of agricultural products to Austria is vulnerable to climate change not only in Austria, but also in countries exporting to Austria (Davis et al., 2021; Hedlund et al., 2022). Nonetheless, about two-thirds of Austria's imports of biomass originate from neighboring countries (Kalt, 2021), which, except for Mediterranean region, may largely face similar risks as Austrian production (ESPON, 2012; Knox et al., 2016; Hasegawa et al., 2022). On the other hand, the provision of tropical products (e.g., coffee, cocoa, palm oil, soybeans) is more vulnerable due to the concentration of production in a limited number of countries and their vulnerability to climate change (European Environment Agency, 2021). 16 % of Austria's imports of food and feed, in raw biomass equivalent, originate from regions that are moderately to highly vulnerable and do not yet appear to be ready to adapt to climate change (trade data from Roux et al. (2023), updated as explained in 2.A.15, vulnerability from Kan et al. (2023)). The supply of wood in Austria is slightly vulnerable to climate change, as about 5 % of Austria's imported raw material input (RMI) of wood originate from regions that are moderately to highly vulnerable and do not yet appear ready to adapt to climate change (Kan et al., 2023; Roux et al., 2023). However, for certain wood products, the share of imports from vulnerable regions can be substantially higher. Diversifying origins and reducing the overall demand for vulnerable products decreases the vulnerability of supply in Austria, while at the same time decreasing environmental pressures on global ecosystems.

Demand for European agricultural exports is also expected to increase in response to climate change, as a response to reduced yields in other regions of the world (Janssens et al., 2020). This increased demand could potentially increase land demand and pressure on ecosystems in Austria (*limited evidence, high agreement*).

### 2.2.2. Impacts of climate change on regulating ecosystem services and related benefits

### Impacts of climate change on carbon sinks in ecosystems

The impacts of climate change on forests and other ecosystems are described in Chapter 1. Here, we build on Chapter 1, by describing the consequences for carbon sinks in forests and other ecosystems. The simulation under a strong warming scenario (GWL 2.0°C–GWL 3.0°C by 2050 and close to GWL 4.0°C by 2100) showed an increase in soil organic carbon (SOC) when conifers are replaced by deciduous tree species (Jandl et al., 2021). In the 150-year simulation, the majority of forest sites become suitable for deciduous forests. The build-up of a large soil organic carbon stock is driven by the harvesting pressure on the remaining coniferous forests. Deciduous forests were less in demand and developed under a regime of light forest intervention. However, towards the end of the century, when the temperature level is far above present levels, soil organic carbon stocks declined (Jandl et al., 2021). Highly intensive production of wood biomass will decrease the carbon stock in forest soils by 3 % in 80 years (Repo et al., 2015).

Impacts of climate change on the carbon sink function of wetlands are outlined in Section 1.6.1. Peat accumulating wetlands are the ecosystems that store the most carbon per area. The effect of land use-related drivers of carbon loss on peat accumulating wetlands, namely drainage and agricultural and forestry management are estimated to be more severe than the effects of climate change. For German peatland sites, there is high confidence that drainage and drying out turns peatlands from a carbon sink to a large source of greenhouse gases with emissions averaging 25 tCO<sub>2</sub>eq/ha/ yr, with peaks up to >60 tCO<sub>2</sub>eq/ha/yr (Tiemeyer et al., 2016; BMLRT, 2022). For Austria, the actual amount and location of agriculturally used drained peatland areas are understudied (Hogl et al., 2023). The Austrian NIR 2023 estimates 13,000 ha of grassland on organic soils, while arable land on organic soils is not listed. In contrast, a recent study estimates 31,000 ha of peatland area under mostly intensive agricultural grassland or cropland management (Hogl et al.,

2023). For Austria quantitative modeling studies on the impact of climate change on drained, agriculturally used peat soils and the carbon sinks remain a research gap.

### Impacts of climate change on the regulation of urban micro-climate

Climate change affects regulating ecosystem services in urban areas. Rising temperatures, long lasting drought periods and altered precipitation patterns as observed in Austria (Chimani et al., 2016; Haslinger et al., 2022), can make trees more susceptible to diseases and can affect the ability of vegetation and green spaces in cities to filter pollutants, reduce the heat island effect, and regulate local temperatures, leading to increased energy demand for cooling, reduced air quality, and heat-related health issues. These impacts hamper the resilience of cities to environmental changes (Lafortezza and Sanesi, 2019). Urban forests, parks, or simply tree-lined streets - urban greens in general - are known to facilitate adaptation to climate change in terms of regulating temperature and water budget. This improves the public health and well-being of urban populations and contributes to sustainable living in terms of positive effects on humans, microclimate, soil and biodiversity (high confidence). More details on heat in urban areas can be found in below, in Cross-Chapter Box 2 and in Chapter 3.

#### Impacts of climate change on biochemical cycles

Ecosystem processes regulating the exchange of nutrients (including mineralization) and soil formation may be hampered by climate change, especially as altered soil humidity and temperatures will also affect soil biota. Results for Austria also take advantage of the specific sites established under the Long-Term Ecological Research network (Mirtl et al., 2015). Investigations for alpine grassland (Seeber et al., 2022) demonstrate that climate change influences the way carbon (C) and nitrogen (N) pools increase as a consequence of extensified grassland management. This effect is primarily driven by rising temperatures, as precipitation is not considered a limiting factor in this context. In forest soils, Kengdo et al. (2023) demonstrate increased carbon inputs due to increased fine root turnover. These cases provide proof that climate change would help to alleviate temperature limitations on many biological processes that support regulating services. However, warming of forest soils also decreases microbial carbon and nitrogen use efficiencies (Tian et al., 2023), thereby reducing regulation - an observation that is in line with microbial biomass accumulation observed with soil cooling (Schnecker et al., 2023). More research is needed to establish the effect of additional stressors on biological systems, such as droughts (especially in areas already now experiencing water scarcity) or ozone damage to vegetation, given that the presence of ozone has been observed to reduce drought stress in oak trees (Peron et al., 2021). Similarly, the effects of drought on microbial communities that had been exposed to elevated temperatures and CO<sub>2</sub> concentrations in a treatment experiment were strongly alleviated (Metze et al., 2023). All these results come from experiments at Austrian sites or with Austrian substrates and demonstrate a high level of scientific activity, especially in recent years, but will only gradually allow conclusions to be drawn about climate change impacts on ecosystems and their ability to regulate nutrient flows, especially as the observations demonstrate that combinations of ecosystem stressors do in several instances not add up, but rather cancel each other out.

## Impacts of climate change on mass movement and erosion control

Climate change will affect the frequency and intensity of gravitational processes through changes in precipitation amounts, duration and snow cover, or frost dynamics (limited evidence, high agreement). Permafrost dynamics in alpine environments influence rock-slope stability through shear stresses and reduced shear resistance of rock masses (Krautblatter et al., 2013; Mamot et al., 2021). Warmer temperatures and longer growing seasons lead to improved tree growth, but longer dry periods also mean additional stress (Schuldt et al., 2020). As tree species have different tolerances to stress situations, new competitive conditions will arise and the tree species composition will alter the protective effect of mountain forests. In addition, the protective effect of forests will be reduced by intensified disturbance regimes caused by temperature driven agents, such as increased bark beetle infestations in Norway spruce dominated forests at higher altitudes (Maroschek et al., 2015), and a higher chance for stand replacing forest fires due to longer drought periods (Müller et al., 2020) or more wind throws caused by extreme wind events (Costa et al., 2021). Protective functions against landslides and snow avalanches are negatively impacted by the stochastic creation of small gaps under warming scenarios with an increase in mean annual temperature up to 4°C across the Alps (Maroschek et al., 2015, 2024) (limited evidence, high agreement). The simulated shift of precipitation events from summer to winter and spring will promote wet snow avalanches in the mountains and the occurrence of flood-relevant snowmelt events, limiting the capacity of forests to control erosion control and reduce mass movements (Glade, 2020). Soil erosion could also increase in agricultural areas, which are usually not covered by vegetation in winter and spring (Glade, 2020). Forest vegetation can influence permanent or deep landslides through drainage in the catchment and directly on the landslide mass (Rickli et al., 2004), but the soil-stabilizing effect of tree roots is not strong in relation to the location of the sliding surfaces in the catchment area (Frehner et al., 2005; Bollinger et al., 2008; Losey and Wehrli, 2013). In general, forest vegetation will presumably exhibit a higher protection from precipitation events with longer duration but lower intensity than from to short-term-high-intensity events (Kleemayr et al., 2020). Direct effects of climate change might have a stronger influence on the future supply of erosion control than management or natural disturbances (Seidl et al., 2019). The impacts of avalanches are difficult to predict with simulation models (in terms of time, space and size) as the influencing factors vary greatly over a very small geographical area and the effect of forests is hardly understood (Studeregger et al., 2020) (see also Sections 1.6.2, 1.7 and 7.4.2).

### Impacts of climate change on wind and fire protection

Climate change affects the capacity of ecosystems to protect society from wind and fire as they act as natural barriers, reducing wind speed, regulate the local climate and prevent erosion. There is high confidence that climate change will increase the frequency and intensity of disturbances, which will affect the species composition, forest structure, forest cover and the amount of flammable material in the forest (Hlásny et al., 2021; Berčák et al., 2023). Windstorm damage in European forests has increased over the past century (Seidl et al., 2011; Schelhaas et al., 2015; Gregow et al., 2017; Patacca et al., 2023) and this trend is expected to continue (Ikonen et al., 2017; Seidl et al., 2017) especially due to the high vulnerability of flat-rooted Norway spruce dominated forests (Bourke et al., 2023). The combination of a high amount of salvage wood from wind disturbances and long drought periods (Senf et al., 2020) further increases the predisposition of these forests to bark beetle infestations (Hlásny et al., 2021) and wildfires (Müller et al., 2020). This will in particular reduce the capacity of the protection forests in Austria to secure the living environment as well as critical infrastructure and settlements in the wildland urban interface (see also Sections 1.2.2, 1.6.2 and 7.4.2).

### Impacts of climate change on the water flow regulation and flood control

Flood hazards are predicted to increase (Chapter 1, Table 1.A.3). Many ecosystems, particularly rivers and riparian forests, depend on floods to function. Ecosystems also regulate high and low flows in two major ways: They mitigate flood risk by providing retention space in the direct vicinity of rivers, and they prevent floods at the catchment scale through vegetation cover and soil type. In addition, drought in the catchment area may affect the water holding capacity of soils (Vári et al., 2022). Anthropogenic interventions such as land-use change, wetland drainage, and river regulation, have had major impacts on this regulating function in rivers and surrounding ecosystems (Jungwirth et al., 2002; Mergili and Duffy, 2022, for the river Lech). In comparison the direct impact of climate change on the capacity of ecosystems to regulate water flow and floods is probably small. Depending on the scenario and its assumptions about future precipitation, various effects can be expected, such as various types of ecosystem degradation due to increased heat and wildfires, or exceeding the ecosystem capacity to hold floods. However, there is little to no information for Austria.

#### Impacts of climate change on pollination

There is high confidence that climate change leads to a shift in the range of species, resulting in reconfiguration of species communities and decoupling of community interactions (Zulka and Götzl, 2015; Rollin et al., 2022). In general, an increase in mean annual temperature has a negative effect on the abundance of wild pollinators (Vasiliev and Greenwood, 2021). Bumblebees and hoverflies are particularly affected (Minachilis et al., 2021), but the degree of ecological specialization of species is also negatively correlated with the risk of extinction (Roberts et al., 2011). As a result of global warming, higher temperature improves foraging by reducing the energy cost of maintaining optimal body temperature, which has been shown for example, for honey bees (Abou-Shaara et al., 2017; Usha et al., 2020) (high confidence). On the other hand, there is high confidence that nectar abundance and total sugar content in nectar are negatively correlated with heat and drought exposure of plants (Borghi et al., 2019) as well as pollen quality and quantity (Zinn et al., 2010). The

associated effects are changes in the area and isolation of suitable habitat with appropriate forage plants and significant declines, especially in specialized pollinator species, but also in generalist species (Rollin et al., 2022).

Directly affected are related ecosystem services such as pollination services and pest control. The value of pollination alone is estimated near to EUR<sub>2023</sub> 442.5 million in Austria, or 9.9 % of the total value of agricultural crop products (Zulka and Götzl, 2015). However, these consequences of climate change can be buffered by targeted management measures and nature-based solutions, such as the development of near-natural elements or habitats with appropriate forage plants in agricultural landscapes (Roberts et al., 2011; Zulka and Götzl, 2015). More near-natural areas increase the possibility of plant and animal migration and thus also of pollinators, while fewer natural areas make this process more difficult.

Low air quality and climate change also increase allergies (Bergmann, 2016). Climate change (mostly due to temperature and  $CO_2$  increase) will affect plant growth and pollen production. Plants will react to heat stress and ozone by producing more pollen or producing pollen more frequently, and the single pollen particles will often have more allergenic proteins on their surface (*medium confidence*). Climate change will also affect the introduction and wider spread of new and sometimes also highly allergenic plants (like *Ambrosia artemisiifolia*) (Cheng et al., 2023).

### Impacts of climate change on pest control, invasive species, vectors, and related health issues

There is *high confidence* that climate change increases risks to ecosystems, biodiversity, food security, and human and animal health (Cissé et al., 2022). Typical examples are neobiota species. Out of the 3,462 vascular plants occurring in Austria, 10.6 % are naturalized neophytes (Schratt-Ehrendorfer et al., 2022). If temporarily occurring species are included, a total of 1,463 non-native vascular plants have been recorded in Austria (Kalusová et al., 2024). Climate and global change are expected not only to open new pathways and territories for invasion, but also weaken the resilience of ecosystems and native species, thus facilitating the establishment of invasive species (IUCN, 2021). These species may

have the same impact as natural hazards and are very costly to eradicate or contain (Turbelin et al., 2023). One example in forests is ash dieback caused by an alien fungal pathogen (Marçais et al., 2022). In agriculture, the main route of introduction of invasive plants has been through deliberate imports as ornamental or crop plants (*medium evidence, high agreement*). A small percentage of these are considered invasive (AGES, 2023a) such as common ragweed (*Ambrosia artemisiifolia*) and jimsonweed (*Datura stramonium*), which are rapidly spreading over Austria (Follak et al., 2017, 2023). These plants not only cause yield and quality losses, but also health burdens, such as increased allergenic load and prolonged allergy seasons (Ambrosia) or poisoning along the food chain (Datura).

There is *high confidence* that climate change also affects blood-sucking arthropods (mosquitoes, biting midges, ticks, etc.), which are vectors for several pathogens (viruses, bacteria, protozoa) that cause mainly zoonoses (Rocklöv and Dubrow, 2020; Thomson and Stanberry, 2022). A disease vector is an organism, most often an arthropod, that transports a pathogen (viruses, bacteria, protozoa) from one host organism to another. Hotter summers and warmer winters (see Section 1.2.2) lead to increasing vector densities (Sonnberger et al., 2020; Duscher et al., 2022), due to higher overwintering success or multiple generations per year, expansion of endemic areas (e.g., tick-borne encephalitis, bluetongue disease), accelerated replication rate of pathogens in vectors, and more frequent years with favorable conditions for the transmission cycle (e.g., Rubel, 2021).

In Austria, native arthropods are potential vectors of emerging pathogens, that have caused several disease outbreaks, such as West Nile virus (Aberle et al., 2018), Usutu virus (Brugger and Rubel, 2009) or bluetongue virus (Purse et al., 2005; Lebl et al., 2013; Brugger et al., 2016). In addition, invasive vector species are increasingly establishing themselves in Austria. Examples include mosquitoes of the genus *Aedes* or *Anopheles* (Lebl et al., 2013; Bakran-Lebl et al., 2022), tick species such as *Hyalomma marginiatum* (Duscher et al., 2022), and sandflies of the genus *Phlebotominae* (Poeppl et al., 2013). These species are potential vectors for tropical and subtropical diseases caused by Chikungunya virus, Dengue virus, yellow fever virus, Crimean-Congo haemorrhagic fever virus or Leishmania parasites.

#### Cross-Chapter Box 2. Health and climate change

#### Hans-Peter Hutter; Hanns Moshammer

To gain a better understanding of how we might address the massive effects of climate change on human health in Austria, we look at the direct and indirect effects of climate change on human health, discuss beneficial and adverse health consequences of some mitigation (and adaptation) measures, and finally describe the carbon footprint of our health care system.

#### The health impact of climate change

There are many ways in which climate change affects health (APCC, 2014, 2018; Weitensfelder et al., 2023). The media and the public are mostly aware of the direct health impact of heat waves. Indeed, extreme weather events such as heat waves, storms, heavy rains and droughts, are likely to increase due to climate change thus intensifying their direct impact on human health and well-being (see Section 1.1.2). From a medical point of view, the most important impact of climate change on health to date is through heat and heat waves (*high confidence*). In 2022, 419 (CI 109; 741) heat related deaths in Austria were estimated (Ballester et al., 2023). The most vulnerable to heat stress are the elderly, and even more so very old and chronically ill people. In addition, very young children and pregnant women are more vulnerable (Kuehn and McCormick, 2017; Chersich et al., 2020; Jiao et al., 2023). But outdoor workers, especially those with physically demanding jobs, also suffer from the heat (Cheung et al., 2016; Ferrari et al., 2023). Hot days and even more so, hot nights reduce regeneration and sleep quality. Therefore, general work performance suffers (*medium evidence, high agreement*). Other vulnerable groups are the socioeconomically disadvantaged with poor housing and lack of access to cool areas, health care, and urban green spaces. This also affects ethnic minorities (Gronlund, 2014; Ellena et al., 2020; De Schrijver et al., 2023).

Not only heat but also extreme cold is detrimental to health. Nowadays, more people die on cold days than on hot days (Gasparrini et al., 2017; Statistik Austria, 2022d; Masselot et al., 2023). As temperatures rise, heat-related deaths are predicted to increase and cold-related deaths to decrease (*high confidence*). According to Martínez-Solanas et al. (2021), heat-related deaths will exceed cold-related deaths in Austria by the end of the century and especially under high-emission scenarios (CCBox 2 Figure 1). However, estimates of cold-related deaths heavily depend on assumptions about effect-thresholds (Arbuthnott et al., 2018) and own data analyses demonstrate that increases in heat deaths already outweigh decreases in cold deaths in Austria (Moshammer, 2023).

The seasonal mortality pattern is first of all characterized by stronger prevalence of cardiovascular diseases, but also infectious diseases play a role (Madaniyazi et al., 2022; Statistik Austria, 2022d). Food-borne bacterial diseases peak in summer and viral respiratory infections in winter (*high confidence*). However, the relevance of temperature (high or low) is far from clear (Liu et al., 2020), making it difficult to assess the impact of a warming climate on these infectious diseases.

While the effects of temperature extremes on well-being, productivity, and physical health are already well understood, the long-term effects on mental health are still understudied in Austria (Weitensfelder et al., 2023). These mental health effects concern post-traumatic stress disorder (PTSD) after catastrophic weather events (Neria et al., 2008), increased suicide rates with increasing temperatures (Kim et al., 2019), but also climate anxiety (Clayton, 2020; Gago et al., 2024). Hickman et al. (2021) report surprisingly high numbers of climate anxiety among young adults, while no representative data are available for Austria. However, based on a convenience sample, Raile and Rieken (2021) report a high percentage of worries about climate change in Austria (81 %) with 41 % 'somewhat' and 16 % 'strongly' or 'very strongly' burdened by these worries. The societal polarization around climate change issues is also likely to reduce human well-being. There is evidence that climate change increases polarization (Wong-Parodi and Feygina, 2021), and that polarization in turn increases stress and anxiety (Fraser et al., 2022). In addition to the political polarization, many



young people suffer from the impression that the elder generations and the government do not take care of them and do not take their worries seriously (Hickman, 2020; Hickman et al., 2021).



Regarding mortality during heat waves, there are indications of ongoing adaptation: As average temperatures rise, the threshold above which an increase in daily mortality is observed also rises (Weitensfelder and Moshammer, 2020) (*limited evidence, high agreement*). Population-wide adaptation through a selection process may also contribute to the rising temperature threshold for daily mortality: Individuals most vulnerable to high temperatures have already died in previous heat waves, leaving the remaining population more resistant to upcoming heat waves (little evidence except for short-term mortality displacement up to several months, but high plausibility).

In addition to the short-term and long-term impacts of extreme weather events, there is a more complex relationship between human health and climate change associated with atmospheric transport and chemistry, soil and water ecosystems, forestry, agriculture, and human energy and transport infrastructure. Many of these changes will have repercussions on human health (APCC, 2018). A recent ARCP project examined the impacts of climate change on ozone concentration in Austria, including impacts on health (Rieder et al., 2023). With drier and hotter summers, average ozone concentrations will likely increase. However, regional ozone concentrations and especially the local peak of ozone concentrations are mainly affected by national emission scenarios: Lower local emissions of non-methane VOCs as well as nitrogen oxides will impact Austria's national ozone levels significantly more than global emission and climate trends. In the follow-up project Future\_Capacity (Formayer et al., 2024a), interactions between ozone concentrations and high temperatures on mortality and morbidity were also examined in a systematic literature review (Wallner and Moshammer, 2023). A total of 45 relevant studies were identified, most of which examined short-term effects on mortality (*high confidence*). They examined either the effect of high ozone levels on the strength of the heat-mortality association or the effect of hot days on the ozone-mortality association. Only few directly examined multiplicative interactions. Most studies demonstrated a clear (over-additive) effect of ozone and heat.

Climate and meteorology affect air quality by impacting the distribution of air masses and atmospheric chemistry. Temperature and pollutants jointly affect health. Meteorological conditions determine whether aerosols accumulate over time or if they are removed from the atmosphere by dry or wet deposition (Seinfeld and Pandis, 1998; van Zanten et al., 2017). The occurrence of inversion days in southern and central Austria, as well as inversion intensities and magnitudes, decreased from 1961 to 2017 due to climate change (Hiebl and Schöner, 2018). This should imply reduced particulate matter (PM) pollution and better air quality. On the other hand, an increasing amount of secondary aerosols can be expected because NH<sub>3</sub> is an important precursor of fine particulate formation in the atmosphere (Spangl et al., 2006; Xu and Penner, 2012; Anderl et al., 2016; Backes et al., 2016a, 2016b), and increasing NH<sub>3</sub> emissions were found in Austria from 1990 to 2017 (Anderl et al., 2019). Since the fraction of secondary aerosols in total aerosol concentration can be high (Spangl et al., 2006), a better understanding of PM pollution is required to reduce this major environmental health threat (Hendriks et al., 2013). Overall, the effects of climate warming on the change in PM pollution will vary by region and season and there is still a lot of uncertainty in model estimates. However, a recent global study (Gomez et al., 2023) estimates that annual PM2.5 levels will be about 10  $\mu$ g/m<sup>3</sup> higher in about 100 years due to GHG-induced warming for Europe and especially for the Alpine Region. While these increases are moderate compared to other parts of the world, notably Africa and South America, the agreement between the different models applied is comparatively good for the European region.

Due to hotter and drier weather events, forest fires are becoming more frequent, more intensive, and widespread. Wildfire smoke is characterized by high levels of PM with very small particle sizes and gases such as nitrogen oxides (Youssouf et al., 2014). Exposure to these pollutants in areas surrounding a wildfire can cause acute effects (e.g., irritation of the eyes, respiratory mucosa) or lead to the onset or exacerbation of acute and chronic respiratory diseases. Furthermore, exposure to forest-fire smoke is associated with cardiovascular diseases and mortality. Currently, wildfires are not yet perceived as a grave risk to health in Austria, but this could change in case of severe climate change (see Cross-Chapter Box 1 and Section 2.2.2).

Low air quality and climate change also increase allergies (Bergmann, 2016) and exposure to pollutants (ozone) can enhance allergic reactions to pollen (Berger et al., 2020), while pollen is predicted to increase in number, exposure duration, and allergenic potential (see Section 2.2.2). Climate change (mostly due to temperature and  $CO_2$  increases) will have an effect on plant growth and pollination. Plants will react to heat stress and ozone, by producing more pollen or producing pollen more often, and the single pollen particles will often also present more allergenic proteins on their surface. Climate change will also affect the introduction and spread of new and potentially highly allergenic plants, like ambrosia (see Sections 1.6.4 and 2.2.2). So-called thunderstorm asthma is increasingly observed: Pollen break up and allergens are released (Thien et al., 2018; Fuchsig and Scholl-Bürgi, 2022; Huang et al., 2022).

Following the 'One Health' approach, the effects of climate change on the health of plants, animals and humans can be assessed together. In that regard, we refer to the detailed discussions in Section 2.2. Especially cultural ecosystem services have a direct bearing on human health and well-being.

In addition to immediate health risks from drowning, trauma, and infection, floods pose longer-term health risks from damaged homes, the need to relocate, or moldy homes. While data on immediate health effects are available in Austria, long-term effects including psychological and mental health impacts are still severely understudied. Therefore, only general conclusions can be drawn based on data from other countries, which indicate that severe effects might occur, especially on mental health (Fernandez et al., 2015; Matthews et al., 2019; Charlson et al., 2021; Schürr et al., 2023).

Hurricanes, tropical storms and tropical cyclones are not frequent in Europe. However, wind speeds of extra-tropical cyclones in Europe can reach 'hurricane force' according to the Beaufort wind scale (Goldman, 2014). Storms can affect human health either through direct effects, including injury or death during the storm from flying debris or falling trees, or indirect effects following the storm including reduced access to health care or medicines, exposure to carbon monoxide (from gasoline powered electrical generators), electrocution, and psychological impacts (Goldman, 2014).

Health related risks of droughts can be direct due to lack of (clean) water for drinking or hygiene measures. However, the danger of drinking water shortage due to droughts is currently not yet a major issue in Austria. Indirect effects are the adverse effects on agricultural outputs, which mainly lead to socio-economic consequences for producers.

While in Austria there is still room for adaptation to mitigate the above-mentioned consequences for health and livability, increasingly large parts of the world do not have this option. When an increasing number of people around the world face impoverishment, loss of resources for livelihood, and a mounting vulnerability to – among others – infectious diseases, this might ultimately also have consequences for public health in Austria (APCC, 2018) (*limited evidence, high agreement*).

#### Health impacts of adaptation and mitigation measures

The number of climate change mitigation policies and adaptation measures increased in recent decades. Many adaptation measures will be effective immediately and contribute to reducing human vulnerability to climate variability. The benefits of other mitigations carried out today will not be evidenced until the coming decades because of the long residence time of GHG in the atmosphere.

Measure	Description	Mitigation potential	Health benefits
Healthy diet	Reduced meat consumption in accordance with dietary recommendations	Up to 20 % reduction in individual footprint due to food intake (see also Table 2.3. and Section 2.3.2.)	20 % reduction of deaths caused by dietary factors
Physically active mobility for Vienna, Graz and Linz	Green Mobility: Targets for the modal share of trips for the years 2020/2025 are almost achieved. E.g., Vienna: Share of car trips reduced from 40 to 28 %	290,000 tCO <sub>2</sub> eq reduction	27 deaths per year per 100,000 prevented
	Green Exercise: Targets exceeded; e.g., Vienna: Further reduction to 18 %	530,000 tCO <sub>2</sub> eq reduction	58 deaths per year per 100,000 prevented
Insulation of houses	Better insulation of houses will reduce heat loss in winter and, if well done, can mitigate heating up in summer	Reduced energy needed both for heating and for cooling	Increased comfort and well- being. Especially relevant for low-income households in poor housing conditions
Health care sector	More targeted use of antibiotics (antibiotic stewardship)	Antibiotics use for human health needs accounts for 1 ktCO <sub>2</sub> eq/yr	Indiscriminate use of antibiotics increases the risk of antibiotic resistance and has lots of adverse side effects
	Reduced use of metered dose inhalers (MDI)	Propellants from MDI account for 13 (2005) to 25.9 (2015) ktCO <sub>2</sub> eq/yr	Dry powder inhalers can be used without propellants and their use is often easier

CCBox 2 Table 1	Estimated health gains from selected mitigation measures (APCC, 2018; Weisz et al., 2020
	Estimated median gams norm servered mitigation medsares (ni ee, zo ro, meisz et al, zozo

Mitigating climate change is often seen as an activity with local costs and global benefits. However, many mitigation measures also have a range of significant health benefits, both locally and in the near term, mainly because they also reduce other airborne emissions, for example,  $NO_x$  and particulate matter (APCC, 2018; Wolkinger et al., 2018; Rieder et al., 2023). Examples of these win-win situations include a shift from motorized to physically active traveling (walking, cycling, and partial use of public transport), healthy (less meat) diets, or better insulation of houses (CCBox 2 Table 1).

However, when considering climate change mitigation and adaptation measures as a whole, unintended consequences might arise, resulting in an actual drawback for human health. Therefore, it might be difficult to create a coherent framework for planning interventions. For example, burning wood instead of gas or oil for heating might in certain cases reduce the carbon footprint (see Section 2.3.2), but will often increase particulate matter air pollution (Haluza et al., 2012). The technology for low emission furnaces is available, but current limit values still allow much higher emissions (Umweltbundesamt, 2023b). Replacing fossil fuels with biofuels is likely to reduce particulate matter pollution but increase emissions of nitrogen dioxide (Hutter et al., 2015). The balance of these effects remains controversial. For Austria, especially the AQUELLA project (Bauer et al., 2007, 2009; Jankowski et al., 2009) has demonstrated the important contribution of smoke from biomass burning to particulate air pollution confirming more global concerns (Sigsgaard et al., 2015).

Air pollution contributes substantially to the burden of disease. Acute health effects of air pollution have also been demonstrated with Austrian data examining different endpoints and pollutants (Moshammer et al., 2013, 2019, 2020; Neuberger et al., 2013). Chronic effects have been demonstrated in Europe by the large ESCAPE project (Beelen et al., 2015), which also included a large cohort from Vorarlberg. The ESCAPE data have been recently merged with those from other studies in the ELAPSE study to demonstrate health effects even at low levels of pollution (Brunekreef et al., 2021). Two recent overviews (in German) of long-term and short-term health effects of air pollutants have been provided by Weinmayr (2023) and Moshammer (2023), respectively.

Another example is the impact of urban design strategies (see Section 3.2.3), while urban greening substantially influences the distribution of solar energy in urban areas, various health co-benefits are expected from these activities, such as reduced obesity and cardiovascular disease and improved mental health through increased physical activity, cooling spaces (e.g., shaded areas), and social connectivity (*limited evidence, high agreement*). On the other hand, other adverse health effects such as pollen allergies have to be taken into account, as allergies might increase with increased urban green space.

Adaptation strategies and measures can be planned directly to improve health. Climate change adaptation measures that are not primarily intended to improve health may have health-related side effects. Any adaptation measure (e.g., in forestry, agriculture, energy, industry, tourism) should therefore also be assessed regarding possible, unintended (negative) side effects on population health.

#### The carbon footprint of the Austrian health care system and mitigation options

The Austrian health care industry as a whole is responsible for 7 % of the carbon footprint of Austria<sup>1</sup> (Weisz et al., 2020). Their estimates for 2010 are presented in CCBox 2 Figure 2. While medical goods and services account for about half of the total carbon footprint, direct energy use and transport account for roughly the other half, with direct energy use declining and transport emissions increasing. Reducing most of those emissions can be addressed with mitigation measures covered in Section 2.3 (food production), Chapter 3 (buildings, infrastructure, transport of goods and persons or information and communication technology) and Chapter 4 (industry, energy).



**CCBox 2 Figure 2** The carbon emissions of the Austrian health care system in 2010 according to Weisz et al. (2020): Pharmaceuticals, medical goods and services, other goods and services, food and catering, facility services, energy (for heating and cooling of buildings mostly), transport (of patients, staff and visitors), and others (including financing, administration, etc.) in hospitals, medical retail, ambulatory services and other parts of the health care system.

<sup>1</sup> This value encompasses the emissions of the goods and services consumed by the health system, and overlaps with various sectors.

However, there are at least some possible mitigation measures that are specific to the health care system. For example, some anesthetic gases (Koch and Pecher, 2020; Richter et al., 2020; Lehmann and Sander, 2021) and propellants of asthma inhalers (Rauchenwald et al., 2020; Wilkinson and Woodcock, 2022) have a large climate forcing potential. Weisz et al. (2020) estimate the greenhouse gas emissions of all anesthetic gases to be 21.4 ktCO<sub>2</sub>eq in 2015. At the same time, metered dose inhalers account for 25.9 ktCO<sub>2</sub>eq. While the former displayed a declining trend, emissions from inhalers increased (nearly doubled) since 2005. In addition to the emissions associated with the manufacturing of these medical devices (which are accounted for in the supply chain), the climate effects of the pharmaceuticals when released into the air must be considered.

Some health care providers are starting to implement reduction strategies. Some hospitals have declared a phase-out of the narcotic gases with the highest greenhouse potential, or are planning to install carbon filters to capture and reuse these gases. Dry-powder inhalers are often a viable alternative to the use of propellants in metered-dose inhalers. At present, it is difficult to establish industry-wide emission reduction targets, because data on current consumption and trends in Austria are not available.

A high-quality, sustainable health care system that focuses on prevention rather than disease would improve human health at scale, while further reducing the carbon footprint of health care. Sustainable health care would reduce unnecessary and emission-prone diagnostic and therapeutic interventions, with X-rays being a prime candidate for the former and antibiotics for the latter. Primary prevention and quality of life provisions are intrinsically more carbon neutral than the current emphasis on end-of-life, high-tech treatments and their associated supply chains.

Today, however, the evolution towards a sustainable health care system in Austria is in its infancy. Even with 'health check-ups' (now: preventive check-ups), the main aim is to discover illnesses in good time, not to maintain health. Our current health care system's orientation, facilities and financing are still primarily geared to the care of the sick and the treatment of illness (Green et al., 2002; Fiandaca, 2017). While treatment of illness is important, preventive care, avoid-ance of illness and maintenance of good health are not truly realized in the current system. While Austria proclaims goals such as 'Health in all Policies' as a strategy to introduce emphasis on prevention (Fonds Gesundes Österreich, 2015), the necessary budgetary commitments are still lacking (Rechnungshof Österreich, 2023).

## 2.2.3. Impacts of climate change on cultural ecosystem services and related benefits

Cultural ecosystem services are broadly defined as the non-material benefits that result from human interaction with the natural environment (Reid et al., 2005). Human well-being greatly depends on cultural ecosystem services, as they provide numerous non-material benefits that contribute to physical and mental health (Nowak-Olejnik et al., 2022; Schirpke et al., 2022). For example, landscapes, terrestrial and freshwater ecosystems, and natural features provide recreational activities, aesthetic experiences, inspiration and spiritual enrichment, and cultural identity (Table 2.A.1). While studies indicated the importance of maintaining cultural and (semi-)natural landscapes for the provision of cultural ecosystem services (Schirpke et al., 2018b; Lavorel et al., 2020; Schirpke et al., 2022), only few studies have explicitly addressed effects of climate change on cultural ecosystem services (Inglis and Vukomanovic, 2020; Mucioki et al., 2021; Schirpke and Ebner, 2022). Most studies focused on recreational activities, while aesthetic, symbolic and inspirational values, education and identity have been less examined (Schirpke et al., 2022, 2024). In the following, only those cultural ecosystem services are presented, for which climate change impacts were reported in the literature relevant to Austria, while there are research needs for educational, inspirational, spiritual, and existence values (Schirpke and Tasser, 2024; Schirpke et al., 2024).

### Impacts of climate change on recreational activities

Recreational activities comprise all kinds of outdoor activities practiced in natural and semi-natural ecosystems for recreational purposes. In Austria, particularly mountain areas and lakes offer manifold opportunities for outdoor recreational activities (Figure 2.6) (Schirpke et al., 2018a; Pröbstl-Haider et al., 2021), such as hiking, cycling, climbing, swimming, boating, skiing, and many others. Global warming has predominantly negative impacts on recreational activities (the economic relevance for the Austrian tourism industry is described in Sections 4.3.3 and 4.3.2). There is *high confidence* that warmer temperatures reduce opportunities for winter sport activities such as alpine skiing and cross-country skiing due to a shorter winter season and insufficient snow cover (see Sections 1.3, 4.3.3 and 7.4), particularly at lower elevated sites (Tervo-Kankare et al., 2013; Steiger et al., 2021). Due to a projected decrease of snow days to 50 % under a GWL 2.0°C scenario, it is expected that winter sport activities will shift upwards to higher elevations (Endler and Matzarakis, 2011). Moreover, people indicated lower intention to engage in recreational winter sport activities under a scenario with low natural snow (GWL not specified) (Frühauf et al., 2020). In summer, the increased potential for mountain hazards due to melting permafrost, such as rock falls and debris flows, may restrict the use of alpine trails and climbing routes, and hence, recreational activities such as hiking, mountaineering, climbing, and mountain biking (Pröbstl-Haider et al., 2016) (medium confidence). Moreover, the increased number of disturbances in mountain forests limits the access to forests due to increased salvage logging and increases the vulnerability of touristic infrastructure due to a reduced protective effect of forests (Lecina-Diaz et al., 2024; Müller, 2011). In contrast, warmer temperatures will extend the season for water-based activities (e.g., swimming, bathing, boating) at large lakes (see also Sections 1.4.2 and 4.3.3): Under GWL 2.0°C, a 40 % increase in warm days with temperatures above 25°C, and a doubling of hot days with temperatures above 30°C are expected by 2050 (Pröbstl-Haider et al., 2021). The increasing number of warm days will also increase recreational opportunities at mountain lakes (Schirpke and Ebner, 2022). However, climate change leads to more frequent low water levels (Soja et al., 2013), which reduces opportunities for water-based recreation (Wieland and Martinis, 2020). Warmer water temperatures also promote the occurrence of algae, cercariae, cyanobacteria (Gallina et al., 2011), which reduce the quality of water-based activities and increase health risks, especially in lowland lakes. Hot summers and increased evaporation will make lake water warmer and often more saline. This will increase the risk for swimmers to acquire wound infections from Vibrio cholerae (Le Roux et al., 2015; Hirk et al., 2016).



**Figure 2.6** Potential climate change impacts on the provision of cultural ecosystem services (classification of impacts based on terciles). Impact values were derived by identifying potential impacts on multiple underlying indicators based on current literature, multiplied with ecosystem service supply and weighted with the spatial distribution of temperature and precipitation under GWL 5.0°C. Please note that the potential impacts on all cultural ecosystem services are negative. For methodology, see Schirpke and Tasser (2024). Numbers indicate major biogeographical regions (from <a href="data.qv.at/katalog/dataset/naturraumzonen">data.qv.at/katalog/dataset/naturraumzonen</a>): (1) Pannonian plains and hills, (2) Southern Alpine foothills, (3) Northern granite and gneiss highlands, (4) Northern Alpine foothills, (5) Northern Alps – east, (6) Central Alps – southeast, (7) Klagenfurt Basin, (8) Southern Alps, (9) Central Alps – central, (10) Northern Alps – west.

In general, there is *high confidence* that climate change induces a shift in the spatial and temporal demand for recreational ecosystem services (Vierikko and Yli-Pelkonen, 2019; Pröbstl-Haider et al., 2021; Wilkins et al., 2021). There will be an upshift in elevation where possible, i.e., high-mountain regions will become more attractive in summer due to cooler temperatures and in winter due to less uncertainty in snow cover compared to lower elevation mountain destinations (see Sections 1.3 and 4.3.4). Moreover, shoulder seasons, i.e., periods between peak and off-peak seasons, will become more attractive for recreational activities than hot summer periods.

### Impacts of climate change on cultural and heritage value

Quantitative assessments of climate change impacts on heritage, aesthetic, and symbolic values under global warming levels are not specified, as the literature mostly consists of qualitative studies or studies using worst-case scenarios.

Landscapes and natural features provide various cultural and heritage values, that are closely connected to the management of landscapes, especially cultural landscapes. As a result of climate change, it can be expected that agricultural use will be intensified at lower elevations where the climate is favorable or where irrigation is possible, while the spatial extent of forests will increase at higher elevations due to an upshift of the tree line and accelerated natural reforestation processes on abandoned grasslands. Both developments lead to a decline of cultural landscapes and thus reduce related spiritual and cultural values (Figure 2.6) (Pröbstl-Haider et al., 2015; von Heßberg et al., 2021) (limited evidence, high agreement). Furthermore, there is high confidence that, under increasing drought conditions, it may not be possible to maintain traditional agroforestry systems that are of high cultural value, such as larch meadows or orchard meadows (Zoderer et al., 2016; Flinzberger et al., 2020), because larch trees will lose their habitat at lower altitudes (Obojes et al., 2018), while the productivity of orchard trees will be reduced (Hammel and Arnold, 2012).

### Impacts of climate change on aesthetic value

The aesthetic value of a landscape depends on both biophysical characteristics of the environment and human perceptions (Daniel, 2001). Usually, this ecosystem service focuses on the visual characteristics of the landscape such as individual types of ecosystems, natural features, and spatial landscape patterns (Schirpke et al., 2016). While some studies worldwide explicitly address climate change impacts on aesthetic values, and report negative impacts, for example, induced by species shifts (Inglis and Vukomanovic, 2020), or invasive species and the disappearance of snow patches (Mameno et al., 2022), there are only few studies in the Austrian or Alpine-wide context. In general, there is high confidence that climate change impacts are strongly linked to changes in land-cover composition arising from global warming or changes in precipitation patterns (Figure 2.6). For example, the expansion of forests at higher elevations leads to a decline in aesthetic value due to a reduction of the viewing depth and lower preferences associated with forest ecosystems (Schirpke et al., 2016, 2017). Moreover, a higher number of forest disturbances (e.g., wind throw areas, burnt areas) might reduce the aesthetic value of forests (Fleischer et al., 2017). The retreat of glaciers and snowfields reduces the aesthetic value of landscapes due to the loss of highly appreciated landscape elements (Schirpke et al., 2021). It also leads to an increase in debris, dry and desolate environments, which is perceived negatively by most visitors (Salim et al., 2021). In lakes, direct and indirect effects of climate change are expected to have negative impacts on visual water quality (high confidence), mainly due to higher levels of eutrophication and declining ecosystem quality, depending on the level of human use and the specific social-ecological conditions of the lake (Schirpke and Ebner, 2022). While remote lakes and high elevation lakes are more exposed to changing climatic conditions, human use is more likely to impact cultural ecosystem services at easily accessible and lower-elevation lakes (Ebner et al., 2022).

### Impacts of climate change on symbolic value

Wild plant and animal species can provide a source of inspiration for people in different ways and may be used for symbolic representations (Schirpke et al., 2018b; Rüdisser et al., 2019). Climate change is expected to reduce suitable habitats for plants and animals that are symbolic for the European Alps, such as edelweiss, gentian, Alpine ibex, chamois, and marmot (Figure 2.6, see Section 1.6) (*medium evidence, high agreement*). For example, habitats of alpine plant species that occur in specific ecosystems and cold climates such as the edelweiss will be reduced or lost due to climate change (Grabherr, 2009; Maghiar et al., 2021). Symbolic animals such as chamois suffer from a decline in forage quality as a consequence of global warming (Reiner et al., 2022; Corlatti et al., 2023). In the case of marmots, the thinner snowpack, which reduces the insulation effect of snow during the snow season, as well as changes in social and demographic structures due to a longer summer season, both have negative effects on the juvenile survival (Rézouki et al., 2016; Glad and Mallard, 2022).

### Impacts of climate change on health related to cultural ecosystem services

Human well-being greatly depends on cultural ecosystem services, which provide important non-material benefits that contribute to mental and physical health (Nowak-Olejnik et al., 2022; Schirpke et al., 2022). A change of ecosystems also has a variety of effects on their health-supporting services. Most research on the recreational effects of ecosystems focuses on either greenery or biodiversity of natural surroundings, with some, but still insufficient supporting recreational influences of both. On the one hand, green landscapes have been shown to be associated with mental health and well-being, via various processes (Markevych et al., 2017): Greenery reduces harmful influences on (mental and physical) health, such as air pollution, noise, or heat, but it also provides a setting for building new capacities (via physical activity and social cohesion) and is also linked to the restoration of capacities. It has been shown that attention restoration (Kaplan, 1995) and a resulting slight enhancement of cognitive performance happens especially well in nature surroundings (Stevenson et al., 2018), although the results can sometimes be ambiguous (Ohly et al., 2016). Natural green surroundings also reduce stress (Ulrich et al., 1991) and even pain (Ulrich, 1984), thus contributing to well-being, mental health and recovery. Overall, and after controlling for socio-economic factors, living in green surroundings is associated with not only better physical health, but also better mental health (Maas et al., 2009; Triguero-Mas et al., 2015), although these studies are usually not able to identify causal mechanisms (Sandifer et al., 2015, for an overview). In addition, greenery leads to a cooler microclimate, which is important for reducing aggression and violence: Recent studies have shown increases in violent crime with rising temperatures (Hu et al., 2017; Miles-Novelo and Anderson, 2019). Whether nature and greenery indirectly promote social cohesion is a matter of discussion, but at least no direct connection between greenery and social cohesion could be found (Triguero-Mas et al., 2015). Similarly, no direct connection between greenery and physical activity could be confirmed (Triguero-Mas et al., 2015), but it has been shown that exercise, assessed by walking, is more restorative in a natural environment than in an urban one (Hartig et al., 2003).

Whether biodiversity has benefits for mental health beyond those of greenery or nature is rather denied, with overall ambiguous results (Marselle et al., 2021). Overall, the association of biodiversity with (mental) health and recovery can be seen as equivalent to that of greenery (via reducing harm, restoring capacities and building capacities), but it can also show some harmful effects on human health, e.g., via allergens (Marselle et al., 2021). Regarding specifics of biodiversity, single studies suggest that species diversity is positively related to well-being, especially bird diversity (Cameron et al., 2020; Methorst et al., 2021), but partly also plant richness (Fuller et al., 2007).

# 2.3. Transformations and measures in the land use system to mitigate and adapt to climate change

The land use system not only contributes to GHG emissions, but also provides carbon sinks when carbon stocks in biomass, soils, and harvested wood products increase. Human activities may strive to mitigate emissions, increase sinks, and/or maintain system functions by adaptation strategies.

The following subchapters assess activities aimed at realizing such mitigation potentials, as well as activities aimed at adapting to climate change. The subchapters build upon the Avoid-Shift-Improve approach (Cross-Chapter Box 4), which gained particular attention in the IPCC AR6 (Creutzig et al., 2022b). For reasons of simplicity, we merge 'shift' and 'avoid' activities, as it is not always straightforward to unambiguously separate these two with regard to the land system. Thus, we discern two sections on the effects of Avoid-Shift-Improve measures namely:

Section 2.3.1, detailing on 'improve' activities, which aim at an increased efficiency (of material or energy use) of providing goods or services. These activities aim to reduce emissions without focusing on changing the quantity produced and consumed of a certain good or service. They may also aim at optimizing outputs under conditions of climate change, by adapting to climate change.

Section 2.3.2 detailing on 'avoid and shift' activities that either reduce resource extraction and consumption, or provide services with different, less emission-intensive resources. An example for the first would be the reduction of food losses, for the second the shift from animal proteins to plantbased proteins. In addition, Section 2.3.3 provides an assessment of nature-based climate solutions for adaptation and mitigation, including the technical mitigation potentials of existing land-system wide CDR approaches (Cross-Chapter Box 6).

Details on accompanying effects, such as the transformation from land to built-up area, are covered elsewhere in this report (e.g., Section 3.2.2). Table 2.3 consolidates the knowledge on mitigation potentials of land-based avoid-shift-improve activities in Austria. Mitigation potentials have a large uncertainty range, e.g., due to limited knowledge on implied transformation processes or feasibility parameters. Caveats, limitations, and research gaps, which are specific to a measure are listed below the corresponding row in the table. General limitations which hold across all measures are discussed below the table.

 Table 2.3
 Overview of mitigation potentials of transformations in the Austrian land use systems. Detailed mitigation potentials of each considered study are listed in 2.A.16.

Total potential* and sub-effects	Evidence on biogeophysical and technical feasibility and barriers	Evidence on economic potential, socio-cultural and institutional barriers and enabling conditions	Mitigation potential compared to nowadays (ca. 2020) [MtCO <sub>2</sub> eq/yr]	Mitigation potential compared to reference scenario** [MtCO2eq/yr]	Confidence	Number of independent studies (independent models) for Austria
-------------------------------------	---	--	---	---	------------	---

\*The total potential is the sum of the sub-effects

\*\* Referring to the reference scenario in each respective study – see limitations below and Appendix A2.16 for the list of studies.

Mitigation potentials represent estimated reductions in greenhouse gas emissions and increases in carbon sinks.

Positive values represent reduced emissions or increased sinks. Negative values represent increased emissions or reduced sinks.

#### Improve measures:

Change in tree species composition: Switching to native non-coniferous species in forests – See Section 2.3.1

Long-term potential (Average ca. 2020–2150)							
Total potential	low	low	-9.5 to -10.6	+1.1 to +1.9	low	2 (1)	
Reduced processing emissions	n.a.	n.a.	n.a.	n.a.	n.a.	0	
Effect on the sink in ecosystems (forest)	low	low	-0.1 to -1.2	+4.1 to +4.9	low	2 (1)	
Effect on the sink in harvested wood products (HWP)	low	low	-2.8	-1.3	low	2 (1)	
Effect on avoided emissions via substitution	low	low	-6.6	-1.7	low	2 (1)	
Short-term potential (Avera	ige 2020–2060)						
Total potential	low	low	-3.7 to +0.3	-0.3 to +1.1	low	2 (1)	
Effect on the sink in ecosystems (forest)	low	low	+0.2 to +4.2	-0.5 to +0.9	low	2 (1)	
Effect on the sink in harvested wood products (HWP)	low	low	-0.7	+0.1	low	2 (1)	
Effect on avoided emissions via substitution	low	low	-3.2	+0.1	low	2 (1)	

Caveats: The effect on the forest sink materializes only under a high GWL (between GWL 4.0°C and GWL 5.0°C) and primarily beyond 2080. The interaction with increased disturbances from climate change is poorly known. Potentials of forestry measures are assessed as yearly averages over a long timeframe, up to 130 years, due to important non-linear dynamics (e.g., sink saturation and decarbonization of alternatives to wood). The comparison to 2020 (column 4) shows the difference in sinks/avoided emissions between the scenario with changed tree species composition, in a long-term average, and the value for 2020 in the reference scenario from the respective study. The increased emissions of the long-term average compared to 2020 are due to the decreasing trend in C sinks and avoided emissions, independently of the taken measures. The change in tree species composition moderately compensates for this overall trend (column 5). The potentials outlined here refer mostly to technical potentials from a current perspective. Time periods differ by studies.

Total potential* and sub-effects	Evidence on biogeophysical and technical feasibility and barriers	Evidence on economic potential, socio-cultural and institutional barriers and enabling conditions	Mitigation potential compared to nowadays (ca. 2020) [MtCO2eq/yr]	Mitigation potential compared to reference scenario** [MtCO2eq/yr]	Confidence	Number of independent studies (independent models) for Austria
-------------------------------------	---	--	--	---	------------	---

Shorter rotation cycles: Immediate rapid rejuvenation of old stands by shortening the end-use age, to limit the risk of storm damage – See Section 2.3.1

Long-term potential (Average ca. 2020–2150)								
Total potential	low	low	-14.1 to -14.6	-2.6 to -3.4	low	2 (1)		
Effect on emissions	n.a.	low	n.a.	n.a.	n.a.	0		
Effect on the sink in ecosystems (forest)	low	low	-7.6 to -8.1	-2.4 to -3.2	low	2 (1)		
Effect on the sink in harvested wood products (HWP)	low	low	-1.9	-0.4	low	2 (1)		
Effect on avoided emissions via substitution	low	low	-4.6	+0.2	low	2 (1)		
Short-term potential (Avera	ige 2020–2060)							
Total potential	low	low	-1.5 to -10.1	-0.7 to -6.8	low	2 (1)		
Effect on the sink in ecosystems (forest)	low	low	-1.1 to -9.7	-4.4 to -10.5	low	2 (1)		
Effect on the sink in harvested wood products (HWP)	low	low	+0.9	+1.7	low	2 (1)		
Effect on avoided emissions via substitution	low	low	-1.3	+2	low	2 (1)		

Caveats: In total over all effects, this measure increases GHG emissions over the considered time period, mainly due to the reduced sink in forests. The interaction with increased disturbances from climate change is poorly known. Potentials of forestry measures are assessed as yearly averages over a long timeframe, up to 130 years, due to important non-linear dynamics (e.g., sink saturation and decarbonization of alternatives to wood). Time periods vary between studies. The potentials outlined here are mostly biogeophysical potentials, assessed from a current perspective.

#### Other Adaptive Forest Management – See Section 2.3.1

#### No data available, as no study exists for Austria.

Including: Fuel management strategies like selective thinning, prescribed burning, mechanical clearings, increasing structural diversity of stands, continuous forest cover, reduce game density, reduce stand density, etc.

#### Total technical mitigation measures in agriculture – See Section 2.3.1

Total potential	low	low	n.a.	+0.3 to +1.6	medium	2
Effect on agricultural emissions	low	low	n.a.	+0.2 to +1.2	low	1
Effect on agricultural soils and C sequestration	low	low	n.a.	+0.07 to +0.49	medium	1

Caveats: Aggregate of the technical measures in agriculture, including those quantified in more detail below (agricultural inputs, livestock farming, rewetting peatlands under agricultural use). (i) Effect on agricultural emissions: Aggregate of technical mitigation measures targeting CH<sub>4</sub> and N<sub>2</sub>O emissions, including manure management, enteric fermentation, soil emissions, and fertilizer application, and assuming 80–100 % dissemination. (ii) Effect on agricultural soils and C sequestration: Including reduced tillage, non-inversion tillage on arable land (no data), broadening of crop rotations – increased soil biomass via field forage crops and legumes (no data), broadening of crop rotations – increased protein supply via forage legumes, implementation of catch/cover crops and greenings, use of manure instead of nitrogen fertilizer, amendment of organic substances such as biochar (no data), reducing cultivation on organic soils and reduced draining, organic farming with increased carbon sequestration measures, agroforestry (increasing tree cover in agricultural areas, silvopasture) (no data), and hedgerows. The effect on agricultural soils and C sequestration reflects mainly increased C sinks, but cannot be fully separated from decreased emissions and substitution effects of the above listed measures (about 0.1 to 0.2 MtCO,eg/yr). Data comes from a meta-analysis, hence the *medium confidence*.

Total potential* and sub-effects	Evidence on biogeophysical and technical feasibility and barriers	Evidence on economic potential, socio-cultural and institutional barriers and enabling conditions	Mitigation potential compared to nowadays (ca. 2020) [MtCO2eq/yr]	Mitigation potential compared to reference scenario** [MtCO2eq/yr]	Confidence	Number of independent studies (independent models) for Austria

#### Reducing GHG emissions by the rewetting of peatlands under agricultural use – See Section 2.2.2

Total effect on GHG emissions	low	low	+0.001 to +0.388*		low	2
----------------------------------	-----	-----	-------------------	--	-----	---

\*Upper estimate for potential (expert judgement) assuming that rewetting of 50 % of 31,000 ha agriculturally used peatland area (Hogl et al., 2023) is possible, with a mean emission reduction of 25 tCO<sub>2</sub>eq/ha/yr (BMLRT, 2022).

Caveats: Estimates assume optimal restoration, with zero emissions after rewetting. Estimates don't include additional carbon sequestration due to regained carbon sink function after rewetting. Estimation of average emission reduction is based on German data; Austrian emission data are lacking. Knowledge on the amount of peatland area and future conditions for rewetting is low.

Reducing GHG emission	Reducing GHG emissions from agricultural inputs (see sub-measures below the table) – See Section 2.3.1								
Total effect on emissions, sinks and substitution	medium	medium	n.a.	+0.05 to +0.330	medium	1			

Caveats: The study is a meta-analysis, hence the *medium confidence*.

Including: Reducing mineral fertilizers – increasing legume forage, reducing emissions from machinery upstream, improving fuel efficiency, reducing emissions from pesticides, systemically applied precision agriculture techniques, reducing N-fertilizer application in lanes, reduction of N-losses by close-to-ground application of slurry, reduction of energy demand in agricultural buildings

### Reducing GHG emissions from livestock farming; excluding the reduction of livestock (see sub-measures below the table) – See Section 2.3.1

Total effect on						
emissions, sinks and	medium	medium	n.a.	+0.015 to +0.8	medium	1
substitution						

Caveats: The study is a meta-analysis, hence the medium confidence.

Including: Optimization of housing, increase of pasture feed intake, reduction of the protein content in feed rations of pigs and poultry, reduction of the protein content in feed rations of bulls, reduction of the protein content in feed rations of cows, refining herd management and feeding protocols, solid manure instead of liquid manure systems (no data), utilization of excrements via biogas plants, feed additives (no data), increase livestock yield and efficiency, extend the productive life, improve animal health.

### Measures to reduce GHG emissions of processing, transport, and other processes in agricultural and wood supply-chains – See Section 2.3.2

No data available, as no study exists for Austria.

Increasing self-sufficiency for agriculture and forestry products and other trade related measures; effect on global GHG emissions embodied in consumption – See Section 2.3.2 and 2.A.12.

No data available, as no study exists for Austria.

#### Shift and avoid measures:

Shift towards healthier and plant-based (EAT-Lancet) diets and according restructuring of agricultural production (incl. reduction of livestock products) – See Section 2.3.2 (protein transition, calories), and in Section 2.3.3 (enabled afforestation/ renaturation)

Total potential	high	medium	+6.3 to +11.1	+5.0 to +15.4	high	6
Effect on agricultural emissions	high	low	+3.6 to + 3.7	+2.5 to +5.6	high	6
Effect on the sink in ecosystems (afforestation / renaturation on freed area	low	low	+2.7 to +7.4	+2.5 to +9.8	low	3
Effect on avoided emissions via substitution (additional production of bioenergy and bio-based materials on freed area	n.a	n.a	n.a	n.a	n.a	0

Total potential* and sub-effects	Evidence on biogeophysical and technical feasibility and barriers	Evidence on economic potential, socio-cultural and institutional barriers and enabling conditions	Mitigation potential compared to nowadays (ca. 2020) [MtCO <sub>2</sub> eq/yr]	Mitigation potential compared to reference scenario** [MtCO2eq/yr]	Confidence	Number of independent studies (independent models) for Austria
-------------------------------------	---	--	---	---	------------	---

Caveats: The sensitivity to the characterization metric needs to be evaluated. Socio-cultural transformation pathways are not described in the studies. Research gaps exist on transformation capacities, enabling conditions, and the contribution of institutions. Little is known about the regional-level and farm-level impacts and changes required for a shift from animal-based to plant-based food production. It is unclear to what degree this shift can be achieved through adjustments in farming practices and what fundamental changes to production systems would be needed in addition to that. Likewise, there is a clear research gap on the degree to which such a shift would induce farm closures and changes in rural structures. Impacts on ecosystem services, biodiversity, and landscapes are partly unclear. Available studies assume continuous population and emission growth. Hence, the mitigation potential comparing to the current situation (column 4) is smaller than when comparing to a reference scenario (column 5).

#### Halving food losses - See Sections 2.3.2 and 2.3.3

Total potential	medium	medium	+0.2 to +1.1	+2.1 to +4.7	medium	2
Effect on agricultural emissions	medium	medium	-0.4 to +0.5	+1.6	medium	1
Effect on the sink in ecosystems (afforestation / renaturation on freed area	low	medium	+0.6	+0.5 to +3.1	low	2
Effect on avoided emissions via substitution (additional production of bioenergy and bio-based materials on freed area	n.a	n.a.	n.a.	n.a.	n.a.	0

Caveats: Data on food waste and losses are poor (research gap). Estimating quantities and avoidability is a major challenge. Halving food losses appears as a plausible assumption. Research gaps on transformational pathways exist. Data on avoidability at various processing stages in the value chain are lacking. The increased agricultural emissions compared to present day are due to the assumed population growth (column 4). Halving food waste compensates for this overall trend (column 5).

#### Reducing the overall use and harvest of wood to build up C stocks in forests - 'Vorratsaufbau' - See Sections 2.3.2 and 2.3.3

Long-term potential (Average ca. 2020–2150)						
Total potential	low	low	-2.6 to -4.2	+7.0 to +8.4	low	2 (1)
Effect on the sink in ecosystems (forest)	low	low	+1.4 to +4.4	+9.3 to +11.1	low	2 (1)
Effect on the sink in harvested wood products (HWP)	low	low	-2.2 to -3.4	-0.7	low	2 (1)
Effect on avoided emissions via substitution	low	low	-6.4 to -0.6	-1.6 to -2.0	low	2 (1)
Short-term potential (Avera	ige 2020–2060)					
Total potential	low	low	+1.4 to +5.4	+6.2 to +9.1	low	2 (1)
Effect on the sink in ecosystems (forest)	low	low	+4.7 to +12.0	+8.7 to +12.3	low	2 (1)
Effect on the sink in harvested wood products (HWP)	low	low	-1.9 to -2.8	-1.1 to -1.2	low	2 (1)
Effect on avoided emissions via substitution	low	low	-0.5 to -4.7	-1.4 to -2.0	low	2 (1)

Total potential* and sub-effects	Evidence on biogeophysical and technical feasibility and barriers	Evidence on economic potential, socio-cultural and institutional barriers and enabling conditions	Mitigation potential compared to nowadays (ca. 2020) [MtCO2eq/yr]	Mitigation potential compared to reference scenario** [MtCO2eq/yr]	Confidence	Number of independent studies (independent models) for Austria
Caveats: The interaction with increased disturbances from climate change is poorly known. Scenarios have 2010 as base year and assume harvest of 20 million m <sup>3</sup> /yr in the reference scenario and of 15 million m <sup>3</sup> /yr on average between 2020 and 2150, excluding logging residues and natural losses ('Erntefestmeter'). Converted to 'Vorratsfestmeter' (i.e., including logging residues and natural losses), these volumes correspond to 30 million m <sup>3</sup> /yr in the reference scenario and 25 million m <sup>3</sup> in the scenario with reduced harvest ('Vorratsaufbau'). Note that for 2020 the harvest volume assumed in the Care4Paris study (BEW, 2020) for the 'Vorratsaufbau' is close to the harvest level in 2016–2021 reported in the latest Austrian						

Forest Inventory (25 million m<sup>3</sup> 'Vorratsfestmeter'). The comparison to 2020 (column 4) shows the difference between the sinks/avoided emissions between the scenario with reduced harvest, in a long-term average, and the value for 2020 in the reference scenario from the respective study (Braun et al., 2016; BFW, 2020). The increased emissions of the long-term average compared to 2020 are due to the decreasing trend in C sinks and avoided emissions, independently of the taken measures. The reduction in harvest strongly compensates for this overall trend (column 5). The potentials outlined here refer mostly to technical potentials from a current perspective. Time periods differ by studies.

Increase cascadic uses of wood: Prioritizing long-life harvested wood products over energy – See Section 2.3.2						
Total effect	Low	low		+0.1 to +0.5	low	2
Caveats: The comparability of the two studies is low. One study compares a scenario where policies either favor material uses or energy. The other study shows two case studies in which chips are used for boards and lignorellulosic ethanol for ethylene instead of energy						

Table 2.3 provides an overview of the mitigation potentials described in the scientific literature specifically applicable to Austria. The direct comparability of the studies is limited and any interpretation of the figures shown needs to consider the respective boundary conditions, baselines, methodological assumptions and resulting limitations, especially when attempting to provide information about combined potentials. It is strongly recommended to refer to the individual studies (listed in 2.A.16) and to examine their individual limitations, potential methodological differences, and research gaps more generally:

Limitations of individual approaches concern the quality including temporal, spatial, and categorial levels of aggregation and uncertainty ranges of the input data used. The boundary conditions and scope of the respective studies may differ from the purpose of describing mitigation potentials in this table, and therefore the adequacy may be limited. Potentials may differ whether measured against current conditions or future situations, in the latter case also strongly depending on projections and scenario assumptions. Some figures are based on rough, back-of-the-envelope calculations, which carry the risk of oversimplification. Moreover, much of the outlined potentials are theoretical, based on biogeophysical feasibility alone, without factoring in technical, socio-cultural, or political constraints that will affect real-world implementation, although some of the outlined strategies do take these aspects into account. In terms of economic feasibility, additional uncertainties arise from the use of price valuations in studies from different years. All studies assume the greenhouse gas characterization metric GWP100, which is also the metric adopted in the UNFCCC. Choosing a different metric such as GWP20 or GWP\* would affect the results (see 2.A.10).

Methodological differences may also be the result of different original objectives of the studies used. This may lead to differences in the underlying assumptions (e.g., different market prices of end products, energy prices, political willingness for implementation) and to discrepancies between the baseline parameters against which mitigation potentials may be measured. Mitigation potentials derived from different studies may be mutually exclusive and thus not additive, adding complexity to the combination of individual mitigation potentials. Especially for measures requiring land area, careful consideration of land use priorities and constraints in terms of availability and additionality is needed. Sub-measures are often aggregated into larger measures, and sometimes grouped into bundles, but this is often done without a consistent or systematic approach. Furthermore, interaction effects between mitigation measures can result in unwanted impacts on other economic sectors. Some of the emissions reduction strategies might be effective immediately, while others have a delayed effect. The most prominent example is carbon sinks, which can play a crucial role in both short and long-term mitigation, but are effective over decades (e.g., 30-150 years) and have considerable uncertainties in their magnitude and stability. Such factors are treated inconsistently in different studies. Carbon stocks are reversible and depend on future activities. As carbon sinks can be affected not only by human intervention but also by natural biophysical processes, the impacts of climate change

itself need to be considered, again leaving margins for different and contradicting approaches.

Research gaps exist in our understanding of the processes leading to the emission and sequestration of GHGs on land. Obviously, the table above only accounts for mitigation potentials that are known and have been addressed in research studies. Due to the complexity of biological processes and the small flux densities over large areas, many factors determining these fluxes remain poorly quantified and may not have been adequately studied. This is even more the case for factors that modify the fluxes (mitigation measures). Known examples are the emission behavior of organic soils or the feedback of climate change on many of the processes. Other research gaps relate to the synergies and trade-offs of mitigation measures with other sustainability objectives or their effectiveness under climate change.

### 2.3.1. 'Improve' measures

### Adaptive forest management – 'Improved' forest management to mitigate and adapt to climate change

Austrian forests, which had been carbon sinks varying around -15 MtCO<sub>2</sub>/yr for several decades, lately have been turning into a carbon source (+5.4 MtCO<sub>2</sub>/yr for 2023, which meant a source also on a five-year average: +0.2 MtCO<sub>2</sub>/yr, see Figure 2.2 and Table 2.A.2). This is the result of the net effects of removals of atmospheric CO<sub>2</sub> through biomass increment and carbon losses caused by natural disturbances, forest management (timber harvest, thinning) and changes in carbon stocks in soils, influenced by inflows and decomposition of organic matter. To maintain or enhance this sink, particularly in the light of expected future climate impacts, adaptive forest management (AFM) approaches to climate change propose three main principles (Bolte et al., 2009; Brang et al., 2014; Spathelf et al., 2018): (i) Increasing resistance to disturbances by improving the stability of forest stands, (ii) promoting resilience of forests for a rapid restoration of forest functions to desired conditions after disturbances, and (iii) promoting the adaptability of forest stands by selecting appropriate species during regeneration, changing species shares during mixture regulation, increasing the diversity and diversification of forests to facilitate the transition to new forest states (Spathelf et al., 2018). Optimizing the rotation length is also discussed for AFM, depending on site characteristics and market demands, to reduce the windthrow risk due to a limitation of the top height reached (Schelhaas et al., 2015) or reduce the risk of a higher mortality of not adapted tree species and to keep the increment rate on a high level (adaptation and focus on substitution). However, on average between 2020 and 2150, shortening rotation cycles would decrease the carbon sink in forests by 2.4 to 3.2 MtCO<sub>2</sub>eq/yr compared to a reference scenario. In addition the sink in harvested wood products will be decreased by 0.4 MtCO<sub>2</sub>eq/yr and the avoided emissions from substituting fossil fuels and abiotic materials will increase by 0.2 MtCO<sub>2</sub>eq/yr (see Table 2.3 and 2.A.16 for individual studies) (BFW, 2020; Ledermann et al., 2022). Changing tree species composition to species that are expected to perform better under a changed climate (Schelhaas et al., 2015) can avoid risks associated with specific pests, diseases or disturbances (Seidl et al., 2009) or drought-intolerant species at the stand or landscape level. According to the results of the Care4Paris project, changing the composition of species to native non-coniferous species would increase the carbon sink in forests by 4.1 to 4.9 MtCO<sub>2</sub>eq/yr on average between 2020 and 2150 compared to a reference scenario. On the other hand, the sink in harvested wood products would be reduced by 1.3 MtCO<sub>2</sub>eq/yr and the avoided emissions from the substitution of fossil fuels and abiotic materials by 1.7 MtCO<sub>2</sub>eq/yr (see Table 2.3 and 2.A.16 for individual studies) (BFW, 2020; Ledermann et al., 2022). Advanced planting with shade-tolerant tree species in mature stands can support tree species change, as harvesting mature stands and regeneration them with new species might result in lower forest carbon stocks in the short term (Kindermann, 2021). In this context, assisted migration strategies transfer more adapted or adaptive tree species (including non-natives) to regions outside their current natural range that are characterized by a suitable climate, thus artificially extending the range distribution of more resilient tree species (Gömöry et al., 2020; Mauri et al., 2023). Assisted migration of tree species has the potential to increase carbon sinks and might allow larger harvest volumes that remain below increment, but uncertainties are large and risks are manifold (Pötzelsberger et al., 2020; Chakraborty et al., 2024). Fuel management strategies like selective thinning, prescribed burning (along railroad embankments, on grasslands or meadows), mechanical clearing or grazing with goat, sheep, or donkey can reduce the fuel loads and thus the probability of large and stand-replacing forest fires (Ascoli et al., 2023; Müller et al., 2020).

There is strong evidence that tree species richness and high genetic diversity positively influence the adaptive capacity of forests to climate change (Brang et al., 2014; Spathelf et al., 2018). Increasing the structural diversity of stands by retaining ecosystem legacies (e.g., seed trees, deadwood, stand remnants) after disturbances will enhance the restoration capacity (Spathelf et al., 2018). In addition, the heterogeneous ownership structure and different management strategies help to promote diversely structured forest ecosystems in Austria. There is high confidence that management for continuous forest cover can help maintain a favorable microclimate in forest stands and provide continuous natural regeneration (Spathelf et al., 2015; Vacek et al., 2019; Zellweger et al., 2020). However, the density and distribution of game populations is a key factor that in many cases jeopardizes a successful natural regeneration without protective measures (Schodterer, 2022). Moderate reductions in stand density have been shown to effectively mitigate drought stress (Elkin et al., 2015), but the effects vary across forest sites, species, and context (Castagneri et al., 2022). As forest management adjusts the tree species composition to increase the resistance and resilience of ecosystems while maintaining high levels of productivity, the supply to the timber, pulp and paper market will change in the long run (increasing the share of deciduous tree species). This development will entail transformation of the timber industry (Klein et al., 2016; Lantz et al., 2022). For more details about climate change adaptation, see Chapters 4 and 5 in the APCC Special Report on land use and climate change (Baumgarten et al., 2024; Kraxner et al., 2024).

Emissions linked to energy use along the supply chain of wood (see Section 2.1.1) are mostly influenced by road maintenance, biomass harvesting techniques, forwarding and biomass transport and depend on the considered tree species, age classes, rotation length, wood assortments, and site quality parameters (Klein et al., 2016). Mitigation measures that focus on these parameters are: Efficient transport, reduced fuel consumption or shifting to alternative energy sources, improved road and skidding line network.

### Agricultural management to mitigate and adapt to climate change

Emission reductions are linked to the activities responsible for the relevant emissions (see Section 2.1.1). In agriculture, an important mitigation option is to increase SOC (Table 2.1). The literature, as summarized by Kraxner et al. (2024), shows a wide range of measures, such as non-inversion tillage on arable land, changes in crop rotations including increased biomass supply from forage grasses and legumes, the integration of catch/cover crops and intercropping and greenings, the use of manure instead of mineral nitrogen fertilizers, or the addition of organic substances such as biochar. Systemic crop management approaches have been suggested to combine benefits of individual measures (Kraxner et al., 2024).

Adding N fertilizer to soils drives  $N_2O$  emissions from soils, so reducing fertilizer application also reduces emissions. Minimizing N inputs may take advantage of more efficient fertilizers such as nitrification inhibitors or precision agriculture to reduce  $N_2O$  emissions by up to 38 % (Winiwarter et al., 2018), possibly with minor yield losses (Foldal et al., 2024). The application of biochar has also been described to efficiently reduce  $N_2O$  emissions (Liu et al., 2018; Borchard et al., 2019). The potential mitigation effects of systemically applied precision agriculture techniques (PAT) have a wide range (*limited evidence, medium agreement*). High investment costs, high demand for expertise, and a lack of training materials are barriers to the use of PAT (Kraxner et al., 2024).

The whole production chain is also relevant (Osterburg et al., 2013; Grassauer et al., 2021). For Germany, there are reports on the trickle-down effects of reducing N inputs from mineral fertilizer, e.g., through taxation (Isermeyer et al., 2021). Optimizing feed rations and livestock protein intake can reduce emissions associated with feed production and N fertilization (Pierer et al., 2016). A mitigation potential of >30,000 tCO<sub>2</sub>eq/yr is estimated at costs below EUR 100/t for pig farming and between EUR 100-200/t for ruminant farming (Fritz et al., 2022; Sinabell et al., 2023) (medium confidence). Increasing legume forage reduces the need for protein feed imports and synthetic fertilizers (Stagnari et al., 2017), potentially abating emissions by  $>15,000 \text{ tCO}_2\text{eq/yr}$ at costs between EUR 100-200/t (Fritz et al., 2022; Sinabell et al., 2023). Also, managing grasslands at medium intensity and increasing pasture feed intake can be cost-effective options, mitigating >45,000 tCO<sub>2</sub>eq/yr at costs below EUR 100/t (Lorenz et al., 2019a; Fritz et al., 2022) (medium confidence). In addition, more efficient use of machinery offers an economically viable potential of mitigating >7,500 tCO<sub>2</sub>eq/yr at costs below EUR 100/t (Fritz et al., 2022; Sinabell et al., 2023). For fossil fuel use, efficiency improvements and biofuels or electrified machinery can play a minor role, but no technique is available for substantial mitigation.

There are several mitigation options to tackle emissions from livestock farming (see also compilation by Kraxner et al., 2024). Feed additives can reduce enteric  $CH_4$  formation in cattle by up to 30 % (no cost assessment available) (Lewis

AAR2

et al., 2013; Abecia et al., 2018; Van Wesemael et al., 2019). Indirect mitigation measures, such as refining herd management and feeding protocols, improving livestock management practices and animal health exhibit a mitigation potential of >20,000 tCO<sub>2</sub>eq/yr and costs below EUR 100/t. Extending the productive life of cows may further mitigate >20,000 tCO<sub>2</sub>eq/yr at costs below EUR 100/t (Dallago et al., 2021; Fritz et al., 2022; Sinabell et al., 2023). An option with high mitigation potential, but far-reaching socio-economic consequences for livestock producers and the entire value chain, is the (further) reduction of livestock numbers based on assumptions of reduced food losses, reduced protein intake or a shift towards plant-based diets and agriculture (see Section 2.3.2). Nevertheless, Austrian livestock production systems appear rather efficient in terms of GHG emissions in the international context due to higher N efficiency and lower N<sub>2</sub>O emissions, the dominance of dual-purpose breeds in cattle production and hence the allocation of GHGs to milk and beef, lower use of imported feedstuffs which are loaded with large amounts of GHGs from land use change. Weiss and Leip (2012) calculate average Austrian GHG emissions from meat and dairy production per unit of output to be among the lowest in the European Union. For example, average emissions are about 1 kgCO<sub>2</sub>eq/kg for Austrian cows' milk compared to an EU average of about 1.3-1.7 kgCO<sub>2</sub>eq/ kg cows' milk and more than 2.5 kgCO<sub>2</sub>eq/kg for cows' milk from Cyprus. Mitigation measures applicable to all livestock species are the optimization of housing and farm manure systems (e.g., Hörtenhuber et al., 2010; Sajeev et al., 2018).

Austria is still a long way from the number of animals per farm or per area found in other industrialized countries. However, even in Austria, there is a trend towards concentration and intensification with increasing problems of odor nuisance (Weitensfelder et al., 2019; Oettl et al., 2022) and bio-aerosols, while ammonia is also a precursor of fine particulate matter in the atmosphere (Liu et al., 2023).

The adaptation of agriculture to climate change requires first and foremost adapting current land use. For arable land, important adaptation measures are, similar to mitigation measures, the broadening of crop rotations, the diversification of crops and the integration of catch crops (protecting soils from drying out) and undersowings, which contribute to humus build-up (measures are described in Jandl et al., 2024c). Quddoos et al. (2023) show that adaptation options help to considerably reduce the adverse effects of climate change for Austrian farms. Similar to cropland, grassland productivity might also profit from moderately warmer temperatures as long as sufficient water is available (Haslmayr et al., 2018; Schaumberger et al., 2019; Volk et al., 2021). However, this requires the adaptation of cutting frequencies and fertilization to changing climatic conditions. Changing climatic conditions might allow an increased use of alpine pastures, but negative effects on biodiversity must be considered.

Climate change may affect the spread of harmful organisms. Universal predictions of pest and disease dynamics and universal control strategies are not possible, but have to be developed specifically. One basic option is to diversify crop rotations (which, e.g., can stop the spread of the maize rootworm: Feusthuber et al., 2017; Falkner et al., 2019). Adequate crop rotation with early, effective weed control and a monitoring system throughout the growing season can prevent the establishment and spread of invasive plants (AGES, 2023b, 2023a). New technologies, such as drone technology, are being used to control Datura spec. in crops (Riegler-Nurscher et al., 2023). Useful organisms such as pollinators may be lost, but improved agricultural management, such as near-natural elements or habitats with appropriate forage plants in agricultural landscapes, can counteract such losses (Roberts et al., 2011; Zulka and Götzl, 2015).

To adapt to climate change-induced water shortage, numerous adaptation measures are discussed (Jandl et al., 2024c): Active and efficient irrigation (where water sources are available), soil-related measures such as reduced tillage, permanent ground cover, and windbreaks are well-established means. Furthermore, fertilization should be adapted to changing nutrient dynamics, water balance and seasonal fertilization requirements. In practice, barriers to widespread implementation lie in farmers' risk attitudes and management practices, market mechanisms and institutional design (Mitter et al., 2019; Hanger-Kopp and Palka, 2022).

Climate change-induced summer heat waves and droughts affect growing conditions for crops. This requires the breeding and establishment of new crop species or varieties. In particular, plant breeding research particularly has to focus on heat and cold tolerance, efficient use of nutrients and water, and resistance to pests (VLK, 2019). In grasslands, the use of drought-resistant varieties is becoming increasingly important, which will also be reflected in breeding programs (Krautzer and Graiss, 2015). New crop species like chickpea (Neugschwandtner et al., 2013) or amaranth (Gimplinger et al., 2007) may become competitive. Rising temperatures, including milder winter conditions, allow the planting of winter forms of traditionally spring-sown crops like pea (Neugschwandtner et al., 2020), faba bean (Neugschwandtner et al., 2015, 2023a), and poppy (Neugschwandtner et al., 2023b). For winter wheat, facultative cultivars are a promising solution to avoid insufficient vernalization during very mild winter seasons (Koppensteiner et al., 2022).

Heat stress also poses a significant threat to livestock in terms of health, animal performance (e.g., feed conversion, daily weight gain, and milk production), well-being, and economic implications for farmers (Hörtenhuber et al., 2020; Herzog et al., 2021; Schauberger et al., 2021). The implementation of adaptation measures is crucial to reduce the adverse effects of heat stress in confined livestock facilities. Such techniques include cooling the inlet air, refining management protocols, and optimizing feeding strategies (Vitt et al., 2017; Mikovits et al., 2019; Schauberger et al., 2020; Herzog et al., 2021). For more details on climate change adaptation, see Chapter 4 of the APCC Special Report on land use and climate change (Baumgarten et al., 2024) and Smit and Wandel, 2006)..

The implementation of mitigation and adaptation measures at the local (farm) level requires adequate financial and technological resources, as well as the information and organization skills to manage these resources accordingly (Smit and Wandel, 2006); without these, adaptation success may be limited. Zeilinger et al. (2023) confirm such under-adaptation to climate change for Austrian farms in arable regions. Recent literature suggests that developing the adaptive capacity of farms is a promising policy tool to enhance adaptation (Vanschoenwinkel et al., 2020). Mitter et al. (2019) identify prerequisites for adaptation intentions of Austrian farmers, including awareness of effective adaptation measures, acceptance of personal responsibility for their farms, and positive adaptation costs.

### 2.3.2. 'Shift and avoid' measures

### Protein transition – shift towards healthy, plant-based diets

There is *high confidence* that the composition of food produced and consumed has a highly significant impact on per capita carbon emissions, driven in particular by the number of farmed livestock and the amount of related products (Jandl et al., 2024c). Past and recent trends in diets in Austria are presented in 2.A.8. Dietary choices impact decisions on agricultural land use, livestock management, processing, transport, and ultimately disposal, and are therefore important for climate protection. Consequently, major potentials from a transformation of diets result from reducing the total amount of products produced and consumed (including total food intake and food waste), as well as shifting the type of products (degree of processing, transport requirements, seasonality) and the kind of products consumed (i.e., dietary choices such as vegan vs. livestock-based products).

This section tackles the role of choices in agricultural food products (protein transition, i.e., a shift to diets with decreased shares of animal products), with other options being discussed in Section 2.3.2. There is high confidence that dietary change cannot be understood as the sole responsibility of end consumers, and that achieving dietary change through a bundle of measures involving all actors across the food system could facilitate an effective and just protein transition (Harwatt, 2019; Willett et al., 2019; Vermeulen et al., 2020; Roux et al., 2022; Mempel et al., 2023; Penker et al., 2023; Voigt et al., 2024) (see 2.A.9 and Table 6.1 for policy instruments related to diets across the food system). Life cycle assessments typically show large differences in the GHG emissions of agricultural products, with livestock-based products having substantially higher emissions than plantbased products when related to the energy or protein content of the food as a functional unit (Table 2.4). This is true for both estimates per unit of kcal and per unit of protein (Tilman and Clark, 2014). The main reasons include the low feed-to-product conversion rates with correspondingly large land requirements and, especially in the case of ruminant livestock, CH<sub>4</sub> emissions from enteric fermentation. Consequently, ruminant products typically embed the most emissions among all animal products for typical GWP100 estimates. A discussion about GWP100 vs. GWP\* can be found in 2.A.10.

Austria is among the world leaders in per capita meat consumption (including bones, tendons parts, and trim fat), with an annual per capita intake of 34.2 kg of pork, 12.8 kg of poultry, and 10.4 kg of beef in 2021 (Statistik Austria, 2022e). The study by Frey and Bruckner (2021) highlights that beef consumption accounts for 38 % of the greenhouse gas emissions from Austria's agricultural product consumption. These data, combined with the high emissions per unit of livestock products, indicate significant per-capita greenhouse gas emissions from the typical Austrian diet (*high confidence*).

Shifting towards the so-called Planetary Health Diet developed by the EAT-Lancet Commission (Willett et al., 2019), i.e., a dietary standard with high shares of vegetarian food that takes into account planetary boundaries and human health, would reduce the demand for animal products and correspondingly restructure agricultural production. It would yield a mitigation potential of 5.0 to 15.4 MtCO<sub>2</sub>eq/yr (compared to individual reference scenarios), including reduced agricultural emissions and enabled afforestation (Schlatzer and Lindenthal, 2020; Lauk et al., 2022; Sun et al., 2022; Le Noë et al., 2023a; Preinfalk et al., 2024) (see Table 2.3 and 2.A.16 for the individual studies). From a consumption-based perspective, shifting to the Planetary Health Diet (EAT-Lancet) would reduce GHG emissions embodied in Austria's consumption of food by ca. 9.3 MtCO<sub>2</sub>eq/yr, or 1.02 tCO<sub>2</sub>eq/cap/yr (Semba et al., 2020). A shift to vegan diets increases the mitigation potential by ca. 2 MtCO<sub>2</sub>eq/yr compared to a planetary health diet (Le Noë et al., 2023a). At the European level, Westhoek et al. (2014) estimate that a 50 % reduction in meat, dairy and eggs from a 2004 reference diet would reduce GHG emissions by 25–40 %.

Note that the diet proposed by the EAT-Lancet Commission (Willett et al., 2019) is beneficial in terms of a reduced carbon footprint and for the health of consumers (Ekmekcioglu et al., 2018). There is *high confidence* that shifting towards plant-based diets, especially reducing red and processed meat, reduces the risk of various diseases such as colorectal cancer and cardiovascular diseases (Chan et al., 2011; Yip et al., 2013; Bouvard et al., 2015; WHO, 2015; Farchi et al., 2017). The risk of colorectal cancer increases by 17 % for every 100 g of red meat consumed per day and by 18 % for every 50 g of processed meat consumed per day (Bouvard et al., 2015).

Furthermore, the production and consumption of agricultural commodities, such as livestock products, feed, or oil crops for food and bioenergy, is the world's leading driver of deforestation, prompting both climate change and the transmission of zoonotic diseases from wildlife to humans (Faust et al., 2018; Pendrill et al., 2022). Averting epidemics of wildlife-borne infectious diseases requires a socio-ecological transformation, a redesign of the global food system and a transition to diets that are plant-based or low in animal products (Wegner et al., 2022).

Meat substitutes. Livestock-based meat as protein source can be replaced by several substitutes, such as legumes and processed plant-based meat (PBM) (He et al., 2020), insect-based meat (IBM), and (in vitro) cultivated meat (CM) (Sinke et al., 2023). These innovations are poised to capture a significant share of the global meat market, estimated at 60 % by 2040 (Gerhard et al., 2019; He et al., 2020). Despite optimistic market and investment scenarios, regulatory, social, economic, ecological, and agronomic challenges (e.g., Codex Alimentarius Austriacus) have to be solved for these products (Eker et al., 2019; Kim et al., 2020; Moran and Blair, 2021; Noguerol et al., 2021; Siddiqui et al., 2022). Table 2.4 compares the environmental impacts of conventional meat (beef, pork, and chicken) and various meat substitutes, based on Life Cycle Assessments. Note that these assessments are based on current data and refer to relatively new processes, so there are large uncertainties and significant room for improvement. Meat substitutes exhibit notable ecological advantages, offering a viable strategy for transitioning away from high-impact beef production towards more environmentally sustainable meat options. Notably, the elevated energy requirements associated with CM and PBM production underscore the imperative for utilizing green energy sources to mitigate  $CO_2$  emissions (Sinke et al., 2023).

Table 2.4 Comparison of the environmental impact of conventional meat (beef, pork, and chicken) (Nijdam et al., 2012; Willett et al., 2019; Sinke et al., 2023), cultured meat (Sinke et al., 2023) plant-based meat (Nijdam et al., 2012; Saget et al., 2021), and insect-based meat (Smetana et al., 2016), related to 1 kg meat. The protein content of meat was assumed at 22 %. Values for plant proteins are adapted from Poore and Nemecek (2018). Negative values show a carbon sink.

Specific environmen- tal impact related to meat production	Energy [MJ/kg]	GHG Emissions [kgCO₂eq/kg]	Land use [m²/kg]			
Animal meat						
Beef	60–105	9.9–141	7–462			
Pork	28–40	4.4–12.3	8.8–16.5			
Chicken	24–26	2.2–9.9	5.1–8.8			
Processed meat alternatives						
Cultured Meat	25–481	1.9–24.8	0.19–6.9			
Plant-based meat	143	1.0–12.3	2–3			
Insect-based meat	2–64	0.17–3.2	0.07-1.03			
	Plant prote	eins				
Tofu		1.6–5.6	1.8–4.9			
Peas		0.6–1.7	2.8–14.2			
Other pulses		1.0–3.8	9.9–41.3			
Nuts		-3.7–3.8	4.5-26.6			

There are a number of trade-offs that need to be considered to enable a just transition to fewer livestock and animal products. While there are multiple alternative uses for arable land used for feed (e.g., food, fuel or fiber production), there are fewer alternative uses of permanent grasslands. Options include feedstock used in biogas plants or green biorefineries (Höltinger et al., 2014). A plausible option for marginal permanent grasslands is reforestation. Reforestation of grasslands is likely to increase total carbon sequestration in soils and biomass (Schirpke et al., 2017) but can create pressures on landscape quality or biodiversity, especially for extensively managed grasslands such as alpine meadows (Hussain et al., 2019). Dietary changes are also likely to have substantial impacts on farm incomes. Introducing the EAT-Lancet diet in the EU-27 has been estimated to increase agricultural income by 71 % in the long run (2050) due to shifts to high value-added crops such as fruits, vegetables and nuts (Rieger et al., 2023). In the short-run, however, there may be income losses (-12 % in Germany for partial EAT-Lancet implementations due to its specialization on livestock production) (Rieger et al., 2023). Similar impacts can be expected for Austria, although for permanent grassland regions, the shift to vegetables and fruits production is more costly and the income losses may be more pronounced. For further discussion on co-benefits and trade-offs related to dietary change and livestock reductions, see 2.A.10.

**Pets.** In Austria, there are approximately 2 million domestic cats and 766,000 dogs (FEDIAF EuropeanPetFood, 2023). These pets primarily subsist on meat-based diets (Swanson et al., 2013; Okin, 2017; Martens et al., 2019; Pedrinelli et al., 2022; Scherer et al., 2022). The IPCC report recognizes carnivorous pets as a significant factor in potential GHG mitigation, estimating their contribution at approximately 0.8 MgCO<sub>2</sub>eq/cap (Creutzig et al., 2022b). It is important to acknowledge that a substantial portion of pet food in the industrial sector is derived from animal by-products, which are inedible residuals obtained through animal rendering processes (Alexander et al., 2022; Acuff et al., 2021; Pedrinelli et al., 2022). Nevertheless, it should be

noted that not all pet foods exclusively rely on by-products, and that some of them may contribute to GHG emissions (*robust evidence, low agreement*).

### Non-protein related food measures – 'avoid' measures

Non-protein-related food measures refer to all changes beyond changing protein sources in human diets (see above) and switching to products produced under climate-friendly management and technologies (see above). These measures include reducing the total amount of products consumed, i.e., avoiding excess calorie consumption, reducing food losses, and changing the origin and processing degrees of food products. Figure 2.7 shows the food system emissions from the two groups, pre- and post-agricultural production and agricultural sector emissions, according to the national GHG inventory for the years 1990 and 2020 (Umweltbundesamt, 2022). Both groups of emission sources are in the same order of magnitude with diverging development patterns.

In 2019, 34.5 % of adults in Austria were overweight and 16.5 % were obese. The number of obese people has increased since 2006 (Statistik Austria, 2022c). Reducing overall food intake, especially the production and consumption of high-calorie products, would hence not only spare agricultural land and GHG emissions, but also help to reduce obesity and its associated diseases, such as ischemic heart disease, stroke, hypertension, type 2 diabetes



**Figure 2.7** Agricultural, pre-production (red box) and post-production emissions (ktCO<sub>2</sub>eq) of the Austrian food system by component in 1990 and 2020 (data: <u>fao.org/faostat/en/#data/GPP</u>; and Umweltbundesamt (2022)). A monochromatic version of the figure can be found in 2.A.11.

mellitus, colorectal cancer, postmenopausal breast cancer, endometrial cancer, kidney cancer and osteoarthritis (Lette et al., 2016). Estimates of the calorie intake in Austria range from 2,673 kcal/cap/day (3,100 kcal/cap/day including food waste) for the year 2022 (Pecher et al., 2024), to 3,117 kcal/ cap/day (3,610 kcal/cap/day including food waste) for the year 2015 (Lauk et al., 2022), and 3,429 kcal/cap/day (excluding food losses) for the year 2010 (Sun et al., 2022). A large-scale survey of Austrians in the year 2014-2016 estimated the total calorie intake at 1,815 kcal/cap/day (women) and 2,453 kcal/cap/day (men; both excluding food waste), which is lower than other estimates but still above dietary recommendations for about 50 % of the Austrian population (Rust et al., 2017). Most estimates are substantially above FAO recommendations for a balanced average Austrian diet of 2,533 kcal for the year 2020 (FAOSTAT, 2023).

Thus, **adapting current diets** has a strong mitigation potential (*robust evidence, medium agreement*) and leads to synergies with human health. However, it is not straightforward to disentangle reductions in calorie uptake ('avoid') from changes in diet composition ('shift', see above), because even calorie reduction scenarios require assumptions on individual diet components, i.e., whether they reduce diet components proportionally or disproportionally (e.g., stronger reductions of products with large GHG emissions).

Linearly translating the current to the recommended levels (i.e., -25 % in calories uptake) according to FAO for Austria would reduce agricultural emissions by ca. 2 MtCO<sub>2</sub>eq/ yr, or ca 0.2 tCO<sub>2</sub>eq/cap/yr. Hiç et al. (2016) estimate the average emissions related to food surplus are at 0.12 tCO<sub>2</sub>eq/ cap/yr. Kim et al. (2020) estimate country-level GHG emissions for different diets and show ranges for Germany from <0.5 to >1.5 tCO<sub>2</sub>eq/cap/yr. These ranges are comparable to the mitigation potentials estimated for the Planetary Health Diet suggested by the EAT-Lancet commission (see above and Table 2.3), which also includes a calorie reduction, e.g., to a maximum of about 2,600 kcal/cap/day in Semba et al. (2020). Other authors concur with these estimates, reporting substantial sustainability impacts from over-consumption similar to consumers' food waste (Alexander et al., 2017). Creutzig et al. (2022a), however, assume small mitigation effects from calorie reductions due low GHG emissions from high-calorie products such as sugar.

The potential for carbon sequestration on land no longer needed for agricultural production – or needed for nourishment of a growing future world population to prevent forest loss and other land use changes – need to be added to these emission savings and are shown in Table 2.3.

Food losses occur throughout the value chain, while food waste refers to the consumption stage (Parfitt et al., 2010). Neither can be fully avoided, but substantially reduced. Estimates of mitigation potentials are smaller than for changes in overall diets (robust evidence, medium agreement), but reducing food losses and waste is considered an important and 'no-regret' climate mitigation measure (Springmann et al., 2018). Halving food losses in Austria would result in a mitigation potential of 2.1 to 4.7 MtCO<sub>2</sub>eq/yr (compared to a reference scenario), including reduced agricultural emissions as well as a carbon sink due to enabled afforestation, assuming that these products would not have been produced (see Table 2.3 and 2.A.16 for the individual studies) (Le Noë et al., 2023a; Preinfalk et al., 2024). For comparison, Parfitt et al. (2010) estimate unavoidable household food waste in the UK to be around 20 %. Therefore, if similar reductions were to be applied to Austria, the mitigation potential would be 0.14 tCO<sub>2</sub>eq/cap/yr. In Austria food waste amounts to 1.1 Mt/yr, of which 49 % comes from private households, 16 % from farms, 11 % from food production, 16 % from gastronomy and 8 % from trade (Luck and Obersteiner, 2021). Springmann et al. (2018) estimate the effect of waste and loss reduction on emissions to be about one-fifth that of dietary changes (the latter including a change in diet mix). Similar results are reported by global food system studies for Europe (Kalt, 2015; Mayer et al., 2022; Röös et al., 2022) However, specific data on food losses and food waste in Austria are scattered. Scherhaufer et al. (2016) summarize studies showing total losses of about 1.1 million tons of biomass per year, which would be about 0.12 t/cap/yr, with about half of the losses occurring in households, and the rest distributed among agricultural production, processing and retail, and outside-home catering. Crippa et al. (2021) estimate end-of-life emissions in the food system to be about 1.39 MtCO<sub>2</sub>eq/yr, or about 6 % of Austrian food system emissions in 2015.

Transport between different locations of production, processing, consumption, and disposal along the food value chain offers further mitigation potentials. Crippa et al. (2021) estimate transport emissions in the Austrian food system at about 2.49 MtCO<sub>2</sub>eq/yr, or 12 % of total food system emissions. Only part of these emissions are avoidable, with typical examples of high GHG emissions and thus mitigation potentials for air-transport of fresh products. Local food systems are defined by small geographical distances between the steps of the food value chain. Despite potential emissions at other stages along the value chain (Schönhart et al., 2009). Therefore, their overall potential must be regarded as being small (*high confidence*).

### A sufficiency-oriented bioeconomy – 'shift and avoid'

Like bioeconomy strategies in general, the Austrian Bioeconomy Strategy implemented in 2019 aims to reduce climate impacts by shifting resource use away from abiotic (especially fossil fuel-based) towards a higher share of biomass use. In Austria, biomass currently accounts for only 23 % of direct material consumption (38 Mt/yr), the rest being mineral, fossil and metallic resources (BMK, 2020). There is high confidence that a shift towards a bioeconomy needs to be complemented by additional measures to avoid the degradation of domestic and international resources and ecosystems. A key reason is that Austria, typical of European countries, already appropriates a high fraction (56 %) of its domestic potential net primary productivity through biomass use and land conversion (Gingrich et al., 2015), and net imports of agricultural products contribute to additional pressures on ecosystems abroad (Haberl et al., 2012). Such additional measures need to ensure (a) sustainable production processes, (b) efficiency improvements in the provision of energy and material services and well-being, and (c) reductions in production and consumption of material and energy sources, i.e., sufficiency (Rockström et al., 2021; Creutzig et al., 2022a). The prioritization of individual elements in bioeconomy strategies varies across stakeholders (Hausknost et al., 2017; Stern et al., 2018). Scientific debates have focused on substitution and efficiency gains, while sufficiency options in a bioeconomy have received much less attention, especially with regard to their social feasibility (Schipfer et al., 2024).

Many factors influence the climate impacts and sustainability of wood harvest and use, illustrated by red triangles in Figure 2.8. The mitigation effect of wood use ultimately results from the interplay of three major elements: (i) Changes in C-stocks in forests due to wood harvest, (ii) changes in socioeconomic wood pools (harvested wood products), and (iii) the substitution of emission-intensive products and services with wood products (see also 2.A.13). There is a large body of literature assessing the interplay of these elements (Bhan et al., 2021; Cowie et al., 2021), which in principle suggests two different approaches (see Box 1.1 in Formayer et al., 2024b, for a detailed comparison). One quantifies actual (observed) GHG emissions over time (e.g., the UNFCCC emissions account scheme; see 2.A.5) (Jandl et al., 2024c), while the other compares different counterfactual scenarios and calculates substitution effects as well as 'carbon opportunity costs' (Norton et al., 2019; Erb et al., 2022; Fehrenbach et al., 2022; Soimakallio et al., 2022; Peng et al., 2023).

The climate effect of substitution is particularly complex to assess (Formayer et al., 2024b, Box 1.1). In direct comparisons, the substitution of biomass for fossil fuels appears to be disadvantageous for biomass in terms of emission reduction, due to its high carbon emission factor. The combustion of solid biomass releases about 29.9 tC/TJ of carbon stored in the biomass, compared to 17.2 tC/TJ for natural gas liquids, 20.2 tC/TJ for diesel oil, and 27.6 tC/TJ for lignite. These 'smoke-stack' comparisons do not take into account the renewable nature of biomass and therefore provide only partial insights (Cowie et al., 2021). More comprehensive accounts therefore consider all greenhouse gas fluxes in ecosystems and society. The comparative climate impact of biomass use then depends on the actually achieved reduction of the product or process under consideration (substitution), which often is jeopardized by overcompensation and rebound effects (York, 2012; Harmon, 2019; Leturcq, 2020), as well as on the comparative emissions impact of the respective material or energy use. The latter is shaped by both the emissions intensity of the substituted product or process and the emissions impact of additionally harvested biomass, both of which are variable over time (Buschbeck and Pauliuk, 2022). Global results show that in many cases, natural succession of ecosystems has a more beneficial climate impact than growing crops for bioenergy - second-generation bioenergy production compete over natural succession only when the displacement factor is very high, i.e., when very emission-intensive processes are substituted (Kalt et al., 2019). Once wood is harvested, extending the lifetime of wood products (e.g., using wood in buildings rather than burning it as bioenergy) reduces their climate impact, because carbon is stored in socio-economic structures for some time before it is released to the atmosphere (Allianz Nachhaltige Universitäten in Österreich, 2021). However, comparing the relative climate impacts of using versus not using wood is particularly complex, because long-term stock-flow dynamics in both ecosystems and socio-economic structures affect the net emissions impact (Seppälä et al., 2019). Results for Germany point to the high temporal variability of displacement factors for wood use (Buschbeck and Pauliuk, 2022). Thus, the relative performance of different wood use strategies depends on the temporal scope of analysis and planning (medium evidence, high agreement).



Figure 2.8 Carbon stocks and flows related to wood use and associated sustainability challenges. NPP refers to Net Primary Productivity.

There is no comprehensive assessment of displacement factors for Austria. Some researchers focus on the beneficial potential of providing biomass to substitute fossil fuels and abiotic materials and creating socio-economic carbon pools in harvested wood products (Braun et al., 2016; Jandl et al., 2018), while others argue that, particularly in the short term, wood harvest should be constrained to maintain net forest carbon sequestration due to the lack of readily available alternatives for carbon removal and to enhance biodiversity in forests (Bellassen and Luyssaert, 2014; Searchinger et al., 2018; Luick et al., 2021; Erb et al., 2022; Norton et al., 2022). For a description of contrasting positions and their main arguments, see Box on climate neutrality of wood use in Chapter 1 of the APCC Special Report on land use and climate change (Formayer et al., 2024b).

The overall global climate impact of wood use in construction and buildings is currently debated in the literature, with very optimistic (Mishra et al., 2022) and more cautious estimates coexisting (Johnston and Radeloff, 2019). However, there is consensus that from a climate change mitigation perspective, material uses of wood are favorable over energy uses (up to a factor 2 in a centennial perspective), as this result appears in practically all modeled conditions (Braun et al., 2016; Kalt et al., 2016; Fehrenbach et al., 2022). Furthermore, the 'cascading' use of wood is a powerful strategy, emphasizing that wood should first be used for material purposes - ideally over long periods - and only burned or pyrolyzed at the end of the product's lifecycle (high confidence) - although interpretations of what 'cascadic' means in practice vary across stakeholders (Ludvig et al., 2021). Estimates of the mitigation potential of using wood for materials rather than energy range from 0.1 to 0.5 MtCO<sub>2</sub>eq/yr (Braun et al., 2016; Kalt et al., 2016) (see Table 2.3 and 2.A.16 for individual studies). The lifetime of wood use varies across product categories (Braun et al., 2016), and increasing lifetimes can result in higher wood shares in a given product category, thus leading to C accumulation in socio-economic stocks. Longer lifetimes of wood products may also allow for lower annual wood demand and thus lower harvest volumes. Increased material wood use and increasing shares of long-lived products (as opposed to, e.g., wood bioenergy use) contributed to a long-term increase in the pool of harvested wood products of 115 MtC between 1830-2010 in Austria, while forest C pools increased by 240 MtC over the same period (Gingrich et al., 2016). For future trends, Braun et al. (2016) project a continuation of C accumulation in harvested wood product pools until 2100, corresponding to 180-350 MtCO<sub>2</sub>eq across scenarios. In their assessment, forest C stocks will increase in the scenario with less wood extraction, and decline in the scenario with higher wood extraction, while conversely, the effects of product substitution will be higher when more wood is used. Substitution effects are subject to time dynamics, and their mitigation effects decrease with the decarbonization of the economy (Buschbeck and Pauliuk, 2022; Jandl et al., 2024b).

In particular, the increased use of timber in construction, a sector which is currently responsible for 8 % and 14 % of Austria's  $CO_2$  and material footprint, respectively (BMK, 2020), can both substitute highly emission-intensive products (e.g., steel) and remain in use for particularly long lifetimes, i.e., decades or more. However, if increased construction timber cannot be derived from current domestic wood processing chains, it would require more imports. In global assessments, there is high confidence that a shift towards more construction timber would require large-scale land use changes and intensive tree plantations (e.g., Hertwich et al., 2019; Mishra et al., 2022). For Austria, Kalt (2018) estimates that by ramping up the share of wood in residential building construction to 50 % (from currently 22 %), the C-stock expansion in timber buildings could amount to 9.2 MtC by 2050, and further savings of 2-4.2 MtC would be realized by substituting emissions for the provision of mineral construction materials. Kalt (2018) considers such a scenario theoretically feasible and without impact on international forest C pools (due to increased harvest volumes), as the additional wood demand would remain far below Austrian sawn wood and panelboard exports. However, the absolute emissions impact of using wood in construction depends on the carbon impact of wood harvesting on forests and the emissions intensity of the construction sector (Buschbeck and Pauliuk, 2022); a recent study for Austria finds that the climate change mitigation effect of increased wood use in buildings is in most cases overshadowed by the forgone carbon sink (carbon opportunity costs) in forests (Maierhofer et al., 2024) (see Section 3.2.2). Currently, stakeholders from the wood sector do not consider an increase in the share of timber in residential construction to be likely, because of institutional constraints (Vihemäki et al., 2019). In addition to wood, other bio-based materials have been proposed for sustainable construction, such as mycelium, hempcrete, flax boards, straw, and reed (Yadav and Agarwal, 2021), but no assessment is available for Austria.

In addition to increasing the share of wood in construction, further emissions could be avoided in the residential sector through demand-side (or sufficiency) strategies (see Sections 3.3.1 and 3.2.2): In a global meta-study, Creutzig et al. (2022a) quantify the potential reduction of emissions through building design, size and use at 25 % (10–40 %) of global residential emissions. Kalt (2018) includes a scenario of moderately declining per-capita floor space, but the additional emissions savings potentials that might be realized through demand-side strategies, such as ending the decline in household size (Ivanova and Büchs, 2022) or fostering co-housing (Ivanova et al., 2020), have not been systematically quantified for Austria.

After its lifetime, timber from demolished buildings can again serve as raw material for material or energy use. Kalcher et al. (2017) quantify scenarios for Austria and find that, in their 'standard' scenario, the amount of wood available annually from the demolition of residential buildings alone will double by 2100. However, they also stress the poor data availability on residential buildings in Austria.

Bioenergy use has increased in recent years and is currently the largest renewable energy use in Austria, accounting for 247 PJ, or 17.3 % of total primary energy supply, while the fraction in domestic primary energy production is 47.5 % (BMK, 2023). Increased use of bioenergy, as the last step in the wood use cascade, and derived from agricultural residues, can play a certain role in achieving renewable energy targets (robust evidence, medium agreement). The APCC Special Report on land use and climate change (Jandl et al., 2024c) concludes that no reliable estimate on sustainable bioenergy potentials can be made at present due to the many trade-offs that are difficult to account for. In the existing literature, bioenergy potentials for Austria are estimated to reach 310-420 PJ by 2030-2050 (Krutzler et al., 2016; Anca-Couce et al., 2021), but these estimates do not explicitly consider system-level interactions between wood harvest and forest biomass stocks (Erb et al., 2016; Olsson et al., 2019; Pingoud et al., 2018). For instance, Kalt et al. (2019) show that at the global level, in terms of climate change mitigation, bioenergy from short rotation coppice is only competitive with natural succession on abandoned agricultural land over longer time frames (>30 yr) and with high displacement factors.

Measures to save energy, such as insulation in residential buildings, have been described as being favorable over increased bioenergy use (Peng et al., 2023; Jandl et al., 2024c). Through efficiency gains and insulation, the demand for wood energy can be reduced in absolute terms: Kranzl et al. (2018) estimate that by 2050, wood in residential final energy use will decrease by 25 % compared to 2020, as building renovation reduces overall energy demand for residential heating. There are knowledge gaps regarding the future supply of timber in Austria, integrating the complex interplay between, on the one hand, climate change and forest management choices that impact the composition of tree species, and on the other hand, technological change or adaptation in the milling industry that might change the use or processing potentials of different wood qualities. Furthermore, inter- and transdisciplinary collaboration will be required to robustly assess the potentials and impacts of sufficiency measures in the context of bioeconomy strategies (Schipfer et al., 2024).

#### 2.3.3. Nature-based (climate) solutions

### Nature-based solutions to increase the carbon sink of managed ecosystems

Nature-based Solutions (NbS) are actions that protect, sustainably manage and restore natural and modified ecosystems to provide multiple benefits for human well-being and biodiversity (Cohen-Shacham et al., 2016; Keesstra et al., 2018; Oral et al., 2020). NbS have large potential benefits for the mitigation of and adaptation to climate change (*high confidence*).

Under certain conditions, land ecosystems can take up carbon and thus counterbalance the accumulation of CO<sub>2</sub> in the atmosphere. Besides soils, woody biomass is the largest carbon pool in terrestrial ecosystems both in Austria and on a global scale (Jandl et al., 2024c). Woody vegetation, particularly in forests, has acted as a net carbon sink in Austria since the 19th century (Gingrich et al., 2016, 2022), due to both the expansion of area under woody vegetation and the increase of timber stocks in existing forest areas. Timber stocks in existing forests increase due to an increase in average age. This occurs in protected forests, but also in managed forests if management intensity is low (e.g., as observed in small-scale forest ownerships). There is high confidence that both the expansion of woody area and the increase in timber stocks continue to have potentials for future carbon sequestration, but are constrained by ecological limitations and land-use competition (Jandl et al., 2024c). At the same time, they may have co-benefits with biodiversity (Oettel and Lapin, 2021).

Adapting forest management can contribute to reducing the gap between actual and potential biomass stocks (high confidence) (see Chapter Box 2.2). The effect has been quantified globally (Houghton and Nassikas, 2018; Cook-Patton et al., 2020; Walker et al., 2022; Mo et al., 2023; Roebroek et al., 2023), but not for Austria. For Austria, Braun et al. (2016) find that in a scenario of reduced timber extraction from all forests, the overall cumulative mitigation potential in the forest sector would range between 600 and 800 MtCO<sub>2</sub> until 2100, compared to any scenario of increased wood use. This difference comprises an increase in the forest sink of 1100 to 1300 MtCO<sub>2</sub>, a decrease in the HWP sink of 100 to 200 MtCO<sub>2</sub>, and a decrease in avoided emissions from fossil fuels of 300 to 400 MtCO<sub>2</sub>. A similar finding is presented in Weiss et al. (2020) (see also Kirchmeir et al., 2022). In order to achieve a net mitigation effect, the reduction in wood harvest in Austria needs to coincide with strategies to

reduce the overall material and energy use and to limit imports (at the EU level), to avoid leakage of wood harvest to other countries and an increased use of other materials and fuels (Meyfroidt and Lambin, 2009; Meyfroidt et al., 2022). According to the Weiss et al. (2020) study, between 2020 and 2150, the contribution of the forest-based sector to national GDP under a reduced harvest scenario would be about 80 % of the contribution under a reference scenario. This corresponds to ca. EUR 1.77 billion less per year, for an average mitigation effect of 7 MtCO<sub>2</sub>/yr over the same period, i.e., ca. EUR 250/tCO<sub>2</sub> (Weiss et al., 2020).

Expanding the forest area in Austria is theoretically possible. Forests currently cover 48 % (2016/2021) of the domestic land, compared to 38 % in 1935 according to the Forest inventory (Waldinventur). Since 1981/1985, forest area has increased mainly due to the expansion of non-managed forests, by 0.135 Mha as opposed to 0.023 Mha for managed forests (Haszprunar et al., 1989). In addition, there are areas covered by trees that are not counted as forest. Due to competition with other land uses such as agriculture, the realistic potential of forest expansion appears to be limited, if demand for agricultural area is not reduced (Kraxner et al., 2024). See Table 2.3 for mitigation potentials of afforestation enabled by a protein transition and reduced food losses. The extent to which forest expansion is considered in climate-change mitigation scenarios is sometimes not explicitly quantified (e.g., Kirchner et al., 2016). But it can be assumed that the C sequestration effect of afforestation is small for time frames up to 2040. Increasing the area under trees in agricultural and urban settings, e.g., through agroforestry and urban trees, could contribute to sequestering additional C in managed ecosystems, with reported values for aboveground carbon storage ranging from 11-27.4 tC/ha in cities and up to 67.5 tC/ha in agroforestry systems (Nowak and Crane, 2002; Strohbach and Haase, 2012; Dorendorf et al., 2015; Bertsch-Hoermann et al., 2021; Golicz et al., 2021). If adequately implemented, such measures can achieve co-benefits for health, biodiversity and climate change adaptation (APCC, 2018; Jandl et al., 2024c) (medium evidence, high agreement). Biodiversity impacts can be positive when natural ecosystems such as mires and forests are restored, and negative when extensively-used agroecosystems are intensified or replaced by plantations (see Cross-Chapter Box 6). Expanding fast-growing tree plantations on previous cropland is one way to increase carbon pools, as they accumulate biomass very rapidly (Gonçalves et al., 2021) and increase the average carbon density per unit area compared to cropland. This can be done for example with en-
ergy plantations such as coppice forests of pure even-aged stands of fast-growing species, managed with a high density in very short rotations, sometimes fertilized and/or irrigated (Stojanović et al., 2017). Trade-offs with food production are possible. Highly pure or mixed even-aged forests managed only for carbon sequestration without considering the quality of timber and associated higher market prices (e.g., applying thinning from below, not removing forked or damaged trees of poor quality, maintaining wood of trees species with low market price) often do not meet the management objectives of forest owners. Compensation payments to forest owners have been shown to increase interest in biomass production for carbon sequestration (Ovando et al., 2019; Winkel et al., 2022).

Agroforestry has a medium potential to mitigate climate change (medium evidence, high agreement). Agroforestry systems currently cover 1.9 % of used agricultural land in Austria, dominated by sylvopastoral systems (Den Herder et al., 2017). Converting subalpine grasslands to sylvoarable systems can result in carbon sequestration, but reduces harvest due to reduced biomass growth under tree cover: In the Austrian Eisenwurzen, a scenario analysis found sequestration rates of 3.4 tC/ha/yr by 2050 while yields were more than halved (Bertsch-Hoermann et al., 2021). There is no assessment of biomass carbon pools in agricultural hedgerows for Austria, but a study for the Netherlands (Van Den Berge et al., 2021) concludes that trees in hedgerows show high and persistent increments of 0.7-4.3 tC/km/yr in stems alone, with basal areas of 22.1-44.9 m²/km. Their exclusion hence represents a major gap in national greenhouse gas inventories. However, no time-frame is given for this estimate. Wenzel et al. (2023) quantify the effect of hedgerows on SOC in top soils in Upper Austria, and show that SOC increases under hedgerows for decades, albeit at declining rates, resulting in higher SOC of 16-106 MgC/ha and 35-119 MgC/ha under hedgerows aged 1-30 and 31-70 years, respectively.

There is an ongoing debate, *with medium evidence* and *low agreement*, whether it is more advantageous to combine forest production and pasture (by cattle) on the same area (forest pasture) or separate the two land uses by creating open pure pasture areas and allowing the surrounding forests to become increasingly dense and enhance the carbon sink (Stöckli, 2013). On the one hand, the heterogeneous forest structure typical of forest pastures has positive effects on species and landscape diversity, as well as on tourism (Schirpke et al., 2019; Anderle et al., 2022). On the other hand, the forest pasture reduces the growth and yield of the

forests, hampers the successful natural regeneration of tree species, and can cause damage to the residual trees, making them vulnerable to pests and diseases (Noack et al., 2010) and reducing the potential carbon sink. Conflicts of interest emerge between agriculture, forestry and tourism when cooperation between actors and interest groups is limited (Johann, 2013).

The potential for carbon sequestration through rewetting of agricultural peat soils and uncertainties about the actual amount and use of agriculturally used peatland represent research gaps. The national greenhouse gas inventory reports 13,000 ha of organic soils in the category 'grassland remaining grassland' (Umweltbundesamt, 2024).

# Nature-based solutions to adapt to climate change and disaster risk reduction

Nature-based solutions are not only linked to climate change mitigation, but also entail multiple synergies and co-benefits for ecosystem services, contributing to multiple global agendas, including disaster risk reduction, climate change adaptation, and sustainable development (Martin et al., 2021). There is high confidence that taking advantage of nature-based solutions (NbS) helps to mitigate the negative impacts of climate change in urban settings and make cities more resilient to environmental changes (Faivre et al., 2017; Lafortezza and Sanesi, 2019). Urban forests, parks or simply tree-lined streets - urban greens in general - are known to facilitate adaptation to climate change in terms of temperature and water budget regulation (see urban heat islands in Section 3.2.3). Urban greening must consider climate-resilient tree species that can cope with future conditions to secure its benefits (Foldal et al., 2022). For example, non-native tree species already present in the urban Alpine area of Austria and neighboring countries were recorded and assessed for their drought tolerance (Herbsthofer et al., 2020). However, in terms of climate change mitigation, the contribution of urban green strategies remains small to moderate (Strohbach and Haase, 2012; Ariluoma et al., 2021) unless further indirect effects on avoided energy demand for cooling buildings are considered (Quaranta et al., 2021). See Section 3.2.2 for such indirect effects related to the built environment and urban planning.

Nature-based solutions entail strong health co-benefits (*high confidence*). In particular, increasing the number, distribution and density of green trees in urban areas improves urban public health and well-being of urban populations and contributes to sustainable living in terms of positive effects

on humans, microclimate, soil and biodiversity (Haluza, 2024; Spörl et al., 2024). Nature-based solutions also support several ecosystem services and their health co-benefits described in Section 2.2 (food security, air pollution, psychological health, etc.).

Beyond urban areas, the development of near-natural areas increases the possibility of migration of plants and animals and thus also of pollinators, counteracting the negative impacts on pollination. There is *limited evidence*, but *high agreement* that nature-based adaptation measures such as the restoration of wetlands, revitalization of rivers, no-tillage and cover-cropping in agriculture, and urban green gardens may restore natural flood control capacity – by providing retention space and improving soil water holding capacity (Seddon, 2022). Despite several decades of experience with river restoration, evidence on its effectiveness is limited, as evaluation is difficult due to the context specificity of projects and realizations at diverse scales with differing objectives (e.g., Kurth and Schirmer, 2014 for Switzerland; Nilsson et al., 2016 for other European instances). Moreover, implementation is still often obstructed by the path dependencies of gray measures, such as straightened riverbeds, which have long lifetimes and are costly to reverse (Hanger-Kopp et al., 2022; Seebauer et al., 2023).

#### Chapter Box 2.2. Carbon sequestration potentials of forests

Forests cover 48 % of the Austrian territory, of which 83.7 % are commercial forests. As forests are able to accumulate and store substantial amounts of carbon (Erb et al., 2018; Arneth et al., 2019), their potential to sequester atmospheric carbon dioxide and provide carbon sinks is relevant for the national GHG budget (Umweltbundesamt, 2023a; Jandl et al., 2024c). This box discusses the conditions for long-term carbon storage and the underlying assumptions needed for estimation. It (i) quantifies current stocks and their historical development, (ii) introduces management strategies to maintain and increase stocks, (iii) discusses theoretical and observed carbon densities, and (iv) examines the pathways to arrive at such quantities. For simplicity, land conversions (land use change to forests or from forests to other uses) are not considered here.

The C budget of forests is characterized by stocks and by stock changes. We only consider above- and below-ground woody biomass, including deadwood, since additional carbon pools such as soil organic carbon are still poorly understood (Le Noë et al., 2023b). Stock estimates (Table 2.1) are derived from the volumetric data of the Austrian Forest Inventory using an average conversion factor of 0.345 tC/m<sup>3</sup> (2.9 m<sup>3</sup>=1 tC), a factor taken from 1990 forest inventory data as converted to total carbon (including branches, leaves, roots) by Umweltbundesamt (for 1990, 973 million m<sup>3</sup> and 339 MtC) and which we maintain here for all comparisons. Stock changes reflect C uptake from the atmosphere (e.g., increment) and biomass C removal (harvest, decomposition and depletion). In Austrian commercial forests, the forest C stock increased from 80.7 tC/ha in the period 1961–70 to 121.0 tC/ha in 2016–18 (Gschwantner, 2019) (*high confidence*). This increase is the result of removals being lower than increment in most years (Figure 2.3).

Forest management options aimed at maintaining forest functions include selective cutting (both thinning and harvesting) as well as tree planting. Management may be directed to favor growth (to extract forest products, but also to remove atmospheric  $CO_2$  efficiently), which requires relatively young forests (60–140 years) and management interventions such as 2–3 thinnings before the final harvest. By contrast, management can also aim to maximize carbon stocks in forests, by selecting tree species that can accumulate high stocks and have low damage risks, by making thinnings to allow long rotation times, making the final harvest before reaching the degradation phase and keeping regeneration times short.

In order to estimate C sequestration potentials, it is important to understand (although challenging to assess) what carbon densities can eventually be achieved in forests. Ecological considerations originate from biophysical principles using parameters such as the net primary production (Saugier et al., 2001) and applying the 'potential vegetation concept', i.e., the vegetation that can be expected under current or pre-industrial climatic and soil conditions, as a benchmark (Erb et al., 2018; Bastos et al., 2021). Actual biomass stocks are understood to be depleted due to the removal of organic material, such as wood or deadwood, and this depletion establishes a stock gap. Global studies of existing forest biomass stocks at the landscape level report that their carbon storage is 24–39 % below the natural potential in the temperate zone (Erb et al., 2018; Walker et al., 2022; Mo et al., 2023).

Data from existing primeval forests in Europe help to quantify this natural potential. Such forests are sparse and small, so results are uncertain and only partially representative. In a meta-study, Keith et al. (2024) present data from 18 sites in Central Europe characterized as primary forests (none in Austria). Several of these sites had carbon stocks exceeding 200 tC/ha, and only 3 sites had less than 150 tC/ha. For the Bialowieza forest in Poland, potential vegetation has been estimated at 189 tC/ha, while current C stocks are considered to be 123 tC/ha, even though the forest has been left unmanaged for centuries (Matuszkiewicz et al., 2021). In Austria, the small area of Rothwald showed an average stock of 175 tC/ha (timber stock decreases of more than 44 tC/ha were observed during a wind throw in 1966) while surrounding managed stands had 266 tC/ha (Schrempf, 1986). Similarly, the Bavarian Forest National Park (Nationalpark Bayerischer Wald) exhibits lower timber stocks (without deadwood) of 96 tC/ha in the unmanaged core zone, compared to 138 tC/ha in the managed zone next to the core zone (Heurich and Sinner, 2005).

Optimum management options from a forestry perspective have been outlined by Zeide (2004), differentiating between optimizing for harvest or for carbon stocks. A steady-state of forest development needs to cover the carbon pool in all stages of the forest cycle at the landscape scale, as well as site- and tree-specific growth and mortality rates caused by disturbances or due to physical age limits of trees (Wirth et al., 2009; Oettel et al., 2020; Ehbrecht et al., 2021; Kobler et al., 2024). Management decisions allow a managed forest to exceed the C storage of potential vegetation. Specifically, the planting of selective tree species (Douglas-fir, *Pseudotsuga menziesii*) (Spannlang, 2022) or the possibility to minimize calamities (Vanomsen, 2006; Kautz and Delb, 2023; Hurteau et al., 2024) are essential factors to increase carbon stocks beyond primeval forests.

C sequestration potentials have been quantified either as a carbon gap to an ecological potential or as an impact of forestry management. There is *robust evidence*, but *low agreement* on which approach is appropriate. When quantifying a C sink, it is not important whether that sink is man-made or natural. The difference becomes relevant for specific management decisions, and for the longer-term robustness of such a sink. The temporal aspect is also decisive for sink dynamics. Luyssaert et al. (2008) show that also old-growth forests further accumulate carbon over centennial time spans (which could explain differences to the potential C stocks observed by Matuszkiewicz et al. (2021)), despite disturbances and calamities (wind, insects, fire). Thus, C sequestration will also occur at the same time scale of centuries.

Specific modeling of such developments over time for Austria in the CareforParis study (BFW, 2020; Ledermann et al., 2022) estimated a peak stock of 193 tC/ha by 2130 for an RCP8.5 scenario (GWL 4.34°C for 2081–2100) assuming average extreme events combined with a management target to increase stocks. In contrast, in a scenario based on current management but with massive calamities, stocks were estimated at 55 tC/ha in 2150 (data presented includes standing deadwood). Here, calamities provide a decisive management risk, with potential impacts also for strategies aimed at maximizing C stocks. The lack of understanding of this risk is a research gap, which will remain challenging to close given the time scales of forest growth.

The above modeling approaches consider calamities as well as C-stocks in the form of standing deadwood. To put into perspective, in 2023, 1.03 MtC forest were destroyed by bark beetles and 0.74 MtC by windthrow (BFW, 2023), together about 0.4 tC/ha if evenly distributed over Austria's forested area. That is 2 ‰ of the biomass C stock, but around 20 % of the annual increment (see Figure 2.3b). The Austrian Forest Inventory measures 7.1 tC/ha deadwood for Austrian forests (Oettel et al., 2020), with values in the small areas of unmanaged natural forest reserves reaching 7.9–37.6 tC/ha, at half lives of several decades (Hararuk et al., 2020; Seibold et al., 2021), with large observed variation (Harmon et al., 2020).

Further challenges in estimating the future development of Austrian forests are related to the legacy of multiple uses in the past (Gingrich et al., 2007, 2021, 2022), which led to a dominance of spruce as a single tree species. Adapting forests to climate change requires, i.e., altering the tree composition (Wessely et al., 2024), which means replacing part of the existing forest stock. Using forests as carbon sinks adds to the complexity, especially if biodiversity is also taken into account (for diverging views on management impacts, see Luick et al., 2021; Jandl et al., 2024a). Assumptions about the use of harvested wood products to substitute more carbon-intensive materials or practices (Weiss et al., 2020) are excluded here and may conflict with the forests' C sequestration potential, see Table 2.3. For a systematic comparison of two differing perspectives on net C effects of forest management and wood use, see Box 1.1 in Jandl et al. (2024c).

## 2.4. Conflicts, lock-ins and enabling conditions – focusing on the example of agricultural land use

The land use system includes a network of actors, activities and institutions along the value chain, from production to food waste, embedded in ecological, social, political, cultural and economic contexts (Penker et al., 2023). Climate change and sustainable development are challenges to society that require action at local, national, transboundary and global scales. Forestry and agricultural legislation, urban and spatial planning, and nature conservation provide the (legal) framework for action and guidance. Climate change and environmental crises, partly driven by dynamics outside the land use sectors, can exacerbate existing conflicts or create new ones (Formayer et al., 2024b). Ongoing soil sealing and urban sprawl has been exacerbating the competition for and conflicts around biologically productive land (see Section 3.2.2).

Focusing mainly on the example of agricultural land use, this section looks into barriers, conflicts, and lock-ins as well as enabling conditions for a transformation. For illustrative purposes, forest-related literature is added in some parts.

### 2.4.1. Barriers, conflicts, and lock-ins

Land is a limited and scarce resource (Haberl and Erb, 2017). The land system, globally as well as in Austria (Formayer et al., 2024b), is coined by strong (intrinsic) trade-offs, targetand interest conflicts (*high confidence*). Land use, e.g., in forests and agroecosystems, has different societal functions, such as production, recreation, or nature conservation, entailing substantial goal conflicts (Bontempi et al., 2023). Moreover, people project different values onto land, from direct or instrumental use values to indirect or intrinsic values (Meyfroidt et al., 2022) and relational values (Leopold, 1949; Himes and Muraca, 2018).

The current specific allocation of the scarce resource land to different uses (Gerber et al., 2018) results from the interaction of a wide variety of stakeholders (land owners, land users, consumers, processors, retailers, public sector), natural site conditions, historical development, market conditions, and technological possibilities. Changes in the allocation of land use and the associated rights and duties to land have environmental, economic and societal effects bearing high conflict potentials. For example, maximizing the carbon sink through large scale reforestation might result in a fundamentally changed landscape and might negatively affect other ecosystem services such as food production or water provision (Jackson et al., 2005; Ferreira et al., 2018), and impact current land users such as farmers by decreasing or completely suppressing their current livelihoods. Such trade-offs are frequent in the land system (Meyfroidt et al., 2022). Negotiating the values and interests among stakeholders (Ellis et al., 2019), and measures to reduce the overall demand for certain ecosystem services (Haberl and Erb, 2017) can help to mitigate trade-offs. In contrast to the strong influence of sector-specific goals and interests, spatial planning struggles to fully assert its organizing and coordinating functions, significantly limiting its capacity to mitigate trade-offs across sectors and ensure climate-friendly decision-making (Svanda and Zech, 2023). There is medium evidence and high agreement that such trade-offs mostly affect less powerful, often also less wealthy, stakeholders (Howe et al., 2014; Pichler et al., 2022).

The outcomes of such land use conflicts are often reinforced by path dependencies and 'environmental amnesia' making them difficult to be reverted (Meyfroidt et al., 2022; Aigner et al., 2023). As noted in the APCC Special Report on land use and climate change (Jandl et al., 2024c), there is high confidence that a fundamental conflict unfolds between economic growth and environmental conservation. Under the current conditions of production and consumption, growing economic performance in Austria and worldwide is closely linked to increasing resource and energy consumption (BMK, 2020) and is associated with negative environmental impacts such as pressure on water resources, rising greenhouse gas emissions (at least in a multiannual average), biodiversity loss (Otero et al., 2020; Moranta et al., 2022; Santika et al., 2024) and soil sealing. This leads to an implicit preference for substitution-pathways and efficiency-pathways over conservation (restoration) or sufficiency pathways (Birch et al., 2010; Allain et al., 2022; Boyer et al., 2023; Eversberg et al., 2023; Fleischmann et al., 2024). However, there is high confidence that the effects of efficiency gains, including, e.g., yield increases, are often partly compensated or even overcompensated due to rebound effects, i.e., increases in consumption (Paul et al., 2019; García et al., 2020; Erb et al., 2023b; Haberl and Erb, 2023; Preinfalk et al., 2024).

In Austria, both, economic performance and resource and energy consumption are growing, albeit resource and energy consumption increase at a lower rate than GDP (relative decoupling) (BMK, 2020). There is *limited evidence*, but *high agreement*, that a climate-friendly land system cannot be achieved without fundamental, comprehensive structural changes, which involve industry and trade, technological innovations and individual approaches (IPCC, 2019a; Penker et al., 2023). The fundamental problem with regard to environmental problems is that landowners (e.g., farmers, forest owners) are not paid for delivering environmental services (as they are mostly public goods) and focus their production activities on private goods such as food, timber products, and recreation services (Cooper et al., 2009; Winkel et al., 2022). Long-term investment decisions by private actors (e.g., livestock housing) or cultural values and norms (e.g., on 'good farming' standards) can reduce the acceptance for alternative production systems. Consequently, the acceptance for environmentally oriented production is comparatively low, if not made mandatory or subsidized by the government. Fostering payments for ecosystem services, better monitoring of ecosystem supply and demand, enhanced policy integration, and improved bottom-up participation and learning among ecosystem service innovators are therefore important strategies (Winkel et al., 2022). Non-acceptance is often reinforced through lobbying by interest groups or even protests (Nature Food, 2024) of the respective sectors, which render it difficult to integrate climate and environmental policies with sectoral policies (Plank et al., 2021).

Difficulties might also arise to change the diverse drivers of current land use. One example is the apparent difficulty in reducing the share of animal products in diets, in Austria and in the EU as a whole, despite the apparent benefits and synergies between climate and biodiversity protection and human health (see Section 2.3.2). To avoid facing the conflicts related to a transformation towards less animal protein production and consumption, dietary change is often hidden behind consumer-narratives (Vermeulen et al., 2020; Duluins and Baret, 2024), preventing structural discussions on solutions to incentivize and support producers in the protein transition. A media analysis of three Austrian newspapers published between 2000 and 2019 shows evolving informal institutions, such as behavioral norms embracing meat-free or reduced-meat diets and no debates about changes in formal institutions, such as regulations and other policy interventions (Hundscheid et al., 2022). Hundscheid et al. (2022) attribute this lack of debates about changes in formal institutions to the deeply entrenched cultural significance of meat, which is actively defended by actors safeguarding the status quo. Kortleve et al. (2024) conclude that in the EU agricultural subsidy system, animal husbandry receives far more support (>80 % of all subsidies) than plant production. On the other hand, animal production, which is to a large degree export oriented, is of key economic importance and a significant reduction in animal husbandry would impact companies as well as numerous family farms. Farms are mostly faced with a small number of comparatively large retail chains and processors and consequently have little bargaining power in the value chain (Jäger et al., 2024), and consumers are mostly not prepared to pay the corresponding higher prices. Thus, while farmers often have to implement environmental and climate mitigation measures, they cannot simply pass on the resulting costs to retail chains or consumers.

The mechanisms of international markets are also relevant in this context. National regulations need to respect the European frame and international trade agreements (e.g., WTO), and the development of international competitiveness of the strongly export-oriented Austrian agricultural, forestry and timber sectors (see Section 2.1.1) is a key factor for the development of (environmental) regulations. A sole focus on consumer responsibility and market-based mechanisms would affect the social justice dimensions, i.e., the distribution of costs and benefits of the transformation towards climate-friendly structures. From a consumer perspective, there is high confidence that unhealthy foods or 'empty calories' (Tilman and Clark, 2014) are cheaper, more readily available, and also constantly featured in advertising compared to healthier alternatives, which contradicts a socially just transformation towards climate-friendly living and is associated with higher C-emissions on-average (Rao et al., 2013; Ravensbergen et al., 2015; Penne and Goedemé, 2021). A 'climate-social policy' aimed at exploiting synergies between reduction of social inequalities and climate change mitigation has not yet been formulated for Austria (Armutskonferenz et al., 2021). Structures of social inequality are key, for example for dietary practices (Penker et al., 2023), and can result in food poverty or obesity due to the unaffordability of healthy food in terms of quantity and quality (Bonaccio et al., 2012).

A further key sustainability conflict in the current land production system relates to working conditions (*medium evidence, high agreement*). Many economic activities related to the land system (production, processing, retail) rely on low or comparatively low wages and/or profits, and precarious working conditions are widespread (Penker et al., 2023). Additionally, the number of serious accidents at work is still very high in the Austrian agriculture and forestry, despite improved technology and prevention measures (Kogler et al., 2015). Barriers to the implementation of resource-conserving or emission-reducing measures often lie in higher

AAR2

labor or capital inputs, which increase production costs, and in reduced land productivity (Jäger et al., 2024), although this is not always the case (Fontana et al., 2013). In this context, it should be noted that farms in Austria are comparatively small and mostly run as family farms, where labor is usually a scarce resource.

Further barriers and lock-ins can be found in existing and contradictory policy frameworks and funding systems, within the CAP direct payments and even within agri-environmental-climate schemes (Kletzan-Slamanig et al., 2022; Jäger et al., 2024).

#### 2.4.2. Enabling the transformation

A successful transformation strategy must consider all these barriers, conflicts and lock-ins, including the international nature of markets and policies. 'Climate-friendly' land use changes, e.g., in agriculture or forestry, affect not only land use and the land users themselves, but also society via several socially relevant public goods and bads (Sauvage, 2014). Many of these goods (e.g., agrobiodiversity, cultivated landscapes, recreation and experience in nature) and bads (e.g., nitrogen and phosphorus inputs resulting in greenhouse gas emissions or water pollution) have no market (as opposed to food, animal feed, timber, etc.) and are therefore not coordinated through market mechanisms (Cooper et al., 2009). (Agri-environment-climate) policy interventions represent a means to compensate for management and opportunity costs, as is already implemented in Austria, e.g., under the agri-environmental program ÖPUL. This program offers a broad range of measures aimed at mitigating greenhouse gas emissions and supporting agriculture in adapting to new conditions, such as greening, manure storage, or manure spreading. In the forestry sector, several CDR measures are also proposed to improve the overall mitigation, but there are no policy instruments in place to compensate forest landowners for potential financial losses and foster strategies to increase sufficiency, recycling and product-lifetime of timber products (Jandl et al., 2024b). Financial compensation for the provision of public goods or the avoidance of public bads provides income opportunities for land users, resulting in negative net taxes on farm production in Austria and other OECD countries - in contrast to positive net taxes in most countries of the Global South (Eastwood et al., 2010). For a successful transformation, social questions of poverty and inequality have to be effectively addressed at the national and international levels (Club of Rome, 2022; Hoffmann et al., 2024).

The Austrian Court of Auditors recommended the definition of quantifiable objectives, assessment indicators and the collection and use of data on the state and change of the environment for effective agri-environmental schemes - ÖPUL (Österreichischer Rechnungshof, 2013). According to the follow-up report, these recommendations were not or only partially implemented (Österreichischer Rechnungshof, 2016). In 2024, at the European level, the European Court of Auditors (ECA) identified a gap between farming incentives and the EU's green targets (European Court of Auditors, 2024). A key finding of the ECA is that, with the exception of organic farming, the European Commission will not be able to measure the contribution of the Common Agricultural Policy, which accounts for just under a third of the EU's total budget for 2021-2027, to its climate goals (European Court of Auditors, 2024). Therefore, a comprehensive provision of knowledge and information is a prerequisite for a successful transformation. Nevertheless, research efforts and innovations (such as agrivoltaics) are currently not well transferred to land use stakeholders. Education systems play a key role in ensuring that knowledge is passed on effectively, e.g., to farmers but also to consumers (Jäger et al., 2024). The development of suitable information and indicator systems is a key aspect and can be enhanced (Wolfslehner and Vacik, 2011; Grima et al., 2023). Austria-supported developments such as the further elaboration of the Agricultural Knowledge and Innovation Systems (AKIS) or the transformation of the European Farm Accountancy Data Network (FADN) into a Farm Sustainability Data Network (FSDN) are positive examples. However, providing information is hardly sufficient to steer transformations such as reducing the consumption and production of meat and avoiding food waste (Jäger et al., 2024). Further progress needs to identify and develop synergies between technological innovations (digitalization, precision farming or robotics), regional development and the mitigation of greenhouse gas emissions (Kraxner et al., 2024; Schipfer et al., 2024).

Austria lacks a systemic approach to sufficiency strategies, reducing consumption and production levels through avoid-shift-improve approaches (Plank et al., 2021)(see Sections 6.5.1, 6.5.4 and Table 6.2). Such an approach would take as its starting point the highly integrated, complex-dynamic nature of socio-ecological dynamics, acknowledging that land systems are cross-cutting and involve various policy areas (including environment and climate, health, agriculture, consumer protection). There is a need for a climate protection law that contains strategic targets in line with the Paris climate goals as well as effective sanction mechanisms and requirements for improvement in order to protect Austria's climate goals against regression (Hollaus et al., 2023) (*medium evidence, high agreement*) (see Section 6.2.3).

It becomes clear that only an integrative policy development allows for an orchestrated embedment of climate protection into various policy areas. International assessment reports conclude that food, energy and water security as well as biodiversity conservation rank high on the Agenda 2030 for Sustainable Development, rendering the promotion of synergies between and across sectoral policies a key strategy (IPBES, 2018). However, the current policy design and formulation in Austria is often sectoral. Moreover, the long-standing concentration of environmental and agricultural agendas in one ministry (2000-2020) has contributed to the concealment of potential conflicts of interest (Penker et al., 2023). Promising developments include a new organization on sustainable food (AGES, 2024) that bridges the three Ministries of Health, Agriculture and Climate, as well as the Sustainable Food Initiative by the Austrian Promotional Bank (AWS, 2024).

Abandoning the current attribution of responsibility for climate protection, which is largely placed on consumers by policymakers and businesses, has the potential to create enabling conditions for a sustainability transformation (Penker et al., 2023). Public food procurement (meals in schools, hospitals, canteens) or reducing taxes on plantbased foods are examples of transformative policies that can support the co-benefits for climate friendly and healthy food supply (see Table 2.A.4 for further measures along the entire food system). Ensuring the resilience and resistance of forests through diverse forest management practices, implementing integrative measures to maintain biodiversity and promoting payments for non-marketable goods and services are key elements for a sustainable transformation. The measures described in the report of the UniNetz initiative provide a wide array of measures in this direction (Vacik et al., 2022). Identifying and engaging with potential repercussions of climate policies on land users can enhance the legitimacy of such effective regime destabilizing measures and can thus contribute to resolving lock-ins (Hundscheid et al., 2024). An integrative food policy relies on combining different policy areas and has been shown to foster decentralized self-organization, entrepreneurship and social learning (Penker et al., 2023), which could open windows for transformative innovations.

Polycentric governance is proposed as an appropriate way to deal with resource management problems, in which decision-making centers take each other into account in competitive and cooperative relationships and have recourse to conflict resolution mechanisms (Arneth et al., 2019). Polycentric governance aims to open the scope of action for context-specific solutions and coordinated interaction between actors at different levels (local, regional, national, and global) and to foster decentralized self-organization, entrepreneurship and social learning. Such approaches have been shown to allow for flexibility and context-specificity in the light of the uncertainties inherent in, for example, the food system and for successful adaptation (medium evidence, high agreement) and require horizontal as well as vertical integration of policy areas or levels (Plank et al., 2021). The need for a common nutrition policy is discussed at both European and national level (IPBES, 2019; SAPEA, 2020; Penker et al., 2023), as is the need to use all governance instruments in the field of nutrition (SAPEA, 2020; WBAE, 2020). See Section 6.4 for more details on governance mechanisms.

A climate-friendly coordination between the production of food, feed, fiber and fuel, and between production, conservation and restoration can be supported by scaling up local land use planning to the regional or even to the national level (Svanda and Zech, 2023). Further suggestions from the spatial planning literature for achieving climate goals involve legal requirements for sectoral plans to contribute to climate-friendly structures, avoiding financial incentives for climate-harmful land use structures (e.g., commuter bonus 'Pendlerpauschale'), or new incentives for unsealing former industrial areas or parking lots (Svanda and Zech, 2023) (*medium evidence, high agreement*) (see Sections 3.2, 3.4, 6.5.1 and Table 6.2).

Environmental programs can reduce conflicts by financially supporting the provision of public goods such as climate-resilient soils, mitigation, biodiversity, and other land-based ecosystem services. However, utilitarian and transactional approaches to trade-offs are likely to depreciate the conservation of biodiversity and ecosystems, for example when forests are mostly used for timber production or carbon sequestration, neglecting a wide array of humannature relations and associated values. New approaches are emerging to address trade-offs considering both social and ecological aspects and the values of different people, while counteracting uneven power relations among actors (Ellis et al., 2019). Other recent examples involve nature-positive scenarios based on the Future Nature Framework (Pereira et al., 2020).

Here, a research gap emerges as well: Chapter 6 of the APCC Special Report on land use and climate change (Jäger et al., 2024) concludes that in the field of emission develop-

ments of the LULUCF sector, there are hardly any cross-sectoral scenario analyses that quantify and map the different land use types and their feedbacks on other sectors for Austria (Jäger et al., 2024). This gap has not yet been filled, but such scenario analyses would be of great importance for the development of a climate protection strategy in the area of land use in Austria and for the analysis of possible land use conflicts between sectors.

Transforming or building new structures is seen as essential to achieving climate goals (Aigner et al., 2023). A holistic view of the agri-food sector, for example one that looks at farmers' land use decisions, the structure of the food industry, the retail sector and consumer demand in an integrated way (Meynard et al., 2017), proves to be beneficial (*high confidence*), as noticed in Chapter 6 of the APCC Special Report on land use and climate change (Jäger et al., 2024) and the APCC Special Report on climate-friendly living (Penker et al., 2023). Adoption also depends on whether the envisaged measures are in line with the values and goals of the farmers, foresters and other entrepreneurs (de Sainte Marie, 2014; Schermer et al., 2018; Walder and Kantelhardt, 2018) and if schemes exist to compensate land users for their additional expenses, since market compensation is often not given due to the public-good-character of ecosystem services.

Literature shows that substantial progress can be made when transformation pathways are developed together with the land users (Kraxner et al., 2017; Berthet et al., 2019; Meyfroidt et al., 2022). There is high confidence that stakeholder involvement in policy development is a prerequisite for effective implementation of climate policies (e.g., Renn, 2006; Newig et al., 2008; O'Faircheallaigh, 2010; Prutsch et al., 2018). For government policies to be effective in practice, their implementation by firms is a key precondition and thus needs a systemic design of government measures (requirements, processing, information, advice, documentation effort, conditions attached to subsidies) (Van Herzele et al., 2013; Darnhofer et al., 2017). Furthermore, it is important to note that many regulations that farmers or forest owners must adhere to are formulated not only by the government but also by private companies. Therefore, the establishment of a governance system that includes all participants in the value chain is essential, warranting a fair exchange of information and a balanced distribution of power within the chain.

#### REFERENCES

- Abecia, L., Martínez-Fernandez, G., Waddams, K., Martín-García,
   A. I., Pinloche, E., Creevey, C. J., et al. (2018). Analysis of the
   Rumen Microbiome and Metabolome to Study the Effect of
   an Antimethanogenic Treatment Applied in Early Life of Kid
   Goats. *Frontiers in Microbiology* 9. Available at: <a href="https://www.frontiersin.org/articles/10.3389/fmicb.2018.02227">https://www.frontiersin.org/articles/10.3389/fmicb.2018.02227</a> (Accessed
   September 20, 2023).
- Aberle, S. W., Kolodziejek, J., Jungbauer, C., Stiasny, K., Aberle, J. H., Zoufaly, A., et al. (2018). Increase in human West Nile and Usutu virus infections, Austria, 2018. *Eurosurveillance* 23. <u>https://doi.org/10.2807/1560-7917.ES.2018.23.43.1800545</u>
- Abou-Shaara, H. F., Owayss, A. A., Ibrahim, Y. Y., and Basuny, N. K. (2017). A review of impacts of temperature and relative humidity on various activities of honey bees. *Insect. Soc.* 64, 455–463. <u>https://doi.org/10.1007/s00040-017-0573-8</u>
- Acuff, H. L., Dainton, A. N., Dhakal, J., Kiprotich, S., and Aldrich, G. (2021). Sustainability and Pet Food: Is There a Role for Veterinarians? *Vet Clin North Am Small Anim Pract* 51, 563–581. https://doi.org/10.1016/j.cvsm.2021.01.010
- AGES (2023a). Ambrosia Infos der AGES über Schaderreger. AGES. Available at: <u>https://www.ages.at/pflanze/pflanzengesundheit/schaderreger-von-a-bis-z/ambrosia</u> (Accessed September 19, 2023).
- AGES (2023b). Gemeiner Stechapfel Infos der AGES über Schaderreger. AGES. Available at: <u>https://www.ages.at/pflanze/pflanzengesundheit/schaderreger-von-a-bis-z/gemeiner-stechapfel</u> (Accessed September 19, 2023).
- AGES (2024). AGES Servicestelle für nachhaltige Lebensmittelund Ernährungssysteme. Available at: <u>https://akademie.ages.at/</u> <u>auftaktveranstaltung\_sustainable\_food\_systems</u>
- Aigner, E., Görg, C., Madner, V., Muhar, A., Novy, A., Posch, A., et al. (2022). "Kapitel II: Summary for Policymakers", in APCC Special Report: Strukturen für ein klimafreundliches Leben (APCC SR Klima-freundliches Leben), eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum). <u>https://doi.org/10.2139/ssrn.4277678</u>
- Aigner, E., Görg, C., Madner, V., Novy, A., Steininger, K. W., Nabernegg, S., et al. (2023). "Kapitel 1. Einleitung: Strukturen für ein klimafreundliches Leben", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 173–194. <u>https://doi.org/10.1007/978-3-662-66497-1\_5</u>
- Ainsworth, E., and Long, S. (2020). 30 years of free-air carbon dioxide enrichment (FACE): What have we learned about future crop productivity and its potential for adaptation? *Glob Change Biol* 27, 27–49. <u>https://doi.org/10.1111/gcb.15375</u>
- Albrich, K., Rammer, W., and Seidl, R. (2020). Climate change causes critical transitions and irreversible alterations of mountain forests. *Global Change Biology* 26, 4013–4027. <u>https://doi. org/10.1111/gcb.15118</u>
- Alexander, P., Berri, A., Moran, D., Reay, D., and Rounsevell, M. D. A. (2020). The global environmental paw print of pet food. *Glob*al Environmental Change 65, 102153. <u>https://doi.org/10.1016/j.</u> <u>gloenvcha.2020.102153</u>
- Alexander, P., Brown, C., Arneth, A., Finnigan, J., Moran, D., and Rounsevell, M. D. A. (2017). Losses, inefficiencies and waste in the global food system. *Agricultural Systems* 153, 190–200. <u>https://doi.org/10.1016/j.agsy.2017.01.014</u>

- Alikadic, A., Pertot, I., Eccel, E., Dolci, C., Zarbo, C., Caffarra, A., et al. (2019). The impact of climate change on grapevine phenology and the influence of altitude: A regional study. *Agricultural* and Forest Meteorology 271, 73–82. <u>https://doi.org/10.1016/j.</u> agrformet.2019.02.030
- Allain, S., Ruault, J.-F., Moraine, M., and Madelrieux, S. (2022). The 'bioeconomics vs bioeconomy' debate: Beyond criticism, advancing research fronts. *Environmental Innovation and Societal Transitions* 42, 58–73. https://doi.org/10.1016/j.eist.2021.11.004
- Allianz Nachhaltige Universitäten in Österreich ed. (2021). Uni-NEtZ-Optionenbericht: Österreichs Handlungsoptionen für die Umsetzung der UN-Agenda 2030 für eine lebenswerte Zukunft. UniNEtZ – Universitäten und Nachhaltige Entwicklungsziele. Vienna. Available at: https://www.uninetz.at/optionenbericht
- AMA (2022). Daten & Fakten der AgrarMarkt Austria für den Bereich Getreide und Ölsaaten. Stand Oktober 2022. Available at: <u>https://www.ama.at/marktinformationen/getreide-und-ol-</u> <u>saaten/marktbericht#17212</u>
- Anca-Couce, A., Hochenauer, C., and Scharler, R. (2021). Bioenergy technologies, uses, market and future trends with Austria as a case study. *Renewable and Sustainable Energy Reviews* 135, 110237. https://doi.org/10.1016/j.rser.2020.110237
- Anderl, M., Haider, S., Zethner, G., Kropsch, M., Pöllinger, A., and Zentner, E. (2016). Maßnahmen zu sekundären Partikeln aus der Landwirtschaft: Ammoniakemissionen. Wien: Umweltbundesamt Wien.
- Anderl, M., Simone Haider, Elisabeth Kampel, Traute Köther, Christoph Lampert, Daniela Perl, et al. (2019). Austria's Informative Inventory Report (IIR) 2019. Submission under the UNECE Convention on Long-range Transboundary Air Pollution and Directive (EU) 2016/2284 on the reduction of national emissions of certain atmospheric pollutants. Vienna: Umweltbundesamt.
- Anderle, M., Paniccia, C., Brambilla, M., Hilpold, A., Volani, S., Tasser, E., et al. (2022). The contribution of landscape features, climate and topography in shaping taxonomical and functional diversity of avian communities in a heterogeneous Alpine region. *Oecologia* 199, 499–512. <u>https://doi.org/10.1007/s00442-022-05134-7</u>
- APCC (2014). Österreichischer Sachstandsbericht Klimawandel 2014 (AAR14)., ed. Austrian Panel on Climate Change (APCC).
   Wien, Österreich: Verlag der Österreichischen Akademie der Wissenschaften. Available at: <u>https://doi.org/10.1553/aar14</u>
- APCC (2018). Österreichischer Special Report Gesundheit, Demographie und Klimawandel (ASR18). Austrian Panel on Climate Change (APCC). Wien, Österreich: Verlag der ÖAW. Available at: <u>https://austriaca.at/8427-0</u> (Accessed September 20, 2022).
- Arbuthnott, K., Hajat, S., Heaviside, C., and Vardoulakis, S. (2018). What is cold-related mortality? A multi-disciplinary perspective to inform climate change impact assessments. *Environment International* 121, 119–129. <u>https://doi.org/10.1016/j.envint.2018.08.053</u>
- Ariluoma, M., Ottelin, J., Hautamäki, R., Tuhkanen, E.-M., and Mänttäri, M. (2021). Carbon sequestration and storage potential of urban green in residential yards: A case study from Helsinki. Urban Forestry & Urban Greening 57, 126939. <u>https://doi. org/10.1016/j.ufug.2020.126939</u>
- Armutskonferenz, ATTAC, and BEIGEWUM eds. (2021). Klimasoziale Politik: Eine gerechte und emissionsfreie Gesellschaft gestalten., 2. Auflage. Wien: bahoe books. Available at: <u>https://</u>

klimasozial.at/wp-content/uploads/2024/06/Klimabuch\_FinaleFahnen\_2Auflage.pdf

- Arneth, A., Denton, F., Agus, F., Elbehri, A., Erb, K.-H., Elasha, B. O., et al. (2019). "Framing and Context", in *Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems*, eds. P. R. Shukla, J. Skea, E. Calvo Buendia, V. Masson-Delmotte, H.-O. Pörtner, D. C. Roberts, et al. (Geneva: Intergovernmental Panel on Climate Change (IPCC)), 77–129. Available at: <u>https://doi. org/10.1017/9781009157988.003</u> (Accessed January 31, 2020).
- Ascoli, D., Plana, E., Oggioni, S. D., Tomao, A., Colonico, M., Corona, P., et al. (2023). Fire-smart solutions for sustainable wildfire risk prevention: Bottom-up initiatives meet top-down policies under EU green deal. *International Journal of Disaster Risk Reduction* 92, 103715. <u>https://doi.org/10.1016/j.ijdrr.2023.103715</u>
- Austrian National Forest Inventory (2022). Österreichische Waldinventur, Bundes Forschungszentrum für Wald. Available at: <u>https://www.waldinventur.at/#/</u> (Accessed October 2, 2023).
- AWS (2024). Sustainable Food Systems. Austria Wirtschaftsservice. Available at: <u>https://www.aws.at/fileadmin/user\_upload/Down</u> <u>loads/ergaenzende\_Information/aws\_Sustainable\_Foodsystems</u> <u>\_Initiative\_Blitzlicht.pdf</u>
- Backes, A., Aulinger, A., Bieser, J., Matthias, V., and Quante, M. (2016a). Ammonia emissions in Europe, part I: Development of a dynamical ammonia emission inventory. *Atmos. Environ.* 131, 55–66. <u>https://doi.org/10.1016/j.atmosenv.2016.01.041</u>
- Backes, A. M., Aulinger, A., Bieser, J., Matthias, V., and Quante, M. (2016b). Ammonia emissions in Europe, part II: How ammonia emission abatement strategies affect secondary aerosols. Atmos. Environ. 126, 153–161. <u>https://doi.org/10.1016/j.</u> <u>atmosenv.2015.11.039</u>
- BAFU (2014). Grundlagen für die Wasserversorgung 2025. Risiken, Herausforderungen und Empfehlungen. Bern: Bundesamt für Umwelt.
- Bahn, M., Schaumberger, A., Pötsch, E. M., Benar-Friedl, B., Birk, S., Herndl, M., et al. (2023). Auswirkungen von Dürre auf Grünland unter aktuellen und künftigen Klimabedingungen. CCCA Fact Sheet #44. Available at: <u>https://ccca.ac.at/ fileadmin/00\_DokumenteHauptmenue/02\_Klimawissen/Fact-Sheets/44\_climgrasshydro\_202306.pdf</u>
- Bakran-Lebl, K., Pree, S., Brenner, T., Daroglou, E., Eigner, B., Griesbacher, A., et al. (2022). First Nationwide Monitoring Program for the Detection of Potentially Invasive Mosquito Species in Austria. *Insects* 13, 276. <u>https://doi.org/10.3390/insects13030276</u>
- Ballester, J., Quijal-Zamorano, M., Méndez Turrubiates, R. F., Pegenaute, F., Herrmann, F. R., Robine, J. M., et al. (2023). Heat-related mortality in Europe during the summer of 2022. *Nat Med* 29, 1857–1866. <u>https://doi.org/10.1038/s41591-023-02419-z</u>
- Bastos, A., Hartung, K., Nützel, T. B., Nabel, J. E. M. S., Houghton, R. A., and Pongratz, J. (2021). Comparison of uncertainties in land-use change fluxes from bookkeeping model parameterisation. *Earth Syst. Dynam.* 12, 745–762. <u>https://doi.org/10.5194/ esd-12-745-2021</u>
- Bauer, H., Kasper-Giebl, A., and Limbeck, A. (2009). "AQUELLA" Graz Süd PM2.5 Quellenanalyse von PM10- und PM2,5 Belastungen in Graz. Available at: <u>https://publik.tuwien.ac.at/files/</u> <u>PubDat\_183867.pdf</u>

- Bauer, H., Marr, I., Kasper-Giebl, A., Limbeck, A., Handler, M., Jankowski, N., et al. (2007). Endbericht für das Projekt "AQUELLA"
  Wien. Bestimmung von Immissionsbeiträgen in Feinstaubproben. MA 22 3869/03. Available at: <u>https://publik.tuwien.ac.at/files/PubDat\_173989.pdf</u>
- Baumgarten, A., Haslmayr, H.-P., Schwarz, M., Huber, S., Weiss, P., Obersteiner, E., et al. (2021). Organic soil carbon in Austria Status quo and foreseeable trends. *Geoderma* 402, 115214. https://doi.org/10.1016/j.geoderma.2021.115214
- Baumgarten, A., Lapin, K., Schüler, S., Freudenschuss, A., Grüneis, H., Konrad, H., et al. (2024). "Kapitel 4. Anpassungsoptionen in der Landnutzung an den Klimawandel", in APCC Special Report: Landnutzung und Klimawandel in Österreich, eds. R. Jandl, U. Tappeiner, C. B. Foldal, and K. Erb (Berlin, Heidelberg: Springer Spektrum), 217–274. <u>https://doi.org/10.1007/978-3-662-67864-0\_6</u>
- Baylis, M. (2013). Research gaps in understanding how climate change will affect arboviral diseases. *Animal Health Research Re*views 14, 143–146. <u>https://doi.org/10.1017/S1466252313000145</u>
- Beck-O'Brien, M., Egenolf, V., Winter, S., Zahnen, J., and Griesshammer, N. (2022). Alles aus Holz – Rohstoff der Zukunft oder kommende Krise. WWF Deutschland. Available at: <u>https://www.wwf.de/fileadmin/fm-wwf/Publikationen-PDF/</u> <u>Wald/WWF-Studie-Alles-aus-Holz.pdf</u>
- Beelen, R., Hoek, G., Raaschou, -Nielsen Ole, Stafoggia, M., Andersen, Z. J., Weinmayr, G., et al. (2015). Natural-Cause Mortality and Long-Term Exposure to Particle Components: An Analysis of 19 European Cohorts within the Multi-Center ESCAPE Project. *Environmental Health Perspectives* 123, 525–533. <u>https://</u> doi.org/10.1289/ehp.1408095
- Bellassen, V., and Luyssaert, S. (2014). Carbon sequestration: Managing forests in uncertain times. *Nature* 506, 153–155. <u>https:// doi.org/10.1038/506153a</u>
- Berčák, R., Holuša, J., Kaczmarowski, J., Tyburski, Ł., Szczygieł, R., Held, A., et al. (2023). Fire Protection Principles and Recommendations in Disturbed Forest Areas in Central Europe: A Review. *Fire* 6, 310. <u>https://doi.org/10.3390/fire6080310</u>
- Berger, M., Bastl, K., Bastl, M., Dirr, L., Hutter, H.-P., Moshammer, H., et al. (2020). Impact of air pollution on symptom severity during the birch, grass and ragweed pollen period in Vienna, Austria: Importance of O3 in 2010–2018. *Environmental Pollution* 263, 114526. <u>https://doi.org/10.1016/j.envpol.2020.114526</u>
- Bergmann, K.-C. (2016). Weather conditions and climate change have an effect on allergies. *Allergo Journal International* 25, 131– 137. <u>https://doi.org/10.1007/s40629-016-0119-z</u>
- Berthet, E. T., Bretagnolle, V., Lavorel, S., Sabatier, R., Tichit, M., and Segrestin, B. (2019). Applying ecological knowledge to the innovative design of sustainable agroecosystems. *Journal of Applied Ecology* 56, 44–51. <u>https://doi.org/10.1111/1365-2664.</u> <u>13173</u>
- Bertsch-Hoermann, B., Egger, C., Gaube, V., and Gingrich, S. (2021). Agroforestry trade-offs between biomass provision and aboveground carbon sequestration in the alpine Eisenwurzen region, Austria. *Reg Environ Change* 21, 77. <u>https://doi.org/10. 1007/s10113-021-01794-y</u>
- BFW (2020). BFW-Praxisinformation 51: Klimakrise managen.
   Vienna: Austrian Research Centre for Forests (BFW). Available
   at: <u>https://shop.bfw.ac.at/bfw-praxisinformation/bfw-praxisinfo-51-2020.html</u> (Accessed November 14, 2024).

- BFW (2023). Dokumentation der Waldschädigungsfaktoren 2023. Available at: <u>https://www.bfw.gv.at/wp-content/uploads/DWF\_</u> <u>SF\_2023\_TAB\_WEB.pdf</u>
- Bhan, M., Gingrich, S., Roux, N., Le Noë, J., Kastner, T., Matej, S., et al. (2021). Quantifying and attributing land use-induced carbon emissions to biomass consumption: A critical assessment of existing approaches. *Journal of Environmental Management* 286, 112228. https://doi.org/10.1016/j.jenvman.2021.112228
- Biermayr, P. (2022). Erneuerbare Energien 2020. Entwicklung in Österreich. Vienna: BMK. Available at: <u>https://www.bmk.gv.at/</u> <u>themen/energie/publikationen/erneuerbare\_energie20.html</u>
- Birch, K., Levidow, L., and Papaioannou, T. (2010). Sustainable Capital? The Neoliberalization of Nature and Knowledge in the European "Knowledge-based Bio-economy." *Sustainability* 2, 2898–2918. <u>https://doi.org/10.3390/su2092898</u>
- BMK (2020). Ressourcennutzung in Österreich 2020. Band 3. Wien: Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie (BMK). Available at: <u>https://www.bmk.gv.at/themen/klima\_umwelt/nachhaltigkeit/</u><u>ressourceneffizienz/publikationen/bericht2020.html</u> (Accessed January 17, 2025).
- BMK (2022). Österreich auf dem Weg zu einer nachhaltigen und zirkulären Gesellschaft. Die österreichische Kreislaufstrategie. Wien: Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie. Available at: <u>https:// www.bmk.gv.at/themen/klima\_umwelt/abfall/Kreislaufwirtschaft/strategie.html</u>
- BMK (2023). Energie in Österreich. Zahlen, Daten, Fakten. Vienna: Ministry of Climate Action, Environment, Energy, Mobility, Innovation and Technology (BMK). Available at: <u>https://www. bmk.gv.at/dam/jcr:da4e9dfd-f51c-44b8-894c-9b049a8336cb/ BMK\_Energie\_in\_OE2023\_barrierefrei.pdf</u> (Accessed November 5, 2024).
- BMLF (2023a). Grüner Bericht 2023. Die Situation der österreichischen Land- und Forstwirtschaft. Vienna: Bundesministerium für Land- und Forstwirtschaft, Regionen und Wasserwirtschaft. Available at: <u>https://gruenerbericht.at/cm4/jdownload/send/2-gr-bericht-terreich/2586-gb2023</u>
- BMLF (2023b). Österreichischer Waldbericht 2023. Wien: Bundesministerium für Land- und Forstwirtschaft, Regionen und Wasserwirtschaft. Available at: <u>https://info.bml.gv.at/themen/wald/ wald-in-oesterreich/oesterreichischer-waldbericht-2023.html</u> (Accessed September 21, 2023).
- BMLF (2024a). Grüner Bericht 2024. Die Situation der österreichischen Land- und Forstwirtschaft. Wien: Bundesministerium für Land- und Forstwirtschaft, Regionen und Wasserwirtschaft. Available at: <u>https://gruenerbericht.at/cm4/jdownload/download/2-gr-bericht-terreich/2591-gb2024</u>
- BMLF (2024b). Holzeinschlag in Österreich vergangene Jahre. Available at: <u>https://info.bml.gv.at/themen/wald/wald-in-oes-terreich/wald-und-zahlen/Holzeinschlag.html</u> (Accessed November 8, 2024).
- BMLRT (2021a). Nationaler Gewässerbewirtschaftungsplan 2021. Vienna: Bundesministerium für Landwirtschaft, Regionen und Tourismus. Available at: https://info.bml.gv.at/themen/wasser/ wasser-oesterreich/wasserrecht\_national/wasserrechtliche\_ kundmachungen/ngp-2021.html
- BMLRT (2021b). Wasserschatz Österreichs. Grundlagen für nachhaltige Nutzungen des Grundwassers. Wien, Österreich: Bundesministerium für Landwirtschaft, Regionen und Tour-

ismus. Available at: <u>https://info.bml.gv.at/dam/jcr:75a703dd-</u> 9c25-452a-ac06-5240abbd118a/Bericht\_Wasserschatz.pdf

- BMLRT (2022). Moorstrategie Österreich 2030+. Wien: Bundesministerium für Landwirtschaft, Regionen und Tourismus. Available at: <u>https://info.bml.gv.at/service/publikationen/wasser/moorstrategie-oesterreich-2030.html</u> (Accessed September 21, 2023).
- Bollinger, D., Bonnard, C., and Keusen, H. (2008). Beurteilung der Wirkung von Schutzmassnahmen gegen Naturgefahren als Grundlage für ihre Berücksichtigung in der Raumplanung, Teil D: Rutschungen. Available at: <u>https://www.gra-nat.ch/system/ media/1802/original/PLANAT\_2008\_-\_Teil\_D\_\_Rutschungen.pdf?1421155295</u>
- Bolte, A., Ammer, C., Löf, M., Madsen, P., Nabuurs, G.-J., Schall, P., et al. (2009). Adaptive forest management in central Europe: Climate change impacts, strategies and integrative concept. *Scandinavian Journal of Forest Research* 24, 473–482. <u>https:// doi.org/10.1080/02827580903418224</u>
- Bonaccio, M., Bonanni, A. E., Castelnuovo, A. D., Lucia, F. D., Donati, M. B., Gaetano, de G., et al. (2012). Low income is associated with poor adherence to a Mediterranean diet and a higher prevalence of obesity: cross-sectional results from the Moli-sani study. *BMJ Open 2*, e001685. <u>https://doi.org/10.1136/bmjopen-2012-001685</u>
- Bönecke, E., Breitsameter, L., and Brüggemann, N. (2020). Decoupling of impact factors reveals the response of German winter wheat yields to climatic changes. *Glob Change Biol* 26, 3601– 3626. <u>https://doi.org/10.1111/gcb.15073</u>
- Bontempi, A., Venturi, P., Del Bene, D., Scheidel, A., Zaldo-Aubanell, Q., and Zaragoza, R. M. (2023). Conflict and conservation: On the role of protected areas for environmental justice. *Global Environmental Change* 82, 102740. <u>https://doi. org/10.1016/j.gloenvcha.2023.102740</u>
- Borchard, N., Schirrmann, M., Cayuela, M. L., Kammann, C., Wrage-Mönnig, N., Estavillo, J. M., et al. (2019). Biochar, soil and land-use interactions that reduce nitrate leaching and N2O emissions: A meta-analysis. *Science of The Total Environment* 651, 2354–2364. <u>https://doi.org/10.1016/j.scitotenv.2018.10.060</u>
- Borghi, M., Perez de Souza, L., Yoshida, T., and Fernie, A. R. (2019). Flowers and climate change: a metabolic perspective. *New Phytologist* 224, 1425–1441. <u>https://doi.org/10.1111/nph.16031</u>
- Bossio, D. A., Cook-Patton, S. C., Ellis, P. W., Fargione, J., Sanderman, J., Smith, P., et al. (2020). The role of soil carbon in natural climate solutions. *Nat Sustain* 3, 391–398. <u>https://doi. org/10.1038/s41893-020-0491-z</u>
- Böttcher, H., Verkerk, H., Gusti, M., Havlik, P., and Schneider, U. (2011). Analysis of potential and costs of LULUCF use by EU Member States. Final Report submitted to the European Commission. Laxenburg, Austria: DG Climate Action, IIASA. Available at: <u>https://pure.iiasa.ac.at/9754</u>
- Bourke, R., Yousefpour, R., and Hanewinkel, M. (2023). Under storm risk, economic productivity of Norway spruce (Picea abies (L.) H. Karst) in monoculture shows sharper decline than in mixture with European beech (Fagus sylvatica L.). Annals of Forest Science 80, 19. <u>https://doi.org/10.1186/s13595-023-01182-y</u>
- Bouvard, V., Loomis, D., Guyton, K. Z., Grosse, Y., Ghissassi, F. E., Benbrahim-Tallaa, L., et al. (2015). Carcinogenicity of consumption of red and processed meat. *The Lancet Oncology* 16, 1599–1600. <u>https://doi.org/10.1016/S1470-2045(15)00444-1</u>

- Boyer, M., Kusche, F., Hackfort, S., Prause, L., and Engelbrecht-Bock, F. (2023). The making of sustainability: ideological strategies, the materiality of nature, and biomass use in the bioeconomy. *Sustain Sci* 18, 675–688. <u>https://doi.org/10.1007/ s11625-022-01254-4</u>
- Boysen, L. R., Lucht, W., Gerten, D., Heck, V., Lenton, T. M., and Schellnhuber, H. J. (2017). The limits to global-warming mitigation by terrestrial carbon removal. *Earth's Future* 5, 463–474. <u>https://doi.org/10.1002/2016EF000469</u>
- Brang, P., Spathelf, P., Larsen, J. B., Bauhus, J., Boncčina, A., Chauvin, C., et al. (2014). Suitability of close-to-nature silviculture for adapting temperate European forests to climate change. *Forestry: An International Journal of Forest Research* 87, 492–503. <u>https://doi.org/10.1093/forestry/cpu018</u>
- Braun, M., Fritz, D., Weiss, P., Braschel, N., Büchsenmeister, R., Freudenschuß, A., et al. (2016). A holistic assessment of greenhouse gas dynamics from forests to the effects of wood products use in Austria. *Carbon Management* 7, 271–283. <u>https://doi.org /10.1080/17583004.2016.1230990</u>
- Brielmann, H., Döberl, G., Weiß, S., and Grath, J. (2023). PFAS in Österreichs Grundwasser: Verbreitung, Bewertung und Rolle von Altstandorten als potentielle Quellen. Österreichische Wasser- und Abfallwirtschaft 75, 491–502. <u>https://doi.org/10.1007/ s00506-023-00976-8</u>
- Brugger, K., Köfer, J., and Rubel, F. (2016). Outdoor and indoor monitoring of livestock-associated Culicoides spp. to assess vector-free periods and disease risks. *BMC Vet Res* 12, 88. <u>https:// doi.org/10.1186/s12917-016-0710-z</u>
- Brugger, K., and Rubel, F. (2009). Simulation of climate-change scenarios to explain Usutu-virus dynamics in Austria. *Preventive Veterinary Medicine* 88, 24–31. <u>https://doi.org/10.1016/j.</u> prevetmed.2008.06.023
- Brunekreef, B., Strak, M., Chen, J., Andersen, Z. J., Atkinson, R., Bauwelinck, M., et al. (2021). Mortality and Morbidity Effects of Long-Term Exposure to Low-Level PM2.5, BC, NO2, and O3: An Analysis of European Cohorts in the ELAPSE Project. *Res Rep Health Eff Inst* 2021, 208. <u>https://pmc.ncbi.nlm.nih.gov/</u> articles/PMC9476567/pdf/hei-2021-208.pdf
- Buschbeck, C., and Pauliuk, S. (2022). Required displacement factors for evaluating and comparing climate impacts of intensive and extensive forestry in Germany. *Carbon Balance Manage* 17, 14. <u>https://doi.org/10.1186/s13021-022-00216-8</u>
- Caffarra, A., Rinaldi, M., Eccel, E., Rossi, V., and Pertot, I. (2012). Modelling the impact of climate change on the interaction between grapevine and its pests and pathogens: European grapevine moth and powdery mildew. *Agriculture, Ecosystems & Environment* 148, 89–101. <u>https://doi.org/10.1016/j.agee.2011.</u> <u>11.017</u>
- Cameron, R. W. F., Brindley, P., Mears, M., McEwan, K., Ferguson, F., Sheffield, D., et al. (2020). Where the wild things are! Do urban green spaces with greater avian biodiversity promote more positive emotions in humans? *Urban Ecosyst* 23, 301–317. https://doi.org/10.1007/s11252-020-00929-z
- Casas-Mulet, R., Matthews, E., Geist, J., Durance, I., and Cable, J. (2021). Negative effects of parasite exposure and variable thermal stress on brown trout (Salmo trutta) under future climatic and hydropower production scenarios. *Climate Change Ecology* 2, 100039. <u>https://doi.org/10.1016/j.ecochg.2021.100039</u>
- Castagneri, D., Vacchiano, G., Hacket-Pain, A., DeRose, R. J., Klein, T., and Bottero, A. (2022). Meta-analysis Reveals Different Competition Effects on Tree Growth Resistance and Resilience

to Drought. *Ecosystems* 25, 30-43. <u>https://doi.org/10.1007/</u> s10021-021-00638-4

- Chakraborty, D., Ciceu, A., Ballian, D., Benito Garzón, M., Bolte, A., Bozic, G., et al. (2024). Assisted tree migration can preserve the European forest carbon sink under climate change. *Nature Climate Change* 14, 845–852. <u>https://doi.org/10.1038/s41558-024-02080-5</u>
- Chamberlain, D. E., Negro, M., Caprio, E., and Rolando, A. (2013). Assessing the sensitivity of alpine birds to potential future changes in habitat and climate to inform management strategies. *Biological Conservation* 167, 127–135. <u>https://doi.org/10.1016/j. biocon.2013.07.036</u>
- Chan, D. S. M., Lau, R., Aune, D., Vieira, R., Greenwood, D. C., Kampman, E., et al. (2011). Red and Processed Meat and Colorectal Cancer Incidence: Meta-Analysis of Prospective Studies. *PLOS ONE* 6, e20456. <u>https://doi.org/10.1371/journal.pone.0020456</u>
- Charlson, F., Ali, S., Benmarhnia, T., Pearl, M., Massazza, A., Augustinavicius, J., et al. (2021). Climate Change and Mental Health: A Scoping Review. *International Journal of Environmental Research and Public Health* 18, 4486. <u>https://doi.org/10.3390/</u> <u>ijerph18094486</u>
- Cheng, X., Frank, U., Zhao, F., Capella, J. R., Winkler, J. B., Schnitzler, J.-P., et al. (2023). Plant growth traits and allergenic potential of Ambrosia artemisiifolia pollen as modified by temperature and NO2. *Environmental and Experimental Botany* 206, 105193. <u>https://doi.org/10.1016/j.envexpbot.2022.105193</u>
- Chersich, M. F., Pham, M. D., Areal, A., Haghighi, M. M., Manyuchi, A., Swift, C. P., et al. (2020). Associations between high temperatures in pregnancy and risk of preterm birth, low birth weight, and stillbirths: systematic review and meta-analysis. *BMJ* 371, m3811. <u>https://doi.org/10.1136/bmj.m3811</u>
- Cheung, S. S., Lee, J. K. W., and Oksa, J. (2016). Thermal stress, human performance, and physical employment standards. *Appl. Physiol. Nutr. Metab.* 41, S148–S164. <u>https://doi.org/10.1139/ apnm-2015-0518</u>
- Chimani, B., Heinrich, B., Hofstätter, M., Kerschbaumer, M., Kienberger, S., Leuprecht, A., et al. (2016). Endbericht ÖKS15 – Klimaszenarien für Österreich – Daten – Methoden – Klimaanalyse. Wien. Available at: <u>https://doi.org/10.60669/wgxm-4927</u>
- Chirichella, R., Stephens, P. A., Mason, T. H. E., and Apollonio, M. (2021). Contrasting Effects of Climate Change on Alpine Chamois. J Wildl Manag 85, 109–120. <u>https://doi.org/10.1002/jwmg.21962</u>
- Cissé, G. (2019). Food-borne and water-borne diseases under climate change in low- and middle-income countries: Further efforts needed for reducing environmental health exposure risks. *Acta Tropica* 194, 181–188. <u>https://doi.org/10.1016/j.actatropica.2019.03.012</u>
- Cissé, G., McLeman, R., Adams, H., Aldunce, P., Bowen, K., Campbell-Lendrum, D., et al. (2022). Health, Wellbeing, and the Changing Structure of Communities. *Climate Change 2022: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*, 1041–1170. <u>https://doi.org/10.1017/</u> 9781009325844.009.1042
- Clara, M., and Müller-Rechberger, H. (2023). Spurenstoffe in Gewässern und bei der Abwasserreinigung: Vorschläge der Europäischen Kommission zur Aktualisierung der Umweltqualitätsnormenrichtlinie und der kommunalen Abwasserricht-

linie. Österr Wasser- und Abfallw 75, 540-548. https://doi. org/10.1007/s00506-023-00972-y

- Clayton, S. (2020). Climate anxiety: Psychological responses to climate change. *Journal of Anxiety Disorders* 74, 102263. <u>https:// doi.org/10.1016/j.janxdis.2020.102263</u>
- Club of Rome ed. (2022). *Earth for All A Survival Guide for Humanity*. München: Oekom. Available at: <u>https://www.oekom.de/</u> <u>buch/earth-for-all-9783962383879</u>
- Cohen-Shacham, E., Janzen, C., Maginnis, S., and Walters, G. eds. (2016). Nature-based solutions to address global societal challenges. Gland: IUCN. Available at: <u>https://portals.iucn.org/library/node/46191</u> (Accessed October 2, 2023).
- Cook-Patton, S. C., Leavitt, S. M., Gibbs, D., Harris, N. L., Lister, K., Anderson-Teixeira, K. J., et al. (2020). Mapping carbon accumulation potential from global natural forest regrowth. *Nature* 585, 545–550. <u>https://doi.org/10.1038/s41586-020-2686-x</u>
- Cooper, T., Hart, K., and Baldock, D. (2009). The Provision of Public Goods through Agriculture in the European Union. Report Prepared for DG Agriculture and Rural Development, Contract No 30-CE-0233091/00-28. London: Institute for European Environmental Policy. Available at: <u>https://ieep.eu/wp-content/uploads/2010/01/final\_pg\_report.pdf</u>
- Corlatti, L., Palme, R., Valencak, T. G., and Gomez, K. M. (2023). Season-dependent impact of forage quality on stress in alpine chamois. *Ecology and Evolution* 13, e10045. <u>https://doi. org/10.1002/ece3.10045</u>
- Costa, M., Marchi, N., Bettella, F., Bolzon, P., Berger, F., and Lingua, E. (2021). Biological Legacies and Rockfall: The Protective Effect of a Windthrown Forest. *Forests* 12, 1141. <u>https://doi.org/10.3390/f12091141</u>
- Cowie, A. L., Berndes, G., Bentsen, N. S., Brandão, M., Cherubini, F., Egnell, G., et al. (2021). Applying a science-based systems perspective to dispel misconceptions about climate effects of forest bioenergy. GCB Bioenergy 13, 1210–1231. <u>https://doi.org/10.1111/gcbb.12844</u>
- Creutzig, F., Niamir, L., Bai, X., Callaghan, M., Cullen, J., Díaz-José, J., et al. (2022a). Demand-side solutions to climate change mitigation consistent with high levels of well-being. *Nat. Clim. Chang.* 12, 36–46. https://doi.org/10.1038/s41558-021-01219-y
- Creutzig, F., Roy, J., Devine-Wright, P., Díaz-José, J., Geels, F. W., Grubler, A., et al. (2022b). "Demand, services and social aspects of mitigation", in *IPCC 2022: Climate Change 2022: Mitigation* of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, eds. P. R. Shukla, J. Skea, R. Slade, A. A. Khourdajie, R. van Diemen, D. McCollum, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press). <u>https://doi.</u> org/10.1017/9781009157926.007
- Crippa, M., Solazzo, E., Guizzardi, D., Monforti-Ferrario, F., Tubiello, F. N., and Leip, A. (2021). Food systems are responsible for a third of global anthropogenic GHG emissions. *Nature Food* 2, 198–209. https://doi.org/10.1038/s43016-021-00225-9
- Dalhaus, T., Schlenker, W., Blanke, M. M., Bravin, E., and Finger, R. (2020). The Effects of Extreme Weather on Apple Quality. *Sci Rep* 10, 7919. <u>https://doi.org/10.1038/s41598-020-64806-7</u>
- Dallago, G. M., Wade, K. M., Cue, R. I., McClure, J. T., Lacroix, R., Pellerin, D., et al. (2021). Keeping Dairy Cows for Longer: A Critical Literature Review on Dairy Cow Longevity in High Milk-Producing Countries. *Animals* 11, 808. <u>https://doi.org/10.3390/ani11030808</u>

- Daniel, T. C. (2001). Whither scenic beauty? Visual landscape quality assessment in the 21st century. *Landscape and Urban Planning* 54, 267–281. <u>https://doi.org/10.1016/S0169-2046(01)</u> 00141-4
- Darnhofer, I., Schermer, M., Steinbacher, M., Gabillet, M., and Daugstad, K. (2017). Preserving permanent mountain grasslands in Western Europe: Why are promising approaches not implemented more widely? *Land Use Policy* 68, 306–315. <u>https://doi.org/10.1016/j.landusepol.2017.08.005</u>
- Davis, K. F., Downs, S., and Gephart, J. A. (2021). Towards food supply chain resilience to environmental shocks. *Nat Food* 2, 54–65. <u>https://doi.org/10.1038/s43016-020-00196-3</u>
- de Sainte Marie, C. (2014). Rethinking agri-environmental schemes. A result-oriented approach to the management of species-rich grasslands in France. *Journal of Environmental Planning and Management* 57, 704–719. <u>https://doi.org/10.1080/09</u> <u>640568.2013.763772</u>
- De Schrijver, E., Royé, D., Gasparrini, A., Franco, O. H., and Vicedo-Cabrera, A. M. (2023). Exploring vulnerability to heat and cold across urban and rural populations in Switzerland. *Environ. Res.: Health* 1, 025003. <u>https://doi.org/10.1088/2752-5309/</u> <u>acab78</u>
- Den Herder, M., Moreno, G., Mosquera-Losada, R. M., Palma, J. H. N., Sidiropoulou, A., Santiago Freijanes, J. J., et al. (2017). Current extent and stratification of agroforestry in the European Union. Agriculture, Ecosystems & Environment 241, 121–132. https://doi.org/10.1016/j.agee.2017.03.005
- Deutsch, C. A., Tewksbury, J. J., Tigchelaar, M., Battisti, D. S., Merrill, S. C., Huey, R. B., et al. (2018). Increase in crop losses to insect pests in a warming climate. *Science* 361, 916–919. <u>https:// doi.org/10.1126/science.aat3466</u>
- Dorendorf, J., Eschenbach, A., Schmidt, K., and Jensen, K. (2015). Both tree and soil carbon need to be quantified for carbon assessments of cities. Urban Forestry & Urban Greening 14, 447– 455. <u>https://doi.org/10.1016/j.ufug.2015.04.005</u>
- Dou, Y., da Silva, R. F. B., Yang, H., and Liu, J. (2018). Spillover effect offsets the conservation effort in the Amazon. *J. Geogr. Sci.* 28, 1715–1732. https://doi.org/10.1007/s11442-018-1539-0
- Duchenne-Moutien, R. A., and Neetoo, H. (2021). Climate Change and Emerging Food Safety Issues: A Review. *Journal of Food Protection* 84, 1884–1897. <u>https://doi.org/10.4315/JFP-21-141</u>
- Duluins, O., and Baret, P. V. (2024). A systematic review of the definitions, narratives and paths forwards for a protein transition in high-income countries. *Nat Food* 5, 28–36. <u>https://doi.org/10.1038/s43016-023-00906-7</u>
- Duscher, G. G., Kienberger, S., Haslinger, K., Holzer, B., Zimpernik, I., Fuchs, R., et al. (2022). Hyalomma spp. in Austria—The Tick, the Climate, the Diseases and the Risk for Humans and Animals. *Microorganisms* 10, 1761. <u>https://doi.org/10.3390/microorganisms10091761</u>
- Eastwood, R., Lipton, M., and Newell, A. (2010). "Chapter 65 Farm Size", in *Handbook of Agricultural Economics*, (Elsevier), 3323– 3397. <u>https://doi.org/10.1016/S1574-0072(09)04065-1</u>
- Ebner, M., Schirpke, U., and Tappeiner, U. (2022). How do anthropogenic pressures affect the provision of ecosystem services of small mountain lakes? *Anthropocene* 38, 100336. <u>https://doi.org/10.1016/j.ancene.2022.100336</u>
- Ebrahimi, E., Manschadi, A. M., Neugschwandtne, R. W., Eitzinger, J., Thaler, S., and Kaul, H.-P. (2016). Assessing the impact of climate change on crop management in winter wheat a case

study for Eastern Austria. J AGR SCI 154, 1153–1170. <u>https://doi.org/10.1017/S0021859616000083</u>

- Eccel, E., Zollo, A. L., Mercogliano, P., and Zorer, R. (2016). Simulations of quantitative shift in bio-climatic indices in the viticultural areas of Trentino (Italian Alps) by an open source R package. *Computers and Electronics in Agriculture* 127, 92–100. <u>https://doi.org/10.1016/j.compag.2016.05.019</u>
- Egarter Vigl, L., Schmid, A., Moser, F., Balotti, A., Gartner, E., Katz, H., et al. (2018). Upward shifts in elevation – a winning strategy for mountain viticulture in the context of climate change? *E3S Web Conf.* 50, 02006. <u>https://doi.org/10.1051/e3sconf/20185002006</u>
- Ehbrecht, M., Seidel, D., Annighöfer, P., Kreft, H., Köhler, M., Zemp, D. C., et al. (2021). Global patterns and climatic controls of forest structural complexity. *Nat Commun* 12, 519. <u>https:// doi.org/10.1038/s41467-020-20767-z</u>
- Eisenmenger, N., Wiedenhofer, D., Schaffartzik, A., Giljum, S., Bruckner, M., Schandl, H., et al. (2016). Consumption-based material flow indicators — Comparing six ways of calculating the Austrian raw material consumption providing six results. *Ecological Economics* 128, 177–186. <u>https://doi.org/10.1016/j.</u> ecolecon.2016.03.010
- Eitzinger, J., Trnka, M., Semeradova, D., Thaler, S., Svobodova, E., Hlavinka, P., et al. (2013). Regional climate change impacts on agricultural crop production in Central and Eastern Europe – hotspots, regional differences and common trends. *Journal* of Agricultural Science 151, 787–812. <u>https://doi.org/10.1017/</u> <u>S0021859612000767</u>
- Eker, S., Reese, G., and Obersteiner, M. (2019). Modelling the drivers of a widespread shift to sustainable diets. *Nature Sustain.* 2, 725–735. <u>https://doi.org/10.1038/s41893-019-0331-1</u>
- Ekmekcioglu, C., Wallner, P., Kundi, M., Weisz, U., Haas, W., and Hutter, H.-P. (2018). Red meat, diseases, and healthy alternatives: A critical review. *Critical Reviews in Food Science and Nutrition* 58, 247–261. <u>https://doi.org/10.1080/10408398.2016.115</u> <u>8148</u>
- Elkin, C., Giuggiola, A., Rigling, A., and Bugmann, H. (2015). Short- and long-term efficacy of forest thinning to mitigate drought impacts in mountain forests in the European Alps. *Ecological Applications* 25, 1083–1098. <u>https://doi.org/10.1890/14-0690.1</u>
- Ellena, M., Breil, M., and Soriani, S. (2020). The heat-health nexus in the urban context: A systematic literature review exploring the socio-economic vulnerabilities and built environment characteristics. *Urban Climate* 34, 100676. <u>https://doi.org/10.1016/j.</u> <u>uclim.2020.100676</u>
- Ellis, E. C., Pascual, U., and Mertz, O. (2019). Ecosystem services and nature's contribution to people: negotiating diverse values and trade-offs in land systems. *Current Opinion in Environmental Sustainability* 38, 86–94. <u>https://doi.org/10.1016/j.cos-</u> ust.2019.05.001
- Endler, C., and Matzarakis, A. (2011). Climatic potential for tourism in the Black Forest, Germany – winter season. *International Journal of Biometeorology* 55, 339–351. <u>https://doi.org/10.1007/</u> s00484-010-0342-0
- Erb, K., Haberl, H., Le Noë, J., Tappeiner, U., Tasser, E., and Gingrich, S. (2022). Changes in perspective needed to forge 'noregret' forest-based climate change mitigation strategies. *GCB Bioenergy* 14, 246–257. <u>https://doi.org/10.1111/gcbb.12921</u>
- Erb, K.-H., Fetzel, T., Plutzar, C., Kastner, T., Lauk, C., Mayer, A., et al. (2016). Biomass turnover time in terrestrial ecosystems

- Erb, K.-H., Gratzer, G., Haberl, H., Tappeiner, U., Tasser, E., and Gingrich, S. (2023a). Energie aus Holzbiomasse: begrenzte Klimaschutz-Potenziale. CCCA Fact Sheet #41. Graz: CCCA. Available at: <u>https://ccca.ac.at/fileadmin/00\_DokumenteHauptmenue/02\_Klimawissen/FactSheets/41\_FS\_holzbiomasse\_ 202303.pdf</u>
- Erb, K.-H., Haberl, H., and Gingrich, S. eds. (2023b). Renewable energy potentials: Biomass/bioenergy., in *W. Winiwarter & V. J. Bruckman (Eds.), Energy Transition for a sustainable society*, (Wien: Austrian Academy of Sciences Press), 29–35. <u>https://doi.org/10.1553/KIOESOP\_013</u>
- Erb, K.-H., Kastner, T., Plutzar, C., Bais, A. L. S., Carvalhais, N., Fetzel, T., et al. (2018). Unexpectedly large impact of forest management and grazing on global vegetation biomass. *Nature* 553, 73–76. https://doi.org/10.1038/nature25138
- Erb, K.-H., Matej, S., Haberl, H., and Gingrich, S. (2024). Sustainable land systems in the Anthropocene: Navigating the global land squeeze. *One Earth* 7, 1170–1186. <u>https://doi.org/10.1016/j. oneear.2024.06.011</u>
- ESPON (2012). Climate change and Europe's regions. *ESPON*. Available at: <u>https://www.espon.eu/climate-2012</u> (Accessed September 19, 2023).
- European Commission (2022). In focus: How can the EU help those touched by energy poverty? Available at: <u>https://commission.europa.eu/news/focus-how-can-eu-help-those-touchedenergy-poverty-2022-02-16\_en</u>
- European Court of Auditors (2024). Common Agricultural Policy Plans – Greener, but not matching the EU's ambitions for the climate and the environment. Special Report. Luxembourg: European Court of Auditors. Available at: <u>https://www.eca.europa.</u> <u>eu/en/publications/SR-2024-20</u>
- European Environment Agency (2017). Forest: growing stock, increment and fellings. Available at: <u>https://www.eea.europa. eu/data-and-maps/indicators/forest-growing-stock-increment-and-fellings-3/assessment</u> (Accessed September 18, 2023).
- European Environment Agency (2021). Global climate change impacts and the supply of agricultural commodities to Europe. European Environment Agency. Available at: <u>https://www.eea.</u> <u>europa.eu/en/analysis/publications/global-climate-change-impacts-and</u>
- Eversberg, D., Holz, J., and Pungas, L. (2023). The bioeconomy and its untenable growth promises: reality checks from research. *Sustain Sci* 18, 569–582. <u>https://doi.org/10.1007/s11625-022-01237-5</u>
- Faivre, N., Fritz, M., Freitas, T., de Boissezon, B., and Vandewoestijne, S. (2017). Nature-Based Solutions in the EU: Innovating with nature to address social, economic and environmental challenges. *Environmental Research* 159, 509–518. <u>https://doi.org/10.1016/j.envres.2017.08.032</u>
- Falkner, K., Mitter, H., Moltchanova, E., and Schmid, E. (2019). A zero-inflated Poisson mixture model to analyse spread and abundance of the Western Corn Rootworm in Austria. Agricultural Systems 174, 105–116. <u>https://doi.org/10.1016/j.agsy.2019.</u> 04.010
- FAOSTAT (2023). Food Security and Nutrition / Suite of Food Security Indicators. Available at: <u>https://www.fao.org/faostat/en/</u><u>#data/FS/metadata</u> (Accessed November 17, 2024).
- Farchi, S., Sario, M. D., Lapucci, E., Davoli, M., and Michelozzi, P. (2017). Meat consumption reduction in Italian regions: Health

co-benefits and decreases in GHG emissions. *PLOS ONE* 12, e0182960. <u>https://doi.org/10.1371/journal.pone.0182960</u>

- Faust, C. L., McCallum, H. I., Bloomfield, L. S. P., Gottdenker, N. L., Gillespie, T. R., Torney, C. J., et al. (2018). Pathogen spillover during land conversion? *Ecology Letters* 21, 471–483. <u>https://doi. org/10.1111/ele.12904</u>
- FEDIAF EuropeanPetFood (2023). Annual Report 2023. Bruxelles: European Pet Food Industry Federation. Available at: <u>https://europeanpetfood.org/wp-content/uploads/2023/07/FEDIAF\_Annual-Report\_2023.pdf</u>
- Fehrenbach, H., Bischoff, M., Böttcher, H., Reise, J., and Hennenberg, K. J. (2022). The Missing Limb: Including Impacts of Biomass Extraction on Forest Carbon Stocks in Greenhouse Gas Balances of Wood Use. *Forests* 13, 365. <u>https://doi.org/10.3390/</u> f13030365
- Fernandez, A., Black, J., Jones, M., Wilson, L., Salvador-Carulla, L., Astell-Burt, T., et al. (2015). Flooding and Mental Health: A Systematic Mapping Review. *PLOS ONE* 10, e0119929. <u>https:// doi.org/10.1371/journal.pone.0119929</u>
- Ferrari, G. N., Leal, G. C. L., Thom de Souza, R. C., and Galdamez, E. V. C. (2023). Impact of climate change on occupational health and safety: A review of methodological approaches. *Work* 74, 485–499. <u>https://doi.org/10.3233/WOR-211303</u>
- Ferreira, J., Lennox, G. D., Gardner, T. A., Thomson, J. R., Berenguer, E., Lees, A. C., et al. (2018). Carbon-focused conservation may fail to protect the most biodiverse tropical forests. *Nature Clim Change* 8, 744–749. <u>https://doi.org/10.1038/s41558-018-0225-7</u>
- Feusthuber, E., Mitter, H., Schönhart, M., and Schmid, E. (2017). Integrated modelling of efficient crop management strategies in response to economic damage potentials of the Western Corn Rootworm in Austria. Agricultural Systems 157, 93–106. <u>https:// doi.org/10.1016/j.agsy.2017.07.011</u>
- Fiandaca, M. S. (2017). Systems healthcare: a holistic paradigm for tomorrow'. BMC Systems Biology 11, 142. <u>https://doi.org/10. 1186/s12918-017-0521-2</u>
- Ficke, A. D., Myrick, C. A., and Hansen, L. J. (2007). Potential impacts of global climate change on freshwater fisheries. *Rev Fish Biol Fisheries* 17, 581–613. <u>https://doi.org/10.1007/s11160-007-9059-5</u>
- Finger, R., Lazzarotto, P., and Calanca, P. (2010). Bio-economic assessment of climate change impacts on managed grassland production. Agricultural Systems 103, 666–674. <u>https://doi.org/10. 1016/j.agsy.2010.08.005</u>
- Fleischer, P., Pichler, V., Fleischer Jr, P., Holko, L., Máliš, F., Gömöryová, E., et al. (2017). Forest ecosystem services affected by natural disturbances, climate and land-use changes in the Tatra Mountains. *Clim. Res.* 73, 57–71. <u>https://doi.org/10.3354/ cr01461</u>
- Fleischmann, B., Mayer, A., Görg, C., and Pichler, M. (2024). Negotiating biophysical limits in the European Union's bioeconomy: a critical analysis of two conflicts over regulating biomass use in EU policy. *Sustain Sci* 19, 1935–1948. <u>https://doi.org/10.1007/ s11625-024-01543-0</u>
- Flinzberger, L., Zinngrebe, Y., and Plieninger, T. (2020). Labelling in Mediterranean agroforestry landscapes: a Delphi study on relevant sustainability indicators. *Sustain Sci* 15, 1369–1382. <u>https://doi.org/10.1007/s11625-020-00800-2</u>
- Foldal, C. B., Hollosi, B., Kodym, A., and Öllerer, B. (2022). Die Grüne Stadt aus forstlicher Sicht. Wien: Bundesforschungszentrum für Wald (BFW), Zentralanstalt für Geodynamik und

Meteorologie (ZAMG). Available at: <u>https://www.bfw.gv.at/</u> wp-content/uploads/Die-Gru%CC%88ne-Stad-aus-forstlicher-<u>Sicht.pdf</u>

- Foldal, C. B., Kittinger, M., Haas, E., and Zechmeister-Boltenstern, S. (2024). Policy measures effectively reduce soil nitrous oxide emissions with minor trade-offs in crop yield. *European J Soil Science* 75, e13475. <u>https://doi.org/10.1111/ejss.13475</u>
- Follak, S., Hochfellner, L., and Schwarz, M. (2023). Aufgepasst! Der Stechapfel wird immer mehr zum Problem. *Der Pflanzenarzt* 76, 30–31.
- Follak, S., Schleicher, C., Schwarz, M., and Essl, F. (2017). Major emerging alien plants in Austrian crop fields. Weed Research 57, 406–416. <u>https://doi.org/10.1111/wre.12272</u>
- Fonds Gesundes Österreich (2015). Health in all Policies in der Praxis der Gesundheitsförderung. Satellitenveranstaltung – Jugendhilfe trifft Gesundheitsförderung., in Tagungsband der 17. Österreichischen Gesundheitsförderungskonferenz, Salzburg 2015, (Salzburg). Available at: <u>https://fgoe.org/sites/fgoe.org/ files/2017-10/2017-08-25%202.pdf</u>
- Fontana, V., Radtke, A., Bossi Fedrigotti, V., Tappeiner, U., Tasser, E., Zerbe, S., et al. (2013). Comparing land-use alternatives: Using the ecosystem services concept to define a multi-criteria decision analysis. *Ecological Economics* 93, 128–136. <u>https://doi. org/10.1016/j.ecolecon.2013.05.007</u>
- Formayer, H., and Goler, R. (2013). "Auswirkungen des Klimawandels auf die klimatische Eignung für den Weinbau in Österreich und Europa", in *Weinbau und Klimawandel. Erste Analysen aus Österreich und führenden internationalen Weinbaugebieten*, (Wien: Verlag der Österreichischen Akademie der Wissenschaften).
- Formayer, H., Hohenwallner-Ries, D., Moshammer, H., Thaler, T., Kratzer, A., Krimm, H., et al. (2024a). Social vulnerability and adaptive capacity to risks from the impacts of future heat waves and air pollution. ACRP. Available at: <u>https://www.klimafonds.</u> gv.at/wp-content/uploads/2024/12/C163521-ACRP13-Future\_ Capacity-KR20AC0K18151-EB.pdf
- Formayer, H., Jandl, R., Bohner, A., Eitzinger, J., Erb, K.-H., Haas, W., et al. (2024b). "Kapitel 1. Ziele, Herangehensweise und Kontext", in APCC Special Report: Landnutzung und Klimawandel in Österreich, eds. R. Jandl, U. Tappeiner, C. B. Foldal, and K. Erb (Berlin, Heidelberg: Springer Spektrum), 57–105. <u>https:// doi.org/10.1007/978-3-662-67864-0\_3</u>
- Fraser, T., Aldrich, D. P., Panagopoulos, C., Hummel, D., and Kim, D. (2022). The harmful effects of partisan polarization on health. *PNAS Nexus* 1, pgac011. <u>https://doi.org/10.1093/pnasnexus/pgac011</u>
- Frehner, M., Wasser, B., and Schwitter, R. (2005). Nachhaltigkeit und Erfolgskontrolle im Schutzwald. Wegleitung für Pflegemassnahmen in Wäldern mit Schutzfunktion. Bern: Bundesamt für Umwelt, Wald und Landschaft (BUWAL). Available at: <u>https://www.bafu.admin.ch/bafu/de/home/themen/wald/ publikationen-studien/publikationen/nachhaltigkeit-und-erfolgskontrolle-im-schutzwald.html</u>
- Frenck, G., Leitinger, G., Obojes, N., Hofmann, M., Newesely, C., Deutschmann, M., et al. (2018). Community-specific hydraulic conductance potential of soil water decomposed for two Alpine grasslands by small-scale lysimetry. *Biogeosciences* 15, 1065– 1078. <u>https://doi.org/10.5194/bg-15-1065-2018</u>
- Frey, V., and Bruckner, M. (2021). The global carbon footprint of Austria's consumption of agricultural (food and nonfood) prod-

ucts. Ecological Economic Papers 41, 1–14. <u>https://doi.org/10.</u> 57938/7cd7ee70-111e-4784-80e9-84965bf2a371

- Fritz, C., Grassauer, F., and Terler, G. (2022). Estimating greenhouse gas abatement costs: method and application using the example of an increased pasture feed intake on cattle farms in Austria. Abschätzung von Treibhausgas-Vermeidungskosten: Methodik und Anwendung am Beispiel eines erhöhten Weidefutteranteils auf rinderhaltenden Betrieben in Österreich. Austrian Journal of Agricultural Economics and Rural Studies 30. <u>https://doi.org/ DOI 10.15203/OEGA\_30.4</u>
- Frühauf, A., Niedermeier, M., and Kopp, M. (2020). Intention to Engage in Winter Sport in Climate Change Affected Environments. Frontiers in Public Health 8. <u>https://doi.org/10.3389/ fpubh.2020.598297</u>
- Fuchsig, H., and Scholl-Bürgi, S. (2022). Welchen Einfluss hat der Klimawandel auf die Gesundheit von Kindern und Jugendlichen?'. Pädiatrie & Pädologie 57, 178–183. <u>https://doi.org/10. 1007/s00608-022-01002-y</u>
- Fuller, R. A., Irvine, K. N., Devine-Wright, P., Warren, P. H., and Gaston, K. J. (2007). Psychological benefits of greenspace increase with biodiversity. *Biology Letters* 3, 390–394. <u>https://doi.org/10.1098/rsbl.2007.0149</u>
- Gago, T., Sargisson, R. J., and Milfont, T. L. (2024). A meta-analysis on the relationship between climate anxiety and wellbeing. *Journal of Environmental Psychology* 94, 102230. <u>https://doi.org/ 10.1016/j.jenvp.2024.102230</u>
- Gale, P., Brouwer, A., Ramnial, V., Kelly, R., Kosmider, R., Fooks, A., et al. (2010). Assessing the impact of climate change on vector-borne viruses in the EU through the elicitation of expert opinion. *Epidemiology & Infection* 138, 214–225. <u>https://doi.org/10.1017/S0950268809990367</u>
- Gallina, N., Anneville, O., and Beniston, M. (2011). Impacts of extreme air temperatures on cyanobacteria in five deep peri-Alpine lakes. *Journal of Limnology* 70, 186–196. <u>https://doi.org/' 10.4081/jlimnol.2011.186</u>
- García, V. R., Gaspart, F., Kastner, T., and Meyfroidt, P. (2020). Agricultural intensification and land use change: assessing country-level induced intensification, land sparing and rebound effect. *Environ. Res. Lett.* 15, 085007. <u>https://doi.org/10.1088/1748-9326/ab8b14</u>
- Gasparrini, A., Guo, Y., Sera, F., Vicedo-Cabrera, A. M., Huber, V., Tong, S., et al. (2017). Projections of temperature-related excess mortality under climate change scenarios. *The Lancet Planetary Health* 1, e360–e367. <u>https://doi.org/10.1016/S2542-5196(17)</u> <u>30156-0</u>
- Gerber, J.-D., Hengstermann, A., and Viallon, F.-X. (2018). "Land policy: How to deal with scarcity of land", in *Instruments of Land Policy*, eds. J.-D. Gerber, T. Hartmann, and A. Hengstermann (London: Routledge).
- Gerhard, C., Suhlmann, G., Ziemßen, F., Donnan, D., Mirku Warschun, and Kühnle, H.-J. (2019). How will Cultured Meat and Meat Alternatives Disrupt the Agricultural and Food Industry? A.T. Kearney. Available at: <u>https://www.kearney. com/documents/291362523/291366549/How+Will+Cultured+Meat+and+Meat+Alternatives+Disrupt+the+Agricultural+and+Food+Industry.pdf</u>
- Gimplinger, D. M., Dobos, G., Schönlechner, R., and Kaul, H.-P. (2007). Yield and quality of grain amaranth (Amaranthus sp.) in Eastern Austria. *Plant, Soil and Environment* 53, 105–112. <u>https://doi.org/10.17221/2224-PSE</u>

- Gingrich, S., Erb, K.-H., Krausmann, F., Gaube, V., and Haberl, H. (2007). Long-term dynamics of terrestrial carbon stocks in Austria: a comprehensive assessment of the time period from 1830 to 2000. *Reg Environ Change* 7, 37–47. <u>https://doi.org/10.1007/</u> s10113-007-0024-6
- Gingrich, S., Lauk, C., Kastner, T., Krausmann, F., Haberl, H., and Erb, K.-H. (2016). "A Forest Transition: Austrian Carbon Budgets 1830–2010", in *Social Ecology*, eds. H. Haberl, M. Fischer-Kowalski, F. Krausmann, V. Winiwarter, H. Haberl, M. Fischer-Kowalski, et al. (Cham: Springer), 417–431. Available at: <u>http://link.springer.com/10.1007/978-3-319-33326-7\_20</u> (Accessed July 10, 2023).
- Gingrich, S., Lauk, C., Krausmann, F., Erb, K.-H., and Le Noë, J. (2021). Changes in energy and livestock systems largely explain the forest transition in Austria (1830–1910). *Land Use Policy* 109, 105624. <u>https://doi.org/10.1016/j.landusepol.2021.105624</u>
- Gingrich, S., Magerl, A., Matej, S., and Le Noë, J. (2022). Forest Transitions in the United States, France and Austria: dynamics of forest change and their socio- metabolic drivers. *Journal of Land Use Science* 17, 113–133. <u>https://doi.org/10.1080/174742</u> <u>3X.2021.2018514</u>
- Gingrich, S., Niedertscheider, M., Kastner, T., Haberl, H., Cosor, G., Krausmann, F., et al. (2015). Exploring long-term trends in land use change and aboveground human appropriation of net primary production in nine European countries. *Land Use Policy* 47, 426–438. <u>https://doi.org/10.1016/j.landusepol.2015.04.027</u>
- Gingrich, S., Weiss, P., Wenzel, W., Anderl, M., Bruckner, M., Díaz-Pinés, E., et al. (2024). "Kapitel 2. Auswirkungen der Landnutzung und -bewirtschaftung sowie naturnaher Ökosysteme auf den Klimawandel: Biophysikalische Effekte, Treibhausgasemissionen und Kohlenstoffspeicher", in APCC Special Report: Landnutzung und Klimawandel in Österreich, eds. R. Jandl, U. Tappeiner, C. B. Foldal, and K. Erb (Berlin, Heidelberg: Springer Spektrum), 107–162. <u>https://doi.org/10.1007/978-3-662-67864-0\_4</u>
- Glad, A., and Mallard, F. (2022). Alpine marmot (Marmota marmota) distribution evolution under climate change: The use of species distribution models at a local scale in the western Pyrenees massif (France). *Ecological Informatics* 69, 101646. <u>https:// doi.org/10.1016/j.ecoinf.2022.101646</u>
- Glade, T. ed. (2020). ExtremA 2019. Aktueller Wissensstand zu Extremereignissen alpiner Naturgefahren in Österreich. Vienna: Vienna University Press. Available at: <u>https://library.oapen.org/ handle/20.500.12657/43512</u>
- Goldman, A. (2014). The health impacts of windstorms: a systematic literature review'. *Public Health* 128, 3–28. <u>https://doi.org/10.1016/j.puhe.2013.09.022</u>
- Golicz, K., Ghazaryan, G., Niether, W., Wartenberg, A. C., Breuer, L., Gattinger, A., et al. (2021). The Role of Small Woody Landscape Features and Agroforestry Systems for National Carbon Budgeting in Germany. *Land* 10, 1028. <u>https://doi.org/10.3390/</u> land10101028
- Gomez, J., Allen, R. J., Turnock, S. T., Horowitz, L. W., Tsigaridis, K., Bauer, S. E., et al. (2023). The projected future degradation in air quality is caused by more abundant natural aerosols in a warmer world. *Commun Earth Environ* 4, 1–11. <u>https://doi.org/10.1038/s43247-023-00688-7</u>
- Gömöry, D., Krajmerová, D., Hrivnák, M., and Longauer, R. (2020). Assisted migration vs. close-to-nature forestry: what are the prospects for tree populations under climate change? *Lesnicky Casopis* 66, 63–70. <u>https://doi.org/10.2478/forj-2020-0008</u>

- Gonçalves, A. C., Malico, I., and Sousa, A. M. O. (2021). "Energy Production from Forest Biomass: An Overview", in *Forest Biomass – From Trees to Energy*, eds. A. C. Gonçalves, A. M. O. Sousa, and I. Malico. Available at: <u>https://www.intechopen.com/ chapters/73136</u> (Accessed October 3, 2023).
- Grabherr, G. (2009). Biodiversity in the high ranges of the Alps: Ethnobotanical and climate change perspectives. *Global Environmental Change* 19, 167–172. <u>https://doi.org/10.1016/j.gloen-vcha.2009.01.007</u>
- Grassauer, F., Herndl, M., Nemecek, T., Guggenberger, T., Fritz, C., Steinwidder, A., et al. (2021). Eco-efficiency of farms considering multiple functions of agriculture: Concept and results from Austrian farms. *Journal of Cleaner Production* 297, 126662. <u>https://doi.org/10.1016/j.jclepro.2021.126662</u>
- Green, A. R., Carrillo, J. E., and Betancourt, J. R. (2002). Why the disease-based model of medicine fails our patients'. *Western Journal of Medicine* 176, 141–143.
- Gregow, H., Laaksonen, A., and Alper, M. E. (2017). Increasing large scale windstorm damage in Western, Central and Northern European forests, 1951–2010. *Sci Rep* 7, 46397. <u>https://doi.org/10.1038/srep46397</u>
- Grima, N., Jutras-Perreault, M.-C., Gobakken, T., Ole Ørka, H., and Vacik, H. (2023). Systematic review for a set of indicators supporting the Common International Classification of Ecosystem Services. *Ecological Indicators* 147, 109978. <u>https://doi. org/10.1016/j.ecolind.2023.109978</u>
- Griscom, B. W., Adams, J., Ellis, P. W., Houghton, R. A., Lomax, G., Miteva, D. A., et al. (2017). Natural climate solutions. *Proceedings of the National Academy of Sciences* 114, 11645–11650. <u>https://doi.org/10.1073/pnas.1710465114</u>
- Gronlund, C. J. (2014). Racial and Socioeconomic Disparities in Heat-Related Health Effects and Their Mechanisms: a Review. *Curr Epidemiol Rep* 1, 165–173. <u>https://doi.org/10.1007/s40471-014-0014-4</u>
- Gschwantner, T. (2019). Holzvorrat auf neuem Höchststand. Wien: Bundesforschungszentrum für Wald. Available at: <u>https://bfw.ac.at/cms\_stamm/050/PDF/BFW-Praxisinfo50\_waldinventur\_fertig\_web.pdf</u>
- Haberl, H. (2015). Competition for land: A sociometabolic perspective. *Ecological Economics* 119, 424–431. <u>https://doi.org/10. 1016/j.ecolecon.2014.10.002</u>
- Haberl, H., and Erb, K.-H. (2017). "Land as a planetary boundary: a socioecological perspective", in *Handbook on Growth* and Sustainability, eds. P. A. Victor and B. Dolter (Cheltenham, UK: Edward Elgar Publishing). Available at: <u>https://doi.org/ 10.4337/9781783473564.00021</u> (Accessed March 21, 2022).
- Haberl, H., and Erb, K.-H. eds. (2023). Energy savings and rebounds., in W. Winiwarter & V. J. Bruckman (Eds.), Energy Transition for a sustainable society, (Wien: Austrian Academy of Sciences Press), 11–16. <u>https://doi.org/10.1553/KIOESOP\_013</u>
- Haberl, H., Erb, K.-H., and Krausmann, F. (2014). Human Appropriation of Net Primary Production: Patterns, Trends, and Planetary Boundaries. *Annual Review of Environment and Resources* 39, 363–391. <u>https://doi.org/10.1146/annurev-environ-121912-094620</u>
- Haberl, H., Kastner, T., Schaffartzik, A., Ludwiczek, N., and Erb, K.-H. (2012). Global effects of national biomass production and consumption: Austria's embodied HANPP related to agricultural biomass in the year 2000. *Ecological Economics* 84, 66–73. <u>https://doi.org/10.1016/j.ecolecon.2012.09.014</u>

- Haines-Young, R., and Potschin, M. (2018). Common International Classification of Ecosystem Services (CICES) V5. 1. Guidance on the Application of the Revised Structure. *Fabis Consulting*. Available at: <u>https://cices.eu/content/uploads/sites/8/2018/01/</u> Guidance-V51-01012018.pdf
- Hallouin, T., Bruen, M., Christie, M., Bullock, C., and Kelly-Quinn, M. (2018). Challenges in Using Hydrology and Water Quality Models for Assessing Freshwater Ecosystem Services: A Review. *Geosciences* 8, 45. <u>https://doi.org/10.3390/geosciences8020045</u>
- Haluza, D. (2024). Klimawandel und Gesundheit (Folge 3): Natur stärkt die urbane Klimaresilienz. ASU Arbeitsmedizin Sozialmedizin Umweltmedizin 2024, 161–164. <u>https://doi.org/10. 17147/asu-1-342890</u>
- Haluza, D., Kaiser, A., Moshammer, H., Flandorfer, C., Kundi, M., and Neuberger, M. (2012). Estimated health impact of a shift from light fuel to residential wood-burning in Upper Austria. *J Expo Sci Environ Epidemiol* 22, 339–343. <u>https://doi.org/ 10.1038/jes.2012.27</u>
- Hammel, K., and Arnold, T. (2012). Understanding the Loss of Traditional Agricultural Systems: A Case Study of Orchard Meadows in Germany. *JAFSCD*, 119–136. <u>https://doi.org/10.5304/ jafscd.2012.024.011</u>
- Handisyde, N., Ross, L., Badjeck, M.-C., and Allison, E. (2006). The Effects of Climate Change on World Aquaculture: A Global Perspective; Final Technical Report. Aquaculture and Fish Genetics Research Programme, Stirling Institute of Aquaculture, DFID: Stirling, UK.
- Hanger-Kopp, S., Kahil, T., Burek, P., Offenzeller, M., Lindinger, H., Sedy, K., et al. (2024). Climate change induced water stress – participatory modeling to identify risks and opportunities in Austrian regions (WaterStressAT) – final report. Laxenburg: International Institute for Applied Systems Analysis (IIASA). Available at: <u>https://www.klimafonds.gv.at/wp-content/uploads/2024/09/C060958-ACRP12-WaterStressAT-KR19AC0K17504-EB.pdf</u>
- Hanger-Kopp, S., and Palka, M. (2022). Decision spaces in agricultural risk management: a mental model study of Austrian crop farmers. *Environ Dev Sustain* 24, 6072–6098. <u>https://doi. org/10.1007/s10668-021-01693-6</u>
- Hanger-Kopp, S., Thaler, T., Seebauer, S., Schinko, T., and Clar, C. (2022). Defining and operationalizing path dependency for the development and monitoring of adaptation pathways. *Global Environmental Change* 72, 102425. <u>https://doi.org/10.1016/j.gloenvcha.2021.102425</u>
- Hannah, L., Roehrdanz, P. R., Ikegami, M., Shepard, A. V., Shaw, M. R., Tabor, G., et al. (2013). Climate change, wine, and conservation. *Proceedings of the National Academy of Sciences* 110, 6907–6912. https://doi.org/10.1073/pnas.1210127110
- Haokip, S. W., Shankar, K., and Lalrinngheta, J. (2020). Climate change and its impact on fruit crops. *J Pharmacogn Phytochem* 9, 435–438. <u>https://www.phytojournal.com/archives/2020.v9.i1.10464/climate-change-and-its-impact-on-fruit-crops</u>
- Hararuk, O., Kurz, W. A., and Didion, M. (2020). Dynamics of dead wood decay in Swiss forests. *For. Ecosyst.* 7, 36. <u>https://doi. org/10.1186/s40663-020-00248-x</u>
- Harmon, M. E. (2019). Have product substitution carbon benefits been overestimated? A sensitivity analysis of key assumptions. *Environ. Res. Lett.* 14, 065008. <u>https://doi.org/10.1088/1748-9326/ab1e95</u>
- Harmon, M. E., Fasth, B. G., Yatskov, M., Kastendick, D., Rock, J., and Woodall, C. W. (2020). Release of coarse woody detritus-re-

lated carbon: a synthesis across forest biomes. *Carbon Balance Manage* 15, 1. <u>https://doi.org/10.1186/s13021-019-0136-6</u>

- Hartig, T., Evans, G. W., Jamner, L. D., Davis, D. S., and Gärling, T. (2003). Tracking restoration in natural and urban field settings. *Journal of Environmental Psychology* 23, 109–123. <u>https://doi. org/10.1016/S0272-4944(02)00109-3</u>
- Harwatt, H. (2019). Including animal to plant protein shifts in climate change mitigation policy: a proposed three-step strategy. *Climate Policy* 19, 533–541. <u>https://doi.org/10.1080/14693062.</u> 2018.1528965
- Hasegawa, T., Wakatsuki, H., Ju, H., Vyas, S., Nelson, G. C., Farrell, A., et al. (2022). A global dataset for the projected impacts of climate change on four major crops. *Sci Data* 9, 58. <u>https://doi.org/10.1038/s41597-022-01150-7</u>
- Haslinger, K., Schöner, W., Abermann, J., Laaha, G., Andre, K., Olefs, M., et al. (2022). Contradictory signal in future surface water availability in Austria: increase on average vs. higher probability of droughts, EGUsphere [preprint]. Available at: <u>https://doi.org/10.5194/egusphere-2022-191</u> (Accessed July 24, 2023).
- Haslmayr, H.-P., Baumgarten, A., Schwarz, M., Huber, S., Prokop, G., Sedy, K., et al. (2018). BEAT – Bodenbedarf für die Ernährungssicherung in Österreich. Bundesministerium für Nachhaltigkeit und Tourismus. Available at: <u>https://dafne.at/content/report\_</u> release/aa85879d-af0f-4273-a1e2-b7f1d7178d41\_0.pdf
- Haszprunar, J., Niebauer, O., and Reitter, A. (1989). Österreichische Forstinventur – Auswertung 1981–1985. Available at: <u>https://bfw.ac.at/rz/bfwcms\_mobil.web?dok=2313</u>
- Hausknost, D., Schriefl, E., Lauk, C., and Kalt, G. (2017). A Transition to Which Bioeconomy? An Exploration of Diverging Techno-Political Choices. *Sustainability* 9, 669. <u>https://doi.org/ 10.3390/su9040669</u>
- He, J., Evans, N. M., Liu, H., and Shao, S. (2020). A review of research on plant-based meat alternatives: Driving forces, history, manufacturing, and consumer attitudes. *Comprehensive Reviews in Food Science and Food Safety* 19, 2639–2656. <u>https://</u> doi.org/10.1111/1541-4337.12610
- Hedlund, J., Carlsen, H., Croft, S., West, C., Bodin, Ö., Stokeld, E., et al. (2022). Impacts of climate change on global food trade networks. *Environ. Res. Lett.* 17, 124040. <u>https://doi.org/10. 1088/1748-9326/aca68b</u>
- Hempel, S., Menz, C., Pinto, S., Galán, E., Janke, D., Estellés, F., et al. (2019). Heat stress risk in European dairy cattle husbandry under different climate change scenarios – uncertainties and potential impacts. *Earth Syst. Dynam* 10, 859–884. <u>https://doi. org/10.5194/esd-10-859-2019</u>
- Hendriks, C., Kranenburg, R., Kuenen, J., van Gijlswijk, R., Kruit, R. W., Segers, A., et al. (2013). The origin of ambient particulate matter concentrations in the Netherlands. *Atmos. Environ.* 69, 289–303. <u>https://doi.org/10.1016/j.atmosenv.2012.12.017</u>
- Herbsthofer, I., Lapin, K., and Meisel, P. (2020). ALPTREES | Sustainable use and management of non native trees in the alpine region. Available at: <u>https://www.bfw.gv.at/en/alptrees-non-native-trees/</u> (Accessed October 2, 2023).
- Hertwich, E. G., Ali, S., Ciacci, L., Fishman, T., Heeren, N., Masanet, E., et al. (2019). Material efficiency strategies to reducing greenhouse gas emissions associated with buildings, vehicles, and electronics—a review. *Environ. Res. Lett.* 14, 043004. <u>https://doi.org/10.1088/1748-9326/ab0fe3</u>
- Herzog, A., Winckler, C., Hörtenhuber, S. J., and Zollitsch, W. (2021). Environmental impacts of implementing basket fans for

heat abatement in dairy farm. Animal 15, 100274. https://doi. org/10.1016/j.animal.2021.100274

- Heurich, M., and Sinner, K. F. (2005). Die Wälder des Nationalparks Bayerischer Wald: Ergebnisse der Waldinventur 2002/2003 im geschichtlichen und waldökologischen Kontext. Grafenau.
- Hiç, C., Pradhan, P., Rybski, D., and Kropp, J. P. (2016). Food Surplus and Its Climate Burdens. *Environ. Sci. Technol.* 50, 4269– 4277. <u>https://doi.org/10.1021/acs.est.5b05088</u>
- Hickman, C. (2020). We need to (find a way to) talk about ... Eco-anxiety. Journal of Social Work Practice 34, 411–424. https://doi.org/10.1080/02650533.2020.1844166
- Hickman, C., Marks, E., Pihkala, P., Clayton, S., Lewandowski, R. E., Mayall, E. E., et al. (2021). Climate anxiety in children and young people and their beliefs about government responses to climate change: a global survey. *The Lancet Planetary Health* 5, e863–e873. <u>https://doi.org/10.1016/S2542-5196(21)00278-3</u>
- Hiebl, J., and Schöner, W. (2018). Temperature inversions in Austria in a warming climate–changes in space and time. *Meteorologische Zeitschrift* 27, 309–323.
- Himes, A., and Muraca, B. (2018). Relational values: the key to pluralistic valuation of ecosystem services. *Current Opinion in Environmental Sustainability* 35, 1–7. <u>https://doi.org/10.1016/j. cosust.2018.09.005</u>
- Hirk, S., Huhulescu, S., Allerberger, F., Lepuschitz, S., Rehak, S., Weil, S., et al. (2016). Necrotizing fasciitis due to Vibrio cholerae non-O1/non-O139 after exposure to Austrian bathing sites. *Wien Klin Wochenschr* 128, 141–145. <u>https://doi.org/10.1007/ s00508-015-0944-y</u>
- Hlásny, T., König, L., Krokene, P., Lindner, M., Montagné-Huck, C., Müller, J., et al. (2021). Bark Beetle Outbreaks in Europe: State of Knowledge and Ways Forward for Management. *Curr Forest*ry Rep 7, 138–165. <u>https://doi.org/10.1007/s40725-021-00142-x</u>
- Hochrainer-Stigler, S., Balkovič, J., Silm, K., and Timonina-Farkas, A. (2019). Large scale extreme risk assessment using copulas: an application to drought events under climate change for Austria. *Computational Management Science* 16, 651–669. <u>https://doi.org/10.1007/s10287-018-0339-4</u>
- Hoffmann, M., Spittler, N., Aigner, E., Körner, D., and Hinterberger, F. eds. (2024). *Earth4All: Austria*. Earth4All. Available at: <u>https://www.clubofrome.org/wp-content/uploads/2024/07/</u> <u>Earth4All-Austria-EN.pdf</u>
- Hogl, K., Eckart, L., Glatzel, S., Kantelhardt, J., Kroisleitner, C., Nordbeck, R., et al. (2023). Governance options for climate smart agriculture on Austrian peatlands. Final Report. Vienna: University of Natural Resources and Life Sciences, Vienna (BOKU), Institute of Forest, Environmental and Natural Resource Policy. Available at: <u>https://projektdatenbank.net/finalreport/8366/</u>
- Hollaus, B., Madner, V., and Schulev-Steindl, E. (2023). "Kapitel 11. Recht", in APCC Special Report: Strukturen für ein klima-freundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 351–388. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_15</u> (Accessed October 4, 2023).
- Höltinger, S., Schmidt, J., Schönhart, M., and Schmid, E. (2014). A spatially explicit techno-economic assessment of green biorefinery concepts. *Biofuels, Bioproducts and Biorefining* 8, 325– 341. <u>https://doi.org/10.1002/bbb.1461</u>
- Hörtenhuber, S. J., Schauberger, G., Mikovits, C., Schönhart, M., Baumgartner, J., Niebuhr, K., et al. (2020). The Effect of Climate Change-Induced Temperature Increase on Performance and

Environmental Impact of Intensive Pig Production Systems. Sustainability 12. <u>https://doi.org/10.3390/su12229442</u>

- Hörtenhuber, S., Lindenthal, T., Amon, B., Markut, T., Kirner, L., and Zollitsch, W. (2010). Greenhouse gas emissions from selected Austrian dairy production systems—model calculations considering the effects of land use change. *Renewable Agriculture and Food Systems* 25, 316–329. <u>https://doi.org/10.1017/</u> <u>S1742170510000025</u>
- Houghton, R. A., and Nassikas, A. A. (2018). Negative emissions from stopping deforestation and forest degradation, globally. *Global Change Biology* 24, 350–359. <u>https://doi.org/10.1111/ gcb.13876</u>
- Howe, C., Suich, H., Vira, B., and Mace, G. M. (2014). Creating win-wins from trade-offs? Ecosystem services for human well-being: A meta-analysis of ecosystem service trade-offs and synergies in the real world. *Global Environmental Change* 28, 263–275. https://doi.org/10.1016/j.gloenvcha.2014.07.005
- Hu, X., Wu, J., Chen, P., Sun, T., and Li, D. (2017). Impact of climate variability and change on crime rates in Tangshan, China. *Science of The Total Environment* 609, 1041–1048. <u>https://doi. org/10.1016/j.scitotenv.2017.07.163</u>
- Huang, F., Wang, D. H., Foo, C. T., Young, A. C., Fok, J. S., and Thien, F. (2022). The Melbourne epidemic thunderstorm asthma event 2016: a 5-year longitudinal study. *Asia Pacific Allergy* 12, e38. <u>https://doi.org/10.5415/apallergy.2022.12.e38</u>
- Hundscheid, L., Voigt, C., Bergthaler, D., Plank, C., Wurzinger, M., and Melcher, A. H. (2024). Policy mix for the sustainable protein transition in Austria – Addressing repercussions of regime shifts as a prerequisite for acceleration. *Environmental Innovation and Societal Transitions* 51, 100819. <u>https://doi.org/ 10.1016/j.eist.2024.100819</u>
- Hundscheid, L., Wurzinger, M., Gühnemann, A., Melcher, A. H., and Stern, T. (2022). Rethinking meat consumption – How institutional shifts affect the sustainable protein transition. *Sustainable Production and Consumption* 31, 301–312. <u>https://doi.org/10.1016/j.spc.2022.02.016</u>
- Hurteau, M. D., Goodwin, M. J., Marsh, C., Zald, H. S., Collins, B., Meyer, M., et al. (2024). Managing fire-prone forests in a time of decreasing carbon carrying capacity. *Frontiers in Ecol & Environ* 22, e2801. <u>https://doi.org/10.1002/fee.2801</u>
- Hussain, R. I., Walcher, R., Eder, R., Allex, B., Wallner, P., Hutter, H.-P., et al. (2019). Management of mountainous meadows associated with biodiversity attributes, perceived health benefits and cultural ecosystem services. *Sci Rep* 9, 14977. <u>https://doi. org/10.1038/s41598-019-51571-5</u>
- Hutter, H.-P., Kundi, M., Moshammer, H., Shelton, J., Krüger, B., Schicker, I., et al. (2015). Replacing Fossil Diesel by Biodiesel Fuel: Expected Impact on Health. Archives of Environmental & Occupational Health 70, 4–9. <u>https://doi.org/10.1080/19338244</u> .2013.787962
- Ikonen, V.-P., Kilpeläinen, A., Zubizarreta-Gerendiain, A., Strandman, H., Asikainen, A., Venäläinen, A., et al. (2017). Regional risks of wind damage in boreal forests under changing management and climate projections. *Can. J. For. Res.* 47, 1632–1645. <u>https://doi.org/10.1139/cjfr-2017-0183</u>
- Imperio, S., Bionda, R., Viterbi, R., and Provenzale, A. (2013). Climate Change and Human Disturbance Can Lead to Local Extinction of Alpine Rock Ptarmigan: New Insight from the Western Italian Alps. *PLOS ONE* 8, e81598. <u>https://doi.org/10.1371/journal.pone.0081598</u>

- Inglis, N. C., and Vukomanovic, J. (2020). Climate change disproportionately affects visual quality of cultural ecosystem services in a mountain region. *Ecosystem Services* 45, 101190. <u>https://</u> doi.org/10.1016/j.ecoser.2020.101190
- IPBES (2018). "Summary for Policymakers", in *The IPBES assessment report on land degradation and restoration*, eds. R. Scholes, L. Montanarella, A. Brainich, N. Barger, B. Brink, M. Cantele, et al. (Bonn, Germany: Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services). Available at: <a href="https://www.researchgate.net/publication/330703000\_The\_IPBES\_assessment\_report\_on\_land\_degradation\_and\_restoration\_Secretariat\_of\_the\_Intergovernmental\_Science-Policy\_Platform\_on\_Biodiversity\_and\_Ecosystem\_Services\_2018/references</a>
- IPBES (2019). Summary for policymakers of the global assessment report on biodiversity and ecosystem services., eds. S. Díaz, J. Settele, E. S. Brondízio, H. T. Ngo, M. Guèze, J. Agard, et al. Bonn Germany: IPBES Secretariat. <u>https://doi.org/10.5281/zenodo.3553579</u>
- IPCC (2019a). "Summary for Policymakers", in Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems, eds. P. R. Shukla, J. Skea, E. Calvo Buendia, V. Masson-Delmotte, H.-O. Pörtner, D. C. Roberts, et al. (Geneva: IPCC). Available at: https://doi.org/10.1017/9781009157988.001
- IPCC (2019b). Refinement to the 2006 IPCC Guidelines for National Greenhouse Gas Inventories — IPCC. Available at: <u>https://www.ipcc.ch/report/2019-refinement-to-the-2006-ipcc-guidelines-for-national-greenhouse-gas-inventories/</u>
- Isermeyer, F., Heidecke, C., and Osterburg, B. (2021). Integrating agriculture into carbon pricing. AgEcon Search. <u>https://doi.org/10.22004/AG.ECON.310017</u>
- IUCN (2021). Invasive alien species and climate change. *Issues Brief.* Available at: <u>https://www.iucn.org/sites/default/files/2022-04/</u> ias\_and\_climate\_change\_issues\_brief\_2021.pdf
- Ivanova, D., Barrett, J., Wiedenhofer, D., Macura, B., Callaghan, M., and Creutzig, F. (2020). Quantifying the potential for climate change mitigation of consumption options. *Environ. Res. Lett.* 15, 093001. <u>https://doi.org/10.1088/1748-9326/ab8589</u>
- Ivanova, D., and Büchs, M. (2022). Implications of shrinking household sizes for meeting the 1.5 °C climate targets. *Ecological Economics* 202, 107590. <u>https://doi.org/10.1016/j.ecolecon.2022.107590</u>
- Jackson, R. B., Jobbágy, E. G., Avissar, R., Roy, S. B., Barrett, D. J., Cook, C. W., et al. (2005). Trading Water for Carbon with Biological Carbon Sequestration. *Science* 310, 1944–1947. <u>https:// doi.org/10.1126/science.1119282</u>
- Jacobi, N., Haas, W., Wiedenhofer, D., and Mayer, A. (2018). Providing an economy-wide monitoring framework for the circular economy in Austria: Status quo and challenges. *Resources, Conservation and Recycling* 137, 156–166. <u>https://doi.org/10.1016/j.</u> resconrec.2018.05.022
- Jäger, J., Darnhofer, I., Dobernig, K., Kettner-Marx, C., Leitinger, G., Meyer, I., et al. (2024). "Kapitel 6. Landnutzungsentscheidungen: Klimawandelrelevante Strategien, Steuerungsinstrumente und Managementansätze", in APCC Special Report: Landnutzung und Klimawandel in Österreich, eds. R. Jandl, U. Tappeiner, C. B. Foldal, K. Erb, R. Jandl, U. Tappeiner, et al. (Berlin, Heidelberg: Springer Spektrum), 339–380. Available at:

https://doi.org/10.1007/978-3-662-67864-0\_8 (Accessed May 17, 2024).

- Jandl, R., Haeler, E., Kindermann, G., Lapin, K., Oettel, J., and Schüler, S. (2024a). Management and biodiversity conservation in Central European forests. *Trees, Forests and People* 17, 100601. <u>https://doi.org/10.1016/j.tfp.2024.100601</u>
- Jandl, R., Hager, H., Kraxner, F., Ledermann, T., and Weiss, P. (2024b). Carbon dioxide neutrality of sustainably managed forests of Austria. *Ecological Engineering* 209, 107417. <u>https://doi. org/10.1016/j.ecoleng.2024.107417</u>
- Jandl, R., Ledermann, T., Kindermann, G., Freudenschuss, A., Gschwantner, T., and Weiss, P. (2018). Strategies for Climate-Smart Forest Management in Austria. *Forests* 9, 592. <u>https://doi.org/10.3390/f9100592</u>
- Jandl, R., Ledermann, T., Kindermann, G., and Weiss, P. (2021). Soil Organic Carbon Stocks in Mixed-Deciduous and Coniferous Forests in Austria. *Front. For. Glob. Change* 4, 688851. <u>https://doi.org/10.3389/ffgc.2021.688851</u>
- Jandl, R., Tappeiner, U., Foldal, C. B., and Erb, K.-H. eds. (2024c). APCC Special Report: Landnutzung und Klimawandel in Österreich. Berlin, Heidelberg: Springer Spektrum. <u>https://doi.org/</u> 10.1007/978-3-662-67864-0
- Jankowski, N., Sageder, M., Bauer, H., Marr, I., Kasper-Giebl, A., Limbeck, A., et al. (2009). "AQUELLA Linz – Oberösterreich" Aerosolquellenanalyse für Linz – Oberösterreich. Available at: <u>https://www.land-oberoesterreich.gv.at/Mediendateien/For-</u> <u>mulare/Dokumente%20UWD%20Abt\_US/Kurzfassung\_End-</u> <u>bericht\_Aquella\_OOE.pdf</u>
- Janssens, C., Havlík, P., Krisztin, T., Baker, J., Frank, S., Hasegawa, T., et al. (2020). Global hunger and climate change adaptation through international trade. *Nature Climate Change* 10, 829–835. <u>https://doi.org/10.1038/s41558-020-0847-4</u>
- Jiao, A., Sun, Y., Avila, C., Chiu, V., Slezak, J., Sacks, D. A., et al. (2023). Analysis of Heat Exposure During Pregnancy and Severe Maternal Morbidity. *JAMA Network Open* 6, e2332780. <u>https://doi.org/10.1001/jamanetworkopen.2023.32780</u>
- Johann, E. (2013). "The History of Utilization and Management of Commons and Consequences of Current Social Change in the Alpine Region of Austria", in *Cultural Severance and the Environment: The Ending of Traditional and Customary Practice on Commons and Landscapes Managed in Common*, ed. I. D. Rotherham (Dordrecht), 133–146. Available at: https://doi. org/10.1007/978-94-007-6159-9\_9 (Accessed October 3, 2023).
- Johnston, C. M. T., and Radeloff, V. C. (2019). Global mitigation potential of carbon stored in harvested wood products. *Proceedings of the National Academy of Sciences* 116, 14526–14531. <u>https://doi.org/10.1073/pnas.1904231116</u>
- Jungwirth, M., Muhar, S., and Schmutz, S. (2002). Re-establishing and assessing ecological integrity in riverine landscapes. *Freshw. Biol.* 47, 867–887. <u>https://doi.org/10.1046/j.1365-2427.</u> 2002.00914.x
- Juroszek, P., Racca, P., Link, S., Farhumand, J., and Kleinhenz, B. (2020). Overview on the review articles published during the past 30 years relating to the potential climate change effects on plant pathogens and crop disease risks. *Plant Pathology* 69, 179–193. https://doi.org/10.1111/ppa.13119
- Kalcher, J., Praxmarer, G., and Teischinger, A. (2017). Quantification of future availabilities of recovered wood from Austrian residential buildings. *Resources, Conservation and Recycling* 123, 143–152. <u>https://doi.org/10.1016/j.resconrec.2016.09.001</u>

- Kalt, G. (2015). Biomass streams in Austria: Drawing a complete picture of biogenic material flows within the national economy. *Resources, Conservation and Recycling* 95, 100–111. <u>https://doi.org/10.1016/j.resconrec.2014.12.006</u>
- Kalt, G. (2018). Carbon dynamics and GHG implications of increasing wood construction: long-term scenarios for residential buildings in Austria. *Carbon Management* 9, 265–275. <u>https:// doi.org/10.1080/17583004.2018.1469948</u>
- Kalt, G., Höher, M., Lauk, C., Schipfer, F., and Kranzl, L. (2016). Carbon accounting of material substitution with biomass: Case studies for Austria investigated with IPCC default and alternative approaches. *Environmental Science & Policy* 64, 155–163. <u>https://doi.org/10.1016/j.envsci.2016.06.022</u>
- Kalt, G., Kaufmann, L., Kastner, T., and Krausmann, F. (2021). Tracing Austria's biomass consumption to source countries: A product-level comparison between bioenergy, food and material. *Ecological Economics* 188, 107129. <u>https://doi.org/10.1016/j. ecolecon.2021.107129</u>
- Kalt, G., and Kranzl, L. (2012). An assessment of international trade related to bioenergy use in Austria—Methodological aspects, recent developments and the relevance of indirect trade. *Energy Policy* 46, 537–549. <u>https://doi.org/10.1016/j.enpol.2012.04.026</u>
- Kalt, G., Mayer, A., Theurl, M. C., Lauk, C., Erb, K.-H., and Haberl, H. (2019). Natural climate solutions versus bioenergy: Can carbon benefits of natural succession compete with bioenergy from short rotation coppice? GCB Bioenergy 11, 1283–1297. <u>https:// doi.org/10.1111/gcbb.12626</u>
- Kalt, T. (2021). Jobs vs. climate justice? Contentious narratives of labor and climate movements in the coal transition in Germany. *Environmental Politics* 30, 1135–1154. <u>https://doi.org/10.1080/ 09644016.2021.1892979</u>
- Kalusová, V., Čeplová, N., Danihelka, J., Večeřa, M., Pyšek, P., Albert, A., et al. (2024). Alien plants of Europe: an overview of national and regional inventories. *Preslia* 96, 149–182. <u>https://doi.org/10.23855/preslia.2024.149</u>
- Kan, S., Chen, B., and Chen, G. (2023). Globalization of forest land use: Increasing threats on climate-vulnerable regions. *Land Use Policy* 132, 106798. <u>https://doi.org/10.1016/j.landusepol.</u> 2023.106798
- Kaplan, S. (1995). The restorative benefits of nature: Toward an integrative framework. *Journal of Environmental Psychology* 15, 169–182. https://doi.org/10.1016/0272-4944(95)90001-2
- Kautz, M., and Delb, H. (2023). Reduziert das Management von Kronenbrüchen Stehendbefall durch Borkenkäfer an Fichten? Available at: <u>https://www.bfw.gv.at/wp-content/uploads/FS-Aktuell\_1.pdf</u>
- Keeler, B. L., Polasky, S., Brauman, K. A., Johnson, K. A., Finlay, J. C., O'Neill, A., et al. (2012). Linking water quality and well-being for improved assessment and valuation of ecosystem services. *Proc. Natl. Acad. Sci. U.S.A.* 109, 18619–18624. <u>https:// doi.org/10.1073/pnas.1215991109</u>
- Keesstra, S., Nunes, J., Novara, A., Finger, D., Avelar, D., Kalantari, Z., et al. (2018). The superior effect of nature based solutions in land management for enhancing ecosystem services. *Science* of *The Total Environment* 610–611, 997–1009. <u>https://doi.org/ 10.1016/j.scitotenv.2017.08.077</u>
- Keith, H., Kun, Z., Hugh, S., Svoboda, M., Mikoláš, M., Adam, D., et al. (2024). Carbon carrying capacity in primary forests shows potential for mitigation achieving the European Green Deal 2030 target. *Commun Earth Environ* 5, 256. <u>https://doi.org/10.1038/s43247-024-01416-5</u>

- Kengdo, S. K., Ahrens, B., Tian, Y., Heinzle, J., Wanek, W., Schindlbacher, A., et al. (2023). Increase in carbon input by enhanced fine root turnover in a long-term warmed forest soil. *Science* of The Total Environment 855, 158800. <u>https://doi.org/10.1016/</u> j.scitotenv.2022.158800
- Kim, B. F., Santo, R. E., Scatterday, A. P., Fry, J. P., Synk, C. M., Cebron, S. R., et al. (2020). Country-specific dietary shifts to mitigate climate and water crises. *Global Environmental Change* 62, 101926. <u>https://doi.org/10.1016/j.gloenvcha.2019.05.010</u>
- Kim, Y., Kim, H., Gasparrini, A., Armstrong, B., Honda, Y., Chung, Y., et al. (2019). Suicide and Ambient Temperature: A Multi-Country Multi-City Study. *Environmental Health Per*spectives 127, 117007. <u>https://doi.org/10.1289/EHP4898</u>
- Kindermann, G. (2021). Kohlenstoffaufnahmeleistung verschiedener Baumarten. Available at: <u>http://193.170.148.82:8080/bfwpdf/kohlenstoffaufnahmeleistung.pdf</u>
- Kindermann, G., and Ledermann, T. (2023). Comparing results of European Forestry Dynamics Model (EFDM) with the single tree growth model Caldis. Vienna: BFW. Available at: <u>http:// 193.170.148.82/AIS55/Details/fullCatalogue/40004093</u>
- Kirchmeir, H., Huber, M., and Fuchs, A. (2022). Die Rolle des Waldes im Klimaschutz. Wie wird unser Wald klimafit? Klagenfurt: E.C.O. Institut für Ökologie im Auftrag von Mutter Erde – Umweltinitiative Wir für die Welt. Available at: <u>https://www.wwf. at/wp-content/uploads/2021/11/Studie-Die-Rolle-des-Waldesim-Klimaschutz.pdf</u>
- Kirchner, M., Schmidt, J., Kindermann, G., Kulmer, V., Mitter, H., Prettenthaler, F., et al. (2015). Ecosystem services and economic development in Austrian agricultural landscapes — The impact of policy and climate change scenarios on trade-offs and synergies. *Ecological Economics* 109, 161–174. <u>https://doi.org/10.1016/j.ecolecon.2014.11.005</u>
- Kirchner, M., Schönhart, M., and Schmid, E. (2016). Spatial impacts of the CAP post-2013 and climate change scenarios on agricultural intensification and environment in Austria. *Ecological Economics* 123, 35–56. <u>https://doi.org/10.1016/j.ecolecon.2015.12.009</u>
- Kleemayr, K., Cech, T., Fürst, W., Hoch, G., Ledermann, T., Markart, G., et al. (2020). "Schutzwald und Extremereignisse in Österreich", in *ExtremA 2019. Aktueller Wissensstand zu Extremereignissen alpiner Naturgefahren in Österreich*, (Vienna University Press), 623–665.
- Klein, D., Wolf, C., Schulz, C., and Weber-Blaschke, G. (2015). 20 years of life cycle assessment (LCA) in the forestry sector: state of the art and a methodical proposal for the LCA of forest production. *Int J Life Cycle Assess* 20, 556–575. <u>https://doi.org/10.1007/s11367-015-0847-1</u>
- Klein, D., Wolf, C., Schulz, C., and Weber-Blaschke, G. (2016). Environmental impacts of various biomass supply chains for the provision of raw wood in Bavaria, Germany, with focus on climate change. *Science of The Total Environment* 539, 45–60. <u>https://doi.org/10.1016/j.scitotenv.2015.08.087</u>
- Kletzan-Slamanig, D., Köppl, A., Sinabell, F., Kirchmayr, S., Müller, S., Rimböck, A., et al. (2022). Analyse klimakontraproduktiver Subventionen in Österreich. Vienna: Österreichisches Institut für Wirtschaftsforschung (WIFO). Available at: <u>https://www. wifo.ac.at/jart/prj3/wifo/resources/person\_dokument/person\_ dokument.jart?publikationsid=69687&mime\_type=application/pdf</u> (Accessed September 15, 2023).
- Knox, J., Daccache, A., Hess, T., and Haro, D. (2016). Meta-analysis of climate impacts and uncertainty on crop yields in Europe.

*Environ. Res. Lett.* 11, 113004. <u>https://doi.org/10.1088/1748-</u>9326/11/11/113004

- Kobler, J., Hochbichler, E., Pröll, G., and Dirnböck, T. (2024). How to Optimize Carbon Sinks and Biodiversity in the Conversion of Norway Spruce to Beech Forests in Austria? *Forests* 15, 359. https://doi.org/10.3390/f15020359
- Koch, S., and Pecher, S. (2020). Neue Herausforderungen für die Anästhesie durch den Klimawandel. *Anaesthesist* 69, 453–462. <u>https://doi.org/10.1007/s00101-020-00770-1</u>
- Kogler, R., Quendler, E., and Boxberger, J. (2015). Analysis of occupational accidents with agricultural machinery in the period 2008–2010 in Austria. *Safety Science* 72, 319–328. <u>https://doi.org/10.1016/j.ssci.2014.10.004</u>
- Koppensteiner, L. J., Kaul, H.-P., Piepho, H.-P., Barta, N., Euteneuer, P., Bernas, J., et al. (2022). Yield and yield components of facultative wheat are affected by sowing time, nitrogen fertilization and environment. *European Journal of Agronomy* 140, 126591. <u>https://doi.org/10.1016/j.eja.2022.126591</u>
- Kortleve, A. J., Mogollón, J. M., Harwatt, H., and Behrens, P. (2024). Over 80% of the European Union's Common Agricultural Policy supports emissions-intensive animal products. *Nat Food* 5, 288–292. <u>https://doi.org/10.1038/s43016-024-00949-4</u>
- Kovats, R. S., Edwards, S. J., Hajat, S., Armstrong, B. G., Ebi, K. L., and Menne, B. (2004). The effect of temperature on food poisoning: a time-series analysis of salmonellosis in ten European countries. *Epidemiology & Infection* 132, 443–453. <u>https://doi. org/10.1017/S0950268804001992</u>
- Kranzl, L., Müller, A., Maia, I., Büchele, R., and Hartner, M. (2018). Wärmezukunft 2050. Erfordernisse und Konsequenzen der Dekarbonisierung von Raumwärme und Warmwasserbereitstellung in Österreich. Wien. Available at: <u>https://www.igwindkraft.at/mmedia/download/2018.02.05/1517825327514183.pdf</u>
- Krautblatter, M., Funk, D., and Günzel, F. K. (2013). Why permafrost rocks become unstable: a rock-ice-mechanical model in time and space. *Earth Surface Processes and Landforms* 38, 876–887. https://doi.org/10.1002/esp.3374
- Krautzer, B., and Graiss, W. (2015). Futterpflanzenzüchtung und Generhaltung. Was braucht und was bietet das Grünland? Irdning-Donnersbachtal: Höhere Bundeslehr- und Forschungsanstalt Raumberg-Gumpenstein. Available at: <u>https://raumberggumpenstein.at/jdownloads/Forschungsberichte/Umweltressourcen\_im\_Gruenland/Generhalt/2\_2015\_Tagungsbeitrag\_ Futterpflanzenzuechtung\_Expertenforum.pdf</u>
- Kraxner, F., Schepaschenko, D., Fuss, S., Lunnan, A., Kindermann, G., Aoki, K., et al. (2017). Mapping certified forests for sustainable management – A global tool for information improvement through participatory and collaborative mapping. *Forest Policy and Economics* 83, 10–18. <u>https://doi.org/10.1016/j.forpol.2017.04.014</u>
- Kraxner, F., Zollitsch, W., Kottusch, C., Bruckman, V. J., Glatzel, S., Hood-Nowotny, R., et al. (2024). "Kapitel 5. Mitigation des Klimawandels", in APCC Special Report: Landnutzung und Klimawandel in Österreich (APCC SR Land), eds. R. Jandl, U. Tappeiner, C. B. Foldal, and K.-H. Erb (Berlin, Heidelberg: Springer Spektrum), 275–338. <u>https://doi.org/10.1007/978-3-662-67864-0\_7</u>
- Krutzler, T., Wiesenberger, H., Heller, C., Gössl, M., Stranner, G., Storch, A., et al. (2016). Szenario erneuerbare Energie 2030 und 2050. Vienna: Umweltbundesamt. Available at: <u>https://www. umweltbundesamt.at/fileadmin/site/publikationen/REP0576.</u> pdf

- Kuehn, L., and McCormick, S. (2017). Heat Exposure and Maternal Health in the Face of Climate Change. International Journal of Environmental Research and Public Health 14, 853. <u>https://doi.org/10.3390/ijerph14080853</u>
- Kühmaier, M., Kral, I., and Kanzian, C. (2022). Greenhouse Gas Emissions of the Forest Supply Chain in Austria in the Year 2018. *Sustainability* 14, 792. <u>https://doi.org/10.3390/su14020792</u>
- Kurth, A.-M., and Schirmer, M. (2014). Thirty years of river restoration in Switzerland: implemented measures and lessons learned. *Environ Earth Sci* 72, 2065–2079. <u>https://doi.org/10. 1007/s12665-014-3115-y</u>
- Lafortezza, R., and Sanesi, G. (2019). Nature-based solutions: Settling the issue of sustainable urbanization. *Environmental Research* 172, 394–398. <u>https://doi.org/10.1016/j.envres.2018.</u> 12.063
- Lantz, V., McMonagle, G., Hennigar, C., Sharma, C., Withey, P., and Ochuodho, T. (2022). Forest succession, management and the economy under a changing climate: Coupling economic and forest management models to assess impacts and adaptation options. *Forest Policy and Economics* 142, 102781. <u>https://doi.org/ 10.1016/j.forpol.2022.102781</u>
- Lauk, C., Kaufmann, L., Theurl, M. C., Wittmann, F., Eder, M., Hörtenhuber, S., et al. (2022). Demand side options to reduce greenhouse gas emissions and the land footprint of urban food systems: A scenario analysis for the City of Vienna. *Journal of Cleaner Production* 359, 132064. <u>https://doi.org/10.1016/j.jclepro.2022.132064</u>
- Lauk, C., Magerl, A., Le Noë, J., Theurl, M. C., and Gingrich, S. (2024). Analyzing long-term dynamics of agricultural greenhouse gas emissions in Austria, 1830–2018. Science of The Total Environment 911, 168667. <u>https://doi.org/10.1016/j.scitotenv.2023.168667</u>
- Lavorel, S., Rey, P. L., Grigulis, K., Zawada, M., and Byczek, C. (2020). Interactions between outdoor recreation and iconic terrestrial vertebrates in two French alpine national parks. *Ecosystem Services* 45, 101155. <u>https://doi.org/10.1016/j.ecoser.</u> 2020.101155
- Le Noë, J., Gingrich, S., Pichler, M., Roux, N., Kaufmann, L., Mayer, A., et al. (2023a). Combining biophysical modeling and Polanyian theory pleads for a re-embedding of the agricultural system in 2050 in Austria. *Environmental Science & Policy* 139, 228– 239. <u>https://doi.org/10.1016/j.envsci.2022.11.006</u>
- Le Noë, J., Manzoni, S., Abramoff, R., Bölscher, T., Bruni, E., Cardinael, R., et al. (2023b). Soil organic carbon models need independent time-series validation for reliable prediction. *Commun Earth Environ* 4, 158. <u>https://doi.org/10.1038/s43247-023-00830-5</u>
- Le Roux, F., Wegner, K. M., Baker-Austin, C., Vezzulli, L., Osorio, C. R., Amaro, C., et al. (2015). The emergence of Vibrio pathogens in Europe: ecology, evolution, and pathogenesis (Paris, 11– 12th March 2015). *Front. Microbiol.* 6. <u>https://doi.org/10.3389/</u> <u>fmicb.2015.00830</u>
- Lebl, K., Nischler, E. M., Walter, M., Brugger, K., and Rubel, F. (2013). First Record of the Disease Vector Anopheles hyrcanus in Austria. *Journal of the American Mosquito Control Association* 29, 59–60. <u>https://doi.org/10.2987/12-6282.1</u>
- Lecina-Diaz, J., Senf, C., Grünig, M., and Seidl, R. (2024). Ecosystem services at risk from disturbance in Europe's forests. *Global Change Biology* 30, e17242. <u>https://doi.org/10.1111/gcb.17242</u>
- Ledermann, T., Braun, M., Kindermann, G., Jandl, R., Ludvig, A., Schadauer, K., et al. (2022). Effects of Silvicultural Adaptation

Measures on Carbon Stock of Austrian Forests. *Forests* 13, 565. https://doi.org/10.3390/f13040565

- Legler, S. E., Caffi, T., and Rossi, V. (2012). A nonlinear model for temperature-dependent development of Erysiphe necator chasmothecia on grapevine leave. *Plant Pathology* 61, 96–105. https://doi.org/10.1111/j.1365-3059.2011.02498.x
- Lehmann, F., and Sander, M. (2021). Treibhausgase in der Anästhesie: Einfache Wege zur klimafreundlicheren Narkose. Hessisches Ärzteblatt, 444–445.
- Lehmann, J., Mempel, F., and Coumou, D. (2018). Increased Occurrence of Record-Wet and Record-Dry Months Reflect Changes in Mean Rainfall'. *Geophysical Research Letters* 45, 13468–13476. <u>https://doi.org/10.1029/2018GL079439</u>
- Lehmann, P., Ammunét, T., Barton, M., Battisti, A., Eigenbrode, S. D., Jepsen, J. U., et al. (2020). Complex responses of global insect pests to climate warming. *Frontiers in Ecol & Environ* 18, 141–150. https://doi.org/10.1002/fee.2160
- Leidwein, A., Kolar, V., Berthold, H., Strauss, G., Krachler, M. M., Weigl, M., et al. (2014). Food security risks for Austria caused by climate change. Vienna: ACRP (Austrian Climate Research Program). Available at: <u>https://homepage.boku.ac.at/imrannad/ Reports/Food\_Security\_Endbericht.pdf</u>
- Leopold, A. (1949). A Sand County Almanac and Sketches Here And There. New York: Oxford University Press. Available at: <u>https://</u> warwick.ac.uk/fac/arts/english/currentstudents/undergraduate/modules/fulllist/first/en124/leopoldsandcountyalmanacexcerpts.pdf
- Lette, M., Bemelmans, W. J. E., Breda, J., Slobbe, L. C. J., Dias, J., and Boshuizen, H. C. (2016). Health care costs attributable to overweight calculated in a standardized way for three European countries. *Eur J Health Econ* 17, 61–69. <u>https://doi.org/10.1007/ s10198-014-0655-8</u>
- Leturcq, P. (2020). GHG displacement factors of harvested wood products: the myth of substitution. *Sci Rep* 10, 20752. <u>https:// doi.org/10.1038/s41598-020-77527-8</u>
- Lewis, K. A., Tzilivakis, J., Green, A., Warner, D. J., Stedman, A., and Naseby, D. (2013). Review of substances/agents that have direct beneficial effect on the environment: mode of action and assessment of efficacy. *EFSA Supporting Publications* 10, 440E. https://doi.org/10.2903/sp.efsa.2013.EN-440
- Lindner, M., Maroschek, M., Netherer, S., Kremer, A., Barbati, A., Garcia-Gonzalo, J., et al. (2010). Climate change impacts, adaptive capacity, and vulnerability of European forest ecosystems. *Forest Ecology and Management* 259, 698–709. <u>https://doi. org/10.1016/j.foreco.2009.09.023</u>
- Liu, Q., Tan, Z.-M., Sun, J., Hou, Y., Fu, C., and Wu, Z. (2020). Changing rapid weather variability increases influenza epidemic risk in a warming climate. *Environ. Res. Lett.* 15, 044004. <u>https://doi.org/10.1088/1748-9326/ab70bc</u>
- Liu, Q., Zhang, Y., Liu, B., Amonette, J. E., Lin, Z., Liu, G., et al. (2018). How does biochar influence soil N cycle? A meta-analysis. *Plant Soil* 426, 211–225. <u>https://doi.org/10.1007/s11104-018-3619-4</u>
- Liu, Z., Rieder, H. E., Schmidt, C., Mayer, M., Guo, Y., Winiwarter, W., et al. (2023). Optimal reactive nitrogen control pathways identified for cost-effective PM2.5 mitigation in Europe. *Nat Commun* 14, 4246. <u>https://doi.org/10.1038/s41467-023-39900-9</u>
- Lorenz, H., Reinsch, T., Hess, S., and Taube, F. (2019a). Is low-input dairy farming more climate friendly? A meta-analysis of the carbon footprints of different production systems. *Journal*

of Cleaner Production 211, 161–170. <u>https://doi.org/10.1016/j.jclepro.2018.11.113</u>

- Lorenz, R., Stalhandske, Z., and Fischer, E. M. (2019b). Detection of a climate change signal in extreme heat, heat stress, and cold in Europe from observations. *Geophysical Research Letters* 46, 8363–8374. <u>https://doi.org/10.1029/2019GL082062</u>
- Losey, S., and Wehrli, A. (2013). Schutzwald in der Schweiz Vom Projekt SilvaProtect-CH zum harmonisierten Schutzwald. Bern: Bundesamt für Umwelt. Available at: <u>https://www.newsd.</u> <u>admin.ch/newsd/message/attachments/29559.pdf</u>
- Luck, S., and Obersteiner, G. (2021). Status Quo zu Lebensmittelabfällen im Projektgebiet. Vienna: Europäische Union Interreg Österreich-Tschechische Republik. Available at: <u>https://2014-2020.</u> <u>at-cz.eu/data/projects/a/58/1079\_file\_1.pdf</u>
- Ludvig, A., Braun, M., Hesser, F., Ranacher, L., Fritz, D., Gschwantner, T., et al. (2021). Comparing policy options for carbon efficiency in the wood value chain: evidence from Austria. *Journal* of Cleaner Production 292, 125985. <u>https://doi.org/10.1016/ j.jclepro.2021.125985</u>
- Luick, R., Hennenberg, K., Leuschner, C., Grossmann, M., Jedicke, E., Schoof, N., et al. (2021). Urwälder, Natur- und Wirtschaftswälder im Kontext von Biodiversitäts- und Klimaschutz Teil 1: Funktionen für die biologische Vielfalt und als Kohlenstoffsenke und -speicher. *Naturschutz und Landschaftsplanung (NuL)* 53, 12–25. <u>https://doi.org/10.1399/NuL.2021.12.01</u>
- Lupi, F., Herriges, J. A., Kim, H., and Stevenson, R. J. (2023). Getting off the ladder: Disentangling water quality indices to enhance the valuation of divergent ecosystem services. *Proc. Natl. Acad. Sci. U.S.A.* 120, e2120261120. <u>https://doi.org/10.1073/pnas.2120261120</u>
- Luyssaert, S., Schulze, E.-D., Börner, A., Knohl, A., Hessenmöller, D., Law, B. E., et al. (2008). Old-growth forests as global carbon sinks. *Nature* 455, 213–215. <u>https://doi.org/10.1038/nature07276</u>
- Maas, J., Verheij, R. A., de Vries, S., Spreeuwenberg, P., Schellevis, F. G., and Groenewegen, P. P. (2009). Morbidity is related to a green living environment. *J Epidemiol Community Health* 63, 967–973. https://doi.org/10.1136/jech.2008.079038
- Madaniyazi, L., Armstrong, B., Chung, Y., Ng, C. F. S., Seposo, X., Kim, Y., et al. (2022). Seasonal variation in mortality and the role of temperature: a multi-country multi-city study. *International Journal of Epidemiology* 51, 122–133. <u>https://doi.org/10.1093/ije/dyab143</u>
- Maghiar, L. M., Stoica, I. A., and Tanentzap, A. J. (2021). Integrating demography and distribution modeling for the iconic Leontopodium alpinum Colm. in the Romanian Carpathians. *Ecology and Evolution* 11, 12322–12334. <u>https://doi.org/10.1002/ ece3.7864</u>
- Maierhofer, D., Van Karsbergen, V., Potrč Obrecht, T., Ruschi Mendes Saade, M., Gingrich, S., Streicher, W., et al. (2024). Linking forest carbon opportunity costs and greenhouse gas emission substitution effects of wooden buildings: The climate optimum concept. Sustainable Production and Consumption 51, 612–627. https://doi.org/10.1016/j.spc.2024.08.021
- Mäkinen, H., Kasev, J., Trnk, M., Balek, J., Kersebaum, K. C., Nendel, C., et al. (2018). Sensitivity of European wheat to extreme weather. *Field Crops Research* 222, 209–217. <u>https://doi. org/10.1016/j.fcr.2017.11.008</u>
- Mameno, K., Kubo, T., Oguma, H., Amagai, Y., and Shoji, Y. (2022). Decline in the alpine landscape aesthetic value in a national

park under climate change. Climatic Change 170, 35. <u>https://doi.</u> org/10.1007/s10584-022-03322-1

- Mamot, P., Weber, S., Eppinger, S., and Krautblatter, M. (2021). A temperature-dependent mechanical model to assess the stability of degrading permafrost rock slopes. *Earth Surface Dynamics* 9, 1125–1151. <u>https://doi.org/10.5194/esurf-9-1125-2021</u>
- Marçais, B., Kosawang, C., Laubray, S., Kjær, E., and Kirisits, T. (2022). "Chapter 13 Ash dieback", in *Forest Microbiology*, eds.
  F. O. Asiegbu, A. Kovalchuk, F. O. Asiegbu, and A. Kovalchuk, 215–237. Available at: <u>https://doi.org/10.1016/B978-0-323-85042-1.00022-7</u> (Accessed May 17, 2024).
- Markevych, I., Schoierer, J., Hartig, T., Chudnovsky, A., Hystad, P., Dzhambov, A. M., et al. (2017). Exploring pathways linking greenspace to health: Theoretical and methodological guidance. *Environmental Research* 158, 301–317. <u>https://doi.org/10.1016/j.envres.2017.06.028</u>
- Maroschek, M., Rammer, W., and Lexer, M. J. (2015). Using a novel assessment framework to evaluate protective functions and timber production in Austrian mountain forests under climate change. *Regional Environmental Change* 15, 1543–1555. <u>https:// doi.org/10.1007/s10113-014-0691-z</u>
- Maroschek, M., Seidl, R., Poschlod, B., and Senf, C. (2024). Quantifying patch size distributions of forest disturbances in protected areas across the European Alps. *Journal of Biogeography* 51, 368–381. https://doi.org/10.1111/jbi.14760
- Marques, A., Martins, I. S., Kastner, T., Plutzar, C., Theurl, M. C., Eisenmenger, N., et al. (2019). Increasing impacts of land use on biodiversity and carbon sequestration driven by population and economic growth'. *Nature Ecology & Evolution* 3, 628–637. <u>https://doi.org/10.1038/s41559-019-0824-3</u>
- Marselle, M. R., Hartig, T., Cox, D. T. C., de Bell, S., Knapp, S., Lindley, S., et al. (2021). Pathways linking biodiversity to human health: A conceptual framework. *Environ Int* 150, 106420. <u>https://doi.org/10.1016/j.envint.2021.106420</u>
- Martens, P., Su, B., and Deblomme, S. (2019). The Ecological Paw Print of Companion Dogs and Cats. *Bioscience* 69, 467–474. https://doi.org/10.1093/biosci/biz044
- Martin, J. G. C., Scolobig, A., Linnerooth-Bayer, J., Liu, W., and Balsiger, J. (2021). Catalyzing Innovation: Governance Enablers of Nature-Based Solutions. *Sustainability* 13, 1971. <u>https://doi.org/10.3390/su13041971</u>
- Martínez-Solanas, È., Quijal-Zamorano, M., Achebak, H., Petrova, D., Robine, J.-M., Herrmann, F. R., et al. (2021). Projections of temperature-attributable mortality in Europe: a time series analysis of 147 contiguous regions in 16 countries. *The Lancet Planetary Health* 5, e446–e454. <u>https://doi.org/10.1016/S2542-5196(21)00150-9</u>
- Masselot, P., Mistry, M., Vanoli, J., Schneider, R., Iungman, T., Garcia-Leon, D., et al. (2023). Excess mortality attributed to heat and cold: a health impact assessment study in 854 cities in Europe. *The Lancet Planetary Health* 7, e271–e281. <u>https://doi.org/10.1016/S2542-5196(23)00023-2</u>
- Matthews, V., Longman, J., Berry, H. L., Passey, M., Bennett-Levy, J., Morgan, G. G., et al. (2019). Differential Mental Health Impact Six Months After Extensive River Flooding in Rural Australia: A Cross-Sectional Analysis Through an Equity Lens. *Front. Public Health* 7. https://doi.org/10.3389/fpubh.2019.00367
- Matulla, C., Schmutz, S., Melcher, A., Gerersdorfer, T., and Haas, P. (2007). Assessing the impact of a downscaled climate change simulation on the fish fauna in an Inner-Alpine River. *Int J*

Biometeorol 52, 127-137. <u>https://doi.org/10.1007/s00484-007-0107-6</u>

- Matuszkiewicz, J. M., Affek, A. N., and Kowalska, A. (2021). Current and potential carbon stock in the forest communities of the Białowieża Biosphere Reserve. *Forest Ecology and Management* 502, 119702. <u>https://doi.org/10.1016/j.foreco.2021.119702</u>
- Mauri, A., Girardello, M., Forzieri, G., Manca, F., Beck, P. S. A., Cescatti, A., et al. (2023). Assisted tree migration can reduce but not avert the decline of forest ecosystem services in Europe. *Global Environmental Change* 80, 102676. <u>https://doi.org/10.1016/j.gloenvcha.2023.102676</u>
- Mayer, A., Kalt, G., Kaufmann, L., Röös, E., Muller, A., Weisshaidinger, R., et al. (2022). Impacts of Scaling up Agroecology on the Sustainability of European Agriculture in 2050. *Euro-Choices* 21, 27–36. <u>https://doi.org/10.1111/1746-692X.12373</u>
- Mehdi-Schulz, B., Zoboli, O., Schürz, C., Strenge, E., Lima, E. M., Parajka, J., et al. (2024). The impacts of climate change on nitrogen losses to the environment in Austria: A dual model analysis across spatial and temporal scales to support policy decisions. *Science of The Total Environment* 918, 170730. <u>https://doi.org/10.1016/j.scitotenv.2024.170730</u>
- Meisner, J. D., Goodier, J. L., Regier, H. A., Shuter, B. J., and Christie, W. J. (1987). An Assessment of the Effects of Climate Warming on Great Lakes Basin Fishes. *Journal of Great Lakes Research* 13, 340–352. <u>https://doi.org/10.1016/S0380-1330(87)71656-6</u>
- Mempel, F., Corbera, E., Labajos, B. R., and Challies, E. (2023). From railroad imperialism to neoliberal reprimarization: Lessons from regime-shifts in the Global Soybean Complex. *Environment and Planning E: Nature and Space* 7, 559–582. <u>https://doi.org/10.1177/25148486231201216</u>
- Mergili, M., and Duffy, R. (2022). "Fluvial Geomorphology and River Restoration: Tiroler Lech Nature Park", in *Landscapes* and Landforms of Austria, ed. C. Embleton-Hamann (Cham: Springer), 253–264. Available at: <u>https://doi.org/10.1007/978-3-</u> 030-92815-5\_16 (Accessed April 21, 2024).
- Methorst, J., Rehdanz, K., Mueller, T., Hansjürgens, B., Bonn, A., and Böhning-Gaese, K. (2021). The importance of species diversity for human well-being in Europe. *Ecological Economics* 181, 106917. <u>https://doi.org/10.1016/j.ecolecon.2020.106917</u>
- Metze, D., Schnecker, J., Canarini, A., Fuchslueger, L., Koch, B. J., Stone, B. W., et al. (2023). Microbial growth under drought is confined to distinct taxa and modified by potential future climate conditions. *Nat Commun* 14, 5895. <u>https://doi.org/10. 1038/s41467-023-41524-y</u>
- Meyfroidt, P., de Bremond, A., Ryan, C. M., Archer, E., Aspinall, R., Chhabra, A., et al. (2022). Ten facts about land systems for sustainability. *PNAS* 119, e2109217118. <u>https://doi.org/10.1073/ pnas.2109217118</u>
- Meyfroidt, P., and Lambin, E. F. (2009). Forest transition in Vietnam and displacement of deforestation abroad. *Proceedings of* the National Academy of Sciences 106, 16139–16144. <u>https://doi.org/10.1073/pnas.0904942106</u>
- Meynard, J.-M., Jeuffroy, M.-H., Le Bail, M., Lefèvre, A., Magrini, M.-B., and Michon, C. (2017). Designing coupled innovations for the sustainability transition of agrifood systems. *Agricultur*al Systems 157, 330–339. <u>https://doi.org/10.1016/j.agsy.2016.</u> 08.002
- Mikovits, C., Zollitsch, W., Hortenhuber, S. J., Baumgartner, J., Niebuhr, K., Piringer, M., et al. (2019). Impacts of global warming on confined livestock systems for growing-fattening pigs: simulation of heat stress for 1981 to 2017 in Central Europe.

Int J Biometeorol 63, 221–230. <u>https://doi.org/10.1007/s00484-018-01655-0</u>

- Miles-Novelo, A., and Anderson, C. A. (2019). Climate Change and Psychology: Effects of Rapid Global Warming on Violence and Aggression. *Curr Clim Change Rep* 5, 36–46. <u>https://doi. org/10.1007/s40641-019-00121-2</u>
- Minachilis, K., Kougioumoutzis, K., and Petanidou, T. (2021). Climate change effects on multi-taxa pollinator diversity and distribution along the elevation gradient of Mount Olympus, Greece. *Ecological Indicators* 132, 108335. <u>https://doi.org/10.1016/j.ecolind.2021.108335</u>
- Mirtl, M., Bahn, M., Battin, T., Borsdorf, A., Dirnböck, T., Englisch, M., et al. (2015). Research for the Future – LTER-Austria White Paper 2015 – On the status and orientation of process oriented ecosystem research, biodiversity and conservation research and socio-ecological research in Austria. Austrian Society for Lont-term Ecological Research. Available at: <u>https://www.lteraustria.at/wp\_lter/wp-content/uploads/2015/03/LTER\_Austria\_WhitePaper\_2015\_en.pdf</u>
- Mishra, A., Humpenöder, F., Churkina, G., Reyer, C. P. O., Beier, F., Bodirsky, B. L., et al. (2022). Land use change and carbon emissions of a transformation to timber cities. *Nat Commun* 13, 4889. <u>https://doi.org/10.1038/s41467-022-32244-w</u>
- Mitter, H., Larcher, M., Schönhart, M., Stöttinger, M., and Schmid, E. (2019). Exploring Farmers' Climate Change Perceptions and Adaptation Intentions: Empirical Evidence from Austria. *Environmental Management* 63, 804–821. <u>https://doi.org/10.1007/</u> <u>s00267-019-01158-7</u>
- Mitter, H., and Schmid, E. (2019). Computing the economic value of climate information for water stress management exemplified by crop production in Austria. Agricultural Water Management 221, 430–448. https://doi.org/10.1016/j.agwat.2019.04.005
- Mitter, H., Schmid, E., and Schneider, U. A. (2015a). Modelling impacts of drought and adaptation scenarios on crop production in Austria. *Journal of the Austrian Society of Agricultural Economics* 24, 223–232.
- Mitter, H., Schönhart, M., Meyer, I., Mechtler, C., Schmid, E., Sinabell, F., et al. (2015b). "Agriculture", in *Economic Evaluation of Climate change impacts Development of a cross-sectoral framework and results for Austria*, eds. K. W. Steininger, M. König, B. Bednar-Friedl, L. Kranzl, W. Loibl, and F. Prettenthaler (Cham, Switzerland: Springer).
- Mo, L., Zohner, C. M., Reich, P. B., Liang, J., De Miguel, S., Nabuurs, G.-J., et al. (2023). Integrated global assessment of the natural forest carbon potential. *Nature* 624, 92–101. <u>https://doi.org/10.1038/s41586-023-06723-z</u>
- Moran, D., and Blair, K. J. (2021). Review: Sustainable livestock systems: anticipating demand-side challenges. *Animal* 15, 100288. <u>https://doi.org/10.1016/j.animal.2021.100288</u>
- Moranta, J., Torres, C., Murray, I., Hidalgo, M., Hinz, H., and Gouraguine, A. (2022). Transcending capitalism growth strategies for biodiversity conservation. *Conservation Biology* 36, e13821. <u>https://doi.org/10.1111/cobi.13821</u>
- Moriondo, M., Bindi, M., Fagarazzi, C., Ferisse, R., and Trombi, G. (2010). Framework for high-resolution climate change impact assessment on grapevines at a regional scale. *Regional environmental change* 11, 553–567. <u>https://doi.org/10.1007/s10113-</u> 010-0171-z
- Moshammer, H. (2023). Akute Wirkungen von Luftschadstoffen auf den Menschen. Umweltmed – Hygiene – Arbeitsmed, 137–141.

- Moshammer, H., Hutter, H.-P., and Kundi, M. (2013). Which metric of ambient ozone to predict daily mortality? *Atmospheric Environment* 65, 171–176. <u>https://doi.org/10.1016/j.atmosenv.</u> 2012.10.032
- Moshammer, H., Panholzer, J., Ulbing, L., Udvarhelyi, E., Ebenbauer, B., and Peter, S. (2019). Acute Effects of Air Pollution and Noise from Road Traffic in a Panel of Young Healthy Adults. *International Journal of Environmental Research and Public Health* 16, 788. <u>https://doi.org/10.3390/ijerph16050788</u>
- Moshammer, H., Poteser, M., Kundi, M., Lemmerer, K., Weitensfelder, L., Wallner, P., et al. (2020). Nitrogen-Dioxide Remains a Valid Air Quality Indicator. *International Journal of Environmental Research and Public Health* 17, 3733. <u>https://doi.org/ 10.3390/ijerph17103733</u>
- Möth, S., Redl, M., Winter, S., Hüttner, F., and Steinkellner, S. (2023). Efficiency of inorganic fungicides against the formation of Erysiphe necator chasmothecia in vineyards. *Pest Manag Sci* 79, 3080–3089. <u>https://doi.org/10.1002/ps.7487</u>
- Mucioki, M., Sowerwine, J., Sarna-Wojcicki, D., Lake, F. K., and Bourque, S. (2021). Conceptualizing Indigenous Cultural Ecosystem Services (ICES) and Benefits under Changing Climate Conditions in the Klamath River Basin and Their Implications for Land Management and Governance'. *Journal of Ethnobiology* 41, 313–330. <u>https://doi.org/10.2993/0278-0771-41.3.313</u>
- Müller, M. (2011). How natural disturbance triggers political conflict: Bark beetles and the meaning of landscape in the Bavarian Forest. *Global Environmental Change* 21, 935–946. <u>https://doi.org/10.1016/j.gloenvcha.2011.05.004</u>
- Müller, M. M., Vilardell, L. V., Vacik, H., Mayer, C., Mayr, S., Carrega, P., et al. (2020). Forest fires in the Alps State of knowledge, future challenges and options for an integrated fire management White Paper for policy makers. <u>https://doi.org/10.13140/RG.2.2.15609.42081</u>
- Nature Food (2024). Not just farmers' protests. *Nat Food* 5, 93. https://doi.org/10.1038/s43016-024-00939-6
- Naumann, G., Spinoni, J., Vogt, J., and Barbosa, P. (2015). Assessment of drought damages and their uncertainties in Europe. *Environ. Res. Lett.* 10, 124013. <u>https://doi.org/10.1088/1748-9326/10/12/124013</u>
- Neria, Y., Nandi, A., and Galea, S. (2008). Post-traumatic stress disorder following disasters: a systematic review. *Psychological Medicine* 38, 467–480. <u>https://doi.org/10.1017/S0033291707001353</u>
- Neuberger, M., Moshammer, H., and Rabczenko, D. (2013). Acute and Subacute Effects of Urban Air Pollution on Cardiopulmonary Emergencies and Mortality: Time Series Studies in Austrian Cities. *International Journal of Environmental Research* and Public Health 10, 4728–4751. <u>https://doi.org/10.3390/</u> ijerph10104728
- Neugschwandtner, R. W., Bernhuber, A., Kammlander, S., Wagentristl, H., Klimek-Kopyra, A., and Kaul, H.-P. (2020). Yield structure components of autumn- and spring-sown pea (Pisum sativum L.). Acta Agriculturae Scandinavica, Section B — Soil & Plant Science 70, 109–116. <u>https://doi.org/10.1080/09064710.</u> 2019.1676463
- Neugschwandtner, R. W., Bernhuber, A., Kammlander, S., Wagentristl, H., Klimek-Kopyra, A., Lošák, T., et al. (2023a). Effect of Two Seeding Rates on Nitrogen Yield and Nitrogen Fixation of Winter and Spring Faba Bean. *Plants* 12, 1711. <u>https://doi.org/ 10.3390/plants12081711</u>
- Neugschwandtner, R. W., Dobos, G., Wagentristl, H., Lošák, T., Klimek-Kopyra, A., and Kaul, H.-P. (2023b). Yield and Yield

Components of Winter Poppy (Papaver somniferum L.) Are Affected by Sowing Date and Sowing Rate under Pannonian Climate Conditions. *Agriculture* 13, 997. <u>https://doi.org/10.3390/</u>agriculture13050997

- Neugschwandtner, R. W., Wichmann, S., Gimplinger, D. M., Wagentristl, H., and Kaul, H.-P. (2013). Chickpea performance compared to pea, barley and oat in Central Europe: Growth analysis and yield. *Turkish Journal Of Field Crops* 18, 179–184.
- Neugschwandtner, R. W., Ziegler, K. V., Kriegner, S., Wagentristl, H., and Kaul, H.-P. (2015). Nitrogen yield and nitrogen fixation of winter faba beans. ACTA AGR SCAND B-S P. 65, 658–666. <u>https://doi.org/10.1080/09064710.2015.1042028</u>
- Neuwirth, C., and Hofer, B. (2013). Spatial sensitivity of grassland yields to weather variations in Austria and its implications for the future. *Applied Geography* 45, 332–341. <u>https://doi.org/10.1016/j.apgeog.2013.08.010</u>
- Newig, J., Gaube, V., Berkhoff, K., Kaldrack, K., Kastens, B., Lutz, J., et al. (2008). The Role of Formalisation, Participation and Context in the Success of Public Involvement Mechanisms in Resource Management. Syst Pract Action Res 21, 423–441. <u>https:// doi.org/10.1007/s11213-008-9113-9</u>
- Niedrist, G. H. (2023). Substantial warming of Central European mountain rivers under climate change. *Reg Environ Change* 23, 43. <u>https://doi.org/10.1007/s10113-023-02037-y</u>
- Nijdam, D., Rood, T., and Westhoek, H. (2012). The price of protein: Review of land use and carbon footprints from life cycle assessments of animal food products and their substitutes. *Food Policy* 37, 760–770. <u>https://doi.org/10.1016/j.foodpol.2012.08.002</u>
- Nilsson, C., Aradottir, A. L., Hagen, D., Halldórsson, G., Høegh, K., Mitchell, R. J., et al. (2016). Evaluating the process of ecological restoration. *E&S* 21, 41. <u>https://doi.org/10.5751/ES-08289-210141</u>
- Noack, F. A. W., Manthey, M., Ruitenbeek, J. H., and Mohadjer, M. R. M. (2010). Separate or mixed production of timber, livestock and biodiversity in the Caspian Forest. *Ecological Economics* 70, 67–76. <u>https://doi.org/10.1016/j.ecolecon.2010.07.033</u>
- Noguerol, A. T., Pagán, M. J., García-Segovia, P., and Varela, P. (2021). Green or clean? Perception of clean label plant-based products by omnivorous, vegan, vegetarian and flexitarian consumers. *Food Research International* 149, 110652. <u>https://doi. org/10.1016/j.foodres.2021.110652</u>
- Norton, M., Baldi, A., Buda, V., Carli, B., Cudlin, P., Jones, M. B., et al. (2019). Serious mismatches continue between science and policy in forest bioenergy. *GCB Bioenergy* 11, 1256–1263. <u>https://doi.org/10.1111/gcbb.12643</u>
- Norton, M., Walloe, L., Brack, D., Booth, M., and Jones, M. B. (2022). Time is of the essence when it comes to forest bioenergy. GCB Bioenergy 14, 108–109. <u>https://doi.org/10.1111/gcbb. 12905</u>
- Nowak, D. J., and Crane, D. E. (2002). Carbon storage and sequestration by urban trees in the USA. *Environmental Pollution* 116, 381–389. <u>https://doi.org/10.1016/S0269-7491(01)00214-7</u>
- Nowak-Olejnik, A., Schirpke, U., and Tappeiner, U. (2022). A systematic review on subjective well-being benefits associated with cultural ecosystem services. *Ecosystem Services* 57, 101467. https://doi.org/10.1016/j.ecoser.2022.101467
- Obojes, N., Meurer, A. K., Newesely, C., Tasser, E., Oberhuber, W., Mayr, S., et al. (2022). Swiss stone pine growth benefits less from recent warming than European larch at a dry-inner alpine forest line as it reacts more sensitive to humidity. *Agricultural*

and Forest Meteorology 315, 108788. <u>https://doi.org/10.1016/j.</u> agrformet.2021.108788

- Obojes, N., Meurer, A., Newesely, C., Tasser, E., Oberhuber, W., Mayr, S., et al. (2018). Water stress limits transpiration and growth of European larch up to the lower subalpine belt in an inner-alpine dry valley. *New Phytologist* 220, 460–475. <u>https:// doi.org/10.1111/nph.15348</u>
- Obojes, N., Tasser, E., Newesely, C., Mayr, S., and Tappeiner, U. (2020). Comparing sap flow of European larch with evergreen conifers at different elevations in an inner-alpine dry valley. *Acta Hortic.*, 113–120. <u>https://doi.org/10.17660/ActaHortic.</u> <u>2020.1300.15</u>
- Oettel, J., and Lapin, K. (2021). Linking forest management and biodiversity indicators to strengthen sustainable forest management in Europe. *Ecological Indicators* 122, 107275. <u>https://doi.org/10.1016/j.ecolind.2020.107275</u>
- Oettel, J., Lapin, K., Kindermann, G., Steiner, H., Schweinzer, K.-M., Frank, G., et al. (2020). Patterns and drivers of deadwood volume and composition in different forest types of the Austrian natural forest reserves. *Forest Ecology and Management* 463, 118016. https://doi.org/10.1016/j.foreco.2020.118016
- Oettl, D., Ferrero, E., Moshammer, H., Weitensfelder, L., Kropsch, M., and Mandl, M. (2022). Recent developments in odour modelling and assessment in five provinces in Austria. *Air Qual Atmos Health* 15, 1647–1657. <u>https://doi.org/10.1007/s11869-022-</u> 01207-x
- O'Faircheallaigh, C. (2010). Public participation and environmental impact assessment: Purposes, implications, and lessons for public policy making. *Environmental Impact Assessment Review* 30, 19–27. <u>https://doi.org/10.1016/j.eiar.2009.05.001</u>
- Ohly, H., White, M. P., Wheeler, B. W., Bethel, A., Ukoumunne, O. C., Nikolaou, V., et al. (2016). Attention Restoration Theory: A systematic review of the attention restoration potential of exposure to natural environments. *Journal of Toxicology and Environmental Health, Part B* 19, 305–343. <u>https://doi.org/10.1080/10937404.2016.1196155</u>
- Okin, G. S. (2017). Environmental impacts of food consumption by dogs and cats. *PLoS ONE* 12, e0181301. <u>https://doi.org/10.1371/journal.pone.0181301</u>
- Olsson, L. H., Barbosa, H., Bhadwal, S., Cowie, A., Delusca, K., Flores-Renteria, D., et al. (2019). "Land degradation", in *Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems,* eds. P. R. Shukla, J. Skea, E. Calvo Buendia, V. Masson-Delmotte, H.-O. Pörtner, D. C. Roberts, et al., 77–129. Available at: <u>https://www.ipcc.ch/site/assets/uploads/sites/4/2019/11/07</u> <u>Chapter-4.pdf</u> (Accessed January 31, 2020).
- Oral, H. V., Carvalho, P., Gajewska, M., Ursino, N., Masi, F., Hullebusch, E. D. V., et al. (2020). A review of nature-based solutions for urban water management in European circular cities: a critical assessment based on case studies and literature. *Blue-Green Systems* 2, 112–136. <u>https://doi.org/10.2166/bgs.2020.932</u>
- Orság, M., Meitner, J., Fischer, M., Svobodová, E., Kopp, R., Mareš, J., et al. (2023). Estimating Heat Stress Effects on the Sustainability of Traditional Freshwater Pond Fishery Systems under Climate Change. Water 15, 1523. <u>https://doi.org/10.3390/</u> w15081523
- Osterburg, B., Kätsch, S., and Wolff, A. (2013). Szenarioanalysen zur Minderung von Treibhausgasemissionen der deutschen Landwirtschaft im Jahr 2050. Braunschweig: Johann Heinrich

von Thünen-Institut. Available at: <u>https://www.econstor.eu/</u> handle/10419/90149 (Accessed September 20, 2023).

- Österreichischer Rechnungshof (2013). Bericht des Rechnungshofes. Agrarumweltprogramm ÖPUL 2007. Österreichischer Rechnungshof. Available at: <u>https://www.rechnungshof.gv.at/</u> <u>rh/home/Agrarumweltprogramm\_OEPUL\_2007.pdf</u>
- Österreichischer Rechnungshof (2016). Bericht des Rechnungshof. Agrarumweltprogramm ÖPUL 2007; Follow-up-Überprüfung. Österreichischer Rechnungshof. Available at: <u>https://www. rechnungshof.gv.at/rh/home/home/home\_7/Agrarumweltprogramm\_OePUL\_2007\_Follow\_up\_Ueberpruefung\_1.pdf</u>
- Otero, I., Farrell, K. N., Pueyo, S., Kallis, G., Kehoe, L., Haberl, H., et al. (2020). Biodiversity policy beyond economic growth. *Con*servation Letters 13, e12713. <u>https://doi.org/10.1111/conl.12713</u>
- Ovando, P., Beguería, S., and Campos, P. (2019). Carbon sequestration or water yield? The effect of payments for ecosystem services on forest management decisions in Mediterranean forests. Water Resources and Economics 28, 100119. <u>https://doi.org/10.1016/j.wre.2018.04.002</u>
- Parfitt, J., Barthel, M., and Macnaughton, S. (2010). Food waste within food supply chains: quantification and potential for change to 2050. *Philosophical Transactions of the Royal Society B: Biological Sciences* 365, 3065–3081. <u>https://doi.org/10.1098/</u> rstb.2010.0126
- Patacca, M., Lindner, M., Lucas-Borja, M. E., Cordonnier, T., Fidej, G., Gardiner, B., et al. (2023). Significant increase in natural disturbance impacts on European forests since 1950. *Global Change Biology* 29, 1359–1376. <u>https://doi.org/10.1111/gcb.16531</u>
- Paul, C., Techen, A.-K., Robinson, J. S., and Helming, K. (2019). Rebound effects in agricultural land and soil management: Review and analytical framework. *Journal of Cleaner Production* 227, 1054–1067. https://doi.org/10.1016/j.jclepro.2019.04.115
- Pecher, C., Marsoner, T., and Tasser, E. (2024). Regional food self-sufficiency potential in the European Alpine space. *Sci Rep* 14, 9527. <u>https://doi.org/10.1038/s41598-024-60010-z</u>
- Pedrinelli, V., Teixeira, F. A., Queiroz, M. R., and Brunetto, M. A. (2022). Environmental impact of diets for dogs and cats. *Sci. Rep.* 12, 18510. <u>https://doi.org/10.1038/s41598-022-22631-0</u>
- Pendrill, F., Gardner, T. A., Meyfroidt, P., Persson, U. M., Adams, J., Azevedo, T., et al. (2022). Disentangling the numbers behind agriculture-driven tropical deforestation. *Science* 377, eabm9267. <u>https://doi.org/10.1126/science.abm9267</u>
- Pendrill, F., Persson, U. M., Godar, J., Kastner, T., Moran, D., Schmidt, S., et al. (2019). Agricultural and forestry trade drives large share of tropical deforestation emissions. *Global Environmental Change* 56, 1–10. <u>https://doi.org/10.1016/j.gloenvcha.</u> 2019.03.002
- Peng, L., Searchinger, T. D., Zionts, J., and Waite, R. (2023). The carbon costs of global wood harvests. *Nature* 620, 110–115. <u>https://doi.org/10.1038/s41586-023-06187-1</u>
- Penker, M., Brunner, K.-M., and Plank, C. (2023). "Kapitel 5. Ernährung", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 245–269. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_9</u> (Accessed October 4, 2023).
- Penne, T., and Goedemé, T. (2021). Can low-income households afford a healthy diet? Insufficient income as a driver of food insecurity in Europe. *Food Policy* 99, 101978. <u>https://doi.org/10.1016/j.foodpol.2020.101978</u>

- Pereira, L. M., Davies, K. K., Den Belder, E., Ferrier, S., Karlsson-Vinkhuyzen, S., Kim, H., et al. (2020). Developing multiscale and integrative nature–people scenarios using the Nature Futures Framework. *People and Nature* 2, 1172–1195. <u>https:// doi.org/10.1002/pan3.10146</u>
- Peron, A., Kaser, L., Fitzky, A. C., Graus, M., Halbwirth, H., Greiner, J., et al. (2021). Combined effects of ozone and drought stress on the emission of biogenic volatile organic compounds from Quercus robur L. *Biogeosciences* 18, 535–556. <u>https://doi. org/10.5194/bg-18-535-2021</u>
- Pichler, M., Schmid, M., and Gingrich, S. (2022). Mechanisms to exclude local people from forests: Shifting power relations in forest transitions. *Ambio* 51, 849–862. <u>https://doi.org/10.1007/ s13280-021-01613-y</u>
- Pierer, M., Amon, B., and Winiwarter, W. (2016). Adapting feeding methods for less nitrogen pollution from pig and dairy cattle farming: abatement costs and uncertainties. *Nutr Cycl Agroeco*syst 104, 201–220. <u>https://doi.org/10.1007/s10705-016-9767-0</u>
- Pingoud, K., Ekholm, T., Sievänen, R., Huuskonen, S., and Hynynen, J. (2018). Trade-offs between forest carbon stocks and harvests in a steady state – A multi-criteria analysis. *Journal* of Environmental Management 210, 96–103. <u>https://doi.org/10. 1016/j.jenvman.2017.12.076</u>
- Pinior, B., Firth, C. L., Loitsch, A., Stockreiter, S., Hutter, S., Richter, V., et al. (2018). Cost distribution of bluetongue surveillance and vaccination programmes in Austria and Switzerland (2007– 2016). VetRecord 182, 257. https://doi.org/10.1136/vr.104448
- Plank, C., Haas, W., Schreuer, A., Irshaid, J., Barben, D., and Görg, C. (2021). Climate policy integration viewed through the stakeholders' eyes: A co-production of knowledge in social-ecological transformation research. *Environmental Policy and Governance* 31, 387–399. <u>https://doi.org/10.1002/eet.1938</u>
- Pletterbauer, F., Melcher, A. H., Ferreira, T., and Schmutz, S. (2015). Impact of climate change on the structure of fish assemblages in European rivers. *Hydrobiologia* 744, 235–254. <u>https://doi. org/10.1007/s10750-014-2079-y</u>
- Poeppl, W., Obwaller, A. G., Weiler, M., Burgmann, H., Mooseder, G., Lorentz, S., et al. (2013). Emergence of sandflies (Phlebotominae) in Austria, a Central European country. *Parasitol Res* 112, 4231–4237. https://doi.org/10.1007/s00436-013-3615-9
- Poore, J., and Nemecek, T. (2018). Reducing food's environmental impacts through producers and consumers. *Science* 360, 987– 992. <u>https://doi.org/10.1126/science.aaq0216</u>
- Pötzelsberger, E., Spiecker, H., Neophytou, C., Mohren, F., Gazda, A., and Hasenauer, H. (2020). Growing Non-native Trees in European Forests Brings Benefits and Opportunities but Also Has Its Risks and Limits. *Curr Forestry Rep* 6, 339–353. <u>https://doi. org/10.1007/s40725-020-00129-0</u>
- Preinfalk, E., Bednar-Friedl, B., Mayer, J., Lauk, C., and Mayer, A. (2024). Sustainability transitions in the agri-food system: Evaluating mitigation potentials, economy-wide effects, co-benefits and trade-offs for the case of Austria. *Ecological Economics* 226, 108357. <u>https://doi.org/10.1016/j.ecolecon.2024.108357</u>
- Pröbstl-Haider, U., Dabrowska, K., and Haider, W. (2016). Risk perception and preferences of mountain tourists in light of glacial retreat and permafrost degradation in the Austrian Alps. *Journal of Outdoor Recreation and Tourism* 13, 66–78. <u>https:// doi.org/10.1016/j.jort.2016.02.002</u>
- Pröbstl-Haider, U., Hödl, C., Ginner, K., and Borgwardt, F. (2021). Climate change: Impacts on outdoor activities in the summer

and shoulder seasons. Journal of Outdoor Recreation and Tourism 34, 100344. https://doi.org/10.1016/j.jort.2020.100344

- Pröbstl-Haider, U., Kelemen-Finan, J., Haider, W., Schauppenlehner, T., Melzer, V., and Mostegl, N. (2015). "Will Climate Change Influence the Attractiveness of Cultural Landscapes in Austria?", in *Tourism and Leisure*, eds. H. Pechlaner, E. Smeral, H. Pechlaner, E. Smeral, H. Pechlaner, E. Smeral, et al. (Wiesbaden: Springer Gabler), 355–370. <u>https://doi.org/10.1007/978-3-658-06660-4\_22</u>
- Prutsch, A., Steurer, R., and Stickler, T. (2018). Is the participatory formulation of policy strategies worth the effort? The case of climate change adaptation in Austria. *Reg Environ Change* 18, 271–285. <u>https://doi.org/10.1007/s10113-017-1204-7</u>
- Purse, B. V., Mellor, P. S., Rogers, D. J., Samuel, A. R., Mertens, P. P. C., and Baylis, M. (2005). Climate change and the recent emergence of bluetongue in Europe. *Nat Rev Microbiol* 3, 171–181. https://doi.org/10.1038/nrmicro1090
- Quaranta, E., Dorati, C., and Pistocchi, A. (2021). Water, energy and climate benefits of urban greening throughout Europe under different climatic scenarios. *Sci Rep* 11, 12163. <u>https://doi.org/10.1038/s41598-021-88141-7</u>
- Quddoos, A., Salhofer, K., and Morawetz, U. B. (2023). Utilising farm-level panel data to estimate climate change impacts and adaptation potentials. J Agricultural Economics 74, 75–99. <u>https://</u> doi.org/10.1111/1477-9552.12490
- Rabitsch, W., Milasowszky, N., Nehring, S., Wiesner, C., Wolter, C., and Essl, F. (2013). The times are changing: temporal shifts in patterns of fish invasions in central European fresh waters. *Journal of Fish Biology* 82, 17–33. <u>https://doi.org/10.1111/j.1095-8649.2012.03457.x</u>
- Raile, P., and Rieken, B. (2021). Eco Anxiety die Angst vor dem Klimawandel: Psychotherapiewissenschaftliche und ethnologische Zugänge. Münster: Waxmann. Available at: <u>https://www.waxmann.com/buecher/?tx\_p2waxmann\_buchliste%5bbuchnr%</u> 5d=4372&ttx\_p2waxmann\_buchliste%5baction%5d=show
- Rajczak, J., and Schär, C. (2017). Projections of future precipitation extremes over Europe: A multimodel assessment of climate simulations. *Journal of Geophysical Research: Atmospheres* 122, 10773–10800. https://doi.org/10.1002/2017JD027176
- Rao, M., Afshin, A., Singh, G., and Mozaffarian, D. (2013). Do healthier foods and diet patterns cost more than less healthy options? A systematic review and meta-analysis. *BMJ Open* 3, e004277. https://doi.org/10.1136/bmjopen-2013-004277
- Rauchenwald, V., Rollins, M. D., Ryan, S. M., Voronov, A., Feiner, J. R., Šarka, K., et al. (2020). New Method of Destroying Waste Anesthetic Gases Using Gas-Phase Photochemistry. *Anesthesia & Analgesia* 131, 288. <u>https://doi.org/10.1213/ANE.</u> 000000000004119
- Ravensbergen, E. A., Waterlander, W. E., Kroeze, W., and Steenhuis, I. H. (2015). Healthy or Unhealthy on Sale? A cross-sectional study on the proportion of healthy and unhealthy foods promoted through flyer advertising by supermarkets in the Netherlands. *BMC Public Health* 15, 470. <u>https://doi.org/10.1186/ s12889-015-1748-8</u>
- Rechnungshof Österreich (2023). Gesundheitsförderung und Prävention. Bericht des Rechnungshofes. Reihe BUND 2023/1. Vienna: Rechnungshof Österreich. Available at: <u>https://www. rechnungshof.gv.at/rh/home/home/2023-1\_Gesundheitsfoerderung.pdf</u>
- Redl, M., Sitavanc, L., and Spangl, B. (2021). Potential and actual ascospore release of Erysiphe necator chasmothecia in Austria.

J Plant Dis Prot 128, 239-248. <u>https://doi.org/10.1007/s41348-020-00376-0</u>

- Rehnus, M., Bollmann, K., Schmatz, D. R., Hackländer, K., and Braunisch, V. (2018). Alpine glacial relict species losing out to climate change: The case of the fragmented mountain hare population (Lepus timidus) in the Alps. *Global Change Biology* 24, 3236–3253. <u>https://doi.org/10.1111/gcb.14087</u>
- Reid, W. V., Mooney, H. A., Cropper, A., Capistrano, D., Carpenter, S. R., Chopra, K., et al. (2005). *Ecosystems and human well-being Synthesis: A Report of the Millennium Ecosystem Assessment.*, ed.Millennium Ecosystem Assessment. Washington D.C.: Island Press. Available at: <u>https://library.wur.nl/WebQuery/wurpubs/340442</u> (Accessed September 19, 2022).
- Reimoser, F., and Reimoser, S. (2016). Long-term trends of hunting bags and wildlife populations in Central Europe. *Beiträge zur Jagd- und Wildforschung* 41, 29–43.
- Reiner, R., Zedrosser, A., Zeiler, H., Hackländer, K., and Corlatti, L. (2022). Habitat and climate shape growth patterns in a mountain ungulate. *Ecology and Evolution* 12, e8650. <u>https:// doi.org/10.1002/ece3.8650</u>
- Reist, J. D., Wrona, F. J., Prowse, T. D., Power, M., Dempson, J. B., Beamish, R. J., et al. (2006). General Effects of Climate Change on Arctic Fishes and Fish Populations. *ambi* 35, 370–380. <u>https://doi.org/10.1579/0044-7447(2006)35[370:GEOCCO]2.0.</u> <u>CO;2</u>
- Renn, O. (2006). Participatory processes for designing environmental policies. Land Use Policy 23, 34–43. <u>https://doi.org/10. 1016/j.landusepol.2004.08.005</u>
- Repo, A., Böttcher, H., Kindermann, G., and Liski, J. (2015). Sustainability of forest bioenergy in Europe: land-use-related carbon dioxide emissions of forest harvest residues. *GCB Bioenergy* 7, 877–887. <u>https://doi.org/10.1111/gcbb.12179</u>
- Revermann, R., Schmid, H., Zbinden, N., Spaar, R., and Schröder, B. (2012). Habitat at the mountain tops: how long can Rock Ptarmigan (Lagopus muta helvetica) survive rapid climate change in the Swiss Alps? A multi-scale approach. J Ornithol 153, 891–905. <u>https://doi.org/10.1007/s10336-012-0819-1</u>
- Rézouki, C., Tafani, M., Cohas, A., Loison, A., Gaillard, J. M., Allainé, D., et al. (2016). Socially mediated effects of climate change decrease survival of hibernating Alpine marmots. *Journal of Animal Ecology* 85, 761–773. <u>https://doi.org/10.1111/1365-2656.12507</u>
- Richter, H., Weixler, S., and Schuster, M. (2020). Der CO2 -Fußabdruck der Anästhesie. Wie die Wahl volatiler Anästhetika die CO2 -Emissionen einer anästhesiologischen Klinik beeinflusst. *Anästh Intensivmed*. <u>https://doi.org/10.19224/ai2020.154</u>
- Rickli, C., Graf, F., Gerber, W., Frei, M., and Böll, A. (2004). Der Wald und seine Bedeutung bei Naturgefahren geologischen Ursprungs. Forum für Wissen. Available at: <u>https://www.dora.</u> <u>lib4ri.ch/wsl/islandora/object/wsl%3A13860/datastream/PDF/</u> <u>download/Rickli-2004-Der\_Wald\_und\_seine\_Bedeutung-%</u> <u>28published\_version%29.pdf</u>
- Rieder, H. E., Bednar-Friedl, B., Moshammer, H., Mayer, M., Knittel, N., Preinfalk, E., et al. (2023). Evaluating the effects of climate warming and precursor emission changes on the attainment of the Austrian ozone standard. ACRP. Available at: <u>https://www.klimafonds.gv.at/wp-content/uploads/sites/16/</u> B960252-ACRP11-ATtain-O3-KR18AC0K14686-EB.pdf
- Rieger, J., Freund, F., Offermann, F., Geibel, I., and Gocht, A. (2023). From fork to farm: Impacts of more sustainable diets in

the EU-27 on the agricultural sector. Journal of Agricultural Economics 74, 764–784. https://doi.org/10.1111/1477-9552.12530

- Riegler-Nurscher, P., Schmidbauer, L., Eder, E., and Treiblmeier, M. (2023). Method for detecting datura in high-resolution UAV images of soybean fields. WSRC 11. Available at: <u>https://www. legumehub.eu/wsrc11-2023/</u>
- Rigo, F., Paniccia, C., Anderle, M., Chianucci, F., Obojes, N., Tappeiner, U., et al. (2024). Relating forest structural characteristics to bat and bird diversity in the Italian Alps. *Forest Ecology* and Management 554, 121673. <u>https://doi.org/10.1016/j.foreco.2023.121673</u>
- Roberts, S., Potts, S., Biesmeijer, K., Kuhlmann, M., Kunin, W., and Ohlemüller, R. (2011). Assessing continental-scale risks for generalist and specialist pollinating bee species under climate change. *BioRisk* 6, 1–18. https://doi.org/10.3897/biorisk.6.1325
- Rocklöv, J., and Dubrow, R. (2020). Author Correction: Climate change: an enduring challenge for vector-borne disease prevention and control. *Nat Immunol* 21, 695. <u>https://doi.org/10.1038/</u> <u>s41590-020-0692-7</u>
- Rockström, J., Beringer, T., Hole, D., Griscom, B., Mascia, M. B., Folke, C., et al. (2021). We need biosphere stewardship that protects carbon sinks and builds resilience. *Proceedings of the National Academy of Sciences* 118, e2115218118. <u>https://doi.org/10.1073/pnas.2115218118</u>
- Roe, S., Streck, C., Obersteiner, M., Frank, S., Griscom, B., Drouet, L., et al. (2019). Contribution of the land sector to a 1.5 °C world. *Nature Climate Change* 9, 817–828. <u>https://doi.org/10.1038/</u> s41558-019-0591-9
- Roebroek, C. T. J., Duveiller, G., Seneviratne, S. I., Davin, E. L., and Cescatti, A. (2023). Releasing global forests from human management: How much more carbon could be stored? *Science* 380, 749–753. <u>https://doi.org/10.1126/science.add5878</u>
- Rollin, O., Aguirre-Gutiérrez, J., Yasrebi-de Kom, I. A. R., Garratt, M. P. D., de Groot, G. A., Kleijn, D., et al. (2022). Effects of ozone air pollution on crop pollinators and pollination. *Global Environmental Change* 75, 102529. <u>https://doi.org/10.1016/j.gloenvcha.2022.102529</u>
- Röös, E., Mayer, A., Muller, A., Kalt, G., Ferguson, S., Erb, K.-H., et al. (2022). Agroecological practices in combination with healthy diets can help meet EU food system policy targets. *Science of The Total Environment* 847, 157612. <u>https://doi.org/10.1016/j.scitotenv.2022.157612</u>
- Rössler, G. (2014). Wuchsleistungsvergleiche zwischen Vor- und Folgebeständen langjähriger Fichten-Dauerversuchsflächen. Dissertation – Institut für Waldwachstum (WAFO). Vienna: BOKU-Universitätfür Bodenkultur. Available at: <u>https://zidapps. boku.ac.at/abstracts/download.php?dataset\_id=12608& property\_id=107</u>
- Roux, N., Kaufmann, L., Bhan, M., Le Noe, J., Matej, S., Laroche, P., et al. (2022). Embodied HANPP of feed and animal products: Tracing pressure on ecosystems along trilateral livestock supply chains 1986–2013. *Science of The Total Environment* 851, 158198. <u>https://doi.org/10.1016/j.scitotenv.2022.158198</u>
- Roux, N., Kaufmann, L., Matej, S., Kastner, T., Bondeau, A., Haberl, H., et al. (2023). Product level dataset on embodied human appropriation of net primary production. *Data in Brief* 51, 109725. <u>https://doi.org/10.1016/j.dib.2023.109725</u>
- Rubel, F. (2021). "Climate change and tick-borne encephalitis in the Greater Alpine region", in *Climate, ticks and disease*, ed. P. Nuttal (Wallingford: CABI), 354–359. Available at: <u>http://www.</u>

cabi.org/cabebooks/ebook/20210488022 (Accessed May 23, 2024).

- Rubenstein, M. A., Weiskopf, S. R., Bertrand, R., Carter, S. L., Comte, L., Eaton, M. J., et al. (2023). Climate change and the global redistribution of biodiversity: substantial variation in empirical support for expected range shifts. *Environmental Evidence* 12, 7. <u>https://doi.org/10.1186/s13750-023-00296-0</u>
- Rüdisser, J., Schirpke, U., and Tappeiner, U. (2019). Symbolic entities in the European Alps: Perception and use of a cultural ecosystem service. *Ecosystem Services* 39, 100980. <u>https://doi. org/10.1016/j.ecoser.2019.100980</u>
- Rust, P., Hasenegger, V., and König, J. (2017). Österreichischer Ernährungsbericht 2017. Vienna: Department für Ernährungswissenschaften der Universität Wien. Available at: <u>https:// nutrition.univie.ac.at/forschung/forschungsbereich-ernaehrungsverhalten-vulnerabler-personengruppen-ass-prof-drpetra-rust/projekte/wie-isst-oesterreich-ernaehrungsbericht-2017/</u>
- Saget, S., Porto Costa, M., Santos, C. S., Vasconcelos, M., Styles, D., and Williams, M. (2021). Comparative life cycle assessment of plant and beef-based patties, including carbon opportunity costs. Sustainable Production and Consumption 28, 936–952. <u>https://doi.org/10.1016/j.spc.2021.07.017</u>
- Sajeev, E. P. M., Amon, B., Ammon, C., Zollitsch, W., and Winiwarter, W. (2018). Evaluating the potential of dietary crude protein manipulation in reducing ammonia emissions from cattle and pig manure: A meta-analysis. *Nutr Cycl Agroecosyst* 110, 161–175. <u>https://doi.org/10.1007/s10705-017-9893-3</u>
- Salim, E., Ravanel, L., and Gauchon, C. (2021). Aesthetic perceptions of the landscape of a shrinking glacier: Evidence from the Mont Blanc massif. *Journal of Outdoor Recreation and Tourism* 35, 100411. <u>https://doi.org/10.1016/j.jort.2021.100411</u>
- Salinari, F., Giosuè, S., Tubiello, F. N., Rettori, A., Rossi, V., Spanna, F., et al. (2006). Downy mildew (Plasmopara viticola) epidemics on grapevine under climate change. *Global Change Biology* 12, 1299–1307. https://doi.org/10.1111/j.1365-2486.2006.01175.x
- Sandifer, P. A., Sutton-Grier, A. E., and Ward, B. P. (2015). Exploring connections among nature, biodiversity, ecosystem services, and human health and well-being: Opportunities to enhance health and biodiversity conservation. *Ecosystem Services* 12, 1–15. <u>https://doi.org/10.1016/j.ecoser.2014.12.007</u>
- Santika, T., Nelson, V., Flint, M., MacEwen, M., Cerretelli, S., and Brack, D. (2024). Leverage points for tackling unsustainable global value chains: market-based measures versus transformative alternatives. *Sustain Sci* 19, 285–305. <u>https://doi. org/10.1007/s11625-023-01430-0</u>
- Santillán, D., Garrote, L., Iglesias, A., and Sotes, V. (2020). Climate change risks and adaptation: new indicators for Mediterranean viticulture. *Mitig Adapt Strateg Glob Change* 25, 881–899. <u>https://doi.org/10.1007/s11027-019-09899-w</u>
- SAPEA (2020). A sustainable food system for the European Union: Evidence review report (1.2)., 1st Edn. Berlin: SAPEA. Available at: <u>https://zenodo.org/record/3745959/files/sustainable-food-</u> system-report.pdf?download=1 (Accessed October 2, 2023).
- Saugier, B., Roy, J., and Mooney, H. A. (2001). "Estimations of global terrestrial productivity: Converging toward a single number?", in *Terrestrial Global Productivity*, eds. J. Roy, B. Saugier, and H. A. Mooney (San Diego, San Francisco, New York, Boston, London, Sydney, Tokyo: Academic Press), 543–557. Available at: <u>https://www.researchgate.net/publication/284993456\_Esti-</u> mations\_of\_global\_terrestrial\_productivity\_Converging\_to-

ward\_a\_single\_number?enrichId=rgreq-1a906667351e697bba 361a9c20824c4a-XXX&enrichSource=Y292ZXJQYWdlOzI4N Dk5MzQ1NjtBUzo1NDA2OTMzMzQ0MjE1MDRAMTUwN

- Sauvage, J. (2014). The Stringency of Environmental Regulations and Trade in Environmental Goods. Paris: OECD Publishing. https://doi.org/10.1787/5jxrjn7xsnmq-en
- Schaffartzik, A., Eisenmenger, N., Krausmann, F., and Weisz, H. (2014). Consumption-based Material Flow Accounting. *Journal of Industrial Ecology* 18, 102–112. <u>https://doi.org/10.1111/jiec.12055</u>
- Schauberger, G., Hennig-Pauka, I., Zollitsch, W., Hortenhuber, S. J., Baumgartner, J., Niebuhr, K., et al. (2020). Efficacy of adaptation measures to alleviate heat stress in confined livestock buildings in temperate climate zones. *Biosystems engineering* 200, 157– 175. <u>https://doi.org/10.1016/j.biosystemseng.2020.09.010</u>
- Schauberger, G., Schönhart, M., Zollitsch, W., Hortenhuber, S. J., Kirner, L., Mikovits, C., et al. (2021). Economic Risk Assessment by Weather-Related Heat Stress Indices for Confined Livestock Buildings: A Case Study for Fattening Pigs in Central Europe. Agriculture 11, 122. <u>https://doi.org/10.3390/agriculture</u> <u>11020122</u>
- Schaumberger, A., Pötsch, E., and Schweiger, M. (2019). Veränderungen der Vegetationsdynamik unter zukünftigen Klimabedingungen. 21. Alpenländisches Expertenforum 2019, 35–44.
- Schelhaas, M.-J., Nabuurs, G.-J., Hengeveld, G., Reyer, C., Hanewinkel, M., Zimmermann, N. E., et al. (2015). Alternative forest management strategies to account for climate change-induced productivity and species suitability changes in Europe. *Reg Environ Change* 15, 1581–1594. <u>https://doi.org/10.1007/s10113-015-0788-z</u>
- Scherer, L., Rueda, O., and Smetana, S. (2022). "Environmental impacts of meat and meat replacements", in *Meat and Meat Replacements: An Interdisciplinary Assessment of Current Status and Future Directions*, eds. H. L. Meiselman and J. M. Lorenzo (Cambridge: Woodhead Publishing), 365–397.
- Scherhaufer, S., Hrad, M., Unger, N., and Obersteiner, G. (2016). Datenlage zu Lebensmittelabfallmengen in Österreich. Zusammenfassung von Studien des ABF-BOKU. Vienna: Universität für Bodenkultur Department für Wasser – Atmosphäre – Umwelt, Institut für Abfallwirtschaft. Available at: <u>https://www.nachhaltigkeit.steiermark.at/cms/dokumente/ 12600736\_139338025/fc59d169/ABF-BOKU\_Zusammenf\_ LM\_in\_%C3%96sterreich\_2016.pdf</u>
- Schermer, M., Stotten, R., Strasser, U., Meißl, G., Marke, T., Förster, K., et al. (2018). The Role of Transdisciplinary Research for Agricultural Climate Change Adaptation Strategies. *Agronomy* 8, 237. <u>https://doi.org/10.3390/agronomy8110237</u>
- Schieler, K. (1997). Methode der Zuwachsberechnung der Österreichischen Waldinventur. Available at: <u>http://zidapp.boku.ac.at/</u> <u>abstracts/search\_abstract.php?paID=3&paLIST=0&paSID</u> =794 (Accessed November 8, 2024).
- Schindler, D. E. (2017). Warmer climate squeezes aquatic predators out of their preferred habitat. *Proceedings of the National Academy of Sciences* 114, 9764–9765. <u>https://doi.org/10.1073/</u> pnas.1712818114
- Schipfer, F., Burli, P., Fritsche, U., Hennig, C., Stricker, F., Wirth, M., et al. (2024). The circular bioeconomy: a driver for system integration. *Energ Sustain Soc* 14, 34. <u>https://doi.org/10.1186/ s13705-024-00461-4</u>
- Schirpke, U., Altzinger, A., Leitinger, G., and Tasser, E. (2019). Change from agricultural to touristic use: Effects on the aes-

thetic value of landscapes over the last 150 years. *Landscape and Urban Planning* 187, 23–35. <u>https://doi.org/10.1016/j.landurb-plan.2019.03.004</u>

- Schirpke, U., and Ebner, M. (2022). Exposure to global change pressures and potential impacts on ecosystem services of mountain lakes in the European Alps. *Journal of Environmental Management* 318, 115606. <u>https://doi.org/10.1016/j.jenv-</u> man.2022.115606
- Schirpke, U., Ebner, M., and Tappeiner, U. (2024). Effects of climate-related environmental changes on non-material benefits from human-nature interactions: A literature review. *Ecosys*tem Services 69, 101650. <u>https://doi.org/10.1016/j.ecoser.2024.</u> 101650
- Schirpke, U., Kohler, M., Leitinger, G., Fontana, V., Tasser, E., and Tappeiner, U. (2017). Future impacts of changing landuse and climate on ecosystem services of mountain grassland and their resilience. *Ecosystem Services* 26, 79–94. <u>https://doi. org/10.1016/j.ecoser.2017.06.008</u>
- Schirpke, U., Meisch, C., Marsoner, T., and Tappeiner, U. (2018a). Revealing spatial and temporal patterns of outdoor recreation in the European Alps and their surroundings. *Ecosystem Ser*vices 31, 336–350. <u>https://doi.org/10.1016/j.ecoser.2017.11.017</u>
- Schirpke, U., Meisch, C., and Tappeiner, U. (2018b). Symbolic species as a cultural ecosystem service in the European Alps: insights and open issues. *Landscape Ecology* 33, 711–730. <u>https:// doi.org/10.1007/s10980-018-0628-x</u>
- Schirpke, U., Scolozzi, R., and Tappeiner, U. (2022). Not too small to benefit society: insights into perceived cultural ecosystem services of mountain lakes in the European Alps'. *Ecology and Society* 27. <u>https://doi.org/10.5751/ES-12987-270106</u>
- Schirpke, U., and Tasser, E. (2024). Potential impacts of climate change on ecosystem services in Austria. *Ecosystem Services* 68, 101641. <u>https://doi.org/10.1016/j.ecoser.2024.101641</u>
- Schirpke, U., Timmermann, F., Tappeiner, U., and Tasser, E. (2016). Cultural ecosystem services of mountain regions: Modelling the aesthetic value. *Ecological Indicators* 69, 78–90. <u>https://doi. org/10.1016/j.ecolind.2016.04.001</u>
- Schirpke, U., Zoderer, B. M., Tappeiner, U., and Tasser, E. (2021). Effects of past landscape changes on aesthetic landscape values in the European Alps. *Landscape and Urban Planning* 212, 104109. https://doi.org/10.1016/j.landurbplan.2021.104109
- Schlatzer, M., and Lindenthal, T. (2020). Einfluss von unterschiedlichen Ernährungsweisen auf Klimawandel und Flächeninanspruchnahme in Österreich und Übersee (DIETCCLU). Endbericht von StartClim2019.B in StartClim2019: Weitere Beiträge zur Umsetzung der österreichischen Anpassungsstrategie (Projekt-Endbericht, non peer-reviewed). Auftraggeber: BML-FUW, BMWF, ÖBf, Land Oberösterreich.
- Schlink, A. C., Nguyen, M.-L., and Viljoen, G. J. (2010). Water requirements for livestock production: a global perspective. *Rev. sci. tech. Off. int. Epiz* 29, 603–619. <u>https://doi.org/10.20506/</u> <u>rst.29.3.1999</u>
- Schnecker, J., Spiegel, F., Li, Y., Richter, A., Sandén, T., Spiegel, H., et al. (2023). Microbial responses to soil cooling might explain increases in microbial biomass in winter. *Biogeochemistry* 164, 521–535. <u>https://doi.org/10.1007/s10533-023-01050-x</u>
- Schodterer, H. (2022). Wildeinflussmonitoring 2019–2021: Mischwälder stärker betroffen. Vienna: Bundesforschungszentrum für Wald. Available at: <u>https://www.bfw.gv.at/pressemeldungen/wildeinflussmonitoring-2019-2021/</u> (Accessed October 3, 2023).

- Schönhart, M., Mitter, H., Schmid, E., Heinrich, G., and Gobiet, A. (2014). Integrated analysis of climate change impacts and adaptation measures in Austrian agriculture. *Ger. J. Agric. Econ* 63, 156–176. https://doi.org/10.22004/ag.econ.253157
- Schönhart, M., and Nadeem, I. (2015). Direct climate change impacts on cattle indicated by THI models. Advances in Animal Biosciences 6, 17–20. <u>https://doi.org/10.1017/S2040470014000430</u>
- Schönhart, M., Penker, M., and Schmid, E. (2009). Sustainable Local Food Production and Consumption: Challenges for Implementation and Research. *Outlook Agric* 38, 175–182. <u>https://doi. org/10.5367/00000009788632313</u>
- Schönhart, M., Schaupenlehner, T., Kuttner, M., Kirchner, M., and Schmid, E. (2016). Climate change impacts on farm production, landscape appearance, and the environment: Policy scenario results from an integrated field-farm-landscape model in Austria. Agricultural Systems 145, 39–50. <u>https://doi.org/10.1016/j. agsy.2016.02.008</u>
- Schönhart, M., Trautvetter, H., Parajka, J., Blaschke, A. F., Hepp, G., Kirchner, M., et al. (2018). Modelled impacts of policies and climate change on land use and water quality in Austria. *Land Use Policy* 76, 500–514. <u>https://doi.org/10.1016/j.landusepol.2018.02.031</u>
- Schratt-Ehrendorfer, L., Niklfeld, H., Schröck, C., and Stöhr, O. (2022). Rote Liste der Farn- und Blütenpflanzen Österreichs. Dritte, völlig neu bearbeitete Auflage. Available at: <u>https://www. zobodat.at/publikation\_articles.php?id=509508</u> (Accessed November 7, 2024).
- Schrempf, W. (1986). Waldbauliche Untersuchungen im Fichten-Tannen-Buchen-Urwald Rothwald und in Urwald-Folgebestaenden. Vienna: Universität für Bodenkultur, Institut für Waldbau. Available at: <u>https://agris.fao.org/search/en/providers/ 123819/records/64735b6f08fd68d546027bd0</u> (Accessed May 27, 2024).
- Schuldt, B., Buras, A., Arend, M., Vitasse, Y., Beierkuhnlein, C., Damm, A., et al. (2020). A first assessment of the impact of the extreme 2018 summer drought on Central European forests. *Basic and Applied Ecology* 45, 86–103. <u>https://doi.org/10.1016/j. baae.2020.04.003</u>
- Schürr, A., Elbel, J., Hieronimi, A., Auer, I., Coenen, M., and Böse-O'Reilly, S. (2023). Mental health in adolescents after experiencing a flood event in Bavaria, Germany—A qualitative interview study. *Front. Public Health* 11. <u>https://doi.org/10.3389/ fpubh.2023.1210072</u>
- Searchinger, T. D., Beringer, T., Holtsmark, B., Kammen, D. M., Lambin, E. F., Lucht, W., et al. (2018). Europe's renewable energy directive poised to harm global forests. *Nat Commun* 9, 3741. https://doi.org/10.1038/s41467-018-06175-4
- Searchinger, T. D., Estes, L., Thornton, P. K., Beringer, T., Notenbaert, A., Rubenstein, D., et al. (2015). High carbon and biodiversity costs from converting Africa's wet savannahs to cropland. *Nature Clim Change* 5, 481–486. <u>https://doi.org/10.1038/ nclimate2584</u>
- Seddon, N. (2022). Harnessing the potential of nature-based solutions for mitigating and adapting to climate change. *Science* 376, 1410–1416. <u>https://doi.org/10.1126/science.abn9668</u>
- Seebauer, S., Thaler, T., Hanger-Kopp, S., and Schinko, T. (2023). How path dependency manifests in flood risk management: observations from four decades in the Ennstal and Aist catchments in Austria. *Reg Environ Change* 23, 31. <u>https://doi.org/10.1007/ s10113-023-02029-y</u>

- Seeber, J., Tasser, E., Rubatscher, D., Loacker, I., Lavorel, S., Robson, T. M., et al. (2022). Effects of land use and climate on carbon and nitrogen pool partitioning in European mountain grasslands. *Science of The Total Environment* 822, 153380. <u>https://doi.org/10.1016/j.scitotenv.2022.153380</u>
- Seibold, S., Rammer, W., Hothorn, T., Seidl, R., Ulyshen, M. D., Lorz, J., et al. (2021). The contribution of insects to global forest deadwood decomposition. *Nature* 597, 77–81. <u>https://doi.org/10.1038/s41586-021-03740-8</u>
- Seidl, R., Schelhaas, M.-J., and Lexer, M. J. (2011). Unraveling the drivers of intensifying forest disturbance regimes in Europe. *Global Change Biology* 17, 2842–2852. <u>https://doi.org/10.1111/j.1365-2486.2011.02452.x</u>
- Seidl, R., Schelhaas, M.-J., Lindner, M., and Lexer, M. J. (2009). Modelling bark beetle disturbances in a large scale forest scenario model to assess climate change impacts and evaluate adaptive management strategies. *Reg Environ Change* 9, 101–119. <u>https:// doi.org/10.1007/s10113-008-0068-2</u>
- Seidl, R., Thom, D., Kautz, M., Martin-Benito, D., Peltoniemi, M., Vacchiano, G., et al. (2017). Forest disturbances under climate change. *Nature Climate Change* 7, 395–402. <u>https://doi. org/10.1038/nclimate3303</u>
- Seidl, R., von Wirth, T., and Krütli, P. (2019). Social acceptance of distributed energy systems in Swiss, German, and Austrian energy transitions. *Energy Research & Social Science* 54, 117–128. <u>https://doi.org/10.1016/j.erss.2019.04.006</u>
- Seinfeld, J. H., and Pandis, S. N. (1998). From air pollution to climate change. Atmospheric Chemistry and Physics 1326.
- Seitner, D., Tripolt, T., Stüger, H. P., and Kiefer, I. (2022). Risikobarometer Gesundheit und Umwelt 2022 – Welche Risiken die Österreicher:innen bewegen. AGES – Österreichische Agentur für Gesundheit und Ernährungssicherheit GmbH. Available at: <u>https://wissenaktuell.ages.at/download/0/0/9c6f4e975830808f-0bab83757e4103d087f462d2/fileadmin/AGES2015/Wissen-Aktuell/Wissen\_aktuell\_2022/Risikobarometer2022\_ BE.pdf (Accessed July 26, 2022).</u>
- Semba, R. D., de Pee, S., Kim, B., McKenzie, S., Nachman, K., and Bloem, M. W. (2020). Adoption of the 'planetary health diet' has different impacts on countries' greenhouse gas emissions. *Nat Food* 1, 481–484. <u>https://doi.org/10.1038/s43016-020-0128-4</u>
- Senf, C., Buras, A., Zang, C. S., Rammig, A., and Seidl, R. (2020). Excess forest mortality is consistently linked to drought across Europe. Nat Commun 11, 6200. <u>https://doi.org/10.1038/s41467-020-19924-1</u>
- Senf, C., and Seidl, R. (2021). Storm and fire disturbances in Europe: Distribution and trends. *Global Change Biology* 27, 3605–3619. <u>https://doi.org/10.1111/gcb.15679</u>
- Seppälä, J., Heinonen, T., Pukkala, T., Kilpeläinen, A., Mattila, T., Myllyviita, T., et al. (2019). Effect of increased wood harvesting and utilization on required greenhouse gas displacement factors of wood-based products and fuels. *Journal of Environmental Management* 247, 580–587. <u>https://doi.org/10.1016/ j.jenvman.2019.06.031</u>
- Settele, J., Bishop, J., and Potts, S. G. (2016). Climate change impacts on pollination. *Nature Plants* 2, 16092. <u>https://doi.org/10.1038/ nplants.2016.92</u>
- Shields, S., and Orme-Evans, G. (2015). The Impacts of Climate Change Mitigation Strategies on Animal Welfare. *Animals* 5, 361–394. <u>https://doi.org/10.3390/ani5020361</u>
- Siddiqui, S. A., Zannou, O., Karim, I., Kasmiati, Awad, N. M. H., Gołaszewski, J., et al. (2022). Avoiding Food Neophobia and

Increasing Consumer Acceptance of New Food Trends—A Decade of Research. *Sustainability* 14, 10391.

- Sigsgaard, T., Forsberg, B., Annesi-Maesano, I., Blomberg, A., Bølling, A., Boman, C., et al. (2015). Health impacts of anthropogenic biomass burning in the developed world. *European Respiratory Journal* 46, 1577–1588. <u>https://doi.org/10.1183/</u> 13993003.01865-2014
- Sinabell, F., Falkner, K., Streicher, G., Schönhart, M., Schmid, E., and Fritz, C. (2023). Austrian Agriculture 2030 – 2040 – 2050. Scenarios and Sensitivity Analyses on Land Use, Crop and Livestock Production. Vienna: Austrian Institute of Economic Research (WIFO). Available at: <u>https://raumberg-gumpenstein.</u> at/jdownloads/sonstiges/34\_2023\_klilasz-endbericht\_final.pdf (Accessed November 14, 2024).
- Sinke, P., Swartz, E., Sanctorum, H., van der Giesen, C., and Odegard, I. (2023). Ex-ante life cycle assessment of commercial-scale cultivated meat production in 2030. *The International Journal* of Life Cycle Assessment 28, 234–254. <u>https://doi.org/10.1007/ s11367-022-02128-8</u>
- Smetana, S., Palanisamy, M., Mathys, A., and Heinz, V. (2016). Sustainability of insect use for feed and food: Life Cycle Assessment perspective. *Journal of Cleaner Production* 137, 741–751. <u>https://doi.org/10.1016/j.jclepro.2016.07.148</u>
- Smit, B., and Wandel, J. (2006). Adaptation, adaptive capacity and vulnerability. *Global Environmental Change* 16, 282–292. <u>https://doi.org/10.1016/j.gloenvcha.2006.03.008</u>
- Sohngen, B., and Sedjo, R. (2005). Impacts of climate change on forest product markets: Implications for North American producers. *The Forestry Chronicle* 81, 669–674. <u>https://doi.org/ 10.5558/tfc81669-5</u>
- Soimakallio, S., Böttcher, H., Niemi, J., Mosley, F., Turunen, S., Hennenberg, K. J., et al. (2022). Closing an open balance: The impact of increased tree harvest on forest carbon. *GCB Bioener*gy 14, 989–1000. <u>https://doi.org/10.1111/gcbb.12981</u>
- Soja, G., Züger, J., Knoflacher, M., Kinner, P., and Soja, A.-M. (2013). Climate impacts on water balance of a shallow steppe lake in Eastern Austria (Lake Neusiedl). *Journal of Hydrology* 480, 115–124. <u>https://doi.org/10.1016/j.jhydrol.2012.12.013</u>
- Sonnberger, K., Duscher, G. G., Fuehrer, H.-P., and Leschnik, M. (2020). Current trends in canine dirofilariosis in Austria—do we face a pre-endemic status? *Parasitol Res* 119, 1001–1009. https://doi.org/10.1007/s00436-019-06576-4
- Spangl, W., Nagl, C., Schneider, J., and Kaiser, A. (2006). Herkunftsanalyse der PM10-Belastung in Österreich. Ferntransport und regionale Beiträge. Wien: Umweltbundesamt.
- Spannlang, R. (2022). Elefantenbeine. Forstzeitung. Available at: <u>https://www.forstzeitung.at/content/holz/forstzeitung/de/</u> <u>markt/2022/10/douglasien-in-oesterreich.html</u> (Accessed November 7, 2024).
- Spathelf, P., Bolte, A., and Maaten, van der E. C. D. (2015). Is Closeto-Nature Silviculture (CNS) an adequate concept to adapt forests to climate change? *Landbauforschung – applied agricultural and forestry research*, 161–170. <u>https://doi.org/10.3220/</u> LBF1452526188000
- Spathelf, P., Stanturf, J., Kleine, M., Jandl, R., Chiatante, D., and Bolte, A. (2018). Adaptive measures: integrating adaptive forest management and forest landscape restoration. *Annals of Forest Science* 75, 55. <u>https://doi.org/10.1007/s13595-018-0736-4</u>
- Spiecker, H., Mielikäinen, K., Köhl, M., and Skovsgaard, J. P. eds. (1996). Growth Trends in European Forests: Studies from 12

- Spörl, P., Trimmel, S., Haluza, D., Sauerbrey, S., Irrgeher, J., Prohaska, T., et al. (2024). Façade Greening for Healthy Urban Air: An Umbrella Review on Particulate Matter Reduction, Challenges, and Future Directions. *Sustainability* 16, 446. <u>https://doi.org/ 10.3390/su16010446</u>
- Springmann, M., Clark, M., Mason-D'Croz, D., Wiebe, K., Bodirsky, B. L., Lassaletta, L., et al. (2018). Options for keeping the food system within environmental limits. *Nature* 562, 519–525. <u>https://doi.org/10.1038/s41586-018-0594-0</u>
- Springmann, M., Godfray, H. C. J., Rayner, M., and Scarborough, P. (2016). Analysis and valuation of the health and climate change cobenefits of dietary change. *Proceedings of the National Academy of Sciences* 113, 4146–4151. <u>https://doi.org/10.1073/ pnas.1523119113</u>
- Stagnari, F., Maggio, A., Galieni, A., and Pisante, M. (2017). Multiple benefits of legumes for agriculture sustainability: an overview. *Chem. Biol. Technol. Agric.* 4, 2. <u>https://doi.org/10.1186/</u> <u>s40538-016-0085-1</u>
- Statistik Austria (2022a). 2021 wurden 4 920 Tonnen Speisefisch in Österreich produziert. Available at: <u>https://www.statistik.at/</u> <u>fileadmin/announcement/2022/12/20221216Aquakulturpro-</u> <u>duktion2021.pdf</u>
- Statistik Austria (2022b). Agrarstrukturerhebung 2020: Land- und forstwirtschaftliche Betriebe und deren Strukturdaten. Endgültige Ergebnisse. Wien: Statistik Austria. Available at: <u>https:// www.statistik.at/fileadmin/publications/SB\_1-17\_AS2020.pdf</u>
- Statistik Austria (2022c). Austrian Health Survey 2019. Vienna: Statistik Austria. Available at: <u>https://www.statistik.at/en/statis-</u> <u>tics/population-and-society/health/health-determinants/over-</u> <u>weight-and-obesity</u>
- Statistik Austria (2022d). Pressemitteilung: 12.756-054/22 Rund 9% aller Sterbefälle im Jahr 2021 aufgrund von COVID-19. Available at: <u>https://www.statistik.at/fileadmin/announcement/</u> 2022/05/20220303Todesursachen2021.pdf
- Statistik Austria (2022e). Versorgungsbilanzen für tierische Produkte 2021. Wien: Statistik Austria. Available at: <u>https://www. statistik.at/fileadmin/user\_upload/SB\_1-26\_tier\_2022.pdf</u>
- Statistik Austria (2023a). Anbauübersicht 2012 bis 2022. Tabelle. *STATISTIK AUSTRIA*. Available at: <u>https://www.statistik.at/</u> <u>statistiken/land-und-forstwirtschaft/pflanzenbau/acker-</u> <u>bau-dauergruenland</u> (Accessed May 17, 2024).
- Statistik Austria (2023b). Landwirtschaftliche Gesamtrechnung. Available at: <u>https://www.statistik.at/statistiken/land-und-forstwirtschaft/land-und-forstwirtschaftliche-oekonomie-und-preise/landwirtschaftliche-gesamtrechnung/landwirtschaftliche-gesamtrechnung-auf-nationaler-ebene</u>
- Statistik Austria (2024). Irrigation. *STATISTICS AUSTRIA*. Available at: <u>https://www.statistik.at/en/statistics/agriculture-and-forestry/agricultural-and-forestry-production-methods/irrigation</u> (Accessed November 7, 2024).
- Steiger, R., Damm, A., Prettenthaler, F., and Pröbstl-Haider, U. (2021). Climate change and winter outdoor activities in Austria. *Journal of Outdoor Recreation and Tourism* 34, 100330. <u>https:// doi.org/10.1016/j.jort.2020.100330</u>
- Stern, T., Ploll, U., Spies, R., Schwarzbauer, P., Hesser, F., and Ranacher, L. (2018). Understanding Perceptions of the Bioeconomy in Austria—An Explorative Case Study. *Sustainability* 10, 4142. <u>https://doi.org/10.3390/su10114142</u>

- Stevenson, M. P., Schilhab, T., and Bentsen, P. (2018). Attention Restoration Theory II: a systematic review to clarify attention processes affected by exposure to natural environments. *Journal of Toxicology and Environmental Health, Part B* 21, 227–268. https://doi.org/10.1080/10937404.2018.1505571
- Steyrer, G. (2020). Dokumentation der Waldschädigungsfaktoren. *BFW*. Available at: <u>https://www.bfw.gv.at/dokumentation-wald-</u> schaedigungsfaktoren/ (Accessed April 1, 2024).
- Stöckli, V. (2013). Grundsätze einer nachhaltigen Waldweide im Gebirgswald. Bündner Wald 4, 25–30.
- Stojanović, M., Sánchez-Salguero, R., Levanič, T., Szatniewska, J., Pokorný, R., and Linares, J. C. (2017). Forecasting tree growth in coppiced and high forests in the Czech Republic. The legacy of management drives the coming Quercus petraea climate responses. *Forest Ecology and Management* 405, 56–68. <u>https:// doi.org/10.1016/j.foreco.2017.09.021</u>
- Strimitzer, L., Wicek, B., and Nemesthothy, K. (2022). Holzströme Österreichs 2020. Available at: <u>https://www.klimaaktiv.at/fileadmin/Bibliothek/Publikationen/2022\_Holzstroeme\_oesterreich.pdf</u>
- Strohbach, M. W., and Haase, D. (2012). Above-ground carbon storage by urban trees in Leipzig, Germany: Analysis of patterns in a European city. *Landscape and Urban Planning* 104, 95–104. <u>https://doi.org/10.1016/j.landurbplan.2011.10.001</u>
- Studeregger, A., Podesser, A., Mitterer, C., Fischer, J.-T., Ertl, W., Nairz, P., et al. (2020). "Lawinen", in ExtremA 2019. Aktueller Wissensstand zu Extremereignissen alpiner Naturgefahren in Österreich, (Vienna University Press), 512–533. Available at: https://info.bml.gv.at/themen/wald/wald-und-naturgefahren/ wildbach--und-lawinenverbauung/kooperationen-zusammenarbeit/extrema\_2019.html
- Sun, Z., Scherer, L., Tukker, A., Seth, A. S.-L., Bruckner, M., Gibbs, H. K., et al. (2022). Dietary change in high-income nations alone can lead to substantial double climate dividend. *Nature Food* 3, 29–37. <u>https://doi.org/10.1038/s43016-021-00431-5</u>
- Svanda, N., and Zech, S. (2023). "Kapitel 19. Raumplanung", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 529– 546. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_23</u> (Accessed May 9, 2024).
- SVS (2024). Forstliche Sicherheitsbestimmungen. Vienna: Sozialversicherungsanstalt der Selbständigen (SVS). Available at: <u>https://www.svs.at/cdscontent/load?contentid=10008.7666</u> 82&version=1651216044 (Accessed May 15, 2025).
- Swanson, K. S., Carter, R. A., Yount, T. P., Aretz, J., and Buff, P. R. (2013). Nutritional Sustainability of Pet Foods. Advances in Nutrition 4, 141–150. <u>https://doi.org/10.3945/an.112.003335</u>
- Tafani, M., Cohas, A., Bonenfant, C., Gaillard, J.-M., and Allainé, D. (2013). Decreasing litter size of marmots over time: a life history response to climate change? *Ecology* 94, 580–586. <u>https:// doi.org/10.1890/12-0833.1</u>
- Tappeiner, U., Tasser, E., Leitinger, G., Cernusca, A., and Tappeiner, G. (2008). Effects of Historical and Likely Future Scenarios of Land Use on Above- and Belowground Vegetation Carbon Stocks of an Alpine Valley. *Ecosystems* 11, 1383–1400. <u>https:// doi.org/10.1007/s10021-008-9195-3</u>
- Tasser, E., Leitinger, G., and Tappeiner, U. (2017). Climate change versus land-use change—What affects the mountain landscapes more? Land Use Policy 60, 60–72. <u>https://doi.org/10.1016/j.</u> landusepol.2016.10.019

- Tasser, E., Schirpke, U., Zoderer, B. M., and Tappeiner, U. (2020). Towards an integrative assessment of land-use type values from the perspective of ecosystem services. *Ecosystem Services* 42, 101082. <u>https://doi.org/10.1016/j.ecoser.2020.101082</u>
- Tasser, E., Unterthurner, B., Agreiter, A., Aukenthaler, H., Gerstgrasser, L., Giardino, M., et al. (2023). Long-term game species dynamic as indicator for changing landscape quality. *Science of The Total Environment* 874, 162375. <u>https://doi.org/10.1016/j. scitotenv.2023.162375</u>
- Tasser, E., Unterthurner, B., Agreiter, A., Gerstgrasser, L., Giardino, M., Tappeiner, U., et al. (2024). Drivers of spatio-temporal population dynamics of game species in a mountain landscape. *Sci Rep* 14, 2740. <u>https://doi.org/10.1038/s41598-024-53019-x</u>
- Tello-García, E., Huber, L., Leitinger, G., Peters, A., Newesely, C., Ringler, M.-E., et al. (2020). Drought- and heat-induced shifts in vegetation composition impact biomass production and water use of alpine grasslands. *Environmental and Experimental Botany* 169, 103921. <u>https://doi.org/10.1016/j.envexpbot.2019.103921</u>
- Tello-García, E., Mandolini, E., Tasser, E., Probst, M., Peintner, U., Tappeiner, U., et al. (2023). Red clover productivity under drought: Are soil microbes a burden or a treasure? *Environmental and Experimental Botany* 214, 105486. <u>https://doi.org/10.</u> <u>1016/j.envexpbot.2023.105486</u>
- Tervo-Kankare, K., Hall, C. M., and Saarinen, J. (2013). Christmas Tourists' Perceptions to Climate Change in Rovaniemi, Finland. *Tourism Geographies* 15, 292–317. <u>https://doi.org/10.1080/ 14616688.2012.726265</u>
- Thaler, S., Eitzinger, J., Trnka, M., and Dubrovsky, M. (2012). Impacts of climate change and alternative adaptation options on winter wheat yield and water productivity in a dry climate in Central Europe. *The Journal of Agricultural Science* 150, 537–555. <u>https://doi.org/10.1017/S0021859612000093</u>
- Thaler, T. (2021). Just retreat—how different countries deal with it: examples from Austria and England. *J Environ Stud Sci* 11, 412–419. https://doi.org/10.1007/s13412-021-00694-1
- Thien, F., Beggs, P. J., Csutoros, D., Darvall, J., Hew, M., Davies, J. M., et al. (2018). The Melbourne epidemic thunderstorm asthma event 2016: an investigation of environmental triggers, effect on health services, and patient risk factors. *The Lancet Planetary Health* 2, e255–e263. <u>https://doi.org/10.1016/S2542-5196(18)30120-7</u>
- Thomson, M. C., and Stanberry, L. R. (2022). Climate Change and Vectorborne Diseases. N Engl J Med 387, 1969–1978. <u>https:// doi.org/10.1056/NEJMra2200092</u>
- Tian, Y., Schindlbacher, A., Malo, C. U., Shi, C., Heinzle, J., Kwatcho Kengdo, S., et al. (2023). Long-term warming of a forest soil reduces microbial biomass and its carbon and nitrogen use efficiencies. Soil Biology and Biochemistry 184, 109109. <u>https:// doi.org/10.1016/j.soilbio.2023.109109</u>
- Tiemeyer, B., Albiac Borraz, E., Augustin, J., Bechtold, M., Beetz, S., Beyer, C., et al. (2016). High emissions of greenhouse gases from grasslands on peat and other organic soils. *Global Change Biology* 22, 4134–4149. <u>https://doi.org/10.1111/gcb.13303</u>
- Tilman, D., and Clark, M. (2014). Global diets link environmental sustainability and human health. *Nature* 515, 518–522. <u>https:// doi.org/10.1038/nature13959</u>
- Triguero-Mas, M., Dadvand, P., Cirach, M., Martínez, D., Medina, A., Mompart, A., et al. (2015). Natural outdoor environments and mental and physical health: relationships and mecha-

- Trnka, M., Rötter, R., Ruiz-Ramos, M., Kersebaum, K. C., Olesen, J. E., Žalud, Z., et al. (2014). Adverse weather conditions for European wheat production will become more frequent with climate change. *Nature Clim Change* 4, 637–643. <u>https://doi.org/ 10.1038/nclimate2242</u>
- Turbelin, A. J., Cuthbert, R. N., Essl, F., Haubrock, P. J., Ricciardi, A., and Courchamp, F. (2023). Biological invasions are as costly as natural hazards. *Perspectives in Ecology and Conservation* 21, 143–150. https://doi.org/10.1016/j.pecon.2023.03.002
- Udali, A., Andrighetto, N., Grigolato, S., and Gatto, P. (2021). Economic Impacts of Forest Storms—Taking Stock of After-Vaia Situation of Local Roundwood Markets in Northeastern Italy. *Forests* 12, 414. <u>https://doi.org/10.3390/f12040414</u>
- Ulrich, R. S. (1984). View Through a Window May Influence Recovery from Surgery. Science 224, 420–421. <u>https://doi.org/ 10.1126/science.6143402</u>
- Ulrich, R. S., Simons, R. F., Losito, B. D., Fiorito, E., Miles, M. A., and Zelson, M. (1991). Stress recovery during exposure to natural and urban environments. *Journal of Environmental Psycholo*gy 11, 201–230. <u>https://doi.org/10.1016/S0272-4944(05)80184-</u> 7
- Umweltbundesamt (2022). Austria's National Inventory Report 2022. Vienna: Umweltbundesamt. Available at: <u>https://www. umweltbundesamt.at/fileadmin/site/publikationen/rep0811.</u> <u>pdf</u>
- Umweltbundesamt (2023a). Austria's National Inventory Report 2023. Vienna: Umweltbundesamt. Available at: <u>https://www. umweltbundesamt.at/fileadmin/site/publikationen/rep0852.</u> <u>pdf</u>
- Umweltbundesamt (2023b). Biomasseheizungen und Luftqualität. Available at: <u>https://www.umweltbundesamt.at/studien-re-ports/publikationsdetail?pub\_id=2477&cHash=6106500e772</u> <u>abe683860fa5b4d8f224d</u>
- Umweltbundesamt (2024). Austria's National Inventory Report 2024. Vienna: Umweltbundesamt. Available at: <u>https://www. umweltbundesamt.at/studien-reports/publikationsdetail?pub\_</u> id=2527&cHash=4f93d43917ca9a3ce6e2ba4c1191ca1a
- Umweltbundesamt (2025a). Austria's Annual Greenhouse Gas Inventory 1990–2023. Submission under Regulation (EU) No 2018/1999. Vienna: Umweltbundesamt. Available at: <u>https:// www.umweltbundesamt.at/fileadmin/site/publikationen/rep</u> 0952.pdf
- Umweltbundesamt (2025b). SEKTOR LANDNUTZUNG, LANDNUTZUNGSÄNDERUNG UND WALD (LULUCF), TREIBHAUSGAS-INVENTUR 2025. Available at: <u>https://</u> www.umweltbundesamt.at/fileadmin/site/aktuelles/2025/hintergrundinfo\_lulucf\_submission\_2025.pdf
- Unterberger, C., Brunner, L., Nabernegg, S., Steininger, K. W., Stabentheiner, E., Monschein, S., et al. (2018). Spring frost risk for regional apple production under a warmer climate. *PLOS ONE* 13, e0200201. <u>https://doi.org/10.1371/journal.pone.</u> 0200201
- Usha, Mishra, V., and Devi, M. S. (2020). Effect of environmental factors on the foraging activities of major bee pollinators. *Journal of entomology and zoology studies* 8, 450–454. <u>https://www.entomoljournal.com/archives/2020/vol8issue3/PartG/8-3-85-547.pdf</u>
- Vacek, Z., Vacek, S., Slanař, J., Bílek, L., Bulušek, D., Štefančík, I., et al. (2019). Adaption of Norway spruce and European beech

forests under climate change: from resistance to close-to-nature silviculture. *Central European Forestry Journal* 65, 129–144. https://doi.org/10.2478/forj-2019-0013

- Vacik, H., Mayr, S., Jandl, R., Hackl, J., Gratzer, G., Pülzl, H., et al. (2022). Österreichs Handlungsoptionen zur Umsetzung der UN-Agenda 2030 für eine lebenswerte Zukunft. UniNeTZ. Available at: <u>https://www.uninetz.at/optionenbericht\_downloads/SDG\_15\_Option\_15\_04.pdf</u>
- Van Den Berge, S., Vangansbeke, P., Baeten, L., Vanhellemont, M., Vanneste, T., De Mil, T., et al. (2021). Biomass increment and carbon sequestration in hedgerow-grown trees. *Dendrochronologia* 70, 125894. <u>https://doi.org/10.1016/j.dendro.2021.125894</u>
- Van Herzele, A., Gobin, A., Van Gossum, P., Acosta, L., Waas, T., Dendoncker, N., et al. (2013). Effort for money? Farmers' rationale for participation in agri-environment measures with different implementation complexity. *Journal of Environmental Management* 131, 110–120. <u>https://doi.org/10.1016/j.jenvman.2013.09.030</u>
- Van Wesemael, D., Vandaele, L., Ampe, B., Cattrysse, H., Duval, S., Kindermann, M., et al. (2019). Reducing enteric methane emissions from dairy cattle: Two ways to supplement 3-nitrooxypropanol. *Journal of Dairy Science* 102, 1780–1787. <u>https:// doi.org/10.3168/jds.2018-14534</u>
- van Zanten, M. C., Wichink Kruit, R. J., Hoogerbrugge, R., Van der Swaluw, E., and van Pul, W. A. J. (2017). Trends in ammonia measurements in the Netherlands over the period 1993– 2014. Atmos. Environ. 148, 352–360. <u>https://doi.org/10.1016/ j.atmosenv.2016.11.007</u>
- Vanomsen, P. (2006). Der Einfluss der Durchforstung auf die Verankerung der Fichte hinsichtlich ihrer Sturmresistenz. Zürich: ETH. <u>https://doi.org/10.3929/ETHZ-A-005195897</u>
- Vanschoenwinkel, J., Moretti, M., and Van Passel, S. (2020). The effect of policy leveraging climate change adaptive capacity in agriculture. *European Review of Agricultural Economics* 47, 138– 156. <u>https://doi.org/10.1093/erae/jbz007</u>
- Vári, Á., Kozma, Z., Pataki, B., Jolánkai, Z., Kardos, M., Decsi, B., et al. (2022). Disentangling the ecosystem service 'flood regulation': Mechanisms and relevant ecosystem condition characteristics. *Ambio* 51, 1855–1870. <u>https://doi.org/10.1007/s13280-022-01708-0</u>
- Vasiliev, D., and Greenwood, S. (2021). The role of climate change in pollinator decline across the Northern Hemisphere is underestimated. *Science of The Total Environment* 775, 145788. <u>https://doi.org/10.1016/j.scitotenv.2021.145788</u>
- Vermeulen, S. J., Park, T., Khoury, C. K., and Béné, C. (2020). Changing diets and the transformation of the global food system. Annals of the New York Academy of Sciences 1478, 3–17. https://doi.org/10.1111/nyas.14446
- Vierikko, K., and Yli-Pelkonen, V. (2019). Seasonality in recreation supply and demand in an urban lake ecosystem in Finland. Urban Ecosystems 22, 769–783. <u>https://doi.org/10.1007/s11252-019-00849-7</u>
- Vihemäki, H., Ludvig, A., Toivonen, R., Toppinen, A., and Weiss, G. (2019). Institutional and policy frameworks shaping the wooden multi-storey construction markets: a comparative case study on Austria and Finland. *Wood Material Science & Engineering* 14, 312–324. https://doi.org/10.1080/17480272.2019.1641741
- Vilkka, L. (2021). *The Intrinsic Value of Nature*. Leiden, Netherlands: Brill.
- Vitt, R., Weber, L., Zollitsch, W., Hörtenhuber, S. J., Baumgartner, J., Niebuhr, K., et al. (2017). Modelled performance of energy sav-

ing air treatment devices to mitigate heat stress for confined livestock buildings in Central Europe. *Biosystems Engineering* 164, 85–97. <u>https://doi.org/10.1016/j.biosystemseng.2017.09.013</u>

- VLK (2019). Klimawandel und Landwirtschaft: Anpassungsstrategien im Bereich Ackerbau. Berlin: Verband der Landwirtschaftskammern (VLK). <u>https://doi.org/10.4126/FRL01-006414662</u>
- Voigt, C., Hundscheid, L., Plank, C., and Pichler, M. (2024). Meat politics. Analysing actors, strategies and power relations governing the meat regime in Austria. *Geoforum* 154, 104048. <u>https://doi.org/10.1016/j.geoforum.2024.104048</u>
- Volk, M., Suter, M., Wahl, A.-L., and Bassin, S. (2021). Subalpine grassland productivity increased with warmer and drier conditions, but not with higher N deposition, in an altitudinal transplantation experiment. *Biogeosciences* 18, 2075–2090. <u>https:// doi.org/10.5194/bg-18-2075-2021</u>
- von Heßberg, A., Jentsch, A., Berauer, B., Ewald, J., Fütterer, S., Görgen, A., et al. (2021). Almen in Zeiten des Klimawandels
  Schutz der Artenvielfalt durch (Wieder-) Beweidung? Die Fallstudie Brunnenkopfalm im Ammergebirge. *Naturschutz* und Landschaftsplanung 53, 28–36. <u>https://doi.org/10.1399/</u> <u>NuL.2021.03.02</u>
- Vršič, S., Šuštar, V., Pulko, B., and Šumenjak, T. K. (2014). Trends in climate parameters affecting winegrape ripening in northeastern Slovenia. *Climate Research* 58, 257–266. <u>https://doi.org/ 10.3354/cr01197</u>
- Vršič, S., and Vodovnik, T. (2012). Reactions of grape varieties to climate changes in North East Slovenia. *Plant, Soil and Environment* 58 (2012), 34–41. <u>https://doi.org/10.17221/352/2011-PSE</u>
- Walder, P., and Kantelhardt, J. (2018). The Environmental Behaviour of Farmers – Capturing the Diversity of Perspectives with a Q Methodological Approach. *Ecological Economics* 143, 55–63. <u>https://doi.org/10.1016/j.ecolecon.2017.06.018</u>
- Walker, W. S., Gorelik, S. R., Cook-Patton, S. C., Baccini, A., Farina, M. K., Solvik, K. K., et al. (2022). The global potential for increased storage of carbon on land. *Proc. Natl. Acad. Sci. U.S.A.* 119, e2111312119. <u>https://doi.org/10.1073/pnas.2111312119</u>
- Wallner, P., and Moshammer, H. (2023). Social Vulnerability and Heat: Systematic Literature Search and Narrative Synthesis. Available at: <u>https://wegcwp.uni-graz.at/discc-at/wp-content/</u> <u>uploads/sites/10/2023/10/working\_paper1\_merged.pdf</u>
- WBAE (2020). Politik für eine nachhaltigere Ernährung Eine integrierte Ernährungspolitik entwickeln und faire Ernährungsumgebungen gestalten. Wissenschaftlicher Beirat für Agrarpolitik, Ernährung und gesundheitlichen Verbraucherschutz beim BMEL. Available at: <u>https://www.bmel.de/SharedDocs/ Archiv/Downloads/wbae-gutachten-nachhaltige-ernaehrung. pdf?\_\_blob=publicationFile&v=3</u>
- Wegner, G. I., Murray, K. A., Springmann, M., Muller, A., Sokolow, S. H., Saylors, K., et al. (2022). Averting wildlife-borne infectious disease epidemics requires a focus on socio-ecological drivers and a redesign of the global food system. *eClinicalMedicine* 47, 101386. <u>https://doi.org/10.1016/j.eclinm.2022.101386</u>
- Weinmayr, G. (2023). Auswirkungen von Luftverschmutzung in Europa und die WHO-Richtlinien: eine Chance, die Gesundheit der Bevölkerung zu schützen und sogar zu verbessern. Umweltmed – Hygiene – Arbeitsmed 28, 117–135.
- Weiss, F., and Leip, A. (2012). Greenhouse gas emissions from the EU livestock sector: A life cycle assessment carried out with the CAPRI model. Agriculture, Ecosystems & Environment 149, 124–134. https://doi.org/10.1016/j.agee.2011.12.015

- Weiss, P., Braun, M., Fritz, D., Gschwantner, T., Hesser, F., Jandl, R., et al. (2020). Endbericht zum Projekt CareforParis. Klimaund Energiefonds Wien. Vienna: Umweltbundesamt. Available at: <u>https://www.waldgeschichten.com/wp-content/uploads/</u> 2021/11/B670274-ACRP9-CareforParis-KR16AC0K13154-EB. pdf
- Weisz, U., Pichler, P.-P., Jaccard, I. S., Haas, W., Matej, S., Bachner, F., et al. (2020). Carbon emission trends and sustainability options in Austrian health care. *Resources, Conservation and Recycling* 160, 104862. <u>https://doi.org/10.1016/j.resconrec.2020.104862</u>
- Weitensfelder, L., Hutter, H.-P., Lemmerer, K., Poteser, M., Wallner, P., and Moshammer, H. (2023). "Health Problems in the European Alps Under Climate Change", in Oxford Research Encyclopedia of Climate Science. <u>https://doi.org/10.1093/acre-fore/9780190228620.013.828</u>
- Weitensfelder, L., and Moshammer, H. (2020). Evidence of Adaptation to Increasing Temperatures. International Journal of Environmental Research and Public Health 17, 97. <u>https://doi.org/10.3390/ijerph17010097</u>
- Weitensfelder, L., Moshammer, H., Öttl, D., and Payer, I. (2019). Exposure-complaint relationships of various environmental odor sources in Styria, Austria. *Environ Sci Pollut Res* 26, 9806– 9815. <u>https://doi.org/10.1007/s11356-019-04410-z</u>
- Wenzel, W. W., Philipsen, F. N., Herold, L., Kingsland-Mengi, A., Laux, M., Golestanifard, A., et al. (2023). Carbon sequestration potential and fractionation in soils after conversion of cultivated land to hedgerows. *Geoderma* 435, 116501. <u>https://doi.org/10.1016/j.geoderma.2023.116501</u>
- Wessely, J., Essl, F., Fiedler, K., Gattringer, A., Hülber, B., Ignateva, O., et al. (2024). A climate-induced tree species bottleneck for forest management in Europe. *Nat Ecol Evol* 8, 1109–1117. <u>https://doi.org/10.1038/s41559-024-02406-8</u>
- Westhoek, H., Lesschen, J. P., Rood, T., Wagner, S., De Marco, A., Murphy-Bokern, D., et al. (2014). Food choices, health and environment: Effects of cutting Europe's meat and dairy intake. *Global Environmental Change* 26, 196–205. <u>https://doi.org/10.1016/j.gloenvcha.2014.02.004</u>
- WHO (2015). IARC Monographs evaluate consumption of red meat and processed meat. Available at: <u>https://www.iarc.who.</u> <u>int/wp-content/uploads/2018/07/pr240\_E.pdf</u>
- Wieland, M., and Martinis, S. (2020). Large-scale surface water change observed by Sentinel-2 during the 2018 drought in Germany. *International Journal of Remote Sensing* 41, 4742–4756. <u>https://doi.org/10.1080/01431161.2020.1723817</u>
- Wilkins, E. J., Chikamoto, Y., Miller, A. B., and Smith, J. W. (2021). Climate change and the demand for recreational ecosystem services on public lands in the continental United States. *Global Environmental Change* 70, 102365. <u>https://doi.org/10.1016/j.gloenvcha.2021.102365</u>
- Wilkinson, A., and Woodcock, A. (2022). The environmental impact of inhalers for asthma: A green challenge and a golden opportunity. *British Journal of Clinical Pharmacology* 88, 3016– 3022. <u>https://doi.org/10.1111/bcp.15135</u>
- Willett, W., Rockström, J., Loken, B., Springmann, M., Lang, T., Vermeulen, S., et al. (2019). Food in the Anthropocene: the EAT-Lancet Commission on healthy diets from sustainable food systems. *The Lancet* 393, 447–492. <u>https://doi.org/10.1016/ S0140-6736(18)31788-4</u>
- Winiwarter, W., Höglund-Isaksson, L., Klimont, Z., Schöpp, W., and Amann, M. (2018). Technical opportunities to reduce glob-

al anthropogenic emissions of nitrous oxide. *Environ. Res. Lett.* 13, 014011. <u>https://doi.org/10.1088/1748-9326/aa9ec9</u>

- Winkel, G., Lovrić, M., Muys, B., Katila, P., Lundhede, T., Pecurul, M., et al. (2022). Governing Europe's forests for multiple ecosystem services: Opportunities, challenges, and policy options. *Forest Policy and Economics* 145, 102849. <u>https://doi.org/10.1016/j. forpol.2022.102849</u>
- Wirth, C., Messier, C., Bergeron, Y., Frank, D., and Fankhänel, A. (2009). "Old-Growth Forest Definitions: a Pragmatic View", in *Old-Growth Forests*, eds. C. Wirth, G. Gleixner, M. Heimann, C. Wirth, G. Gleixner, and M. Heimann (Berlin, Heidelberg: Springer), 11–33. Available at: <u>http://link.springer.com /10.1007/978-3-540-92706-8\_2</u> (Accessed May 15, 2024).
- Wolfslehner, B., and Vacik, H. (2011). Mapping indicator models: From intuitive problem structuring to quantified decision-making in sustainable forest management. *Ecological Indicators* 11, 274–283. <u>https://doi.org/10.1016/j.ecolind.2010.05.004</u>
- Wolkinger, B., Haas, W., Bachner, G., Weisz, U., Steininger, K. W., Hutter, H.-P., et al. (2018). Evaluating Health Co-Benefits of Climate Change Mitigation in Urban Mobility. *International Journal of Environmental Research and Public Health* 15, 880. <u>https://doi.org/10.3390/ijerph15050880</u>
- Wong-Parodi, G., and Feygina, I. (2021). Engaging People on Climate Change: The Role of Emotional Responses. *Environmental Communication* 15, 571–593. <u>https://doi.org/10.1080/17524032</u>. 2020.1871051
- Xu, L., and Penner, J. E. (2012). Global simulations of nitrate and ammonium aerosols and their radiative effects. *Atmospheric Chemistry and Physics* 12, 9479–9504. <u>https://doi.org/10.5194/</u> acp-12-9479-2012
- Yadav, M., and Agarwal, M. (2021). Biobased building materials for sustainable future: An overview. *Materials Today: Proceedings* 43, 2895–2902. <u>https://doi.org/10.1016/j.matpr.2021.01.165</u>
- Yip, C. S. C., Crane, G., and Karnon, J. (2013). Systematic review of reducing population meat consumption to reduce greenhouse gas emissions and obtain health benefits: effectiveness and models assessments. *Int J Public Health* 58, 683–693. <u>https://doi.org/10.1007/s00038-013-0484-z</u>
- York, R. (2012). Do alternative energy sources displace fossil fuels? *Nature Clim Change* 2, 441–443. <u>https://doi.org/10.1038/</u> nclimate1451
- Youssouf, H., Liousse, C., Roblou, L., Assamoi, E.-M., Salonen, R. O., Maesano, C., et al. (2014). Non-Accidental Health Impacts of Wildfire Smoke. *International Journal of Environmen*tal Research and Public Health 11, 11772–11804. <u>https://doi.org/10.3390/ijerph111111772</u>
- Zaller, J., Oswald, A., Wildenberg, M., Burtscher-Schaden, H., Nadeem, I., Formayer, H., et al. (2013). Potential to reduce pesticides in intensive apple production through management practices could be challenged by climatic extremes. *Science of The Total Environment* 872, 162237. <u>https://doi.org/10.1016/j.scitotenv.2023.162237</u>
- ZAMG (2020). Klimaänderungssignale Trockenheit für die Land-, Wasser- und Forstwirtschaft um Bundesland Salzburg. Available at: <u>https://www.salzburg.gv.at/umweltnaturwasser\_/Documents/Umwelt/20201130\_LandSBG\_PlanungstoolTrockenheit\_EndberichtZAMG.pdf</u>
- Zeide, B. (2004). Optimal stand density: a solution. *Can. J. For. Res.* 34, 846–854. <u>https://doi.org/10.1139/x03-258</u>
- Zeilinger, J., Niedermayr, A., Quddoos, A., and Kantelhardt, J. (2023). Identifying under-adaptation of farms to climate change.

*European Review of Agricultural Economics* 50, 497–528. <u>https://doi.org/10.1093/erae/jbac025</u>

- Zellweger, F., De Frenne, P., Lenoir, J., Vangansbeke, P., Verheyen, K., Bernhardt-Römermann, M., et al. (2020). Forest microclimate dynamics drive plant responses to warming. *Science* 368, 772–775. https://doi.org/10.1126/science.aba6880
- Zinn, K. E., Tunc-Ozdemir, M., and Harper, J. F. (2010). Temperature stress and plant sexual reproduction: uncovering the weakest links. *Journal of Experimental Botany* 61, 1959–1968. <u>https:// doi.org/10.1093/jxb/erq053</u>
- Zoderer, B. M., Lupo Stanghellini, P. S., Tasser, E., Walde, J., Wieser, H., and Tappeiner, U. (2016). Exploring socio-cultural values of ecosystem service categories in the Central Alps: the influence of socio-demographic factors and landscape type. *Reg Environ*

*Change* 16, 2033–2044. <u>https://doi.org/10.1007/s10113-015-</u>0922-y

- Zu, K., Wang, Z., Zhu, X., Lenoir, J., Shrestha, N., Lyu, T., et al. (2021). Upward shift and elevational range contractions of subtropical mountain plants in response to climate change. *Science* of *The Total Environment* 783, 146896. <u>https://doi.org/10.1016/j. scitotenv.2021.146896</u>
- Zulka, K. P., and Götzl, M. (2015). "Ecosystem Services: Pest Control and Pollination", in *Economic Evaluation of Climate Change Impacts: Development of a Cross-Sectoral Framework and Results for Austria*, eds. K. W. Steininger, M. König, B. Bednar-Friedl, L. Kranzl, W. Loibl, F. Prettenthaler, et al. (Cham: Springer International Publishing), 169–189. <u>https://doi.org/10.1007/978-3-319-12457-5\_10</u>




# Chapter 3 Built environments and mobility



### Second Austrian Assessment Report on Climate Change | AAR2

### Chapter 3 Built environments and mobility

This publication was subject to an anonymous international peer review process guided by several review editors, the AAR2 co-chairs and the APCC Steering Committee.

The views or opinions expressed in this publication are not necessarily those of the institutions with whose support this work was carried out. The publisher and the supporting institutions do not guarantee the accuracy or continued availability of urls to external websites or third-party websites mentioned in this publication and accept no responsibility for the accuracy or appropriateness of any content contained therein.

#### **Recommended citation**

S. Peer, W. Streicher, G. Wagner, et al. (2025): Built environments and mobility.
In "Second Austrian Assessment Report on Climate Change (AAR2) of the Austrian Panel on Climate Change (APCC)".
[D. Huppmann, M. Keiler, K. Riahi, H. Rieder (eds.)]. Austrian Academy of Sciences Press, Vienna, Austria | doi: doi.org/10.1553/aar2-ch3

You can download the recommended citation in the formats Endnote (.ris) and BibTex (.bib) at <u>aar2.ccca.ac.at/citation</u>.

Vienna, June 2025 ISBN: 978-3-7001-5110-4 | <u>austriaca.at/5110-4</u> (full report)

#### Production

Publisher: Austrian Academy of Sciences Press, Vienna Typesetting: Crossdesign Werbeagentur GmbH, Graz Cover: paulaidaanton | Agentur für digiloges Design Graphic design Figure 3.2: paulaidaanton | Agentur für digiloges Design Al-based tools such as DeepL were used to support the linguistic revision.

Contact address for product safety: Austrian Academy of Sciences Press Dr. Ignaz Seipel-Platz 2 A-1010 Wien Tel.: +43-1-51581-3420 E-Mail: verlag@oeaw.ac.at

#### Funding acknowledgement

The AAR2 was funded by the Climate and Energy Fund under the 14th Austrian Climate Research Program (ACRP).

#### **Copyright and license**

© 2025 Austrian Panel on Climate Change (APCC) This report is published under a Creative Commons CC-BY 4.0-License. <u>creativecommons.org/licenses/by/4.0</u>





The AAR2 is a report by the Austrian Panel on Climate Change (APCC). https://aar2.ccca.ac.at/en

#### **Coordinating Lead Authors**

Stefanie Peer Wolfgang Streicher Gernot Wagner

#### **Lead Authors**

Wolfgang Amann Felix Creutzig Manfred Gronalt Holger Heinfellner Bernhard Hofko Stefan Kienberger Manfred Kleidorfer Lukas Kranzl Kerstin Krellenberg Matthias Landgraf Tajda Obrecht Alexander Passer Franziska Sielker Vincent van Karsbergen

#### **Contributing Authors**

Antonia Golab Michael Gruber Johannes Horak Marcus Hummel Christoph Loschan Philipp Mascherbauer Andreas Müller Nirav Patel Alexandra Pintilie

#### **Chapter Scientist**

Vincent van Karsbergen

#### **Review Editors**

Jörn Birkmann David McCollum University of Stuttgart Oak Ridge National Laboratory

University of Innsbruck | UIBK

#### Liaisons from AAR2 Technical Support Unit

Anika Donner	University of Innsbruck   UIBK
Kay Helfricht	University of Innsbruck   UIBK

Vienna University of Economics and Business | WU University of Innsbruck | UIBK Columbia Business School, New York

Institute for Real Estate, Construction and Housing, Vienna | IIBW Potsdam Institute for Climate Impact Research | PIK BOKU University Environment Agency Austria | UBA Technical University Vienna | TU Wien GeoSphere Austria University of Innsbruck | UIBK Technical University Vienna | TU Wien University of Vienna Evias Rail & Graz University of Technology | TU Graz Graz University of Technology | TU Graz Graz University of Technology | TU Graz Technical University Vienna | TU Wien University of Technology | TU Graz

Technical University Vienna | TU Wien Technical University Vienna | TU Wien Stadtklimatologie und Umwelt, Magistrat Linz e-think energy research Technical University Vienna | TU Wien e-think energy research Technical University Vienna | TU Wien Technical University Vienna | TU Wien

### **TABLE OF CONTENTS**

Exec	cutive	summary	175			
3.1.	Chapter introduction					
3.2.	Cities, settlements, and spatial planning					
	3.2.1.	Cities and settlements	178			
	Chapt	er Box 3.1				
	Clima	te change adaptation plan 'Zukunft Linz'				
	('Futu	re Linz')	180			
	3.2.2.	Land-use and spatial planning	181			
		Strategic context in Austria	181			
		Key development trends and risk factors	181			
		Reducing land consumption in times of				
		population and economic growth	182			
		Spatial energy and infrastructure				
		planning	185			
	3.2.3.	Climate impacts and risks	186			
		Urban heat islands	186			
		Urban water impacts	187			
		Integrated mitigation and adaptation				
		strategies	188			
3.3.	Buildi	ngs	188			
	3.3.1.	Statistics and scenarios of technologies				
		and renewables in buildings	188			
		Current status	188			
		Greenhouse gas reduction and climate				
		change mitigation scenarios	190			
		Improve, Shift and Avoid measures in the building sector	192			
	Chapt	er Box 3.2				
Energy storage in buildings – technologies						
	and n	eeds	193			

	3.3.2.	Construction products 1	94				
	3.3.3. Legal reforms – barriers and						
		perspectives 1	96				
		Impact of EU legislation 1	96				
		Existing legal regulations and measures. 1	96				
		Pending legal reforms 1	97				
		Preconditions for achieving building					
		decarbonization by 2040 1	98				
3.4.	Trans	port and infrastructure 1	98				
	3.4.1.	Introduction 1	98				
	3.4.2.	Person mobility 2	.01				
		Status quo and trends 2	01				
	Chapt	ter Box 3.3					
	Flexib	ility potential of electro-mobility in					
	Austri	a2	03				
		Mitigation strategies 2	.04				
		Scenarios	.05				
		Governance and acceptability 2	.05				
	3.4.3.	Freight transport 2	.06				
		Status quo and trends 2	.06				
		Mitigation strategies 2	.08				
		Scenarios 2	.09				
	3.4.4.	Transport infrastructure 2	.09				
		Railway infrastructure 2	10				
		Road infrastructure 2	11				
		Risks and adaptation possibilities 2	12				
Con	clusior	າ2	14				
Refe	erences	s 2	15				
Glos	sary	aar2.ccca.ac.at/glossa	ary				
List	of abbr	eviations <u>aar2.ccca.ac.at/abbreviation</u>	ons				
Full	report.	<u>aar2.ccca.ac.at/rep</u>	ort				

#### **EXECUTIVE SUMMARY**

Buildings and transportation provide essential services, while direct energy use accounts for 36 % and 34 % of Austria's total end-use energy consumption in 2023, contributing 9 % and 29 % of total national greenhouse gas (GHG) emissions, respectively (*high confidence*). Including indirect GHG emissions from district heating and electricity used in buildings and mobility (e.g., heat pumps, electric vehicles) in their respective sectors would increase these shares, accounting for 3.4 % (buildings) and 0.3 % (transport) of the total national emissions in 2022 (*high confidence*). {3.3.1, 3.4.1}

Sustainable cities, villages, and settlements feature short distances for work, leisure, schools, and daily necessities as well as appropriate infrastructure for diverse modes of active mobility. Settlement areas have significant potential for GHG emissions savings, with urban households emitting about one-half to one-third of suburban ones due to smaller physical footprints, shorter distances traveled, and lower car dependency (*high confidence*). 'Urban' and 'rural' here align along a continuum, with continued suburbanization representing the most direct climate policy challenge. This points to the need to avoid green-field development in favor of maintaining and fostering compact settlements (*high confidence*). {3.2, 3.4.2}

Compact urban development and densification can be achieved by improved utilization of existing planning instruments and the development of strong multi-level governance and coordination (*high confidence*). Challenges include urban heat island (UHI) effects and preferences for single-family homes and car ownership (*high confidence*). Addressing these deeply ingrained lifestyles – rooted in prevailing social norms and perceived convenience – is a crucial challenge for advancing decarbonization in the housing and transport sector (*high confidence*). {3.2, 3.4.2}

Adapting Austria's built environment to climate change entails prioritizing resilient water management, greening initiatives for local ambient air cooling and water retention, protection of infrastructure from landslides and flooding as well as robust information, communication and early warning systems (*high confidence*). This necessitates integrating climate considerations into spatial planning as well as building design guidelines and standards, including adaptable building codes and infrastructure regulations (*high confidence*). Localized limits to adaptation may arise under extreme warming scenarios (*low confidence*). {3.2.3, 3.3.3, 3.4.4}

Absolute building GHG emissions decreased by about 43 % between 1990 and 2022, despite increases in the number and size of housing units (*high confidence*). This decoupling was due to rapid deployment of renewable energy sources and district heating, better insulated buildings and less need for heating due to higher ambient temperatures (*high confidence*). {3.3.1}

Main pathways for further reduction of building-sector GHG emissions include replacement of oil and gas burners by renewable energy driven systems, implementing airheat recovery in mechanical ventilation systems, building retrofitting by further improving insulation levels, adopting circular design principles and the use of low-emission construction materials - while considering embedded environmental impacts across their entire lifecycle (high confidence). Main uncertain countervailing factors stem from potential population growth - according to official Austrian projections - and a potential shortage of installers qualified to deploy transition technologies based on renewables. Climate change is projected to decrease heating and moderately increase cooling demand in buildings - reducing net energy consumption and increasing the need for combined heating and cooling systems (adaptation) (high confidence). {3.3.1, 3.3.2}

Electrification of buildings and road transport – with heat pumps and electric vehicles pivotal – will lead to a dominant role for electricity as the main source of energy in both sectors (*high confidence*). Supporting policies, regulations, and infrastructure investments are expected to accelerate this transition (*high confidence*). {3.3.1, 3.4.2, 3.4.3}

Road transport accounts for 99 % of GHG emissions in the Austrian transport sector (excluding international flights), with freight transport contributing 35 % in 2023 (*high confidence*). After peaking in 2005, GHG emissions in transport saw a steady rise from 2012 onward. Substantial reductions occurred only in recent years, beginning in 2020, largely driven by external factors such as the COVID-19 pandemic and Russia's invasion of Ukraine. Nevertheless, in 2023, transport-related emissions remained 42 % above the 1990 level (*high confidence*). {3.4.1, 3.4.3} An expected increasing demand for freight transport, combined with limited potential for rail modal shifts, and a still limited availability of competitive low-carbon trucking alternatives render it difficult to achieve full transport sector decarbonization by 2040 (*medium evidence, high agreement*). {3.4.1, 3.4.3}

A combination of 'pull' and 'push' measures provides the most effective strategy for decarbonizing transport and housing. Push measures, including higher fuel taxes or bans on fossil-fuel-based heating systems, are essential for creating disincentives for carbon-intensive choices. Pull measures - such as affordable public transport, improved cycling infrastructure, and incentives for building renovations or switching from fossil-fuel heating to renewable alternatives - further encourage behavioral change by improving the attractiveness of low-carbon choices. Pull measures can also help in improving the public acceptability of push measures, in particular by mitigating potential negative distributional impacts of push measures (medium confidence). Aligned with the ASI (Avoid/Shift/ Improve) framework, these measures can reduce demand for emission-intensive activities and construction (avoid), promote sustainable alternatives like public transport and heat pumps (shift), and improve efficiency through technological advances, such as electric vehicles, deep building retrofits, circular design and the utilization of low-emission materials – while considering embedded lifecycle impacts (improve) (*medium confidence*). {3.3.1, 3.3.2, 3.3.3, 3.4.2, 3.4.3, Cross-Chapter Box 4}

Reduced dependence on motorized transport has multiple co-benefits and positive spillover effects: Lower energy and resource requirements for vehicle production and recycling, improved traffic safety, less noise and local pollution, improved public space quality due to parking and road space freed up for alternative uses (e.g., green space), and improved health through the use of active mobility (*high confidence*). {3.4.1, 3.4.2, 3.4.3}

GHG emissions from constructing, maintaining and renovating buildings and infrastructure present a major challenge (*high confidence*). Key levers to drive reductions are revised building codes, infrastructure regulations (e.g., minimum parking requirements) as well as tendering and procurement policies that take embedded emissions and circular design approaches into account (*medium confidence*). {3.3.1, 3.3.2, 3.3.3, 3.4.4}

#### 3.1. Chapter introduction

Climate change affects the built environment and transport, both via long-term trends (e.g., rising temperatures) and via extreme weather events like floods, droughts, and storms. Spatial planning, transport and building infrastructure are thus relevant for adapting to climate change and also play a crucial role in cutting greenhouse gas (GHG) emissions (Creutzig et al., 2015; Haberl et al., 2023). This requires a reconsideration of traditional village and urban configurations on the one hand, and building codes and mobility cultures on the other hand.

While assessing the sectors fundamental to the built environment and mobility in the context of climate change and sustainability, Figure 3.1 outlines the various, often interconnected disciplines considered. Hence, this chapter starts with a macro view of the impact of cities and settlements on the climate, and vice versa, as well as the potential to use spatial planning to reduce GHG emissions (Section 3.2). After detailing key developments and projections (Section 3.2.1), the chapter focuses on reducing land consumption in times of population and economic growth (Section 3.2.2), an important lever to lessen overall climate impacts and exposures, and impacts and risks of climate change within urban environments (Section 3.2.3). Section 3.3 focuses on buildings and their past, present, and future climate impacts and exposures from the perspectives of building technology and renewables (3.3.1), construction material (3.3.2) and legislation (3.3.3). Section 3.4 does the same for the passenger (3.4.2) and freight transport sector (3.4.3), as well as the related road and rail infrastructure (3.4.4). Notably for infrastructure, adaptation to climate change plays a crucial role. The spatial structure shapes behavioral choices, in particular with respect to buildings and mobility. Accordingly, urban and regional spatial planning, buildings and transport systems are closely related to demand-side climate change mitigation. Therefore, it is critical to understand household-level GHG emissions-footprints as linked to broader land-use, building, and mobility choices.

Chapter 3 is closely linked to the other chapters of the report. It builds on the estimates of Chapter 1 regarding changes in the frequency of extreme weather events in order to assess potential adaptations to climate change in spatial planning, buildings, and transport (infrastructure). The emphasis on spatial planning in Chapter 3 has implications for land available for other uses (agriculture, forests, etc.), which are the focus of Chapter 2. Chapter 3 and Chapter 4 are particularly related via the energy sector, as electrification is a



Figure 3.1 Disciplines involved in the study of climate change and solutions, specifically for climate change mitigation and adaptation, as considered in this chapter. Each discipline adds its own perspective and limits to the analysis. Given the complexity and open system nature of urban issues, it is necessary to consider and move between different perspectives.

main strategy for the decarbonization of the building and transport sectors. Further links exist to other topics of Chapter 4, including the volume of freight transport correlating closely with economic activities, tourism and work being a main source of travel demand, or teleworking as a form of work that does not require daily commuting. Links to Chapter 5 arise from demand patterns and lifestyle choices that impact the desirability and acceptance of different forms of housing and mobility behavior, and of different types of spatial surroundings; the distributional impacts of buildings, mobility, and spatial planning policies link these two chapters. Following the understanding of the recent IPCC report (Creutzig et al., 2022; IPCC, 2022b), demand-side mitigation includes both socio-behavioral shifts (the lifestyle domain of Chapter 5 of this report) and physical infrastructures as major modifiers enabling or prohibiting certain lifestyles (the domain of this chapter). Due to the involvement of different governmental levels (EU, national, provincial, municipal, local) in spatial planning, buildings, and transport, Chapter 6 provides an important governance/political context to Chapter 3. Chapter 7, meanwhile, intersects with Chapter 3 along multiple dimensions, including spatial and infrastructure planning, mobility, and buildings, as they are related to mountain regions. With its overarching focus on transformation pathways, Chapter 8 encompasses sectoral scenarios in particular for buildings and transportation.

# 3.2. Cities, settlements, and spatial planning

This section evaluates how urban areas and spatial planning influence climate change mitigation and adaptation. Section 3.2.1 discusses trends and projections in urban development, while Section 3.2.2 explores land-use strategies for reducing greenhouse gas (GHG) emissions. Section 3.2.3 focuses on climate risks to cities and settlements, addressing the resilience of infrastructure and adaptive capacity. Additionally, Chapter Box 3.1 highlights the 'Zukunft Linz' climate adaptation plan as a case study in urban resilience.

#### 3.2.1. Cities and settlements

Cities and urban settlements are both impacted by and likewise directly contribute to climate change, via a host of interwoven demographic, economic, land use and infrastructural changes (Rosenzweig et al., 2010; Weichselbaumer et al., 2022). Worldwide, roughly 75 % of global GHG emissions come from urban and suburban settlements, attributable primarily to residential buildings, followed by commercial/industrial buildings, transport, and waste/sewage (e.g., Satterthwaite, 2008; Hickman and Banister, 2015).

Meanwhile, both Europe and Austria are experiencing continued urbanization: In Europe approximately 75 % live in urban settlements with more than 10,000 inhabitants today (UN DESA, 2019), a figure that is expected to rise to 83.7 % by 2050. Austria represents a lower rate of urbanization, at 59.2 % of its 8.7 million inhabitants (in 2019), expected to rise to 71 % in 2050, according to UN DESA (2019). Recent statistics show that the total population of Austria has increased to 9.2 million in 2024 (Statistik Austria, 2024b).

Further analysis points to the importance of a nuanced view of urbanization, e.g., drawing clear distinctions between the dense urban core with multi-family housing and efficient, multimodal transport infrastructure, and the continuing suburbanization at the urban periphery of larger cities and between cities and small(er) settlements, often with single-family housing and limited mobility choices (Steinegger, 2023). This increasingly pressures local administrations to ensure adequate, affordable, and climate-resilient housing while simultaneously expanding infrastructure to support more climate-friendly mobility (see Section 3.2.3). Divergent regional population growth rates call for detailed plans and planning instruments for both mitigation and adaptation, while also necessitating cross-regional coordination and management. The long-term nature of spatial planning creates particular urgency, given Austria's climate neutrality goal for 2040.

Climate change affects cities and urban settlements most directly via intensity and frequency of extreme weather events, including heavy rain fall, droughts, and heat (see also Section 1.2). Heat events are becoming more frequent in Austria (see Figure 3.2 for increase in hot days, i.e., with a daily maximum temperature over 30°C). This is of particular significance for the built environment, exacerbated by urban heat island (UHI) effects (Oke and Fuggle, 1972). Meanwhile, cold stress in winter is reduced by increasing temperature, as, e.g., illustrated in Figure 3.3, showing historic trends in hot and ice days in Vienna (see Section 3.3.1 and Section 4.5.2 for the impacts on space heating and cooling). These phenomena are exacerbated by continued surface sealing and loss of open and green space, which impairs natural cooling and water retention functions. There is wide regional variation in exposure, susceptibility, and coping capacities both across and within cities, especially those with



**Figure 3.2** Hot days yearly average in the nine provincial capitals measured in the past decades and forecasted for two different climate scenarios (with and without global climate protection measures, i.e., GWL 2.0°C and GWL 4.0°C, resp.) (GeoSphere Austria, 2022).

high socio-spatial differentiation, e.g., access to public transport and green spaces (Krellenberg et al., 2017). In Vienna, for example, socio-spatial inequalities have increased in the context of rising labor inequality and an uneven restructuring of the urban housing market (Kadi et al., 2022). There is some evidence of 'green gentrification', referring to the fact that the renaturation of urban spaces can lead to rising housing prices and socio-demographic upgrading, and thereby to the (in)direct displacement of underprivileged groups, who already suffer most from environmental injustices, in the private rental segment (Friesenecker et al., 2023).

Overall, detailed geospatial information on urban vulnerability in Austria is limited, with some exceptions, e.g., Linz (see Chapter Box 3.1) and Vienna (e.g., Weatherpark GmbH, 2021, with a grid resolution of 10 m). The 'Urban Heat Island Strategy City of Vienna' (Magistrat der Stadt Wien, 2015) demonstrates various actions and their implementation that reduce heat in the summer months, including strategic and technical measures, to improve microand neighborhood climates. The 'Heat Action Plan' (Stadt Wien, 2022) identifies 'vulnerable' people and groups, such as elderly and socially isolated people, people in need of care, people with chronic care-dependency, and people with chronic or mental illnesses, pregnant women, young children, or people living and working in particularly difficult conditions. The change adaptation in 'Graz Action Plan' (Stadt Graz, 2018) defines action fields but does not address vulnerable areas or place-based adaptation measures.

Making Austrian cities and the broader built environment 'climate-fit' involves both climate adaptation and mitigation efforts which are highly interwoven and need both be considered in integrated urban planning and development processes. Strong linkages exist to both, the transport sector (Section 3.4) and the building sector (Section 3.3), in terms of retrofitting existing structures and buildings but also to make efficient use of public space in terms of de-sealing, multifunctional uses and greening, which likewise enhance the quality of life in a broader sense. In addition, new development projects should be planned and designed from the ground up to be climate-fit by implementing integrated sustainable development approaches, thus contributing to the necessary sustainable urban transformations. Overall, compact cities with reduced mobility needs through short-



**Figure 3.3** Frequency of hot days (maximum daily temperature exceeding 30°C) and ice days (maximum daily temperature below 0°C) recorded in Vienna (Stadt Wien, 2024).

er distances (e.g., the '15-minute city' concept) and more inclusive planning and design of public space (e.g., the 'superblock' approach) to support multiple uses of open space, will allow mitigation and adaptation measures to be considered together, supporting the fundamental transformations needed in urban development and spatial planning (WBGU – Wissenschaftlicher Beirat der Bundesregierung Globale Umweltveränderungen, 2011, 2016).

#### Chapter Box 3.1. Climate change adaptation plan 'Zukunft Linz' ('Future Linz')

'Zukunft Linz' ('Future Linz') forms the basis for climate change adaptation in Linz (Magistrat Linz, 2023), based on detailed, spatially disaggregated risk assessments linked to exposure and vulnerability, taking into account adaptive capacity (Box 3.1 Figure 1). The map shows regions of high priority for heat-related adaptation measures. Additionally, exposure maps exist for hail, fluvial floods, and storms, allowing for the prioritization of adaptation efforts and risk monitoring (BML, 2023). The expertise is housed within Linz's urban climatology and environmental management (UCEM) division, with an urban climatologist and two additional climate change adaptation staff members who help prioritize and coordinate city-wide adaptation efforts across city departments. This includes, e.g., planning for heat events negatively affecting public health and thermal comfort. For example, Linz's medium- to long-term projects include a greening initiative focused on tree planting to address UHI effects, a stream renaturation program to restore natural habitats and prevent flooding, and a heat emergency plan based on spatially disaggregated risk events (Box 3.1 Figure 1).



Box 3.1 Figure 1 Excerpt of the heat risk map of Linz focused on the districts 'Innere Stadt' and 'Bulgariplatzviertel', showing spatially resolved vulnerabilities and adaptive capacities (Horak and Peßenteiner, 2023). Vulnerability considers residents' age, pre-existing medical conditions, and socio-economic status. Adaptive capacity is based on publicly accessible urban green areas >7,500 m<sup>2</sup> and tree cover within walking distance of at most 250 m.

#### 3.2.2. Land-use and spatial planning

#### Strategic context in Austria

Land-use planning takes account of and aims to influence spatial development trends. In 2021, the office of the Austrian Conference on Spatial Planning (ÖROK) published the 'ÖREK 2030', the Austrian Spatial Development Concept for Austria (ÖROK, 2021), elaborated with input by stakeholders from all governmental levels including the state, the federal provinces, districts and municipalities. The strategy is the most important federal document providing a joint vision for future development foci. The ÖREK 2030 outlines the key themes of need for urban and regional transformation for the next 10 years in four pillars (ÖROK, 2021):

- A parsimonious and sparing use of spatial resources;
- The strengthening of the social and spatial cohesion;
- The sustainable and climate sensitive development of economic spaces and systems;
- The development of vertical and horizontal governance.

Land-use planning remains a key tool for enhancing Austria's resilience and adaptive capacity while guiding contemporary development trends, such as multilocality, connectivity, mobility, land use, economic activities, digitalization, and housing. It plays a crucial role in both preventing additional emissions and enabling effective adaptation and mitigation measures (ÖROK, 2021).

The governance of and legal systems in Austria imply that these overall goals can only be achieved through efficient vertical and horizontal governmental coordination (see Section 6.4). There are three main levels: The federal government is only legally responsible for the development of sectoral plans and, through the ÖROK, for the development of the spatial development strategy. Due to the Austrian Constitution, which grants the right to develop their own laws in any policy area that is not regulated at the federal level, the nine federal provinces have the right to develop spatial development laws ('Raumordnungsgesetze') in the absence of a federal spatial planning regulation (BGBl. Nr. 1/1930, 2024, art. 15). As a result, each of the nine federal provinces has its own legal framework. In general, the federal level remains with competences to outline strategy documents as well as sectoral plans. Further, the federal provinces have the obligation to provide the frameworks and legal specifications for local land-use regulation (Gruber et al., 2018). In line with the subsidiarity principle, the right for legal instruments linked to land-use regulation lies with the municipal level. The municipalities in Austria have the responsibility to develop strategic development concepts, zoning plans, and/or building regulation plans (BGBl. Nr. 1/1930, 2024, art. 118). In some federal provinces, development plans and strategies may also be developed at the regional level.

This multi-level governance framework implies that the federal government only can act through strategic guidelines to address many of Austria's developmental changes. To address certain developmental challenges, such as land take, the ÖROK can devise implementation plans - for instance the Land Protection Strategy for Austria ('Bodenstrategie für Österreich') (ÖROK, 2023b). Recently, the limitations of the multi-level governance system have become evident: Although the federal provinces - who hold the legal authority have adopted the Land Protection Strategy, the final version omits the commitment to limit land take to a maximum of 2.5 hectares per day, as originally outlined in the 2020-2024 government program. This case illustrates the complexity of tackling critical challenges in spatial planning within a multi-level governance context. Effective land use monitoring and planning are essential for identifying areas that are exposed and vulnerable to, or likely to be impacted by, climate change-related extreme events, such as severe droughts or intense rainfalls causing floods (see also Cross-Chapter Box 1). Land use planning plays an important role in managing spatial development in a sustainable way.

#### Key development trends and risk factors

Overall, climate change is associated with more extreme weather events (IPCC, 2021a), an increase in the number of hot days, and a shift in seasonal extremes (Fuchs et al., 2015, 2022) (see Section 3.2.1). Impacts differ between urban and rural areas, and between lowland and mountainous areas, which is particularly relevant for Austria (see Section 1.2.2). For example, an increase in extreme rainfall events after long periods of drought can have potentially devastating impacts on agricultural practices in mountainous areas (see Sections 2.2.1, 7.4.2 and Table 2.2) and an increased occurrence of avalanches puts more settlements at risk (Permanent Secretariat of the Alpine Convention, 2017).

The projected increase of annual mean temperatures also suggests that several European regions, especially Austria, will be particularly affected (IPCC, 2022a). The increased amplitude of phenomena associated with extreme hydro-meteorological or compound events, as well as their increased number, can have devastating impacts on existing economic systems, social justice, human health, existing infrastructure, and the built environment in general (see also Cross-Chapter Box 1 and the Cross-Chapter Box 2). In addition to these meteorological developments, soil sealing exacerbates the negative impacts of heat waves, particularly in urban areas, by increasing the UHI effect (see Section 3.2.3).

Demographic development trends, such as ageing regions, do not necessarily coincide with land consumption patterns. The ÖROK estimates a population growth of 3.2 % to 9.22 million inhabitants in 2030 compared to 2021 (ÖROK, 2021). This growth is not equally distributed. On the contrary, some Austrian regions, especially urban areas, continue to grow while others continue to shrink: The ÖREK 2030 estimates that the population of the Vienna Region will grow by more than 15 % by 2040, while some regions, such as the Lavanttal, will shrink by around 10 % (ÖROK, 2021).

It is striking that land consumption continues even in areas with declining population (Dallhammer et al., 2022). While some regions are experiencing shrinkage, Austria's overall population continues to grow, creating demand for additional housing units in urban areas. In rural areas, this new demand is being met primarily with single-family and two-family homes, mostly detached or semi-detached. In urbanized areas, apartment buildings are the norm. These housing demands, as well as industrial developments, result in new soil sealing.

At the same time, demographic development trends will lead to a higher percentage of older people in many regions of Austria. The ÖROK regional forecast up to January 2050 suggests a continuous trend towards an ageing population, especially in rural areas (ÖROK, 2023c). Figure 3.4 illustrates these ageing trends occurring in mountainous areas and rural Austria.

# Reducing land consumption in times of population and economic growth

The European Commission called for target of No-Net-Land-Take by 2050, which was concretized in 2021 with the Soil Protection Strategy for 2030 (European Commission,



#### Population change 2021-2050: Age 65+

Origin of data: ÖROK Regionalprognosen 2021 - Bevölkerung, Statistik Austria - data.statistik.gv.at 2024, European Union's Copernicus Land Monitoring Service 2020 Administrative boundaries: ÖROK Regionalprognosen 2021 - Bevölkerung, Eurostat 2020 und 2021, BEV 2023 (CC BY 4.0); Digitales Landschaftsmodell: BEV 2023 Cartographic implementation: Franziska Sielker and Alexandra Pintilie, TU Wien 2024

Figure 3.4 Projected regional population changes in Austria from 2021 to 2050, focusing on individuals aged 65 and older, based on the regional forecast from ÖROK (Sielker and Pintilie, 2024, based on ÖROK, 2021).

2022c). By 2023, Member States were required to set their objectives for reducing land take. In 2022, the ÖROK introduced a new methodology to measure land take and soil sealing in Austria (ÖROK, 2023d).

In parallel, both on the EU level and in Austria, a process is underway to provide conceptual clarity on the difference between land take and soil sealing. The term 'land take' in the sense of 'land consumption' includes built-up areas as well as urban green areas and land used for agricultural, forestry, or other economic activities. The term 'soil sealing' or 'imperviousness' refers to the situation in which the nature of the soil changes in a way that it becomes an impermeable medium (Marquard et al., 2020; Decoville and Feltgen, 2023). Land take, including buildings for housing, infrastructure, recreational activities as well as cultivation areas, involves the long-term loss of biologically productive soil.

Figure 3.5 visualizes the degree of soil sealing and potential permanent settlement areas. Especially in Austria, which has a high proportion of alpine areas, the concept of permanent settlement area is of great importance. It refers to the inhabited and (economically) usable area available for settlement development, agricultural production, and infrastructure. It consists of a settlement area and a potentially inhabitable area, the latter including agricultural and green areas. The uninhabitable area (non-settlement area) is made up of forest areas, alpine grassland, wasteland, and water bodies. The CORINE Land Cover data as well as population and employment data (see ÖROK, 2015; Statistik Austria, 2023a) form the basis for the delineation of permanent settlement areas. This data was overlayed with information regarding the imperviousness degree. Soil sealing, as mentioned above, refers to areas that are continuously covered with a layer that is completely impermeable to water and air. The data from the Copernicus Land Monitoring Service represent the degree of sealing, in percent and aggregated to a 100 x 100 m grid (European Environment Agency, 2020). This method was chosen to add to the existing ÖROK Atlas (ÖROK, 2023d), which maps the proportion of soil sealing in administrative units. By making use of the granularity of the most recent Copernicus data, Figure 3.5 overlaps soil



### Soil Sealing in Austria (2018)

Origin of data: European Environment Agency, European Union's Copernicus Land Monitoring Service 2020, Statistik Austria - data.statistik.gv.at 2023 Administrative boundaries: Eurostat 2020 and 2021, BEV 2023 (CC BY 4.0); Digitales Gelände- und Landschaftsmodell: BEV 2023, Geoland.at 2015 Cartographic implementation: Franziska Sielker and Alexandra Pintilie, TU Wien 2024

Figure 3.5 Soil sealing in Austria, as observed in the year 2018, showing the extent of land covered by impervious surfaces (Sielker and Pintilie, 2024, based on European Environment Agency, 2020).

#### Soil Sealing in Austria - Federal Province Capitals (2018)





imperviousness with the potentially inhabitable land, allowing for a spatially detailed visualization. To get a sense of this granularity, Figure 3.6 provides zoom views for the capitals of the federal provinces.

Austria's continued excessive land consumption is in stark contrast to the objective of climate resilience, as open spaces continue to be sealed. Using its own methodology, which is also based on data on building land zoning, the ÖROK presented relevant figures in its report on land take and soil sealing in Austria. Accordingly, in 2022, an area of approximately 5,648 km<sup>2</sup> has been changed in a way that the land is no longer available for agricultural and/or forestry production or as natural habitat. This is described as land take in contrast to soil sealing. Land take marks roughly 17.3 % of the permanent settlement areas in Austria. More than half of this area (52 % or 2,964 km<sup>2</sup>) was already sealed in 2022 (ÖROK, 2022). If the land consumption within Austria continues at this pace, conflicts around land use will increase in the inhabitable settlement areas, as is shown in Figure 3.5 and Figure 3.6.

A key takeaway from Figure 3.5 is that in the potentially inhabitable areas of Austria, a diversity of uses has to be realized. In many cases, these are mutually exclusive land use demands, as for example agriculture and housing. Considering the diversity of land use demands in the context of population development, the pressure on urbanized areas in particular will continue to increase. Decisions on new land take, and in particular soil sealing, need to take into account the potential negative impacts in other areas. In the long-term, land take in general and sealed soil in particular has serious implications for natural ecosystems. Biodiversity loss and habitat fragmentation are some of the consequences.

Similarly, as mentioned above, the systems of governance and planning in Austria tend towards a strong executive power with substantial planning-relevant competences at the district level, followed by the provincial level and a relatively weak state level with its main competences in sectoral plans. With regard to soil sealing, local development strategies and zoning are key to prevention. However, there is no overall Austrian strategy that sets targets for local authorities. In addition to land use plans, there are a number of instruments that can support the goal of no-net-land-take. These include, for example, climate checks for potential building land, environmental impact assessments for new construction projects, funding schemes for area recycling and unsealing, and using the opportunity to define settlement boundaries for local development (see also ÖROK, 2023a).

A key challenge for the future is therefore to make better use of existing instruments and to develop incentives to reduce new designations of building land. In order for local authorities to achieve overarching goals such as improving resilience and adaptive capacity, greater multi-level governance coordination may be useful. Multi-level governance discourses, including additional training and education, as well as raising awareness of the role of each local authority in contributing to these overarching goals, will be key.

#### Spatial energy and infrastructure planning

Achieving climate neutrality in the built environment is closely linked to the transformation of energy infrastructure (Camarasa et al., 2022; European Commission: Directorate-General for Energy et al., 2022; Billerbeck et al., 2024; Fallahnejad et al., 2024). This applies to the grids used to transport energy (electricity, district heating and eventually gases) and distribute it to end-users. It also applies to renewable energy generation, some of which is on-site, directly connected to buildings, such as building mounted photovoltaics (PV), or in the form of renewable energy generation sites, such as wind farms or ground-mounted PV. Both components have a strong spatial component and are linked to the planning of settlements in urban and rural contexts. While the second component is more closely related to renewable energy generation (Sections 4.5.2, 4.5.3) and biomass (Sections 2.1.1, 2.2.1, 2.3.2), this section will focus on the first component, i.e., the spatial component of energy grid infrastructure in relation to settlement structure, energy densities and related planning activities.

A key question in the decarbonization of the built environment is which part of the building stock should be connected to which energy infrastructure in order to provide a full supply of renewable energy (also taking into account the potential of measures to reduce energy demand). While more or less all buildings in Austria with a relevant energy demand are connected to the electricity grid, this is not the case for district heating and gas grids. High infrastructure costs for supplying only a low number of connected end-users should be avoided. In this context, spatial planning is important to ensure high connection rates and thus a high economic utilization of infrastructure investments. For district heating grids, the challenge is to identify and implement district heating areas in line with decarbonization plans, connecting climate neutral, renewable heat supply resources with heat demand (Billerbeck et al., 2024; Fallahnejad et al., 2024). For the gas grid infrastructure, spatial energy planning will become even more important, considering that the simple substitution of fossil gas by renewable gases may not be fully possible or very expensive, thus requiring a gradual decommissioning of the gas grid (see, e.g., Zwickl-Bernhard and Auer, 2022).

Recognizing these needs, the revised energy efficiency directive (Directive (EU) 2023/1791) of the European Union foresees the enforcement of local heating and cooling plans at least in municipalities with a population greater than 45,000. These plans should include a strategy for enforcing renewable heating and cooling, waste heat, and district heating and cooling. This should be based on a thorough analysis of heating and cooling systems in the local building stock as well as an analysis of energy efficiency measures. Clearly, these provisions will also need to be implemented in Austria with the corresponding effects.

In some European countries, municipal and spatial energy planning have a long tradition, particularly in Denmark (see, e.g., Chittum and Østergaard, 2014; Büchele, 2019). In addition, there is a considerable tradition of developing tools for spatial energy planning and municipal strategic energy planning, as described, for example, in Mirakyan and De Guio (2013), Stöglehner et al. (2016), Büchele et al. (2019), and Mandel et al. (2023). In addition, several tools have been developed in recent years to support municipalities in particular in the task of municipal heat planning, such as the 'Hotmaps' tool (Hotmaps Project, 2023) or 'Thermos' (Thermos, 2021). However, one of the main bottlenecks for effective spatial energy planning is a solid database, first of all on the buildings, their energy performance and related energy consumption, the resulting energy density and their existing technologies, in particular heating systems. Activities to improve data availability in this respect in Austria in recent years can be summarized along two lines. The first is mainly based on open access data and applies different algorithms to merge these datasets in order to derive the required data on a high spatial resolution, e.g., at the hectare level. Examples include the 'Energiemosaik', the project 'Enerspired cities' (Schardinger et al., 2021) and the 'Austrian Heatmap'1. The second line of activities aims to provide a higher resolution, i.e., at the building level, and also builds on non-open data sources, such as the building registry 'AGWR' (Statistik Austria, 2024a), including information from building energy performance certificates or datasets from chimney sweeps and plumbers. Here, mainly the project 'Spatial Energy Planning'<sup>2</sup>, within the framework of the 'green energy lab' program, developed new methods and approaches for the three regions of Salzburg, Styria and Vienna.

Several municipalities, local and regional authorities have implemented approaches and partly regulatory policy instruments of spatial energy planning. For example, the city of Vienna established the concept of 'climate protection areas' in 2018, and since then, the energy planning department 'MA 20' has been gradually implementing this concept by defining different areas where the use of fossil fuels is restricted by law (see also Erker et al., 2021).

In part, the concept of spatial energy planning also needs to be redefined and placed in the context of full climate neutrality targets, i.e., restricting the use of fossil fuels in some areas makes sense in a policy regime where fossil fuels in general are not restricted. However, since an overall policy regime is needed in which fossil fuels are completely phased out, spatial energy planning needs to support this goal by defining the areas where this phase-out must take place first. Moreover, in such a policy regime, spatial energy planning is also needed to define areas where shared energy infrastructure facilitating the phase-out of fossil fuels should be preferred to individual (even renewable) solutions. This applies, e.g., to renewable district heating and cooling or public transportation infrastructure. Overall, there is a consensus in the scientific community that gases (be it fossil or renewable) should be used for high-temperature, high-exergy applications and not for low-temperature applications such as space and water heating (see, e.g., European Commission: Directorate-General for Energy et al., 2022; Rosenow, 2022).

For spatial energy planning, this implies that gas grids in residential areas should be decommissioned (see also Rodgarkia-Dara et al., 2023). The supply of gases to industrial processes is closely linked to other dimensions of spatial energy planning.

While spatial energy planning has historically relied heavily on one-time data collection efforts, there is now a strong trend toward more innovative data approaches that rely on cross-regional, dynamic, automated, digitalized, continuously updated, and flexible data collection approaches to support planning and policy processes at different spatial scales. There is a strong consensus in the literature on the relevance of spatial energy planning and related energy infrastructure planning in achieving climate neutrality (see, e.g., Büchele, 2019; Büchele et al., 2019; Horak et al., 2022; Dumke et al., 2024; Fallahnejad et al., 2024).

#### 3.2.3. Climate impacts and risks

#### Urban heat islands

Urban settlements are particularly vulnerable to heat island effects, which affect human health (Tong et al., 2021) (see also Cross-Chapter Box 2). The spatial variation of urban temperature is highly specific to urban form (Zekar et al., 2023), with a direct link between surface sealing and surface temperatures, especially at night, as the built environment (asphalt, buildings) stores heat during the day and releases it at night, resulting in tropical nights (not below 20°C) (Rippstein et al., 2023, fig. 3.7). This poses particular health risks, despite behavioral and physiological adaptations over time (Hagen and Weihs, 2023).

Ameliorating extreme heat in urban environments is critical to reducing heat-related mortality (Martilli et al., 2020). Similar to studies on land surface temperature, studies on air temperature stress the impact of vegetation cover and tree canopy presence, which results in a cooling effect and reduces the ambient temperature (Zumwald et al., 2021). Integrated mitigation and adaptation strategies, such as improved building insulation, are key to ameliorating urban heat island effects (see Section 3.2.3 Integrated mitigation and adaptation strategies).

Older and higher-income neighborhoods are typically associated with more vegetation cover and thus less pronounced urban heat island effects. Heat mitigation may need to focus particularly on lower-income communities, where historical underinvestment has resulted in less green space and more surface sealing (Section 3.2.1).

<sup>&</sup>lt;sup>1</sup> <u>austrian-heatmap.gv.at</u> (BMK, 2022a), building also on the methods developed by Müller et al. (2019)

<sup>&</sup>lt;sup>2</sup> waermeplanung.at (SEP, 2021)

#### Urban water impacts

Austria's urban water infrastructure is highly developed, with 95 % of the Austrian population connected to central drainage systems and 93 % to central water supplies. Both combined sewer systems - where wastewater and stormwater flow through a single pipe - and separate sewer systems - where wastewater and stormwater are carried in distinct pipes - are used. While combined sewer systems are efficient during normal conditions, they can cause overflows during heavy rainfall, leading to untreated or partially treated stormwater and wastewater discharges into nearby waterbodies, called combined sewer overflow (CSO) emissions. Using separate sewer systems reduces the risk of CSO but requires additional infrastructure. Over the last 20 years, decentralized stormwater treatment has been increasingly adopted to mitigate the challenges of urban runoff. As Austria's urban water management sector has a net estimated annual energy consumption of 282 GWh and GHG emissions of 278,000 tCO2eq (Zach, 2022) (limited evidence, medium agreement), energy recovery from wastewater is considered a critical strategy to make the sector carbon neutral or even a carbon dioxide (CO<sub>2</sub>) sink.

Relevant climate change impacts on urban water infrastructure are changing precipitation and temperature patterns, including increased occurrences of dry periods, excessive heat, extreme precipitation events, and tropical nights (Arnbjerg-Nielsen et al., 2013; Kourtis and Tsihrintzis, 2021). Detailed information on climate projections for Austrian settlements is provided in Chapter 1 (Section 1.2.2; Figures 1.4, 1.5, 1.7 and 1.8) and Section 2.2.2. While there are seasonal and regional variations, there is *medium to high* confidence that rainfall intensities for typical design events (return period 5-10 years) and extreme events (return >100 years) will grow (see Cross-Chapter Box 1). Rainfall intensities for short-duration events are expected to increase by 10 % per degree of temperature rise. Additionally, as mentioned in Section 3.2.2, soil sealing in urban areas is on the rise, resulting in more paved spaces.

Intensified precipitation and soil sealing together boost surface runoff and hence challenge urban drainage systems. For combined sewer systems, this results in increased runoff to wastewater treatment plants and increased CSO emissions, clearly increasing impacts for already slightly shorter event return periods (*medium evidence, high agreement*). As of 2024, the EU proposal for a revised 'Urban Wastewater Treatment Directive' (European Commission, 2022a) limits combined sewer overflow emissions to be below 2 % of annual wastewater runoff. Currently, several Austrian cities are expected to exceed this limit (Muschalla, 2023), and as climate change impacts are expected to worsen, this will lead to adaptation and investment requirements (*limited evidence, high agreement*).

For both combined and separate systems, an increase in rainfall intensities of extreme events also elevates the risk of pluvial floods. To enhance resilience to these climate impacts, necessary adaptation measures include either (1) expanding existing drainage infrastructure or (2) implementing strategies such as de-sealing surfaces, decentralized stormwater management to reduce runoff into drainage systems, and integrating blue-green infrastructure (e.g., vegetation and stormwater ponds) (Kleidorfer et al., 2018). The second approach seems economically and ecologically superior, however, detailed investment demand is unclear and needs to be assessed. De-coupling and de-sealing of surfaces would improve the water and energy balance by increasing infiltration, evapotranspiration and storage of water (Oral et al., 2020; Probst et al., 2022; Ferreira et al., 2024) (high confidence). Moreover, such adaptation strategies offer the opportunity to transform existing systems by integrating sustainable, ecological solutions and providing aesthetic, social, and economic benefits. However, compared to existing central systems, this can complexify planning and maintenance procedures (Fletcher et al., 2015).

A transition pathway towards sustainable drainage concepts requires mandatory decentralized stormwater management for new and reconstructed buildings and the implementation of adaptation measures into spatial planning, prohibiting sealing of surfaces and providing the necessary data sources for decision making (e.g., risk maps for pluvial flooding) (Mikovits et al., 2017). Decentralized stormwater management systems often require construction on private property, making it crucial to develop strategies and incentives to facilitate the disconnection of existing stormwater inflows from the drainage system. While many cities and settlements already require that new buildings refrain from connecting stormwater to existing drainage systems, there are generally no incentives to encourage retrofitting in older developments. Introducing split fees - where stormwater and wastewater discharges to the drainage system are billed separately - could serve as a potential financial motivator. Alternatively, providing financial support or subsidies to encourage disconnection of currently connected areas from the drainage system is another viable approach.

In addition to runoff reduction, stormwater harvesting and reuse are becoming increasingly important as precipitation patterns change and water resources are reduced due to longer dry periods between rainfall events and increased water demand for irrigation of nature-based-solutions (vegetation) for urban cooling (Funke and Kleidorfer, 2024) (see Section 3.2.3 Urban heat islands). Therefore, investments in long-term storage systems can not only reduce the demand for drinking water, but also reduce peak flows and provide water to urban vegetation to increase cooling during heat waves.

Overall, urban areas need to prepare for extreme events by establishing risk maps, early warning systems and emergency plans, and by collecting and providing the necessary data to develop such plans. Consequently, water management is part of integrated climate change mitigation and adaptation planning (Ürge-Vorsatz et al., 2018) (see below).

#### Integrated mitigation and adaptation strategies

Urban development and spatial planning can support both mitigation and adaptation. While sometimes in conflict, some strategies, such as better insulated buildings, can address both simultaneously (see, e.g., Section 3.2.3 Urban heat islands).

Compact cities are widely understood as a main strategy for designing low-carbon cities (see also Cross-Chapter Box 4 and Cross-Chapter Box 5) (Stöglehner et al., 2016; Stöglehner and Abart-Heriszt, 2022; Statistik Austria, 2024d). Short distances translate to less distance driven by car and/or higher modal shares of active modes of transportation. High density also means that public transit fulfills the economics of density preconditions to become more financially viable (Creutzig, 2014).

Urban infrastructures can reduce energy use and resulting GHG emissions, for example by reducing thermal loss in denser buildings (Borck and Brueckner, 2018). Avoiding urban sprawl, associated with several externalities (Dieleman and Wegener, 2004), can, in turn, be guided macro-economically by increasing fuel prices and marginal costs of motorized transport to obtain a spatially optimal equilibrium (Creutzig, 2014). Compact cities can lead to urban heat islands and sealed surfaces can prevent water storage and effective water management strategies (see Section 3.2.3 Urban heat islands, Urban water impacts).

Two key strategies can help moderate this trade-off. First, at the macro-scale, new urban developments could follow a star-shaped design, with public transit lines radiating out from the city center while preserving green space in between transit axes (Pierer and Creutzig, 2019). Second, particularly in already built environments, a key strategy is to focus on urban space readaptation. This includes changing inefficient use of street space, such as by unsealing and repurposing parking spaces, or thinking in terms of multi-purpose uses. This would discourage private automobility (the main driver of urban transport GHG emissions; see Section 3.4.2) and support the management of both water extremes and heat waves. Other options include vertical greening of buildings, green roofs, increasing tree cover along city streets, and improved long-term water storage.

Public health is another motivation for compact '15-minute cities' (Caprotti et al., 2024). Highly accessible walkable and cyclable urban design is not only a major mitigation option, it also provides more inclusive city services related to well-being (Lwasa et al., 2023). Solutions include planning cities around walkable sub-centers where multiple destinations such as shopping, work, leisure activities, and others, can be reached without driving (Newman and Kenworthy, 2006; Oswald et al., 2020).

#### 3.3. Buildings

This section assesses the building sector's contribution to climate change mitigation and adaptation, emphasizing technology, materials, policy, and legislation. Section 3.3.1 examines energy demand, renewable technologies, and emission reduction pathways in buildings. Section 3.3.2 addresses the decarbonization of construction materials, while Section 3.3.3 reviews legal frameworks, barriers, and opportunities for reducing emissions in the building sector. Chapter Box 3.2 further explores energy storage in buildings, highlighting key technological needs and solutions.

### 3.3.1. Statistics and scenarios of technologies and renewables in buildings

#### Current status

This section outlines the current baseline for the Austrian building sector, based on recent statistics, including data on building numbers and types, energy demand, energy carriers, and greenhouse gas (GHG) emissions. The sector is dominated by single- and two-apartment residential buildings, while nonresidential structures make up 11.6 % of the total building stock (Table 3.1). This section provides an overview of the key assumptions and outcomes of scenarios and measures aimed at reducing energy demand and GHG

Building type → ↓ Ownership	Residential building with 1 apartment	g Residential building with 2 apartments Residential building with 3 or more apartments		Other buildings	
Private persons	1,477,870	287,101	160,002	175,819	
Corporations under public law	11,006 1,868		28,086	42,541	
Non-profit building associations	27,785	1,575	60,242	2,268	
Other legal persons	14,853	2,208	25,400	56,153	
Total by type	1,531,514	292,752	273,730	276,781	
Total overall	2,374,777				

Table 3.1 Number of buildings in Austria in 2021 (Statistik Austria, 2024d).

emissions (refer to Section 3.3.3 for a legislative perspective on the topic).

Austria's housing stock totals 4.91 million apartments (Statistik Austria, 2021). Of these, 4.02 million are classified as primary residences, while the remaining 0.89 million consist of second homes, holiday properties, or vacant units for various reasons. This distribution is key to understanding residential trends and housing market dynamics in Austria.

Almost all subsidies for housing decarbonization in Austria are contingent upon the registration of a primary residence. In the multi-apartment sector, the legal framework plays a crucial role in shaping the conditions for decarbonization efforts. Housing law regulations vary significantly across different types of properties, including owner-occupied apartments, privately rented apartments, housing association rentals, and municipal rentals.

Of the 1.78 million main-residence apartments in detached houses with one or two units, 1.45 million are owner-occupied. In multi-apartment buildings, about a third of the 2.23 million apartments are privately rented, a quarter are housing association rentals, and a fifth are owner-occupied. The remaining apartments are municipal or have another legal status (Statistik Austria, 2023c).

The end-use energy demand presented in Table 3.2 indicates that, according to the statistics of 2022, fossil fuels account for 29 % of the total demand, primarily from gas and oil. Biomass accounts for about 19 %, while district heating has grown significantly to 15.5 %, with a relative increase of 175 % from 1990 to 2022. Electricity demand, which comprises 29.5 % of the total, includes space heating, domestic hot water (such as direct electric heating, heat pump compressors, hot water preparation, and heating system controls, including ventilation and heat recovery), as well as air conditioning, household electricity, and other uses. The 12 % reduction in end-use energy demand from 2021 to 2022 is largely attributed to the milder weather conditions in 2022 compared to 2021. From 2022 to 2023, this demand decreased by a further 6 %, resulting in a 36 % share of total end-use energy consumption, partly due to the increasing replacement of fossil fuel boilers with heat pumps (BMK, 2024a).

Type → ↓ Year	Oil	Coal	Gas	Biomass	Electricity*	District heating*	Ambient heat, etc.**	Total***
1990	93,451	27,578	46,093	60,457	73,412	21,798	2,239	326,143
2005	92,796	4,682	88,876	61,791	103,487	43,050	7,042	402,803
2021	51,143	478	83,024	90,944	118,456	68,098	24,914	437,062
2022	44,025	358	67,438	72,722	113,321	59,730	27,074	384,673
2021–2022	-14 %	-25 %	-19 %	-20 %	-4,3 %	-12 %	+8.7 %	-12 %
1990–2022	-53 %	-99 %	+46 %	+20 %	+54 %	+174 %	+1,109 %	+18 %
Share 2022	11.4 %	0.1 %	17.5 %	18.9 %	29.5 %	15.5 %	7.0 %	100 %

Table 3.2 End-use energy demand in the Austrian building sector, expressed in terajoules [TJ] (Umweltbundesamt, 2024b).

\* GHG emissions from electricity generation and district heating are attributed to the energy and industry sector.

\*\* Geothermal, ambient heat (for heat pumps) and solar thermal.

\*\*\* Including other fuels (combustible waste, peat).

Year $\rightarrow$ $\downarrow$ Main contributor	1990	2021	2022	Rel. change 2021–2022	Rel. change 1990–2022	Share of national GHG emissions 2022
Private households	10.605	7.326	6.154	-16 %	-42 %	8.4 %
thereof stationary	10.414	7.219	6.047	-16 %	-42 %	8.3 %
thereof mobile	0.191	0.108	0.107	-0.9 %	-44 %	0.1 %
Public and private services	2.313	1.537	1.224	-20 %	-47 %	1.7 %
Building sector	12.918	8.863	7.378	-17 %	-43 %	10 %

Table 3.3The GHG emissions from the operation of the Austrian building sector, broken down into private households and commercial buildings,measured in million tons  $CO_2eq$  [MtCO<sub>2</sub>eq] (Umweltbundesamt, 2024b).

Table 3.3 shows GHG emissions and their relative reductions in Austria's building sector between 1990 and 2022. GHG emissions from district heating and electricity generation are allocated to the energy and industry sector. However, when accounting for the GHG emissions linked to the electricity demand of buildings - 26.6 % of total electricity demand for private households, 16.3 % for public and private services as well as 82.5 % for district heating (Umweltbundesamt, 2024b) - the building sector's share of total emissions in 2022 increases from 10 to 13.5 % (high confi*dence*). The share of direct emissions decreased by 20 % to 5.886 MtCO<sub>2</sub>eq from 2022 to 2023, driven by the increased adoption of heat pumps, which shifts fossil fuel demand to electricity need, accounted for in the energy sector. As heat pumps are far more efficient than fossil fuel boilers, this transition reduces the overall emissions. However, the figures for 2023 cannot be determined as the corresponding data is not available yet.

# Greenhouse gas reduction and climate change mitigation scenarios

Relevant climate change impacts for buildings are changes in maximum, minimum and monthly ambient temperature, which reduces space heating demand in winter but increases space cooling demand in summer. All available mitigation scenarios take into account: The projected number of buildings (considering increases due to population growth), the projected thermal state of buildings (new, renovated, divided by building type and age), the thermal renovation rate and thermal quality (the space heating demand after renovation), the proposed share of renewable energy carriers for new and exchanged old heating, ventilation and air-conditioning (HVAC) systems, and the HVAC exchange rate. Using these values, the scenarios calculate a combination of enhanced energy efficiency and a transition to renewable energy sources. Regarding future population development, quantitative international migration scenarios that take into account different climate change scenarios (RCPs) are not available at the moment. Therefore, the predictions of Statistik Austria (2024c), which are based on the current situation, are used for Austria. A more detailed discussion can be found in Section 6.8.2.

While the scenarios outlined in the literature vary in scope, they share a high degree of similarity in their assumptions and outcomes. Some studies focus exclusively on technical aspects and address annual energy balances for buildings without considering seasonal storage or broader geographic areas (Pfeifer et al., 2016; Dobler et al., 2018). Others adopt a more comprehensive approach, encompassing all sectors (Ebenbichler et al., 2018; Umweltbundesamt, 2019; Kranzl et al., 2020; Steininger et al., 2021). Additionally, some studies incorporate pricing mechanisms and macroeconomic factors, extending their analyses to cover Austria in its entirety as well as the dynamics of European energy markets. Brandes et al. (2021) and Luderer et al. (2021) perform scenario calculations for Germany with hourly resolution, incorporating energy storage solutions into their analyses. Ebenbichler et al. (2024) developed a Tyrol-focused scenario on an hourly basis, incorporating short- and long-term storage solutions while considering 2050 projections for Austria, Germany, Italy, France, and the broader European context (see also Chapter Box 3.2 for more information on energy storage in buildings). The official Austrian scenarios WEM (With Existing Measures), WAM (With Advanced Measures) and Transition (reaching nearly fossil free energy system by 2040), are presented in a report by the Environment Agency Austria (Umweltbundesamt, 2023a). In the WAM scenario, fossil fuels are not used directly for space heating or cooling; however, fossil fuels are still included in electricity production. At the time of writing, the input data for the NetZero2040 scenarios (such as renovation rate, thermal quality, exchange rate to renewables) have not yet been published (NetZero2040, 2024).

Projections for Austrian settlements in the scenarios show population growth and a similar renovation rate as today, but with deeper renovations resulting in lower energy demand per square meter of renovated space. Overall, the final energy demand for heating is expected to decrease by 10 to 50 % by 2040/2050, despite population growth and increasing heated area, due to an assumed constant capita per floor area. This reduction is largely driven by thermal renovation and partly driven by climate change. The demand for cooling energy is anticipated to rise significantly, depending on the climate scenario, but remains low in absolute terms (see Figure 3.7 and Figure 4.12). Building design must account for rising temperatures to minimize cooling demand (Frischknecht et al., 2020). HVAC systems will shift from fossil fuels to renewable sources, primarily heat pumps, but also district heating (fossil-free), biomass and, to a lesser extent, direct electric heating. In all back-casting scenarios, building operations can be achieved without fossil fuels. However, achieving this sooner requires a higher renovation rate and faster replacement of HVAC systems. Yet, all forecasting scenarios still include some fossil fuel use by 2050, while all of the backcasting scenarios, by definition, phase out fossil fuels.

Biomass use for heating is constrained by its sustainable potential, primarily sourced from residues in the saw, paper, and pulp industries, with some contributions from wood. The biomass must be allocated across different applications, including combined heat and power (CHP) plants, direct heat use in district heating systems, and decentralized biomass stoves. Additionally, wood is used as a building material, acting as a temporary  $CO_2$  sink. However, the growing demand for biomass in the industry, particularly as a raw material for chemical products, limits its availability for energy use. As a result, the share of biomass in the overall energy system will remain small compared to electricity generated from renewable sources. Some biomass may also come from sawmill waste or imported biomass (see Sections 2.1.1, 2.3.2).

Required adaptation measures for buildings will be the renovation to much lower space heating energy demand values than current building codes require. Additionally, the adaptation to higher ambient temperatures decreases the winter space heating demand (at times when energy costs are high due to reduced renewable energy production) and increases the need for space cooling energy (at times when sufficient renewable energy from PV and hydropower is available) (see Sections 4.5.2, 4.5.3). As a result, further tightening building code requirements remains a topic of debate.

Figure 3.7 shows that across all climate scenarios analyzed (from +1.3°C for RCP 2.6 in 2050 to +4.3°C for RCP 8.5 in 2080 – i.e., the projected future local mean temperature increases, in this case in Innsbruck, compared to today's mean), the reduction in heating demand exceeds the increase in cooling demand for single-family houses (OIB, 2019, 2023; Streicher et al., 2024b). Cooling demand only becomes evident when temperatures rise more than 2°C. Similar findings are, e.g., reported by Schöniger et al. (2023), Suna et al. (2024), and Sonnleithner et al. (2023). Ziemele et





al. (2023) found comparable results for Riga, while Wilbanks et al. (2008) reviewed studies in the U.S., and Elnagar et al. (2023) performed an analysis for cases in Belgium, factoring in renovation rates.

In summary, the literature consistently shows a decrease in winter heating demand during periods of limited renewable energy production, coupled with an increase in summer cooling demand during periods of high renewable energy availability. Therefore, climate change does not pose an increased risk to the energy system in terms of heating and cooling demand. However, the design of flexible HVAC systems that can provide both heating and cooling – such as heat pumps integrated into ceiling- or air-based systems – is a cost-effective adaptation strategy to address climate variability (*high confidence*).

## Improve, Shift and Avoid measures in the building sector

The thermal renovation rates and renovation depth, i.e., space heating demand after renovation (improve) and replacement of heating system rates (shift), are estimated to be between 1.3 and 3 % per year for renovation and 3 and 5 % per year for replacement of HVAC equipment in all scenarios mentioned in the previous section. The lower end of this range reflects historical trends, where most renovations or replacements occur only when components fail, reach the end of their lifecycle, or change ownership. The classification of improve and avoid is done according to the Cross-Chapter Box 4.

Based on these assumptions, the costs for high level renovation or replacement are calculated as the difference between the standard renovation/replacement costs and those for achieving high energy efficiency and renewable energy standards (i.e., additional costs). Using the historic renovation/replacement rate, a reduction of the final energy demand of buildings (including household electricity, domestic hot water, space heating and cooling) is calculated to be approximately 14 % (WEM/WAM) to 30 % (Ebenbichler et al., 2018) by 2050, for different renovation depth assumptions, i.e., the space heating energy consumption assumed in the renovation scenarios after the thermal renovation of a building is varied between current maximum allowable space heating demand values of OIB 6 (thematic guideline number 6 by the 'Österreichisches Institut für Bautechnik') (OIB, 2019) down to half of these values.

In all backcasting scenarios (Tyrol 2050, Transition Scenario, Ebenbichler et al., 2018) 100 % renewable energy supply is achieved. However, in some forward scenarios (WEM/ WAM), the electricity demand for buildings is not fully decarbonized (see the previous Section). In some scenarios, the renovation/replacement rate is increased to achieve reduced energy consumption sooner, e.g., by 2040. In such cases, still functioning building components with residual value must be replaced, which increases the costs. With a thermal renovation rate of 3 % and a high renovation depth, the total final energy demand of buildings could be reduced by about 26 % by 2050 (Steininger et al., 2024) (*high confidence*).

Additionally, the production of essential renovation materials (insulation, windows, shading) and renewable-based HVAC systems (in all cited scenarios primarily heat pumps, but also district heating and biomass) will need to slightly increase. This has to be accompanied by the training and education of a larger workforce, including planners, plumbers, and craftsmen. Streicher et al. (2024a) estimate that the Tyrol 2050 scenario (Ebenbichler et al., 2018) would require an additional 580 and 660 installers and 44 and 36 planners for heat pumps and photovoltaic systems, respectively. These numbers can be multiplied by a factor of 12 to approximate the total Austrian demand (in relation to the population of Tyrol). Furthermore, legislation banning new oil and gas boilers has already been enacted (BGBl. I Nr. 6/2020; BGBl. I Nr. 8/2024). At present, a substantial subsidy for phasing out oil and gas is stimulating market growth (KPC, 2022; BMK, 2024c), and as economies of scale take effect and more products become available, both investment and installation costs are expected to decrease.

Steininger et al. (2024) recommend a combination of push and pull measures to drive policy action (for legislation, see Section 3.3.3). These include stricter building codes, prohibitions on new fossil-based HVAC systems, and incentives to accelerate the transition to fossil-free HVAC solutions. Subsidies should target technologies that are not yet widely available or lack sufficient expertise in production, planning, and installation. Current building codes (OIB, 2019) optimize investment and operating costs, resulting in designs close to passive house standards. However, buildings with higher energy demand (and cost) are allowed if the efficiency of the HVAC system is 30 % higher than a reference system or a rooftop PV is integrated (dual way), even if the efficiency of the building itself is not directly linked to the HVAC system or the renewable energy produced by the building. The WEM and WAM scenarios, as well as the Tyrol 2050 scenario (Ebenbichler et al., 2018), slightly reduce the allowable OIB 6 useful energy demand for space heating; Ebenbichler et al. (2018) even halve the current OIB 6 values. Listed buildings

are not included in OIB 6 as they represent a small proportion of the total building stock. Subsidies for these measures should be phased out once market demand is met or prices fall to sustainable levels. Public acceptance can be improved ensuring sufficient skilled labor and affordable costs to normalize these technologies (similar to the adoption of triple-pane windows in Austria, Germany and Switzerland) (*robust evidence, medium agreement*).

The current standards for snow loads (Austrian Standards, 2022; last adaptation in 2022, ÖNORM B1991-1-3:2022-05) or the publicly available data on flood zones, landslides, etc. in a report from the Austrian Government (BML, 2023) show that the requirements are continuously adapted and used to improve related building standards. Existing buildings may need to be adapted to the changing requirements (*medium evidence, high agreement*).

Avoid (sufficiency) measures in the building sector include, e.g., reducing the square meters of living space per person, reducing heating/increasing cooling set-temperatures, and reducing the number of technical appliances (Zimmermann, 2018). The current average net-living-area in Austria is about 47 m<sup>2</sup>/person. This has increased from about 32 m²/person in the year 1990 (Statistik Austria, 2004, 2022c). The room temperature for the calculation of energy certificates has even been increased from 20 to 22°C in recent years to better reflect reality (e.g., Austrian Standards, 2019). One avoiding strategy is the taxation of long-term vacant apartments, a measure currently implemented in the federal provinces of Tyrol, Styria, and Salzburg, and planned for adoption throughout Austria (Koller et al., 2023; Parlamentsdirektion, 2024b) (robust evidence, medium agreement).

#### Chapter Box 3.2. Energy storage in buildings – technologies and needs

**Introduction:** The growing reliance on variable renewable energy sources has heightened the need for energy storage and demand adjustments. Buildings, due to their size and thermal inertia, represent significant opportunities for flexibility. This chapter box explores (1) current energy storage technologies and demand response in buildings, (2) load-shifting and storage potential in Austria's building stock, and (3) barriers, drivers, and policy requirements. Related e-mobility considerations are addressed in Chapter Box 3.3.

**State-of-the-art energy storage and demand response in buildings:** Thermal energy storage, the most common method in buildings, is used for domestic hot water and space heating. Storage types include (Borri et al., 2021):

- Sensible heat storage: Uses water tanks or the thermal mass of buildings (e.g., concrete floors). Effective for dayto-night storage up to a maximum of 2-3 days, it is widely used for its simplicity and cost-effectiveness, allowing for short term demand shifting in systems with heat pumps or electric boilers (Miara et al., 2014; Bechtel et al., 2020; Fitzpatrick et al., 2020). However, it involves increased heating demands due to thermal losses in water storage systems or slight overheating in winter (and undercooling in summer) within buildings (Pasqui et al., 2023). Despite this, local renewables and off-peak electricity can offset costs and GHG emissions (Mascherbauer et al., 2022b; Schöniger et al., 2024).
- Latent heat storage: Relies on phase-change materials (PCMs) to maintain specific temperatures but faces challenges in efficiency and implementation. Streicher et al. (2008) could find no advantages of phase change materials as energy storage in buildings compared to water or thermal masses.
- Chemical and sorption-based storage: Offers high-capacity, long-term storage with minimal losses, but remains costly and complex (Ding and Riffat, 2013; Bao and Ma, 2022). As a result, this type is virtually absent from the market.

Battery systems, increasingly paired with photovoltaic (PV) installations, align energy generation with household demand. However, their high costs and limited seasonal storage capacity primarily confine their usage to short-term applications, such as daily or weekly energy balancing (Ochs et al., 2021; Li et al., 2023). Advanced hydrogen systems with fuel cells, while flexible, are even less cost-effective. Flexibility and energy storage in Austrian buildings: Buildings equipped with heat pumps and thermal storage systems can stabilize electricity and district heating grids on a daily basis (Schöniger and Morawetz, 2022; Suna et al., 2022). Studies such as by Heidenthaler et al. (2023) and Wolisz et al. (2016) highlight thermal activation and preheating/ cooling as key mechanisms for demand shifting, even over several days. Research by Tosatto et al. (2023) and Magni and Ochs (2021) projects significant load-shifting potential in Austrian regions, while Turner et al. (2015) confirms precooling benefits in lightweight structures. Overall, thermal mass insulated from indoor air proves critical for demand flexibility (Reynders et al., 2013; Masy et al., 2015; Le Dréau and Heiselberg, 2016; Luo et al., 2020).

Nationally, dynamic pricing shows potential to optimize electrified heating systems, but incentives remain insufficient for consumers (Mascherbauer et al., 2022a; Schöniger et al., 2024). Thermal inertia studies estimate 50 % of peak heating loads can be shifted to off-peak periods in post-1980 buildings (Weiß et al., 2019). By 2040, demand-response-optimized heat pump operation could achieve cost reductions of 50-75 % (Amann et al., 2023b).

**Barriers, drivers and policy needs:** Variable electricity tariffs, such as real-time pricing, could incentivize demand-side flexibility (Fitzpatrick et al., 2020). Studies suggest regulations are needed to balance retailer profits and consumer welfare (Guo and Weeks, 2022). While savings from smart systems currently only marginally outweigh their costs, future high renewable energy integration could increase their value.

For flexibility adoption to grow, policies must standardize and subsidize smart systems. By 2050, widespread heat pump use and supportive policies could significantly impact electricity grids and renewable energy integration. Challenges and opportunities in decarbonizing energy are further discussed in Sections 4.5.2 and 4.5.3.

#### 3.3.2. Construction products

Population growth, increasing per capita net floor area, sustainability concerns and changing lifestyles are contributing to a significant increase in both new construction and building renovations (IPCC, 2021b). As a result, the production of construction products used for new buildings and renovations represents 11 % of total global energyand process-related GHG emissions, with steel and cement production contributing more than half of these emissions (Röck et al., 2020). Consequently, there is growing interest in utilizing materials with low embodied environmental impacts. In addition, the reuse and recycling of materials is becoming increasingly important to reduce the demand for new resources. In Austria, 2.4 million buildings - both residential and non-residential (see Table 3.1) - are responsible for a significant portion of GHG emissions. While most of these emissions come from the operational phase of buildings, an increasing share is associated with the production of construction materials, which is typically attributed to the industrial sector or associated with electricity use and therefore falls under the emissions of the energy sector. In order to meet its climate goals, Austria aims to significantly reduce these building emissions by 2040 (see Section 3.3.1), which can only be achieved if both operational and embodied emissions reduced simultaneously.

More than half of the emissions of new buildings can be related to the embodied emissions of materials when considering the entire lifecycle (IEA, 2019; Frischknecht et al., 2020; Röck et al., 2020). In a study by Truger et al. (2022) for Austria, total GHG emissions associated with the building stock increase by a factor of 3 to 4 when the system boundaries are extended to include the entire lifecycle of buildings (including the embodied emissions of buildings), ranging from 7 MtCO<sub>2</sub>eq/year (Table 3.3) of direct operational emissions (i.e., 10 % of national emissions) to 22-31 MtCO<sub>2</sub>eq/year for Austria. In this sense, the most promising decarbonization measures for materials listed in the decarbonization roadmap currently developed by the EU (Le Den et al., 2023) are related to the embodied impacts of the products and materials used in buildings. They can be grouped into the Avoid-Shift-Improve (ASI) framework as follows (see also Cross-Chapter Box 4):

- Avoid: Focus on minimizing resource use by repurposing existing products and optimizing material efficiency. Examples include reusing building components, integrating void formers into concrete slabs, and optimizing structural designs to reduce material demand.
- Shift: Transition to alternative materials that are circular, low-carbon, or bio-based, thereby reducing environmental impact and supporting sustainable practices.

• Improve: Enhance the efficiency and sustainability of current production methods. This involves increasing the share of renewable energy used in manufacturing processes and integrating technologies like carbon capture to reduce emissions.

The recently released UN report 'Building Materials and the Climate: Constructing a New Future' (United Nations Environment Programme and Yale Center for Ecosystems + Architecture, 2023) provides a comprehensive overview of measures to decarbonize construction materials. These efforts are further bolstered by evolving EU legislation, particularly within the framework of the EU Green Deal, which aims to reduce GHG emissions across member states (COM/2019/640 final). Key directives under this framework include initiatives promoting circular economy procurement and the ongoing revision of the Construction Product Regulation (CPR) (European Commission, 2020a). These focus on enhancing environmental sustainability and establishing robust markets for the reuse and recycling of construction materials. In addition to the regulations on the European level, i.e., the Energy Performance of Buildings Directive (Directive (EU) 2023/1791) and the CPR, Austria is preparing its own directives (OIB, 2023), which align with and build on EU measures, as detailed in Section 3.3.3. In addition, more emphasis needs to be placed on equipping building stakeholders with the knowledge and skills needed to effectively implement these decarbonization measures.

Scenarios for the decarbonization of building products that align with the Austrian context include:

- Minimizing the use of new building products by leveraging existing structures. Refurbishing the existing building stock not only reduces the need for new materials but is also essential to reducing carbon emissions from older buildings. This approach is highlighted in the updated Energy Performance of Buildings Directive (EPBD, European Commission, 2024b).
- Utilizing construction products with low embodied emissions (Frischknecht et al., 2019; Röck et al., 2020).
- Improving embodied emissions in material production. Advances in production processes and improved energy mixes can lower emissions (Potrč Obrecht et al., 2021). Alaux et al. (2024) demonstrated how future material production trends may influence the sector's overall carbon footprint.
- Adopting circular economy principles, which can significantly reduce embodied emissions (Ghisellini et al., 2018;

Malabi Eberhardt et al., 2020; Mirzaie et al., 2020) – see

also Cross-Chapter Box 5. However, data on the potential reuse of materials from existing stocks is sparse. In Vienna, researchers are investigating the local potential (Lederer et al., 2020; Lederer and Blasenbauer, 2024), while the project 'Kraisbau', funded by the Austrian Research Promotion Agency (FFG), aims to provide further estimates (Kraisbau, 2022). In this context, more emphasis should be placed on building design to tailor new buildings for circular (re)use (Kanters, 2020; Akhimien et al., 2021).

- Reducing material use through efficient construction techniques: For example, comparing prefabricated wood-frame houses with traditional solid construction methods, such as those using resource-intensive materials like reinforced concrete or masonry, and incorporating void formers to reduce the volume of concrete or other materials required, demonstrates how resource efficiency and reduced embodied carbon can be achieved in construction (Mañes-Navarrete et al., 2024).
- Incorporating carbon storage in materials: The EU's provisional agreement on a certification framework for carbon removal promotes the temporary storage of carbon in durable materials, such as wood-based construction products, that are intended to remain in the building for at least 35 years. However, when a comprehensive, system-wide boundary is applied to the analysis, it becomes clear that the promotion of wood-use in buildings is only advantageous over reduced harvesting under certain conditions (high confidence). Maierhofer et al. (2024) for Austria find that key criteria such as a high carbon intensity of the energy system and the efficient and sustainable use of wood building materials are critical for realizing these benefits. Similar conclusions are reached by Fehrenbach et al. (2022) and Soimakallio et al. (2022), which are in line with assessments using discounting principles (Peng et al., 2023) (see also Section 2.3.2).

Alaux et al. (2024) quantified the potential reductions in GHG emissions for various materials, taking into account decarbonization strategies such as better energy mixes, circular practices, and production advances. While a 10 % reduction seems possible for most materials, wood could even reach 35 % by 2050, but only if the extended scope is not considered, as shown by Maierhofer et al. (2024). Achieving climate goals will ultimately require decarbonizing the entire construction lifecycle, including materials (European Commission, 2020b; Toth et al., 2022) (*robust evidence, medium agreement*).

#### 3.3.3. Legal reforms – barriers and perspectives

Austria is known for its high building standards, which excel in thermal quality, durability, ecological considerations, affordability, and social aspects of housing. There is a strong political commitment to meet international GHG reduction targets, aiming for net zero emissions by 2040 – ten years ahead of the 2050 global benchmark. This ambition is reflected at the federal level in the government program of the ÖVP-Green coalition (2020–2025) (BKA, 2020) and in several federal provinces, such as Vienna. Despite this commitment, experts are skeptical about the feasibility of achieving these goals with the measures currently in place (Habert et al., 2020; Steininger et al., 2024) (*medium confidence*).

#### Impact of EU legislation

Austria, a member of the EU since 1995, implements EU legislation in areas of EU competence, such as energy policy. Although housing is not directly under EU competence, related regulations have a significant impact on the sector, particularly with regard to climate targets. Key EU legislation includes the Energy Performance of Buildings Directive (first introduced in 2002, last revised in 2024) (European Commission, 2024b), as well as supporting measures such as the Energy Efficiency Directive (Directive (EU) 2023/1791), the revised Renewable Energy Directive (Directive (EU) 2023/2413), and the Alternative Fuels Infrastructure Regulation (European Commission, 2024a). The following key EU legislative frameworks are shaping Austria's energy and buildings policy:

- The European Climate Law 2021 (Regulation (EU) 2021/1119) mandates climate neutrality by 2050, a 55 % reduction in GHG emissions by 2030 compared to 1990 levels ('Fit for 55'), and outlines implementation mechanisms. Several of the twelve proposed measures target the construction, housing, and real estate sectors.
- The Energy Performance of Buildings Directive (EPBD) (draft revision, 2021): As part of the Fit for 55 package, this directive proposes a mandatory zero-emission standard for all new buildings starting in 2030 (COM/2021/802 final). It also aims to tackle inefficiency in existing buildings by progressively phasing out the 15 % of properties with the highest energy consumption through rental and sale bans. To boost renovation rates, the directive includes financial support measures to cushion the impact

on low-income households, facilitating compliance while ensuring social equity.

- Inclusion of buildings in the EU Emissions Trading System as of 2026: This expansion, combined with the already existing CO<sub>2</sub> pricing mechanism, will increase the cost of heating with fossil fuels like oil and gas.
- The tightening of the Renewable Energy Directive (Directive (EU) 2018/2001) puts further pressure on national legislation to phase out oil and gas, not only with subsidies but also using regulatory measures.
- The Energy Efficiency Directive (Directive (EU) 2018/ 2002) sets a renovation target of 3 % for public buildings.
- The Effort-Sharing Regulation (Regulation (EU) 2018/ 842) tightens emission reduction targets, imposing stricter national obligations.
- The new Environmental, Social, and Governance (ESG) criteria (COM/2023/314 final) and the EU Taxonomy Regulation (Regulation (EU) 2020/852) are likely to be a game-changer for commercial real estate, as they will make it much more difficult to finance properties with poor thermal efficiency, leading to a reduction in the value of such buildings.

In response to the energy market disruptions caused by Russia's invasion of Ukraine, the European Commission introduced the REPowerEU Plan (European Commission, 2022b). This strategy focuses on diversifying energy sources, accelerating clean energy adoption, and promoting energy savings. Integrated into the EU Recovery and Resilience Facility (RRF), the plan enforces even stricter regulations under the already demanding frameworks. These developments underscore the increasing regulatory pressure on Austria's housing and energy sectors, with significant implications for achieving climate resilience (*high confidence*).

#### Existing legal regulations and measures

Austria has long pursued legal reforms in its building sector to advance climate protection, heavily influenced by EU legislation. Among the key elements of the regulatory framework is the Condominium Act ('Wohnungseigentumsgesetz') (BGBl. I Nr. 70/2002), which introduced reforms such as revised quorum rules to facilitate decision-making within condominium owners' associations. These changes make it easier to implement building decarbonization measures. In addition, mandatory minimum contributions to reserve funds ensure that more financial resources are available, allowing investments to be made without relying solely on external financing.

The Limited-Profit Housing Act ('Wohnungsgemeinnützigkeitsgesetz') (BGBl. Nr. 139/1979), provides the most supportive framework for decarbonization. Limited-Profit Housing Associations (LPHAs, 'Gemeinnützige Bauvereinigungen') benefit from strong financing mechanisms such as the 'maintenance and improvement contribution', which can be as high as EUR<sub>2023</sub> 2.33 per square meter per month. LPHAs can also use energy savings contracting and enforce rent increases after retrofits, making them particularly effective for large-scale energy efficiency projects.

On the other hand, the Tenancy Act ('Mietrechtsgesetz') (BGBl. Nr. 520/1981), has proved resistant to reform due to enduring ideological divisions among political parties. This is particularly problematic in the segment of older rent-protected buildings built before 1945, where energy efficiency has no impact on allowable rents. As a result, it remains difficult to recover retrofit costs through rent adjustments.

Construction law, administered by Austria's federal provinces, has significantly improved energy performance standards primarily in response to EU directives such as the EPBD (European Commission, 2024b). In order to harmonize regional building laws, the OIB has issued thematic guidelines, including guidelines on energy saving and thermal insulation ('Richtlinie 6', OIB, 2019). A new thematic guideline ('Richtlinie 7') on the sustainable use of natural resources is about to be adopted and implemented (OIB, 2023). Other relevant regulations include the Building Energy Performance Certificate Act ('Energieausweis-Vorlage-Gesetz') (BGBl. I Nr. 27/2012) and the Heating and Cooling Costs Billing Act ('Heiz- und Kältekostenabrechnungsgesetz') (BGBl. Nr. 827/1992).

In terms of financial incentives, both the federal and provincial governments offer extensive subsidies for building renovation and heating system replacement. The federal provinces provide around EUR<sub>2023</sub> 500 million anually (Amann, 2019; Amann et al., 2023a). Federal initiatives include the Refurbishment Initiative ('Sanierungsoffensive'), the Out of Oil and Gas Bonus ('Raus aus Öl und Gas Bonus'), and Clean Heating for All Initiative ('Sauber Heizen für Alle'), which can cover up to 100 % of the costs of replacing heating systems for households at risk of poverty. Subsidies totaling EUR<sub>2023</sub> 2.66 billion will be available between 2024 and 2027 (Amann et al., 2023a). In early 2024, the federal government has announced additional support from the Housing Package ('Wohnbaupaket') (Parlamentsdirektion, 2024a). Provincial housing subsidies are based on regional laws and decrees, and federal subsidies are administered under the Law on Domestic Environmental Subsidies ('Umweltförderungsgesetz') (BGBl. I Nr. 152/2023). Recent tax incentives, although less prominent, include measures to benefit homeowners through income tax reductions and businesses through corporate tax adjustments (*medium evidence, high agreement*).

#### Pending legal reforms

One of the most significant policy instruments in the Austrian political system is the Fiscal Equalization Act, which regulates the distribution of funds between the federal state, provinces, and municipalities, usually for a five-year period (see also Section 6.4.1). The latest agreement, reached at the end of 2023, introduced several changes. While housing subsidies have always played an important role in previous financial equalization agreements, almost all authorities have now been transferred to the federal provinces. As a result, the current financial equalization scheme contains few regulations on housing. Notably, a 'future fund' has been established, allocating approximately EUR<sub>2023</sub> 300 million per year for housing-related measures. These funds are available if certain renovation and land consumption targets are met, with an equivalent amount earmarked for increasing the share of renewable energy.

The Renewable Heat Act (BGBl. I Nr. 8/2024) prohibits the installation of oil and gas boilers in new buildings. However, it does not (yet) regulate the replacement of existing fossil fuel boilers. In addition to the Renewable Heat Act, housing law reforms are essential. A particular challenge is the Tenancy Act (or MRG, as introduced above), which has so far been largely resistant to reform. Essential reforms include the mandatory cooperation of tenants in the decarbonization of heating systems and the establishment of a fair distribution of the associated costs between landlords and tenants. The Condominium Act (WEG) also needs further reform to be aligned with decarbonization goals.

Land acquisition regulations ('Grundverkehrsrecht') could be used to reduce building-land hoarding (see Section 3.2). Housing subsidies, both for new construction and renovation, could be more explicitly linked to climate protection goals. The federal provinces have the primary responsibility for ensuring that no one is left behind on the path to climate neutrality. To stimulate climate-friendly investments in an economically sustainable way, policymakers

could implement tiered incentive structures that prioritize long-term energy efficiency and carbon neutrality, as those outlined in the 'COVID-19 investment premium', while ensuring broad accessibility and minimizing deadweight effects (Weyerstraß, 2021).

# Preconditions for achieving building decarbonization by 2040

In addition to a massive reduction in the energy demand of the buildings and a switch to renewable energy sources (see Section 3.3.1), district heating systems need to be decarbonized. Simulations show that if the renovation rate is increased from the current 1.5 to 2.8 % by 2030, the entire thermally inefficient building stock could be completely renovated by 2040 (Amann et al., 2021). At the same time, nearly 700,000 oil-heated apartments will need to be converted by 2035, followed by more than one million gas-heated apartments by 2040 (in terms of registered main residences). Success in this area will depend on a significant increase in the number of buildings heated by district heating, biomass, or heat pumps (Amann, 2023; Streicher et al., 2024a).

While decarbonizing the building sector by 2040 is technically feasible, it depends on creating the right framework conditions. Key challenges include a shortage of skilled workers and the need for increased innovation. In addition, Austria's federal system, characterized by strong provincial autonomy and inadequate cooperation mechanisms, poses a significant obstacle. The current financial equalization framework is insufficient to address the scale of necessary reforms at both the federal and provincial levels. Therefore, new models of cooperation and coordination will be essential (Amann, 2023) (see also Section 6.4.1).

#### 3.4. Transport and infrastructure

This section explores the role of transport and infrastructure in reducing emissions and adapting to climate change. Section 3.4.1 introduces the transport sector's emission trends and provides an overview of this subchapter. Section 3.4.2 focuses on decarbonization in the passenger transport sector, while Section 3.4.3 focuses on decarbonization in the freight transport sector. Section 3.4.4 discusses the GHG emissions embodied in road and rail infrastructure, as well as climate risks for transport infrastructure and related adaptation needs. Chapter Box 3.3 discusses the flexibility potential of electro-mobility in Austria.

#### 3.4.1. Introduction

The transport sector is Austria's second largest emitter of GHGs after industry. In 2023, direct GHG emissions associated with the movement of people and goods within Austria, together with fuel sold domestically, totaled 19.8 million tCO<sub>2</sub>eq, excluding international air traffic (Umweltbundesamt, 2024c). This corresponds to 29 % of Austria's total GHG emissions in 2023 (Umweltbundesamt, 2024c). Notably, transport remains the only sector where GHG emissions have not decreased since the base year 1990 (see Figure 3.8). After a continuous increase in emissions from 2012 onwards, significant reductions were observed only during and after the COVID-19 pandemic, as shown in Figure 3.8. These reductions were primarily driven by external factors: In 2020, GHG emissions fell by 13.5 % in a single year due to the pandemic and related lockdown restrictions that reduced transportation activity (Umweltbundesamt, 2024c). The low emission levels in 2022, similar to those in 2020, were largely influenced by Russia's invasion of Ukraine and the resulting European energy crisis, which drove up energy costs. The further decrease in emissions in 2023 was mainly attributed to lower economic output, which also reduced fuel exports from Austria (Heinfellner et al., 2024b).

In 2022 (the latest year available for Europe-wide comparison), renewable energy sources accounted for 10 % of energy consumption in the Austrian transport sector (European Commission: Eurostat, 2024). While this was slightly above the EU average, it remained significantly below the EU's 2030 target of 29 % (*medium evidence, high agreement*).

In 2023, roughly 34 % of total transport GHG emissions were generated by roadside freight transport, 65 % by roadside passenger transport and the remaining share by rail, shipping and domestic aviation. This highlights that transport emissions are overwhelmingly dominated by road transport. Within this category, diesel-powered passenger cars accounted for the largest share (42 %), followed by heavy-duty vehicles (26 %) and gasoline-powered passenger cars (23 %) (Heinfellner et al., 2024b).

International aviation is excluded from national emission inventories due to the lack of reporting requirements. While national domestic traffic accounted for only 0.2 % of total transport sector GHG emissions in 2023, the inclusion of international flights (originating or landing in Austria) would significantly increase this figure (Heinfellner et al., 2024b). Aviation emissions peaked in 2019 and, despite the dip during COVID-19, showed a strong upward trend af-



GHG emissions of Austria's transport sector 1990-2023 (incl. fuel export)



ter the pandemic. By 2023, aviation emissions had already exceeded the level of 2018 (VCÖ, 2024a). Aviation affects global warming not only through GHG emissions, but also through nitrogen oxide (NO<sub>x</sub>) emissions and the release of pollutants into sensitive atmospheric layers, leading to cloud formation effects. These combined factors amplify the aviation's impact on global warming, making its effect substantially greater than those of the GHG emissions alone (Lee et al., 2021) (*medium confidence*).

The intensity of direct emissions from different modes of transport is illustrated in Figure 3.9, which shows the GHG emissions per passenger kilometer for the main modes of transport using the latest available data from June 2024. There is a striking disparity between rail and road transport. This is largely due to the fact that 81 % of passenger trains run on a catenary system powered entirely by renewable electricity sources, with 95 % of the energy coming from

hydropower and the remaining 5 % from other renewables (ÖBB-Holding AG, 2022). For aviation, Figure 3.9 includes the significant warming effects of GHGs other than  $CO_2$  that are released into sensitive layers of the atmosphere (as explained above).

It is important to note that GHG emissions are not limited to the use of vehicles, but are also emitted during the entire lifecycle of an individual vehicle, including construction, disposal and recycling ('cradle-to-grave' emissions). For battery electric vehicles (BEVs), these indirect emissions include in particular those from energy production (De Blas et al., 2020; Owen et al., 2023) (*high confidence*). In Austria's official GHG reporting, such indirect emissions are not allocated to the transport sector, but to the industry or energy sector (if their source is in Austria). However, for most purchased vehicles, the construction phase takes place outside Austria. In addition, the construction, maintenance,





**Figure 3.9** Greenhouse gas intensity of different transportation modes in Austria as of June 2024: Passenger transportation (PT, in gCO<sub>2</sub>eq per passenger-kilometer) and freight transportation (FT, in gCO<sub>2</sub>eq per ton-kilometer) (Umweltbundesamt, 2024a).

and disposal of road and rail infrastructure contribute significantly to GHG emissions (see Section 3.4.4). For road infrastructure, emissions vary by road type, usage patterns, and surface material (e.g., asphalt versus concrete), adding an estimated 0.4-6.4 % to transport-related emissions (Gruber and Hofko, 2023). Rail infrastructure emits 60 % more than rail operations in Austria - this relatively large offset is due to the low operational emissions of Austrian rail systems, which are largely powered by renewable energy. From a stock-flow perspective, which assesses material stocks of existing infrastructure and the flows required for maintenance, walking and public transport are the most efficient in providing services in terms of resource use and emissions. Road transport, on the other hand, ranks lowest in this respect (Virág et al., 2022b) (medium confidence).

**Risks:** In the transport sector, climate change risks primarily relate to infrastructure disruptions and damage caused by extreme weather events. These lead to significant repair costs and can cause delays and uncertainty for both passenger and freight transport (*medium evidence, high agreement*) (Section 3.4.4).

Impacts and adaptation mechanisms: The main expected impacts of climate change on the transport sector are damage to infrastructure caused by extreme weather events. These can be mitigated through adaptation measures, such as protecting rail and road infrastructure from landslides or treefall, and using more heat-resistant surface materials (medium evidence, high agreement). Climate change is also expected to directly affect transport mode choices: Higher summer temperatures may reduce the attractiveness of active modes such as walking and cycling, while milder temperatures and lack of snow in winters may increase their attractiveness, leading to ambiguous and likely location-specific effects (medium confidence). In addition, destination preferences could shift; for example, lower altitude winter tourism areas in the Alps are likely to become less attractive (Li et al., 2024) (see also Sections 4.3.3, 7.4.2) (medium evidence, high agreement).

Mitigation options: There are numerous mitigation options for both freight and passenger transport that fit into the Avoid-Shift-Improve (ASI) framework (see also the Cross-Chapter Box 4):

- Avoid: Reduce GHG emissions by decreasing the number of trips.
- Shift: Reduce GHG emissions by shifting to transport modes with lower GHG intensity.

• Improve: Reduce GHG emissions by adopting less GHG-intensive technologies, especially in the (remaining) road-based traffic.

There is a broad consensus that no single policy is sufficient, but that comprehensive policy packages are needed to achieve the necessary reductions in GHG emissions associated with the transport of goods and people, as shown in two recent research projects (QUALITY, aPPRAISE) for the case of Austria (Thaller et al., 2021; Dugan et al., 2022; Hössinger et al., 2023) as well as in the recently published 'Maßnahmenbericht Mobilitätswende' (Heinfellner et al., 2024a) (high confidence). In particular, push policies such as pricing mechanisms and speed limits are essential. Similarly, pull policies (e.g., improvements in public transport) and technological advances alone will not be sufficient, especially as the availability of renewable energy remains a limiting factor - likely beyond 2040 (robust evidence, medium agreement). Rising electricity costs also pose a substantial risk to transport electrification by slowing the transition process (medium confidence).

Specific recommendations for Austria have been outlined in the Austrian 2030 Mobility Master Plan (MMP) (BMK, 2021b), on the basis of which strategies for different sub-areas of mobility have been developed and published. The MMP, the report 'Pathways to Zero Carbon Transport Sector' (Angelini et al., 2022) and the scenarios calculated by the Austrian Environmental Agency (Umweltbundesamt, 2023a) all show that a substantial modal shift away from road transport is necessary to meet Austria's climate goals - continued growth in road-based travel (passenger-kilometers) and freight transport (ton-kilometers)<sup>3</sup> is considered unsustainable (medium confidence). Regarding infrastructure, besides limiting the construction of new infrastructure, mitigation options focused on emissions associated with the construction, maintenance and, in some cases, recycling of infrastructure are fairly limited (medium confidence). Some progress has been made in developing materials and technologies to reduce the embodied emissions (see also Section 4.2). Incentives to consider GHG reductions in the tendering process can be seen as a low-hanging fruit and key enabler (also for innovation) in this context (medium evidence, high agreement).

<sup>&</sup>lt;sup>3</sup> Ton-kilometers or [tkm], measures the transportation of one ton of goods over one kilometer, a common metric unit in freight logistics.

Most mitigation options in the transport sector have significant co-benefits (see Section 8.3.1), including reduced pollution, improved public health, improved quality and availability of public spaces (as cars take up about 10 times more space than other mobility modes), and mitigation of urban heat islands (UHIs) (Sovacool et al., 2021; Maier et al., 2023) (see also Section 3.2.3 Integrated mitigation and adaptation strategies) (high confidence). Reducing reliance on private car ownership could not only substantially reduce material demand, including for electric vehicle batteries (see Haas et al., 2025) (see Section 4.2), but also reduce household mobility expenditures (Schönfelder et al., 2016) (high confidence). In addition, higher taxation on vehicle ownership or use (e.g., through road pricing) could generate broader co-benefits by allowing the redistribution of tax revenues for investment in, for example, public transport. Such measures may become essential as electrification reduces public revenues from fuel taxes, which together with associated VAT revenues currently account for about 5 % of Austria's national public tax revenues (BMDV, 2022) (high confidence). Finally, Chapter Box 3.3 describes the potential of electric vehicles (EVs) to address the issue of uneven renewable energy distribution and the resulting need to balance residual loads.

Many of the most effective mitigation options, which are typically fiscal and regulatory in nature, face significant implementation challenges. These difficulties stem from the involvement of multiple governance levels (see Section 6.4.1), vested material interests from industries such as automotive (Gössling et al., 2016; Pichler et al., 2021) and construction industries (see Sections 6.2, 6.4), and deeply entrenched societal norms and perceptions regarding private vehicle use and ownership (Mattioli et al., 2020) (see Sections 6.2, 5.3.1, 5.3.2), which in turn reduce public acceptance of stringent policies (*high confidence*).

Transport is a recurring theme throughout this report and is addressed from a number of perspectives. Chapter 4 examines GHG emissions embedded in infrastructure, energy demand from electrification, and emissions associated with the tourism sector. Chapter 5 looks at inequalities in emissions between socio-economic groups (5.2.2), public acceptance of policies (5.6.3), and the influence of lifestyles and norms (5.3), issues that are also related to mobility. Sections 6.2 and 6.4 discuss how the automotive industry hinders the implementation of effective policies, while Section 6.5.1 outlines the Austrian legal framework for transport. Chapter 7 deals with mobility and tourism in the Alpine region. Chapter 8 presents mitigation scenarios involving the transport sector (Section 8.4), its contribution to achieving the Sustainable Development Goals (SDGs) (Section 8.3.2), and the costs and potentials (Section 8.3.1) for decarbonizing the transport sector.

#### 3.4.2. Person mobility

#### Status quo and trends

Transport performance: In 1990, approximately 76.7 billion passenger kilometers were traveled in Austria. By 2019, the last year before the COVID-19 pandemic, this figure had increased by 41 % to about 107.9 billion passenger kilometers (European Environment Agency, 2022), while the Austrian population grew by only about 16 % over the same period (Statistik Austria, 2022a). Mobility surveys conducted in 1983, 1995, and 2013/2014 show that the average daily travel distance of Austrians has steadily increased from 22 to 29 to 34 kilometers, respectively. Remarkably, the average daily travel time has remained almost unchanged at around 70 minutes (European Environment Agency, 2021b). This increase in passenger kilometers per capita, especially by car, correlates with several trends: The emergence of decentralized settlement structures (see Section 3.2.1), the functional segregation of livelihood functions, the expansion of 'social network geographies', and the continuous expansion of transport infrastructure (see Section 3.4.4). In addition, tourism contributes significantly to travel demand in Austria, as described in Sections 4.3 and 7.4.2.

**Modal split:** In 2022, motorized individual transport – including passenger cars and motorized two-wheelers – accounted for approximately 68 % of total domestic passenger transport performance (Heinfellner et al., 2024b). In 2018, the modal split at the trip-level was distributed as follows: 16 % for walking, 7 % for cycling, 61 % for car, and 16 % for public transport (BMK, 2021b). However, when evaluated by distance traveled, only 3 % of the total distance was covered by active modes (walking and cycling), reflecting their shorter average trip lengths, 27 % by public transport, and 70 % by car (European Commission: Directorate-General for Mobility and Transport, 2023). Despite the dominance of car travel, Austria stands out within the EU for its comparatively high use of public transport (Odyssee-Mure, 2022a).

The modal split is strongly influenced by the spatial characteristics of residential areas. For instance, car usage accounts for only 26 % of trips made by residents of Vienna (Wiener Linien, 2023). However, for commuters crossing Vienna's city limits from the surrounding regions on week-

days, the car modal share increases significantly to 77 % (Magistrat Wien, 2023). Despite the fact that many trips cover relatively short distances, car use remains prevalent: 40 % of all car trips are less than 5 km (BMK, 2016, Fig. 4.5-14). During and after the COVID-19 pandemic, the private car emerged as the 'winner' of the crisis, while active mobility gained attention as a promising alternative, especially at the expense of public transport (Hauger et al., 2022). Early data from post-pandemic years suggest that active mobility has maintained a higher modal share compared to prepandemic years (Bronnenmayer, 2024). The popularity of e-bikes has surged in recent years, with approximately one in two bicycles sold being an e-bike, resulting in a total stock of approximately 1.1 million e-bikes in Austria (BMK, 2023b). On the other hand, newer mobility forms such as e-scooters and car-sharing or car-pooling currently hold a small modal share (primarily in urban areas). Overall, these modes mainly tend to replace trips that would have been made by cycling, walking or public transport, especially in the absence of measures discouraging private car ownership and use (Mock, 2023). Research from Vienna and Zurich indicates that private ownership of e-bikes or e-scooters is more likely to substitute car trips than shared micromobility options (Laa and Leth, 2020; Bieliński et al., 2021; Shibayama et al., 2021; Reck et al., 2022; Mock, 2023) (medium confidence).

Emissions from car travel: Motorized individual transport accounts for nearly 65 % of Austria's total transport emissions, making it the primary source of GHGs in the sector. Domestic GHG emissions from passenger cars and motorized two-wheelers increased from 8.9 million tCO<sub>2</sub>eq in 1990 to 11.5 million tCO<sub>2</sub>eq in 2023. Taking into account price-related fuel exports (fuel purchased in Austria but consumed abroad), these emissions increase further, from 1.0 million tCO<sub>2</sub>eq in 1990 to 2.2 million tCO<sub>2</sub>eq in 2023 (Heinfellner et al., 2024b). However, the volume of price-related fuel exports dropped significantly in 2022 due to high energy costs in Austria. The growing share of EVs in the Austrian fleet is expected to significantly reduce transport-related emissions in the medium term, driven by EU regulations that aim to phase out combustion engine vehicles by 2035 (with some exceptions) (Regulation (EU) 2023/851). By the end of 2023, purely electric vehicles accounted for 20 % of newly registered passenger cars (Statistik Austria, 2024e) and 3.0 % of the total passenger car fleet in Austria (Statistik Austria, 2023b). It is noteworthy that about 80 % of these electric vehicles are company cars. In an EU-wide comparison, Austria ranked fifth in EV adoption in 2022 (EAFO, 2023).

Car ownership: The motorization rate in Austria has increased significantly in recent decades, rising from 391 passenger cars per 1,000 inhabitants in 1990 to 566 per 1,000 inhabitants at the end of 2022 (Statistik Austria, 2023b), a figure close to the EU average (European Commission: Eurostat, 2023). Austria's car fleet is relatively young, with an average age of 8.7 years in 2021, one of the lowest in the EU, despite a gradual increase over time, which is likely to reflect improvements in vehicle durability (ÖAMTC, 2003; ACEA, 2023b). Car ownership patterns vary significantly by settlement type. A substantial proportion of households own more than one car, especially in smaller towns and villages. In municipalities with less than 10,000 inhabitants, 38 % of households owned more than one car in 2019/20, whereas in Vienna this figure was only 7 % in the same period. Notably, car-free households are more common in Vienna, with 47 % of households reporting no car ownership in 2019/20, compared to 41 % a decade earlier. By contrast, only 12 % of households in smaller municipalities reported being car-free (Statistik Austria, 2011, 2022b).

Austria has not yet experienced a significant decline in the acquisitions of driving licenses among young adults, a trend observed in several other countries. In fact, the number of licenses issued to individuals aged 15 to 24 increased slightly from 119,789 in 2019 to 121,667 in 2022 (Statistik Austria, 2023: 'Kfz-Lenkberechtigungen'). This suggests that outside of urban areas - especially Vienna - the 'peak car' phenomenon has arguably not yet been reached in Austria. At the same time, there has been a notable shift in vehicle preferences, with the market share of SUVs increasing dramatically in recent years. The share of SUVs rose from 8.2 % in 2005 to 21.0 % in 2015, and by the first half of 2023, SUVs accounted for 44.5 % of all new car registrations (VCÖ, 2023b). This trend has substantial environmental implications, as SUVs generally have a higher carbon footprint in both production and use than smaller, lighter vehicles.

**Socioeconomic aspects:** On average, households allocate about 12 % of their net income to mobility expenditures, a share that is consistent across income groups but reflects higher absolute spending in wealthier households (Schönfelder et al., 2016; Aschauer et al., 2019). Higher mobility spending is often associated with greater GHG emissions, largely due to the ownership of more and larger vehicles. Air travel consumption is also unevenly distributed, with highly educated, young, and urban individuals in Austria taking a disproportionate share of flights (Falk and Hagsten, 2021) (*medium evidence, high agreement*) (see also Section 5.2.2).

#### Chapter Box 3.3. Flexibility potential of electro-mobility in Austria

Introduction: The expected demand for electricity from electro-mobility has important implications for the design of the electricity market, which is undergoing a transformative shift due to the expansion of renewable energy sources (see also Section 4.5). This shift transforms the market paradigm from one where flexible generation meets inflexible demand to one where flexible demand meets inflexible generation, requiring multiple flexibility options (Gea-Bermúdez et al., 2021; Plaum et al., 2022) (see also Chapter Box 3.2). Flexibility is critical not only for balancing generation and demand (Gade et al., 2022; Saffari et al., 2023), but also for addressing the uneven distribution of renewable energy and subsequent managing of residual loads (Allard et al., 2020). Greater demand-side flexibility can reduce reliance on fossil-fuel power plants, thereby reducing GHG emissions and minimizing curtailment of renewable energy. This, in turn, can encourage further investment in renewable energy sources (Loschan et al., 2023, 2024). In particular, electric vehicles (EVs) have a high flexibility potential (i.e., the difference between the load and the available capacity during the time span of the vehicle's connection to the power grid), even surpassing heat pumps (Karimi-Arpanahi et al., 2022; Kröger et al., 2023) (*medium confidence*).

**Impact of electro-mobility on electricity demand:** In a scenario where 50 % of passenger cars in Austria are battery-electric vehicles (BEVs) (2.5 million cars) with a specific energy demand of 16.60 kWh/100 km (Desai et al., 2023) and an annual mileage of 13,900 km (VCÖ, 2023a), the total annual charging demand would be 5.8 TWh. Fleet forecasts by Angelini et al. (2022), for light- and heavy-duty vehicles indicate an additional charging demand of 0.6 TWh by 2030. This would increase Austria's annual electricity demand, currently totaling 60.7 TWh (E-control, 2023), by 10 % (*medium evidence, high agreement*).

Flexibility potential of electro-mobility: EV demand-side flexibility can be used at the local level for peak shaving (Ioakimidis et al., 2018; Li et al., 2020; Van Kriekinge et al., 2021; Zheng et al., 2021) and valley filling (Zhang et al., 2014a, 2014b; Jian et al., 2017; Ioakimidis et al., 2018), leading to reduced renewable energy curtailment (Haddadian et al., 2015, 2016; Schuller et al., 2015). The chosen charging strategy affects the flexibility potential: For slow charging with lower charging power, vehicles are connected to the charging station for a longer period of time, while fast charging with higher power levels is mainly used when the connection to the charging station is limited to 1–2 hours. The flexibility potential is especially high during night hours (Flammini et al., 2019; Loschan et al., 2023), when the majority of slow charging processes take place (Speth and Plötz, 2024) (*medium evidence, high agreement*). It should, however, be noted that the full flexibility potential offered by EVs is constrained by the requirement to meet transportation needs and by the temporal and spatial availability of the charging infrastructure and the vehicle (Mills and MacGill, 2018).

**Bidirectional technology potential:** In addition to the potential of demand-side flexibility, Vehicle-To-Grid (V2G) technology ('bidirectional charging') allows the vehicle battery to be used to supply electricity to the system, further expanding the range of potential interactions between BEVs and the electricity sector (Misconel et al., 2022). For example, EVs can be used to transport electricity by charging at one location, driving to a destination, and discharging there to act as an electricity supplier, thereby reducing grid use (Khodayar et al., 2012; Verzijlbergh et al., 2014; Nikoobakht et al., 2019). This flexibility can further reduce the need for the expansion of transmission capacity and redispatch measures (Loschan et al., 2023), improve the balance between generation and demand (Bibak and Tekiner-Mogulkoc, 2022), and reduce peak demand (Tan and Wang, 2017). In this way, EVs can provide ancillary services and reserves (Thingvad et al., 2019; Osório et al., 2021; Figgener et al., 2022).

Barriers, drivers and policy needs: While the demand-side flexibility of EVs can support the decarbonization of the electricity market, it may simultaneously lead to increased electricity consumption (Tehrani and Wang, 2015; Gilleran et al., 2021). This could be problematic because the integration of non-dispatchable renewables, combined with the electrification of multiple sectors, leads to increased power flows over the transmission grid, resulting in congestion and consequently higher redispatch costs (Loschan et al., 2023). Compared to the demand-side adjustment of load, large-scale implementation of bidirectional charging, though promising, is facing more barriers such as increased battery wear due to frequent charge-discharge cycles and limited availability of V2G-capable vehicles and bi-directional chargers, which remain expensive (Goncearuc et al., 2024).

Overall, the future flexibility potential is highly dependent on the development of preferred charging strategies for the different applications and mobility patterns of EVs, as well as expansion strategies for public charging infrastructure (*medium evidence, high agreement*). Realizing the flexibility potential of e-mobility in the electricity system will require a robust regulatory framework to address challenges while unlocking opportunities for grid stability and sustainability (Gonzalez Venegas et al., 2021; Sadeghian et al., 2022; Loschan et al., 2023).

#### **Mitigation strategies**

The scientific consensus emphasizes that achieving decarbonization of the transport sector requires a comprehensive mix of awareness-raising, regulatory, infrastructure and fiscal measures (*high confidence*). Integrated policy packages that combine *push* measures (banning, limiting, or disincentivizing carbon-intensive travel) with *pull* measures (making sustainable alternatives like public transport and cycling infrastructure more feasible and attractive) are the most effective approach (e.g., Dugan et al., 2022; Koch et al., 2022; Hössinger et al., 2023; Heinfellner et al., 2024a) (for local policies, see Section 3.2.2).

Avoiding travel: The COVID-19 pandemic demonstrated the potential of teleworking to reduce commuting and teleconferencing to reduce business travel, particularly air travel. However, due to rebound effects, such as increased local errands on home-office days, existing evidence is inconclusive about the extent to which travel avoidance is feasible (Maier et al., 2022) (*medium confidence*).

Effective spatial and settlement planning (Section 3.2.2) has significant potential in the medium to long term to reduce the frequency and distance of trips, while promoting active transport modes (avoid and shift). Strategies include urban 'city of short distances'/'15-minute city' concepts in urban areas and the revitalization of town and village centers in rural or semi-urban regions. These initiatives often focus on reducing car-centric infrastructure through lanes restrictions, road closures, increased parking charges or decreased parking spaces while ensuring daily destinations (e.g., workplaces, shops, and public services) are accessible by walking, cycling, or public transport (see Jandl et al., 2024) (see Section 6.5.1). Regulatory measures such as speed limits and the elimination of minimum requirements for parking facilities further support these goals (high confidence). More broadly, promoting sufficiency through changes in lifestyle (i.e., consumption) and changes in social and cultural norms can enhance acceptance of these measures (see Section 5.6.2).

Shifting away from private motorized vehicles: Efforts to shift travel away from private motorized vehicles include reducing the cost and improving the attractiveness of alternative modes (high confidence). For public transport, this includes simplified and affordable fares, such as the regional/ national versions of the 'Klimaticket' (BMK, 2023c), as well as improved quality (e.g., reliability, frequency, integrated schedules) and expanded spatial and temporal coverage. Demand-responsive transport (DRT) can address mobility gaps in remote areas, consistent with a 'mobility guarantee' approach that minimizes the need for car ownership (Laa et al., 2022). For international travel, advancing high-speed rail, intermodal connections, and addressing capacity constraints (e.g., night trains) are critical, although challenges remain in the medium term (medium confidence). Improving biking and walking infrastructure - especially for shorter trips, which make up the majority - remains vital in both urban and rural areas (e.g., ETH Zürich, 2023) (high confidence).

Electric vehicles as key technological improvement option: Travel demand that cannot be avoided or shifted must rely on low-emission technologies that are energy and cost efficient across their entire lifecycle. For passenger road transport, BEVs best meet these criteria when powered by 100 % renewable electricity (Hoekstra, 2019; Fritz et al., 2021, 2022; BMK, 2022b) (robust evidence, medium agreement). Financial incentives (subsidies, tax reductions), regulatory measures (low emission zones, e-taxi mandates, see also Ajanovic et al. (2021) for the case of Vienna), charging infrastructure, and policies that enable the installation of charging points in shared housing are essential (Kumar and Alok, 2020) (high confidence). Economies of scale are expected to further lower the purchase cost of BEVs, reducing the need for subsidies (Wicki et al., 2022). EU bans on diesel/gasoline car sales by 2035 remain the most effective lever to accelerate this shift. Alternative fuels such as e-fuels - synthetic fuels produced using renewable electricity, water, and carbon dioxide (CO<sub>2</sub>) - and hydrogen are unlikely to play a significant role in passenger transport due to lower efficiency and higher costs - e.g., Morrison et al.

(2024) estimate  $CO_2$  abatement costs of over  $EUR_{2023}$  600 per ton (GHG emissions reduced in  $CO_2$  weight equivalent) for e-fuels versus less than  $EUR_{2023}$  150 per ton for BEVs (see Section 8.3.2). Strimitzer et al. (2022) conducted an efficiency comparison in terms of the distance that can be traveled using 10,000 kWh of renewable electricity, demonstrating the energy efficiency of BEVs at 74 %, far surpassing hydrogen fuel cell vehicles (30 %) and e-fuels (15 %). However, e-fuels are likely to play a critical role in decarbonizing air travel, particularly for large aircraft (*medium confidence*).

Pricing and subsidies: Pricing strategies, in line with the user-/polluter-pays principle are essential for achieving the goals of avoiding trips, shifting to sustainable modes, and relying on technological improvements (Angelini et al., 2022). For instance, the recent 'Maßnahmenbericht Mobilitätswende' shows that increasing the mineral oil tax and introducing road pricing have the highest GHG emission reduction potential among 13 policy measures examined (Heinfellner et al., 2024a). Currently, Austria's road transport cost coverage ratio - covering variable external (e.g., environmental) and infrastructure costs with tolls, taxes and fees charged to users - is only 21 %, which is one of the lowest in Europe (European Commission: Directorate-General for Mobility and Transport et al., 2019). Compared to bans, pricing policies have the advantage of generating tax revenues that can be used to counterbalance negative distributional effects, especially in rural car-dependent regions (Axhausen et al., 2021). A distinction can be made between generic pricing instruments such as carbon pricing (the ETS will be expanded to cover transport from 2027 onwards: ETS 2) (European Union, 2024), and more specific instruments such as city tolls, parking fees and taxes on car ownership. With sufficiently high prices, all of these instruments have a high potential for influencing travel behavior in such a way that substantial CO<sub>2</sub> emission reductions can be generated (high confidence). Removing counterproductive subsidies (e.g., for company cars, commuting, or hybrid SUVs) also has significant potential to reduce emissions (Peneder et al., 2022) (high confidence). Similarly, introducing taxes on kerosene or other flight charges could encourage a modal shift away from air travel and lead to innovations that benefit aircraft efficiency (medium evidence, high agreement).

Exceptional challenges in reducing car modal share: Reducing the car modal share in rural and semi-rural areas remains particularly challenging due to the perceived convenience and reliability of cars (*high confidence*). Spatial planning policies – such as minimum standards for public infrastructure, expansion of public transport, and investment in walking and cycling – can enable shifts in the modal split. Studies suggest that a significant proportion of the population is willing to switch modes under these conditions (Millonig et al., 2022; Peer et al., 2023). In the medium and longer term, these efforts could lead to a reduction in car ownership, especially among households with multiple cars (*medium confidence*).

#### Scenarios

The Environment Agency Austria (EAA) presents possible trajectories for transport-related GHG emissions (Umweltbundesamt, 2023a). In the With Existing Measures (WEM) scenario, only measures (that were) bindingly implemented or legally fixed by January 1, 2022, are considered. These are insufficient to reverse unfavorable trends. The WEM scenario predicts an increase in passenger transport to 126.7 billion passenger kilometers by 2040 and an increase in motorization to 639 cars per 1,000 inhabitants. By 2040, 69 % of passenger kilometers would rely on motorized private transport, with only 56 % of the fleet electrified. This results in GHG emissions of 10 million tCO<sub>2</sub>eq in that year (Umweltbundesamt, 2023b).

In contrast, the EAA's Transition scenario models complete decarbonization of transport by 2040 using a backcasting approach. Measures are tailored in design and intensity to achieve climate neutrality by this deadline.

Motorization stabilizes at around 570 cars per 1,000 people, with the fleet almost entirely electrified by 2040. This eliminates GHG emissions from private transportation and meets climate neutrality targets. However, this requires about 23.4 TWh of renewable electricity (a 35 % increase from the WEM scenario), putting transport in competition with other sectors. Issues around equitable distribution of limited renewable energy remain unresolved, with the transport sector accounting for 20 % of energy use in the 'Transition' scenario by 2040.

#### Governance and acceptability

Legal and institutional challenges – such as the insufficient design and enforcement of existing measures – often hinder effective policy implementation (Jandl et al., 2024) (see Section 6.5.1). In addition, a critical barrier to implementation remains the lack of public acceptance, especially for stringent policy instruments (see also Heinfellner et al., 2024a) (see Section 5.6.3). Public opposition is, however, not limited to strict regulations, as the debates around the '15-minute city' concept show (Caprotti et al., 2024). Strategies for improving public support include (*high confidence*):

- Policy packaging: Combining push and pull measures can mitigate negative distributional effects, such as those felt by rural residents relying on cars for commuting (Thaller et al., 2021; Dugan et al., 2022; Hössinger et al., 2023; Heinfellner et al., 2024a).
- Participatory approaches: Co-design and stakeholder engagement increase legitimacy and acceptance.
- Awareness and co-benefits: Informing the public about policies' benefits can build support (e.g., Peer et al., 2023).
- Strategic implementation: Timing, coalition-building, trials, and gradual rollouts increase acceptance. For example, Stockholm's temporary congestion charge became a permanent policy after public support increased post-trial (Schuitema et al., 2010; Heyen and Wicki, 2024).
- Address spillover effects: Avoid unintended consequences, such as Vienna's parking permit system ('Parkpickerl'), which initially shifted parking demand to adjacent districts until more comprehensive policies were enacted.
- Making sustainable travel the default: Designing sustainable travel options in such a way that they are more convenient to use than less sustainable travel options (Section 5.5.1).

Recent surveys suggest that Austrians may be more open to regulatory and pricing policies than previously thought, with a majority viewing most measures positively or neutrally (Hössinger et al., 2023; Heinfellner et al., 2024a). Lack of acceptance of technological options may also hinder progress towards decarbonization (*high confidence*). For example, a substantial portion of the Austrian population remains reluctant to adopt electric vehicles (Priessner et al., 2018).

#### 3.4.3. Freight transport

#### Status quo and trends

Like passenger transportation, freight transportation is a derived demand that results from the need to move goods. As a result, economic activity, typically measured by GDP, has historically been a reliable predictor of freight volumes and flows. However, changes in the structure of the economy such as the growth in the service sector, changes in supply chains (e.g., just-in-time logistics), and policies affecting shippers and carriers - have weakened this correlation and reduced the accuracy of forecasts (Meersman and Van de Voorde, 2013; ITF, 2023). This decoupling is evident when comparing ton-kilometers per GDP unit across EU nations; Austria is in line with the EU average, but exceeds many Western European countries (Odyssee-Mure, 2022b). Despite this trend, the consensus points to continued growth in freight demand, driven in particular by the expansion of e-commerce.

Emissions, modal split and freight demand: Figure 3.10 shows the evolution of transport performance for road and rail since 1990. In particular, road transport performance has increased considerably over this time period.

Table 3.4 shows the freight transport demand for inland waterways, rail and road in 2023, categorized by transport type. Road freight transport reached a record high of 56,846,395  $\times$  1,000 tkm in 2021, accounting for 71 % of the total freight transport share. By 2023, the road share increased slightly to 72 %. Rail's role in intermodal freight



Figure 3.10 Freight transport performance by rail and road in Austria (1990– 2023) (Heinfellner et al., 2024b).
Modal split in the year 2023 [1,000 tkm]	Inland	Import	Export	Transit	Total	
Inland waterway	509,407	320,922	310,71	45,506	1,186,546	2 %
Rail	3,948,293	4,921,328	3,516,192	7,819,358	20,205,170	26 %
Road	18,034,181	8,725,596	8,614,571	19,754,058	55,128,407	72 %
Overall	21,982,474	13,646,924	12,130,763	27,573,416	76,520,123	100 %

Table 3.4Modal split of freight transport (road/rail) by transport type (inland, import, export and transit) in Austria in the year 2023 [1000 tkm](Statistik Austria, 2024d).

transport is mainly facilitated by 14 inland container terminals across Austria (Schienen-Control, 2023; VABU, 2023).

The contribution of road freight transport to total GHG emissions from transport increased from 31 % in 1990 to 34 % in 2023, but declined significantly from 2022 to 2023 as Austria's economic output also declined. Fuel exports in commercial vehicles account for 12 % of these emissions (European Commission: Eurostat et al., 2023; Schuster et al., 2023; Heinfellner et al., 2024b). Among Austria's transport-intensive industries (e.g., mining, food, and waste), logistics activities account for about 12 % of their total GHG emissions (Miklautsch et al., 2022).

Parcels and urban freight transport: In 2020, approximately 290 million parcels were delivered in Austria, with parcel deliveries increasing by 9 % annually over the last decade. The Viennese chamber of commerce projects an increase to 339 million parcels by 2030 (WKO Wien, 2022). Despite this large volume, the associated GHG emissions have not been explicitly measured. Urban freight accounts for 10 % of urban road transport in Europe, with deliveries estimated to contribute 20 % of freight emissions – the same as maritime transport, although it represents only 3 % of freight activity compared to 70 % for shipping (ITF, 2023).

Technological advancements: There have been significant technological changes in the light- and heavy-duty vehicle markets in recent years. For heavy-duty electric trucks, recent studies show that electrified highways are economically feasible, but feasibility depends on international adoption, which seems unlikely. In addition, the construction of such electrified highways is expected to result in large GHG emissions (Qiu et al., 2022; Colovic et al., 2024) (*medium confidence*).

Battery-operated freight vehicles have been advancing in recent years: In 2024, battery trucks reached a 4 % market share among light-duty vehicles, which are often used for distribution and especially urban freight delivery, and a 1.6 % share among heavy-duty vehicles (i.e., larger truck classes) in Austria (AustriaTech, 2024). Scaling up this adoption will require improved charging infrastructure, especially along highways and at loading sites (*limited evidence*, *high agreement*).

Hydrogen-powered heavy-duty vehicles offer greater range than (existing) battery-operated vehicles and faster refueling (similar to diesel and gasoline). However, high hydrogen prices and limited filling stations (four in Austria) make them economically and logistically unattractive, and this is unlikely to change in the short to medium term (*medium confidence*).

Hydrotreated vegetable oil (HVO) is gaining traction as a transitional fuel due to the limited availability of heavy-duty battery electric trucks and related charging infrastructure. A 220-fold increase in HVO sales compared to 2022 levels occurred following Austria's new national fuel regulation in 2023, which promotes the use of renewable energy (Heinfellner et al., 2024b).

**Energy efficiency:** Low-carbon fuels, including biofuels and synthetic fuels, significantly reduce GHG emissions compared to standard diesel for heavy road vehicles (Benajes et al., 2024). However, compared to battery electric trucks, biofuels require 2–3 times more cumulative energy input (i.e., the sum of all primary energy inputs), while synthetic fuels (or e-fuels) require 5.5–6.5 times more, due to their energy-intensive production processes that use electricity to convert hydrogen and carbon dioxide into a usable energy source (Fritz et al., 2022). This is a major challenge given the expected increase in demand for renewable energy across all sectors (*medium evidence, high agreement*). Lifecycle assessments also indicate that battery electric trucks have the lowest cradle-to-grave GHG emissions (O'Connell et al., 2022) (*medium confidence*).

Adaptation: As discussed in Section 3.4.4 on infrastructure, the increasing likelihood of extreme weather events – such as landslides, floods, and tree falls – and their impacts on infrastructure negatively affect network connectivity and thus accessibility and reliability. With higher global warming levels, such events are expected to become even more frequent (see Section 3.2.3, Sections 1.4.1, 1.6.2, 1.7, and Cross-Chapter Box 1). Particularly in mountainous areas,

AAR2

where alternative routes or modes are not available, prolonged network disruptions can lead to high economic costs if goods are not delivered on time (e.g., Fikar et al., 2016) (*medium evidence, high agreement*). Supply chain diversification offers some potential to mitigate these risks (*medium confidence*).

#### Mitigation strategies

The medium-term mitigation potential for freight transport is considered to be relatively high (*high confidence*). This potential can be divided into three key areas: Avoiding unnecessary freight movements, shifting to more sustainable transport modes, and exploiting technological improvements. A comprehensive policy addressing all three dimensions could include a dynamic road pricing system that takes into account usage, location, time of day and vehicle type and internalizes negative externalities, including not only GHG emissions but also local air pollution, accidents, noise and infrastructure costs (Marcucci et al., 2023) (*high confidence*). In Austria, less than one-fifth of these costs are currently covered by charges and tolls, despite the existence of a motorway toll system (European Commission: Directorate-General for Mobility and Transport et al., 2019).

Avoiding and shortening trips: Optimizing logistics and supply chains (e.g., reducing empty backhauls) can play a key role in reducing freight movements (*medium confidence*). Regionalization of production and distribution can also lead to shorter transport distances (*medium evidence, high agreement*). Adopting circular economy principles, which promote resource efficiency and product reuse, can help minimize the need to transport new goods (*medium confidence*). In urban areas, tailored approaches such as parcel boxes can further contribute to GHG emission reductions (*limited evidence, medium agreement*). However, while these measures could moderate the projected growth in freight volumes, an overall reduction would require strong pricing and/or regulatory measures (*medium confidence*).

Shift from road to rail: Shifting long-haul freight movements from road to rail has significant GHG emissions-savings potential given the low emission factors of railway operations in Austria (see Section 8.3.2 and the recent 'Maßnahmenbericht Mobilitätswende') (Heinfellner et al., 2024a). Despite the relatively high share of rail freight in Austria (27 %), increasing this share to the political target of 40 % by 2040 poses considerable practical challenges. These include the rigidity and limited capacity of the rail system, as well as first and last mile issues. Accessibility and expansion of rail infrastructure, terminals and sidings are critical, with the latter even declining in recent years. Strong policies, such as bans on road freight along corridors with sufficient rail capacity (limited to certain types of freight, times of day, or truck types), could enforce this shift. Improvements in rail speed, reliability, and predictability are also critical. To compensate for rail's higher costs, subsidies – e.g., for personnel or infrastructure usage fees – could be effective, especially in the absence of road freight charges that fully internalize the corresponding external costs (*medium evidence, high agreement*).

Rail modal shift is generally viable for trips over 250– 300 km; shifts for shorter distances are usually unattractive for shippers and carriers. Due to Austria's geography, approximately 80 % of the longer trips start or end abroad. Achieving the 40 % modal share is therefore highly dependent on improving international connectivity, without which only a 34 % share is considered achievable (BMK, 2021b). Cross-border rail operations remain inefficient, hampered by waiting times, limited capacity, and unattractive, difficult-to-schedule services.

In addition, interconnectivity gaps also exist between rail networks and airports, seaports, and major inland waterways. The EU's efforts to address these issues are pronounced, as evidenced by initiatives such as the 'Fourth Railway Package' (European Commission, 2016) and the multimodal Trans-European Transport Network (TEN-T) 'core network' targeting major European axes by 2030 (and a more comprehensive network by 2050) (European Commission, 2020b). It is estimated that failure to complete the TEN-T would result in a loss of 1.8 % of potential economic growth and 10 million person-years of employment across the EU (Schade et al., 2015) (*medium evidence, high agreement*).

Technological options: There is great potential to shift road freight vehicles to low or zero emission technologies. EU regulations, including a Directive on the deployment of alternative fuel infrastructure (Directive 2014/94/EU), the 'Clean Vehicles Directive' (Directive (EU) 2019/1161), and a Regulation on CO<sub>2</sub> emission performance standards for new heavy-duty vehicles (Regulation (EU) 2019/1242), require manufacturers to reduce the fleet-wide average CO<sub>2</sub> emissions of their newly registered trucks per calendar year by 15 % by 2025 and 45 % by 2030 (compared to 2019 levels). These measures have already reduced the CO<sub>2</sub> intensity of the freight vehicle fleet, although growing freight demand has largely offset these gains (*medium evidence, high agreement*) (see Figure 3.8 and Figure 3.10). The relatively young truck fleet in Austria (average age: 6.6 years (ACEA, 2023a), possibly a result of significantly lower toll fees for less polluting vehicles) underlines the potential for a rapid change in propulsion technology, which would lead to considerable GHG emission reductions. This can be seen as low-hanging fruit, and vehicle manufacturers can be seen as enablers of a faster transition. Battery-electric systems, particularly for distances under 300–500 km, offer feasible near-term solutions if supported by regulatory and infrastructure advancements. A key barrier is limited range and long charging times, which could be mitigated by systems that allow battery swapping (e.g., a 'stagecoach' model), especially for international routes (*medium confidence*).

Emissions from urban freight transport can be significantly reduced by establishing zero-emission zones in cities, effectively enforcing  $CO_2$ -free logistics in urban centers. Light-duty electric vehicles (range ~500 km, charging time ~30 minutes) have already already entered the market and could be complemented by (e-)cargo bikes for short-distance and special deliveries (*medium evidence, high agreement*).

#### **Scenarios**

For the Austrian context, there is little evidence on decarbonization scenarios for the freight sector. Sedlacek et al. (2021) calculate two scenarios for the road freight sector, one corresponding to a WEM scenario and one that achieves carbon neutrality by 2040. The scenarios emphasize the role of technological progress in all vehicle categories. A Europe-wide strategy and policy framework is considered essential due to the international nature of freight transport. Despite significant transformation efforts, they conclude that the macroeconomic impact on Austria's GDP is expected to be small (*low confidence*).

#### 3.4.4. Transport infrastructure

Infrastructure supports the physical mobility of goods and people and includes fixed structures (e.g., roads, railways, bicycle and pedestrian paths, inland waterways), access points (e.g., train stations, airports), and parking facilities (both on- and off-street). In 2022, Austria's transportation infrastructure included approximately 128,000 km of roads, 4,965 km of railways, 1,033 railway stations, 6 commercial airports, 8 inland waterway terminals, and 538 rail sidings. Statistics for local infrastructure, such as bicycle paths and parking spaces, are difficult to obtain, but notable examples include 280 km of tram lines and 80 km of subway networks.

Infrastructure can be characterized in terms of its capacity and technical specifications, which in turn affect its (perceived) quality, safety and resilience. Changes in infrastructure alter accessibility, typically affecting the flow of people and goods across routes and transport modes. Especially over time, infrastructure developments can reshape spatial patterns, such as residential and business locations and supply chain structures. For example, the expansion of (high quality) road networks has repeatedly been shown to induce urban sprawl and suburbanization (*robust evidence, medium agreement*) (see also Section 3.2).

The provision of transport infrastructure is associated with significant embodied GHG emissions (see also Section 4.2), which are often underestimated or ignored, leading to a severe undervaluation of the environmental impacts of transport (Facanha and Horvath, 2007; Chester and Horvath, 2008). A distinction can be made between direct and indirect effects:

- Direct effects result from energy consumption and associated GHG emissions for raw material extraction (mining) and production, transportation, production and construction of the transport infrastructure itself, maintenance and eventual disposal. Infrastructure construction requires substantial amounts of material (Haas et al., 2024) (*medium evidence, high agreement*).
- Indirect effects include increased fuel consumption of vehicles due to road surface irregularities, but also changes in travel behavior and freight transport movements induced by infrastructure improvements: In general, infrastructure improvements often increase travel activity, potentially increasing GHG emissions, especially in the case of road infrastructure expansions, where on average a 1 % increase in road kilometers leads to a corresponding 1 % increase in vehicle kilometers traveled (e.g., Garcia-López et al., 2022) (*medium confidence*). Similarly, the availability of convenient parking greatly increases the likelihood of car ownership and use (e.g., Hess, 2001) (*medium confidence*).

Infrastructure can exacerbate the effects of global warming, particularly in urban areas, by reinforcing the UHI phenomenon as described in Section 3.2.1, which is particularly relevant in densely populated areas. In addition, infrastructure covers a significant area of land, contributing to soil sealing and potentially increasing the negative impacts of extreme weather events such as heatwaves and heavy precipitation (see Section 3.2.3). Land-based infrastructure occupies 6.7 % of Austria's permanent settlement area (Umweltbundesamt, 2020). Among transportation modes, rail infrastructure uses 7 m<sup>2</sup> of land per passenger, compared to 100 m<sup>2</sup> per car user.

In addition to being a source of GHG emissions, infrastructure is also strongly affected by climate change and associated weather events – see also Section 3.2.3 for the urban context and Section 7.4.3 for infrastructure in alpine areas. The main weather hazards requiring adaptation are high temperatures and (excessive) precipitation (Fian et al., 2021; Palin et al., 2021) (*medium evidence, high agreement*).

In the following subsections, rail and road infrastructure, the dominant types in Austria, are analyzed in detail. Other infrastructure types also contribute to GHG emissions, but have a comparatively smaller network size and therefore a limited reduction potential. Nevertheless, some statistics and considerations for rail and road infrastructure can be adapted to broader application. For example, trams and subways share characteristics with rail infrastructure due to their rail-bound systems, while on-street parking is consistent with emission factors for road surfaces. A discussion on urban infrastructure planning can be found in Section 3.2.2.

Although local variations and dependencies make it challenging to generalize the need for infrastructure investments, adaptations, or even removal, evidence suggests that welfare benefits of additional infrastructure diminish at high levels of infrastructure availability (Virág et al., 2022a) (*limited evidence, medium agreement*).

#### Railway infrastructure

As of 2023, Austria's rail infrastructure network covers 4,935 km, of which 2,262 km are double track (ÖBB-Holding AG, 2024). Among European countries, Austria ranks second in per capita investment in rail infrastructure, surpassed only by Switzerland (ITF, 2022). The investment plan of the Federal Ministry for Climate Action, Environment, Energy, Mobility, Innovation and Technology allocates EUR<sub>2023</sub> 21.1 billion for the period 2024-2029 for new line construction, line expansion and electrification, freight terminals, digitalization, efficiency improvements, and park-and-ride facilities. The annual budgets include EUR<sub>2023</sub> 780 million for maintenance and EUR<sub>2023</sub> 838 million for reinvestment. Major cost-intensive ongoing projects include the Brenner Base Tunnel, the Koralm Tunnel and Semmering Base Tunnel, while further investment plans are guided by Austria's long-term expansion strategy, the 'Zielnetz 2040' (BMK, 2023a, 2024b).

Existing rail capacity is heavily utilized. In 2023, the Austrian Federal Railways 'ÖBB' handled 12.6 billion traveled passenger kilometers, which corresponds to 7,026 trains operated daily (ÖBB-Holding AG, 2024). In addition, freight transport accounted for 26.1 billion ton-kilometers. Maintenance work limits the extent to which freight transport movements can be shifted to night hours. Expanding rail capacity is complex and cost-intensive, especially for routes that cross mountains or urban areas, requiring costly tunnels and infrastructure. Austria's central location in Europe means that its trans-European connectivity relies heavily on investments from neighboring countries. Without adequate foreign investment, the shift of freight from road to rail is constrained, as intra-Austrian distances often fall below the 300 km threshold typically considered uneconomic for rail freight (see also Section 3.4.3).

The majority of the environmental impacts of railway infrastructure are caused by the rails, ballast, sub-ballast, and civil engineering structures (de Bortoli et al., 2020), with the track substructure being a critical factor (Cheng et al., 2020; Pons et al., 2020). The GHG emissions for ballasted railway tracks reported in this section are calculated based on a detailed bottom-up approach that considers all related upstream and downstream processes, including local conditions (traffic loads, radii, and elevation), used track components, specific supply chains, and emissions from trackwork machinery (Landgraf and Horvath, 2021; Landgraf et al., 2022). In addition, supply chains, asset lifetimes, and maintenance requirements throughout the infrastructure lifecycle are considered (Landgraf and Horvath, 2021) (*medium evidence, high agreement*).

In Austria, the focus is on maintaining and renewing the existing network rather than building new lines. In 2023, for example, 216 km of track were renewed, which corresponds to an annual rate of 2.3 % (ÖBB-Infrastruktur AG, 2024), while new track construction has averaged 12 km per year since 2008 (ÖBB-Holding AG, 2022). This continuous renewal rate offers high potential for innovation, as new solutions can be integrated swiftly into the existing network (*medium evidence, high agreement*).

Annual cradle-to-gate GHG emissions from rail infrastructure – based on the current infrastructure network, asset distribution and renewal rates – amount to 234,730 tCO<sub>2</sub>eq, compared to 144,900 tCO<sub>2</sub>eq from passenger operations and 149,500 tCO<sub>2</sub>eq from freight operations in 2019. Tunnels and aerial structures (bridges) remain the largest contributors to emissions per kilometer (Chang and Kendall, 2011; Landgraf and Horvath, 2021) (*medium evidence, high agreement*).

Track construction and maintenance account for 55– 60 % of GHG emissions within the Austrian rail network (Landgraf and Horvath, 2021). This share is relatively high compared to other countries (Banar and Özdemir, 2015; Jones et al., 2017) because, unlike most other foreign rail operators, ÖBB uniquely sources its traction power from renewable energy sources – 95 % hydropower and 5 % other renewables (ÖBB-Holding AG, 2022). This significantly reduced operational GHG emissions, but increases the relative share attributable to infrastructure.

#### Mitigation strategies

**Planning for future rail demand:** Overall, the shift towards rail is crucial to avoid environmental impacts in the transport sector, as emphasized by the European Green Deal's strategy to 'accelerate the shift to sustainable and smart mobility' (ERA, 2020). However, the development of transport infrastructure development should be limited to what is necessary, based on in-depth traffic volume simulations, which the ASI framework promotes.

Shift to low-emission materials and fuels in rail infrastructure construction and maintenance: The main further mitigation potential lies in shifting to low-emission materials and fuels (ÖBB-Holding AG, 2022). Key areas include steel and concrete production, the adoption of circular economy principles, and the use of fossil-free propulsion for heavy maintenance machinery and transport. In particular, as the extraction and processing of raw materials account for 18 % of the EU's total GHG emissions associated with the consumption of goods and services, climate-friendly procurement and processing practices are crucial (European Environment Agency, 2021a). In addition to further research and support (subsidies, financing) for the market entry of sustainable products, this can be achieved by quantifying and integrating environmental impacts into the public procurement process (BMK, 2021a; UNEP, 2021). This, in turn, may encourage contractors and manufactures to invest in environmentally efficient production processes and services (medium confidence).

Efficiency improvements in rail infrastructure design, usage, and maintenance: Railway infrastructure can be improved by increasing efficiency within the system. The further implementation of the 'European Train Control System' (ETCS) according to Austria's 'National Implementation Plan' (BMVIT, 2017) will increase the efficiency and capacity of the existing railway infrastructure. In addition, ÖBB has already taken a number of measures to mitigate lifecy-

cle costs and GHG emissions (ÖBB-Holding AG, 2022). For example, ballast cleaning allows about 50 % of material to be reduced when reinvesting in existing rail infrastructure (Zeiner et al., 2021). Optimized maintenance planning and improvements in railway infrastructure design can also extend the service life of existing infrastructure (Landgraf and Horvath, 2021).

## Road infrastructure

In 2022, the Austrian road infrastructure consisted of 2,260 km of high-level roads ('Autobahnen' and 'Schnellstraßen', 33,800 km of intermediate-level roads ('Landesstraßen' 'B' and 'L'), and about 92,000 km of low-level roads ('Gemeindestraßen'). Overall, more than 95 % of roads are built with asphalt pavements (bituminous bound, hot mix asphalt (HMA)) and less than 5 % with concrete pavements (cement bound, portland cement concrete (PCC)). However, about one third of the high-level road network is cement-bound. In comparison, Austria has a high-level road network that is 50 % larger per capita than the EU average and about 55 % larger than Germany's (VCÖ, 2024b).

In 2020, EUR<sub>2023</sub> 1,291 million were spent on the high-level road network, 46 % on new construction and 54 % on maintenance. For this, 7.4 million tons of asphalt mixture were produced, a decrease of 10 % from the peak year (2010), while 1.26 million tons of reclaimed asphalt were recovered for recycling. Of the recycled material, 70 % was reused in HMA, while 30 % was diverted to other uses or landfills (Blab et al., 2012; BMK, 2022c; EAPA, 2023).

GHG emissions from road infrastructure vary by material type. This can be assessed using a production-based material flow analysis, which includes the production of (raw) materials (bituminous binders, cement, mineral aggregates, additives), and the transport of materials. HMA production results in 40–50 kgCO<sub>2</sub>eq/t, while PCC emissions range from 75–100 kgCO<sub>2</sub>eq/t, as shown by Gruber and Hofko (2023) with calculations based on Gruber (2023). For HMA, raw materials contribute approximately 50 %, transportation 5 %, and production of asphalt mixture about 45 % of total GHG emissions. In contrast, PCC raw materials account for 95 % of the emissions, with transportation and production accounting for the remainder.

Overall, the impact of asphalt paved roads on GHG emissions is 0.4 % to 4 % of traffic-related emissions for the

low- and high-level road network, respectively, while concrete roads contribute 0.6 % to 6.4 %, respectively (calculations based on Gruber, 2023). Due to continuous dynamic loading of road infrastructure by vehicle traffic, the road surface and structural properties deteriorate over time. As a result, top-layer lifetimes are typically 10-15 years before replacement. This deterioration includes increased surface roughness and increased longitudinal unevenness, leading to increased activation of vehicle damping systems, resulting in energy dissipation and thus increased fuel consumption. Recent studies estimate the potential GHG emission savings from improving the evenness of road surfaces to be between 5-15 % (Louhghalam et al., 2019). For example, road traffic on a 1 km long section of a 3-lane road with 27,000 vehicles per day, of which 10 % are heavy goods vehicles (HGVs), causes approximately 3,000 tCO<sub>2</sub>eq of GHG emissions per year. With a theoretical, rather pessimistic savings potential of 2.5 % by improving longitudinal evenness, 74 tCO<sub>2</sub>eq could be saved per year, which is roughly equivalent to the GHG emissions caused by the rehabilitation of the surface layer. Thus, in this case, the emissions from asphalt production could be offset by reduced fuel consumption already within the first year (Roxon et al., 2019). Even with an 30 % overall share of electric vehicles (cars and HGVs), the corresponding emissions savings potential would still be 59 tCO<sub>2</sub>eq per year (calculations based on Gruber and Hofko, 2023).

#### Mitigation strategies

Avoidance of road infrastructure expansion: Current planning and design standards for road infrastructure drive the need to expand the network and cross-sections of existing roads due to: (a) Future traffic volume growth - mandatory assumptions of a 2–3 % annual traffic growth result in larger, thicker structures, wider cross-sections, and sometimes additional lanes; (b) Design speed requirements - minimum design speed thresholds set for various road classes, increase space and material requirements, as speed limits affect the required curvature radii and lane widths; (c) Capacity thresholds - in periodic checks of existing roads, current capacity overload thresholds prioritize high user service levels over efficiency. Therefore, critically reviewing and adapting all standards and guidelines to minimize future expansions can not only reduce the need for material- and production-based GHG emissions, but also reduce land use and indirect emissions from further traffic attracted to overly capacious infrastructure (Anupriya et al., 2023) (high confidence).

Shifting to low-emission materials in the construction and maintenance of road infrastructure: Incentives in tendering processes can drive reductions in GHG emissions by incorporating best bid criteria that optimize road production and products not only economically and technically, but also ecologically (medium evidence, high agreement). Calculation tools to assess emission reduction potential based on material mix designs and production parameters are already available for the tendering process of the high-level road network operated by ASFINAG (an Austrian public corporation that plans, finances, builds, maintains and collects tolls for the Austrian highways) (ASFINAG, 2024). Dry storage of mineral aggregate, reclaimed asphalt pavement (RAP), and short transport distances have a significant positive impact on the GHG reduction potential of HMA, increasing its importance in achieving minimum overall energy consumption (Hofko et al., 2020) (high confidence). While the addition of RAP to HMA can reduce emissions, excessive RAP content can affect the durability of the road, reducing its service life while increasing the need for rehabilitation and therefore longterm energy consumption. There is currently no consensus on safe RAP limits, highlighting the need for further research to establish reliable thresholds and improve lifecycle analysis.

**Optimization of surface quality:** High surface quality (i.e., longitudinal evenness) improves not only the structural quality (technical lifetime) but also, as discussed above, fleet fuel consumption. Incorporating surface quality models into pavement management systems can therefore further reduce traffic-related GHG emissions (*high confidence*). Continuous assessment of longitudinal evenness is possible with simple means, as recent studies have shown (Gruber and Hofko, 2024).

## Risks and adaptation possibilities

Adaptation to increasing global warming levels is essential for transport infrastructure due to the risks posed by changing weather patterns and their consequences (see also **Cross-Chapter Box 1**), including infrastructure unavailability and reduced network resilience (*high confidence*). Temporary or prolonged unavailability can result in significant economic costs due to delays and necessary adjustments in routes and modes of transport caused by reduced travel time reliability, supply chain disruptions, and other logistical challenges. Urban areas face higher impacts due to denser populations and freight demand, while rural areas – although less affected (with the exception of tourism hotspots; see also Section 7.4) – may suffer severe disruptions due to the lack of alternative routes and modes (no redundancy)

AAR2

(*medium evidence, high agreement*). The complexity of infrastructure operations means that the response of the infrastructure to hazards is rarely linear (*medium evidence, high agreement*). Infrastructure damages can also increase the risk of accidents and potentially lead to loss of property (vehicles, cargo, etc.) if not detected in time, or if warnings are ignored.

In general, proactive adaptation safeguards infrastructure and ensures functionality under increasing global warming levels. Moreover, adaptation of rail and road infrastructure to climate change usually results in net macroeconomic benefits by balancing direct benefits and indirect benefits such as employment, even at rather low damage reduction potentials (Bachner, 2017) (*limited evidence, medium agreement*). So far, Austria's adaptation targets for climate change – as outlined in König et al. (2014) – have not been met in any of the relevant transport and infrastructure dimensions (see Table 48 in Balas et al., 2021) (*high confidence*).

Heavy rain events: Increased likelihood of shorter (high confidence) and longer (low confidence) heavy rainfall events can trigger pluvial and riverine flooding, rock falls, avalanches (rock and snow), and shallow slides, especially in narrow valleys (Huttenlau et al., 2010; Olsson et al., 2012; Löschner et al., 2017; Schlögl and Matulla, 2018; Unterberger et al., 2019). Consequences include infrastructure damage, disruption, and potentially safety issues. Prolonged heavy rain events can increase surface runoff due to increased soil saturation and cause structural failure of unbound base layers, necessitating premature reconstruction (high confidence). In addition, landslides can result from rain-induced deconstruction. Bridge scour and water damage to electronic equipment are other specific consequences associated with heavy rain events. Adapting standard practices with porous surface layers, slightly tilted surfaces (for water drainage), and slope stabilization increases resilience (see also Section 3.2.3 Urban water impacts). Surface elevation in vulnerable areas, such as valleys, creates safe escape routes during floods. Infrastructure improvements such as water retention basins manage excess water. In flood-prone areas, soil monitoring combined with an automatic warning system provides additional safety (see also Sections 1.4.1, 3.2.3 and 7.4.1). Urban water management concepts, such as the 'sponge city' concept, can improve safety during short heavy rain events and retain water from precipitation in soils under sealed surfaces by allowing water to locally percolate through porous surface layers (see also Section 3.2.3).

A recent example of such a heavy rain event occurred in Eastern Austria in September 2024. In addition to an esti-

mated EUR 1.3 billion in damages to businesses and households, immense infrastructure damage occurred, the cost of which cannot yet be determined (as of 2024) (WIFO, 2024). It led to the closure of Austria's main railway line between Vienna and Salzburg ('Weststrecke') for nearly two months due to flooding that damaged tunnels, stations, and electronic systems. Despite being designed in the 1990s according to standards to statistically withstand 100-year flood extremes, the 2024 rainfall exceeded these magnitudes, with water levels equivalent to a 500- or 1000-year event. An example for which infrastructure costs have been quantified are the several heavy rain events in Styria in the summer of 2024, with a total estimated damage of EUR<sub>2023</sub> 34 million (ORF, 2024).

Acute heat: Increased acute heat can cause significant material damage. For road infrastructure, this includes premature failure of asphalt pavements due to excessive softening and permanent deformation (rutting) (Zhang et al., 2022) and failure of concrete pavements due to slab buckling (Kerr, 1994). Acute heat increases the risk of rail buckling, with thermal expansion being particularly problematic in narrow curves such as those often found in mountainous areas. These risks can be partially counteracted by changes in product design and construction techniques. Speed and weight restrictions on roads and railways, and the design of rail infrastructure to avoid narrow curves (e.g., tunnels) are other suitable short- and long-term adaptation measures.

In urban areas, adaptation strategies to reduce UHIs are of utmost importance (see Section 3.2.3 Integrated mitigation and adaptation strategies). Surface adaptations such as lighter-colored pavements and permeable structures are currently being studied to gain experience. Lighter colored surfaces increase reflection and reduce energy dissipation, resulting in faster cooling after sunset. However, during the day, higher reflectance can lead to higher local temperature maxima. Permeable structures contain less mass to act as a heat carrier, and water can be applied and partially stored during extreme temperature events to shift energy conversion from heating to evaporation (Myrup, 1969; Qin et al., 2024) (high confidence). However, changes in road surfaces can only provide partial solutions. Preserving and increasing the amount of natural shade is necessary and provides the most significant improvements to UHIs.

Lack of precipitation: Risks associated with a lack of precipitation primarily affect waterways, as lower water levels reduce freight carrying capacity. However, waterways are only a small part of the Austrian transportation network. In exceptional cases, dry soils can cause shrinkage cracks or even landslides, leading to infrastructure failures. Cost-effective adaptation options for these risks are limited.

#### Regulation, governance, and planning aspects

Beyond the physical adaptation of infrastructure, regulatory and governance frameworks will also need to evolve to meet the demands of building and operating transport infrastructure under changing climate conditions (e.g., see Table 48 in Balas et al., 2021). The acute need for action is underscored by the fact that infrastructure assets built today are expected to remain operational well beyond 2100, a timeframe that is often overlooked in current legislation (see, e.g., Siefer et al., 2019, on railway infrastructure regulations in Germany). Forward-looking scenarios and risk assessment tools, such as maps detailing changes in minimum and maximum surface temperatures and extreme precipitation probabilities, can guide the design of resilient infrastructure to accommodate changing climate patterns (Fasthuber, 2019; Esterl et al., 2022). In particular, the urgent need to re-evaluate flood risk statistics was further illustrated by the extreme flood event in the fall of 2024, which revealed significant gaps in existing predictive models and preparedness measures.

#### Conclusion

Climate change is already impacting Austria's built environment and transport sector, with rising temperatures and extreme weather events threatening infrastructure. To ensure resilience and protect public welfare, adaptation strategies are essential, including improved water management, flood and landslide protection, and greening measures to mitigate heat stress. Soil sealing and excessive land take exacerbate these challenges, highlighting the need for permeable surfaces, sustainable drainage systems, and nature-based solutions. Resilient urban planning must incorporate these elements to ensure long-term sustainability. Meanwhile, Austria's building and transport sectors remain major contributors to climate change, accounting for 70 % of national energy consumption and 38 % of greenhouse gas emissions in 2023, excluding emissions from electricity and district heat generation, which are attributed to the industrial sector. Electrification, driven by heat pumps and electric vehicles, is key to decarbonization, but its full potential depends on defossilizing and decarbonizing both electricity generation and district heating. Maximizing the benefits of the transition requires deep renovations of buildings, widespread adoption of heat pumps, and the use of low-emitting materials alongside circular design principles to minimize environmental impacts and resource consumption.

Compact urban development and reduced car dependency are also critical to reducing emissions, but urban heat island effects and strong preferences for single-family homes and car ownership pose challenges. Mitigation options include urban greening, prioritizing mixed-use neighborhoods, and improving public transport and active mobility infrastructure. Freight transportation is another hurdle, with rising demand and limited low-carbon alternatives delaying full decarbonization. Solutions include optimizing logistics and shifting long-distance transport to rail, while improving cross-border rail connectivity in line with EU transport policy. Dynamic road pricing and stronger incentives targeting the advancement of sustainable transport technologies, including low- and zero-emission trucks, could further accelerate emissions reductions.

Achieving a low-carbon future will require a combination of pull and push measures, including incentives for behavioral change, renewable energy adoption, and efficiency improvements. With a 43 % reduction in GHG emissions from 1990 to 2022, the building sector is already making progress. However, continued emphasis on emission avoidance, renewable solutions, and efficiency gains will remain critical to meeting climate goals. Through integrated planning, technological innovation, and behavioral shifts, Austria can foster more resilient, sustainable, and livable communities.

#### REFERENCES

- ACEA (2023a). Average age of the EU vehicle fleet, by country. ACEA – Eur. Automob. Manuf. Assoc. Available at: <u>https://www.acea.auto/figure/average-age-of-eu-vehicle-fleet-by-country/</u> (Accessed September 20, 2023).
- ACEA (2023b). Vehicles in use in Europe 2023. ACEA European Automobile Manufacturers' Association. Available at: <u>https://</u> <u>www.acea.auto/files/ACEA-report-vehicles-in-use-europe-</u> <u>2023.pdf</u>
- Ajanovic, A., Siebenhofer, M., and Haas, R. (2021). Electric Mobility in Cities: The Case of Vienna. *Energies* 14, 217. <u>https://doi. org/10.3390/en14010217</u>
- Akhimien, N. G., Latif, E., and Hou, S. S. (2021). Application of circular economy principles in buildings: A systematic review. J. Build. Eng. 38, 102041. <u>https://doi.org/10.1016/j.jobe.2020. 102041</u>
- Alaux, N., Schwark, B., Hörmann, M., Ruschi Mendes Saade, M., and Passer, A. (2024). Assessing the prospective environmental impacts and circularity potentials of building stocks: An open-source model from Austria (PULSE-AT). *J. Ind. Ecol.* 28, 1435–1448. <u>https://doi.org/10.1111/jiec.13558</u>
- Allard, S., Mima, S., Debusschere, V., Quoc, T. T., Criqui, P., and Hadjsaid, N. (2020). European transmission grid expansion as a flexibility option in a scenario of large scale variable renewable energies integration. *Energy Econ.* 87, 104733. <u>https://doi. org/10.1016/j.eneco.2020.104733</u>
- Amann, W. (2019). "Strukturwandel in der österreichischen Wohnbauförderung", in Wohnungsgemeinnützigkeit in Recht
   – Wirtschaft – Gesellschaft. Festschrift für Karl Wurm, ed.GBV (Wien: LexisNexis). Available at: <u>https://iibw.at/en/for-</u> schungs-datenbank/233-2019-20 (Accessed February 8, 2025).
- Amann, W. (2023). "Klimaschutz und Föderalismus die Potenziale der Wohnbauförderung", in *Klimaschutz und Föderalismus*. Band 139 der Schriftenreihe des Instituts für Föderalismus, eds. P. Bußjäger and M. Eller (Innsbruck: Studien-Verlag). Available at: <u>https://www.iibw.at/de/forschungs-datenbank/197-2023-8</u> (Accessed February 8, 2025).
- Amann, W., Mundt, A., and Oberhuber, A. (2023a). Berichtstandard Wohnbauförderung 2023. Wien: IIBW, im Auftrag des Landes Wien. Available at: <u>https://iibw.at/de/forschungs-datenbank/415-berichtstandard-wohnbaufoerderung-2023</u> (Accessed February 8, 2025).
- Amann, W., Mundt, A., Reiterer, D., Schieder, W., and Storch, A. (2021). Monitoring-System zu Sanierungsmaßnahmen in Österreich. Wien: IIBW & Umweltbundesamt GmbH. Available at: <u>http://iibw.at/documents/2021%20IIBW\_Umweltbundesamt.</u> <u>%20Sanierungsrate.pdf</u> (Accessed September 26, 2023).
- Amann, W., Preßmair, G., Stipsits, A., Schoisengeier, S., Amann, C., Preisler, A., et al. (2023b). Bewertung der Bauteilaktivierung als Option für Flexibilität im Strommarkt. Wien: BMK. Available at: <u>https://nachhaltigwirtschaften.at/resources/sdz\_pdf/</u> <u>schriftenreihe-2023-50-bauteilaktivierung-netzdienlichkeit.pdf</u> (Accessed February 8, 2025).
- Angelini, A., Heinfellner, H., Pfaffenbichler, P., and Schwingshackl, M. (2022). Transition Mobility 2040. Entwicklung eines Klimaund Energieszenarios zur Abbildung von Klimaneutralität im Verkehr 2040. Vienna: Umweltbundesamt GmbH. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/publikationen/rep0808.pdf</u> (Accessed September 12, 2022).

- Anupriya, Bansal, P., and Graham, D. J. (2023). Congestion in cities: Can road capacity expansions provide a solution? *Transp. Res. Part Policy Pract.* 174, 103726. <u>https://doi.org/10.1016/j.tra.</u> 2023.103726
- Arnbjerg-Nielsen, K., Willems, P., Olsson, J., Beecham, S., Pathirana, A., Bülow Gregersen, I., et al. (2013). Impacts of climate change on rainfall extremes and urban drainage systems: a review. *Water Sci. Technol.* 68, 16–28. <u>https://doi.org/10.2166/ wst.2013.251</u>
- Aschauer, F., Rösel, I., Hössinger, R., Kreis, H. B., and Gerike, R. (2019). Time use, mobility and expenditure: an innovative survey design for understanding individuals' trade-off processes. *Transportation* 46, 307–339. <u>https://doi.org/10.1007/s11116-018-9961-9</u>
- ASFINAG (2024). ASFINAG. Available at: <u>https://www.asfinag.at/</u> (Accessed November 29, 2024).
- Austrian Standards (2019). ÖNORM B 8110-5: 2019 03 15 Wärmeschutz im Hochbau – Teil 5: Klimamodell und Nutzungsprofile. Available at: <u>https://shop.austrian-standards.at/action/ de/public/details/652632/OENORM\_B\_8110-5\_2019\_03\_15</u> (Accessed September 25, 2023).
- Austrian Standards (2022). ÖNORM B 1991-1-3: 2022 05 15 Eurocode 1 Einwirkungen auf Tragwerke Teil 1–3: Allgemeine Einwirkungen Schneelasten Nationale Festlegungen zur ÖNORM EN 1991-1-3, nationale Erläuterungen und nationale Ergänzungen. Available at: <u>https://shop.austrian-standards.at/action/de/public/details/1229841/OENORM\_B\_1991-1-3\_2022\_05\_15</u> (Accessed September 25, 2023).
- AustriaTech (2024). Elektromobilität in Österreich Zahlen, Daten & Fakten | August 2024. Available at: <u>https://www.austriatech.</u> <u>at/assets/Uploads/Publikationen/OLE\_ZDF\_2024\_August\_DE.pdf</u> (Accessed February 1, 2025).
- Axhausen, K. W., Molloy, J., Tchervenkov, C., Becker, F., Hintermann, B., Schoeman, B., et al. (2021). Empirical analysis of mobility behavior in the presence of Pigovian transport pricing. ETH Zurich. <u>https://doi.org/10.3929/ETHZ-B-000500100</u>
- Bachner, G. (2017). Assessing the economy-wide effects of climate change adaptation options of land transport systems in Austria. *Reg. Environ. Change* 17, 929–940. <u>https://doi.org/10.1007/s10113-016-1089-x</u>
- Balas, M., Felderer, A., Völler, S., Zeitz, F., and Margelik, E. (2021). Zweiter Fortschrittsbericht zur österreichischen Strategie zur Anpassung an den Klimawandel. Wien: BMK. Available at: <u>https://www.bmk.gv.at/dam/jcr:4a7614de-cbbc-47b4-bd01-3ac3d079c509/klimawandel-fortschrittsbericht-2021.pdf</u> (Accessed October 2, 2023).
- Banar, M., and Özdemir, A. (2015). An evaluation of railway passenger transport in Turkey using life cycle assessment and life cycle cost methods. *Transp. Res. Part Transp. Environ.* 41, 88– 105. <u>https://doi.org/10.1016/j.trd.2015.09.017</u>
- Bao, H., and Ma, Z. (2022). "Thermochemical energy storage", in *Storing Energy*, (Elsevier), 651–683. <u>https://doi.org/10.1016/</u> B978-0-12-824510-1.00028-3
- Bechtel, S., Rafii-Tabrizi, S., Scholzen, F., Hadji-Minaglou, J.-R., and Maas, S. (2020). Influence of thermal energy storage and heat pump parametrization for demand-side-management in a nearly-zero-energy-building using model predictive control. *Energy Build*. 226, 110364. <u>https://doi.org/10.1016/j.enbuild</u>. 2020.110364
- Benajes, J., García, A., Monsalve-Serrano, J., and Guzmán-Mendoza, M. (2024). A review on low carbon fuels for road vehicles:

The good, the bad and the energy potential for the transport sector. *Fuel* 361, 130647. <u>https://doi.org/10.1016/j.fuel.2023.130647</u>

- BGBl. I Nr. 6/2020 (2020). Ölkesseleinbauverbotsgesetz (ÖKEVG). Available at: <u>https://www.ris.bka.gv.at/Ergebnis.wxe?Suchwor-te=20010894&x=0&y=0&Abfrage=Gesamtabfrage</u> (Accessed May 10, 2024).
- BGBl. I Nr. 8/2024 (2024). Bundesgesetz über die erneuerbare Wärmebereitstellung in neuen Baulichkeiten (Erneuerbare-Wärme-Gesetz – EWG). Available at: <u>https://www.ris.bka.</u> <u>gv.at/eli/bgbl/I/2024/8</u> (Accessed May 10, 2024).
- BGBI. I Nr. 27/2012 (2012). Energieausweis-Vorlage-Gesetz 2012 Bundesrecht konsolidiert, Fassung vom 10.05.2024. Available at: <u>https://www.ris.bka.gv.at/GeltendeFassung.wxe?Abfrage=Bundesnormen&Gesetzesnummer=20007799</u> (Accessed May 10, 2024).
- BGBI. I Nr. 70/2002 (2002). Bundesgesetz über das Wohnungseigentum (Wohnungseigentumsgesetz 2002 – WEG 2002). Fassung vom 09.08.2022. Available at: <u>https://www.ris.bka.gv.at/</u> <u>eli/bgbl/i/2002/70/P0/NOR30002088</u> (Accessed June 8, 2024).
- BGBI. I Nr. 152/2023 (2023). Bundesgesetz über die Förderung von Maßnahmen in den Bereichen der Wasserwirtschaft, der Umwelt, der Altlastensanierung, des Flächenrecyclings, der Biodiversität und der Kreislaufwirtschaft und zum Schutz der Umwelt im Ausland sowie über das österreichische JI/CDM-Programm für den Klimaschutz (Umweltförderungsgesetz – UFG). Fassung vom 21.11.2024. Available at: <u>https://ris.bka.gv.at/GeltendeFassung.wxe?Abfrage=Bundesnormen&Gesetzesnummer=10010755</u> (Accessed November 21, 2024).
- BGBl. Nr. 1/1930 (2024). Bundes-Verfassungsgesetz (B-VG). Fassung vom 30.04.2024. Available at: <u>https://www.ris.bka.gv.at/GeltendeFassung.wxe?Abfrage=Bundesnormen&Gesetz-esnummer=10000138</u>
- BGBl. Nr. 139/1979 (1979). Bundesgesetz vom 8. März 1979 über die Gemeinnützigkeit im Wohnungswesen (Wohnungsgemeinnützigkeitsgesetz – WGG). Available at: <u>https://www.ris.bka.</u> gv.at/GeltendeFassung.wxe?Abfrage=Bundesnormen&Gesetzesnummer=10011509 (Accessed June 8, 2024).
- BGBl. Nr. 520/1981 (1981). Bundesgesetz vom 12. November 1981 über das Mietrecht (Mietrechtsgesetz – MRG). Available at: <u>https://www.ris.bka.gv.at/GeltendeFassung.wxe?Abfrage=Bundesnormen&Gesetzesnummer=10002531</u> (Accessed June 8, 2024).
- BGBl. Nr. 827/1992 (1992). Bundesgesetz über die sparsamere Nutzung von Energie durch verbrauchsabhängige Abrechnung der Heiz-, Warmwasser- und Kältekosten (Heiz- und Kältekostenabrechnungsgesetz – HeizKG) BGBl. Nr. 827/1992. Available at: <u>https://www.ris.bka.gv.at/GeltendeFassung.wxe?Abfrage=Bundesnormen&Gesetzesnummer=10007277</u> (Accessed May 10, 2024).
- Bibak, B., and Tekiner-Mogulkoc, H. (2022). The parametric analysis of the electric vehicles and vehicle to grid system's role in flattening the power demand. *Sustain. Energy Grids Netw.* 30, 100605. <u>https://doi.org/10.1016/J.SEGAN.2022.100605</u>
- Bieliński, T., Kwapisz, A., and Ważna, A. (2021). Electric bike-sharing services mode substitution for driving, public transit, and cycling. *Transp. Res. Part Transp. Environ.* 96, 102883. <u>https:// doi.org/10.1016/j.trd.2021.102883</u>
- Billerbeck, A., Kiefer, C. P., Winkler, J., Bernath, C., Sensfuß, F., Kranzl, L., et al. (2024). The race between hydrogen and heat pumps for space and water heating: A model-based scenar-

AAR2

io analysis. Energy Convers. Manag. 299, 117850. https://doi. org/10.1016/j.enconman.2023.117850

- BKA (2020). Aus Verantwortung für Österreich. Regierungsprogramm 2020 – 2024. Vienna: Bundeskanzleramt Österreich. Available at: <u>https://www.dievolkspartei.at/Download/Regierungsprogramm\_2020.pdf</u> (Accessed September 12, 2022).
- Blab, R., Hoffmann, M., Langer, M., Marchtrenker, S., Niescher, P., Peyerl, M., et al. (2012). *Betonstraßen – das Handbuch: Leitfaden für die Praxis*. Wien: Zement und Beton, Handels- und Werbe GmbH.
- BMDV (2022). Kompensation zukünftiger Einnahmeausfälle des Staates aufgrund der Antriebswende im Straßenverkehr. Berlin: Bundesministerium für Digitales und Verkehr. Available at: <u>https://bmdv.bund.de/SharedDocs/DE/Anlage/G/kompensation-zukuenftiger-einnahmeausfaelle.html</u> (Accessed November 7, 2024).
- BMK (2016). Österreich unterwegs 2013/2014. Ergebnisbericht zur österreichweiten Mobilitätserhebung "Österreich unterwegs 2013/2014." Vienna: Bundesministerium für Verkehr, Innovation und Technologie. Available at: <u>https://www.bmk.gv.at/</u> <u>themen/verkehrsplanung/statistik/oesterreich\_unterwegs/be-</u> richte.html (Accessed November 7, 2024).
- BMK (2021a). Aktionsplan & Kernkriterien. für die Beschaffung nachhaltiger Produkte und Leistungen. Wien: Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie. Available at: <u>https://www.nabe.gv.at/nabe-aktionsplan/</u> (Accessed May 10, 2024).
- BMK (2021b). Austria's 2030 Mobility Master Plan. The new climate action framework for the transport sector: sustainable – resilient – digital. Vienna: Federal Ministry Republic of Austria for Climate Action, Environment, Energy, Mobility, Innovation and Technology. Available at: <u>https://www.bmk.gv.at/dam/ jcr:eaf9808b-b7f9-43d0-9faf-df28c202ce31/BMK\_Mobilitaetsmasterplan2030\_EN\_UA.pdf</u> (Accessed September 6, 2023).
- BMK (2022a). Austrian Heat Map. Available at: <u>https://austri-an-heatmap.gv.at/</u> (Accessed February 5, 2025).
- BMK (2022b). Faktencheck E-Mobilität. Bundesminist. Klimaschutz Umw. Energ. Mobil. Innov. Technol. Available at: <u>https://www. bmk.gv.at/themen/mobilitaet/alternative\_verkehrskonzepte/</u><u>elektromobilitaet/zahlen/faktencheck.html</u> (Accessed May 10, 2024).
- BMK (2022c). Treibhausgasreduktions-Maßnahmen im Gebäudesektor Österreichs 2009–2021. Bundesminist. Klimaschutz Umw. Energ. Mobil. Innov. Technol. Available at: <u>https://www. bmk.gv.at/themen/energie/publikationen/treibhausgasreduktion-wohnbau.html</u> (Accessed September 20, 2023).
- BMK (2023a). Erarbeitungsprozess Zielnetz 2040. Available at: <u>https://www.bmk.gv.at/themen/verkehrsplanung/ausbauplan/</u> <u>zielnetz.html</u> (Accessed October 9, 2023).
- BMK (2023b). Mehr als eine Million E-Bikes in Österreich. klimaaktiv. Available at: <u>https://www.klimaaktivmobil.at/en/</u> fachpersonen/mobilitaet/studien/verkaufszahlen-2022-mehr-<u>als-eine-million-e-bikes-in-oesterreich</u> (Accessed February 5, 2025).
- BMK (2023c). Zahlen, Daten, Fakten. Bundesminist. Klimaschutz Umw. Energ. Mobil. Innov. Technol. Available at: <u>https://www. bmk.gv.at/themen/mobilitaet/1-2-3-ticket/fakten.html</u> (Accessed December 19, 2024).
- BMK (2024a). Energie in Österreich. Zahlen, Daten, Fakten. Vienna: Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie (BMK). Available at:

https://www.bmk.gv.at/dam/jcr:cf217356-e1a4-4fbc-9834-828bafccbae2/BMK\_Energie\_in\_Oe\_2024.pdf

- BMK (2024b). Fachentwurf Zielnetz 2040. Bundesminist. Klimaschutz Umw. Energ. Mobil. Innov. Technol. Available at: <u>https:// www.bmk.gv.at/themen/verkehrsplanung/ausbauplan/zielnetz.</u> <u>html</u> (Accessed May 10, 2024).
- BMK (2024c). Kesseltausch. Available at: <u>https://kesseltausch.at/</u> (Accessed May 10, 2024).
- BML (2023). HORA Natural Hazard Overview & Risk Assessment Austria. Available at: <u>https://hora.gv.at/</u> (Accessed September 25, 2023).
- BMVIT (2017). National Implementation Plan for ERTMS in Austria. Wien: Bundesministerium für Verkehr, Innovation und Technologie. Available at: <u>https://www.bmk.gv.at/en/topics/</u> <u>transport/railways/interop.html</u> (Accessed May 10, 2024).
- Borck, R., and Brueckner, J. K. (2018). Optimal Energy Taxation in Cities. J. Assoc. Environ. Resour. Econ. 5, 481–516. <u>https://doi.org/10.1086/695614</u>
- Borri, E., Zsembinszki, G., and Cabeza, L. F. (2021). Recent developments of thermal energy storage applications in the built environment: A bibliometric analysis and systematic review. *Appl. Therm. Eng.* 189, 116666. <u>https://doi.org/10.1016/j.applthermaleng.2021.116666</u>
- Brandes, J., Haun, M., Wrede, D., Jürgens, P., Kost, C., and Henning, H.-M. (2021). Wege zu einem klimaneutralen Energiesystem – Die deutsche Energiewende im Kontext gesellschaftlicher Verhaltensweisen – Update November 2021: Klimaneutralität 2045. Freiburg: Fraunhofer-Institut für Solare Energiesysteme ISE. Available at: <u>https://www.ise.fraunhofer.de/de/veroeffentlichungen/studien/wege-zu-einem-klimaneutralen-energiesystem-version-2020-21.html</u> (Accessed September 25, 2023).
- Bronnenmayer, H. (2024). Jetzt in Zahlen Radfahren liegt auch 2023 im Trend. Wien: Mobilitätsagentur Wien GmbH. Available at: <u>https://www.fahrradwien.at/2024/03/25/jetzt-in-zahlenradfahren-liegt-auch-2023-im-trend/</u> (Accessed November 7, 2024).
- Büchele, R. (2019). Integrated strategic heating and cooling planning on different spatial levels and political framework conditions. Technische Universität Wien. <u>https://doi.org/10.34726/</u> HSS.2019.69286
- Büchele, R., Kranzl, L., and Hummel, M. (2019). Integrated strategic heating and cooling planning on regional level for the case of Brasov. *Energy* 171, 475–484. <u>https://doi.org/10.1016/j.energy.2019.01.030</u>
- Camarasa, C., Mata, É., Navarro, J. P. J., Reyna, J., Bezerra, P., Angelkorte, G. B., et al. (2022). A global comparison of building decarbonization scenarios by 2050 towards 1.5–2 °C targets. *Nat. Commun.* 13, 3077. <u>https://doi.org/10.1038/s41467-022-29890-5</u>
- Caprotti, F., Duarte, C., and Joss, S. (2024). The 15-minute city as paranoid urbanism: Ten critical reflections. *Cities* 155, 105497. <u>https://doi.org/10.1016/j.cities.2024.105497</u>
- Chang, B., and Kendall, A. (2011). Life cycle greenhouse gas assessment of infrastructure construction for California's high-speed rail system. *Transp. Res. Part Transp. Environ.* 16, 429–434. https://doi.org/10.1016/j.trd.2011.04.004
- Cheng, S., Lin, J., Xu, W., Yang, D., Liu, J., and Li, H. (2020). Carbon, water, land and material footprints of China's high-speed railway construction. *Transp. Res. Part Transp. Environ.* 82, 102314. <u>https://doi.org/10.1016/j.trd.2020.102314</u>

- Chester, M., and Horvath, A. (2008). Environmental Life-cycle Assessment of Passenger Transportation: A Detailed Methodology for Energy, Greenhouse Gas and Criteria Pollutant Inventories of Automobiles, Buses, Light Rail, Heavy Rail and Air v.2. University of California, Berkeley, Institute of Transportation Studies. Available at: <u>https://escholarship.org/uc/item/5670921q</u> (Accessed September 11, 2022).
- Chittum, A., and Østergaard, P. A. (2014). How Danish communal heat planning empowers municipalities and benefits individual consumers. *Energy Policy* 74, 465–474. <u>https://doi.org/10.1016/j.</u> enpol.2014.08.001
- Colovic, A., Marinelli, M., and Ottomanelli, M. (2024). Towards the electrification of freight transport: A network design model for assessing the adoption of eHighways. *Transp. Policy* 150, 106–120. https://doi.org/10.1016/j.tranpol.2024.03.012
- COM/2019/640 final (2019). COMMUNICATION FROM THE COMMISSION TO THE EUROPEAN PARLIAMENT, THE EUROPEAN COUNCIL, THE COUNCIL, THE EUROPE-AN ECONOMIC AND SOCIAL COMMITTEE AND THE COMMITTEE OF THE REGIONS The European Green Deal. Available at: <u>https://eur-lex.europa.eu/legal-content/EN/TXT/</u> <u>?uri=COM%3A2019%3A640%3AFIN</u> (Accessed September 14, 2022).
- COM/2021/802 final (2021). Proposal for a DIRECTIVE OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL on the energy performance of buildings (recast). Available at: <u>https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=celex:</u> <u>52021PC0802</u> (Accessed September 26, 2023).
- COM/2023/314 final (2023). Proposal for a REGULATION OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL on the transparency and integrity of Environmental, Social and Governance (ESG) rating activities. Available at: <u>https://eur-lex. europa.eu/legal-content/EN/TXT/?uri=CELEX%3A52023P-C0314&qid=1732204534132</u> (Accessed November 21, 2024).
- Creutzig, F. (2014). How fuel prices determine public transport infrastructure, modal shares and urban form. *Urban Clim.* 10, 63–76. https://doi.org/10.1016/j.uclim.2014.09.003
- Creutzig, F., Baiocchi, G., Bierkandt, R., Pichler, P.-P., and Seto, K. C. (2015). Global typology of urban energy use and potentials for an urbanization mitigation wedge. *Proc. Natl. Acad. Sci.* 112, 6283–6288. <u>https://doi.org/10.1073/pnas.1315545112</u>
- Creutzig, F., Roy, J., Devine-Wright, P., Díaz-José, J., Geels, F. W., Grubler, A., et al. (2022). "Demand, services and social aspects of mitigation Supplementary Material", in *IPCC*, 2022: Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, eds. P. R. Shukla, J. Skea, R. Slade, A. A. Khourdajie, R. van Diemen, D. McCollum, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press). https://doi.org/10.1017/9781009157926.007
- Dallhammer, E., Gaupp-Berghausen, M., Messinger, I., Schremmer, C., and Mollay, U. (2022). Analyse der Flächeninanspruchnahme bezüglich Raumtypen und Wirkfaktoren in der Ostregion. Wien: ÖIR GmbH.
- De Blas, I., Mediavilla, M., Capellán-Pérez, I., and Duce, C. (2020). The limits of transport decarbonization under the current growth paradigm. *Energy Strategy Rev.* 32, 100543. <u>https://doi. org/10.1016/j.esr.2020.100543</u>
- de Bortoli, A., Bouhaya, L., and Feraille, A. (2020). A life cycle model for high-speed rail infrastructure: environmental inventories and assessment of the Tours-Bordeaux railway in France.

Int. J. Life Cycle Assess. 25, 814–830. <u>https://doi.org/10.1007/s11367-019-01727-2</u>

- Decoville, A., and Feltgen, V. (2023). Clarifying the EU objective of no net land take: A necessity to avoid the cure being worse than the disease. *Land Use Policy* 131, 106722. <u>https://doi.org/ 10.1016/j.landusepol.2023.106722</u>
- Desai, A., Kanika, and Patel, C. R. (2023). The impact of electric vehicle charging infrastructure on the energy demand of a city. *Energy Rep.* 9, 814–823. <u>https://doi.org/10.1016/j.egyr.2023.05.177</u>
- Dieleman, F., and Wegener, M. (2004). Compact City and Urban Sprawl. Built Environ. 30, 308–323. <u>https://doi.org/10.2148/</u> <u>benv.30.4.308.57151</u>
- Ding, Y., and Riffat, S. B. (2013). Thermochemical energy storage technologies for building applications: a state-of-the-art review. *Int. J. Low-Carbon Technol.* 8, 106–116. <u>https://doi.org/10.1093/</u> ijlct/cts004
- Directive 2014/94/EU (2014). Directive 2014/94/EU of the European Parliament and of the Council of 22 October 2014 on the deployment of alternative fuels infrastructure Text with EEA relevance. Available at: <u>https://eur-lex.europa.eu/eli/dir/2014/94/oj</u> (Accessed June 8, 2024).
- Directive (EU) 2018/2001 (2018). Directive (EU) 2018/2001 of the European Parliament and of the Council of 11 December 2018 on the promotion of the use of energy from renewable sources (recast) (Text with EEA relevance.). Available at: <u>http://data.europa.eu/eli/dir/2018/2001/oj/eng</u> (Accessed September 26, 2023).
- Directive (EU) 2018/2002 (2018). Directive (EU) 2018/2002 of the European Parliament and of the Council of 11 December 2018 amending Directive 2012/27/EU on energy efficiency (Text with EEA relevance.). Available at: <u>http://data.europa.eu/eli/dir/2018/2002/oj/eng</u> (Accessed September 26, 2023).
- Directive (EU) 2019/1161 (2019). Directive (EU) 2019/1161 of the European Parliament and of the Council of 20 June 2019 amending Directive 2009/33/EC on the promotion of clean and energy-efficient road transport vehicles (Text with EEA relevance.). Available at: <u>https://eur-lex.europa.eu/eli/dir/2019/1161/oj</u> (Accessed June 8, 2024).
- Directive (EU) 2023/1791 (2023). Directive (EU) 2023/1791 of the European Parliament and of the Council of 13 September 2023 on energy efficiency and amending Regulation (EU) 2023/955 (recast). Available at: <u>http://data.europa.eu/eli/dir/2023/1791/oj/eng</u> (Accessed May 10, 2024).
- Directive (EU) 2023/2413 (2023). Directive (EU) 2023/2413 of the European Parliament and of the Council of 18 October 2023 amending Directive (EU) 2018/2001, Regulation (EU) 2018/1999 and Directive 98/70/EC as regards the promotion of energy from renewable sources, and repealing Council Directive (EU) 2015/652. Available at: <u>http://data.europa.eu/eli/ dir/2023/2413/oj/eng</u> (Accessed May 10, 2024).
- Dobler, C., Pfeifer, D., and Streicher, W. (2018). Reaching energy autonomy in a medium-sized city – three scenarios to model possible future energy developments in the residential building sector. *Sustain. Dev.* 26, 859–869. <u>https://doi.org/10.1002/ sd.1855</u>
- Dugan, A., Mayer, J., Thaller, A., Bachner, G., and Steininger, K. W. (2022). Developing policy packages for low-carbon passenger transport: A mixed methods analysis of trade-offs and synergies. *Ecol. Econ.* 193, 107304. <u>https://doi.org/10.1016/j.ecolecon.2021.107304</u>

- Dumke, H., Fischer, T., Stöglehner, G., and Getzner, M. (2024). "Kapitel 7. Raumplanung und Klimawandel", in APCC Special Report: Landnutzung und Klimawandel in Österreich, eds. R. Jandl, U. Tappeiner, C. B. Foldal, and K. Erb (Berlin, Heidelberg: Springer Spektrum), 381–405. <u>https://doi.org/10.1007/978-3-662-67864-0\_9</u>
- EAFO (2023). European Alternative Fuels Observatory. Country comparison. European Union (EU27). *Eur. Comm.* Available at: <u>https://alternative-fuels-observatory.ec.europa.eu/transport-mode/road/european-union-eu27/country-comparison</u> (Accessed September 19, 2023).
- EAPA (2023). Asphalt in Figures key figures of the European asphalt industry. *EAPA*. Available at: <u>https://eapa.org/asphalt-in-figures/</u> (Accessed September 20, 2023).
- Ebenbichler, R., Hert, A., Streicher, W., Fischer, D., Richtfeld, C., Mailer, M., et al. (2018). Szenarien Tirol 2050. Ressourcen- und Technologieeinsatz-Szenarien Tirol 2050. Available at: <u>https:// www.lebensraum.tirol/wp-content/uploads/2021/08/05\_Ressourcen-und-Technologieeinsatz-Szenarien-Tirol-2050.pdf</u> (Accessed September 25, 2023).
- Ebenbichler, R., Hertl, A., Duncker, T., Hofmann, A., Streicher, W., Mailer, M., et al. (2024). Energiespeicher Tirol 2050. Innsbruck: Energieagentur Tirol. Available at: <u>https://www.energieagentur. tirol/uploads/tx\_bh/3286/energiespeicher-tirol-2050-kurzfassung.pdf</u> (Accessed February 1, 2025).
- E-control (2023). Operating statistics, 2023. Available at: <u>https://</u> www.e-control.at/en/statistik/e-statistik/archiv/betriebsstatistik
- Elnagar, E., Gendebien, S., Georges, E., Berardi, U., Doutreloup, S., and Lemort, V. (2023). Framework to assess climate change impact on heating and cooling energy demands in building stock: A case study of Belgium in 2050 and 2100. *Energy Build*. 298, 113547. <u>https://doi.org/10.1016/j.enbuild.2023.113547</u>
- ERA (2020). Report. Fostering the railway sector through the European Green Deal. ERA. Available at: <u>https://www.era.europa.eu/content/report-fostering-railway-sector-through-europe-an-green-deal\_en</u> (Accessed May 10, 2024).
- Erker, S., Kinsperger, A., Hemis, H., and Vogl, B. (2021). "Energieraumpläne – ein Meilenstein am Weg zur nachhaltigen Energiezukunft Wiens", in *Energieraumplanung – ein zentraler Faktor zum Gelingen der Energiewende*, eds. R. Giffinger, M. Berger, K. Weninger, and S. Zech, 28–37. <u>https://doi.org/10.34726/1022</u>
- Esterl, T., Zegers, A., Spreitzhofer, J., Totschnig, G., Knöttner, S., Strömer, S., et al. (2022). Flexibility in Supply and Demand in the Austrian Electricity System. Wien: Energie-Control. Available at: <u>https://www.e-control.at/documents/1785851/ 1811597/20220207\_Flexibility\_study\_AIT\_EN\_FINAL.pdf/ c4e66653-6e47-12e2-ba04-d80922e69567?t=1653295626322</u> (Accessed November 10, 2024).
- ETH Zürich (2023). D-BAUG Lighthouse Project: E-Bike City. Available at: <u>https://ebikecity.baug.ethz.ch/en/</u> (Accessed September 20, 2023).
- EU (2024). ETS2: buildings, road transport and additional sectors. *Eur. Comm.* Available at: <u>https://climate.ec.europa.eu/</u> <u>eu-action/eu-emissions-trading-system-eu-ets/ets2-buildings-road-transport-and-additional-sectors\_en</u> (Accessed November 29, 2024).
- European Commission (2016). Fourth railway package of 2016. Available at: <u>https://transport.ec.europa.eu/transport-modes/</u> <u>rail/railway-packages/fourth-railway-package-2016\_en</u> (Accessed January 15, 2025).

- European Commission (2020a). Construction Products Regulation acquis. Available at: <u>https://single-market-economy.ec.eu-</u> <u>ropa.eu/sectors/construction/construction-products-regula-</u> <u>tion-cpr/acquis\_en</u> (Accessed September 25, 2023).
- European Commission (2020b). Progress report on implementation of the TEN-T network in 2016–2017. Brussels: European Commission. Available at: <u>https://transport.ec.europa.eu/newsevents/news/progress-ten-t-network-2020-08-26\_en</u> (Accessed September 26, 2023).
- European Commission (2022a). Proposal for a revised Urban Wastewater Treatment Directive. Brussels: Directorate-General for Environment. Available at: <u>https://environment.ec.europa.</u> <u>eu/publications/proposal-revised-urban-wastewater-treat-</u> <u>ment-directive\_en</u> (Accessed September 22, 2023).
- European Commission (2022b). REPowerEU. European Commission. Available at: <u>https://commission.europa.eu/publications/</u> <u>key-documents-repowereu\_en</u> (Accessed May 10, 2024).
- European Commission (2022c). Soil strategy for 2030. Available at: <u>https://environment.ec.europa.eu/topics/soil-and-land/</u> <u>soil-strategy\_en</u> (Accessed February 1, 2025).
- European Commission (2024a). Alternative Fuels Infrastructure European Commission. Available at: <u>https://transport.ec.europa.eu/transport-themes/clean-transport/alternative-fuels-sustainable-mobility-europe/alternative-fuels-infrastructure\_en</u> (Accessed May 10, 2024).
- European Commission (2024b). Energy Performance of Buildings Directive – European Commission. Available at: <u>https://energy.ec.europa.eu/topics/energy-efficiency/energy-efficient-build-ings/energy-performance-buildings-directive\_en</u> (Accessed June 10, 2024).
- European Commission: Directorate-General for Energy, Kranzl, L., Fallahnejad, M., Büchele, R., Müller, A., Hummel, M., et al. (2022). Renewable space heating under the revised Renewable Energy Directive – ENER/C1/2018 494 – final report. LU: Publications Office of the European Union. Available at: <u>https:// data.europa.eu/doi/10.2833/525486</u> (Accessed September 22, 2023).
- European Commission: Directorate-General for Mobility and Transport (2023). EU transport in figures: Statistical pocketbook 2023 – Directorate General for Mobility and Transport. LU: Publications Office of the European Union. Available at: <u>https:// data.europa.eu/doi/10.2832/319371</u> (Accessed November 7, 2024).
- European Commission: Directorate-General for Mobility and Transport, Schroten, A., Wijngaarden, L., Schmidt, M., and Morgan-Price, S. (2019). State of play of internalisation in the European transport sector. LU: Publications Office. Available at: <u>https://data.europa.eu/doi/10.2832/537526</u> (Accessed September 19, 2023).
- European Commission: Eurostat (2023). Number of cars per inhabitant increased in 2021. Available at: <u>https://ec.europa.eu/</u> <u>eurostat/web/products-eurostat-news/w/ddn-20230530-1</u> (Accessed May 10, 2024).
- European Commission: Eurostat (2024). Share of renewables in transport increased slightly in 2022. Available at: <u>https:// ec.europa.eu/eurostat/web/products-eurostat-news/w/ddn-20240205-1</u> (Accessed May 10, 2024).
- European Commission: Eurostat, Jere, N., Sušilović, D., Ford-Alexandraki, E., and Xenellis, G. (2023). Key figures on European transport. Luxemburg: Publications Office of the European Union. Available at: <u>https://data.europa.eu/doi/10.2785/4866</u>

- European Environment Agency (2020). Imperviousness Density 2018 (raster 100 m), Europe, 3-yearly, Aug. 2020. <u>https://doi.org/10.2909/524FA72F-61D7-4364-801E-3E271D7B10BC</u>
- European Environment Agency (2021a). Improving the climate impact of raw material sourcing. Copenhagen: European Environment Agency. Available at: <u>https://www.eea.europa.eu/</u> <u>publications/improving-the-climate-impact-of</u> (Accessed May 10, 2024).
- European Environment Agency (2021b). Transport and environment report 2020 Train or plane? *Eur. Environ. Agency.* Available at: <u>https://www.eea.europa.eu/publications/transport-and-environment-report-2020</u> (Accessed September 19, 2023).
- European Environment Agency (2022). Transport and environment report 2021. *Eur. Environ. Agency*. Available at: <u>https://</u><u>www.eea.europa.eu/publications/transport-and-environmentreport-2021</u> (Accessed September 19, 2023).
- Facanha, C., and Horvath, A. (2007). Evaluation of Life-Cycle Air Emission Factors of Freight Transportation. *Environ. Sci. Tech*nol. 41, 7138–7144. <u>https://doi.org/10.1021/es070989q</u>
- Falk, M. T., and Hagsten, E. (2021). Determinants of CO2 emissions generated by air travel vary across reasons for the trip. *En*viron. Sci. Pollut. Res. 28, 22969–22980. <u>https://doi.org/10.1007/</u> <u>s11356-020-12219-4</u>
- Fallahnejad, M., Kranzl, L., Haas, R., Hummel, M., Müller, A., García, L. S., et al. (2024). District heating potential in the EU-27: Evaluating the impacts of heat demand reduction and market share growth. *Appl. Energy* 353, 122154. <u>https://doi.org/ 10.1016/j.apenergy.2023.122154</u>
- Fasthuber, D. (2019). Auswirkungen und Potentiale der Integration von Elektromobilität in das elektrische Energiesystem Österreichs. TU Wien. <u>https://doi.org/10.34726/HSS.2019.28820</u>
- Fehrenbach, H., Bischoff, M., Böttcher, H., Reise, J., and Hennenberg, K. J. (2022). The Missing Limb: Including Impacts of Biomass Extraction on Forest Carbon Stocks in Greenhouse Gas Balances of Wood Use. *Forests* 13, 365. <u>https://doi.org/10.3390/ f13030365</u>
- Ferreira, J. C., Costa Dos Santos, D., and Campos, L. C. (2024). Bluegreen infrastructure in view of Integrated Urban Water Management: A novel assessment of an effectiveness index. Water Res. 257, 121658. <u>https://doi.org/10.1016/j.watres.2024.121658</u>
- Fian, T., Hörbinger, S, Müller, H, Lehner, S, Matulla, C, Wally, C, et al. (2021). clim\_ect. Klimawirkanalysen entlang der ÖBB-Bahnstrecken. Ergebnisbericht. Wien: Bundesministerium für Verkehr, Innovation und Technologie Abteilung Mobilitäts- und Verkehrstechnologien. Available at: <u>https://projekte. ffg.at/anhang/61ea7c9ed117c\_clim\_ect\_Ergebnisbericht.pdf</u> (Accessed May 10, 2024).
- Figgener, J., Tepe, B., Rücker, F., Schoeneberger, I., Hecht, C., Jossen, A., et al. (2022). The influence of frequency containment reserve flexibilization on the economics of electric vehicle fleet operation. *J. Energy Storage* 53, 105138. <u>https://doi. org/10.1016/J.EST.2022.105138</u>
- Fikar, C., Hirsch, P., Posset, M., and Gronalt, M. (2016). Impact of transalpine rail network disruptions: A study of the Brenner Pass. J. Transp. Geogr. 54, 122–131. <u>https://doi.org/10.1016/j.jtrangeo.2016.05.018</u>
- Fitzpatrick, P., D'Ettorre, F., De Rosa, M., Yadack, M., Eicker, U., and Finn, D. P. (2020). Influence of electricity prices on energy flexibility of integrated hybrid heat pump and thermal storage

systems in a residential building. *Energy Build*. 223, 110142. https://doi.org/10.1016/j.enbuild.2020.110142

- Flammini, M. G., Prettico, G., Julea, A., Fulli, G., Mazza, A., and Chicco, G. (2019). Statistical characterisation of the real transaction data gathered from electric vehicle charging stations. *Electr. Power Syst. Res.* 166, 136–150. <u>https://doi.org/10.1016/j.epsr.2018.09.022</u>
- Fletcher, T. D., Shuster, W., Hunt, W. F., Ashley, R., Butler, D., Arthur, S., et al. (2015). SUDS, LID, BMPs, WSUD and more – The evolution and application of terminology surrounding urban drainage. *Urban Water J.* 12, 525–542. <u>https://doi.org/10.1080/ 1573062X.2014.916314</u>
- Friesenecker, M., Thaler, T., and Clar, C. (2023). Green gentrification and changing planning policies in Vienna? Urban Res. Pract. 17, 393–415. <u>https://doi.org/10.1080/17535069.2023.222</u> 8275
- Frischknecht, R., Balouktsi, M., Lützkendorf, T., Aumann, A., Birgisdottir, H., Ruse, E. G., et al. (2019). Environmental benchmarks for buildings: needs, challenges and solutions—71st LCA forum, Swiss Federal Institute of Technology, Zürich, 18 June 2019. *Int. J. Life Cycle Assess.* 24, 2272–2280. <u>https://doi. org/10.1007/s11367-019-01690-y</u>
- Frischknecht, R., Ramseier, L., Yang, W., Birgisdottir, H., Chae, C. U., Lützkendorf, T., et al. (2020). Comparison of the greenhouse gas emissions of a high-rise residential building assessed with different national LCA approaches – IEA EBC Annex 72. IOP Conf. Ser. Earth Environ. Sci. 588, 022029. <u>https://doi.org/10.1088/1755-1315/588/2/022029</u>
- Fritz, D., Heinfellner, H., and Lambert, S. (2021). Die Ökobilanz von Personenkraftwagen. Vienna: Umweltbundesamt GmbH. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/</u> <u>publikationen/rep0763.pdf</u> (Accessed September 12, 2022).
- Fritz, D., Heinfellner, H., and Lambert, S. (2022). Die Ökobilanz von schweren Nutzfahrzeugen und Bussen. Bewertung ausgesuchter Anwendungsfälle alternativer Antriebskonzepte hinsichtlich Reduktionspotenzial von CO2-Emissionen und Energieverbrauch. Wien: Umweltbundesamt GmbH. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/publikationen/rep0801.pdf</u>
- Fuchs, S., Keiler, M., and Zischg, A. (2015). A spatiotemporal multi-hazard exposure assessment based on property data. *Nat. Hazards Earth Syst. Sci.* 15, 2127–2142. <u>https://doi.org/10.5194/</u> <u>nhess-15-2127-2015</u>
- Fuchs, S., Wenk, M., and Keiler, M. (2022). "Geomorphic Hazards in Austria", in *Landscapes and Landforms of Austria*, ed. C. Embleton-Hamann (Cham: Springer International Publishing), 105–117. https://doi.org/10.1007/978-3-030-92815-5\_6
- Funke, F., and Kleidorfer, M. (2024). Sensitivity of sustainable urban drainage systems to precipitation events and malfunctions. *Blue-Green Syst.* 6, 33–52. <u>https://doi.org/10.2166/bgs.2024.046</u>
- Gade, P. A. V., Skjøtskift, T., Bindner, H. W., and Kazempour, J. (2022). Ecosystem for Demand-side Flexibility Revisited: The Danish Solution. *Electr. J.* 35, 107206. <u>https://doi.org/10.1016/j. tej.2022.107206</u>
- Garcia-López, M.-À., Pasidis, I., and Viladecans-Marsal, E. (2022). Congestion in highways when tolls and railroads matter: evidence from European cities. *J. Econ. Geogr.* 22, 931–960. <u>https:// doi.org/10.1093/jeg/lbab025</u>
- Gea-Bermúdez, J., Jensen, I. G., Münster, M., Koivisto, M., Kirkerud, J. G., Chen, Y., et al. (2021). The role of sector coupling in the green transition: A least-cost energy system development

in Northern-central Europe towards 2050. *Appl. Energy* 289, 116685. <u>https://doi.org/10.1016/j.apenergy.2021.116685</u>

- GeoSphere Austria (2022). Massive Zunahme an Hitzetagen. *Geosph. Austria*. Available at: <u>https://geosphere.at/de/aktuelles/</u><u>news/massive-zunahme-an-hitzetagen</u> (Accessed September 22, 2023).
- Ghisellini, P., Ripa, M., and Ulgiati, S. (2018). Exploring environmental and economic costs and benefits of a circular economy approach to the construction and demolition sector. A literature review. J. Clean. Prod. 178, 618–643. <u>https://doi.org/10.1016/j.jclepro.2017.11.207</u>
- Gilleran, M., Bonnema, E., Woods, J., Mishra, P., Doebber, I., Hunter, C., et al. (2021). Impact of electric vehicle charging on the power demand of retail buildings. *Adv. Appl. Energy* 4, 100062. <u>https://doi.org/10.1016/j.adapen.2021.100062</u>
- Goncearuc, A., De Cauwer, C., Sapountzoglou, N., Kriekinge, G. V., Huber, D., Messagie, M., et al. (2024). The barriers to widespread adoption of vehicle-to-grid: A comprehensive review. *Energy Rep.* 12, 27–41. <u>https://doi.org/10.1016/j.egyr.2024.05.075</u>
- Gonzalez Venegas, F., Petit, M., and Perez, Y. (2021). Active integration of electric vehicles into distribution grids: Barriers and frameworks for flexibility services. *Renew. Sustain. Energy Rev.* 145, 111060. <u>https://doi.org/10.1016/j.rser.2021.111060</u>
- Gössling, S., Cohen, S. A., and Hares, A. (2016). Inside the black box: EU policy officers' perspectives on transport and climate change mitigation. *J. Transp. Geogr.* 57, 83–93. <u>https://doi.org/ 10.1016/j.jtrangeo.2016.10.002</u>
- Gruber, M., and Hofko, B. (2024). Carbin App zur Zustandserfassung der Oberflächenebenheit mittels International Roughness Index (IRI). Vienna. Available at: <u>http://hdl.handle.net/20.500.12708/190707</u>
- Gruber, M., Kanonier, A., Pohn-Weidinger, S., and Schindelegger, A. (2018). Raumordnung in Österreich und Bezüge zur Raumentwicklung und Regionalpolitik. Wien: Geschäftsstelle der Österreichischen Raumordnungskonferenz (ÖROK).
- Gruber, M. R. (2023). Ecological and Economic Aspects of Road Construction and Operation. TU Wien. <u>https://doi.org/10.34726/HSS.2023.79647</u>
- Gruber, M. R., and Hofko, B. (2023). Life Cycle Assessment of Greenhouse Gas Emissions from Recycled Asphalt Pavement Production. Sustainability 15, 4629. <u>https://doi.org/10.3390/ su15054629</u>
- Guo, B., and Weeks, M. (2022). Dynamic tariffs, demand response, and regulation in retail electricity markets. *Energy Econ*. 106, 105774. <u>https://doi.org/10.1016/j.eneco.2021.105774</u>
- Haas, W., Baumgart, A., Eisenmenger, N., Virág, D., Kalt, G., Sommer, M., et al. (2025). How decarbonization and the circular economy interact: Benefits and trade-offs in the case of the buildings, transport, and electricity sectors in Austria. *J. Ind. Ecol.*, 1–15. <u>https://doi.org/10.1111/jiec.13619</u>
- Haas, W., Meyer, I., Virág, D., Kalt, G., Baumgart, A., Eisenmenger, N., et al. (2024). Circular Economy and Decarbonisation: Synergies and Trade-offs. WIFO. Available at: <u>https://www.wifo. ac.at/publication/pid/53260426</u> (Accessed November 10, 2024).
- Haberl, H., Löw, M., Perez-Laborda, A., Matej, S., Plank, B., Wiedenhofer, D., et al. (2023). Built structures influence patterns of energy demand and CO2 emissions across countries. *Nat. Commun.* 14, 3898. <u>https://doi.org/10.1038/s41467-023-39728-3</u>
- Habert, G., Röck, M., Steininger, K., Lupisek, A., Birgisdóttir, H., Desing, H., et al. (2020). Carbon budgets for buildings: harmon-

ising temporal, spatial and sectoral dimensions. *Build. Cities* 1, 429–452. <u>https://doi.org/10.3929/ETHZ-B-000428030</u>

- Haddadian, G., Khalili, N., Khodayar, M., and Shahidehpour, M. (2016). Optimal coordination of variable renewable resources and electric vehicles as distributed storage for energy sustainability. *Sustain. Energy Grids Netw.* 6, 14–24. <u>https://doi. org/10.1016/j.segan.2015.12.001</u>
- Haddadian, G., Khalili, N., Khodayar, M., and Shahiedehpour, M. (2015). Security-constrained power generation scheduling with thermal generating units, variable energy resources, and electric vehicle storage for V2G deployment. *Int. J. Electr. Power Energy Syst.* 73, 498–507. https://doi.org/10.1016/j.ijepes.2015.05.020
- Hagen, M., and Weihs, P. (2023). Mortality during Heatwaves and Tropical Nights in Vienna between 1998 and 2022. *Atmosphere* 14, 1498. https://doi.org/10.3390/atmos14101498
- Hauger, G., Fian, T., Elias, D., Dominko, C., Kreihsler, S., and Scharl, A. (2022). Conversion: Chancen für die Transformation des Mobilitätssystems. Vienna: Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie (BMK). Available at: <u>https://projekte.ffg.at/anhang/628ddb-43d53c5\_Conversion\_Endbericht\_220509\_v2.pdf</u> (Accessed September 12, 2022).
- Heidenthaler, D., Deng, Y., Leeb, M., Grobbauer, M., Kranzl, L., Seiwald, L., et al. (2023). Automated energy performance certificate based urban building energy modelling approach for predicting heat load profiles of districts. *Energy* 278, 128024. <u>https://doi.org/10.1016/j.energy.2023.128024</u>
- Heinfellner, H., Angelini, A., Böhnke, M., Gugele, B., Xaca, C. M., Raimund, W., et al. (2024a). Maßnahmenbericht für eine sozialund klimaverträgliche Mobilitätswende – Sachstand Mobilität 2024. Wien: Umweltbundesamt GmbH. Available at: <u>https:// www.umweltbundesamt.at/fileadmin/site/publikationen/rep</u> 0928.pdf (Accessed October 6, 2024).
- Heinfellner, H., Glatt, A., and Stranner, G. (2024b). Detailbericht zur Nahzeitprognose der Österreichischen Treibhausgas-Emissionen des Verkehrs 2023. Wien: Umweltbundesamt GmbH. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/ publikationen/rep0921.pdf</u> (Accessed October 6, 2024).
- Hess, D. B. (2001). Effect of Free Parking on Commuter Mode Choice: Evidence from Travel Diary Data. *Transp. Res. Rec.* 1753, 35–42. <u>https://doi.org/10.3141/1753-05</u>
- Heyen, D. A., and Wicki, M. (2024). Increasing public support for climate policy proposals: a research agenda on governable acceptability factors. *Clim. Policy* 24, 785–794. <u>https://doi.org/10. 1080/14693062.2024.2330390</u>
- Hickman, R., and Banister, D. (2015). Transport, Climate Change and the City. New York: Routledge & CRC Press. Available at: <u>https://www.routledge.com/Transport-Climate-Change-and-the-City/Hickman-Banister/p/book/9780415660037</u> (Accessed September 22, 2023).
- Hoekstra, A. (2019). The Underestimated Potential of Battery Electric Vehicles to Reduce Emissions. *Joule* 3, 1412–1414. <u>https:// doi.org/10.1016/j.joule.2019.06.002</u>
- Hofko, B., Gruber, M., and Rechberger, H. (2020). Qualitätskriterium Recyclingasphalt. Wien: Technische Universität Wien. Available at: <u>http://hdl.handle.net/20.500.12708/40200</u> (Accessed February 1, 2025).
- Horak, D., Hainoun, A., Neugebauer, G., and Stoeglehner, G. (2022). A review of spatio-temporal urban energy system modeling for urban decarbonization strategy formulation. *Renew.*

Sustain. Energy Rev. 162, 112426. <u>https://doi.org/10.1016/j.</u> rser.2022.112426

- Horak, J., and Peßenteiner, S. (2023). Zukunft Linz Der klimagerechte Weg von Linz zur Anpassung an den Klimawandel. <u>https://doi.org/10.5281/ZENODO.14191053</u>
- Hössinger, R., Peer, S., and Juschten, M. (2023). Give citizens a task: An innovative tool to compose policy bundles that reach the climate goal. *Transp. Res. Part Policy Pract.* 173, 103694. <u>https:// doi.org/10.1016/j.tra.2023.103694</u>
- Hotmaps Project (2023). Hotmaps (EU Horizon 2020 funded). Available at: <u>https://www.hotmaps.eu/map</u> (Accessed September 22, 2023).
- Huttenlau, M., Stötter, J., and Stiefelmeyer, H. (2010). Risk-based damage potential and loss estimation of extreme flooding scenarios in the Austrian Federal Province of Tyrol. Nat. Hazards Earth Syst. Sci. 10, 2451–2473. <u>https://doi.org/10.5194/</u> <u>nhess-10-2451-2010</u>
- IEA (2019). 2019 Global Status Report for Buildings and Construction: Towards a Zero-Emission, Efficient and Resilient Buildings and Construction Sector. Paris: IEA. Available at: <u>https://www. iea.org/reports/global-status-report-for-buildings-and-construction-2019</u>
- Ioakimidis, C. S., Thomas, D., Rycerski, P., and Genikomsakis, K. N. (2018). Peak shaving and valley filling of power consumption profile in non-residential buildings using an electric vehicle parking lot. *Energy* 148, 148–158. <u>https://doi.org/10.1016/J. ENERGY.2018.01.128</u>
- IPCC (2021a). Climate Change 2021 The Physical Science Basis: Working Group I Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change., 1st Edn. Cambridge University Press. <u>https://doi.org/10.1017/9781009157896</u>
- IPCC (2021b). "Summary for Policymakers", in Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, eds. V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press), 3–32. https://doi.org/10.1017/9781009157896.001
- IPCC (2022a). Climate Change 2022 Impacts, Adaptation and Vulnerability: Working Group II Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change., 1st Edn. Cambridge University Press. <u>https://doi.org/10.1017/</u> 9781009325844
- IPCC (2022b). "Summary for Policymakers", in Global Warming of 1.5°C: IPCC Special Report on Impacts of Global Warming of 1.5°C above Pre-industrial Levels in Context of Strengthening Response to Climate Change, Sustainable Development, and Efforts to Eradicate Poverty, eds. V. Masson-Delmotte, P. Zhai, H.-O. Pörtner, D. Roberts, J. Skea, P. R. Shukla, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press), 3–24. https://doi.org/10.1017/9781009157940
- ITF (2022). Transport infrastructure investment and maintenance spending. OECD Data Explor. Available at: <u>https://stats.oecd.org/Index.aspx?DataSetCode=ITF\_INV-MTN\_DATA#</u> (Accessed September 20, 2023).
- ITF (2023). ITF Transport Outlook 2023. Paris: OECD Publishing. https://doi.org/10.1787/b6cc9ad5-en
- Jandl, R., Tappeiner, U., Foldal, C. B., and Erb, K.-H. eds. (2024). APCC Special Report: Landnutzung und Klimawandel in Österreich. Berlin, Heidelberg: Springer Spektrum. <u>https://doi.org/ 10.1007/978-3-662-67864-0</u>

- Jian, L., Zheng, Y., and Shao, Z. (2017). High efficient valley-filling strategy for centralized coordinated charging of large-scale electric vehicles. *Appl. Energy* 186, 46–55. <u>https://doi.org/10.1016/J. APENERGY.2016.10.117</u>
- Jones, H., Moura, F., and Domingos, T. (2017). Life cycle assessment of high-speed rail: a case study in Portugal. *Int. J. Life Cycle* Assess. 22, 410–422. https://doi.org/10.1007/s11367-016-1177-7
- Kadi, J., Banabak, S., and Schneider, A. (2022). Widening gaps? Socio-spatial inequality in the "very" European city of Vienna since the financial crisis. *Cities* 131, 103887. <u>https://doi.org/10.1016/j.cities.2022.103887</u>
- Kanters, J. (2020). Circular Building Design: An Analysis of Barriers and Drivers for a Circular Building Sector. *Buildings* 10, 77. https://doi.org/10.3390/buildings10040077
- Karimi-Arpanahi, S., Jooshaki, M., Pourmousavi, S. A., and Lehtonen, M. (2022). Leveraging the flexibility of electric vehicle parking lots in distribution networks with high renewable penetration. *Int. J. Electr. Power Energy Syst.* 142, 108366. <u>https://doi. org/10.1016/j.ijepes.2022.108366</u>
- Kerr, A. D. (1994). Blowup of a concrete pavement adjoining a rigid structure. Int. J. Non-Linear Mech. 29, 387–396. <u>https://doi. org/10.1016/0020-7462(94)90009-4</u>
- Khodayar, M. E., Wu, L., and Shahidehpour, M. (2012). Hourly Coordination of Electric Vehicle Operation and Volatile Wind Power Generation in SCUC. *IEEE Trans. Smart Grid* 3, 1271– 1279. <u>https://doi.org/10.1109/TSG.2012.2186642</u>
- Kleidorfer, M., Tscheikner-Gratl, F., Vonach, T., and Rauch, W. (2018). What can we learn from a 500-year event? Experiences from urban drainage in Austria. *Water Sci. Technol.* 77, 2146– 2154. https://doi.org/10.2166/wst.2018.138
- Koch, N., Naumann, L., Pretis, F., Ritter, N., and Schwarz, M. (2022). Attributing agnostically detected large reductions in road CO2 emissions to policy mixes. *Nat. Energy* 7, 844–853. <u>https://doi.org/10.1038/s41560-022-01095-6</u>
- Koller, K., Hyben, T., Rittsteuer, F., and Höglinger, K. (2023). Aktuelle Leerstandsabgabe-Gesetze in Österreich. *PwC Leg. Blog.*Available at: <u>https://blog.pwclegal.at/aktuelle-leerstandsabgabe-gesetze-in-oesterreich/</u> (Accessed May 10, 2024).
- König, M., Wolkinger, B., Bednar-Friedl, B., and Felderer, A. (2014). Ein Anpassungsfahrplan für die Österreichische Strassenverkehrsinfrastruktur. Wien: Umweltbundesamt GmbH.
- Kourtis, I. M., and Tsihrintzis, V. A. (2021). Adaptation of urban drainage networks to climate change: A review. *Sci. Total Environ.* 771, 145431. <u>https://doi.org/10.1016/j.scitotenv.2021.</u> 145431
- KPC (2022). Übersicht förderungsfähige Heizungssysteme. Available at: <u>https://www.umweltfoerderung.at/uebersicht-foerderungsfaehige-heizungssysteme</u> (Accessed May 10, 2024).
- Kraisbau (2022). KRAISBAU als Grundstein für eine Zirkuläre Baubranche. Available at: <u>https://www.kraisbau.at/</u> (Accessed November 22, 2024).
- Kranzl, L., Müller, A., Schipfer, F., Smet, K., Grabner, D., Litschauer, K., et al. (2020). Transitioning buildings to full reliance on renewable energy and assuring inclusive and affordable housing. ACRP project Decarb Inclusive final report. (No. 1). Vienna: Klima- und Energiefonds. Available at: <u>https://repositum.tuwien.at/handle/20.500.12708/40258</u>
- Krellenberg, K., Welz, J., Link, F., and Barth, K. (2017). Urban vulnerability and the contribution of socio-environmental fragmentation: Theoretical and methodological pathways. *Prog. Hum. Geogr.* 41, 408–431. <u>https://doi.org/10.1177/0309132516645959</u>

- Kröger, D., Peper, J., and Rehtanz, C. (2023). Electricity market modeling considering a high penetration of flexible heating systems and electric vehicles. *Appl. Energy* 331, 120406. <u>https://doi.org/10.1016/j.apenergy.2022.120406</u>
- Kumar, R. R., and Alok, K. (2020). Adoption of electric vehicle: A literature review and prospects for sustainability. J. Clean. Prod. 253, 119911. <u>https://doi.org/10.1016/j.jclepro.2019.119911</u>
- Laa, B., and Leth, U. (2020). Survey of E-scooter users in Vienna: Who they are and how they ride. *J. Transp. Geogr.* 89, 102874. https://doi.org/10.1016/j.jtrangeo.2020.102874
- Laa, B., Shibayama, T., Brezina, T., Schönfelder, S., Damjanovic, D., Szalai, E., et al. (2022). A nationwide mobility service guarantee for Austria: possible design scenarios and implications. *Eur. Transp. Res. Rev.* 14, 25. <u>https://doi.org/10.1186/s12544-022-00550-5</u>
- Landgraf, M., and Horvath, A. (2021). Embodied greenhouse gas assessment of railway infrastructure: the case of Austria. *Envi*ron. Res. Infrastruct. Sustain. 1, 025008. <u>https://doi.org/10.1088/</u> <u>2634-4505/ac1242</u>
- Landgraf, M., Zeiner, M., Knabl, D., and Corman, F. (2022). Environmental impacts and associated costs of railway turnouts based on Austrian data. *Transp. Res. Part Transp. Environ.* 103, 103168. <u>https://doi.org/10.1016/j.trd.2021.103168</u>
- Le Den, X., Steinmann, J., Kockat, J., Toth, Z., Röck, M., Allacker, K., et al. (2023). Supporting the development of a roadmap for the reduction of whole life carbon of buildings. final report. Brussels: Publications Office of the European Union. Available at: <u>https://data.europa.eu/doi/10.2779/634412</u> (Accessed May 10, 2024).
- Le Dréau, J., and Heiselberg, P. (2016). Energy flexibility of residential buildings using short term heat storage in the thermal mass. *Energy* 111, 991–1002. <u>https://doi.org/10.1016/j.energy.2016.</u> 05.076
- Lederer, J., and Blasenbauer, D. (2024). Material Flow Analysis-Based Sustainability Assessment for Circular Economy Scenarios of Urban Building Stock of Vienna. *Sustainability* 16, 7319. <u>https://doi.org/10.3390/su16177319</u>
- Lederer, J., Gassner, A., Kleemann, F., and Fellner, J. (2020). Potentials for a circular economy of mineral construction materials and demolition waste in urban areas: a case study from Vienna. *Resour. Conserv. Recycl.* 161, 104942. <u>https://doi.org/10.1016/j. resconrec.2020.104942</u>
- Lee, D. S., Fahey, D. W., Skowron, A., Allen, M. R., Burkhardt, U., Chen, Q., et al. (2021). The contribution of global aviation to anthropogenic climate forcing for 2000 to 2018. *Atmos. Environ.* 244, 117834. <u>https://doi.org/10.1016/j.atmosenv.2020.117834</u>
- Li, B., Liu, Z., Wu, Y., Wang, P., Liu, R., and Zhang, L. (2023). Review on photovoltaic with battery energy storage system for power supply to buildings: Challenges and opportunities. *J. Energy Storage* 61, 106763. <u>https://doi.org/10.1016/j.est.2023.106763</u>
- Li, C., Chen, G., and Wang, S. (2024). Urban mobility resilience under heat extremes: Evidence from bike-sharing travel in New York. *Travel Behav. Soc.* 37, 100821. <u>https://doi.org/10.1016/j. tbs.2024.100821</u>
- Li, X., Tan, Y., Liu, X., Liao, Q., Sun, B., Cao, G., et al. (2020). A cost-benefit analysis of V2G electric vehicles supporting peak shaving in Shanghai. *Electr. Power Syst. Res.* 179, 106058. <u>https:// doi.org/10.1016/J.EPSR.2019.106058</u>
- Loschan, C., Auer, H., and Lettner, G. (2024). Synergies and competition: Examining flexibility options in the European electric-

ity market. Int. J. Electr. Power Energy Syst. 159, 109992. <u>https://</u>doi.org/10.1016/j.ijepes.2024.109992

- Loschan, C., Schwabeneder, D., Lettner, G., and Auer, H. (2023). Flexibility potential of aggregated electric vehicle fleets to reduce transmission congestions and redispatch needs: A case study in Austria. *Int. J. Electr. Power Energy Syst.* 146, 108802. https://doi.org/10.1016/j.ijepes.2022.108802
- Löschner, L., Herrnegger, M., Apperl, B., Senoner, T., Seher, W., and Nachtnebel, H. P. (2017). Flood risk, climate change and settlement development: a micro-scale assessment of Austrian municipalities. *Reg. Environ. Change* 17, 311–322. <u>https://doi. org/10.1007/s10113-016-1009-0</u>
- Louhghalam, A., Tootkaboni, M., Igusa, T., and Ulm, F. J. (2019). Closed-Form Solution of Road Roughness-Induced Vehicle Energy Dissipation. J. Appl. Mech. 86, 011003. <u>https://doi. org/10.1115/1.4041500</u>
- Luderer, G., Kost, C., and Sörgel, D. (2021). Deutschland auf dem Weg zur Klimaneutralität 2045 – Szenarien und Pfade im Modellvergleich. Potsdam: Potsdam Institute for Climate Impact Research. <u>https://doi.org/10.48485/pik.2021.006</u>
- Luo, J. (Tom), Joybari, M. M., Panchabikesan, K., Haghighat, F., Moreau, A., and Robichaud, M. (2020). Parametric study to maximize the peak load shifting and thermal comfort in residential buildings located in cold climates. *J. Energy Storage* 30, 101560. https://doi.org/10.1016/j.est.2020.101560
- Lwasa, S., Seto, K. C., Bai, X., Blanco, H., Gurney, K. R., Kılkış, Ş., et al. (2023). "Urban Systems and Other Settlements", in *IPCC*, 2022: Climate Change 2022 – Mitigation of Climate Change: Working Group III Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, eds. P. R. Shukla, J. Skea, R. Slade, A. Al Khourdajie, R. van Diemen, D. McCollum, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press), 861–952. <u>https://doi.org/10.1017/</u> 9781009157926.010
- Magistrat der Stadt Wien (2015). Urban Heat Islands. Strategieplan Wien. Wien: Wiener Umweltschutzabteilung Magistratsabteilung 22. Available at: <u>https://www.wien.gv.at/umweltschutz/</u> <u>raum/pdf/uhi-strategieplan.pdf?trk=public\_post\_commenttext</u> (Accessed June 18, 2024).
- Magistrat Linz (2023). Zukunft Linz Der klimagerechte Weg von Linz zur Anpassung an den Klimawandel. Linz: Magistrat Linz – Planung, Technik und Umwelt. Available at: <u>https://www.linz.</u> <u>at/media/umwelt/Anpassungskonzept\_Zukunft\_Linz\_2023.</u> <u>pdf</u> (Accessed June 18, 2024).
- Magistrat Wien (2023). Kordonerhebung Wien 2022: Ergebnisbericht / Auftraggeber: Magistrat der Stadt Wien MA 18 – Stadtentwicklung und Stadtplanung, Amt der NÖ Landesregierung RU7 Raumordnung und Gesamtverkehrsangelegenheit; Auftragnehmer: Technische Universität Graz, Inst. f. Straßen- und Verkehrswesen, Ingenieurbüro für Verkehrswesen und Verkehrswirtschaft – DI Rittler Christian. Graz, Wien: Technische Universität Graz, MA18 – Stadtentwicklung und Stadtplanung. Available at: https://resolver.obvsg.at/urn:nbn:at:AT-WBR-1248398.
- Magni, M., and Ochs, F. (2021). Energiespeicher Tirol 2050 Stufe II.
- Maier, R., Posch, A., Proß, C., Plakolb, S., and Steininger, K. W. (2023). Cutting social costs by decarbonizing passenger transport. *Transp. Res. Part Transp. Environ.* 122, 103878. <u>https://doi. org/10.1016/j.trd.2023.103878</u>

- Maier, R., Thaller, A., and Fleiß, E. (2022). Telework: A Social Tipping Intervention for Passenger Transportation? SSRN Electron. J. <u>https://doi.org/10.2139/ssrn.4226849</u>
- Maierhofer, D., Van Karsbergen, V., Potrč Obrecht, T., Ruschi Mendes Saade, M., Gingrich, S., Streicher, W., et al. (2024).
  Linking forest carbon opportunity costs and greenhouse gas emission substitution effects of wooden buildings: The climate optimum concept. Sustain. Prod. Consum. 51, 612–627. <u>https:// doi.org/10.1016/j.spc.2024.08.021</u>
- Malabi Eberhardt, L. C., Van Stijn, A., Nygaard Rasmussen, F., Birkved, M., and Birgisdottir, H. (2020). Development of a Life Cycle Assessment Allocation Approach for Circular Economy in the Built Environment. Sustainability 12, 9579. <u>https://doi.org/10.3390/su12229579</u>
- Mandel, T., Worrell, E., and Alibaş, Ş. (2023). Balancing heat saving and supply in local energy planning: Insights from 1970– 1989 buildings in three European countries. *Smart Energy* 12, 100121. <u>https://doi.org/10.1016/j.segy.2023.100121</u>
- Mañes-Navarrete, D., Redón-Santafé, M., and Paya-Zaforteza, I. (2024). Timber structures designed for disassembly: A cornerstone for sustainability in 21st century construction. J. Build. Eng. 96, 110619. <u>https://doi.org/10.1016/j.jobe.2024.110619</u>
- Marcucci, E., Gatta, V., Simoni, M., and Maltese, I. (2023). "Pricing in freight transport", in *Handbook on Transport Pricing and Financing*, eds. A. Tirachini, D. Hörcher, and E. T. Verhoef (Cheltenham: Edward Elgar Publishing), 229–251. <u>https://doi.org/</u> <u>10.4337/9781800375550.00019</u>
- Marquard, E., Bartke, S., Gifreu I Font, J., Humer, A., Jonkman, A., Jürgenson, E., et al. (2020). Land Consumption and Land Take: Enhancing Conceptual Clarity for Evaluating Spatial Governance in the EU Context. Sustainability 12, 8269. <u>https://doi.org/10.3390/su12198269</u>
- Martilli, A., Roth, M., Chow, W. T. L., Demuzere, M., Lipson, M., Krayenhoff, E. S., et al. (2020). Summer average urban-rural surface temperature differences do not indicate the need for urban heat reduction. <u>https://doi.org/10.31219/osf.io/8gnbf</u>
- Mascherbauer, P., Kranzl, L., Yu, S., and Haupt, T. (2022a). Investigating the impact of smart energy management system on the residential electricity consumption in Austria. *Energy* 249, 123665. <u>https://doi.org/10.1016/j.energy.2022.123665</u>
- Mascherbauer, P., Schöniger, F., Kranzl, L., and Yu, S. (2022b). Impact of variable electricity price on heat pump operated buildings. *Open Res. Eur.* 2, 135. <u>https://doi.org/10.12688/openreseurope.15268.1</u>
- Masy, G., Georges, E., Verhelst, C., Lemort, V., and André, P. (2015). Smart grid energy flexible buildings through the use of heat pumps and building thermal mass as energy storage in the Belgian context. *Sci. Technol. Built Environ.* 21, 800–811. <u>https:// doi.org/10.1080/23744731.2015.1035590</u>
- Mattioli, G., Roberts, C., Steinberger, J. K., and Brown, A. (2020). The political economy of car dependence: A systems of provision approach. *Energy Res. Soc. Sci.* 66, 101486. <u>https://doi.org/10.1016/j.erss.2020.101486</u>
- Meersman, H., and Van de Voorde, E. (2013). "The Relationship between Economic Activity and Freight Transport", in *Freight Transport Modelling*, eds. M. Ben-Akiva, H. Meersman, and E. Van de Voorde (Emerald Group Publishing Limited), 15–43. https://doi.org/10.1108/9781781902868-002
- Miara, M., Günther, D., Leitner, Z. L., and Wapler, J. (2014). Simulation of an Air-to-Water Heat Pump System to Evaluate the Impact of Demand-Side-Management Measures on Efficiency

and Load-Shifting Potential. *Energy Technol.* 2, 90–99. <u>https://doi.org/10.1002/ente.201300087</u>

- Miklautsch, P., König, A., and Woschank, M. (2022). Identifying a Country's Freight Transport-Intensive Economic Sectors and Their Logistics Emissions—Method Development and Exemplary Evaluation with Austria. *Sustainability* 14, 15050. <u>https:// doi.org/10.3390/su142215050</u>
- Mikovits, C., Tscheikner-Gratl, F., Jasper-Tönnies, A., Einfalt, T., Huttenlau, M., Schöpf, M., et al. (2017). Decision Support for Adaptation Planning of Urban Drainage Systems. J. Water Resour. Plan. Manag. 143, 04017069. <u>https://doi.org/10.1061/ (ASCE)WR.1943-5452.0000840</u>
- Millonig, A., Rudloff, C., Richter, G., Lorenz, F., and Peer, S. (2022). Fair mobility budgets: A concept for achieving climate neutrality and transport equity. *Transp. Res. Part Transp. Environ.* 103, 103165. <u>https://doi.org/10.1016/j.trd.2021.103165</u>
- Mills, G., and MacGill, I. (2018). Assessing Electric Vehicle storage, flexibility, and Distributed Energy Resource potential. J. Energy Storage 17, 357–366. <u>https://doi.org/10.1016/j.est.2018.01.016</u>
- Mirakyan, A., and De Guio, R. (2013). Integrated energy planning in cities and territories: A review of methods and tools. *Renew. Sustain. Energy Rev.* 22, 289–297. <u>https://doi.org/10.1016/j.rser.</u> 2013.01.033
- Mirzaie, S., Thuring, M., and Allacker, K. (2020). End-of-life modelling of buildings to support more informed decisions towards achieving circular economy targets. *Int. J. Life Cycle Assess.* 25, 2122–2139. <u>https://doi.org/10.1007/s11367-020-01807-8</u>
- Misconel, S., Prina, M. G., Hobbie, H., Möst, D., and Sparber, W. (2022). Model-based step-wise marginal CO<sub>2</sub> abatement cost curves to determine least-cost decarbonization pathways for sector-coupled energy systems. J. Clean. Prod. 368, 133173. <u>https://doi.org/10.1016/j.jclepro.2022.133173</u>
- Mock, M. (2023). Making and breaking links: the transformative potential of shared mobility from a practice theories perspective. *Mobilities* 18, 374–390. <u>https://doi.org/10.1080/17450101.</u> 2022.2142066
- Morrison, K., Miller, J., Rodrigues, P. F., Mulholland, E., Zhou, Y., Baldino, C., et al. (2024). Cleaning up Germany's vehicle stock: Strategies to decarbonize the passenger car fleet. Berlin: ICCT. Available at: <u>https://theicct.org/publication/vehicle-stock-emis-</u> sions-pv-germany-oct24/ (Accessed November 7, 2024).
- Müller, A., Hummel, M., Kranzl, L., Fallahnejad, M., and Büchele, R. (2019). Open Source Data for Gross Floor Area and Heat Demand Density on the Hectare Level for EU 28. *Energies* 12, 4789. <u>https://doi.org/10.3390/en12244789</u>
- Muschalla, D. (2023). Potenzielle Auswirkungen der geplanten EU-AwRL auf das Management von Entwässerungssystemen. Available at: <u>https://www.oewav.at/upload/medialibrary/Folder\_Kanalmanagement\_2023\_FIN5258.pdf</u> (Accessed September 22, 2023).
- Myrup, L. O. (1969). A Numerical Model of the Urban Heat Island. J. Appl. Meteorol. Climatol. 8, 908–918. <u>https://doi.org/10.1175/ 1520-0450(1969)008<0908:ANMOTU>2.0.CO;2</u>
- NetZero2040 (2024). Szenarien zur Erreichung der Klimaneutralität in Österreich bis 2040. *NetZero2040*. Available at: <u>https://</u> <u>www.netzero2040.at</u> (Accessed November 17, 2024).
- Newman, P., and Kenworthy, J. (2006). Urban design to reduce automobile dependence. *Opolis* 2, 35–52.
- Nikoobakht, A., Aghaei, J., Niknam, T., Farahmand, H., and Korpås, M. (2019). Electric vehicle mobility and optimal grid reconfiguration as flexibility tools in wind integrated pow-

- ÖAMTC (2003). ÖAMTC: Pkw-Fuhrpark der Österreicher ist durchschnittlich 7,84 Jahre alt. Available at: <u>https://www.ots.</u> <u>at/presseaussendung/OTS\_20031120\_OTS0062/oeamtc-pkw-fuhrpark-der-oesterreicher-ist-durchschnittlich-784-jahre-alt</u> (Accessed November 7, 2024).
- ÖBB-Holding AG (2022). Nachhaltigkeitsbericht 2021. Wien: ÖBB-Holding AG. Available at: <u>https://static.web.oebb.at/kon-zern/nhb21/index.html</u> (Accessed September 11, 2022).
- ÖBB-Holding AG (2024). ÖBB kompakt 2023/2024. Zahlen Daten Fakten. Wien: ÖBB-Holding AG. Available at: <u>https://presse. oebb.at/de/dam/jcr:3a7eda30-9fee-475c-a40c-7e12350c4689/</u> <u>OEBB\_Zahlen\_2024.pdf</u> (Accessed November 12, 2024).
- ÖBB-Infrastruktur AG (2024). Netzzustandsbericht 2023. Wien: ÖBB-Infrastruktur AG. Available at: <u>https://infrastruktur.</u> <u>oebb.at/dam/jcr:8c9f6fc2-9afb-46d4-a11b-b81326982717/</u> <u>oebb-netzzustandsbericht-2023.pdf</u> (Accessed January 19, 2025).
- Ochs, F., Magni, M., Venturi, E., Tosatto, A., and Streicher, W. (2021). Cost-optimal nZEB HVAC configurations with onsite storage. E3S Web Conf. 246, 07001. <u>https://doi.org/10.1051/e3sconf/202124607001</u>
- O'Connell, A., Pavlenko, N., Bieker, G., and Searle, S. (2022). A comparison of the life-cycle greenhouse gas emissions of European heavy-duty vehicles and fuels. Washington: ICCT. Available at: <u>https://theicct.org/publication/lca-ghg-emissions-hdv-fuels-europe-feb23/</u>
- Odyssee-Mure (2022a). Passenger mobility per capita. Available at: <u>https://www.odyssee-mure.eu/publications/efficiency-by-sec-</u> <u>tor/transport/passenger-mobility-per-capita.html</u> (Accessed May 10, 2024).
- Odyssee-Mure (2022b). Trends in freight traffic. Trends in freight traffic and GDP (EU). Available at: <u>https://www.odyssee-mure.</u> <u>eu/publications/efficiency-by-sector/transport/freight-traffic-</u> <u>trends-gdp-eu-level.html</u> (Accessed May 10, 2024).
- OIB (2019). OIB-Richtlinie 6. Energieeinsparung und Wärmeschutz. Wien: Österreichisches Institut für Bautechnik. Available at: <u>https://www.oib.or.at/de/oib-richtlinien/richtlinien/2019/ oib-richtlinie-6-kostenoptimalit%C3%A4t</u> (Accessed September 25, 2023).
- OIB (2023). OIB-Grundlagendokument zur Ausarbeitung einer OIB-Richtlinie 7 Nachhaltige Nutzung der natürlichen Ressourcen. Wien: Österreichisches Institut für Bautechnik. Available at: <u>https://www.oib.or.at/de/oib-richtlinien/richtlinien/2023/</u> <u>oib-richtlinie-7-grundlagendokument</u> (Accessed May 10, 2024).
- Oke, T. R., and Fuggle, R. F. (1972). Comparison of urban/rural counter and net radiation at night. *Bound.-Layer Meteorol.* 2, 290–308. <u>https://doi.org/10.1007/BF02184771</u>
- Olsson, J., Camhy, D., Hochedlinger, M., Kutschera, P., Schlobinski, S., and Gruber, G. (2012). "Assessment of Combined Sewer Overflows under Climate Change Urban Drainage Pilot Study Linz", in *IWA World Congress on Water, Climate and Energy* 2012, (AIT). Available at: <u>https://publications.ait.ac.at/en/publications/assessment-of-combined-sewer-overflows-under-climate-change-urban</u>
- Oral, H. V., Carvalho, P., Gajewska, M., Ursino, N., Masi, F., Hullebusch, E. D. V., et al. (2020). A review of nature-based solutions for urban water management in European circular cities: a crit-

ical assessment based on case studies and literature. *Blue-Green Syst.* 2, 112–136. <u>https://doi.org/10.2166/bgs.2020.932</u>

- ORF (2024). Rekordkosten durch kaputte Infrastruktur. *steiermark.* ORF.at. Available at: <u>https://steiermark.orf.at/stories/3270640/</u> (Accessed November 10, 2024).
- ÖROK (2015). Dauersiedlungsraum. Available at: <u>https://www.oerok-atlas.at/oerok/files/summaries/74.pdf</u> (Accessed February 1, 2025).
- ÖROK (2021). Das österreichische Raumentwicklungskonzept 2030 – ÖREK 2030 – Raum für Wandel. Österreichischen Raumordnungskonferenz (ÖROK). Available at: <u>https://www.oerek2030.at/</u> (Accessed September 22, 2023).
- ÖROK (2022). Flächeninanspruchnahme und Versiegelung in Österreich (2022). Available at: <u>https://www.oerok.gv.at/raum/ daten-und-grundlagen/ergebnisse-oesterreich-2022</u> (Accessed February 1, 2025).
- ÖROK (2023a). Beispielsammlung zur Bodenstrategie für Österreich. Wien: Österreichische Raumordnungskonferenz (ÖROK). Available at: <u>https://www.oerok.gv.at/fileadmin/user\_upload/Bilder/2.Reiter-Raum\_u\_Region/6\_OEREK\_Umsetzungspakte/Bodenstrategie/Sammlung\_Good\_Practice\_Bodenstrategie\_Final.pdf</u> (Accessed February 1, 2025).
- ÖROK (2023b). Bodenstrategie für Österreich. Available at: <u>https://www.oerok.gv.at/bodenstrategie</u> (Accessed February 1, 2025).
- ÖROK (2023c). ÖROK-Bevölkerungsprognose 2021–2051. ÖROK-Atlas. Available at: <u>https://www.oerok-atlas.at/#indica-tor/65</u> (Accessed September 22, 2023).
- ÖROK (2023d). ÖROK-Monitoring von Flächeninanspruchnahme und Versiegelung. Available at: <u>https://www.oerok.gv.at/monitoring-flaecheninanspruchnahme</u> (Accessed January 27, 2025).
- Osório, G. J., Lotfi, M., Gough, M., Javadi, M., Espassandim, H. M. D., Shafie-khah, M., et al. (2021). Modeling an electric vehicle parking lot with solar rooftop participating in the reserve market and in ancillary services provision. *J. Clean. Prod.* 318, 128503. <u>https://doi.org/10.1016/j.jclepro.2021.128503</u>
- Oswald, S. M., Hollosi, B., Žuvela-Aloise, M., See, L., Guggenberger, S., Hafner, W., et al. (2020). Using urban climate modelling and improved land use classifications to support climate change adaptation in urban environments: A case study for the city of Klagenfurt, Austria. *Urban Clim.* 31, 100582. <u>https://doi. org/10.1016/j.uclim.2020.100582</u>
- Owen, J. R., Kemp, D., Lechner, A. M., Harris, J., Zhang, R., and Lèbre, É. (2023). Energy transition minerals and their intersection with land-connected peoples. *Nat. Sustain.* 6, 203–211. <u>https://doi.org/10.1038/s41893-022-00994-6</u>
- Palin, E. J., Stipanovic Oslakovic, I., Gavin, K., and Quinn, A. (2021). Implications of climate change for railway infrastructure. WIREs Clim. Change 12, e728. <u>https://doi.org/10.1002/ wcc.728</u>
- Parlamentsdirektion (2024a). Konjunkturpaket für den Wohnbau. Wien: Parlament Österreich. Available at: <u>https://www. parlament.gv.at/fachinfos/budgetdienst/Konjunkturpaket-fuerden-Wohnbau</u> (Accessed May 10, 2024).
- Parlamentsdirektion (2024b). Leerstandsabgabe: Ausschuss schickt Gesetzentwurf in Begutachtung. Parlam. Österr. Available at: <u>https://www.parlament.gv.at/aktuelles/news/Leerstandsab-gabe-Ausschuss-schickt-Gesetzentwurf-in-Begutachtung</u> (Accessed May 10, 2024).
- Pasqui, M., Vaccaro, G., Lubello, P., Milazzo, A., and Carcasci, C. (2023). Heat pumps and thermal energy storages centralised management in a Renewable Energy Community. *Int. J. Sus-*

tain. Energy Plan. Manag. 38, 65-82. <u>https://doi.org/10.54337/</u> ijsepm.7625

- Pavić, I., Capuder, T., and Kuzle, I. (2015). Value of flexible electric vehicles in providing spinning reserve services. *Appl. Energy* 157, 60–74. <u>https://doi.org/10.1016/J.APENERGY.2015.07.070</u>
- Peer, S., Gangl, K., Spitzer, F., and van der Werff, E. (2023). Which policy measures can motivate active mobility in rural and semi-rural areas? *Transp. Res. Part Transp. Environ.* 118, 103688. https://doi.org/10.1016/j.trd.2023.103688
- Peneder, M., Bittschi, B., Köppl, A., Mayerhofer, P., and Url, T. (2022). The WIFO Radar of Competitiveness for the Austrian Economy 2021. Wien: WIFO. Available at: <u>https://www.wifo. ac.at/wp-content/uploads/upload-6974/roa\_2022\_02\_wifo\_radar\_.pdf</u>
- Peng, L., Searchinger, T. D., Zionts, J., and Waite, R. (2023). The carbon costs of global wood harvests. *Nature* 620, 110–115. <u>https://doi.org/10.1038/s41586-023-06187-1</u>
- Permanent Secretariat of the Alpine Convention (2017). Climate Chance – How it affects the Alps and what we can do. Available at: <u>https://www.alpconv.org/fileadmin/user\_upload/Publications/AlpineConventionFolder\_Climate\_Change\_2017\_EN\_ reverse.pdf</u>
- Pfeifer, D., Streicher, W., Fischer, D., and Mautner, P. (2016). GIS based bottom-up approach to evaluate the energy demand for the SINFONIA district Innsbruck (AT). Available at: <u>https:// data.europa.eu/doi/10.2790/290244</u> (Accessed September 25, 2023).
- Pichler, M., Krenmayr, N., Schneider, E., and Brand, U. (2021). EU industrial policy: Between modernization and transformation of the automotive industry. *Environ. Innov. Soc. Transit.* 38, 140–152. https://doi.org/10.1016/j.eist.2020.12.002
- Pierer, C., and Creutzig, F. (2019). Star-shaped cities alleviate tradeoff between climate change mitigation and adaptation. *Environ. Res. Lett.* 14, 085011. <u>https://doi.org/10.1088/1748-9326/ab2081</u>
- Plaum, F., Ahmadiahangar, R., Rosin, A., and Kilter, J. (2022). Aggregated demand-side energy flexibility: A comprehensive review on characterization, forecasting and market prospects. *Energy Rep.* 8, 9344–9362. <u>https://doi.org/10.1016/j.egyr.2022.</u> 07.038
- Pons, J. J., Villalba Sanchis, I., Insa Franco, R., and Yepes, V. (2020). Life cycle assessment of a railway tracks substructures: Comparison of ballast and ballastless rail tracks. *Environ. Impact Assess. Rev.* 85, 106444. https://doi.org/10.1016/j.eiar.2020.106444
- Potrč Obrecht, T., Jordan, S., Legat, A., and Passer, A. (2021). The role of electricity mix and production efficiency improvements on greenhouse gas (GHG) emissions of building components and future refurbishment measures. *Int. J. Life Cycle Assess.* 26, 839–851. https://doi.org/10.1007/s11367-021-01920-2
- Priessner, A., Sposato, R., and Hampl, N. (2018). Predictors of electric vehicle adoption: An analysis of potential electric vehicle drivers in Austria. *Energy Policy* 122, 701–714. <u>https://doi.org/10.1016/j.enpol.2018.07.058</u>
- Probst, N., Bach, P. M., Cook, L. M., Maurer, M., and Leitão, J. P. (2022). Blue Green Systems for urban heat mitigation: mechanisms, effectiveness and research directions. *Blue-Green Syst.* 4, 348–376. <u>https://doi.org/10.2166/bgs.2022.028</u>
- Qin, Y., Ghalambaz, S., Sheremet, M., Baro, M., and Ghalambaz, M. (2024). Deciphering Urban Heat Island Mitigation: A Comprehensive Analysis of Application Categories and Research Trends. Sustain. Cities Soc. 101, 105081. <u>https://doi.org/10.1016/ j.scs.2023.105081</u>

- Qiu, K., Ribberink, H., and Entchev, E. (2022). Economic feasibility of electrified highways for heavy-duty electric trucks. *Appl. Energy* 326, 119935. <u>https://doi.org/10.1016/j.apenergy.2022.</u> <u>119935</u>
- Reck, D. J., Martin, H., and Axhausen, K. W. (2022). Mode choice, substitution patterns and environmental impacts of shared and personal micro-mobility. *Transp. Res. Part Transp. Environ*. 102, 103134. <u>https://doi.org/10.1016/j.trd.2021.103134</u>
- Regulation (EU) 2018/842 (2018). Regulation (EU) 2018/842 of the European Parliament and of the Council of 30 May 2018 on binding annual greenhouse gas emission reductions by Member States from 2021 to 2030 contributing to climate action to meet commitments under the Paris Agreement and amending Regulation (EU) No 525/2013 (Text with EEA relevance). Available at: <u>http://data.europa.eu/eli/reg/2018/842/oj/eng</u> (Accessed September 26, 2023).
- Regulation (EU) 2019/1242 (2019). Regulation (EU) 2019/1242 of the European Parliament and of the Council of 20 June 2019 setting CO2 emission performance standards for new heavy-duty vehicles and amending Regulations (EC) No 595/2009 and (EU) 2018/956 of the European Parliament and of the Council and Council Directive 96/53/EC. Available at: <u>https://eur-lex.europa.eu/eli/reg/2019/1242/oj</u> (Accessed June 8, 2024).
- Regulation (EU) 2020/852 (2020). Regulation (EU) 2020/852 of the European Parliament and of the Council of 18 June 2020 on the establishment of a framework to facilitate sustainable investment, and amending Regulation (EU) 2019/2088 (Text with EEA relevance). Available at: <u>http://data.europa.eu/eli/reg/ 2020/852/oj</u>
- Regulation (EU) 2021/1119 (2021). Regulation (EU) 2021/1119 of the European Parliament and of the Council of 30 June 2021 establishing the framework for achieving climate neutrality and amending Regulations (EC) No 401/2009 and (EU) 2018/1999 ('European Climate Law'). Available at: <u>http://data.europa.eu/ eli/reg/2021/1119/oj</u> (Accessed September 6, 2023).
- Regulation (EU) 2023/851 (2023). Regulation (EU) 2023/851 of the European Parliament and of the Council of 19 April 2023 amending Regulation (EU) 2019/631 as regards strengthening the CO2 emission performance standards for new passenger cars and new light commercial vehicles in line with the Union's increased climate ambition (Text with EEA relevance). Available at: <u>http://data.europa.eu/eli/reg/2023/851/oj/eng</u> (Accessed September 19, 2023).
- Reynders, G., Nuytten, T., and Saelens, D. (2013). Potential of structural thermal mass for demand-side management in dwellings. *Build. Environ.* 64, 187–199. <u>https://doi.org/10.1016/j. buildenv.2013.03.010</u>
- Rippstein, V., De Schrijver, E., Eckert, S., and Vicedo-Cabrera, A. M. (2023). Trends in tropical nights and their effects on mortality in Switzerland across 50 years. *PLOS Clim.* 2, e0000162. <u>https://doi.org/10.1371/journal.pclm.0000162</u>
- Röck, M., Saade, M. R. M., Balouktsi, M., Rasmussen, F. N., Birgisdottir, H., Frischknecht, R., et al. (2020). Embodied GHG emissions of buildings – The hidden challenge for effective climate change mitigation. *Appl. Energy* 258, 114107. <u>https://doi. org/10.1016/j.apenergy.2019.114107</u>
- Rodgarkia-Dara, A., Gatzen, C., Sonnen, L., Lane, A., Zwickl-Bernhard, S., and Auer, H. (2023). Rolle der Gasinfrastruktur in einem klimaneutralen Österreich. Wien: BMK. Available at: <u>https://www.bmk.gv.at/themen/energie/publikationen/rollegasinfrastruktur.html</u> (Accessed September 22, 2023).

- Rosenow, J. (2022). Is heating homes with hydrogen all but a pipe dream? An evidence review. *Joule* 6, 2225–2228. <u>https://doi.org/10.1016/j.joule.2022.08.015</u>
- Rosenzweig, C., Solecki, W., Hammer, S. A., and Mehrotra, S. (2010). Cities lead the way in climate–change action. *Nature* 467, 909–911. <u>https://doi.org/10.1038/467909a</u>
- Roxon, J., Botshekan, M., Wanichkul, A., Ulm, F.-J., Nara, S., Anini, B., et al. (2019). Carbin App: A crowdsourced approach for monitoring quality of roads and their environmental impact. Available at: <u>https://pavement.engineering.asu.edu/wp-content/uploads/sites/16/2020/06/2019-Carbin-App.pdf</u> (Accessed February 1, 2025).
- Sadeghian, O., Oshnoei, A., Mohammadi-ivatloo, B., Vahidinasab, V., and Anvari-Moghaddam, A. (2022). A comprehensive review on electric vehicles smart charging: Solutions, strategies, technologies, and challenges. J. Energy Storage 54, 105241. https://doi.org/10.1016/j.est.2022.105241
- Saffari, M., Crownshaw, T., and McPherson, M. (2023). Assessing the potential of demand-side flexibility to improve the performance of electricity systems under high variable renewable energy penetration. *Energy* 272, 127133. <u>https://doi.org/10.1016/j. energy.2023.127133</u>
- Satterthwaite, D. (2008). Cities' contribution to global warming: notes on the allocation of greenhouse gas emissions. *Environ. Urban.* 20, 539–549. <u>https://doi.org/10.1177/0956247808096127</u>
- Schade, W., Krail, M., Hartwig, J., Walther, C., Sutter, D., Killer, M., et al. (2015). Cost of non-completion of the TEN-T. Final Report. Karlsruhe: Fraunhofer Institut für System und Innovationsforschung (ISI). Available at: <u>https://transport.ec.europa. eu/document/download/2db38e71-a588-4a46-9e52-61d587d-1b93e\_en?filename=2015-06-fraunhofer-cost-of-non-completion-of-the-ten-t.pdf&prefLang=et (Accessed February 1, 2025).</u>
- Schardinger, I., Vockner, B., Atzl, C., Mittlböck, M., Biberacher, M., Kinzl, H., et al. (2021). Enerspired Cities – offene harmonisierte Informationsgrundlagen für die energieorientierte Stadtplanung. Wien: BMK.
- Schienen-Control (2023). Regulator and arbitration board. Schienen-Control. Available at: <u>https://www.schienencontrol.gv.at/en/schienencontrol.html</u> (Accessed September 20, 2023).
- Schlögl, M., and Matulla, C. (2018). Potential future exposure of European land transport infrastructure to rainfall-induced landslides throughout the 21st century. *Nat. Hazards Earth Syst. Sci.* 18, 1121–1132. <u>https://doi.org/10.5194/nhess-18-1121-2018</u>
- Schönfelder, S., Sommer, M., Falk, R., Kratena, K., Molitor, R., Clees, L., et al. (2016). COSTS – Leistbarkeit von Mobilität in Österreich. Wien: WIFO. Available at: <u>https://fti-mobilitaetswende.at/de/publikationen/personenmobilitaet/projektberichte/costs.php</u> (Accessed November 7, 2024).
- Schöniger, F., Mascherbauer, P., Resch, G., Kranzl, L., and Haas, R. (2024). The potential of decentral heat pumps as flexibility option for decarbonised energy systems. *Energy Effic.* 17, 26. <u>https://doi.org/10.1007/s12053-024-10206-z</u>
- Schöniger, F., and Morawetz, U. B. (2022). What comes down must go up: Why fluctuating renewable energy does not necessarily increase electricity spot price variance in Europe. *Energy Econ.* 111, 106069. <u>https://doi.org/10.1016/j.eneco.2022.106069</u>
- Schöniger, F., Resch, G., Suna, D., Formayer, H., Pardo-Garcia, N., Hasengst, F., et al. (2023). Securing Austria's Electricity Supply in times of Climate Change; Final report of the ACRP12 project SECURES (2020–2023). Available at: <u>https://www.secures.</u>

at/fileadmin/cmc/Final\_Report\_SECURES.pdf (Accessed May 6, 2024).

- Schuitema, G., Steg, L., and Forward, S. (2010). Explaining differences in acceptability before and acceptance after the implementation of a congestion charge in Stockholm. *Transp. Res. Part Policy Pract.* 44, 99–109. <u>https://doi.org/10.1016/j.tra.</u> 2009.11.005
- Schuller, A., Flath, C. M., and Gottwalt, S. (2015). Quantifying load flexibility of electric vehicles for renewable energy integration. *Appl. Energy* 151, 335–344. <u>https://doi.org/10.1016/J.APENER-GY.2015.04.004</u>
- Schuster, S., Weninger, B., Rudlof, M., and Klinghofer, S. (2023). Verkehrsstatistik 2022. Wien: Statistik Austria. Available at: <u>https://www.statistik.at/fileadmin/user\_upload/Verkehr-2022-barr.pdf</u> (Accessed February 1, 2025).
- Sedlacek, N., Steinacher, I., Hartmann, G., Schrampf, J., Bachner, G., Duelli, S., et al. (2021). CLEARER – Climate Neutral Freight Transport – Studie – Endbericht Final. Wien: Klima- und Energiefonds vertreten durch die Österreichische Forschungsförderungsgesellschaft mbH (FFG). Available at: <u>https://www. econsult.at/images/Artikel/CLEARER-FInal-Report.pdf</u>
- SEP (2021). Spatial Energy Planning. Available at: <u>http://www.waermeplanung.at/</u> (Accessed November 19, 2024).
- Shibayama, T., Sandholzer, F., Laa, B., and Brezina, T. (2021). Impact of COVID-19 lockdown on commuting. *Eur. J. Transp. Infrastruct. Res.* 21, 70–93. <u>https://doi.org/10.18757/EJTIR.2021.</u> 21.1.5135
- Siefer, T., Schütze, C., and Strohbach, M. (2019). Untersuchung der Regelwerke für den Bahnbetrieb auf Schwachstellen hinsichtlich des zu erwartenden Klimawandels: Los 1: Klimawandel. Bonn: Eisenbahn-Bundesamt. Available at: <u>https://www.dzsf. bund.de/SharedDocs/Downloads/DZSF/Veroeffentlichungen/ Forschungsberichte/2018/EBA-Forschungsbericht\_2018-08a.</u> <u>pdf?\_blob=publicationFile&v=6</u> (Accessed September 11, 2022).
- Soimakallio, S., Böttcher, H., Niemi, J., Mosley, F., Turunen, S., Hennenberg, K. J., et al. (2022). Closing an open balance: The impact of increased tree harvest on forest carbon. *GCB Bioener*gy 14, 989–1000. <u>https://doi.org/10.1111/gcbb.12981</u>
- Sonnleithner, M., Hanus, C., Floegl, H., Ipser, C., Passawa, R., Schneider, B., et al. (2023). monumentum ad usum. Erhebung von Nutzungspotenzialen von baukulturellem Erbe in Niederösterreich für gemeinnützige Wohnbauträger. Krems: Donau-Universität Krems. Available at: <u>https://www.noe.gv.at/ noe/Wohnen-Leben/F-2259\_Endbericht.pdf</u> (Accessed November 18, 2024).
- Sovacool, B. K., Kim, J., and Yang, M. (2021). The hidden costs of energy and mobility: A global meta-analysis and research synthesis of electricity and transport externalities. *Energy Res. Soc. Sci.* 72, 101885. <u>https://doi.org/10.1016/j.erss.2020.101885</u>
- Speth, D., and Plötz, P. (2024). Depot slow charging is sufficient for most electric trucks in Germany. *Transp. Res. Part Transp. Environ.* 128, 104078. <u>https://doi.org/10.1016/j.trd.2024.104078</u>
- Stadt Graz (2018). Klimawandel Anpassung in Graz. Aktionsplan 2022. Graz: Stadt Graz Umweltamt. Available at: <u>https://www. umweltservice.graz.at/infos/andere/Klimawandelanpassung\_ Aktionsplan2022.pdf</u> (Accessed June 18, 2024).
- Stadt Wien (2022). Vienna Heat Action Plan. For a cool Vienna. Wien: Vienna City Administration. Available at: <u>https://www.digital.wienbibliothek.at/wbrup/download/pdf/4047970?originalFilename=true</u> (Accessed June 18, 2024).

- Stadt Wien (2024). Jährliche Hitze- und Eistage seit 1955 Stadt Wien. Available at: <u>https://www.data.gv.at/katalog/de/dataset/</u> <u>vie-bdl-biz-ecl-tmp-1955f</u> (Accessed November 18, 2024).
- Statistik Austria (2004). Gebäude- und Wohnungszählung 2001, Hauptergebnisse Tirol. Available at: <u>https://www.statistik.at/</u> <u>fileadmin/publications/Gebaeude-\_und\_Wohnungszaehlung\_</u> <u>2001\_\_Hauptergebnisse\_Tirol.pdf</u> (Accessed September 25, 2023).
- Statistik Austria (2011). Tabelle Mobilität der privaten Haushalte 2009/10 (.ods). Available at: <u>https://www.statistik.at/statistiken/ bevoelkerung-und-soziales/ausgaben-und-ausstattung-privater-haushalte/ausstattung</u> (Accessed November 7, 2024).
- Statistik Austria (2021). Census Austria Results of the Register-based Census. Available at: <u>https://www.statistik.at/services/tools/services/publikationen/detail/1789</u> (Accessed June 18, 2024).
- Statistik Austria (2022a). Bevölkerung zu Jahres-/ Quartalsanfang. Available at: <u>https://www.statistik.at/statistiken/bevoelker-ung-und-soziales/bevoelkerung/bevoelkerungsstand/bev-oelkerung-zu-jahres-/-quartalsanfang</u> (Accessed September 12, 2022).
- Statistik Austria (2022b). Tabelle Mobilität der privaten Haushalte 2019/20 (.ods). Available at: <u>https://www.statistik.at/statistiken/</u> <u>bevoelkerung-und-soziales/ausgaben-und-ausstattung-pri-</u> vater-haushalte/ausstattung (Accessed November 7, 2024).
- Statistik Austria (2022c). Wohnen 2021 Zahlen, Daten und Indikatoren der Wohnstatistik. Available at: <u>https://www.statistik.</u> <u>at/fileadmin/user\_upload/Wohnen-2021.pdf</u> (Accessed September 25, 2023).
- Statistik Austria (2023a). Dauersiedlungsraum. Available at: https://www.data.gv.at/katalog/dataset/5c74388f-200b-3deb-90 43-eeb1e7d7b53a#additional-info
- Statistik Austria (2023b). Tabelle: Kfz-Bestand 2023 (.ods). Available at: <u>https://www.statistik.at/statistiken/tourismus-und-verkehr/</u><u>fahrzeuge/kfz-bestand</u> (Accessed September 19, 2023).
- Statistik Austria (2023c). Wohnen 2022 Zahlen, Daten und Indikatoren der Wohnstatistik. Wien. Available at: <u>https://www. statistik.at/services/tools/services/publikationen/detail/1579</u> (Accessed August 28, 2024).
- Statistik Austria (2024a). Adress-GWR-Online (AGWR). Available at: <u>https://www.statistik.at/datenbanken/adress-gebaeude-und-wohnungsregister/adress-gebaeude-und-wohnungsregister/adress-gwr-online-agwr</u> (Accessed May 10, 2024).
- Statistik Austria (2024b). Bevölkerung zu Jahres-/Quartalsanfang. Available at: <u>https://www.statistik.at/statistiken/bevoelkerung-und-soziales/bevoelkerung/bevoelkerungsstand/bevoelkerung-zu-jahres-/-quartalsanfang</u> (Accessed November 18, 2024).
- Statistik Austria (2024c). Bevölkerungsprognosen für Österreich und die Bundesländer. Available at: <u>https://www.statistik.at/</u> <u>statistiken/bevoelkerung-und-soziales/bevoelkerung/demographische-prognosen/bevoelkerungsprognosen-fuer-oesterreich-und-die-bundeslaender</u> (Accessed September 12, 2022).
- Statistik Austria (2024d). STATcube Statistische Datenbank. Available at: <u>https://www.statistik.at/datenbanken/statcube-sta-tistische-datenbank</u> (Accessed May 10, 2024).
- Statistik Austria (2024e). Tabelle: Fahrzeug-Neuzulassungen Jänner bis Dezember 2023 (.ods). Available at: <u>https://www.statistik.at/statistiken/tourismus-und-verkehr/fahrzeuge/kfz-neuzulassungen</u> (Accessed November 7, 2024).

- Steinegger, W. (2023). *Upgrade suburbia: Graz* 50\_2050. Wien: Buchschmiede von Dataform Media GmbH.
- Steininger, K. W., Mayer, J., Bachner, G., Duelli, S., Frei, E., Grossmann, W., et al. (2021). The economic effects of achieving the 2030 EU climate targets in the context of the Corona crisis: An Austrian perspective. Graz: Wegener Center Verlag. <u>https://doi. org/10.13140/RG.2.2.20973.51685</u>
- Steininger, K. W., Riahi, K., Stagl, S., Kromp-Kolb, H., Kirchengast, G., Rosenfeld, D. C., et al. (2024). Nationaler Energie- und Klimaplan (NEKP) für Österreich – Wissenschaftliche Bewertung der in der Konsultation 2023 vorgeschlagenen Maßnahmen. Wien: Climate Change Centre Austria (CCCA). Available at: https://ccca.ac.at/fileadmin/00\_DokumenteHauptmenue/02\_ Klimawissen/RefNEKP/Bericht/NEKP\_Wissenschaftliche\_Bewertung\_der\_Massnahmen\_der\_Stellungnahmen\_Februar 2024.pdf (Accessed May 10, 2024).
- Stöglehner, G., and Abart-Heriszt, L. (2022). Integrated spatial and energy planning in Styria – A role model for local and regional energy transition and climate protection policies. *Renew. Sustain. Energy Rev.* 165, 112587. <u>https://doi.org/10.1016/j.</u> <u>rser.2022.112587</u>
- Stöglehner, G., Neugebauer, G., Erker, S., and Narodoslawsky, M. eds. (2016). Integrated Spatial and Energy Planning. Supporting Climate Protection and the Energy Turn with Means of Spatial Planning. Cham: Springer. <u>https://doi.org/10.1007/978-3-319-31870-7</u>
- Streicher, W., Beck, S., Hofmann, A., Schuchter, L., and Öttl, S. (2024a). Tiroler Kapazitäten für eine Energiewende im Sektor Gebäude. Standortagentur Tirol.
- Streicher, W., Schultz, J. M., Solé, C., Cabeza, L. F., Bony, J., Citherlet, S., et al. (2008). Final report of Subtask C "Phase Change Materials" The overview. Graz: Graz University of Technology Austria. Available at: <u>https://task32.iea-shc.org/Data/Sites/1/publications/task32-c7.pdf</u>
- Streicher, W., van Karsbergen, V., Thür, A., and Ochs, F. (2024b). Space Heating And Cooling Demand Of Buildings With Climate Change For A Cold And A Hot Climate – Is This A Risk?, in *EuroSun 2024*. Available at: <u>https://proceedings.ises.org/conference/eurosun2024/papers/eurosun2024-0008-Ochs.pdf</u>
- Strimitzer, L., Dolna-Gruber, C., and Baumann, M. (2022). gÖl. Mobil. Einsatz von grünen Treibstoffen in der Mobilität. Wien: BMK. Available at: <u>https://www.bmk.gv.at/dam/jcr:bb133bafef46-48c1-897b-ab003373d0f6/AEA\_Kurzbericht\_gOeLMobil\_UA.pdf</u> (Accessed May 10, 2024).
- Suna, D., Resch, G., Schöniger, F. B., Hasengst, F., Pardo-Garcia, N., Totschnig, G., et al. (2024). "Securing Austria's Electricity Supply in Times of Climate Change", in *Climate.Changes.Security : Navigating Climate Change and Security Challenges in the OSCE Region*, eds. H. Lampalzer and G. Hainzl (Landesverteidigungsakademie (LVAk) / Institut für Friedenssicherung und Konfliktmanagement (IFK)), 197–221. Available at: <u>http://hdl.handle.</u> <u>net/20.500.12708/197501</u> (Accessed November 18, 2024).
- Suna, D., Totschnig, G., Schöniger, F., Resch, G., Spreitzhofer, J., and Esterl, T. (2022). Assessment of flexibility needs and options for a 100% renewable electricity system by 2030 in Austria. *Smart Energy* 6, 100077. https://doi.org/10.1016/j.segy.2022.100077
- Tan, J., and Wang, L. (2017). A Game-Theoretic Framework for Vehicle-to-Grid Frequency Regulation Considering Smart Charging Mechanism. *IEEE Trans. Smart Grid* 8, 2358–2369. <u>https://doi.org/10.1109/TSG.2016.2524020</u>

- Tehrani, N. H., and Wang, P. (2015). Probabilistic estimation of plug-in electric vehicles charging load profile. *Electr. Power Syst. Res.* 124, 133–143. <u>https://doi.org/10.1016/j.epsr.2015.03.010</u>
- Thaller, A., Posch, A., Dugan, A., and Steininger, K. (2021). How to design policy packages for sustainable transport: Balancing disruptiveness and implementability. *Transp. Res. Part Transp. Environ.* 91, 102714. <u>https://doi.org/10.1016/j.trd.2021.102714</u>
- Thermos (2021). Thermos (EU Horizon 2020 funded). Available at: <u>https://www.thermos-project.eu/thermos-tool/tool-access/</u> (Accessed September 22, 2023).
- Thingvad, A., Ziras, C., and Marinelli, M. (2019). Economic value of electric vehicle reserve provision in the Nordic countries under driving requirements and charger losses. J. Energy Storage 21, 826–834. <u>https://doi.org/10.1016/J.EST.2018.12.018</u>
- Tong, S., Prior, J., McGregor, G., Shi, X., and Kinney, P. (2021). Urban heat: an increasing threat to global health. *BMJ* 375, n2467. <u>https://doi.org/10.1136/bmj.n2467</u>
- Tosatto, A., Ochs, F., Streicher, W., Magni, M., and Venturi, E. (2023). Methodology for the calculation of energy scenarios to achieve carbon neutrality in the building stock., in *Proceedings* of Building Simulation 2023: 18th Conference of IBPSA, 2727– 2734. <u>https://doi.org/10.26868/25222708.2023.1489</u>
- Toth, Z., Volt, J., and Steuwer, S. (2022). Roadmap to climate-proof buildings and construction – How to embed whole-life carbon in the EPBD. BPIE (Buildings Performance Institute Europe). Available at: <u>https://build-up.ec.europa.eu/en/resources-and-tools/publications/roadmap-climate-proof-buildingsand-construction-how-embed-whole</u> (Accessed September 26, 2023).
- Truger, B., Nabernegg, S., Lackner, T., Röck, M., Alaux, N., Hoxha, E., et al. (2022). Life cycle GHG emissions of the Austrian building stock: A combined bottom-up and top-down approach. *IOP Conf. Ser. Earth Environ. Sci.* 1078, 012024. <u>https:// doi.org/10.1088/1755-1315/1078/1/012024</u>
- Turner, W. J. N., Walker, I. S., and Roux, J. (2015). Peak load reductions: Electric load shifting with mechanical pre-cooling of residential buildings with low thermal mass. *Energy* 82, 1057–1067. <u>https://doi.org/10.1016/j.energy.2015.02.011</u>
- Umweltbundesamt (2019). GHG Projections and Assessment of Policies and Measures in Austria. Wien: Umweltbundesamt GmbH. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/publikationen/rep0687.pdf</u> (Accessed September 25, 2023).
- Umweltbundesamt (2020). Flächeninanspruchnahme. Available at: <u>https://www.umweltbundesamt.at/umweltthemen/boden/flae-cheninanspruchnahme</u> (Accessed September 20, 2023).
- Umweltbundesamt (2023a). Energie- und Treibhausgasszenarien 2023: WEM, WAM und Transition mit Zeitreihen von 2020 bis 2050. Wien: Umweltbundesamt GmbH. Available at: <u>https:// www.umweltbundesamt.at/fileadmin/site/publikationen/ rep0882.pdf</u>
- Umweltbundesamt (2023b). Klimaschutzbericht 2023. Wien: Umweltbundesamt GmbH. Available at: <u>https://www.um-weltbundesamt.at/studien-reports/publikationsdetail?pub\_id=2485&cHash=d40157f547cb06f523410f72beec62f9</u> (Accessed September 25, 2023).
- Umweltbundesamt (2024a). Emissionsfaktoren für Verkehrsmittel. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/</u> <u>themen/mobilitaet/daten/ekz\_pkm\_tkm\_verkehrsmittel.pdf</u> (Accessed October 6, 2024).

- Umweltbundesamt (2024b). Klimaschutzbericht 2024. Wien: Umweltbundesamt GmbH. Available at: <u>https://www.um-weltbundesamt.at/studien-reports/publikationsdetail?pub\_id=2531&cHash=4c252e1f79d1baad77e26f203d847bd4</u> (Accessed September 25, 2023).
- Umweltbundesamt (2024c). Nahzeitprognose der Österreichischen Treibhausgas-Emissionen für das Jahr 2023. Wien: Umweltbundesamt. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/publikationen/rep0918.pdf</u>
- UN DESA (2019). World Urbanization Prospects: The 2018 Revision. New York: United Nations Department of Economic and Social Affairs. <u>https://doi.org/10.18356/b9e995fe-en</u>
- UNEP (2021). Sustainable Public Procurement: How to Wake the Sleeping Giant – Introducing the United Nations Environment Programme's Approach. Paris: UNEP. Available at: <u>https://wedocs.unep.org/bitstream/handle/20.500.11822/37045/SPPWSG.</u> pdf
- United Nations Environment Programme and Yale Center for Ecosystems + Architecture (2023). Building Materials and the Climate: Constructing a New Future. Available at: <u>https://wedocs. unep.org/xmlui/handle/20.500.11822/43293</u> (Accessed May 10, 2024).
- Unterberger, C., Hudson, P., Wouter Botzen, W. J., Schroeer, K., and Steininger, K. W. (2019). Future Public Sector Flood Risk and Risk Sharing Arrangements: An Assessment for Austria. *Ecol. Econ.* 156, 153–163. <u>https://doi.org/10.1016/j.ecolecon.2018.09.019</u>
- Ürge-Vorsatz, D., Rosenzweig, C., Dawson, R. J., Sanchez Rodriguez, R., Bai, X., Barau, A. S., et al. (2018). Locking in positive climate responses in cities. *Nat. Clim. Change* 8, 174–177. <u>https://doi.org/10.1038/s41558-018-0100-6</u>
- VABU (2023). VABU Verband f
  ür Anschlussbahnunternehmen. Available at: <u>https://www.anschlussbahnen.at/</u> (Accessed September 20, 2023).
- Van Kriekinge, G., De Cauwer, C., Sapountzoglou, N., Coosemans, T., and Messagie, M. (2021). Peak shaving and cost minimization using model predictive control for uni- and bi-directional charging of electric vehicles. *Energy Rep.* 7, 8760–8771. <u>https:// doi.org/10.1016/J.EGYR.2021.11.207</u>
- VCÖ (2023a). VCÖ: Haushalte in Wien und Westösterreich fahren weniger mit dem Auto als der Österreich-Schnitt. Available at: <u>https://vcoe.at/presse/presseaussendungen/detail/vcoehaushalte-in-wien-und-westoesterreich-fahren-weniger-mitdem-auto-als-der-oesterreich-schnitt</u>
- VCÖ (2023b). VCÖ: SUV-Anteil bei Neuwagen so hoch wie noch nie – die meisten SUV in Wien zugelassen – Mobilität mit Zukunft. Available at: <u>https://vcoe.at/presse/presseaussendungen/detail/vcoe-suv-anteil-bei-neuwagen-so-hoch-wienoch-nie-die-meisten-suv-in-wien-zugelassen</u> (Accessed September 19, 2023).
- VCÖ (2024a). VCÖ: Klimaschädliche Emissionen des Flugverkehrs in Österreich im Vorjahr um 40 Prozent gestiegen. Available at: <u>https://vcoe.at/presse/presseaussendungen/detail/vcoe-klimaschaedliche-emissionen-des-flugverkehrs-in-oesterreichim-vorjahr-um-40-prozent-gestiegen</u> (Accessed May 10, 2024).
- VCÖ (2024b). Zahlen und Fakten zur Verkehrsinfrastruktur in Österreich. Available at: <u>https://vcoe.at/service/fragen-und-ant-</u><u>worten/zahlen-und-fakten-zur-verkehrsinfrastruktur-in-oes-</u><u>terreich</u> (Accessed May 10, 2024).
- Verzijlbergh, R., Martínez-Anido, C. B., Lukszo, Z., and Vries, L. de (2014). Does controlled electric vehicle charging substitute

cross-border transmission capacity? *Appl. Energy* 120, 169–180. https://doi.org/10.1016/J.APENERGY.2013.08.020

- Virág, D., Wiedenhofer, D., Baumgart, A., Matej, S., Krausmann, F., Min, J., et al. (2022a). How much infrastructure is required to support decent mobility for all? An exploratory assessment. *Ecol. Econ.* 200, 107511. <u>https://doi.org/10.1016/j.ecolecon.2022.107511</u>
- Virág, D., Wiedenhofer, D., Haas, W., Haberl, H., Kalt, G., and Krausmann, F. (2022b). The stock-flow-service nexus of personal mobility in an urban context: Vienna, Austria. *Environ. Dev.* 41, 100628. <u>https://doi.org/10.1016/j.envdev.2021.100628</u>
- WBGU Wissenschaftlicher Beirat der Bundesregierung Globale Umweltveränderungen (2011). World in Transition. A Social Contract for Sustainability. Flagship Report. Berlin: WBGU. Available at: <u>https://www.wbgu.de/en/publications/publication/world-in-transition-a-social-contract-for-sustainability</u> (Accessed September 22, 2023).
- WBGU Wissenschaftlicher Beirat der Bundesregierung Globale Umweltveränderungen (2016). Humanity on the move: Unlocking the transformative power of cities. WBGU Flagship Report. Berlin: WBGU. Available at: <u>https://www.wbgu.de/ en/publications/publication/humanity-on-the-move-unlocking-the-transformative-power-of-cities</u> (Accessed September 22, 2023).
- Weatherpark GmbH (2021). Stadtklimaanalyse Wien. Klimatologie und Analysekarte Stadtklima. im Auftrag der MA18 der Stadt Wien. Wien. Available at: <u>https://www.digital.wienbibliothek.at/wbrup/download/pdf/4851346</u> (Accessed June 18, 2024).
- Weichselbaumer, R., Damyanovic, D., and Reinwald, F. (2022). The Contribution of Public Spaces to Climate Change Adaptation in Austrian Cities., in Mobility, Knowledge and Innovation Hubs in Urban and Regional Development. Proceedings of REAL CORP 2022, 27th International Conference on Urban Development, Regional Planning and Information Society, eds. M. Schrenk, V. V. Popovich, P. Zeile, P. Elisei, C. Beyer, and J. Ryser (Wien: CORP – Competence Center of Urban and Regional Planning), 767– 780. Available at: https://www.corp.at/fileadmin/proceedings/ CORP2022\_proceedings.pdf (Accessed May 10, 2024).
- Weiß, T., Fulterer, A. M., and Knotzer, A. (2019). Energy flexibility of domestic thermal loads – a building typology approach of the residential building stock in Austria. *Adv. Build. Energy Res.* 13, 122–137. https://doi.org/10.1080/17512549.2017.1420606
- Weyerstraß, K. (2021). Fiscal policies during the Covid-19 crisis in Austria: A macroeconomic assessment. Munich: ifo Institute – Leibniz Institute for Economic Research at the University of Munich. Available at: <u>https://hdl.handle.net/10419/272360</u>
- Wicki, M., Brückmann, G., and Bernauer, T. (2022). How to accelerate the uptake of electric cars? Insights from a choice experiment. J. Clean. Prod. 355, 131774. <u>https://doi.org/10.1016/j.jclepro.2022.131774</u>
- Wiener Linien (2023). Rückblick & Ausblick: mehr Stammkund\*innen als je zuvor, Wiener Linien als sichere und nachhaltige Arbeitgeberin, Öffi-Ausbau geht weiter. Wien. Linien. Available at: <u>http://www.wienerlinien.at/news/rueckblick-2022-ausblick-2023</u> (Accessed September 19, 2023).
- WIFO (2024). Hochwasser verursachte 1,3 Mrd. € Schaden. Available at: <u>https://www.wifo.ac.at/news/hochwasser-verur-</u>sachte-13-mrd-e-schaden/ (Accessed November 10, 2024).
- Wilbanks, T., Bhatt, V., Bilello, D., Bull, S., Ekmann, J., Horak, W., et al. (2008). Effects of Climate Change on Energy Production

and Use in the United States. US Dep. Energy Publ. Available at: https://digitalcommons.unl.edu/usdoepub/12

- WKO Wien (2022). KEP Branchenreport 2022 Wien. Wirschaftskammer Wien. Available at: <u>https://www.wko.at/wien/verkehrbetriebsstandort/kep-branchenreport-2022.pdf</u> (Accessed February 1, 2025).
- Wolisz, H., Blacha, T., Jahangiri, P., Wesseling, M., and Müller, D. (2016). The inherent thermal storage capacity of buildings – potential for load shifting with electricity powered heating systems., in *INNOSTORAGE Conference* 16.02.2016–18.02.2016, *Beer-Sheva, Israel*, 16 Feb 201618 Feb 2016; Aachen (2016). https://doi.org/10.18154/RWTH-2016-03659
- Zach, F. (2022). Unser Wasser macht gutes Klima. Investitionen der Siedlungswasserwirtschaft in die erneuerbare Energieversorgung. Wien: Bundesministerium für Land-und Forstwirtschaft, Regionen und Wasserwirtschaft. Available at: <u>https://info.bml.gv.at/service/publikationen/wasser/unser-wasser-macht-gutes-klima.html</u> (Accessed September 22, 2023).
- Zeiner, M., Landgraf, M., Knabl, D., Antony, B., Barrena Cárdenas, V., and Koczwara, C. (2021). Assessment and Recommendations for a Fossil Free Future for Track Work Machinery. Sustainability 13, 11444. <u>https://doi.org/10.3390/su132011444</u>
- Zekar, A., Milojevic-Dupont, N., Zumwald, M., Wagner, F., and Creutzig, F. (2023). Urban form features determine spatio-temporal variation of ambient temperature: A comparative study of three European cities. *Urban Clim.* 49, 101467. <u>https://doi. org/10.1016/j.uclim.2023.101467</u>
- Zhang, C., Tan, Y., Gao, Y., Fu, Y., Li, J., Li, S., et al. (2022). Resilience assessment of asphalt pavement rutting under climate change. *Transp. Res. Part Transp. Environ.* 109, 103395. <u>https:// doi.org/10.1016/j.trd.2022.103395</u>
- Zhang, K., Xu, L., Ouyang, M., Wang, H., Lu, L., Li, J., et al. (2014a). Optimal decentralized valley-filling charging strategy for electric vehicles. *Energy Convers. Manag.* 78, 537–550. <u>https://doi.org/10.1016/j.enconman.2013.11.011</u>

AAR2

- Zhang, L., Jabbari, F., Brown, T., and Samuelsen, S. (2014b). Coordinating plug-in electric vehicle charging with electric grid: Valley filling and target load following. *J. Power Sources* 267, 584–597. https://doi.org/10.1016/J.JPOWSOUR.2014.04.078
- Zheng, Y., Shao, Z., and Jian, L. (2021). The peak load shaving assessment of developing a user-oriented vehicle-to-grid scheme with multiple operation modes: The case study of Shenzhen, China. Sustain. Cities Soc. 67, 102744. <u>https://doi.org/10.1016/J. SCS.2021.102744</u>
- Ziemele, J., Gendelis, S., and Dace, E. (2023). Impact of global warming and building renovation on the heat demand and district heating capacity: Case of the city of Riga. *Energy* 276, 127567. <u>https://doi.org/10.1016/j.energy.2023.127567</u>
- Zimmermann, P. (2018). Bewertbarkeit und ökobilanzieller Einfluss von Suffizienz im Gebäudebereich – Entwicklung einer Suffizienz-Bewertungsmethodik und Bestimmung des Einflusses von Suffizienz auf die Ökobilanz von Wohngebäuden. München: Technischen Universität München. Available at: <u>https://mediatum.ub.tum.de/doc/1546825/8lffu6uyh751qfb</u> <u>1g4l02304b.MA\_Suffizienz\_Zimmermann\_Ausarbeitung.pdf</u> (Accessed May 10, 2024).
- Zumwald, M., Baumberger, C., Bresch, D. N., and Knutti, R. (2021). Assessing the representational accuracy of data-driven models: The case of the effect of urban green infrastructure on temperature. *Environ. Model. Softw.* 141, 105048. <u>https://doi. org/10.1016/j.envsoft.2021.105048</u>
- Zwickl-Bernhard, S., and Auer, H. (2022). Demystifying natural gas distribution grid decommissioning: An open-source approach to local deep decarbonization of urban neighborhoods. *Energy* 238, 121805. https://doi.org/10.1016/j.energy.2021.121805





Chapter 4

# Provision of goods and services in a climate-resilient economy via materials, energy and work

# Second Austrian Assessment Report on Climate Change | AAR2

# Chapter 4 Provision of goods and services in a climate-resilient economy via materials, energy and work

This publication was subject to an anonymous international peer review process guided by several review editors, the AAR2 co-chairs and the APCC Steering Committee.

The views or opinions expressed in this publication are not necessarily those of the institutions with whose support this work was carried out. The publisher and the supporting institutions do not guarantee the accuracy or continued availability of urls to external websites or third-party websites mentioned in this publication and accept no responsibility for the accuracy or appropriateness of any content contained therein.

#### **Recommended citation**

T. Schinko, B. Smetschka, K. Steininger, et al. (2025): Provision of goods and services in a climate-resilient economy via materials, energy and work. In "Second Austrian Assessment Report on Climate Change (AAR2) of the Austrian Panel on Climate Change (APCC)". [D. Huppmann, M. Keiler, K. Riahi, H. Rieder (eds.)]. Austrian Academy of Sciences Press, Vienna, Austria | doi: doi.org/10.1553/aar2-ch4

You can download the recommended citation in the formats Endnote (.ris) and BibTex (.bib) at <u>aar2.ccca.ac.at/citation</u>.

Vienna, June 2025 ISBN: 978-3-7001-5110-4 | austriaca.at/5110-4 (full report)

#### Production

Publisher: Austrian Academy of Sciences Press, Vienna Typesetting: Crossdesign Werbeagentur GmbH, Graz Cover: paulaidaanton | Agentur für digiloges Design Graphic design Figure 4.1, CCBox 6 Figure 1: paulaidaanton | Agentur für digiloges Design Al-based tools such as DeepL were used to support the linguistic revision.

Contact address for product safety: Austrian Academy of Sciences Press Dr. Ignaz Seipel-Platz 2 A-1010 Wien Tel.: +43-1-51581-3420 E-Mail: verlag@oeaw.ac.at

#### Funding acknowledgement

The AAR2 was funded by the Climate and Energy Fund under the 14th Austrian Climate Research Program (ACRP).

#### **Copyright and license**

© 2025 Austrian Panel on Climate Change (APCC) This report is published under a Creative Commons CC-BY 4.0-License. <u>creativecommons.org/licenses/by/4.0</u>





The AAR2 is a report by the Austrian Panel on Climate Change (APCC). https://aar2.ccca.ac.at/en

#### **Coordinating Lead Authors**

Thomas Schinko Barbara Smetschka Karl Steininger

#### **Lead Authors**

Ernest Aigner Gabriel Bachner Katharina Bohnenberger Anna Burton Oliver Fritz Willi Haas Holger Ott Johannes Schmidt Franziska Schöniger Harald Wieser Caroline Zimm

#### **Contributing Authors**

André Baumgart Benigna Boza-Kiss Alina Brad Simone Gingrich Markus Lehner Ina Meyer Andreas Novy Tajda Obrecht Jonas Peisker Jana Petermann Harald Rieder Anna Viktoria Rohrer Nicolas Roux Christian Schützenhofer Dominik Steinberger-Maierhofer Harald Vacik

#### **Chapter Scientists**

Tobias Eibinger Anna Viktoria Rohrer University of Graz University of Graz

#### **Review Editors**

Reinhard Haas Kathy Reimann Technical University Vienna | TU Wien Agora Industry

#### **Liaisons from AAR2 Technical Support Unit**

Karl Scheifinger	International Institute for Applied Systems Analysis   IIASA
Jan Steinhauser	International Institute for Applied Systems Analysis   IIASA

International Institute for Applied Systems Analysis | IIASA BOKU University University of Graz

Leuphana Universität Lüneburg University of Graz University of Duisburg-Essen Austrian Institute of Economic Research | WIFO Austrian Institute of Economic Research | WIFO BOKU University Montanuniversität Leoben BOKU University BOKU University & Austrian Institute of Technology GmbH | AIT Austrian Institute for SME Research International Institute for Applied Systems Analysis | IIASA

**BOKU University** International Institute for Applied Systems Analysis | IIASA University of Vienna **BOKU University** Montanuniversität Leoben Austrian Institute of Economic Research | WIFO Vienna University of Economics and Business | WU Graz University of Technology | TU Graz International Institute for Applied Systems Analysis | IIASA Paris Lodron University of Salzburg | PLUS **BOKU University** University of Graz **BOKU University** Austrian Institute of Technology GmbH | AIT Graz University of Technology | TU Graz **BOKU University** 

# **TABLE OF CONTENTS**

cutive s	summary	235
Chapt	er introduction	239
4.1.1.	The provisioning system	239
4.1.2.	Sectoral emissions	240
Provis	ioning of goods and services	241
4.2.1.	Status quo in the provisioning of goods and services	241
4.2.2.	Future systems for the provisioning of goods and services using circular economy principles	243
Cross-	Chapter Box 3	
Needs	orientation in provisioning systems	248
Cross-	Chapter Box 4	240
Avoia	- Shirt - Improve: The ASI-framework	249
Circul	ar economy	252
4.2.3.	Necessary transformation in the	
	provisioning of goods and services	254
The ca	ase of tourism and climate change	256
4.3.1.	Overview of the Austrian tourism sector .	256
4.3.2.	The contribution of tourism to GHG emissions	256
4.3.3.	Future systems: Impact of climate change on tourism	257
4.3.4.	Needed transitions in the Austrian	
	tourism sector	258
	Avoid	258
	Shift	258
	Improve	258
Indust	try and manufacturing	259
4.4.1.	Status quo	259
4.4.2.	Current and future systems	259
	Steel	259
	Cement	260
	Chemicals	261
	Pulp and paper	261
	Cutive : Chapt 4.1.1. 4.1.2. Provis 4.2.1. 4.2.2. Cross- Needs Cross- Avoid Cross- Cross- Avoid Cross- Circul 4.2.3. The ca 4.3.1. 4.3.2. 4.3.3. 4.3.4. 4.3.4.	Cuive summary         Chapter introduction         4.1.1. The provisioning system         4.1.2. Sectoral emissions         Provisioning of goods and services         4.2.1. Status quo in the provisioning of goods and services using circular economy principles         Cross-Chapter Box 3           Needs orientation in provisioning systems         Cross-Chapter Box 4           Avoid – Shift – Improve: The ASI-framework         Cross-Chapter Box 5           Circular economy         4.2.3. Necessary transformation in the provisioning of goods and services         The case of tourism and climate change         4.3.1. Overview of the Austrian tourism sector         4.3.2. The contribution of tourism to GHG emissions.         4.3.3. Future systems: Impact of climate change on tourism         4.3.4. Needed transitions in the Austrian tourism sector         A.3.4. Needed transitions in the Austrian tourism sector         Avoid         Shift         Improve         Industry and manufacturing         4.4.1. Status quo         4.4.2. Current and future systems         Steel       Cement         Chemicals       Pulp and paper

	Chapt	ter Box 4.1	
	Integ	rated scenarios for industry	
	decar	bonization in Austria	262
		Other manufacturing sectors	264
		Economy-wide effects	264
	4.4.3.	Climate related risks and adaptation of industry and manufacturing	264
	4.4.4.	Necessary transformation of industry and manufacturing	264
4.5.	Energ	y supply systems	265
	4.5.1.	Characteristics of the current energy system/status quo	265
	4.5.2.	Characteristics of the future energy system	267
		Energy supply security and resilience of systems and infrastructure	269
		Climate change impacts on energy systems	270
	4.5.3.	The necessary transformation: Achieving	272
		a decarbonized energy system	275
4.6.	Decar by car	bonization of hard-to-abate industries rbon capture, geological storage, and	275
4.6.	Decar by cai utiliza	bonization of hard-to-abate industries rbon capture, geological storage, and ation	273
4.6.	Decar by car utiliza Cross Carbo	bonization of hard-to-abate industries rbon capture, geological storage, and ation -Chapter Box 6   on dioxide removal	273 274 275
4.6.	Decar by car utiliza Cross- Carbo 4.6.1.	bonization of hard-to-abate industries rbon capture, geological storage, and ation -Chapter Box 6   on dioxide removal Current status of CCUS technologies	<ul> <li>273</li> <li>274</li> <li>275</li> <li>278</li> </ul>
4.6.	Decar by car utiliza Cross- Carbo 4.6.1. 4.6.2.	bonization of hard-to-abate industries rbon capture, geological storage, and ation -Chapter Box 6   on dioxide removal Current status of CCUS technologies CCUS potential in and for Austria	<ul> <li>273</li> <li>274</li> <li>275</li> <li>278</li> <li>279</li> </ul>
4.6.	Decar by car utiliza Cross Carbo 4.6.1. 4.6.2. 4.6.3.	bonization of hard-to-abate industries rbon capture, geological storage, and ation -Chapter Box 6   on dioxide removal Current status of CCUS technologies CCUS potential in and for Austria Necessary transformations and R&D	<ul> <li>273</li> <li>274</li> <li>275</li> <li>278</li> <li>279</li> <li>281</li> </ul>
4.6.	Decar by car utiliza Cross Carbo 4.6.1. 4.6.2. 4.6.3. Provis	bonization of hard-to-abate industries rbon capture, geological storage, and ation -Chapter Box 6   on dioxide removal Current status of CCUS technologies CCUS potential in and for Austria Necessary transformations and R&D	<ul> <li>274</li> <li>275</li> <li>278</li> <li>279</li> <li>281</li> <li>281</li> </ul>
4.6. 4.7.	Decar by car utiliza Cross- Carbo 4.6.1. 4.6.2. 4.6.3. Provis 4.7.1.	bonization of hard-to-abate industries rbon capture, geological storage, and ation -Chapter Box 6   on dioxide removal Current status of CCUS technologies CCUS potential in and for Austria Necessary transformations and R&D  Sion by work Work and climate change	<ul> <li>273</li> <li>274</li> <li>275</li> <li>278</li> <li>279</li> <li>281</li> <li>281</li> </ul>
4.6.	Decar by car utiliza Cross- Carbo 4.6.1. 4.6.2. 4.6.3. Provis 4.7.1. 4.7.2.	bonization of hard-to-abate industries rbon capture, geological storage, and ation -Chapter Box 6   on dioxide removal Current status of CCUS technologies CCUS potential in and for Austria Necessary transformations and R&D  Sion by work Work and climate change Work in a climate-friendly future perspective.	<ul> <li>274</li> <li>275</li> <li>278</li> <li>279</li> <li>281</li> <li>281</li> <li>283</li> </ul>
4.6.	Decar by car utiliza Cross- Carbo 4.6.1. 4.6.2. 4.6.3. Provis 4.7.1. 4.7.2. 4.7.3.	bonization of hard-to-abate industries rbon capture, geological storage, and ation -Chapter Box 6   on dioxide removal Current status of CCUS technologies CCUS potential in and for Austria Necessary transformations and R&D Necessary transformations and R&D Work and climate change Work in a climate-friendly future perspective Transformation towards climate-friendly work and green employment	<ul> <li>274</li> <li>275</li> <li>278</li> <li>279</li> <li>281</li> <li>281</li> <li>283</li> <li>284</li> </ul>
4.6. 4.7.	Decar by car utiliza Cross Carbo 4.6.1. 4.6.2. 4.6.3. Provis 4.7.1. 4.7.2. 4.7.3.	bonization of hard-to-abate industries rbon capture, geological storage, and ation -Chapter Box 6   on dioxide removal Current status of CCUS technologies CCUS potential in and for Austria Necessary transformations and R&D Necessary transformations and R&D Work and climate change Work in a climate friendly future perspective Transformation towards climate-friendly work and green employment	<ul> <li>274</li> <li>275</li> <li>278</li> <li>279</li> <li>281</li> <li>281</li> <li>283</li> <li>284</li> <li>286</li> </ul>
4.6. 4.7. Refe	Decar by car utiliza Cross Carbo 4.6.1. 4.6.2. 4.6.3. Provis 4.7.1. 4.7.2. 4.7.3.	bonization of hard-to-abate industries rbon capture, geological storage, and ation -Chapter Box 6   on dioxide removal Current status of CCUS technologies CCUS potential in and for Austria Necessary transformations and R&D Necessary transformations and R&D Work and climate change Work in a climate-friendly future perspective Transformation towards climate-friendly work and green employment	<ul> <li>274</li> <li>275</li> <li>278</li> <li>279</li> <li>281</li> <li>281</li> <li>283</li> <li>284</li> <li>286</li> <li>SSSATV</li> </ul>
4.6. 4.7. Refe	Decar by car utiliza Cross Carbo 4.6.1. 4.6.2. 4.6.3. Provis 4.7.1. 4.7.2. 4.7.3.	bonization of hard-to-abate industries rbon capture, geological storage, and ation -Chapter Box 6   on dioxide removal Current status of CCUS technologies CCUS potential in and for Austria Necessary transformations and R&D Necessary transformations and R&D work and climate change Work and climate change Work in a climate-friendly future perspective Transformation towards climate-friendly work and green employments <u>aar2.ccca.ac.at/gloc</u> reviationsaar2.ccca.ac.at/abbrevia	2774 2775 278 279 281 281 283 284 283 284 286 555ary
4.6. 4.7. Refe Glos List App	Decar by car utiliza Cross Carbo 4.6.1. 4.6.2. 4.6.3. Provis 4.7.1. 4.7.2. 4.7.3. erences sary of abbreendix	bonization of hard-to-abate industries rbon capture, geological storage, and ation -Chapter Box 6   on dioxide removal Current status of CCUS technologies CCUS potential in and for Austria Necessary transformations and R&D Necessary transformations and R&D Work and climate change Work in a climate friendly future perspective Transformation towards climate-friendly work and green employment s <u>aar2.ccca.ac.at/abbrevia</u> <u>aar2.ccca.ac.at/abbrevia</u>	273 274 275 278 279 281 281 281 283 284 284 286 ssary tions dix/4

#### **EXECUTIVE SUMMARY**

Several mitigation pathways are available after 2030 to reach net-zero CO<sub>2</sub>eq emissions in 2040 (medium evidence, high agreement). First, it is evident that the infrastructure necessary for energy storage, grids, and public transport will differ from the current infrastructure. This implies the need for swift and determined decisions today so that the necessary infrastructure is available in time, given the considerable construction time for energy and rail infrastructure. Second, the discrepancy in energy service demand, industrial output, and energy efficiency measures causes a large range of primary energy use (237-337 TWh) across the scenarios, underscoring the significance of measures reducing demand. Third, energy import dependency varies considerably (5-39 %). Fourth, wind power and solar PV constitute the backbone of the power sector in all scenarios, satisfying demand, but the respective shares differ (wind: 5-25 %; PV: 6-20 % of gross domestic consumption). Fifth, biomass remains a highly important resource in the energy sector, but to varying degrees (ranging from 14-28 % of gross domestic consumption).  $\{4.4, 4.5\}$ 

Unabated fossil liquid fuels and gases are partly substituted by varying amounts of biogenic fuels, synthetic fuels and fossil fuels with CCS in 2040 in the scenarios (medium evidence, high agreement). While electrification is the most cost-effective mitigation technology in many applications, there is still a substantial need for carbon-neutral fuels and gases in 2040. Their use ranges from to 91–156 TWh in 2040. These fuels are mainly used in industry (35-59 % of all liquid fuels, and gas, and solid use), in transport (8-33 %), in households, services and agriculture (20-34 %) and in power generation and district heating (4–21%). Legacy effects cause the continued use of fossil fuels or the build-up of synthetic fuel use in land transport and heating. Due to delayed action in these sectors, a significant stock of assets based on liquids or gases may remain, causing the costly and inefficient use of these fuels in sectors where electrification could provide emission reductions at lower cost. As an example, a high proportion of liquid fuels in the transport sector will only be necessary if the electrification of land transport remains at its current low level. Sensitivity scenarios show that the phaseout of internal combustion engines in land transport is crucial if climate neutrality is to be achieved quickly. Efficient scenarios show that liquid fuels are only used for aviation and shipping, and that, after overcoming the legacy of the past, no liquid fuels or gases are used for heating. {4.4, 4.5}

The Austrian economy is already at risk due to climate change related impacts. Future climate change as well as socioeconomic development will further intensify these impacts (*limited evidence, high agreement*). International studies provide quantitative climate change impact assessments at national and supranational levels. Subnational and sector-specific assessments are rare, but needed to broaden the evidence base for potential climate change-related impacts in Austria, focusing on the Austrian energy generation and transmission infrastructure, services and industrial sectors. {4.3.3, 4.5.2}

Current provisioning of goods and services is a key driver for high levels of energy and material demand and thus domestic and upstream greenhouse gas emissions (*high confidence*). The Austrian economy rests on growing infrastructure and building stocks to provide services in transport, housing, communication, and finance (*high confidence*). The continued accumulation of infrastructure and buildings and the increased service and activity levels have largely offset technological efficiency gains and inhibited structural shifts towards a service economy (*high confidence*). {4.2.1}

A needs-oriented circular economy that focuses on reducing energy and material use through optimized space and building use, improved spatial planning that halts urban sprawl, shifting motorized private transport to public and active modes of transport, sharing, eco-design, increasing durability, reuse, and repair provides multiple benefits (*high confidence*). Shifting from material-intensive production and construction processes to services (e.g., rental, sharing) has the potential to avoid rebound effects and increase value-added and employment (*medium confidence*). The provisioning of goods and services thereby can become significantly more climate-friendly and -resilient while maintaining high levels of service quality and well-being (*high confidence*). {4.2.2}

Achieving a net-zero-emissions and resilient provisioning system requires changes at multiple levels: End-use technologies, the infrastructures that underpin everyday life, and in the local provisioning systems that shape people's modes of living (*high confidence*). Important decarbonization technologies such as electric vehicles and photovoltaics increase demand for electricity and critical raw materials, which need to be compensated by energy and material efficiency measures that reduce both the volume of material stocks in use and their energy requirements (*high confidence*). This aids decarbonization across all sectors, reduces dependencies on critical raw materials, reduces land consumption, and aligns with the government's circular economy targets (*medium evidence, high agreement*). {4.2.3}

Tourism is of high economic significance for Austria and is associated with substantial emissions (*high confidence*). With about 4.15 % of  $CO_2$  emissions in Austria originating from tourism, and up to 34.1 % when transport-related emissions between origin and destination of foreign and domestic visitors to Austria are included, the sector is a relevant driver of emissions (*medium confidence*). Due to the economic significance of tourism in Austria (7.5 % of GDP in 2019), tourism is an important sector to be considered on the path to a carbon-neutral economy (*high confidence*).  $\{4.3.2\}$ 

The impacts of climate change will substantially alter some of the supply and demand in the tourism sector (*high confidence*). Changing overall climate conditions (e.g., increase in average temperatures, shift in precipitation, shorter snow cover duration) and increasing extreme weather events (e.g., number of heat days, precipitation intensity, storm frequencies/intensity) will change the supply conditions for tourism services and thus require an adaptation of tourism (*high confidence*). The adjustment will need to take place along several dimensions and include seasonal adjustment of supply and demand, changes in the type of products offered and a reduction in the frequency of travel combined with longer stays (*high confidence*). {4.3.3}

A modal shift to public transport and the greening of the destination transport systems are key in mitigating emissions in tourism (*high confidence*). The largest share (75 %) of global total tourism-induced greenhouse gas (GHG) emissions is generated by transport to, from and at the destination (mostly by air and road) (*high confidence*). In Austria, too, the single most important source of GHG emissions related to tourism is personal transport, which makes up 92.2 % of the sector's GHG emissions (*medium confidence*). Consequently, to reduce tourism-related GHG emissions, a higher share of transport to and from as well as at the destination must be carried out by trains and busses and, as public transport capacities are limited, in particular in sub-urban and rural regions, by low (zero-) emission private vehicles (*high confidence*). {4.3.4}

The industry and manufacturing sector is particularly emission intensive, contributing 18 % to GDP and 35 % (26 MtCO<sub>2</sub>eq) to emissions in 2020 (*high confidence*). Emissions are highly concentrated in the iron and steel, chemicals, pulp and paper, and non-metallic minerals industries, which account for 82 % (21 Mt) of the sector's and 29 % of national emissions. From 2014–2020 emissions of this sector fell by 1 % per year, while value added has grown by a rate of 2 % per year (*high confidence*). {4.4.1}

Many technological options are already available to support the defossilization of Austria's industry and manufacturing sector, but key technologies such as hydrogen-based steelmaking and the integration of CCUS require further research and development. Future systems will heavily rely on electricity; thus, the availability of low-carbon electricity is key to prevent outsourcing emissions to the energy sector (*high confidence*). If current production levels are maintained, electricity demand for the direct use in emission-intensive industries will more than double (*high confidence*). {4.4.2}

The defossilization of industry and manufacturing requires an integrated and coordinated economy-wide strategy, as individual incremental improvements are insufficient (*high confidence*). In addition to sufficiency measures (avoid), the key levers are fuel switching (shift), energy/ material efficiency and carbon capture (improve). These changes involve new investments and will generate major new challenges, such as securing sufficient renewable energy supply and restructuring global value chains (*medium confidence*). {4.4.4}

The Austrian energy system is mainly based on fossil fuels and is a large emitter of greenhouse gas emissions (*high confidence*). The Austrian energy system (serving various economic sectors) was responsible for 70 % (44.5 MtCO<sub>2</sub>eq) of all greenhouse gas emissions in Austria in 2023 and fossil fuels had a share of 62 % in energy supply. A major share of fossil fuels is imported (96 % in 2023), implying a high import dependency in terms of primary energy. {4.5.1}

A carbon-neutral and climate-resilient energy system is electrified, hosts high shares of renewables, and is prepared for climate change impacts to maintain system security (*medium evidence, high agreement*). A future carbon-neutral energy system will show significantly higher shares of electricity in end use and larger shares of renewables (*medium evidence, high agreement*), but many different technology mixes allow carbon neutrality to be achieved, e.g., focusing on wind power or on solar PV, or allowing for higher shares of carbon capture and storage versus using green hydrogen (*medium evidence, high agreement*). Systems with high shares of wind and/or solar PV will have a higher degree of exposure and vulnerability to weather-related variability and hazards (*limited evidence, medium agreement*). Therefore, the design of energy systems needs to consider long-term climate trends in addition to weather variability, exposure, and vulnerability to reduce future risks (*limited evidence, medium agreement*). {4.5.2}

The net costs of transitioning to a carbon-neutral energy system are low to negative (limited evidence, medium agreement), most of the necessary technologies are available, and there are in principle sufficient land resources for most pathways to successfully transition to a carbon-neutral energy system (medium confidence). However, major barriers in terms of infrastructure development, including flexibility options to balance intermittent renewable supply and coupling energy-demanding sectors, policy design, technology acceptance and supply-chain bottlenecks, need to be overcome to implement a carbon-neutral energy system (medium evidence, high agreement). In all scenarios, import dependency on energy carriers is reduced, but reductions vary and depend on assumptions about the development of energy demand, in particular in the industrial sector, on assumptions about domestic renewable energy potential and on the expansion of fossil fuels with carbon capture and storage. {4.5.3}

Carbon Capture, Utilization, and Storage (CCUS) technology chains are mature and can be employed to bind  $CO_2$ and thus prevent the release of emissions into the atmosphere (*high confidence*). CCUS refers to technology chains for reducing  $CO_2$  emissions on a relevant scale through geological storage underground or through utilization in industrial processes and products. While geological  $CO_2$  storage (CCS) is a large-scale and permanent-reduction option,  $CO_2$ utilization (CCU) is limited by demand and final product lifetime. {4.6.1}

During the transition to a climate-neutral society, especially regarding hard-to-abate industrial emissions, CCUS will play a crucial role in decarbonizing the Austrian economy (*high confidence*). CCUS can enable emission-critical industries in Austria to achieve  $CO_2$  reduction goals. This requires a national and European transportation network of  $CO_2$ sources and sinks. Combined with bioenergy and direct air capture, geological  $CO_2$  storage can contribute to achieving negative emissions (Carbon Dioxide Removal, CDR) (*medium confidence*). Mineral carbonation and closed  $CO_2$  utilization loops can contribute to the  $CO_2$  reduction goals in the transition to climate neutrality (*medium confidence*). CCUS also comes with new challenges, as it is energy and water intensive and puts pressure on the supply systems. {4.6.2}

The development of CCUS in Austria requires overcoming geological, technological, and societal challenges (*high confidence*). Using the geological subsurface for permanent storage requires public acceptance based on relevant information and a broad societal consensus. Technological feasibility requires that  $CO_2$  emitters and sinks are connected by a transport network, that geological fields are developed while minimizing risks, and that innovative solutions for utilizing  $CO_2$  are found that permanently remove  $CO_2$ . With Austria's estimated capacity, CCUS can contribute to national decarbonization on a relevant scale (*high confidence*). {4.6.3}

Work, including unpaid care and paid employment, dominates daily life in Austria, and the structural dependence of persons and welfare systems on paid work currently tends to impede climate-neutral provisioning as well as climate-friendly living (*high confidence*). High-productive, energy-intensive work is well-paid and structurally linked to low-productive, energy-light, unpaid or low-paid, very essential work (including care) (*limited evidence, medium agreement*). The work-mode of living poses a structural barrier for climate-friendly mobility, food, housing and leisure practices, and encourages energy- and fossil-fuel-intensive practices (*medium confidence*). 46 % of Austrians believe that their current job does not contribute to the green transition (*limited evidence, high agreement*). {4.7.1}

A transition to a climate-friendly society will end the requirement for greenhouse gas emissions from paid employment or unpaid care (*limited evidence, high agreement*). The greening and sharing of paid employment and unpaid care will ensure a high quality of life, good work and climate-friendly living, work-life balance and adequate provisioning of care (*medium confidence*). Under the current business-as-usual scenario, work will be severely affected by the unfolding of the climate crisis, requiring sectoral shifts towards the provisioning of basic goods such as health, care, food, and housing (*limited evidence, medium agreement*). {4.7.2} To avoid climate-damaging production and to overcome labor supply constraints, climate-friendly regulation of labor markets can enable a shift from excess production to climate-friendly provision of essential services (*medium confidence*). A redistribution of paid work and unpaid care and the relocation of jobs to people's living place through economic and labor market policies, can enable climate-friendly work-life patterns (*medium confidence*). More equal distribution of wealth and income, job guarantee and public provision of basic goods (e.g., housing, food and health care) can increase economic security and thus enable green labor market transformations (*medium confidence*). {4.7.3}

#### 4.1. Chapter introduction

Chapter 4 of the AAR2 discusses the role of the Austrian economy in a climate-neutral 'Provisioning System', with the objective of satisfying anticipated human needs and fostering human well-being (see, e.g., Fanning et al., 2020). In conjunction with the Avoid-Shift-Improve framework (see Cross-Chapter Box 4), this demand and service-provisioning perspective (see Cross-Chapter Box 3) allows to identify leverage points, such as a Circular Economy (see Cross-Chapter Box 5), to achieve a climate-neutral and low material footprint economy along the resource conversion chain.

#### 4.1.1. The provisioning system

Here, in Chapter 4, we focus on the provision of goods and services (capital and consumer goods and from other sectors including tourism) and of energy to meet diverse human needs. Figure 4.1 illustrates this understanding of the economy as a provisioning system and how specific elements of it are represented in the AAR2. The provisioning system is embedded in the broader societal sphere and fulfills the ultimate goal of satisfying human needs and thereby generating human well-being, which requires appropriate governance and financial arrangements. The provision of goods and services, for example through manufacturing processes, involves a certain demand for resources (for materials, energy, and work). In doing so, the provisioning system interacts closely with the ecological sphere through the demand for natural resources and as a source of greenhouse gas (GHG) emissions (with mitigation potential through Carbon Capture Storage and Utilization, CCS/CCU, see Section 4.6) and other pollution and impacts on ecosystem health (which plays a key role in land-based Carbon Dioxide Removal, CDR, see Cross-Chapter Box 4). In addition, the provisioning system is affected by climate and global change, presenting opportunities and challenges for adaptation to climate change.

Other chapters of the AAR2 discuss other aspects of the Austrian provisioning system in the context of a climate neutral economy, in particular Chapter 2 (agriculture, land use, ecosystem services) and Chapter 3 (buildings and mobility), Chapter 5 (human needs and wellbeing and individ-



Figure 4.1 The economy as a 'Provisioning System' as represented in the AAR2.

ual decision making at household and company level) and Chapter 6 (governance, finance and policy).

After a brief presentation of Austria's historical GHG emissions at the sectoral level (Section 4.1.2), the provisioning of goods and services (Section 4.2) is the starting point of this chapter. As tourism is an important sector within the Austrian economy, we assess tourism as a driver of climate change but also as a highly affected sector, with significant adaptation needs in the future (Section 4.3). We assess the industrial and manufacturing sectors that provide goods and services for final demand (Section 4.4), with a particular focus on Austria's hard-to-abate industrial sectors and the opportunities and challenges of a more circular economy for material supply. In addition, we assess the scientific evidence on energy supply (Section 4.5) and focus on decarbonization through the storage and use of  $CO_2$  from carbon-intensive industries (Section 4.6). We also review the literature on the framework conditions for work, which are not only shaped by economic development, but are also crucial for the development of low-carbon pathways in production and consumption. Both paid work and unpaid but socially necessary work and activities are important for climate-friendly livelihoods (Section 4.7). Each of these sub-chapters reflects on the respective status quo, the constitution of a potential future climate-resilient and carbon-neutral sub-system, and the transformations needed to get there.

# 4.1.2. Sectoral emissions

GHG emissions in Austria have been fairly stable over the last five decades at 80 MtCO<sub>2</sub>eq/yr, excluding sinks such as land-use, land-use change and forestry (LULUFC). Looking at Austrian sectoral emissions, the transport sector and industry are the largest contributors, with the latter's emissions



**Figure 4.2** Panel (a) Greenhouse gas emissions in Austria by sector (production-based accounting) since 1990 (Umweltbundesamt, 2025) (solid thick line including LULUCF; solid thin line excluding LULUCF; GHG emissions excluding LULUCF since 1950 (dotted line, Gütschow et al., 2016, 2024). Panel (b) Production- vs. consumption-based accounting of GHG emissions excluding sources and sinks from LULUCF (Dorninger et al., 2025). Panel (c) GHG emissions in Austria by regulatory coverage, excluding sources and sinks from LULUCF (Umweltbundesamt, 2024c, 2025). The reference data shown in this figure is available at aar2.ccca.ac.at/explorer.

split into those from industrial processes (the larger share) and those from energy use in industry (Figure 4.2).

When carbon emissions are adjusted for emissions associated with foreign trade in goods and services, Austrian imports are much more carbon-intensive than Austrian exports, making these so-called 'consumption-based' emissions about 50 % higher than the emissions generated within Austrian borders, so-called 'territorial' or 'production-based' emissions. Per-capita emissions in Austria are 10 % higher than the EU27 average and more than 50 % higher than the global average (Ritchie et al., 2024).

Only the last few years show a noticeable decrease in GHG emissions (Nowcast 2024, Umweltbundesamt, 2024b). This reduction of emissions was at first mainly due to effects of the COVID-19 pandemic-induced lockdowns, and in 2022 and 2023 due to a combination of the fossil fuel price increases due to the war of aggression by Russia against Ukraine and climate policy measures by the Austrian government (see Chapter 6) (Eibinger et al., 2024).

#### 4.2. Provisioning of goods and services

This section considers the GHG intensity and climate resilience of the goods and services provided to meet individual and societal needs. Provisioning systems define the range of options available to individuals to meet their needs and thus have a significant influence on the adopted technologies, daily routines, and lifestyles (Plank et al., 2021b; Wieser and Kaufmann, 2023). The provisioning of goods and services contributes to well-being, but it requires both human labor and material stocks such as buildings, transportation infrastructure, and the energy production system, the expansion and maintenance of which can consume large amounts of energy (i.e., fossil fuels) and materials. In addition, access to services, such as thermal comfort in a living space or access to a particular destination, generally requires energy (Haberl, 2018; Kalt et al., 2019).

The goods and services that are made available for final consumption and use, and the way in which they are provided, thus have a major impact on resource demand and the climate footprint of a society. Furthermore, goods and services are essential for meeting needs that are distributed across space, time, and society (Chapter 5). Transforming goods and services to be climate-neutral and adapting them to changing climate conditions is therefore critical to ensuring adequate levels of service for all. In addition to decarbonization measures in 'supply-side' energy and manufacturing systems, which will be discussed below, changes in the volume (avoid), composition (shift), and quality (improve) of goods and services offer a wide range of options for 'demand-side' reductions in GHG emissions (Creutzig et al., 2022b) (see Cross-Chapter Box 4).

# 4.2.1. Status quo in the provisioning of goods and services

Material flow analyses provide a comprehensive overview of the number of resources that have to be extracted each year for the provisioning of goods and services. In Austria, the annual demand for processed materials amounts to 242 Mt, of which 51.2 % are extracted domestically, 40.5 % are imported, and 8.3 % are derived from secondary materials (see Figure 4.3) (BMK, 2024b). Nearly half of these processed materials (46 %) are used for durable goods and infrastructures, following their continued expansion in the recent decades (Jacobi et al., 2018; Wiedenhofer et al., 2021; Schug et al., 2023). The processed materials consist of 74 Mt of biomass, 27 Mt of metals (incl. extractive waste), 105 Mt of non-metallic minerals, and 37 Mt of fossil energy carriers. The expansion, replacement, and maintenance of the built environment in particular has become a major source of material demand (101 Mt), overshadowing the material requirements for technical equipment (4 Mt), transport vehicles (3 Mt), and non-technical products (3 Mt) used by households and firms. Today, the total mass of infrastructures and buildings in Austria is estimated at 540 t/capita, compared to 450 t/capita in Germany (Haberl et al., 2021), and per capita stocks are particularly high in sparsely populated areas (Schug et al., 2023). Important drivers of the expansion of infrastructures and buildings include the growth of the service economy (Deetman et al., 2021; Plank et al., 2021b), their significance for contemporary financial investments (Schaffartzik et al., 2021; Pineault, 2023; Wieser et al., 2023), and spatial development (see Chapter 3).

Increasing material demand for building and maintaining materials stocks (Wiedenhofer et al., 2021; Schug et al., 2023) and high growth rates in some service sectors (especially real estate, information and communication technologies, finance) (Plank et al., 2021b) have largely offset positive contributions to the decarbonization of provisioning systems made possible by structural shifts towards a more service-oriented economy. As a result, the total material footprint decreased only marginally between 2000 and 2015 in Austria (Plank et al., 2021b). A study found that out of 11 high-income countries that were able to reduce consump-



Figure 4.3 The material flows through the economy of Austria from inputs (extraction and imports) to final use (stock building) to outflows of exports and domestic processed outputs (DPO) (waste and emissions excluding oxygen from air) in 2020 (BMK, 2024b).

tion-based  $CO_2$  emissions between 2013 and 2019 (achieving absolute decoupling from their GDP), Austria had the second slowest rate of reduction in consumption-based  $CO_2$ emissions (-0.87 %/yr), which, when extrapolated into the future, is far from sufficient to stay below the temperature limit of the Paris Agreement (Vogel and Hickel, 2023).

Despite the importance of material stocks such as buildings, infrastructure, and machinery in driving material demand and GHG emissions, their expansion has received little attention in Austrian climate policy and discourse to date (cf. Theine et al., 2023) (see Chapter 6). Instead, efforts towards more climate-friendly goods and services have traditionally focused on the development and diffusion of technologies (Aigner et al., 2023b; Haas et al., 2023) (see Chapter 6). Today, 4.6 % of Austria's GDP is related to the production of environmental goods and services. Since 2008, this sector has grown largely in line with the Austrian economy, with comparatively high growth rates in export products, especially in renewable technologies, industrial products (chemical, machinery, information and communications technologies), and waste management. The Austrian economy has been shown to be particularly competitive in the provision of technologies for low-carbon buildings and rail transport (Steininger et al., 2021). However, most firms have so far focused on process innovations, with less attention being paid to the climate-friendliness of goods and services and how they are delivered to customers (Schöggl et al., 2022). The portfolios of Austrian manufacturing firms remain predominantly product-centric, with complementary services (e.g., repair, maintenance) and alternative, service-oriented business models (e.g., contracting, rental) making up only a small share of company revenues and added value to date (Dachs et al., 2014; Mastrogiacomo et al., 2019; Friesenbichler and Kügler, 2022; Wieser et al., 2023). The provisioning of energy-efficient technologies has thus been largely detached from wider concerns about material demand.

Public and non-governmental organizations are core pillars of the provisioning of goods and services in Austria. Public provisioning tends to be significantly more energy and material efficient due to a stronger focus on services and collective needs (Giljum et al., 2016; Vogel et al., 2021; in relation to public transport, see Virág et al., 2022). However, in the private sector, the focus has so far been on improving
procurement and internal processes for resource efficiency (see Klien et al., 2023), rather than on the implications of public infrastructures and services for climate-friendly living (Aigner et al., 2023b; Haas et al., 2023). Significant decarbonization potential exists in several areas of public provisioning, including transport, housing, energy, and land use (Bröthaler et al., 2023).

In addition to challenges in cutting material demand, it has proven difficult to reduce the demand for energy needed to process and utilize goods and services. Energy demand is a key driver of GHG emissions in Austria, stabilizing at a level of around 1,280 PJ final energy (BMK, 2023b). The residential, industrial, and transport sector (Chapter 3) each account for around 30 % of final energy demand (Section 4.5). Despite improvements in energy efficiency, so far only relative decoupling (resource use grows more slowly than GDP) has been achieved for energy to date (BMK, 2023b). Absolute decoupling (declining energy use independent of GDP growth), which would be necessary to achieve ambitious climate goals with the current energy mix and without negative emission technologies, has not yet been achieved in Austria (Figure 4.4a). Reasons for this can be found in the residential and transport sectors and include the rebound effect (Seebauer, 2018), increasing activity and service levels, and the expansion of material stocks in buildings, infrastructure, and vehicle fleet which have outpaced energy efficiency improvements (see Chapter 3). In Austria, for example, higher heating energy demand (Figure 4.4b) is associated with behavior (Kulmer and Seebauer, 2019; Venturi et al., 2023) and an increase in floor area (Holzmann et al., 2013; Narula et al., 2022), despite recent climate- and price-induced reductions in energy demand for heating (WIFO, 2024). Heterogeneous effects across income groups have been observed (Kulmer and Seebauer, 2019), with, for example, energy-poor households compensating for previously unmet thermal comfort needs when insulation leads to lower energy bills for heating (Berger and Höltl, 2019). Similarly, the number of kilometers traveled in passenger cars, which are still predominantly powered by fossil fuels, has increased almost twice as much in Austria (+11.3 %) compared to the EU average (+6 %), while some EU countries have managed to achieve a decrease since 2010 (e.g., The Netherlands -3.8 %) (Eurostat, VCÖ, 2019) (Section 3.4.2). With their high dependence on fossil fuels and increasing demand for thermal comfort (mainly heating) and mobility, these sectors pose challenges for Austria's decarbonization, but several Avoid-Shift-Improve options are available (see Cross-Chapter Box 4 and Chapter 3).

Both private and public provisioning systems are exposed to extreme weather events, particularly floods and droughts, which can cause significant economic damage (Steininger et al., 2016; Unterberger et al., 2019). Empirical evidence on climate adaptation in the private sector is scarce but indicates that levels of awareness are low (Balas et al., 2021) and that the implementation of climate risk management measures is underdeveloped (Meinel and Höferl, 2017; Mitter et al., 2019; Hanger-Kopp and Palka, 2022), with public authorities considered as responsible for the management of physical climate-related risks (Rauter et al., 2020).

# 4.2.2. Future systems for the provisioning of goods and services using circular economy principles

Provisioning systems can become significantly more climate-friendly and -resilient while maintaining high levels of service quality and well-being (Barrett et al., 2022; Creutzig et al., 2022a). Energy services needed for a decent life (Grubler et al., 2018; Rao and Min, 2018; Kikstra et al., 2021) (Section 5.2.1) are substantially lower than the ener-



Figure 4.4 (a) No absolute decoupling of energy demand from GDP growth in Austria as service levels increase; (b) heating demand (example), linked to energy; (c) import dependency. Source: Redrawn Figures 34, 36, 43 (BMK, 2023b).

gy footprint of the current provisioning system. For example, the carbon footprint of the Austrian healthcare system (0.8 tCO<sub>2</sub>/capita) is significantly higher than in most European countries (0.52 tCO<sub>2</sub>/capita in France or 0.42 tCO<sub>2</sub>/capita in Sweden), suggesting significant GHG mitigation opportunities in the provision of high-quality healthcare services (Pichler et al., 2019; Weisz et al., 2020) (see Cross-Chapter Box 2 and Section 5.2). Reducing energy and material demand through changes in provisioning systems can deliver multiple co-benefits, most notably with respect to health and clean air (Creutzig et al., 2022a, 2022b), and system security and import dependence (see Figure 4.4c) (Bento et al., 2024), and enable low-carbon transformations of upstream supply systems (Grubler et al., 2018; Barrett et al., 2022; Creutzig et al., 2022b).

The recent decades have seen a sharp increase in material extraction from the environment, which has multiplied by a factor of 12 from 1900 to 2015. This acceleration of material use can be observed for all major material categories: Biomass, metals, non-metallic minerals, and fossil fuels (Krausmann et al., 2018). This unprecedented growth in resource use has implications for the climate, the environment, and the global supply chain. In this context, the Global Resource Outlook (UNEP IRP, 2024, p. 46) states prominently: "Given that resource use is driving the triple planetary crisis, sustainable resource management is urgently needed".

To mitigate the increase in resource use, the concept of the circular economy has become increasingly popular in recent years (see Cross-Chapter Box 5 and Cross-Chapter Box 4, for more details on circular economy principles). Material Economics (2018) investigated the climate potential of the circular economy in relation to industries in the European Union for 2050. They model (a) the material recirculation opportunity and assume that 75 % of steel, 50 % of aluminum and 56 % of plastics could be recirculated, saving 33 % of CO<sub>2</sub> emissions. Next, (b) their model results show that materials efficiency increases through avoiding losses in production processes (e.g., for aluminum or building materials) could save another 11 % of CO<sub>2</sub> emissions. Finally, (c) circular economy business models are identified as another lever that could reduce 12 % of emissions, particularly through sharing of vehicles (utilization of cars is currently about 2 % in the EU) and buildings (European offices' utilization is only about 40 %). Overall, 56 % of CO<sub>2</sub> emissions in heavy industries can be saved by these circular economy strategies.

A review article (Cantzler et al., 2020) screened more than 300 English-language papers in the fields of industry, waste, energy, buildings, transport, and agriculture that explicitly referred to both circular economy or closely related concepts and climate change mitigation potential. Only 10 % of the studies provided insights into how the circular economy can support climate change mitigation. The highest abatement potentials are found in industry, energy, and transport, medium abatement potentials in waste and buildings, and the lowest abatement potentials in agriculture. In conclusion, substitution of material and process substitution in cement, steel, and vehicle production, together with a shift to renewable energies, are core and essential strategies to decarbonize economies. This study shows that the existing literature provides a rather fragmented picture. Circular economy strategies of lower priority seem to be favored, while refuse, reduce, and rethink strategies are only poorly represented in the applied literature (see Cross-Chapter Box 5).

The most recent Global Resource Outlook (UNEP IRP, 2024) developed scenarios for a sustainability transition to 2060. It focuses on four policy packages, including (1) resource efficiency measures, (2) climate and energy, (3) food and land, and (4) a just transition. The packages address, for example, a resource tax, efficient and sustainable settlements with a compact urban form, and sustainable transport modes (ad. 1), carbon pricing, electrification, and renewable energy (ad. 2), nature protection and restoration, healthy diets, and less food waste (ad. 3), and net carbon revenues to provide a carbon dividend payment that supports reduced inequalities and no net economic loss due to the transition (ad. 4). These policy packages achieve a 30 % reduction in resource use and a 90 % reduction of GHG emissions by 2060 compared to a continuation of historical trends.

For Austria, a recent study models the biophysical economy for the period 2018 to 2040 to investigate how circular economy strategies in the building, transport (passenger and freight) and electricity sectors interact with decarbonization (Haas et al., 2025). For this purpose, two economic projections (moderate growth (1) and no growth (2)) are used in combination with four scenarios. One scenario (R) is a continuation of historical trends with no clear decarbonization, while the second scenario (A) is a decarbonization in the three sectors with strong recycling activities and a weak modal split shift representing circular economy strategies by 2040 (see details for all scenarios in Figure 4.5). A third scenario (B) adds to the decarbonization weak circular economy strategies, which are mainly a moderate reduction of passenger kilometers (p-km) and ton kilometers (t-km) and a moderate reduction of per capita floor area in new building construction. The strong circular economy scenario (C) includes a strong modal split shift and strong reduction in traffic volume, increased car sharing, a complete halt to the expansion of the road network, a halt to new construction on unbuilt land, and a higher share of wood construction in new buildings. The reduction in construction activity allows for a significant reduction in ton kilometers as, in addition to eliminating the transportation of fossil fuels, a large proportion of construction materials no longer need to be transported.

The weak circular economy scenario is based on assumptions found in official documents and studies during the modeling exercise (e.g., Kranzl et al., 2018; BMNT, 2019; Krutzler et al., 2023). The strong circular economy scenario is designed to achieve decarbonization and the targets set out by the Austrian government's circular economy strategy (BMK, 2022).

Methodologically, this research is based on international predecessor studies (Haas et al., 2015; Mayer et al., 2019a; Haas et al., 2020; Wang et al., 2020; Miatto et al., 2024) but goes beyond them in modeling all stocks in the building, transport and electricity sectors. End-use explicit material stocks are essential in modeling, as a transition will sooner or later need to replace societal stocks from heating systems, vehicles, to power plants.

The results in Figure 4.5 are presented for a smooth economic recovery projection after the crises since 2019 (COVID-19 pandemic, Russian war on Ukraine, assuming an annual GDP growth of 1.6 %, projection indexed 1) and a slow recovery with zero GDP growth (projection indexed 2). Reducing domestic material consumption (DMC) from 18.4 t/capita in 2018 to the targeted 14 t/capita in 2030 and about 6 t/capita in 2050 (based on the proclaimed 7 t/capita material footprint target of the government's circular economy strategy) (BMK, 2022) is unachievable with pure decarbonization (yellow line, A1). A combination of decarbonization and weak circular economy (B1) measures based on loop-closing (blue line), would still lead to a material consumption significantly above the official target and much closer to the reference value with continued trends (see Figure 4.5a). The strong circular economy scenario (C1) can contribute its share of the buildings, transport, and electricity sectors to government's circular economy targets. It should be emphasized that the measures needed to achieve these targets are highly controversial amongst policymakers. A halt to new building construction on unbuilt land is highly debated, as is a significant reduction in car use (derStandard, 2024).

The phasing out of fossil fuels through fleet electrification and non-fossil fuel heating systems (e.g., heat pumps), as well as the decarbonization of the electricity sector (green power), will contribute significantly to the achievement of the targets. The combination of a phase-out of new constructions of buildings in the period until 2030 and a strong modal split shift assists to reduce both processed materials and final energy use. While some measures have a strong impact on reducing both energy and material demand (e.g., no construction on unbuilt land), others mainly reduce final energy (fleet electrification) or materials (no new roads). Collectively, they contribute to a reduction in final energy of more than 50 %. The direct final energy use in these three sectors is completely based on renewable energy sources (mainly wind, solar and hydro at present levels). The reduction of material of more than 70 % by 2040 is in line with a reduction of the material footprint of Austria from 22 t/ capita in 2018 to 7 t/capita in 2050.

The transition to a carbon-neutral economy can be facilitated if accompanied by strong circular economy strategies (Creutzig et al., 2024). The electrification of transport and, to some extent, heating (especially through heat pumps) is increasing demand for electricity. In 2018, electricity generation was 244 PJ, of which 158 PJ was based on renewable energy sources. In 2040, when all electricity generation is to be based on renewable sources, the demand is 378 PJ in the decarbonization scenario A1 (2.4 times the demand in 2018). The weak circular economy scenario (B1) requires 354 PJ of renewables (2.2x), while the strong circular economy scenario (C1) requires 308 PJ (2.0x). At the same time, as renewable energy sources increase, the demand for rare earth elements (REEs) increases from 84 t in 2018 to 722 t in the decarbonization scenario (A1) in 2040 and 181 t in the strong circular economy scenario in 2040. In particular, the transition will require REEs for electric vehicles and wind turbines. The availability of REEs is potentially challenging, depending on the global increase in renewable energy (Valero et al., 2018; Li et al., 2020; Klimenko et al., 2021) and potentially complicated by geopolitical tensions and negative social impacts in extraction locations (Creutzig et al., 2024). Similarly, a strong increase in electricity demand, as seen in the decarbonization scenario (A1), will require a significant increase in storage capacity, for example in the transport sector, associated with batteries containing scarce raw materials such as lithium (Diouf and Pode, 2015). This material demand can be mitigated by the strong circular economy scenario (C1), as it significantly reduces the number of battery-driven vehicles, for example through less road transport and car sharing.



**Figure 4.5** (a) Domestic material consumption (DMC, all material extraction + imports – exports) for different decarbonization and circular economy scenarios towards the 2030 and 2050 targets of the Austrian Circular Economy Strategy. Scenarios were applied for two different economic projections, a moderate growth (GDP growth after recovery of 1.6 % and a slow recovery and growth of 0 %). (b) Below, a list of measures for decarbonization and circular economy is presented. These measures were applied in a decarbonization and loop closing (material recovery through recycling) scenario (A1), a decarbonization and weak circular economy scenario (applying officially reported measures) and a decarbonization and strong circular economy scenario (C1) that aims for contributing to the government's circular economy targets, which is summarized in processed materials (PM = DMC + secondary materials) but also reduces the demand for final energy. See assumptions for measures in the legend (Haas et al., 2025).

Although some circular economy measures in scenario C1 may seem extensive, the actual impact on quality of life will be rather low. While the car ownership rate is drastically reduced in a strong circular economy scenario (due to modal split shifts and an increase in car sharing), the distance traveled per capita would be reduced by 6 % compared to 2018. This is reasonable given the recent increase in the popularity of teleworking arrangements that reduce the need for work-related car-based commuting (Hartwig et al., 2022),

and since the phasing out of new buildings in the C1 scenar-

io stops further urban sprawl. Another impact of the transition can be seen on floor space and land consumption. The decarbonization scenario (A1) would imply a continuation and thus an increase in land consumption from 11.5 to 12.4 ha/d in 2040. However, such an increase could be mitigated in the C1 scenario because no new roads and no new buildings on unbuilt land have the potential to stabilize land use without further land-consuming stock additions. In terms of floor space, this means that while we have 72 m<sup>2</sup>/capita for residential and office space in 2018, this would increase to 78 m<sup>2</sup>/capita in 2040 in the decarbonization scenario (A1), but decrease to  $67 \text{ m}^2$ /capita in the strong circular economy scenario (C1). Among the older population (60 years and older), there is an openness to reducing per capita floor space. For example, a German study found that 29 % of homeowners and 11 % of tenants are overburdened and assess their housing conditions as too spacious (Kitzmann, 2023). The Global Resource Outlook estimates a utilization rate of 40 % for office buildings (UNEP IRP, 2024). Numerous non-representative case studies show that office, public and educational buildings stand empty for between 90 to 95 % of their service life (including non-working time like weekends or holidays). Factors like low occupancy of seminar rooms, home office, part-time jobs and high proportion of field service can play a significant role here (Wiegand, 2012) (see also Section 3.3). Another Swedish study finds in a specific case a utilization rate of meeting rooms between 14 and 36 %, with meeting rooms mostly larger than needed, and an office attendance of less than 50 % on weekdays and during working hours, with no overcrowding up to 68 % office attendance (Holmin et al., 2015). Thus, a 7 % reduction in heated floor space for residential buildings and offices does not seem unrealistic. Further potential can be found in improved spatial planning that counteracts urban sprawl (Brenner et al., 2024).

AAR2

From an economic point of view, an assessment of these scenarios shows that the average GDP growth rate for the period 2018–2040 would increase from 1.35 % in the reference scenario to 1.44 % in the C1 scenario. This change is due to the re-investment of savings from reduced construction activity in services, as services have a relatively low import intensity and high domestic wage intensity. For the same reason, employment also shows positive effects (Meyer et al., 2024).

An investigation of 18 individual measures with the overall aim of reducing the consumption of mineral raw materials in line with the sustainability strategies (consistency, efficiency, and sufficiency) from exploration through material processing to semi-finished production finds many positive effects for SDG 12 and its targets (Trummer et al., 2022) (see also Chapter 3). An interaction assessment of the measures related to the individual SDGs to better understand the many synergies and trade-offs between SDGs/targets was not conducted and is still lacking. The Global Resource Outlook finds that 13 SDGs are directly, and 4 are indirectly linked to natural resource use (out of 17 SDGs) (UNEP IRP, 2024).

The transition to a carbon-neutral economy offers both benefits and transition risks, depending on how it is implemented. A narrow implementation of measures for buildings, transport, and electricity sector, as in the A1 decarbonization scenario, could risk being highly vulnerable to supply chain disruptions, as REE demand would steeply increase. In addition, the high demand for electricity from renewable energy sources carries the risk that it is highly dependent on the smooth approval of projects, with little room to participate and respond constructively to public opposition and the availability of a sufficiently skilled workforce. Such a scenario would imply further land consumption, exceeding the long-discussed target of 2.5 ha of land consumption per day. It would fail to meet the targets set out in the government's circular economy strategy and to address the triple planetary crisis of climate change, biodiversity loss, and pollution. In contrast, the strong circular economy scenario (C1) has great potential to achieve circular economy goals, reduce land consumption, and facilitate the transition to renewable energy sources, as energy consumption and demand for REE are significantly lower. In summary, especially if a strong circular economy strategy in combination with demand-side measures is implemented, both the magnitude of the climate challenge and the need for material resources relevant to biodiversity and pollution will be reduced (see Creutzig et al., 2024; Haas et al., 2025).

# Cross-Chapter Box 3. Needs orientation in provisioning systems

#### Caroline Zimm; Benigna Boza-Kiss

A demand-side perspective that focuses on the provisioning of services can help (i) identifying large leverage opportunities along the resource conversion chain and (ii) shifting focus on human needs and well-being.

Conversion losses occur, for example, when energy is transformed from one form to another, such as from primary energy (e.g., crude oil) via final energy (e.g., gasoline) to useful energy (e.g., power to the wheel of the car) to the energy service level (e.g., mobility for a certain distance). First, in the current energy system dominated by thermal combustion (e.g., internal combustion engines, boilers), service energy (i.e., useful exergy as a proxy) accounts for 24 % of primary energy input in Austria (CCBox 3 Figure 1a), up from 11 % in 1960 (Marshall et al., 2024). This means that each unit of energy that is avoided or not needed at end-use levels translates to around four units of primary energy that do not need to be provided (Marshall et al., 2024). When we move towards more (renewable) electricity-based systems, these conversion losses will be reduced substantially. The conversion losses also occur in materials processing, where Austria is connected to global material chains (Section 4.2). Globally, steel production from primary ore shows high conversion efficiency losses (CCBox 3 Figure 1b), even though steel is one of the easier-to-recycle materials and a respective market exists. Challenges related to materials will increase with the need for high-quality materials, which are often not available through recycling, and increasing shares of compound materials used, which cannot be recycled easily. Actions reducing demand have great potential in terms of overall resource implications and other co-benefits, e.g., for energy, GHG reduction potential, costs, well-being, reduced air pollution, supply security, flexibility, positive impact on the trade balance of energy importers (Finn and Brockway, 2023; Bento et al., 2024) (see Figure 4.4c and Cross-Chapter Box 2).



**CCBox 3 Figure 1** (a) Resource efficiency cascades throughout the provisioning system for energy in Austria and (b) for steel globally. Each step of the cascade shows the percentage of the extracted primary resource remaining after conversion losses, starting from primary energy on the right. Arrows indicate the efficiency ratio between primary input and the last conversion step. (c) Resource conversion chains (gray shades) linking energy services (e.g., heating, cooking, and mobility) with primary energy for a typical European household (Source: Adapted from Wilson et al. (2023) based on TWI2050 (2018). Estimates for energy in Austria from Marshall et al. (2024), for global steel from De Stercke (2014)).

Moreover, demand for goods and services can be met with different levels of energy or material input. By shifting the focus away from primary resource inputs (e.g., tons of oil) to services that meet human needs and enable human well-being (e.g., thermal comfort), diverse provisioning opportunities of different resource intensities arise. Energy or materials are not demanded for their own sake, but for the goods and services they enable (CCBox 3 Figure 1c). The design of services and the associated user experience is highly relevant to managing demand (Lovins, 2010; Polaine et al., 2013). This is critical for just energy transitions, as human well-being should not be compromised, or at least maintained, but ideally be improved in the transition process. Focusing on service levels rather than on energy or material inputs can facilitate the achievement of decarbonization and distributional objectives, such as those related to affordability (Sections 5.6, 6.7), energy poverty (Sections 5.3.4, 6.8) and decent living (Sections 5.2.1, 6.8), but a focus on the demand side also helps to achieve improvements in health (see Cross-Chapter Box 2), jobs, and energy security (Bento et al., 2024).

# Cross-Chapter Box 4. Avoid – Shift – Improve: The ASI-framework

Andreas Novy; Ernest Aigner; Willi Haas; Nicolas Roux; Caroline Zimm

Avoiding, shifting and improving are different entry points for achieving reductions in emissions and resource use which have been systematized in the so-called A(void)-S(hift)-I(mprove)-framework (Creutzig et al., 2018). In many ways, the ASI-framework is complementary to a framework based on efficiency, consistency and sufficiency. Here, efficiency (i.e., improve) refers to 'doing things better', consistency (i.e., shift) refers to 'doing things differently' and sufficiency (i.e., avoid) to 'doing less' (Fuchs et al., 2023). The ASI-framework represents a hierarchy, with concerns for 'Avoid' and 'Shift' preceding those of 'Improve' (Arnz et al., 2024), as improve strategies so far have not led to the required GHG emissions reductions (Haberl et al., 2020), while national climate action plans in the EU still hardly use sufficiency-oriented measures (Fuchs et al., 2023).

*Avoid* refers to mitigation options that aim at **reducing or refraining from activities** that are using products and services that are unnecessary in meeting basic human needs (see Cross-Chapter Box 5) (Fuchs et al., 2023). 'Avoid' refers to the reduction of energy and materials either through behavioral change and/or through the redesign of associated *service provisioning systems*, such as the redesign of infrastructures and institutions by means of laws, regulations, nudg-ing, etc. (Aigner et al., 2023b). Sometimes structures enable needs satisfaction without producing goods or using energy and materials (e.g., a compact city) (Aigner et al., 2023b). Examples of mitigation options of 'Avoid' comprise reducing unused living space, calory consumption or transport needs by working from home (or remotely), thereby avoiding the production and consumption of energy and material-intensive harmful and unhealthy food (Lampl et al., 2024).

*Shift* describes a change through substitution of one option for another. This entails a **switch to another cleaner op-tion or service provisioning system.** This could be the switch from car to public transport due to new infrastructural arrangements or price incentives (e.g., climate and repair bonus), replacing fossil fuels by renewables in the energy mix (e.g., gas boiler to heat pump), or moving towards a more plant-based diet.

*Improve* refers to enhancing the efficiency of existing technologies or practices improving their environmental performance. This often means optimizing the efficiency of existing products, services and production processes. Examples are better building insulation, reducing food losses, or higher efficiency in internal combustion engines.

In line with multi-perspectivity, avoid, shift, and improve measures are not mutually exclusive, but **need to be com-bined in portfolios of measures**, to increase their effectiveness (Fuchs et al., 2023; Novy et al., 2023) (see Sections 6.5.4,

**8.6.1**). Transport systems that provide the service of accessibility and mobility can combine compact cities with a shift to public transport and ridesharing (see Section 3.4.2, CCBox 4 Table 1).

As all ASI mitigation options entail synergies and trade-offs (e.g., rebound effects), a careful approach is required that takes into account the interactions between ASI options in order to utilize synergies and mitigate trade-offs (Haas et al. (2023); for a discussion on digitalization, see Santarius et al. (2020)) (see Section 8.3, 8.6).

The ASI-framework is also relevant for GHG-emissions stemming from consumption in general (e.g., GHG emissions from process emissions in the concrete production as part of housing provisioning) (Aigner et al., 2023a, 2023b).

While avoid and shift measures often depend on changes in infrastructure, institutions and prices, improve measures are often compatible with incumbent habits and practices and can therefore be implemented more quickly. However, improve options alone often bear a small mitigation potential compared to avoid and shift measures to achieve decarbonization. They may also lead to rebound effects when efficiency gains induce an increase in production and consumption (Alcott, 2008; Sachverständigenrat für Umweltfragen, 2024).

The ASI-framework aims at reducing environmental pressures rather than solely favoring efficiency gains, acknowledging the absolute biophysical limits of the Earth system and the non-substitutability between different dimensions of sustainability. It is grounded in theories of provisioning, exploring the interplay between production, consumption and distribution. While a 'provisioning system approach' does not examine the material throughputs (Fine, 1994; Bayliss and Fine, 2020), a 'social provisioning perspective' studies the organization of livelihoods and places provisioning at the center of feminist and (social) ecological economics, investigating who provides services (paid and unpaid), by what means, and with what material impacts (Nelson, 1993; Jo, 2011; Jo and Todorova, 2019; Spash, 2024). Human well-being depends on the specific institutional-normative organization of the socio-economic system embedded in biophysical systems with diverse planetary boundaries (O'Neill et al., 2018). Therefore, gender, inequality and power relations affect the potential to mitigate emissions and resource use (Brand-Correa et al., 2018; O'Neill et al., 2018; Schaffartzik et al., 2021). In such a broad understanding of provisioning, actors including households, business, civil society and (semi-)public institutions form provisioning systems (Plank et al., 2021b). They evaluate, judge, and decide on whether and how to modify framework conditions of service-provisioning systems - and in consequence - forms of life (Creutzig et al., 2022a). Structures of supply (e.g., available portfolio of goods) and infrastructures (e.g., village centers with a high quality of livability) interact with social and cultural norms for service choices (i.e., diet choice, modal split).

The ASI framework encourages a more systematic way of thinking about different measures by **directly targeting well-being** and needs satisfaction when formulating and comparing mitigation strategies. The underlying eudaimonic understanding of a good life puts human needs at the center, including thermal comfort, nutrition, or mobility (Creutzig et al., 2016, 2018) (see Cross-Chapter Box 3). It aims to change supply structures and modify infrastructures and choice architectures to change demand for goods and services, thereby differentiating between the overall amount of resource use (e.g., liters of fuel), the required *service level* (e.g., the use of energy for heating or cooking) and the related *provisioning systems* (e.g., the energy, food or mobility system covering production, consumption and distribution) that are available to provide the different services (Creutzig et al., 2022a).

The ASI framework, interested in what people need for a good life, has labeled its policy proposals as demand-side solutions, i.e., service provisioning systems with the specific objective to conceptualize **demand**, **preferences and choic-es not as given but as malleable** (Creutzig et al., 2018). Globally, demand-side solutions could collectively reduce emissions by 40–70 % by 2050 (IPCC, 2023a). Demand-side solutions widen the available portfolio of measures. First, they nudge behavioral changes within given infrastructures and institutions by exploring consumer preferences and deriving feasible (often incremental) solutions under given framework conditions (see Chapter 5). Second, they aim at deeper transformations of norms and objectives due to different infrastructures, institutions and prices (Meadows, 1999; Abson et al., 2017; Aigner et al., 2023a) (see Section 8.6.3).

Some examples to guide the categorization of options are provided in CCBox 4 Table 1.

CCBox 4 Table 1 Selected examples of measures across sectors categorized according to the ASI framework (assessment by authors based on Creutzig et al., 2018, 2022a, 2024; Aigner et al., 2023; Arnz et al., 2024). For a systematic list of measures and their assessment, see the respective chapters, as well as Table 6.2.

	Service	Avoid	Shift	Improve			
Transport	Accessibility Mobility	Compact cities	Electric two-, three- and	Eco-driving			
	MODIIIty	Integration of transport and land- use planning	Mode shift from car to	Light-weight vehicles			
		Teleworking	cycling, walking, or public transit	More efficient gasoline engines			
		Workplace near to place of resi- dence		Ride sharing			
Buildings	Shelter	Avoid empty living space	Combined heat and power	Building insulation			
		Change temperature set-points	Heat pumps, district heat-	Improved raw materials			
		Efficient floor space allocation and reuse of existing buildings	Invertor air conditioning	Condensing boilers			
		Passive house (avoiding demand for heating/cooling)	Solar thermal	Incremental insulation options			
		Shading and solar radiation aware architecture		Energy-efficient appliances			
		Smaller houses		Other building materials			
Manufactured products and services	Clothing Appliances	Prohibition of certain fast fashion business practices (e.g., throwing	Shift to recycled materials, low-carbon materials for	Long-lasting fabric, appli- ances			
		away non-used goods in on- line-shopping)	buildings and infrastructure	New manufacturing pro- cesses and equipment use			
		Reuse and repair (that is: Not producing new goods, thereby avoiding emissions and material)		Use of low-carbon fabrics			
		Sharing economy					
Food	Nutrition	Calories in line with daily needs	Healthy fresh (seasonal) food to replace processed	More efficient fridges and freezers			
		Smaller fridges and freezers	food Shift from ruminant meat to other protein sources where appropriate	Reuse food waste and food waste reduction			
Work	Commuting	Limit the availability of car park- ing lots	Provide sharing options at workplace	Shift to electric mobility for commuting			
		Reduce the number of days worked per week to avoid non-needed commuting	Shift to public transport or cycling	Subsize the electrification of company cars			
		Work-places nearby the living places					

# Cross-Chapter Box 5. Circular economy

#### Willi Haas, André Baumgart

Steadily increasing extraction and processing of material resources (fossil fuels, minerals, non-metallic minerals and biomass) are the main drivers of the three planetary crises climate change, biodiversity loss and pollution (Geissdoerfer et al., 2017; UNEP IRP, 2024). To better deal with these increasing resource flows, the concept of circular economy has gained popularity in the past two decades and is being continuously developed (Kirchherr et al., 2023). The importance of and interest in circular economy concepts derive primarily from the fact that they can be used as a basis for developing solutions to environmental and climate policy concerns that will also provide economic opportunities, while relying on key economic principles such as the efficient use of resources. Increased resource efficiency through a circular economy essentially means striving for higher value creation with lower use of natural resources (Geng et al., 2019). The definition of the circular economy matters for circularity assessments and prioritizing measures. In the last two decades, according to Korhonen et al. (2018), practitioners have been the dominant driving force behind the circular economy concept. Thus, policymakers, businesses, business consultants, business associations, and business foundations play a key role in this field (People's Republic of China, 2008; Ellen Macarthur Foundation, 2013, 2015; COM/2015/0614 final; McKinsey & Company, 2016; Japanese Ministry of the Environment, 2018; de Wit et al., 2019). Consequently, these actors promote definitions that serve their interests with little attention to consistency across products, supply chains, materials, or scales, and in the worst case risk greenwashing and image loss (Nobre and Tavares, 2021). But also within the scientific community, circular economy "means many different things to different people" (Kirchherr et al., 2017, p. 229). In a follow-up article, Kirchherr et al. (2023) found 221 definitions in 6,566 studies in a review of circular economy conceptualization.

A definition that received broad recognition (more than 3,800 citations) comes from Korhonen et al. (2018, p. 39): "Circular economy is an economy constructed from societal production-consumption systems that maximizes the service produced from the linear nature-society-nature material and energy throughput flow. This is done by using cyclical materials flows, renewable energy sources and cascading-type energy flows. Successful circular economy contributes to all the three dimensions of sustainable development. Circular economy limits the throughput flow to a level that nature tolerates and utilizes ecosystem cycles in economic cycles by respecting their natural reproduction rates".

In line with this definition, there are several circularity strategy frameworks, often referred to as 10R (Potting et al., 2017; Blomsma et al., 2019; Morseletto, 2020) – because of the ten strategies all starting with an 'R' – which aim to minimize the consumption of raw materials and the generation of waste and emissions. The 10R strategies (sometimes referred to as 9Rs due to different counting) range from narrowing (refuse, reduce, redesign), slowing (reuse, repair, refurbish, remanufacture, repurpose), to closing loops (recycle, recover) of material flows. Strategies are prioritized according to their levels of circularity, with smarter product manufacturing and use, such as product sharing, generally being preferred over extending product lifetime because more users can be served by one product (narrowing strategy with high circularity). Next in order is lifetime extension (slowing), followed by material recycling (closing loops), in which product integrity is lost. Last is incineration, in which energy is recovered with the highest quality material reduction (Potting et al., 2017).

Thus, the 10R strategies follow the same guiding principle as the Avoid-Shift-Improve framework (Muñoz et al., 2024). While Avoid-Shift-Improve is primarily focused on reducing GHG emissions, the 10R strategies are focused on reducing raw material inputs from nature and waste (Reike et al., 2018). However, the two concepts are inherently linked, as the material perspective of the 10R framework, which includes energy materials (fossil and biomass fuels) and the extraction, processing, and use of materials as products and infrastructure, is a major cause for energy use and GHG emissions (see CCBox 5 Figure 1). As a direct result of material and energy use reduction, the circular economy approach also reduces land consumption and biodiversity loss.

The circular economy is thought to contribute to resource security and resilient value chains by redirecting resource flows from waste management to the productive economy (IEA, 2022). However, there is a risk that circular economy strategies focus too narrowly on closing loops, respectively recycling, as this promises a zero-waste world and thus allows unrestrained growth of stocks and flows, rendering circular economy impossible and failing to address the enormous strain to reduce and slow material and energy use (Haas et al., 2025).

While circular economy offers strategies to reduce material and energy use as well as waste and emissions, it encounters surmountable thermodynamic limits. According to the first law of thermodynamics, mass and energy are constant in a closed system (Fischer-Kowalski et al., 2011). Therefore, if the global socio-economic system is a circular economy that does not draw on the resources of the biosphere, this is not feasible with growing stocks of materials. Materials cannot be created out of nothing, so any system with growing stocks will need inputs of primary materials, even if it were possible to completely reuse or recycle all old materials and waste. As stocks, even if constant, have to be maintained due to deterioration, wear and losses, a fully circular economy is impossible. The second law of thermodynamics on entropy means that material qualitative characteristics are altered in irreversible production processes (Giampietro, 2022). Further, energy and material requirements for recovery, processing and recycling rise disproportionally the more dispersed and mixed end-of-life wastes are. This is because waste materials of all households for example need to be returned to the specialized production sites and have to comply with quality and purity requirements of high-tech production processes or for performance reasons, e.g., in construction (Ayres, 1999; Reck and Graedel, 2012; Cullen, 2017).

All these limitations need to be seriously considered when discussing circular economy potentials against the backdrop of material and energy use dynamics at different levels. According to the latest Global Resources Outlook report (UNEP IRP, 2024), global domestic material consumption (DMC) amounted to 96 Gt/yr in 2019. With 9.5 Gt/yr of secondary materials having re-entered the production chain, global input circularity was about 9 % (UNEP IRP, 2024). Since 1900, non-circular input flows have increased 16-fold, while the global input cycling rate has decreased from 43 to 27 % (Haas et al., 2020). At EU level, with a DMC of 6.4 Gt/yr and a recycling of 0.8 Gt/yr in 2022, the circularity rate was about 11 % (Eurostat, 2023).



three strategies to reduce GHG emissions while the circular economy approaches set priorities for 10R strategies to reduce material and energy in terms of their related inflows of raw materials from and outflows wastes and emissions to nature. Source: Own visualization.

To increase circularity, circular economy concepts have been integrated into various international policies and strategies, including the Sustainable Development Goals (SDGs), where SDG 12 'Responsible consumption and production' is primary concerned with sustainable use of natural resources and waste reduction. Likewise, the European Union adopted the 'EU Action Plan for the Circular Economy' in 2015 (COM/2015/0614 final), which has gained new impetus in the wake of the Green Deal to achieve climate neutrality (COM/2019/640 final), but has also been criticized for being holistic in words but 'end-of-pipe' in action (Calisto Friant et al., 2021), with fragmented and competing goals reducing effectiveness (Domenech and Bahn-Walkowiak, 2019), and finally for a lack of high-quality data to assess progress towards a far-reaching circular economy (Morseletto and Haas, 2023). The Austrian government has set the ambitious goal to achieve climate neutrality by 2040 at the latest (BMK, 2023a). The 'Austrian Circularity Strategy' aims to reduce resource use and waste by promoting circular economy strategies (BMK, 2022). The strategy envisions an annual per capita DMC of 14 t in 2030 and a per capita material footprint (MF=DMC including upstream material requirements) of 7 t/yr in 2040, which corresponds to a DMC of roughly 5–6 t/capita. However, Austria, as a country with a high per capita stock level (Haberl et al., 2021), had a DMC of around 18 t/capita in 2018, of which 28 % was used in the building sector, 19 % in the transport sector, and 2 % in the electricity sector (Haas et al., 2025). This leaves more than half for other sectors, including manufacturing and agriculture (see Section 4.2.2 and Chapter 5).

Circular economy can support climate policy and the transition to carbon neutrality, as the accumulation of material in societal stocks in particular is key to energy reduction, since the expansion, maintenance and operation of stocks are major drivers of energy use. Since green energy will remain a bottleneck for at least the next decades (see challenges discussed in Ang et al., 2022), the circular economy's reduction potential is urgently needed.

The goals of the Austrian Circularity Strategy are interwoven with goals from other national strategies: Soil sealing remains an issue in Austria, with the average daily increase in hectares of land taken being significantly above the official 2.5 ha/d target formulated as part of the Master Plan Rural Development 2017 (Prokop, 2019). Reducing the amount of land sealing implies a reduction in newly constructed houses or a slowdown in further transport network expansion and consequently puts a restraint on resource use. Reductions in material and energy use lead to reductions in GHG emissions. Together, a reduction in land sealing and associated land fragmentation and mitigation of GHG emissions benefit biodiversity and the climate (Wiedenhofer et al., 2015; Koemle et al., 2018).

# 4.2.3. Necessary transformation in the provisioning of goods and services

Considering the large stocks of materials in use in infrastructures, buildings, and appliances, major investments are required to reduce fossil fuel consumption needed for their use and to adapt these stocks to changing climatic conditions (Bachner, 2017; Bachner et al., 2019a; Bröthaler et al., 2023). At the same time, the modernization and replacement of existing material provisioning structures, alongside necessary investments in new manufacturing facilities (see Section 4.4) and energy supply infrastructures (see Section 4.5), would significantly increase electricity and material demand in the short- to medium-term. Furthermore, an overemphasis on the diffusion of low-carbon technologies can create long-term dependencies on critical raw materials such as lithium. This calls for a balanced approach to transformation that accounts for the different timeframes within which net reductions in GHG emissions can be achieved. Energy and material efficiency measures in the provisioning of goods and services with high potential for short-term reductions in demand, thus play a vital role in the transformation process (Barrett et al., 2022; Creutzig et al., 2022a; Langevin et al., 2023). Avoid measures come with both short- and long-term potential, thus focusing on them offers great transformation opportunities. Within all Avoid-Shift-Improve options, the interplay between individual actions and existing structures can hamper the potential (see Chapters 5 and 6; Cross-Chapter Box 4).

To realize the potential for reductions in energy and material demand outlined in the previous section, and to improve the resilience of provisioning systems to climate-related shocks, provisioning systems would need to undergo transformative change at multiple levels: Shifting to local provisioning systems that enable climate-friendly modes of living, establishing infrastructures and buildings that facilitate climate-friendly practices in everyday life, and deploying more climate-friendly end-use technologies. To date, the latter have received most attention in climate mitigation strategies (see Section 4.2.1). Integrative frameworks such as the circular economy can support actions across all three levels and thereby contribute to more systemic and needs-oriented responses (see Cross-Chapter Box 5 and Cross-Chapter Box 4) to climate mitigation.

Local provisioning systems for climate-friendly modes of living: Provisioning systems can play an important role in giving rise to daily needs in the first place. The quality, design, and spatial organization of provisioning systems can influence decisions as diverse as starting a family, relocating, or choosing a certain lifestyle - decisions that can have longterm repercussions for both climate mitigation and adaptation (Schäfer et al., 2012; Greene, 2018). Compact city design, polycentric provisioning systems, and multifunctional areas can reduce dependencies on unsustainable technologies and practices (e.g., car dependency) (Svanda and Zech, 2023). Design aspects need to go beyond spatial planning to include institutional arrangements with regards to, for example, work or decision making (Sections 4.7 and 8.5.4). Within the design of systems that cater to certain needs lies the largest power for avoid and shift activities that reduce the overall system size (see Cross-Chapter Box 4).

Infrastructures and buildings for everyday life: At the level of everyday practices, the circular economy, the sharing economy, and digitalization are three interrelated 'megatrends' that offer cross-sectoral opportunities to improve more infrastructures and buildings more efficiently (Creutzig et al., 2022b). Synergies may exist between the circular economy and resilience, but evidence is limited to date (Kennedy and Linnenluecke, 2022). New, digitally-enabled and more service-oriented business models are at the center of many circular economy and sharing economy initiatives, although assessments of their mitigation potential still remain fragmented. Despite the greater service orientation, there is still a significant gap between the focus on business models and specific product lifecycles in these economic models on the one hand, and the broader solution space considered in energy and material services perspectives on the other (Carmona et al., 2017; Whiting et al., 2022; Wieser et al., 2023). Less attention is paid, for example, to public provisioning and inequalities in access to goods and services. The mitigation potential of circular and sharing business models is therefore contingent on the careful management of rebound effects (Kjaer et al., 2019; Castro et al., 2022; Metic and Pigosso, 2022). As a regulator, procurer, and owner of large infrastructure providers, the public sector can play a key role in supporting the transition to climate-neutral and -resilient infrastructures and buildings (Kubeczko and Krisch, 2023). It has been shown that policies to meet Austria's circular economy strategy (BMK, 2022) and the formulated targets for land take (Prokop, 2019), have to limit the construction of road infrastructure and buildings, as they are major consumers of materials and are related to increasing energy consumption (see Figure 4.5). Thus, a strong circular economy strategy focused on narrowing material resource flows and essentially targeting infrastructures and buildings in the transport, building and electricity sectors, can reduce material consumption in these sectors by more than 60 % and their energy consumption by more than 50 % (Haas et al., 2025).

Environmental end-use technologies: The diffusion of climate-friendly end-use technologies has been critical to climate mitigation to date (Geels, 2023). As the development of important end-use technologies such as electric cars, heat pumps, photovoltaics, and biomass district heating in Austria shows, innovation and diffusion processes typically take several decades, frequently have an infrastructure component, and involve significant changes in user practices (Geels and Johnson, 2018; Kulmer et al., 2022). Supporting such non-incremental innovation requires a balanced combination of supply-side 'technology-push', demand-side 'market-pull', and systemic intervention measures (Pitelis et al., 2020; Aflaki et al., 2021). Where technologies have reached a high level of maturity and become available in markets, coordinated efforts to improve cost-performance ratios, coalition building, and market shaping can create self-reinforcing positive feedback effects to accelerate technology diffusion (Roberts et al., 2018; Victor et al., 2019; Kulmer et al., 2022; Andersen et al., 2023; Geels and Ayoub, 2023). Cost-performance improvements can be achieved through economies of scale, learning-by-doing, and increased availability of complementary technologies (Victor et al., 2019). Shared visions and targets, public procurement, empowerment of actor coalitions, visibility of frontrunners, and economic incentives for adoption can increase support for climate-friendly technologies (Roberts et al., 2018; Termeer and Dewulf, 2019). End-use technologies lie at the core of improve and also shift activities (Cross-Chapter Box 4).

# 4.3. The case of tourism and climate change

# 4.3.1. Overview of the Austrian tourism sector

The Austrian tourism sector is of considerable importance to the Austrian economy. Prior to the COVID-19 pandemic, Austria recorded more than 46.2 million guests with around 152.7 million overnight stays in 2019 (Statistik Austria, 2022). In 2019, the direct value-added effects of tourism amounted to EUR<sub>2023</sub> 26.45 billion according to the Tourism Satellite Account (TSA), which corresponds to a share of 5.5 % of the gross domestic product in 2019 and EUR<sub>2023</sub> 36.2 billion (7.5 % of GDP in 2019) if the indirect value-added effects of tourism are also included; direct tourism employment in 2019 accounted for 6.2 % of total nationwide employment (Fritz et al., 2021).

As peripheral Alpine regions in particular are economically dependent on the tourism sector, a failure to successfully implement adaptation and mitigation strategies to climate change in the immediate future will pose significant economic risks (Pröbstl-Haider et al., 2020).

# 4.3.2. The contribution of tourism to GHG emissions

Tourism is a significant contributor to global warming. According to Lenzen et al. (2018), global tourism emitted approximately 4.5 GtCO<sub>2</sub>eq in 2013, accounting for about 8 % of total GHG emissions. This includes direct emissions from visitor consumption (2.9 GtCO<sub>2</sub>eq) and indirect emissions from the upstream supply chain (1–2 GtCO<sub>2</sub>eq). Notably, about half of the emissions are attributed to transport, with a higher share originating from high-income countries due to increased air travel consumption. Gössling et al. (2023, p. 2) note that "adding aviation's non-CO<sub>2</sub> contribution to climate change ... increases tourism's contribution to global warming to 10 % in 2013".

For Austria, estimates of tourism-induced emissions are provided by three sources: Lenzen et al. (2018), Neger et al. (2021), and Prettenthaler et al. (2021).

Lenzen et al. (2018): The authors provide (so far unpublished) estimates indicating that GHG emissions from the Austrian tourism sector amounted to 12.85 MtCO<sub>2</sub>eq in 2013. The accounting relies on Tourism Satellite Account data and is destination-based (DBA), i.e., it includes

emissions from domestic tourism, inbound tourism, and local services for outbound travel (e.g., travel from home to the airport). For international air transport, only emissions from national carriers for inbound, outbound, and stop-over services are included, as well as emissions from other transport providers registered in Austria.

Additionally, Lenzen et al. (2018) apply a residence-based accounting (RBA) approach that considers all direct and indirect travel-related emissions (transport, accommodation, food, etc.) generated by Austrian residents. These emissions are estimated at 15.63 MtCO<sub>2</sub>eq, indicating that Austria is a 'net origin' country: Emissions caused by Austrian residents traveling abroad exceed those induced by international travelers visiting Austria by 2.78 MtCO<sub>2</sub>eq (21.7 %).

Neger et al. (2021): This study uses a destination-based accounting (DBA) approach. They estimate that the Austrian tourism sector directly emitted 4.3 MtCO<sub>2</sub>eq in 2016, which corresponds to 4.15 % of total Austrian CO<sub>2</sub>eq emissions. Notably, this estimate does not include indirect emissions from tourism activities (scope 2 and 3 emissions). Thus, the reported direct emissions are significantly lower than the estimates of Lenzen et al. (2018).

For the year 2018, the authors also calculate transport-related emissions from foreign visitors to Austria which amount to 35 MtCO<sub>2</sub>eq (and thus more than eight times the DBA emissions, corresponding to a share in total CO<sub>2</sub>eq emissions of 34.1 %). Almost 95 % of these emissions are generated by visitors arriving in Austria by air from outside Europe. Together with the DBA estimates of 4.3 MtCO<sub>2</sub>eq for 2016, this implies that about 99 % of all emissions directly caused by visitors to Austrian destinations are related to transport. However, such a high share of transport-related emissions may depend on the estimation method used (e.g., if international air carriers are included), but also on the share of international visitors arriving mainly by car or air. Other studies have found much lower shares: For Norway, Sun et al. (2022) estimate that 56 % of total carbon emissions are transport-related, the same share was reported by Kitamura et al. (2020) for Japan.

• Prettenthaler et al. (2021): This study reports direct transport-related emissions from residents traveling to destinations within and outside Austria (RBA) at 11.5 Mt CO<sub>2</sub>eq for 2016, with 95 % of these emissions concerning travel abroad. However, as this estimate does not include emissions from accommodation and food, it is not directly comparable with the estimates from other studies.

As the estimates by Lenzen et al. (2018) and by Neger et al. (2021) differ not only in scope (direct and indirect emissions for the former, only direct emissions for the latter), but also in magnitude, a high degree of uncertainty regarding tourism-induced emissions for Austria is evident. The challenges of estimating tourism emissions' footprint are also high-lighted by Gössling et al. (2023). An emissions inventory for Austrian tourism is needed, which must shed light on tourism-induced emissions from different perspectives (DBA, RBA, producer-based accounting) and the emission sources.

# 4.3.3. Future systems: Impact of climate change on tourism

Austria's tourism industry is particularly vulnerable to climate change due to its dependence on the natural environment, which is one of the most important factors motivating international tourists to visit the country (Nature Reloaded, 2015). Rising temperatures, shifting weather patterns, and extreme weather events are reshaping tourism offerings and demand. This will consequently change tourism employment, both in terms of regionality and seasonal shifts, which may require different skill sets, training and education (Scott et al., 2019). Shifts in tourism employment will depend on changes in tourism product offerings. A detailed analysis of the impacts of climate change on the Austrian tourism industry is provided in the Austrian Special Report 2019 'Tourismus und Klimawandel' (Pröbstl-Haider et al., 2020), which discusses the demand- and supply-side effects of climate change.

Higher temperatures could make summer vacations in the Austrian alpine regions more attractive for both domestic and international visitors seeking cooler destinations, thus shifting their choice of destinations away from southern European destinations suffering from intense heat waves, rising sea temperatures, wildfires and water scarcity. In addition, more moderate temperatures in the shoulder months of the summer season will extend the season. On the other hand, winter tourism in the Alps is expected to lose demand due to higher average temperatures and less precipitation in the form of snow (Steiger et al., 2021). Consequently, the main adaptation strategy will be to at least partially compensate for the loss of income from alpine snow-related winter tourism by increasing tourism revenue from spring to fall (Pröbstl-Haider et al., 2020).

Austrian cities and rural, non-alpine tourism regions are also experiencing various impacts of climate change, particularly rising temperatures, but also extreme weather events. These changes affect the environment, local economies, and overall quality of life for residents and visitors alike.

- More frequent and intense heat waves in cities can exacerbate the urban heat island effect, making them significantly warmer than surrounding rural areas (Loibl et al., 2014). This can lead to increased energy demand for cooling, higher health risks for vulnerable populations, and a greater strain on infrastructure. Additionally, higher temperatures can worsen air quality by increasing the formation of ground-level ozone, which poses health risks, particularly for individuals with pre-existing health conditions (European Climate and Health Observatory, 2021; Mayer et al., 2022; Umweltbundesamt, 2024a).
- Rising temperatures may initially appear to benefit rural tourism, but extreme heat may deter visitors. Rising air temperatures lead to higher water temperatures in lakes, which can negatively affect aquatic ecosystems (Dokulil et al., 2021). Warmer temperatures can disrupt local ecosystems, leading to changes in species composition and loss of biodiversity. Further, changes in water quality and levels can affect recreational activities such as swimming, boating, and fishing; thereby reducing the attractiveness of tourist destinations. Rising temperatures can lead to more intense and frequent precipitation events, increasing the risk of flooding that damages infrastructure, homes and businesses, and deters tourists.

In winter, the Austrian ski industry is expected to be negatively affected by rising temperatures (see Section 1.3 on snow coverage). Studies on the impacts of climate change on ski tourism show a significant shortening of the ski season, less precipitation, lower snow depth, increased need and difficulty for artificial snowmaking, and a decrease in glaciers in Austria (Olefs et al., 2017; Helfricht et al., 2019; Steiger et al., 2020; Steiger and Scott, 2020; Abegg and Mayer, 2023; François et al., 2023a; Salim et al., 2023; Mayer and Abegg, 2024) (see Section 7.4 for details on climate change impacts on ski destinations). Visitors may also be deterred by the lack of snow-covered landscapes. In this respect, guest numbers in ski resorts fluctuate more than the number of overnight stays (Steiger, 2010; Falk, 2013), presumably because day visitors can react more spontaneously to weather and snowfall than vacation guests (Steiger, 2011). Many Austrian ski resorts, especially those at lower altitudes, will therefore have to find alternative solutions for increasingly difficult winter operations, as it may no longer be technically possible, economically viable, or preferable/acceptable to guests

and local communities to provide sufficient snow cover in some places (see Section 7.4). However, there are also challenges in high-altitude ski resorts, particularly related to the retreat of permafrost and glaciers (Mayer and Abegg, 2024).

Furthermore, the climate crisis could significantly reduce the importance of the tourism industry in the context of a general loss of prosperity (Scott and Gössling, 2022).

# 4.3.4. Needed transitions in the Austrian tourism sector

### Avoid

A reduction in the **number of trips** would decrease overall GHG emissions. In the case of business trips, digitization facilitates such a reduction, as meetings are increasingly organized online (Kagermann, 2015). The COVID-19 pandemic provided incentives for firms to invest in equipment for online videoconferencing, thereby saving on travel costs. Furthermore, an increasing number of conferences are organized online or in a hybrid mode (Lenzen et al., 2018; Hanaei et al., 2022). A reduction in the number of holiday trips, accompanied by an increase in vacation duration to compensate for the loss of recreational value, could help to curb tourism-induced emissions (Pröbstl-Haider et al., 2020; Gössling and Higham, 2021).

Furthermore, as the largest share of emissions originates from mobility (Dubois et al., 2011; Pröbstl-Haider et al., 2020), visitors from **long-distance markets** traveling by air could be discouraged to spend their holidays in Austria, e.g., by refraining from marketing activities in these countries. However, global net emissions benefits are only achieved if long distance trips to (Austrian) destinations are not substituted by long distance trips to other destinations outside Austria.

### Shift

In line with the above, climate change mitigation efforts may take the form of a shift in the target groups of the Austrian tourism industry, for example by focusing on guests that have more opportunities to travel by sustainable means of transport. This is essential, as air traveling produces 31 times more  $CO_2$  emissions than train travel, and traveling with a combustion engine car produces about 15 times more  $CO_2$ / km compared to the train alternative (VCÖ, 2022). Touristic service providers and tourism operators may encourage the proposed shifts by guiding guest behavior, for example by introducing price or convenience incentives (Gössling and Higham, 2021; Bursa et al., 2022; Blättler et al., 2024). Such substitution could also be incentivized by internalizing the environmental costs of carbon emissions.

Public transportation networks, active mobility options (including bike and e-bike rental systems) or car sharing systems can be an attractive and more sustainable option for mobility within destinations (Pisoni et al., 2022; Edberg, 2024) (see Chapter 3 on spatial planning and mobility). However, this can only be achieved through infrastructure investments.

A shift in the mode of transport is also key to reducing emissions from outgoing tourism: Reducing the number of long-haul trips by plane and increasing domestic travel by train, for example, already drastically reduces GHG emissions (Umweltbundesamt, 2018; VCÖ, 2022, vacation in Austria by train: 15 kg CO<sub>2</sub>/capita/d vs long-haul trip by plane: 454 kg CO<sub>2</sub>/capita/d).

#### Improve

In addition to avoiding emission-intensive tourism and shifting to a less carbon-intensive mode of the tourism production function, the current supply of tourism services, including transport to and from the destination, needs to be improved to reduce emissions (Gössling and Higham, 2021; Gössling et al., 2023). If individual travel cannot be fully shifted to public transport (both because of the mobility preferences of the travelers and insufficient rail transport capacities), the use of electric cars should be promoted by investing in charging infrastructure. If, at the same time, electricity generation is shifted to renewable energy, CO<sub>2</sub> emissions will be reduced (Holmberg and Erdemir, 2019). Numerous other transition measures have been listed and discussed for the different segments of the tourism industry (accommodation, restaurants, transport, summer and winter outdoor activities, events etc.).

Suggested improvements focus on energy efficiency (related to tourism buildings and infrastructure), waste reduction and recycling concepts, water management, intensification of natural planting to limit ground erosion, and greening of buildings to create natural shade and cooling systems (see, e.g., Pröbstl-Haider et al., 2020; Gössling and Higham, 2021). In the accommodation sector, building insulation can be improved and measures can be taken to increase energy efficiency in terms of room temperature, hot water use, and lighting (Energieinstitut der Wirtschaft GmbH, 2012). This is particularly important in city destinations, as energy demand for cooling buildings will increase as climate change continues (Prettenthaler et al., 2008; Dowling, 2013; Berger et al., 2014a, 2014b). In addition, measures such as increasing green spaces, creating green roofs, and improving urban planning can be proposed to reduce the urban heat island effect (Aram et al., 2019).

In recent years, the Austrian tourism industry has already made a considerable shift away from non-renewable energy sources (Umweltbundesamt, 2023b): Between 2008 and 2019, the share of renewable energy in the accommodation and restaurants sector has increased from 36 to 54 %, which is above both the Austrian average of 34 % and the Austrian government's target of 50 % for 2030 (Pröbstl-Haider et al., 2020). This ongoing shift to renewable energy is necessary at the expense of non-renewable sources (see Federal Statistics Office, 2023, for where this is not the case). In Austria, several federal government programs offer financial support for companies to encourage this shift.

In winter tourism destinations, technological innovations are needed to find new ways to intensify snowmaking due to lower precipitation and higher winter temperatures, while improving the efficiency of water storage and other processes to reduce the growing demand for energy. The economic benefits should be carefully weighed against potential ecological impacts and problems associated with snowmaking, such as energy consumption, resource scarcity, and biological degradation (Aigner et al., 2024) (also see Section 2.2 on ecosystem services).

However, the potential for adaptation is limited. Due to regional climate conditions, businesses and skiing resorts, especially at lower altitudes (Marty, 2013), have already closed due to economic unviability (Falk, 2013). As climate change progresses, regional 'tipping points' will be crossed, making alpine skiing impossible in certain (mostly lower-lying) alpine regions (Ginkel et al., 2020). The exact point at which skiing becomes impossible depends on several factors, including the intensity and pace of warming, the availability of technological adaptations, local geography, and snowmaking capacity (Damm et al., 2014; François et al., 2023a).

#### 4.4. Industry and manufacturing

# 4.4.1. Status quo

The industry and manufacturing sector<sup>1</sup> covers a wide range of production processes needed for the transformation of materials and energy into intermediate and finished goods. Manufacturing firms in Austria tend to use state-of-the-art technology, but energy intensity is relatively high compared to other Western European countries due to the high share of energy-intensive metal and paper industries (Diendorfer et al., 2021). Between 2014–2020, the sector contributed 18 % to Austria's value added, growing by 2 %/yr. Due to its high emission intensity, the sector's share of Austrian emissions was significantly higher over the same period, but stable (fluctuating between 32–35 %), with absolute emissions of 26 MtCO<sub>2</sub>eq in 2020 (Statistik Austria, 2023a, 2023c). Despite increasing output, absolute emissions in industry have remained relatively stable since 2005, mainly because of energy efficiency improvements (Umweltbundesamt, 2023a), indicating a relative decoupling from economic growth.

Emissions are highly concentrated in a small number of sectors and firms. In 2020, more than 80 % of the sector's emissions (or 29 % of total national emissions) came from four groups of products (see Figure 4.6): Basic metals (12 MtCO<sub>2</sub>eq, 16 % of national emissions), other non-metallic mineral products (5 MtCO<sub>2</sub>eq, 7 %), chemicals and chemical products (2.2 MtCO<sub>2</sub>eq, 3 %), and paper and paper products (1.9 MtCO<sub>2</sub>eq, 2 %) (Statistik Austria, 2023f, 2023e, 2023c).

# 4.4.2. Current and future systems

The following sections present sectoral perspectives and economy-wide effects in the Austrian context. More details on mitigation options and emission reduction potentials per sector are given in Table 4.A.1. Global trends for each sector can be found in the appendix as well.

### Steel

In Austria, the metal sector is relatively large, with steel being the most important product. In 2020, the sector 'manufacture of basic metals' generated 1.1 % of the economy's value added (EUR<sub>2023</sub> 4.45 billion) and employed more than 37,000 people (Statistik Austria, 2023a, 2023b). Emissions come mainly from the use of coal/coke as a reducing agent and from the high temperatures required for steel production (see Figure 4.6 for details). Steel output has been stable since 2018, between 7–8 Mt/yr (WSA, 2023a), of which 90 % is produced via the blast furnace-basic oxygen furnace (BF-BOF) route (WSA, 2023b). The specific energy demand of crude steel in Austria is 4.9 MWh/t, its emission factor is 1.7 tCO<sub>2</sub>eq/t which is below the global and EU averages

<sup>&</sup>lt;sup>1</sup> In this section the industry and manufacturing sector is defined as NACE sectors 10–32, if not stated differently.



Figure 4.6 Production-based direct GHG emissions and energy inputs by energy product (covering also feedstocks) and sector. The sector 'manufacture of coke and refined petroleum products' was excluded from 'other manufacturing'. Sources: Physical Energy Flow Accounts (Statistik Austria, 2023f) and Air Emissions Accounts (Statistik Austria, 2023c).

(5.5 and 1.9 respectively) (Alton et al., 2022; Rahnama Mobarakeh and Kienberger, 2022). 88 % of emissions are process-related and 12 % are combustion-based (Rahnama Mobarakeh and Kienberger, 2022).

Emissions can be reduced by increasing circularity, especially by increasing the proportion of secondary steel (avoid). It is estimated that scrap-based steel production emits only one-eighth of the carbon compared to the conventional BF-BOF route (Wang et al., 2021). In the long term, the Austrian steel sector plans to switch to a hydrogen-based direct reduced iron-electric arc furnace (DRI-EAF) route (using natural gas as a bridging fuel until hydrogen becomes economical; shift). At constant output, this would require an additional 30–33 TWh of electricity (50 % of today's demand in Austria) (Mayer et al., 2019b; Alton et al., 2022). Efficiency improvements (improve) can further reduce emissions, but emissions are already close to the best available technology (BAT) of 1.6 tCO<sub>2</sub>eq/t (Material Economics, 2019).

Mayer et al. (2019b) derive Austria-specific unit costs of  $EUR_{2023}$  551/t of steel for a low-cost specification (Plasma Direct Reduction) and  $EUR_{2023}$  760/t for a high-cost specification (H<sub>2</sub> based DRI), compared to the current BF-BOF route at  $EUR_{2023}$  500/t. However, these figures are highly sensitive to electricity price assumptions. Other estimates range between  $EUR_{2023}$  460–1,020/t, covering (in order of increas-

ing cost): Recycling via EAF, gas-based DRI-EAF, gas-based DRI-EAF with carbon capture and storage (CCS), hydrogen-based DRI-EAF, smelting reduction with CCS as well as material efficiency and circularity (Material Economics, 2019; IEA, 2020).

Austria-specific 'greenfield' investment requirements for carbon-neutral steel production technologies are about  $EUR_{2023}$  1,340/t of annual capacity (corresponding to capital expenditures (CAPEX) of about  $EUR_{2023}$  120/t) (Mayer et al., 2019b). In the international literature, the range is  $EUR_{2023}$  670–1,050/t (Chiappinelli et al., 2021). Investment requirements are estimated to be 60–65 % higher for the implementation of new processes or CCS, but only 25 % higher (or even below baseline levels in the long term) for a circular economy pathway (Material Economics, 2019).

# Cement

In Austria, the sector 'manufacture of other non-metallic mineral products' directly generated 0.8 % of the value added in 2020 (EUR<sub>2023</sub> 3.1 billion), employing 30,500 people (Statistik Austria, 2023a, 2023b). Emissions come mainly from geogenic process-emissions and the combustion of gas and waste for high-temperature heat in cement production (Rahnama Mobarakeh and Kienberger, 2022) (see Figure 4.6). In 2020, Austria produced 5.2 Mt of cement and is equipped with highly efficient dry kiln factories, producing at 0.84 MWh/t of cement, generating 0.5 tCO<sub>2</sub>/t cement (Rahnama Mobarakeh and Kienberger, 2022).

In the Austrian context, short/medium term solutions are improve-measures such as circular economy (including recycling) and material efficiency increases, further increases in energy efficiency, the use of waste fuel with higher heating value content as well as biomass. Specifically, the share of biomass in alternative waste fuels (AWFs) could be increased (depending on availability; see Section 2.3.2) until AWFs are replaced by biomethane or hydrogen (fuels with higher calorific values) by mid-century. However, this would only reduce heat-related emissions. Since a large share (60 %) of CO<sub>2</sub> emissions are process-related (coming from the chemical reaction to decompose limestone (CaCO<sub>3</sub>) into lime (CaO) and CO<sub>2</sub>), CCS/CCU (improve) is currently seen as an option to reach carbon neutrality in the long term (Plaza et al., 2020; Bashmakov et al., 2022; Rahnama Mobarakeh and Kienberger, 2022) (see also Section 4.6). Specifically, 'cement electrolysis' with CCUS is currently being researched (mostly in the US), indicating both technical and economic viability (Mowbray et al., 2023). Energy demand for the non-metallic minerals sector in Austria is expected to increase by 25-90 % (2-9 TWh), mainly due to higher electricity demand for CCS. Natural gas and coal are expected to be replaced by hydrogen (Alton et al., 2022).

Production costs of carbon-neutral cement are estimated to be 70–115 % higher than the reference technology when using CCS (Material Economics, 2019). The cheapest alternative technology is oxyfuel CCS with cost estimates between  $EUR_{2023}$  100–110/t clinker. Other technologies range between  $EUR_{2023}$  130–150/t clinker. In general, higher CAPEX, higher fixed operating costs, and the higher electricity and steam requirements drive the cost increase (Gardarsdottir et al., 2019). Investment requirements are estimated to increase by 22–49 %, with a circular economy pathway demanding the lowest additional investment (Material Economics, 2019).

# Chemicals

In Austria, the sector 'manufacture of chemicals and chemical products' contributed with 0.9 % (EUR<sub>2023</sub> 3.85 billion) to total value added, employing 18,600 people in 2020 (Statistik Austria, 2023a, 2023b). Energy demand and emissions are mainly driven by low and high temperature heat provision via natural gas (for details, see Figure 4.6).

Specifically for Austria, the methanol-to-olefin (MTO) process appears to be a promising option (Alton et al., 2022), which could replace naphtha (i.e., a fossil resource) as feedstock (improve). However, this requires methanol, which could be produced from biomass (depending on availability; see Section 2.3.2 on a sufficiency oriented bioeconomy) or hydrogen and captured CO<sub>2</sub> from other sectors (e.g., cement). Such production of the feedstock would increase energy intensity, but (within the system boundaries of the chemicals industry) the process could become a carbon sink, by absorbing captured carbon from other sectors. Ammonia could be completely synthesized from hydrogen. Following the decarbonization pathways described in (Alton et al., 2022), the total energy demand of the Austrian chemicals sector (28 TWh in 2020) is projected to increase strongly, reaching between 53-63 TWh in 2050, of which 8-9 TWh will be electricity, the rest hydrogen and biofuels as feedstock. This is due to the higher energy intensity of the processes compared to the status quo (particularly the MTO process). Taking into account the electricity required for hydrogen production, electricity demand would increase by 42 TWh.

Switching to low-carbon production routes in the chemicals sector is estimated to increase costs as the feedstock has to be synthesized. For plastics, cost increases are estimated at 20–43 % (Material Economics, 2019). For ammonia and methanol, cost increases are expected to range from 25–90 % depending on the technology and feedstock, with electrification options being the most expensive and CCU-based options less so (IEA, 2020). Investment requirements in the plastics industry are estimated to increase by 122–199 % (Material Economics, 2019).

# Pulp and paper

In Austria, the sector 'manufacture of paper and paper products' accounted for 0.7 % (EUR<sub>2023</sub> 2.76 billion) of national value added in 2020 and employed 17,000 people (Statistik Austria, 2023a, 2023b). Direct emissions mainly come from natural gas combustion for low-temperature heat for drying processes (Figure 4.6 gives details). In 2020, 1.6 Mt of pulp and 5.6 Mt of paper and board were produced in Austria (CEPI, 2023).

Short to medium term mitigation options in Austria are efficiency improvements and fuel switching (to electricity) (improve). The emission intensity can still be decreased, which is currently  $0.38 \text{ tCO}_2/\text{t}$  paper (compared to the EU average of 0.33). In addition, heat management (recovery of

waste heat) can be further improved and alternative drying technologies used. Natural gas in boilers could be replaced by biofuels (depending on availability, see Section 2.3.2 on a sufficiency oriented bioeconomy), or non-electric boilers could be replaced by electric boilers (improve). In the long term, further electrification and installation of heat pumps and an increase in the share of biofuels in combined heat and power (CHP) can replace natural gas (the only source of direct emissions). The low process emissions from lime kiln (3 %) could be phased out through CCS/CCU (Rahnama Mobarakeh and Kienberger, 2022) (improve). In terms of energy demand, electricity demand could increase significantly from 2 to 7 TWh if heat pumps are used on a large scale, replacing 6 TWh of natural gas (Alton et al., 2022).

Cost estimates for decarbonizing the pulp and paper industry are scarce. Abatement costs from full biomass firing are estimated to be around  $EUR_{2023}$  54/tCO<sub>2</sub>eq (Bashmakov et al., 2022). Economy-wide estimates of investment requirements for decarbonizing low-temperature heat range from  $EUR_{2023}$  320–570 million (with the cheapest option being heat pumps) and the conversion of CHP plants is estimated to require  $EUR_{2023}$  140–230 million (Diendorfer et al., 2021).

# Chapter Box 4.1. Integrated scenarios for industry decarbonization in Austria

The study by Alton et al. (2022) presents two mitigation pathways for the industry and manufacturing sector in Austria up to 2050, with the goal of achieving full decarbonization. The results for the four energy-intensive sectors steel, cement, chemicals, and pulp and paper are summarized in Table 4.1. The first scenario, called 'Pathway of Industry' (POI), was developed based on an intensive stakeholder dialogue and represents their expected developments until 2030, with extrapolations until 2050. The second scenario, called 'Zero Emission' (ZEM), is based on a back-casting approach and reaches carbon neutrality by 2050, assuming breakthrough technologies. In the steel sector, both scenarios assume a shift to direct reduction (at constant steel output), using mainly (natural-, bio- and synthetic) gas in the POI scenario and mainly hydrogen in the ZEM scenario. In the chemicals sector, a shift from fossil resources to hydrogen, biofuels, biomass, and electricity is assumed. Due to the assumed methanol-to-olefins process, the production of methanol increases strongly, which also implies a higher energy demand for synthesizing feedstocks. The cement industry (included in non-metallic minerals) is assumed to deploy CCS technologies (amine scrubbing in POI and oxyfuel in ZEM), which increases electricity demand. In the **pulp and paper** sector, emissions are reduced by increasing the share of biofuels and using hydrogen in POI, whereas in ZEM electrification through heat pumps leads to zero emissions and lower energy demand. Overall, emissions from industry and manufacturing are reduced by 98 % (to 0.6 Mt in POI) and 106 % (to -1.6 Mt in ZEM). Compared to 2020, energy demand increases by 47 % in POI and 50 % in ZEM, with the largest increases in the chemicals sector (+227 % in POI and +294 % in ZEM).

	2020					2050 POI						2050 ZEM								
	Energy demand [TWh]				[Mt CO <sub>2</sub> eq]	2eq] Energy demand [TWh]				[Mt CO <sub>2</sub> eq]			Energy demand [TWh]				[Mt CO <sub>2</sub> eq]			
	Electricity	Hydrogen	Hydrogen transformation losses	Other fuels and feedstocks	Total	Emissions	Electricity	Hydrogen	Hydrogen transformation losses	Other fuels and feedstocks	Total	Excl. hydrogen	Incl. hydrogen	Electricity	Hydrogen	Hydrogen transformation losses	Other fuels and feedstocks	Total	Excl. hydrogen	Incl. hydrogen
Steel	2.6	-	-	34.2	36.8	12.8	4.8	7.1	3.0	15.6	30.5	0.7	0.8	4.8	19.6	8.4	4.3	37.1	0.8	1.2
Chemicals	4.4	-	-	11.5	15.9	3.9	9.3	19.2	8.2	15.3	51.9	-2.4	-2.0	8.3	23.2	10.0	21.1	62.6	-4.8	-4.3
Non-metallic minerals	1.9	-	-	7.8	9.7	5.2	7.2	2.3	1.0	6.0	16.5	1.0	1.0	4.9	1.3	0.6	5.3	12.1	0.8	0.9
Pulp and paper	2.3	-	-	19.5	21.7	2.2	2.3	4.1	1.8	16.0	24.3	0.0	0.1	7.1	0.0	0.0	12.2	19.3	0.1	0.1
Rest of manufacturing	12.7	-	-	17.4	30.1	5.4	24.9	6.4	2.7	10.2	44.3	0.6	0.7	23.4	3.4	1.4	12.2	40.4	0.4	0.5
All manufacturing industries	23.8	0.0	0.0	90.4	114.2	29.5	48.6	39.1	16.7	63.1	167.5	-0.2	0.6	48.5	47.5	20.4	55.2	171.5	-2.6	-1.6
2050 POI/2020	+104 %			-30 %	+47 %	-98 %														
2050 ZEM/2020	+104 %			-39 %	+50 %	-106 %														

Table 4.1 Current (2020) and future (2050) energy demand, emissions for the four sectors steel, chemicals, non-metallic minerals, and pulp and paper, as well as for rest of manufacturing in Austria for two scenarios (POI and ZEM) (Source: Alton et al., 2022; Nagovnak et al., 2024). Note that numbers do not match Figure 4.7 due to differences in system boundaries.

#### Other manufacturing sectors

The GHG emissions from the remaining manufacturing subsectors (2 MtCO<sub>2</sub>eq) account for 8 % of emissions from industry and manufacturing (Diendorfer et al., 2021) (see Figure 4.6). The emissions can be attributed to a large and heterogeneous number of manufacturing firms, mostly operating in the construction industry, followed by mechanical engineering, food and beverages, and woodworking industries. The same sectors are also the main emitters of GHG emissions, although other relevant processes include mining, the processing of non-ferrous metals, vehicle construction, and textiles. Due to the lower energy intensity of these sectors, only nine manufacturing plants are covered by the EU-ETS.

Although relatively less energy intensive overall, a significant share of the energy demand in the 'other manufacturing' sector is due to energy-intensive processes, requiring targeted mitigation measures. In the construction sector, by far the largest technical decarbonization potential rests in the electrification of engines or the switch to carbon-neutral gas for the operation of machinery and off-road vehicles (improve). In other sectors, significant reductions in GHG emissions could be achieved by installing heat pumps (shift) to meet the energy demand for process heat (<200°C), which is particularly high in the food and mining industries, and for room heating, for which natural gas remains a common energy source in machine engineering and the woodworking industries (Diendorfer et al., 2021).

# Economy-wide effects

Production costs and prices of carbon-neutral materials and goods are expected to be higher, yet the direct impact on consumers is estimated to be small. Price increases for typical final demand goods (such as plastic bottles, cars or houses) are estimated at 1-5 %, despite large price increases for materials (cement: 35-115 %, steel: 10-50 %, and primary chemicals: 15-115 %) (Bataille et al., 2018; Material Economics, 2019; Bashmakov et al., 2022). The effects on consumption possibilities (welfare) could be stronger if changes in income or employment are included. Bachner et al. (2020a), analyzing the transition of the European steel sector, show income (GDP) effects for Austria between +0.5 % and -1.5 %, depending on the assumption of production factor availability (idle or not). In particular, capital owners could benefit from distributional effects due to higher capital intensities (Bachner et al., 2020b). Regarding the timing of decarbonization, early action may be less costly in terms of foregone GDP per avoided ton of  $CO_2$  (Bachner et al., 2020b; Steininger et al., 2021). For labor-market implications, see Section 4.7.

# 4.4.3. Climate related risks and adaptation of industry and manufacturing

Industry and manufacturing are affected by physical risks related to climate (change). Energy-intensive firms are often located along rivers for access to water and waterways, making them vulnerable to water scarcity and fluvial flood risk (Bashmakov et al., 2022). For Austria, Bachner et al. (2023) estimate the current direct flood risk via sectoral capital stock damages for the whole economy, represented by 72 sectors. Out of a total of 18 manufacturing sectors, 12 are among the top 30 sectoral losers, with the highest direct risks in the 'fabricated metal products', 'chemicals and chemical products', 'basic metals' and 'food products'. As the flood risk in Austria is expected to increase (see Cross-Chapter Box 1), a sectoral analysis of the future flood risk in Austria is necessary. Heat-induced labor productivity losses are also an important risk, both globally (Dasgupta et al., 2021) and for Austria (Urban and Steininger, 2015). Next to direct risks, indirect risks via supply chains are also a concern for industry (Bednar-Friedl et al., 2022b). Extreme weather events abroad (e.g., floods), but also gradual climate change (e.g., rising temperatures and resulting losses in labor productivity) can trigger negative economic effects domestically (Knittel et al., 2020, 2024).

In terms of adaptation, European industry has been found to have a low level of adaptation, with SMEs often lacking sufficient knowledge (Bednar-Friedl et al., 2022a). Little is known about adaptation in Austria's industry sector. Key adaptation options for industry are knowledge creation, research and development (R&D), and risk management, with governments and international communities seen as key actors for adaptation (Bednar-Friedl et al., 2022b). Some industries could benefit from adaptation, as demand for materials such as concrete or steel could increase, for example for flood protection or sea walls (Bashmakov et al., 2022). Relocation could also change exposure to natural hazards.

# 4.4.4. Necessary transformation of industry and manufacturing

Direct and indirect electricity demand is expected to grow strongly (Lechtenböhmer et al., 2016; Geyer et al., 2021; Nagovnak et al., 2024). Direct demand is estimated to increase by 110-130 % at constant production levels (Gever et al., 2021; Alton et al., 2022). Thus, in addition to avoid measures, material efficiency (i.e., reducing material demand), and energy efficiency improvements, the transition to a fully renewable electricity supply is key to a system-wide decarbonization (Material Economics, 2019; Rissman et al., 2020; Geyer et al., 2021; Edelenbosch et al., 2024). Under insufficient domestic supply (see Table 4.A.3 for the range of estimates), imports could be a complementary strategy. However, uncertainties regarding the stability of the electricity grid and the flexibility of the energy system have been identified as key concerns (Bachner et al., 2020b). Due to these uncertainties, industries could relocate to regions with high renewable energy availability at low prices (Samadi et al., 2023; Verpoort et al., 2024) and high potential for CCS (Bashmakov et al., 2022), thus requiring a societal debate on this 'renewables pull' effect (Verpoort et al., 2024).

Estimated investment requirements for decarbonizing steel, plastics, ammonia, and cement in the EU are 0.2 % of gross fixed capital formation, but for individual firms, investment in (often costlier) low-carbon technologies might be too risky due to high uncertainties (see Section 4.4.2 for sectoral investment requirements). A particular feature is the high share of 'process emissions' (emissions from non-combustion processes in production), which require fundamentally new tailor-made processes and equipment (Odelenbosch et al., 2022). These are mostly available at relatively high levels of technological readiness (Alton et al., 2022), but some key technologies and their integration into existing process chains require further R&D (Gailani et al., 2024). Crucially, investment incentives are often weak due to the characteristics of these industries, the risk of asset obsolescence, and uncertainties about future technological and commercial viability (Wesseling et al., 2017; Bataille et al., 2018; Material Economics, 2019; Zhang et al., 2022). However, there are opportunities for first-movers to gain competitive advantages if the appropriate market framework is established (Karkatsoulis et al., 2016; Bachner et al., 2020b; Zhang et al., 2022). To achieve a decarbonized industry, a mix of measures is needed, in particular investment support schemes, contracts-for-difference, market-based instruments (CO<sub>2</sub> pricing, border carbon adjustment) and a strengthening of circular economy principles (Edelenbosch et al., 2024) (see also Chapter 6).

In industrial SMEs, where the energy intensity of core production processes is low, support processes (e.g., lighting, ventilation, heating) can account for a significant share of energy demand and have been found to be the most promising intervention points for achieving reductions in energy use (Johansson et al., 2019). The main firm-level barriers in industrial SMEs relate to lack of information and the insufficient financial viability of investments and alternative procurement models such as energy contracting (Fresner et al., 2017; Johansson et al., 2019; Hrovatin et al., 2021). A cost-effective policy mix to reduce energy demand in industrial SMEs provides tailored support through a combination of subsidized energy audits and subsidies for investments in technologies (Johansson et al., 2019). For more energy-intensive medium-sized enterprises, supporting the implementation of energy management systems, facilitating learning networks, and regulatory policies can increase the adoption rate of energy efficiency measures (Thollander et al., 2015; Paramonova and Thollander, 2016; Jalo et al., 2021; Johansson et al., 2022).

### 4.5. Energy supply systems

# 4.5.1. Characteristics of the current energy system/status quo

Although fossil fuel use declined from 75 to 66 % of total primary energy consumption when comparing the period 2003-2007 to 2019-2023, the Austrian energy system is still largely based on fossil fuels (Statistik Austria, 2024a) (see Figure 4.7a). Coal use shows the strongest decline (-35 %), followed by fossil oil use (-18 %). Gas consumption decreases slightly over the whole period (-6 %). The decline in fossil fuel use is made possible by a stabilization of gross energy use (improve measures), and a switch from fossil fuels to biomass (53 % increase), ambient heat (216 % increase), wind and solar PV (707 % increase), and hydropower (10 % increase) (shift measures) (see Figure 4.7a). As a consequence of the high use of fossil fuels, about 58 % of primary energy was imported on average in the period 2019-2023 (see Figure 4.7b). This affects Austria's energy trade balance and comes with several other challenges related to, inter alia energy security, affordability, and flexibility (see Section 4.2). At the pace of the energy transition until 2021, it is impossible to achieve a carbon-neutral Austria by 2040 (Auel and Schmidt, 2023).

The sector 'production' slightly increased its final energy consumption (+4 %), 'households', 'services' and 'other' decreased final energy consumption by 1 %, and 'transport' slightly decreased by 2 % when comparing 2003–2007 with 2019–2023 (see Figure 4.7c and d). In 2019–2023, natural gas was mainly used in 'production' (58 %), followed by 'households', 'services' and 'other' (39 %). Natural gas use in the 'transport' sector was low (at around 3 %). Compared to the period 2003–2007, 'industry' and 'production' increased their share by 8 % at the cost of 'households', 'services' and 'others'.

Fossil oil is mainly used in transport (82 %), and this share even increased from 72 % in the period 2003–2007. 'Households', 'services', and 'other' have reduced their use of fossil oil for heating (the share decreased from 21 to 14 %). Overall, fossil oil consumption in the 'transport' sector decreased slightly from 96 TWh (2003–2007) to 89 TWh (2019–2023), taking into account the very weak years in terms of transportation consumption due to COVID-19 restrictions (Hartwig et al., 2022; Staehle et al., 2022). Coal is mainly used in the 'production' sector (97 %), mainly for steel production. Compared to the period 2003– 2007, this share increased by 17 %, as 'households', 'services', and 'other' have significantly reduced their coal consumption, mainly due to the fading out of coal as a heating fuel (shift measure). Coal use in the 'transport' sector has virtually disappeared.

The use of renewables has increased in absolute terms in all sectors, but the share of the 'transport' sector in all renewables increased from about 5 % in the period 2003–2007 to 10 % in the period 2019–2023, and it decreased in the 'production' sector from 34 to 30 %, while the sector 'other' remained constant.

As a consequence of the slow pace of the energy transition, fossil fuel combustion is still by far the largest emitter



**Figure 4.7** (a) Gross domestic energy consumption by fuel in Austria for 2003–2023 based on Austrian energy balance. (b) Import shares by fuel. (c) Structure of final energy demand by sector. (d) Final energy use by fuel and sector. The numbers on the right side of (a) and (c) show the growth rate in percent from the period 2003–2007 to 2019–2023, e.g., coal use has declined by 35 % in the period. The category 'other' includes households, services and agriculture. (Source: Energiebilanzen, Statistik Austria, 2024a).

of greenhouse gases in Austria (66 % in 2018–2022, compared to 71 % in 2003–2007) (Umweltbundesamt, 2025).

Austria's energy infrastructure is characterized by geographical and technological peculiarities. In the electricity sector, hydropower accounts for a large share of electricity generation in most of the federal provinces, with the exception of Burgenland and Vienna; wind power plants are located mainly in Burgenland and Lower Austria, and to a lesser extent in Styria. The Austrian electricity grid has a length of about 250,000 km. As far as the gas infrastructure is concerned, there are currently 47,000 km of gas network in Austria. In both cases, the vast majority is at the distribution network level. The Austrian gas storage facilities are located in Upper Austria and Salzburg (about 75 %) and in the east of Lower Austria (about 25 % of the total Austrian storage volume). Due to its location in Central Europe, Austria is an important energy transit country and the related infrastructure is also relevant for neighboring countries (BMK, 2024a).

# 4.5.2. Characteristics of the future energy system

A future carbon-neutral energy system can rely on a combination of domestic energy supply technologies and imported energy carriers, i.e., wind power, solar PV, biomass, geothermal energy, hydropower and imported hydrogen, other synthetic fuels and fossil fuels with carbon capture and storage (shift measures, strong co-benefits between these mitigation options and SDG 7) (see Table 4.A.3). However, there is little agreement on the best mix of these technologies, as many different technological pathways allow achieving climate neutrality in the Austrian energy system, and the underlying scenario processes make significantly different assumptions about the potentials of different technologies (see Figure 4.8). However, there is a consensus that renewable energies, especially wind power and solar PV, need to be strongly expanded, and that the electrification of land transport and heat supply is crucial. Both measures are a common driver of decarbonization in all available scenarios up to 2030, but the scenarios differ more after 2030. Electrification causes a substantial decrease of gross domestic energy consumption of 50-100 TWh by 2040 (BMK, 2024a; Schmidt et al., 2025). Under the assumption of unlimited potentials and from an energy system perspective, wind power fits seasonal demand and hydropower supply patterns better than solar PV. Systems with a higher share of wind power therefore have a higher level of reliability at the same cost or the same level of reliability at a lower cost (Scholz et al., 2017; Maeder et al.,



Figure 4.8 Gross domestic energy use by energy carrier in Austrian net-zero scenarios. The reference data shown in this figure is available at <u>aar2</u>. <u>ccca.ac.at/explorer</u>.

2021; Wehrle et al., 2021; Schmidt et al., 2025). Furthermore, if synthetic fuels are available as imports, domestic electricity generation can be reduced, as the production of these fuels, especially for use in hard-to-abate industries, is electricity intensive (Schmidt et al., 2025). Similarly, biomass is a versatile fuel that can replace the use of synthetic fuels, but sustainable potentials are limited in Austria (Schmidt et al., 2025). Notably, in climate neutrality scenarios, biomass will be used in different sectors than today: Its use will be reduced in domestic heating as it is replaced by heat pumps and thermal insulation, while biomass will be instead upgraded to biogenic methane for use in high-temperature processes in industry (the scenario data used in this analysis is available at <u>aar2.ccca.ac.at/explorer</u>).

The electrification of energy services supports the achievement of mitigation goals, as it allows for highly efficient delivery of energy services and the direct use of variable renewable electricity generation (Luderer et al., 2022; Sejkora et al., 2022; Schmidt et al., 2025) (the scenario data used in this analysis is available at aar2.ccca.ac.at/explorer). Therefore, final consumption of electricity increases in all scenarios. Electrification means that sectors that previously operated in isolation will become much more integrated (Brown et al., 2018), such as the power sector, mobility, and heating and cooling, as formerly fossil-fueled processes in these sectors are electrified, for instance using heat pumps and electric cars (Schmidt et al., 2025). Further energy efficiency measures such as high insulation rates and building standards for new construction, as well as measures to reduce the level of energy services on the demand side, such as a reduction in heated area or driven distances, will help limit the necessary infrastructure expansion on the supply side (avoid measures) (the scenario data used in this analysis is available at <u>aar2.ccca.ac.at/explorer</u>) (see Sections 4.2, 3.3, 3.4). In the transportation sector, demand-side measures are generally very important, especially if a very rapid electrification of cars and trucks, far beyond current penetration rates, is not achieved (Schmidt et al., 2025) (see Section 3.4). However, changes in demand for land transport services in a fully electrified land transportation sector will only have a limited impact on total gross domestic energy consumption due to the high efficiency of electric drives (Schmidt et al., 2025). In contrast, a major reduction in energy demand by industry is the most important driver of reductions in gross domestic energy use in decarbonization scenarios (Schmidt et al., 2025).

Scenarios of a climate neutral Austrian energy system indicate that hydrogen, synthetic gases and fuels will be used primarily in industry, air and ship transportation, and partly to balance the power grid (see Figure 4.9; the scenario data used in this analysis is available at aar2.ccca.ac.at/explorer), as other uses are too costly (Ueckerdt et al., 2021). Hydrogen and synthetic methane play an important role in industry, where their use can only be avoided by expanding carbon capture and storage in combination with fossil fuels. As intensive carbon capture and storage pathways have not been extensively assessed in integrated energy system scenarios in Austria, the use of hydrogen, synthetic gases and fuels in industry is high in the available scenarios (Schmidt et al., 2025) (the scenario data used in this analysis is available at aar2.ccca.ac.at/explorer). However, there are significant differences between the scenarios: First, if Austria does not produce raw iron, but uses either scrap steel or imports direct reduced iron pellets (shift and avoid measure), the demand for hydrogen in the industry sector decreases significantly (Sections 4.2 and 4.4). Second, if the availability of these fuels is limited or if they are expensive, the scenarios instead show increased electrification of industry, higher utilization of hydrogen than synthetic methane, and increased use of biomass upgraded to biomethane. Furthermore, synfuels are relevant for replacing fossil fuels in air and ship transport (see Section 3.4). In district heating and power generation, the use of hydrogen or synthetic methane depends on the scenario assumptions and the variability between scenarios is high. In land transport, synthetic fuels are used extensively only in scenarios where uptake of electrification is exogenously limited. For households and services, synthetic gas is exclusively used because of the lock-in of gas-based infrastructure for heating, i.e., the long lifespan of today's heating infrastructure requires the use of some synthetic gas as a substitute for fossil gas. Hydrogen, synthetic fuels, and gas are either imported or produced domestically, depending on relative cost assumptions and assumptions about the expansion of domestic renewables. In all scenarios, how-



Figure 4.9 Use of hydrogen, synthetic gas and synthetic fuels in different applications in the energy system (74 scenarios of a climate neutral Austrian energy system; the scenario data used in this analysis is available at <u>aar2.ccca.ac.at/explorer</u>).

ever, dependence on imports of energy carriers is reduced compared to today because the use of hydrogen, synthetic gases and fuels is less than today's fossil fuel consumption. At the same time, material dependencies may arise in the renewable energy sector (see Cross-Chapter Box 5). A further reduction in demand also leads to a further reduction in import dependency (Schmidt et al., 2025).

In terms of **infrastructure**, the expansion of the electricity grid is seen of high importance for the integration of increasing shares of renewable electricity in the energy system. Costs for this are estimated at EUR<sub>2023</sub> 8.75 billion by 2034 (BMK, 2024a). In the gas sector, a parallel infrastructure for renewable methane and hydrogen is planned and associated with costs of EUR<sub>2023</sub> 1.9 billion until 2050 (BMK, 2024a).

# Energy supply security and resilience of systems and infrastructure

A central question is how the Austrian decarbonized energy system has to be designed while maintaining security of supply and resilience, considering all aspects of mitigation and adaptation. Current energy systems have relatively large buffers as fossil fuels are easily storable. Renewables-based energy systems show a reduction in these buffers and an increased demand for flexibility (Kondziella and Bruckner, 2016; Suna et al., 2022). In electrified systems with large shares of variable renewables, the size of buffers and associated costs may increase due to higher variability of supply. At the same time, the acceptable risk of electricity system failure may decrease, as electrification will interconnect many different sectors.

Flexibility options in the carbon-neutral electricity system can be provided by (a) sector coupling (for more detailed information on energy storage in buildings and mobility, see Chapter Boxes 3.2 and 3.3), (b) short- and long-term storage, (c) carbon-neutral firm capacity, (d) demand-side management (e.g., industrial or residential load shifting) (Märkle-Huß et al., 2018; Christensen et al., 2020; Mascherbauer et al., 2022; Schwaiger et al., 2023), and (e) exports/imports (Resch et al., 2020; Haas et al., 2022; Suna







Figure 4.10 Flexibility/storage needs in the power sector in different scenarios with increasing shares of variable renewable energy. The values are based on 14 scenarios and different years and depicted for daily (a), weekly (b), monthly (c), and annual (d) flexibility needs. The status quo (2020) is based on Suna et al. (2022), 2030 is based on Suna et al. (2022) and Schöniger et al. (2023), 2040 is based on Schmidt et al. (2025), 2050 is based on Schöniger et al. (2023). Source: Own illustration.

et al., 2022; Lienhard et al., 2023; López Prol et al., 2023). There are different timescales for flexibility needs and options (see Figure 4.10), which will have different dynamics in the future. Scenarios with a higher solar PV generation share show higher daily (hourly fluctuations within a day) and annual (monthly fluctuations within a year) flexibility demand than scenarios with a higher wind generation share due to the strong intra-day and seasonal characteristic of solar PV generation. Scenarios with a higher wind share show more fluctuations of the renewable electricity generation on a weekly and monthly timescale (Suna et al., 2022; Schöniger et al., 2023; Schmidt et al., 2025).

Harnessing the synergies of seasonal compatibilities of different energy demand and supply components can increase system stability (Jerez et al., 2013; Ramsebner et al., 2021; López Prol et al., 2024). Reducing energy demand in the building sector (see Section 3.3), for instance, reduces the need for seasonal storage, as there is a natural seasonal mismatch between heating demand (peak in winter) and renewable energy supply (surplus in summer) in Austria, which can be reduced in this way. Climate change impacts on demand and supply patterns might improve the seasonal correlation of renewable energy supply and the stability of energy demand due to reduced heating demand.

The transition to a more decentralized electricity system implies changes in system vulnerability that need to be further understood. Reducing energy demand is a no-regret measure to reduce the necessity of energy infrastructure and related risks (Kranzl et al., 2015) (see Section 4.2). Electricity systems with higher shares of renewable energies show lower sensitivity to fuel price shocks in Austria (Totschnig et al., 2017) and lower fuel import dependency (see scenario descriptions in Section 4.5.2).

In sectors where they can be used efficiently, fuels such as hydrogen, renewable gases, and other e-fuels might play a role and have to be domestically generated or imported as well as stored in the future energy system (see Sections 3.4, 4.2, 4.4). The efficient use of available renewable energy sources requires the use of e-fuels in hard-to-electrify sectors (Sections 4.6 and 3.4) and the use of synergies provided by electrolysis, for example the provision of electricity system flexibility (Reiter and Lindorfer, 2015; Greiml et al., 2021) and waste heat (Böhm et al., 2021).

# Climate change impacts on energy systems

The energy supply system will become increasingly linked to the climate system due to higher electrification and large shares of weather-dependent renewable energy (Cronin et al., 2018; Emodi et al., 2019; Yalew et al., 2020; Simoes et al., 2021; De Felice et al., 2023; Kapica et al., 2024). Climate change impacts the energy system in the fields of (i) energy supply, (ii) energy demand, and (iii) energy infrastructure due to natural hazards (Kranzl et al., 2015). Long investment cycles of energy infrastructure assets require timely adaptation of planning, building and operation to avoid lock-in effects (Seto et al., 2016; Fisch-Romito et al., 2021).

Climate change is expected to increase hazards that impact different parts of the energy system on both the supply and demand side: While heat waves and droughts particularly impact thermal electricity generation (cooling requirements) (Kranzl et al., 2010) and hydropower, transmission and other renewable generation technologies are more risk-sensitive to cold waves, wildfires, flooding, heavy snow, ice storms, and windstorms (Yalew et al., 2020). If energy infrastructure is not designed to be resilient, supply risks can increase, as higher weather dependence and more infrastructure damage from extreme events can lead to a stressed system (Mikellidou et al., 2018; Perera et al., 2020; Doss-Gollin et al., 2023). Climate change impacts on energy demand, supply, and infrastructure through extreme events (e.g., wind, droughts, cold waves, heat waves, floodings, wildfires, storms, and compound effects thereof) (Brás et al., 2023) or changes in long-term trends can lead to cascading effects and energy supply disruptions (Rübbelke and Vögele, 2011; Hossain et al., 2021). Energy systems, especially the increasingly important electricity system, are highly interconnected geographically, which can lead to large-scale spatial disruptions in the event of failure. Possible adaptation measures include reducing energy demand (Section 4.2) and cooling water demand, increasing the degree of interconnection of the energy system, increasing back-up capacity (flexible generation or storage), spatial and technological diversification of energy sources and flexibility options (Yalew et al., 2020) (see above), and integrating impacts of climate change on energy demand and supply into energy strategies and business decisions. An example of mitigation co-benefits is the improvement of energy efficiency in buildings, which makes them less vulnerable to climate change impacts (Zachariadis and Hadjinicolaou, 2014) (see Section 3.3). Energy system design in terms of generation mix, interconnection grade, and availability of flexibility options has a greater impact than weather variability between different years or climate scenarios (Totschnig et al., 2017; Bloomfield et al., 2021).

Table 4.A.2 gives an overview of climate change impacts on electricity demand and supply components in Austria, as identified in the current literature. The supply side is affected by temperature effects on thermal power plant efficiency and cooling requirements, and by changes in meteorological variables such as precipitation/evaporation, radiation, and wind speed on renewable electricity generation patterns (Kranzl et al., 2015). There is no clear trend regarding the impacts on the supply side (wind, solar PV, and hydropower generation) in Austria. Depending on the climate scenarios considered, either increasing or decreasing annual mean electricity generation is projected and has to be differentiated regionally (see Section 7.4.2 Energy). The impacts on hydropower are expected to be stronger than on wind and solar generation.

Additionally to Table 4.A.2, Figure 4.11 shows modeled relative changes of annual renewable electricity generation and annual heating and cooling demand for different global warming levels (GWL) compared to the reference period 1981–2010 based on Formayer et al. (2023) and Schöniger et al. (2023, 2024).

There is *low agreement* on the effects of climate change on the sign of change of overall hydropower generation in Austria (Stanton et al., 2016; Wagner et al., 2017), with some studies showing increasing (e.g., Simoes et al., 2021; Formayer et al., 2023) and others showing decreasing generation (e.g. Kranzl et al., 2010, 2015; Eitzinger et al., 2014; Schleypen et al., 2019). However, there is *high agreement*  that hydropower will show earlier runoff in spring, higher generation during winter, and lower generation during summer due to a change in the snowmelt period (Kranzl et al., 2010; Finger et al., 2012; Kling et al., 2012; Fuchs et al., 2013; Totschnig et al., 2017; Wagner et al., 2017; Blöschl et al., 2018; Formayer et al., 2023; Wechsler et al., 2023) (see Figure 4.12). This smoothed seasonality of hydropower is expected to decrease the need for seasonal storage (Kranzl et al., 2010; Blöschl et al., 2011). However, negative energy system impacts may arise from extreme events (droughts or flooding). For hydropower, an increase in interannual variability is observed, which poses a risk to the highly hydropower-based Austrian electricity system. However, the climate change impact on interannual hydropower variability is expected to be lower than the naturally high interannual variability of hydropower in Austria (until 2050) (Blöschl et al., 2018). (Pumped) storage hydropower plants are expected to be less affected by climate impacts due to their balancing capacity, which allows for some adaptation (Kranzl et al., 2010) (see Section 7.4.2). However, impacts of glacier melt dynamics are expected to limit the adaptive capacity of hydropower production with glacier catchments (Finger et al., 2012) and to affect the whole energy system in Austria (see Section 7.4.2).

On the demand side, climate change mainly affects the energy demand for heating and cooling (see Table 4.A.2).



**Figure 4.11** Climate change impacts on the Austrian energy system for different global warming levels (GWL) compared to the mean of the reference period 1981–2010. (a) Relative changes of the annual electricity generation of PV, wind onshore, and hydro run-of-river and (b) on annual cooling and heating demand are shown. For each GWL, the box shows the 20 weather years in which that level is reached in the underlying climate scenario (2026–2045 for GWL 2.0°C and 2073–2092 for GWL 4.0°C). Findings: For wind and hydro, a slightly increasing annual generation is shown, for PV no trend is visible. There is a clear relative increase in cooling and decrease in heating demand. Despite the strong relative increase of cooling demand, there is an overall negative net effect on heating and cooling energy demand in Austria due to higher absolute heating than cooling demand. (Own illustration, data points are based on the RCP8.5 scenario of Formayer et al. (2023) and Schöniger et al. (2024)).

With increasing global warming levels, an increasing trend in cooling demand and a decreasing trend in heating demand are projected for Austria (see also Sections 3.2.3, 3.3.1). This sign of change is robust across studies and models (Bachner et al., 2013; Berger et al., 2014b; Kranzl et al., 2014a; Steininger et al., 2015; Damm et al., 2017; Bird et al., 2019; Ramsebner et al., 2021). This implies a more even seasonal profile of electric load by decreasing demand during the cold seasons and increasing the load during warm seasons. As heating demand is still expected to be significantly higher than cooling demand in absolute terms, the net effect of climate change on energy demand in Austria is found to be negative even under the strongest GWL considered (Bachner et al., 2013; Totschnig et al., 2017; Formayer et al., 2023). Kranzl et al. (2015) found an 18-20 % reduction in electricity demand for heating and cooling by 2050. Climate change impacts and adaptation on the demand side (e.g., peaking need for electric cooling during heat waves) can have significant effects on supply-side requirements (Totschnig et al., 2017; Rode et al., 2021; Viguié et al., 2021). Literature for Austria shows that critical situations for the electricity system are still mainly expected during winter, but heat waves are shown to pose an increasing risk to the Austrian and Central European electricity system (Schöniger et al., 2023; Gruber, 2024). If future energy system design and policies consider these climate impacts to build a resilient



**Figure 4.12** Climate change impact on seasonal patterns of run-ofriver hydropower generation in Austria for global warming levels (GWL) 1.5°C, 2.0°C, and 4.0°C compared to the reference period 1981–2010. (Own illustration; based on Formayer et al. (2023): SDM\_ICHEC-EC-EARTH\_rcp45/85\_r1i1p1\_KNMI-RACMO22E climate projections for GWL and ERA5-Land for reference period; GWL 1.5°C and 2.0°C contain data from the RCP4.5 and RCP8.5 scenario, GWL 4.0°C only from RCP8.5 scenario; 7-days rolling mean capacity factors are displayed).

energy system (van der Wiel et al., 2019), energy supply disruptions and their cascading effects can be avoided.

There is a research gap for quantitative climate change impacts on critical energy infrastructure in Austria. For other countries, peak capacity losses for thermal power plants in the range of 1.1-4.6 % at the end of the century have been calculated (Sathaye et al., 2013). The issue of increased capacity losses in transmission lines and substations due to higher temperature levels is found to account for up to an additional 8 % and 3.6 % of peak capacity (Sathaye et al., 2013), with a particular high risk in no-wind conditions at high temperatures. Grid-connected energy infrastructure can be impacted by hazards such as flows, floods, landslides, permafrost thawing, rockslides and other extreme meteorological events such as heavy precipitation, high temperatures, or strong winds (Schaeffer et al., 2012). Possible adaptation measures include incorporating future cooling demands into power infrastructure design parameters, or rerouting transmission in situations of local impacts where possible.

In terms of macroeconomic effects, the shift of the electricity load towards the summer is expected to result in gains in the overall energy system on the one hand, and costs due to increased cooling demand and electricity generation costs (e.g., increased peak load production, reduced hydropower during dry periods) on the other hand. Climate change impacts in the electricity sector are found to increase annual costs in the sector by about EUR<sub>2023</sub> 302.22-840.96 million in 2050 and, due to spillover effects, reduce GDP by EUR<sub>2023</sub> 216.81 million in the 2030s and by EUR<sub>2023</sub> 613.64 million in the 2050s (Kranzl et al., 2014b, 2015). For heating and cooling, a positive annual net effect of EUR<sub>2023</sub> 39.42 million (2016-2045) and EUR<sub>2023</sub> 65.7 million (2036-2065), respectively, was calculated (Kranzl et al., 2014b). Based on the same dynamics, Steininger et al. (2020) report average annual costs of EUR<sub>2023</sub> 347.42 million (2030) (2050: EUR<sub>2023</sub> 926.44-1,304.33 million - low border RCP4.5, high border RCP8.5). All costs are calculated without considering infrastructure failures or blackout risks due to extreme events and natural hazards. In general, there is a literature gap on costs linked to energy infrastructure damages and large-scale energy supply disruptions in Austria. In the electricity system, costs of such large-scale energy supply disruptions are estimated to be orders of magnitude higher than those associated with general electricity generation and can reach the future costs of an entire year's electricity consumption (Kranzl et al., 2014b). Reichl et al. (2013) estimate the costs of a power outage to the Austrian economy on a weekday morning in summer to

be about  $EUR_{2023}$  504.35–576.4 million (6 h duration) and  $EUR_{2023}$  1,080.75–1,585.1 million (24 h duration), with the energy-intensive manufacturing sector bearing the largest share of these disruption costs.

# 4.5.3. The necessary transformation: Achieving a decarbonized energy system

Even under stringent demand-side reductions, the supply side of the energy system and the related grids will have to be strongly transformed to reach climate neutrality: Infrastructure will have to be partly expanded and partly built back (Schmidt et al., 2025) (the scenario data used in this analysis is available at <u>aar2.ccca.ac.at/explorer</u>). The technologies to harvest the low-hanging fruits of decarbonization, i.e., electrification of heating and mobility, and renewable electricity generation via solar PV and wind power, are readily available on global markets, but the expansion speed in Austria will have to increase considerably (Schmidt et al., 2025), although the maximum historical expansion of solar PV in 2023 is in line with climate neutrality goals (Statistik Austria, 2024a). However, to decarbonize industry, technologies and commodities are required, which are currently not available at scale, and where rapid upscaling will be challenging, such as for hydrogen production (Odenweller et al., 2022; IEA, 2023). For Austria, additional annual investment costs of a fully carbon-neutral economy are estimated to be in the range of EUR<sub>2023</sub> 6.2-10.8 billion, of which EUR<sub>2023</sub> 2.3-5 billion are for energy production, EUR<sub>2023</sub> 2.7-4 billion for transport, and EUR<sub>2023</sub> 0.7–0.9 billion for buildings (see Table 8.1). Relative costs depend, however, largely on uncertain assumptions about future fossil fuel prices, renewable energy (IEA, 2023) and capital costs (Bachner et al., 2019b).

The land resources required for the pathways to a carbon-neutral Austria do not technically limit the transition. This is particularly the case in scenarios where biomass use is not increased substantially above current levels: There is sufficient technical potential for wind power and solar PV (see Table 4.A.3) compared to biomass from agriculture or forestry, where complex trade-offs between energetic and other uses of natural resources exist (see Section 2.3.2). In particular, wind power has low direct land-use requirements (Turkovska et al., 2024), while solar PV land-use efficiency can be increased by using agri-photovoltaic schemes (Mikovits et al., 2024). However, biodiversity can be affected by energy system infrastructure: Birds, bats, and non-volant mammals are impacted by wind power, but adverse impacts can be mitigated to some extent, e.g., by the choice of turbine locations or by turning off turbines in periods with high risks for bat collisions (McKenna et al., 2025). Depending on prior land-use and management of PV parks, solar PV installations can be beneficial or detrimental to plant and pollinator diversity (Blaydes et al., 2021; Nordberg et al., 2021). While there are in principle sufficient raw materials available to enable a global energy transition by 2050 (Wang et al., 2023), their mobilization is nevertheless challenging and can entail important trade-offs, such as conflicts with traditional or indigenous populations (Owen et al., 2023) or negative environmental impacts (Berthet et al., 2024). A circular economy can reduce the needs for mining substantially (see Cross-Chapter Box 5), and ecologically responsible mining and sourcing can reduce environmental impacts (Sonter et al., 2023). Overall, the net requirements for mining will decrease as a result of the energy transition, as the reduction in fossil fuel mining overcompensates for the additional mining of raw materials used in low-carbon technologies (Nijnens et al., 2023). The risks, potentials, and uncertainties of carbon capture and storage technologies are discussed in more detail in Section 4.6.

In contrast, limited policy support is a crucial barrier in Austria: Implemented targets and corresponding policy support are still too low to achieve a fully carbon neutral Austria by 2040 (Umweltbundesamt, 2023b). Furthermore, even the federal targets for renewable energy expansion may not be met due to a mismatch with corresponding targets by federal states, e.g., for wind power (Baumann et al., 2021; BMK, 2023c). Even if policy support was sufficient on all governance levels, projects may fail due to a lack of acceptance by the local population (Scherhaufer et al., 2017; Sposato and Hampl, 2018; Seidl et al., 2019). The latter barrier may be partly overcome through compensation payments or participatory planning methods (Kapeller and Biegelbauer, 2020; McKenna et al., 2025). Additionally, power grid infrastructure is also subject to public opposition (Friedl and Reichl, 2016), but will have to be significantly expanded in carbon-neutral scenarios for the Austrian energy system (BMK, 2023c). Regulatory innovations such as energy communities can support the establishment of decentralized and decarbonized energy systems in a participatory manner (Haas et al., 2022). However, renewable energy subsidy schemes can increase income inequality if policies are not designed to distribute the potential welfare gains and losses from renewable energy uptake in Austria evenly (interaction with SDG10). The speed of the transition is also limited by the necessary expertise of the workforce, which can suffer from detrimental boom and bust cycles and needs

corresponding policy solutions (Briggs et al., 2022). Of all the required supply-side technologies, solar PV is the most on track, both in terms of the realized expansion as well as in terms of the match between federal and federal province level goals (BMK, 2023c). However, as the generation profile of solar PV is unfavorable in Austrian conditions, overbuilding solar PV may imply the use of more integration technologies (Wehrle et al., 2021; Schmidt et al., 2025).

Table 4.A.3 and Table 4.A.4 show the potential of different renewable electricity potentials and cost in Austria.

# 4.6. Decarbonization of hard-to-abate industries by carbon capture, geological storage, and utilization

Carbon Capture, Utilization and Storage (CCUS) refers to a chain of technologies to prevent  $CO_2$  (carbon) emissions by capturing carbon dioxide ( $CO_2$ ) from suitable point sources, such as industrial facilities, and then permanently storing the captured  $CO_2$  in the geological subsurface or using it as a feedstock for products (IPCC, 2005; Bui et al., 2018; Lehner, 2021; Ott, 2023). In combination with bioenergy (BECCS)

or direct air capture (DACCS) technologies, CCS also offers the potential for negative  $CO_2$  emissions and thus the direct and permanent removal of  $CO_2$  from the atmosphere (Bui et al., 2018; IPCC, 2023b); this is referred to as Carbon Dioxide Removal (CDR). A schematic overview is provided in Figure 4.13. CDR methods that do not involve CCUS technologies are not covered in this chapter. The reader is referred to the Cross-Chapter Box 6.

Analysis of sectoral climate policies shows that currently planned measures are not sufficient to reduce emissions to the extent necessary to achieve the climate goals of the Paris Agreement (Nascimento et al., 2022). In this context, CCUS is seen as an essential component for achieving climate goals (Bui et al., 2018; IPCC, 2023b). According to recent assessments for the EU, this would result in an average demand for carbon capture by CCS of 230–430 Mt/yr in 2030 and 930– 1,200 Mt/yr in 2050, while the scale of CCU deployment could cover a wide range of 47–800 Mt/yr in 2050 (Butnar et al., 2020). Consequently, the European Commission has identified the deployment of CCU and CCS technologies during this decade as critical (COM/2020/562 final). CCUS can play a crucial role in decarbonizing the Austrian economy, especially with regard to hard-to-abate industrial



Figure 4.13 Potential carbon sources, sinks, and a qualitative indication of capture effort and fixation periods. Modified from CaCTUS project proposal (ACRP, Klima + Energie Fonds, 2021) (Global CCS Institute, 2021; CaCTUS, 2024).

emissions. In particular, CCUS can support  $CO_2$ -intensive industries in Austria to meet  $CO_2$  reduction targets while maintaining their competitiveness, which requires a clear definition of which emissions are generally avoidable and which should be classified as hard-to-abate (BMF, 2024; Hochmeister et al., 2024a; Ott and Kulich, 2024). In addition, Austria's biomass potential likely plays a role in offsetting and achieving negative emissions.

### Cross-Chapter Box 6. Carbon dioxide removal

Harald Vacik; Alina Brad; Simone Gingrich; Tajda Obrecht; Holger Ott; Alexander Passer; Jana Petermann; Nicolas Roux; Harald Rieder; Dominik Steinberger-Maierhofer

Main message: Carbon Dioxide Removal (CDR) describes negative emission technologies that offset unavoidable current and historical emissions to the atmosphere. The assessment of the potential of each CDR option relies in part on the same energy and area resources, therefore the options should be considered as interdependent. The successful implementation of CDR options also depends on several external factors, such as technological developments, the regulatory framework, public perception, existing land use and energy demand under changing climate conditions.

Definitions and principles: CDR techniques encompass all anthropogenic activities that remove carbon dioxide  $(CO_2)$  from the atmosphere and durably store it in geological, terrestrial and marine reservoirs or in products. CDR has two objectives: (i) to offset unavoidable anthropogenic  $CO_2$  emissions to the atmosphere, and (ii) to create negative  $CO_2$  emissions, i.e., offsetting historical anthropogenic emissions. Numerous existing and emerging CDR technologies are known, focusing on natural and technological means. They include existing and potential anthropogenic enhancement of biological or geochemical  $CO_2$  sinks such as afforestation, reforestation, land restoration, and technical means such as bioenergy or direct air capture in combination with carbon capture and storage (BECCS and DACCS), biochar production (PyCCS) and chemical weathering in terrestrial or aquatic systems, but exclude natural  $CO_2$  uptake not directly caused by human activities. For a positive CDR effect, the total quantity of atmospheric  $CO_2$  physically removed and permanently stored must be higher than the total quantity of  $CO_2$  emitted during the process.

Potentials of CDR options for Austria: The potential of the CDR options can only be assessed on a case-by-case basis. The many interdependencies require an estimate of the CDR potentials individually as presented in CCBox 6 Figure 1, and the implementation in Austria would therefore require a careful consideration of the best combination of options and the mutual impairment of their potential. The CDR options, their potential scale of application, their costs and effectiveness, as well as their interference with other land uses and  $CO_2$  storage options are described in detail in Chapters 1, 2, 3 and 4. Considered aspects include technical readiness, land and other resource needs, geographical and geological constraints, primary energy demand, time scale of sequestration, environmental, social, and health impacts and potential risks, trade-offs and co-benefits. The technical and plausible CDR potential is presented in CCBox 6 Figure 1 for the current potential (considering the current capacities and constraints until 2030) and the future potential (up to 2050). Additionally, a qualitative assessment has been made for the required area and energy demand, the public acceptance and any constraints regarding the legal framework for implementing the CDR option.

The potential for the long-term carbon dioxide removal of different land uses has been assessed for soil management, agroforestry and renaturation, as well as different forest management options (afforestation, adaptive or extensive forest management). Two CDR options consider a land use change, where technical measures, reducing food waste or special diets allow to gain some area for afforestation measures. Among the forest-related CDR options, different management strategies (e.g., change of tree species composition, shorter rotation) are considered. Depending on the CDR option (Table 2.3, Sections 2.3.1 and 2.3.3), the total mitigation effects are within a range of  $-1.2 \text{ MtCO}_2$ /ha and year and  $+8.0 \text{ MtCO}_2$ /ha and year compared to the reference of the year 2020.

	\ /		2
ŀ	۱ŀ	۱ŀ	٢2

	Technical / plausible potential	(need p	Assum for area, e ublic acce	Co-benefits / trade-offs			
CDR Options	CO <sub>2</sub> /yr until 2030/2050 + Max (CO <sub>2</sub> )	Area	Energy	Costs	Accept- ance	Legal adaptations	
BECCS / DACCS capture CO <sub>2</sub> for storage and utilization	CCS: 3 to 25 Mt CO <sub>2</sub> /yr ** Total onshore storage capacity of several hundred Mt CO <sub>2</sub> in Austria ** CCU: 1 to 5 Mt CO <sub>2</sub> /yr **	+	+++	+++	no	required	High demand for energy from biomass might raise conflicts with other uses of timber and pulp wood (figures and references see Section 4.6)
Soil management use plants timber for biochar in soils	0.9 to 3.7 t CO <sub>2</sub> /ha/yr Based on global estimates 2030: 1 to 3.3 Gt CO <sub>2</sub> /yr 2050: 1.1 to 4.9 Gt CO <sub>2</sub> /yr	0	+	++	limited	not required	High competition on the market for timber and pulp wood as well as biomass for energy production (figures and references see Section 2.3)
Afforestation (LUC) of agricultural land through a protein transition (lancet diet) and halving food waste	Compared to 2020: 3.3 to 8.0 Mt CO <sub>2</sub> /yr Compared to Reference Scenario: 3.0 to 12.9 Mt CO <sub>2</sub> /yr	+	+	+	limited	not required	Loss of argicultural land and loss of biodiversity, lower income of livestoci farmers, changes in landscape (figures and references see Section 2.3)
Agroforestry (LUC) increasing sinks through short rotation coppices, hedges and by technical mitigation measures on agricultural land	Compared to reference scenario: 0.1 to 0.4 Mt CO <sub>2</sub> /yr	++	++	++	limited	not required	Loss of argicultural land, partly increased biodiversity, changes in landscape (figures and references see Section 2.3)
Renaturation of wetlands under agricultural use	1 to 100 kt $CO_2/yr$ (GWP100) avg45 t $CO_2/ha/yr$ up to >-60 t $CO_2/ha/yr$	÷	0	++	high	not required	Changes in biodiversity and landscape (drainag of peatlands/rewetting on extensively managee peatsites) reduction in feedstock and market crops (figures and references see Section 2.3)
Adaptive forest management change of tree species composition	Compared to 2020: Sink is decreased by -0.1 to -1.2 Mt CO <sub>2</sub> /yr <sup>v</sup> Compared to reference scenario: 4.1 to 4.9 Mt CO <sub>2</sub> /yr <sup>v</sup>	0	0	+	limited	not required	Improved biodiversity, reduced vulnerability to climate change impacts less availability of demanded coniferous timber assortments (figures and references see Section 2.3)
Extensive forest management reducing the overall use and harvest of wood	Compared to 2020: 1.4 to 4.4 Mt CO <sub>2</sub> /yr <sup>v</sup> Compared to reference scenario: 9.3 to 11.1 Mt CO <sub>2</sub> /yr <sup>v</sup>	0	0	+	limited	not required	Improved biodiversity, less availability of timber, less farmers' income and decreased rural employment (figures and references see Section 2.3)
<ul> <li>* only if limited by geological</li> <li>* reductions through industrial CCUS not considered</li> <li>* average values (2010-2150)</li> </ul>	0 no additional requirements for a + small additional requirements fo ++ medium additional requirement +++ high additional requirements	area, energy o or area, energ nts for area, e for area, ene	or investments ly or investme nergy or inves rgy or investn	nts stments nents			

CCBox 6 Figure 1 Assessment of the potential of Carbon Dioxide Removal (CDR) options in Austria.

BECCS and DACCS refer to geological storage of  $CO_2$  captured from the atmosphere and are thus derivatives of CCS (see Section 4.6). While classical CCS projects capture  $CO_2$  from point sources, atmospheric  $CO_2$  is less concentrated and captured by biomass (BE) or by direct air capture (DA). It should be noted that the required renewable biomass is only available on a decentralized basis, and long transport distances must be expected for centralized combustion and subsequent sequestration. Technically,  $CO_2$  can only be removed from the atmosphere with a comparably high energy input, which is why DA has not yet reached the necessary technological maturity. BECCS, including geological storage, is in principle available as a technology for achieving negative emissions, but faces a time challenge in terms of political, legal and technical implementation (Section 4.6). The provision of biomass is also likely to be a limiting element in the BECCS chain (Section 2.3.2).

Several factors influence the potential of biochar application to agricultural soils, such as pyrolysis production, soil conditions, climate, field management, biomass availability, making it difficult to estimate its CDR potential. Therefore, the reported global potential ranges from 0.65 to 35 GtCO<sub>2</sub>eq/yr depend on the applied models and their assumptions (Tisserant and Cherubini, 2019). The IPCC (2023a) provides ranges of 0.1-4.9 GtCO<sub>2</sub>/yr (WG1) and 0.1-6.6 GtCO<sub>2</sub>/yr (WG3) in AR6 globally, Smith (2016) provides an estimate of 0.9-3.7 tCO<sub>2</sub>/ha/yr, which is used in CCBox 6 Figure 1.

In 2024, the European Council and the European Parliament reached a political agreement to establish an EU-level voluntary certification framework for permanent carbon removals, carbon farming and carbon storage in products to facilitate the emission reduction activities in the EU (Regulation (EU) 2024/3012). The timing of carbon uptake and storage is important, as there is a need to remove large quantities of carbon from the atmosphere within the next few decades. A large potential for a permanent carbon storage is seen in the construction sector with the use of organic construction materials during the use phase of the built environment; a carbon uptake at the beginning of the life-cycle is seen as less mature and efficient and there are challenges related to certification schemes in buildings and the recycling of timber products (Kuittinen et al., 2023). In any case, CDR techniques will have cost implications: In the construction industry, price increases of 20–30 % for steel and 20–80 % for cement can be expected (Material Economics, 2019).

Research on public perceptions of CDR options is limited, but reveals low levels of public awareness and knowledge (Smith et al., 2023). Generally, conventional CDR methods such as afforestation are favored over novel approaches such as DACCS, which often encounter uncertainty (Bellamy et al., 2016; Dumbrell et al., 2016; Jobin and Siegrist, 2020; Brutschin et al., 2024). With respect to land-based CDR methods, concerns about land use, biodiversity, and environmental impacts are more frequently raised. BECCS also raises concerns about land competition, food prices, biodiversity, and the sustainability of biomass. Research on public perception of CCS shows safety concerns about leakage and seismic risks of underground  $CO_2$  storage (Cox et al., 2020). In addition, public scepticism appears to be higher in wealthier countries, as studies have shown lower confidence in the potential of science and technology to address climate change in these countries. This would call for more country-specific approaches to the deployment of potentially useful technologies (Brutschin et al., 2024). There is also a need for a clear, transparent and robust definition of what counts as hard-to-abate and a clear delineation of residual emissions in the net zero year. These should indicate how much and by whom emissions should be reduced by the various CDR options by 2040, as well as the corresponding storage, utilization, and removal capacities needed (Brad et al., 2024).

Main challenges with CDR in Austria: CDR options could be used in all sectors to offset unavoidable anthropogenic  $CO_2$  emissions into the atmosphere, as well as to offset historical anthropogenic emissions to ensure climate change mitigation. Although they can serve to compensate for residual emissions that are hard-to-abate, CDR options are not a substitute for emission reductions to reach climate targets (*medium confidence*) (see Section 4.6). Correspondingly, an important political challenge is to send robust political signals that CDR cannot compensate for any delay in phasing out fossil fuels, especially given the limited and uncertain CDR potentials (Brad et al., 2024).

CDR options in Austria include land-based biological (i.e., afforestation, reforestation, ecosystem restoration, and soil management) and technical processes to capture and permanently store  $CO_2$ , such as BECCS and DACCS. They differ in terms of their technical and plausible potentials, synergies and trade-offs, and technical readiness levels (*medium confidence*) (see Sections 2.3 and 4.6).

CDR options face several limitations; the availability of biomass for CDR may conflict with other land uses for energy production, construction purposes, pulp/paper industry, or conservation efforts; and long transport distances for the provision of renewable energy from biomass are to be expected. Geological storage capacities may also compete with other subsurface technologies, such as natural gas storage and future hydrogen storage operations (e.g., for steel production), and energetic uses, such as geothermal energy production and storage (Hochmeister et al., 2024a, 2024b; Ott and Kulich, 2024; BMK, 2024a; Kulich and Ott, 2025). Carbon capture and storage is currently prohibited on Austrian territory, but is currently under reconsideration by the Austrian CMS (see Section 4.6).

In the construction sector, there are several challenges related to the certification schemes in buildings and the recycling of timber products. Trade-offs exist also in storing carbon in buildings in comparison to other options (Maierhofer et al., 2024) and only a combination of various models and sectors together with the supply and demand as well as life cycle assessments of the whole supply chain of each CDR method can show that the total quantity of atmospheric  $CO_2$  removed and permanently stored is greater than the total quantity of  $CO_2$  emitted to the atmosphere (*medium confidence*) (see Sections 4.6 and 2.3).

The implementation of CDR requires future research to address the monitoring of CDR, such as imports of timber, reuse of material, and the definition of temporal aspects, such as the duration of  $CO_2$  removal. It requires a robust common definition of permanent carbon storage, as the EU Commission's definition, for example, is incompatible with the terminology used in the UNFCCC and IPCC frameworks.

### 4.6.1. Current status of CCUS technologies

 $CO_2$  capture and transport: CCS requires  $CO_2$  in a pure state and in large quantities from so-called point sources or  $CO_2$ -intensive industries such as cement or steel production (Paltsev et al., 2021). Several technologies are available to capture  $CO_2$  at the industrial site. These are largely mature (technology readiness level TRL 9) and are based, for example, on chemical absorption (e.g., amine scrubbing) or adsorption (Dziejarski et al., 2023). Existing industrial plants can be retrofitted for  $CO_2$  capture, or special processes tailored to specific industrial processes can be used (Fröhlich et al., 2019).  $CO_2$  capture is energy intensive, increasing energy demand in production processes, and water intensive, potentially putting pressure on water supply systems. Carbon capture largely determines the cost of CCUS operations (Kearns et al., 2021).

Because CCS places high demands on geological formations, the captured  $CO_2$  must be transported over long distances to suitable storage sites, usually by pipeline or ship. The same applies to the use of  $CO_2$ , which can be centralized at production sites. Transport distances from the capture site to the storage site range from a few kilometers to several hundred kilometers onshore or offshore (Ausfelder et al., 2020). Shorter transport distances are generally more economical and energy efficient but limit the potential for CCUS deployments due to the accessibility of suitable storage sites.  $CO_2$  transport by pipeline is in commercial use (NPC, 2019; Kearns et al., 2021) and is therefore at TRL 9.  $CO_2$  transport by waterways has not yet been used for CCUS and is therefore at a lower TRL 4–7 (Ausfelder et al., 2020; Al Baroudi et al., 2021; Kearns et al., 2021). The combined cost of  $CO_2$  transport and storage is generally assumed to be a uniform  $EUR_{2023}$  9.91 per t $CO_2$  (Smith et al., 2021), which is reasonable in some regions, but not in Europe. Europe has the highest transport and storage costs in the baseline scenario of around  $EUR_{2023}$  34.7 per t $CO_2$ , as offshore storage is preferred. However, even in Europe, onshore storage is expected to be more economic, in the range of  $EUR_{2023}$  15.87 per t $CO_2$ .

Geological storage: Suitable geological formations are found at depths greater than 800 m, where  $CO_2$  is in a dense state that allows efficient use of pore space for storage (Angus et al., 1987; Benson et al., 2005). In this state, however, the  $CO_2$  still has a lower density than the reservoir water and therefore experiences buoyancy; to prevent the injected  $CO_2$ from migrating into higher layers containing drinking water or even into the atmosphere, a trap structure in combination with an impermeable caprock, e.g., an overlying claystone formation, is required (IPCC, 2005). Corresponding geological formations are known from oil and gas fields, which are suitable reservoirs for geological  $CO_2$  storage, as are deep saline aquifers. In a suitable storage facility,  $CO_2$  is permanently bound by physical and chemical processes (Xu et al.,
2003; De Silva et al., 2015; Krevor et al., 2015; Brandstätter et al., 2025), providing high storage security (Benson et al., 2005). Despite technical maturity, geological reservoirs are individually different and require site-specific development. Numerical reservoir modeling and uncertainty analysis are required to ensure safe storage. Monitoring options must be in place to detect CO<sub>2</sub> leakage during the injection phase and beyond (IPCC, 2005; Directive 2009/31/EC). Upon closure of a storage site, legal obligations for monitoring and corrective, preventive and remediation measures should be transferred to the competent authority (Directive 2009/31/ EC). Geological CO<sub>2</sub> storage in hydrocarbon reservoirs and saline aquifers is commercially applied and therefore mature (Benson et al., 2005; Bui et al., 2018). However, storage options in unconventional geological reservoirs such as coal seams and volcanic (ultramafic) rocks are less developed (Benson et al., 2005; Kearns et al., 2021). There are currently about 30 ongoing CCS projects, 11 of which are capturing more than 1 Mt/yr of CO<sub>2</sub> for long-term storage (Global CCS Institute, 2021, 2023). Although these rates are still far from the actual problem of anthropogenic CO<sub>2</sub> emissions, CCS technology is successfully applied and available as a critical technology.

Carbon utilization: Various industries and their products depend on carbon as a feedstock. Carbon supply has been primarily based on petroleum, coal (coke), and natural gas (European Commission: Directorate-General for Energy, 2020). Thus, alternative carbon sources are required for a sustainable and renewable circular economy. One option is CCU, which is based on either renewable sources or holding CO<sub>2</sub> in closed loops (Hepburn et al., 2019). In CCU processes, CO<sub>2</sub> is used as a C1-building block to synthesize carbon-containing products. However, most of these products rerelease CO<sub>2</sub> during their utilization (e.g., e-fuels) or at the end of their lifetime (e.g., plastics). Therefore, CCU contributes to an atmospheric CO<sub>2</sub> reduction only when the CO<sub>2</sub> has been taken from the air, CO<sub>2</sub> is held in a closed loop, or products are generated that bind CO<sub>2</sub> for at least several hundred years (Hepburn et al., 2019).

CCU products of great relevance are sustainable aviation fuels that can either be produced from biomass feedstock in different process routes or synthesized from  $CO_2$  with green hydrogen (Richter et al., 2018) (for biomass potentials see Section 2.3.2 on sufficiency oriented bioeconomy). Mineral carbonation, i.e., the production of carbonates from  $CO_2$ and metal oxides, is a CCU option with a durable fixation of  $CO_2$  (Olajire, 2013). Appropriate metal oxides can be found in primary rocks (e.g., serpentine) or also in secondary feedstocks (e.g., slags and ashes) (Wang et al., 2018). The carbonates can be utilized in the pulp and paper industry or as fillers in construction materials. Another promising CCU option is to retain the  $CO_2$  in a closed loop (Rosenfeld et al., 2020). One potential application of such looping processes is in the steel industry until it transitions to carbon-neutral production.

# 4.6.2. CCUS potential in and for Austria

In 2023, Austria's total GHG emissions amounted to 68.6  $MtCO_2eq$  (Umweltbundesamt, 2023a). The energy and industry sectors are responsible for more than 43 % of total emissions (CO<sub>2</sub>eq), with decentralized CO<sub>2</sub> sources such as agriculture and transportation accounting for 39 % (Umweltbundesamt, 2023a). In Austria, the steel and cement industries emit about 14.1  $MtCO_2/yr$  into the atmosphere. An overview of the largest industrial CO<sub>2</sub> sources and potential geological sinks in Austria is shown in Figure 4.14.

Geological storage capacity: An Austria-wide assessment of the CCUS potential is currently under development (CaCTUS, 2024). Significant potential storage regions are located in the sedimentary basins of the Upper Austrian Molasse Zone and the Vienna Basin as well as in the Styrian Basin (Hochmeister et al., 2024a, 2024b; Ott and Kulich, 2024; Kulich and Ott, 2025).

Domestic hydrocarbon fields: Hydrocarbon reservoirs are largely developed and well known, and therefore represent the fastest route to geological storage. The theoretical storage capacity of depleted hydrocarbon fields is estimated by Scharf and Clemens (2006) to be at 465 MtCO<sub>2</sub>. A recent Austria-wide study (CaCTUS, 2024; Hochmeister et al., 2024a, 2024b; Ott and Kulich, 2024; Kulich and Ott, 2025) takes a more conservative approach and shows that the largest hydrocarbon reservoirs represent a total effective storage potential of 250 MtCO<sub>2</sub>, taking into account fields with a storage capacity of more than 2 Mt. Excluding the highly perforated oil horizons, the largest gas and gas condensate fields still offer a total effective storage potential of >100 MtCO<sub>2</sub>. Assuming operating times of 10 to 30 years (European Commission, 2024), the storage capacities result in possible storage rates of 3-5 MtCO<sub>2</sub>/yr. Alternative uses of this underground space, such as hydrogen storage or geothermal energy recovery, are also taken into account.

**Domestic aquifers:** Saline aquifers have been hardly studied, with the exception of the Aderklaa conglomerate in the Vienna Basin; a theoretical storage capacity of 1,000 MtCO<sub>2</sub> has been calculated by Heinemann and Scharf (2004). Saline



Figure 4.14 Potential hard-to-abate carbon sources >0.1 MtCO<sub>2</sub> emissions in 2021 according to E-PRTR or ETS, and potential geological sinks. Montanuniversität Leoben (Hochmeister et al., 2024b, 2024a; Ott and Kulich, 2024; Kulich and Ott, 2025).

aquifers urgently need to be explored. In general, however, aquifers can be expected to have many times the storage capacity of hydrocarbon reservoirs.

Storage capacity inside Austria: Depending on the exact definition of 'hard-to-abate  $CO_2$  emissions' (BMF, 2024), which range from 7 to 14 Mt/yr, and with the estimated storage capacity including all uncertainties, it can be stated that the storage capacity in Austria is sufficient for several decades.

Storage outside Austria: Securing and developing storage capacity outside Austria depends on factors that are difficult to estimate in terms of time and realism. These factors include the development of a suitable transport network, international agreements and the availability of storage capacity (see, e.g., Baltac et al., 2023). It can be assumed that there is sufficient storage capacity in the North Sea, estimated to be in the order of several 100 billion tons in total (Karvounis and Blunt, 2021), but it will take time to develop and will only be available to a limited extent in the medium term. For example, the expansion of the Norwegian flagship project Northern Lights (Northern Lights, 2024) is estimated at only 5 Mt/yr by 2030. Austria is competing with other European countries. In any case, it can be assumed that large-scale storage outside Austria will be a later option (2040/2050).  $CO_2$  storage in Austria therefore seems necessary, does not conflict with other European storage activities and would likely reduce Austrian emissions much faster.

CCU potential in Austria: No data are currently available specifically for Austria. In any case, CCU associated with fossil GHG emissions does not contribute to the removal of carbon from the atmosphere. CCU products produced from fossil  $CO_2$  will eventually release GHG into the atmosphere at the end of their life cycle. However, carbon is needed for selected fuels and as a feedstock in the chemical industry. In the long term, CCU is an indispensable part of a future economic system. The availability and further development of CCU technologies and the necessary amounts of renewable energy for their operation must be ensured for the circularity of the carbon economy in a defossilized economic system (Purr and Garvens, 2021). Consequently, the potential of CCU in general, and for Austria in particular, is limited to  $CO_2$  captured directly from the atmosphere or biomass as a carbon source. In addition, closed carbon cycles and carbonates as long-term carbon fixation options contribute to the  $CO_2$  utilization potential. Against this background, a first estimate of the CCU potential in Austria is between 1 and 5 MtCO<sub>2</sub>/yr, which has to be confirmed in the ongoing study (CaCTUS, 2024).

# 4.6.3. Necessary transformations and R&D

The development of CCUS requires a national and European  $CO_2$  transport network from  $CO_2$  sources to  $CO_2$  sinks, exploration and development of the geological subsurface as well as solutions for the storage and utilization of  $CO_2$ . With the capacities explored and estimated to date in Austria, CCUS and geological storage in particular can make a significant contribution to national decarbonization. CCUS-specific R&D requires financial resources, an appropriate legal framework, public acceptance and a national CCUS roadmap for Austria (BMF, 2024; Hochmeister et al., 2024a; Ott and Kulich, 2024).

Site-specific R&D on CCS: Despite technological maturity, geological reservoirs require exploration and site-specific R&D for predictive modeling of  $CO_2$  migration and reliable monitoring concepts for capacity and safety analysis. Therefore, geological  $CO_2$  storage is not an off-the-shelf technology and should be considered early in the CCS implementation process.

Phased implementation of CCS: For geological storage, the timeframes to 2030 and 2040 are very short. This is because site development usually takes several years (Bui et al., 2018). Having storage capacity or export infrastructure in place by then will require rapid action and a strong commitment to implementation. The three domestic and international geological storage options discussed above are likely to have different development times and should therefore be phased over time. A phased approach allows for a timely entry into decarbonization through CCS and sustainable availability of storage volumes and rates in line with Austria's Carbon Management Strategy (CMS) and negative emissions ambitions. All options and their time constraints should be considered, e.g., Phase 1: Development of depleted hydrocarbon fields for a timely entry into decarbonization through CCS. Exploration of aquifers: Even in scenarios where aquifers are ultimately not developed due to the existence of alternatives, the precautionary principle requires the state be aware of possible future options and their storage capacity. Securing storage capacities outside Austria in accordance with the residual emissions to be expected in the future. Phase 2: Depending on the situation: Development of additional storage capacities in domestic aquifers or abroad.

**Public perception:** CCS projects do not usually fail for technical reasons, but because of a lack of social acceptance and the resulting political will (Witte, 2021). Early and targeted communication of the opportunities and risks of a technology offers the opportunity to bring negatively charged issues back to a factual level and should generally be given higher priority. In this way, the information level of society can be raised to a level at which meaningful opinion-forming can take place, which should then be seen as a prerequisite for creating acceptance (Cremer et al., 2008).

CCU: With regard to CCU, sufficient availability of renewable energy is a prerequisite for the climate-neutral operation of industrial processes. Cooperation with international CCU sites, which are expected to develop in areas with optimal conditions for renewable electricity generation, could be an opportunity for a circular carbon economy. In addition, direct air capture technologies and sustainable biomass feedstocks (e.g., cascaded biomass, microalgae) need to be further developed and optimized to achieve carbon neutrality (Purr and Garvens, 2021). The optimization of CCU processes in terms of energy efficiency, cost reduction through scaling and learning curves, and the development of novel catalytic, electrochemical, and photochemical CCU pathways pave the way for their beneficial implementation in a future defossilized economy (Klankermayer et al., 2016). Finally, a regulatory framework for CCU needs to be established, in particular for the crediting of CCU products in the EU Emissions Trading System (Böhringer and Lange, 2013; BMF, 2024).

# 4.7. Provision by work

#### 4.7.1. Work and climate change

Work, including unpaid care and paid employment, dominates daily life in Austria, and the structural dependence of persons and welfare systems on paid work currently tends to impede climate-neutral provisioning as well as climate-friendly living (Pettinger, 2017; Kreinin and Aigner, 2022; Aigner et al., 2023b; Bärnthaler and Gough, 2023; Gerold et al., 2023; Hoffmann et al., 2023; Mason and Büchs, 2023) (high confidence). 90 % of the Austrian population (>10 years) is involved in care work (on average 3.5 h/d per household), while only 49 % of the Austrian population are involved in paid work (on average 8.25 h/d) (Statistik Austria, 2024b). The current unequal distribution of paid and unpaid work for the care of people (children, elderly, those in need of care) is strongly influenced by gender-specific divisions of labor (Kuhl et al., 2011; Bauhardt, 2013; Dengler and Lang, 2019, 2022; Knobloch, 2019; Cohen and MacGregor, 2020). Work time preferences follow social norms, with 17.5-30 % of Austrian workers defined as overemployed, i.e., who want to reduce paid working hours (Gerold and Nocker, 2018). Although unpaid care (incl. caring for oneself, household, community, society as well as nature) and paid work are indispensable for the functioning of provisioning systems and the formal economy (Biesecker et al., 2000; Biesecker and Hofmeister, 2006; Cleveland et al., 2017; Smetschka et al., 2023c), work has received little attention in climate science. While many individuals depend on income from paid employment, work is also a source of material livelihood, social integration, the creation of meaning and the development of self-identity (Feigl and Wukovitsch, 2018). In its current form, employment causes consumption, is biophysically intense and has ambivalent health impacts (Gerold et al., 2023). The climate effects of employment can be assessed by considering the goods and services as outputs from work, the tasks and activities workers perform at the workplace, the impact of employment conditions on the sustainability of workers' and their households' lifestyles, and the efficiency of the production process in relation to socially desired provisioning (Bohnenberger, 2022a).

High-productive, energy-intensive work is well-paid and structurally linked to low-productive, energy-light, unpaid or low-paid, very essential work (limited evidence, medium agreement). Large parts of paid work do not meet the climate-mitigation requirements (Hoffmann and Spash, 2021; Hofbauer et al., 2023), for instance, when emissions are allocated to work according to income or production (Aigner et al., 2023b; Hoffmann et al., 2023). 60 % of production-based emissions are related to exported goods and services, and thus increase the consumption-based emissions of other countries (Aigner et al., 2023b). Production-based emissions per job and per value-added unit are unevenly distributed across sectors, geography, institutional form (state, private, non-for-profit), firm size, income, working-hours, and occupation (Gabelberger et al., 2020; François et al., 2023b; Heyen, 2023; Oberholzer, 2023). Employment in manufacturing, industry and transport (Hardt et al., 2020, 2021) is associated with high emissions. Employment in interpersonal care services causes lower emissions, but many of these activities are based on the upstream production of goods (Hardt et al., 2020; Aigner and Lichtenberger, 2021; Hardt et al., 2021; Hoffmann et al., 2023; Smetschka et al., 2023a). Due to their interpersonal nature, care and service sectors face barriers to fossil-fuel driven productivity, which leads to a devaluation of climate-friendly care that structurally links the two spheres (Baumol, 1967; Hardt et al., 2020; Aigner and Lichtenberger, 2021). Public provisioning, such as health and care, is a key part of the economy, and can lead to labor shortages and competition between the respective sectors. Social and economic marginalization and devaluation of unpaid care through the dominance of paid work takes place, which leads to time-competition between the two forms of work. Spending time on employment and market-based economic production reduces the time available for voluntary and alternative subsistence activities that satisfy needs with usually low-carbon emissions (Dengler and Strunk, 2018; Aulenbacher et al., 2021).

The work mode of living poses a structural barrier for climate-friendly mobility, food, housing as well as leisure practices, and fosters energy- and fossil-fuel-intensive practices (Hofbauer et al., 2023; Bohnenberger, 2022b; Gerold et al., 2023) (medium confidence). Both work acceleration (Rosa, 2005) and time pressure (Sullivan and Gershuny, 2018) affect climate-related decision making (Rinderspacher, 1985; Shove et al., 2009; Rosa et al., 2015; Schor, 2016; Großer et al., 2020; Chung, 2022). Employment and care are key motivations for the use of individual mobility in Austria and are therefore inherently related to emissions in the mobility sectors (see Smetschka et al., 2023b; Wiedenhofer et al., 2023). More than 50 % of travel distances of employees in Austria are work-related (44 % commuting and 9 % business related) (Tomschy et al., 2016). Although there have been changes in the work-induced consumption patterns, an overall reduction of work-induced emissions cannot be observed: In Germany, for example, increasing labor force participation increased household consumption footprints, offsetting technical efficiency improvements and declining paid work time per employee (Wiedenhofer et al., 2023). Similarly, mobility footprints (e.g., through teleworking) faced a (time) rebound effect as the number of trips declined but distance of commutes increased (Wiedenhofer et al., 2023). In addition to paid work, changes in unpaid care responsibilities also shift the modal split towards motor vehicles (Tomschy et al., 2016). A shift towards climate-friendly modes of living

requires a reconfiguration of working time to enable caring responsibilities (Hanbury et al., 2023). Some studies show that free time is mainly used for care activities (Gerold et al., 2017), but workers perceive additional recreational time from working time reduction as a key benefit (Hidasi et al., 2023; Antal et al., 2024).

46 % of Austrians believe that their current job does not contribute to the green transition (limited evidence, high agreement), which can be associated with contemporary societal conflicts entailing multiple actors and levels (European Commission: Directorate-General for Employment, Social Affairs and Inclusion, 2022). Workers at 'green' workplaces are more likely to engage in pro-environmental behaviors, while workers at workplaces that lack requirements for environmental knowledge in daily tasks, or where workers do not see their own impact or have no control over their own tasks, are less motivated to engage in such behaviors (Ciocirlan, 2017, 2023). Societal conflicts entail conflicts related to transitioning to climate-friendly work-life patterns, but also a lack of transitioning to green jobs (Fritz and Eversberg, 2024). This leads to anti-environmental actions, such as the defense of climate-damaging industries, but also to pro-environmental phenomena such as climate quitting (Brand and Wissen, 2021). Public infrastructure, industrial lobbying and state investments form a powerful consortium that establish the institutions, infrastructure, and rules upon which individuals form their preferences for need-satisfiers such as mode of commuting or job preservation in the automobile sector. Historically, this has increased car dependency (Mattioli et al., 2020). Today, a tendency towards acknowledging the climate implications in public investment decisions can be observed in housing or mobility, but not in the domain of work. Work-related institutions currently disregard both the impacts of climate crisis on work and the mitigation requirements for shaping work through labor law, social security systems, public services and industrial relations (Müller et al., 2019; Kopf, 2021; Bohnenberger, 2023) (limited evidence, high agreement).

# 4.7.2. Work in a climate-friendly future perspective

A transition to a climate-friendly society will end the requirement for greenhouse gas emissions from paid employment or unpaid care (*limited evidence, high agreement*). This involves a complete shift away from climate-damaging products and services to climate-friendly outputs, incl. sectoral shifts and an end to employment-induced climate-harming consumption. More work will be undertaken in person-related and resource-light sectors and activities, while energy- and resource-intensive sectors will be smaller, reducing the dependency of workers on climate-harming jobs. An overall trend of anxiety and political instability currently inhibits an efficient market or publicly planned distribution of workplaces to more essential work. This can be superseded by limiting the role of paid work for livelihoods and subsistence. Thereby, the reduction of working hours in paid work, can enable a better adaptation to climate impacts, e.g., when shifting working hours to less hot time periods, reducing work intensity through additional breaks, shifting work away from overheated locations, providing cooling systems, including greening of workplaces (Schulte et al., 2016; Masuda et al., 2021; Bauer et al., 2022; Boltz et al., 2023; Bühn and Voss, 2023).

Greening and sharing of paid employment and unpaid care will ensure quality of life, good work and climate-friendly living, work-life balance and sufficient provisioning of care (medium confidence). A reduction in the volume of paid work, especially energy- and resource-intensive work, is necessary to avoid exceeding ecological limits (Seidl and Zahrnt, 2019; Hoffmann and Spash, 2021). The achievement of social goals is contingent upon the decoupling of income and social security from employment and paid work (Petschow et al., 2018; Kubon-Gilke, 2019). A more equal distribution of paid work and unpaid care reduces time pressure and thereby enables the sharing of goods and climate-friendly lifestyles (Smetschka et al., 2019, 2024b). Compensatory forms of consumption (Røpke, 1999; Shove et al., 2009) which are triggered by dissatisfying forms of work will cease to exist.

Given the current business-as-usual, work will be severely impacted by the unfolding of the climate-crisis (Parsons et al., 2021) (Chapters 1, 2, 3), requiring sectoral shifts to the provisioning of basic goods such as health, care, food, and housing (limited evidence, medium agreement). To ensure the climate-resilience of work-systems, transformative adaptation measures ensure workers' security against the backdrop of an increasing number of heat waves and other extreme weather events (medium confidence). This shift will coincide with demographic trends that will drastically increase the demand for care (Famira-Mühlberger and Firgo, 2019). Impacts of climate change include the destruction of production sites through natural hazard events such as fires, avalanches, and floods (see Chapters 1, 2, 3), as well as declining labor productivity due to heat stress (Parsons et al., 2021). Perspectives from employees confirm this for

the Austrian health sector (Brugger et al., 2024a). For the past two decades, Parsons et al. (2022) estimate an annual labor loss of 20 hours per capita in Austria, a figure that is expected to rise substantially in the future (Parsons et al., 2022). Moreover, these impacts will disproportionately affect physically intensive workers and vulnerable groups (e.g., the elderly, people with disabilities), which in turn will lead to increasing care responsibilities, including new diseases (Chapters 1, 2, 3), and declining efficiency of current medication and general methods in the health sector (Bühn and Voss, 2023) (see Cross-Chapter Box 2).

# 4.7.3. Transformation towards climate-friendly work and green employment

To avoid climate-damaging production and to overcome labor supply constraints, climate-friendly regulation of labor markets can enable a shift from excess production to climate-friendly provisioning of essential services (Bärnthaler and Gough, 2023) (medium confidence). Production in energy- and emission-intensive sectors needs to be reduced, including hours worked in the respective sectors (Aigner et al., 2023b). Such a shift would also address future rising demand for provisioning of care (Famira-Mühlberger and Firgo, 2019). Climate-friendly public investments can (i) strengthen climate-friendly employment, (ii) satisfy societal needs, and (iii) and ensure a socially acceptable transformation (Krisch et al., 2020; Schultheiß et al., 2021). Climate-neutrality plans at the European and national level require comprehensive changes, especially in the production sector (Gabelberger et al., 2020; Steininger et al., 2021; Meinhart et al., 2022). In the transformation phase to a climate-friendly system, the volume of labor is likely to remain constant due to the necessary reconstruction of the infrastructure (Aiginger, 2016). Overall, climate mitigation policies are expected to increase employment until 2030 (Alexandri et al., 2024). Shifting to renewable energy creates additional demand for labor in the short and long run, and value added in the respective sectors until 2030 (Goers et al., 2020), leading to more equal and less polarized European labor markets (Alexandri et al., 2024). The shift towards renewable energy faces a shortage of low-skilled labor although the transition will mainly increase medium- and high-skilled employment. Shifting the tax burden from human labor to energy and resource consumption could smooth this transition (Köppl and Schratzenstaller, 2019).

Workplace codetermination and greened industrial relations can avoid climate-damaging production and support a shift to climate-friendly production processes and sustainable employment (Initiative Wege aus der Krise, 2019; Aigner et al., 2023b) (medium confidence). Employees and their representatives are aware of the need for transformation (e.g., Littig, 2017; Niedermoser, 2017; AK Wien, 2023, 2024; AK Wien and ÖGB, 2024), as are some companies (e.g., Global 2000 et al., 2017; BMK, 2022). Climate-friendly employment can be achieved through conversion of business models to more climate-friendly products and essential services (UNDP, 2015; Hoffmann and Spash, 2021; Hofbauer et al., 2023). Potential employment impacts have been explored, for example, for the phase-out of the internal combustion engine (Sala et al., 2020; Wissen et al., 2020). Unions and green collective bargaining can facilitate transitional assistance policies (SOC/747-EESC-2022-01-01, 2023; OECD, 2024). Guaranteeing material security and a fair distribution of transformation efforts is crucial (Specht-Prebanda, 2020; Wissen et al., 2020; ETUC, 2021; ÖGB, 2021). A just transition can be supported by stronger unionization in green sectors, strengthening unions as pro-environmental actors and improving working conditions in these sectors. A lack of compatibility of work activities and the business purpose with workers' values leads to moral stress and lower job satisfaction, which may explain the increasing climate quitting in climate-harming occupations (Net Positive Employee Barometer, 2023) and societal polarization about the green transition (Smetschka et al., 2024a). Climate-friendly provisioning systems (e.g., mobility, food) at the workplace may, however, positively contribute to climate protection at the workplace and beyond (limited evidence, medium agreement). This includes limiting business flights and limiting the serving of GHG-intensive foods such as meat or dairy products (limited evidence, medium agreement) and providing infrastructures for climate-friendly commuting, such as bicycle stations or incentives for low-carbon modes of transport for commuting (e.g., climate ticket, active mobility) (Heinfellner et al., 2024).

A redistribution of paid employment and unpaid care as well as relocating workplaces to people's living place through economic and labor market policies can enable climate-friendly work-life patterns (*medium confidence*). Shorter working hours are considered a suitable measure (Cieplinski et al., 2021; Vetter, 2023) to (i) facilitate a climate-friendly life outside of paid employment (Schor, 2005; Knight et al., 2013; Larsson et al., 2019), and to (ii) distribute the volume of employment more evenly (Antal, 2014; Gerold and Nocker, 2015; Seidl and Zahrnt, 2019; Figerl et al., 2021). The specific effects of reduced working hours on individual behavior are ambiguous (Hanbury et al., 2019, 2023; Hofbauer et al., 2023). A shift in societal values toward a more balanced work-life relationship can reshape expectations about the meaning of work (Aichholzer et al., 2019) and the desire for shorter working hours (Csoka, 2018; Arbeitszeiten im Fokus - Daten, Gestaltung, Bedarfe, 2021). Making paid and unpaid care more attractive by raising wages, reducing working hours, and increasing visibility, can create incentives to shift time to low-energy and low-emission care activities (Aigner and Lichtenberger, 2021) and reduce gendered inequalities in care work (Pirklbauer and Wukovitsch, 2019; Gottschlich and Katz, 2023). Climate-friendly time policy (Reisch and Bietz, 2014) and care-oriented time policy (Heitkötter et al., 2009) focus on time as a lever for political design and a more equitable distribution of work between the sexes. Time policy can support climate-friendly lifestyles (Hartard et al., 2006; Rau, 2015; Schor, 2016) through reduced time pressure, deceleration, and reduced mental load that foster climate-friendly decisions in everyday life (Hofmeister and Mölders, 2021), green volunteering (Winker, 2021), substituting resource-intensive practices with time-intensive ones in many lifeworlds (Buhl et al., 2017), positive health co-benefits (APCC, 2019; Dengler et al., 2024), structural conditions such as reduced commuting through spatial, urban and transport planning (see Section 3.3.2), and temporal support for care work.

A more equal distribution of wealth and income, job-guarantees and public provisioning of basic goods (housing, food, mobility) can increase economic security and thus enable green labor market transformations as individual material consequences of switching jobs are reduced (Dukelow and Murphy, 2022; Lee and Koch, 2023) (medium confidence). The current coupling of income, social security, recognition, and participation in employment constrains the scope for climate policy (Hoffmann and Paulsen, 2020; Bohnenberger and Schultheis, 2021; Gerold et al., 2023). Therefore, measures to increase economic security during transitional phases, including an ecologically driven planned economic decline, include an unconditional basic income, greater self-sufficiency (Littig and Spitzer, 2011; Paech, 2012), or the provision of comprehensive services of general interest (Coote and Percy, 2020; Bohnenberger and Schultheis, 2021; Büchs, 2021; Gough, 2022; Bohnenberger, 2023; Rehm et al., 2023). An examination of public spending through the lens of gender and climate justice presents an opportunity to achieve emission reductions in the care sector (Schalatek, 2012). Time banks, for instance, show a potential to connect care work and employment, thereby creating more socially and climate-friendly working time quotas (Schor, 2016; Bader et al., 2021).

Climate mainstreaming is required in all dimensions of labor and social policy, including labor law and unemployment insurance (limited evidence, medium agreement). Although employment is an important element of climate policy (Seidl and Zahrnt, 2019; Bohnenberger, 2022a; Kreinin and Aigner, 2022; Gerold et al., 2023), work has not been a focal point for climate-friendly interventions. There has been a focus on (re-)education and green skills particularly in engineering and managerial skills (Vona et al., 2015, 2018; Vona, 2018) and action plans are being developed mainly for the energy transition (for Austria, see BMK (2023a) and for the global level, IRENA and ILO (2021)). For details on (re-)education and green skills needs in Austria, see Dorr et al. (2023) and AMS (2024). Climate-related knowledge and skills also increase legitimacy (Pichler et al., 2021). Re-education in climate-harming occupations carries the risk of lock-ins in unsustainable jobs (Bohnenberger, 2022a). It is more promising to provide reskilling and upskilling support for jobs in care, education, and resource-light sectors, where labor shortages exist. In addition, in many instances, special education and, consequently, an adaptation of the curricula is needed (Dorr et al., 2023; AMS, 2024). Brugger et al. (2024b) might provide one best-case example for the health sector. A complete shift to climate-friendly work systems requires a comprehensive transformation of work, as emphasized by literature on Just Transition (ILO, 2015; TUDCN, 2019); Sustainable Work (UNDP, 2015; Barth et al., 2016; Littig et al., 2018) and Post-Work (Frayne, 2016; Hoffmann and Paulsen, 2020). Greening work requires a structural change of welfare systems towards a sustainable welfare state (Bohnenberger and Schultheis, 2021; Bohnenberger, 2023; Hirvilammi et al., 2023), which is currently being considered within Austria's social security outlook (BMSGPK, 2024). This implies integrating climate mainstreaming across all sectors and institutions of social security (Bohnenberger, 2021), including unemployment insurance and employment policy, where initial approaches have already been developed for the Austrian labor market service (Neier et al., 2022, 2024; Smetschka et al., 2024a). Work not only structures daily lives and production processes, but is also subject to a wide range of laws and political conflicts (Smetschka et al., 2024a). Achieving a transformation of work through targeted policies thus requires critical reflection and consideration of related actors and institutions, including social partners, unions, employer associations, social insurance and social movements (e.g., women's rights movements) (Smetschka et al., 2024a).

# REFERENCES

- Abegg, B., and Mayer, M. (2023). The exceptional year of 2022: "deathblow" to glacier summer skiing in the Alps? *Front. Hum. Dyn.* 5, 1154245. <u>https://doi.org/10.3389/fhumd.2023.1154245</u>
- Abson, D. J., Fischer, J., Leventon, J., Newig, J., Schomerus, T., Vilsmaier, U., et al. (2017). Leverage points for sustainability transformation. *Ambio* 46, 30–39. <u>https://doi.org/10.1007/s13280-016-0800-y</u>
- Aflaki, S., Basher, S. A., and Masini, A. (2021). Technology-push, demand-pull and endogenous drivers of innovation in the renewable energy industry. *Clean Techn Environ Policy* 23, 1563– 1580. <u>https://doi.org/10.1007/s10098-021-02048-5</u>
- Aichholzer, J., Friesl, C., Hajdinjak, S., and Kritzinger, S. eds. (2019). Quo vadis, Österreich? Wertewandel zwischen 1990 und 2018. Wien: Czernin Verlag.
- Aiginger, K. (2016). New Dynamics for Europe: Reaping the Benefits of Socio-ecological transition. Part I: Synthesis. Available at: <u>https://www.wifo.ac.at/jart/prj3/wifo/resources/person\_dokument/person\_dokument.jart?publikationsid=58791&mime\_ type=application/pdf</u>
- Aigner, E., Görg, C., Madner, V., Muhar, A., Posch, A., Steininger, K. W., et al. (2023a). "Summary for Policymakers," in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 19–33. <u>https:// doi.org/10.1007/978-3-662-66497-1\_2</u>
- Aigner, E., Görg, C., Madner, V., Novy, A., Steininger, K. W., Nabernegg, S., et al. (2023b). "Kapitel 1. Einleitung: Strukturen für ein klimafreundliches Leben," in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 173–194. <u>https://doi.org/10.1007/978-3-662-66497-1\_5</u>
- Aigner, E., and Lichtenberger, H. (2021). "Pflege: Sorglos? Klimasoziale Antworten auf die Pflegekrise," in *Klimasoziale Politik: Eine gerechte und emissionsfreie Gesellschaft gestalten*, eds. BEIGEWUM, Attac, and Armutskonferenz (bahoe books), 175–183.
- Aigner, G., Steiger, R., and Mayer, M. (2024). Snowmaking in Austria: Energy consumption, water turnover, CO2 emissions. CISS 9, 028. <u>https://doi.org/10.36950/2024.4ciss028</u>
- AK Wien (2023). AK Wien, die Interessenvertretung für Arbeitnehmer. Arbeiterkammer Wien. Available at: <u>https://wien.arbeiterkammer.at/index.html</u> (Accessed September 14, 2023).
- AK Wien (2024). Eine Zukunft für die Vielen: Der Plan der Arbeiterkammer für den sozialen und ökologischen Umbau. Vienna: Verlag Arbeiterkammer Wien. Available at: <u>https://wien.arbeiterkammer.at/interessenvertretung/arbeitundsoziales/sozialstaat/AK-Umbauplan\_Langfassung\_2024.pdf</u> (Accessed October 22, 2024).
- AK Wien, and ÖGB (2024). Gemeinsam bewegen. Available at: https://emedien.arbeiterkammer.at/viewer/image/AC08718425 \_2024\_06/2/#topDocAnchor
- Al Baroudi, H., Awoyomi, A., Patchigolla, K., Jonnalagadda, K., and Anthony, E. J. (2021). A review of large-scale CO2 shipping and marine emissions management for carbon capture, utilisation and storage. *Applied Energy* 287, 116510. <u>https://doi.org/10.1016/j.apenergy.2021.116510</u>

- Alcott, B. (2008). The sufficiency strategy: Would rich-world frugality lower environmental impact? *Ecological Economics* 64, 770–786. https://doi.org/10.1016/j.ecolecon.2007.04.015
- Alexandri, E., Antón, J.-I., and Lewney, R. (2024). The impact of climate change mitigation policies on European labour markets. *Ecological Economics* 216, 108022. <u>https://doi.org/10.1016/j.</u> ecolecon.2023.108022
- Alton, V., Binderbauer, P., Cvetkovska, R., Drexler-Schmid, G., Gahleitner, B., Geyer, R., et al. (2022). Pathway To Industrial Decarbonisation. Scenarios For The Development Of The Industrial Sector In Austria. Vienna, Austria. Available at: <u>https:// www.nefi.at/files/media/Pdfs/NEFI\_Szenarienbericht\_v15</u> <u>WHY\_Design.pdf</u> (Accessed June 1, 2022).
- AMS (2024). Die New-Skills-Gespräche des AMS Österreich 2022–2024. Available at: <u>https://forschungsnetzwerk.ams.at/</u> <u>dam/jcr:beac615c-7f54-438f-abc1-04609a2bf2d9/AMS\_re-</u> port\_177\_-\_New\_Skills\_Sammelband.pdf
- Andersen, A. D., Geels, F. W., Coenen, L., Hanson, J., Korsnes, M., Linnerud, K., et al. (2023). Faster, broader, and deeper! Suggested directions for research on net-zero transitions. Oxford Open Energy 2, oiad007. <u>https://doi.org/10.1093/ooenergy/oiad007</u>
- Ang, T.-Z., Salem, M., Kamarol, M., Das, H. S., Nazari, M. A., and Prabaharan, N. (2022). A comprehensive study of renewable energy sources: Classifications, challenges and suggestions. *Energy Strategy Reviews* 43, 100939. <u>https://doi.org/10.1016/j.esr.2022.100939</u>
- Angus, S., Armstrong, B., and Reuck, K. M. de eds. (1987). International Thermodynamic Tables of the Fluid State Carbon Dioxide. *International Union of Pure and Applied Chemistry*, 656–732.
- Antal, M. (2014). Green goals and full employment: Are they compatible? *Ecological Economics* 107, 276–286. <u>https://doi.org/10.1016/j.ecolecon.2014.08.014</u>
- Antal, M., Lehmann, B., Guimaraes, T., Halmos, A., and Lukács, B. (2024). Shorter hours wanted? A systematic review of workingtime preferences and outcomes. *International Labour Review*, ilr.12406. <u>https://doi.org/10.1111/ilr.12406</u>
- APCC (2019). Österreichischer Special Report Gesundheit, Demographie und Klimawandel (ASR18). Austrian Panel on Climate Change (APCC). Wien, Österreich: Verlag der ÖAW. Available at: https://austriaca.at/8427-0 (Accessed September 20, 2022).
- Aram, F., García, E. H., Solgi, E., and Mansournia, S. (2019). Urban green space cooling effect in cities. *Heliyon* 5, e01339. <u>https:// doi.org/10.1016/j.heliyon.2019.e01339</u>
- Arbeitszeiten im Fokus Daten, Gestaltung, Bedarfe (2021). Wien: Forba, AK. Available at: <u>https://www.forba.at/wp-content/uploads/2021/01/210129\_AK-Arbeitszeiten\_im\_Fokus2021.pdf</u>
- Arnz, M., Göke, L., Thema, J., Wiese, F., Wulff, N., Kendziorski, M., et al. (2024). Avoid, Shift or Improve passenger transport? Impacts on the energy system. *Energy Strategy Reviews* 52, 101302. <u>https://doi.org/10.1016/j.esr.2024.101302</u>
- Auel, K., and Schmidt, P. (2023). "Austria: Weathering the Storm and Greening the Economy?," in *Climate Change and the Future* of Europe: Views from the Capitals, eds. M. Kaeding, J. Pollak, and P. Schmidt (Cham: Springer International Publishing), 3–6. https://doi.org/10.1007/978-3-031-23328-9\_1
- Aulenbacher, B., Lutz, H., and Schwiter, K. (2021). Gute Sorge ohne gute Arbeit? Live-in-Care in Deutschland, Österreich und der Schweiz., 1. Auflage. Weinheim: Beltz Juventa.
- Ausfelder, F., Baltac, S., Berly, T., Beuttler, C., Bittleston, S., Blanco, H., et al. (2020). Special Report on Carbon Capture Utilisation

and Storage. IEA. Available at: <u>https://iea.blob.core.windows.</u> net/assets/181b48b4-323f-454d-96fb-0bb1889d96a9/CCUS\_ in\_clean\_energy\_transitions.pdf (Accessed August 21, 2023).

- Ayres, R. U. (1999). The second law, the fourth law, recycling and limits to growth. *Ecological Economics* 29, 473–483. <u>https://doi.org/10.1016/S0921-8009(98)00098-6</u>
- Bachner, G. (2017). Assessing the economy-wide effects of climate change adaptation options of land transport systems in Austria. *Reg Environ Change* 17, 929–940. <u>https://doi.org/10.1007/ s10113-016-1089-x</u>
- Bachner, G., Bednar-Friedl, B., Birk, S., Feichtinger, G., Gobiet, A., Gutschi, C., et al. (2013). Impacts of Climate Change and Adaptation in the Electricity Sector – The Case of Austria in a Continental European Context (EL.ADAPT). Graz: Wegener Center Scientific Report 51-2013, Wegener Center Verlag.
- Bachner, G., Bednar-Friedl, B., and Knittel, N. (2019a). How does climate change adaptation affect public budgets? Development of an assessment framework and a demonstration for Austria. *Mitig Adapt Strateg Glob Change* 24, 1325–1341. <u>https://doi. org/10.1007/s11027-019-9842-3</u>
- Bachner, G., Knittel, N., Poledna, S., Hochrainer-Stigler, S., and Reiter, K. (2023). Revealing indirect risks in complex socioeconomic systems: A highly detailed multi-model analysis of flood events in Austria. *Risk Analysis* 44, 229–243. <u>https://doi.org/10.1111/risa.14144</u>
- Bachner, G., Mayer, J., and Steininger, K. W. (2019b). Costs or benefits? Assessing the economy-wide effects of the electricity sector's low carbon transition – The role of capital costs, divergent risk perceptions and premiums. *Energy Strategy Reviews* 26, 100373. <u>https://doi.org/10.1016/j.esr.2019.100373</u>
- Bachner, G., Mayer, J., Steininger, K. W., Anger-Kraavi, A., Smith, A., and Barker, T. S. (2020a). Uncertainties in macroeconomic assessments of low-carbon transition pathways – The case of the European iron and steel industry. *Ecological Economics* 172, 106631. https://doi.org/10.1016/j.ecolecon.2020.106631
- Bachner, G., Wolkinger, B., Mayer, J., Tuerk, A., and Steininger, K. W. (2020b). Risk assessment of the low-carbon transition of Austria's steel and electricity sectors. *Environmental Innovation* and Societal Transitions 35, 309–332. <u>https://doi.org/10.1016/j.eist.2018.12.005</u>
- Bader, C., Moser, S., Neubert, S. F., Hanbury, H. A., and Lannen, A. (2021). Free Days for Future? Bern, Switzerland: Centre for Development and Environment, University of Bern, Switzerland. <u>https://doi.org/10.48350/157757</u>
- Balas, M., Felderer, A., Völler, S., Zeitz, F., and Margelik, E. (2021). Zweiter Fortschrittsbericht zur österreichischen Strategie zur Anpassung an den Klimawandel. Wien: BMK. Available at: <u>https://www.bmk.gv.at/dam/jcr:4a7614de-cbbc-47b4-bd01-3ac3d079c509/klimawandel-fortschrittsbericht-2021.pdf</u> (Accessed October 2, 2023).
- Baltac, S., Pusceddu, E., O'Sullivan, C., Galbraith-Olive, H., and Henderson, C. (2023). Unlocking Europe's CO2 Storage Potential: Analysis of Optimal CO2 Storage in Europe. Available at: <u>https://www.catf.us/resource/unlocking-europes-co2-stor-</u> age-potential/ (Accessed August 10, 2023).
- Bärnthaler, R., and Gough, I. (2023). Provisioning for sufficiency: envisaging production corridors. Sustainability: Science, Practice and Policy 19, 2218690. <u>https://doi.org/10.1080/15487733.</u> 2023.2218690
- Barrett, J., Pye, S., Betts-Davies, S., Broad, O., Price, J., Eyre, N., et al. (2022). Energy demand reduction options for meeting na-

tional zero-emission targets in the United Kingdom. *Nat Energy* 7, 726–735. <u>https://doi.org/10.1038/s41560-022-01057-y</u>

- Barth, T., Jochum, G., and Littig, B. eds. (2016). Nachhaltige Arbeit. Soziologische Beiträge zur Neubestimmung der gesellschaftlichen Naturverhältnisse. Frankfurt: Campus Verlag.
- Bashmakov, I. A., Nilsson, L. J., Acquaye, A., Bataille, C., Cullen, J. M., de la Rue du Can, S., et al. (2022). Industry. Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, 1161–1244. <u>https://doi.org/10.1017/9781009157926.013</u>
- Bataille, C., Åhman, M., Neuhoff, K., Nilsson, L. J., Fischedick, M., Lechtenböhmer, S., et al. (2018). A review of technology and policy deep decarbonization pathway options for making energy-intensive industry production consistent with the Paris Agreement. *Journal of Cleaner Production* 187, 960–973. <u>https:// doi.org/10.1016/j.jclepro.2018.03.107</u>
- Bauer, S., Bux, K., Dieterich, F., Gabriel, K., Kienast, C., Klar, S., et al. (2022). Klimawandel und Arbeitsschutz. Dortmund/Berlin/ Dresden: Bundesanstalt für Arbeitsschutz und Arbeitsmedizin (BAuA). <u>https://doi.org/10.21934/BAUA:BERICHT20220601</u>
- Bauhardt, C. (2013). Wege aus der Krise? Green New Deal—Postwachstumsgesellschaft—Solidarische Ökonomie: Alternativen zur Wachstumsökonomie aus feministischer Sicht. GENDER – Zeitschrift für Geschlecht, Kultur und Gesellschaft 5, 9–26.
- Baumann, M., Dolna-Gruber, C., Goritschnig, W., Pauritsch, G., and Rohrer, M. (2021). Klima- und Energiestrategien der Länder. Wien: Austrian Energy Agency. Available at: <u>https:// static1.squarespace.com/static/5b978be0697a98a663136c47/t/ 606e8e68f395266edb4fee99/1617858166401/IGW-BL+Vergleich\_Endbericht\_Final.pdf</u> (Accessed September 14, 2023).
- Baumol, W. J. (1967). Macroeconomics of Unbalanced Growth: The Anatomy of Urban Crisis. *The American Economic Review* 57, 415–426.
- Bayliss, K., and Fine, B. (2020). A Guide to the Systems of Provision Approach: Who Gets What, How and Why. Cham: Springer International Publishing. <u>https://doi.org/10.1007/978-3-030-</u> 54143-9
- Bednar-Friedl, B., Biesbroek, R., Schmidt, D. N., Alexander, P., Børsheim, K. Y., Carnicer, J., et al. (2022a). "Europe," in *Climate Change 2022: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*, eds. H. O. Pörtner, D. C. Roberts, M. Tignor, E. S. Poloczanska, K. Mintenbeck, A. Alegría, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press), 1817–1927.
- Bednar-Friedl, B., Knittel, N., Raich, J., and Adams, K. M. (2022b). Adaptation to transboundary climate risks in trade: Investigating actors and strategies for an emerging challenge. WIREs Climate Change 13, e758. <u>https://doi.org/10.1002/wcc.758</u>
- Bellamy, R., Chilvers, J., and Vaughan, N. E. (2016). Deliberative Mapping of options for tackling climate change: Citizens and specialists "open up" appraisal of geoengineering. *Public Underst Sci* 25, 269–286. <u>https://doi.org/10.1177/0963662514548628</u>
- Benson, S., Cook, P., Anderson, J., Bachu, S., Nimir, H., Basu, B., et al. (2005). "Chapter 5 – Underground geological storage," in *IPCC Special Report on Carbon Dioxide Capture and Storage*, eds. B. Metz, O. Davidson, H. de Coninck, M. Loos, and L. Meyer (Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press), 197–265.

- Bento, N., Grubler, A., Boza-Kiss, B., De Stercke, S., Krey, V., Mc-Collum, D. L., et al. (2024). Leverage demand-side policies for energy security. *Science* 383, 946–949. <u>https://doi.org/10.1126/ science.adj6150</u>
- Berger, T., Amann, C., Formayer, H., Korjenic, A., Pospichal, B., Neururer, C., et al. (2014a). Impacts of urban location and climate change upon energy demand of office buildings in Vienna, Austria. *Building and Environment* 81, 258–269. <u>https://doi. org/10.1016/j.buildenv.2014.07.007</u>
- Berger, T., Amann, C., Formayer, H., Korjenic, A., Pospischal, B., Neururer, C., et al. (2014b). Impacts of climate change upon cooling and heating energy demand of office buildings in Vienna, Austria. *Energy and Buildings* 80, 517–530. <u>https://doi. org/10.1016/j.enbuild.2014.03.084</u>
- Berger, T., and Höltl, A. (2019). Thermal insulation of rental residential housing: Do energy poor households benefit? A case study in Krems, Austria. *Energy Policy* 127, 341–349. <u>https://doi.org/10.1016/j.enpol.2018.12.018</u>
- Berthet, E., Lavalley, J., Anquetil-Deck, C., Ballesteros, F., Stadler, K., Soytas, U., et al. (2024). Assessing the social and environmental impacts of critical mineral supply chains for the energy transition in Europe. *Global Environmental Change* 86, 102841. <u>https://doi.org/10.1016/j.gloenvcha.2024.102841</u>
- Biesecker, A., and Hofmeister, S. (2006). Die Neuerfindung des Ökonomischen: ein (re)produktionstheoretischer Beitrag zur sozial-ökologischen Forschung. München: Oekom-VerlG.
- Biesecker, A., Mathes, M., Schön, S., and Scurrell, B. eds. (2000). Vorsorgendes Wirtschaften: auf dem Weg zu einer Ökonomie des guten Lebens; eine Publikation aus dem Netzwerk Vorsorgendes Wirtschaften. Bielefeld: Kleine Verlag.
- Bird, D. N., De Wit, R., Schwaiger, H. P., Andre, K., Beermann, M., and Žuvela-Aloise, M. (2019). Estimating the daily peak and annual total electricity demand for cooling in Vienna, Austria by 2050. Urban Climate 28, 100452. <u>https://doi.org/10.1016/j. uclim.2019.100452</u>
- Blättler, K., Wallimann, H., and von Arx, W. (2024). Free public transport to the destination: A causal analysis of tourists' travel mode choice. *Transportation Research Part A: Policy and Practice* 187, 104166. <u>https://doi.org/10.1016/j.tra.2024.104166</u>
- Blaydes, H., Potts, S. G., Whyatt, J. D., and Armstrong, A. (2021). Opportunities to enhance pollinator biodiversity in solar parks. *Renewable and Sustainable Energy Reviews* 145, 111065. <u>https://doi.org/10.1016/j.rser.2021.111065</u>
- Blomsma, F., Pieroni, M., Kravchenko, M., Pigosso, D. C. A., Hildenbrand, J., Kristinsdottir, A. R., et al. (2019). Developing a circular strategies framework for manufacturing companies to support circular economy-oriented innovation. *Journal of Cleaner Production* 241, 118271. <u>https://doi.org/10.1016/j.jclepro.2019.118271</u>
- Bloomfield, H. C., Brayshaw, D. J., Troccoli, A., Goodess, C. M., De Felice, M., Dubus, L., et al. (2021). Quantifying the sensitivity of european power systems to energy scenarios and climate change projections. *Renewable Energy* 164, 1062–1075. <u>https:// doi.org/10.1016/j.renene.2020.09.125</u>
- Blöschl, G., Blaschke, A. P., Haslinger, K., Hofstätter, M., Parajka, J., Salinas, J., et al. (2018). Auswirkungen der Klimaänderung auf Österreichs Wasserwirtschaft – ein aktualisierter Statusbericht. Österreichische Wasser- und Abfallwirtschaft 70, 462–473. https://doi.org/10.1007/s00506-018-0498-0
- Blöschl, G., Schöner, W., Kroiß, H., P. Blaschke, A., Böhm, R., Haslinger, K., et al. (2011). Anpassungsstrategien an den Kli-

mawandel für Österreichs Wasserwirtschaft – Ziele und Schlussfolgerungen der Studie für Bund und Länder. Österreichische Wasser- und Abfallwirtschaft 63, 1–10. <u>https://doi.</u> org/10.1007/s00506-010-0274-2

- BMF (2024). Carbon Management. Available at: <u>https://www.bmf.</u> gv.at/themen/klimapolitik/carbon\_management.html
- BMK (2022). Österreich auf dem Weg zu einer nachhaltigen und zirkulären Gesellschaft. Die österreichische Kreislaufstrategie. Wien: Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie. Available at: <u>https:// www.bmk.gv.at/themen/klima\_umwelt/abfall/Kreislaufwirtschaft/strategie.html</u>
- BMK (2023a). Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie (BMK). Available at: <u>https://www.bmk.gv.at/public.html</u> (Accessed September 14, 2023).
- BMK (2023b). Energie in Österreich. Zahlen, Daten, Fakten. Vienna: Ministry of Climate Action, Environment, Energy, Mobility, Innovation and Technology (BMK). Available at: <u>https://www. bmk.gv.at/dam/jcr:da4e9dfd-f51c-44b8-894c-9b049a8336cb/ BMK\_Energie\_in\_OE2023\_barrierefrei.pdf</u> (Accessed November 5, 2024).
- BMK (2023c). Integrierter österreichischer Netzinfrastrukturplan (ÖNIP). Vienna. Available at: <u>https://www.bmk.gv.at/themen/ energie/energieversorgung/netzinfrastrukturplan.html</u> (Accessed September 14, 2023).
- BMK (2024a). Integrierter österreichischer Netzinfrastrukturplan (ÖNIP). Vienna: BMK. Available at: <u>https://www.bmk.gv.at/</u> <u>dam/jcr:d4d6888b-8e57-4ec6-87c3-0ffec220d83f/Integrierteroesterreichischer-Netzinfrastrukturplan.pdf</u> (Accessed May 6, 2024).
- BMK (2024b). Ressourcennutzung in Österreich 2024. Band 4. Vienna: Ministry of Climate Action, Environment, Energy, Mobility, Innovation and Technology (BMK).
- BMNT (2019). Integrated National Energy and Climate Plan for Austria. 2021–2030. pursuant to Regulation (EU) 2018/1999 of the European Parliament and of the Council on the Governance of the Energy Union and Climate Action. Vienna: Federal Ministry for Sustainability and Tourism, Austria. Available at: <u>https://energy.ec.europa.eu/system/files/2020-03/at\_final\_ necp\_main\_en\_0.pdf</u> (Accessed May 7, 2024).
- BMSGPK (2024). Sozialbericht 2024. Band II: Sozialpolitische Analysen. Vienna: Bundesministerium für Soziales, Gesundheit, Pflege und Konsumentenschutz (BMSGPK). Available at: <u>https://www.parlament.gv.at/dokument/XXVII/III/1146/imfname\_1621786.pdf</u> (Accessed May 6, 2024).
- Böhm, H., Moser, S., Puschnigg, S., and Zauner, A. (2021). Power-to-hydrogen & district heating: Technology-based and infrastructure-oriented analysis of (future) sector coupling potentials. *International Journal of Hydrogen Energy* 46, 31938–31951. <u>https://doi.org/10.1016/j.ijhydene.2021.06.233</u>
- Bohnenberger, K. (2021). "Die Zukunft sozialer Sicherungssysteme: Sechs Kriterien Nachhaltiger Sozialpolitik.," in Soziale Sicherungssysteme im Umbruch: Beiträge zur sozial-ökologischen Transformation, eds. D. Rodenhäuser, H. Vetter, B. Held, and H. Diefenbacher (Marburg: Metropolis-Verlag).
- Bohnenberger, K. (2022a). Greening work: labor market policies for the environment. *Empirica* 49, 347–368. <u>https://doi.org/10. 1007/s10663-021-09530-9</u>

- Bohnenberger, K. (2022b). Is it a green or brown job? A Taxonomy of Sustainable Employment. *Ecological Economics* 200, 107469. <u>https://doi.org/10.1016/j.ecolecon.2022.107469</u>
- Bohnenberger, K. (2023). Peaks and gaps in eco-social policy and sustainable welfare: A systematic literature map of the research landscape. *European Journal of Social Security* 25, 328–346. https://doi.org/10.1177/13882627231214546
- Bohnenberger, K., and Schultheis, J. (2021). "Sozialpolitik für eine klimagerechte Gesellschaft," in *Klimasoziale Politik. Eine* gerechte und emissionsfreie Gesellschaft gestalten, eds. Die Armutskonferenz, Attac, and Beigewum (Wien: Bahoe Books), 71–84.
- Böhringer, C., and Lange, A. (2013). "European Union's Emissions Trading System," in *Encyclopedia of Energy, Natural Resource,* and Environmental Economics, ed. J. F. Shogren (Elsevier), 155– 160. <u>https://doi.org/10.1016/B978-0-12-375067-9.00070-X</u>
- Boltz, S., Klotz, M., and Wetzstein, A. (2023). Auswirkungen der Klimakrise auf Sicherheit und Gesundheit bei der Arbeit. Available at: <u>https://forum.dguv.de/ausgabe/1-2023/artikel/</u> <u>auswirkungen-der-klimakrise-auf-sicherheit-und-gesundheit-bei-der-arbeit#:~:text=Wenn%20es%20beispielsweise%20</u> <u>zukünftig%20zu,und%20einem%20erhöhten%20Unfallrisiko%20führen</u>. (Accessed May 6, 2024).
- Brad, A., Schneider, E., Maneka, D., Hirt, C., and Gingrich, S. (2024). The politics of carbon management in Austria: Emerging fault lines on carbon capture, storage, utilization and removal. *Energy Research & Social Science* 116, 103697. <u>https:// doi.org/10.1016/j.erss.2024.103697</u>
- Brand, U., and Wissen, M. (2021). *The Imperial Mode of Living: Everyday Life and the Ecological Crisis of Capitalism*. Brooklyn: Verso Books.
- Brand-Correa, L. I., Martin-Ortega, J., and Steinberger, J. K. (2018). Human Scale Energy Services: Untangling a 'golden thread'. *Energy Research & Social Science* 38, 178–187. <u>https://doi.org/10.1016/j.erss.2018.01.008</u>
- Brandstätter, B., Arnold, P., and Ott, H. (2025). CO2 Trapping Curves from Digital Rock Physics: Application to Austrian Rock Types.
- Brás, T. A., Simoes, S. G., Amorim, F., and Fortes, P. (2023). How much extreme weather events have affected European power generation in the past three decades? *Renewable and Sustainable Energy Reviews* 183, 113494. <u>https://doi.org/10.1016/j. rser.2023.113494</u>
- Brenner, A.-K., Haas, W., Krüger, T., Matej, S., Haberl, H., Schug, F., et al. (2024). What drives densification and sprawl in cities? A spatially explicit assessment for Vienna, between 1984 and 2018. *Land Use Policy* 138, 107037. <u>https://doi.org/10.1016/j. landusepol.2023.107037</u>
- Briggs, C., Atherton, A., Gill, J., Langdon, R., Rutovitz, J., and Nagrath, K. (2022). Building a 'Fair and Fast' energy transition? Renewable energy employment, skill shortages and social licence in regional areas. *Renewable and Sustainable Energy Transition* 2, 100039. <u>https://doi.org/10.1016/j.rset.2022.100039</u>
- Bröthaler, J., Getzner, M., Müller, H. L., Plank, L., Miess, M., Niedertscheider, M., et al. (2023). Öffentliche Investitionen für den Klimaschutz in Österreich: Potenziale des öffentlichen Vermögens. Kammer für Arbeiter und Angestellte für Wien. Available at: <u>https://repositum.tuwien.at/handle/20.500.12708/187853</u> (Accessed September 6, 2023).

- Brown, T., Schlachtberger, D., Kies, A., Schramm, S., and Greiner, M. (2018). Synergies of sector coupling and transmission reinforcement in a cost-optimised, highly renewable European energy system. *Energy* 160, 720–739. <u>https://doi.org/10.1016/j.</u> energy.2018.06.222
- Brugger, K., Dinhof, K., Schmidt, A., Aigner, E., and Fischer, M. (2024a). Gesundes Arbeiten trotz Klimawandel. Befragung zu Klimawandel und Gesundheit unter Gesundheitsberufen. Wien: Gesundheit Österreich. Available at: <u>https://jasmin.goeg.</u> <u>at/id/eprint/3777</u>
- Brugger, K., Horváth, I., Marent, J., and Schmidt, A. E. (2024b). Handbuch zur Stärkung der Klimakompetenz in den Gesundheitsberufen. Wien: Gesundheit Österreich. Available at: <u>https://jasmin.goeg.at/id/eprint/3362</u>
- Brutschin, E., Baum, C. M., Fritz, L., Low, S., Sovacool, B. K., and Riahi, K. (2024). Drivers and attitudes of public support for technological solutions to climate change in 30 countries. *Environ. Res. Lett.* 19, 114098. <u>https://doi.org/10.1088/1748-9326/ ad7c67</u>
- Büchs, M. (2021). Sustainable welfare: How do universal basic income and universal basic services compare? *Ecological Economics* 189, 107152. <u>https://doi.org/10.1016/j.ecolecon.2021.</u> <u>107152</u>
- Buhl, J., Schipperges, M., and Liedtke, C. (2017). "Die Ressourcenintensität der Zeit und ihre Bedeutung für nachhaltige Lebensstile," in *Verbraucherwissenschaften*, eds. P. Kenning, A. Oehler, L. A. Reisch, and C. Grugel (Wiesbaden: Springer FGabler), 295–311. https://doi.org/10.1007/978-3-658-10926-4\_16
- Bühn, S., and Voss, M. (2023). Klimawandel und Gesundheit Auswirkungen auf die Arbeitswelt. Berlin: CPHP. Available at: <u>https://www.arbeit-sicher-und-gesund.de/fileadmin/PDFs/ klug-gutachten-klimawandel-und-gesundheit-auswirkungen-auf-die-arbeitswelt\_stand-feb-2023.pdf</u> (Accessed May 6, 2024).
- Bui, M., Adjiman, C. S., Bardow, A., Anthony, E. J., Boston, A., Brown, S., et al. (2018). Carbon capture and storage (CCS): the way forward. *Energy Environ. Sci.* 11, 1062–1176. <u>https://doi.org/10.1039/C7EE02342A</u>
- Bursa, B., Mailer, M., and Axhausen, K. W. (2022). Travel behavior on vacation: transport mode choice of tourists at destinations. *Transportation Research Part A: Policy and Practice* 166, 234– 261. <u>https://doi.org/10.1016/j.tra.2022.09.018</u>
- Butnar, I., Cronin, J., and Pye, S. (2020). Review of Carbon Capture Utilisation and Carbon Capture and Storage in future EU decarbonisation scenarios. UCL Energy Institute. Available at: <u>https://zeroemissionsplatform.eu/wp-content/uploads/Report-Review-of-CCU-and-CCS-in-future-EU-decarbonisation-scenarios.pdf</u> (Accessed August 21, 2023).
- CaCTUS (2024). Projekt-Website des ACRP-Projekts CaCTUS CARBON CAPTURE, TRANSFORMATION, UTILIZATION & STORAGE. Available at: <u>https://project-cactus.at/</u> (Accessed August 21, 2023).
  - Wolf-Zöllner, P., Böhm, H., Veseli, A., Hochmeister, S., Kulich, J., Fazeni-Fraisl, K., Lehner, M., Kienberger, T., Ott, H., Fleischhacker, J., Sachs, N., & Kapfer, M. (2025). CaCTUS Carbon Capture & Transformation, Utilization and Storage: CCUS in Österreich Potenziale, Technologien, Folgenabschätzung und Einordnung der rechtlichen Rahmenbedingungen. Berg- und hüttenmännische Monatshefte: BHM; <u>https://link.springer.com/article/10.1007/s00501-025-01571-y</u>

Two policy briefs are available at: <u>https://ccca.ac.at/news/detail/</u> cactus-publikation-und-policy-briefs-erschienen

- Calisto Friant, M., Vermeulen, W. J. V., and Salomone, R. (2021). Analysing European Union circular economy policies: words versus actions. *Sustainable Production and Consumption* 27, 337–353. https://doi.org/10.1016/j.spc.2020.11.001
- Cantzler, J., Creutzig, F., Ayargarnchanakul, E., Javaid, A., Wong, L., and Haas, W. (2020). Saving resources and the climate? A systematic review of the circular economy and its mitigation potential. *Environ. Res. Lett.* 15, 123001. <u>https://doi.org/ 10.1088/1748-9326/abbeb7</u>
- Carmona, L. G., Whiting, K., Carrasco, A., Sousa, T., and Domingos, T. (2017). Material Services with Both Eyes Wide Open. *Sustainability* 9, 1508. <u>https://doi.org/10.3390/su9091508</u>
- Castro, C. G., Trevisan, A. H., Pigosso, D. C. A., and Mascarenhas, J. (2022). The rebound effect of circular economy: Definitions, mechanisms and a research agenda. *Journal of Cleaner Production* 345, 131136. <u>https://doi.org/10.1016/j.jclepro.2022.131136</u>
- CEPI (2023). KEY STATISTICS 2022. European pulp & paper industry. Brussels, Belgium: Confederation of European Paper Industries. Available at: <u>https://www.cepi.org/wp-content/uploads/2023/07/2022-Key-Statistics-FINAL.pdf</u>
- Chiappinelli, O., Gerres, T., Neuhoff, K., Lettow, F., Coninck, de H., Felsmann, B., et al. (2021). A green COVID-19 recovery of the EU basic materials sector: identifying potentials, barriers and policy solutions. *Climate Policy* 21, 1328–1346.
- Christensen, T. H., Friis, F., Bettin, S., Throndsen, W., Ornetzeder, M., Skjølsvold, T. M., et al. (2020). The role of competences, engagement, and devices in configuring the impact of prices in energy demand response: Findings from three smart energy pilots with households. *Energy Policy* 137, 111142. <u>https://doi.org/10.1016/j.enpol.2019.111142</u>
- Chung, H. (2022). A Social Policy Case for a Four-Day Week. Journal of Social Policy 51, 551–566. <u>https://doi.org/10.1017/</u> S0047279422000186
- Cieplinski, A., D'Alessandro, S., and Guarnieri, P. (2021). Environmental impacts of productivity-led working time reduction. *Ecological Economics* 179, 106822. <u>https://doi.org/10.1016/j.ecolecon.2020.106822</u>
- Ciocirlan, C. E. (2017). Environmental Workplace Behaviors: Definition Matters. Organization & Environment 30, 51–70. <u>https://doi.org/10.1177/1086026615628036</u>
- Ciocirlan, C. E. (2023). Have me do, and I'll always be true: Exploring the match between green employees and their jobs. *Journal* of Cleaner Production 383, 135471. <u>https://doi.org/10.1016/j.jclepro.2022.135471</u>
- Cleveland, D. A., Phares, N., Nightingale, K. D., Weatherby, R. L., Radis, W., Ballard, J., et al. (2017). The potential for urban household vegetable gardens to reduce greenhouse gas emissions. *Landscape and Urban Planning* 157, 365–374. <u>https://doi. org/10.1016/j.landurbplan.2016.07.008</u>
- Cohen, M., and MacGregor, S. (2020). Towards a feminist green new deal for the UK: A PAPER FOR THE WBG COMMIS-SION ON A GENDER-EQUAL ECONOMY. Women's Budget Group. Available at: <u>https://www.research.manchester.ac.uk/</u> <u>portal/files/170845257/Cohen\_and\_MacGregor\_Feminist\_ Green\_New\_Deal\_2020.pdf</u>
- COM/2015/0614 final (2015). COMMUNICATION FROM THE COMMISSION TO THE EUROPEAN PARLIAMENT, THE COUNCIL, THE EUROPEAN ECONOMIC AND SOCIAL COMMITTEE AND THE COMMITTEE OF THE REGIONS

Closing the loop – An EU action plan for the Circular Economy. Available at: <u>https://eur-lex.europa.eu/legal-content/EN/</u> <u>TXT/?uri=CELEX%3A52015DC0614</u> (Accessed September 14, 2022).

- COM/2019/640 final (2019). COMMUNICATION FROM THE COMMISSION TO THE EUROPEAN PARLIAMENT, THE EUROPEAN COUNCIL, THE COUNCIL, THE EUROPE-AN ECONOMIC AND SOCIAL COMMITTEE AND THE COMMITTEE OF THE REGIONS The European Green Deal. Available at: <u>https://eur-lex.europa.eu/legal-content/EN/TXT/</u> <u>?uri=COM%3A2019%3A640%3AFIN</u> (Accessed September 14, 2022).
- COM/2020/562 final (2020). COMMUNICATION FROM THE COMMISSION TO THE EUROPEAN PARLIAMENT, THE COUNCIL, THE EUROPEAN ECONOMIC AND SOCIAL COMMITTEE AND THE COMMITTEE OF THE REGIONS Stepping up Europe's 2030 climate ambition Investing in a climate-neutral future for the benefit of our people. Available at: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX-%3A52020DC0562 (Accessed September 14, 2023).
- Coote, A., and Percy, A. (2020). *The Case for Universal Basic Services*. Cambridge, UK/Medford, MA: Polity.
- Cox, E., Spence, E., and Pidgeon, N. (2020). Public perceptions of carbon dioxide removal in the United States and the United Kingdom. *Nat. Clim. Chang.* 10, 744–749. <u>https://doi.org/10. 1038/s41558-020-0823-z</u>
- Cremer, C., Esken, A., Fischedick, M., Gruber, E., Idrissova, F., Kuckshinrichs, W., et al. (2008). Sozioökonomische Begleitforschung zur gesellschaftlichen Akzeptanz von Carbon Capture and Storage (CCS) auf nationaler und internationaler Ebene. Wuppertal: Wuppertal Institut, Forschungzentrum Jülich (STE), Fraunhofer Institut (ISI), BSR Sustainability GmbH. Available at: <u>https://epub.wupperinst.org/frontdoor/deliver/index/docId/2989/file/2989\_Akzeptanz\_CCS.pdf</u>
- Creutzig, F., Fernandez, B., Haberl, H., Khosla, R., Mulugetta, Y., and Seto, K. C. (2016). Beyond Technology: Demand-Side Solutions for Climate Change Mitigation. *Annual Review of Envi*ronment and Resources 41, 173–198. <u>https://doi.org/10.1146/</u> annurev-environ-110615-085428
- Creutzig, F., Niamir, L., Bai, X., Callaghan, M., Cullen, J., Díaz-José, J., et al. (2022a). Demand-side solutions to climate change mitigation consistent with high levels of well-being. *Nat. Clim. Chang.* 12, 36–46. <u>https://doi.org/10.1038/s41558-021-01219-y</u>
- Creutzig, F., Roy, J., Devine-Wright, P., Díaz-José, J., Geels, F. W., Grubler, A., et al. (2022b). "Demand, services and social aspects of mitigation.," in *IPCC 2022: Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*, eds. P. R. Shukla, J. Skea, R. Slade, A. A. Khourdajie, R. van Diemen, D. McCollum, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press). <u>https://doi.org/10.1017/9781009157926.007</u>
- Creutzig, F., Roy, J., Lamb, W. F., Azevedo, I. M. L., Bruine de Bruin, W., Dalkmann, H., et al. (2018). Towards demand-side solutions for mitigating climate change. *Nature Climate Change* 8, 260– 263. <u>https://doi.org/10.1038/s41558-018-0121-1</u>
- Creutzig, F., Simoes, S. G., Leipold, S., Berrill, P., Azevedo, I., Edelenbosch, O., et al. (2024). Demand-side strategies key for mitigating material impacts of energy transitions. *Nat. Clim. Chang.* 14, 561–572. https://doi.org/10.1038/s41558-024-02016-z

- Cronin, J., Anandarajah, G., and Dessens, O. (2018). Climate change impacts on the energy system: a review of trends and gaps. *Climatic Change* 151, 79–93. <u>https://doi.org/10.1007/s10584-018-2265-4</u>
- Csoka, B. (2018). 31 Stunden sind genug. *A&W blog.* Available at: <u>https://awblog.at/31-stunden-sind-genug</u> (Accessed September 20, 2021).
- Cullen, J. M. (2017). Circular Economy: Theoretical Benchmark or Perpetual Motion Machine? *J of Industrial Ecology* 21, 483–486. <u>https://doi.org/10.1111/jiec.12599</u>
- Dachs, B., Biege, S., Borowiecki, M., Lay, G., Jäger, A., and Schartinger, D. (2014). Servitisation of European manufacturing: evidence from a large scale database. *The Service Industries Journal* 34, 5–23. https://doi.org/10.1080/02642069.2013.776543
- Damm, A., Köberl, J., and Prettenthaler, F. (2014). Does artificial snow production pay under future climate conditions? – A case study for a vulnerable ski area in Austria. *Tourism Management* 43, 8–21. <u>https://doi.org/10.1016/j.tourman.2014.01.009</u>
- Damm, A., Köberl, J., Prettenthaler, F., Rogler, N., and Töglhofer, C. (2017). Impacts of +2 °C global warming on electricity demand in Europe. *Climate Services* 7, 12–30. <u>https://doi.org/10.1016/j. cliser.2016.07.001</u>
- Dasgupta, S., Van Maanen, N., Gosling, S. N., Piontek, F., Otto, C., and Schleussner, C.-F. (2021). Effects of climate change on combined labour productivity and supply: an empirical, multi-model study. *The Lancet Planetary Health* 5, e455–e465. <u>https://doi. org/10.1016/S2542-5196(21)00170-4</u>
- De Felice, M., Koolen, D., Kanellopoulos, K., Busch, S., and Zucker, A. (2023). Climate variability on Fit for 55 European power systems. *PLoS ONE* 18, e0289705. <u>https://doi.org/10.1371/journal.pone.0289705</u>
- De Silva, G. P. D., Ranjith, P. G., and Perera, M. S. A. (2015). Geochemical aspects of CO2 sequestration in deep saline aquifers: A review. *Fuel* 155, 128–143. <u>https://doi.org/10.1016/j. fuel.2015.03.045</u>
- De Stercke, S. (2014). Dynamics of Energy Systems: A Useful Perspective. Laxenburg, Austria: International Institute for Applied Systems Analysis (IIASA). Available at: <u>https://pure.iiasa.ac.at/</u> id/eprint/11254/1/IR-14-013.pdf
- de Wit, M., Verstraeten-Jochemsen, J., Hoogzaad, J., Kubbinga, B., von Daniels, C., Steenmeijer, M., et al. (2019). The Circularity Gap Report 2019. The Platform for Accelerating the Circular Economy (PACE). Available at: <u>https://circulareconomy.europa.eu/platform/sites/default/files/circularity\_gap\_report\_2019.</u> <u>pdf</u> (Accessed December 9, 2023).
- Deetman, S., de Boer, H. S., Van Engelenburg, M., van der Voet, E., and van Vuuren, D. P. (2021). Projected material requirements for the global electricity infrastructure – generation, transmission and storage. *Resources, Conservation and Recycling* 164, 105200. <u>https://doi.org/10.1016/j.resconrec.2020.105200</u>
- Dengler, C., Dornis, N., and Heck, L. (2024). Klimafreundliche und gesundheitsfördernde Aspekte von Zeitwohlstand. Policy Brief. Wien: Gesundheit Österreich. Available at: <u>https://jasmin.goeg.</u> <u>at/id/eprint/3423/1/GOEGPolicyBrief-Zeitwohlstand\_bf.pdf</u> (Accessed May 6, 2024).
- Dengler, C., and Lang, M. (2019). "Feminism Meets Degrowth. Sorgearbeit in einer Postwachstumsgesellschaft," in Ökonomie des Versorgens. Feministisch-kritische Wirtschaftstheorien im deutschsprachigen Raum., (Weinheim: Beltz Juventa), 305–330.
- Dengler, C., and Lang, M. (2022). Commoning Care: Feminist Degrowth Visions for a Socio-Ecological Transformation. *Feminist*

*Economics* 28, 1–28. <u>https://doi.org/10.1080/13545701.2021.19</u> 42511

- Dengler, C., and Strunk, B. (2018). The Monetized Economy Versus Care and the Environment: Degrowth Perspectives On Reconciling an Antagonism. *Feminist Economics* 24, 160–183. <u>https://doi.org/10.1080/13545701.2017.1383620</u>
- derStandard (2024). Regierung lässt Pläne gegen Bodenversiegelung platzen. Available at: <u>https://www.derstandard.at/story/</u> <u>3000000212620/regierung-laesst-plaene-gegen-bodenversiegelung-platzen</u>
- Diendorfer, C., Gahleitner, B., Dachs, B., Kienberger, T., Nagovnak, P., Böhm, H., et al. (2021). Klimaneutralität Österreichs bis 2040: Beitrag der österreichischen Industrie. Wien: Austrian Institute of Technology, EnergieVerbundTechnik, Energieinstitut, Austrian Energy Agency. Available at: <u>https://www.bmk.gv.at/ themen/klima\_umwelt/gruene-industriepolitik/ziele.html</u> (Accessed August 24, 2023).
- Dillman, K. J., Czepkiewicz, M., Heinonen, J., and Davíðsdóttir, B. (2023). Ecological intensity of social provisioning in mobility systems: A global analysis. *Energy Research & Social Science* 104, 103242. <u>https://doi.org/10.1016/j.erss.2023.103242</u>
- Diouf, B., and Pode, R. (2015). Potential of lithium-ion batteries in renewable energy. *Renewable Energy* 76, 375–380. <u>https://doi.org/10.1016/j.renene.2014.11.058</u>
- Directive 2009/31/EC (2009). Directive 2009/31/EC of the European Parliament and of the Council of 23 April 2009 on the geological storage of carbon dioxide and amending Council Directive 85/337/EEC, European Parliament and Council Directives 2000/60/EC, 2001/80/EC, 2004/35/EC, 2006/12/EC, 2008/1/EC and Regulation (EC) No 1013/2006 (Text with EEA relevance). Available at: <u>https://eur-lex.europa.eu/eli/dir/2009/31/oj</u> (Accessed September 14, 2023).
- Dokulil, M. T., de Eyto, E., Maberly, S. C., May, L., Weyhenmeyer, G. A., and Woolway, R. I. (2021). Increasing maximum lake surface temperature under climate change. *Climatic Change* 165, 56. https://doi.org/10.1007/s10584-021-03085-1
- Domenech, T., and Bahn-Walkowiak, B. (2019). Transition Towards a Resource Efficient Circular Economy in Europe: Policy Lessons From the EU and the Member States. *Ecological Economics* 155, 7–19. <u>https://doi.org/10.1016/j.ecolecon.2017.11.001</u>
- Dorninger, C., Ginrich, S., Haas, W., Brad, A., Schneider, E., and Wiedenhofer, D. (2025). Slow and unequal reduction in Austrian household GHG footprints between 2000 and 2020. Available at: <u>https://dx.doi.org/10.2139/ssrn.5052026</u>
- Dorr, A., Heckl, E., Marcher, A., Petzlberger, K., and Depperschmidt, C. (2023). Green Jobs in Österreich mit Fokus auf Lehrausbildungen. Berufsprofile, Kompetenzen, Beschäftigungschancen. Arbeitsmarktservice Österreich (AMS). Available at: <u>https://forschungsnetzwerk.ams.at/dam/jcr:d1e9fcfa-0e7d-4596-a048-edfc1f10327d/AMS\_report\_174\_-\_Green\_ Jobs\_in\_Oesterreich\_Fokus\_Lehrausbildungen.pdf</u>
- Doss-Gollin, J., Amonkar, Y., Schmeltzer, K., and Cohan, D. (2023). Improving the Representation of Climate Risks in Long-Term Electricity Systems Planning: a Critical Review. *Curr Sustainable Renewable Energy Rep* 10, 206–217. <u>https://doi.org/10.1007/ s40518-023-00224-3</u>
- Dowling, P. (2013). The impact of climate change on the European energy system. *Energy Policy* 60, 406–417. <u>https://doi.org/10.1016/j.enpol.2013.05.093</u>
- Dubois, G., Peeters, P., Ceron, J.-P., and Gössling, S. (2011). The future tourism mobility of the world population: Emission

growth versus climate policy. *Transportation Research Part A: Policy and Practice* 45, 1031–1042. <u>https://doi.org/10.1016/j.tra.2009.11.004</u>

- Dukelow, F., and Murphy, M. P. (2022). Building the Future from the Present: Imagining Post-Growth, Post-Productivist Ecosocial Policy. *Journal of Social Policy* 51, 504–518. <u>https://doi.org/10.1017/S0047279422000150</u>
- Dumbrell, N. P., Kragt, M. E., and Gibson, F. L. (2016). What carbon farming activities are farmers likely to adopt? A best-worst scaling survey. *Land Use Policy* 54, 29–37. <u>https://doi.org/10.1016/j.</u> <u>landusepol.2016.02.002</u>
- Dziejarski, B., Krzyżyńska, R., and Andersson, K. (2023). Current status of carbon capture, utilization, and storage technologies in the global economy: A survey of technical assessment. *Fuel* 342, 127776. https://doi.org/10.1016/j.fuel.2023.127776
- Edberg, K. (2024). E-biking within a transitioning transport system: the quest for flexible mobility. *Mobilities* 19, 428–443. <u>https://doi.org/10.1080/17450101.2023.2259111</u>
- Edelenbosch, O. Y., Hof, A. F., Van Den Berg, M., De Boer, H. S., Chen, H.-H., Daioglou, V., et al. (2024). Reducing sectoral hardto-abate emissions to limit reliance on carbon dioxide removal. *Nat. Clim. Chang.* 14, 715–722. <u>https://doi.org/10.1038/s41558-024-02025-y</u>
- Eibinger, T., Manner, H., and Steininger, K. (2024). Memo: Die Entwicklung der österreichischen Treibhausgasemissionen seit 2021. Graz: Wegener Center für Klima und Globalen Wandel. Available at: https://wegccloud.uni-graz.at/s/jEz4oGERocGrQe6
- Eitzinger, J., Haberl, H., Amon, B., Blamauer, B., Essl, F., Gaube, V., et al. (2014). "Land- und Forstwirtschaft, Wasser, Ökosysteme und Biodiversität," in Österreichischer Sachstandsbericht Klimawandel 2014 (AAR14), Austrian Panel on Climate Change (APCC), ed. Austrian Panel on Climate Change (APCC) (Vienna: Verlag der Österreichischen Akademie der Wissenschaften), 771–856.
- Ellen Macarthur Foundation (2013). Towards the Circular Economy Vol. 1: An Economic and business rationale for an accelerated transition. Available at: <u>https://www.ellenmacarthurfoundation.org/towards-the-circular-economy-vol-1-an-economicand-business-rationale-for-an</u>
- Ellen Macarthur Foundation (2015). Growth Within: A Circular Economy Vision for a Competitive Europe. Available at: <u>https:// emf.thirdlight.com/link/8izw1qhml4ga-404tsz/@/preview/1?o</u>
- Emodi, N. V., Chaiechi, T., and Beg, A. B. M. R. A. (2019). The impact of climate variability and change on the energy system: A systematic scoping review. *Science of The Total Environment* 676, 545–563. <u>https://doi.org/10.1016/j.scitotenv.2019.04.294</u>
- Energieinstitut der Wirtschaft GmbH (2012). Energiekennzahlen in Dienstleistungsgebäuden. Energieinstitut der Wirtschaft GmbH. Available at: <u>https://www.energieinstitut.net/de/system/files/0903\_final\_dienstleistungsgebaude\_20120530.pdf</u> (Accessed June 7, 2024).
- ETUC (2021). ETUC position: A Just Transition Legal Framework to complement the Fit for 55 package. European Trade Union Confederation. Available at: <u>https://www.etuc.org/sites/default/ files/document/file/2021-12/ETUC%20position%20for%20</u> <u>a%20Just%20Transition%20Legal%20Framework%20to%20</u> <u>Complement%20the%20Fit%20for%2055%20Package\_1.pdf</u>
- European Climate and Health Observatory (2021). Ground-level ozone effects on human health under the changing climate. Available at: https://climate-adapt.eea.europa.eu/en/observato-

ry/evidence/health-effects/ground-level-ozone/ground-levelozone (Accessed June 7, 2024).

- European Commission (2024). Guidance document 1: CO2 storage life cycle and risk management framework. Brussels: Publikations Office of the European Union. Available at: <u>https://</u> <u>climate.ec.europa.eu/document/download/951d14ea-ce0f-4753-92dd-35ba88920888\_en?filename=ccs-implementation\_</u> <u>gd1\_en\_0.pdf</u>
- European Commission: Directorate-General for Employment, Social Affairs and Inclusion (2022). Fairness perceptions of the green transition. Brussels: Publications Office of the European Union. Available at: <u>https://data.europa.eu/doi/10.2767/651172</u>
- European Commission: Directorate-General for Energy (2020). *EU energy in figures: Statistical pocketbook 2020*. Publications Office of the European Union. Available at: <u>https://data.europa.</u> eu/doi/10.2833/29877 (Accessed August 21, 2023).
- Eurostat (2023). Circular economy material flows. Available at: <u>https://ec.europa.eu/eurostat/statistics-explained/index.php?ti-tle=Circular\_economy\_-material\_flows</u> (Accessed May 7, 2024).
- Eurostat, VCÖ (2019). In Österreich ist Autoverkehr seit 2010 doppelt so stark gestiegen wie im EU-Schnitt. VCÖ – Mobilität mit Zukunft. Available at: <u>https://vcoe.at/presse/presseaussendungen/detail/in-oesterreich-ist-autoverkehr-seit-2010-doppeltso-stark-gestiegen-wie-im-eu-schnitt</u> (Accessed September 12, 2023).
- Falk, M. (2013). Impact of Long-Term Weather on Domestic and Foreign Winter Tourism Demand. *International Journal of Tourism Research* 15, 1–17. <u>https://doi.org/10.1002/jtr.865</u>
- Famira-Mühlberger, U., and Firgo, M. (2019). Zum künftigen Bedarf an Pflegepersonal in den stationären und mobilen Diensten. WIFO Monatsberichte (monthly reports) 92, 149–157.
- Fanning, A. L., O'Neill, D. W., and Büchs, M. (2020). Provisioning systems for a good life within planetary boundaries. *Global Environmental Change* 64, 102135. <u>https://doi.org/10.1016/j.gloenvcha.2020.102135</u>
- Federal Statistics Office (2023). Nutzung erneuerbarer Energien im Tourismus und der Gesamtwirtschaft. Available at: <u>https://www. bfs.admin.ch/bfs/en/home/statistics/tourism/monetary-aspects/sustainability-indicators.assetdetail.26625787.html</u>
- Feigl, G., and Wukovitsch, F. eds. (2018). AK Wohlstandsbericht 2018. Wien: AK Wien. Available at: <u>https://emedien.arbeiterkammer.at/viewer/resolver?urn=urn:nbn:at:at-akw:g-2251600</u>
- Figerl, J., Tamesberger, D., and Theurl, S. (2021). Umverteilung von Arbeit(-szeit). Eine (Netto)Kostenschätzung für ein staatlich gefördertes Arbeitszeitverkürzungsmodell. *Momentum Quarterly – Zeitschrift für sozialen Fortschritt* 10, 1–65. <u>https://doi.org/10.15203/momentumquarterly.vol10.no1.p3-19</u>
- Fine, B. (1994). Towards a political economy of food. Review of International Political Economy 1, 519–545. <u>https://doi.org/10. 1080/09692299408434297</u>
- Finger, D., Heinrich, G., Gobiet, A., and Bauder, A. (2012). Projections of future water resources and their uncertainty in a glacierized catchment in the Swiss Alps and the subsequent effects on hydropower production during the 21<sup>st</sup> century. *Water Resources Research* 48, 2011WR010733. <u>https://doi.org/10.1029/2011WR010733</u>
- Finn, O., and Brockway, P. E. (2023). Much broader than health: Surveying the diverse co-benefits of energy demand reduction in Europe. *Energy Research & Social Science* 95, 102890. <u>https:// doi.org/10.1016/j.erss.2022.102890</u>

- Fischer-Kowalski, M., Krausmann, F., Giljum, S., Lutter, S., Mayer, A., Bringezu, S., et al. (2011). Methodology and Indicators of Economy-wide Material Flow Accounting: State of the Art and Reliability Across Sources. *Journal of Industrial Ecology* 15, 855–876. https://doi.org/10.1111/j.1530-9290.2011.00366.x
- Fisch-Romito, V., Guivarch, C., Creutzig, F., Minx, J. C., and Callaghan, M. W. (2021). Systematic map of the literature on carbon lock-in induced by long-lived capital. *Environ. Res. Lett.* 16, 053004. <u>https://doi.org/10.1088/1748-9326/aba660</u>
- Formayer, H., Maier, P., Nadeem, I., Leidinger, D., Lehner, F., Schöniger, F., et al. (2023). SECURES-Met – A European wide meteorological data set suitable for electricity modelling (supply and demand) for historical climate and climate change projections. <u>https://doi.org/10.5281/ZENODO.7907883</u>
- François, H., Samacoïts, R., Bird, D. N., Köberl, J., Prettenthaler, F., and Morin, S. (2023a). Climate change exacerbates snow-water-energy challenges for European ski tourism. *Nature Climate Change* 13, 935–942. <u>https://doi.org/10.1038/s41558-023-</u> 01759-5
- François, M., Mertens de Wilmars, S., and Maréchal, K. (2023b). Unlocking the potential of income and wealth caps in postgrowth transformation: A framework for improving policy design. *Ecological Economics* 208, 107788. <u>https://doi.org/10.1016/ j.ecolecon.2023.107788</u>
- Frayne, D. (2016). Stepping outside the circle: the ecological promise of shorter working hours. *Green Letters* 20, 197–212. <u>https:// doi.org/10.1080/14688417.2016.1160793</u>
- Fresner, J., Morea, F., Krenn, C., Aranda Uson, J., and Tomasi, F. (2017). Energy efficiency in small and medium enterprises: Lessons learned from 280 energy audits across Europe. *Journal of Cleaner Production* 142, 1650–1660. <u>https://doi.org/10.1016/j.jclepro.2016.11.126</u>
- Friedl, C., and Reichl, J. (2016). Realizing energy infrastructure projects – A qualitative empirical analysis of local practices to address social acceptance. *Energy Policy* 89, 184–193. <u>https:// doi.org/10.1016/j.enpol.2015.11.027</u>
- Friesenbichler, K. S., and Kügler, A. (2022). Servitization across countries and sectors: Evidence from world input-output data. *Economic Systems* 46, 101014. <u>https://doi.org/10.1016/j.ecosys.</u> 2022.101014
- Fritz, M., and Eversberg, D. (2024). Mentalities, classes and the four lines of conflict in the social-ecological transformation. *Eur Polit Sci* 23, 39–55. https://doi.org/10.1057/s41304-023-00457-2
- Fritz, O., Burton, A., Ehn-Fragner, S., Streicher, G., Laimer, P., and Orsolic, I. (2021). Bericht über die Bedeutung, Entwicklung und Struktur von Tourismus und Freizeitwirtschaft in Österreich im Jahr 2020. Österreichisches Institut für Wirtschaftsforschung, Statistik Austria. Available at: <u>https://www.wifo.ac.at/ jart/prj3/wifo/resources/person\_dokument/person\_dokument.</u> <u>jart?publikationsid=67060&mime\_type=application/pdf#:~:text=Der%20Marktanteil%20%C3%96sterreichs%20an%20 den,2020%20sch%C3%A4tzungsweise%2035%2C5%20Mrd</u>
- Fröhlich, T., Bloemer, S., Münter, D., and Brischke, L.-A. (2019). CO2-Quellen für die PtX-Herstellung in Deutschland – Technologien, Umweltwirkung, Verfügbarkeit. ifeu paper 03/2019.
- Fuchs, D., Lorek, S., Mamut, P., and Rossmoeller, A. (2023). "Fifty Shades of Sufficiency: Semantic Confusion and No Policy," in *Routledge Handbook of Environmental Policy*, eds. H. Jörgens, C. Knill, and Y. Steinebach (London: Routledge).
- Fuchs, M., Zeller, E., Joeppen, A., Weilguni, H., and Kling, H. (2013). "Das Wasserkraftpotenzial Österreichs im Spannungs-

feld von Umweltpolitik und Klimawandel," in *Wasserkraftprojekte*, ed. S. Heimerl (Wiesbaden: Springer Vieweg), 98–105. https://doi.org/10.1007/978-3-658-00996-0\_13

- Gabelberger, F., Kettner-Marx, C., Peneder, M., and Streicher, G. (2020). Landkarte der "(De-)Karbonisierung" für den produzierenden Bereich in Österreich. Vienna: Austrian Institute of Economic Research (WIFO). Available at: <u>https://emedien.arbeiterkammer.at/viewer/api/v1/records/AC16063422/files/source/ AC16063422.pdf</u> (Accessed September 13, 2023).
- Gailani, A., Cooper, S., Allen, S., Pimm, A., Taylor, P., and Gross, R. (2024). Assessing the potential of decarbonization options for industrial sectors. *Joule* 8, 576–603. <u>https://doi.org/10.1016/j.</u> joule.2024.01.007
- Gardarsdottir, S. O., De Lena, E., Romano, M., Roussanaly, S., Voldsund, M., Pérez-Calvo, J.-F., et al. (2019). Comparison of Technologies for CO<sub>2</sub> Capture from Cement Production—Part 2: Cost Analysis. *Energies* 12, 542. <u>https://doi.org/10.3390/ en12030542</u>
- Geels, F. W. (2023). Demand-side emission reduction through behavior change or technology adoption? Empirical evidence from UK heating, mobility, and electricity use. One Earth 6, 337–340. https://doi.org/10.1016/j.oneear.2023.03.012
- Geels, F. W., and Ayoub, M. (2023). A socio-technical transition perspective on positive tipping points in climate change mitigation: Analysing seven interacting feedback loops in offshore wind and electric vehicles acceleration. *Technological Forecasting and Social Change* 193, 122639. <u>https://doi.org/10.1016/j. techfore.2023.122639</u>
- Geels, F. W., and Johnson, V. (2018). Towards a modular and temporal understanding of system diffusion: Adoption models and socio-technical theories applied to Austrian biomass district-heating (1979–2013). *Energy Research & Social Science* 38, 138–153. https://doi.org/10.1016/j.erss.2018.02.010
- Geissdoerfer, M., Savaget, P., Bocken, N. M. P., and Hultink, E. J. (2017). The Circular Economy – A new sustainability paradigm? *Journal of Cleaner Production* 143, 757–768. <u>https://doi. org/10.1016/j.jclepro.2016.12.048</u>
- Geng, Y., Sarkis, J., and Bleischwitz, R. (2019). How to globalize the circular economy. *Nature* 565, 153–155. <u>https://doi.org/10.1038/</u> <u>d41586-019-00017-z</u>
- Gerold, S., Hoffmann, M., and Aigner, E. (2023). Towards a critical understanding of work in ecological economics: A postwork perspective. *Ecological Economics* 212, 107935. <u>https://doi. org/10.1016/j.ecolecon.2023.107935</u>
- Gerold, S., and Nocker, M. (2015). Reduction of Working Time in Austria. A Mixed Methods Study Relating a New Work Time Policy to Employee Preferences. WWWforEurope Working Paper, No. 97. Wien: WWWforEurope: Welfare, Wealth and Work for Europe.
- Gerold, S., and Nocker, M. (2018). More Leisure or Higher Pay? A Mixed-methods Study on Reducing Working Time in Austria. *Ecological Economics* 143, 27–36. <u>https://doi.org/10.1016/j.ecolecon.2017.06.016</u>
- Gerold, S., Soder, M., and Schwendinger, M. (2017). Arbeitszeitverkürzung in der Praxis. Innovative Modelle in österreichischen Betrieben. Wirtschaft und Gesellschaft 43, 177–204.
- Geyer, R., Knöttner, S., Diendorfer, C., Drexler-Schmid, G., and Alton, V. (2021). 100% Renewable Energy for Austria's Industry: Scenarios, Energy Carriers and Infrastructure Requirements. *Applied Sciences* 11, 1819. <u>https://doi.org/10.3390/app11041819</u>

- Giampietro, M. (2022). "The entropic nature of the economic process: A scientific explanation of the blunder of circular economy," in *The Impossibilities of the Circular Economy. Separating Aspirations from Reality*, eds. H. Lehmann, C. Hinske, V. de Margerie, and A. Slaveikova Nikolova (London: Routledge).
- Giljum, S., Wieland, H., Lutter, S., Bruckner, M., Wood, R., Tukker, A., et al. (2016). Identifying priority areas for European resource policies: a MRIO-based material footprint assessment. *Journal of Economic Structures* 5, 17. <u>https://doi.org/10.1186/ s40008-016-0048-5</u>
- Ginkel, K. C. H. van, Botzen, W. J. W., Haasnoot, M., Bachner, G., Steininger, K. W., Hinkel, J., et al. (2020). Climate change induced socio-economic tipping points: review and stakeholder consultation for policy relevant research. *Environ. Res. Lett.* 15, 023001. <u>https://doi.org/10.1088/1748-9326/ab6395</u>
- Global 2000, Greenpeace, and WWF Österreich (2017). Betreff: Appell der Wirtschaft für Energiewende und Klimaschutz. Available at: <u>https://www.global2000.at/sites/global/files/Ap-pell\_Brief.pdf</u>
- Global CCS Institute (2021). Global Status of CCS 2020. Australia. Available at: <u>https://www.globalccsinstitute.com/wp-content/uploads/2021/03/Global-Status-of-CCS-Report-English.pdf</u> (Accessed August 21, 2023).
- Global CCS Institute (2023). Global Status of CCS 2022. Available at: <u>https://status22.globalccsinstitute.com/wp-content/uploads/2023/03/GCCSI\_Global-Report-2022\_PDF\_FINAL-01-03-23.pdf</u>
- Goers, S., Schneider, F., Steinmüller, H., and Tichler, R. (2020). Wirtschaftswachstum und Beschäftigung durch Investitionen in Erneuerbare Energien. Volkswirtschaftliche Effekte durch Investitionen in ausgewählte Produktions- und Speichertechnologien. Linz: Energieinstitut an der Johannes Kepler Universität Linz. Available at: <u>https://energieinstitut-linz.at/wp-content/</u> <u>uploads/2020/10/Energieinstitut-VWL-Effekte-durch-Investitionen-in-EE-Langfassung.pdf</u>
- Gössling, S., Balas, M., Mayer, M., and Sun, Y.-Y. (2023). A review of tourism and climate change mitigation: The scales, scopes, stakeholders and strategies of carbon management. *Tourism Management* 95, 104681. <u>https://doi.org/10.1016/j.tourman.2022.104681</u>
- Gössling, S., and Higham, J. (2021). The Low-Carbon Imperative: Destination Management under Urgent Climate Change. Journal of Travel Research 60, 1167–1179. <u>https://doi.org/10.1177/</u> 0047287520933679
- Gottschlich, D., and Katz, C. (2023). Beziehungsarbeit ist angesagt: Care-Prinzipien und sozial-ökologische Transformation. *politische ökologie* 172, 91–96.
- Gough, I. (2022). Two Scenarios for Sustainable Welfare: A Framework for an Eco-Social Contract. *Social Policy and Society* 21, 460–472. https://doi.org/10.1017/S1474746421000701
- Greene, M. (2018). Socio-technical transitions and dynamics in everyday consumption practice. *Global Environmental Change* 52, 1–9. https://doi.org/10.1016/j.gloenvcha.2018.05.007
- Greiml, M., Fritz, F., and Kienberger, T. (2021). Increasing installable photovoltaic power by implementing power-to-gas as electricity grid relief – A techno-economic assessment. *Energy* 235, 121307. https://doi.org/10.1016/j.energy.2021.121307
- Großer, E., Jorck, G. von, Kludas, S., Mundt, I., and Sharp, H. (2020). Sozial-ökologische Infrastrukturen – Rahmenbedingungen für Zeitwohlstand und neue Formen von Arbeit. Öko-

logisches Wirtschaften – Fachzeitschrift 33, 14–16. https://doi. org/10.14512/OEW350414

- Gruber, K. (2024). Bedeutung der Berücksichtigung unterschiedlicher Wetterjahre und Klimamodelle in der Energiesystemmodellierung. Available at: <u>https://www.tugraz.at/fileadmin/ user\_upload/tugrazExternal/f560810f-089d-42d8-ae6d-8e82a-8454ca9/files/lf/Session\_A1/111\_LF\_Gruber.pdf</u> (Accessed May 6, 2024).
- Grubler, A., Wilson, C., Bento, N., Boza-Kiss, B., Krey, V., McCollum, D. L., et al. (2018). A low energy demand scenario for meeting the 1.5 °C target and sustainable development goals without negative emission technologies. *Nature Energy* 3, 515– 527. <u>https://doi.org/10.1038/s41560-018-0172-6</u>
- Gütschow, J., Busch, D., and Pflüger, M. (2024). The PRIMAP-hist national historical emissions time series (1750–2023) v2.6. https://doi.org/10.5281/ZENODO.13752654
- Gütschow, J., Jeffery, M. L., Gieseke, R., Gebel, R., Stevens, D., Krapp, M., et al. (2016). The PRIMAP-hist national historical emissions time series. *Earth Syst. Sci. Data* 8, 571–603. <u>https:// doi.org/10.5194/essd-8-571-2016</u>
- Haas, R., Auer, H., and Resch, G. (2022). Heading towards democratic and sustainable electricity systems – the example of Austria. *Renew. Energy Environ. Sustain.* 7, 20. <u>https://doi.org/10.1051/rees/2022009</u>
- Haas, W., Baumgart, A., Eisenmenger, N., Virág, D., Kalt, G., Sommer, M., et al. (2025). How decarbonization and the circular economy interact: Benefits and trade-offs in the case of the buildings, transport, and electricity sectors in Austria. *J of Industrial Ecology*, 1–15. <u>https://doi.org/10.1111/jiec.13619</u>
- Haas, W., Krausmann, F., Wiedenhofer, D., and Heinz, M. (2015). How circular is the global economy? An assessment of material flows, waste production and recycling in the EU and the world in 2005. *Journal of Industrial Ecology* 19, 765–777. <u>https://doi. org/10.1111/jiec.12244</u>
- Haas, W., Krausmann, F., Wiedenhofer, D., Lauk, C., and Mayer, A. (2020). Spaceship earth's odyssey to a circular economy – a century long perspective. *Resources, Conservation and Recycling* 163, 105076. <u>https://doi.org/10.1016/j.resconrec.2020.105076</u>
- Haas, W., Muhar, A., Dorninger, C., and Gugerell, K. (2023). "Kapitel 23. Synthese: Pfade zur Transformation struktureller Bedingungen für ein klimafreundliches Leben," in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 613–647. Available at: https://doi.org/10.1007/978-3-662-66497-1\_27
- Haberl, H. (2018). The stock-flow-service nexus: implications for socio-ecological transformations toward sustainability and circularity. Available at: <u>https://doi.org/10.5281/zenodo.4147188</u> (Accessed October 29, 2024).
- Haberl, H., Wiedenhofer, D., Schug, F., Frantz, D., Virág, D., Plutzar, C., et al. (2021). High-Resolution Maps of Material Stocks in Buildings and Infrastructures in Austria and Germany. *Environ. Sci. Technol.* 55, 3368–3379. <u>https://doi.org/10.1021/acs.est.0c05642</u>
- Haberl, H., Wiedenhofer, D., Virág, D., Kalt, G., Plank, B., Brockway, P., et al. (2020). A systematic review of the evidence on decoupling of GDP, resource use and GHG emissions, part II: synthesizing the insights. *Environ. Res. Lett.* 15, 065003. <u>https:// doi.org/10.1088/1748-9326/ab842a</u>
- Hanaei, S., Takian, A., Majdzadeh, R., Maboloc, C. R., Grossmann, I., Gomes, O., et al. (2022). Emerging Standards and the Hy-

brid Model for Organizing Scientific Events During and After the COVID-19 Pandemic. *Disaster Med. Public Health Prep.* 16, 1172–1177. <u>https://doi.org/10.1017/dmp.2020.406</u>

- Hanbury, H., Bader, C., and Moser, S. (2019). Reducing Working Hours as a Means to Foster Low(er)-Carbon Lifestyles? An Exploratory Study on Swiss Employees. *Sustainability* 11, 2024. https://doi.org/10.3390/su11072024
- Hanbury, H., Illien, P., Ming, E., Moser, S., Bader, C., and Neubert, S. (2023). Working less for more? A systematic review of the social, economic, and ecological effects of working time reduction policies in the global North. Sustainability: Science, Practice and Policy 19, 2222595. <u>https://doi.org/10.1080/15487733.2023</u> .2222595
- Hanger-Kopp, S., and Palka, M. (2022). Decision spaces in agricultural risk management: a mental model study of Austrian crop farmers. *Environ Dev Sustain* 24, 6072–6098. <u>https://doi. org/10.1007/s10668-021-01693-6</u>
- Hardt, L., Barrett, J., Taylor, P. G., and Foxon, T. J. (2020). Structural Change for a Post-Growth Economy: Investigating the Relationship between Embodied Energy Intensity and Labour Productivity. *Sustainability* 12, 962. <u>https://doi.org/10.3390/ su12030962</u>
- Hardt, L., Barrett, J., Taylor, P. G., and Foxon, T. J. (2021). What structural change is needed for a post-growth economy: A framework of analysis and empirical evidence. *Ecological Economics* 179, 106845. <u>https://doi.org/10.1016/j.ecolecon.2020?</u> <u>106845</u>
- Hartard, S., Schaffer, A., and Stahmer, C. eds. (2006). *Die Halbtagsgesellschaft. Konkrete Utopie für eine zukunftsfähige Gesellschaft.* Baden-Baden: Nomos Verlag.
- Hartwig, L., Hössinger, R., Susilo, Y. O., and Gühnemann, A. (2022). The Impacts of a COVID-19 Related Lockdown (and Reopening Phases) on Time Use and Mobility for Activities in Austria—Results from a Multi-Wave Combined Survey. Sustainability 14, 7422. https://doi.org/10.3390/su14127422
- Heinemann, Z., and Scharf, C. (2004). "CO Sinks in Oil/Gas Fields & Aquifers: Technologies, Challenges & Potentials," in CO2 Capture and Sequestration In Future International R&D Programmes, ed.Bundesministerium für Verkehr, Innovation und Technologie (Vienna: Bundesministerium für Verkehr, Innovation und Technologie), 61–72. Available at: <u>https://nachhaltigwirtschaften.at/resources/nw\_pdf/0424\_ccs.pdf</u>
- Heinfellner, H., Hartl, A., Schwendinger, M., Nocker, M., Wickenbrock, J., Hubin, C., et al. (2024). Praxishandbuch Nachhaltiges betriebliches Mobilitätsmanagement: So gelingt die Transformation zu einer nachhaltigen, innovativen und inklusiven Mobilität in Unternehmen., 1. Auflage 2024, eds. C. Falkinger and A. Lindinger. Wien: Linde Verlag Ges.m.b.H.
- Heitkötter, M., Jurczyk, K., and Lange, A. eds. (2009). Zeit für Beziehungen? Zeit und Zeitpolitik für Familien. Opladen: B. Budrich.
- Helfricht, K., Huss, M., Fischer, A., and Otto, J.-C. (2019). Calibrated Ice Thickness Estimate for All Glaciers in Austria. *Frontiers* in Earth Science 7, 68. <u>https://doi.org/10.3389/feart.2019.00068</u>
- Hepburn, C., Adlen, E., Beddington, J., Carter, E. A., Fuss, S., Mac Dowell, N., et al. (2019). The technological and economic prospects for CO2 utilization and removal. *Nature* 575, 87–97. <u>https://doi.org/10.1038/s41586-019-1681-6</u>
- Heyen, D. A. (2023). Social justice in the context of climate policy: systematizing the variety of inequality dimensions, social impacts, and justice principles. *Climate Policy* 23, 539–554. <u>https://</u> doi.org/10.1080/14693062.2022.2142499

- Hidasi, K., Venczel, T., and Antal, M. (2023). Working time reduction: Employers' perspectives and eco-social implications – ten cases from Hungary. *European Journal of Social Security* 25, 426–444. <u>https://doi.org/10.1177/13882627231214547</u>
- Hirvilammi, T., Häikiö, L., Johansson, H., Koch, M., and Perkiö, J. (2023). Social Policy in a Climate Emergency Context: Towards an Ecosocial Research Agenda. *Journal of Social Policy* 52, 471. <u>https://doi.org/10.1017/S0047279423000053</u>
- Hochmeister, S., Kühberger, L., Kulich, J., Ott, H., and Kienberger, T. (2024a). A methodology for the determination of future Carbon Management Strategies: A case study of Austria. *IJSEPM* 41, 108–124. <u>https://doi.org/10.54337/ijsepm.8280</u>
- Hochmeister, S., Kühberger, L., Kulich, J., Ott, H., and Kienberger, T. (2024b). Carbon Management für ein klimaneutrales Österreich. *Elektrotech. Inftech.* 141, 299–306. <u>https://doi.org/ 10.1007/s00502-024-01235-8</u>
- Hofbauer, J., Gerold, S., Klaus, D., and Wukovitsch, F. (2023). "Kapitel 7. Erwerbsarbeit," in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer, Spektrum), 285–307. <u>https://doi.org/10.1007/978-3-662-66497-1\_11</u>
- Hoffmann, M., Pantazidou, M., and Smith, T. (2023). Critiques of work: The radical roots of degrowth. <u>https://doi.org/10.31235/ osf.io/m9q2s</u>
- Hoffmann, M., and Paulsen, R. (2020). Resolving the 'jobs-environment-dilemma'? The case for critiques of work in sustainability research. *Environmental Sociology* 6, 343–354. <u>https:// doi.org/10.1080/23251042.2020.1790718</u>
- Hoffmann, M., and Spash, C. L. (2021). The impacts of climate change mitigation on work for the Austrian economy. Institute for Multilevel Governance and Development. SRE – Discussion Papers No. 10/2021. Vienna: WU Vienna University of Economics and Business. Available at: <u>https://research.wu.ac.at/ws/ files/19015690/sre-disc-2021\_10.pdf</u> (Accessed September 15, 2023).
- Hofmeister, S., and Mölders, T. eds. (2021). Für Natur sorgen? Dilemmata feministischer Positionierungen zwischen Sorge- und Herrschaftsverhältnissen. Opladen Berlin Toronto: Verlag Barbara Budrich. <u>https://doi.org/10.3224/84742424</u>
- Holmberg, K., and Erdemir, A. (2019). The impact of tribology on energy use and CO2 emission globally and in combustion engine and electric cars. *Tribology International* 135, 389–396. <u>https://doi.org/10.1016/j.triboint.2019.03.024</u>
- Holmin, J., Levison, E., and Oehme, S. (2015). The utilization of office spaces and its impact on energy use.
- Holzmann, A., Adensam, H., Kratena, K., and Schmid, E. (2013). Decomposing final energy use for heating in the residential sector in Austria. *Energy Policy* 62, 607–616. <u>https://doi.org/10.1016/j.enpol.2013.07.027</u>
- Hossain, E., Roy, S., Mohammad, N., Nawar, N., and Dipta, D. R. (2021). Metrics and enhancement strategies for grid resilience and reliability during natural disasters. *Applied Energy* 290, 116709. https://doi.org/10.1016/j.apenergy.2021.116709
- Hrovatin, N., Cagno, E., Dolšak, J., and Zorić, J. (2021). How important are perceived barriers and drivers versus other contextual factors for the adoption of energy efficiency measures: An empirical investigation in manufacturing SMEs. *Journal of Cleaner Production* 323, 129123. <u>https://doi.org/10.1016/j.jclepro.2021.129123</u>

- IEA (2020). Energy Technology Perspectives 2020. International Energy Agency (IEA). Available at: <u>https://www.iea.org/reports/energy-technology-perspectives-2020#overview</u> (Accessed November 7, 2024).
- IEA (2022). World Energy Outlook 2022. International Energy Agency (IEA). Available at: <u>https://iea.blob.core.windows.net/</u> <u>assets/830fe099-5530-48f2-a7c1-11f35d510983/WorldEner-</u> <u>gyOutlook2022.pdf</u> (Accessed September 14, 2023).
- IEA (2023). Tracking Clean Energy Progress 2023. IEA. Available at: <u>https://www.iea.org/reports/tracking-clean-energy-progress-2023</u> (Accessed September 14, 2023).
- ILO (2015). Guidelines for a just transition towards environmentally sustainable economies and societies for all. Switzerland: ILO International Labour Organization.
- Initiative Wege aus der Krise (2019). Just Transition Klimaschutz demokratisch gestalten! Vienna: Initiative Wege aus der Krise. Available at: <u>https://www.wege-aus-der-krise.at/images/Just\_Transition\_final.pdf</u> (Accessed September 13, 2023).
- IPCC (2005). IPCC Special Report on Carbon Dioxide Capture and Storage. Prepared by Working Group III of the Intergovernmental Panel on Climate Change., eds. B. Metz, O. Davidson, H. de Coninck, M. Loos, and L. Meyer. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- IPCC (2023a). Climate Change 2022 Mitigation of Climate Change: Working Group III Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change., 1<sup>st</sup> Edn, eds.
   P. R. Shukla, J. Skea, R. Slade, A. A. Khourdajie, A. Hasija, J. Malley, et al. Cambridge, UK and New York, NY, USA: Cambridge University Press. <u>https://doi.org/10.1017/9781009157926</u>
- IPCC (2023b). Climate Change 2023: Synthesis Report. Contribution of Working Groups I, II and III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change., eds. Core Writing Team, H. Lee, and J. Romero. Geneva, Switzerland: IPCC. Available at: <u>https://doi.org/10.59327/IPCC/AR6-9789291691647</u> (Accessed September 29, 2023).
- IRENA, and ILO (2021). Renewable Energy and Jobs Annual Review 2021. Abu Dhabi, Genf: International Renewable Energy Agency, International Labour Organization. Available at: <u>https://www.irena.org/publications/2021/Oct/Renewable-En-</u> ergy-and-Jobs-Annual-Review-2021
- Jacobi, N., Haas, W., Wiedenhofer, D., and Mayer, A. (2018). Providing an economy-wide monitoring framework for the circular economy in Austria: Status quo and challenges. *Resources, Conservation and Recycling* 137, 156–166. <u>https://doi.org/10.1016/j.</u> <u>resconrec.2018.05.022</u>
- Jalo, N., Johansson, I., Kanchiralla, F. M., and Thollander, P. (2021). Do energy efficiency networks help reduce barriers to energy efficiency? -A case study of a regional Swedish policy program for industrial SMEs. *Renewable and Sustainable Energy Reviews* 151, 111579. <u>https://doi.org/10.1016/j.rser.2021.111579</u>
- Japanese Ministry of the Environment (2018). Circulating and Ecological Economy. Available at: <u>https://www.env.go.jp/content/900453287.pdf</u> (Accessed May 7, 2024).
- Jerez, S., Trigo, R. M., Sarsa, A., Lorente-Plazas, R., Pozo-Vázquez, D., and Montávez, J. P. (2013). Spatio-temporal Complementarity between Solar and Wind Power in the Iberian Peninsula. *Energy Procedia* 40, 48–57. <u>https://doi.org/10.1016/j.egypro.2013.08.007</u>
- Jo, T.-H. (2011). Social Provisioning Process and Socio-Economic Modeling. *The American Journal of Economics and Sociology* 70, 1094–1116. <u>https://doi.org/10.1111/j.1536-7150.2011.00808.x</u>

- Jo, T.-H., and Todorova, Z. (2019). "Social Provisioning Process: A heterodox view of the economy," in *The Routledge Handbook of Heterodox Economics: Theorizing, Analyzing, and Transforming Capitalism*, eds. T.-H. Jo, L. Chester, C. D'Ippoliti, T.-H. Jo, L. Chester, and C. D'Ippoliti (London: Routledge), 29–40.
- Jobin, M., and Siegrist, M. (2020). Support for the Deployment of Climate Engineering: A Comparison of Ten Different Technologies. *Risk Analysis* 40, 1058–1078. <u>https://doi.org/10.1111/ risa.13462</u>
- Johansson, I., Johnsson, S., and Thollander, P. (2022). Impact evaluation of an energy efficiency network policy programme for industrial SMEs in Sweden. *Resources, Environment and Sustainability* 9, 100065. <u>https://doi.org/10.1016/j.resenv.2022.100065</u>
- Johansson, I., Mardan, N., Cornelis, E., Kimura, O., and Thollander, P. (2019). Designing Policies and Programmes for Improved Energy Efficiency in Industrial SMEs. *Energies* 12, 1338. <u>https:// doi.org/10.3390/en12071338</u>
- Kagermann, H. (2015). "Change Through Digitization—Value Creation in the Age of Industry 4.0," in *Management of Permanent Change*, eds. H. Albach, H. Meffert, A. Pinkwart, and R. Reichwald (Wiesbaden: Springer Gabler), 23–45. <u>https://doi.org/10.1007/978-3-658-05014-6\_2</u>
- Kalt, G., Wiedenhofer, D., Görg, C., and Haberl, H. (2019). Conceptualizing energy services: A review of energy and well-being along the Energy Service Cascade. *Energy Research & Social Science* 53, 47–58. <u>https://doi.org/10.1016/j.erss.2019.02.026</u>
- Kapeller, S., and Biegelbauer, P. (2020). How (not) to solve local conflicts around alternative energy production: Six cases of siting decisions of Austrian wind power parks. *Utilities Policy* 65, 101062. <u>https://doi.org/10.1016/j.jup.2020.101062</u>
- Kapica, J., Jurasz, J., Canales, F. A., Bloomfield, H., Guezgouz, M., De Felice, M., et al. (2024). The potential impact of climate change on European renewable energy droughts. *Renewable and Sustainable Energy Reviews* 189, 114011. <u>https://doi. org/10.1016/j.rser.2023.114011</u>
- Karkatsoulis, P., Capros, P., Fragkos, P., Paroussos, L., and Tsani, S. (2016). First-mover advantages of the European Union's climate change mitigation strategy. *Int. J. Energy Res.* 40, 814–830. <u>https://doi.org/10.1002/er.3487</u>
- Karvounis, P., and Blunt, M. J. (2021). Assessment of CO2 geological storage capacity of saline aquifers under the North Sea. *International Journal of Greenhouse Gas Control* 111, 103463. <u>https://doi.org/10.1016/j.ijggc.2021.103463</u>
- Kearns, D., Liu, H., and Consoli, C. (2021). Technology Readiness and Costs of CCS. Global CCS Institute. Available at: <u>https:// www.globalccsinstitute.com/wp-content/uploads/2022/03/</u> <u>CCE-CCS-Technology-Readiness-and-Costs-22-1.pdf</u> (Accessed August 21, 2023).
- Kennedy, S., and Linnenluecke, M. K. (2022). Circular economy and resilience: A research agenda. *Business Strategy and the En*vironment 31, 2754–2765. <u>https://doi.org/10.1002/bse.3004</u>
- Kikstra, J. S., Mastrucci, A., Min, J., Riahi, K., and Rao, N. D. (2021). Decent living gaps and energy needs around the world. *Environ. Res. Lett.* 16, 095006. <u>https://doi.org/10.1088/1748-9326/ac1c27</u>
- Kirchherr, J., Reike, D., and Hekkert, M. (2017). Conceptualizing the circular economy: An analysis of 114 definitions. *Re*sources, Conservation and Recycling 127, 221–232. <u>https://doi.org/10.1016/j.resconrec.2017.09.005</u>
- Kirchherr, J., Yang, N.-H. N., Schulze-Spüntrup, F., Heerink, M. J., and Hartley, K. (2023). Conceptualizing the Circular Economy (Revisited): An Analysis of 221 Definitions. *Resources, Con-*

servation and Recycling 194, 107001. https://doi.org/10.1016/j. resconrec.2023.107001

- Kitamura, Y., Ichisugi, Y., Karkour, S., and Itsubo, N. (2020). Carbon Footprint Evaluation Based on Tourist Consumption toward Sustainable Tourism in Japan. Sustainability 12, 2219. <u>https://doi.org/10.3390/su12062219</u>
- Kitzmann, R. (2023). Home swapping as instrument for more housing sufficiency! *International Journal of Housing Policy*, 1–18. <u>https://doi.org/10.1080/19491247.2023.2269619</u>
- Kjaer, L. L., Pigosso, D. C. A., Niero, M., Bech, N. M., and McAloone, T. C. (2019). Product/Service-Systems for a Circular Economy: The Route to Decoupling Economic Growth from Resource Consumption? *Journal of Industrial Ecology* 23, 22–35. <u>https://doi.org/10.1111/jiec.12747</u>
- Klankermayer, J., Wesselbaum, S., Beydoun, K., and Leitner, W. (2016). Selective Catalytic Synthesis Using the Combination of Carbon Dioxide and Hydrogen: Catalytic Chess at the Interface of Energy and Chemistry. *Angew Chem Int Ed Engl* 55, 7296– 7343. <u>https://doi.org/10.1002/anie.201507458</u>
- Klien, M., Sommer, M., and Weingärtler, M. (2023). naBe-Aktionsplan: Wirkungsanalyse der nachhaltigen öffentlichen Beschaffung in Österreich. Vienna: Österreichisches Institut für Wirtschaftsforschung. Available at: <u>https://www.wifo.ac.at/ publication/176069/</u>
- Klimenko, V. V., Ratner, S. V., and Tereshin, A. G. (2021). Constraints imposed by key-material resources on renewable energy development. *Renewable and Sustainable Energy Reviews* 144, 111011. <u>https://doi.org/10.1016/j.rser.2021.111011</u>
- Kling, H., Fuchs, M., and Paulin, M. (2012). Runoff conditions in the upper Danube basin under an ensemble of climate change scenarios. *Journal of Hydrology* 424–425, 264–277. <u>https://doi.org/10.1016/j.jhydrol.2012.01.011</u>
- Knight, K. W., Rosa, E. A., and Schor, J. B. (2013). Could working less reduce pressures on the environment? A cross-national panel analysis of OECD countries, 1970–2007. *Global Environmental Change* 23, 691–700. <u>https://doi.org/10.1016/j.gloenvcha.2013.02.017</u>
- Knittel, N., Bednar-Friedl, B., Jury, M. W., Lincke, D., and Hinkel, J. (2024). Beyond Borders: Quantifying climate risks in Austria's global trade relations with varying levels of adaptation.
- Knittel, N., Jury, M. W., Bednar-Friedl, B., Bachner, G., and Steiner, A. K. (2020). A global analysis of heat-related labour productivity losses under climate change—implications for Germany's foreign trade. *Climatic Change* 160, 251–269. <u>https://doi.org/10.1007/s10584-020-02661-1</u>
- Knobloch, U. ed. (2019). Ökonomie des Versorgens: feministisch-kritische Wirtschaftstheorien im deutschsprachigen Raum., 1. Auflage. Weinheim: Beltz Juventa.
- Koemle, D., Zinngrebe, Y., and Yu, X. (2018). Highway construction and wildlife populations: Evidence from Austria. *Land Use Policy* 73, 447–457. <u>https://doi.org/10.1016/j.landusepol.2018.02.021</u>
- Kondziella, H., and Bruckner, T. (2016). Flexibility requirements of renewable energy based electricity systems a review of research results and methodologies. *Renewable and Sustainable Energy Reviews* 53, 10–22. <u>https://doi.org/10.1016/j.rser.2015.07.199</u>
- Kopf, J. (2021). Die Extraportion Mut für den Klimaschutz. derStandard. Available at: <u>https://www.derstandard.at/story/</u>2000126748307/die-extraportion-mut-fuer-den-klimaschutz (Accessed October 27, 2021).

- Köppl, A., and Schratzenstaller, M. (2019). "Ein Abgabensystem, das (Erwerbs-)Arbeit fördert," in *Tätigsein in der Postwachstumsgesellschaft*, eds. I. Seidl and A. Zahrnt (Marburg: Metropolis), 207–225.
- Korhonen, J., Nuur, C., Feldmann, A., and Birkie, S. E. (2018). Circular economy as an essentially contested concept. *Journal* of Cleaner Production 175, 544–552. <u>https://doi.org/10.1016/j.jclepro.2017.12.111</u>
- Kranzl, L., Formayer, H., Hirner, R., Hummel, M., Müller, A., Schicker, I., et al. (2014a). The Impact of Climate Change and Energy Efficiency on Heating and Cooling Energy Demand and Load., ed.Institut für Elektrizitätswirtschaft und Energieinnovation (IEE) Technische Universität Graz (TUG) (Graz).
- Kranzl, L., Haas, R., Kalt, G., Müller, A., Nakicenovic, N., Redl, C., et al. (2010). Energie-der-Zukunft. KlimAdapt. Ableitung von prioritären Maßnahmen zur Adaption des Energiesystems an den Klimawandel. Report. Vienna: Energy Economics Group, TU Wien, BOKU Vienna.
- Kranzl, L., Müller, A., Maia, I., Büchele, R., and Hartner, M. (2018). Wärmezukunft 2050. Erfordernisse und Konsequenzen der Dekarbonisierung von Raumwärme und Warmwasserbereitstellung in Österreich Kurzfassung. Wien.
- Kranzl, L., Totschnig, G., Müller, A., Bachner, G., and Bednar-Friedl, B. (2015). "Electricity," in *Economic Evaluation of Climate Change Impacts*, eds. K. W. Steininger, M. König, B. Bednar-Friedl, L. Kranzl, W. Loibl, and F. Prettenthaler (Cham: Springer International Publishing), 257–278. <u>https://doi.org/10. 1007/978-3-319-12457-5\_14</u>
- Kranzl, L., Totschnig, G., Müller, A., Hummel, M., Loibl, W., Schicker, I., et al. (2014b). Auswirkungen des Klimawandels auf die Energie- und Stromversorgung in Österreich. CCCA Fact Sheet #10. Available at: <u>https://ccca.ac.at/fileadmin/00\_DokumenteHauptmenue/02\_Klimawissen/FactSheets/10\_energie\_ v5\_02112015.pdf</u>
- Krausmann, F., Lauk, C., Haas, W., and Wiedenhofer, D. (2018).
  From resource extraction to outflows of wastes and emissions: The socioeconomic metabolism of the global economy, 1900– 2015. *Global Environmental Change* 52, 131–140. <u>https://doi.org/10.1016/j.gloenvcha.2018.07.003</u>
- Kreinin, H., and Aigner, E. (2022). From "Decent work and economic growth" to "Sustainable work and economic degrowth": a new framework for SDG 8. *Empirica* 49, 281–311. <u>https://doi. org/10.1007/s10663-021-09526-5</u>
- Krevor, S., Blunt, M. J., Benson, S. M., Pentland, C. H., Reynolds, C., Al-Menhali, A., et al. (2015). Capillary trapping for geologic carbon dioxide storage – From pore scale physics to field scale implications. *International Journal of Greenhouse Gas Control* 40, 221–237. https://doi.org/10.1016/j.jiggc.2015.04.006
- Krisch, A., Novy, A., Plank, L., Schmidt, A. E., and Blaas, W. (2020). Die Leistungsträgerinnen des Alltagslebens. Covid-19 als Brennglas für die notwendige Neubewertung von Wirtschaft, Arbeit und Leistung. Wien. Available at: <u>http://hdl.handle.net/20.500.12708/40167</u> (Accessed January 17, 2025).
- Krutzler, T., Schindler, I., and Wasserbaur, R. (2023). Energie- und Treibhausgas-Szenario Transition 2040. Available at: <u>https:// www.umweltbundesamt.at/fileadmin/site/publikationen/ rep0880.pdf</u>
- Kubeczko, K., and Krisch, A. (2023). "Kapitel 22: Netzgebundene Infrastrukturen," in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg:

Springer), 591-609. <u>https://doi.org/10.1007/978-3-662-66497-</u> 1\_17

- Kubon-Gilke, G. (2019). "Soziale Sicherung in der Postwachstumsgesellschaft," in *Tätigsein in der Postwachstumsgesellschaft*, eds.
  I. Seidl and A. Zahrnt (Marburg: Metropolis), 193–206.
- Kuhl, M., Maier, F., and Mill, H. T. (2011). The gender dimensions of the green new deal—A study commissioned by the Greens/ EFA. Available at: <u>https://www.greens-efa.eu/en/article/document/the-gender-dimensions-of-the-green-new-deal</u> (Accessed December 20, 2021).
- Kuittinen, M., Zernicke, C., Slabik, S., and Hafner, A. (2023). How can carbon be stored in the built environment? A review of potential options. *Architectural Science Review* 66, 91–107. <u>https:// doi.org/10.1080/00038628.2021.1896471</u>
- Kulich, J., and Ott, H. (2025). CCS Capacity in Austria and its Competitive Usage of the Subsurface. SSRN Journal. <u>https://doi.org/10.2139/ssrn.5068885</u>
- Kulmer, V., and Seebauer, S. (2019). How robust are estimates of the rebound effect of energy efficiency improvements? A sensitivity analysis of consumer heterogeneity and elasticities. *Energy Policy* 132, 1–14. <u>https://doi.org/10.1016/j.enpol.2019.05.001</u>
- Kulmer, V., Seebauer, S., Hinterreither, H., Kortschak, D., Theurl, M. C., and Haas, W. (2022). Transforming the s-shape: Identifying and explaining turning points in market diffusion curves of low-carbon technologies in Austria. *Research Policy* 51, 104371. <u>https://doi.org/10.1016/j.respol.2021.104371</u>
- Lamb, W. F., and Steinberger, J. K. (2017). Human well-being and climate change mitigation. WIREs Climate Change 8, e485. <u>https://doi.org/10.1002/wcc.485</u>
- Lampl, C., Schmidt, A. E., and Aigner, E. (2024). Ernährungsarmut in Österreich als Barriere für eine gesunde und klimafreundliche Ernährung: Status quo und Handlungsoptionen. Wien: Gesundheit Österreich. Available at: <u>https://jasmin.goeg.at/id/</u> <u>eprint/3695/</u> (Accessed June 7, 2024).
- Langevin, J., Satre-Meloy, A., Satchwell, A. J., Hledik, R., Olszewski, J., Peters, K., et al. (2023). Demand-side solutions in the US building sector could achieve deep emissions reductions and avoid over \$100 billion in power sector costs. One Earth 6, 1005–1031.
- Larsson, J., Nässén, J., and Lundberg, E. (2019). "Work-time reduction for sustainable lifestyles," in *Routledge Handbook of Global Sustainability Governance*, (Routledge).
- Lechtenböhmer, S., Nilsson, L. J., Åhman, M., and Schneider, C. (2016). Decarbonising the energy intensive basic materials industry through electrification – Implications for future EU electricity demand. *Energy* 115, 1623–1631. <u>https://doi.org/10.1016/j.energy.2016.07.110</u>
- Lee, J., and Koch, M. (2023). The role of work and social protection systems in social-ecological transformations: Insights from deliberative citizen forums in Sweden. *European Journal of Social Security* 25, 408–425. <u>https://doi. org/10.1177/13882627231204989</u>
- Lehner, M. (2021). Carbon Capture and Utilization (CCU). Fact Sheet #32. CCCA. Available at: <u>https://ccca.ac.at/fileadmin/00</u> <u>DokumenteHauptmenue/02\_Klimawissen/FactSheets/32\_carbon\_capture\_and\_utilization\_ccu\_202104.pdf</u> (Accessed August 21, 2023).
- Lenzen, M., Sun, Y.-Y., Faturay, F., Ting, Y.-P., Geschke, A., and Malik, A. (2018). The carbon footprint of global tourism. *Nature Clim Change* 8, 522–528. <u>https://doi.org/10.1038/s41558-018-0141-x</u>

- Li, J., Peng, K., Wang, P., Zhang, N., Feng, K., Guan, D., et al. (2020). Critical Rare-Earth Elements Mismatch Global Wind-Power Ambitions. One Earth 3, 116–125. <u>https://doi.org/10.1016/j.</u> oneear.2020.06.009
- Lienhard, N., Mutschler, R., Leenders, L., and Rüdisüli, M. (2023). Concurrent deficit and surplus situations in the future renewable Swiss and European electricity system. *Energy Strategy Re*views 46, 101036. <u>https://doi.org/10.1016/j.esr.2022.101036</u>
- Littig, B. (2017). "Umweltschutz und Gewerkschaften eine langsame, aber stetige Annäherung," in Gewerkschaften und die Gestaltung einer sozial-ökologischen Gesellschaft, eds. U. Brand and K. Niedermoser (Wien: ÖGB Verlag), 195–204.
- Littig, B., Barth, T., and Jochum, G. (2018). Nachhaltige Arbeit ist mehr als green jobs. ArbeitnehmerInnenvertretungen und die sozial-ökologische Transformation der gegenwärtigen Arbeitsgesellschaft. WISO – Wirtschafts- und sozialpolitische Zeitschrift 41, 63–77.
- Littig, B., and Spitzer, M. (2011). Arbeit neu. Erweiterte Arbeitskonzepte im Vergleich. Düsseldorf: Hans-Böckler-Stiftung. Available at: <u>https://www.boeckler.de/de/faust-detail.htm?</u> <u>sync\_id=HBS-004939</u>
- Loibl, W., Tötzer, T., Köstl, M., Nabernegg, S., Steininger, K. W., Themeßl, M., et al. (2014). The Impact of Climate Change on Thermal Comfort in Austrian Cities. CCCA Factsheet #8e. CCCA.
- López Prol, J., De Llano Paz, F., Calvo-Silvosa, A., Pfenninger, S., and Staffell, I. (2024). Wind-solar technological, spatial and temporal complementarities in Europe: A portfolio approach. *Energy* 292, 130348. <u>https://doi.org/10.1016/j.energy.2024.130348</u>
- López Prol, J., Steininger, K. W., Williges, K., Grossmann, W. D., and Grossmann, I. (2023). Potential gains of long-distance trade in electricity. *Energy Economics* 124, 106739. <u>https://doi. org/10.1016/j.eneco.2023.106739</u>
- Lovins, A. B. (2010). Integrative Design: A Disruptive Source of Expanding Returns to Investments in Energy Efficiency. Rocky Mountain Institute. Available at: <u>https://rmi.org/insight/integrative-design-a-disruptive-source-of-expanding-returns-to-investments-in-energy-efficiency/</u> (Accessed May 6, 2024).
- Luderer, G., Madeddu, S., Merfort, L., Ueckerdt, F., Pehl, M., Pietzcker, R., et al. (2022). Impact of declining renewable energy costs on electrification in low-emission scenarios. *Nat Energy* 7, 32–42. <u>https://doi.org/10.1038/s41560-021-00937-z</u>
- Maeder, M., Weiss, O., and Boulouchos, K. (2021). Assessing the need for flexibility technologies in decarbonized power systems:
  A new model applied to Central Europe. *Applied Energy* 282, 116050. <u>https://doi.org/10.1016/j.apenergy.2020.116050</u>
- Maierhofer, D., Van Karsbergen, V., Potrč Obrecht, T., Ruschi Mendes Saade, M., Gingrich, S., Streicher, W., et al. (2024).
  Linking forest carbon opportunity costs and greenhouse gas emission substitution effects of wooden buildings: The climate optimum concept. Sustainable Production and Consumption 51, 612–627. https://doi.org/10.1016/j.spc.2024.08.021
- Märkle-Huß, J., Feuerriegel, S., and Neumann, D. (2018). Largescale demand response and its implications for spot prices, load and policies: Insights from the German-Austrian electricity market. *Applied Energy* 210, 1290–1298. <u>https://doi. org/10.1016/j.apenergy.2017.08.039</u>
- Marshall, Z., Heun, M. K., Brockway, P. E., Aramendia, E., Steenwyk, P., Relph, T., et al. (2024). Country-Level Primary-Final-Useful (CL-PFU) energy and exergy database v1.2, 1960– 2020. <u>https://doi.org/10.5518/1199</u>

- Marty, C. (2013). "Climate Change and Snow Cover in the European Alps," in *The Impacts of Skiing and Related Winter Recreational Activities on Mountain Environments*, eds. C. Rixen and A. Rolando (BENTHAM SCIENCE PUBLISHERS), 30–44. https://doi.org/10.2174/9781608054886113010005
- Mascherbauer, P., Kranzl, L., Yu, S., and Haupt, T. (2022). Investigating the impact of smart energy management system on the residential electricity consumption in Austria. *Energy* 249, 123665. <u>https://doi.org/10.1016/j.energy.2022.123665</u>
- Mason, N., and Büchs, M. (2023). Barriers to adopting wellbeing-economy narratives: comparing the Wellbeing Economy Alliance and Wellbeing Economy Governments. Sustainability: Science, Practice and Policy 19, 2222624. <u>https://doi.org/10.1080/ 15487733.2023.2222624</u>
- Mastrogiacomo, L., Barravecchia, F., and Franceschini, F. (2019). A worldwide survey on manufacturing servitization. Int J Adv Manuf Technol 103, 3927–3942. <u>https://doi.org/10.1007/ s00170-019-03740-z</u>
- Masuda, Y. J., Garg, T., Anggraeni, I., Ebi, K., Krenz, J., Game, E. T., et al. (2021). Warming from tropical deforestation reduces worker productivity in rural communities. *Nat Commun* 12, 1601. <u>https://doi.org/10.1038/s41467-021-21779-z</u>
- Material Economics (2018). The Circular Economy, a Powerful Force for Climate Mitigation: Transformative innovation for prosperous and low-carbon industry. Material Economics. Available at: https://materialeconomics.com/node/14
- Material Economics (2019). Industrial Transformation 2050. Pathways to Net-Zero Emissions from EU Heavy Industry. Cambridge: University of Cambridge Institute for Sustainability Leadership (CISL). Available at: <u>https://www.cisl.cam.ac.uk/</u> <u>system/files/documents/material-economics-industrial-transformation-2050.pdf</u> (Accessed November 7, 2024).
- Material Economics (2020). Preserving Value in EU Industrial Materials: A value perspective on the use of steel, plastics, aluminium. Material Economics. Available at: <u>https://materialeconomics.com/node/15</u>
- Mattioli, G., Roberts, C., Steinberger, J. K., and Brown, A. (2020). The political economy of car dependence: A systems of provision approach. *Energy Research & Social Science* 66, 101486. <u>https://doi.org/10.1016/j.erss.2020.101486</u>
- Mayer, A., Haas, W., Wiedenhofer, D., Krausmann, F., Nuss, P., and Blengini, G. A. (2019a). Measuring Progress towards a Circular Economy: A Monitoring Framework for Economy-wide Material Loop Closing in the EU28: Progress towards a Circular Economy in the EU28. *Journal of Industrial Ecology* 23, 62–76. https://doi.org/10.1111/jiec.12809
- Mayer, J., Bachner, G., and Steininger, K. W. (2019b). Macroeconomic implications of switching to process-emission-free iron and steel production in Europe. *Journal of Cleaner Production* 210, 1517–1533. <u>https://doi.org/10.1016/j.jclepro.2018.11.118</u>
- Mayer, M., and Abegg, B. (2024). Development of summer skiing days in Austrian glacier ski areas in the first two decades of the twenty-first century. *International Journal of Biometeorology* 68, 547–564. https://doi.org/10.1007/s00484-022-02371-6
- Mayer, M., Schreier, S. F., Spangl, W., Staehle, C., Trimmel, H., and Rieder, H. E. (2022). An analysis of 30 years of surface ozone concentrations in Austria: temporal evolution, changes in precursor emissions and chemical regimes, temperature dependence, and lessons for the future. *Environ. Sci.: Atmos.* 2, 601– 615. <u>https://doi.org/10.1039/D2EA00004K</u>

- McKenna, R., Lilliestam, J., Heinrichs, H. U., Weinand, J., Schmidt, J., Staffell, I., et al. (2025). System impacts of wind energy developments: Key research challenges and opportunities. *Joule* 9, 101799. <u>https://doi.org/10.1016/j.joule.2024.11.016</u>
- McKinsey & Company (2016). The circular economy: Moving from theory to practice. Available at: <u>https://www.mckinsey. com/~/media/McKinsey/Business%20Functions/Sustainability/Our%20Insights/The%20circular%20economy%20Moving%20from%20theory%20to%20practice/The%20circular%20 economy%20Moving%20from%20theory%20to%20practice. ashx (Accessed May 7, 2024).</u>
- Meadows, D. (1999). Leverage Points: Places to Intervene in a System. Hartland, VT, USA: The Sustainability Institute. Available at: <u>https://www.donellameadows.org/wp-content/userfiles/ Leverage\_Points.pdf</u> (Accessed January 17, 2025).
- Meinel, U., and Höferl, K.-M. (2017). Non-Adaptive Behavior in the Face of Climate Change: First Insights from a Behavioral Perspective Based on a Case Study among Firm Managers in Alpine Austria. Sustainability 9, 1132. <u>https://doi.org/10.3390/ su9071132</u>
- Meinhart, B., Gabelberger, F., Sinabell, F., and Streicher, G. (2022). Transformation und "Just Transition" in Österreich. Wien: Österreichisches Institut für Wirtschaftsforschung (WIFO). Available at: <u>https://www.wifo.ac.at/jart/prj3/wifo/resources/ person\_dokument/person\_dokument.jart?publikationsid= 68029&mime\_type=application/pdf</u>
- Metic, J., and Pigosso, D. C. A. (2022). Research avenues for uncovering the rebound effects of the circular economy: A systematic literature review. *Journal of Cleaner Production* 368, 133133. <u>https://doi.org/10.1016/j.jclepro.2022.133133</u>
- Meyer, I., Sinabell, F., Streicher, G., Bohner, A., and Spiegel, H. (2024). "Carbon Sequestration in Austrian Soils: Environmental Significance and Economic Effects of Selected Measures," in *Biodiversity and Climate. Tackling Global Footprints.*, eds. E. Civel, C. de Perthuis, J. E. Milne, M. S. Andersen, and H. Ashiabo (Cheltenham, UK: Edward Elgar Publishing), 74–87.
- Miatto, A., Emami, N., Goodwin, K., West, J., Taskhiri, M. S., Wiedmann, T., et al. (2024). Australia's circular economy metrics and indicators. *Journal of Industrial Ecology* n/a. <u>https://doi.org/10.1111/jiec.13458</u>
- Mikellidou, C. V., Shakou, L. M., Boustras, G., and Dimopoulos, C. (2018). Energy critical infrastructures at risk from climate change: A state of the art review. *Safety Science* 110, 110–120. <u>https://doi.org/10.1016/j.ssci.2017.12.022</u>
- Mikovits, C., Krexner, T., Kral, I., Bauer, A., Schauppenlehner, T., Schönhart, M., et al. (2024). The Contribution of Agrivoltaics to Reaching Climate Neutrality in Austria: Combined Simulation of Electricity and Crop Outputs of Agrivoltaic Systems. <u>https:// doi.org/10.2139/ssrn.4743081</u>
- Mitter, H., Larcher, M., Schönhart, M., Stöttinger, M., and Schmid, E. (2019). Exploring Farmers' Climate Change Perceptions and Adaptation Intentions: Empirical Evidence from Austria. *Envi*ronmental Management 63, 804–821. <u>https://doi.org/10.1007/</u> s00267-019-01158-7
- Morseletto, P. (2020). Targets for a circular economy. *Resources*, *Conservation and Recycling* 153, 104553. <u>https://doi.org/10.1016/j.resconrec.2019.104553</u>
- Morseletto, P., and Haas, W. (2023). A call for high-quality data to foster a decisive transformation towards a circular economy. *Resources, Conservation and Recycling* 197, 107092. <u>https://doi. org/10.1016/j.resconrec.2023.107092</u>

- Mowbray, B. A. W., Zhang, Z. B., Parkyn, C. T. E., and Berlinguette, C. P. (2023). Electrochemical Cement Clinker Precursor Production at Low Voltages. ACS Energy Lett. 8, 1772–1778. <u>https:// doi.org/10.1021/acsenergylett.3c00242</u>
- Müller, A., Krucsay, B., Keil, C., Pimminger, F., Glowinska, I., Brandl, J., et al. (2019). Von A wie Arbeit bis Z wie Zukunft. Arbeiten und Wirtschaften in der Klimakrise. Wien/Berlin: Periskop/I.L.A. Kollektiv.
- Muñoz, S., Hosseini, M. R., and Crawford, R. H. (2024). Towards a holistic assessment of circular economy strategies: The 9R circularity index. *Sustainable Production and Consumption* 47, 400–412. https://doi.org/10.1016/j.spc.2024.04.015
- Nagovnak, P., Schützenhofer, C., Mobarakeh, M. R., Cvetkovska, R., Stortecky, S., Hainoun, A., et al. (2024). Assessment of technology-based options for climate neutrality in Austrian manufacturing industry. *Heliyon* 10, e25382. <u>https://doi.org/10.1016/j. heliyon.2024.e25382</u>
- Narula, K., Ploiner, C., Getzinger, G., and Patel, M. K. (2022). Impact of energy efficiency and decarbonisation policies for buildings: A comparative assessment of Austria and Switzerland. *Energy and Buildings* 268, 112175. <u>https://doi.org/10.1016/j. enbuild.2022.112175</u>
- Nascimento, L., Kuramochi, T., Iacobuta, G., den Elzen, M., Fekete, H., Weishaupt, M., et al. (2022). Twenty years of climate policy: G20 coverage and gaps. *Climate Policy* 22, 158–174. <u>https://doi.org/10.1080/14693062.2021.1993776</u>
- Nature Reloaded (2015). Vienna: Österreich Werbung. Available at: <u>https://www.austriatourism.com/fileadmin/user\_upload/</u> <u>Media\_Library/Downloads/Marke/thema-austriantime.pdf</u>
- Neger, C., Prettenthaler, F., Gössling, S., and Damm, A. (2021). Carbon intensity of tourism in Austria: Estimates and policy implications. *Journal of Outdoor Recreation and Tourism* 33, 100331. <u>https://doi.org/10.1016/j.jort.2020.100331</u>
- Neier, T., Kreinin, H., Gerold, S., Heyne, S., Laa, E., and Bohnenberger, K. (2024). Navigating labor-market transitions: an eco-social policy toolbox for public employment services. Sustainability: Science, Practice and Policy 20, 2386799. <u>https://doi.org/10.1080/15487733.2024.2386799</u>
- Neier, T., Kreinin, H., Heyne, S., Laa, E., and Bohnenberger, K. (2022). Sozial-ökologische Arbeitsmarktpolitik. Vienna: Austrian Chamber of Labour. Available at: <u>https://noe.arbeiterkammer.at/service/zeitschriftenundstudien/arbeitundwirtschaft/</u> <u>Sozial-oekologische\_Arbeitsmarktpolitik.html</u> (Accessed September 15, 2023).
- Nelson, R. R. ed. (1993). National innovation systems: a comparative analysis. New York: Oxford University Press.
- Net Positive Employee Barometer (2023). From Quiet Quitting To Conscious Quitting. Available at: <u>https://www.paulpol-man.com/wp-content/uploads/2023/02/MC\_Paul-Polman\_Net-Positive-Employee-Barometer\_Final\_web.pdf</u> (Accessed May 6, 2024).
- Niedermoser, K. (2017). "Gewerkschaften und die ökologische Frage – historische Entwicklungen und aktuelle Herausforderungen," in Gewerkschaften und die Gestaltung einer sozial-ökologischen Gesellschaft, eds. U. Brand and K. Niedermoser (Wien: ÖGB Verlag).
- Nijnens, J., Behrens, P., Kraan, O., Sprecher, B., and Kleijn, R. (2023). Energy transition will require substantially less mining than the current fossil system. *Joule* 7, 2408–2413. <u>https://doi.org/10.1016/j.joule.2023.10.005</u>

- Nobre, G. C., and Tavares, E. (2021). The quest for a circular economy final definition: A scientific perspective. *Journal of Cleaner Production* 314, 127973. <u>https://doi.org/10.1016/j.jcle-</u> pro.2021.127973
- Nordberg, E. J., Julian Caley, M., and Schwarzkopf, L. (2021). Designing solar farms for synergistic commercial and conservation outcomes. *Solar Energy* 228, 586–593. <u>https://doi.org/10.1016/j. solener.2021.09.090</u>
- Northern Lights (2024). Northern Lights About the Longship project. Available at: <u>https://norlights.com/about-the-longshipproject/</u> (Accessed May 6, 2024).
- Novy, A., Haderer, M., Kubeczko, K., Aigner, E., Bärnthaler, R., Brand, U., et al. (2023). "Kapitel 2. Perspektiven zur Analyse und Gestaltung von Strukturen klimafreundlichen Lebens," in *APCC Special Report: Strukturen für ein klimafreundliches Leb*en, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 195–213. https://doi.org/10.1007/978-3-662-66497-1\_6
- NPC (2019). Meeting the Dual Challenge A Roadmap to At-Scale Deployment of Carbon Capture, Use, and Storage. National Petroleum Council. Available at: <u>https://dualchallenge.npc.org/</u> <u>downloads.php</u> (Accessed August 21, 2023).
- Oberholzer, B. (2023). Post-growth transition, working time reduction, and the question of profits. *Ecological Economics* 206, 107748. <u>https://doi.org/10.1016/j.ecolecon.2023.107748</u>
- Odelenbosch, O., van den Berg, M., Hof, A., de Boer, H. S., Daioglou, V., Dekker, M., et al. (2022). Mitigating greenhouse gas emissions in hard-to-abate sectors. The Hague: PBL Netherlands Environmental Assessment Agency. Available at: <u>https:// www.pbl.nl/en/publications/mitigating-greenhouse-gas-emissions-in-hard-to-abate-sectors</u> (Accessed August 1, 2023).
- Odenweller, A., Ueckerdt, F., Nemet, G. F., Jensterle, M., and Luderer, G. (2022). Probabilistic feasibility space of scaling up green hydrogen supply. *Nat Energy* 7, 854–865. <u>https://doi.org/10.1038/s41560-022-01097-4</u>
- OECD (2024). OECD Employment Outlook 2024: The Net-Zero Transition and the Labour Market. Paris: OECD Publishing. https://doi.org/10.1787/ac8b3538-en
- ÖGB (2021). Klimapolitik aus ArbeitnehmerInnen- Perspektive. Vienna: Austrian Trade Union Federation (ÖGB). Available at: <u>https://www.oegb.at/content/dam/oegb/downloads/forderung-spapiere/Klima-Positionspapier%20des%20%C3%96GB\_Kli-mapolitik%20aus%20ArbeitnehmerInnen-Perspektive.pdf</u>
- Olajire, A. (2013). A Review of Mineral Carbonation Technology in Sequestration of CO2. Journal of Petroleum Science and Engineering 109, 364–392. <u>https://doi.org/10.1016/j.petrol.2013.03.013</u>
- Olefs, M., Girstmair, A., Hiebl, J., Koch, R., and Schoener, W. (2017). An area-wide snow climatology for Austria since 1961 based on newly available daily precipitation and air temperature grids. *Geophysical Research Abstracts* 19. Available at: <u>https://meetingorganizer.copernicus.org/EGU2017/EGU2017-12249.pdf</u>
- O'Neill, D. W., Fanning, A. L., Lamb, W. F., and Steinberger, J. K. (2018b). A good life for all within planetary boundaries. *Nature Sustainability* 1, 88–95. <u>https://doi.org/10.1038/s41893-018-0021-4</u>
- Ott, H. (2023). Carbon Capture and Storage (CCS). Fact Sheet #43. CCCA. Available at: <u>https://ccca.ac.at/fileadmin/00\_Doku-menteHauptmenue/02\_Klimawissen/FactSheets/43\_carbon\_capture\_storage\_202304.pdf</u> (Accessed August 21, 2023).

- Ott, H., and Kulich, J. (2024). CCS: Chancen und Risiken einer umstrittenen Technologie. *Berg Huettenmaenn Monatsh* 169, 553–559. https://doi.org/10.1007/s00501-024-01508-x
- Owen, J. R., Kemp, D., Lechner, A. M., Harris, J., Zhang, R., and Lèbre, É. (2023). Energy transition minerals and their intersection with land-connected peoples. *Nat Sustain* 6, 203–211. <u>https://doi.org/10.1038/s41893-022-00994-6</u>
- Paech, N. (2012). Befreiung vom Überfluss: Auf dem Weg in die Postwachstumsökonomie. München: Oekom Verlag.
- Paltsev, S., Morris, J., Kheshgi, H., and Herzog, H. (2021). Hard-to-Abate Sectors: The role of industrial carbon capture and storage (CCS) in emission mitigation. *Applied Energy* 300, 117322. <u>https://doi.org/10.1016/j.apenergy.2021.117322</u>
- Paramonova, S., and Thollander, P. (2016). Energy-efficiency networks for SMEs: Learning from the Swedish experience. *Renewable and Sustainable Energy Reviews* 65, 295–307. <u>https://doi. org/10.1016/j.rser.2016.06.088</u>
- Parsons, L. A., Masuda, Y. J., Kroeger, T., Shindell, D., Wolff, N. H., and Spector, J. T. (2022). Global labor loss due to humid heat exposure underestimated for outdoor workers. *Environ. Res. Lett.* 17, 014050. <u>https://doi.org/10.1088/1748-9326/ac3dae</u>
- Parsons, L. A., Shindell, D., Tigchelaar, M., Zhang, Y., and Spector, J. T. (2021). Increased labor losses and decreased adaptation potential in a warmer world. *Nat Commun* 12, 7286. <u>https://doi.org/10.1038/s41467-021-27328-y</u>
- People's Republic of China (2008). Circular Economy Promotion Law. Available at: <u>https://www.greenpolicyplatform.org/sites/</u> <u>default/files/downloads/policy-database/CHINA)%20Circu-</u> <u>lar%20Economy%20Promotion%20Law%20(2008).pdf</u> (Accessed May 7, 2024).
- Perera, A. T. D., Nik, V. M., Chen, D., Scartezzini, J.-L., and Hong, T. (2020). Quantifying the impacts of climate change and extreme climate events on energy systems. *Nat Energy* 5, 150–159. <u>https://doi.org/10.1038/s41560-020-0558-0</u>
- Petschow, U., Lange, S., Hofmann, D., Pissarskoi, E., aus dem Moore, N., Korfhage, T., et al. (2018). Gesellschaftliches Wohlergehen innerhalb planetarer Grenzen. Der Ansatz einer vorsorgeorientierten Postwachstumsposition. Dessau-Roßlau: Umweltbundesamt. Available at: <u>https://www.umweltbundesamt.de/sites/default/files/medien/1410/publikationen/uba\_texte\_89\_2018\_vorsorgeorientierte\_postwachstumsposition.pdf</u> (Accessed February 15, 2021).
- Pettinger, L. (2017). Green collar work: Conceptualizing and exploring an emerging field of work. *Sociology Compass* 11, e12443. <u>https://doi.org/10.1111/soc4.12443</u>
- Pichler, M., Krenmayr, N., Maneka, D., Brand, U., Högelsberger, H., and Wissen, M. (2021). Beyond the jobs-versus-environment dilemma? Contested social-ecological transformations in the automotive industry. *Energy Research & Social Science* 79, 102180. <u>https://doi.org/10.1016/j.erss.2021.102180</u>
- Pichler, P.-P., Jaccard, I. S., Weisz, U., and Weisz, H. (2019). International comparison of health care carbon footprints. *Environ. Res. Lett.* 14, 064004. https://doi.org/10.1088/1748-9326/ab19e1
- Pineault, É. (2023). A Social Ecology of Capital. London: Pluto Press. https://doi.org/10.2307/jj.168342
- Pirklbauer, S., and Wukovitsch, F. (2019). Nachhaltige Arbeit ein interessenpolitischer Blick aus der Arbeiterkammer (Österreich). WSI-Mitteilungen 72, 59–63. <u>https://doi.org/10.5771/0342-300X-2019-1-59</u>
- Pisoni, E., Christidis, P., and Navajas Cawood, E. (2022). Active mobility versus motorized transport? User choices and benefits

for the society. *Science of The Total Environment* 806, 150627. https://doi.org/10.1016/j.scitotenv.2021.150627

- Pitelis, A., Vasilakos, N., and Chalvatzis, K. (2020). Fostering innovation in renewable energy technologies: Choice of policy instruments and effectiveness. *Renewable Energy* 151, 1163–1172. <u>https://doi.org/10.1016/j.renene.2019.11.100</u>
- Plank, B., Eisenmenger, N., and Schaffartzik, A. (2021a). Do material efficiency improvements backfire?: Insights from an index decomposition analysis about the link between CO2 emissions and material use for Austria. *Journal of Industrial Ecology* 25, 511–522. https://doi.org/10.1111/jiec.13076
- Plank, C., Liehr, S., Hummel, D., Wiedenhofer, D., Haberl, H., and Görg, C. (2021b). Doing more with less: Provisioning systems and the transformation of the stock-flow-service nexus. *Ecological Economics* 187, 107093. <u>https://doi.org/10.1016/j.ecole-</u> <u>con.2021.107093</u>
- Plaza, M. G., Martínez, S., and Rubiera, F. (2020). CO2 Capture, Use, and Storage in the Cement Industry: State of the Art and Expectations. *Energies* 13, 5692. <u>https://doi.org/10.3390/ en13215692</u>
- Polaine, A., Løvlie, L., and Reason, B. (2013). Service Design: From Insight to Implementation., 1<sup>st</sup> Edn. Brooklyn, NY: Rosenfeld Media.
- Potting, J., Hekkert, M. P., Worrell, E., and Hanemaaijer, A. (2017). Circular Economy: Measuring Innovation in the Product Chain. Den Haage: PBL Netherlands Environmental Assessment Agency. Available at: <u>https://dspace.library.uu.nl/handle/1874/358310</u> (Accessed May 7, 2024).
- Prettenthaler, F., Gössling, S., Damm, A., Neger, C., Köberl, J., and Haas, W. (2021). "Berücksichtigung der globalen Entwicklung," in *Tourismus und Klimawandel*, eds. U. Pröbstl-Haider, D. Lund-Durlacher, M. Olefs, and F. Prettenthaler (Berlin, Heidelberg: Springer Spektrum), 195–208. <u>https://doi.org/10.1007/978-3-662-61522-5\_11</u>
- Prettenthaler, F., Töglhofer, C., Habsburg-Lothringen, C., and Türk, A. (2008). "Klimabedingte Änderung des Heiz- und Kühlenergiebedarfs für Österreich," in *Studien zum Klimawandel in Österreich: Heizen und Kühlen im Klimawandel*, eds. F. Prettenthaler and A. Gobiet (Vienna: Verlag der Österreichischen Akademie der Wissenschaften (ÖAW)), 29–49.
- Pröbstl-Haider, U., Lund-Durlacher, D., Olefs, M., and Prettenthaler, F. eds. (2020). *Tourismus und Klimawandel*. Berlin, Heidelberg: Springer Spektrum.
- Prokop, G. (2019). Bodenverbrauch in Österreich. BMK, UBA. Available at: <u>https://www.naturland-noe.at/download/stu-</u> <u>die%20uba%20bodenverbrauch.pdf</u>
- Purr, K., and Garvens, H.-J. (2021). Diskussionsbeitrag zur Bewertung von Carbon Capture and Utilization. Dessau-Roßlau: Umweltbundesamt. Available at: <u>https://www.umweltbundesamt.</u> <u>de/publikationen/diskussionsbeitrag-zur-bewertung-von-carbon-capture</u> (Accessed August 21, 2023).
- Rahnama Mobarakeh, M., and Kienberger, T. (2022). Climate neutrality strategies for energy-intensive industries: An Austrian case study. *Cleaner Engineering and Technology* 10, 100545. <u>https://doi.org/10.1016/j.clet.2022.100545</u>
- Ramsebner, J., Linares, P., and Haas, R. (2021). Estimating storage needs for renewables in Europe: The correlation between renewable energy sources and heating and cooling demand. *Smart Energy* 3, 100038. https://doi.org/10.1016/j.segy.2021.100038

- Rao, N. D., and Min, J. (2018). Less global inequality can improve climate outcomes. Wiley Interdisciplinary Reviews: Climate Change 9, e513. <u>https://doi.org/10.1002/wcc.513</u>
- Rau, H. (2015). "Time use and resource consumption," in *International Encyclopedia of the Social and Behavioural Sciences*, ed. J.
   D. Wright (Oxford: Elsevier).
- Rauter, M., Kaufmann, M., Thaler, T., and Fuchs, S. (2020). Flood risk management in Austria: Analysing the shift in responsibility-sharing between public and private actors from a public stakeholder's perspective. *Land Use Policy* 99, 105017. <u>https:// doi.org/10.1016/j.landusepol.2020.105017</u>
- Reck, B. K., and Graedel, T. E. (2012). Challenges in Metal Recycling. Science 337, 690–695. <u>https://doi.org/10.1126/science.1217501</u>
- Regulation (EU) 2024/3012 (2024). Regulation (EU) 2024/3012 of the European Parliament and of the Council of 27 November 2024 establishing a Union certification framework for permanent carbon removals, carbon farming and carbon storage in products. Available at: <u>http://data.europa.eu/eli/reg/2024/3012/</u> oj
- Rehm, M., Huwe, V., and Bohnenberger, K. (2023). Klimasoziale Transformation – Klimaschutz und Ungleichheitsreduktion wirken Hand in Hand. Gütersloh: Bertelsmann Stiftung.
- Reichl, J., Schmidthaler, M., and Schneider, F. (2013). The value of supply security: The costs of power outages to Austrian households, firms and the public sector. *Energy Economics* 36, 256– 261. <u>https://doi.org/10.1016/j.eneco.2012.08.044</u>
- Reike, D., Vermeulen, W. J. V., and Witjes, S. (2018). The circular economy: New or Refurbished as CE 3.0? — Exploring Controversies in the Conceptualization of the Circular Economy through a Focus on History and Resource Value Retention Options. *Resources, Conservation and Recycling* 135, 246–264. https://doi.org/10.1016/j.resconrec.2017.08.027
- Reisch, L. A., and Bietz, S. (2014). Zeit für Nachhaltigkeit Zeiten der Transformation: mit Zeitpolitik gesellschaftliche Veränderungsprozesse steuern. München: oekom.
- Reiter, G., and Lindorfer, J. (2015). Evaluating CO2 sources for power-to-gas applications – A case study for Austria. *Journal of CO2 Utilization* 10, 40–49. <u>https://doi.org/10.1016/j. jcou.2015.03.003</u>
- Resch, G., Totschnig, G., Suna, D., Schöniger, F., Geipel, J., and Liebmann, L. (2020). "Assessment of Prerequisites and Impacts of a Renewable-Based Electricity Supply in Austria by 2030," in *Accelerating the Transition to a 100% Renewable Energy Era*, ed. T. S. Uyar (Cham: Springer International Publishing), 99–111. https://doi.org/10.1007/978-3-030-40738-4\_4
- Richter, S., Braun-Unkhoff, M., Naumann, C., and Riedel, U. (2018). Paths to alternative fuels for aviation. *CEAS Aeronautical Journal* 9, 389–403. <u>https://doi.org/10.1007/s13272-018-0296-1</u>
- Rinderspacher, J. P. (1985). *Gesellschaft ohne Zeit: individuelle Zeitverwendung und soziale Organisation der Arbeit.* Frankfurt [am Main]; New York: Campus.
- Rissman, J., Bataille, C., Masanet, E., Aden, N., Morrow, W. R., Zhou, N., et al. (2020). Technologies and policies to decarbonize global industry: Review and assessment of mitigation drivers through 2070. *Applied Energy* 266, 114848. <u>https://doi. org/10.1016/j.apenergy.2020.114848</u>
- Ritchie, H., Rosado, P., and Roser, M. (2024). CO<sub>2</sub> and Greenhouse Gas Emissions. *Our World in Data*. Available at: <u>https://our-worldindata.org/co2-and-greenhouse-gas-emissions</u>

- Roberts, C., Geels, F. W., Lockwood, M., Newell, P., Schmitz, H., Turnheim, B., et al. (2018). The politics of accelerating low-carbon transitions: Towards a new research agenda. *Energy Research & Social Science* 44, 304–311. <u>https://doi.org/10.1016/j.</u> <u>erss.2018.06.001</u>
- Rode, A., Carleton, T., Delgado, M., Greenstone, M., Houser, T., Hsiang, S., et al. (2021). Estimating a social cost of carbon for global energy consumption. *Nature* 598, 308–314. <u>https://doi. org/10.1038/s41586-021-03883-8</u>
- Røpke, I. (1999). The dynamics of willingness to consume. *Ecological Economics* 28, 399–420. <u>https://doi.org/10.1016/S0921-8009(98)00107-4</u>
- Rosa, H. (2005). Beschleunigung: die Veränderung der Zeitstrukturen in der Moderne., 1. Aufl. Frankfurt am Main: Suhrkamp.
- Rosa, H., Paech, N., Habermann, F., Haug, F., Wittmann, F., and Kirschenmann, L. (2015). Zeitwohlstand: wie wir anders arbeiten, nachhaltig wirtschaften und besser leben., 2. Auflage, ed.Konzeptwerk Neue Ökonomie. München: Oekom Verlag.
- Rosenfeld, D. C., Böhm, H., Lindorfer, J., and Lehner, M. (2020). Scenario analysis of implementing a power-to-gas and biomass gasification system in an integrated steel plant: A techno-economic and environmental study. *Renewable Energy* 147, 1511– 1524.
- Rübbelke, D., and Vögele, S. (2011). Impacts of climate change on European critical infrastructures: The case of the power sector. *Environmental Science & Policy* 14, 53–63. <u>https://doi. org/10.1016/j.envsci.2010.10.007</u>
- Sachverständigenrat für Umweltfragen ed. (2024). Suffizienz als "Strategie des Genug": Eine Einladung zur Diskussion. Berlin: Sachverständigenrat für Umweltfragen SRU. Available at: <u>https://www.umweltrat.de/SharedDocs/Downloads/DE/04\_</u> <u>Stellungnahmen/2020\_2024/2024\_03\_Suffizienz.html</u> (Accessed May 16, 2024).
- Sala, A., Lütkemeyer, M., Birkmaier, A., Martineau, S., Schieder, P., Bruckmüller, T., et al. (2020). E-MAPP 2. E-Mobility – Austrian Prouction Potential, Qualification and Traning needs. Wien: Fraunhofer Austria Research GmbH, TU Wien, Smart Mobility Power GmbH. Available at: <u>https://www.klimafonds.</u> gv.at/wp-content/uploads/sites/16/2020\_E-MAPP2\_-FhA\_ TU\_SMP\_v2.3.pdf
- Salim, E., Mayer, M., Sacher, P., and Ravanel, L. (2023). Visitors' motivations to engage in glacier tourism in the European Alps: comparison of six sites in France, Switzerland, and Austria. *Journal of Sustainable Tourism* 31, 1373–1393. <u>https://doi.org/</u>10.1080/09669582.2022.2044833
- Samadi, S., Fischer, A., and Lechtenböhmer, S. (2023). The renewables pull effect: How regional differences in renewable energy costs could influence where industrial production is located in the future. *Energy Research & Social Science* 104, 103257. https://doi.org/10.1016/j.erss.2023.103257
- Santarius, T., Pohl, J., and Lange, S. (2020). Digitalization and the Decoupling Debate: Can ICT Help to Reduce Environmental Impacts While the Economy Keeps Growing? *Sustainability* 12, 7496. <u>https://doi.org/10.3390/su12187496</u>
- Sathaye, J. A., Dale, L. L., Larsen, P. H., Fitts, G. A., Koy, K., Lewis, S. M., et al. (2013). Estimating impacts of warming temperatures on California's electricity system. *Global Environmental Change* 23, 499–511. <u>https://doi.org/10.1016/j.gloenvcha.2012.12.005</u>
- Schaeffer, R., Szklo, A. S., Pereira De Lucena, A. F., Moreira Cesar Borba, B. S., Pupo Nogueira, L. P., Fleming, F. P., et al. (2012).

Energy sector vulnerability to climate change: A review. *Energy* 38, 1–12. <u>https://doi.org/10.1016/j.energy.2011.11.056</u>

- Schäfer, M., Jaeger-Erben, M., and Bamberg, S. (2012). Life Events as Windows of Opportunity for Changing Towards Sustainable Consumption Patterns? J Consum Policy 35, 65–84. <u>https://doi.org/10.1007/s10603-011-9181-6</u>
- Schaffartzik, A., Pichler, M., Pineault, E., Wiedenhofer, D., Gross, R., and Haberl, H. (2021). The transformation of provisioning systems from an integrated perspective of social metabolism and political economy: a conceptual framework. *Sustain Sci* 16, 1405–1421. <u>https://doi.org/10.1007/s11625-021-00952-9</u>
- Schalatek, L. (2012). "Gender und Klimafinanzierung: Doppeltes Mainstreaming für Nachhaltige Entwicklung," in Geschlecht
  – Macht – Klima. Feministische Perspektiven auf Klima, gesellschaftliche Naturverhältnisse und Gerechtigkeit, eds. G. Çağlar, M. do Mar Castro Varela, and H. Schwenken (Opladen, Berlin & Toronto: Barbara Budrich), 137–167.
- Scharf, C., and Clemens, T. (2006). CO2-Sequestration Potential in Austrian Oil and Gas Fields. <u>https://doi.org/10.2118/100176-MS</u>
- Scherhaufer, P., Höltinger, S., Salak, B., Schauppenlehner, T., and Schmidt, J. (2017). Patterns of acceptance and non-acceptance within energy landscapes: A case study on wind energy expansion in Austria. *Energy Policy* 109, 863–870. <u>https://doi.org/10.1016/j.enpol.2017.05.057</u>
- Schleypen, J. R., Dasgupta, S., Borsky, S., Jury, M., Ščasný, M., and Bezhanishvili, L. (2019). D2.4 Impacts on Industry, Energy, Services, and Trade. Deliverable of the H2020 COACCH project. Available at: <u>https://www.coacch.eu/wp-content/uploads/2018/03/D2.4\_revMAR2022.pdf</u>
- Schmidt, J., Mitter, H., Baumann, M., Boza-Kiss, B., Huppmann, D., Wehrle, S., et al. (2025). Need for speed: Co-creating scenarios for climate neutral energy systems in Austria in 2040. *Energy Policy* 198, 114493. <u>https://doi.org/10.1016/j.enpol.2024.114493</u>
- Schöggl, J.-P., Stumpf, L., Rusch, M., and Baumgartner, R. J. (2022). Die Umsetzung der Kreislaufwirtschaft in österreichischen Unternehmen – Praktiken, Strategien und Auswirkungen auf den Unternehmenserfolg. Österr Wasser- und Abfallw 74, 51–63. <u>https://doi.org/10.1007/s00506-021-00828-3</u>
- Scholz, Y., Gils, H. C., and Pietzcker, R. C. (2017). Application of a high-detail energy system model to derive power sector characteristics at high wind and solar shares. *Energy Economics* 64, 568–582. <u>https://doi.org/10.1016/j.eneco.2016.06.021</u>
- Schöniger, F., Resch, G., Suna, D., Formayer, H., Pardo-Garcia, N., Hasengst, F., et al. (2023). Securing Austria's Electricity Supply in times of Climate Change; Final report of the ACRP12 project SECURES (2020–2023). Available at: <u>https://www.secures.</u> <u>at/fileadmin/cmc/Final\_Report\_SECURES.pdf</u> (Accessed May 6, 2024).
- Schöniger, F., Resch, G., Suna, D., Widhalm, P., Totschnig, G., Pardo-Garcia, N., et al. (2024). SECURES-Energy: Hourly electricity demand and supply profiles for historical climate and climate change projections in Europe until 2100. <u>https://doi.org/10.5281/ZENODO.8355364</u>
- Schor, J. B. (2005). Sustainable Consumption and Worktime Reduction. Journal of Industrial Ecology 9, 37–50. <u>https://doi.org/</u> <u>10.1162/1088198054084581</u>
- Schor, J. B. (2016). Wahrer Wohlstand: mit weniger Arbeit besser leben., Deutsche Erstausgabe. München: Oekom.
- Schug, F., Frantz, D., Wiedenhofer, D., Haberl, H., Virág, D., Van Der Linden, S., et al. (2023). High-resolution mapping of 33

years of material stock and population growth in Germany using Earth Observation data. *J of Industrial Ecology* 27, 110–124. https://doi.org/10.1111/jiec.13343

- Schulte, P. A., Bhattacharya, A., Butler, C. R., Chun, H. K., Jacklitsch, B., Jacobs, T., et al. (2016). Advancing the framework for considering the effects of climate change on worker safety and health. *J Occup Environ Hyg* 13, 847–865. <u>https://doi.org/10.108</u> 0/15459624.2016.1179388
- Schultheiß, J., Feigl, G., Pirklbauer, S., and Wukovitsch, F. (2021). AK-Wohlstandsbericht 2021. Analyse des gesellschaftlichen Fortschritts in Österreich 2017–2022. Materialien zu Wirtschaft und Gesellschaft. Wien: AK Wien. Available at: <u>https://www.arbeiterkammer.at/interessenvertretung/wirtschaft/verteilungsgerechtigkeit/AK-Wohlstandsbericht\_2021.pdf</u>
- Schwaiger, C., Trautner, T., and Bleicher, F. (2023). Energy-Cost-Optimized Strategies for Discrete Mechanical Manufacturing. *Procedia CIRP* 118, 1022–1027. <u>https://doi.org/10.1016/j.procir.2023.06.175</u>
- Scott, D., and Gössling, S. (2022). A review of research into tourism and climate change – Launching the annals of tourism research curated collection on tourism and climate change. Annals of Tourism Research 95, 103409. <u>https://doi.org/10.1016/j.</u> annals.2022.103409
- Scott, D., Hall, C. M., and Gössling, S. (2019). Global tourism vulnerability to climate change. *Annals of Tourism Research* 77, 49–61. <u>https://doi.org/10.1016/j.annals.2019.05.007</u>
- Seebauer, S. (2018). The psychology of rebound effects: Explaining energy efficiency rebound behaviours with electric vehicles and building insulation in Austria. *Energy Research & Social Science* 46, 311–320. <u>https://doi.org/10.1016/j.erss.2018.08.006</u>
- Seidl, I., and Zahrnt, A. eds. (2019). Tätigsein in der Postwachstumsgesellschaft. Marburg: Metropolis-Verlag.
- Seidl, R., von Wirth, T., and Krütli, P. (2019). Social acceptance of distributed energy systems in Swiss, German, and Austrian energy transitions. *Energy Research & Social Science* 54, 117–128. https://doi.org/10.1016/j.erss.2019.04.006
- Sejkora, C., Kühberger, L., Radner, F., Trattner, A., and Kienberger, T. (2022). Exergy as criteria for efficient energy systems Maximising energy efficiency from resource to energy service, an Austrian case study. *Energy* 239, 122173. <u>https://doi.org/10.1016/j.energy.2021.122173</u>
- Seto, K. C., Davis, S. J., Mitchell, R. B., Stokes, E. C., Unruh, G., and Ürge-Vorsatz, D. (2016). Carbon Lock-In: Types, Causes, and Policy Implications. *Annu. Rev. Environ. Resour.* 41, 425–452. <u>https://doi.org/10.1146/annurev-environ-110615-085934</u>
- Shove, E., Trentmann, F., and Wilk, R. R. eds. (2009). *Time, con-sumption and everyday life: practice, materiality and culture*. Oxford; New York: Routledge.
- Simoes, S. G., Amorim, F., Siggini, G., Sessa, V., Saint-Drenan, Y.-M., Carvalho, S., et al. (2021). Climate proofing the renewable electricity deployment in Europe – Introducing climate variability in large energy systems models. *Energy Strategy Reviews* 35, 100657. <u>https://doi.org/10.1016/j.esr.2021.100657</u>
- Smetschka, B., Aigner, E., and Bohnenberger, K. (2024a). "Structures and Time for Climate-friendly Work," in Nachhaltige Arbeitswelten: Überlegungen zu einer zukunftsfähigen Gestaltung von Arbeit, eds. A. Raschauer and N. Tomaschek (Münster: Waxmann).
- Smetschka, B., Aigner, E., and Mader, K. (2024b). "Strukturen für einen klimafreundlichen Alltag und Potenziale für Klima-, Sorge- und Geschlechtergerechtigkeit," in Arbeits- und Lebens-

welten im sozial-ökologischen Umbau, eds. H. Eichmann, U. Filipič, S. Nowak, and S. Pirklbauer (Wien: AK Wien).

- Smetschka, B., Gaube, V., and Mader, K. (2023a). Time to care— Care for time—How spending more time for care than consumption helps to mitigate climate change. *Frontiers in Sustainability* 4. <u>https://doi.org/10.3389/frsus.2023.1070253</u>
- Smetschka, B., Hofbauer, J., Penker, M., Jany, A., Frey, H., and Wiedenhofer, D. (2023b). "Kapitel 3. Überblick Handlungsfelder," in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 217–226. https://doi.org/10.1007/978-3-662-66497-1\_7
- Smetschka, B., Mader, K., Simsa, R., and Wiedenhofer, D. (2023c). "Kapitel 8. Sorgearbeit für die eigene Person, Haushalt, Familie und Gesellschaft," in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 309–327. <u>https://doi.org/10.1007/978-3-662-66497-1\_12</u>
- Smetschka, B., Wiedenhofer, D., Egger, C., Haselsteiner, E., Moran, D., and Gaube, V. (2019). Time Matters: The Carbon Footprint of Everyday Activities in Austria. *Ecological Economics* 164, 106357. <u>https://doi.org/10.1016/j.ecolecon.2019.106357</u>
- Smith, E., Morris, J., Kheshgi, H., Teletzke, G., Herzog, H., and Paltsev, S. (2021). The cost of CO2 transport and storage in global integrated assessment modeling. *International Journal of Greenhouse Gas Control* 109, 103367. <u>https://doi.org/10.1016/j. ijggc.2021.103367</u>
- Smith, P. (2016). Soil carbon sequestration and biochar as negative emission technologies. *Global Change Biology* 22, 1315–1324. <u>https://doi.org/10.1111/gcb.13178</u>
- Smith, S., Geden, O., Nemet, G. F., Gidden, M., Lamb, W. F., Powis, C., et al. (2023). State of Carbon Dioxide Removal – 1<sup>st</sup> Edition. <u>https://doi.org/10.17605/OSF.IO/W3B4Z</u>
- SOC/747-EESC-2022-01-01 (2023). Green collective bargaining. European Economic and Social Committee. Available at: <u>https://webapi2016.eesc.europa.eu/v1/documents/EESC-2022-06018-00-02-AC-TRA-EN.docx/content</u> (Accessed May 6, 2024).
- Sonter, L. J., Maron, M., Bull, J. W., Giljum, S., Luckeneder, S., Maus, V., et al. (2023). How to fuel an energy transition with ecologically responsible mining. *Proceedings of the National Academy of Sciences* 120, e2307006120. <u>https://doi.org/10.1073/</u> pnas.2307006120
- Spash, C. L. (2024). Foundations of social ecological economics: the fight for revolutionary change in economic thought. Manchester: Manchester University Press.
- Specht-Prebanda, M. (2020). ISW-Betriebsrätebefragung 2019: Betriebsklima und Unternehmenskultur. Linz: ISW – Institut für Sozial- und Wirtschaftswissenschaften. Available at: <u>https:// www.zeitschriftwiso.at/fileadmin/user\_upload/HP\_ISW-Betriebsraetebefragung\_2019.pdf</u>
- Sposato, R. G., and Hampl, N. (2018). Worldviews as predictors of wind and solar energy support in Austria: Bridging social acceptance and risk perception research. *Energy Research & Social Science* 42, 237–246. <u>https://doi.org/10.1016/j.erss.2018.03.012</u>
- Staehle, C., Mayer, M., Kirchsteiger, B., Klaus, V., Kult-Herdin, J., Schmidt, C., et al. (2022). Quantifying changes in ambient NOx, O3 and PM10 concentrations in Austria during the COVID-19 related lockdown in spring 2020. *Air Qual Atmos Health* 15, 1993–2007. <u>https://doi.org/10.1007/s11869-022-01232-w</u>

- Stanton, M. C. B., Dessai, S., and Paavola, J. (2016). A systematic review of the impacts of climate variability and change on electricity systems in Europe. *Energy* 109, 1148–1159. <u>https://doi.org/10.1016/j.energy.2016.05.015</u>
- Statistik Austria (2022). Ankünfte & Nächtigungen. Available at: <u>https://www.statistik.at/statistiken/tourismus-und-verkehr/</u> <u>tourismus/beherbergung/ankuenfte-naechtigungen</u> (Accessed September 14, 2022).
- Statistik Austria (2023a). Hauptaggregate der VGR: Bruttowertschöpfung nach ÖNACE-Abteilungen (Q). Vienna, Austria. Available at: <u>https://www.statistik.at/statistiken/volkswirtschaft-und-oeffentliche-finanzen/volkswirtschaftliche-gesamtrechnungen/bruttoinlandsprodukt-und-hauptaggregate</u>
- Statistik Austria (2023b). Hauptaggregate der VGR: Erwerbstätigkeit (Personen, Inlandskonzept) nach ESVG 2010, nach ÖNACE-Abteilungen (Q). Vienna, Austria. Available at: <u>https://</u> <u>www.statistik.at/statistiken/volkswirtschaft-und-oeffentlichefinanzen/volkswirtschaftliche-gesamtrechnungen/bruttoinlandsprodukt-und-hauptaggregate</u>
- Statistik Austria (2023c). Luftemissionsrechnung. Luftemissionsrechnung ab 2008. Vienna, Austria. Available at: <u>https://www. statistik.at/statistiken/energie-und-umwelt/umwelt/luftemissionsrechnung</u>
- Statistik Austria (2023d). Materialflussrechnung. Available at: <u>https://www.statistik.at/statistiken/energie-und-umwelt/um-welt/materialflussrechnung</u> (Accessed May 7, 2024).
- Statistik Austria (2023e). Nutzenergieanalyse. Nutzenergieanalyse ab 2005. Vienna, Austria. Available at: <u>https://www.statistik.at/</u> <u>statistiken/energie-und-umwelt/energie/nutzenergieanalyse</u>
- Statistik Austria (2023f). Physische Energieflussrechnung. Physikalische Verwendungstabelle der Energieträgerflüsse in Terajoule. Vienna, Austria. Available at: <u>https://www.statistik.at/ statistiken/energie-und-umwelt/energie/physische-energieflussrechnungen</u>
- Statistik Austria (2023g). Umweltgesamtrechnungen Modul Umweltorientierte Produktion und Dienstleistung (EGSS) 2021. Projektbericht. Wien: Statistik Austria. Available at: <u>https://www. statistik.at/fileadmin/user\_upload/EGSS\_2021\_20230511.pdf</u> (Accessed October 2, 2023).
- Statistik Austria (2024a). Energiebilanzen. Available at: <u>https://</u> <u>www.statistik.at/statistiken/energie-und-umwelt/energie/ener-</u> <u>giebilanzen</u> (Accessed September 14, 2023).
- Statistik Austria (2024b). Zeitverwendung 2021/22 Ergebnisse der Zeitverwendungserhebung. Vienna: Statistik Austria. Available at: <u>https://www.statistik.at/fileadmin/user\_upload/</u> ZVE\_2021-22\_barrierefrei.pdf
- Steiger, R. (2010). The impact of climate change on ski season length and snowmaking requirements in Tyrol, Austria. *Climate Research* 43, 251–262. <u>https://doi.org/10.3354/cr00941</u>
- Steiger, R. (2011). The impact of snow scarcity on ski tourism: an analysis of the record warm season 2006/2007 in Tyrol (Austria). *Tourism Review* 66, 4–13. <u>https://doi.org/10.1108/</u> 16605371111175285
- Steiger, R., Posch, E., Tappeiner, G., and Walde, J. (2020). The impact of climate change on demand of ski tourism a simulation study based on stated preferences. *Ecological Economics* 170, 106589. <u>https://doi.org/10.1016/j.ecolecon.2019.106589</u>
- Steiger, R., Pröbstl-Haider, U., Prettenthaler, F., Damm, A., Falk, M., and Neger, C. (2021). "Outdooraktivitäten und damit zusammenhängende Einrichtungen im Winter," in *Tourismus und Klimawandel*, eds. U. Pröbstl-Haider, D. Lund-Durlacher, M. Olefs,

and F. Prettenthaler (Berlin, Heidelberg: Springer Spektrum), 109–122. https://doi.org/10.1007/978-3-662-61522-5\_6

- Steiger, R., and Scott, D. (2020). Ski tourism in a warmer world: Increased adaptation and regional economic impacts in Austria. *Tourism Management* 77, 104032. <u>https://doi.org/10.1016/j. tourman.2019.104032</u>
- Steininger, K. W., Bednar-Friedl, B., Formayer, H., and König, M. (2016). Consistent economic cross-sectoral climate change impact scenario analysis: Method and application to Austria. *Climate Services* 1, 39–52. <u>https://doi.org/10.1016/j.cliser.2016.02.003</u>
- Steininger, K. W., Bednar-Friedl, B., Knittel, N., Kirchengast, G., Nabernegg, S., Williges, K., et al. (2020). Klimapolitik in Österreich: Innovationschance Coronakrise und die Kosten des Nicht-Handelns.
- Steininger, K. W., König, M., Bednar-Friedl, B., Kranzl, L., Loibl, W., and Prettenthaler, F. eds. (2015). Economic Evaluation of Climate Change Impacts. Development of a Cross-Sectoral Framework and Results for Austria. Berlin: Springer.
- Steininger, K. W., Mayer, J., Bachner, G., Duelli, S., Frei, E., Grossmann, W., et al. (2021). The economic effects of achieving the 2030 EU climate targets in the context of the Corona crisis: An Austrian perspective. Graz: Wegener Center Verlag. <u>https://doi. org/10.13140/RG.2.2.20973.51685</u>
- Sullivan, O., and Gershuny, J. (2018). Speed-Up Society? Evidence from the UK 2000 and 2015 Time Use Diary Surveys. Sociology 52, 20–38. <u>https://doi.org/10.1177/0038038517712914</u>
- Sun, Y.-Y., Gössling, S., Hem, L. E., Iversen, N. M., Walnum, H. J., Scott, D., et al. (2022). Can Norway become a net-zero economy under scenarios of tourism growth? *Journal of Cleaner Production* 363, 132414. <u>https://doi.org/10.1016/j.jclepro.2022.132414</u>
- Suna, D., Totschnig, G., Schöniger, F., Resch, G., Spreitzhofer, J., and Esterl, T. (2022). Assessment of flexibility needs and options for a 100% renewable electricity system by 2030 in Austria. Smart Energy 6, 100077. https://doi.org/10.1016/j.segy.2022.100077
- Svanda, N., and Zech, S. (2023). "Kapitel 19. Raumplanung," in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 529– 546. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_23</u> (Accessed May 9, 2024).
- Termeer, C. J. A. M., and Dewulf, A. (2019). A small wins framework to overcome the evaluation paradox of governing wicked problems. *Policy and Society* 38, 298–314. <u>https://doi.org/10.108</u> 0/14494035.2018.1497933
- Theine, H., Regen, L., Perez Delgado, V. D., Reitan, C., and Novy, A. (2023). "Kapitel 20: Mediendiskurse und -strukturen," in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. A. Posch, K. W. Steininger, E. Aigner, A. Posch, K. W. Steininger, and E. Aigner (Berlin, Heidelberg), 413–436. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_18</u>
- Thollander, P., Kimura, O., Wakabayashi, M., and Rohdin, P. (2015). A review of industrial energy and climate policies in Japan and Sweden with emphasis towards SMEs. *Renewable and Sustainable Energy Reviews* 50, 504–512. <u>https://doi.org/10.1016/j. rser.2015.04.102</u>
- Tisserant, A., and Cherubini, F. (2019). Potentials, Limitations, Co-Benefits, and Trade-Offs of Biochar Applications to Soils for Climate Change Mitigation. Land 8, 179. <u>https://doi.org/10. 3390/land8120179</u>

- Tomschy, R., Herry, M., Sammer, G., Klementschitz, R., Riegler, S., Follmer, G., et al. (2016). Österreich unterwegs 2013/2014. Ergebnisbericht zur österreichweiten Mobilitätserhebung "Österreich unterwegs 2013/2014". Vienna: Bundesministerium für Verkehr, Innovation und Technologie. Available at: <u>https://www.bmk.gv.at/dam/jcr:fbe20298-a4cf-46d9-bbee-01ad771a7fda/oeu\_2013-2014\_Ergebnisbericht.pdf</u> (Accessed February 6, 2025).
- Totschnig, G., Hirner, R., Müller, A., Kranzl, L., Hummel, M., Nachtnebel, H.-P., et al. (2017). Climate change impact and resilience in the electricity sector: The example of Austria and Germany. *Energy Policy* 103, 238–248. <u>https://doi.org/10.1016/j.enpol.2017.01.019</u>
- Trummer, P., Ammerer, G., and Scherz, M. (2022). Sustainable Consumption and Production in the Extraction and Processing of Raw Materials—Measures Sets for Achieving SDG Target 12.2. Sustainability 14, 10971. <u>https://doi.org/10.3390/su</u> <u>141710971</u>
- TUDCN (2019). The Contribution of Social Dialogue to the 2030 Agenda: Promoting a Just Transition towards sustainable economies and societies for all. Brussels: TUDCN – Trade Union Development Cooperation Network. Available at: <u>https://www. ituc-csi.org/social-dialogue-for-sdgs-promoting-just-transition</u>
- Turkovska, O., Gruber, K., Klingler, M., Klöckl, C., Ramirez Camargo, L., Regner, P., et al. (2024). Methodological and reporting inconsistencies in land-use requirements misguide future renewable energy planning. *One Earth* 7, 1741–1759. <u>https://doi.org/10.1016/j.oneear.2024.09.010</u>
- TWI2050 The World in 2050 (2018). Transformations to Achieve the Sustainable Development Goals. Report Prepared by The World in 2050 Initiative. Laxenburg, Austria: International Institute for Applied Systems Analysis (IIASA). Available at: <u>https://pure.iiasa.ac.at/id/eprint/15347/1/TWI2050\_Report</u> 081118-web-new.pdf
- Ueckerdt, F., Bauer, C., Dirnaichner, A., Everall, J., Sacchi, R., and Luderer, G. (2021). Potential and risks of hydrogen-based e-fuels in climate change mitigation. *Nat. Clim. Chang.* 11, 384–393. <u>https://doi.org/10.1038/s41558-021-01032-7</u>
- Umweltbundesamt (2018). Treibhausgas-Bilanz unterschiedlicher Urlaubstypen. Available at: <u>https://www.umweltbundesamt.at/</u> <u>aktuelles/presse/news2018/news20181213</u>
- Umweltbundesamt (2023a). Austria's Annual Greenhouse Gas Inventory 1990–2021: Submission under Regulation (EU) no 2018/1999. Vienna: Umweltbundesamt. Available at: <u>https:// www.umweltbundesamt.at/fileadmin/site/publikationen/rep</u> 0841.pdf
- Umweltbundesamt (2023b). Projektionsbericht Treibhausgas-Emissionen (WEM) 2023. Available at: <u>https://reportnet.</u> <u>europa.eu/public/dataflow/890</u> (Accessed September 14, 2023).
- Umweltbundesamt (2024a). Air information current ozone levels. Available at: <u>https://www.umweltbundesamt.at/en/services/</u> <u>services-en/ozon-levels</u> (Accessed June 7, 2024).
- Umweltbundesamt (2024b). Nahzeitprognose der Österreichischen Treibhausgas-Emissionen für das Jahr 2023. Wien: Umweltbundesamt. Available at: <u>https://www.umweltbundesamt.at/fil-</u> eadmin/site/publikationen/rep0918.pdf
- Umweltbundesamt (2024c). Treibhausgas-Emissionen nach CRF. Österreichische Emissionen von Treibhausgasen nach CRT-Sektoren (UNFCCC). Available at: <u>https://www.data.gv.at/katalog/</u> en/dataset/78bd7b69-c1a7-456b-8698-fac3b24f7aa5

- Umweltbundesamt (2025). Austria's Annual Greenhouse Gas Inventory 1990–2023. Submission under Regulation (EU) No 2018/1999. Vienna: Umweltbundesamt. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/publikationen/rep0952.pdf</u> (Accessed June 2, 2025).
- UNDP ed. (2015). Bericht über die menschliche Entwicklung 2015. Arbeit und menschliche Entwicklung., Deutsche Ausgabe. Berlin: Berliner Wissenschafts-Verlag GmbH.
- UNEP (2024). Global Resources Outlook 2024: Bend the trend. Pathways to a liveable planet as resource use spikes. Available at: <u>https://www.unep.org/resources/Global-Resource-Outlook-2024</u> (Accessed April 15, 2024).
- UNEP IRP (2024). Global Resources Outlook 2024: Bend the trend. Pathways to a Liveable Planet as Resource Use Spikes. United Nations Environment Programme. Available at: <u>https://wedocs.unep.org/20.500.11822/44901</u>
- Unterberger, C., Hudson, P., Wouter Botzen, W. J., Schroeer, K., and Steininger, K. W. (2019). Future Public Sector Flood Risk and Risk Sharing Arrangements: An Assessment for Austria. *Ecological Economics* 156, 153–163. <u>https://doi.org/10.1016/j.</u> <u>ecolecon.2018.09.019</u>
- Urban, H., and Steininger, K. W. (2015). "Manufacturing and Trade: Labour Productivity Losses," in *Economic Evaluation of Climate Change Impacts: Development of a Cross-Sectoral Framework and Results for Austria*, eds. K. W. Steininger, M. König, B. Bednar-Friedl, L. Kranzl, W. Loibl, and F. Prettenthaler (Cham: Springer Climate), 301–322. <u>https://doi.org/10.1007/978-3-319-12457-5\_16</u>
- Valero, A., Valero, A., Calvo, G., and Ortego, A. (2018). Material bottlenecks in the future development of green technologies. *Renewable and Sustainable Energy Reviews* 93, 178–200. <u>https:// doi.org/10.1016/j.rser.2018.05.041</u>
- van der Wiel, K., Bloomfield, H. C., Lee, R. W., Stoop, L. P., Blackport, R., Screen, J. A., et al. (2019). The influence of weather regimes on European renewable energy production and demand. *Environ. Res. Lett.* 14, 094010. <u>https://doi.org/10.1088/1748-9326/ab38d3</u>
- VCÖ (2022). Klima-Emissionen der Verkehrsmittel im Vergleich. VCÖ – Mobilität mit Zukunft. Available at: <u>https://vcoe.at/ser-vice/fragen-und-antworten/klima-emissionen-der-verkehrs-mittel-im-vergleich</u> (Accessed September 14, 2023).
- Venturi, E., Ochs, F., and Dermentzis, G. (2023). Identifying the influence of user behaviour on building energy consumption based on model-based analysis of in-situ monitoring data. *Journal of Building Engineering* 64, 105717. <u>https://doi.org/10.1016/j.jobe.2022.105717</u>
- Verpoort, P. C., Gast, L., Hofmann, A., and Ueckerdt, F. (2024). Impact of global heterogeneity of renewable energy supply on heavy industrial production and green value chains. *Nat Energy* 9, 491–503. <u>https://doi.org/10.1038/s41560-024-01492-z</u>
- Vetter, A. (2023). Sorgen f
  ür das Lebensnotwendige: Feministische Postwachstumsperspektiven. politische ökologie 172, 81–86.
- Victor, D. G., Geels, F. W., and Sharpe, S. (2019). Accelerating the Low Carbon Transition. The Case for Stronger, More Targeted and Coordinated International Action. Brookings Institution. Available at: <u>https://www.brookings.edu/wp-content/ uploads/2019/12/Coordinatedactionreport.pdf</u> (Accessed September 15, 2023).
- Viguié, V., Juhel, S., Ben-Ari, T., Colombert, M., Ford, J. D., Giraudet, L. G., et al. (2021). When adaptation increases energy

demand: A systematic map of the literature. *Environ. Res. Lett.* 16, 033004. <u>https://doi.org/10.1088/1748-9326/abc044</u>

- Virág, D., Wiedenhofer, D., Baumgart, A., Matej, S., Krausmann, F., Min, J., et al. (2022). How much infrastructure is required to support decent mobility for all? An exploratory assessment. *Ecological Economics* 200, 107511. <u>https://doi.org/10.1016/j. ecolecon.2022.107511</u>
- Vogel, J., and Hickel, J. (2023). Is green growth happening? An empirical analysis of achieved versus Paris-compliant CO<sub>2</sub>– GDP decoupling in high-income countries. *The Lancet Planetary Health* 7, e759–e769. <u>https://doi.org/10.1016/S2542-5196(23)00174-2</u>
- Vogel, J., Steinberger, J. K., O'Neill, D. W., Lamb, W. F., and Krishnakumar, J. (2021). Socio-economic conditions for satisfying human needs at low energy use: An international analysis of social provisioning. *Global Environmental Change* 69, 102287. <u>https://doi.org/10.1016/j.gloenvcha.2021.102287</u>
- Vona, F. (2018). Job losses and political acceptability of climate policies: why the 'job-killing' argument is so persistent and how to overturn it. *Climate Policy* 19, 524–532. <u>https://doi.org/10.1080</u> /14693062.2018.1532871
- Vona, F., Marin, G., Consoli, D., and Popp, D. (2015). Green Skills. SSRN Journal. <u>https://doi.org/10.2139/ssrn.2639099</u>
- Vona, F., Marin, G., Consoli, D., and Popp, D. (2018). Environmental Regulation and Green Skills: An Empirical Exploration. *Journal of the Association of Environmental and Resource Economists* 5, 713–753. <u>https://doi.org/10.1086/698859</u>
- Wagner, T., Themeßl, M., Schüppel, A., Gobiet, A., Stigler, H., and Birk, S. (2017). Impacts of climate change on stream flow and hydro power generation in the Alpine region. *Environ Earth Sci* 76, 4. <u>https://doi.org/10.1007/s12665-016-6318-6</u>
- Wang, F., Dreisinger, D. B., Jarvis, M., and Hitchins, T. (2018). The technology of CO2 sequestration by mineral carbonation: current status and future prospects. *Canadian Metallurgical Quarterly* 57, 46–58. <u>https://doi.org/10.1080/00084433.2017.137522</u> 1
- Wang, H., Schandl, H., Wang, X., Ma, F., Yue, Q., Wang, G., et al. (2020). Measuring progress of China's circular economy. *Resources, Conservation and Recycling* 163, 105070. <u>https://doi.org/10.1016/j.resconrec.2020.105070</u>
- Wang, P., Ryberg, M., Yang, Y., Feng, K., Kara, S., Hauschild, M., et al. (2021). Efficiency stagnation in global steel production urges joint supply- and demand-side mitigation efforts. *Nat Commun* 12, 2066. https://doi.org/10.1038/s41467-021-22245-6
- Wang, S., Hausfather, Z., Davis, S., Lloyd, J., Olson, E. B., Liebermann, L., et al. (2023). Future demand for electricity generation materials under different climate mitigation scenarios. *Joule* 7, 309–332. <u>https://doi.org/10.1016/j.joule.2023.01.001</u>
- Wechsler, T., Stähli, M., Jorde, K., Zappa, M., and Schaefli, B. (2023). The future of Alpine Run-of-River hydropower production: Climate change, environmental flow requirements, and technical production potential. *Science of The Total Environment* 890, 163934. <u>https://doi.org/10.1016/j.scitotenv.2023.163934</u>
- Wehrle, S., Gruber, K., and Schmidt, J. (2021). The cost of undisturbed landscapes. *Energy Policy* 159, 112617. <u>https://doi.org/10.1016/j.enpol.2021.112617</u>
- Weisz, U., Pichler, P.-P., Jaccard, I. S., Haas, W., Matej, S., Bachner, F., et al. (2020). Carbon emission trends and sustainability options in Austrian health care. *Resources, Conservation and Recycling* 160, 104862. <u>https://doi.org/10.1016/j.resconrec.2020.104862</u>

- Wesseling, J. H., Lechtenböhmer, S., Åhman, M., Nilsson, L. J., Worrell, E., and Coenen, L. (2017). The transition of energy intensive processing industries towards deep decarbonization: Characteristics and implications for future research. *Renewable and Sustainable Energy Reviews* 79, 1303–1313. <u>https://doi. org/10.1016/j.rser.2017.05.156</u>
- Whiting, K., Carmona, L. G., and Carrasco, A. (2022). The resource service cascade: A conceptual framework for the integration of ecosystem, energy and material services. *Environmental Development* 41, 100647. <u>https://doi.org/10.1016/j.envdev.2021.100647</u>
- Wiedenhofer, D., Fishman, T., Plank, B., Miatto, A., Lauk, C., Haas, W., et al. (2021). Prospects for a saturation of humanity's resource use? An analysis of material stocks and flows in nine world regions from 1900 to 2035. *Global Environmental Change* 71, 102410. <u>https://doi.org/10.1016/j.gloenvcha.2021.102410</u>
- Wiedenhofer, D., Plank, B., and Antal, M. (2023). Declining household greenhouse gas footprints in Germany: Decomposing the contributions of working time, consumption, mobility, energy efficiency and decarbonization between 2000 – 2019. <u>https:// doi.org/10.21203/rs.3.rs-3409954/v2</u>
- Wiedenhofer, D., Steinberger, J. K., Eisenmenger, N., and Haas, W. (2015). Maintenance and Expansion: Modeling Material Stocks and Flows for Residential Buildings and Transportation Networks in the EU25. J of Industrial Ecology 19, 538–551. <u>https://</u> doi.org/10.1111/jiec.12216
- Wiegand, D. (2012). "MoreSpace Organisation der Raumnutzung über die Zeit," in *Stadt: Gestalten: Festschrift für Klaus Semsroth*, (Vienna: Springer), 199–204. <u>https://doi.org/10.1007/978-</u> <u>3-7091-1057-7\_37</u>
- Wieser, H., Bachinger, K., Fluch, J., Meitz, S., Oberholzner, T., and Reiter, J. (2023). Kreislauforientierte Dienstleistungssysteme für das produzierende Gewerbe (PRO\_Service): Forschungs- und Entwicklungsbedarf im Übergang von Produkten zu integrierten Systemlösungen für die Kreislaufwirtschaft. Gleisdorf: Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie (BMK).
- Wieser, H., and Kaufmann, P. (2023). "Kapitel 14. Die Versorgung mit Gütern und Dienstleistungen," in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 413–436. Available at: https://doi.org/10.1007/978-3-662-66497-1\_18

- WIFO (2024). Temperaturkorrigierte Einsparungen des Gasverbrauchs. Available at: <u>https://energie.wifo.ac.at/analysis/gassavings</u> (Accessed May 6, 2024).
- Wilson, C., Caroline Zimm, and Arnulf Grubler (2023). "Energy Services-led Transformation," in *Handbook of Energy Transitions*, ed. K. Araújo (London: Routledge).
- Winker, G. (2021). Solidarische Care-Ökonomie. Revolutionäre Realpolitik für Care und Klima. Bielefeld: Transcript.
- Wissen, M., Pichler, M., Maneka, D., Krenmayr, N., Högelsberger, H., and Brand, U. (2020). "Zwischen Modernisierung und sozial-ökologischer Konversion. Konflikte um die Zukunft der österreichischen Automobilindustrie," in Abschied von Kohle und Auto? Sozial-ökologische Transformationskonflikte um Energie und Mobilität, eds. K. Dörre, M. Holzschuh, J. Köster, and J. Sittel (Frankfurt am Main/New York: Campus), 223–266.
- Witte, K. (2021). Social Acceptance of Carbon Capture and Storage (CCS) from Industrial Applications. *Sustainability* 13, 12278. <u>https://doi.org/10.3390/su132112278</u>
- WSA (2023a). Total production of crude steel. Available at: <u>https://worldsteel.org/steel-topics/statistics/annual-produc-</u> <u>tion-steel-data/?ind=P1\_crude\_steel\_total\_pub/AUT</u> (Accessed August 4, 2023).
- WSA (2023b). World Steel in Figures 2023. Available at: <u>https://worldsteel.org/steel-topics/statistics/world-steel-in-fig-ures-2023/</u> (Accessed August 4, 2023).
- Xu, T., Apps, J. A., and Pruess, K. (2003). Reactive geochemical transport simulation to study mineral trapping for CO 2 disposal in deep arenaceous formationsL. *J. Geophys. Res.* 108. <u>https:// doi.org/10.1029/2002JB001979</u>
- Yalew, S. G., van Vliet, M. T. H., Gernaat, D. E. H. J., Ludwig, F., Miara, A., Park, C., et al. (2020). Impacts of climate change on energy systems in global and regional scenarios. *Nat Energy* 5, 794–802. <u>https://doi.org/10.1038/s41560-020-0664-z</u>
- Zachariadis, T., and Hadjinicolaou, P. (2014). The effect of climate change on electricity needs – A case study from Mediterranean Europe. *Energy* 76, 899–910. <u>https://doi.org/10.1016/j.energy.</u> 2014.09.001
- Zhang, A., Alvi, M. F., Gong, Y., and Wang, J. X. (2022). Overcoming barriers to supply chain decarbonization: Case studies of first movers. *Resources, Conservation and Recycling* 186, 106536. <u>https://doi.org/10.1016/j.resconrec.2022.106536</u>





**Chapter 5** 

# Navigating demand-side transformations to achieve net zero through human decisions and behavior

# Second Austrian Assessment Report on Climate Change | AAR2

# Chapter 5 Navigating demand-side transformations to achieve net zero through human decisions and behavior

This publication was subject to an anonymous international peer review process guided by several review editors, the AAR2 co-chairs and the APCC Steering Committee.

The views or opinions expressed in this publication are not necessarily those of the institutions with whose support this work was carried out. The publisher and the supporting institutions do not guarantee the accuracy or continued availability of urls to external websites or third-party websites mentioned in this publication and accept no responsibility for the accuracy or appropriateness of any content contained therein.

#### **Recommended citation**

E. Blanco, I. Otto, S. Pahl, et al. (2025): Navigating demand-side transformations to achieve net zero through human decisions and behavior. In "Second Austrian Assessment Report on Climate Change (AAR2) of the Austrian Panel on Climate Change (APCC)". [D. Huppmann, M. Keiler, K. Riahi, H. Rieder (eds.)]. Austrian Academy of Sciences Press, Vienna, Austria | doi: <u>doi.org/10.1553/aar2-ch5</u>

You can download the recommended citation in the formats Endnote (.ris) and BibTex (.bib) at aar2.ccca.ac.at/citation.

Vienna, June 2025 ISBN: 978-3-7001-5110-4 | <u>austriaca.at/5110-4</u> (full report)

## Production

Publisher: Austrian Academy of Sciences Press, Vienna Typesetting: Crossdesign Werbeagentur GmbH, Graz Cover: paulaidaanton | Agentur für digiloges Design Graphic design Figures 5.3, 5.6: paulaidaanton | Agentur für digiloges Design Al-based tools such as DeepL were used to support the linguistic revision.

Contact address for product safety: Austrian Academy of Sciences Press Dr. Ignaz Seipel-Platz 2 A-1010 Wien Tel.: +43-1-51581-3420 E-Mail: verlag@oeaw.ac.at

Funding acknowledgement

The AAR2 was funded by the Climate and Energy Fund under the 14th Austrian Climate Research Program (ACRP).

Copyright and license © 2025 Austrian Panel on Climate Change (APCC) This report is published under a Creative Commons CC-BY 4.0-License. creativecommons.org/licenses/by/4.0





The AAR2 is a report by the Austrian Panel on Climate Change (APCC). https://aar2.ccca.ac.at/en

<b>Coordinating Lead Authors</b>			
Esther Blanco	University of Innsbruck   UIBK		
llona M. Otto	University of Graz		
Sabine Pahl	University of Vienna		
Lead Authors			
Benigna Boza-Kiss	International Institute for Applied Systems Analysis   IIASA		
Thomas Brudermann	University of Graz		
Katharina Gangl	Institute for Advanced Studies   IHS		
Elisabeth Gsottbauer	University of Innsbruck   UIBK		
Michael Kirchler	University of Innsbruck   UIBK		
Michael Kuhn	International Institute for Applied Systems Analysis   IIASA		
Stefan Nabernegg	University of Graz		
Johannes Reichl	Johannes Kepler University Linz & Energy Institute at Johannes Kepler		
	University Linz		
Ulrich Remus	University of Innsbruck   UIBK		
Contributing Authors			
Franz L. Schönburg	Energy Institute at Johannes Kepler University Linz		
Rene Schwaiger	University of Innsbruck   UIBK		
Review Editors			
Stefan Pauliuk	University of Freiburg		
Bas van Ruijven	International Institute for Applied Systems Analysis   IIASA		
Chapter Scientist			
Stefan Nabernegg	University of Graz		
Linicons from AAD2 Tashnisal Support Unit			
Maria Holzingor	International Institute for Applied Systems Applysis 1985		
	International Institute for Applied Systems Analysis   IIASA		
verena kauchenwalū	international institute for Applied Systems Analysis   IIASA		

# **TABLE OF CONTENTS**

Exec	Executive summary 31		
5.1.	Chapt	er introduction	314
5.2.	. Well-being and inequality related to climate mitigation and adaptation		
	5 2 1	Welfare well-being and decent living	317
	J.Z.1.	Well being and welfare perspectives and	517
		measures	317
		Impact of well-being improvement on climate change	319
		Impact of climate change on well-being .	320
		Trade-offs, synergies, and co-benefits related to the energy and mobility	
		transition	321
	5.2.2.	Socio-economic implications for	
		emission footprints	322
5.3.	Mitiga	ation and adaptation decisions of	
citizens and households: Lifestyles and			
	consu	mption	325
	5.3.1.	Human behavior and social norms	325
		Behavioral biases and decision heuristics	
		influence sustainable behavior	325
		Behavioral change and habit formation – suggestions and pathways	327
	5.3.2.	Collective action	328
		Socio-economic and socio-psychological	
		background of participating individuals .	329
		Context and information	329
		Objectives and constraints	330
		Design of institutions for sustained	
		collective action	331
		Social feedback	331
	5.3.3.	Interaction with technology and	
		infrastructure	332
	5.3.4.	Social inequalities in energy supply and	
		related policies	333
		Energy poverty and vulnerability factors.	334

	Regulatory framework, measures, and their impact	335
	Initiatives of the private sector to tackle energy poverty in Austria	336
	Identification of priorities to combat	
	energy poverty	337
5.4. Orga	anizations as agents of change	337
5.4.1	. Firms as agents of change	338
5.4.2	<ol> <li>Higher education and research, NGOs, and civil society</li> </ol>	340
5.4.3	8. Energy communities	341
	Primary objectives and benefits	342
	Status of energy communities in Austria	342
5.5. Circular and sharing economy and the role of digitalization 343		
5.5.1	Drivers, barriers, and measures to foster the circular economy and sharing economy	344
5.5.2	<ol> <li>Digitalization in a circular and sharing economy</li> </ol>	347
5.5.3	<ol> <li>Risks associated with circular and sharing economy, in particular digital rebound</li> </ol>	349
5.6. Dist	ributional justice, social tipping points	
and	social acceptance	351
5.6.1	. Distributional justice	352
5.6.2	2. Social tipping points	353
5.6.3	B. Social acceptability of decarbonization	354
	Engaging the public	356
Chaj Beha	pter Box 5.1   avioral aspects of responses to climate	
risks	5	357
Referenc	es	359
Glossary.	<u>aar2.ccca.ac.at/glos</u>	sary
List of abl	breviations <u>aar2.ccca.ac.at/abbreviati</u>	ions
Appendix Full report	taar2.ccca.ac.at/append	IX/5
runiepul	aaiz.ccca.ac.at/16	501

#### **EXECUTIVE SUMMARY**

There has been strong societal climate concern and demand for climate action in Austria (high confidence) {5.6.3}. The social acceptability of policies strongly determines effective implementation (high confidence) {5.6.3}. Accompanying measures to alleviate fears of negative effects (e.g., 'Klimabonus' to alleviate fears of loss in purchasing power) as well as genuine societal engagement and co-creation approaches (e.g., civil consultation council) will yield fairer and more effective measures (limited evidence, high agreement). These measures together are likely to enhance the social acceptability of climate policies and remove barriers for changing lifestyles and consumption (medium evidence, high agreement) {5.1, 5.4, 5.6.3}. Similarly, enabling factors for the circular economy include defining circular solutions as the default (positive lock-in), the use of enabling policy systems, such as strong information tools (e.g., labels), phasing out high-emission technology, and social pressure (medium evidence, high agreement) {5.5}. In addition, supporting renewable energy technologies, energy efficiency, carbon pricing and similar policy interventions designed to tackle climate change risks have uneven societal impacts. This raises the challenge of providing structural, long-term approaches that prevent any exacerbation of existing social inequalities and promote a fair and inclusive energy transition instead (medium evidence, high agreement) {5.3.4}.

Demands arising from lifestyles and consumption are hard to change due to existing infrastructure (e.g., some locations lack railways or safe bike lanes) {5.3.3}, economic disincentives (e.g., subsidies for high-emissions products and services), and social norms (e.g., energy-intensive goods such as big cars or big living spaces socially rewarded as status goods) (high confidence) {5.3.1, 5.3.2}. Policy and infrastructure changes as well as cultural, behavioral and normative changes can increase the rate of emission reductions (high confidence) and can also facilitate adaptation to climate change impacts (high confidence). Austrian consumption-based emissions are about 50 % higher than production-based emissions, with production-based emissions varying strongly across regions and consumption-based emissions across income (high confidence). Production-based emissions are lower in Austrian urban areas than in rural areas. However, whether urbanization leads to higher or lower consumption-based emissions depends on how cities are designed and the availability of climate-friendly consumption options  $\{5.2.2\}$ .

Austrian households in the highest income decile cause substantially larger emissions than others (*medium evidence, high agreement*) {5.2.2}. The top 20 % of Austrian households emit more than the bottom 50 %. Energy poverty vulnerability is particularly important and intertwined with a range of social factors, including unemployment, gender inequality, educational attainments, ethnicity, housing status, disabilities, and age. Notably, women are particularly vulnerable, largely due to factors such as the gender pay gap, their lower participation in the labor market, and the uneven distribution of caregiving responsibilities within families (*high confidence*) {5.3.4}.

Firms, civil society organizations, and education organizations can foster behavioral change towards net-zero lifestyles by providing leadership and implementing change within their organizations (medium evidence, high agreement) {5.4}. For example, collective action can constitute an important bottom-up component of both the energy transition (e.g., through the formation of energy or sharing communities) and adaptation efforts at the (local) community level. Collective action schemes have been demonstrated as being highly effective and stimulate wider and longer-term actions when certain identifiable pre-conditions are met for participation, organization and its embeddedness in broader policies. Crucial aspects include instilling a sense of community, empowerment, and trust that go beyond private gains and includes a broad range of participants; embedding collective action in complementary policies; and avoiding initiatives that are perceived as elitist and, therefore, create polarization and climate backlash in broader society {5.3.2}.

Behavioral, lifestyle and consumption changes related to emissions not regulated under the EU ETS will likely entail stronger EU-wide reductions of emissions in the short term than changes related to emissions covered by the EU ETS (*high confidence*). Thus, focusing on goals such as reducing the use of fossil fuels in home heating systems, making dietary changes, altering land use, etc. has a stronger short-term potential for emission reduction than tackling topics regulated by the ETS (e.g., switching to green energy providers, reducing flights within the EU) (*high confidence*) {5.1}. Once the EU ETS 2 becomes binding, equivalent comments can be made for emissions under that second carbon market within the EU. There are already numerous good examples of circular economy practices by citizens and businesses alike (high confidence) {5.5}. These are framed by the adopted government strategy (BMK, 2022) based on international circularity concepts but are often not evaluated based on their effectiveness in reducing emissions (medium evidence, high agreement). Firms can be agents of change to support the net-zero transition: Austrian firms are already investing substantial funds to adapt to the negative effects of climate change and are making efforts to decouple their productivity from emissions {5.4.1}. Green-washing and other forms of information asymmetries undermine consumer efforts to shift consumption patterns in accordance with their proenvironmental preferences. Measures aimed at improving information for consumers and reducing exploitative practices by firms could enhance the share of climate-friendly products in the economy {5.4.1}. Circular Economy approaches promote change at the systems level, where consumers have a role to play, but where they can only achieve major results if regulations, material and ICT infrastructures support them (high confidence) {5.5}. Similarly, implementing circular economy solutions requires strong business solutions that drive this change (limited evidence, high agreement).

The energy requirement to provide the so called 'decent living standards' (DLS) for all citizens in Austria for all critical dimensions implies an average energy demand per person per year of 32 GJ (*medium confidence*) {5.2}. These standards include shelter of proper quality and temperature, sufficient mobility, the right quality and quantity of food, education, participation, and health services. This value falls on the upper end of the international range (9–36 GJ per person per year) and is equivalent to 15–30 % of the current total domestic energy consumption {5.2.1}. Austria has the leverage to reduce GHG emissions by adopting technology and reducing inequality (*high confidence*) {5.2.2, 5.3.4}, while maintaining service levels and reducing the emission footprint {5.2.2}. Policies, regulations, and technology standards targeted to lower rebound effects from technology adoption and promote digitalization to support the transition to the circular economy could enable stronger CO<sub>2</sub> mitigation objectives {5.5} (*high confidence*).

Human responses to climate risks can amplify or reduce the expected impacts (*medium evidence, high agreement*) {Chapter Box 5.1}. Disaster and crisis management plans, management plans for shortages of energy and natural resources, or extreme weather events can limit the expected impacts (*limited evidence, medium agreement*). Communication and compliance strategies of management plans must integrate behavior science principles to be effective (e.g., defaults for emergency notifications, support from role-models in society) (*high confidence*). These could build on the existing risk management structures (e.g., training courses as part of work safety on climate impact responses, local evacuation plans, early warning systems).

Experts believe that the net-zero emissions system can be achieved in Austria by mid-century, because we have the necessary technology, resources, knowledge, and skills (*limited evidence, medium agreement*) {5.6.2}. Regulations, technology standards, expansion of infrastructure supporting low-carbon lifestyles, incentives, and, in some cases, removing disincentives and including sanctions are needed to speed up the necessary changes and ensure a just and fair transformation process. Sharing knowledge about business and civil society forerunners, best-practice examples, and co-benefits for citizens can increase the public support for the process.

## 5.1. Chapter introduction

### Addressing the challenge

Limiting global warming requires substantial transitions (Otto et al., 2020a) with rapid and deep emission reductions in all sectors to reach net-zero greenhouse gas (GHG) emissions by mid-century (IPCC, 2018; Doblas-Reves et al., 2021). Austrian as well as global GHG emissions must decrease at rates of over 7 % per year. Responding to the challenges ahead of us requires a fundamental transformation in human behavior and lifestyles, infrastructure, energy supply, agriculture, and land use. To make low-carbon lifestyles attractive and preferred options for citizens, as well as to ensure the long-term effect of such changes, comprehensive structural changes are urgently needed in all areas of life (Aigner et al., 2023b). These cannot happen, however, without the support of society, individuals, communities, and different stakeholder groups. Therefore, different levels of analysis must be considered, ranging from those of individuals and social groupings to those of legal and governance frameworks, as well as the interactions between them. The assessment in this chapter places a focus on how to trigger social, demand-side change to create net-zero societies and addresses two main conceptual approaches.

We focus on the decision-making processes behind current choices and investigate the human dimension of the climate and energy system. As a complement to Chapters 1, 2, 3 and 4, which mainly address aggregate policy impacts and projections, we focus on the human decisions that generate such aggregate impacts. The need to satisfy people's demands for safe and comfortable shelter, mobility that allows them to work and participate, consumer goods and reliable lighting services, as well as access to essential public services, underlies the societal demand for materials and energy. These services, in turn, also satisfy higher human needs, such as for autonomy, mastery, and belonging. The levels and types of consumption determine the demand for resources, imply a provisioning system, and ultimately impact the climate and other environmental systems. As described in Cross-Chapter Box 3, technological and social innovations that lead to an improvement in the efficiency of service provision in response to Austrian demand can have multiplicative and cascading impacts. Chapter 4 describes the provisioning systems and solutions that have been developed to meet the demand for services and amenities, while Chapter 3 reviews the housing and mobility-related service, material, and energy demands. These, in turn, are driven by the human needs, lifestyles, and behaviors in society assessed in this chapter.

Second, climate change is generated by people; thus, the drive to reduce emissions and address adaptation to climate change impacts needs to come from people as well. People have different roles in societies, as citizens, but also as parts of households, larger communities, firms, governments, or NGOs (Creutzig et al., 2022b). Some individuals have very high agency as community leaders, opinion leaders in traditional or social media, or business and industry managers (Otto et al., 2020b; Hampton and Whitmarsh, 2023). All these entities are shaped by people and their individual or collective constructs (e.g., identities, values, norms). In this chapter, we cover the elements influencing decisions by households to decarbonize (Section 5.3) and the actors organized in different forms of firms (Section 5.4.1), organizations in higher education and research, NGOs, and activist groups (Section 5.4.2) or energy communities (Section 5.4.3). Governmental organizations have such idiosyncratic relevance that they are covered separately in Chapter 6. The order of these sub-chapters does not reflect a prioritized need to change regarding different actors or groupings. In Section 5.5, we put the different actors from 5.2, 5.3, and 5.4 together under an overarching conceptualization of the combined circular economy and sharing economy. We assess emission implications of inequality and wealth in society (Section 5.2.2), and finally we review socio-psychological factors associated with policy support, areas for social tipping and the social acceptability of decarbonization measures (Section 5.6).

Two approaches in particular shape our overall conceptual perspective, the integrative COM-B Model and tipping points. Some commentators contrast 'behavior change' and 'system change' and present a trade-off between these two perspectives on change. In this chapter, behavior change and system change present two complementary perspectives (United Nations Environment Programme, 2020). Research results highlight the interactions and feedback loops between individual choices and changes in the social structure, technology, and infrastructure. Taking an overarching perspective allows synergies to occur between top-down and bottom-up processes. We do not take an either-or perspective; instead, we posit that change happens at all levels of society from that of the individual to social to societal, using the framing of the COM-B model and social tipping points to describe these processes.
### Capabilities, opportunities and motivation for behavior change (COM-B model)

The COM-B model describes proximal and distal factors that drive behavior (Michie et al., 2011). Proximal factors consist of (a) capabilities (C) that allow people to participate in an activity (e.g., knowledge or skills such as cycling), (b) opportunity (O) to engage in the behavior (i.e., external factors that enable the behavior such as access to a bike, safe cycle lanes), and (c) motivation (M) (i.e., automatic or reflective psychological processes that trigger behavior such as habits or values). The model also describes mechanisms of change in the form of interventions, such as education, modeling, incentivization and finally embeds this into systemic and structural change such as fiscal measures, regulation and environmental planning. We propose this model as background for the chapter because it is able to integrate analysis across different levels, from individual to societal while describing specific mechanisms of change.

#### Social tipping points

Another key framework is that of social tipping points, which describes how social norms, behaviors, and technologies can spread rapidly from a minority group to the majority of society (Tàbara et al., 2018; Farmer et al., 2019). A tipping point in both social and natural systems refers to a critical threshold in the system control parameter. Once this is exceeded by a small amount, the factors mentioned above can influence a crucial system feature of relevance. This leads to qualitative change in the system once a reference time has passed, allowing for the emergence of the effect (Lenton et al., 2008). Tipping points are related to the existence of multiple stable states. Systems can transition between these, implying that it is difficult to push the system back to the previous state once a new stable state has been achieved (Winkelmann et al., 2022). However, not all rapid changes lead to systemic change. In the literature, the term 'tipping elements' is used to refer to system components where rapid change is possible and that can push the evolution of the entire system to a new, qualitatively different state (Lenton et al., 2008). Social tipping can involve both desirable and undesirable dynamics (Stadelmann-Steffen et al., 2021). One important research question is how such rapid change dynamics can be used to navigate human societies toward the net-zero emissions system. Otto et al. (2020a) proposed that social tipping interventions could be solutions leading to rapid and deep systemic transformation. This process

involves targeting sensitive intervention points (Farmer et al., 2019) and leverage points, which indicate places where intervention in a complex system is possible and where a minor shift in one thing can produce big changes in everything else (Meadows, 1999).

In an expert elicitation process, scientists and stakeholders were asked to scope interventions in Austria that have the potential to activate a systemic tipping process, thus reducing GHG emissions at the national level. The results of that exercise are presented in Section 5.6.

## Demand-side solutions and the EU Emissions Trading System (ETS)

One point that deserves special attention in this introduction is the varied relevance of demand-side solutions to different sectors, depending on their coverage under the Emissions Trading System (ETS) in the EU. Reductions of emissions in sectors currently under the EU ETS1 will not reduce the overall cap of the emission market. It will simply reduce the price of the emissions in the market, which will be then traded and used elsewhere in the economy, either within Austria or another EU country. Thus, to create more effective, short-term impacts on aggregate EU emissions, it is more effective to address elements that do not fall under the jurisdiction of the ETS system. These elements include direct oil consumption (e.g., heating in households, public buildings or firms, and road transportation with combustion engines) and impacts of dietary changes (e.g., increasing plant-based food or reducing the consumption of beef). For the elements that do not fall under the ETS system jurisdiction, any emission reductions are translated directly to emissions that are avoided. For the elements that do fall under the ETS system jurisdiction, only the reduction in the cap defined by the EU Commission (and, therefore, the associated number of polluting permits) entails a reduction in aggregate emissions. While the EU Commission might be more inclined to reduce the cap of the ETS for lower aggregate emissions under the trading system (and thus lower permit prices), as considered in the Market Stability Reserve, estimates suggest that 1 ton of additional abatement under the ETS could lead to a net long-term emissions reduction of as little as 0.4-0.8 tons (Perino, 2018), even if large uncertainties remain (Perino, 2019; Rosendahl, 2019).

<sup>&</sup>lt;sup>1</sup> i.e., electricity and heat generation, industrial manufacturing and aviation sectors. See Chapter Box 6.1 for more details on the sectors currently covered under the ETS as well as sectors planned to be covered under the ETS2.

This has led to calls for policy mixes to achieve more ambitious progress for emissions under the ETS (Flachsland et al., 2020; van den Bergh et al., 2021). The changes made in Phase 4 of the EU ETS, introducing the Market Stability Reserve, and policies complementing emission abatements with those in the EU ETS, such as those supporting renewable energies, carbon taxes, energy efficiency measures and voluntary abatement efforts, are reducing the number of allowances issued in the long run and thus reducing total emissions (even if this is probably not on a 1:1 basis). More broadly, one could expect that having households and firms shifting to decarbonization will facilitate the political and social acceptability of more ambitious climate policies, which are discussed in further detail in Chapter 6.

#### Carbon pricing and societal fairness

At least in the short and medium term, further increases in energy prices are expected as a result of reductions in the total number of allowances planned by the EU ETS. These contribute to meeting the net-zero emission objective for 2050. These expected increases, as well as the crucial importance of fairness and people's perceptions of fairness, justify the introduction of the topic of social inequalities in energy supply in this chapter. Energy efficiency measures by households and firms are still highly relevant for climate policy. The social and political feasibility of an ambitious climate policy is challenged in a (likely) scenario involving higher energy prices (at least in a transition period), and more generally when higher prices for carbon-intensive products and services are expected (see also Chapter 4, esp. 4.5). How well households and businesses are prepared to increase their energy efficiency will determine how 'painful' higher energy prices will be for medium-term welfare and prosperity, as illustrated by the higher energy prices we experienced in 2022 due to the Ukrainian war. (In-) equalities are naturally a much broader relevant topic in the context of climate impacts, mitigation and adaptation measures, and are addressed in Section 5.2 on well-being. Some of the hesitancy to implement ambitious climate policies is related to the impacts that these could have on well-being and distributive issues, and particularly regarding impacts on disadvantaged households.

#### What can be done, starting today?

This chapter puts a special emphasis on a pragmatic view of how to foster social transformations to achieve net-zero emissions, *starting from* the current set of infrastructure (and technology), social norms, and laws. This means seriously considering the social and political acceptability of policies. Changes in social norms, infrastructures, services and regulations can 'unleash' people, as well as construct new preferences and values (Sunstein, 2019; Aigner et al., 2023b). Unleashing entails that what once was unsayable is said and what was once unthinkable is done. The emergence of new preferences and values takes us into a domain where more progress towards net zero could take place. And these preferences and values can be measured to evaluate our progress towards some of the transformative scenarios described in Chapter 8, which place Austrian society within the threshold of the new climate boundaries.

Finally, this chapter assesses the state of knowledge on how we can encourage people, firms, and other organizations to change their climate-relevant behavior, moving beyond the more conventional policy prescriptions of taxes and command-and-control. It thus describes an accompanying set of policies and broader measures that can be developed to reduce climate impacts and prepare people for less painful decarbonization processes, but still protect both people's right to free choice and well-being, maximizing social fairness and co-benefits.

### 5.2. Well-being and inequality related to climate mitigation and adaptation

Universalizing climate-friendly living requires reassessing the way local, regional and global systems operate for society, the economy and the natural environment. In line with the novel chapter of 'Demand, services and social aspects' in the IPCC AR6 (Creutzig et al., 2022b), we use the well-being and provisioning perspective to frame the human-side of climate mitigation and adaptation.

Well-being for all and reduced inequality within planetary boundaries is attainable through implementing a system-wide demand-side transformation that also achieves climate mitigation and adaptation (Oswald et al., 2020; Creutzig et al., 2022b) (*medium confidence*). The APCC Special Report on Climate-friendly Living connected the social sciences, norms, cultures and individual behavior with technology, infrastructure and structural set-up to assess the opportunities and drivers of climate-friendly living using four perspectives that show different aspects of the dimensions of change (Aigner et al., 2023a). (1) The *Market perspective* builds on the polluter pays principle and getting prices right to achieve decarbonization through market forces (*high con*- fidence). Climate friendly choices become attractive when they become default and cheaper than the climate-harmful option. However, non-rational behavior and just climate transition are not well explained (Corvino, 2023) (robust evidence, medium agreement). (2) The Innovation perspective acknowledges the value of - mostly radical - innovation not only for technology, but for society, infrastructure, business models, among others. It also calculates with exnovation, i.e., elimination of harmful solutions (high confidence). For example, Mock (2024) explains the need for exnovation of car-based transportation to enable climate-friendly mobility system, e.g., in Austria (see Section 3.4). She explains the normative orientation underlining the SDGs, minimum thresholds (such as the decent living standards), as well as pledged targets and principles that mandate climate-friendly or socially beneficial solutions. (3) The Provisioning perspective assumes basic or fair needs to be provided with increased efficiency. Efficiency increase can originate from a multitude of sources, such as technical, organizational or behavioral improvements to provide the same or improved services as before to establish climate-friendly solutions (Grübler, 1998; Rao and Min, 2018a; Saunders et al., 2021) (high confidence). (4) The Society-Nature perspective uses knowledge on the intertwined co-existence of human and biophysical systems. Any change in one will directly affect the other.

In this section, well-being and climate-friendly living conditions are assessed mainly along the provisioning systems perspective (linking closely with the framework drawn in Chapter 4, Figure 4.1), to assess how to achieve a good and satisfied life that is within sufficiency levels without compromising well-being so that the provisioning leads to a reduction of the climate pressures.

#### 5.2.1. Welfare, well-being and decent living

Policies are expected to benefit society and thus to take into account the effects on the well-being of all (Ilmola-Sheppard et al., 2020; Kikstra et al., 2021). The traditional approach to ever increasing economic growth, productivity and higher incomes is overturned by realizing that these do not directly lead to life satisfaction (Vogel et al., 2021). Using the provisioning systems perspective, the ultimate goal of ensuring well-being and increasing equality should be achieved with levelized services of high quality. Providing these material services is linked to drastically lower energy and resource demands, the transformation of service provision will have positive environmental, social and economic consequences.

#### Well-being and welfare perspectives and measures

Well-being definitions and metrics are diverse to capture multiple facets of the concept (see Lutz et al., 2021 for an overview), of which only a few are mentioned here. Hedonic and eudaimonic approaches of well-being (VanderWeele, 2017; VanderWeele et al., 2020) relate to different ways to achieve satisfaction. In the hedonic approach, well-being is linked to positive affect (hedonic happiness)<sup>2</sup> or life satisfaction (evaluative happiness) (e.g., Diener et al., 2018; Stone and Krueger, 2018; Krekel et al., 2020); whereas the latter describes whether an individual has the capability towards self-realization and full functionality. Hedonic concepts relate to what economists would describe as utility (Jones and Klenow, 2016) - and can thus be understood to be outcome-oriented, whereas eudaimonic concepts are often based on capabilities (Nussbaum and Sen, 1993) or needs-satisfier relationships, using dimensions or standards of life provisioning, as in the decent living standards (DLS) (Doyal and Gough, 1984; Max-Neef, 1991; Gough, 2015; Rao and Min, 2018a). There is an expectation to measure well-being to show changes and relationships, e.g., with climate change or climate mitigation. The indicators and indices range from measuring objective aspects, such as the physical and social satisfiers that enable a good life, to focusing on subjective aspects, including perceived functions, feelings and an evaluation of satisfaction (subjective well-being). Establishing the bidirectional relationship between well-being and climate impacts is highly necessary. Objective elements of well-being can be linked more directly to GHG emissions by assessing the amount of energy and resources to satisfy basic, acceptable or sufficient levels of needs and wants. Through them, it is possible to assess the mitigating outcome of improving service provision. The metrices are compared in Figure 5.1, indicating Austrian relative performance compared to the most comprehensive region available (e.g., World, OECD) for each measure.

• OECD Better-life framework: 'How's Life?' (OECD, 2020) covers both inputs of material conditions (e.g., income, employment, housing) and outcomes such as health, education, work-life balance, political governance, civic engagement, social engagement, personal safety as well as subjective well-being. According to these

<sup>&</sup>lt;sup>2</sup> Note that hedonic and eudaimonic have a different meaning from how they are used in psychology.

measures Austria outperforms the average in jobs, health, environmental quality, social connections, safety and life satisfaction.

- Statistik Austria (2021): 'Wie geht's Österreich 2000– 2020' (Wegscheider-Pichler et al., 2021) is similar to the OECD Better-life framework and covers material aspects, quality-of-life aspects (including measures of subjective well-being) and environmental aspects of well-being.
- World Happiness Report (Helliwell et al., 2023): For the time frame 2020–2022, Austria scores 7.097 on a 1–10 scale, ranks 11/137 (95 % c.i. for rank 8–15) (Helliwell et al., 2023, fig. 2.1), exhibits a happiness gap (top vs. bottom half of population) of 2.653 and thus ranks 18/137 (95 % c.i. 9–33) in ascending order of the gap (Helliwell et al., 2023, fig. 2.2).
- Subjective Wellbeing Index (Ivanović et al., 2022): Based on the Subjective Wellbeing Index (SWB), Austria is experiencing a high (and highest ranking) level of well-be-

ing amongst Central European countries (~6 % above the CEE average) and has experienced an increase across both rural and urban areas between 2008 and 2018.

Years of Good Life (YoGL) (Lutz et al., 2021): The YoGL indicator combines life satisfaction and health, cognitive functioning, out of poverty aspects of well-being with age-gender-education specific life expectancy. By counting expected life years for which minimum levels of all well-being indicators are met, YoGL is a measure of remaining 'good' lifetime. Backcasting methods show that YoGL in Austria at age 20 has increased from around 34 (36) years for men (women) in 1950, to 51 (50) years in 2015 (Striessnig et al., 2021). This places Austria somewhat at the lower end of values experienced in other highly developed countries but high above the global average. Cast against the respective levels of total life expectancy, this implies that while 64 % (73 %) were spent in good life in 1950, these shares increased to 80 % (85 %) in 2015.



Figure 5.1 Overview and comparison of well-being measures and indexes. The left panel depicts the content of the well-being indexes in terms of their coverage of subjective (left) to physical, i.e., objective (right) provisioning. On the right panel the values for Austria are contrasted to the largest comparable reference region for which data is available. Note that the decent living standards dimensions are presented in physical terms here, in terms of floorspace per capita for housing, motorized passenger-km per year for personal mobility, kilocalories (kcal) for full daily food intake and for meat consumption per person per day. A: Austria, OECD: Countries of the Organization of Economic Cooperation and Development, EU: European Union, W: World, CEE: Central Eastern European countries (Source: Own compilation using data from: Rao et al., 2019; Millward-Hopkins et al., 2020; OECD, 2020; Lutz et al., 2021; Wegscheider-Pichler et al., 2021; Ivanović et al., 2022; Helliwell et al., 2024).

• The decent living standards (DLS) framework is a basic universal basket of satisfiers to ensure a minimum, but not affluent quantity and quality of nutrition, safe shelter with minimum space and thermal comfort, sufficient and in-house water for drinking and basic ablutions, improved sanitation, lighting, clean cooking fuels, cold storage, access to the internet and broadcast media, and the use of motorized transport, including public transit, access to health care and education to support both physical and social well-being (Rao and Min, 2018a; Rao et al., 2019; Millward-Hopkins et al., 2020). While DLS dimensions are a set of finite satiable human needs that are universal, the number of resources and energy to provide these needs have geographical (e.g., colder climate), cultural (e.g., local dishes), norm (e.g., transport mode), and personal (e.g., being an introvert or extrovert) aspects (high confidence). The energy needed to ensure DLS for the whole population of Austria is much smaller than the current final energy demand, between 15-30 % per capita energy demand in 2023 (Millward-Hopkins et al., 2020; Kikstra et al., 2021). This indicates that providing everyone with basic well-being is possible with significantly lower environmental impact, while the beyond-DLS emissions provide more affluent or even superfluous lifestyles (see also Section 5.2.2).

Altogether, this evidence speaks to Austria scoring high on most well-being measures and dimensions on a global measure but with some variability in 'performance' as compared to other (European) countries with similar levels of development, and sizeable socio-economic inequality in some of the measures (*high confidence*).

### Impact of well-being improvement on climate change

Technical energy efficiency (in energy per GDP) has been increasing in all global regions since at least the year 2000, as has the per capita energy intensity in most of the regions (except for Latin America and the Middle East) (Saunders et al., 2021). In Austria, energy efficiency measures have saved 1.73 Mtoe energy in the residential housing sector and 0.74 Mtoe in transport over the last 20 years (2000–2021). In spite of other drivers (e.g., numbers and sizes of homes, more appliances, and more travel), modal splits have counteracted some of these savings, leading to an overall total increase in energy demand since 2000 and rather stabilized in recent years (Odyssee-MURE, 2024) (see Section 4.2.1). To provide

a comprehensive framework for assessing the impacts of energy transitions on well-being, Köppl and Schleicher (2018), Schinko et al. (2021), and Sommer et al. (2021) developed a functionality approach relating energy services to the satisfaction of human needs (functionalities: Shelter, access to people, goods, services and places, and forms of other life support). At the same time, this approach was amenable to an economic general equilibrium analysis. Bachner et al. (2021) applied this approach in a study on the implications of a net-zero transition scenario by 2050 on the well-being of people in Austria. They found that well-being was enhanced in the scenario, although conventional economic indicators, such as the GDP, could potentially deteriorate.

Austrian citizens on average have access to well above the so-called 'basic levels' of the physical and societal dimensions of the decent level standards (Figure 5.2). As described in Section 4.2.2 material provisioning systems in Austria have the potential for sufficiency, efficiency and infrastructural transformation (see Section 5.3.3 for the latter) to provide well-being. According to a growing body of literature, human needs steadily decouple from energy or resource demands (Steinberger and Roberts, 2010). Beyond a certain energy level (a global average of 60 GJ per person or 1 tCO<sub>2</sub>eq in 2005), development indicators such as the Human Development Index and specific indicators, such as life expectancy, literacy, schooling do not increase much, i.e., reach a plateau (Preston, 2007; Steinberger and Roberts, 2010; Akizu-Gardoki et al., 2018, 2020). Although Austria has also started to decouple well-being from energy consumption mostly due to technology improvements, when considering embedded energy, the process is not steady and smaller than in most European countries (Akizu-Gardoki et al., 2018). Changes in lifestyle and sufficiency could close these gaps in Austria, which does not lead to constraints in well-being and benefits. In line with the reduction of the combined private and public floor space per person in an enhanced environmental (strong circular economy) scenario (Section 4.2.2), the WAM scenario ('With Additional Measures') of the 2020 National Building Renovation Plan of Austria (OIB, 2020) with further sufficiency measures (Alaux et al., 2024a), assumes a reduction of residential floor space from 46 m<sup>2</sup> in 2023 to 44 m<sup>2</sup> in 2050. Also in mobility, the Austrian 2030 Mobility Master Plan considers sufficiency as one of the three principles of Austria's future mobility system (BMK, 2021). The energy required to ensure DLS (see previous subchapter) in Austria means a minimum of 32 GJ final energy per person per year on average, which is at the upper end of the international range (9-36 GJ per person per year). DLS for all in Austria requires no more than 15–30 % of the current total domestic energy consumption. In terms of material footprint of DLS, the international range is 3–14 tons material flow per person per year, while direct and indirect stocks are on average 32 and 11 tons per person, respectively (Vélez-Henao and Pauliuk, 2023). Austria has the leverage to reduce GHG emissions by adopting technology and reducing inequality (*high confidence*) (see Sections 5.2.2 and 5.3.4), while maintaining service levels and reducing the emission footprint (see Sections 5.2.2).

However, there are still vertical and horizontal inequalities that may suffer from a lack of adequate provision of many of the services. The latter refers to inequalities between individuals or households within a group (discussed, for example, in Douenne (2020) and Nabernegg (2021)). Statistik Austria has shown that in 2022, households with lower incomes were exposed to above-average energy costs in their homes (Statistik Austria, 2024a, 2024b), thus at risk of not even reaching DLS in the heating dimension. Vertical inequalities are inequalities between groups, such as people of different genders or ethnicities, who share a common identity, background, circumstance. Laa et al. (2022) find that organizing and financing public transportation in rural Austria remains a challenge. Eisfeld and Seebauer (2022)



Figure 5.2 Comparison of the final energy demands per capita per year to satisfy the decent living standards (DLS) along key dimensions. Blue columns show the energy demand required to provide the minimum services of DLS for all in Austria, as opposed to the average level of current services derived from statistics and surveys, assuming the same Austrian technological and structural intensities. Intensities include operational and life-cycle energy based on public data (Source: BMK, 2016; Kikstra et al., 2021; Statistik Austria, 2021b; European Environment Agency, 2023; Central Intelligence Agency, 2024; European Commission, 2024).

identified self-restricting populations to compensate their inability to pay for appropriate heating and cooling, noting large heterogeneity in the 'decent levels' across dimensions (e.g., regionally, urban/rural divide, income levels, and other social factors). However, when considering the level of provision of a 'good life' (i.e., beyond the basic and universal decent living standards), the energy and material needs are much higher.

The provision of health, education, and other public services at higher levels is associated with the higher achievement of well-being than the provision of other functions (Vogel et al., 2021).

#### Impact of climate change on well-being

There is a broad consensus among scientists that climate change impacts (and climate policies) affect the well-being of individuals and communities (Fleurbaey, 2009; Stiglitz et al., 2018; Rao and Wilson, 2022; Dang et al., 2024) (see also Section 6.8) through extreme weather events, such as heat waves and floods, as well as through gradual changes, including those in European countries. Such impacts occur both due to material (income) losses and changes in health and, importantly, mental health. These are captured by direct measures of well-being such as life satisfaction. In the following, we assess evidence for the impacts of climate change on health, mental health, and well-being (see also Cross-Chapter Box 2); the economic impacts are assessed in Chapter 4. We emphasize the severe lack of evidence for Austria and, therefore, draw on international evidence placing focus on European countries.

Health: There is high level of agreement that climate change poses a severe threat to population health (e.g., Ebi et al., 2021; Romanello et al., 2021; Hensher, 2023), thereby greatly increasing the social costs of carbon (Bressler, 2021; Bressler et al., 2021) and the direct health care costs (Khan et al., 2016), with only limited scope for adaptation (Deschênes and Greenstone, 2011; Barreca et al., 2015, 2016; Bressler et al., 2021) (medium evidence, high agreement). Impacts occur through several channels and most notably temperature increases and extreme heat events (Gasparrini et al., 2017; Karlsson and Ziebarth, 2018; Bressler et al., 2021; Vicedo-Cabrera et al., 2021; Adélaïde et al., 2022; Masiero et al., 2022), pollution (but to a lesser extent in industrialized countries) (Fuller et al., 2022), flood risks, impacts on nutrition through higher food prices and food security (Lake et al., 2012), neonatal health (Cil and Kim, 2022; Palma et al., 2022), chronic diseases (Vandenberghe and Albrecht, 2018),

new vector-borne diseases, and violence (Otrachshenko et al., 2021) (see also Cross-Chapter Box 2); and these impacts may compound over time due to reductions in individual and institutional capacity for adapting further (Borghi and Kuhn, 2024).

Mental health: Mental health impacts of climate change can appear in the form of post-traumatic disorder and anxiety in response to losses experienced during (repeated) extreme events and as a result of the more gradual accumulation of climate anxiety or the (negative) impact of change on other socioeconomic determinants of mental health (e.g., income, access to affordable goods and services, limited uncertainty). The latter shows a strong socioeconomic gradient (Fritze et al., 2008; Hrabok et al., 2020; Lawrance et al., 2022) (high confidence). Mental health impacts on adolescent and child populations, both in response to traumatic events (Ma et al., 2022) and general climate anxiety, have been demonstrated for high-income countries (Hickman et al., 2021; Martin et al., 2022). A supportive parenting style, family and social support, and especially peer group and school support have been found to be protective, but significant variance in the context and circumstances has been observed (high con*fidence*). There is evidence that the effects are cumulative, but findings on whether these impacts occur in childhood are mixed (McBride et al., 2021; Vergunst and Berry, 2022). Further evidence shows that climate anxiety is widespread among scientists (Head and Harada, 2017).

While climate or eco-anxiety and eco-depression have been associated with lower mental well-being, these factors may also affect people's motivation to engage in pro-environmental behavior and activism, with conflicting results reported for eco-anxiety (Stanley et al., 2021; Ogunbode et al., 2022). In contrast, climate anger is positively correlated with both well-being and climate action (Stanley et al., 2021). The relationships are shaped by the context, with GDP having a positive effect on the relationship between climate anxiety – pro-environmental behavior and climate action – and personal experience, media, and social norms having more mixed effects (Ogunbode et al., 2022). The role of coping and coping styles is crucial here (Clayton, 2020). See also Cross-Chapter Box 2.

Human capital accumulation: Some evidence is available that heat exposure inhibits educational outcomes and, thus, cumulatively inhibits the accumulation of human capital as an important source of well-being in both low- and high-income countries (Graff Zivin et al., 2018, 2020; Park et al., 2020, 2021; Park, 2022). These effects can be mitigated by changing school infrastructure (air conditioning) and are disproportionately more evident in disadvantaged areas (*high confidence*).

Subjective well-being – life satisfaction and happiness: Evidence shows a positive impact of environmental amenities and a negative impact of temperature, pollution, and climate impact at aggregate (national) levels. However, impacts vary by local circumstances; for example, pollution impacts do not show a significant impact in London, while amenities do (Krekel et al., 2020, and studies reviewed therein).

Trade-offs, synergies, and co-benefits related to the energy and mobility transition

Impacts of a transition on well-being: There is a broad consensus that climate change mitigation and the ensuing life-style changes (e.g., in food or mobility) have important co-benefits at the local level, increasing well-being as measured based on objective material constituents (Creutzig et al., 2022a) or improvements in health or mental health (e.g., Chang et al., 2017; Deng et al., 2018; Gao et al., 2018; Karlsson et al., 2020) (Cross-Chapter Box 2) (high confidence). Estimates of the value of co-benefits are high. In simulation studies, these can offset large parts of mitigation costs and/or justify the use of stringent climate policies (for lower air pollution in Europe, see Schucht et al. (2015); globally, see Scovronick et al. (2019); for dietary change globally, see Springmann et al. (2016)). Many of these co-benefits can be viewed through the lens of SDGs (see Section 8.3.1). Evidence is insufficient (and, thus, of low confidence), however, for the social dimensions of well-being, personal security, social cohesion, and political stability (Creutzig et al., 2022a). A parallel large body of evidence exists that pro-environmental behavior is associated with higher subjective measures of well-being (Zawadzki et al., 2020). The positive impact on well-being is thus broadly robust to the measure and can be of high confidence. Nonetheless, transition measures at a local level, such as the introduction of wind parks, can result in lower well-being and opposition (e.g., Pohl et al., 2018, 2021). This issue calls for enhanced planning, communication, and local stakeholder involvement (e.g., Friedl and Reichl, 2016). Finally, Mock et al. (2019) provided case study evidence for 11 environmental initiatives from five European countries, including Austria, demonstrating a positive impact of collective environmental action on psychological well-being.

Energy-service perspective and role of public services: Any far-reaching energy transition will imply major restructuring of infrastructures and, in many cases, a shift from private to public services. This implies that any evaluation of transformative measures should not be carried out by comparing individual measures/policies with a given nexus of service provisioning but should instead account for a change in the whole nexus. For this purpose, it is useful to think in terms of categories of energy or transport services rather than of specific means of energy consumption or modes of transport (Sovacool, 2011; Fell, 2017; Grubler et al., 2018) (see Section 4.2). The transition of the infrastructures underlying energy or mobility services must ensure coherent and comprehensive access to these services across space, time, and system boundaries, as well as inclusive access across all strata of the population. Such systemic interventions are expected to have positive impacts on well-being. Kletzan et al. (2006) describe these impacts for mobility and heating in Austria; Mathiesen et al. (2015) for smart energy grids; Steg and Gifford (2005) and Lucas et al. (2007) for transport policy scenarios in the UK; Steininger and Bachner (2014) for commuter car-sharing in Austria; Cloutier et al. (2017) for sustainable commute in US cities; and Buehler et al. (2019) for integrated urban-rural public transport systems in Austria, Germany, and Switzerland. This initial evidence would need to be corroborated (limited evidence, high agreement).

Energy security improvement with energy demand reduction: Energy security is a top priority for Austria because of the high dependence on imported energy. Disruptions due to wars, economic conflict, climate change, and price surges can threaten the long-term and the short-term provision of energy and related critical functions. As described in Chapter 4, Bento et al. (2024) showed that demand interventions, such as building renovation and electrification of mobility are more impactful than conventional supply-side approaches, such as fuel substitution or import diversification, in Germany, Spain and other countries studied.

There is considerable complementarity in the design of service provisioning systems and infrastructures for enhancing well-being. This allows for the potentially high leverage of joint and consistent adaptations to these systems, but also heightens path dependency and underlines organizational and budgetary challenges. This is because coordination across different government domains is required as well as single, major transformations besides incremental improvements, which are both challenging (see Section 1.4). A related topic is the need for systems to be designed coherently to meet the needs of all population strata in both urban and rural communities (Pearsall et al., 2021). High income populations ensure their well-being with unproportionally large energy and material footprint, as opposed to public, easily accessible infrastructure (evidence for the United Kingdom, Baltruszewicz et al., 2023).

A broad and large-scale energy system transformation offers an opportunity to redesign public services and welfare systems in other areas than energy, for example changes in the healthcare, education, urban development, water provision systems and many others that will be affected directly or indirectly (Weisz et al., 2020 for Austria; Hensher, 2023).

#### 5.2.2. Socio-economic implications for emission footprints

Understanding the distribution of GHG emissions among different actors and entities is relevant (i) for deriving the respective emission responsibilities, potentials, and targets for emission reductions, as well as (ii) for designing policy interventions.

Austria's territorial per-capita emissions are slightly above the EU-27 average and substantially above the global average (Umweltbundesamt, 2022b). Unlike territorial (production-based) emissions, GHG footprints (consumption-based emissions) do not take into account emissions from the economic activities of a specific entity. Instead, they consider the emissions from the final demand of this entity (Steininger et al., 2016b) (see Section 4.1.1). Austria's GHG footprint is about 1.5 times larger than its territorial emissions, because the emissions embodied in imports exceed the Austrian emissions embodied in exports (Muñoz and Steininger, 2010; Giljum, 2018; Steininger et al., 2018; Nabernegg et al., 2023) (high confidence). Furthermore, national policies are available that address not only territorial emissions, but also consumption-based emissions (Steininger et al., 2018; Nabernegg et al., 2019) (limited evidence, high agreement). In terms of a global safe and just operating space for GHG emissions, Austria exceeded its fair share by a factor of 3.4 in 2015 (Fanning et al., 2022; University of Leeds, 2023). Furthermore, current emissions can be compared with a remaining Austrian GHG budget (see Section 8.2). If the Austrian emissions continue to be produced in the future as they were in 2022, the Austrian GHG budget that enables it to stay within a 1.5°C warming range without a temperature overshoot and to have a 66 % probability of reaching the target would be exhausted by mid-2025 (CCCA, 2022).

At the regional scale, territorial emissions vary considerably between Austrian federal provinces, ranging from 4.2–14.5 tCO<sub>2</sub>eq per capita in Vienna and Upper Austria, see Figure 5.3a (Umweltbundesamt, 2022a). This is mainly determined by the level of emission-intensive industrial production in the region. In contrast, GHG footprints are more evenly distributed across the provinces, as consumption patterns do not differ substantially between these regions (Muñoz et al., 2020). The most relevant differences identified by Muñoz et al. (2020) concern direct household emissions from transport and heating, which are linked to the degree of urbanization in each region (see also Chapter 3).

At the global level, the relationship between urbanization and GHG emissions depends on each country's development stage, with urban areas in low- to middle-income countries displaying higher per-capita emissions, while urban areas in high-income countries are often associated with lower per-capita emissions than rural or suburban areas (Lwasa et al., 2023). While urban areas may provide services such as housing and mobility with less energy (see Section 3.2), urbanization may increase income and consumption levels, which are associated with emissions (Jones and Kammen, 2014). Whether urbanization leads to higher or lower emissions, therefore, depends on how cities are designed and what climate-friendly consumption options are available to the citizens (Lwasa et al., 2023).

In Austria, evidence shows that cities have lower territorial per-capita emissions, for example, for Vienna and Graz compared to the Austrian or federal-province average (Pichler and Steininger, 2019; Stadt Graz, 2022; Umweltbundesamt, 2022a; Nabernegg et al., 2023). However, territorial emissions in Linz are an order of magnitude higher than the national average due to large, emission-intensive industries within the geographical city borders. Regarding GHG footprints, these patterns are less clear, as a study on Vienna indicates that this city has a lower GHG footprint than the Austrian average (Eisenmenger et al., 2022), while the footprints of Graz and Linz are estimated to be above the Austrian average (Pichler and Steininger, 2019; Stadt Graz, 2022; Nabernegg et al., 2023).

Emissions are also unequally distributed across individuals and households with different incomes. At the global level, income and emissions are strongly correlated (Bruckner et al., 2022; Chancel, 2022). Chancel (2022) estimated that "the bottom 50 % of the world population emitted 12 % of global GHG emission in 2019, whereas the top 10 % emitted 48 % of the total". Although a relatively small group of high-income individuals is responsible for a substantial share of emissions, researchers have largely focused on low-income groups when determining the impact of poverty reduction on emissions (Hubacek et al., 2017; Rao and Min, 2018b; Bruckner et al., 2022) or the impact of climate policies on low-income households (also see Section 5.3.4). Evidence for the GHG footprints of high-income individuals is largely lacking (Otto et al., 2019).



Figure 5.3 Emissions per capita across regions and income groups. Panel (a): Territorial emissions and GHG footprints per capita in 2019 across global regions, Austrian provinces, and cities in 2019 (Steininger et al., 2018; Pichler and Steininger, 2019; Stadt Graz, 2022; Umweltbundesamt, 2022a; Nabernegg et al., 2023). Panel (b): Distribution of GHG footprints per capita across Austrian income deciles and the highest percentile (Theine et al., 2022; Nabernegg et al., 2023).

For Austria, studies on the distribution of emissions across households reveal a substantial but more equal distribution of GHG footprints than global GHG footprints (see Figure 5.3b) (Greenpeace, 2020; Muñoz et al., 2020; Nabernegg, 2021; Theine et al., 2022; Nabernegg et al., 2023) (*medium evidence, high agreement*). Theine et al. (2022) estimate that the GHG footprint of the highest income decile is 4.1 times larger than that of the lowest income decile. Household GHG footprints, therefore, closely follow increasing expenditure in income deciles, with mobility expenditure being a key driver. However, emission estimates for lowand high-income groups based on survey data are subject to higher levels of under-representation and under-reporting (Flachaire et al., 2022); therefore, the inequalities might be even larger.

In a similar study on German households, Schuster and Otto (2022) showed that the lifestyle GHG emissions in the lowest and highest emission groups can differ by a magnitude of ten. Income, education, age, gender, and regional differences resulted in distinct emission profiles (Schuster and Otto, 2022). The authors analyzed lifestyle  $CO_2$  emissions in sectors including housing, transportation, and consumption. In the housing sector, the most important sources of emissions were identified as heating and electricity use. The per capita living space increased with income and most of the respondents lived on their own or in two-person households. Excessive living space per person is ineffective from an energetic perspective, but changing housing preferences through political means or incentives is difficult to achieve. There are, however, interesting examples of policies and measures that encourage the use of smaller living spaces, which could also be implemented in Austria. These include the moving bonus offered by the city of Frankfurt when moving out of a social housing flat (Wuppertal Institut für Klima, Umwelt, Energie, 2024), provided that the current flat is too large and the new flat is smaller, or eco-villages that often involve small private spaces but larger shared spaces (Wiest et al., 2022; Jany et al., 2023). Reductions in emissions in the housing sector are likely to occur as access to fossil fuel-free heating systems improves, as people switch to renewable energy sources in general, and as house insulation and energy-efficient home appliances improve (see Section 3.3).

In the transportation sector, it is noticeable that the mobility of the majority of the population is relatively low. The remainder, however, have extremely mobile lifestyles. The highest proportion of lifestyle GHG emissions from the wealthiest individuals globally is due to extreme mobility, including frequent flying with private jets and in business cabins (Otto et al., 2019). Additionally, travel and cars are still status symbols, and not only for the wealthiest members of society. The highest emissions in the transportation sector arise in the 30-49 age group (Schuster and Otto, 2022). This could be because this age group has a fixed income, its members often advance in their careers and private life (start a family, acquire larger living spaces, commute longer distances, etc.), but also because they are trying to establish themselves socially through the use of status symbols. Research points out that policies need to incorporate elements of social justice by targeting and curbing the emissions of the wealthy, since they can afford clean technologies, and their lifestyle choices have important downstream influences on the aspirations and lifestyles of people in other social classes (Otto et al., 2019). In addition, men have a higher footprint in the transport sector than women.

In the sector of consumption, which includes food and non-food products, the CO<sub>2</sub> emissions from nutrition are not group-specific. Dietary habits are not income-specific, and, in the German sample, the respondents across different regions and milieus had relatively similar eating habits (Schuster and Otto, 2022). Reducing meat consumption in different social groups could contribute the most to reducing CO<sub>2</sub> emissions. Other variables, including education and cultural background, did not influence emissions from food consumption in the German sample. Emissions due to non-food consumption were, however, again strongly correlated with income. The authors also pointed out a few gender differences in the emission patterns. To give an example, men tended to consume more meat and had therefore higher food consumption emissions, however, women had slightly higher emissions from clothing purchases than men (Schuster and Otto, 2022).

These multiple lines of evidence show that privileged members of society consume disproportionately more resources and energy to maintain their lifestyles than less privileged members. In addition, the least privileged members, who are usually less mobile and often live in marginal areas and under poorer conditions, are disproportionately more heavily affected by environmental pollution and climate extremes including for example heat waves (Brimicombe et al., 2024). The poorest social groups also often perform work that exposes them more to the impacts of climate change, including occupations in agriculture and construction. These inequalities must be taken into account. Defining a universal, irreducible, and essential set of material conditions for ensuring basic human well-being, along with indicators and quantitative thresholds that reflect local customs and preferences, can help to go beyond simply eradicating poverty and encourage the social participation of the least privileged members of society. Redistributive policy instruments, setting standards and bans on most polluting and harmful activities and technologies can help to achieve more equitable outcomes.

# 5.3. Mitigation and adaptation decisions of citizens and households: Lifestyles and consumption

#### 5.3.1. Human behavior and social norms

Changing the GHG footprint of Austrian society is crucial for delivering the transformative change needed to meet national climate objectives. Up to 62 % of the Austrian GHG footprint are related to private household consumption with the main components of mobility (20 %), residential energy use and housing (17 %), and food (9 %) (Aigner et al., 2023c). While these emissions occur within the current provisioning system, behavioral and lifestyle choices are central determinants for their reduction. Citizens' impact is even greater if the influence on governments, business, and finance are considered (Newell et al., 2021). Despite the role of government institutions for infrastructure provision, researchers in the behavioral sciences take theoretical perspectives to look at the role of social determinants, bounded rationality conditions including nudging, as well as monetary incentives and to explore fast decarbonization processes (Newell et al., 2021). These elements affect or are affected by attitudes, values, and beliefs as well as broader cognitive, affective, and social processes (O'Brien and Klein, 2017; O'Brien and Yazdani Aliabadi, 2020).

Behavioral biases and social norms play a critical role in hindering or promoting climate mitigation actions (Nielsen et al., 2021). Behavioral biases such as the tendency to maintain the status quo bias and present bias can help explain the insufficient behavioral change observed. For instance, individuals may prefer to stick to fossil fuels for transportation, rather than explore alternative modes, due to the familiarity and convenience of their current behavior. Additionally, individuals may tend to prioritize immediate gains over longterm benefits, making them less willing to invest in sustainable lifestyle changes (i.e., present bias). Moreover, individual consumption patterns and the social comparison aspect of consumption are also relevant to the limited uptake of sustainable lifestyles (Cialdini and Jacobson, 2021). Research has shown that GHG emissions are positively related to income and wealth, indicating that those with a higher socioeconomic status tend to consume more and have a greater environmental impact (Ravallion et al., 2000; Theine et al., 2022). Evidence-based policy interventions can be designed to address the barriers that impede behavioral change and support more sustainable ways of living (e.g., in line with the COM-B model). By developing targeted interventions that are tailored to meet the needs of specific groups, such as households with different socioeconomic statuses, values or motives, policymakers can increase the chance that their efforts will lead to changes in behavior (Nielsen et al., 2020a, 2020b), and even more so, because European National Energy and Climate Plans (NECPs) lack political interventions that ensure sufficiency (Zell-Ziegler et al., 2021; Lage et al., 2023). Examples include nudging techniques and behavioral interventions that guide individuals towards more sustainable choices, taxes on environmentally harmful products that provide a financial incentive for individuals to choose more sustainable options, and subsidies that encourage the adoption of sustainable technologies and behaviors.

One technical note must be addressed in this sub-chapter: It mainly touches on the literature with an international rather than an Austrian context, as only few studies with a particular Austrian context on behavioral biases and heuristics exist. Exceptions include the Austrian APCC Report on Structures for Climate-Friendly Living, that emphasizes the need for institutional and infrastructural changes supporting individual behavioral and lifestyle changes (Aigner et al., 2023b) (see also COM-B model). However, it is realistic to assume that findings from international studies (mainly conducted in Western countries) are informative for understanding the Austrian context<sup>3</sup>. Nevertheless, these circumstances clearly outline the need for more research and more randomized control trials of specific policy interventions in the Austrian context.

# Behavioral biases and decision heuristics influence sustainable behavior

Reducing global GHG emissions can be described as a global public goods problem (Barrett, 2007). While the scientif-

<sup>&</sup>lt;sup>3</sup> See, for instance, a study on loss aversion in 53 countries, showing loss aversion also among most Austrian participants (Wang et al., 2017).

ic evidence clearly indicates that it would be beneficial for everyone to cooperate on a global level across countries, individual polluters face free rider incentives, which tempt countries and individuals to take advantage of others' efforts while making no effort themselves. In this case, the free rider benefits from an expected world that has low GHG emissions, without making any effort (high agreement). As a consequence, climate mitigation actions are not undertaken at sufficiently high levels, and GHG emissions increase further (Hasson et al., 2010; Blanco et al., 2020). The 'tragedy of the commons' problem is also closely related to the problem described above, and the literature outlines the need of institutions to diversify (i.e., institutions that add to local and regional institutions by focusing on global challenges). Ostrom et al. (1999) offer evidence that indicates how to overcome some of these problems.

With respect to climate adaptation policies, however, the structure is slightly different, as this problem can be subsumed under a private good problem. In this case, the adapter (e.g., country, individual) primarily (and sometimes solely) benefits from the implemented adaptation strategy (Hasson et al., 2010; Blanco et al., 2020; McEvoy et al., 2022) (*high agreement*).

While the global GHG mitigation problem can only be solved by international cooperation and by governments and other stakeholders cooperating, the behavior of individuals and communities play an important role as well. They play a role as citizens, because they vote for certain parties (governments) that are more or less in favor of climate mitigation and adaptation policies. Importantly, a country's total carbon emissions are strongly determined by its carbon footprint (see Section 5.2.2), namely emissions that cannot be influenced directly by individual behavior. According to the Federal Ministry of Climate Action, Environment, Energy, Mobility, Innovation, and Technology, the gray footprint amounts to approximately 25 % of Austria's GHG emissions. In addition, a country's  $CO_2$  footprint is also defined by its citizens' behavior, including consumption, mobility, and the choice of renewable energy purchase (Ivanova et al., 2020; Creutzig et al., 2022b) (robust evidence).

Thus, in addition to structural factors, various behavioral biases can hinder people's willingness to adapt their behavior by preventing individuals from taking appropriate action.

First, confirmation bias is the human tendency to seek information that confirms our prior beliefs or knowledge and to disregard information that opposes our prior beliefs or knowledge (Kappes et al., 2020). This bias is often fueled by a lack of accurate knowledge or exposure to biased information, which can be disseminated through certain media outlets or political parties (Lazer et al., 2018). Confirmation bias can play a significant role in hindering sustainable behavior change, as it causes individuals to selectively seek out information that confirms their pre-existing beliefs or values, while ignoring or discounting information that challenges these. For instance, a significant number of people are not interested in knowing the extent of their carbon footprint and even admittedly ignore such information before making environmentally relevant choices (Thunström et al., 2014; Reisch et al., 2021) (*medium evidence*).

Second, present bias refers to the tendency of individuals to value the present more highly than the future. This bias can lead people to overvalue the immediate costs of climate mitigation policies, such as a carbon tax or changes in individual diets or heating systems, while placing less weight on the uncertain future benefits, such as the reduced frequency of extreme weather events (Weber, 2010). A study in Germany shows that people with present bias consume more electricity than people without such bias (Werthschulte and Löschel, 2021) (*medium evidence*).

Third, loss aversion refers to the tendency of people to strongly prefer avoiding losses compared to acquiring gains of the same magnitude (Kahneman et al., 1991). Loss aversion and the related status quo bias can hinder efforts to increase sustainable behavior (Nicolson et al., 2017; Ghesla et al., 2020). This is an additional reason that people may be reluctant to change their habits: They perceive the potential losses of altering their behavior as greater than the future benefits of behavioral change. In addition, the future environmental benefits will only materialize if a large portion of the world's population and governments take an adequate number of actions soon. This could result in individuals maintaining their carbon-intensive behaviors rather than adopting more sustainable alternatives (*robust evidence*).

Fourth, ingroup and outgroup biases in the form of the perceptual bias (Zhao and Luo, 2021) and pluralistic ignorance (Sparkman et al., 2022) might play a critical role as well. In countries or societies with strong polarization in public opinions, ingroup and outgroup biases can be particularly damaging to sustainable behavior, as individuals may become more entrenched in their beliefs and less willing to consider opposing viewpoints. For instance, Sparkman et al. (2022) investigated individuals' perceptions of national concern about the climate crisis and support for climate mitigation policies in the US. They documented that the vast majority of Americans underestimate the national support for climate mitigation policies and for climate concern, and these misguided beliefs might undermine their own action and engagement (*robust evidence*).

Fifth, social norms are collectively agreed-upon rules about how to behave in specific situations (Bicchieri, 2005), and social norms have been shown to influence a number of climate change-related behaviors (for a summary, see Cialdini and Jacobson, 2021). Examples come from littering (Cialdini et al., 1990), energy conservation (Allcott, 2011; Andor et al., 2020), and water conservation (Ferraro et al., 2011; Schultz et al., 2016). A recent meta-study of 430 primary studies and 10 meta-studies by Bergquist et al. (2023), shows that social comparison/norms next to financial incentives were the most effective tools (robust evidence). Although norm-based information is known to be effective in promoting desirable behaviors, it can also have negative consequences, such as the boomerang effect, where individuals revert to undesirable behaviors, and the backfire effect, where individuals adopt the exact opposite behavior, as evidenced by Gangl et al.'s (2022c) study in Austria where a norm-based intervention resulted in increased (rather than decreased) littering in community-building waste disposal areas.

# Behavioral change and habit formation – suggestions and pathways

Environmental behavior and behavioral change can be influenced by a variety of measures. These range from classic public policies such as taxes to information provision, and newer policy approaches, including behavioral nudges and comprehensive behavior change programs similar to those used in public health (e.g., Allcott, 2011; Spotswood, 2016; Carlsson et al., 2021). The concept of 'nudging' is central in a behavioral approach to public policy and accounts for the fact that decision-making can be influenced by subtle modifications to the decision context (via 'exploiting' behavioral biases) without reducing options or changing economic incentives (Thaler and Sunstein, 2009; Schubert, 2017; Nisa et al., 2019). There is evidence and agreement that the provision of social norm-based information - providing information on what 'most people' do or approve of - can successfully initiate behavioral change in many settings, including energy and water consumption, recycling, and the provision of public goods, to name only a few (Rege and Telle, 2004; Allcott, 2011; Abrahamse and Steg, 2013; Delmas et al., 2013; Ferraro and Price, 2013) (see also above) (robust evidence). Nudges altering the choice architecture include defaults for green energy providers (Ebeling and Lotz, 2015) and related energy choices such as defaults for heating and cooling settings (Brown et al., 2013), as well as more prominent ordering and positioning of green food alternatives (Garnett et al., 2020; Gravert and Kurz, 2021). There is evidence that choice architecture interventions can promote climate change mitigation actions, but, for the most part, only make small to medium contributions to mitigation with median effect sizes of up to 10 % (Hummel and Maedche, 2019; Nisa et al., 2019; Van Der Linden and Goldberg, 2020; Meier et al., 2022; Mertens et al., 2022). Classical paternalistic methods such as taxes, bans, and subsidies show larger effects, but acceptance among citizens is variable (Dechezleprêtre et al., 2022) (robust evidence). Systematic cost-benefit evaluations of behavioral interventions are lacking in the scientific literature, but Benartzi et al. (2017) have provided evidence for the advantages of behavioral change nudging approaches compared to traditional policy approaches.

Moreover, impacts of nudges are heterogeneous, depending on the characteristics of the person involved, other situational factors, and the persistence of effects varies (Bao and Ho, 2015; Brandon et al., 2017; Mertens et al., 2022) (medium evidence). One implication of this literature is that interventions may need to be tailored to the populations or settings of interest. Thus, it is important to pre-test interventions in the specific context. Another consideration is that the effectiveness of nudges may be temporary, with their impact potentially diminishing over time. As many environmental behaviors have a habitual component, including energy and water use, meat consumption, and commuting, it is crucial to factor in the time stability of effects in intervention design (Verplanken and Whitmarsh, 2021). Research in related domains (health behavior) indicate that monetary incentives in the form of rewards can foster good and stop bad habits (Gneezy and Rustichini, 2000; Charness and Gneezy, 2009; Royer et al., 2015; Rohde and Verbeke, 2017; Carrera et al., 2018). Such incentives, however, bear a risk of undermining intrinsic motivation (Gneezy et al., 2011; Bowles and Polanía-Reyes, 2012) (robust evidence). Results from information-based interventions and nudges on habit formation are mixed and depend on the context in which they are applied (medium evidence). Gravert and Olsson Collentine (2021) find no effect of information on sustained public transport ridership, while Ayres et al. (2013) and Byrne et al. (2021) find that information provision can overall promote shortand long-term behavior changes in energy and water use, but with modest to small contributions to resource savings. In addition, nudging techniques have been criticized recently, also due to publication bias (DellaVigna and Linos, 2022;

Maier et al., 2022). Moreover, the answer to the question of whether nudges are unethical is far from clear, with arguments both in favor and against it (Sunstein, 2015), although proponents have pointed out that the existing choice architecture may also be problematic from an ethical perspective.

At the same time, there is strong evidence that monetary incentives can also change pro-environmental behaviors and contribute to the formation of sustainable habits (Maki et al., 2016), also for mobility which is generally more difficult to change (Gravert and Olsson Collentine, 2021) (*robust evidence*). However, incentives are also debated due to concerns that they might primarily work in the short term and potentially undermine intrinsic motivation as stated above (Frey and Oberholzer-Gee, 1997). Overall, evidence suggests that, while information provision and nudges can work to a certain degree (with zero effect as well) (as outlined in DellaVigna and Linos, 2022; Maier et al., 2022), their immediate impact is smaller than that of monetary incentives (*robust evidence*). Recent research, however, suggests that combining interventions (i.e., monetary incentives with information and nudges) can help make individual interventions more effective (Drews et al., 2020; Khanna et al., 2021).

#### 5.3.2. Collective action

Individuals do not make decisions in isolated self-reference but are subject to the social structure, which is made up of (i) institutions (i.e., informal norms, customs and traditions, and formal constitutions and rules); (ii) organizations; and (iii) the technosphere (i.e., infrastructure and technology) in which they are embedded. This take is increasingly accepted across all social sciences (Williamson, 1998; Dietz et al., 2003; Axsen and Kurani, 2012; Charness and Sutter, 2012; Bamberg et al., 2018; Otto et al., 2020b). Agency, in this context understood to be the capacity to change energy and resource use patterns, lies on the spectrum between individual and collective (also see figure 1 in Otto et al., 2020b). By pooling their resources and efforts under collec-



Figure 5.4 Determinants of collective action along the formation-organization-impact line.

tive agency, individuals are able to attain outcomes that they mould not be able to achieve individually. Collective agency e often includes the build-up of longer-term organizational e or physical infrastructure and can thus be understood to be of a strategic nature. It is typically aimed at (i) enhancing and managing the provision of local public goods, i.e., club a goods or commons, such as food- or car-sharing initiatives, ioint investments in renewable energy infrastructures (energy infrastructures)

goods or commons, such as food- or car-sharing initiatives, joint investments in renewable energy infrastructures (energy communities), or the provision of shared infrastructures for adaptation (communal flood defenses, support systems in case of heat waves); and/or (ii) generating political clout.

The scope for establishing and sustaining successful collective action with transformational impact depends on a number of factors, which we group along the formation-organization-impact line for the purpose of assessment (see Figure 5.4). Domains include the socio-economic background of (possibly) participating individuals as well as the *context* and information that determine the formation; objectives and constraints and the consequential design of institutions that enable sustained organizational effectiveness; and the role of *feedback* for transformative impact. For each of these domains, enabling and inhibiting factors are identified as well as the scope for enabling policies (Figure 5.4). We restrict evidence to high-income, mostly European contexts, including the few available Austrian studies. The more limited evidence on collective action undertaken against climate transitions (e.g., wind parks or grid structures) is assessed followingly on negative feedback. The specific assessment of energy communities as well as NGOs and civil society can be found in Sections 5.4.3 and 5.4.2, respectively.

## Socio-economic and socio-psychological background of participating individuals

The willingness to engage in community action – and for many of the studies quoted, the willingness to invest financial and time resources into community renewable energy infrastructures within a number of European countries – has been shown to increase with age, income/wealth, and education (Dóci and Vasileiadou, 2015; Yildiz et al., 2015; Bauwens and Eyre, 2017; Ebers Broughel and Hampl, 2018; Curtin et al., 2019) (*high confidence*). In addition, the willingness to engage is found to vary according to the residential context (higher engagement in rural communities in Germany, but no significant impact in Austria) (Kalkbrenner and Roosen, 2016; Ebers Broughel and Hampl, 2018) (*medium confidence*) and to increase with the intensity of energy usage (Bauwens and Eyre, 2017) (*medium evidence, low agree-* ment). Pro-environmental beliefs and trust in the technical effectiveness of renewable energy solutions (or collective efficacy more generally) also play a strong role (Bamberg et al., 2015; Dóci and Vasileiadou, 2015; Kalkbrenner and Roosen, 2016; Mignon and Bergek, 2016; Ebers Broughel and Hampl, 2018; Howe et al., 2019; Reichl et al., 2021) (high confidence). Finally, lack of financial resources, financial competency, and risk aversion tend to inhibit investments in communal energy (Ebers Broughel and Hampl, 2018; Curtin et al., 2019) but not necessarily the willingness to engage in the form of in-kind contributions (Kalkbrenner and Roosen, 2016) (medium evidence, high agreement). Related evidence on food sharing initiatives shows that members are also recruited from the educated middle classes, although income and wealth play a lesser role, because most contributions are based on time and effort, factors which otherwise rely less on financial resources. Trust and the belief in convincing solutions also matters (survey findings in Nikravech et al., 2020) (high confidence).

#### Context and information

Context matters for the formation and successful implementation of collective action in terms of the scale and scope of the underlying environmental commons problem (e.g., in the context of adaptation) (Adger et al., 2005), of the broader socio-economic background into which private or collective actions are embedded (Steg, 2023), and of the policy context (Villamayor-Tomas et al., 2019).

Salience: Generally, individuals are more inclined to undertake collective action if climate change is perceived as real and salient (Bateman and O'Connor, 2016; Brink and Wamsler, 2019; Mildenberger et al., 2019; van Valkengoed et al., 2021; Steg, 2023; Wannewitz and Garschagen, 2023) (high confidence). Scholars debate, however, whether the experience of local extreme weather events is enhancing collective action to increase future protection or rather stifling it due to the perceived or real loss of resources and capacity and a sense of low self-efficacy (Adger et al., 2009; Adger, 2016; Wannewitz and Garschagen, 2023) (robust evidence, low agreement). A study from Germany based on a representative population sample, points out that the experience of seasonal temperature change is associated with stronger personal climate change concerns as well as the willingness to mitigate climate change, although with a weaker effect (Pfeifer and Otto, 2023).

Socio-economic context: With respect to the formation of collective action to promote sustainability, peer effects play a strong role. Specifically, the presence of local social capital, as embraced by community initiatives (aimed at sustainability but also at other objectives), combines environmental beliefs and beliefs in the efficacy of renewable energy schemes and other such initiatives. This combination boosts such schemes by also pulling along community members with weaker environmental beliefs (Dóci and Vasileiadou, 2015; Bauwens, 2019; Broska, 2021) (high confidence). In contrast, a lack of norm and belief conformity as well as trust, due to, for example, socio-economic heterogeneity and anonymity, may generate particular challenges in urban environments (Wannewitz and Garschagen, 2023). In settings in which close spatial proximity is not important, these issues can be overcome, however, by using internet-based platforms and communication (Nikravech et al., 2020).

Policy context: The embedded nature of collective action in clear-cut complementary regulatory and legal frameworks helps to foster the foundation and functionality of collective action (Villamayor-Tomas et al., 2019). This occurs in two ways. First, collective action is promoted through reductions in uncertainty and transaction costs (affecting coordination) call for organizational support and policies aimed at the sharing and diffusion of information. Second, it is promoted through changes in payoffs and their perception and an increase in normative consonance (affecting prisoner's dilemma and zero-sum situations), which call for support in terms of framing but also hard policies such as regulations and taxes. Case study evidence shows that typically more than one of these influencing channels need to be activated to encourage collective action, implying that policies need to be coherent across governance scales (central, local, individual) and degrees of agency (formal, collective, individual) (Villamayor-Tomas et al., 2019) (high confidence).

This hints at a strong role for communication, personal and social factors (Biresselioglu et al., 2020; Nikravech et al., 2020). Cases included Austria, where the reliability of information provided by neutral parties, the role of convincing narratives, the provision of knowledge on energy efficiency, a focus of policies on long-term rather than short-term costs, and higher knowledge/better information on the part of media representatives were specifically called for.

Supporting policies are leveraged by a psychological signal given through policy institutions and policymakers about more widely shared pro-environmental values and beliefs (Steg, 2023). In contrast, direct policy interventions made to meet similar targets may also discourage efforts by substituting private and community initiative and signaling

that there is 'no need' (Reichl et al., 2021). Thus, stronger financial incentives in Italy as opposed to Austria to garner photovoltaic investments have been found to crowd in economic motivations but also to crowd out motivation for collective energy (Braito et al., 2017).

**Contextual information:** Good information with respect to the underlying socio-technological setting and political context and the associated payoffs has been shown to strongly enable collective action (Mignon and Bergek, 2016; Geiger et al., 2017; Ebers Broughel and Hampl, 2018; Curtin et al., 2019) (*high confidence*). Ebers Broughel and Hampl (2018) show that financially-relevant information on community renewable energy projects raised the willingness to invest in Austria and Switzerland, and false beliefs stifled it. For Austria specifically, there is evidence that, while individual attitudes and behavioral intentions are rather pro-climate, considerable knowledge gaps, misunderstandings, and barriers also exist (Thaller and Brudermann, 2020), a finding that is consistent with evidence from Germany (Fischer et al., 2019).

Information also matters regarding the values shared by (potential) participants. Climate concerns in others are often underestimated (Steg, 2023) (*high confidence*). Thus, the lack of information on shared environmental beliefs may lock collective (and private) action into a 'spiral of silence' (Donsbach et al., 2014; Geiger and Swim, 2016; Amel et al., 2017; Bouman et al., 2021). Geiger et al. (2021a) show that (a lack of) information about the political identity of other group members in association with (false) beliefs about their willingness to contribute to environmental objectives has a strong impact on personal behaviors in the lab. There is also a risk that inaccurate meta-beliefs may boost polarization and stifle pro-environmental behavior.

#### Objectives and constraints

The stated objectives of collective action at the community level typically embrace (1) the provision and sustenance of local public goods that either help to serve climate mitigation or climate adaptation objectives at a community level (Bisaro and Hinkel, 2016; Wannewitz and Garschagen, 2023) or (2) the provision of information/education, lobbying and other forms of activism at a civil society level (e.g., Boucher et al., 2021; Fritsche and Masson, 2021). These can be understood to translate to varying degree into individual objectives, namely (a) material, in particular financial returns; (b) immaterial returns, such as a sense of identity that comes with a shared vision and set of values; and (c) a sense of self-empowerment, as well as a more general communal sense of belonging (Dóci and Vasileiadou, 2015; Bamberg et al., 2018; Schulte et al., 2020; Broska, 2021; Fritsche and Masson, 2021; Vesely et al., 2021) (*high confidence*). There is clear evidence that tackling climate action can in and of itself increase well-being (Zawadzki et al., 2020; Creutzig et al., 2022a).

While collective actions of type (2) are mostly associated with immaterial returns of type (b), these have been shown to be leveraged by emotions both positive (sense of individual empowerment, empathy) and negative (anger, sense of injustice) (e.g., Brown et al., 2019; Landmann and Rohmann, 2020; Furlong and Vignoles, 2021). Type (1) actions are associated with both types of returns (a) and (b), but the extent to which material or immaterial considerations dominate is inconclusive and context-dependent (Walker, 2011; Sagebiel et al., 2014; Bauwens, 2019; Schulte et al., 2020). According to evidence on renewable energy communities in Belgium, financial returns dominate in large communities, whereas shared values dominate in smaller communities with social interaction (Bauwens, 2019). Heterogeneity in the objectives across members within each initiative has also been shown to matter (Bauwens, 2016). Evidence from behavioral experiments shows that cooperation enhances adaptation to environmental uncertainty; high inequalities among the resource users, however, hinder cooperation (Heinz et al., 2022).

#### Design of institutions for sustained collective action

The importance of institutions, and in particular sharing rules, punishment/reward structures, and the information flows that support them as facilitators of sustained collective action, have been documented by recent experimental economic and a variety of other social science studies.

Role of sharing rules and punishment/reward structures: Blanco et al. (2018, 2021) show that externally sponsored collective action tends to be more successful when personalized and/or effort-related payments are made compared to equal sharing. Focusing on risky settings, Blanco et al. (2020) show that public and private insurance against losses related to a collective resource are substitutes, implying that collective action (as a form of public self-insurance) is weaker in settings where individuals can rely on private protection (and vice versa). A laboratory experiment on car sharing in Austria demonstrates that both legal and community sanctions tend to raise contributions (Hartl and Hofmann, 2022). Endogenous choice of institutional rules: Dannenberg and Gallier (2020) survey evidence on the extent to which and how social groups choose their own institutions to overcome commons problems and find that institutions are self-chosen to a larger extent by more cooperative groups. However, they may not always be implemented. Thus, there is no conclusive evidence on the superiority of self-governance. Dannenberg et al. (2014) provide experimental evidence on the formation of coalitions. While the endogenous choice of mechanisms that sanction deviation enhances participation, large participation also stifles commitment. Gallier et al. (2019) provide experimental evidence on parochialism vs. efficiency: While there is a revealed preference for parochialism, no impact was found on choosing relatively more efficient arrangements.

**Socio-economic heterogeneity:** A diverse socio-economic background of group members tends to render agreement on sharing and punishment/reward structures more difficult (Araral, 2009). In such cases, a strengthening of the immaterial rewards to collective actions is helpful to raise incentives when individual objectives are heterogeneous (Bauwens, 2016). Collective action can be sustained over the longer term only if it meets criteria of distributive and procedural fairness (Adger, 2016; Steg, 2023) and is associated with mutual trust among participants and in the responsible parties (Steg, 2023; Wannewitz and Garschagen, 2023) (*high confidence*).

Information on compliance: Functioning of collective action is affected by information on compliance, particularly when it comes to containing the incentive to get a free ride on others' efforts. Bühren and Dannenberg (2021) show how successful collective action hinges on information about the composition of the group with respect to free riders vs. cooperators. Hartl and Hofmann (2022) show that public information about compliance (by way of a 'badge') shapes the sanctioning mechanism in the context of their car-sharing experiment.

#### Social feedback

**Positive:** One type of positive feedback process drives the emergence of the collective itself. Here, the attainment of social empowerment (or effective collective agency) itself can become an important objective (Bamberg et al., 2018), which has scope for creating effective feedback and increasing ambition. Thus, the establishment of citizen power plants in Austria has been found to enhance empowerment in a number of ways, although this process tends to be biased towards the status quo (of established schemes and structures) and better-resourced individuals and communities (Schreuer, 2016). Learning within the collective, the strengthening of norms and identity and the resulting desire for consistent behavior, as well as a spillover from group efficacy into self-efficacy, may trigger and reinforce further collective action involving new themes and participants (Garmendia and Stagl, 2010). This learning may also trigger changes in private environment-related behaviors (Jugert et al., 2016; Sloot et al., 2018; Broska, 2021; Jans, 2021). However, a recent meta-analysis finds no evidence of this on average, suggesting that positive spillovers are the most likely if the (presumed) spillover can be traced back to normative goals and was triggered by an autonomy-supportive intervention (Geiger et al., 2021b). A second type of positive feedback relates to the spillover of action to other individuals, a process that may lead to social tipping, as discussed in greater detail in Section 5.6.2.

Negative (risk of polarization and collective counter-action): Processes towards pro-social collective behaviors are often accompanied by and may even trigger anti-social behavior, where market integration, economic diversification, and strengthened group identities have been identified as key triggers of the simultaneous development of pro-and anti-social behavior (Basurto et al., 2016). There is ample evidence that beliefs and values correlate closely with (party) political world views (Hornsey et al., 2016; Ziegler, 2017), although findings by Reichl et al. (2021) suggest a more nuanced view, at least in Europe, according to whether conservative/socio-democratic views pertain to economic issues or social issues. In the latter study, conservatism implies less environmental concern (*robust evidence, medium agreement*).

Incentives to strengthen group identity (Akerlof and Kranton, 2000; Fritsche and Masson, 2021) may thus lead to polarization, especially if identities are established in a conspicuous or moralizing way (Kurz et al., 2020; Furlong and Vignoles, 2021). In particular, communication styles, deliberate political messaging, and influencing activity in interaction with biased media and in particular social media reporting have a strong reinforcing role here (Farrell, 2016; Chinn et al., 2020; Falkenberg et al., 2022). The role of educational differences is more highly contested and may depend on the specific generation (Kahan et al., 2012; Ross et al., 2019). Countering polarization tendencies requires broad and inclusive population-based initiatives, careful policy balancing and interventions (Fritsche and Masson, 2021; Tindall et al., 2022; van de Grift and Cuppen, 2022),

and effective communication strategies (Chu and Yang, 2018) (also see Section 5.6.3).

Collective action may also be directed against renewable energy projects, and in particular the placement of wind or solar farms. While part of such opposition may be understood as 'not in my backyard' (NIMBY) behavior that may be mitigated by appropriate compensation, recent evidence, including from Austria (Scherhaufer et al., 2017), points at more complex motivations behind group formation, involving issues of local identity. Overcoming collective resistance, or even better avoiding it in the first place, then requires more holistic patterns of community involvement and ownership, both metaphorically and literally (Devine-Wright, 2011; Petrova, 2013; Botetzagias et al., 2015; Boyle et al., 2019; O'Neil, 2021) (*high confidence*).

# 5.3.3. Interaction with technology and infrastructure

Many of the changes discussed above are within the agency of individual citizens. However, there are also limits to individual agency, and legal, policy, and infrastructure changes must take place to enable individual behavioral and lifestyle changes (Wiest et al., 2022). Using technology will always have an impact on the energy and material demand, as well as on pollution (Coccia, 2022). On the other hand, technology can be a means of providing or consuming in an easier, faster, or otherwise more preferable way than without technology. As such, technology does not hinder the net-zero transition (IPCC, 2022), instead it supports the development and roll-out of renewable technologies, development of new business models, as well as expansion of infrastructure for renewables and low-carbon lifestyles.

Technology is 'a means to do things', consisting of technical artifacts ('hardware') and a disembodied element of knowledge ('software') (Grübler, 1998). Infrastructure covers physical objects and structures such as roads, railroads, power supplies, buildings, and factories. Both technology and infrastructure, i.e., the elements of the technosphere, influence social structures and the way society operates (Spaargaren, 1997), and contrasts the aggregate of all artificial objects with the natural world (Herrmann-Pillath, 2018). Just as social norms impose constraints on human behavior and decisions, on the one hand, they structure human interactions and provide opportunities, on the other hand. The co-evolution and interaction of the technosphere and the policy systems, accompanied with social norms create a carbon lock-in that is hard to break (Unruh, 2002). Thus, the technosphere can be viewed as humanly designed constructs that provide certain opportunities, but also limit certain choices of individuals operating at different geographical and time scales (Donges et al., 2017). The lifespan of large infrastructural objects extends sometimes to more than 50 years, which is comparable to the lifespan of one human generation (Otto et al., 2020b). Infrastructure such as roads or railways enable or disable certain individual and political choices.

Empirical data show that the adoption of energy efficient technologies and small-scale renewables can contribute to 30–70 % of the GHG emissions reduction potential in buildings (Creutzig et al., 2022b). Technological advances have led to significant improvements around the world and in Austria (Odyssee-MURE project, 2021; Saunders et al., 2021). The availability of materials and rare metals for the net-zero transition might hinder the transition speed, and the implementation of circular economy principles might lower the demand for new materials and metals.

Technology has multiple connections to sustainability and a clear human dimension. Certain technologies provide alternative, more sustainable solutions than others. This is achieved by improving their technical efficiencies or by lowering their carbon content directly or indirectly (through their lifecycle). This can improve the climate or environmental impact of an energy service, such as for mobility and living space (see Cross-Chapter Box 4). For example, the specific consumption of cars in Austria has improved by 0.753 % per year between 2000 and 2019, with an overall efficiency gain of the transport sector of 0.529 % per year measured by ODEX. The ODEX combines the energy efficiency trends of the different modes of transport (cars, trucks and light vehicles, bus, motorcycles, air, waterways, rail) (Odyssee-MURE project, 2021). This makes Austria slightly slower in improvement than the EU average (0.6 % per year improvement). On the other hand, due to the increase in vehicle size, modal shifts and extension of distances, technical efficiency improvements were completely offset. Technology can contribute to mitigating climate change by enabling the avoidance of a given service or of an associated energy or material consumption. For instance, digital technologies can be used to optimize room temperatures, leading to lower energy consumption as long as people's behavior doesn't counteract the technical gains. Or teleworking technologies can lead to as large as 80 % lower CO<sub>2</sub> emissions from commuting (Hook et al., 2020). Finally, a system of technologies can fully *shift* the solution space, such as rolling-out passive buildings, shifting industrial practices, or inducing a modal shift in personal mobility to non-motorized or shared mobility (e.g., walking, biking, car-sharing, bike-rental).

The interaction between the technosphere and human decisions is critical to the ultimate environmental implications of the technology and technology systems (e.g., buildings) in all cases where the technology is used, adopted or accepted (Huckebrink and Bertsch, 2021). The influence of lifestyle has been studied only to a limited extent and more Austrian results would be needed. In a study of the Community Energy Saving Program (CESP) in the UK, the average energy saving was 30 % compared to the pre-retrofit condition, but behavioral adjustments varied widely (Elsharkawy and Rutherford, 2015). In another UK study, electricity savings ranged from 0 to 50 %, depending on how the occupants used the new technologies (Araya Mejías et al., 2021). The improvement of energy services through technology and/or lifestyle and behavior, needs to be further evaluated specifically for Austria. Beyond the introduction of new and additional policies and/or technologies, exnovation (Scherhaufer et al., 2017) and negative learning should also be explored and better understood (see Section 5.2). Exnovation leads to discontinuation of ineffective solutions, allows for adaptation with new or changed behaviors and therefore improves techno-human impact on the environment.

Subsidy programs aimed at, for example, extending public transportation networks or car-sharing facilities, as well as building codes and standards, such as shared laundry and freezer space in apartment buildings, can support positive interactions between technology and human behavior. On the other hand, policies and financing mechanisms for specific technologies create market distortions and can lead to an energy efficiency gap (Saunders et al., 2021); therefore, these should be used with caution, for short time scales, and with a conscious decision to consider the non-intended impacts. Promoting smaller living or working spaces in new buildings could also help limit the sprawl of technology and infrastructure (Wiest et al., 2022). It is essential that new infrastructural projects are aligned with the net-zero objectives and that they support, or at least are not in conflict with, the net-zero emission system goals (Aigner et al., 2023b).

#### 5.3.4. Social inequalities in energy supply and related policies

Austria is steadily shifting to a renewable energy system, but this transition results in significant inequalities. Social inequality in the context of the energy supply arises from the combination of inequalities in income, energy prices,

Three of these indicators

and energy efficiency in the residential sector. Embedded in the issue of social inequality, it is of interest to define minimum standards in energy efficiency in the residential sector, secure energy supply, and affordable energy (services; electricity and heat) for particularly vulnerable parts of the population. In such a context, energy poverty can be seen as a manifestation of social inequalities that can lead to social exclusion, the origin of which is complex and multidimensional (*high confidence*).

This chapter addresses social inequality in the energy supply and places a special focus on the issue of energy poverty, providing descriptions of particularly vulnerable groups, the related legal frames and policies, and measures that can be used to combat this problematic issue at the national level.

#### Energy poverty and vulnerability factors

Energy is a fundamental resource in modern societies, the lack of which can diminish wealth and living standards, forcing individuals and households into situations of marginalization, and even jeopardizing their physical health (Brunner and Mandl, 2014; Demski et al., 2019; Shyu, 2021) (*high confidence*). Struggles caused by the lack of access to energy services or disproportionate energy costs are known as *energy poverty* (Primc et al., 2021; Salman et al., 2022).

Despite efforts and advancements to alleviate energy poverty in Austria, a legal definition for this term is still lacking (Dobbins et al., 2019; Bouzarovski et al., 2021). In this context, the Austrian Ministry for Social Affairs (BMASGK) commissioned a comprehensive expert study to determine the magnitude of this phenomenon in Austria in 2018 and create the most suitable definition of energy poverty for the Austrian context. As a result, definitions of 'energy poverty' and 'at risk of energy poverty' were proposed for those households 'at risk of poverty or social exclusion' (AROPE) being unable to afford essential energy services (i.e., power, heating, and hot water supply) (Matzinger et al., 2018). In Austria, energy poverty is acknowledged to be a complex issue, which can only be measured by using a diversity of indicators. Initially, Austrian statistical records on energy poverty focused mainly on excessive energy expenditure (Statistik Austria, 2017, 2019). The 2021 publication added the affordability dimension to consider those households which showed low energy expenditures because they sacrificed a basic level of energy use to cover other elementary needs (Statistik Austria, 2021a). The latest reports even consider eight indicators to draw a more accurate picture (C/2020/9600; Statistik Austria, 2022).

Three of these indicators are the most highly recognized causes of energy poverty, namely poor energy efficiency, low income, and high energy bills (Boardman, 2013), although further dimensions have been suggested (e.g., labor market situation, welfare state type) (Middlemiss, 2022) (high confidence). Vulnerability to energy poverty is associated with many social characteristics like unemployment, gender inequality, ethnicity, tenancy, impairment, and age (i.e., the elderly are more at risk) (Snell et al., 2015; O'Sullivan et al., 2017; Churchill and Smyth, 2020; Sánchez et al., 2020; Ivanova and Middlemiss, 2021; Petrova and Simcock, 2021; Primc et al., 2021; Wang et al., 2021). The novel perspective of the foundational economy that emphasizes the actual burden born by households using the residual income approach sheds different light onto energy poverty and aims for more inclusive climate change mitigation policies (Calafati et al., 2021; Russell et al., 2022) (limited evidence, medium agreement).

Austria has one of the lowest energy poverty rates in Europe (EPAH, 2022) (high confidence). Prior to the energy price hike of 2022, 3.2 % of all Austrian households indicated that they were unable to keep houses adequately warm, whereas 3.3 % had high energy costs while living under risk of poverty (Statistik Austria, 2022, 2024a). Furthermore, energy-poor households are often in arrears with a debt (Butler and Sherriff, 2017). Households that are traditionally disproportionately affected by high energy expenses in Austria reflect people who have a minimum compulsory education (nearly threefold that of non-energy poor), live alone (twofold of non-energy poor), and are elderly (32 % higher). Household composition and size are indeed important determinants of energy poverty (Robinson, 2019; Seebauer, 2021; Sunikka-Blank and Galvin, 2021; Eisfeld and Seebauer, 2022). One-person households constitute a large majority (i.e., 71 % considering the high energy expenses approach and 56 % by limited affordability) (Statistik Austria, 2022, 2024a). Energy poverty also has a gender dimension (Robinson, 2019; Clancy et al., 2020; Sánchez et al., 2020; Sunikka-Blank and Galvin, 2021). Various factors contribute to the higher vulnerability of women, such as the gender pay gap, labor market participation, and the uneven distribution of family caregiver duties (Mort, 2019). Despite the COVID-19 outbreak, energy poverty remained stable until 2022<sup>4</sup>; however, due to rising energy prices, Statistik

<sup>&</sup>lt;sup>4</sup> When this assessment report was last updated, EU-SILC data were only available up to the survey year 2022 (carried out in the first half of that year)

Austria (2022) started a special survey on a quarterly basis, which also included energy affordability. Data show a considerable increase in energy-poor households. While in Q4 2021, dwelling energy affordability affected 6.0 % of Austrian households, this value rose to 12.1 % in Q4 2022, which constituted an unprecedented record value for Austria. Recovery happened at a slow pace, but in Q4 2023, the portion of households unable to afford essential warmth decreases to 7.5 % due to lower energy prices (Statistik Austria, 2024c).

The linkage between energy efficiency and energy poverty has been clearly recognized (Boardman, 2013); however, dwelling energy efficiency (e.g., measured by means of Energy Performance Certificates) is not part of national statistics reports on energy poverty. Using the Low-Income-Low-Energy-Efficiency indicator could reveal a more precise picture and serve as a basis for more accurate policy interventions that counteract any rebound effect (Abbasi et al., 2022). Dwelling characteristics like type of building, year of construction, floor space, and ownership status have an impact on energy consumption. In 2022, the vast majority of energy poor households, not able to afford adequate warmth, lived in apartment buildings (73 %), had floor space up to 80 m<sup>2</sup> (66 %) and paid rent (75 %). Regarding energy expenses, 45 % of energy poor households dwelt within old buildings constructed before 1960 (Statistik Austria, 2022, 2024a).

Tenants often have few options to invest in energy efficiency measures, while landlords rarely invest in retrofitting despite of the economic viability of investments in such energy efficiency upgrade, a situation known as the energy efficiency gap (Galvin and Sunikka-Blank, 2018). Tenancy regulation in Austria tend to protect tenants; however, the split incentives dilemma in rental housing hinders advancements in energy efficiency in the building sector (Seebauer et al., 2019). Nonetheless, this issue has a lower effect on social rental housing since social housing cooperatives are not profit-seeking (Kranzl et al., 2020). Social housing in Austria serves a social mandate and provides housing with a limited-profit margin (IIBW, 2016; Klien and Streicher, 2021), where social housing providers have a great potential to implement coordinated interventions and prevent energy poverty (Croon et al., 2024). In Austria, the share of social housing as main residence is 7 % city-owned and 17 % cooperative owned (Statistik Austria, 2022, 2024a). Estimates of the Austrian Federation of Limited-Profit Housing Associations (GBV, 2021) suggest that energy poverty could affect around 2-3 % of all cooperative (social rental) housing tenants (i.e., between 13,000-20,000 households). Nonetheless, integrating both climate and social housing policies seems

AAR2

necessary to achieve decarbonization goals while reducing social inequalities (Seebauer et al., 2019). After all, income and GHG emissions have a strong correlation (see Section 5.2.2).

Apart from energy poverty, high energy prices have an impact on food and mobility poverty; however, these synergetic effects are rarely addressed (Essletzbichler et al., 2023; Penker et al., 2023) (*limited evidence, medium agreement*). Politically, food security is considered an accomplished goal in Austria (BKA, 2020) and thus an adequate monitoring of social inequalities related to food consumption, health, and climate impacts is missing (Penker et al., 2023). Transportation cost constitutes an over-average economic burden for households under risk of poverty compared with the rest of the population (Frey et al., 2023). Synergetic effects are to be considered in order to avoid the perpetuation of existing social inequalities and rather promote an active contribution to the transition goals by all parts of society (Essletzbichler et al., 2023).

#### Regulatory framework, measures, and their impact

The EU's European Green Deal (COM/2019/640 final) prioritizes a fair and just transition, with regulatory frameworks ensuring affordable and inclusive energy services (see Section 6.8.1). This includes addressing energy poverty with an EU-wide definition (C/2023/408) and a new urban mobility framework (European Commission, 2021) to promote sustainable development and reduce social inequalities.

In Austria, the regulatory framework (BGBl. I Nr. 110/2010; BGBl. I Nr. 150/2021) requires energy suppliers and authorities to consider vulnerable households when implementing energy-saving measures, to raise awareness, and to reduce overall number of affected households (BGBl. Nr. 140/1979). There are restrictions on power disconnections for utility bill arrears (ElWOG, GWG), and energy consumers gained the right to pay by installments in May 2022 (para. 82 BGBl. I Nr. 110/2010). The Energy Efficiency Act (BGBl. I Nr. 72/2014) established the coordination Office on Energy Poverty (KEA) to facilitate stakeholder collaboration for effective energy poverty reduction. The dialogue is crucial amid the increasing energy poverty, highlighting the need for targeted, comprehensive measures (see Section 6.3).

Climate change mitigation measures aim to ease the transition to cleaner energy for households, focusing on: (i) building renovation, (ii) regional development planning, (iii) renewable energies, (iv) energy consumption, (v) mobility, and (vi) natural hazard protection. The Austrian government has introduced support programs in energy and mobility to counteract inflation (BMNT, 2019; IEA, 2020; Jensen and Carvalho Fachada, 2021). However, these measures have varying impacts on societal groups, with concerns raised about short-term relief over long-term benefits, especially for vulnerable groups (Seebauer, 2021; Fink et al., 2022b). While these climate change mitigation measures are discussed in Chapter 3, Chapter 4, and Chapter 6, we outline the most relevant ones in terms of their effects on vulnerable groups.

**Building renovation** measures encompass government schemes supporting various improvements such as loft, floor, and wall insulation, along with the installation of other energy efficiency features. Grants cover the initial costs of rooftop photovoltaic or solar thermal systems (also part of renewable energies, below), heating system replacements, draught insulation with heat recovery, and offer more benefits for middle-class households with sufficient savings for energy-efficient retrofits (Seebauer, 2021). However, those in rental dwellings often face the 'landlord-tenant dilemma', where long-term building renovation investments are disadvantageous for both parties (Petrov and Ryan, 2021; Fink et al., 2022b, 2022a) (*high confidence*).

**Regional development** planning measures support energy-efficient urban planning, building revegetation, and green roofs/facades, but these can be costly for vulnerable households (landlord-tenant dilemma). They enhance property value but risk green gentrification (see Section 3.2.1), pushing vulnerable groups to more affordable areas (Kadi and Matznetter, 2022). The full extent of this impact is a research gap (Seebauer et al., 2021) (*limited evidence, medium agreement*).

Renewable energy measures support citizen energy communities, energy counseling, and expanding electric vehicle charging infrastructure. These efforts are effective for reaching vulnerable groups, especially when incentives are tailored. Information campaigns are more effective when combined with actionable solutions, and peer counseling has shown promise in reaching vulnerable groups who may not seek public support otherwise (Seebauer et al., 2021) (*limited evidence, medium agreement*).

Measures related to **energy consumption** include carbon taxes and tax exemptions for maintenance, repair, and overhaul, as well as measures designed to simplify the availability of spare parts, equipment, and information (Seebauer et al., 2021; Österreichs E-Wirtschaft, 2023). Energy taxes have a regressive effect on the lowest income segments, which bear the higher burden in relative terms. Those with higher income can easily afford shifting to sustainable systems, and higher taxes on fossil fuels play no substantial role in practical terms (Becker et al., 2021; Mayer et al., 2021). On the other hand, people in the lowest income segments lack the ability to build up savings and thus cannot afford to shift to cleaner solutions (Petrov and Ryan, 2021; Seebauer et al., 2021). The relatively higher tax burden is even greater for those in the lowest income quintile disregarded by tax exemptions (Bernhofer, 2021). The 'Klimabonus' (i.e., Austrian climate dividend) has an important redistributing effect, but the accuracy of its social impact over time has not yet been determined and can be considered as research gap (Bolte et al., 2024) (see Section 6.5.1) (*limited evidence, medium agreement*).

Mobility support schemes are designed to enhance public transportation availability, promote climate-friendly modes, offer electric vehicle rebates, and facilitate sharing initiatives and commuting allowances. This includes expanding cycle networks and making cycling more appealing. While mobility is often overlooked in energy poverty discussions and definitions, it is gaining importance. Electricity and heat are key factors, but mobility is increasingly significant. Public investments in this area benefit low-income households the most, as about half of those in the lowest income bracket do not own a vehicle (Berger et al., 2020; Blanck et al., 2020; Blanck and Kreye, 2021; Seebauer et al., 2021) (limited evidence, medium agreement). Austria prioritizes improving public transport infrastructure and schedule frequency (Austrian Federal Chancellery, 2020). However, the European Commission (2023) indicates a rising trend of mobility poverty, affecting 40 % of Austrians living below the poverty line.

Natural hazard protection measures place a focus on civil protection, emergency planning, and hazardous site remediation, as do public awareness campaigns. Climate change introduces additional challenges, particularly impacting vulnerable groups. Heatwaves and increased temperature exposure have health risks for residents in densely populated areas with limited green spaces (Reischl et al., 2018; Seebauer et al., 2021) (Cross-Chapter Box 2).

#### Initiatives of the private sector to tackle energy poverty in Austria

Energy supply companies in Austria have launched a variety of initiatives intended to alleviate and tackle energy poverty. Support under these initiatives is offered in different ways, such as by offering professional energy counseling, granting alternative payment facilities, and/or funding for renewal of domestic appliances. Eligible households apply for this support by contacting one of several social organizations, which serve as a link between households and energy suppliers (Gouveia, 2019). Since 2009, around 16,000 individuals from all over Austria have received support from the "Electricity Assistance Fund" initiated by Caritas, which is endowed by energy suppliers like VERBUND (2022) and Salzburg AG. Similar initiatives can be found at the regional level like the Energie AG Oberösterreich and Linz AG's Energy Solidarity Budget in Upper Austria (Energie AG, 2021). In February 2023, in response to the energy price crisis, Caritas Austria, Volkshilfe, and the Ministry of Climate Protection (BMK, 2023) launched a similar initiative offering aid to low-income household in order to favor decarbonization by reducing their energy consumption.

#### Identification of priorities to combat energy poverty

Various political measures have been implemented to address energy poverty, but a unified national strategy is crucial for long-term solutions (Seebauer et al., 2019) (*medium evidence, high agreement*). Professional energy counseling has been a starting point, but this may not suffice if it is limited to advising on energy savings (Kyprianou et al., 2019). Immediate energy counseling is necessary, offering shortterm relief for temporary hardships or acute financial crises. This could involve replacing inefficient appliances, introducing one-time subsidies, placing limits on disconnections, or allowing retrofitting in cases of urgent health risks (International Commission on the Futures of Education, 2021).

Medium-term measures include relief such as green electricity exemptions and appliance replacements for improved efficiency. Long-term strategies involve building refurbishments like upgrading windows and doors to enhance energy efficiency as well as supporting the establishment of citizen energy communities (E-Control, 2012; Kyprianou et al., 2019; Rechnungshof Österreich, 2020; European Commission, 2022). Implementing these measures not only creates or sustains jobs but also reduces illness, revitalizes neighborhoods, and promotes social inclusion (Heeman et al., 2022).

Historically, the degree of energy poverty in Austria has remained stable, despite political efforts (Statistik Austria, 2022) (*robust evidence, medium agreement*). Structural, long-term approaches are needed to prevent energy poverty. Thermal refurbishments of buildings, as demonstrated in projects like SINFONIA in Innsbruck, are essential (Peper and Failla, 2021). Raising awareness in energy-poor households can lead to behavioral changes and cost savings, while replacing old appliances and heating systems can result in significant energy savings (Słupik et al., 2021). It is also important to state that improving the thermal properties of buildings is the key to cutting energy consumption (Boltz and Pichler, 2014; Słupik et al., 2021). Efforts such as the 'Sauber Heizen für Alle' grant targeting low-income households demonstrate the potential of such efforts to reduce inequality and eradicate energy poverty (Umweltförderung, 2024). Improving housing conditions and upgrading appliances have been shown to effectively reduce energy consumption and costs, leading to  $CO_2$  emissions reductions (Papantonis et al., 2022).

Overall, addressing energy poverty presents opportunities for synergy with climate change mitigation goals. Tackling poverty and energy poverty can have positive climate impacts, enhancing the quality of life while reducing CO<sub>2</sub> emissions through improved energy efficiency (Austrian Federal Chamber of Labour, 2021).

#### 5.4. Organizations as agents of change

Beyond individuals and households, many organizations are making choices that define our aggregate impacts as a society. Organizations are powerful agents of change, because decisions on how to run these organizations define their direct CO<sub>2</sub> footprints, but also because they affect the barriers and enabling factors for individuals and households in society. Similarly, organizations can be references in climate transformations, illustrating more sustainable ways of operating to their members (including employees). This can also have an impact on organization stakeholders. For example, Creutzig et al. (2022b) highlight the relevance of middle actors - such as building managers, landlords, consultants, technology installers, or car dealers - for influencing the behavioral patterns of other people and the associated climate impacts. Moreover, social processes can multiply and enforce individually held beliefs and empower people in organizations.

This section first covers firms, then moves to higher education and research institutions, NGOs, and civil society organizations, and finishes with communities. The sub-section on firms focuses on the within-firm operational decisions that define  $CO_2$  footprints as well as the economic incentives that encourage firms to be greener. See Chapter 4 to learn more about the broader implications of firms' decisions on system-wide structures. The sub-section on higher education and research organizations, NGOs, and civil society reviews the important roles of largely non-commercial entities. The sub-section on communities considers meaningful aggregates of people in close proximity such as neighborhoods and people coming together with similar aims such as interest groups. Such communities can emerge bottom-up against existing governance structures or they can be encouraged and deliberately fostered in a more topdown manner.

#### 5.4.1. Firms as agents of change

There is a widespread assumption in the political debate, the media, and among business stakeholders that a trade-off between the financial and environmental performance of firms is inevitable. It is common to find opposition to recommendations and regulations that can strengthen environmental standards, as well as to discussions about greening the business sector. At the same time, empirical evidence suggests that the voluntary certification of good environmental practice, such as the Eco Management and Auditing Scheme (EMAS), ISO 14001 or the European Eco-label (EU Flower), is widespread and that many firms voluntarily adhere to high standards of performance that go beyond environmental relations. According to the KPMG Survey on Sustainability Reporting (KPMG, 2022), 96 % of the world's largest 250 firms by revenue, based on the Fortune 500 ranking, report on sustainability or environmental, social, and governance issues. Referring to climate, 64 % recognize climate change as a risk to their business, and 80 % report carbon targets. The majority of them (56 %) link their targets to the global 2°C target in the Paris Agreement, and the majority (59 %) intend to reach their targets only through emission reductions (34 % by combining emission reductions and carbon credits, and 7 % not specifying their method). The adoption of the Task Force on Climate-related Financial Disclosures (TCFD) has increased substantially in recent years. This standard was established in 2015 by the Financial Stability Board in response to the threat of climate change to the stability of the global financial system and, in 2022, was adopted by 61 % of the largest companies (KPMG, 2022). Considering the top 100 companies by revenue in 58 countries, territories, and jurisdictions, in Europe 82 % of these report on sustainability, which is an increase of 5 % over the previous year. In Austria, the percentage of the top 100 firms reporting according to standards of the Green Reporting Initiative (GRI) is 89 % (among the top 10 internationally). The share of adherence to local stock exchange guidelines is lower and does not stand out internationally. Similar comments apply to the use of Sustainability Accounting Standards Board, developed in 2011 to guide companies on investor-focused sustainability disclosure. Overall, it is natural to ask why firms engage in such voluntary practices in the first place, as this empirical observation seems inconsistent with the recurrently repeated narrative seen in the business forums that environmental action by firms negatively impacts their financial performance.

A large body of international literature shows that there is no general empirical support for the assumption that being green is harmful to firms (Blanco et al., 2009; Molina-Azorín et al., 2009; Albertini, 2013; Dixon-Fowler et al., 2013; Endrikat et al., 2014; Velte, 2022) (robust evidence, medium agreement). The wide range of data published for different sectors, types of firms, and countries does not support the view that environmental action generally comes with a trade-off regarding the financial bottom line. Indeed, the general picture that appears is that improved environmental performance has a small but positive impact on the financial performance of firms. Evidence suggests that increased operational efficiency and lower cost of debt are the mediating factors explaining the positive impact (see, e.g., Gao and Wan, 2023). Firms can do better in terms of economic and financial performance by being greener, provided that the measures implemented are proactive and based on re-designing processes (rather than end-of-pipe technologies), moderate in terms of abatement levels, and supported by investors and consumers (Blanco et al., 2009; Dixon-Fowler et al., 2013).

Sustainability concerns are prominent among Austrian job seekers and consumers alike. Statistics show that 65 % of Austrians are willing to pay more for climate-friendly food, and 78 % support labeling foods to show their environmental impact, indicating a strong market for green products in Austria (European Investment Bank, 2023). Accordingly, many Austrian companies have switched to advertising the environmental friendliness of their products and services rather than focusing on price or quality. While so-called 'green claims' are becoming increasingly popular, accusations of 'greenwashing' are also on the rise (CMS, 2023). An EY study revealed that only 46 % of Austria's top companies publish sustainability reports, and merely one-third of these firms have their environmental data externally audited. This rate is considerably lower than, for example, in Germany, where 73 % of companies undergo such audits (Ernst & Young Global Limited, 2022). Upon closer examination, explicit instances of greenwashing are evident in Austria. The Austrian Association for Consumer Information (VKI) actively targets this issue by performing 'greenwashing checks'. This initiative allows consumers to report misleading green claims. The VKI assesses these claims and, if deemed deceptive, publishes the findings on its website, potentially damaging the reputation of the involved company. Moreover, the VKI can take legal action against companies making unsubstantiated or vague environmental claims, reinforcing its commitment to enforcing truthful advertising in compliance with Austrian law. In Austria, several greenwashing cases have led to legal actions. For instance, after a VKI complaint, a brewery was found guilty of falsely advertising its beer as 'CO2-neutral brewed' and was ordered to re-label its products. Another case involved an association of dairy farmers that claimed their products supported climate protection but withdrew their ad after receiving criticism. Additionally, the recyclability of a ballpoint pen was contested due to the impracticality of recycling it in Austria (CMS, 2023). In another case, an Austrian court ruled that Austrian Airlines AG engaged in greenwashing by advertising 'CO<sub>2</sub>-neutral' flights between Vienna and Venice using 100 % sustainable aviation fuel (SAF), despite the fact that SAF constituted only about 0.1 % of global aviation fuel and the actual use in these flights was only up to 5 %. The court ordered the airline to correct the misleading advertisements by sharing accurate information on its social media platforms (Lahey, 2023).

Moreover, the use of sustainability labels and logos has risen in Austria. Such labels typically require a certification process, which must be transparent and objective. The VKI has criticized some Austrian labels for failing to genuinely substantiate their sustainability claims due to non-transparent and objective processes (CMS, 2023). A prime example of the importance of product labeling can be seen in the food industry in Austria. Although, transport accounts for only about 6 % of the total emissions from the global food system (Poore and Nemecek, 2018), Austrian consumers value regional products, which they view as being of higher quality and more eco-friendly. To support this, Austria introduced new laws in 2022 requiring origin labeling on processed foods and communal catering for meat, milk, and eggs, which were effective as of mid-2023. Additionally, since July 2022, Austrian farmers have had to disclose the origin of their products in the supply chain, impacting their food marketing strategies (CMS, 2023).

It is relevant to emphasize that there are also costs associated to 'no-action' strategies by firms. Total annual net damages from climate change for Austrian firms across multiple sectors - agriculture, forestry, energy, building management, tourism, water supply, waste management, sales, transport, and urban green spaces - have been estimated to range from EUR 2.5-5.2 billion by around 2030. These figures include damages associated with natural disasters and health impacts. Projections for 2050 suggest that these combined costs could escalate to between EUR 4.3-10.8 billion annually (Steininger et al., 2015, 2016a, 2020). Most critically: How do Austrian firms themselves perceive the impacts of these threats, and what are their responses? According to a large-scale survey by the European Investment Bank (EIB) from 2022 and 2023, 64 % of Austrian firms reported that they feel exposed to physical risks from climate change (EU: 57 %), and 43 % have set climate targets (EU: 41 %). Furthermore, Austrian firms are more proactive than the EU average in addressing climate risks, with a higher percentage of firms stating that they implement mitigation measures while simultaneously planning or investing in climate adaptation strategies. The implemented mitigation measures by the Austrian firms include enhancing new green technologies or products (47 % in Austria vs. 32 % in the EU), investing in energy efficiency (76 % vs. 57 %), generating renewable energy (61 % vs. 37 %), and focusing on waste minimization and recycling (77 % vs. 64 %), as well as sustainable transport options (54 % vs. 43 %). Overall, 94 % of Austrian firms in the survey stated to implement some form of mitigation, compared to 88 % in the EU. Additionally, 30 % of the Austrian firms stated to have developed adaptation strategies (EU: 14 %), 37 % to have set actual adaptation measures (EU: 20 %), and 11 % to use insurance to hedge risks (EU: 10 %). A significant 47 % of Austrian firms in the sample are engaged in any adaptation activities, surpassing the EU average of 33 %.

Austrian firms increasingly view climate transformation as an opportunity, whereby 38 % saw benefits in 2023, up from around 32 % in 2022, percentages that are higher than EU averages for both years (European Investment Bank, 2023). Evidence from the previous 2020–2021 EIB survey showed that roughly one-third of Austrian firms believe the transition will positively affect their demand (34 %, EU: 34 %) and reputation (33 %, EU: 38 %), with potential negative impacts on the supply chain noted by 27 % (EU: 25 %). Of the surveyed Austrian firms, 29 % employ dedicated climate staff, surpassing the EU average of 23 %. The primary barrier identified by the firms for further investment in climate action is the uncertainties surrounding taxation and regulatory frameworks (European Investment Bank, 2021). Additionally, Austrian companies recognize the importance of meeting sustainability goals in recruitment. This is important as 72 % of Austrians under the age of 30 consider a prospective employer's climate impact important when job hunting, with 21 % ranking it as a top priority (European Investment Bank, 2023). Austrian family firms have adopted a diverse array of environmental strategies. These include energy conservation measures, optimizing transportation routes to minimize traffic, sustainable resource utilization, using alternative packaging materials, and engaging in collaborative projects with the agricultural sector. The primary motivation behind these environmental practices appears to be compliance with eco-certification standards (Kuttner et al., 2021). Lastly, significant barriers to sustainability integration have been found to include resource constraints and organizational inertia, which prevent effective resource allocation that supports the achievement of sustainability goals (Kiesnere and Baumgartner, 2019).

As business owners, investors, and employees, individuals can have an impact on advancing towards greener societies. The debate on whose responsibility it is to take action on climate neutrality, with firms, governments, and citizens acting as mutually exclusive agents of change, is not well-founded and quickly becomes a circular argument. Firms argue that they do not sell more green products or turn greener in operations, because consumers are not ready to pay the additional price premia; consumers argue that they cannot afford additional price premia and that governments are responsible to guarantee a healthy environment; and governments argue that they are not pushing for stronger green guidelines and regulations for firms because they do not have the support from citizens to do so and would lose power in upcoming elections. All these actors are interrelated rather than mutually exclusive. Again, the empirical observation that 'green firms' have emerged and are thriving in the Austrian economy is a reality check against the most pessimistic accounts of alternative, lower emission impact, economic arrangements. However, an open question is the extent to which greener operations can become a strong feature of mainstream business, enabling them to affect a shift in the overall operation of the economy. This is particularly relevant for 'discount firms' which operate on very small profit margins and typically serve the less-favored sectors in society. However, there is reason to believe that the shift can spread to the majority of the business sector: The KPMG survey on sustainability reporting shows over the years that large global companies tend to lead the way in sustainability reporting, with the top 250 internationally ahead of the top 100 at the country level, and these ahead of others (KPMG,

2022). Thus, looking at the sustainability performance and evolution of the larger companies can be a useful gauge for identifying broader trends that are eventually adopted more widely.

Although evidence from companies and organizations across the EU suggests that behavioral interventions such as nudging can effectively promote green behavior (e.g., among employees) (see, e.g., Brown et al., 2013; Handgraaf et al., 2013; Mehrwert Berlin, 2022), relatively few examples are known of Austrian firms employing such techniques to achieve sustainability goals. A notable exception is a field experiment conducted during the summer of 2021 by the Institute for Higher Studies (IHS) in partnership with the ARA, an Austrian packaging collection and recycling company. This study tested different designs of waste collection bins against a control group with no bins in 90 locations in Leoben, Steyr, and Krems. The findings demonstrate that collection bins enhance public waste separation, especially those with images of pristine nature (Gangl et al., 2022b).

# 5.4.2. Higher education and research, NGOs, and civil society

Higher education and research organizations are important agents for change through educating people and building relevant competencies in new generations (see Section 6.8.3 for other educational contexts), allocating research funding to societal challenges, and through their leadership and role model function in society (medium evidence, high agreement). Taking the example of higher education, these function much like other big organizations (see Section 5.4.1), albeit in a different, less commercial context (at least in Europe) and their governance and goals are different from commercial organizations (Kezar, 2001). Leal Filho et al. (2014) describe a drive for integrated, interdisciplinary, and transversal application of sustainability in the higher education sector in areas ranging from education to operations, but important gaps were identified (e.g., in business schools, see Csillag et al., 2022) and also between the strategies expressed and actions taken (e.g., Brennan et al., 2015). In Austria, as of January 2024, 20 universities were organized in the Alliance of Sustainable Universities in Austria a network to promote sustainability in higher education and society (nachhaltigeuniversitaeten.at).

GHG emission reduction and sustainability initiatives at Austrian universities include introducing emission monitoring and reduction programs (Hölbling et al., 2023). These include reducing air-based business travels, introducing more flexibility by combining or extending business trips to reduce the distance traveled or the number of flights, increasing the choice of plant-based options at university canteens, offering only plant-based options at university and student events, encouraging commuting via public transportation or bikes, as well as investing in renewable energy harvesting facilities in university buildings, and reducing overall emissions in heating, cooling, and electricity. Some of these initiatives were initiated by student groups and as a result of student protests at universities. The Climate Change Center Austria (CCCA) is a network of universities and non-university research organizations founded in 2011 that serves as a joint contact point for science, politics, administration, and the public, sharing climate change knowledge and expertise and has produced a national energy and climate reference plan in 2019 (NECP). Besides operational activities of universities, endeavors to reform research and education have the potential to multiply knowledge and competencies and trigger action, e.g., the certificate of the Alliance on Education for Sustainable Development (nachhaltigeuniversitaeten.at/zertifikat) or university internal working groups on Sustainability Science (e.g., boku.ac.at/en/nachhaltigkeit/ sustainability-in-research/wg-sustainability-research).

Pioneering roles can also be played by citizen groups, NGOs, and other civil society organizations (see Section 6.4.2) (*medium evidence, high agreement*). For example, Fridays for Future has been active in Austria since 2018. Further examples for initiatives include co-housing groups, energy communities (see Section 5.4.3), and other forms of initiatives organized around sharing and mutual support principles. These are discussed in more detail in the subsequent sections (Komendantova, 2021). An interesting example can be found in the City of Graz, which launched the KLIMAEURO+ Program in 2024. This program provides small amounts of funding for groups and organizations, enabling them to organize local initiatives and events that directly reduce GHG emissions or provide information and training (Stadt Graz, 2024).

At the national Austrian level, two programs ought to be mentioned in this context because they complement and foster action by community groups and citizens. Climate and energy model regions (Klima- und Energiemodellregionen, KEM) and climate change adaptation model regions (Klimawandel-Anpassungsmodellregionen, KLAR!) are administered by the Austrian government's climate and energy fund. KEM is focused on decentralizing and defossilizing the energy supply, whereas KLAR! is focused on climate adaptation. The KEM approach integrates governance and implementation elements, combining top-down and bottom-up dynamics for multi-level energy transitions through partnerships across scales and societal learning (Irshaid et al., 2021). Komendantova et al. (2018) highlight that KEM activities appear to focus on informing and awareness-raising measures but are limited in participation in actual decision making. This is an important starting point but leaves room for more sophisticated participation forms according to Arnstein's (1969) ladder of participation. Within the KLAR! portfolio of activities, Babcicky and Seebauer (2021) have investigated the interplay of physical and social factors in determining flooding outcomes and demonstrated the value of adding psychological indicators on top of traditionally used socio-demographic proxies. In sum, integrating multi-level perspectives from psychological to systemic and building on the existing awareness-raising initiatives holds great potential for achieving actual energy and climate transitions.

#### 5.4.3. Energy communities

Community-based, bottom-up initiatives have an increasing potential in the energy transition and can tap into new low-carbon opportunities (Pathak et al., 2022). Energy community-like initiatives have appeared in Europe sporadically, such as energy efficiency cooperatives in the UK or smallscale wind and biomass projects run by communities in the 1970s in Denmark and Austria (Walker et al., 2010). Individual intentions are often driven by collective willingness or actions, as described in Section 5.3.2. The business case has improved with the legal foundations established recently at the EU level (Fina and Monsberger, 2023; Wierling et al., 2023) (limited evidence, medium agreement). The end-user is expected to become a key actor in future energy systems and to contribute to the levels of generation, aggregation, distribution, storage, and consumption, as well as supply of the energy itself or energy services (e.g., by providing electric mobility or other energy services). Energy communities establish space and structures that fulfill these functions and/ or help stakeholder cooperation.

The EU has defined 'energy communities' since 2019 as collaborative citizen-driven energy actions that contribute to the clean energy transition, advancing energy efficiency or the renewable energy supply within local communities (Roberto et al., 2023). These must be available for participation and effective control by citizens, local authorities, and SMEs. Citizen energy communities (CECs) are mandated in Article 2(11) of the Internal Electricity Market Directive



Figure 5.5 Integration of energy communities in the energy systems (Klaassen and Van der Laan, 2019).

(Directive (EU) 2019/944) and in Article 2(70) of the Recast Internal Gas Market Directive (COM/2021/803). Renewable energy communities (RECs) are defined and regulated in Article 2(16) of the recast Renewable Energy Directive 2018/2001 (COM/2021/557 final). The participants' main economic activities cannot be in the energy sector, and, as such, they can enter and withdraw their affiliation at any time. The differences between CECs and RECs are based on a few features, however, they are largely similar. CECs have a wider scope, allowing citizen participation across the electricity sector, and there is no limit to their size, while RECs focus on renewables only, and types of stakeholders are limited, as they are expected to be located in the vicinity of the project. RECs must be autonomous, leading to a more democratic set-up in decision-making, while this is not a requirement for CECs (REScoop.EU, 2019).

#### Primary objectives and benefits

Upscaling demand-side technologies depends on addressing the technical, financial, regulatory, and structural barriers at a local level and with public engagement, processes in which RECs and CECs are instrumental (Von Wirth et al., 2018; Fouladvand et al., 2020; Cabeza et al., 2022). According to the Directives, the main objective of creating and operating these communities is to generate social and environmental benefits rather than financial profits. Several authors find that potential participants are motivated by an economic argument in Europe and beyond (Rogers et al., 2008; Komendantova, 2021; Hackbarth and Löbbe, 2022) (medium confidence). The price provided by a locally owned system can often be lower than general market prices, depending on the agreements between community members. Citizen energy communities often join for joint bulk purchases or combined activities, such as building renovation. This allows them to negotiate better prices. At the same time, others emphasize the prevalence or co-occurrence of the pull-effect of social factors (Masson and Fritsche, 2021; Vesely et al., 2021), social norms, trust, and environmental concerns (Kalkbrenner and Roosen, 2016; Sloot et al., 2019). A willingness to participate in or support the local community can also drive projects (Fischer et al., 2021) (medium evidence, high agreement). As summarized in Figure 5.5, citizens can enter the energy system at any level, become part of the energy communities and cooperate in them in varied ways. These bottom-up local organizations not only implement projects directly, but have a demonstrative effect, raising public awareness and monitoring value (Azarova et al., 2019).

#### Status of energy communities in Austria

Austria had already introduced the concept of energy communities before the EU mandates. Collective self-consumption (CSC) was already provisioned in 2017 in the amendment of the Electricity Act (ElWOG) (BGBl. I Nr. 110/2010). This law enabled private and non-private CSC to be established and operated (Frieden et al., 2020), with a limitation placed on the participation by large companies and energy companies (Gemeinschaft schafft Energie, 2023). Austria was a frontrunner in enacting the final legislative transposition of the EU Directives in summer 2021. As a result, for example, owners of multi-apartment buildings can share electricity, which was previously hardly possible. Wierling et al. (2023) identified 384 energy communities in Austria, applying a slightly broader understanding than that of the legal text, which makes it the fifth most abundant country in Europe after Germany, the Netherlands, Denmark, and Ireland.

The Coordination Office for Energy Communities was set up in 2021 in Austria. This collaborates with public advisory institutions in the federal provinces, helping them set up energy communities. Energy communities in Austria can access funds to increase energy efficiency and the penetration of renewables. Some municipalities and federal provinces have transformed their support schemes for single installations (e.g., PV in general) to support either larger (usually commercial) scale investments or energy community solutions. The Austrian government provides up to four million Euro to support the establishment of energy communities. On federal province level, energy communities are supported (e.g., in Styria) locally. For instance, the City of Graz introduced financial support in 2016 that extended to 2020 (Frieden et al., 2020).

Compliance is monitored with reference to the legal requirements for energy communities. The energy community should provide data and information to the regulatory authority, which performs random or case-by-case compliance checks. The regulatory authority should publish an annual report on the energy communities established in Austria, and in particular on the number and regional distribution of energy communities.

Despite the empowerment potentials of decentralized energy systems, not all societal groups are equally positioned to benefit from energy community policies, with issues of energy justice taking place within initiatives, between initiatives and related actors, as well as beyond initiatives (Villavicencio Calzadilla and Mauger, 2018; Van Bommel and Höffken, 2021).

Energy communities are seen as essential elements of a future-proof, sustainable energy system (Cabeza et al., 2022; Pathak et al., 2022) and particularly in Austria (Azarova

et al., 2019; Cejka et al., 2021; Fina and Monsberger, 2023; Wierling et al., 2023) (*limited evidence, high agreement*). However, whether this will effectively be the case is dependent on their uptake during the next years, as this will define their system-wide impact.

## 5.5. Circular and sharing economy and the role of digitalization

Circular economy (CE) and sharing economy (SE) are overlapping business models that benefit climate mitigation by providing for personal needs with reduced amounts of energy and material. CE "designs and manages both, processes and outputs to maximize ecosystem functioning and human wellbeing" (Murray et al., 2017, p. 369) (see Cross-Chapter Box 5). SE refers to an 'economic system in which goods are not necessarily owned by consumers, and services are not necessarily offered by an enterprise, but can be shared between different consumers and used by other consumers' (Heinrichs, 2013; Acquier et al., 2017). Austria's level of circularity is slightly better than global average (9.7 % vs. 8.6 % respectively), which could be increased four-fold with relevant technological and structural changes (de Wit et al., 2019), but its ultimate success depends on consumer commitment (Ghisellini et al., 2018; Holzer et al., 2023) (medium confidence). CE and SE are also attractive, because they offer opportunities for new investments as well as redesigned lifestyles (Hörtenhuber et al., 2010; Aigner et al., 2023a) (limited evidence, high agreement). In this chapter, we focus on the aspect of the end-users, such as on the private and business consumers, as opposed to the suppliers as in Chapter 4 (Section 4.2). This aspect complements the assessment of how to achieve circularity (see Sections 4.2, 3.3.2 and 3.4.4) with the involvement of the users, whose role is not explored enough (BMNT, 2019) (medium evidence, high agreement). Building on the European and Austrian strategies (e.g., European Commission, 2020; BMK, 2022), various behavioral changes and changes in social norms and practices are necessary to actually achieve emission reductions. Austria adopted the 10R framework (Potting et al., 2017) in the Austrian Circularity Strategy (BMK, 2022) as presented in Table 5.1. The ten levels of circularity indicate how to retain raw materials optimally in the cycle with R1 'refuse' highlighting the importance of simply consuming nothing. Thereby, these strategies are in line with the ASI framework (Cross-Chapter Box 4), where the R1 and R2 levels avoid a demand for energy and materials at

the design phase, the R6–R10 levels shift the system level efficiency by changing products or services, and the R3–5 and R7 levels **improve** the efficiency of the product and service by reducing the energy/material input necessary. Related individual behaviors and practical examples in Austria are also indicated in Table 5.1. In this sub-chapter, we highlight the current lack of methodological strong research, which implies that more research funds are needed to study how to encourage end-users to adopt new consumption habits and decisions.

Although all Rs are relevant for the user side, the consumer behavior most commonly associated with the circular economy is the correct disposal of waste and recycling. In the micro census, 94.4 % of Austrians report that they collect waste separately (Neubauer, 2020). Generally, the public perception (in contrast to the mixed discussion in the media) of the circular and sharing economy is positive (Hartl et al., 2024). However, the amount of waste is increasing, and waste is not collected perfectly. In particular, concerning food waste (about 1/3 of food is wasted in Austria), regular public outrage occurs although the amount of food waste is not declining, indicating a high intention-behavior-gap in the society (Schanes et al., 2018) (high confidence). Although evidence on the determining attitudes of food waste and possible interventions is still rare, the existing evidence suggests that saving money is more important than environmental attitudes (Schanes et al., 2018).

Concerning ways to define individual behaviors related to the sharing economy, these can be distinguished between (a) business-to-consumer models (e.g., car sharing organized by an enterprise), (b) peer-to-peer exchange via one responsible individual based on a platform (e.g., couch surfing), and (c) consumption within self-regulated communities without a distinct authority (e.g., community gardens) (Hartl and Hofmann, 2019). Whereas pragmatic and economic reasons can be examined to predict the private demand for shared services and products, the identification as an environmentalist can predicts whether a private person supplies shared services or products (e.g., the willingness to share their own car) (Hartl et al., 2020).

Table 5.1 gives a selection of examples of the direct role of consumers in Austria to realize the Rs of the circular and sharing economy. Alaux et al. (2024b) (as detailed in Chapter 3) shows the potential emission reductions from, for example, building materials that can be gained by applying circularity, which can only be achieved if consumer decisions and behavior change occur simultaneously. By transforming both the consumer and production sides, it is possible to capture carbon (e.g., via wood-based construction of passive-housing) with tenants with low-carbon behaviors, which is particularly relevant for Austria (Sikkema et al., 2023).

# 5.5.1. Drivers, barriers, and measures to foster the circular economy and sharing economy

Circular economy is a systems level change, where consumers have a major role to play. It is insufficient to accomplish infrastructural and production-side changes, while it is also not enough for the consumers to be willing to recycle and share more. These need to concur. Almost all Austrians (98 %) believe that consumers have the largest role to play to reduce food waste (European Commission, 2015). Several of the R10 strategies have been in use for decades, for example, recycling, but these might have unexpected incentives and rebounds (Santarius and Soland, 2018; Brock et al., 2021).

Consumers' deliberate choices are preceded by behavioral intentions that, in turn, depend on their attitudes, perceived social pressures, or subjective norms, as well as their perception of control over their behavior (Ajzen, 1991). This explains why, in many instances of the circular economy implementation, there is a positive end-user attitude, but no collectively significant level of action has occurred (medium evidence, high agreement). Tröger and Panhuber (2023) found that 60 % of Austrians agree that there should be a ban on destroying clothes that are still in good condition and that these should be available for purchase. Still, secondhand shopping is not a common practice among the population, and sharing clothing is only relevant for luxury apparel (Pantano and Stylos, 2020). A recent study in the Netherlands (Koch and Vringer, 2023) showed that 60 % of the respondents were willing to buy secondhand clothing, but only less than 10 % actually do so. Nonetheless, secondhand business models are believed to generate overall more consumption because of moral licensing. The consumer decides to take advantage of products and services at three key moments: At the points of purchase, use, and disposal (Wastling et al., 2018).

Based on the combination of system preparedness and the intention-behavior approach described above, the key barriers to circular economy strategies in Austria are the following (De Jesus and Mendonça, 2018; Zero Waste Austria, 2021; Zibell et al., 2021; BMK, 2022):

	Circular economy strategies	Examples of corresponding individual behavior and decision	General practical examples of existing initiatives	Practical examples in food systems	Practical examples of ICT platforms
R1	Refuse: Abandoning redundant products and services	Anti-consumption attitude: Buying, demanding, gifting nothing	Avoiding packaging: <u>zerowasteaustria.at/</u> <u>verpackungsfrei-einkaufen-</u> <u>in-wien.html</u>	Transformation to vegan or buying nothing or only region- al food, e.g., <u>Ama Bio Siegel</u>	<u>Codecheck.info</u> – ecoGator – Evocco app – ToxFox
R2	Rethink: Increasing the intensity of the use of products/ services	Sharing on peer- to-peer or business level/enhancing multifunctions, e.g., home-office	Peer-to-peer lift-sharing: <u>blablacar.de</u>	'Host a dinner' initiative: <u>Eatwith.com</u>	<u>Airbnb.at</u> – <u>share-now.com</u> – <u>at.getaround.com</u> – <u>fragnebenan.com</u> – <u>peerby.com</u> – <u>grover.com</u>
R3	Reduce: Increasing efficiency in manu- facturing or reducing environmental impact	Buying, demanding and giving products and services with lower footprint	Adopting sustainable slow fashion: <u>dariadeh.com/pages/</u> <u>philosophy</u>	Visit vegetarian restaurants; Vienna is among the 6 most vegetarian cities in Europe (McDonagh, 2021)/Purchase energy efficient food process- ing devices	<u>GreenMeter.eu</u>
R4	<b>Reuse:</b> Using a discarded product again in its original function	Buying products in reusable packaging and returning to the service provider	Informal retail (flea markets) and second hand shops attract 20 % of purchases in clothing, 12 % in books and games (no downloads) (Statista, 2023)	Buying milk in glass bottles with return services, reintro- duced by Berglandmilch (2020)	<u>Vinted.at</u> – <u>Nebenan.de</u> – <u>peerby.com</u> – <u>Restado.de</u> – <u>Kaputt.de</u>
R5	<b>Repair:</b> Correcting a defective product so it can be used in its original function	Repairing broken electronic goods, shoes etc.	<u>zerowasteaustria.at/</u> verpackungsfrei-einkaufen- in-wien.html	Viennese Food Bank guidance on distribution of 'best before' products (Wiener Tafel, 2016)	Shop.fairphone.com – ifixit.com – kaputt.de – Shift.eco – wertgarantie.de – ReplaceDirect.de
R6	<b>Refurbish:</b> Restoring an old product to bring it up to date	Collecting and giving away old products, willingness to buy upcycled products	Creating new fashion based on old fashion: <u>Karja.at</u>	TooGoodToGo App where con- sumers can buy food products that are close to their end-date at a much lower price	Amazon Renewed – Circularcomputing.com – <u>Mudjeans.com</u> – <u>refurbed.at</u>
R7	<b>Remanufacture:</b> Using parts of a discarded product in a new product with the same function	Cooking with left- overs, accepting re- use of products from upcycled materials	Remanufacturing of construc- tion materials from demoli- tions: <u>Restado.de</u> is the largest marketplace for reclaimed con- struction materials in Europe	The company '2nd chance cereal' produce granola made from unsold bread (Sofia project, 2020)	<u>excessmaterials-</u> exchange.com – <u>Roetz-bikes.com</u>
R8	<b>Repurpose:</b> Using a discarded product or its parts to create a new product or service for a new purpose	Using products for a new purpose	Various design handicrafts, e.g., selling on Etsy and promoted by municipalities	Products based on food waste: <u>Unverschwendet.at</u> . Note there are traditional meals that use, e.g., stale bread (French toast) (Peschel and Aschemann- Witzel, 2020)	No example found
R9	<b>Recycle:</b> Processing the materials to ob- tain a new (higher or lower grade) quality	Willingness to pre- sort, recycle at home or collect in separate waste collection system	Large and established busi- nesses in collaboration with municipal services. On the individual level, cost savings benefit those that reduce their municipal waste through waste selection.	Production of compost: wien.gv.at/umwelt/ma48/ beratung/muelltrennung/ biogener-abfall/kompost.html	<u>Renewi.com</u>
R10	Recover: Incinerat- ing materials with energy recovery	Willingness to pre- sort and collect waste in the avail- able system (e.g., bring to a waste park) to incinerate it	Waste incineration though a separation system is also high in Austria (as opposed to landfill) producing heat and electricity. Also complying with the aims of local sourcing.	Food incineration	No example found

 Table 5.1
 Definition of the circular economy levels and their corresponding behavior and practical examples specifically in Austria based on Potting et al. (2017) and BMK (2022). Source: general and food examples own collection; Information and Communication Technologies (ICT) (Zeiss, 2019).

- Technological and infrastructure barriers include the availability of and access to the traditional as opposed to the circular economy alternative, for example, to mobility-as-a-service, public transport, or sharing system (see Section 3.4). Due to the system lock-ins, it is typically easier to use the less sustainable solutions, and these require extra efforts to change (*high confidence*). The distribution of space favors cars, which makes them easy to access. A way to break lock-ins is to ensure that the more sustainable options (e.g., those relevant for CE and SE) become the default options, and thus the decisions to use or apply them become straightforward (Niedderer et al., 2014) (*limited evidence, high agreement*). For example, redesigning city centers with pedestrian areas will make walking self-evident instead of car use (see Section 3.2).
- Legislation also limits the scale of the circular and sharing economy on the consumer side by reducing the opportunity, usually providing for the safety of the consumer (e.g., best-used-before date, obligations of packaging, restrictions on donating food). In the same way, wholesalers and supermarkets establish their own internal rules on, for example, quality criteria (e.g., size, color) and dispose of the low-quality or non-standard products (e.g., fruits and vegetables) before they get to the consumers (*medium confidence*).
- Financial and market barriers limit the purchase of organic and often regional products because only negative (e.g., labor costs) but not positive (e.g., health, environment) costs are factored in, which reduce the competitiveness of the product (*medium confidence*). Economic decisions also affect wholesalers and supermarkets, who optimize packaging and visual appeal (e.g., shiny apples), assuming that this will increase consumer satisfaction (Zero Waste Austria, 2021).
- Consumers are encouraged to purchase and dispose of products through advertisements, peer-pressure, perceived style (e.g., matching colors, seasonal products such as decoration, chocolate Easter eggs, Christmas cookies), and planned product obsolescence. In addition, buying nothing, repairing (e.g., clothes), avoiding (e.g., a car), or reducing (e.g., a small flat) strategies induce shame and involve stigmatization, as others might think one is poor (Anderluh et al., 2023) (*limited evidence, medium agreement*). Making the CE alternatives attractive (e.g., asking celebrities to advertise the sustainable solution) could counteract the traditional triggers. For example, the French government and the French Environmental Agency (ADEME) produced and streamed a national

communication campaign to popularize non-consumption and repairing products just ahead of the 2023 Black Friday, titled 'What if we asked ourselves the right questions before consuming?'. However, empirical assessments of the effectiveness of such initiatives are rare.

 Lack of literacy and skills among consumers. Knowledge and skills (e.g., which products can be repaired and how) and the likelihood of using technological systems (e.g., app-based e-bike sharing), as well as opportunities to acquire this knowledge and skills, are often missing (one exemption is a regulation that requires financial advisors to ask investors about their environmental preferences and to inform them accordingly (Seifert et al., 2024)).

Drivers of consumer-adoption of circular economy solutions make it easier for the end-user to carry out the desired behavior and harder to carry out the undesired one (Niedderer et al., 2014) (limited evidence, high agreement). Several factors can influence access and ease of using circular or sharing solutions in Austria. (1) Legal and regulatory requirements and standards lead the consumer towards the CE solutions by default. For example, all building and infrastructure development in Austria could be supplied from demolished old building materials (de Wit et al., 2019), which could be mandated by building standards, reducing the need to actually make an end-user decision. (2) Available and easily accessible infrastructure could channel users towards using more circular economy solutions. Presenting environmental food choices as a default option increases, for example, the tendency to make vegetarian choices (Hansen et al., 2021). An increase in density of the public transport infrastructure and closeness of shared mobility (cars, bikes, scooters) has been found to reduce car kilometers traveled. For instance, Mulalic and Rouwendal (2020) provided real-life evidence that the establishment and extension of the underground network in Copenhagen, Denmark, led to a reduction of the total number of cars of 3.4 %, while a certain (most strongly affected) district saw a car ownership reduction of up to 14 %. These findings are in line with the theoretical direct effects found by Holmgren (2020), and may grow to 10–39 % in the long term (*limited evidence, high agreement*). Digitalization has drastically increased the solution spectrum for both business models and peer-to-peer sharing solutions benefitting from faster and more direct involvement (Räisänen et al., 2021). (3) Consumers can be further motivated to use CE strategies (e.g., repair, refurbish, reuse) when it is easy, socially acceptable, and affordable to do. The cost reduction of public transportation shown in field experiments indicated that usage can be increased by cheap or free travel (Gravert and Olsson Collentine, 2021), although this does not necessarily reduce car usage. On the contrary, taxes can boost the uptake of non-environmentally friendly solutions through price signals (Porter and Linde, 1995). Requiring the provision of information on environmentally friendly consumption options at the point of sale can motivate pro-environmental choices (Seifert et al., 2024).

Recognizing the behavioral drivers of circular solutions also increases engagement with these solutions through, for example, attitudes or perceived social pressure (see above). Regarding the first one, providing information in the form of a product signature (e.g., labels) increases trust and knowledge about products (however see the issue of greenwashing). Labels informing people about regional and/or organic food products are well-known to Austrians, and about 61 % of respondents know the AMA Biosiegel and 42 % the EU-BIO-LOGO (Agrar Markt Austria, 2023). Regarding the latter, peer pressure, social norms, and social identity influence purchase choices, product use, or product disposal, as well as the habit of using public transport (Franssens et al., 2021). One promising intervention for reducing litter that needs further investigation is motivating people to pick up litter from others (e.g., during hiking trips or even in cities) (Rosenthal and Yu, 2022). Studies suggest that identification as an environmentalist predicts the willingness of car owners to offer carpooling, whereas economic and pragmatic reasons predict decisions to consume car-sharing services (Hartl et al., 2020). Feedback in combination with gamification (e.g., growing/shrinking tree for slow/fast driving) is also an efficient way to motivate people to use less energy while driving cars (Dahlinger et al., 2018), heating (Carrico and Riemer, 2011; Staddon et al., 2016), or taking long showers (Tiefenbeck et al., 2018).

Successful interventions are based on a thorough high-quality empirical analysis of drivers and barriers of behavior (see COM-B model by Michie et al., 2011). Therefore, the servicing system (e.g., availability of local products) needs to be upscaled, and the intention-behavior gap should be tackled at the same time. Individuals need knowledge and skills (i.e., competence) to perform a specific behavior, and they need to be motivated emotionally and rationally to pursue a specific behavioral goal. Finally, the financial, physical, and social contexts and infrastructure also need to encourage individuals to perform a specific behavior. However, there is currently a lack of rigorous large-scale development of and tests on efficient interventions due to a lack of sufficient financial research funds in the behavioral sciences.

### 5.5.2. Digitalization in a circular and sharing economy

Digitalization is without a doubt seen by many economic actors as a potential silver bullet for solving environmental problems and for achieving net-zero emission economies (Ellen MacArthur Foundation, 2019; Bolton et al., 2022; Schober and Mattke, 2022). For example, digitalization has been said to reduce energy consumption, because it encourages the use of using cloud services (virtualization), the collection, monitoring, and optimization of data flows across all product life cycle levels, and connects people and businesses that would otherwise not have been connected (Zeiss et al., 2021; Bolton et al., 2022). However, even though digitalization might indeed support the move towards CE, it has multiple side effects that can transform efficiency gains into increased resource consumption (Coroamă and Mattern, 2019; Santarius et al., 2020; Pouri, 2021). For example, video streaming results in the consumption of even more video hours, thus increasing energy use (Santarius et al., 2020), which, in turn, increases emissions from data centers (Lord et al., 2022). Even in the case of teleworking, where the CO<sub>2</sub> savings from reduced commuting appear to be large, a number of rebound effects have been reported, such as short-term effects (e.g., increases in ICT consumption) and long-term effects (e.g., larger houses, greater distance to work, other energy-intensive purchases) that offset or even eliminate the energy savings (O'Brien and Yazdani Aliabadi, 2020). Thus, the most important guiding principle for using digitalization to reduce CO<sub>2</sub> impacts is 'digital sufficiency', which Santarius et al. (2022, p. 2) describes as "any strategy that directly aims at decreasing the absolute level of resource and energy use by reducing the levels of production and consumption" for the dimensions of hardware, software, user, and economy. As such, the issue is not only to produce fewer devices and make them more energy-efficient, but also to reduce the energy footprint of software, increase users' sustainable behavior when using digital devices and, finally, to support a transition to sufficiency-oriented business models that promote public and community goods (Santarius et al., 2022).

After a summary of ICT opportunities, this section provides a much-needed counter-perspective to the many overly optimistic accounts in practice-oriented reports by consultancies such as Deloitte and McKinsey, which see innovative ICTs such as artificial intelligence (AI) and Big Data as a panacea for many environmental problems. To obtain a realistic picture of the overall effects of digitalization efforts on CE, the following sub-section briefly reviews the current state of the use of ICT in the circular economy and the corresponding debate on its effects, placing a special focus on digital rebound effects (Coroamă and Mattern, 2019; Schmelzle et al., 2022).

In the circular economy, managing information flows (market, actor, material object, and activity-related information flows) (Zeiss, 2019) is the key for leveraging positive impacts (Kirchherr et al., 2018). For example, extending the life of products through repair requires a knowledge of their condition, location, and repairability. Recent advances in IT, such as sensor-based technologies that generate information or predictive analytics, can leverage such information and offer opportunities to integrate information into material flows that can be highly transformative in CE application scenarios (Zeiss, 2019). In the following section, typical ICT support for such information flows is reviewed.

- Big Data analytics and artificial intelligence which be used to analyze large sets of data can reveal insights and patterns that help identify opportunities for circular practices, such as recycling or refurbishment (Berg and Wilts, 2018; Ellen MacArthur Foundation, 2019). Information systems can provide decision-makers with real-time data and insights and report key circular economy performance indicators, helping organizations set targets, monitor progress, and improve their circular practices (Saidani et al., 2019). As such, these tools help actors make sense of the smarter use and lifespan extension of material objects by providing relevant information at the right time (Zeiss, 2019). One example is GreenMeter.eu, an app designed to educate people about biodiversity in their backyard and to provide directly applicable recommendations for biodiversity conservation or restoration in specific locations. Furthermore, approaches taken to assess individual consumption habits are increasingly digitalized. Examples are personal carbon accounting, which records and calculates CO<sub>2</sub> amounts and other GHGs that a person or household emits into the atmosphere through their daily activities and consumption patterns. The aim of personal carbon accounting is to raise awareness for the individual carbon footprint and to encourage more environmentally friendly behavior (Bolton et al., 2022).
- Product lifecycle information systems capture data in areas ranging from design and manufacturing to usage and disposal. These data help businesses optimize product design for durability, repairability, and recyclability (Berg

and Wilts, 2018; Zeiss et al., 2021; Bolton et al., 2022). For example, information systems can support virtual simulations and modeling for circular product design, analyzing factors like recyclability and disassembly. These can simulate product lifecycles, material choices, and endof-life scenarios to optimize for circular outcomes. For example, digital twin technology creates virtual replicas of physical products, enabling the real-time monitoring, maintenance, and optimization of product performance throughout the lifecycle (Zeiss et al., 2021). This helps create products that are more conducive to the circular economy (Bolton et al., 2022). ICT can also aid in the proper collection, dismantling, and recycling of electronic waste, ensuring that valuable resources are recovered and hazardous materials are handled safely. ICT can facilitate efficient reverse logistics processes, ensuring that the products and materials used are collected, sorted, and transported to appropriate recycling or refurbishment facilities (Bolton et al., 2022; Wilson et al., 2022).

- Task-supporting information systems support the transformation of material objects to extend their lives or the lives of their components and raw materials. CE practices, however, expect actors to actively participate in complex and cognitively demanding tasks, such as repairing a smartphone or recycling manufacturing waste (Zeiss, 2019). For example, platforms such as <u>kaputt.de</u> help stakeholders identify the defect, find suitable solutions (e.g., self-repair, professional repair service, replacement/ disposal), and carry out the actual repair by providing activity-based information, such as repair and disassembly instructions, cost estimates, or a list of required tools (Zeiss, 2019).
- Integrating Internet of Things (IoT) into CE will provide means to track and monitor the condition and performance of products throughout their lifecycles (Wilts and Berg, 2017; Zeiss et al., 2021). These data can be used to optimize maintenance schedules and extend product lifespans by facilitating the return, repair, or recycling of products (Berg and Wilts, 2018; Bolton et al., 2022). Against this backdrop, information systems can create material passports that document the composition and origin of materials in products. These passports facilitate easier recycling and support the use of recycled materials in new products, but also facilitate predictive maintenance, optimizing resource use and reducing waste (Hoosain et al., 2020).
- Information systems can also be used to optimize inventory levels, reducing overstocking and waste. By tracking

product usage and demand, businesses can better plan production and distribution. In the supply chain, information systems provide transparency, allowing stakeholders to track the flow of materials, products, and components throughout their lifecycles. This visibility helps identify potential points of waste generation and supports responsible sourcing. These tools can also be used to support waste management and recycling processes by optimizing collection routes, tracking recycling rates, and identifying opportunities for improvement (Zeiss et al., 2021). Blockchain technology can enhance transparency in supply chains, allowing consumers to trace the origin and lifecycle of products. This may help in verifying sustainable and circular practices (Wilts and Berg, 2017). For example, the Dutch start-up Circularise (circularise.com) created a decentralized communication protocol based on blockchain to improve data quality and availability in circular value networks without releasing datasets or actor identities (Zeiss et al., 2021).

- Information systems can enable the creation of digital platforms and marketplaces that support circular business models such as sharing and renting (rethinking), and remanufacturing (Berg and Wilts, 2018; Wilson et al., 2020). These platforms connect consumers and businesses interested in circular practices. For example, online marketplaces for reuse facilitate the exchange or resale of used products and materials by connecting sellers and buyers, making it easier to find secondhand goods and extend the life of products. ICT enables the shift from traditional ownership models to 'Product-as-a-Service' models. Instead of buying products outright, consumers can access products and services on-demand. This encourages manufacturers to design for durability and easy repair, since they retain ownership of the product.
- Matchmaking information systems for CE practices help stakeholders find and compare supply and demand for physical material objects and negotiate and contract terms and conditions of the proposed transaction (Zeiss, 2019). Thus, such systems create transparency, reduce complexity, build trust, and act as legal advisors. Typical examples are sharing platforms, such as Vinted (ex Kreiderkreisel, <u>vinted.at</u>) or <u>willhaben.at</u>, where users can present secondhand clothes and other products with the aim to sell, exchange, or give away items for free (rethink and reuse).
- Collaborative information systems allow different stakeholders, including businesses, governments, and NGOs, to share knowledge, collaborate on circular initiatives,

and drive collective action (Zeiss, 2019). Information systems can be utilized to disseminate information about the circular economy to the public, raising awareness and promoting responsible consumption habits. Cloud-based platforms enable collaborative sharing and access to data, fostering collaboration between stakeholders involved in circular economy initiatives (Bolton et al., 2022). An example is PACE (<u>pacecircular.org</u>) a European-wide platform which serves as a networking and knowledge-sharing hub and brings together various stakeholders from the public and private sectors, including businesses, governments, NGOs, academia, and other organizations. These stakeholders collaborate to exchange best practices, innovative ideas, and solutions related to circular economy practices, policies, and technologies.

The impact of Green IS on environmental sustainability is two-fold (Kranz et al., 2015). Direct effects arise when ICT itself directly contributes to reducing the negative impact on the environment by reducing the consumption of natural resources over the entire life cycle from production to recycling, while indirect effects help societies and companies (e.g., in the areas of production, logistics, energy, and mobility) to operate more eco-efficiently. They can even be systemic, leading to medium- and long-term changes in economic structures and behaviors that support more eco-efficient practices (Henkel and Kranz, 2018). Please note that, due to their novelty, there is little empirical research on the impact of innovative ICT, such as Big Data, AI, and Blockchain, on the environment in various application areas, not only in Austria but worldwide (Lange et al., 2023). As such, it is not surprising that empirical studies on general rebound effects in the Austrian context are also rare, and those with an explicit ICT reference are almost non-existent. For example, a recent review on ICT-induced rebound effects shows that only 4 % of studies are from Austria, 4 % from Switzerland and 12 % from Germany, 14 % from the UK, and 9 % have an EU-wide perspective (Font Vivanco et al., 2022).

### 5.5.3. Risks associated with circular and sharing economy, in particular digital rebound

Business models that engage consumers in behaviors supporting the circular economy also engender backfiring risks. A study from France showed that secondhand fashion offers tempt consumers to buy more than regular fashion offers would; the pro-environmental image of secondhand fashion offers a justification for buying more (Parguel et al., 2017). Generally, there is a risk that consumers choose one easy sustainable option (e.g., waste separation) to justify their other unsustainable behaviors (Brudermann, 2022). Another risk is, for instance, if car-sharing offers lead to a switch away from public transport instead of a switch away from a private car. However, estimates from Italy suggest that the substitution rate of private cars is five times higher on average than that of public transport (Ceccato et al., 2021). The sharing economy is based on constant financial flows that allow access to goods and services (e.g., rent for a car), which is a risk for households with unstable income flows (Anderluh et al., 2023).

As the use of ICT in the circular economy relies heavily on electronic devices, the increasing use of ICT systems can lead to increasing energy levels. Predictions range from 1.9 % of global electricity use by ICT (Malmodin and Lundén, 2018) to as high as 6.3 % (Andrae and Edler, 2015). While it is difficult to arrive at definite numbers, researchers agree that, despite efficiency gains due to improvements in data centers and devices, energy use for ICT infrastructure could continue to increase (Hittinger and Jaramillo, 2019; Freitag et al., 2021). In addition, environmental and social issues can arise from the production of ICT due to disposal and recycling practices for electronic equipment, the exploitation of limited resources, and insufficient environmental standards (Lange et al., 2023). For example, ICT will also lead to an increase in electronic waste (e-waste), if the products are not managed properly. Although ICT can support the circular economy and recycling in other industries, the hardware is not yet built entirely out of recycled or renewable materials (Santarius et al., 2022). Against this background, digitalization in manufacturing only brings about marginal efficiency improvements (Lange et al., 2023). There are also more optimistic projections, such as the SMARTer 2030 (GeSI, 2020), which states that ICT could contribute to a 20 % reduction in global CO<sub>2</sub> emissions by 2030. However, these often do not consider rebound effects (Freitag et al., 2021), such as the following:

• Due to Big Data analysis, AI and video streaming data volumes are increasing (Freitag et al., 2021), which, in turn, leads to the exponential growth of data storage, processing, and transmission. Hence, the increasing reliance on IT infrastructure, data centers, and digital technologies leads to higher energy consumption and environmental impacts associated with manufacturing and operating these systems (Morley et al., 2018; Lord et al., 2022; Zhang and Wei, 2022). The massively growing volume of

data completely or partially offsets the previous savings potential, even though the energy and resource efficiency of data processing and transmission is constantly being improved (Malmodin, 2020). For example, model training for AI applications and even more so model inferencing (Biewald, 2019) has been predicted to significantly contribute to  $CO_2$  emissions (Strubell et al., 2019; Freitag et al., 2021). Thus, calls are increasing for AI to be developed not only for sustainability (Schober and Mattke, 2022), but also so AI itself becomes more eco-friendly (Schwartz et al., 2020; Santarius et al., 2022).

- The increasing use of IoT is another factor, which might increasingly contribute to the carbon footprint of ICT in the future (Freitag et al., 2021). There are estimates that the IoT semiconductor production alone could increase 18-fold to 722 TWh in 2025 (Statista 2020, cited by Freitag et al., 2021). For example, in smart home scenarios smart meters are used to provide transparency on energy usage, also with the aim to save household energy due to intelligent redistribution of energy (Pohl et al., 2021). However, smart home technology might also lead to adopting new services (e.g., security services) or intensifying current services (e.g., media entertainment), which, in turn, promotes even more energy consumption (Wilson et al., 2017; Freitag et al., 2021). Against this backdrop, efficiency gains are highly dependent on various context factors, which are difficult to assess in model calculations, such as size of households, adoption patterns due to privacy and security risks (Wunderlich et al., 2019), frequency of data transmissions, and direct effects such as resource consumption in the production process and resource disposal. Thus, in the example of smart meter rollouts, even optimistic assessments tend to show small efficiency gains overall, if any (Gährs et al., 2021), making a 'blind' technological rollout without a thorough impact analysis across the product lifecycle rather questionable (Veit and Thatcher, 2023).
- While IT can enable transparency and traceability in supply chains, it can also contribute to large rebound effects if not managed carefully. An example is **blockchain technology** which is supposed to enhance transparency in the supply chain (Kouhizadeh and Sarkis, 2018), but has serious environmental implications. Many blockchain networks, and especially those that use proof-of-work (PoW) consensus mechanisms (such as Bitcoin), require significant computational power to validate transactions and create new blocks. This leads to high energy consumption, as powerful hardware continuously performs com-
plex calculations, with increasing carbon footprints comparable to whole nations (Freitag et al., 2021). Moreover, the generation of e-waste due to the need to frequently replace hardware, increase the frequency of software updates, and centralize mining activities in regions with cheap electricity are not only problems related to blockchain technology, but general issues of ICT use. Newer consensus mechanisms, such as proof-of-stake (PoS) and other variations, may offer potential solutions that reduce energy consumption compared to PoW. Additionally, efforts are being made to develop more energy-efficient blockchain protocols and to transition existing networks to more sustainable consensus mechanisms, but opinions are mixed about whether these efforts might work (De Vries, 2019).

- As indicated above, while IT may enable new business models and services that extend the life span of products through repair, refurbishment, or sharing, it can also lead to higher consumption in other areas, leading to rebound effects that cancel out some, if not all, potential savings (Freitag et al., 2021; Lange et al., 2023). For example, a shared mobility platform can encourage people to travel more frequently, leading to higher overall energy and resource consumption (Lord et al., 2022). Moreover, induction effects may result from the introduction of ICT applications, because they create new options for consumption and production and cause an increase in the demand for energy and resources (Lange et al., 2021). These negative side effects increase economic growth overall, thus increasing energy demands, which offsets potential energy and resource savings (Bieser and Hilty, 2018). The transportation example is a good example of such complex, higher-order effects that introduce new problems in transportation and logistics. These include not only increasing emissions from data centers, but also wider psychological and societal problems, such as de-skilling and a greater supply of low-skilled workers, coordination problems when working with multiple platforms simultaneously, distractions from interacting with a mobile app while driving (Lord et al., 2022), and so on. Additionally, the implementation of IT systems in the circular economy may create lock-in effects, where organizations become dependent on specific technologies or platforms (Lehdonvirta, 2022). This can restrict flexibility and hinder the adoption of more sustainable alternatives in the future (Lange et al., 2023).
- Another area of concern is the negative impact on society. Measures applied to track and measure the carbon

footprint of products or organizations, or apps and software used by individuals to track and measure their green habits, often involve the detailed, continuous monitoring and collection of various kinds of behavioral data. Therefore, a question arises regarding the appropriate level of such data monitoring and collection measures. This is directly related to issues of surveillance (Zuboff, 2015; Lange et al., 2023), data ownership, security, and privacy (Bolton et al., 2022; Hammi et al., 2022), as well as issues of algorithmic bias (Kordzadeh and Ghasemaghaei, 2022). Recent examples in the UK involve the use of automated ICT surveillance to enforce clean air zones, who has come under criticism due to faulty cameras, scam websites, and automated infringement processes.

 There are also important side effects on the macroeconomic level (Gillingham et al., 2016). Such economy-wide rebound effects occur "when the market adjustments and innovation processes after energy efficiency improvements lead to an overall increase in energy use within an economy" (Kulmer and Seebauer, 2019). A recent simulation study with data from Austrian households found that a 10 % efficiency improvement in the fossil fuel consumption may result in an economy-wide rebound effect of 65 %, but a comparatively weak direct rebound effect of 8–12 % (Kulmer and Seebauer, 2019). Thus, studies on rebound effects should not only focus on direct but also on higher-order effects.

To mitigate these rebound effects, it is important to monitor and assess the impact of ICT on the entire life cycle regarding its use in a circular economy. This particularly concerns wider societal implications, such as the perils of surveillance and control (Benlian et al., 2022; Lange et al., 2023).

# 5.6. Distributional justice, social tipping points and social acceptance

We have reviewed how social change happens and what cognitive, emotional, and social principles and dynamics may facilitate or hinder social change processes in order to integrate this vital level of analysis into that of natural sciences, technology and engineering and institutional design. We have adopted a pragmatic approach, tackling the question of 'what can be done and how'. We have intentionally avoided the question of 'what should be done' and the distributional justice implications of taking different approaches to answer this question. We address some of the equity and ethical considerations of climate action below. This is followed by an assessment of social tipping points in Austria towards net zero. We finish by reflecting upon the relevance of the social acceptance of policies for the political feasibility of the net-zero transition.

#### 5.6.1. Distributional justice

For the last 30 years, climate action discussions have focused strongly on 'what should be done'. This has led to the definition of the threshold objectives by climate scientists of the +1.5°C or +2°C, and the associated reductions in climate emissions (see, e.g., IPCC, 2018). The question of 'what should be done' is also the question that has received the most attention in economics, with the Nobel Prize going to Nordhaus on the DICE model being a clear highlight. The question is however modified into 'what should be done to maximize social welfare' and has produced estimates of social costs of carbon as a solution (Nordhaus, 2017, 2019). Thus, the analytical framework and answers that economists take are very different from those of other climate scientists, and some commentators have pointed out that our economy is being run on economic models that fundamentally do not consider natural laws (e.g., Stanton et al., 2009; Weitzman, 2009; Kemp et al., 2022; Rising et al., 2022).

The way each of these questions is answered highlights very different ethical stances. The question on 'what should be done' often implies a stronger ethical commitment to the preservation of human lives at 'whatever cost', while the economists typically aggregate human happiness but disregard equity considerations associated with this (which can nevertheless be done by considering more complex social welfare functions than the typical aggregate discounted utilitarian). While the question on 'what should be done' implies as an objective to avoid the loss of human life and suffering in parts of the world that will be badly hit by climate change (e.g., projection of deaths from heat waves in India or displacement of coastal populations in Bangladesh), the question of 'what should be done to maximize social welfare' is answered by creating scenarios of the 'efficient' level of effort that is worth taking. This entails changes in western economies and lifestyles; e.g., is foregoing the idea of owning a car going 'too far' in climate effort? Or is reducing the usage of air conditioning in European countries and therefore potentially have higher heat-wave deaths going 'too far' in climate action? This is not to say that economic analyses do not generally consider the value of human lives, or human lives in countries projected to be more badly hit by climate change. But certainly, there is no common agreement that these need to be preserved at 'whatever cost' in terms of a loss of comfort, change in lifestyle, or aggregation of all efforts in terms of 'reduced social efficiency', as is commonly presented in the economic jargon.

The underlying ethical stances on distributional justice are very different. Distributional justice in economics is a common underdog to social efficiency (aggregating human well-being without considering the equity aspects to it). Distributional justice is at the center of the ethical debate for aiming for thresholds of  $+1.5^{\circ}$ C or  $+2^{\circ}$ C. The disparities between these two views have been a barrier to climate action for decades. The standpoint on distributional justice in Austria ought to be a political decision reflecting social views.

There are also critical ethical aspects of triggering social transformations of any kind, including towards net zero, that need to be explicitly addressed as part of well-functioning democratic systems. Any behavioral intervention should be challenged by asking ethical questions related to how much these modify the choice architecture, individual freedoms, or preferences, and by considering its equity and distributional implications. Reducing information asymmetries by using labels or other information interventions might help citizens make more informed choices and are widely accepted as appropriate (e.g., Sunstein, 2019). Some social welfare enhancing interventions, however, such as using a different default option in new energy contracts or interventions influencing social preferences, might be considered manipulative and ethically questionable by some groups (see, e.g., Bohr, 2016; Küppers, 2024). Indeed, information nudges are more highly accepted than choice architecture nudges (Reynolds et al., 2019; Lohmann et al., 2022), and their acceptance depends on the perceived effectiveness and fairness (John et al., 2023). In any case, studies overwhelmingly indicate that in many countries worldwide, the public largely approve nudges in the environmental, health, and safety domains (Jung and Mellers, 2016; Reisch and Sunstein, 2016; Reisch et al., 2017; Loibl et al., 2018; Hagmann et al., 2019; Sunstein, 2019; Dudás and Szántó, 2021; Krisam et al., 2021; John et al., 2023) (high confidence) (see also Section 6.5.1).

Still, open questions exist about the *legitimacy* of such interventions and measures, related to the question of 'who has the right to modify what people desire?'; on the *transparency* of such interventions, referring to 'do people have a right to know that such interventions are in place?'; and on the *accountability* of implementing such interventions, meaning 'how can people call for responsibilities on the outcomes or side-effects of such interventions? Increasing transparency has not undermined nudge effectiveness in previous research (Bruns et al., 2018) and, if anything, has improved nudge acceptance (Lohmann et al., 2022). Some of these concerns are common to other regulatory policies: While command-and-control is coercive, the harms and risks posed by the impacts of global warming provide an ethical and potentially also legal argument for limiting some aspects of individual freedom and choice. However, it is essential that any new policies explicitly build on the principles of legitimacy, transparency, and accountability that build the foundation of the Austrian democratic system. Carefully considering the ethical aspects of interventions to navigate demand-side solutions is a critical factor in determining their social acceptability.

#### 5.6.2. Social tipping points

In lifestyle change processes, it is essential to understand how social norms and the spread of new behaviors can lead to reaching social tipping points, where previously minority lifestyles and choices ('early adopters') reach the majority of society. For these changes to occur, they need the support of people who make policies, regulations, set technology standards, and make infrastructural changes enabling climate-friendly lifestyles for all. Social tipping points provide a useful framework for asking where rapid social change in lowering anthropogenic GHG emissions can happen and what kind of interventions can trigger such changes (Otto et al., 2020a). In an expert elicitation process, Nabernegg and Otto (2025) asked climate scientists in Austria from different fields to propose candidates for social tipping elements that are likely to help achieve a climate-neutral society in Austria. The authors synthesized the replies in a workshop with selected scientists and conducted interviews with stakeholders from different fields to assess the plausibility of insights. Overall, 33 scientists and stakeholders participated in the process. The study by Nabernegg and Otto (2025) provides initial insights into the applicability of the concept of social tipping in Austria, but further research is needed to shed light on the specific processes and interventions needed.

The results from Nabernegg and Otto (2025) are presented in Figure 5.6 providing 7 important areas where rapid change in GHG emission reduction is possible. Corresponding 14 interventions that are likely to trigger such changes are provided in Table 5.A.1 in the Appendix, including the number of responses. The most frequent suggestions were in the area of norms and value changes, such as changed aspirations for single-family homes and cars as status symbols to increase the social status of climate-friendly lifestyles, and technology cost reduction for renewable energy production, energy storage and e-mobility. Interestingly, the answers in this area did not concern technology breakthroughs or technologies that have not yet been invented but were about providing sufficient support for the implementation of existing technologies for renewable energy harvesting and the development of decentralized solutions and the corresponding infrastructure. The experts and stakeholders also emphasized the role of legislation and the implementation of a growth independent economic system in Austria that include the recognition of the limits of economic growth as well as the limits of measures related to technological solutions and market-based instruments. Interestingly, the expert elicitation participants were rather confident that the suggested candidates for social tipping elements can lead to a rapid decarbonization in Austria, but they were somewhat less confident that the suggested candidates for social tipping elements would actually be implemented. Figure 5.6 locates the proposed areas for rapid change on structural layers of Austrian society, with a manifestation time that varies from very rapid (<1 year) to very slow (>30 years). Although changes in social norms and customs occur very slowly (Williamson, 1998), some of these changes have already been initiated in Austrian society. Given the rapid emission reductions required to avoid catastrophic climate change, it is necessary to act in each of these areas as soon as possible, regardless of their time of manifestation.

The expert elicitation process identified the following most important barriers or negative social tipping elements that would delay the achievement of climate neutrality in Austria: The possible election of an anti-climate government, a prolonged or geographically spreading war in Ukraine or other world regions, the advertisement of fossil fuel products and the fossil industry lobby, a lack of financial incentives for supporting climate neutral options, climate skepticism and denial, and finally misinformation, a low awareness of the problem, and a prioritization of short-term interests. Finally, the role of public discourse and media was mentioned as an important factor for fostering rapid social change.

On the contrary, failure to achieve a net-zero transition by mid-century at the global level could result in severe disruptions to ecosystems, societies, and economies (Steffen et al., 2018). Gowdy (2020) warns that the unstable climate conditions and massive loss of biodiversity we are currently experiencing could even lead to the end of human civiliza-



Figure 5.6 Areas of social tipping with the potential to drive a rapid transition to net zero in the Austrian human society based on Nabernegg and Otto (2025). The processes they represent unfold across levels of social structure on widely different timescales (Williamson, 1998; Otto et al., 2020a), ranging from the fast dynamics of market exchanges and resource flows on sub-annual timescales to the slow decadal- to centennial-scale changes on the level of customs, values, and social norms. Confidence that the described phenomenon is a social tipping area is indicated by (+++) high confidence (>4.5 out of 6), (++) medium confidence (3.0–4.5 out of 6); (+) low confidence (<3.0 out of 6).

tion as we know it. If the atmosphere warms 3–4°C by 2100, and eventually as much as 8°C or more, the planet could return to the unstable climate conditions of the Pleistocene, when agriculture was impossible. Under such conditions, human societies could revert to hunting and gathering (*low confidence*). At present, however, most of the literature on climate-induced social collapse is conceptual (e.g., Steel et al., 2024) or based on literature reviews (e.g., Richards et al., 2021). Comprehensive risk assessment studies are currently lacking.

#### 5.6.3. Social acceptability of decarbonization

Social acceptance for socioeconomic changes that accompany the reduction and elimination of GHG emissions is of great importance, as resistance from the public can hinder or slow down such a transition. A majority of Austrians believe that it is the responsibility of national governments to address climate change (European Parliamentary Research Service, 2021; Vlasceanu et al., 2024). Specifically, 64 % of Austrians state that they are in favor of stricter government measures imposing behavioral changes to address the climate emergency, 66 % believe that they are more concerned about the climate emergency than their government, and 83 % say they want to replace short-distance flights by fast, low-polluting trains in collaboration with neighboring countries (European Investment Bank, 2021). Recent survey results indicate a declining relevance of climate change concerns in social and geopolitical developments after 2022 and up to the completion of this assessment report (European Commission. Directorate General for Climate Action., 2023; Hajek et al., 2024; Schwaiger et al., 2025). Therefore, one should be cautious about extrapolating the previous high levels of concern and demand for climate action into the future. For Austria, the transition to reduce emissions and climate neutrality poses several challenges to various sectors in the economy and will also have substantial impact on people's lives, likely impacting various groups in society differently (Meinhart et al., 2022). It could thus enhance the acceptability of climate policies if these adhere to the Just Transition Mechanism as proposed by the European Union, which aims to "ensure that the transition towards a climate neutral economy happens in a fair way, leaving no one behind" (European Commission, 2025) (see Section 6.8.1). Results from Thaller et al. (2023) indicate that the public acceptance of climate policies depends on them being perceived as being fair, effective, and only minimally intrusive.

AAR2

Many studies suggest that it is important to garner public acceptance and understanding for technologies that can enable the transition to a sustainable future, including renewable energy and low-carbon household technologies. Sposato and Hampl (2018) found that public acceptance and support for renewable energy technologies, such as solar panels and wind turbines, play a crucial role in their successful deployment and widespread adoption. Similarly, Batel (2020) highlighted the significance of public understanding and acceptance of low-carbon household technologies, such as energy-efficient appliances and smart home systems, in achieving sustainable consumption patterns.

In addition to renewable energy technologies, there is growing interest in climate engineering approaches, and particularly carbon capture and storage (CCS) technologies. Arning et al. (2019) emphasized the importance of public perception and acceptance of CCS technologies, which can play a role in mitigating GHG emissions from industrial processes and power generation. Public awareness and understanding of the potential benefits and risks associated with these technologies are crucial for their successful implementation and long-term viability. Pianta et al. (2021) further highlighted the need for effective communication strategies to address public concerns and promote public acceptance of climate engineering approaches. Academic experts perceive the feasibility of CCS technologies as medium-high (Kerner et al., 2023).

Besides the fact that climate neutrality will likely be strongly influenced by technological change, many of those technologies will still involve people, and the extent of abatement will depend on whether technologies are adopted and how technologies are used by individuals. Exadaktylos and van den Bergh (2021) pointed out that various behavioral biases can influence energy-related decisions and, in turn, lead to substantial and harmful rebound effects where energy savings are offset due to incorrect use of the new technologies. A study providing empirical evidence for the proposed hypothesis is that of Brandon et al. (2022), who showed that, against their expectations, smart thermostats had no economically significant effect on the energy use of households due to the inadequate usage of those devices by the individual users (i.e., users were likely to override beneficial default settings of those devices). Ways to overcome such inefficiencies could be effective education for energy efficiency and targeted behavioral interventions which can tackle behavioral irregularities more directly.

There is a large amount of evidence and high levels of agreement that economic incentives in the form of carbon and energy taxes are often rejected by the public (McCright and Dunlap, 2011; Baldwin and Lammers, 2016; Baranzini and Carattini, 2017); consequently, there is growing interest in designing such economic policy instruments to tackle not only efficiency but to increase its social acceptance (Klenert et al., 2018) (see Section 6.5.1). Among others, carbon tax support has been shown to be undermined by fairness concerns (Jagers et al., 2021), beliefs in the ineffectiveness of carbon taxes (Steg et al., 2006), references to the policy as a 'tax' (Kallbekken et al., 2011; Cherry et al., 2012; Baranzini and Carattini, 2017; Carattini et al., 2018) and distrust in the government (Beuermann and Santarius, 2006). The use of the tax revenue is another of the relevant factors for the support of carbon pricing (Baranzini and Carattini, 2017). Acceptability increases if tax revenues are earmarked to environmental spending (Thalmann, 2004; Baranzini and Carattini, 2017; Beiser-McGrath and Bernauer, 2019) or re-distributed similar to the 'Klimabonus' in Austria (Kallbekken et al., 2011; as suggested by Baranzini and Carattini, 2017; Beiser-McGrath and Bernauer, 2019; Carattini et al., 2019; Woerner et al., 2024). But in any case, the visibility of the use of the tax revenue to the public seems to be critical (Mildenberger et al., 2022).

Behavioral interventions have the capacity to induce lifestyle changes indirectly by fostering the social acceptability of ambitious climate policies such as carbon pricing. Tackling effects regarding references to the carbon pricing policies as a 'tax' (Kallbekken et al., 2011; Cherry et al., 2012), persuasive messages (Kaplowitz and McCright, 2015), informational interventions (Dechezleprêtre et al., 2022; Douenne and Fabre, 2022), information about compensatory measures and distributive issues, such as refunding tax revenues (Beiser-McGrath and Bernauer, 2019; Jagers et al., 2019, 2021; Douenne and Fabre, 2022), and learning about the benefits of welfare-enhancing carbon taxes (Janusch et al., 2021) have been shown to potentially influence the acceptability of and support for carbon pricing. Vlasceanu et al. (2024) conducted a 'mega-study' across 63 countries to investigate the effects of behavioral interventions on pro-environmental attitudes, beliefs, and behavior. The authors assessed the influence of 11 expert-crowdsourced interventions based on competing theoretical frameworks in the behavioral sciences on four climate mitigation outcomes: beliefs, policy support, sharing information on social media, and an effortful tree-planting behavioral task. The interventions' overall impacts were very low and mixed, with their effectiveness depending on the types of interventions and outcome measures. Across all countries, writing a letter to a member of a future generation was most effective in increasing support. The overall Austrian sample in this study consisted of 502 participants and was representative of the population in terms of age and gender. In Austria, among the 11 different interventions tested, the dynamic social norms intervention proved to be the most effective for increasing support for climate policies compared to a control group. Specifically, this study informed participants that "more and more people are becoming concerned about climate change" and highlighted changing country-level norms. Almost all of this evidence is based hypothetical scenarios and self-reports, and thus caution should be placed when interpreting the reliability of the findings (see, e.g., Kormos and Gifford, 2014).

Recently, studies on increasing the social acceptance of climate policies have started using real-life measures of policy support and making large data collection efforts in general populations, enhancing the reliability of the evidence. For example, Dechezleprêtre et al. (2022) examined support for climate policies in 20 countries responsible for 72 % of global  $CO_2$  emissions. Among others, support for climate policies was measured by examining the willingness of individuals to support a petition for urgent climate action, highlighting the need for immediate emission cuts to prevent environmental harm. Specifically, when respondents were provided with detailed information about the impacts of climate change, their willingness to support this petition increased. On the other hand, explaining climate policies increased their stated (survey measure) but not their revealed support (petition) for policies.

Moreover, command and control policies may be more easily accepted by the public than carbon taxes, which can be seen as more coercive and may therefore be subject to greater public opposition (e.g., Baranzini and Carattini, 2017).

These examples demonstrate the importance of public acceptance and understanding in driving the adoption and effective deployment of sustainable technologies and policies. Building trust, addressing misconceptions, and engaging the public in decision-making processes could contribute to creating a supportive environment for the successful implementation of these technologies and the overall transition to a low-carbon future.

#### Engaging the public

Engaging the public entails different dimensions, including communication, deliberation, education, and citizen science. Education and communication as well as the scope for co-creation are crucial for overcoming anxiety, barriers to adoption, and polarization by fostering an understanding for overall well-being (see Section 5.2.1). For example, this understanding may be that a loss of private transport may be compensated by improved public transport options and additional public space in cities or that material restrictions are compensated by immaterial health or the benefits of psychological well-being (Rauch and Steiner, 2006; Brand-Correa et al., 2018; Dür and Keller, 2018; Brudermann et al., 2019). The importance of communicating about climate change and raising public awareness about climate change and climate action has become an active area of research. Climate change communication has to meet the public's needs and help individuals navigate through the inherent complexities of the phenomenon. Moser (2010) mentions three basic goals of climate change communication: To educate and inform, motivate engagement and action, and change social norms and cultural values. One tool that can play an important role in achieving the aforementioned goals is 'framing', that is, describing a problem in different ways to preferentially give rise to different choices or preferences (Tversky and Kahneman, 1981). For instance, a prime example of framing in the context of climate change is given in a study by Lockwood (2011), who investigated how the alternative framing of climate policies would affect public support in the UK. Frames for three different policy areas were tested: Expanding renewable energy, promoting residential energy efficiency, and financial assistance for developing countries. For instance, renewable energy was framed as either a strategy for energy security, as an economic opportunity, or as the 'control frame' of fighting climate change. Among all policy areas, the only significant framing effect was found for promotion of renewable energy. When this topic was framed as a strategy for energy security, it gathered considerably more support than under the climate change frame, while the economic opportunity frame attracted the lowest support. This is a noteworthy finding, since the economic opportunity framing of climate policies has recently gained popularity in the public debate.

Communication is also relevant to adaptation to climate change, such as responding to increased instances of flooding, wildfire, and water shortages. Studies from other countries like Australia and Canada started to develop communication schemes that could prevent human causes of wildfire and prepare residents for the event of a wildfire (Remenick, 2018). Water shortages need to be accompanied by efficient (to-be-developed) communication strategies that motivate citizens to comply with usage schemes or to save energy if a shortage of hydropower occurs (Schultz et al., 2016). Concerning flooding, communication strategies require the efficient promotion of prevention strategies (e.g., purchase of insurances, organization of volunteers) and correct behavior during the event (e.g., compliance with safety regulations).

More generally, many studies show that information campaigns to inform the public how to take concrete action on climate change, such as campaigns on energy conservation, are effective for changing individual behavior patterns in that they help people translate lessons communicated, such as energy tips, into actual use (Farrow et al., 2018; Andor et al., 2022). Although most campaigns strive to change individual behavior, some studies have also shown their effectiveness in influencing community or even national energy consumption (e.g., Carrico et al., 2015).

Moreover, many studies have demonstrated that deliberation can affect how people think about complex societal problems and policy issues, such as climate change and climate policy. For instance, experimental evidence was gathered by MacKenzie and Caluwaerts (2021), who compared support for climate action policies between groups of people (i.e., a group that deliberated the policies with each other and a group that did not). They found that those who deliberated became more supportive of government action, particularly with respect to tax policies (such as a gas tax). One way to implement such deliberation in practice is by using so-called 'Climate Assemblies' in which citizens representing different groups of populations come together to jointly debate climate change policies. Such assemblies have been successfully implemented, for instance, in the UK, Denmark, Germany, Finland, France, Scotland, Spain, and recently also in Austria. In Austria, a hundred randomly selected citizens who were representative of all regions of Austria as well as of different parts of society formed such a Climate Assembly and were tasked with answering the question 'What can Austria do to become climate-friendly by 2040?'. In the Assembly, small groups worked together over six weekends to produce concrete policy proposals for dealing with climate change in Austria. In total, the Assembly developed 93 proposals to address climate change, including a land sealing ban, the abolition of fossil energy subsidies, and higher taxes for climate-damaging vehicles (see Section 6.4.2). Similar comments apply to the design of policy bundles by the public on specific topics, such as transportation (Hössinger et al., 2023). After taking the perspective of a 'task solver' in policy design, citizens propose ambitious bundles of policies, reaching in most cases (60 %) the emission reduction targets of 70 %.

Similarly, education is a strong pillar of public engagement (see Section 6.8.3). Action for climate protection will only be successful when a range of opportunities for different groups in society are provided to improve climate literacy, the understanding of science, awareness, hope, and concern, as well as hopefully and subsequently enable people to make more informed decisions on matters related to climate change. For instance, young people in schools can learn about the science of climate change and understand the impacts of their actions on climate change. In Austria, various educational initiatives have been implemented such as the Austrian Education Energy Initiative (ETSIT) to increase energy literacy and the k.i.d.Z.21 program targeting teenagers to foster climate literacy. Both initiatives were implemented among primary and secondary school students, and the results indicate that they raise awareness and concern about the problem at hand and influence student's intention about energy-saving actions and climate-friendly behaviors (Deisenrieder et al., 2020; Keller et al., 2022).

Lastly, citizen science offers promising opportunities for the future deployment of public engagement campaigns. The European Commission is defining citizen science as 'general public engagement in scientific research activities where citizens actively contribute to science either with their intellectual effort, or surrounding knowledge, or their tools and resources'. This is still developed to only a very limited extent.

#### Chapter Box 5.1. Behavioral aspects of responses to climate risks

This box emphasizes behavioral aspects related to responses to climate hazards such as floods, heat waves or wild fires once they manifest. In this way, it can be understood as reflections on adaptation strategies once climate damages manifest. For more information related to climate risks on mental health, please see Section 5.2.1 and Cross-Chapter Box 2, for risks related to migration due to climate impacts in other countries, see Section 6.8.2. For the interested reader, the EU-wide report for responses to crises, including but not restricted to climate-related crises, 'Safer Together – Strength-ening Europe's Civilian and Military Preparedness and Readiness' was released in late 2024 (Niinistö, 2024) and is highly relevant when considering responses to climate catastrophes.

Human responses to disasters can either reduce the damages in crisis situations, whenever there is a display of cooperation, coordination, mutual help, etc., or increase the damages if there is a panic response, unleashed conflict, focus on self-interest, panic buying, etc. (Niinistö, 2024). Therefore, it is very important to set the right institutions and governance structures that can facilitate cooperation and coordination during disasters, such as risk management plans, organizational network that coordinate disaster response, communication networks, information and data exchange, etc. This should include adaptation to climate change, for example in urban and rural planning in relation to heat stress or unusual precipitation events. In the literature, disaster risk reduction (DRR) policies covering, for example, goals, recognition of critical signals, communication and coordination between public and non-profit actors, as well as with citizens are seen to increase successful crisis response (Steigenberger, 2016; Kuhlicke et al., 2020; Krogh and Lo, 2023). Nonetheless, empirical comparisons on a country level show that exposure to frequent and severe disasters does not lead to improved DRR policies (Nohrstedt et al., 2021). Case studies show that the success of these crises plans increases with training, expertise-building, planning and plan enactment, leadership, and personal acquaintance and trust between actors (Steigenberger, 2016; Seebauer and Babcicky, 2018; Krogh and Lo, 2023). Future research should establish more empirical evidence on successful DRR and might adopt more holistic strategies such as adopting the concept of safety culture, also for the context of natural hazards DRRs (Marshall, 2020).

During a crisis, behavioral interventions can be used to enhance compliance with risk reduction policies, e.g., energy saving, water saving (Gangl et al., 2022a; European Commission et al., 2023). There is a high agreement in the literature that cheap informational nudges effectively promote energy savings (Buckley, 2020). Effective communication depends, for instance, on clear and concrete goals, including implementation intentions, i.e., if-then-plans (Ahn et al., 2021), simple heuristics, i.e., rule-of-thumb (Tversky and Kahneman, 1981, 1986) from trusted sources and role-models (Craig and McCann, 1978). Increasing the salience of specific behaviors through visual signals, such as traffic light symbols (Dobber et al., 2023) and reminders which use feedback (Legault et al., 2020) or targeted social norms (Bonan et al., 2020; Kaestner and Vance, 2022) can also increase compliance with rules.

Nonetheless, future behavioral research should test whether announcements or early warning systems also work as expected in acute crisis situations and how positive and negative spill-over effects can be utilized to increase the efficacy of crisis communication. For instance, communication policies can target companies using the strategy to create positive spill-over effects to households (European Commission et al., 2023). The report on preparedness and readiness to crises to the President of the European Commission offers a broad range of recommendations for crisis management within and across EU states, aiming to mitigate the negative effects of crises, including natural hazards and other climate change related events (Niinistö, 2024).

#### REFERENCES

- Abbasi, M. H., Abdullah, B., Castaño-Rosa, R., Ahmad, M. W., Rostami, A., and Cullen, J. (2022). Planning energy interventions in buildings and tackling fuel poverty: Can two birds be fed with one scone? *Energy Research & Social Science* 93, 102841. <u>https:// doi.org/10.1016/j.erss.2022.102841</u>
- Abrahamse, W., and Steg, L. (2013). Social influence approaches to encourage resource conservation: A meta-analysis. *Global En*vironmental Change 23, 1773–1785. <u>https://doi.org/10.1016/j.gloenvcha.2013.07.029</u>
- Acquier, A., Daudigeos, T., and Pinkse, J. (2017). Promises and paradoxes of the sharing economy: An organizing framework. *Technological Forecasting and Social Change* 125, 1–10. <u>https:// doi.org/10.1016/j.techfore.2017.07.006</u>
- Adélaïde, L., Chanel, O., and Pascal, M. (2022). Health effects from heat waves in France: an economic evaluation. *Eur J Health Econ* 23, 119–131. <u>https://doi.org/10.1007/s10198-021-01357-2</u>
- Adger, W. N. (2016). Place, well-being, and fairness shape priorities for adaptation to climate change. *Global Environmental Change* 38, A1–A3. <u>https://doi.org/10.1016/j.gloenvcha.2016.03.009</u>
- Adger, W. N., Arnell, N. W., and Tompkins, E. L. (2005). Successful adaptation to climate change across scales. *Global Environmental Change* 15, 77–86. <u>https://doi.org/10.1016/j.gloenvcha.</u> <u>2004.12.005</u>
- Adger, W. N., Dessai, S., Goulden, M., Hulme, M., Lorenzoni, I., Nelson, D. R., et al. (2009). Are there social limits to adaptation to climate change? *Climatic Change* 93, 335–354. <u>https://doi. org/10.1007/s10584-008-9520-z</u>
- Agrar Markt Austria (2023). Bio in Zahlen. Available at: https:// bioinfo.at/bio-in-zahlen (Accessed September 13, 2023).
- Ahn, J. N., Hu, D., and Vega, M. (2021). Changing pace: Using implementation intentions to enhance social distancing behavior during the COVID-19 pandemic. *Journal of Experimental Psychology: Applied* 27, 762–772. <u>https://doi.org/10.1037/</u> <u>xap0000385</u>
- Aigner, E., Görg, C., Krisch, A., Madner, V., Muhar, A., Novy, A., et al. (2023a). "Technical Summary", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 105–170. Available at: https://doi.org/10.1007/978-3-662-66497-1\_4 (Accessed October 4, 2023).
- Aigner, E., Görg, C., Madner, V., Muhar, A., Posch, A., Steininger, K. W., et al. (2023b). "Summary for Policymakers," in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 19–33. <u>https:// doi.org/10.1007/978-3-662-66497-1\_2</u>
- Aigner, E., Görg, C., Madner, V., Novy, A., Steininger, K. W., Nabernegg, S., et al. (2023c). "Kapitel 1. Einleitung: Strukturen für ein klimafreundliches Leben", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 173–194. <u>https://doi.org/10.1007/978-3-662-66497-1\_5</u>
- Ajzen, I. (1991). The theory of planned behavior. Organizational Behavior and Human Decision Processes 50, 179–211. <u>https://doi.org/10.1016/0749-5978(91)90020-T</u>

- Akerlof, G. A., and Kranton, R. E. (2000). Economics and Identity. The Quarterly Journal of Economics 115, 715–753. <u>https://doi.org/10.1162/003355300554881</u>
- Akizu-Gardoki, O., Bueno, G., Wiedmann, T., Lopez-Guede, J. M., Arto, I., Hernandez, P., et al. (2018). Decoupling between human development and energy consumption within footprint accounts. *Journal of Cleaner Production* 202, 1145–1157. <u>https:// doi.org/10.1016/j.jclepro.2018.08.235</u>
- Akizu-Gardoki, O., Kunze, C., Coxeter, A., Bueno, G., Wiedmann, T., and Lopez-Guede, J. M. (2020). Discovery of a possible Well-being Turning Point within energy footprint accounts which may support the degrowth theory. *Energy for Sustainable Development* 59, 22–32. <u>https://doi.org/10.1016/j.esd.2020.09.</u> 001
- Alaux, N., Obrecht, T., Saade, M., and Passer, A. (2024a). A Need for Sufficiency in the Built Environment: Prospective Life Cycle Assessment of the Austrian Building Stock. *IOP Conf. Ser.: Earth Environ. Sci.* 1363, 012002. <u>https://doi.org/10.1088/1755-1315/1363/1/012002</u>
- Alaux, N., Schwark, B., Hörmann, M., Ruschi Mendes Saade, M., and Passer, A. (2024b). Assessing the prospective environmental impacts and circularity potentials of building stocks: An open-source model from Austria (PULSE-AT). J of Industrial Ecology 28, 1435–1448. <u>https://doi.org/10.1111/jiec.13558</u>
- Albertini, E. (2013). Does Environmental Management Improve Financial Performance? A Meta-Analytical Review. Organization & Environment 26, 431–457. <u>https://doi.org/10.1177/ 1086026613510301</u>
- Allcott, H. (2011). Social norms and energy conservation. Journal of public Economics 95, 1082–1095. <u>https://doi.org/10.1016/j.jpubeco.2011.03.003</u>
- Amel, E., Manning, C., Scott, B., and Koger, S. (2017). Beyond the roots of human inaction: Fostering collective effort toward ecosystem conservation. *Science* 356, 275–279. <u>https://doi.org/ 10.1126/science.aal1931</u>
- Anderluh, A., Moser, M., Pellegrini, T., Richter, L., Zahra, M., Hackl, R., et al. (2023). CE4ALL – Kreislaufwirtschaft für alle. Incentive, Enabler und Inhibitoren für einen nachhaltigen Konsum. Wien: Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie (BMK).
- Andor, M. A., Gerster, A., and Peters, J. (2022). Information campaigns for residential energy conservation. *European Economic Review* 144, 104094. <u>https://doi.org/10.1016/j.euroecorev.2022.</u> <u>104094</u>
- Andor, M. A., Gerster, A., Peters, J., and Schmidt, C. M. (2020). Social Norms and Energy Conservation Beyond the US. *Journal of Environmental Economics and Management* 103, 102351. <u>https://doi.org/10.1016/j.jeem.2020.102351</u>
- Andrae, A. S. G., and Edler, T. (2015). On Global Electricity Usage of Communication Technology: Trends to 2030. *Challenges* 6, 117–157. <u>https://doi.org/10.3390/challe6010117</u>
- Araral, E. (2009). What Explains Collective Action in the Commons? Theory and Evidence from the Philippines. World Development 37, 687–697. <u>https://doi.org/10.1016/j.worlddev.2008.</u> 08.002
- Araya Mejías, O., Montalvo, C., García-Berrocal, A., Cubillo, M., and Gordaliza, D. (2021). Energy Savings after Comprehensive Renovations of the Building: A Case Study in the United Kingdom and Italy. *Energies* 14, 6460. <u>https://doi.org/10.3390/ en14206460</u>

- Arning, K., Offermann-van Heek, J., Linzenich, A., Kaetelhoen, A., Sternberg, A., Bardow, A., et al. (2019). Same or different? Insights on public perception and acceptance of carbon capture and storage or utilization in Germany. *Energy Policy* 125, 235– 249. <u>https://doi.org/10.1016/j.enpol.2018.10.039</u>
- Arnstein, S. R. (1969). A Ladder Of Citizen Participation. Journal of the American Institute of Planners 35, 216–224. <u>https://doi.org/10.1080/01944366908977225</u>
- Austrian Federal Chamber of Labour (2021). The EU renovation wave: Decarbonisation of the building sector as the key measure to combat energy poverty. Brussels: AK Europa. Available at: <u>https://www.akeuropa.eu/sites/default/files/2021-04/Renovationwave.pdf</u>
- Austrian Federal Chancellery (2020). Out of a Sense of Responsibility for Austria: Government Programme 2020–2024. Vienna: Bundeskanzleramt.
- Axsen, J., and Kurani, K. S. (2012). Social Influence, Consumer Behavior, and Low-Carbon Energy Transitions. *Annual Review of Environment and Resources* 37, 311–340. <u>https://doi.org/10.1146/annurev-environ-062111-145049</u>
- Ayres, I., Raseman, S., and Shih, A. (2013). Evidence from two large field experiments that peer comparison feedback can reduce residential energy usage. *The Journal of Law, Economics, & Or*ganization 29, 992–1022. <u>https://doi.org/10.1093/jleo/ews020</u>
- Azarova, V., Cohen, J., Friedl, C., and Reichl, J. (2019). Designing local renewable energy communities to increase social acceptance: Evidence from a choice experiment in Austria, Germany, Italy, and Switzerland. *Energy Policy* 132, 1176–1183. <u>https:// doi.org/10.1016/j.enpol.2019.06.067</u>
- Babcicky, P., and Seebauer, S. (2021). People, not just places: Expanding physical and social vulnerability indices by psychological indicators. *J Flood Risk Management* 14, e12752. <u>https://doi. org/10.1111/jfr3.12752</u>
- Bachner, G., Mayer, J., Fischer, L., Steininger, K. W., Sommer, M., Köppl, A., et al. (2021). Application of the concept of "functionalities" in macroeconomic modelling frameworks – Insights for Austria and methodological lessons learned. EconTrans working paper #4. WIFO Working Papers. Available at: <u>https://www. econstor.eu/handle/10419/237062</u> (Accessed August 31, 2023).
- Baldwin, M., and Lammers, J. (2016). Past-focused environmental comparisons promote proenvironmental outcomes for conservatives. Proc. Natl. Acad. Sci. U.S.A. 113, 14953–14957. <u>https:// doi.org/10.1073/pnas.1610834113</u>
- Baltruszewicz, M., Steinberger, J. K., Paavola, J., Ivanova, D., Brand-Correa, L. I., and Owen, A. (2023). Social outcomes of energy use in the United Kingdom: Household energy footprints and their links to well-being. *Ecological Economics* 205, 107686. <u>https://doi.org/10.1016/j.ecolecon.2022.107686</u>
- Bamberg, S., Rees, J. H., and Schulte, M. (2018). "8 Environmental protection through societal change: What psychology knows about collective climate action—and what it needs to find out", in *Psychology and Climate Change*, eds. S. Clayton and C. Manning (Academic Press), 185–213. <u>https://doi.org/10.1016/B978-0-12-813130-5.00008-4</u>
- Bamberg, S., Rees, J., and Seebauer, S. (2015). Collective climate action: Determinants of participation intention in community-based pro-environmental initiatives. *Journal of Environmental Psychology* 43, 155–165. <u>https://doi.org/10.1016/j.jenvp.</u> 2015.06.006
- Bao, J., and Ho, B. (2015). Heterogeneous Effects of Informational Nudges on Pro-social Behavior. The B.E. Journal of Economic

Analysis & Policy 15, 1619–1655. <u>https://doi.org/10.1515/be-jeap-2014-0125</u>

- Baranzini, A., and Carattini, S. (2017). Effectiveness, earmarking and labeling: testing the acceptability of carbon taxes with survey data. *Environ Econ Policy Stud* 19, 197–227. <u>https://doi.org/10.1007/s10018-016-0144-7</u>
- Barreca, A., Clay, K., Deschênes, O., Greenstone, M., and Shapiro, J. S. (2015). Convergence in Adaptation to Climate Change: Evidence from High Temperatures and Mortality, 1900–2004. *American Economic Review* 105, 247–251. <u>https://doi.org/10. 1257/aer.p20151028</u>
- Barreca, A., Clay, K., Deschenes, O., Greenstone, M., and Shapiro, J. S. (2016). Adapting to Climate Change: The Remarkable Decline in the US Temperature-Mortality Relationship over the Twentieth Century. *Journal of Political Economy* 124, 105–159. https://doi.org/10.1086/684582
- Barrett, S. (2007). Why Cooperate? The Incentive to Supply Global Public Goods. Oxford University Press, USA.
- Basurto, X., Blanco, E., Nenadovic, M., and Vollan, B. (2016). Integrating simultaneous prosocial and antisocial behavior into theories of collective action. *Sci. Adv.* 2, e1501220. <u>https://doi. org/10.1126/sciadv.1501220</u>
- Batel, S. (2020). Research on the social acceptance of renewable energy technologies: Past, present and future. *Energy Research & Social Science* 68, 101544. <u>https://doi.org/10.1016/j.erss.2020.101544</u>
- Bateman, T. S., and O'Connor, K. (2016). Felt responsibility and climate engagement: Distinguishing adaptation from mitigation. *Global Environmental Change* 41, 206–215. <u>https://doi.org/ 10.1016/j.gloenvcha.2016.11.001</u>
- Bauwens, T. (2016). Explaining the diversity of motivations behind community renewable energy. *Energy Policy* 93, 278–290. <u>https://doi.org/10.1016/j.enpol.2016.03.017</u>
- Bauwens, T. (2019). Analyzing the determinants of the size of investments by community renewable energy members: Findings and policy implications from Flanders. *Energy Policy* 129, 841–852. <u>https://doi.org/10.1016/j.enpol.2019.02.067</u>
- Bauwens, T., and Eyre, N. (2017). Exploring the links between community-based governance and sustainable energy use: Quantitative evidence from Flanders. *Ecological Economics* 137, 163–172. https://doi.org/10.1016/j.ecolecon.2017.03.006
- Becker, M., Pfeifer, G., and Schweikert, K. (2021). Price Effects of the Austrian Fuel Price Fixing Act: A Synthetic Control Study. *Energy Economics* 97, 105207. <u>https://doi.org/10.1016/j.eneco.</u> 2021.105207
- Beiser-McGrath, L. F., and Bernauer, T. (2019). Could revenue recycling make effective carbon taxation politically feasible? *Sci. Adv.* 5, eaax3323. <u>https://doi.org/10.1126/sciadv.aax3323</u>
- Benartzi, S., Beshears, J., Milkman, K. L., Sunstein, C. R., Thaler, R. H., Shankar, M., et al. (2017). Should Governments Invest More in Nudging? *Psychol Sci* 28, 1041–1055. <u>https://doi. org/10.1177/0956797617702501</u>
- Benlian, A., Wiener, M., Cram, W. A., Krasnova, H., Maedche, A., Möhlmann, M., et al. (2022). Algorithmic Management: Bright and Dark Sides, Practical Implications, and Research Opportunities. Business and Information Systems Engineering. <u>https:// doi.org/10.1007/s12599-022-00764-w</u>
- Bento, N., Grubler, A., Boza-Kiss, B., De Stercke, S., Krey, V., Mc-Collum, D. L., et al. (2024). Leverage demand-side policies for energy security. *Science* 383, 946–949. <u>https://doi.org/10.1126/ science.adj6150</u>

- Berg, H., and Wilts, H. (2018). Digital platforms as market places for the circular economy – requirements and challenges. *Nach-haltigkeitsManagementForum* 27, 1–9. <u>https://doi.org/10.1007/ s00550-018-0468-9</u>
- Berger, J., Strohner, L., and Thomas, T. (2020). Klimainstrumente im Vergleich: Herausforderungen in Hinblick auf ökologische, ökonomische und soziale Nachhaltigkeit. Vienna: EcoAustria – Institute for Economic Research. Available at: <u>https://www. econstor.eu/handle/10419/227419</u>
- Bergquist, M., Thiel, M., Goldberg, M. H., and Van Der Linden, S. (2023). Field interventions for climate change mitigation behaviors: A second-order meta-analysis. *Proc. Natl. Acad. Sci. U.S.A.* 120, e2214851120. <u>https://doi.org/10.1073/pnas.2214851120</u>
- Bernhofer, D. (2021). Die Verteilungseffekte einer CO2-Bepreisung im Haushaltssektor: Ein Überblick über die aktuelle empirische Literatur. Wirtschaft und Gesellschaft-WuG 47, 225–239. <u>https:// doi.org/10.59288/wug472.61</u>
- Beuermann, C., and Santarius, T. (2006). Ecological tax reform in Germany: handling two hot potatoes at the same time. *Energy Policy* 34, 917–929. https://doi.org/10.1016/j.enpol.2004.08.045
- BGBl. I Nr. 72/2014 (2024). Bundesgesetz über die Verbesserung der Energieeffizienz bei Haushalten, Unternehmen und dem Bund sowie Energieverbrauchserfassung und Monitoring (Bundes-Energieeffizienzgesetz – EEffG) StF: BGBl. I Nr. 72/2014 idF. BGBl. I Nr. 29/2024.
- BGBI. I Nr. 110/2010 (2010). Bundesgesetz, mit dem die Organisation auf dem Gebiet der Elektrizitätswirtschaft neu geregelt wird (Elektrizitätswirtschafts- und -organisationsgesetz 2010 ElWOG 2010) StF: BGBI. I Nr. 110/2010 idF. BGBI. I Nr. 145/2023.
- BGBI. I Nr. 150/2021 (2024). Bundesgesetz über den Ausbau von Energie aus erneuerbaren Quellen (Erneuerbaren-Ausbau-Gesetz – EAG) StF: BGBI. I Nr. 150/2021 idF. BGBI. I Nr. 27/2024.
- BGBl. Nr. 140/1979 (2022). Bundesgesetz vom 8. März 1979, mit dem Bestimmungen zum Schutz der Verbraucher getroffen werden (Konsumentenschutzgesetz – KSchG) StF: BGBl. Nr. 140/1979 idF. BGBl. BGBl. I Nr. 109/2022.
- Bicchieri, C. (2005). The grammar of society: The nature and dynamics of social norms. Cambridge University Press.
- Bieser, J. C. T., and Hilty, L. M. (2018). "An Approach to Assess Indirect Environmental Effects of Digitalization Based on a Time-Use Perspective", in *Advances and New Trends in Environmental Informatics. Progress in IS*, eds. HJ. Bungartz, d Kranzlmüller, W. Weinberg, J. Weismüller, and V. Wohlgemuth (Springer), 67–78. https://doi.org/10.1007/978-3-319-99654-7\_5
- Biewald, L. (2019). Deep Learning and Carbon Emissions. *Towards Data Science*. Available at: <u>https://towardsdatascience.com/</u> <u>deep-learning-and-carbon-emissions-79723d5bc86e</u>
- Biresselioglu, M. E., Demir, M. H., Demirbag Kaplan, M., and Solak, B. (2020). Individuals, collectives, and energy transition: Analysing the motivators and barriers of European decarbonisation. *Energy Research & Social Science* 66, 101493. <u>https://doi.org/10.1016/j.erss.2020.101493</u>
- Bisaro, A., and Hinkel, J. (2016). Governance of social dilemmas in climate change adaptation. *Nature Clim Change* 6, 354–359. <u>https://doi.org/10.1038/nclimate2936</u>
- BKA (2020). Aus Verantwortung für Österreich. Regierungsprogramm 2020 – 2024. Vienna: Bundeskanzleramt Österreich. Available at: <u>https://www.bundeskanzleramt.gv.at/dam/jcr:</u> <u>7b9e6755-2115-440c-b2ec-cbf64a931aa8/RegProgramm-lang.</u> <u>pdf</u> (Accessed September 12, 2022).

- Blanck, R., and Kreye, K. (2021). Verteilungswirkungen ausgewählter klimapolitischer Maßnahmen im Bereich Mobilität. Berlin: Öko-Institut e.V.
- Blanck, R., Kreye, K., and Zimmer, W. (2020). Impulse für mehr Klimaschutz und soziale Gerechtigkeit in der Verkehrspolitik. Berlin: Öko-Institut e.V.
- Blanco, E., Dutcher, E. G., and Haller, T. (2020). Social dilemmas with public and private insurance against losses. *Journal of Economic Behavior & Organization* 180, 924–937. <u>https://doi.org/10.1016/j.jebo.2019.02.008</u>
- Blanco, E., Haller, T., and Walker, J. M. (2018). Provision of environmental public goods: Unconditional and conditional donations from outsiders. *Journal of Environmental Economics and Management* 92, 815–831. <u>https://doi.org/10.1016/j.jeem.2017.10.002</u>
- Blanco, E., Rey-Maquieira, J., and Lozano, J. (2009). THE ECO-NOMIC IMPACTS OF VOLUNTARY ENVIRONMENTAL PERFORMANCE OF FIRMS: A CRITICAL REVIEW. Journal of Economic Surveys 23, 462–502. <u>https://doi.org/10.1111/j.1467-6419.2008.00569.x</u>
- Blanco, E., Struwe, N., and Walker, J. M. (2021). Experimental evidence on sharing rules and additionality in transfer payments. *Journal of Economic Behavior & Organization* 188, 1221–1247. <u>https://doi.org/10.1016/j.jebo.2021.06.012</u>
- BMK (2016). Österreich unterwegs 2013/2014. Ergebnisbericht zur österreichweiten Mobilitätserhebung "Österreich unterwegs 2013/2014". Vienna: Bundesministerium für Verkehr, Innovation und Technologie. Available at: <u>https://www.bmk.gv.at/ dam/jcr:fbe20298-a4cf-46d9-bbee-01ad771a7fda/oeu\_2013-2014\_Ergebnisbericht.pdf</u> (Accessed February 6, 2025).
- BMK (2021). Austria's 2030 Mobility Master Plan. The new climate action framework for the transport sector: sustainable – resilient – digital. Vienna: Federal Ministry Republic of Austria for Climate Action, Environment, Energy, Mobility, Innovation and Technology. Available at: <u>https://www.bmk.gv.at/dam/ jcr:eaf9808b-b7f9-43d0-9faf-df28c202ce31/BMK\_Mobilitaetsmasterplan2030\_EN\_UA.pdf</u> (Accessed September 6, 2023).
- BMK (2022). Österreich auf dem Weg zu einer nachhaltigen und zirkulären Gesellschaft. Die österreichische Kreislaufstrategie. Wien: Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie. Available at: <u>https:// www.bmk.gv.at/themen/klima\_umwelt/abfall/Kreislaufwirtschaft/strategie.html</u>
- BMK (2023). Neue Förderaktion mit der Caritas: Gratis Energiesparberatung & Gerätetausch für Einkommensschwache. BMK Infothek. Available at: <u>https://infothek.bmk.gv.at/neuefoerderaktion-mit-der-caritas-gratis-energiesparberatunggeraetetausch-fuer-einkommensschwache/</u>
- BMNT (2019). Integrated National Energy and Climate Plan for Austria. 2021–2030. pursuant to Regulation (EU) 2018/1999 of the European Parliament and of the Council on the Governance of the Energy Union and Climate Action. Vienna: Federal Ministry for Sustainability and Tourism, Austria. Available at: <u>https://energy.ec.europa.eu/system/files/2020-03/at\_final\_ necp\_main\_en\_0.pdf</u> (Accessed May 7, 2024).
- Boardman, B. (2013). Fixing fuel poverty: challenges and solutions. London: Routledge.
- Bohr, J. (2016). The 'climatism' cartel: why climate change deniers oppose market-based mitigation policy. *Environmental Politics* 25, 812–830. <u>https://doi.org/10.1080/09644016.2016.1156106</u>

- Bolte, S., Klackl, J., Hansen, J., Jonas, E., and Uhl-Hädicke, I. (2024). The role of political orientation and value framing in carbon pricing acceptance: Evidence from a representative sample. *Journal of Environmental Psychology* 96, 102330. <u>https://doi. org/10.1016/j.jenvp.2024.102330</u>
- Bolton, P., Lacy, P., Spence, M., Xu, M., Chen, L., Choi, S., et al. (2022). Digital Circular Economy For Net Zero. SSRN Electronic Journal. <u>https://doi.org/10.2139/SSRN.4092608</u>
- Boltz, W., and Pichler, F. (2014). Getting it Right: Defining and Fighting Energy Poverty in Austria. *The ICER Chronicle. Edition 2*, 19–24.
- Bonan, J., Cattaneo, C., d'Adda, G., and Tavoni, M. (2020). The interaction of descriptive and injunctive social norms in promoting energy conservation. *Nat Energy* 5, 900–909. <u>https://doi.org/10.1038/s41560-020-00719-z</u>
- Borghi, J., and Kuhn, M. (2024). A health economics perspective on behavioural responses to climate change across geographic, socio-economic and demographic strata. *Environ. Res. Lett.* 19, 081001. <u>https://doi.org/10.1088/1748-9326/ad5d0c</u>
- Botetzagias, I., Malesios, C., Kolokotroni, A., and Moysiadis, Y. (2015). The role of NIMBY in opposing the siting of wind farms: evidence from Greece. *Journal of Environmental Planning and Management* 58, 229–251. <u>https://doi.org/10.1080/ 09640568.2013.851596</u>
- Boucher, J. L., Kwan, G. T., Ottoboni, G. R., and McCaffrey, M. S. (2021). From the suites to the streets: Examining the range of behaviors and attitudes of international climate activists. *Energy Research & Social Science* 72, 101866. <u>https://doi.org/10.1016/j.erss.2020.101866</u>
- Bouman, T., Steg, L., and Perlaviciute, G. (2021). From values to climate action. *Current Opinion in Psychology* 42, 102–107. <u>https:// doi.org/10.1016/j.copsyc.2021.04.010</u>
- Bouzarovski, S., Thomson, H., and Cornelis, M. (2021). Confronting energy poverty in Europe: A research and policy agenda. *Energies* 14, 858. <u>https://doi.org/10.3390/en14040858</u>
- Bowles, S., and Polanía-Reyes, S. (2012). Economic Incentives and Social Preferences: Substitutes or Complements? *Journal* of Economic Literature 50, 368–425. <u>https://doi.org/10.1257/ jel.50.2.368</u>
- Boyle, K. J., Boatwright, J., Brahma, S., and Xu, W. (2019). NIM-BY, not, in siting community wind farms. *Resource and Energy Economics* 57, 85–100. <u>https://doi.org/10.1016/j.reseneeco.</u> 2019.04.004
- Braito, M., Flint, C., Muhar, A., Penker, M., and Vogel, S. (2017). Individual and collective socio-psychological patterns of photovoltaic investment under diverging policy regimes of Austria and Italy. *Energy Policy* 109, 141–153. <u>https://doi.org/10.1016/j. enpol.2017.06.063</u>
- Brand-Correa, L. I., Martin-Ortega, J., and Steinberger, J. K. (2018). Human Scale Energy Services: Untangling a 'golden thread'. *Energy Research & Social Science* 38, 178–187. <u>https://doi.org/10.1016/j.erss.2018.01.008</u>
- Brandon, A., Clapp, C., List, J., Metcalfe, R., and Price, M. (2022). The Human Perils of Scaling Smart Technologies: Evidence from Field Experiments. Cambridge, MA: National Bureau of Economic Research. <u>https://doi.org/10.3386/w30482</u>
- Brandon, A., Ferraro, P. J., List, J. A., Metcalfe, R. D., Price, M. K., and Rundhammer, F. (2017). Do the effects of nudges persist? theory and evidence from 38 natural field experiments. Cambridge, MA: National Bureau of Economic Research. <u>https:// doi.org/10.3386/w23277</u>

- Brennan, L., Binney, W., Hall, J., and Hall, M. (2015). Whose Job Is That? Saving the Biosphere Starts at Work. *Journal of Nonprofit* & Public Sector Marketing 27, 307–330. <u>https://doi.org/10.1080/</u> 10495142.2015.1053348
- Bressler, R. D. (2021). The mortality cost of carbon. *Nat Commun* 12, 4467. https://doi.org/10.1038/s41467-021-24487-w
- Bressler, R. D., Moore, F. C., Rennert, K., and Anthoff, D. (2021). Estimates of country level temperature-related mortality damage functions. *Sci Rep* 11, 20282. <u>https://doi.org/10.1038/ s41598-021-99156-5</u>
- Brimicombe, C., Runkle, J. D., Tuholske, C., Domeisen, D. I. V., Gao, C., Toftum, J., et al. (2024). Preventing heat-related deaths: The urgent need for a global early warning system for heat. *PLOS Clim* 3, e0000437. <u>https://doi.org/10.1371/journal.pclm.0000437</u>
- Brink, E., and Wamsler, C. (2019). Citizen engagement in climate adaptation surveyed: The role of values, worldviews, gender and place. *Journal of Cleaner Production* 209, 1342–1353. <u>https:// doi.org/10.1016/j.jclepro.2018.10.164</u>
- Brock, J., Volcovici, V., and Geddie, J. (2021). The recycling myth: A plastic waste solution littered with failure. *Reuters*. Available at: <u>https://www.reuters.com/investigates/special-report/environment-plastic-oil-recycling/</u> (Accessed September 13, 2023).
- Broska, L. H. (2021). It's all about community: On the interplay of social capital, social needs, and environmental concern in sustainable community action. *Energy Research & Social Science* 79, 102165. <u>https://doi.org/10.1016/j.erss.2021.102165</u>
- Brown, K., Adger, W. N., Devine-Wright, P., Anderies, J. M., Barr, S., Bousquet, F., et al. (2019). Empathy, place and identity interactions for sustainability. *Global Environmental Change* 56, 11–17. <u>https://doi.org/10.1016/j.gloenvcha.2019.03.003</u>
- Brown, Z., Johnstone, N., Haščič, I., Vong, L., and Barascud, F. (2013). Testing the effect of defaults on the thermostat settings of OECD employees. *Energy Economics* 39, 128–134. <u>https:// doi.org/10.1016/j.eneco.2013.04.011</u>
- Bruckner, B., Hubacek, K., Shan, Y., Zhong, H., and Feng, K. (2022). Impacts of poverty alleviation on national and global carbon emissions. *Nat Sustain* 5, 311–320. <u>https://doi.org/10.1038/ s41893-021-00842-z</u>
- Brudermann, T. (2022). Die Kunst der Ausrede: Warum wir uns lieber selbst täuschen, statt klimafreundlich zu leben. München: Oekom Verlag.
- Brudermann, T., Aschemann, R., Füllsack, M., and Posch, A. (2019). Education for Sustainable Development 4.0: Lessons Learned from the University of Graz, Austria. *Sustainability* 11, 2347. <u>https://doi.org/10.3390/su11082347</u>
- Brunner, K.-M., and Mandl, S. (2014). "Energy Consumption and Social Inequality: The Problem of Fuel Poverty", in *Energy con*sumption: impacts of human activity, current, and future challenges, environmental and socio-economic effects, ed. S. Reiter (New York: Nova Science Publishers, Inc).
- Bruns, H., Kantorowicz-Reznichenko, E., Klement, K., Jonsson, M. L., and Rahali, B. (2018). Can nudges be transparent and yet effective? *Journal of Economic Psychology* 65, 41–59. <u>https://doi. org/10.1016/j.joep.2018.02.002</u>
- Buckley, P. (2020). Prices, information and nudges for residential electricity conservation: A meta-analysis. *Ecological Economics* 172, 106635. <u>https://doi.org/10.1016/j.ecolecon.2020.106635</u>
- Buehler, R., Pucher, J., and Dümmler, O. (2019). Verkehrsverbund: The evolution and spread of fully integrated regional public transport in Germany, Austria, and Switzerland. *International*

*Journal of Sustainable Transportation* 13, 36–50. <u>https://doi.org/</u> 10.1080/15568318.2018.1431821

- Bühren, C., and Dannenberg, A. (2021). The demand for punishment to promote cooperation among like-minded people. *European Economic Review* 138, 103862. <u>https://doi.org/10.1016/j.</u> <u>euroecorev.2021.103862</u>
- Butler, D., and Sherriff, G. (2017). 'It's normal to have damp': Using a qualitative psychological approach to analyse the lived experience of energy vulnerability among young adult households. *Indoor and Built Environment* 26, 964–979. <u>https://doi.org/10.1177/1420326X17708018</u>
- Byrne, D. P., Nauze, A. L., and Martin, L. A. (2021). An Experimental Study of Monthly Electricity Demand (In)elasticity. *The Energy Journal* 42, 205–222. <u>https://doi.org/10.5547/ 01956574.42.2.dbyr</u>
- C/2020/9600 (2020). Commission Recommendation (EU) 2020/ 1563 of 14 October 2020 on energy poverty. Official Journal of the European Union. Available at: <u>https://eur-lex.europa.eu/ legal-content/EN/TXT/?uri=CELEX%3A32020H1563</u>
- C/2023/408 (2023). Commission Recommendation (EU) 2023/2407 of 20 October 2023 on energy poverty. Available at: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=OJ% 3AL\_202302407
- Cabeza, L. F., Bai, Q., Bertoldi, P., Kihila, J. M., Lucena, A. F. P., Mata, É., et al. (2022). Buildings. Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. https://doi.org/10.1017/9781009157926.011
- Calafati, L., Froud, J., Haslam, C., Johal, S., and Williams, K. (2021). Diversity in leading and laggard regions: living standards, residual income and regional policy. *Cambridge Journal of Regions*, *Economy and Society* 14, 117–139. <u>https://doi.org/10.1093/ cjres/rsaa027</u>
- Carattini, S., Carvalho, M., and Fankhauser, S. (2018). Overcoming public resistance to carbon taxes. WIREs Climate Change 9, e531. <u>https://doi.org/10.1002/wcc.531</u>
- Carattini, S., Kallbekken, S., and Orlov, A. (2019). How to win public support for a global carbon tax. *Nature* 565, 289–291. <u>https://</u> <u>doi.org/10.1038/d41586-019-00124-x</u>
- Carlsson, F., Gravert, C., Johansson-Stenman, O., and Kurz, V. (2021). The Use of Green Nudges as an Environmental Policy Instrument. *Review of Environmental Economics and Policy* 15, 216–237. https://doi.org/10.1086/715524
- Carrera, M., Royer, H., Stehr, M., and Sydnor, J. (2018). Can financial incentives help people trying to establish new habits? Experimental evidence with new gym members. *Journal of health economics* 58, 202–214. <u>https://doi.org/10.1016/j.jhealeco.2018.02.010</u>
- Carrico, A. R., and Riemer, M. (2011). Motivating energy conservation in the workplace: An evaluation of the use of group-level feedback and peer education. *Journal of Environmental Psychology* 31, 1–13. <u>https://doi.org/10.1016/j.jenvp.2010.11.004</u>
- Carrico, A. R., Vandenbergh, M. P., Stern, P. C., and Dietz, T. (2015). US climate policy needs behavioural science. *Nature Clim Change* 5, 177–179. <u>https://doi.org/10.1038/nclimate2518</u>
- CCCA (2022). 1,5° C: Wieviel Treibhausgase dürfen wir noch emittieren? Hintergrundpapier zu globalen und nationalen Treibhausgasbudgets. Wien: CCCA. Available at: <u>https://ccca.</u> <u>ac.at/thg-budget</u>
- Ceccato, R., Chicco, A., and Diana, M. (2021). Evaluating car-sharing switching rates from traditional transport means through

logit models and Random Forest classifiers. *Transportation Planning and Technology* 44, 160–175. <u>https://doi.org/10.1080/</u>03081060.2020.1868084

- Cejka, S., Zeilinger, F., Stefan, M., Zehetbauer, P., Veseli, A., Burgstaller, K., et al. (2021). "Implementation and Operation of Blockchain-Based Energy Communities Under the New Legal Framework", in Smart Cities, Green Technologies, and Intelligent Transport Systems, eds. C. Klein, M. Helfert, K. Berns, and O. Gusikhin (Cham: Springer International Publishing), 3–30. https://doi.org/10.1007/978-3-030-89170-1\_1
- Central Intelligence Agency (2024). The World Factbook Austria. *The World Factbook*. Available at: <u>https://www.cia.gov/theworld-factbook/countries/austria</u> (Accessed October 28, 2024).
- Chancel, L. (2022). Global carbon inequality over 1990–2019. *Nature Sustainability* 5, 931–938. <u>https://doi.org/10.1038/s41893-</u> 022-00955-z
- Chang, K. M., Hess, J. J., Balbus, J. M., Buonocore, J. J., Cleveland, D. A., Grabow, M. L., et al. (2017). Ancillary health effects of climate mitigation scenarios as drivers of policy uptake: a review of air quality, transportation and diet co-benefits modeling studies. *Environ. Res. Lett.* 12, 113001. <u>https://doi.org/10.1088/1748-9326/aa8f7b</u>
- Charness, G., and Gneezy, U. (2009). Incentives to exercise. *Econometrica* 77, 909–931. https://doi.org/10.3982/ECTA7416
- Charness, G., and Sutter, M. (2012). Groups Make Better Self-Interested Decisions. *Journal of Economic Perspectives* 26, 157– 176. <u>https://doi.org/10.1257/jep.26.3.157</u>
- Cherry, T. L., Kallbekken, S., and Kroll, S. (2012). The acceptability of efficiency-enhancing environmental taxes, subsidies and regulation: An experimental investigation. *Environmental Science* & Policy 16, 90–96. <u>https://doi.org/10.1016/j.envsci.2011.11.007</u>
- Chinn, S., Hart, P. S., and Soroka, S. (2020). Politicization and Polarization in Climate Change News Content, 1985–2017. *Science Communication* 42, 112–129. <u>https://doi.org/10.1177/ 1075547019900290</u>
- Chu, H., and Yang, J. Z. (2018). Taking climate change here and now – mitigating ideological polarization with psychological distance. *Global Environmental Change* 53, 174–181. <u>https://doi.org/10.1016/j.gloenvcha.2018.09.013</u>
- Churchill, S. A., and Smyth, R. (2020). Ethnic diversity, energy poverty and the mediating role of trust: Evidence from household panel data for Australia. *Energy Economics* 86, 104663. <u>https:// doi.org/10.1016/j.eneco.2020.104663</u>
- Cialdini, R. B., and Jacobson, R. P. (2021). Influences of social norms on climate change-related behaviors. *Current Opinion* in Behavioral Sciences 42, 1–8. <u>https://doi.org/10.1016/j.cobeha.2021.01.005</u>
- Cialdini, R. B., Reno, R. R., and Kallgren, C. A. (1990). A focus theory of normative conduct: Recycling the concept of norms to reduce littering in public places. *Journal of personality and social psychology* 58, 1015–1026. <u>https://psycnet.apa.org/doi/10.1037/</u> 0022-3514.58.6.1015
- Cil, G., and Kim, J. (2022). Extreme temperatures during pregnancy and adverse birth outcomes: Evidence from 2009 to 2018 U.S. national birth data. *Health Economics* 31, 1993–2024. <u>https:// doi.org/10.1002/hec.4559</u>
- Clancy, J., Özerol, G., Mohlakoana, N., Feenstra, M., and Sol Cueva, L. (2020). "Engendering the Energy Transition: Setting the Scene", in *Engendering the Energy Transition*, eds. J. Clancy, G. Özerol, N. Mohlakoana, M. Feenstra, and L. Sol Cueva (Cham: Palgrave Macmillan), 3–9.

- Clayton, S. (2020). Climate anxiety: Psychological responses to climate change. *Journal of Anxiety Disorders* 74, 102263. <u>https:// doi.org/10.1016/j.janxdis.2020.102263</u>
- Cloutier, S., Karner, A., Breetz, H. L., Toufani, P., Onat, N., Patel, S., et al. (2017). Measures of a Sustainable Commute as a Predictor of Happiness. *Sustainability* 9, 1214. <u>https://doi.org/10.3390/</u> <u>su9071214</u>
- CMS (2023). CMS Green Globe: Austria. Available at: <u>https://cms.law/en/are/publication/cms-green-globe/austria</u>
- Coccia, M. (2022). New technological trajectories to reduce fossil-fuel pollution and support sustainable socioeconomic systems. In Review. <u>https://doi.org/10.21203/rs.3.rs-2323975/v1</u>
- COM/2019/640 final (2019). COMMUNICATION FROM THE COMMISSION TO THE EUROPEAN PARLIAMENT, THE EUROPEAN COUNCIL, THE COUNCIL, THE EUROPE-AN ECONOMIC AND SOCIAL COMMITTEE AND THE COMMITTEE OF THE REGIONS The European Green Deal. Available at: <u>https://eur-lex.europa.eu/legal-content/EN/TXT/</u> <u>?uri=COM%3A2019%3A640%3AFIN</u> (Accessed September 14, 2022).
- COM/2021/557 final (2021). Proposal for a DIRECTIVE OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL amending Directive (EU) 2018/2001 of the European Parliament and of the Council, Regulation (EU) 2018/1999 of the European Parliament and of the Council and Directive 98/70/EC of the European Parliament and of the Council as regards the promotion of energy from renewable sources, and repealing Council Directive (EU) 2015/652. Available at: <u>https://eur-lex.europa.</u> <u>eu/legal-content/EN/TXT/?uri=CELEX%3A52021PC0557</u> (Accessed September 14, 2023).
- COM/2021/803 (2021). Proposal for a DIRECTIVE OF THE EU-ROPEAN PARLIAMENT AND OF THE COUNCIL on common rules for the internal markets in renewable and natural gases and in hydrogen. Available at: <u>https://eur-lex.europa.eu/ legal-content/en/ALL/?uri=CELEX%3A52021PC0803</u>
- Coroamă, V. C., and Mattern, F. (2019). Digital Rebound-Why Digitalization Will Not Redeem Us Our Environmental Sins., in *Proceedings of the 6<sup>th</sup> International Conference on ICT for Sustainability(ICT4S 2019)*. Available at: <u>https://vs.inf.ethz.ch/</u> <u>publ/papers/CoroamaMattern2019-DigitalRebound.pdf</u>
- Corvino, F. (2023). The forward-looking polluter pays principle for a just climate transition. *Critical Review of International Social and Political Philosophy*, 1–28. <u>https://doi.org/10.1080/13698</u> <u>230.2023.2243729</u>
- Craig, C. S., and McCann, J. M. (1978). Assessing Communication Effects on Energy Conservation. *J CONSUM RES* 5, 82. <u>https:// doi.org/10.1086/208718</u>
- Creutzig, F., Niamir, L., Bai, X., Callaghan, M., Cullen, J., Díaz-José, J., et al. (2022a). Demand-side solutions to climate change mitigation consistent with high levels of well-being. *Nat. Clim. Chang.* 12, 36–46. <u>https://doi.org/10.1038/s41558-021-01219-y</u>
- Creutzig, F., Roy, J., Devine-Wright, P., Díaz-José, J., Geels, F. W., Grubler, A., et al. (2022b). "Demand, services and social aspects of mitigation Supplementary Material", in *IPCC, 2022: Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*, eds. P. R. Shukla, J. Skea, R. Slade, A. A. Khourdajie, R. van Diemen, D. McCollum, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press). https://doi.org/10.1017/9781009157926.007

- Croon, T. M., Hoekstra, J. S. C. M., and Dubois, U. (2024). Energy poverty alleviation by social housing providers: A qualitative investigation of targeted interventions in France, England, and the Netherlands. *Energy Policy* 192, 114247. <u>https://doi.org/ 10.1016/j.enpol.2024.114247</u>
- Csillag, S., Király, G., Rakovics, M., and Géring, Z. (2022). Agents for sustainable futures? The (unfulfilled) promise of sustainability at leading business schools. *Futures* 144, 103044. <u>https://doi.org/10.1016/j.futures.2022.103044</u>
- Curtin, J., McInerney, C., Gallachóir, B. Ó., and Salm, S. (2019). Energizing local communities—What motivates Irish citizens to invest in distributed renewables? *Energy Research & Social Science* 48, 177–188. <u>https://doi.org/10.1016/j.erss.2018.08.020</u>
- Dahlinger, A., Tiefenbeck, V., Ryder, B., Gahr, B., Fleisch, E., and Wortmann, F. (2018). The impact of numerical vs. symbolic eco-driving feedback on fuel consumption – A randomized control field trial. *Transportation Research Part D: Transport and Environment* 65, 375–386. <u>https://doi.org/10.1016/j.trd.2018.</u> 09.013
- Dang, H.-A. H., Hallegatte, S., and Trinh, T.-A. (2024). Does global warming worsen poverty and inequality? An updated review. *Journal of Economic Surveys* 38, 1873–1905. <u>https://doi.org/ 10.1111/joes.12636</u>
- Dannenberg, A., and Gallier, C. (2020). The choice of institutions to solve cooperation problems: a survey of experimental research. *Exp Econ* 23, 716–749. <u>https://doi.org/10.1007/s10683-019-09629-8</u>
- Dannenberg, A., Lange, A., and Sturm, B. (2014). Participation and Commitment in Voluntary Coalitions to Provide Public Goods. *Economica* 81, 257–275. <u>https://doi.org/10.1111/ecca.12073</u>
- De Jesus, A., and Mendonça, S. (2018). Lost in Transition? Drivers and Barriers in the Eco-innovation Road to the Circular Economy. *Ecological Economics* 145, 75–89. <u>https://doi.org/10.1016/j. ecolecon.2017.08.001</u>
- De Vries, A. (2019). Renewable Energy Will Not Solve Bitcoin's Sustainability Problem. *Joule* 3, 893–898. <u>https://doi.org/10.1016/j.joule.2019.02.007</u>
- de Wit, M., Haas, W., Steenmeijer, M., Virág, D., van Barneveld, J., and Verstraeten-Jochemsen, J. (2019). The Circularity Gap Report Austria. Circle Economy, ARA. Available at: <u>https://cdn. prod.website-files.com/5e185aa4d27bcf348400ed82/637266e 4ae79e3f4124a0312\_Circularity%20Gap%20Report%20Austria.pdf</u>
- Dechezleprêtre, A., Fabre, A., Kruse, T., Planterose, B., Chico, A. S., and Stantcheva, S. (2022). Fighting climate change: International attitudes toward climate policies. OECD Publishing, Paris: OECD Economics Department Working Papers. <u>https://doi.org/10.1787/3406f29a-en</u>
- Deisenrieder, V., Kubisch, S., Keller, L., and Stötter, J. (2020). Bridging the Action Gap by Democratizing Climate Change Education—The Case of k.i.d.Z.21 in the Context of Fridays for Future. *Sustainability* 12, 1748. <u>https://doi.org/10.3390/su12051748</u>
- DellaVigna, S., and Linos, E. (2022). RCTs to Scale: Comprehensive Evidence from Two Nudge Units. *ECTA* 90, 81–116. <u>https://doi.org/10.3982/ECTA18709</u>
- Delmas, M. A., Fischlein, M., and Asensio, O. I. (2013). Information strategies and energy conservation behavior: A meta-analysis of experimental studies from 1975 to 2012. *Energy Policy* 61, 729–739. <u>https://doi.org/10.1016/j.enpol.2013.05.109</u>
- Demski, C., Thomas, G., Becker, S., Evensen, D., and Pidgeon, N. (2019). Acceptance of energy transitions and policies: Public

conceptualisations of energy as a need and basic right in the United Kingdom. *Energy Research & Social Science* 48, 33–45. https://doi.org/10.1016/j.erss.2018.09.018

- Deng, H.-M., Liang, Q.-M., Liu, L.-J., and Anadon, L. D. (2018). Co-benefits of greenhouse gas mitigation: a review and classification by type, mitigation sector, and geography. *Environ. Res. Lett.* 12, 123001. https://doi.org/10.1088/1748-9326/aa98d2
- Deschênes, O., and Greenstone, M. (2011). Climate Change, Mortality, and Adaptation: Evidence from Annual Fluctuations in Weather in the US. American Economic Journal: Applied Economics 3, 152–185. https://doi.org/10.1257/app.3.4.152
- Devine-Wright, P. (2011). Public engagement with large-scale renewable energy technologies: breaking the cycle of NIMBYism. WIREs Climate Change 2, 19–26. <u>https://doi.org/10.1002/</u> wcc.89
- Diener, E., Oishi, S., and Tay, L. (2018). Advances in subjective well-being research. *Nat Hum Behav* 2, 253–260. <u>https://doi.org/10.1038/s41562-018-0307-6</u>
- Dietz, T., Ostrom, E., and Stern, P. C. (2003). The Struggle to Govern the Commons. Science 302, 1907–1912. <u>https://doi.org/10.1126/science.1091015</u>
- Directive (EU) 2019/944 (2019). Directive (EU) 2019/944 of the European Parliament and of the Council of 5 June 2019 on common rules for the internal market for electricity and amending Directive 2012/27/EU. Available at: <u>https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX:32019L0944</u>
- Dixon-Fowler, H. R., Slater, D. J., Johnson, J. L., Ellstrand, A. E., and Romi, A. M. (2013). Beyond "Does it Pay to be Green?" A Meta-Analysis of Moderators of the CEP–CFP Relationship. J Bus Ethics 112, 353–366. <u>https://doi.org/10.1007/s10551-012-</u> 1268-8
- Dobber, T., Kruikemeier, S., Votta, F., Helberger, N., and Goodman, E. P. (2023). The effect of traffic light veracity labels on perceptions of political advertising source and message credibility on social media. *Journal of Information Technology & Politics*, 1–16. https://doi.org/10.1080/19331681.2023.2224316
- Dobbins, A., Fuso Nerini, F., Deane, P., and Pye, S. (2019). Strengthening the EU response to energy poverty. *Nature Energy* 4, 2–5. <u>https://doi.org/10.1038/s41560-018-0316-8</u>
- Doblas-Reyes, F. J., Sörensson, A. A., Almazroui, M., Dosio, A., Gutowski, W. J., Haarsma, R., et al. (2021). "Linking Global to Regional Climate Change", in *Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*, eds. V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press), 1363–1512. https://doi.org/10.1017/9781009157896.012
- Dóci, G., and Vasileiadou, E. (2015). "Let's do it ourselves" Individual motivations for investing in renewables at community level. *Renewable and Sustainable Energy Reviews* 49, 41–50. <u>https://</u> <u>doi.org/10.1016/j.rser.2015.04.051</u>
- Donges, J. F., Lucht, W., Müller-Hansen, F., and Steffen, W. (2017). The technosphere in Earth System analysis: A coevolutionary perspective. *The Anthropocene Review* 4, 23–33. <u>https://doi.org/ 10.1177/2053019616676608</u>
- Donsbach, W., Salmon, C. T., and Tsfati, Y. eds. (2014). The Spiral of Silence: New Perspectives on Communication and Public Opinion. New York: Routledge.

- Douenne, T. (2020). The Vertical and Horizontal Distributive Effects of Energy Taxes: A Case Study of a French Policy. *The Energy Journal* 41, 231–254. https://doi.org/10.5547/01956574.41.3.tdou
- Douenne, T., and Fabre, A. (2022). Yellow Vests, Pessimistic Beliefs, and Carbon Tax Aversion. American Economic Journal: Economic Policy 14, 81–110. <u>https://doi.org/10.1257/pol.20200092</u>
- Doyal, L., and Gough, I. (1984). A theory of human needs. *Critical Social Policy* 4, 6–38. <u>https://doi.org/10.1177/02610183840040</u> <u>1002</u>
- Drews, S., Exadaktylos, F., and van den Bergh, J. C. (2020). Assessing synergy of incentives and nudges in the energy policy mix. *Energy Policy* 144, 111605. <u>https://doi.org/10.1016/j.enpol.</u> 2020.111605
- Dudás, L., and Szántó, R. (2021). Nudging in the time of coronavirus? Comparing public support for soft and hard preventive measures, highlighting the role of risk perception and experience. *PLoS ONE* 16, e0256241. <u>https://doi.org/10.1371/journal.pone.0256241</u>
- Dür, M., and Keller, L. (2018). Education for Sustainable Development through International Collaboration. A Case Study on Concepts and Conceptual Change of School-Students from India and Austria on Gender Equality and Sustainable Growth. *Education Sciences* 8, 187. <u>https://doi.org/10.3390/educsci8040</u> 187
- Ebeling, F., and Lotz, S. (2015). Domestic uptake of green energy promoted by opt-out tariffs. *Nature Climate Change* 5, 868–871. https://doi.org/10.1038/nclimate2681
- Ebers Broughel, A., and Hampl, N. (2018). Community financing of renewable energy projects in Austria and Switzerland: Profiles of potential investors. *Energy Policy* 123, 722–736. <u>https:// doi.org/10.1016/j.enpol.2018.08.054</u>
- Ebi, K. L., Vanos, J., Baldwin, J. W., Bell, J. E., Hondula, D. M., Errett, N. A., et al. (2021). Extreme Weather and Climate Change: Population Health and Health System Implications. *Annual Review of Public Health* 42, 293–315. <u>https://doi.org/10.1146/annurev-publhealth-012420-105026</u>
- E-Control (2012). Energy poverty in Austria. Vienna: Energie-Control Austria. Available at: <u>https://www.e-control.at/</u> <u>documents/1785851/1811597/Energiearmut\_Definitionen+</u> <u>und+Indikatoren\_14082013\_en.pdf/e91ed391-719f-469a-</u> <u>b16e-6f74703de96b?t=1438105557837</u>
- Eisenmenger, N., Kaufmann, L., Kalt, G., Dorninger, C., Perkovic, M., Lederer, J., et al. (2022). CO2- und Material-Fußabdruck für Wien: Eine Analyse der nachhaltigen Ressourcennutzung in Wien vor dem Hintergrund der Reduktionsziele in der Smart Klima City Strategie. Wien: Universität für Bodenkultur. Available at: <u>https://www.wien.gv.at/wirtschaft/standort/pdf/mocam-kurz.pdf</u>
- Eisfeld, K., and Seebauer, S. (2022). The energy austerity pitfall: Linking hidden energy poverty with self-restriction in household use in Austria. *Energy Research & Social Science* 84, 102427. <u>https://doi.org/10.1016/j.erss.2021.102427</u>
- Ellen MacArthur Foundation (2019). Artificial intelligence and the circular economy – AI as a tool to accelerate the transition. Available at: <u>https://emf.thirdlight.com/file/24/GgC25OAGB</u> <u>vwdiFGgtzZGVXuZsz/Artificial%20intelligence%20and%20</u> <u>the%20circular%20economy.pdf</u> (Accessed September 9, 2022).
- Elsharkawy, H., and Rutherford, P. (2015). Retrofitting social housing in the UK: Home energy use and performance in a pre-Community Energy Saving Programme (CESP). *Energy and Buildings* 88, 25–33. https://doi.org/10.1016/j.enbuild.2014.11.045

- Endrikat, J., Guenther, E., and Hoppe, H. (2014). Making sense of conflicting empirical findings: A meta-analytic review of the relationship between corporate environmental and financial performance. *European Management Journal* 32, 735–751. <u>https:// doi.org/10.1016/j.emj.2013.12.004</u>
- Energie AG (2021). Group Annual Report 2020/2021. Linz, Austria: Energie AG Oberösterreich. Available at: <u>https://reports.</u> <u>energieag.at/2021/annual-report/\_assets/downloads/gesamtenergieag-ar2021.pdf</u>
- EPAH (2022). National indicators. *Energy Poverty Advisory Hub.* Available at: <u>https://energy-poverty.ec.europa.eu/epah-indica-tors</u>
- Ernst & Young Global Limited (2022). EY-Analyse zur Nachhaltigkeitsberichterstattung in Österreich 2022. Available at: <u>https://www.ey.com/de\_at/news/2022/08/ey-analyse-nhb-oesterreich-2022</u>
- Essletzbichler, J., Miklin, X., Volmary, H., and Ornetzeder, M. (2023). "Kapitel 17. Soziale und räumliche Ungleichheit", in *APCC Special Report: Strukturen für ein klimafreundliches Leben*, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 481–497. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_21</u> (Accessed October 4, 2023).
- European Commission (2015). Flash Eurobarometer. Food waste and date marking. Available at: <u>https://europa.eu/eurobarometer/api/deliverable/download/file?deliverableId=51631</u>
- European Commission (2020). Circular economy action plan. Available at: <u>https://environment.ec.europa.eu/strategy/circu-lar-economy-action-plan\_en</u>
- European Commission (2021). The New EU Urban Mobility Framework. Available at: <u>https://eur-lex.europa.eu/legal-content/EN/</u> <u>TXT/?uri=CELEX%3A52021DC0811&qid=1644666550402</u>
- European Commission (2022). In focus: How can the EU help those touched by energy poverty? Available at: <u>https://commission.europa.eu/news/focus-how-can-eu-help-those-touchedenergy-poverty-2022-02-16\_en</u>
- European Commission (2023). "Annex 8: Fair Transition to Climate Neutrality", in 2023 country report – Austria, eds. European Commission and Directorate-General for Economic and Financial Affairs (Publication Office of the European Union). Available at: https://data.europa.eu/doi/10.2765/650497
- European Commission (2024). EU Building Stock Observatory. Available at: <u>https://building-stock-observatory.energy.ec.euro-pa.eu/database/</u> (Accessed October 28, 2024).
- European Commission (2025). The Just Transition Mechanism: making sure no one is left behind. Available at: <u>https://commission.europa.eu/strategy-and-policy/priorities-2019-2024/</u> <u>european-green-deal/finance-and-green-deal/just-transitionmechanism\_en</u>
- European Commission. Directorate General for Climate Action. (2023). *Climate change: report*. LU: Publications Office. Available at: <u>https://data.europa.eu/doi/10.2834/653431</u> (Accessed February 11, 2025).
- European Commission, Joint Research Centre, Blasco, A., Alt, M., and Gangl, K. (2023). *Energy savings at home and work: Behavioural interventions to tackle the energy crisis*. Publications Office of the European Union. Available at: <u>https://data.europa. eu/doi/10.2760/407488</u> (Accessed October 29, 2024).
- European Environment Agency (2023). Textiles in Europe's circular economy. European Environment Agency. Available at: https://

www.eea.europa.eu/publications/textiles-in-europes-circulareconomy (Accessed October 28, 2024).

- European Investment Bank (2021). European firms and climate change 2020/2021. Evidence from the EIB Investment Survey 2021. Available at: <u>https://www.eib.org/attachments/publica-tions/eibis\_2020\_report\_on\_climate\_change\_en.pdf</u>
- European Investment Bank (2023). What drives firms' investment in climate action? Evidence from the 2022–2023 EIB Investment Survey. Available at: <u>https://www.eib.org/attachments/</u> <u>lucalli/20230114\_what\_drives\_firms\_investment\_in\_climate\_</u> <u>change\_en.pdf</u>
- European Parliamentary Research Service (2021). EU progress on climate action-How are the Member States doing? European Parliamentary Research Service. Available at: <u>https://www. europarl.europa.eu/RegData/etudes/BRIE/2021/696186/</u> EPRS\_BRI(2021)696186\_EN.pdf
- Exadaktylos, F., and van den Bergh, J. (2021). Energy-related behaviour and rebound when rationality, self-interest and willpower are limited. *Nat Energy* 6, 1104–1113. <u>https://doi.org/10. 1038/s41560-021-00889-4</u>
- Falkenberg, M., Galeazzi, A., Torricelli, M., Di Marco, N., Larosa, F., Sas, M., et al. (2022). Growing polarization around climate change on social media. *Nat. Clim. Chang.* 12, 1114–1121. <u>https://doi.org/10.1038/s41558-022-01527-x</u>
- Fanning, A. L., O'Neill, D. W., Hickel, J., and Roux, N. (2022). The social shortfall and ecological overshoot of nations. *Nat Sustain* 5, 26–36. <u>https://doi.org/10.1038/s41893-021-00799-z</u>
- Farmer, J. D., Hepburn, C., Ives, M. C., Hale, T., Wetzer, T., Mealy, P., et al. (2019). Sensitive intervention points in the post-carbon transition. *Science* 364, 132–134. <u>https://doi.org/10.1126/ science.aaw7287</u>
- Farrell, J. (2016). Corporate funding and ideological polarization about climate change. *Proceedings of the National Academy of Sciences* 113, 92–97. <u>https://doi.org/10.1073/pnas.1509433112</u>
- Farrow, K., Grolleau, G., and Mzoughi, N. (2018). Less is more in energy conservation and efficiency messaging. *Energy Policy* 122, 1–6. https://doi.org/10.1016/j.enpol.2018.07.007
- Fell, M. J. (2017). Energy services: A conceptual review. Energy Research & Social Science 27, 129–140. <u>https://doi.org/10.1016/j.erss.2017.02.010</u>
- Ferraro, P. J., Miranda, J. J., and Price, M. K. (2011). The Persistence of Treatment Effects with Norm-Based Policy Instruments: Evidence from a Randomized Environmental Policy Experiment. *American Economic Review* 101, 318–322. <u>https://doi.org/10. 1257/aer.101.3.318</u>
- Ferraro, P. J., and Price, M. K. (2013). Using nonpecuniary strategies to influence behavior: Evidence from a large-scale field experiment. *Review of Economics and Statistics* 95, 64–73. <u>https:// doi.org/10.1162/REST\_a\_00344</u>
- Fina, B., and Monsberger, C. (2023). Measures and business model approaches to facilitate the diffusion and integration of energy communities—A special focus on the Austrian case. *Energy Strategy Reviews* 49, 101161. <u>https://doi.org/10.1016/j.esr.2023.</u> 101161
- Fink, M., Mayrhuber, C., and Rocha-Akis, S. (2022a). Maßnahmenpakete gegen Teuerung: Potentielle Wirkung auf die privaten Haushalte. Vienna: Austrian Institute of Economic Research. https://www.wifo.ac.at/wwa/pubid/69633
- Fink, M., Mayrhuber, C., and Rocha-Akis, S. (2022b). Steigende Lebenshaltungskosten und Armut. Vienna: Austrian Institute of Economic Research. : <u>https://www.wifo.ac.at/wwa/pubid/69632</u>

- Fischer, B., Gutsche, G., and Wetzel, H. (2021). Who wants to get involved? Determining citizen willingness to participate in German renewable energy cooperatives. *Energy Research & Social Science* 76, 102013. <u>https://doi.org/10.1016/j.erss.2021.102013</u>
- Fischer, H., Amelung, D., and Said, N. (2019). The accuracy of German citizens' confidence in their climate change knowledge. *Nat. Clim. Chang.* 9, 776–780. <u>https://doi.org/10.1038/s41558-019-0563-0</u>
- Flachaire, E., Lustig, N., and Vigorito, A. (2022). Underreporting of Top Incomes and Inequality: A Comparison of Correction Methods using Simulations and Linked Survey and Tax Data. *Review of Income and Wealth* 69, 1033–1059. <u>https://doi. org/10.1111/roiw.12618</u>
- Flachsland, C., Pahle, M., Burtraw, D., Edenhofer, O., Elkerbout, M., Fischer, C., et al. (2020). How to avoid history repeating itself: the case for an EU Emissions Trading System (EU ETS) price floor revisited. *Climate Policy* 20, 133–142. <u>https://doi.org/10.1080/14693062.2019.1682494</u>
- Fleurbaey, M. (2009). Beyond GDP: The Quest for a Measure of Social Welfare. *Journal of Economic Literature* 47, 1029–1075. <u>https://doi.org/10.1257/jel.47.4.1029</u>
- Font Vivanco, D., Freire-González, J., Galvin, R., Santarius, T., Walnum, H. J., Makov, T., et al. (2022). Rebound effect and sustainability science: A review. *Journal of Industrial Ecology* 26, 1543–1563. <u>https://doi.org/10.1111/JIEC.13295</u>
- Fouladvand, J., Mouter, N., Ghorbani, A., and Herder, P. (2020). Formation and Continuation of Thermal Energy Community Systems: An Explorative Agent-Based Model for the Netherlands. *Energies* 13, 2829. <u>https://doi.org/10.3390/en13112829</u>
- Franssens, S., Botchway, E., De Swart, W., and Dewitte, S. (2021). Nudging Commuters to Increase Public Transport Use: A Field Experiment in Rotterdam. *Front. Psychol.* 12, 633865. <u>https:// doi.org/10.3389/fpsyg.2021.633865</u>
- Freitag, C., Berners-Lee, M., Widdicks, K., Knowles, B., Blair, G. S., and Friday, A. (2021). The real climate and transformative impact of ICT: A critique of estimates, trends, and regulations. *Patterns* 2, 100340. <u>https://doi.org/10.1016/j.patter.2021.100340</u>
- Frey, B. S., and Oberholzer-Gee, F. (1997). The Cost of Price Incentives: An Empirical Analysis of Motivation Crowding- Out. *The American Economic Review* 87, 746–755.
- Frey, H., Brezina, T., and Emberger, G. (2023). "Kapitel 6. Mobilität", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 271–284. https://doi.org/10.1007/978-3-662-66497-1\_10
- Frieden, D., Tuerk, A., Neumann, C., D'Herbemont, S., and Roberts, J. (2020). Collective self-consumption and energy communities: Trends and challenges in the transposition of the EU framework. <u>https://doi.org/10.13140/RG.2.2.25685.04321</u>
- Friedl, C., and Reichl, J. (2016). Realizing energy infrastructure projects – A qualitative empirical analysis of local practices to address social acceptance. *Energy Policy* 89, 184–193. <u>https:// doi.org/10.1016/j.enpol.2015.11.027</u>
- Fritsche, I., and Masson, T. (2021). Collective climate action: When do people turn into collective environmental agents? *Current Opinion in Psychology* 42, 114–119. <u>https://doi.org/10.1016/j. copsyc.2021.05.001</u>
- Fritze, J. G., Blashki, G. A., Burke, S., and Wiseman, J. (2008). Hope, despair and transformation: Climate change and the promotion of mental health and wellbeing. *International Journal of Mental Health Systems* 2, 13. <u>https://doi.org/10.1186/1752-4458-2-13</u>

- Fuller, R., Landrigan, P. J., Balakrishnan, K., Bathan, G., Bose-O'Reilly, S., Brauer, M., et al. (2022). Pollution and health: a progress update. *The Lancet Planetary Health* 6, e535–e547. https://doi.org/10.1016/S2542-5196(22)00090-0
- Furlong, C., and Vignoles, V. L. (2021). Social Identification in Collective Climate Activism: Predicting Participation in the Environmental Movement, Extinction Rebellion. *Identity* 21, 20–35. <u>https://doi.org/10.1080/15283488.2020.1856664</u>
- Gährs, S., Weiß, J., Bluhm, H., Dunkelberg, E., and Katner, J. (2021). Erkenntnisse zu Umweltwirkungen von Smart Metern: Erfahrungen aus dem Einsatz von Smart Metern in Europa. Dessau-Roßlau: Umweltbundesamt. Available at: <u>https://www. umweltbundesamt.de/sites/default/files/medien/5750/publikationen/2021-05-06\_cc\_34-2021\_umweltwirkungen\_smart\_ meter.pdf</u>
- Gallier, C., Goeschl, T., Kesternich, M., Lohse, J., Reif, C., and Römer, D. (2019). Leveling up? An inter-neighborhood experiment on parochialism and the efficiency of multi-level public goods provision. *Journal of Economic Behavior & Organization* 164, 500–517. <u>https://doi.org/10.1016/j.jebo.2019.05.028</u>
- Galvin, R., and Sunikka-Blank, M. (2018). Economic Inequality and Household Energy Consumption in High-income Countries: A Challenge for Social Science Based Energy Research. *Ecological Economics* 153, 78–88. <u>https://doi.org/10.1016/j.ecolecon.2018.07.003</u>
- Gangl, K., Abstiens, K., Gsottbauer, E., Kirchler, E., Riener, G., Seifert, M., et al. (2022a). Energiekrise was tun? Verhaltenswissenschaftliche Empfehlungen. Vienna, Austria: Institute for Advanced Studies. Available at: <u>https://irihs.ihs.ac.at/id/eprint/6234</u> (Accessed October 29, 2024).
- Gangl, K., Spitzer, F., and Walter, A. (2022b). Abfalltrennung und Littering im öffentlichen Raum. Ein verhaltensökonomisches Feldexperiment. Vienna, Austria: Institute for Advanced Studies. Available at: <u>https://irihs.ihs.ac.at/id/eprint/6097/7/ihsreport-2022-gangl-spitzer-walter-abfalltrennung-littering-imoeffentlichen-raum.pdf</u>
- Gangl, K., Walter, A., and Van Lange, P. A. M. (2022c). Implicit reminders of reputation and nature reduce littering more than explicit information on injunctive norms and monetary costs. *Journal of Environmental Psychology* 84, 101914. <u>https://doi.org/10.1016/j.jenvp.2022.101914</u>
- Gao, J., Kovats, S., Vardoulakis, S., Wilkinson, P., Woodward, A., Li, J., et al. (2018). Public health co-benefits of greenhouse gas emissions reduction: A systematic review. *Science of The Total Environment* 627, 388–402. <u>https://doi.org/10.1016/j.scitotenv.</u> 2018.01.193
- Gao, L., and Wan, L. (2023). Does corporate environmental responsibility contribute to financial performance? A dual path analysis through operational efficiency and the cost of debt. *Corp Soc Responsibility Env* 30, 308–323. https://doi.org/10.1002/csr.2356
- Garmendia, E., and Stagl, S. (2010). Public participation for sustainability and social learning: Concepts and lessons from three case studies in Europe. *Ecological Economics* 69, 1712–1722. https://doi.org/10.1016/j.ecolecon.2010.03.027
- Garnett, E. E., Marteau, T. M., Sandbrook, C., Pilling, M. A., and Balmford, A. (2020). Order of meals at the counter and distance between options affect student cafeteria vegetarian sales. *Nat Food* 1, 485–488. <u>https://doi.org/10.1038/s43016-020-0132-8</u>
- Gasparrini, A., Guo, Y., Sera, F., Vicedo-Cabrera, A. M., Huber, V., Tong, S., et al. (2017). Projections of temperature-related excess mortality under climate change scenarios. *The Lancet*

Planetary Health 1, e360-e367. <u>https://doi.org/10.1016/S2542-5196(17)30156-0</u>

- GBV (2021). Energiekosten und Energiearmut von Mieterinnen und Mietern im gemeinnützigen Wohnungsbestand. Vienna: Austrian Federation of Limited-Profit Housing Assosiations. Available at: <u>https://gbv-aktuell.at/fileadmin/import/images/</u>2021/11/Researchbrief.pdf
- Geiger, N., and Swim, J. K. (2016). Climate of silence: Pluralistic ignorance as a barrier to climate change discussion. *Journal of Environmental Psychology* 47, 79–90. <u>https://doi.org/10.1016/j.jenvp.2016.05.002</u>
- Geiger, N., Swim, J. K., and Fraser, J. (2017). Creating a climate for change: Interventions, efficacy and public discussion about climate change. *Journal of Environmental Psychology* 51, 104–116. <u>https://doi.org/10.1016/j.jenvp.2017.03.010</u>
- Geiger, N., Swim, J. K., Gasper, K., Fraser, J., and Flinner, K. (2021a). How do I feel when I think about taking action? Hope and boredom, not anxiety and helplessness, predict intentions to take climate action. *Journal of Environmental Psychology* 76, 101649. https://doi.org/10.1016/j.jenvp.2021.101649
- Geiger, S. J., Brick, C., Nalborczyk, L., Bosshard, A., and Jostmann, N. B. (2021b). More green than gray? Toward a sustainable overview of environmental spillover effects: A Bayesian meta-analysis. *Journal of Environmental Psychology* 78, 101694. <u>https://doi. org/10.1016/j.jenvp.2021.101694</u>
- Gemeinschaft schafft Energie (2023). Österreichische Koordinationsstelle für Energiegemeinschaften. Available at: https://energiegemeinschaften.gv.at/ (Accessed September 14, 2023).
- GeSI (2020). #SMARTer2030 ICT Solutions for 21<sup>st</sup> Century Challenges. Brussels, Belgium: Global e-Sustainability initiative. Available at: <u>http://smarter2030.gesi.org</u>.
- Ghesla, C., Grieder, M., Schmitz, J., and Stadelmann, M. (2020). Pro-environmental incentives and loss aversion: A field experiment on electricity saving behavior. *Energy Policy* 137, 111131. <u>https://doi.org/10.1016/j.enpol.2019.111131</u>
- Ghisellini, P., Ripa, M., and Ulgiati, S. (2018). Exploring environmental and economic costs and benefits of a circular economy approach to the construction and demolition sector. A literature review. *Journal of Cleaner Production* 178, 618–643. <u>https://doi.org/10.1016/j.jclepro.2017.11.207</u>
- Giljum, S. (2018). Die Notwendigkeit einer konsumbezogenen Betrachtung der Treibhausgasemissionen Österreichs. CCCA Fact Sheet #21. Available at: <u>https://ccca.ac.at/fileadmin/00\_DokumenteHauptmenue/02\_Klimawissen/FactSheets/21\_konsumbasierte\_Treibhausgasemissionen.pdf</u>
- Gillingham, K., Rapson, D., and Wagner, G. (2016). The Rebound Effect and Energy Efficiency Policy. *Review of Environmental Economics and Policy* 10, 68–88. <u>https://doi.org/10.1093/reep/ rev017</u>
- Gneezy, U., Meier, S., and Rey-Biel, P. (2011). When and Why Incentives (Don't) Work to Modify Behavior. *Journal of Economic Perspectives* 25, 191–210. <u>https://doi.org/10.1257/jep.25.4.191</u>
- Gneezy, U., and Rustichini, A. (2000). A Fine is a Price. *The Journal* of Legal Studies 29, 1–17. <u>https://doi.org/10.1086/468061</u>
- Gough, I. (2015). Climate change and sustainable welfare: the centrality of human needs. Cambridge Journal of Economics 39, 1191–1214. <u>https://doi.org/10.1093/cje/bev039</u>
- Gouveia, C. (2019). Best practices in STEP project countries: Report. Available at: <u>https://www.stepenergy.eu/wp-content/uploads/2020/08/STEP-847080.-Best-practices-in-STEP-projectcountries.-D2.4.pdf</u>

- Gowdy, J. (2020). Our hunter-gatherer future: Climate change, agriculture and uncivilization. *Futures* 115, 102488. <u>https://doi.org/10.1016/j.futures.2019.102488</u>
- Graff Zivin, J., Hsiang, S. M., and Neidell, M. (2018). Temperature and Human Capital in the Short and Long Run. Journal of the Association of Environmental and Resource Economists 5, 77–105. <u>https://doi.org/10.1086/694177</u>
- Graff Zivin, J., Song, Y., Tang, Q., and Zhang, P. (2020). Temperature and high-stakes cognitive performance: Evidence from the national college entrance examination in China. *Journal of Environmental Economics and Management* 104, 102365. <u>https://doi.org/10.1016/j.jeem.2020.102365</u>
- Gravert, C., and Kurz, V. (2021). Nudging à la carte: a field experiment on climate-friendly food choice. *Behav. Public Policy* 5, 378–395. <u>https://doi.org/10.1017/bpp.2019.11</u>
- Gravert, C., and Olsson Collentine, L. (2021). When nudges aren't enough: Norms, incentives and habit formation in public transport usage. *Journal of Economic Behavior & Organization* 190, 1–14. <u>https://doi.org/10.1016/j.jebo.2021.07.012</u>
- Greenpeace (2020). Klima Ungerechtigkeit in Österreich: Eine Studie zur ungleichen Verteilung von CO2-Ausstoss nach Einkommensschichten. Wien: Greenpeace in Zentral- und Osteuropa. Available at: <u>https://mobilitaet.greenpeace.at/assets/uploads/assets/uploads/GP\_ReportKlimaungerechtigkeitAT.pdf</u> <u>?\_gl=1\*008zr6\*\_gcl\_au\*MTQxMDg1MzY2OC4xNzM4NjU3</u> NjUy\*\_ga\*OTgzNDcxNzcxLjE3Mzg2NTc2NTI.\*\_ga YKYRVRWTCR\*MTczODY1NzY1MS4xLjAuMTczODY1Nz Y1MS42MC4wLjA.</u>
- Grübler, A. (1998). Technology and Global Change., 1<sup>st</sup> Edn. Cambridge University Press. <u>https://doi.org/10.1017/CBO978131</u> 6036471
- Grubler, A., Wilson, C., Bento, N., Boza-Kiss, B., Krey, V., McCollum, D. L., et al. (2018). A low energy demand scenario for meeting the 1.5 °C target and sustainable development goals without negative emission technologies. *Nature Energy* 3, 515– 527. <u>https://doi.org/10.1038/s41560-018-0172-6</u>
- Hackbarth, A., and Löbbe, S. (2022). "What motivates private households to participate in energy communities? A literature review and German case study", in *Energy Communities*, (Elsevier), 153–166. <u>https://doi.org/10.1016/B978-0-323-91135-1.00026-2</u>
- Hagmann, D., Ho, E. H., and Loewenstein, G. (2019). Nudging out support for a carbon tax. *Nat. Clim. Chang.* 9, 484–489. <u>https:// doi.org/10.1038/s41558-019-0474-0</u>
- Hajek, P., Siegl, A., and Dziendziel, N. (2024). Wahltagsbefragung NRW 2024: Wählermotive, -strukturen und -ströme. Vienna, Austria: Peter Hajek Public Opinion Strategies. Available at: <u>https://www.peterhajek.at/wp-content/uploads/2024/09/</u> PULS24\_ATV\_Wahltagsbefragung\_NRW2024.pdf
- Hammi, B., Zeadally, S., Khatoun, R., and Nebhen, J. (2022). Survey on Smart Homes: Vulnerabilities, Risks, and Countermeasures. *Comput. Secur.* 117, 102677. <u>https://doi.org/10.1016/j.cose.2022.</u> 102677
- Hampton, S., and Whitmarsh, L. (2023). Choices for climate action: A review of the multiple roles individuals play. *One Earth* 6, 1157–1172. <u>https://doi.org/10.1016/j.oneear.2023.08.006</u>
- Handgraaf, M. J. J., Van Lidth De Jeude, M. A., and Appelt, K. C. (2013). Public praise vs. private pay: Effects of rewards on energy conservation in the workplace. *Ecological Economics* 86, 86–92. <u>https://doi.org/10.1016/j.ecolecon.2012.11.008</u>

- Hansen, P. G., Schilling, M., and Malthesen, M. S. (2021). Nudging healthy and sustainable food choices: three randomized controlled field experiments using a vegetarian lunch-default as a normative signal. *Journal of Public Health* 43, 392–397. <u>https:// doi.org/10.1093/pubmed/fdz154</u>
- Hartl, B., and Hofmann, E. (2019). "Sharing economy", in A Research Agenda for Economic Psychology, eds. K. Gangl and E. Kirchler (Edward Elgar Publishing). <u>https://doi.org/10.4337/ 9781788116060.00016</u>
- Hartl, B., and Hofmann, E. (2022). The social dilemma of car sharing – The impact of power and the role of trust in community car sharing. *International Journal of Sustainable Transportation* 16, 526–540. <u>https://doi.org/10.1080/15568318.2021.1912224</u>
- Hartl, B., Kamleitner, B., and Holub, S. (2020). Take me on a ride: The role of environmentalist identity for carpooling. *Psychology* and Marketing 37, 663–676. <u>https://doi.org/10.1002/mar.21340</u>
- Hartl, B., Marth, S., Hofmann, E., and Penz, E. (2024). 'One bad apple won't spoil the bunch' – Representations of the "sharing economy." *Journal of Business Research* 176, 114580. <u>https://doi. org/10.1016/j.jbusres.2024.114580</u>
- Hasson, R., Löfgren, Å., and Visser, M. (2010). Climate change in a public goods game: Investment decision in mitigation versus adaptation. *Ecological Economics* 70, 331–338. <u>https://doi.org/10.1016/j.ecolecon.2010.09.004</u>
- Head, L., and Harada, T. (2017). Keeping the heart a long way from the brain: The emotional labour of climate scientists. *Emotion, Space and Society* 24, 34–41. <u>https://doi.org/10.1016/j.emospa.2017.07.005</u>
- Heeman, J., Faassen, E., Rogulj, I., Pizzini, G., Anagnostopoulos, F., Oikonomou, V., et al. (2022). Status of energy poverty and policies to address it in CEE/SEE countries. Amsterdam: Institute for European Energy and Climate Policy Stichting – IEECP. Available at: <u>https://build-up.ec.europa.eu/sites/default/files/ content/ieecp\_policies-to-decarbonise-residential-buildings\_ ws1\_report\_2022.pdf</u>
- Heinrichs, H. (2013). Sharing Economy: A Potential New Pathway to Sustainability. GAIA – Ecological Perspectives for Science and Society 22, 228–231. <u>https://doi.org/10.14512/gaia.22.4.5</u>
- Heinz, S., Otto, I. M., Tan, R., Jin, Y., and Glebe, T. W. (2022). Cooperation Enhances Adaptation to Environmental Uncertainty: Evidence from Irrigation Behavioral Experiments in South China. *Water* 14, 1098. <u>https://doi.org/10.3390/w14071098</u>
- Helliwell, J. F., Layard, R., Sachs, J. D., Neve, J.-E. D., Aknin, L. B., and Wang, S. (2023). World Happiness Report 2023. New York: Sustainable Development Solutions Network. Available at: <u>https://worldhappiness.report/ed/2023/</u> (Accessed August 31, 2023).
- Helliwell, J. F., Layard, R., Sachs, J. D., Neve, J.-E. D., Aknin, L. B., and Wang, S. eds. (2024). World Happiness Report 2024. Oxford: University of Oxford: Wellbeing Research Centre. Available at: <u>https://worldhappiness.report/ed/2024/</u> (Accessed August 31, 2023).
- Henkel, C., and Kranz, J. (2018). Pro-Environmental Behavior and Green Information Systems Research – Review, Synthesis and Directions for Future Research. *ICIS 2018 Proceedings*. Available at: <u>https://aisel.aisnet.org/icis2018/green/Presentations/3</u> (Accessed September 12, 2022).
- Hensher, M. (2023). Climate change, health and sustainable healthcare: The role of health economics. *Health Economics* 32, 985– 992. https://doi.org/10.1002/hec.4656

- Herrmann-Pillath, C. (2018). The Case for a New Discipline: Technosphere Science. *Ecological Economics* 149, 212–225. <u>https:// doi.org/10.1016/j.ecolecon.2018.03.024</u>
- Hickman, C., Marks, E., Pihkala, P., Clayton, S., Lewandowski, R. E., Mayall, E. E., et al. (2021). Climate anxiety in children and young people and their beliefs about government responses to climate change: a global survey. *The Lancet Planetary Health* 5, e863–e873. <u>https://doi.org/10.1016/S2542-5196(21)00278-3</u>
- Hittinger, E., and Jaramillo, P. (2019). Internet of things: Energy boon or bane? *Science* 364, 326–328. <u>https://doi.org/10.1126/ science.aau8825</u>
- Hölbling, S., Kirchengast, G., and Danzer, J. (2023). Unmasking mobility patterns: international travel behavior and emissions of scientists in a higher research institution. *IJSHE* 24, 355–371. <u>https://doi.org/10.1108/IJSHE-03-2023-0081</u>
- Holmgren, J. (2020). The effect of public transport quality on car ownership – A source of wider benefits? *Research in Transportation Economics* 83, 100957. <u>https://doi.org/10.1016/j.re-</u> <u>trec.2020.100957</u>
- Holzer, D., Popowicz, M., Rauter, R., Silberschneider, K., and Stern, T. (2023). Parallel universes, one circular goal: An empirical study comparing Austrian wood- and plastic-based industries. *Sustainable Production and Consumption* 43, 46–61. <u>https://doi. org/10.1016/j.spc.2023.10.014</u>
- Hook, A., Court, V., Sovacool, B. K., and Sorrell, S. (2020). A systematic review of the energy and climate impacts of teleworking. *Environmental Research Letters* 15, 093003. <u>https://doi.org/10.1088/1748-9326/ab8a84</u>
- Hoosain, M. S., Paul, B. S., Raza, S. M., and Ramakrishna, S. (2020).
   "Material Passports and Circular Economy", in *An Introduction to Circular Economy*, (Singapore: Springer), 131–158. <u>https://doi.org/10.1007/978-981-15-8510-4\_8</u>
- Hornsey, M. J., Harris, E. A., Bain, P. G., and Fielding, K. S. (2016). Meta-analyses of the determinants and outcomes of belief in climate change. *Nature Clim Change* 6, 622–626. <u>https://doi. org/10.1038/nclimate2943</u>
- Hörtenhuber, S., Lindenthal, T., Amon, B., Markut, T., Kirner, L., and Zollitsch, W. (2010). Greenhouse gas emissions from selected Austrian dairy production systems—model calculations considering the effects of land use change. *Renewable Agriculture and Food Systems* 25, 316–329. <u>https://doi.org/10.1017/ S1742170510000025</u>
- Hössinger, R., Peer, S., and Juschten, M. (2023). Give citizens a task: An innovative tool to compose policy bundles that reach the climate goal. *Transportation Research Part A: Policy and Practice* 173, 103694. <u>https://doi.org/10.1016/j.tra.2023.103694</u>
- Howe, L. C., MacInnis, B., Krosnick, J. A., Markowitz, E. M., and Socolow, R. (2019). Acknowledging uncertainty impacts public acceptance of climate scientists' predictions. *Nat. Clim. Chang.* 9, 863–867. <u>https://doi.org/10.1038/s41558-019-0587-5</u>
- Hrabok, M., Delorme, A., and Agyapong, V. I. O. (2020). Threats to Mental Health and Well-Being Associated with Climate Change. *Journal of Anxiety Disorders* 76, 102295. <u>https://doi. org/10.1016/j.janxdis.2020.102295</u>
- Hubacek, K., Baiocchi, G., Feng, K., and Patwardhan, A. (2017). Poverty eradication in a carbon constrained world. *Nat Commun* 8, 912. <u>https://doi.org/10.1038/s41467-017-00919-4</u>
- Huckebrink, D., and Bertsch, V. (2021). Integrating Behavioural Aspects in Energy System Modelling—A Review. *Energies* 14, 4579. <u>https://doi.org/10.3390/en14154579</u>

- Hummel, D., and Maedche, A. (2019). How effective is nudging? A quantitative review on the effect sizes and limits of empirical nudging studies. *Journal of Behavioral and Experimental Economics* 80, 47–58. <u>https://doi.org/10.1016/j.socec.2019.03.005</u>
- IEA (2020). Austria 2020: Energy Policy Review. Paris: IEA. Available at: <u>https://www.iea.org/reports/austria-2020</u>
- IIBW (2016). Third United Nations Conference on Housing and Sustainable Urban Development (HABITAT III). Case Study – The Austrian System of Social Housing., (Quito). Available at: <u>https://www.bmeia.gv.at/fileadmin/user\_upload/Habitat\_III\_ Case\_Study\_Austrian\_System\_of\_Social\_Housing.pdf</u>
- Ilmola-Sheppard, L., Strelkovskii, N., Rovenskaya, E., Abramzon, S., and Bar, R. (2020). A systems description of the national well-being system. Version 1.0. Laxenburg, Austria: International Institute for Applied Systems Analysis. <u>https://pure.iiasa. ac.at/16318</u>
- International Commission on the Futures of Education (2021). *Reimagining our futures together: a new social contract for education*. Paris: Educational and Cultural Organization of the United Nations. <u>https://doi.org/10.54675/ASRB4722</u>
- IPCC (2018). Global Warming of 1.5°C. An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty., eds. V. Masson-Delmotte, P. Zhai, H.-O. Pörtner, D. Roberts, J. Skea, P. R. Shukla, et al. Cambridge, UK and New York, NY, USA: Cambridge University Press. <u>https://doi. org/10.1017/9781009157940</u>
- IPCC (2022). Climate Change 2022 Mitigation of Climate Change: Working Group III Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change., 1<sup>st</sup> Edn, eds.
   P. R. Shukla, J. Skea, R. Slade, A. A. Khourdajie, A. Hasija, J. Malley, et al. Cambridge, UK and New York, NY, USA: Cambridge University Press. <u>https://doi.org/10.1017/9781009157926</u>
- Irshaid, J., Mochizuki, J., and Schinko, T. (2021). Challenges to local innovation and implementation of low-carbon energy-transition measures: A tale of two Austrian regions. *Energy Policy* 156, 112432. <u>https://doi.org/10.1016/j.enpol.2021.112432</u>
- Ivanova, D., Barrett, J., Wiedenhofer, D., Macura, B., Callaghan, M., and Creutzig, F. (2020). Quantifying the potential for climate change mitigation of consumption options. *Environ. Res. Lett.* 15, 093001. <u>https://doi.org/10.1088/1748-9326/ab8589</u>
- Ivanova, D., and Middlemiss, L. (2021). Characterizing the energy use of disabled people in the European Union towards inclusion in the energy transition. *Nature Energy* 6, 1188–1197. <u>https:// doi.org/10.1038/s41560-021-00932-4</u>
- Ivanović, V., Šimac, B., and Trako Poljak, T. (2022). Subjective Wellbeing in Rural and Urban Central Europe: Evidence from the European Social Survey (2008 and 2018). *International Journal of Sociology* 52, 397–419. <u>https://doi.org/10.1080/00207659.</u> 2022.2089806
- Jagers, S. C., Lachapelle, E., Martinsson, J., and Matti, S. (2021). Bridging the ideological gap? How fairness perceptions mediate the effect of revenue recycling on public support for carbon taxes in the United States, Canada and Germany. *Review of Policy Research* 38, 529–554. <u>https://doi.org/10.1111/ropr.12439</u>
- Jagers, S. C., Martinsson, J., and Matti, S. (2019). The impact of compensatory measures on public support for carbon taxation: an experimental study in Sweden. *Climate Policy* 19, 147–160. <u>https://doi.org/10.1080/14693062.2018.1470963</u>

- Jans, L. (2021). Changing environmental behaviour from the bottom up: The formation of pro-environmental social identities. *Journal of Environmental Psychology* 73, 101531. <u>https://doi.org/ 10.1016/j.jenvp.2020.101531</u>
- Janusch, N., Kroll, S., Goemans, C., Cherry, T. L., and Kallbekken, S. (2021). Learning to accept welfare-enhancing policies: an experimental investigation of congestion pricing. *Exp Econ* 24, 59–86. <u>https://doi.org/10.1007/s10683-020-09650-2</u>
- Jany, A., Bukowski, M., Heindl, G., and Kreissl, K. (2023). "Kapitel
  4. Wohnen", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy,
  A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 227–244. <u>https://doi.org/10.1007/978-3-662-66497-1\_8</u>
- Jensen, L., and Carvalho Fachada, J. (2021). Climate action in Austria: Latest state of play. European Parliamentary Research Service. Available at: <u>https://www.europarl.europa.eu/RegData/</u> <u>etudes/BRIE/2021/696186/EPRS\_BRI(2021)696186\_EN.pdf</u>
- John, P., Martin, A., and Mikołajczak, G. (2023). Support for behavioral nudges versus alternative policy instruments and their perceived fairness and efficacy. *Regulation & Governance* 17, 363–371. https://doi.org/10.1111/rego.12460
- Jones, C. I., and Klenow, P. J. (2016). Beyond GDP? Welfare across Countries and Time. American Economic Review 106, 2426– 2457. <u>https://doi.org/10.1257/aer.20110236</u>
- Jones, C., and Kammen, D. M. (2014). Spatial Distribution of U.S. Household Carbon Footprints Reveals Suburbanization Undermines Greenhouse Gas Benefits of Urban Population Density. *Environ. Sci. Technol.* 48, 895–902. <u>https://doi.org/10.1021/es 4034364</u>
- Jugert, P., Greenaway, K. H., Barth, M., Büchner, R., Eisentraut, S., and Fritsche, I. (2016). Collective efficacy increases pro-environmental intentions through increasing self-efficacy. *Journal of Environmental Psychology* 48, 12–23. <u>https://doi.org/10.1016/j.jenvp.2016.08.003</u>
- Jung, J. Y., and Mellers, B. A. (2016). American attitudes toward nudges. Judgm. decis. mak. 11, 62–74. <u>https://doi.org/10.1017/ S1930297500007592</u>
- Kadi, J., and Matznetter, W. (2022). The long history of gentrification in Vienna, 1890–2020. *City* 26, 450–472. <u>https://doi.org/ 10.1080/13604813.2022.2054221</u>
- Kaestner, K., and Vance, C. (2022). On Home Energy Reports and Boomerangs: Evidence from Austria., (Kiel, Hamburg: ZBW – Leibniz Information Centre for Economics). Available at: <u>https://hdl.handle.net/10419/264122</u> (Accessed October 29, 2024).
- Kahan, D. M., Peters, E., Wittlin, M., Slovic, P., Ouellette, L. L., Braman, D., et al. (2012). The polarizing impact of science literacy and numeracy on perceived climate change risks. *Nature Clim Change* 2, 732–735. <u>https://doi.org/10.1038/nclimate1547</u>
- Kahneman, D., Knetsch, J. L., and Thaler, R. H. (1991). Anomalies: The Endowment Effect, Loss Aversion, and Status Quo Bias. *Journal of Economic Perspectives* 5, 193–206. <u>https://doi.org/10.1257/jep.5.1.193</u>
- Kalkbrenner, B. J., and Roosen, J. (2016). Citizens' willingness to participate in local renewable energy projects: The role of community and trust in Germany. *Energy Research & Social Science* 13, 60–70. <u>https://doi.org/10.1016/j.erss.2015.12.006</u>
- Kallbekken, S., Kroll, S., and Cherry, T. L. (2011). Do you not like Pigou, or do you not understand him? Tax aversion and revenue recycling in the lab. *Journal of Environmental Eco*-

nomics and Management 62, 53-64. <u>https://doi.org/10.1016/j.jeem.2010.10.006</u>

- Kaplowitz, S. A., and McCright, A. M. (2015). Effects of policy characteristics and justifications on acceptance of a gasoline tax increase. *Energy Policy* 87, 370–381. <u>https://doi.org/10.1016/j.enpol.2015.08.037</u>
- Kappes, A., Harvey, A. H., Lohrenz, T., Montague, P. R., and Sharot, T. (2020). Confirmation bias in the utilization of others' opinion strength. *Nat Neurosci* 23, 130–137. <u>https://doi.org/10.1038/ s41593-019-0549-2</u>
- Karlsson, M., Alfredsson, E., and Westling, N. (2020). Climate policy co-benefits: a review. *Climate Policy* 20, 292–316. <u>https://doi. org/10.1080/14693062.2020.1724070</u>
- Karlsson, M., and Ziebarth, N. R. (2018). Population health effects and health-related costs of extreme temperatures: Comprehensive evidence from Germany. *Journal of Environmental Economics and Management* 91, 93–117. <u>https://doi.org/10.1016/j.jeem.2018.06.004</u>
- Keller, L., Riede, M., Link, S., Hüfner, K., and Stötter, J. (2022). Can Education Save Money, Energy, and the Climate?—Assessing the Potential Impacts of Climate Change Education on Energy Literacy and Energy Consumption in the Light of the EU Energy Efficiency Directive and the Austrian Energy Efficiency Act. Energies 15, 1118. https://doi.org/10.3390/en15031118
- Kemp, L., Xu, C., Depledge, J., Ebi, K. L., Gibbins, G., Kohler, T. A., et al. (2022). Climate Endgame: Exploring catastrophic climate change scenarios. *Proc. Natl. Acad. Sci. U.S.A.* 119, e2108146119. <u>https://doi.org/10.1073/pnas.2108146119</u>
- Kerner, C., Thaller, A., and Brudermann, T. (2023). Carbon dioxide removal to combat climate change? An expert survey on perception and support. *Environ. Res. Commun.* 5, 041003. <u>https:// doi.org/10.1088/2515-7620/accc72</u>
- Kezar, A. J. (2001). Understanding and facilitating organizational change in the 21<sup>st</sup> century : recent research and conceptualizations. San Francisco: Jossey-Bass. Available at: <u>https://files.eric.ed.gov/fulltext/ED457711.pdf</u>
- Khan, S. A. R., Zaman, K., and Zhang, Y. (2016). The relationship between energy-resource depletion, climate change, health resources and the environmental Kuznets curve: Evidence from the panel of selected developed countries. *Renewable and Sustainable Energy Reviews* 62, 468–477. <u>https://doi.org/10.1016/j. rser.2016.04.061</u>
- Khanna, T. M., Baiocchi, G., Callaghan, M., Creutzig, F., Guias, H., Haddaway, N. R., et al. (2021). A multi-country meta-analysis on the role of behavioural change in reducing energy consumption and CO2 emissions in residential buildings. *Nat Energy* 6, 925–932. <u>https://doi.org/10.1038/s41560-021-00866-x</u>
- Kiesnere, A., and Baumgartner, R. (2019). Sustainability Management in Practice: Organizational Change for Sustainability in Smaller Large-Sized Companies in Austria. Sustainability 11, 572. <u>https://doi.org/10.3390/su11030572</u>
- Kikstra, J. S., Mastrucci, A., Min, J., Riahi, K., and Rao, N. D. (2021). Decent living gaps and energy needs around the world. *Environ. Res. Lett.* 16, 095006. <u>https://doi.org/10.1088/1748-9326/ac1c27</u>
- Kirchherr, J., Piscicelli, L., Bour, R., Kostense-Smit, E., Muller, J., Huibrechtse-Truijens, A., et al. (2018). Barriers to the Circular Economy: Evidence From the European Union (EU). *Ecological Economics* 150, 264–272. <u>https://doi.org/10.1016/j.ecolecon.</u> 2018.04.028
- Klaassen, E., and Van der Laan, M. (2019). USEF White Paper. Energy & Flexibility Services for Citizens Energy Commu-

nities. USEF. Available at: <u>https://www.usef.energy/app/up-loads/2019/02/USEF-White-Paper-Energy-and-Flexibility-Services-for-Citizens-Energy-Communities-final-CM.pdf</u>

- Klenert, D., Mattauch, L., Combet, E., Edenhofer, O., Hepburn, C., Rafaty, R., et al. (2018). Making carbon pricing work for citizens. *Nature Clim Change* 8, 669–677. <u>https://doi.org/10.1038/</u> s41558-018-0201-2
- Kletzan, D., Köppl, A., Kratena, K., Schleicher, S., and Wüger, M. (2006). Towards sustainable consumption: Economic modelling of mobility and heating for Austria. *Ecological Economics* 57, 608–626. <u>https://doi.org/10.1016/j.ecolecon.2005.05.014</u>
- Klien, M., and Streicher, G. (2021). Ökonomische Wirkungen des gemeinnützigen Wohnbaus. Vienna: Österreichisches Institut für Wirtschaftsforschung. Available at: <u>https://www.wifo.ac.at/</u> <u>publication/55838/</u>
- Koch, J., and Vringer, K. (2023). Hoe 'circulair' zijn Nederlandse consumenten? Een empirisch onderzoek naar gedrag, bereidheid en de potentiële milieuwinst van circulair consumeren. Den Haag: PBL Planbureau voor de Leefomgeving.
- Komendantova, N. (2021). Transferring awareness into action: A meta-analysis of the behavioral drivers of energy transitions in Germany, Austria, Finland, Morocco, Jordan and Iran. *Energy Research & Social Science* 71, 101826. <u>https://doi.org/10.1016/j.erss.2020.101826</u>
- Komendantova, N., Riegler, M., and Neumueller, S. (2018). Of transitions and models: Community engagement, democracy, and empowerment in the Austrian energy transition. *Energy Research & Social Science* 39, 141–151. <u>https://doi.org/10.1016/j.</u> <u>erss.2017.10.031</u>
- Köppl, A., and Schleicher, S. (2018). What Will Make Energy Systems Sustainable? Sustainability 10, 2537. <u>https://doi.org/ 10.3390/su10072537</u>
- Kordzadeh, N., and Ghasemaghaei, M. (2022). Algorithmic bias: review, synthesis, and future research directions. *European Journal of Information Systems* 31, 388–409. <u>https://doi.org/10.1080/ 0960085X.2021.1927212</u>
- Kormos, C., and Gifford, R. (2014). The validity of self-report measures of proenvironmental behavior: A meta-analytic review. *Journal of Environmental Psychology* 40, 359–371. <u>https://doi.org/10.1016/j.jenvp.2014.09.003</u>
- Kouhizadeh, M., and Sarkis, J. (2018). Blockchain Practices, Potentials, and Perspectives in Greening Supply Chains. Sustainability 10. https://doi.org/10.3390/su10103652
- KPMG (2022). Big shifts, small steps: Survey of Sustainability Reporting 2022. KPMG International. Available at: <u>https://assets. kpmg.com/content/dam/kpmg/xx/pdf/2023/04/big-shiftssmall-steps.pdf</u>
- Kranz, J., Kolbe, L. M., Koo, C., and Boudreau, M. C. (2015). Smart energy: where do we stand and where should we go? *Electronic Markets* 25, 7–16. <u>https://doi.org/10.1007/s12525-015-0180-3</u>
- Kranzl, L., Müller, A., Schipfer, F., Smet, K., Grabner, D., Litschauer, K., et al. (2020). Transitioning buildings to full reliance on renewable energy and assuring inclusive and affordable housing. ACRP project Decarb Inclusive final report. (No. 1). Vienna: Klima- und Energiefonds. Available at: <u>https://repositum.tuwien.at/handle/20.500.12708/40258</u>
- Krekel, C., MacKerron, G., and Helliwell, J. F. (2020). "How Environmental Quality Affects Our Happiness", in *World Happiness Report 2020*, (Sustainable Development Solutions Network), 95–112. Available at: <u>https://www.jstor.org/stable/resrep25851.8</u> (Accessed August 30, 2023).

- Krisam, M., Maier, M., Janßen, R., and Krisam, J. (2021). What do Germans really think about health-nudges? *BMC Public Health* 21, 821. https://doi.org/10.1186/s12889-021-10808-7
- Krogh, A. H., and Lo, C. (2023). Robust emergency management: The role of institutional trust in organized volunteers. *Public Administration* 101, 142–157. <u>https://doi.org/10.1111/padm.12894</u>
- Kuhlicke, C., Seebauer, S., Hudson, P., Begg, C., Bubeck, P., Dittmer, C., et al. (2020). The behavioral turn in flood risk management, its assumptions and potential implications. *WIREs Water* 7, e1418. <u>https://doi.org/10.1002/wat2.1418</u>
- Kulmer, V., and Seebauer, S. (2019). How robust are estimates of the rebound effect of energy efficiency improvements? A sensitivity analysis of consumer heterogeneity and elasticities. *Energy Policy* 132, 1–14. <u>https://doi.org/10.1016/j.enpol.2019.05.001</u>
- Küppers, A. (2024). 'Climate-Soviets,' 'Alarmism,' and 'Eco-Dictatorship': The Framing of Climate Change Scepticism by the Populist Radical Right Alternative for Germany. *German Politics* 33, 1–21. <u>https://doi.org/10.1080/09644008.2022.2056596</u>
- Kurz, T., Prosser, A. M. B., Rabinovich, A., and O'Neill, S. (2020). Could Vegans and Lycra Cyclists be Bad for the Planet? Theorizing the Role of Moralized Minority Practice Identities in Processes of Societal-Level Change. *Journal of Social Issues* 76, 86–100. <u>https://doi.org/10.1111/josi.12366</u>
- Kuttner, M., Feldbauer-Durstmüller, B., and Mitter, C. (2021). Corporate social responsibility in Austrian family firms: socioemotional wealth and stewardship insights from a qualitative approach. *JFBM* 11, 238–253. <u>https://doi.org/10.1108/JFBM-04-2019-0028</u>
- Kyprianou, I., Serghides, D. K., Varo, A., Gouveia, J. P., Kopeva, D., and Murauskaite, L. (2019). Energy poverty policies and measures in 5 EU countries: A comparative study. *Energy and Buildings* 196, 46–60. <u>https://doi.org/10.1016/j.enbuild.2019.05.003</u>
- Laa, B., Shibayama, T., Brezina, T., Schönfelder, S., Damjanovic, D., Szalai, E., et al. (2022). A nationwide mobility service guarantee for Austria: possible design scenarios and implications. *Eur. Transp. Res. Rev.* 14, 25. <u>https://doi.org/10.1186/s12544-022-00550-5</u>
- Lage, J., Thema, J., Zell-Ziegler, C., Best, B., Cordroch, L., and Wiese, F. (2023). Citizens call for sufficiency and regulation — A comparison of European citizen assemblies and National Energy and Climate Plans. *Energy Research & Social Science* 104, 103254. <u>https://doi.org/10.1016/j.erss.2023.103254</u>
- Lahey, S. (2023). Austrian Airlines Found Guilty in Greenwashing Case over Carbon Neutral Flight Claims. *ESG Today*. Available at: <u>https://www.esgtoday.com/austrian-airlines-found-guiltyin-greenwashing-over-carbon-neutral-flight-claims/</u>
- Lake, I. R., Hooper, L., Abdelhamid, A., Bentham, G., Boxall, A. B. A., Draper, A., et al. (2012). Climate Change and Food Security: Health Impacts in Developed Countries. *Environmental Health Perspectives* 120, 1520–1526. <u>https://doi.org/10.1289/ ehp.1104424</u>
- Landmann, H., and Rohmann, A. (2020). Being moved by protest: Collective efficacy beliefs and injustice appraisals enhance collective action intentions for forest protection via positive and negative emotions. *Journal of Environmental Psychology* 71, 101491. <u>https://doi.org/10.1016/j.jenvp.2020.101491</u>
- Lange, S., Kern, F., Peuckert, J., and Santarius, T. (2021). The Jevons paradox unravelled: A multi-level typology of rebound effects and mechanisms. *Energy Research and Social Science* 74, 101982. <u>https://doi.org/10.1016/j.erss.2021.101982</u>

- Lange, S., Santarius, T., Dencik, L., Diez, T., Ferreboeuf, H., Hankey, S., et al. (2023). Digital Reset, Redirecting Technologies for the Deep Sustainability Transformation. München: oekom Verlag. https://doi.org/10.14512/9783987262463
- Lawrance, E. L., Thompson, R., Newberry Le Vay, J., Page, L., and Jennings, N. (2022). The Impact of Climate Change on Mental Health and Emotional Wellbeing: A Narrative Review of Current Evidence, and its Implications. *International Review of Psychiatry* 34, 443–498. <u>https://doi.org/10.1080/09540261.2022.21</u> 28725
- Lazer, D. M. J., Baum, M. A., Benkler, Y., Berinsky, A. J., Greenhill, K. M., Menczer, F., et al. (2018). The science of fake news. *Sci*ence 359, 1094–1096. <u>https://doi.org/10.1126/science.aao2998</u>
- Legault, L., Bird, S., Powers, S. E., Sherman, A., Schay, A., Hou, D., et al. (2020). Impact of a Motivational Intervention and Interactive Feedback on Electricity and Water Consumption: A Smart Housing Field Experiment. *Environment and Behavior* 52, 666– 692. <u>https://doi.org/10.1177/0013916518811433</u>
- Lehdonvirta, V. (2022). Cloud Empires: How Digital Platforms Are Overtaking the State and How We Can Regain Control. Cambridge, Massachusetts; London, England: The MIT Press. https://doi.org/10.7551/mitpress/14219.001.0001
- Lenton, T. M., Held, H., Kriegler, E., Hall, J. W., Lucht, W., Rahmstorf, S., et al. (2008). Tipping elements in the Earth's climate system. *Proc. Natl. Acad. Sci. U.S.A.* 105, 1786–1793. <u>https://doi. org/10.1073/pnas.0705414105</u>
- Lockwood, M. (2011). Does the framing of climate policies make a difference to public support? Evidence from UK marginal constituencies. *Climate Policy* 11, 1097–1112. <u>https://doi.org/10. 1080/14693062.2011.579301</u>
- Lohmann, P. M., Gsottbauer, E., Doherty, A., and Kontoleon, A. (2022). Do carbon footprint labels promote climatarian diets? Evidence from a large-scale field experiment. *Journal of Environmental Economics and Management* 114, 102693. <u>https://doi.org/10.1016/j.jeem.2022.102693</u>
- Loibl, C., Sunstein, C. R., Rauber, J., and Reisch, L. A. (2018). Which Europeans like nudges? Approval and controversy in four European countries. *Journal of Consumer Affairs* 52, 655– 688. https://doi.org/10.1111/joca.12181
- Lord, C., Bates, O., Friday, A., McLeod, F., Cherrett, T., Martinez-Sykora, A., et al. (2022). The sustainability of the gig economy food delivery system (Deliveroo, UberEATS and Just-Eat): Histories and futures of rebound, lock-in and path dependency. *International Journal of Sustainable Transportation* 17, 490–502. https://doi.org/10.1080/15568318.2022.2066583
- Lucas, K., Marsden, G., Brooks, M., and Kimble, M. (2007). Assessment of Capabilities for Examining Long-Term Social Sustainability of Transport and Land Use Strategies. *Transportation Research Record* 2013, 30–37. https://doi.org/10.3141/2013-05
- Lutz, W., Striessnig, E., Dimitrova, A., Ghislandi, S., Lijadi, A., Reiter, C., et al. (2021). Years of good life is a well-being indicator designed to serve research on sustainability. *Proceedings of the National Academy of Sciences* 118, e1907351118. <u>https://doi.org/ 10.1073/pnas.1907351118</u>
- Lwasa, S., Seto, K. C., Bai, X., Blanco, H., Gurney, K. R., Kılkış, Ş., et al. (2023). "Urban Systems and Other Settlements", in *IPCC*, 2022: Climate Change 2022 Mitigation of Climate Change: Working Group III Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, eds. P. R. Shukla, J. Skea, R. Slade, A. Al Khourdajie, R. van Diemen, D. McCollum, et al. (Cambridge, UK and New York, NY, USA:

Cambridge University Press), 861–952. <u>https://doi.org/10.1017/</u> 9781009157926.010

- Ma, T., Moore, J., and Cleary, A. (2022). Climate change impacts on the mental health and wellbeing of young people: A scoping review of risk and protective factors. *Social Science & Medicine* 301, 114888. <u>https://doi.org/10.1016/j.socscimed.2022.114888</u>
- MacKenzie, M. K., and Caluwaerts, D. (2021). Paying for the future: deliberation and support for climate action policies. *Journal of Environmental Policy & Planning* 23, 317–331. <u>https://doi.org/10.1080/1523908X.2021.1883424</u>
- Maier, M., Bartoš, F., Stanley, T. D., Shanks, D. R., Harris, A. J. L., and Wagenmakers, E.-J. (2022). No evidence for nudging after adjusting for publication bias. *Proc. Natl. Acad. Sci. U.S.A.* 119, e2200300119. <u>https://doi.org/10.1073/pnas.2200300119</u>
- Maki, A., Burns, R. J., Ha, L., and Rothman, A. J. (2016). Paying people to protect the environment: A meta-analysis of financial incentive interventions to promote proenvironmental behaviors. *Journal of Environmental Psychology* 47, 242–255. <u>https:// doi.org/10.1016/j.jenvp.2016.07.006</u>
- Malmodin, J. (2020). The power consumption of mobile and fixed network data services – The case of streaming video and downloading large files., in *Electronics Goes Green*, (Berlin: Fraunhofer IZM), 87–96.
- Malmodin, J., and Lundén, D. (2018). The energy and carbon footprint of the global ICT and E & M sectors 2010–2015. Sustainability 10, 3027. <u>https://doi.org/10.3390/su10093027</u>
- Marshall, T. M. (2020). Risk perception and safety culture: Tools for improving the implementation of disaster risk reduction strategies. *International Journal of Disaster Risk Reduction* 47, 101557. https://doi.org/10.1016/j.ijdrr.2020.101557
- Martin, G., Reilly, K., Everitt, H., and Gilliland, J. A. (2022). Review: The impact of climate change awareness on children's mental well-being and negative emotions a scoping review. *Child and Adolescent Mental Health* 27, 59–72. <u>https://doi.org/10.1111/camh.12525</u>
- Masiero, G., Mazzonna, F., and Santarossa, M. (2022). The effect of absolute versus relative temperature on health and the role of social care. *Health Economics* 31, 1228–1248. <u>https://doi.org/ 10.1002/hec.4507</u>
- Masson, T., and Fritsche, I. (2021). We need climate change mitigation and climate change mitigation needs the 'We': a stateof-the-art review of social identity effects motivating climate change action. *Current Opinion in Behavioral Sciences* 42, 89– 96. https://doi.org/10.1016/j.cobeha.2021.04.006
- Mathiesen, B. V., Lund, H., Connolly, D., Wenzel, H., Østergaard, P. A., Möller, B., et al. (2015). Smart Energy Systems for coherent 100% renewable energy and transport solutions. *Applied Energy* 145, 139–154. <u>https://doi.org/10.1016/j.apenergy.2015.01.075</u>
- Matzinger, S., Heizmann, K., and David, E. (2018). Studie zur Eruierung einer Definition von Energiearmut in Österreich aus Sicht der sozialwirtschaftlichen und energiewirtschaftlichen Praxis. Vienna: Federal Ministry of Social Affairs, Health, Care and Consumer Protection. Available at: www.sozialministerium.at/dam/jcr:d7d414ea-a6c5-49db-9b83-ac2e806b6e98/studie\_energiearmut\_endversion.pdf
- Max-Neef, M. (1991). Human Scale Development: Conception, Application and Further Reflections. New York: Zed Books Ltd.
- Mayer, J., Dugan, A., Bachner, G., and Steininger, K. W. (2021). Is carbon pricing regressive? Insights from a recursive-dynamic CGE analysis with heterogeneous households for Austria.

Energy Economics 104, 105661. <u>https://doi.org/10.1016/j.ene-</u>co.2021.105661

- McBride, S. E., Hammond, M. D., Sibley, C. G., and Milfont, T. L. (2021). Longitudinal relations between climate change concern and psychological wellbeing. *Journal of Environmental Psycholo*gy 78, 101713. <u>https://doi.org/10.1016/j.jenvp.2021.101713</u>
- McCright, A. M., and Dunlap, R. E. (2011). The Politicization of Climate Change and Polarization in the American Public's Views of Global Warming, 2001–2010. *The Sociological Quarterly* 52, 155–194. <u>https://doi.org/10.1111/j.1533-8525.2011.01198.x</u>
- McDonagh, S. (2021). These are the six most vegetarian cities in Europe. *euronews*. Available at: <u>https://www.euronews.com/green/2021/04/10/these-are-the-six-most-vegetarian-cities-in-europe</u> (Accessed September 13, 2023).
- McEvoy, D. M., Haller, T., and Blanco, E. (2022). The Role of Non-Binding Pledges in Social Dilemmas with Mitigation and Adaptation. *Environ Resource Econ* 81, 685–710. <u>https://doi.org/10.1007/s10640-021-00645-y</u>
- Meadows, D. (1999). Leverage Points: Places to Intervene in a System. Hartland, VT, USA: The Sustainability Institute. Available at: <u>https://www.donellameadows.org/wp-content/userfiles/ Leverage\_Points.pdf</u> (Accessed January 17, 2025).
- Mehrwert Berlin (2022). Wer radelt am meisten? Available at: https://www.wer-radelt-am-meisten.de/#faq
- Meier, J., Andor, M. A., Doebbe, F. C., Haddaway, N. R., and Reisch, L. A. (2022). Review: Do green defaults reduce meat consumption? *Food Policy* 110, 102298. <u>https://doi.org/10.1016/j.foodpol.2022.102298</u>
- Meinhart, B., Gabelberger, F., Sinabell, F., and Streicher, G. (2022). Transformation und "Just Transition" in Österreich. Wien: Österreichisches Institut für Wirtschaftsforschung (WIFO). Available at: https://www.wifo.ac.at/jart/prj3/wifo/resources/ person\_dokument/person\_dokument.jart?publikationsid= 68029&mime\_type=application/pdf
- Mertens, S., Herberz, M., Hahnel, U. J. J., and Brosch, T. (2022). The effectiveness of nudging: A meta-analysis of choice architecture interventions across behavioral domains. *Proc. Natl. Acad. Sci. U.S.A.* 119, e2107346118. <u>https://doi.org/10.1073/</u> pnas.2107346118
- Michie, S., Van Stralen, M. M., and West, R. (2011). The behaviour change wheel: A new method for characterising and designing behaviour change interventions. *Implementation Sci* 6, 42. <u>https://doi.org/10.1186/1748-5908-6-42</u>
- Middlemiss, L. (2022). Who is vulnerable to energy poverty in the Global North, and what is their experience? *Wiley Interdisciplinary Reviews: Energy and Environment* 11, e455. <u>https://doi.org/10.1002/wene.455</u>
- Mignon, I., and Bergek, A. (2016). System- and actor-level challenges for diffusion of renewable electricity technologies: an international comparison. *Journal of Cleaner Production* 128, 105–115. https://doi.org/10.1016/j.jclepro.2015.09.048
- Mildenberger, M., Lachapelle, E., Harrison, K., and Stadelmann-Steffen, I. (2022). Limited impacts of carbon tax rebate programmes on public support for carbon pricing. *Nat. Clim. Chang.* 12, 141–147. <u>https://doi.org/10.1038/s41558-021-012</u> <u>68-3</u>
- Mildenberger, M., Lubell, M., and Hummel, M. (2019). Personalized risk messaging can reduce climate concerns. *Global Envi*ronmental Change 55, 15–24. <u>https://doi.org/10.1016/j.gloenvcha.2019.01.002</u>

- Millward-Hopkins, J., Steinberger, J. K., Rao, N. D., and Oswald, Y. (2020). Providing decent living with minimum energy: A global scenario. *Global Environmental Change* 65, 102168. <u>https://doi.org/10.1016/j.gloenvcha.2020.102168</u>
- Mock, M. (2024). The temporalities of shared mobility practices from an environmental perspective. Soziologie und Nachhaltigkeit Sonderband 3. <u>https://doi.org/10.17879/sun-2024-</u> 5272
- Mock, M., Omann, I., Polzin, C., Spekkink, W., Schuler, J., Pandur, V., et al. (2019). "Something inside me has been set in motion": Exploring the psychological wellbeing of people engaged in sustainability initiatives. *Ecological Economics* 160, 1–11. <u>https://doi.org/10.1016/j.ecolecon.2019.02.002</u>
- Molina-Azorín, J. F., Claver-Cortés, E., López-Gamero, M. D., and Tarí, J. J. (2009). Green management and financial performance: a literature review. *Management Decision* 47, 1080–1100. <u>https:// doi.org/10.1108/00251740910978313</u>
- Morley, J., Widdicks, K., and Hazas, M. (2018). Digitalisation, energy and data demand: The impact of Internet traffic on overall and peak electricity consumption. *Energy Research & Social Science* 38, 128–137. <u>https://doi.org/10.1016/j.erss.2018.01.018</u>
- Mort, H. (2019). A Review of Energy and Gender Research in the Global North. Vienna, Austria: Technical University of Vienna. Available at: <u>https://www.tuwien.at/fileadmin/Assets/dienstleister/abteilung\_genderkompetenz/gender\_in\_der\_Forschung/GEECCO\_Results/Additional\_resources\_and\_literature\_reviews/GEECCO\_WP6\_Literatur\_Review\_Energy\_ Gender\_Research\_Global\_North.pdf</u>
- Moser, S. C. (2010). Communicating climate change: history, challenges, process and future directions. WIREs Climate Change 1, 31–53. <u>https://doi.org/10.1002/wcc.11</u>
- Mulalic, I., and Rouwendal, J. (2020). Does improving public transport decrease car ownership? Evidence from a residential sorting model for the Copenhagen metropolitan area. *Regional Science and Urban Economics* 83, 103543. <u>https://doi.org/10.1016/</u> j.regsciurbeco.2020.103543
- Muñoz, P., and Steininger, K. W. (2010). Austria's CO2 responsibility and the carbon content of its international trade. *Ecological Economics* 69, 2003–2019. <u>https://doi.org/10.1016/j.ecolecon.2010.05.017</u>
- Muñoz, P., Zwick, S., and Mirzabaev, A. (2020). The impact of urbanization on Austria's carbon footprint. *Journal of Cleaner Production* 263, 121326. <u>https://doi.org/10.1016/j.jclepro.2020.</u> <u>121326</u>
- Murray, A., Skene, K., and Haynes, K. (2017). The Circular Economy: An Interdisciplinary Exploration of the Concept and Application in a Global Context. J Bus Ethics, 369–380. <u>https://doi.org/10.1007/s10551-015-2693-2</u>
- Nabernegg, S. (2021). Emission distribution and incidence of national mitigation policies among households in Austria. Graz: University of Graz, Department of Economics. Available at: https://unipub.uni-graz.at/obvugrveroeff/id/9638164
- Nabernegg, S., Bednar-Friedl, B., Muñoz, P., Titz, M., and Vogel, J. (2019). National Policies for Global Emission Reductions: Effectiveness of Carbon Emission Reductions in International Supply Chains. *Ecological Economics* 158, 146–157. <u>https://doi. org/10.1016/j.ecolecon.2018.12.006</u>
- Nabernegg, S., and Otto, I. (2025). Social tipping dynamics for reaching net zero in Austria.

- Nabernegg, S., Steininger, K. W., and Lackner, T. (2023). Consumption- and production-based emissions: Updates for Austria. Graz: Wegener Center Verlag.
- Neubauer, M. (2020). Umweltbedingungen, Umweltverhalten 2019. Ergebnisse des Mikrozensus. Wien: Statistik Austria. Available at: https://www.statistik.at/fileadmin/publications/Umweltbedingungen\_Umweltverhalten\_2019\_Ergebnisse\_des\_ Mikrozensus.pdf
- Newell, P., Twena, M., and Daley, F. (2021). Scaling behaviour change for a 1.5-degree world: challenges and opportunities. *Glob. Sustain.*, e22. <u>https://doi.org/10.1017/sus.2021.23</u>
- Nicolson, M., Huebner, G. M., Shipworth, D., and Elam, S. (2017). Tailored emails prompt electric vehicle owners to engage with tariff switching information. *Nat Energy* 2, 17073. <u>https://doi.org/10.1038/nenergy.2017.73</u>
- Niedderer, K., Mackrill, J., Clune, S., Lockton, D., Ludden, G., Morris, A., et al. (2014). Creating Sustainable Innovation through Design for Behaviour Change: Full Project Report. <u>https://doi.org/10.13140/2.1.4817.4409</u>
- Nielsen, K. S., Clayton, S., Stern, P. C., Dietz, T., Capstick, S., and Whitmarsh, L. (2021). How psychology can help limit climate change. *American Psychologist* 76, 130–144. <u>https://doi. org/10.1037/amp0000624</u>
- Nielsen, K. S., Stern, P. C., Dietz, T., Gilligan, J. M., Van Vuuren, D. P., Figueroa, M. J., et al. (2020a). Improving Climate Change Mitigation Analysis: A Framework for Examining Feasibility. One Earth 3, 325–336. <u>https://doi.org/10.1016/j.oneear.2020.08.007</u>
- Nielsen, K. S., Van Der Linden, S., and Stern, P. C. (2020b). How Behavioral Interventions Can Reduce the Climate Impact of Energy Use. *Joule* 4, 1613–1616. <u>https://doi.org/10.1016/j.joule.</u> 2020.07.008
- Niinistö, S. (2024). Safer Together: Strengthening Europe's Civilian and Military Preparedness and Readiness. Available at: <u>https:// commission.europa.eu/document/download/5bb2881f-9e29-42f2-8b77-8739b19d047c\_en?filename=2024\_Niinisto-report\_ Book\_VEpdf</u>
- Nikravech, M., Kwan, V., Dobernig, K., Wilhelm-Rechmann, A., and Langen, N. (2020). Limiting food waste via grassroots initiatives as a potential for climate change mitigation: a systematic review. *Environ. Res. Lett.* 15, 123008. <u>https://doi. org/10.1088/1748-9326/aba2fe</u>
- Nisa, C. F., Bélanger, J. J., Schumpe, B. M., and Faller, D. G. (2019). Meta-analysis of randomised controlled trials testing behavioural interventions to promote household action on climate change. *Nature communications* 10, 4545. <u>https://doi.org/ 10.1038/s41467-019-12457-2</u>
- Nohrstedt, D., Mazzoleni, M., Parker, C. F., and Di Baldassarre, G. (2021). Exposure to natural hazard events unassociated with policy change for improved disaster risk reduction. *Nat Commun* 12, 193. <u>https://doi.org/10.1038/s41467-020-20435-2</u>
- Nordhaus, W. (2019). Climate Change: The Ultimate Challenge for Economics. American Economic Review 109, 1991–2014. <u>https://doi.org/10.1257/aer.109.6.1991</u>
- Nordhaus, W. D. (2017). Revisiting the social cost of carbon. *Proc. Natl. Acad. Sci. U.S.A.* 114, 1518–1523. <u>https://doi.org/10.1073/</u> pnas.1609244114
- Nussbaum, M., and Sen, A. (1993). The Quality of Life. Oxford; New York: Oxford University Press.
- O'Brien, E., and Klein, N. (2017). The tipping point of perceived change: Asymmetric thresholds in diagnosing improvement

versus decline. Journal of Personality and Social Psychology 112, 161–185. https://doi.org/10.1037/pspa0000070

- O'Brien, W., and Yazdani Aliabadi, F. (2020). Does telecommuting save energy? A critical review of quantitative studies and their research methods. *Energy and Buildings* 225, 110298. <u>https:// doi.org/10.1016/j.enbuild.2020.110298</u>
- Odyssee-MURE (2024). Energy efficiency trends and policies. Austria Energy Profile. Available at: <u>https://www.odyssee-mure.</u> <u>eu/publications/efficiency-trends-policies-profiles/austria.html</u> (Accessed October 28, 2024).
- Odyssee-MURE project (2021). Sectoral Profile Transport. Available at: <u>https://www.odyssee-mure.eu/publications/efficiency-by-sector/transport/transport-eu.pdf</u>
- OECD (2020). How's Life? 2020: Measuring Well-being. Paris: OECD Publishing. https://doi.org/10.1787/9870c393-en
- Ogunbode, C. A., Doran, R., Hanss, D., Ojala, M., Salmela-Aro, K., van den Broek, K. L., et al. (2022). Climate anxiety, wellbeing and pro-environmental action: correlates of negative emotional responses to climate change in 32 countries. *Journal of Environmental Psychology* 84, 101887. <u>https://doi.org/10.1016/j.jenvp.2022.101887</u>
- OIB (2020). OIB Guideline 6: Energy conservation and thermal protection. Vienna, Austria: Austrian Institute of Construction Engineering (OIB). Available at: <u>https://energy.ec.</u> <u>europa.eu/document/download/84846ac4-c2fa-43bf-baad-20952579e2b7\_en?filename=at\_2020\_ltrs\_en\_0.pdf</u> (Accessed February 21, 2025).
- O'Neil, S. G. (2021). Community obstacles to large scale solar: NIMBY and renewables. J Environ Stud Sci 11, 85–92. <u>https://doi.org/10.1007/s13412-020-00644-3</u>
- Österreichs E-Wirtschaft (2023). Maßnahmen zur Abfederung von Energiearmut. Available at: <u>https://oesterreichsenergie.</u> <u>at/publikationen/ueberblick/detailseite/massnahmen-zur-abfederung-von-energiearmut</u> (Accessed April 4, 2024).
- Ostrom, E., Burger, J., Field, C. B., Norgaard, R. B., and Policansky, D. (1999). Revisiting the Commons: Local Lessons, Global Challenges. *Science* 284, 278–282. <u>https://doi.org/10.1126/science.284.5412.278</u>
- O'Sullivan, K. C., Howden-Chapman, P., Sim, D., Stanley, J., Rowan, R. L., Clark, I. K. H., et al. (2017). Cool? Young people investigate living in cold housing and fuel poverty. A mixed methods action research study. SSM-Population Health 3, 66–74. <u>https:// doi.org/10.1016/j.ssmph.2016.12.006</u>
- Oswald, Y., Owen, A., and Steinberger, J. K. (2020). Large inequality in international and intranational energy footprints between income groups and across consumption categories. *Nat Energy* 5, 231–239. <u>https://doi.org/10.1038/s41560-020-0579-8</u>
- Otrachshenko, V., Popova, O., and Tavares, J. (2021). Extreme Temperature and Extreme Violence: Evidence from Russia. *Economic Inquiry* 59, 243–262. <u>https://doi.org/10.1111/ecin.12936</u>
- Otto, I. M., Donges, J. F., Cremades, R., Bhowmik, A., Hewitt, R. J., Lucht, W., et al. (2020a). Social tipping dynamics for stabilizing Earth's climate by 2050. *Proc. Natl. Acad. Sci. U.S.A.* 117, 2354–2365. https://doi.org/10.1073/pnas.1900577117
- Otto, I. M., Kim, K. M., Dubrovsky, N., and Lucht, W. (2019). Shift the focus from the super-poor to the super-rich. *Nature Clim Change* 9, 82–84. <u>https://doi.org/10.1038/s41558-019-0402-3</u>
- Otto, I. M., Wiedermann, M., Cremades, R., Donges, J. F., Auer, C., and Lucht, W. (2020b). Human agency in the Anthropocene. *Ecological Economics* 167, 106463. <u>https://doi.org/10.1016/j. ecolecon.2019.106463</u>

- Palma, A., Petrunyk, I., and Vuri, D. (2022). Prenatal air pollution exposure and neonatal health. *Health Economics* 31, 729–759. https://doi.org/10.1002/hec.4474
- Pantano, E., and Stylos, N. (2020). The *Cinderella* moment: Exploring consumers' motivations to engage with renting as collaborative luxury consumption mode. *Psychology and Marketing* 37, 740–753. https://doi.org/10.1002/mar.21345
- Papantonis, D., Tzani, D., Burbidge, M., Stavrakas, V., Bouzarovski, S., and Flamos, A. (2022). How to improve energy efficiency policies to address energy poverty? Literature and stakeholder insights for private rented housing in Europe. *Energy Research & Social Science* 93, 102832. <u>https://doi.org/10.1016/j.erss.2022.102832</u>
- Parguel, B., Lunardo, R., and Benoit-Moreau, F. (2017). Sustainability of the sharing economy in question: When second-hand peer-to-peer platforms stimulate indulgent consumption. *Technological Forecasting and Social Change* 125, 48–57. <u>https://doi.org/10.1016/j.techfore.2017.03.029</u>
- Park, R. J. (2022). Hot Temperature and High-Stakes Performance. Journal of Human Resources 57, 400–434. <u>https://doi.org/10.3368/jhr.57.2.0618-9535R3</u>
- Park, R. J., Behrer, A. P., and Goodman, J. (2021). Learning is inhibited by heat exposure, both internationally and within the United States. *Nat Hum Behav* 5, 19–27. <u>https://doi.org/10.1038/</u> s41562-020-00959-9
- Park, R. J., Goodman, J., Hurwitz, M., and Smith, J. (2020). Heat and Learning. *American Economic Journal: Economic Policy* 12, 306–339. https://doi.org/10.1257/pol.20180612
- Pathak, M., Slade, R., Shukla, P. R., Skea, J., Pichs-Madruga, R., and Ürge-Vorsatz, D. (2022). "Technical Summary", in Climate Change 2022 – Mitigation of Climate Change: Working Group III Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, eds. P. R. Shukla, J. Skea, R. Slade, R. Al Khourdajie, R. van Diemen, D. McCollum, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press), 51–148. https://doi.org/10.1017/9781009157926.002
- Pearsall, H., Gutierrez-Velez, V. H., Gilbert, M. R., Hoque, S., Eakin, H., Brondizio, E. S., et al. (2021). Advancing equitable health and well-being across urban-rural sustainable infrastructure systems. *npj Urban Sustain* 1, 26. <u>https://doi.org/10.1038/s42949-021-00028-8</u>
- Penker, M., Brunner, K.-M., and Plank, C. (2023). "Kapitel 5. Ernährung", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 245–269. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_9</u> (Accessed October 4, 2023).
- Peper, S., and Failla, M.-C. (2021). Sinfonia. Comparison of measured and calculated energy demand and assessment of the results. Available at: <u>https://passiv.de/downloads/05\_sinfonia\_monioring\_innsbruck\_final\_report.pdf</u>
- Perino, G. (2018). New EU ETS Phase 4 rules temporarily puncture waterbed. *Nature Clim Change* 8, 262–264. <u>https://doi.org/10.1038/s41558-018-0120-2</u>
- Perino, G. (2019). Reply: EU ETS and the waterbed effect. *Nat. Clim. Chang.* 9, 736–736. <u>https://doi.org/10.1038/s41558-019-0580-z</u>
- Peschel, A. O., and Aschemann-Witzel, J. (2020). Sell more for less or less for more? The role of transparency in consumer response to upcycled food products. *Journal of Cleaner Production* 273, 122884. <u>https://doi.org/10.1016/j.jclepro.2020.122884</u>

- Petrov, I., and Ryan, L. (2021). The landlord-tenant problem and energy efficiency in the residential rental market. *Energy Policy* 157, 112458. <u>https://doi.org/10.1016/j.enpol.2021.112458</u>
- Petrova, M. A. (2013). NIMBYism revisited: public acceptance of wind energy in the United States. WIREs Climate Change 4, 575–601. <u>https://doi.org/10.1002/wcc.250</u>
- Petrova, S., and Simcock, N. (2021). Gender and energy: domestic inequities reconsidered. Social & Cultural Geography 22, 849– 867. https://doi.org/10.1080/14649365.2019.1645200
- Pfeifer, L., and Otto, I. M. (2023). Changing seasonal temperature offers a window of opportunity for stricter climate policy. *Environmental Science & Policy* 140, 35–45. <u>https://doi.org/10.1016/j.envsci.2022.11.010</u>
- Pianta, S., Rinscheid, A., and Weber, E. U. (2021). Carbon Capture and Storage in the United States: Perceptions, preferences, and lessons for policy. *Energy Policy* 151, 112149. <u>https://doi. org/10.1016/j.enpol.2021.112149</u>
- Pichler, C., and Steininger, K. W. (2019). Das Treibhausgasbudget für die Stadt Graz: Die aktuell der Stadt Graz zuzurechnenden Treibhausgas-Emissionen als Basis für sowohl Mitigationspfade als auch die Zurechnung des Carbon Budgets für Graz. Wissenschaftlicher Bericht 84-2019. Graz: Wegener Center Verlag. Available at: <u>https://wegcwww.uni-graz.at/publ/wegcreports/2019/WCV-WissBer-Nr84-CPichlerKSteininger-Nov2019.pdf</u>
- Pohl, J., Frick, V., Hoefner, A., Santarius, T., and Finkbeiner, M. (2021). Environmental saving potentials of a smart home system from a life cycle perspective: How green is the smart home? *Journal of Cleaner Production* 312, 127845. <u>https://doi. org/10.1016/J.JCLEPRO.2021.127845</u>
- Pohl, J., Gabriel, J., and Hübner, G. (2018). Understanding stress effects of wind turbine noise – The integrated approach. *Energy Policy* 112, 119–128. <u>https://doi.org/10.1016/j.enpol.2017.</u> 10.007
- Poore, J., and Nemecek, T. (2018). Reducing food's environmental impacts through producers and consumers. *Science* 360, 987– 992. <u>https://doi.org/10.1126/science.aaq0216</u>
- Porter, M. E., and Linde, C. V. D. (1995). Toward a New Conception of the Environment-Competitiveness Relationship. *Journal* of Economic Perspectives 9, 97–118. <u>https://doi.org/10.1257/jep.</u> 9.4.97
- Potting, J., Hekkert, M. P., Worrell, E., and Hanemaaijer, A. (2017). Circular Economy: Measuring Innovation in the Product Chain. Den Haage: PBL Netherlands Environmental Assessment Agency. Available at: <u>https://dspace.library.uu.nl/handle/ 1874/358310</u> (Accessed May 7, 2024).
- Pouri, M. J. (2021). Eight impacts of the digital sharing economy on resource consumption. *Resources, Conservation and Recycling* 168, 105434. <u>https://doi.org/10.1016/j.resconrec.2021.105434</u>
- Preston, S. H. (2007). The changing relation between mortality and level of economic development. *International Journal of Epidemiology* 36, 484–490. <u>https://doi.org/10.1093/ije/dym075</u>
- Primc, K., Dominko, M., and Slabe-Erker, R. (2021). 30 years of energy and fuel poverty research: A retrospective analysis and future trends. *Journal of Cleaner Production* 301, 127003. <u>https://</u> doi.org/10.1016/j.jclepro.2021.127003
- Räisänen, J., Ojala, A., and Tuovinen, T. (2021). Building trust in the sharing economy: Current approaches and future considerations. *Journal of Cleaner Production* 279, 123724. <u>https://doi.org/10.1016/j.jclepro.2020.123724</u>

- Rao, N. D., and Min, J. (2018a). Decent Living Standards: Material Prerequisites for Human Wellbeing. *Soc Indic Res* 138, 225–244. https://doi.org/10.1007/s11205-017-1650-0
- Rao, N. D., and Min, J. (2018b). Less global inequality can improve climate outcomes. Wiley Interdisciplinary Reviews: Climate Change 9, e513. <u>https://doi.org/10.1002/wcc.513</u>
- Rao, N. D., Min, J., and Mastrucci, A. (2019). Energy requirements for decent living in India, Brazil and South Africa. *Nat Energy* 4, 1025–1032. <u>https://doi.org/10.1038/s41560-019-0497-9</u>
- Rao, N. D., and Wilson, C. (2022). Advancing energy and well-being research. *Nat Sustain* 5, 98–103. <u>https://doi.org/10.1038/ s41893-021-00775-7</u>
- Rauch, F., and Steiner, R. (2006). School development through Education for Sustainable Development in Austria. *Environmental Education Research* 12, 115–127. <u>https://doi.org/10.1080/ 13504620500527782</u>
- Ravallion, M., Heil, M., and Jalan, J. (2000). Carbon emissions and income inequality. Oxford Economic Papers 52, 651–669. <u>https://econpapers.repec.org/RePEc:oup:oxecpp:v:52:y:2000:i:</u> <u>4:p:651-69</u>
- Rechnungshof Österreich (2020). Energiewirtschaftliche Maßnahmen gegen Energiearmut. Bericht des Rechnungshofes. Wien. Available at: <u>https://www.rechnungshof.gv.at/rh/home/ home/Energiearmut.pdf</u>
- Rege, M., and Telle, K. (2004). The impact of social approval and framing on cooperation in public good situations. *Journal* of Public Economics 88, 1625–1644. <u>https://doi.org/10.1016/ S0047-2727(03)00021-5</u>
- Reichl, J., Cohen, J. J., Klöckner, C. A., Kollmann, A., and Azarova, V. (2021). The drivers of individual climate actions in Europe. *Global Environmental Change* 71, 102390. <u>https://doi.org/10.1016/j.gloenvcha.2021.102390</u>
- Reisch, L. A., and Sunstein, C. R. (2016). Do Europeans like nudges? Judgm. decis. mak. 11, 310–325. <u>https://doi.org/10.1017/ S1930297500003740</u>
- Reisch, L. A., Sunstein, C. R., and Gwozdz, W. (2017). Viewpoint: Beyond carrots and sticks: Europeans support health nudges. *Food Policy* 69, 1–10. <u>https://doi.org/10.1016/j.foodpol.2017.01.007</u>
- Reisch, L. A., Sunstein, C. R., and Kaiser, M. (2021). What do people want to know? Information avoidance and food policy implications. *Food Policy* 102, 102076. <u>https://doi.org/10.1016/</u> j.foodpol.2021.102076
- Reischl, C., Rauter, R., and Posch, A. (2018). Urban vulnerability and adaptation to heatwaves: a case study of Graz (Austria). *Climate Policy* 18, 63–75. <u>https://doi.org/10.1080/ 14693062.2016.1227953</u>
- Remenick, L. (2018). The Role of Communication in Preparation for Wildland Fire: A Literature Review. *Environmental Communication* 12, 164–176. <u>https://doi.org/10.1080/17524032.2017. 1346519</u>
- REScoop.EU (2019). Q&A: What are 'citizen' and 'renewable' energy communities? Available at: <u>https://www.rescoop.eu/uploads/</u> <u>rescoop/downloads/QA-What-are-citizens-energy-communi-</u> <u>ties-renewable-energy-communities-in-the-CEP.pdf</u> (Accessed May 7, 2024).
- Reynolds, J. P., Archer, S., Pilling, M., Kenny, M., Hollands, G. J., and Marteau, T. M. (2019). Public acceptability of nudging and taxing to reduce consumption of alcohol, tobacco, and food: A population-based survey experiment. *Social Science & Medicine* 236, 112395. <u>https://doi.org/10.1016/j.socscimed.2019.112395</u>

- Richards, C. E., Lupton, R. C., and Allwood, J. M. (2021). Re-framing the threat of global warming: an empirical causal loop diagram of climate change, food insecurity and societal collapse. *Climatic Change* 164, 49. <u>https://doi.org/10.1007/s10584-021-02957-w</u>
- Rising, J., Tedesco, M., Piontek, F., and Stainforth, D. A. (2022). The missing risks of climate change. *Nature* 610, 643–651. <u>https://doi.org/10.1038/s41586-022-05243-6</u>
- Roberto, R., Ferruzzi, G., Negro, V., and Noussan, M. (2023). Mapping of Energy Community Development in Europe: State of the Art and Research Directions. *Energies* 16, 6554. <u>https://doi.org/10.3390/en16186554</u>
- Robinson, C. (2019). Energy poverty and gender in England: A spatial perspective. *Geoforum* 104, 222–233. <u>https://doi.org/ 10.1016/j.geoforum.2019.05.001</u>
- Rogers, J. C., Simmons, E. A., Convery, I., and Weatherall, A. (2008). Public perceptions of opportunities for community-based renewable energy projects. *Energy Policy* 36, 4217–4226. <u>https:// doi.org/10.1016/j.enpol.2008.07.028</u>
- Rohde, K. I., and Verbeke, W. (2017). We like to see you in the gym—A field experiment on financial incentives for short and long term gym attendance. *Journal of Economic Behavior* & Organization 134, 388–407. <u>https://doi.org/10.1016/j.jebo.</u> 2016.12.012
- Romanello, M., McGushin, A., Di Napoli, C., Drummond, P., Hughes, N., Jamart, L., et al. (2021). The 2021 report of the Lancet Countdown on health and climate change: code red for a healthy future. *The Lancet* 398, 1619–1662. <u>https://doi. org/10.1016/S0140-6736(21)01787-6</u>
- Rosendahl, K. E. (2019). EU ETS and the waterbed effect. *Nat. Clim. Chang.* 9, 734–735. <u>https://doi.org/10.1038/s41558-019-0579-5</u>
- Rosenthal, S., and Yu, M. S. C. (2022). Anticipated guilt and anti-littering civic engagement in an extended norm activation model. *Journal of Environmental Psychology* 80, 101757. <u>https:// doi.org/10.1016/j.jenvp.2022.101757</u>
- Ross, A. D., Rouse, S. M., and Mobley, W. (2019). Polarization of Climate Change Beliefs: The Role of the Millennial Generation Identity. *Social Science Quarterly* 100, 2625–2640. <u>https://doi. org/10.1111/ssqu.12640</u>
- Royer, H., Stehr, M., and Sydnor, J. (2015). Incentives, Commitments, and Habit Formation in Exercise: Evidence from a Field Experiment with Workers at a Fortune-500 Company. *American Economic Journal: Applied Economics* 7, 51–84. <u>https://doi. org/10.1257/app.20130327</u>
- Russell, B., Beel, D., Rees Jones, I., and Jones, M. (2022). Placing the Foundational Economy: An emerging discourse for post-neoliberal economic development. *Environ Plan A* 54, 1069–1085. <u>https://doi.org/10.1177/0308518X221098745</u>
- Sagebiel, J., Müller, J. R., and Rommel, J. (2014). Are consumers willing to pay more for electricity from cooperatives? Results from an online Choice Experiment in Germany. *Energy Research & Social Science* 2, 90–101. <u>https://doi.org/10.1016/j.erss.2014.04.003</u>
- Saidani, M., Yannou, B., Leroy, Y., Cluzel, F., and Kendall, A. (2019). A taxonomy of circular economy indicators. *Journal of Cleaner Production* 207, 542–559. <u>https://doi.org/10.1016/j.jcle-pro.2018.10.014</u>
- Salman, M., Zha, D., and Wang, G. (2022). Assessment of energy poverty convergence: A global analysis. *Energy* 255, 124579. <u>https://doi.org/10.1016/j.energy.2022.124579</u>

- Sánchez, C. S.-G., Fernández, A. S., and Peiró, M. N. (2020). Feminisation of energy poverty in the city of Madrid. *Energy and Buildings* 223, 110157. <u>https://doi.org/10.1016/j.enbuild.2020.110157</u>
- Santarius, T., Bieser, J. C. T., Frick, V., Höjer, M., Gossen, · Maike, Lorenz, ·, et al. (2022). Digital sufficiency: conceptual considerations for ICTs on a finite planet. *Annals of Telecommunications* 78, 277–295. <u>https://doi.org/10.1007/s12243-022-00914-x</u>
- Santarius, T., Pohl, J., and Lange, S. (2020). Digitalization and the Decoupling Debate: Can ICT Help to Reduce Environmental Impacts While the Economy Keeps Growing? *Sustainability* 12, 7496. <u>https://doi.org/10.3390/su12187496</u>
- Santarius, T., and Soland, M. (2018). How Technological Efficiency Improvements Change Consumer Preferences: Towards a Psychological Theory of Rebound Effects. *Ecological Economics* 146, 414–424. <u>https://doi.org/10.1016/j.ecolecon.2017.12.009</u>
- Saunders, H. D., Roy, J., Azevedo, I. M. L., Chakravarty, D., Dasgupta, S., De La Rue Du Can, S., et al. (2021). Energy Efficiency: What Has Research Delivered in the Last 40 Years? *Annu. Rev. Environ. Resour.* 46, 135–165. <u>https://doi.org/10.1146/annurev-environ-012320-084937</u>
- Schanes, K., Dobernig, K., and Gözet, B. (2018). Food waste matters – A systematic review of household food waste practices and their policy implications. *Journal of Cleaner Production* 182, 978–991. https://doi.org/10.1016/j.jclepro.2018.02.030
- Scherhaufer, P., Höltinger, S., Salak, B., Schauppenlehner, T., and Schmidt, J. (2017). Patterns of acceptance and non-acceptance within energy landscapes: A case study on wind energy expansion in Austria. *Energy Policy* 109, 863–870. <u>https://doi.org/10.1016/j.enpol.2017.05.057</u>
- Schinko, T., Weifner, A., and Köppl, A. (2021). The interaction of energy services, breakthrough technologies, and human need satisfaction. EconTrans working paper #1. Vienna: Austrian Institute of Economic Research (WIFO). Available at: <u>https:// www.econstor.eu/handle/10419/237059</u> (Accessed August 31, 2023).
- Schmelzle, F., Kunkel, S., Beier, M. M. und G., and IASS, I. for A. S. S. (2022). Digitalisierte Industrie und Nachhaltigkeit – Zwischen Synergie und Dissonanz. *IM 38* 2022, 7–11. <u>https:// doi.org/10.30844/I40M\_22-1\_7-11</u>
- Schober, L.-M., and Mattke, J. (2022). AI for Sustainability in Organisations: A Literature Review., in AMCIS 2022 Proceedings, 2022. Available at: <u>https://aisel.aisnet.org/amcis2022/sig\_green/ sig\_green/15/</u> (Accessed September 9, 2022).
- Schreuer, A. (2016). The establishment of citizen power plants in Austria: A process of empowerment? *Energy Research & Social Science* 13, 126–135. <u>https://doi.org/10.1016/j.erss.2015.12.003</u>
- Schubert, C. (2017). Green nudges: Do they work? Are they ethical? *Ecological economics* 132, 329–342. <u>https://doi.org/10.1016/j.ecolecon.2016.11.009</u>
- Schucht, S., Colette, A., Rao, S., Holland, M., Schöpp, W., Kolp, P., et al. (2015). Moving towards ambitious climate policies: Monetised health benefits from improved air quality could offset mitigation costs in Europe. *Environmental Science & Policy* 50, 252–269. https://doi.org/10.1016/j.envsci.2015.03.001
- Schulte, M., Bamberg, S., Rees, J., and Rollin, P. (2020). Social identity as a key concept for connecting transformative societal change with individual environmental activism. *Journal of Environmental Psychology* 72, 101525. <u>https://doi.org/10.1016/j.jenvp.2020.101525</u>
- Schultz, P. W., Messina, A., Tronu, G., Limas, E. F., Gupta, R., and Estrada, M. (2016). Personalized Normative Feedback and the

Moderating Role of Personal Norms: A Field Experiment to Reduce Residential Water Consumption. *Environment and Behavior* 48, 686–710. <u>https://doi.org/10.1177/0013916514553835</u>

- Schuster, A., and Otto, I. M. (2022). Understanding Socio-metabolic Inequalities Using Consumption Data from Germany. *Capitalism Nature Socialism* 34, 97–118. <u>https://doi.org/10.1080/104</u> <u>55752.2022.2140066</u>
- Schwaiger, R., Holzknecht, A., Huber, J., Kirchler, M., Blanco, E., and Benda, J. (2025). Visible Tax Dividends and Support for Carbon Pricing – Evidence from Austria. Available at: <u>https:// osf.io/zv4p5/</u> (Accessed February 24, 2025).
- Schwartz, R., Dodge, J., Smith, N. A., and Etzioni, O. (2020). Green AI. Communications of the ACM 63, 54–63. <u>https://doi.org/10.1145/3381831</u>
- Scovronick, N., Budolfson, M., Dennig, F., Errickson, F., Fleurbaey, M., Peng, W., et al. (2019). The impact of human health co-benefits on evaluations of global climate policy. *Nat Commun* 10, 2095. <u>https://doi.org/10.1038/s41467-019-09499-x</u>
- Seebauer, S. (2021). How to make building renovation work for low-income renters: Preferences for distributional principles and procedural options in Austria. *Energy Research & Social Science* 82, 102270. <u>https://doi.org/10.1016/j.erss.2021.102270</u>
- Seebauer, S., and Babcicky, P. (2018). Trust and the communication of flood risks: comparing the roles of local governments, volunteers in emergency services, and neighbours. J Flood Risk Management 11, 305–316. <u>https://doi.org/10.1111/jfr3.12313</u>
- Seebauer, S., Friesenecker, M., and Eisfeld, K. (2019). Integrating climate and social housing policy to alleviate energy poverty: an analysis of targets and instruments in Austria. *Energy Sources, Part B: Economics, Planning, and Policy* 14, 304–326. <u>https://doi. org/10.1080/15567249.2019.1693665</u>
- Seebauer, S., Lückl, A., Köberl, J., and Kulmer, V. (2021). Soziale Folgen des Klimawandels in Österreich. Vienna: BMSGPK Federal Ministry of Social Affairs. Available at: <u>https://www. sozialministerium.at/Themen/Soziales/Soziale-Themen/Soziales-und-Klimawandel.html</u> (Accessed January 17, 2025).
- Seifert, M., Spitzer, F., Haeckl, S., Gaudeul, A., Kirchler, E., Palan, S., et al. (2024). Can information provision and preference elicitation promote ESG investments? Evidence from a large, incentivized online experiment. *Journal of Banking & Finance* 161, 107114. <u>https://doi.org/10.1016/j.jbankfin.2024.107114</u>
- Shyu, C.-W. (2021). A framework for 'right to energy' to meet UN SDG7: Policy implications to meet basic human energy needs, eradicate energy poverty, enhance energy justice, and uphold energy democracy. *Energy Research & Social Science* 79, 102199. https://doi.org/10.1016/j.erss.2021.102199
- Sikkema, R., Styles, D., Jonsson, R., Tobin, B., and Byrne, K. A. (2023). A market inventory of construction wood for residential building in Europe – in the light of the Green Deal and new circular economy ambitions. *Sustainable Cities and Society* 90, 104370. https://doi.org/10.1016/j.scs.2022.104370
- Sloot, D., Jans, L., and Steg, L. (2018). Can community energy initiatives motivate sustainable energy behaviours? The role of initiative involvement and personal pro-environmental motivation. *Journal of Environmental Psychology* 57, 99–106. <u>https:// doi.org/10.1016/j.jenvp.2018.06.007</u>
- Sloot, D., Jans, L., and Steg, L. (2019). In it for the money, the environment, or the community? Motives for being involved in community energy initiatives. *Global Environmental Change* 57, 101936. <u>https://doi.org/10.1016/j.gloenvcha.2019.101936</u>

- Słupik, S., Kos-Łabędowicz, J., and Trzęsiok, J. (2021). How to Encourage Energy Savings Behaviours? The Most Effective Incentives from the Perspective of European Consumers. *Energies* 14, 8009. <u>https://doi.org/10.3390/en14238009</u>
- Snell, C., Bevan, M., and Thomson, H. (2015). Justice, fuel poverty and disabled people in England. *Energy Research & Social Science* 10, 123–132. <u>https://doi.org/10.1016/j.erss.2015.07.012</u>
- Sommer, M., Köppl, A., Schleicher, S. P., Bachner, G., Mayer, J., Fischer, L., et al. (2021). The concept of "functionalities" in a macroeconomic modelling framework – Insights for Austria. EconTrans working paper #3. Vienna: Austrian Institute of Economic Research (WIFO). Available at: <u>https://www.econstor.eu/ handle/10419/237061</u> (Accessed August 31, 2023).
- Sovacool, B. K. (2011). Conceptualizing urban household energy use: Climbing the "Energy Services Ladder." *Energy Policy* 39, 1659–1668. <u>https://doi.org/10.1016/j.enpol.2010.12.041</u>
- Spaargaren, G. (1997). The ecological modernization of production and consumption: essays in environmental sociology. Available at: <u>https://api.semanticscholar.org/CorpusID:153347780</u>
- Sparkman, G., Geiger, N., and Weber, E. U. (2022). Americans experience a false social reality by underestimating popular climate policy support by nearly half. *Nat Commun* 13, 4779. <u>https://doi.org/10.1038/s41467-022-32412-y</u>
- Sposato, R. G., and Hampl, N. (2018). Worldviews as predictors of wind and solar energy support in Austria: Bridging social acceptance and risk perception research. *Energy Research & Social Science* 42, 237–246. <u>https://doi.org/10.1016/j.erss.2018.03.012</u>
- Spotswood, F. ed. (2016). Beyond behaviour change: Key issues, Interdisciplinary Approaches and Future Directions. Bristol University Press, Policy Press.
- Springmann, M., Godfray, H. C. J., Rayner, M., and Scarborough, P. (2016). Analysis and valuation of the health and climate change cobenefits of dietary change. *Proceedings of the National Academy of Sciences* 113, 4146–4151. <u>https://doi.org/10.1073/ pnas.1523119113</u>
- Staddon, S. C., Cycil, C., Goulden, M., Leygue, C., and Spence, A. (2016). Intervening to change behaviour and save energy in the workplace: A systematic review of available evidence. *Energy Research & Social Science* 17, 30–51. <u>https://doi.org/10.1016/j.</u> <u>erss.2016.03.027</u>
- Stadelmann-Steffen, I., Eder, C., Harring, N., Spilker, G., and Katsanidou, A. (2021). A framework for social tipping in climate change mitigation: What we can learn about social tipping dynamics from the chlorofluorocarbons phase-out. *Energy Research & Social Science* 82, 102307. <u>https://doi.org/10.1016/j. erss.2021.102307</u>
- Stadt Graz (2022). Klimaschutzplan Graz Teil 1 Eröffnungsbilanz. Graz: Stadt Graz. Available at: <u>https://www.graz.at/cms/</u> <u>dokumente/10387901\_8106610/9a82ba0e/Beilage%20GR%20</u> <u>Klimaschutzplan\_Graz\_Teil1\_Er%C3%B6ffnungsbilanz\_</u> <u>20220310.pdf</u> (Accessed November 18, 2024).
- Stadt Graz (2024). Klimaeuro+ Förderung der Stadt Graz. *Graz.* Available at: <u>https://www.graz.at/cms/beitrag/10426938/982</u> <u>1093/Klimaeuro.html</u> (Accessed May 8, 2024).
- Stanley, S. K., Hogg, T. L., Leviston, Z., and Walker, I. (2021). From anger to action: Differential impacts of eco-anxiety, eco-depression, and eco-anger on climate action and wellbeing. *The Journal of Climate Change and Health* 1, 100003. <u>https://doi.org/10.1016/j.joclim.2021.100003</u>
- Stanton, E. A., Ackerman, F., and Kartha, S. (2009). Inside the integrated assessment models: Four issues in climate economics.

Climate and Development 1, 166–184. https://doi.org/10.3763/ cdev.2009.0015

- Statista (2023). Most common second-hand purchases by category in Austria 2023. *Statista*. Available at: <u>https://www.statista.com/</u> <u>forecasts/1001262/most-common-second-hand-purchases-by-</u> <u>category-in-austria</u> (Accessed September 13, 2023).
- Statistik Austria (2017). Haushaltsenergie und Einkommen mit besonderem Fokus auf Energiearmut: 2014. Vienna: Statistik Austria. Available at: <u>https://www.statistik.at/fileadmin/publications/Haushaltsenergie\_und\_Einkommen\_mit\_besonderem\_</u> Fokus\_auf\_Energiearmut\_2014.pdf
- Statistik Austria (2019). Energiearmut in Österreich 2016: Haushaltsenergie und Einkommen. Vienna: Statistik Austria. Available at: <u>https://www.statistik.at/fileadmin/publications/</u> Energiearmut\_in\_OEsterreich\_2016\_-\_Haushaltsenergie\_ und\_Einkommen.pdf
- Statistik Austria (2021a). Erweiterte Betrachtung der Energiearmut in Österreich 2018: Hohe Energiekosten bzw. Nicht-Leistbarkeit von Energie für Wohnen. Vienna: Statistik Austria. Available at: https://www.statistik.at/fileadmin/publications/Erweiterte\_Betrachtung\_der\_Energiearmut\_in\_OEsterreich\_2018.pdf
- Statistik Austria (2021b). Konsumerhebung 2019/20. Vienna. Available at: <u>https://www.statistik.at/ueber-uns/erhebungen/personen-und-haushaltserhebungen/konsumerhebung</u> (Accessed August 20, 2021).
- Statistik Austria (2022). Dimensionen der Energiearmut in Österreich: Hohe Energiekosten bzw. Nicht-Leistbarkeit von Energie für Wohnen. Vienna: Statistik Austria. Available at: <u>https://</u> <u>www.statistik.at/fileadmin/publications/Dimensionen-der-Energiearmut-2020-2021\_barrierefrei.pdf</u>
- Statistik Austria (2024a). Dimensionen der Energiearmut in Österreich 2021/22: Indikatorenüberblick und detaillierte Betrachtung. Vienna: Verlag Österreich. Available at: <u>https://www.statistik.at/</u>fileadmin/user\_upload/Energiearmut-2021\_22\_barr\_Web.pdf
- Statistik Austria (2024b). Energy poverty 2022: Heating according to 3.2% of households not affordable. Statistik Austria. Available at: <u>https://www.statistik.at/fileadmin/announcement/2024/03/20240318Energiearmut2022EN.pdf</u> (Accessed November 6, 2024).
- Statistik Austria (2024c). So geht's uns heute: die sozialen Krisenfolgen im vierten Quartal 2023 – Schwerpunkt: Ernährungsarmut: Ergebnisse einer Statistik-Austria-Befragung. Vienna: BMSGPK. Available at: <u>https://www.statistik.at/fileadmin/ user\_upload/Bericht-Soziale-Krisenfolgen-QU4-2023.pdf</u>
- Steel, D., Mintz-Woo, K., and DesRoches, C. T. (2024). Collapse, social tipping dynamics, and framing climate change. *Politics, Philosophy & Economics* 23, 230–251. <u>https://doi.org/10.1177/ 1470594X231196432</u>
- Steffen, W., Rockström, J., Richardson, K., Lenton, T. M., Folke, C., Liverman, D., et al. (2018). Trajectories of the Earth System in the Anthropocene. *Proc. Natl. Acad. Sci. U.S.A.* 115, 8252–8259. <u>https://doi.org/10.1073/pnas.1810141115</u>
- Steg, L. (2023). Psychology of Climate Change. Annual Review of Psychology 74, 391–421. <u>https://doi.org/10.1146/annurevpsych-032720-042905</u>
- Steg, L., Dreijerink, L., and Abrahamse, W. (2006). Why are Energy Policies Acceptable and Effective? *Environment and Behavior* 38, 92–111. <u>https://doi.org/10.1177/0013916505278519</u>
- Steg, L., and Gifford, R. (2005). Sustainable transportation and quality of life. *Journal of Transport Geography* 13, 59–69. <u>https:// doi.org/10.1016/j.jtrangeo.2004.11.003</u>

- Steigenberger, N. (2016). Organizing for the Big One: A Review of Case Studies and a Research Agenda for Multi-Agency Disaster Response. *Contingencies & Crisis Mgmt* 24, 60–72. <u>https://doi.org/10.1111/1468-5973.12106</u>
- Steinberger, J. K., and Roberts, J. T. (2010). From constraint to sufficiency: The decoupling of energy and carbon from human needs, 1975–2005. *Ecological Economics* 70, 425–433. <u>https://doi.org/10.1016/j.ecolecon.2010.09.014</u>
- Steininger, K., Bednar-Friedl, B., Knittel, N., Kirchengast, G., Nabernegg, S., Williges, K., et al. (2020). Klimapolitik in Österreich: Innovationschance Coronakrise und die Kosten des Nicht-Handelns. Graz: Wegener Center Verlag. <u>https://doi.org/ 10.25364/23.2020.1</u>
- Steininger, K. W., and Bachner, G. (2014). Extending car-sharing to serve commuters: An implementation in Austria. *Ecological Economics* 101, 64–66. <u>https://doi.org/10.1016/j.ecolecon.2014.03.001</u>
- Steininger, K. W., Bednar-Friedl, B., Formayer, H., and König, M. (2016a). Consistent economic cross-sectoral climate change impact scenario analysis: Method and application to Austria. *Climate Services* 1, 39–52. <u>https://doi.org/10.1016/j.cliser.2016.02.003</u>
- Steininger, K. W., König, M., Bednar-Friedl, B., Kranzl, L., Loibl, W., and Prettenthaler, F. (2015). Economic Evaluation of Climate Change Impacts: Development of a Cross-Sectoral Framework and Results for Austria. Cham: Springer. Available at: <u>https:// link.springer.com/book/10.1007/978-3-319-12457-5</u> (Accessed November 8, 2024).
- Steininger, K. W., Lininger, C., Meyer, L. H., Muñoz, P., and Schinko, T. (2016b). Multiple carbon accounting to support just and effective climate policies. *Nature Clim. Change* 6, 35–41. <u>https://doi.org/10.1038/nclimate2867</u>
- Steininger, K. W., Munoz, P., Karstensen, J., Peters, G. P., Strohmaier, R., and Velázquez, E. (2018). Austria's consumption-based greenhouse gas emissions: Identifying sectoral sources and destinations. *Global Environmental Change* 48, 226–242. <u>https:// doi.org/10.1016/j.gloenvcha.2017.11.011</u>
- Stiglitz, J., Fitoussi, J., and Durand, M. eds. (2018). For Good Measure: Advancing Research on Well-being Metrics Beyond GDP. Paris: OECD Publishing. <u>https://doi.org/10.1787/9789264307278-en</u>
- Stone, A. A., and Krueger, A. B. (2018). "Understanding subjective well-being", in For Good Measure: Advancing Research on Well-being Metrics Beyond GDP, eds. J. Stiglitz, J. Fitoussi, and M. Durand (Paris: OECD Publishing), 163–201. <u>https://doi.org/10.1787/9789264307278-9-en</u>
- Striessnig, E., Reiter, C., and Dimitrova, A. (2021). Global improvements in Years of Good Life since 1950. Vienna Yearbook of Population Research 19, 141–168. <u>https://doi.org/10.1553/populationyearbook2021.res1.2</u>
- Strubell, E., Ganesh, A., and McCallum, A. (2019). Energy and Policy Considerations for Deep Learning in NLP, (Florence, Italy). <u>https://doi.org/10.48550/arXiv.1906.02243</u>
- Sunikka-Blank, M., and Galvin, R. (2021). Single parents in cold homes in Europe: How intersecting personal and national characteristics drive up the numbers of these vulnerable households. *Energy Policy* 150, 112134. <u>https://doi.org/10.1016/j.enpol.2021.112134</u>
- Sunstein, C. R. (2015). Nudging and Choice Architecture: Ethical Considerations. *Yale Journal on Regulation* Forthcoming. Available at: <u>https://ssrn.com/abstract=2551264</u>

- Sunstein, C. R. (2019). How change happens. Cambridge, Massachusetts London: MIT Press.
- Tàbara, D., Frantzeskaki, N., Hölscher, K., Pedde, S., Kok, K., Lamperti, F., et al. (2018). Positive tipping points in a rapidly warming world. *Current Opinion in Environmental Sustainabili*ty 31, 120–129. https://doi.org/10.1016/j.cosust.2018.01.012
- Thaler, R. H., and Sunstein, C. R. (2009). Nudge: improving decisions about health, wealth and happiness., Revised edition. London: Penguin Books.
- Thaller, A., and Brudermann, T. (2020). "You know nothing, John Doe" – Judgmental overconfidence in lay climate knowledge. *Journal of Environmental Psychology* 69, 101427. <u>https://doi.org/ 10.1016/j.jenvp.2020.101427</u>
- Thaller, A., Fleiß, E., Brohmer, H., Köstenbaumer, D., Posch, A., and Athenstaedt, U. (2023). When perceived fairness and acceptance go hand in hand–Drivers of regulatory and economic policies for low-carbon mobility. *PLOS Climate* 2, e0000157. https://doi.org/10.1371/journal.pclm.0000157
- Thalmann, P. (2004). The Public Acceptance of Green Taxes: 2 Million Voters Express Their Opinion. *Public Choice* 119, 179–217. <u>https://doi.org/10.1023/B:PUCH.0000024165.18082.db</u>
- Theine, H., Humer, S., Moser, M., and Schnetzer, M. (2022). Emissions inequality: Disparities in income, expenditure, and the carbon footprint in Austria. *Ecological Economics* 197, 107435. <u>https://doi.org/10.1016/j.ecolecon.2022.107435</u>
- Thunström, L., van 't Veld, K., Shogren, J. F., and Nordström, J. (2014). On strategic ignorance of environmental harm and social norms. *Revue d'économie politique* 124, 195–214. <u>https:// doi.org/10.3917/redp.242.0195</u>
- Tiefenbeck, V., Goette, L., Degen, K., Tasic, V., Fleisch, E., Lalive, R., et al. (2018). Overcoming Salience Bias: How Real-Time Feedback Fosters Resource Conservation. *Management Science* 64, 1458–1476. <u>https://doi.org/10.1287/mnsc.2016.2646</u>
- Tindall, D., Stoddart, M. C. J., and Dunlap, R. E. eds. (2022). Handbook of Anti-Environmentalism. Cheltenham, UK, Northampton, MA, USA: Edward Elgar Publishing. <u>https://doi.org/10.4337/9781839100222</u>
- Tröger, N., and Panhuber, L. (2023). (Sustainable) Fashion Consumption in Austria: A High Level of Awareness, but Potential to do More. Wien: Arbeiterkammer für Arbeiter und Angestellte für Wien, Greenpeace. Available at: <u>https://wien.arbeiterkammer.at/interessenvertretung/konsument/Sustanaible\_Fashion\_</u> Consumption\_in\_Austria.pdf (Accessed February 6, 2025).
- Tversky, A., and Kahneman, D. (1981). The Framing of Decisions and the Psychology of Choice. *Science* 211, 453–458. <u>https://doi.org/10.1126/science.7455683</u>
- Tversky, A., and Kahneman, D. (1986). The Framing of Decisions and the Evaluation of Prospects. *Studies in Logic and the Foundations of Mathematics* 114, 503–520. <u>https://doi.org/10.1016/</u> <u>S0049-237X(09)70710-4</u>
- Umweltbundesamt (2022a). Bundesländer Luftschadstoff-Inventur 1990–2020. Regionalisierung der nationalen Emissionsdaten auf Grundlage von EU-Berichtspflichten (Datenstand 2022). Vienna: Umweltbundesamt GmbH. Available at: <u>https://www. umweltbundesamt.at/fileadmin/site/publikationen/rep0817.</u> <u>pdf</u>
- Umweltbundesamt (2022b). Klimaschutzbericht 2022. Vienna: Umweltbundesamt GmbH. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/publikationen/rep0816.pdf</u>
- Umweltförderung (2024). Sauber Heizen für Alle 2024. Available at: <u>https://www.umweltfoerderung.at/privatpersonen/sauber-</u>

heizen-fuer-alle-2024/unterkategorie-ein-und-zweifamilienhaus

- United Nations Environment Programme (2020). Emissions Gap Report 2020. Nairobi. Available at: <u>https://www.unep.org/emissions-gap-report-2020</u>
- University of Leeds (2023). A Good Life For All Within Planetary Boundaries. Available at: <u>https://goodlife.leeds.ac.uk/national-trends/country-trends/#AUT</u> (Accessed September 15, 2023).
- Unruh, G. C. (2002). Escaping carbon lock-in. *Energy Policy* 30, 317–325. https://doi.org/10.1016/S0301-4215(01)00098-2
- Van Bommel, N., and Höffken, J. I. (2021). Energy justice within, between and beyond European community energy initiatives: A review. *Energy Research & Social Science* 79, 102157. <u>https://doi. org/10.1016/j.erss.2021.102157</u>
- van de Grift, E., and Cuppen, E. (2022). Beyond the public in controversies: A systematic review on social opposition and renewable energy actors. *Energy Research & Social Science* 91, 102749. <u>https://doi.org/10.1016/j.erss.2022.102749</u>
- van den Bergh, J., Castro, J., Drews, S., Exadaktylos, F., Foramitti, J., Klein, F., et al. (2021). Designing an effective climate-policy mix: accounting for instrument synergy. *Climate Policy* 21, 745–764. https://doi.org/10.1080/14693062.2021.1907276
- Van Der Linden, S., and Goldberg, M. H. (2020). Alternative meta-analysis of behavioral interventions to promote action on climate change yields different conclusions. *Nat Commun* 11, 3915. <u>https://doi.org/10.1038/s41467-020-17613-7</u>
- van Valkengoed, A. M., Steg, L., and Perlaviciute, G. (2021). Development and validation of a climate change perceptions scale. *Journal of Environmental Psychology* 76, 101652. <u>https://doi. org/10.1016/j.jenvp.2021.101652</u>
- Vandenberghe, D., and Albrecht, J. (2018). Tackling the chronic disease burden: are there co-benefits from climate policy measures? *Eur J Health Econ* 19, 1259–1283. <u>https://doi.org/10.1007/ s10198-018-0972-4</u>
- VanderWeele, T. J. (2017). On the promotion of human flourishing. *Proceedings of the National Academy of Sciences* 114, 8148–8156. <u>https://doi.org/10.1073/pnas.1702996114</u>
- VanderWeele, T. J., Trudel-Fitzgerald, C., Allin, P., Farrelly, C., Fletcher, G., Frederick, D. E., et al. (2020). Current recommendations on the selection of measures for well-being. *Preventive Medicine* 133, 106004. <u>https://doi.org/10.1016/j.ypmed.2020.</u> <u>106004</u>
- Veit, D. J., and Thatcher, J. B. (2023). Digitalization as a problem or solution? Charting the path for research on sustainable information systems. *Journal of Business Economics* 93, 1231–1253. <u>https://doi.org/10.1007/s11573-023-01143-x</u>
- Vélez-Henao, J. A., and Pauliuk, S. (2023). Material Requirements of Decent Living Standards. *Environ. Sci. Technol.* 57, 14206– 14217. <u>https://doi.org/10.1021/acs.est.3c03957</u>
- Velte, P. (2022). Meta-analyses on Corporate Social Responsibility (CSR): a literature review. *Manag Rev Q* 72, 627–675. <u>https:// doi.org/10.1007/s11301-021-00211-2</u>
- VERBUND AG (2022). VERBUND Electricity assistance fund with Caritas. Available at: <u>https://www.verbund.com/en-at/about-verbund/responsibility/social-issues/electricity-assistance-fund</u>
- Vergunst, F., and Berry, H. L. (2022). Climate Change and Children's Mental Health: A Developmental Perspective. *Clinical Psychological Science* 10, 767–785. <u>https://doi.org/10.1177/ 21677026211040787</u>

AAR2

- Verplanken, B., and Whitmarsh, L. (2021). Habit and climate change. Current Opinion in Behavioral Sciences 42, 42–46. <u>https://doi.org/10.1016/j.cobeha.2021.02.020</u>
- Vesely, S., Masson, T., Chokrai, P., Becker, A. M., Fritsche, I., Klöckner, C. A., et al. (2021). Climate change action as a project of identity: Eight meta-analyses. *Global Environmental Change* 70, 102322. <u>https://doi.org/10.1016/j.gloenvcha.2021.102322</u>
- Vicedo-Cabrera, A. M., Scovronick, N., Sera, F., Royé, D., Schneider, R., Tobias, A., et al. (2021). The burden of heat-related mortality attributable to recent human-induced climate change. *Nat. Clim. Chang.* 11, 492–500. <u>https://doi.org/10.1038/s41558-021-01058-x</u>
- Villamayor-Tomas, S., Thiel, A., Amblard, L., Zikos, D., and Blanco, E. (2019). Diagnosing the role of the state for local collective action: Types of action situations and policy instruments. *Environmental Science & Policy* 97, 44–57. <u>https://doi.org/10.1016/j.</u> <u>envsci.2019.03.009</u>
- Villavicencio Calzadilla, P., and Mauger, R. (2018). The UN's new sustainable development agenda and renewable energy: the challenge to reach SDG7 while achieving energy justice. *Journal* of Energy & Natural Resources Law 36, 233–254. <u>https://doi.org/ 10.1080/02646811.2017.1377951</u>
- Vlasceanu, M., Doell, K. C., Bak-Coleman, J. B., Todorova, B., Berkebile-Weinberg, M. M., Grayson, S. J., et al. (2024). Addressing climate change with behavioral science: A global intervention tournament in 63 countries. *Sci. Adv.* 10, eadj5778. <u>https://doi.org/10.1126/sciadv.adj5778</u>
- Vogel, J., Steinberger, J. K., O'Neill, D. W., Lamb, W. F., and Krishnakumar, J. (2021). Socio-economic conditions for satisfying human needs at low energy use: An international analysis of social provisioning. *Global Environmental Change* 69, 102287. <u>https://doi.org/10.1016/j.gloenvcha.2021.102287</u>
- Von Wirth, T., Gislason, L., and Seidl, R. (2018). Distributed energy systems on a neighborhood scale: Reviewing drivers of and barriers to social acceptance. *Renewable and Sustainable Energy Re*views 82, 2618–2628. https://doi.org/10.1016/j.rser.2017.09.086
- Walker, G. (2011). The role for 'community' in carbon governance. WIREs Climate Change 2, 777–782. <u>https://doi.org/10.1002/</u> wcc.137
- Walker, G., Devine-Wright, P., Hunter, S., High, H., and Evans, B. (2010). Trust and community: Exploring the meanings, contexts and dynamics of community renewable energy. *Energy Policy* 38, 2655–2663. https://doi.org/10.1016/j.enpol.2009.05.055
- Wang, M., Rieger, M. O., and Hens, T. (2017). The Impact of Culture on Loss Aversion. *Journal of Behavioral Decision Making* 30, 270–281. <u>https://doi.org/10.1002/bdm.1941</u>
- Wang, Q., Kwan, M.-P., Fan, J., and Lin, J. (2021). Racial disparities in energy poverty in the United States. *Renewable and Sustainable Energy Reviews* 137, 110620. <u>https://doi.org/10.1016/</u> j.rser.2020.110620
- Wannewitz, M., and Garschagen, M. (2023). Collective adaptation to climate change. *Current Opinion in Environmental Sustainability* 61, 101248. <u>https://doi.org/10.1016/j.cosust.2022.101248</u>
- Wastling, T., Charnley, F., and Moreno, M. (2018). Design for Circular Behaviour: Considering Users in a Circular Economy. Sustainability 10, 1743. https://doi.org/10.3390/su10061743
- Weber, E. U. (2010). What shapes perceptions of climate change? WIREs Climate Change 1, 332–342. <u>https://doi.org/10.1002/</u> wcc.41
- Wegscheider-Pichler, A., Prettner, C., Lamei, N., and Unger, W. (2021). Wie geht's Österreich? 2021. Wien: Statistik Austria.

Available at: <u>https://www.statistik.at/fileadmin/publications/</u> Wie\_geht\_s\_OEsterreich\_2021.pdf

- Weisz, U., Pichler, P.-P., Jaccard, I. S., Haas, W., Matej, S., Bachner, F., et al. (2020). Carbon emission trends and sustainability options in Austrian health care. *Resources, Conservation and Recycling* 160, 104862. <u>https://doi.org/10.1016/j.resconrec.2020.104862</u>
- Weitzman, M. L. (2009). On Modeling and Interpreting the Economics of Catastrophic Climate Change. *Review of Economics* and Statistics 91, 1–19. <u>https://doi.org/10.1162/rest.91.1.1</u>
- Werthschulte, M., and Löschel, A. (2021). On the role of present bias and biased price beliefs in household energy consumption. *Journal of Environmental Economics and Management* 109, 102500. <u>https://doi.org/10.1016/j.jeem.2021.102500</u>
- Wiener Tafel (2016). Wiener Tafel- Geruchs- und Geschmackslabor: "Ist das noch gut?". Available at: <u>https://wienertafel.at/</u> <u>wp-content/uploads/2021/09/PA\_Kinderuni2016\_Dasistdoch-gut\_2016.pdf</u>
- Wierling, A., Schwanitz, V. J., Zeiss, J. P., Von Beck, C., Paudler, H. A., Koren, I. K., et al. (2023). A Europe-wide inventory of citizen-led energy action with data from 29 countries and over 10000 initiatives. *Sci Data* 10, 9. <u>https://doi.org/10.1038/ s41597-022-01902-5</u>
- Wiest, F., Gamarra Scavone, M., Newell, M., Otto, I., and Ringsmuth, A. (2022). Scaling Up Ecovillagers' Lifestyles Can Help to Decarbonise Europe. *Sustainability* 14, 13611. <u>https://doi.org/</u> 10.3390/su142013611
- Williamson, O. E. (1998). The Institutions of Governance. The American Economic Review 88, 75–79.
- Wilson, C., Hargreaves, T., and Hauxwell-Baldwin, R. (2017). Benefits and risks of smart home technologies. *Energy Policy* 103, 72–83. <u>https://doi.org/10.1016/j.enpol.2016.12.047</u>
- Wilson, C., Kerr, L., Sprei, F., Vrain, E., and Wilson, M. (2020). Potential Climate Benefits of Digital Consumer Innovations. Annual Review of Environment and Resources 45, 113–144. <u>https:// doi.org/10.1146/annurev-environ-012320-082424</u>
- Wilson, M., Paschen, J., and Pitt, L. (2022). The circular economy meets artificial intelligence (AI): understanding the opportunities of AI for reverse logistics. *Management of Environmental Quality: An International Journal* 33, 9–25. <u>https://doi.org/10.1108/MEQ-10-2020-0222</u>
- Wilts, H., and Berg, H. (2017). The Digital Circular Economy: Can the Digital Transformation Pave the Way for Resource-Efficient Materials Cycles? *IJESNR* 7, 555725. <u>https://doi.org/10.19080/</u> <u>IJESNR.2017.07.555725</u>
- Winkelmann, R., Donges, J. F., Smith, E. K., Milkoreit, M., Eder, C., Heitzig, J., et al. (2022). Social tipping processes towards climate action: A conceptual framework. *Ecological Economics* 192, 107242. https://doi.org/10.1016/j.ecolecon.2021.107242
- Woerner, A., Imai, T., Pace, D., and Schmidt, K. (2024). DP19590 How to Increase Public Support for Carbon Pricing with Revenue Recycling. Paris & London: CERP Press. Available at: <u>https://cepr.org/publications/dp19590</u> (Accessed October 29, 2024).
- Wunderlich, P., Veit, D. J., and Sarker, S. (2019). Adoption of Sustainable Technologies: A Mixed-Methods Study of German Households. *MIS Quarterly: Management Information Systems* 43, 673–691. <u>https://doi.org/10.25300/MISQ/2019/12112</u>
- Wuppertal Institut für Klima, Umwelt, Energie (2024). European Sufficiency Policy Database. *Energie-Suffizienz*. Available at: <u>https://energysufficiency.de/en/policy-database-en/</u> (Accessed June 9, 2024).

- Yildiz, Ö., Rommel, J., Debor, S., Holstenkamp, L., Mey, F., Müller, J. R., et al. (2015). Renewable energy cooperatives as gatekeepers or facilitators? Recent developments in Germany and a multidisciplinary research agenda. *Energy Research & Social Science* 6, 59–73. <u>https://doi.org/10.1016/j.erss.2014.12.001</u>
- Zawadzki, S. J., Steg, L., and Bouman, T. (2020). Meta-analytic evidence for a robust and positive association between individuals' pro-environmental behaviors and their subjective wellbeing. *Environ. Res. Lett.* 15, 123007. <u>https://doi.org/10.1088/1748-9326/abc4ae</u>
- Zeiss, R. (2019). Information Flows in Circular Economy Practices., in *ECIS*, (Stockholm-Uppsala, Sweden). Available at: <u>https://kups.ub.uni-koeln.de/9759/1/Information%20Flows%20in%20</u> <u>Circular%20Economy%20Practices\_revised\_v1.0.pdf</u> (Acccessed September 9, 2022).
- Zeiss, R., Ixmeier, A., Recker, J., and Kranz, J. (2021). Mobilising information systems scholarship for a circular economy: Review, synthesis, and directions for future research. *Information Systems Journal* 31, 148–183. https://doi.org/10.1111/isj.12305
- Zell-Ziegler, C., Thema, J., Best, B., Wiese, F., Lage, J., Schmidt, A., et al. (2021). Enough? The role of sufficiency in European energy and climate plans. *Energy Policy* 157, 112483. <u>https://doi. org/10.1016/j.enpol.2021.112483</u>
- Zero Waste Austria (2021). Food Waste Legislation and Action in Austria. Vienna: Zero Waste Austria – Verein zur Schonung von

Ressourcen. Available at: https://www.zerowasteaustria.at/uploads/5/8/6/4/58648241/eng\_food\_waste\_in\_austria\_11\_20\_. pdf (Accessed February 6, 2025).

- Zhang, X., and Wei, C. (2022). The economic and environmental impacts of information and communication technology: A state-of-the-art review and prospects. *Resources, conservation* and recycling 185, 106477-. <u>https://doi.org/10.1016/j.resconrec.2022.106477</u>
- Zhao, J., and Luo, Y. (2021). A framework to address cognitive biases of climate change. *Neuron* 109, 3548–3551. <u>https://doi.org/ 10.1016/j.neuron.2021.08.029</u>
- Zibell, L., Beznea, A., Torres, P., and Sikora, I. (2021). Expanding the knowledge base around the role of consumers in the circular economy. Promoting circular behaviour in textiles and electronics. Ricardo for European Environment Agency. Available at: <u>https://www.eea.europa.eu/publications/influencing-consumer-choices-towards-circularity/role-of-consumers-in-the-ce/ view</u>
- Ziegler, A. (2017). Political orientation, environmental values, and climate change beliefs and attitudes: An empirical cross country analysis. *Energy Economics* 63, 144–153. <u>https://doi.org/10.1016/j.eneco.2017.01.022</u>
- Zuboff, S. (2015). Big other: Surveillance Capitalism and the Prospects of an Information Civilization. *Journal of Information Technology* 30, 75–89. <u>https://doi.org/10.1057/jit.2015.5</u>





**Chapter 6** 

# Climate governance: Political, legal, economic and societal aspects

# Second Austrian Assessment Report on Climate Change | AAR2

# Chapter 6 Climate governance: Political, legal, economic and societal aspects

This publication was subject to an anonymous international peer review process guided by several review editors, the AAR2 co-chairs and the APCC Steering Committee.

The views or opinions expressed in this publication are not necessarily those of the institutions with whose support this work was carried out. The publisher and the supporting institutions do not guarantee the accuracy or continued availability of urls to external websites or third-party websites mentioned in this publication and accept no responsibility for the accuracy or appropriateness of any content contained therein.

#### **Recommended citation**

A. Brad, D. Ennöckl, C. Kettner, et al. (2025): Climate governance: Political, legal, economic and societal aspects.
In "Second Austrian Assessment Report on Climate Change (AAR2) of the Austrian Panel on Climate Change (APCC)".
[D. Huppmann, M. Keiler, K. Riahi, H. Rieder (eds.)]. Austrian Academy of Sciences Press, Vienna, Austria | doi: doi.org/10.1553/aar2-ch6

You can download the recommended citation in the formats Endnote (.ris) and BibTex (.bib) at <u>aar2.ccca.ac.at/citation</u>.

Vienna, June 2025 ISBN: 978-3-7001-5110-4 | <u>austriaca.at/5110-4</u> (full report)

#### Production

Publisher: Austrian Academy of Sciences Press, Vienna Typesetting: Crossdesign Werbeagentur GmbH, Graz Cover: paulaidaanton | Agentur für digiloges Design Graphic design Table 6.2: paulaidaanton | Agentur für digiloges Design Al-based tools such as DeepL were used to support the linguistic revision.

Contact address for product safety: Austrian Academy of Sciences Press Dr. Ignaz Seipel-Platz 2 A-1010 Wien Tel.: +43-1-51581-3420 E-Mail: verlag@oeaw.ac.at

#### Funding acknowledgement

The AAR2 was funded by the Climate and Energy Fund under the 14th Austrian Climate Research Program (ACRP).

#### **Copyright and license**

© 2025 Austrian Panel on Climate Change (APCC) This report is published under a Creative Commons CC-BY 4.0-License. <u>creativecommons.org/licenses/by/4.0</u>





The AAR2 is a report by the Austrian Panel on Climate Change (APCC). https://aar2.ccca.ac.at/en

#### **Coordinating Lead Authors**

Alina Brad Daniel Ennöckl Claudia Kettner

#### **Lead Authors**

Ulrich Brand Sarah Haider-Nash Nikolaus Handig Daniel Hausknost Miriam Hofer Roman Hoffmann Lars Keller Mathias Kirchner Daniela Kletzan-Slamanig Karina Knaus Halliki Kreinin Thomas Leoni Danval Maneka Andrea Möller **Ilse Schindler** Margit Schratzenstaller Hendrik Theine Erika Wagner

#### **Contributing Authors**

Ernest Aigner Mel Arnecke Richard Bärnthaler Daniela Ecker Jarmo Kikstra Michael Miess

Andreas Novy Nicolas Roux Thomas Schinko

## **Review Editors**

Dragana Damjanovic Marco Pütz

### Chapter Scientist Danyal Maneka

University of Vienna

# Liaisons from AAR2 Technical Support Unit

Maria Holzinger	International Institute for Applied Systems Analysis   IIASA
Verena Rauchenwald	International Institute for Applied Systems Analysis   IIASA

University of Vienna BOKU University Austrian Institute of Economic Research | WIFO

University of Vienna University for Continuing Education Krems **BOKU University** Vienna University of Economics and Business | WU University of Graz International Institute for Applied Systems Analysis | IIASA University of Innsbruck | UIBK **BOKU University** Austrian Institute of Economic Research | WIFO Austrian Energy Agency | AEA Research Institute for Sustainability at GFZ RIFS Potsdam University of Applied Sciences Wiener Neustadt University of Vienna University of Vienna Environment Agency Austria | UBA Austrian Institute of Economic Research | WIFO Johannes Kepler University Linz | JKU Johannes Kepler University Linz | JKU

Leuphana University Lüneburg Democracy Centre Vienna University of Leeds Johannes Kepler University Linz | JKU International Institute for Applied Systems Analysis | IIASA King Abdullah University of Science and Technology & Environment Agency Austria | UBA Vienna University of Economics and Business | WU BOKU University International Institute for Applied Systems Analysis | IIASA

University of Vienna Swiss Federal Institute for Snow and Avalanche Research | SLF

# **TABLE OF CONTENTS**

Executive summary			387
6.1. Introduction			389
6.2. The structural and dynamic context of climate governance			
	(limiti	ng) growth	391
6.3.	<ul> <li>Regulatory framework of adaptation and mitigation</li> <li>39</li> </ul>		
	6.3.1.	International: Requirements of the Paris Agreement	396
	6.3.2.	The EU regulatory context	396
	6.3.3.	The national regulatory context	397
6.4.	Institu	utions and actors of climate governance	
	in Aus	stria	398
	6.4.1.	Institutions	399
		Federalism	399
		Social partnership	400
		Institutional focus on climate change adaptation	400
	6.4.2.	Actors	400
		Political parties	401
		Organized interest groups	402
		Climate organizations and movements	403
6.5.	Policy	instruments for climate change	
mitigation and adaptation		404	
	6.5.1.	Mitigation instruments	408
		Regulatory instruments for climate change mitigation	408
		Planning instruments for energy and climate change mitigation policies	409
		Economic instruments for climate change mitigation	410
Chapter Box 6.1			
Carbon pricing in the EU and Austria 412			
		Other policy instruments for climate mitigation	413
		Instruments compensating negative distributional effects of climate policies	414

6.5.2.	Adaptation instruments	416
	Regulatory instruments for climate change adaptation	416
	Planning instruments for climate change adaptation	416
	Economic instruments for climate	
	change adaptation	417
	Other policy instruments for climate change adaptation	417
6.5.3.	Climate Policy Integration	417
6.5.4.	Integrative perspective on Austria's climate policy instruments	418
6.6. Clima	te policy litigation	418
6.6.1.	Human rights, climate lawsuits and	
	environmental proceedings	419
6.6.2.	Climate liability	419
6.7. Budg	et, green public finances, investments	
and c	ost of (in)action	419
6.7.1.	Investments for climate mitigation and adaptation	419
6.7.2.	Cost of (in)action	421
6.7.3.	Greening public finances in Austria	422
6.8. Socie	tal aspects	424
6.8.1.	Just transition	424
6.8.2.	Migration	426
	Climate change and migration linkages	426
	Climate impacts on migration to and within Europe	427
	Climate impacts on migration to and	
	within Austria	427
	Political discourses on climate change	
	and migration in Austria	428
6.8.3.	Climate Change Education	428
6.8.4.	Media representation of climate change in Austria	430
Reference	s	432
Glossary	aar2.ccca.ac.at/glos	sary
List of abb	reviations <u>aar2.ccca.ac.at/abbreviat</u>	tions
Appendix	<u>aar2.ccca.ac.at/appen</u> d	dix/6
Full report	<u>aar2.ccca.ac.at/re</u>	port
### **EXECUTIVE SUMMARY**

Climate policies currently in place are insufficient to achieve the targets set out for Austria (*high confidence*). For Paris-aligned policymaking at all levels of government, it is essential to integrate climate policy comprehensively, both vertically and horizontally (*high confidence*). This integration should also encompass a clear prioritization of climate policies in case of target conflicts. In addition to mitigation policies, adaptation instruments are necessary to offset the negative effects of unavoidable climate change. Measures supporting a just transition are crucial for the successful implementation of climate policies. They can alleviate undesired social hardships and improve social and political acceptance (*high confidence*). {6.5}

The GHG emission reductions achievable through the current policy instruments are insufficient to meet the EU legal targets in line with the Paris climate goals (*high con-fidence*). International treaties related to climate protection have no direct effect ('nicht unmittelbar wirksam') on the legal framework for climate protection in Austria. This is evident in the reluctance of Austrian policymakers to implement a Climate Protection Act ('Klimaschutzgesetz'), which would have specified sectoral targets for GHG reduction up to 2030 and beyond. Austria's legally binding GHG emission reduction targets are derived from the effort sharing provisions set forth in EU legislation. The framework of EU energy and environmental law serves as the primary driver for Austrian climate law (*high confidence*). {6.3}

It is essential to consider the wider structural context in which climate governance is embedded when assessing Austrian climate governance (*high confidence*). To enable deep decarbonization, climate policy must address the structural context of climate governance, including the politico-juridical framework and wider societal (nature) relations. Given that such interventions inevitably result in winners and losers and may disrupt established routines, they meet resistance. Failure to address these resulting transformation conflicts could lead to backlash against climate action (*high confidence*). {6.2}

Although climate governance in Austria includes a large variety of instruments to foster effective mitigation and adaptation measures, these have not been sufficiently implemented by political actors so far (*high confidence*). This is exemplified by a reluctance of policymakers to phase out climate counterproductive subsidies and to green the tax system. Powerful economic and political interest groups in Austria are obstructing effective climate action and thus preventing the full potential of climate governance from being realized (*high confidence*). As a result, an implementation gap has emerged – ambitious climate targets exist without sufficient measures and instruments in place to achieve them (*high confidence*). Implemented instruments do not alter existing power relations, growth logics, and the dominant mode of production and living (*high confidence*). This is reflected in the lack of sufficiency-oriented 'avoid' policy instruments capable of significantly reducing demand for energy, materials and land. {6.4}

The significant and rising costs associated with climate inaction will exceed the budgetary costs of effective climate action (*high confidence*). Reaching net-zero and adequate adaptation will require considerable investments, with the projected benefits exceeding the costs (*high confidence*). The additional investment required is estimated at between 1.3 % and 4.8 % of GDP per year. Public budgets, in terms of both revenue and expenditure, are a crucial lever in this process. However, they are currently underutilized and undermined by counterproductive measures such as climate harmful subsidies (*high confidence*). In addition, effective climate action also requires the mobilization of private funds to bridge the investment gap. This can be achieved by providing adequate standards for green finance and reducing the carbon exposure of the economy (*high confidence*). {6.7}

Climate change losses in Austria, currently averaging EUR<sub>2019</sub> 2 billion annually, are projected to rise significantly by 2030 (EUR<sub>2019</sub> 2.5–5.2 billion) and by 2050 (EUR<sub>2019</sub> 4.3–10.8 billion) (estimates from 2020) (*limited evidence, high agreement*). Austria is currently impacted by climate-related damage and loss, and future climate change along with socioeconomic development will intensify these effects (*limited evidence, high agreement*). {6.7.2}

Just transition policies should address the asymmetric impacts of climate policy on employment and workers across sectors and regions. Furthermore, they must consider the wider inequalities and vulnerabilities exacerbated by the transition to climate neutrality (*high confidence*). While the concept of a just transition, retaining its labor-centric origin, remains a point of contention, with interpretations spanning from narrow socio-technical shifts to comprehensive social-ecological transformations of the social relations of production, the introduction of such policies is crucial to secure public acceptance and support for rapid decarbonization (*high confidence*). {6.8.1}

The links between climate change and migration are largely disregarded in the political debate in Austria, and significance is mainly attributed when climate change results in increased international migration to Austria (*limited evidence, high agreement*). The evidence base remains limited, revealing ambiguous effects on whether climate change will drive increased migration to Europe, with no clear indication of its specific impacts on Austria. {6.8.2}

Education is universally recognized as a key instrument for achieving all SDGs, and this is particularly true for Climate Change Education and SDG 13 'Climate Action' (*high confidence*). However, education itself needs to be transformed in order to fully exploit its transformative potential for sustainable and climate-friendly living in Austria. This applies to the entire educational spectrum, starting in early childhood and extending through adulthood, in both formal and non-formal settings. The contributions of all players and stakeholders to climate change education must be promptly and fundamentally transformed and strengthened (*high confidence*). {6.8.3}

Media representation of climate change likely increased in Austria since the mid-2000s. While dominant frames focus on the negative economic consequences of climate action, parts of the media represent climate change-skeptical positions. In contrast, several perspectives, such as the need for fundamental transformation of economy and society beyond market-based approaches and individual responsibility, as well as the question of social inequality tend to be underrepresented in the media discourse (*limited evidence, high agreement*). {6.8.4}

## 6.1. Introduction

Effective climate action combines legal, economic, political and societal aspects. This chapter assesses these aspects from an interdisciplinary perspective. It begins with the structural context of climate governance (Section 6.2). In a second step, the legal framework and the international commitments for Austrian climate policy are addressed (Section 6.3). The following section (6.4) assesses the role of institutions and actors. This is followed by an integrated assessment of existing policy instruments for climate change mitigation and adaptation (Section 6.5).<sup>1</sup>Section 6.6 assesses the legal remedies of climate liability by reviewing selected climate lawsuits. Public finances, investment requirements<sup>2</sup> and costs of (in)action for mitigation and adaptation are addressed in Section 6.7. The final section (6.8) examines the literature on various societal aspects, such as distribution and justice, migration, education, and media in the context of the climate crisis.

# 6.2. The structural and dynamic context of climate governance

Climate governance refers to the various interactions of political, economic and social institutions to steer social systems towards climate change mitigation and adaptation. These interactions are embedded into societal structures, such as political and legal frameworks, societal (nature) relations, and the political economy of modern societies, as well as the conflict dynamics and antagonisms that arise within them. Achieving the deep decarbonization necessary to meet the goal outlined in the Paris Agreement requires addressing these structures and dynamics that both enable and hinder such efforts (Novy et al., 2023a) (see Section 8.3.3) (robust evidence, medium agreement). The literature on sustainability transformations points to three core problem complexes that need to be addressed in order to understand which aspects of society need to be changed and why deep decarbonization is such a contested process (Newell, 2015; Scoones et al., 2015; Brand, 2016; Görg et al., 2017; Stirling, 2019; Morgan, 2020; Pichler, 2023):

(1) The first complex includes deep-rooted structures such as the modern growth economy, unequal and unsustainable societal (nature) relations, and associated mental infrastructures and political institutions that shape and impede transformation (*robust evidence, medium agreement*).

Political economy and political ecology research show that climate governance is embedded in a broader structural context that encompasses modern society, its economy, and the state (Perreault et al., 2015; Paterson and P-Laberge, 2018). These societal structures comprise both material and ideational dimensions and serve as both facilitators and entrenched obstacles to achieving deep decarbonization (Hirth et al., 2023). First, modern economies are seen as inherently driven by competition, economic growth, and profit-seeking (Feola, 2020; Hausknost, 2020), which poses fundamental challenges to a low carbon transformation (see Fouquet, 2019; Buch-Hansen and Carstensen, 2021) (see Cross-Chapter Box 7).<sup>3</sup> An economy prioritizing profit-seeking over basic human needs is identified as a main driver of ecological disruptions (Görg et al., 2017), mainly because competition forces producers to strive for unsustainable growth in order to maintain competitiveness (O'Connor, 1991; Saitō, 2016; Bärnthaler et al., 2021). This implies an expansionary dynamic that is also considered as a cause of geopolitical and geoeconomic tensions (Arrighi and Silver, 2001; Harvey, 2003; Scheffran, 2023; Brand and Wissen, 2024) and, since the late 18th century, is a driver of and itself driven by the large-scale use of fossil fuels (Malm, 2013; Ortiz, 2020; Christophers, 2022). Second, various forms of intra-societal inequality as well as unequal north-south relations have also been found to be entangled with structures of growth-oriented and fossil-fuel-based economies (Hickel, 2017; Daggett, 2018; Brand and Wissen, 2021). Inequalities such as class and unequal gender relations as well as racist divisions serve as structural features reinforcing and stabilizing unsustainable modes of producing and living, particularly as they enable to shift and thus externalize negative social and environmental impacts of economic activities to other places and populations (Lessenich, 2019; Sovacool et al., 2020; Brand and Wissen, 2021). These very inequalities are also identified as main drivers of vulnerability and lack of adaptation to the effects of climate change (Cappelli,

<sup>&</sup>lt;sup>1</sup> The focus here is on providing a general overview, while more detailed accounts and information on the effects of these policies on GHG emissions are given by the sectoral chapters of this report.

<sup>&</sup>lt;sup>2</sup> Green finance in particular is addressed in Cross Chapter Box 8.

<sup>&</sup>lt;sup>3</sup> This view contradicts standard economics, which sees these very characteristics as key levers for decarbonization (Hawken et al., 2000; GCEC, 2014; World Bank, 2023a).

2023). Third, structural unsustainability also has an ideational dimension, as it shapes widely shared conceptions of modern societies or 'mental infrastructures' (Welzer, 2011), that is, deep-rooted understandings of a good life, freedom (Bergthaller, 2017), progress and emancipation (Blühdorn, 2022). Fourth, different perspectives exist towards the extent structural unsustainability also shapes modern political institutions such as the state and liberal democracy.<sup>4</sup> On the one hand, there is a broad body of literature considering the state as the key for a deep decarbonization (Mazzucato, 2014; Eckersley, 2021; Babić and Dixon, 2023). Due to its capacity to allocate and distribute resources, facilitate innovation, regulate production and consumption, manage free-rider problems, and enforce legal frameworks, the state is seen as essentially capable to steer societies towards sustainability. On the other hand, scholars argue that growth imperatives and related deeply rooted societal relations and orientations shape and stabilize the modern state and liberal democracy. The functionality secured, e.g., by taxes, and legitimacy generated for instance by the welfare state and liberal democratic procedures of states co-evolved with and partially rely on a growth-based fossil economy (Hausknost, 2020; Pichler et al., 2020; Koch, 2022; Mitchell, 2023). Consequently, doubts arise about the state's steering capacity, as it is perceived not as a neutral regulator but as an asymmetric terrain (Jessop, 2007), that tends to favor strategies aligned with the dominant fossil fuel-based growth paradigm (Silvester and Fisker, 2023). This inclination results on the one hand in institutional dependencies on fossil fuels, reinforcing carbon lock-ins (Unruh, 2002). On the other hand, it constitutes a structural limitation of transformative state action to a narrow corridor of ecological modernization (Johnstone and Newell, 2018; Hausknost, 2020). For example, due to the automotive industry's central role in Austria's export-oriented industrial economy, its substantial contribution to the tax basis of the state, its role for the formation of labor relations and wage dynamics, and a persistent connotation of automobility with a good life within the Austrian population, state action to move beyond car-centered mobility systems as well as a conversion and downscaling of the automotive industry is relatively difficult to achieve (Pichler et al., 2021a). Therefore, climate governance as an issue specific arrangement embedded in these wider institutional arrangements is not only reactive and responsible for developing solutions to respond to the risks caused or compounded by climate hazards but is also constitutive of them and contributes to the emergence of risks (see Cross-Chapter Box 1). This implies that the state and other institutional contexts vary in their openness to ambitious climate action. For example, Austria's membership in the EU means that the structures and processes of climate governance in Austria are embedded in the European legislative architecture. This framework would allow for more ambitious climate governance objectives and measures than what has been pursued by Austrian governments so far (Nash and Steurer, 2019).

(2) The second complex entails more dynamic aspects such as interest and power constellations, as well as resulting conflicts that arise over deep decarbonization (*high confidence*).

Actors are differently positioned within unsustainable and unequal societal (nature) relations. As a result, they have different, sometimes even conflicting, interests, visions and policy preferences with regard to climate governance, and different capacities to shape it (van der Ven, 2016; Marquardt, 2017; Schneider et al., 2023). Accordingly, social science research on sustainability transformations pays much attention to the contested nature of climate action. First, the obstructive and delaying role of incumbents and (more recently) other powerful actors (e.g., in politics and the media) has been well studied (Geels, 2014; Meckling et al., 2015; Newell, 2019; Turnheim and Sovacool, 2020; Ekberg et al., 2022; Brulle et al., 2024; Plehwe et al., 2024). Sociological research highlights, e.g., how the strategies of powerful actors interact with dominant non-transformative attitudes and unsustainable practices within societies characterized by cleavages and class divisions (Fritz and Eversberg, 2024). Popular perseverance is at times seen as a major obstacle. This includes a protracted tendency of large parts of the citizenry to prefer the convenience of unsustainable lifestyles and to reject climate policy proposals, partly despite growing awareness of the climate crisis (Blühdorn, 2020; Beckert, 2024). Scholars of International Relations and International Political Economy highlight the international and global dimensions of contested transformations and show how manifest geopolitical and geoeconomic tensions shape climate governance (Paterson, 2020; Scheffran, 2023; Brand and Wissen, 2024). This multifaceted contestation of climate action is reflected in a burgeoning literature on transformation conflicts at multiple sites and scales (Brand and Wissen, 2024; Dörre et al., 2024; Eversberg et al., 2024; Fritz and Eversberg, 2024; Herring et al., 2024; Kalt, 2024),

<sup>&</sup>lt;sup>4</sup> The state is treated in more detail than other aspects here because much of the recent literature emphasizes its central place in governing the climate crisis.

and on how these conflicts may lead to backlash against climate action (Patterson, 2023).<sup>5</sup>

(3) The third complex includes the structurally and power-shaped, and inherently conflictive character of policies and policymaking (*robust evidence, medium agreement*).

The prioritization, formulation and implementation of climate-related policies, as well as their content, are framed and shaped by structural inequality, unsustainability, power and conflict. This is highlighted by critical policy studies (Pichler, 2023; Schneider et al., 2023) and has three main consequences: First, policies are understood as the outcome of conflicts and compromises between diverging interests and visions. Contestations between forces of inertia and change implies a conflict orientation of climate policy (Niedertscheider et al., 2018; Plank et al., 2021), as effective climate policy needs to counter powerful interests by building strategic state capacity (Meckling and Nahm, 2022). Second, policies differ in the extent to which they reproduce or transform the structural context of unsustainable societal (nature) relations. This is crucial, because the effectiveness of climate policies depends on their ability to change structures of unsustainability (Novy et al., 2023a). Third, it is highlighted that policymakers are more likely to adopt policies that align with existing societal structures, dominant interests, and power relations than more transformative policies, as these non-disruptive policies avoid challenging dominant growth imperatives and their associated interests (Markusson et al., 2018). Such institutional biases may partly explain the lack of sufficiency or 'avoid' policies in Austria and the EU, i.e., those focusing on reducing demand rather than merely increasing efficiency (Zell-Ziegler et al., 2021; Jarre et al., 2024; Brad et al., 2025) (see also Cross-Chapter Box 4). Fourth, to address popular resistances and increase the acceptability and legitimacy of climate policies, considerations of justice and participation need to be firmly embedded in climate governance. This is all the more important given the unequal distribution of responsibility for climate change, the unequal distribution of its impacts, and the unequal distribution of the means to adapt to it (Newell et al., 2022).

# Cross-Chapter Box 7. Degrowth and other beyond growth concepts: The desirability and feasibility of (limiting) growth

Jarmo S. Kikstra, Ernest Aigner, Richard Bärnthaler, Halliki Kreinin, Danyal Maneka, Michael Miess, Andreas Novy, Nicolas Roux, Thomas Schinko

## Context

Critiques of relentless economic growth have a long history. Active academic debates on steady-state economics (Daly, 1973) and attempts to model and illustrate the limits to growth (Meadows et al., 1972) are already more than half a century old. The debate is fierce and continuing, discussing the feasibility and desirability of different elements of growth (Meadows et al., 1992; Nordhaus et al., 1992). Here, the core question is what kind of economic structures are conducive to improving well-being while achieving ecological stability and sustainability (Hickel and Hallegatte, 2022). A strong consensus exists on the need to leave behind GDP as an indicator of social progress (Stiglitz et al., 2009; Costanza et al., 2014), but there are different frameworks to analyze what this means for policymaking and the organization of economies around the world. It includes recognizing that while more energy use is required to provide decent living standards for all that are currently deprived (Kikstra et al., 2021), there is a large potential for simultaneous reduction of energy demand, which can bring multiple interacting benefits (Grubler et al., 2018; Creutzig et al., 2022). Moreover, moving beyond growth requires replacing the existing economic growth paradigm with alternative social imaginaries and narratives (Fournier, 2008; Latouche, 2009). Skepticism towards green-growth strategies is widespread among climate-policy researchers (King et al., 2023), and resource and emissions decoupling in the past decade have been highly insufficient

<sup>&</sup>lt;sup>5</sup> However, studies also find evidence for broad popular support for climate policy focused on green investment and industrial policy as well as measures compensating unpopular climate policy instruments like carbon pricing in different European countries (Abou-Chadi et al., 2024).

to be on track to meet climate targets, both global targets and targets for Austria (Haberl et al., 2020; Vogel and Hickel, 2023), across many biophysical pressures (CCBox 7 Figure 1).

## Beyond growth frameworks

Beyond growth frameworks see most current economic systems as unsustainable and unable to sufficiently increase human well-being within a socially just transition process unless they address perverse growth incentives. Approaches to structural economic changes, tools, and preferred policy interventions differ between frameworks. Definitions used here are in CCBox 7 Table 1. This box focuses on degrowth, a prominent growth-critical approach (Kallis et al., 2018). Degrowth and green growth are contrasted as alternative climate mitigation strategies (CCBox 7 Table 1) to identify common and differing insights on what type of growth is feasible and desirable. A key distinguishing feature is the role of 'avoid' and sufficiency strategies for climate mitigation, which are valorized much more heavily in degrowth (see also Cross-Chapter Box 4).

Framework	Definition
Agrowth	An argument that sees GDP as an unreliable indicator for progress and argues that we should not focus on it.
Beyond growth	An umbrella term for terms and approaches that are critical of continued economic growth.
Decent Living Standards	"A set of material requirements that are essential for human flourishing" (Rao and Min, 2018).
Degrowth	A research field and socio-political movement focusing on a targeted, redistributive and democrati- cally planned downscaling of production and consumption, primarily in industrialized countries, as a way to achieve sustainability and well-being, while minimizing key risks.
Doughnut economics	"A multi-criteria boundary tool to support economic planning by identifying environmental limits and social thresholds and is agnostic towards GDP growth" (Raworth, 2017), as it does not see GDP as a good indicator for environmental or social progress.
Feminist ecological economics	An approach to economics that centers household and community (re)production and interrela- tions with nature, for instance through care and unpaid work, and justice related to gender, ecologi- cal degradation, decolonization, and global inequities.
Foundational economy	An approach to restructuring the economy to prioritize the provisioning of everyday human needs like food, housing, health services and transport within planetary limits.
Green growth	A set of policies that aim to bring together and simultaneously stimulate sustainable development and economic growth.
Post-growth	A broad set of approaches that challenge the necessity, feasibility, and desirability of continuous economic growth, especially in advanced economies.
Steady-state economics	A concept focused on maintaining a sustainable non-growing economy, including a constant level of resource use, population, and capital stock, with a focus on social welfare not economic growth.
Sufficiency	A set of measures and daily practices that avoid demand for energy, materials, land and water while delivering human well-being for all within planetary boundaries.
Well-being economy	A set of measures that delivers on five core needs for ecological and human well-being: Dignity, connection, nature, fairness, and participation.

#### CCBox 7 Table 1 Overview of relevant terms.

## (De)growth in the context of climate change mitigation and broader sustainable development

Many proposals have attempted to quantify what might be considered 'enough' and 'too much'. Proposed minimum requirements for human flourishing have been based on theories of the human needs (Max-Neef, 1991; Sen, 2012; Lamb and Steinberger, 2017) that "avoid serious harm and are universalizable, objective, empirically grounded, non-substitutable and satiable" (Gough, 2015). For one example, the Decent Living Standards (Rao and Min, 2018), minimum energy requirements are less than half of current global energy use (*medium confidence*) (Millward-Hopkins et al., 2020; Kikstra et al., 2021). Total levels consistent with decent living standards for all are higher due to inequality (Millward-Hopkins, 2022) but can be lowered with improvements in service provisioning systems and reduced within-country inequality (Kikstra et al., 2024a). The minimum material footprint to support decent living standards has been estimated at 3–13 t per capita per year (Vélez-Henao and Pauliuk, 2023) (*medium confidence*), which is also less than half of the current annual material footprint, estimated at 24–33 t per capita in Austria pre-COVID (Giljum et al., 2019; Eisenmenger et al., 2020; Lenzen et al., 2022; Plank et al., 2022; UNEP IRP, 2023; Statistik Austria, 2024) (*high confidence*). The federal government's Circularity Strategy has recently set a 7 tMF/cap target for 2050, which expresses the intention to lower resource use levels (see Cross-Chapter Box 5).

Ceilings on activity are more often linked to large-scale environmental thresholds and risks such as planetary boundaries, tipping points, and climate impacts more broadly. Maximum activity levels are then most often linked to carbon budgets corresponding to global warming levels (see Chapter Box 1.1), through estimates of carbon intensities of certain activities, and through the concepts of consumption and production corridors (Di Giulio and Fuchs, 2014; Bärnthaler and Gough, 2023). Additionally, ethical arguments have been made to determine upper limits based on preferences and perceptions of fair levels of income inequality (Osberg and Smeeding, 2006), and a recent survey distinguished the 'rich' from the 'super-rich' in the Netherlands at between EUR 1 million and 3 million total personal wealth (Robeyns et al., 2021).

Whether it is desirable and feasible for climate mitigation strategies to downscale a particular activity depends strongly on the sector and country, noting different levels of development, inequality, and needs. Less necessary, and ecologically destructive sectors may include short-distance flights (Dobruszkes et al., 2022) and other inefficient and harmful mobility (Muller et al., 2011; Cattaneo et al., 2022), housing-as-asset (zu Ermgassen et al., 2022), leisure (Smetschka and Wiedenhofer, 2023), fast fashion (Niinimäki et al., 2020), affluent energy overconsumption (Büchs et al., 2023), and animal-based foods (Kozicka et al., 2023). At such a policy level, many of these proposals are supported by both degrowth and green growth analyses, although the approaches to identifying and embedding them differ.

While advocates for degrowth strategies typically call for a clear break with standard economic doctrine, there are elements of agreement with conclusions from standard economic analyses. A full accounting of the external costs of energy calls for rapidly shrinking fossil-based energy production (Muller et al., 2011). Induced technological change explicitly calls for a shift from 'dirty' to 'clean' sectors (Acemoglu et al., 2012), with demand reduction playing a significant role. IEA (2020), for example, shows how, by 2050, 50 % of global steel demand could be met by reuse and recycling, under a 1.5°C decarbonization pathway.

Challenges to green growth strategies have been more extensively researched than challenges to degrowth. For example, efficiency improvements can lead to rebound effects, i.e., induce increased demand, partially offsetting environmental gains (Gillingham et al., 2016; Moshiri and Aliyev, 2017; Raimund, 2023). Much less is known about how strong challenges, including possible rebound effects (Sorrell et al., 2020), to achieving a beyond growth pathway could be, with limited evidence available about the effects of systematically targeted downscaling on employment and productivity, on international competitiveness and financial stability in the case of unilateral or cooperative implementation, or on the financial ability to finance the energy transition, or on the extent to which the transition to renewables induces productivity gains and resulting economic growth (Arkolakis and Walsh, 2023).

From a macro-perspective, to achieve sustainability, economic activity needs to be sufficiently decoupled from emissions as well as from material, energy, water and land resources. To mitigate climate change, the emission intensity of economic activity and energy consumption needs to decline faster with higher economic growth and the same emission reduction target, which may be an indicator of a higher macroeconomic challenge. At the same time, to stabilize at any temperature, CO<sub>2</sub> emissions need to go to zero, ultimately requiring full decoupling. Even if a degrowth scenario achieves absolute decoupling from economic growth in both GHG emissions and final energy, this does not necessarily mean that the year-on-year change in emissions intensity or final energy intensity is different from a green growth scenario (CCBox 7 Figure 1f). At the same time, higher energy demand in a green growth scenario requires a larger energy supply system, which requires more energy and metals such as copper for the energy system itself (Granier et al., 2007; Lesk et al., 2022; Slameršak et al., 2022; Wang et al., 2023), as illustrated by the scenario highlighted in CCBox 7 Figure 1f, and may thus come with higher environmental pressures beyond emissions, potentially lowering feasibility of decoupling from all environmental pressures (*limited evidence, medium agreement*). Economic growth is also related to biodiversity loss through greater resource consumption and higher emissions (Otero et al., 2020), while the combination of degrowth and efficiency gains can lead to simultaneous emissions and health benefits (Bodirsky et al., 2022).



CCBox 7 Figure 1 Past trends in (a) Biophysical pressures and the exceedance of ecological thresholds for Austria; (b) Trends in social provisioning and the meeting of social minima for Austria; (c) Share of thresholds crossed in 2015 across all countries with Austria highlighted compared to a state where sustainability would be achieved. All data from panel a–c is from Fanning et al. (2022). (d, e, and g) A modeling example comparing the need for upscaling renewables (d) under a 1.5°C consistent target for Australia for a 'continuing GDP growth' versus a 'slowing GDP growth' scenario, with their respective (e) total energy use, GDP, and emissions pathways, and (g) emissions and energy intensities. All data from panel (f) based on Kikstra et al., (2024b). (g) Theoretical representation of a degrowth logic for climate change mitigation compared to common mitigation modeling approaches, based on Li et al. (2023).

Integrated modeling approaches that adequately reflect the dynamics of a global degrowth transition across sectors in coherent global integrated frameworks with high regional detail do not yet exist, but many partial studies have provided explorations and preliminary quantifications. Exploratory country-level modeling indicates that an emissions reduction can be combined with reductions in consumption and increases in social prosperity (D'Alessandro et al., 2020), while lowering the challenges of upscaling renewables and carbon prices fast enough to meet ambitious climate targets (Kikstra et al., 2023) (CCBox 7 Figure 1f). Global studies are few, and range from simplified energy-emissions modeling, to illustrate the space for developing new process-based mitigation pathways (Keyßer and Lenzen, 2021), to first versions of process-based modeling of the conflict between economic growth, climate policy and resource sustainability, with a green-growth scenario failing to meet climate goals and a post-growth scenario meeting emissions objectives (Nieto et al., 2020). For food and land systems, a modeling study showed the possibility of achieving a steady-state, net-zero GHG food system that improves nutritional outcomes (Bodirsky et al., 2022). A few studies have also looked at post-growth modeling using stock-flow consistent models (Victor, 2012; Jackson and Victor, 2020; Sers, 2022). While these models tend to remain conceptual rather than empirically calibrated, one recent working paper described the possibility of finding higher inflation in 'the North' and somewhat slower economic growth in 'the South' due to reduced international trade if 'the North' unilaterally shifted to degrowth while 'the South' continued to focus on economic development (Leoni et al., 2023).

As discussed above, qualitative literature exists on the feasibility, desirability, and economic dynamic effects within and across countries for a transformation beyond growth (Kallis et al., 2018; Kerschner et al., 2018; Hickel et al., 2022). However, attempts to quantify feasibility and desirability are mainly explorative (D'Alessandro et al., 2020; Jackson and Victor, 2020; Nieto et al., 2020; Keyßer and Lenzen, 2021; Moyer, 2023; Kikstra et al., 2024b). Instead, quantifications demonstrate the need for 'avoid' strategies by showing the shortcomings of green growth strategies (Vogel and Hickel, 2023) or by comparing high and low resource demand scenarios (Grubler et al., 2018; Sugiyama et al., 2024).

## **Key elements**

In short, the degrowth approach:

- Emphasizes empirical evidence on the difficulties of decoupling GDP-growth from growth in emissions, energy, and material footprints, and is concerned about the speed and scale needed to achieve climate targets without the relying on speculative levels of deployment of negative-emission technology;
- Problematizes the limited number of mitigation strategies available due to growth imperatives and concerns about rebound effects without concurrent sufficiency-based policies;
- Links emission and resource-use reductions in the spheres of consumption and production;
- Aims at guaranteeing and potentially expanding necessary forms of consumption/production, for instance through
  collective forms of service provisioning, while downscaling less necessary and ecologically destructive forms of production/consumption, for instance through sufficiency and 'avoid' strategies;
- Raises a key question for democratic deliberation and democratic planning: What do we as a society need more of and what do we need less of to live a good life within planetary boundaries?
- Takes a eudaimonic approach to well-being and welfare, assuming universal objective basic needs or decent living standards, whose fulfilment 'for all' is prioritized over growth to satisfy wants or preferences when the two conflict;
- Centers on national and international justice implications, problematizing the disproportionate share of the Earth's resources that richer countries appropriate both in terms of common resources such as carbon emissions into the atmosphere, and through processes of unequal exchange (e.g., Dorninger et al., 2021);
- Intends to move our socio-economic-ecological system from scarcity to (radical) abundance by re-connecting humans to each other, land, nature, and a culture of care, while de-commodifying and de-colonizing the socio-economic system.

For climate mitigation, compared to green growth, degrowth approaches:

- May reduce geophysical and technological feasibility concerns, and therewith enable faster emissions reductions, due to a reduced speed and scale of upscaling renewables for the same emissions reduction target compared to a green growth approach;
- Reduce environmental pressures beyond climate change due to reduced resource needs (high confidence);
- Come with a stronger focus on improving eudaimonic human well-being and focus less on preferences;
- May come with higher sociopolitical feasibility concerns due to rapid economic, sociocultural, and economic breaks with past trends.

# 6.3. Regulatory framework of adaptation and mitigation

## 6.3.1. International: Requirements of the Paris Agreement

The Paris Agreement is currently the central international treaty for combating climate change, committing signatories to limit global warming to well below 2°C and possibly to 1.5°C above pre-industrial levels. Unlike its predecessor, the Kyoto Protocol, it contains neither numerical reduction targets nor a sanction mechanism (Binder and Ritter, 2023). It is not directly applicable in Austria. Other treaties and conventions also have an indirect impact on climate protection in Austria, such as the General Agreement on Tariffs and Trade (Mayr et al., 2021) or the Convention on International Civil Aviation (Karimi-Schmidt, 2019).

Article 3 of the Paris Agreement requires all parties to contribute to the reduction targets. These contributions are to be determined by the states themselves but must be based on an ambitious level and include a steady increase with the aim of achieving the objectives of the agreement. The EU and its Member States submitted a joint Nationally Determined Contribution and updated their initial commitment in December 2020 to a net domestic GHG emission reduction of at least 55 % by 2030 compared to 1990. The EU Effort Sharing Regulation (ESR) partially transfers this commitment to Member States by setting separate binding annual GHG emission targets for each Member State for emissions not covered by the EU Emissions Trading System (EU ETS) (Handig and Stangl, 2023).

Among other commitments, Article 7 of the Paris Agreement sets out the goal of enhancing adaptive capacity, strengthening resilience and reducing vulnerability to climate change in order to contribute to sustainable development and ensure adequate adaptation in the context of the temperature goal.

## 6.3.2. The EU regulatory context

The European Climate Law (Regulation (EU) 2021/1119) establishes a framework for reducing GHG emissions by sources and increasing removals of GHG by sinks. It sets a binding interim target of climate neutrality by 2050. At the same time, it aims to make progress in implementing adaptation strategies. In order to achieve the binding target of climate neutrality, the binding target within the EU is to reduce GHG emissions by at least 55 % by 2030 compared to 1990 (Frenz, 2022; Handig and Stangl, 2023). The European Climate Change Act is a central part of the EU's Fit for 55 package, a set of proposals to revise and update EU legislation. Most proposals have already been adopted (see Table 6.1). EU law is the driving force behind Austrian national climate legislation (Ennöckl, 2020; Hofer, 2024a) (*high confidence*).

The EU's climate change strategy is based on reducing GHG emissions, increasing the share of renewable energy and energy efficiency, and addressing the climate change impacts of land use and land-use change. The overall GHG emission reduction target is divided between sectors covered by the EU ETS and sectors covered by the ESR. The EU ETS currently covers emissions from energy-intensive industry, energy utilities and domestic aviation, and will be extended to include emissions from maritime transport in 2027. These sectors are subject to an EU-wide reduction target of 62 % for 2030 compared to 2005 (Directive (EU) 2023/959) (see Section 6.5.1). For the ESR sectors, an overall emission reduction target of 40 % applies for 2030, and specific targets have been defined for individual Member States. Austria is obliged to achieve a reduction of 48 % by 2030 compared

Measures	New legal measure / Revision of existing legal measure (n/r)	Legal act
Stronger emissions trading system including aviation	n/r	Directive (EU) 2023/959
Extending emissions trading to maritime, road transport, and buildings	n/r	Directive (EU) 2023/959
Restructuring taxation of energy products and electricity (not adopted)	n/r	COM/2021/563 final
New Carbon Border Adjustment Mechanism (CBAM)	n	Regulation (EU) 2023/956
Updated ESR	r	Regulation (EU) 2023/1319
Updated Land Use, Land Use Change and Forestry (LULUCF) Regulation	r	Regulation (EU) 2023/839
Updated Renewable Energy Directive (RED)	r	Directive (EU) 2023/2413
Updated Energy Efficiency Directive (EED)	r	Regulation (EU) 2023/955
Stricter $CO_2$ -performance for cars and vans	r	Regulation (EU) 2023/851
New infrastructure for alternative fuels	n	Regulation (EU) 2023/1804
ReFuelEU: more sustainable aviation fuels	n	Regulation (EU) 2023/2405
FuelEU: Cleaner maritime Fuels	n/r	Regulation (EU) 2023/1805
Reduction of methane emissions in energy sector	n	Regulation (EU) 2024/1787
Updated energy performance of buildings	r	Directive (EU) 2024/1275
Social Climate Fund	n	Regulation (EU) 2023/955
Decarbonization of gas markets and promotion of hydrogen	r	Directive (EU) 2024/1788

Tab	le 6	.1 (	Dverview	of F	it fo	r 55	legal	measures
-----	------	------	----------	------	-------	------	-------	----------

to 2005 (Regulation (EU) 2023/857). EU law provides for a governance mechanism in this area (Dederer, 2021). Integral to this are integrated national energy and climate plans for ten-year periods, which the Member States are required to prepare and report to the European Commission. They are obliged to implement effective instruments to reduce GHG emissions, often based on European directives on buildings, vehicles, energy efficiency, renewable energy, etc. In addition, a separate EU-wide emissions trading system ('ETS 2') will be introduced in 2027 for buildings and transport, as well as for energy supply and manufacturing not covered by the EU ETS. A Social Climate Fund (SCF) will be established, including revenues from ETS auctioning, with the general objective of supporting vulnerable households as well as measures and investments to reduce emissions in the road transport and buildings sectors (Regulation (EU) 2023/955). An EU-wide legislative framework to reduce methane emissions in the energy sector complements efforts to achieve significant GHG emission reductions (Regulation (EU) 2024/1787).

Article 5 of the European Climate Change Act requires the European Commission and Member States to develop strategies for adaptation to climate change. The new EU Adaptation Strategy was adopted in 2021 and aims to achieve a climate-resilient Union by 2050 by promoting smarter, faster and more systemic adaptation and strengthening international action on adaptation. Through 'smarter adaptation', the EU aims to improve data and risk assessment tools to advance knowledge on adaptation and manage uncertainty. Under the goal of 'faster adaptation', the EU focuses on the need for faster and more comprehensive adaptation by accelerating the development and deployment of adaptation solutions. The objective of 'more systemic' adaptation refers to the need for policy developments at all levels and sectors, taking into account the cross-cutting priorities of local resilience, integration of adaptation into macro-fiscal policies and implementation of nature-based solutions for adaptation (COM/2021/82 final).

## 6.3.3. The national regulatory context

The Austrian legal framework for climate protection is largely based on EU legislation, which is mainly determined by the European Climate Change Act, the ETS Directive, the ESR and the LULUCF Regulation, and supplemented by numerous sector-specific regulations and directives. Purely national regulations are of secondary importance. Member States have policy space mainly in those sectors that are currently not covered by the EU ETS (transport, buildings, agriculture). It is precisely in these areas that national legislative measures have so far been taken only selectively (Ennöckl, 2020).

Within the existing division of powers between the federal government and the federal provinces in the Federal Constitutional Law, there is no explicit area of responsibility for climate protection measures. Climate protection, as well as environmental protection in general, is a cross-cutting issue (Handstanger, 2022; Ennöckl, 2023a) (*high confi*- dence). The Constitutional Sustainability Act ('BVG Nachhaltigkeit') (BGBl. I Nr. 111/2013) contains a constitutional commitment to sustainability and thus to climate protection in all areas of government action. It sets a mandate for action for all responsible actors in legislation and implementation (Sander and Schlatter, 2014). Insufficient efforts in these areas have not yet been taken up by the courts. The legal effectiveness of the Constitutional Sustainability Act is therefore considered to be very low (Ennöckl, 2023a; Kirchmair and Krempelmeier, 2023) (medium confidence). In Austria, an appropriate design of the legal framework can make a significant contribution to the transition to a climate-neutral society and economy. However, this requires addressing and resolving deeper structural problems of the law, including the reorganization of the legal institutions and infrastructures that co-constitute and regulate the political economy. Thus, necessary legal changes affect not only environmental law, but also essential parts of economic law in general and the overall legal order of fundamental socio-economic structures (Lachmayer and Müller, 2023) (see Section 6.2).

In 2011, Austria was an early adopter of a Climate Protection Act ('Klimaschutzgesetz') at the national level. The Climate Protection Act does not include any competences, strategies or instruments for the reduction of GHG emissions, but rather consists of a mere mechanism to distribute the emission reduction targets in 2020 set by EU and international law to different sectors (Ennöckl, 2023b). The law is therefore mainly described as a 'mandate or negotiation law' (Ennöckl, 2020; Schulev-Steindl et al., 2020). Adaptation to climate change plays a minor role in the Austrian Climate Protection Act.

Since the GHG reduction targets contained in the law expired in 2020, Austria has been de facto without a functioning nationally determined climate protection act. The Federal Energy Efficiency Act (BGBl. I Nr. 72/2014) provides for energy efficiency targets at the national level, which should contribute to achieving climate neutrality by 2040 (Suchanek et al., 2023). Although the goal of achieving climate neutrality by 2040, as envisaged in the Government Program 2020, has not yet been explicitly enshrined in legislation, some legal provisions refer to this target. For example, § 35 (12) (a) of the Federal Energy Efficiency Act states that the goal is to contribute to achieving climate neutrality by 2040.

Concerning the expansion of renewable energies, Austria has set a target of producing 5 TWh of renewable gas and raising electricity production from renewable sources on a yearly balance<sup>6</sup> to 100 % until 2030 (Ennser, 2022). It has introduced a set of measures to accelerate the deployment of renewable energy production. Regarding energy efficiency, Austria aims to achieve a linear decrease in final energy consumption up to a maximum of 920 PJ in 2030. Various laws, e.g., on renewable gas are under consultation but not yet decided or will have to be amended once the directives from the Fit for 55 directives come into force.

# 6.4. Institutions and actors of climate governance in Austria

This section assesses Austrian climate governance strategies, their institutional design and the actors involved. It analyses the interests and power relations involved in governance processes that may limit their effectiveness. The concept of governance refers to a mode of political steering that includes, but is not limited to, state actors. It can involve multiple scales (from local to international) and a variety of private, public and civil society actors and institutions. It is typically conceived as a consequence of the failure of states to 'govern' complex societal problems, such as climate change, through vertical 'top-down' regulation alone (Jordan et al., 2003; Biermann et al., 2012; Brand, 2016; Bäckstrand and Kronsell, 2017). Climate governance often focuses on technological (and sometimes social) innovations and their diffusion across economies and societies. However, as the climate crisis is a time-bound problem, a complementary focus on exnovation, i.e., the "intended dismantling of practices, products, technologies and infrastructure" (Krüger and Pellicer-Sifres, 2020, p. 117) becomes ever more important. This includes, for example, phasing out fossil energy infrastructures such as coal-fired power generation, as well as phasing out fossil subsidies and socio-technical systems such as the internal combustion engine and its various applications. In this report, the relationship between innovation and exnovation in climate policy and governance is systematically reflected in the Avoid-Shift-Improve (ASI) framework (see Cross-Chapter Box 4). Governance of exnovation is necessary to shift societies from fossil to sustainable socio-technical systems. It is also essential to implement forms of climate governance that aim to avoid unnecessary and harmful consumption patterns (robust evidence, medium agreement).

<sup>&</sup>lt;sup>6</sup> This means that the percentage is to be determined based on the total quantity of energy consumed in a year and does not have to be achieved in all hours, days or months of the year.

Austria has long had a reputation as a leader in environmental policy (Lauber, 1997). However, despite this reputation, Austria's weak climate policy record contrasts with its strong declared climate ambitions (Melidis and Russel, 2020). Therefore, there is a strong mismatch between Austria's (powerfully produced) self-image as a country with relatively high environmental standards and the actual unsustainability in many sectors. The general climate policy consensus in the Austrian state apparatuses is to deal with climate change in a mode of 'ecological modernization', i.e., to change the energy and resource base without changing social structures and power relations as well as the prevailing modes of production and consumption. This ties in with the clientelist character of Austrian climate policy, in which certain state apparatuses often act in favor of certain groups, e.g., farmers or the business sector (Steurer et al., 2020; Nash and Steurer, 2023) (medium confidence).

In the last 20 years, the Austrian government and parliament have formulated three climate strategies and a Climate Protection Act (Nash and Steurer, 2023) (see Section 6.3.3), which has been amended several times. However, climate policy in Austria remains fragmented and incremental and has so far failed to achieve its goals (Niedertscheider et al., 2018; APCC, 2023; Tebecis, 2023). Climate governance has not been a coherent government priority (Scherhaufer and Clar, 2021). As in other European countries, climate governance in Austria has so far mainly relied on 'soft' measures - so-called 'New Environmental Policy Instruments' - which include voluntary measures and market-based instruments, that do not constitute deep interventions in the economic structure of society or in consumer practices (Wurzel et al., 2013; Feichtinger et al., 2021). It remains deeply committed to a perspective of technological change and market-driven innovation to stimulate 'sustainable' or 'green' growth, while there is no vision for sufficiency-oriented governance, i.e., policy frameworks and governance mechanisms that address the systemic drivers of the climate crisis (Novy et al., 2023b). So far, climate governance has not attempted to steer the scale, nature or scope of final consumption (e.g., through mandatory carbon labeling or choice editing) (Maniates, 2010). The incremental nature of Austrian governance can be exemplified by the automobile sector, where government responses to the climate crisis have so far relied on market-based solutions, technological fixes (e.g., shifting to electric vehicles instead of public transport), international competitiveness, and technological innovation, and have failed to achieve the much-needed downscaling and conversion of production and consumption (Krenmayr et al., 2020; Pichler et al., 2021b) (*robust evidence, medium agreement*).

## 6.4.1. Institutions

In this chapter, 'institutions' refers to the political, legal, and economic rules and forms of organizing climate governance. Institutions are located at and traverse multiple levels of scale, from local/municipal to regional, national, European, and international (Pierre and Peters, 2020). Here we focus on the national and federal provincial level. At the national level, the establishment of a Ministry for Climate Protection, Environment, Energy, Mobility, Innovation and Technology (BMK) in 2020 was a symbolically important step for climate governance in Austria (Nash and Steurer, 2023), both because climate was explicitly mentioned in the title of the ministry and because it was given priority over the other areas covered by the ministry. Moreover, the integration of the key sectors of transport and energy into the ministry's remit brings important levers for climate change mitigation (Eberl et al., 2020).

## Federalism

The fragmented structure of the Austrian government and its respective agencies remains an obstacle for the realization of ambitious climate policies at the national level. In particular, Austrian federalism is more of a hindrance than an enabler for the formulation and implementation of ambitious climate policies. Conflicting interests between the federal and federal provincial levels favor either agreements with unambitious targets, unclear responsibilities, a lack of sanction mechanisms, or no agreement at all (Steurer et al., 2020; Scherhaufer and Clar, 2021) (medium confidence). The recently adopted Fiscal Equalization Agreement for the period 2024-2028 does not provide for substantial reforms to strengthen intergovernmental climate governance in Austria (Bittschi et al., 2024). In terms of institutional design, national climate councils, as provided for in the European Climate Change Act, can play an important role in convening stakeholders and providing expert scientific advice on climate policy to relevant authorities (Evans and Duwe, 2021). They can also monitor relevant policy developments (EEA, 2021a). Despite the encouragement to do so, not all Member States have yet established an independent climate advisory body.7

<sup>&</sup>lt;sup>7</sup> These are ESABCC, Denmark, Estonia, Finland, France, Germany, Greece, Ireland, Luxembourg, the Netherlands, Slovenia, Sweden, Iceland and the United Kingdom.

## Social partnership

The Austrian social partnership, an important mechanism for well-designed economic and social policies, has not yet reached the heights of climate policy (Tobin, 2017; Niedertscheider et al., 2018; Brand and Niedermoser, 2019) (medium confidence). The social partnership was designed to reconcile conflicting social interests in an era of rapid fossil-based economic and socio-metabolic expansion, and as such may not be ideally suited in its current form to manage the decarbonization of a complex economic structure. Empirical research has even identified the Austrian social partnership as a force obstructing climate action (Brand and Pawloff, 2014). However, in recent times, important initiatives have been launched to 'green' the Austrian social partnership, e.g., the initiative 'Growth in Transition' that started in 2008 or a more recent initiative to strengthen the perspective of a comprehensive social-ecological transformation within the Chamber of Labor (Arnecke, 2024).

## Institutional focus on climate change adaptation

There is evidence that Austria's governance system is better suited to enabling effective adaptation to climate change than mitigation (Clar and Steurer, 2014; Steurer and Clar, 2018) (medium confidence). Clar and Steurer (2014, p. 23), for example, argue that Austrian federalism "could prove helpful in mediating between national guidance and local adaptation". However, the existing institutional set-up seems to be more geared towards conventional strategies to help individual and collective actors cope with or gradually adapt to climate change (Fekete et al., 2022), e.g., via sophisticated disaster risk management systems or adapted building standards (Nordbeck et al., 2019). In contrast, it seems much less prepared for what Fedele et al. (2019, p. 116) call "transformative adaptation", that is "fundamental systems' changes that address root causes of vulnerability to climate change" (cf. Brand and Wissen, 2024) (high confidence).

Natural hazard management and climate change adaptation – internationally, but also in Austria – are currently largely uncoordinated in dealing with some of the same climate-related risks (see Cross-Chapter Box 1). In terms of risk analysis and decision making, natural hazard management focuses strongly on experience, while climate change adaptation focuses mainly on possible future developments (Schinko et al., 2017; Leitner et al., 2020) (*high confidence*). As climate change impacts intensify in the future and adaptation limits may be reached, current incremental risk management measures may no longer be sufficient to maintain societal goals and values (Dow et al., 2013; Preston et al., 2013; Schinko et al., 2024) (*high confidence*). To improve the effectiveness and efficiency of climate-related risk management, the two currently largely independent approaches need to be more closely integrated into a holistic approach – a concept known as 'climate risk management' (Jones et al., 2014; Mechler et al., 2014; Schinko et al., 2017) (*high confidence*). In order to achieve this integration, it is crucial to first understand the current, often isolated governance structures in specific countries or regions and thus identify the potential synergies, but also conflicts, at the interface between these two policy areas (Leitner et al., 2020) (*high confidence*).

Moreover, historically, risk management strategies have been developed for single hazards (Leitner et al., 2020), while advancing climate change will increasingly cause compound climate-related risks (e.g., Zscheischler et al., 2018) (*medium confidence*). To overcome existing path dependencies (Hanger-Kopp et al., 2022; Seebauer et al., 2023) of single-hazard risk management arrangements, which may ultimately cause barriers to transformative adaptation efforts and lead to the intensification of existing societal inequalities, Thaler et al. (2023) argue for a compound risk governance system and management practice (*medium confidence*).

The various institutional deficits with regard to both mitigation and adaptation point to the need for substantial changes, not only in existing policies, but also in the polity, i.e., the (political, legal and economic) institutional framework within which mitigation and adaptation policies are formulated and implemented (Pichler, 2023).

### 6.4.2. Actors

Actors relevant to climate policy and policymaking can be found both inside and outside institutions and can be collectively organized in institutions such as political parties. As in many other democratic industrialized countries today, the existing constellation of actors in Austria harbors the danger of 'climate delay' or 'climate policy dismantling', i.e., the possibility of obstructing and delaying already agreed climate policies and decarbonization targets, as well as cutting, reducing or abolishing existing policies (Lamb et al., 2020a; Ekberg et al., 2022; Paterson et al., 2024) (see Section 8.5.1). After a phase of progressive climate policy formulation starting around 2019, the EU and many of its Member States, including Austria (Arnecke, 2024), are currently facing an era of reluctance and restraint with regard to the further development of climate policy (Plehwe et al., 2024) (*high confidence*).

## **Political parties**

The positions and strategies of political parties on climate change in Austria vary greatly. In the legislative period of 2019–2024, the government has set the target of climate neutrality by 2040. There are, however, substantial differences among parties when it comes to concrete measures to achieve this goal, despite a formal commitment of all parties to the overall objective of climate neutrality (Kirchengast et al., 2019a) (*high confidence*).

Ambitious climate action - mostly within the corridor of ecological modernization - is demanded primarily by the Austrian Green Party. Accordingly, it is also the party most clearly identified with the issue of climate change in Austria, by experts and the electorate (Dolezal, 2016; Abstiens et al., 2021). While the Green Party has benefited considerably from the issue salience of climate change in the parliamentary elections of 2019 and in several regional elections, more recent elections, e.g., to the European Parliament, have seen Green losses in an atmosphere of increasing societal polarization and strong gains of far-right parties across Europe, including in Austria (Cunningham et al., 2024; Schwörer, 2024). While this provides an ambiguous picture of the impact of polarization and far-right popularity on the salience of climate change as an electoral issue and the performance of the Green Party (Buzogány and Scherhaufer, 2018), it also suggests that the salience of the climate change issue in recent years positively influences the Green Party's performance, both on the European (Pearson and Rüdig, 2020) as well as the national level, where the Greens entered a governing coalition as the junior partner of the conservative People's Party (ÖVP) after the snap elections in 2019 (Eberl et al., 2020) (medium confidence).

Representing business as well as agrarian interests, the ÖVP is located at the blocking end of the party spectrum when it comes to ambitious climate action. With an emphasis on the principles of voluntariness and 'technology openness', a construal of the climate crisis as a question of technological innovation and a framing of more binding and far-reaching measures as a threat to the national economy and prosperity, it has repeatedly hindered the implementation of more ambitious climate policy. In recent years, it has done so from a powerful position, leading two governments and dominating key state apparatuses such as the BMF and the Chancellery (Steurer et al., 2022; Auel and Schmidt, 2023; Pearson, 2024) (*medium confidence*). One area of climate policy where the ÖVP has recently taken a more proactive role is Carbon Capture and Storage (CCS) and Carbon Capture Utilization (CCU). In spring 2023, the ÖVP-led BMF initiated the development of a carbon management strategy, aimed at establishing a regulatory and subsidy framework for CCS and CCU in Austria. This integration of CCS and CCU into Austria's climate strategy is in line with the party's inclination towards techno-solutionism. It poses crucial political challenges in terms of addressing the risk that unwarranted expectations of CCS and CCU undermine other mitigation efforts, in terms of limiting CCS to residual emissions that are truly hard-to-abate, and dealing with the persistent risks and uncertainties of these technologies (Brad et al., 2024).

On a programmatic level, the Social Democratic Party (SPÖ) is committed to global or European multilateral agreements and stresses that global responsibility requires local action. It sees climate protection as a challenge that requires a profound restructuring of modes of production and consumption as well as the economy and society at large. However, in a 2019 analysis of party positions on climate change, the alignment of the SPÖ's climate policy with the Paris Agreement was rated as "unclear", partly due to shortcomings in spatial planning (Kirchengast et al., 2019b). An infamous example of these shortcomings is the Lobautunnel, a proposed underground segment of a bypass highway beneath a Viennese nature reserve. The tunnel not only represents the SPÖ's reluctance to abandon emission and resource-intensive pathways like car-centered mobility, but it is also emblematic of how climate-related issues are still highly contentious within the party. While the Viennese party stands firmly behind the project, others in the party, including the federal chair and the party youth, intermittently opposed it. Looking beyond mobility, the federal chair has introduced plans for a climate transformation fund ('Klima-Trafo') in early 2024. This fund is intended to foster and secure sustainable investments, e.g., in renewables or innovative start-ups (Arnecke, 2024).

For the resurgent far-right Austrian Freedom Party (FPÖ), the protection of nature and immediate environment is a salient and fundamental component of ideology and legislative activity. This is nourished by a long-established ethno-nationalist ecological framework of anti-anthropocentrism and organicism (Voss, 2019), and sometimes serves more strategic ends (Tosun and Debus, 2022). It also entails a programmatic embrace of a domestic green energy transition as a way to strengthen national sovereignty (Selk and Kemmerzell, 2022) and sometimes even seems to extend to climate mitigation (Ruser and Machin, 2019; Ćetković and Hagemann, 2020). However, climate policy sceptic positions in the guise of economic pragmatism and even outright denial of anthropogenic climate change are still firmly anchored within the party (Forchtner, 2020; Malm, 2021) (medium evidence, high agreement). This is reflected not only on the programmatic level (Kirchengast et al., 2019a), but also in parliamentary decision making - both at the national and EU level - where the FPÖ has repeatedly opposed mitigation measures (Ćetković and Hagemann, 2020), most infamously in its vote against the Paris Agreement (Austrian Parliament, 2016). While this can be interpreted as an act against multilateralism rather than climate obstruction (Selk and Kemmerzell, 2022), there are also domestic issues where the FPÖ has positioned itself against measures that allegedly harm the so-called 'common man', symbolized, e.g., by the figure of the car driver, who is claimed to suffer from increasing restrictions imposed by climate change policies (Ruser and Machin, 2019). This points to a populist strategy dividing the political space into a self-righteous (urban) and climate-conscious 'elite', imposing their climate agenda on a victimized 'people'. Climate action thus appears as an elitist project directed against the majority in the FPO's discourse (Buzogány and Mohamad-Klotzbach, 2022). This also points to a more general characterization of right-wing parties as unbridled defenders of a resource and emission intensive 'imperial mode of living' (Eversberg, 2018; Brand and Wissen, 2024).

The liberal party Neos is strongly committed to multilateral and transnational efforts to mitigate climate change in which it wants Austria to lead by example. Crucially, Neos claims that effective climate protection can best be achieved within an ecologically modernized market economy that promotes technological innovation. While the party perceives state regulation as an obstacle to economic development in many respects, climate protection measures form a notable exception here. For Neos, the decarbonization of the market economy does also require regulatory action by the state, e.g., in the form of a restructuring of subsidies. Importantly, however, this does not imply an expansion of the current regulatory framework or the state's ability to intervene in the economy. Instead, climate protection measures such as a  $CO_2$  tax should be 'revenue-neutral', i.e., accompanied by deregulation measures that offset the additional costs for Austrian companies. Many of the climate protection measures proposed by the federal government are supported by the Neos. While they joined the Greens in opposing the Lobautunnel, Neos supported the Renewable Energy Act and the Energy Efficiency Act (Arnecke, 2024).

## Organized interest groups

The neo-corporatist character of Austrian politics places specific interest groups in a key position in climate policymaking, by formally establishing links between the state and a limited set of social partner organizations (Tálos and Hinterseer, 2019). While the social partnership is strong when it comes to economic and social issues, it is weak when ambitious climate policies need to be formulated and implemented (Brand and Pawloff, 2014; Niedertscheider et al., 2018). However, there is a range of positions on climate change among the social partners. The Chamber of Labor is the most ambitious social partner trying to pursue ambitious climate policies, despite its occasional advocacy for the expansion of fossil infrastructure to secure employment (Arnecke, 2024) (*high confidence*).

Historically, the relationship between trade unions and environmentalists in Austria has been complicated and conflictive (Brand and Niedermoser, 2019). In the past, trade unions have repeatedly opposed climate protection measures, which they believed would hinder economic development and undermine their bargaining power. In doing so, they have even joined forces with energy companies to secure cheap energy prices (Soder et al., 2018). Despite the complex relationship between trade unions and environmental goals, newly emerging alliances between trade unions and environmental groups have been identified as potential new societal alliances for social-ecological transformation (Soder et al., 2018). It should also be noted, that some 'traditional' trade union policies, e.g., working time reduction, can be considered as 'implicit climate policy', as they are not understood as measures to combat climate change but can contribute to climate-friendly living (Brand and Niedermoser, 2017; Liebig, 2019) (see Section 4.7.3).8 In recent years, the climate crisis has become an increasingly central issue for the Austrian Trade Union Federation (ÖGB), whose 2023 program includes fourteen pages on climate change and a 'just transition'. Here, the ÖGB emphasizes the risks of climate change itself (e.g., working in increased heat) as well as its mitigation (e.g., layoffs due to stricter regulation). Despite their warnings, they also point out the potential long-term synergies between their fundamental goals of improving wages, working conditions and job security on one side and climate protection on the other.

<sup>&</sup>lt;sup>8</sup> The potential positive contribution of a (wage) worktime reduction to enabling climate-friendly living depends on the design and mix of respective policies (Hardt et al., 2020, 2021).

However, it is important to note the sectors and class fractions represented by the ÖGB. While some branches, such as the sectoral union for transport and services (vida), have common interests with climate movements like Fridays for Future (FFF) and in fields like public transport even engage in a joint campaign called 'We ride together', conflicts exist in individual cases such as plans for airport expansion. In other sectors, climate politics is even more contentious. The sectoral union for construction and wood, for example, has also entered into a coalition with FFF, demanding, among other things, subsidies for renovation. In response to the severe crisis in the construction industry, however, the union has since joined the WKO in calling for an economic stimulus package that includes resource-intensive measures such as subsidies for single-family homes (Arnecke, 2024). Meanwhile, in sectors such as the automotive industry, the contradiction between the immediate interests of workers and environmentalists is even more obvious (Pichler et al., 2021a). Where they exist, union's climate strategies tend to remain within the corridor of ecological modernization, as they tend to support climate action only if the primary goals of employment, competitiveness, and economic growth are not compromised (Brand and Niedermoser, 2019) (medium confidence).

The Austrian Economic Chamber (WKO) and the Federation of Austrian Industries (IV) represent different business factions with different interests regarding climate policy. For instance, oil and gas producers and renewable energy producers are both represented by the WKO, yet they have conflicting interests regarding an energy transition. However, WKO and IV have been identified as powerful veto players, blocking the implementation of ambitious climate policy (Abstiens et al., 2021) (high confidence). A concrete example is the development of the 2011 Climate Protection Act, where the WKO played a central role in limiting ambition, especially of GHG emission reduction targets (Brand and Pawloff, 2014; Nash and Steurer, 2021). This obstructive role is also reflected in the perceptions of the members of the Citizens' Climate Council. Evaluations of the body show that members were disappointed with the IV and the WKO, who were late in submitting their inputs, were the only stakeholders not to attend in person but online, and attended only one round of discussions with the citizens, while all the other stakeholders stayed for two rounds (Buzogány et al., 2022). A more ambivalent role seems to be played by the Austrian Chamber of Agriculture. While sometimes described as 'pro-environment', it sided with the other social partners in slowing down the transition to renewable energies. In support of agribusiness interests, it favored subsidies for biofuels and lobbied against feed-in-tariffs for wind or solar power (Tobin, 2017; Arnecke, 2024).

### Climate organizations and movements

Although Austria has established environmental organizations, policymakers and social partners only partially view them as legitimate partners in policymaking (Hermann et al., 2015). However, in recent years, as in many other countries around the world, the FFF movement has become a key civil society actor in Austrian climate politics, ushering in a new phase in the history of the Austrian environmental and climate movement (Daniel and Deutschmann, 2020; Arnecke, 2024). In Germany, FFF has been identified as supportive of representative liberal democracy, but also critical of incumbent power relations (Buzogány and Scherhaufer, 2022). Although putting pressure on politicians was the main motive for participants taking part in a 2019 FFF demonstration in Austria (Daniel and Deutschmann, 2020), they also showed little trust in the government to act appropriately to counter climate change (Buzogány and Mikecz, 2019). As well as demanding from politicians to make decisions in-line with climate science and international agreements such as the Paris Agreement, FFF Austria has been involved in specific conflicts, opposing fossil infrastructure projects such as the Lobautunnel and the proposed third runway at Vienna Airport, and supporting the expansion of wind farms in the Waldviertel region of Lower Austria (Arnecke, 2024).

The Lobautunnel conflict also led to the founding of the Austrian branch of the Last Generation protest movement, which is best known for its civil disobedience (Arnecke, 2024). As with the German branch of the movement (Rucht, 2023), Last Generation's strategy has been highly conflictual and has led to polarized responses, including repressive measures against climate activists (Simsa, 2024). The movement ceased its activities in 2024.

In addition to demonstrations, NGOs and movement actors have also used climate litigation as a strategy to influence climate policy change (see also Section 6.6). For example, a 2017 court case to prevent the construction of a third runway at Vienna Airport was initiated by individuals together with an NGO, the Anti-Aircraft Noise Society (Setzer and Bangalore, 2017). In 2020, Greenpeace and numerous other plaintiffs filed a human rights lawsuit with the Constitutional Court based on preferential tax conditions for air travel compared to train travel. The case is currently pending before the European Court of Human Rights (Wallner, 2022). A central mainstream actor in recent years has been the 'Klimavolksbegehren'. Using a formal direct-democratic instrument for citizens to bring issues to the attention of parliament, which is obliged to consider and report on the issue, climate activists initiated a popular petition on climate policy in 2020. It was signed by 380,590 Austrian citizens (5.9 % of the population entitled to vote), making it the 20th most-supported public petition in Austrian history and leading to its subsequent consideration in parliament (Wallner, 2022). Of the four demands made by the petition – binding and sector-specific  $CO_2$  budgets, a constitutional amendment guaranteeing climate protection, eco-social tax reform, and a citizens' climate assembly – the last two were taken up (Arnecke, 2024).

The Citizens' Climate Assembly was a representative mini-public initiated in 2021. Although designated independent and impartial, the assembly was associated with the minister for climate action from the Green Party from the outset (Buzogány et al., 2022). A key output of the assembly was a set of policy recommendations that demanded more ambitious climate policy than the status quo across a very comprehensive range of policy areas, although these have not been discussed in parliament and the main feedback loop in the political process has been a detailed response from the relevant ministries (Clar et al., 2023). The assembly was praised as a good example of a deliberative forum at the national level (Clar et al., 2023). It was built upon important procedural aspects - specialized facilitation, a scientific advisory board, and possibilities for participative deliberation by participants on both procedural and substantive aspects (Buzogány et al., 2022), all of which were highlighted as beneficial by participants (Praprotnik et al., 2022). The extent to which participants were concerned about climate change did not change over the course of the assembly, which participants put down to being able to direct their concerns in a meaningful direction (Praprotnik et al., 2022). This empowerment continued after the assembly concluded, with former members founding an official organization to continue their work. Criticism of the assembly was that the stakeholder board lacked meaningful engagement (Buzogány et al., 2022). Furthermore, despite the intention to constitute a representative proxy of the Austrian population, the assembly members were found to be more concerned about climate change and more interested in politics than the general population (Praprotnik et al., 2022). In particular, the exclusion of people who had not been vaccinated against COVID-19 from participation excluded a group overrepresented among FPÖ supporters (Arnecke, 2024).

## 6.5. Policy instruments for climate change mitigation and adaptation

This subchapter addresses policy instruments for climate mitigation and adaptation in Austria, offering only a brief overview of specific instruments, with detailed discussions provided in the sector-specific Chapters 2, 3 and 4.

Effective climate policies enable structural change towards climate-friendly living and producing, and increase resilience in the face of unavoidable climate change. This includes transforming unsustainable consumption and production patterns (see Section 6.2). Key levers to achieve this goal are demand-side mitigation policies, and associated instruments focused on increasing sufficiency, i.e., reducing demand for energy, materials and land while ensuring high levels of well-being within planetary boundaries. These policies, also characterized as 'avoid'-instruments, work by preventing demand altogether, distinguishing them from instruments that 'shift' demand to other means or merely 'improve' existing provisioning structures (see Cross-Chapter Box 4).

'Avoid' instruments span across various categories (regulatory, economic, etc.) and sectors. In the transport sector, for instance, teleworking incentives such as subsidies for energy bills, help reduce commuting needs. Some instruments contribute both to avoiding and shifting demand, e.g., banning domestic flights can reduce (air) travel while also shifting traffic from air to rail. Evidence from Austria and other EU countries indicates that such 'avoid' instruments are significantly underrepresented in both current policy mixes and medium- to long-term policy plans (Zell-Ziegler et al., 2021; Novy et al., 2023b; Jarre et al., 2024; Olesen and Vikkelsø, 2024; Brad et al., 2025).

Moreover, the design of policy instruments is subject to a variety of problems, including imperfect information, undefined property rights, multiple externalities, market power, vested interests, unobservable or immeasurable behavior, monitoring problems, uncertainty, lack of stakeholder support, and constraints on available administrative capacity (Bennear and Stavins, 2007). This motivates a second-best policy framework using a combination of regulatory, economic and other instruments to address the identified problems and explains the widespread use of climate policy packages. EU Member States reported 2,052 climate change policies or measures in 2021, of which 44 % were economic instruments and 41 % regulatory instruments (EEA, 2021b). Table 6.2 Summary table on the status of policy instruments in Austria (content based on expert estimations by the Lead Authors of AAR2).

#### Part A: Mitigation

	Agriculture / forestry / other land use	Transport (excluding aviation)	Buildings	Non-ETS industries	ETS industries	Energy supply	Aviation
		National emission tra	ading system			EU ETS	
			Energy taxes				Fuel taxes
	Carbon border adjustments				Carbon borde	er adjustments	
Economic	EU level border-meat-adjustment tax			VAT exemption for PV systems	VAT on inter- national flights		
taxes and ETS	Elimination of reduced VAT or higher taxes rate for meat and dairy products	Road pricing trucks	Vacancy fee		Passenger taxes		
	Higher taxes for emission- intensive products	Road pricing cars / city to <b>ll</b>	Secondary residence fee				
	Fertilizer tax	Vehicle taxes	Real estate tax				
	Soil sealing taxes	Parking fees					

		R&D subsidies							
	Investment subsidies	Subsidies for low emission mobility	Subsidies for change of heating systems	Environmental protection subsidies	Investment subsidies				
	(Restructuring of) subsidies		Subsidies for thermal refurbishment		Feed-in premiums				
Economic	Payments for eco-system services								
instruments:	Premiums for the reduction of livestock								
subsidies	De-sealing premiums								
	Earmarked compensation payments directed to re-naturation								
		Commuting subsidies							
		Subsidies for company cars							

	Reduction of rezoning of green spaces	Spatial planning						
			Energy planning		Energy planning			
Planning instruments		Unsealing / greening	Re-densification	Re-naturation of industrial sites				
		Expansion and electrification of public transport	Adaptation of infrastructure		Expansion & adaptation of grid infrastructure			
		Infrastructure for e-mobility						
				Product passes				

	Non-Financial disclosure regulation									
	Qualification programs									
	Information campaigns									
Other	F	iscal equalization		Fiscal equalization						
matrumenta		Labeling								
	Reform of communal tax									
	Guideline for fertilizer applications									

	Agriculture / forestry / other land use	Transport (excluding aviation)	Buildings	Non-ETS industries	ETS industries	Energy supply	Aviation
	LULUCF regulation	CO2 fleet emission limits / ban of fossil drives	Building codes	Product design standards		Renewable quotas	Ban of short flights
	Feedstuff regulation	Speed limits	Moratoria on expansion of natural gas network	Obligatory energy audits		Regulations for energy communities	Ban of private jets
	Supply chain responsibility regulation	Vehicle access restrictions	Ban of fossil heating systems (new buildings)	Obligatory wast	e heat utilization	Sustainability standards for biofuels	
		Environmental impact assessment		En	vironment impacts assessm	ent	
	Regulation on deforestation-free products	Parking space management	Ban of fossil heating systems (old buildings)		IPPC standards		
	Afforestation obligation	Obligatory shift of longer freight transport to rail	Adaptation of residential property law		CCU/S regulation		
Regulatory	Fertilizer regulations	Obligatory transport on rail for certain goods	Adaptation of heritage protection				
instruments	Restrictions regarding menus in public canteens	Mobility budgets	Adaptation of tenancy law				
	Conservation of biodiversity, nature restoration law	Adaptation of building regulations					
	Regulation on the sustainable use of plant protection products	Moratoria on building of high-ranking roads	Refurbishment of public buildings				
	Maximum livestock-to-land ratio	Right to telework					
	Re-naturation of forests through reduced timber extraction						
	Guidelines for soil protection						
	Advertising bans for meat	Advertising bans for cars					Advertising ban for airline
	Pi	ublic procurement					
	Change in the forestry act		Parking space obligation				

	Coupons for vulnerable consumers	Subsidies for public transport for low- income households	Subsidies for changing heat- ing systems and refurbishment for low-income households	Financial support for affected regions	
		Lump-sum pag	yments ('Klimabonus')	Reductions in corporate taxes	
Just			Energy efficiency programs	fficiency programs Reductions in labor taxes	
transition instruments / compensation			Consumer protection and information	Employment subsidies	
compensation instruments			Preferential loans / energy efficient mortgages	Active labor market policies	
			Curbing monthly back payments of tenants for		

Implemented but insufficient 77%

Implemented and (fairly) sufficient 9% Counterproductive instrument 3%

## Part B: Adaptation

	Agriculture / forestry / other land use	Transport (excluding aviation)	Buildings	Non-ETS industries	ETS industries	Energy supply	Aviation	
	Subsidies for resilient crops		Subsidies for passive cooling					
Economic instruments	Subsidies for water-saving management systems		Subsidies for renovation					
	Subsidies for cover crops							
			Consideration of adaption i					
-				SSUES III EIA				
	Heat protection (labor law)	J		Heat protecti	on (labor law)			
Regulatory			Adaption of tenancy law (Responsibility and Entitlement)					
msnoments			Adaption of residential property law					
			Adaption of heritage protection					
			Spatial planning					
			Heat protecti	on (nassive & active cooling				
-			Elood protection		y, unsealing)			
	Avalanche protection							
Planning instruments			Avaianc	the protection				
		Resilient water	Avalanc Torre nanagement: upgrade of draina	ne protection ent control age infrastructure, stormwa	ter harvesting and re-use			
		Resilient water	Avalanc Torre management: upgrade of draina	he protection ent control age infrastructure, stormwa Supply chain	ter harvesting and re-use			
	Strip cropping ("Begrünungsstreifen / Agroforststreifen")	Resilient water	Cooling and water retention through greening, de-seal- ing, and implementation of "sponge cities"	the protection ent control age infrastructure, stormwa Supply chain	ter harvesting and re-use diversification			
	Strip cropping ("Begrünungsstreifen / Agroforststreifen")	Resilient water	Cooling and water retention through greening, de-seal- ing, and implementation of "sponge cities"	he protection ent control age infrastructure, stormwa Supply chain	ter harvesting and re-use diversification			
	Strip cropping ("Begrünungsstreifen / Agroforststreifen")	Resilient water	Cooling and water retention through greening, de-seal- ing, and implementation of "sponge cities"	he protection ent control age infrastructure, stormwa Supply chain	ter harvesting and re-use diversification			

Non-financial disclosure regulation

Not implemented / missing

Implemented but insufficient 97%

For both mitigation and adaptation, this subchapter focuses on regulatory, planning, economic and other instruments such as information and behavioral instruments. Regulatory instruments are highly sector-specific, ranging from building codes to feedstock regulations to fossil fuel bans. For planning instruments, this section focuses on energy planning. Regarding economic instruments, the focus is on instruments that aim at mitigation and adaptation by steering relative prices.<sup>9</sup> Regarding mitigation, instruments to compensate for potentially negative distributional impacts of mitigation policies are also addressed.

Table 6.2 provides an overview of important climate policy instruments in Austria in different sectors, distinguishing instruments according to their implementation status and their alignment with climate policy objectives. Instruments are further divided into different types (economic, regulatory, planning, just transition and other instruments). Table 6.2 (A) displays mitigation instruments while Table 6.2 (B) presents adaptation instruments.

### 6.5.1. Mitigation instruments

# Regulatory instruments for climate change mitigation

There are numerous examples in policy practice where regulatory instruments have been used effectively, e.g., to increase energy efficiency, control toxic substances, or manage natural resources such as forests or fisheries (Bennear and Stavins, 2007). Furthermore, regulatory instruments such as bans and limits have been identified as key levers to achieve phase-out and exnovation of unsustainable structures and practices (Heyen et al., 2017; Green, 2018; Rosenbloom et al., 2020; Pichler et al., 2021b; Rinscheid et al., 2021). However, regulatory instruments also come with general policy issues (see introduction of Section 6.5). This is illustrated by the mixed track record of the European Union's Corporate Average Fuel Economy (CAFE) targets. Political influence from car manufacturing lobbies (Oki, 2021) has played a role in undermining the standard's effectiveness (for more information on the influence of obstructive actors, see Sections 6.2 and 6.4.2). This interference contributed to missed emissions targets in the mobility sector and ultimately led to a reduction in overall welfare (Reynaert, 2021).

Standardization is crucial for both market and non-market interactions among individuals or groups for two key reasons: (i) without a basic level of standardization, it becomes highly costly for individuals to compare goods and services, making informed purchasing decisions difficult (standards are therefore essential for reasonable price formation); and (ii) even in the absence of formal or centralized regulation (laws, directives, etc.), implicit or decentralized forms of regulation still emerge through social interactions, including habits, customs, and practices (Pindyck and Rubinfeld, 2018).

## Transport

The legal framework for the transport sector in Austria has not yet been adapted to the requirements of sustainable transport in passenger and freight transport (see Section 3.4). There is a lack of regulatory instruments in transport law aimed at avoiding CO<sub>2</sub> emissions. Current measures, such as speed limits and driving bans, are mainly directed at reducing local emissions, e.g., particulate matter, and thus can only contribute indirectly to the reduction of GHG emissions (Schulev-Steindl et al., 2021). In permit procedures for construction of new road infrastructure, GHG emission reduction aspects only play a minor, if not negligible, role (Graber, 2023). There are also few legally binding requirements for public transportation (Fitz, 2023a). Amendments to building codes in some federal provinces have introduced requirements for the installation of EV infrastructure (i.e., charging infrastructure) in new buildings (Cejka, 2024) and have also facilitated the installation of charging infrastructure in apartment buildings (Fidler, 2023). Furthermore, an amendment to road traffic law introduced provisions to promote cycling and walking (Hugeneck, 2023).

Currently, certain waste shipments must be made by rail or by other means of transportation with equal or lower pollutant or GHG potential. However, this obligation does not apply equally to primary raw materials. Therefore, evaluating the instrument in terms of its impact on the recycling

Economic policy goes beyond measures to influence prices and, consequently, shape economic activities. It encompasses a wide range of measures of government interventions to proactively steer structural change. While some types of measures merely alter the framework conditions under which producers and consumers act (i.e., they do not aim at direct intervention), others intend to intervene more directly into economic activities (Peters, 2015). Many recent studies highlight the contribution of economic policy to the transformation to a climate-friendly economy (Blanchard et al., 2022). This includes, for example, trade policy (Wolf, 2021), industrial policy (Rodrik, 2014; Eder and Schneider, 2018; Pichler et al., 2021b), financial and monetary policy (Krogstrup and Oman, 2019; Cahen-Fourot, 2021), (re-)distributional policy (Theine et al., 2022; Kapeller et al., 2023), and labor market and employment policies (Bohnenberger, 2022; Hofbauer et al., 2023).

of secondary raw materials could be an important step in reducing the use of natural resources (Krasznai, 2023).

## Industries and energy supply

GHG emissions from installations are primarily regulated through the economic instrument of emissions trading (see Chapter Box 6.1). Whether Member States can take additional measures in the context of the EU ETS is controversial (Hofer, 2023a). The GHG intensity of installations currently plays little or no role in plant permit procedures (Ennöckl and Sander, 2023; Hofer, 2023a). Regulatory instruments currently focus mainly on energy efficiency and GHG emissions of households and individuals; for industrial emissions and electricity generation, economic instruments are currently favored by policy makers. One exception is the Environmental Impact Assessment (EIA) Act: For example, projects subject to EIA must be designed to limit GHG emissions according to the state of the art or best available techniques. However, for plants included in the EU ETS, no emission limits for direct GHG emissions of GHG subject to emission trading apply (Baumgartner and Niederhuber, 2023; Hofer, 2023b).

The ban on geological storage of  $CO_2$ , which came into force in 2011 and is currently under discussion, does not apply to the sequestration and transport of  $CO_2$  or to research projects, and the storage volumes permitted for research projects are so small that this also hampers the implementation of research projects (Tiefenthaler, 2022).

## Buildings

Various regulations on EU, national and federal provincial level regard energy efficiency, electricity and heating of buildings. For example, the Renewable Heat Act ('Erneuerbare-Wärme-Gesetz') bans the installation of gas heating in new buildings (BGBl. I Nr. 8/2024). The building regulations of the federal provinces integrate mitigation aspects, e.g., by establishing the principle that buildings and all their parts must be planned and constructed in such a way, that the amount of energy required during use is limited according to the state of the art. Most regulations on sustainable and low-emission buildings have been implemented because of EU legislation. More comprehensive, possibly nationwide, regulations are needed to strengthen mitigation efforts (Häusler and Schlenk, 2021) (see Section 3.3.3). Concerning the energy-efficient renovation of buildings protected by heritage law, the competent authority has to consider aspects of ecological sustainability in general and energy efficiency and renewable energy sources in particular when deciding on permits to modify such buildings.

### Agriculture

In the agricultural sector, too, the national regulatory framework is largely determined by EU law (The Common Agricultural Policy, CAP) (Eckhardt, 2022). According to EU law, the Austrian CAP strategy must include agri-environmental and climate commitments as well as voluntary schemes for climate, environment and animal welfare ('eco-schemes'), and is complemented by fertilizer regulations and afforestation obligations.

## Planning instruments for energy and climate change mitigation policies

Spatial planning laws of the federal provinces either directly address climate protection as a spatial planning objective or implicitly include mitigation aspects by referring to the protection of the environment and natural resources in general. In addition, some provinces explicitly refer to the expansion of renewable energies as a primary objective in their spatial planning laws (Stöglehner, 2023a). Nevertheless, land consumption in Austria is high (Pories and Pfeiffer, 2024). This section focuses on energy planning, while spatial planning is addressed in detail in Section 3.2.

## Energy Planning

In addition to lengthy permitting processes, inconsistencies resulting from a lack of coherent nationwide spatial planning for renewable energy infrastructure are barriers to the rapid expansion of renewable energy. Both federal provinces and municipalities are required to designate areas for renewable energy development and infrastructure according to their competencies (Storr, 2024). Efficiency losses thus may result from varying spatial capacity designated for renewable energy projects at different levels. The amended Renewable Energy Directive (Directive (EU) 2023/2413) requires Member States to identify potentials and define areas for accelerated approval of renewable energy projects in order to accelerate the transformation process (Handig and Rathmayer, 2024). Additionally, based on an amendment to the EIA Act and only for wind turbines, in the absence of both local and supra-local spatial energy planning in the federal provinces with regard to the designation of sites for wind turbines, the

consent of the municipality alone is now considered sufficient for the granting of a permit for a wind turbine. This means that the lack of designated areas for wind turbines is no longer a legal obstacle to accelerating the expansion of wind turbines (Altenburger et al., 2024).

The legal framework aims to resolve the tension between the expansion of renewable energy and the protection of biodiversity by, among other things, compensating for biodiversity losses through the designation of specific compensation areas, i.e., areas that must be designated and permanently secured as compensation for biodiversity interventions. However, there is currently no detailed legal framework for compensation areas (Wagner and Ecker, 2013).

Currently, Austrian spatial planning legislation does not consistently and comprehensively address the spatial dimensions of energy consumption and energy supply (Stöglehner, 2023a, 2023b). In 2014, the Austrian Conference on Spatial Planning (ÖROK) developed an expert paper that contains the relevant action requirements for the transition to renewable energies. Here, improving the legal framework for spatial planning has great potential to support the energy transition (see Section 3.2.2).

### Strategic Environmental Assessment

A Strategic Environmental Assessment (SEA), i.e., the assessment of the environmental impacts of a project at the planning stage<sup>10</sup>, must be carried out for major planning projects and programs with significant expected environmental impacts (Alge et al., 2019). This includes significant effects of the plan or program on climatic factors (Stöglehner, 2023b).

Without further specifying the scope and depth of the assessments, the European legal framework for the acceleration of renewable energy (Council Regulation (EU) 2022/2577) strongly emphasizes the role of SEAs in energy planning. According to Article 6 of the EU Acceleration Regulation, the EIA and the assessment of species protection under Article 12 (1) of the Habitats Directive (Council Directive 92/43/ EEC) can be omitted if a project is carried out in an area designated for the expansion of renewable energies, provided that this area has been designated for this purpose after an SEA has been carried out (Handig and Rathmayer, 2024). It is therefore important that the SEA adequately addresses all environmental issues that would otherwise have been considered at project level (e.g.,  $CO_2$  emissions) and that public participation in the process is ensured (Stangl, 2024).

#### Nature Impact Assessment

As biodiversity and climate protection are inseparably linked (see Section 1.6.4), the Nature Impact Assessment (NIA) resulting from Article 6 of the Habitats Directive (Council Directive 92/43/EEC) provides an instrument for balancing interests and defining compensatory measures effectively and uniformly in all Member States when expanding renewable energy (Wagner and Ecker, 2019).

The current EU framework for the expansion of renewable energies provides for exemptions from NIAs for renewable energy installations by affirming the overriding public interest in the expansion of renewable energies and allowing compensation for biodiversity losses, while ensuring that the design of compensation areas is consistent with the overall system of EU habitat and species protection law (Handig and Rathmayer, 2024; Stangl, 2024).

# Economic instruments for climate change mitigation

Economic instruments encompass fiscal and other economic incentives and disincentives to incorporate environmental costs and benefits into household and firm budgets. They include pricing instruments (e.g., carbon taxes and emissions trading), subsidies and fiscal transfers.

Carbon pricing instruments are used specifically to influence the incentive structures of production and consumption activities. They focus on market interactions and the efficient allocation of scarce resources. In the presence of negative externalities, pricing instruments are theoretically the most efficient policy intervention to internalize negative external costs that would otherwise be borne by the public (Metcalf and Weisbach, 2009; Ellerman et al., 2010; Pigou, 2013). Second order implications, such as distributional or leakage effects, require additional instruments. Economic instruments are increasingly used internationally in climate policy and have proven to be effective (Bayer and Aklin, 2020; Gugler et al., 2021; Koch et al., 2022; World Bank, 2024) (medium confidence). However, available evidence shows that the degree of environmental effectiveness varies across sectors and countries, suggesting that a policy mix of pricing instruments and command and control approaches is needed (Haites, 2018; Shmelev and Speck, 2018; Green, 2021; van den Bergh et al., 2021) (high confidence).

<sup>&</sup>lt;sup>10</sup> While the EIA assesses the environmental impact of a specific project (project level), the SEA examines effects of plans and programs on the environment (planning level).

The scope, stringency, coherence and timeframe of the use of economic instruments for climate change mitigation in Austria can be strengthened considerably (COM/2021/555 final; OECD, 2021a; Steininger et al., 2024) (*high confidence*). Also, the menu of readily available and affordable low-carbon substitution possibilities in response to the use of economic instruments is moderate to high as well as steadily increasing due to social and technological innovation.

Implementing economic instruments – in particular carbon pricing in the form of emissions trading or emission taxes – faces trade-offs due to other potentially conflicting policy objectives (e.g., economic competitiveness) and unintended consequences such as negative distributional impacts. Carbon pricing approaches require additional measures to minimize or compensate for these impacts, for which robust strategies exist (e.g., the Austrian 'Klimabonus'). Analyses for Austria (Kirchner et al., 2019; Eisner et al., 2021; Mayer et al., 2021) show the importance of revenue recycling to compensate for the regressivity of carbon taxes (see Section on compensation instruments) (*high confidence*).

Carbon pricing instruments have limits. These can be structural, when flexible response options are not (sufficiently) available (e.g., public transport in rural areas), or informational, when affected actors lack sufficient knowledge about relevant data, such as underestimated total cost of ownership of private cars (Andor et al., 2020), when actors exhibit present bias (Heutel, 2015)<sup>11</sup>, or in the presence of principal-agent problems, e.g., between landlords and tenants (Seebauer, 2021), or due to lack of public trust in carbon-pricing mechanisms and general distrust of policy (Ewald et al., 2022). Thus, the presence of other types of market failures beyond externalities requires the use of further specifically designed instruments formed into policy packages (*high confidence*).

Carbon pricing instruments have the advantage of generating revenues that can be used either to mitigate negative impacts on competitiveness or income distribution, or to subsidize investments and research and development (R&D), etc. Both auctioning revenues from ETS and ESR could be substantial sources of public revenue to support just transition and modernization funds, and transparent earmarking could increase public acceptance (Klenert et al., 2018). Setting price signals is a normative endeavor with significant distributional impacts and prone to creating resistance from political opponents, incumbent firms or (parts of) the public. This may result in decisions to weaken the stringency of the political intervention (Michaelowa, 1998; Segerson et al., 2024).

Although subsidies can provide important incentives and can result in societal benefits in the case of positive externalities (e.g., R&D, nature conservation) compared to carbon-pricing instruments, they represent a second-best solution. While they are politically easier to implement, they also have several shortcomings - they burden public budgets, the additionality of the funded investments is not ensured, they may have negative distributional impacts; by reducing prices, they may lead to increased production and consumption, and there are significant barriers to subsidy removal, due to the vested interests of beneficiary groups (Michaelowa, 1998; Kärnä et al., 2020; Renström et al., 2021; Heyl et al., 2022; Segerson et al., 2024). Reducing environmentally counterproductive support measures is the other side of the coin and is necessary to correct the pricing of externalities, not only to bring prices closer to the true social cost of carbon, but also to avoid conflicting and inefficient use of several policy instruments (OECD, 2005; Skovgaard and van Asselt, 2018; Elgouacem, 2020; Kletzan-Slamanig et al., 2022). This underlines the dual role of economic instruments, which not only correct externalities, but can also preserve or transform (market) structures (Segerson et al., 2024).

Emissions from the non-ETS sectors (especially buildings and transport) are currently addressed by various types of instruments at EU and Member State level. While the building sector is mainly addressed by regulatory instruments (regional building codes; see Section on regulatory instruments for mitigation), economic instruments like subsidies for thermal renovation and the substitution of fossil heating systems also play a role in combination with incentives provided by energy taxation or exogenously caused energy price increases. In 2027, both sectors will be included in the new emission trading system (ETS 2). GHG and air-pollutant emissions of mobile sources (passenger cars, duty vehicles) are addressed with economic instruments, especially pricing approaches (registration and circulation taxes, mineral oil tax, tolls, parking fees). In addition, investment subsidies and tax benefits for electric vehicles have been increasingly provided in recent years. The combination of various economic instruments for transport is justified by affecting not only the regular use but also investment decisions, thus

<sup>&</sup>lt;sup>11</sup> Present bias is the tendency to settle for a smaller present reward rather than wait for a larger future reward, in a trade-off situation.

compensating the present bias of consumers (Heutel, 2015) and also considering other externalities. Additionally, EU directives regulate fleet fuel efficiencies.

However, the rate of change in the building sector to date has been insufficient, and for the Austrian transport sector the use of economic instruments had no statistically significant impact on GHG emissions (Koch et al., 2022), due to a lack of (effective) regulation, rebound effects overcompensating fuel efficiency gains, and the existence of opposing incentives provided by fossil fuel subsidies and other counterproductive measures (e.g., commuter subsidies, company car taxation). The latter is particularly relevant for aviation, where regulation is either insufficient (inclusion of only intra-EEA flights in the EU ETS, mostly free allocation of allowances, too moderate airline ticket levies) or non-existent (general tax exemption for international aviation for fuels and VAT). Furthermore, efficiency gains have been overcompensated by the continuous growth in air traffic.

Industry and energy generation have been included in the EU ETS since its introduction. Emissions from these sectors in Austria have decreased by 7.2 % between 2005 and 2022 (Umweltbundesamt, 2024), indicating a certain effectiveness of the instrument, at least regarding emissions from energy generation. The price incentive from the ETS was complemented by other policies like energy taxation (albeit with tax refunds for energy intensive industries), subsidies for renewables, energy efficiency and R&D. From 2026 onwards, a new instrument on EU level, the CBAM, will apply a carbon price on emission-intensive goods entering the EU in line with the reduction/phase-out of free allowances in the EU ETS.

In the agricultural sector, the only economic instruments currently aimed at reducing GHG emissions are subsidies for those agri-environmental measures within the Austrian rural development program that have been shown to effectively reduce GHG emissions (Freudenschuß et al., 2010; Foldal et al., 2019; BML, 2024; Kraxner et al., 2024) (*medium confidence*), i.e., greening of arable land, reduced tillage and mulching, reduced application of mineral fertilizer, environmentally friendly and biodiversity-promoting management, and organic agriculture. However, the overall impact of all subsidies in the rural development program may lead to a rebound effect, as they can result in increased production output and consequently higher GHG emissions.

## Chapter Box 6.1. Carbon pricing in the EU and Austria

## The EU Emissions Trading System (EU ETS)

The EU implemented the EU ETS as its flagship climate policy instrument in 2005 (Directive 2003/87/EC). Initially the scheme only covered  $CO_2$  emissions from energy supply and emission-intensive industry.<sup>12</sup> However, in 2012, it was expanded to include domestic aviation (Directive 2008/101/EC), and by 2027 maritime shipping will also be integrated into the system (Directive (EU) 2023/959). In 2021, the EU ETS covered approximately 38 % of the EU's GHG emissions. In addition to EU members, Norway, Iceland and Liechtenstein have been participating directly in the EU ETS since 2008 (EEA Joint Committee, 2007), and the Swiss ETS has been linked to it since 2020 (European Commission and Swiss Confederation, 2017). The EU ETS is a cap-and-trade system: First, an emission limit ('cap') is set for the regulated entities, and then emission allowances (certificates) are distributed to them, either through free allocation or auction. These allowances must be surrendered to cover emissions and can be traded between entities, which, according to economic theory, leads to least-cost abatement.

The design of the EU ETS has been adjusted several times. While the definition of the cap and the allocation of emission allowances to sectors and installations was initially the responsibility of the Member States, there has been an EU-wide cap and allocation process since 2013. This cap decreases by an annual linear reduction factor, which has been increased twice to reflect the increasing ambition of the EU's climate targets. Moreover, the allocation process has shifted towards auctioning, particularly for the electricity sector. For this sector, allowances are generally auctioned, while emit-

<sup>&</sup>lt;sup>12</sup> The set of industry activities covered by the EU ETS has increased since 2013. Moreover, in addition to CO<sub>2</sub> emissions for certain activities also other GHGs were included (e.g., nitrous oxides from the production of certain chemicals).

ters in sectors potentially exposed to carbon leakage receive free allocation based on benchmarks. For all other sectors, a mix of free allocation and auctioning applies. In the future, the CBAM will gradually replace free allocation (Regulation (EU) 2023/956). For a long time, the EU ETS was characterized by low allowance prices due to an oversupply of allowances following the economic crisis in 2009. As a result, quantity management provisions (the 'Market Stability Reserve', MSR) have been introduced in the EU ETS to create a stable and continuously increasing price signal (Decision (EU) 2015/1814; Directive (EU) 2023/959). In case of an oversupply of allowances in the market, allowances will be placed in the MSR. If the oversupply does not decrease, these allowances will eventually be canceled. Conversely, in case of a shortage of allowances in the market or if the price increases excessively, allowances are released from the MSR.

## The national emission trading system in Austria ('Nationale CO<sub>2</sub>-Bepreisung')

To reduce energy-related  $CO_2$  emissions not covered by the EU ETS, 17 European countries have so far implemented carbon taxes (World Bank, 2023b). Austria has, however, followed the approach of Germany and introduced a national emission trading system for these sectors in October 2022 (BGBl. I Nr. 10/2022), in the context of the Recovery and Resilience Facility. As in Germany, the carbon price follows a pre-defined pathway increasing from EUR 30/tCO<sub>2</sub> in 2022 to EUR 55/tCO<sub>2</sub> in 2025. However, in Austria the price is automatically adjusted in case of substantial price fluctuations. This adjustment led to a price of EUR 32.5/tCO<sub>2</sub> in 2023 (instead of the initially planned EUR 35/tCO<sub>2</sub>) due to energy price increases caused by the war in Ukraine. In contrast to the downstream approach of the EU ETS, the design of the national ETS follows an upstream approach, which means that the suppliers of fuels will ultimately have to surrender the certificates and pass on the costs to their consumers. As there is no emission cap in place, the national ETS functions like a carbon tax.

For households, revenues from the national ETS are recycled via regionally differentiated lump-sum payments ('Klimabonus'), distinguishing between four types of regions according to the quality of public transport and settlement density.<sup>13</sup> For companies, hardship and carbon leakage provisions apply. Agriculture is currently exempt from the national ETS. In 2027, the national ETS will be replaced by a newly established European emissions trading system ('ETS 2') covering energy-related  $CO_2$  emissions from transport and buildings as well as from energy supply and manufacturing not covered by the EU ETS.

### Other policy instruments for climate mitigation

#### Information instruments

Information-based instruments address the consumption side of mitigation and aim to support consumers in making sustainable choices (Jordan et al., 2013; van den Bergh et al., 2021). They include labels, education, ratings, rankings, and campaigns.

Environmental labeling and information schemes (ELIS) are a set of instruments aiming at 'empowering' consumers by providing information on the environmental impacts of products and services (Gruère, 2013; Klintman, 2016).

ELIS in Austria are largely harmonized with EU and OECD standards and with the International Organization for Standardization (ISO) 14020 series of labeling standards. Many ecolabels have an implicit link to climate mitigation (organic agriculture, wood and fiber sourcing, etc.), but there are few (if any) third-party ecolabels in Austria certifying a low carbon-footprint. Recently, first-party claims of climate neutrality by companies have mushroomed and increasingly face accusations of 'greenwashing' or 'carbonwashing' (In and Schumacher, 2021; Kubat and Tokić, 2024) (*high confidence*).

Beyond organic farming, which is based on an EU regulatory framework, evidence of environmental effective-

<sup>&</sup>lt;sup>13</sup> In 2023, adults received a payment between EUR 110 and EUR 220. In 2022, all adults (independently of their area of residence) received payment of EUR 250 to also compensate for the strong energy price increases. Children receive(d) half the amounts of adults.

ness of ecolabels is scarce (Meis-Harris et al., 2021). Recent NGO criticism blamed eco-certification schemes of being weak, ineffective and in some cases fraudulent (Vogt, 2019; Greenpeace International, 2021). Ecolabels typically generate market niches, but do not contribute to transforming the respective market segment as such. A general criticism of information-based instruments is that they help create sustainable options, but do not eliminate unsustainable ones (Hausknost, 2014; Bärnthaler, 2023).

Government-sponsored mandatory climate score labeling schemes have been proposed as a more effective alternative to voluntary labeling (Lemken et al., 2021). However, this would cross the line to being a regulatory instrument. Overall, evidence suggests that substantial changes in consumption-based emissions and individual behavior will require regulatory frameworks and strong public policies rather than reliance on voluntary labeling (Dubois et al., 2019).

## Behavioral instruments

Behavioral public policy, as understood here, includes all means and modes of public policy aimed at influencing human behavior by using insights from behavioral economics, behavioral sciences, psychology or neurosciences (Straßheim, 2020) (see Section 5.3.1). Nudges are the most prominent type of behavioral public policy. Nudging has been criticized for targeting individual behavior only, and for neglecting the societal determinants of choice (Mock, 2023). Understanding (market) behavior as 'social practice' that is embedded in and constrained by societal structures suggests the need for policy instruments that intervene in these structures (rather than addressing individual behavior) and, as a result, "restrict or eliminate choice" (Mock, 2023, p. 381) where necessary. Examples include the reduction of automobile-centered infrastructure and the sustainable reorganization of interconnected social domains like work, housing, education, and food. This category also includes regulatory experiments, which are "an instrument which deliberately deviates from the existing regulatory framework to try out new rules in a real-world setting" (Bauknecht and Kubeczko, 2024, p. 45). The idea is to create a legal free space in which new ways of regulating technological and social innovations can be tried out. This can take the form of 'regulatory sandboxes' that provide legal exemptions to experiment with new ways of doing things, or of 'regulatory innovation experiments' that aim to adapt an existing regulatory framework to new societal (and technological) challenges. In Austria, an example of the latter is the *Energy.Free.Room* program created by the BMK (Bovera and Lo Schiavo, 2022). For the European energy sector, the EU has published a *state of play of regulatory experimentation in the EU* (Gangale et al., 2023). In Germany, the planned *Real-world Laboratory Act* (BMWK, 2023) aims to provide a legal basis for experimenting with new approaches. Regulatory experimentation could be a crucial instrument for expanding and mainstreaming niche innovations, but also for adapting existing regulatory frameworks more quickly to the challenge of the climate crisis.

## Enabling instruments

Enabling instruments aim to facilitate the transformation of social practices by providing the respective material, organizational and legal infrastructures for new ways of doing things (e.g., recycling schemes, new legal forms of shared ownership or use of products and utilities) (Grubb et al., 2020). Enabling instruments are essential for experimenting with and implementing social innovations and structures for climate-friendly living (APCC, 2023).

# Instruments compensating negative distributional effects of climate policies

The necessity of compensation measures is mostly addressed in the literature regarding the introduction or increase of environmental taxes in general and carbon taxes in particular, which are often shown to be regressive in high-income countries, particularly when electricity and heating are taxed (less regarding fuel taxes) (*high confidence*). Such compensation measures aim to mitigate undesirable distributional effects of climate mitigation measures, which has also been shown to be an important determinant of their public acceptance (see Köppl and Schratzenstaller (2023) for an overview of recent empirical research regarding carbon taxation). Regulatory and planning instruments will also have distributional impacts that may require compensation measures (*limited evidence, high agreement*).

The section is based on empirical findings for Austria regarding the incidence of the burden of climate measures (Kirchner et al., 2019; Eisner et al., 2021; Kettner-Marx et al., 2021; Mayer et al., 2021; Six and Lechinger, 2021; Budgetdienst des Österreichischen Parlaments, 2022) and regarding the effects of potential compensation measures for Austria (Kettner et al., 2024). Simulations of social-ecological tax reform scenarios in Austria confirm, on the one hand, that carbon pricing can contribute to reducing  $CO_2$  emissions, spur renewable energy use and improve energy efficiency (*medium evidence, high agreement*). On the other hand, this literature confirms a trade-off between equity and efficiency with respect to revenue recycling (see Table 6.3) (*medium evidence, high agreement*): Reducing labor costs increases GDP and employment, but does not increase low household incomes or improve equality. The opposite is true for lump-sum payments. One potential compensation measure not considered in Table 6.3 are green vouchers. Their distributional effect is likely similar to that of direct payments, but may reduce potential rebound effects better (Büchs et al., 2021). Currently, no study exists that assesses the impact of using carbon tax revenues for green vouchers for Austria.

In its 2022 social-ecological tax reform, the Austrian government has opted for lump-sum payments ('Klimabonus') to compensate households for the introduction of the national ETS. These per capita climate bonus payments do not differ according to household income but according to the region of residence (with the exception of the first year, 2022, when all adults received equal payments of EUR 250 to also compensate for the strong increase in energy prices due to the war in Ukraine); children receive half of the payments of adults (BGBI. I Nr. 10/2022; BGBI. I Nr. 11/2022).

In addition to studies focusing on carbon pricing, there is a growing literature on compensation measures regarding housing and increasing energy prices in general, with a particular focus on energy poverty. Interestingly, not much research in this area seems to be currently available for the sector of private mobility.

Regarding the housing sector, most research focuses on how to mitigate burdens on energy poor or low-income households (see also Section 5.6.3). While many compensation measures exist in the housing sector such as financial interventions, energy efficiency programs, consumer protection and information (Pye et al., 2017), research highlights the challenge of financing retrofits and/or energy efficiency measures and the impact of these measures on energy use and energy poverty (Berger and Höltl, 2019; Seebauer et al., 2019; Bertoldi et al., 2021; Bourgeois et al., 2021; Seebauer, 2021; George et al., 2023). There is agreement on the need for compensation measures that can alleviate the burden of both upfront costs, financing costs and the tenant/landlord dilemma (medium evidence, high agreement). Studies show that targeting energy poverty may require (i) public funding schemes that operate at the building level (to avoid displacement) and focus on energy-inefficient buildings with households at poverty risk (Berger and Höltl, 2019); (ii) addressing the tenant/landlord dilemma by implementing personalized support, limiting tenant's back payments for renovation measures, higher subsidies for retrofitting measures for buildings with poor households, and a better coordination of climate change initiatives to target energy-poor households (Seebauer et al., 2019; George et al., 2023); and (iii) combining a polluter-pay principle with an ability-to-pay approach for low-income tenants (Seebauer, 2021).

The recent increase in energy prices, especially for natural gas, due to the war in Ukraine has resulted in numerous compensation measures in Austria and other European

SDG-Targets	SDG-Targets No compens tion		ect payment dividend)	Reduced value added tax		e Reduc	ed labor osts	Investments for climate mitigation	Mult comper meas	tiple nsation ures <sup>1</sup>	
13.2 Reduced CO <sub>2</sub> emissions	[6], [8]	[2],	[3], [5], [4], [6]	[4], [5], [6]		[5]	, [6]	[3], [7]	[1], [2],	[1], [2], [3], [8]	
7.2 Renewable energies	[6]		[6]	[6]			[6]	[7]	No d	No data	
7.3 Energy efficiency	No data		No data	No data		No	data	No data	No d	No data	
8.1 Economic growth	[6], [8]	[3]	], [4], [5], [6]	[4]	[5], [6	j] [5]	, [6]	[3], [7]	[1], [2],	[1], [2], [3], [8]	
8.5 Employment	[6], [8]		[3], [5], [6]	[6]	[5]	[5]	, [6]	[6]	[1], [2], [3]	[8]	
10.1 Income of low-income households	[6]	[2]	], [4], [5], [6]	[4]	[5], [6	[5], [6] [5], [6]		No data	[2	2]	
10.4 Increase equality	[6]	[2]	], [4], [5], [6]	[4]	[5], [6	5]	, [6]	No data	[2	2]	
Evaluation scheme:	Strong negative	Medium	Weak	Neut	ral	Weak positive	Medium	n Strong positive			

 Table 6.3
 Effects of carbon pricing with different compensation mechanisms in Austria.

<sup>1</sup> Multiple compensation measures refer to comprehensive climate mitigation packages

Sources: Authors' own visualization based on [1] Goers and Schneider (2019); [2] Großmann et al. (2019); [3] Großmann et al. (2020); [4] Mayer et al. (2021); [5] Kettner et al. (2024); [6] Kirchner et al. (2019); [7] Kratena and Schleicher (1999); [8] Schneider et al. (2010).

countries. For Austria, Kettner et al. (2023) show that most of the implemented compensation measures have negative effects on climate mitigation, especially those that directly reduce energy prices. Only energy efficiency subsidies, investment subsidies for energy independence and support of climate-friendly mobility show synergies between climate protection and mitigating social burdens. A study by Baumgartner et al. (2022) recommends compensation measures that have synergies with climate mitigation, i.e., investment subsidies for building retrofits and heating boiler exchanges, alternative energy supply and energy use (e.g., anergy, geothermal), promotion of public transport and active mobility, town center revitalization, and energy efficiency subsidies. Baumgartner et al. (2022) argue that the Austrian commuter allowance should not be increased to avoid disincentives.

## 6.5.2. Adaptation instruments

# Regulatory instruments for climate change adaptation

Article 5 of the European Climate Law requires the European Commission and the Member States to develop strategies for adaptation to climate change. Member States must adopt national adaptation strategies and implement them (Handig and Stangl, 2023). Every five years, starting in 2023, the European Commission will assess the compatibility of national measures with the integrated national energy and climate plans and the biennial progress reports to be prepared under the Governance Regulation (Regulation (EU) 2018/1999).

The Austrian Climate Protection Act only refers to adaptation issues in § 4 (2): The National Climate Protection Committee shall, among other things, advise on adaptation to the unavoidable consequences of climate change (Ennöckl, 2023b). Otherwise, only a few areas of law explicitly address adaptation to climate change (Bertel, 2023, 2024).

In 2024, the BMK published the third Austrian Strategy for Adaptation to Climate Change (BMK, 2024a). The document provides a framework for adaptation policy in 14 fields of activity: Agriculture, forestry, water balance and water management, tourism, energy (with a focus on the electricity sector), construction and housing, protection against natural hazards, disaster management, health, ecosystems and biodiversity, transport infrastructure including mobility aspects, spatial planning, economy, cities (urban open and green spaces). Recommendations for action to achieve climate change adaptation are given for each area of activity. However, these are hardly reflected at the legal level. Only in exceptional cases does legislation take climate change adaptation into account. Legal research on adaptation issues is also scarce.

One example, relevant for certain projects in the Non-ETS and ETS industries, energy supply and transport sector, is the EIA Act. According to § 6 (1) and § 3 (5), the environmental impact declaration or individual decision of the competent authority must consider a project's vulnerability to the impacts of climate change, especially due to its location (Baumgartner and Niederhuber, 2023). Also, the Water Act obliges the competent authority to include the projected impacts of climate change on the occurrence of floods in its flood risk assessments and flood risk management plans.

Furthermore, some legal provisions at national or federal state level address climate change adaptation in their objectives, e.g., stating that in order to maintain an economically healthy, efficient and comprehensive agricultural and forestry sector in a functional rural area, adaptation to climate change should be taken into consideration according to § 1 (1) Agricultural Act (BGBl. Nr. 375/1992), or that the effects of climate change on electricity supply should be taken into account by appropriate measures as stated in § 1 (3) of the Vienna Electricity Sector Act (LGBl. Nr. 46/2005). § 37 (1) of the Burgenland Spatial Planning Act (LGBl. Nr. 49/2019) further specifies that shopping centers and supermarkets shall be designed to meet the requirements of climate change mitigation and adaptation.

According to the Austrian Adaptation Strategy, measures should be implemented in the transport and energy sectors to strengthen the resilience of infrastructure to the impacts of climate change and to ensure security of supply, e.g., by incorporating adaptation requirements into the relevant regulatory framework. In the building sector, legislative changes are also needed to ensure the consideration and integration of adaptation requirements in the relevant regulatory frameworks. This includes, e.g., amendments to tenancy, heritage and residential property laws to facilitate the transformation of existing buildings.

### Planning instruments for climate change adaptation

Spatial planning law and building regulations as well as disaster control law contain provisions relevant to climate change adaptation, as they provide a legal framework for natural disaster scenarios (e.g., floods, droughts, debris flows), establish warning systems or risk zones (Stöglehner, 2023a, 2023b). However, most legal provisions do not explicitly mention climate change adaptation, nor do they currently emphasize the impacts of climate change.

The Austrian Adaptation Strategy proposes a stronger integration of water management planning and water demand into spatial planning for the purpose of adaptation. Such a proposal has not yet been taken up.

Equally important in terms of this strategy is the planning of ecologically significant open spaces (near-natural spaces, habitat corridors) and the anchoring of cold air corridors in spatial planning.

Introducing a mandatory Nature and Climate Impact Assessment for all legal acts with regard to the adaptation and mitigation goals would contribute to ensuring that climate change adaptation is taken into account in all new laws within the framework of the above-mentioned fields of action (Wagner et al., 2021).

# Economic instruments for climate change adaptation

Economic instruments to promote climate adaptation include market-based instruments (i.e., monetary incentives such as taxes, subsidies or permits), as well as financial and risk-financing instruments. Instruments that have been identified by Bräuninger et al. (2011) as being suitable for climate adaptation include land use taxes to reduce the vulnerability from sealed soil or guarantees to reduce borrowing costs for mitigation measures. Agrawala et al. (2008) focus on insurance, environmental markets and public-private partnerships as viable options to incentivize private action for adaptation. Economic instruments can play an important role and complement and reinforce existing instruments for land use planning and flood protection (Filatova, 2014; Ackerschott et al., 2023). A particular role is played by risk financing instruments, which do not tackle adaptation directly, but pool risks with the aim of redistributing direct losses (e.g., from flooding) and reducing consequential losses such as bankruptcy. A challenge for the use of market-based economic instruments for adaptation is the definition of measurable and tangible adaptation units (such as 'saved wealth' or 'saved health' as proposed by Bräuninger et al. (2011)).

# Other policy instruments for climate change adaptation

In contrast to mitigation, adaptation policy instruments typically do not address individuals as consumers or private decision-makers outside their professional roles. Therefore, the market-related categories of information and behavioral instruments do not apply to adaptation in the same way. In Austria, adaptation measures strongly rely on public action regarding initiation, financing and implementation, and mostly address public authorities, industry actors and practitioners (Knittel and Bednar-Friedl, 2016). At the same time, adaptation measures tend to be voluntary and weakly institutionalized, lacking binding rules of implementation (Clar and Steurer, 2019).

Academic literature increasingly emphasizes that adaptation can no longer be understood as an isolated policy objective, but that the adaptation and transformation of societies are deeply interlinked (Mach and Siders, 2021; Werners et al., 2021; Fekete et al., 2022). Climate change is already beginning to transform the socio-economic structure of Austria, e.g., winter tourism is losing ground as the economic backbone of alpine provinces, and agriculture and forestry are experiencing increasing transformational stress. Adaptation, conceived as an attempt to adjust the status quo to a changing climate, may reach its limits once losses and damages undermine human well-being beyond the local level (Mechler et al., 2020; Schinko et al., 2022). There is a lack of policy instruments reflecting the need for comprehensive and transformative climate risk management in response to a rapidly changing alpine and lowland climate (Schinko et al., 2022) (high confidence).

Researchers have proposed some ideas to strengthen the effectiveness, scope, reach and depth of adaptive measures in Austria and their transformative potential:

- The establishment of a 'National Climate Risk Council' to create missing decision-making structures (Leitner et al., 2020)
- The extension of the Austrian Disaster Fund ('Katastrophenfonds') to include private sector anticipatory adaptation (Leitner et al., 2020)
- The strengthening of participatory adaptation instruments, including participatory budgeting, which has yielded successful adaptation measures in Vienna (Thaler and Seebauer, 2019; Ahn et al., 2023)
- Greater reliance on nature-based solutions (including restoration and protection of ecosystem functions) (Dubo et al., 2022; Schinko et al., 2022; Turner et al., 2022)

## 6.5.3. Climate Policy Integration

Climate policies are more effective when they are part of policies in the respective policy fields (e.g., agriculture, transport, buildings) than when they are separate policies. Climate Policy Integration (CPI) is primarily concerned with the integration of climate change into all policy areas (Adelle and Russel, 2013). Four different categories of analysis have been identified for CPI: (i) normative CPI, (ii) CPI as a process of governing, (iii) CPI as a policy outcome (Jordan and Lenschow, 2010), and (iv) CPI through reflexive learning, given that diverging interests - which need to be acknowledged - and the often contentious nature of politics and the fragmented nature of the polity (i.e., political institutions) require complex processes to enable CPI (Plank et al., 2021) (see Section 6.2). The CPI label, however, risks glossing over this conflictual nature of substantive decarbonization, as has been shown in the literature for Austria and the EU (Kettner and Kletzan-Slamanig, 2018; Niedertscheider et al., 2018; Plank et al., 2021) (high confidence).

Social science literature shows that coherent climate policies face a structurally fragmented apparatus, which formulates and implements policies, as well as competing policy targets and often unsustainable policies. Therefore, climate policy integration is much more a question of political priorities and power than of well-designed policies (see Sections 6.2 and 6.4).

With regard to climate mitigation, Austria's federal structure has been identified as a factor hindering CPI (Steurer and Clar, 2015; Steurer et al., 2020) (see Section 6.4.1), with the challenge of vertical coordination compounding the already difficult challenge of horizontal CPI. Successful CPI requires the institutional prioritization of decarbonization as an overarching policy goal, for example by elevating climate protection in a constitutional framework or introducing climate protection as a human right.

No analyses of CPI in the context of adaptation in Austria are available so far. However, a mapping of adaptation governance for Switzerland (Braunschweiger et al., 2018) shows that while strategic decisions and funding are usually located at the federal level, the actual implementation takes place at the regional or local level, underlining the common division of labor in multi-level governance systems.

Using the national adaptation strategy as a framework, all federal provinces have developed regional adaptation strategies following a dialogue process and the involvement of federal province's representatives in the development of the national strategy. The implementation of adaptation measures, however, is not obligatory or mandated by law. Thus, policy-making for adaptation involves to a large extent informal, cooperation-oriented, 'soft' governance modes and coordination formats, and vertical cooperation or the adherence of stakeholders to non-binding agreements (Probst et al., 2019). Actual implementation seems to be particularly driven by problem pressure, i.e., experienced climate change impacts, rather than governance arrangements (Clar and Steurer, 2019).

## 6.5.4. Integrative perspective on Austria's climate policy instruments

Austria's official emission scenarios (Umweltbundesamt, 2023a, 2023b) show that the country's climate neutrality target for 2040 cannot be achieved through existing and currently planned policy measures alone. This illustrates the need for comprehensive policy mixes and that climate issues must be mainstreamed across all policy areas to enhance effectiveness of climate policy. Achieving this requires attention to the interactions and overlaps among various measures (high confidence) (see Section 6.5.3). With the window to reach the Paris Agreement goals rapidly closing, climate governance and policy must prioritize not only the objectives of (technological) innovation, but also the goals of 'exnovation, that seek to rapidly phase out fossil-based technologies, products and practices (David, 2017; Davidson, 2019). Austrian climate governance to date is mainly oriented on adding sustainable technologies to an unsustainable structure, rather than implementing a targeted strategy of 'creative destruction' (Kivimaa and Kern, 2016) (see Section 6.5.1). This points to a lack of transformative ambition within the current Austrian climate governance, which is further highlighted by the absence of sufficiency instruments that could avoid energy and resource demand by transforming provisioning systems (Zell-Ziegler et al., 2021; Jarre et al., 2024; Brad et al., 2025) (see Section 6.5.1).

The preceding sections assessed low hanging fruits, challenges, and barriers for Austrian climate policy from an interdisciplinary perspective. Achieving targeted climate action calls for balancing various policy objectives, in particular effectiveness, justice, and efficiency.

### 6.6. Climate policy litigation

Climate policy litigation denotes legal remedies aimed at obliging the state and/or corporations to reduce GHG emissions. These lawsuits often aim not only to achieve legal success in court, but also to raise public awareness and initiate political developments to strengthen efforts in climate change mitigation (Fitz, 2023b). So far, in Austria there has been a clear focus on lawsuits against the state, a recent example being the case of twelve children and adolescents who invoked their constitutional children's rights.

# 6.6.1. Human rights, climate lawsuits and environmental proceedings

The Austrian constitution does not explicitly refer to climate change or climate protection (Ennöckl, 2023a). However, comprehensive environmental protection forms a state objective<sup>14</sup>, and there is widespread agreement that climate protection is covered by this provision. The legal consequences of this state objective are still controversial (Kirchmair and Krempelmeier, 2023), culminating in the much-criticized decision of the Constitutional Court, which qualified the refusal of a permit for a third runway at Vienna's airport on the grounds of climate protection as arbitrary and unlawful (Madner and Schulev-Steindl, 2017; Merli, 2017). Moreover, human rights, as enshrined in the Austrian constitution (e.g., the right to life and the right to respect for private and family life), oblige the state to take sufficient measures to protect humans from the consequences of climate change, as the European Court of Human Rights confirmed in a lawsuit against Switzerland (Ennöckl et al., 2024; Hofer, 2024b; Hollaus, 2024) (high confidence).

To date, the Constitutional Court has not ruled substantively on the existence and scope of positive state obligations regarding climate change. Most lawsuits failed on formal grounds. Moreover, the court ruled that there was no constitutional claim for issuing an ordinance regarding a ban on sales of fossil fuels and heating oil (Gärner et al., 2023). Due to the high formal hurdles for filing a climate-related lawsuit, there is a deficit in legal protection under the Austrian constitution that may violate the European Convention on Human Rights (Hofer, 2024b; Hollaus, 2024) (*high confidence*). A pending case with potentially far-reaching consequences is Müllner vs. Austria before the European Court of Human Rights, in which a person with multiple sclerosis is demanding climate protection on the grounds that their disease symptoms increase with rising temperatures.

The right to participation in environmental proceedings according to the Aarhus Convention, especially for NGOs, has – despite the 'Aarhus-Beteiligungs-Gesetz' from 2018 – not yet been fully implemented in the Austrian legal framework, both at the level of administrative proceedings and in court trials (Handig, 2023). The question arises as to whether and by whom deficits in climate protection can be claimed against the state in general and in individual procedures for the approval of GHG-intensive projects. So far, climate change as a public interest (Üblagger, 2024) has mostly gained importance in administrative and judicial decisions on the expansion of renewable energy, when balancing different public interests affected by a renewable energy project, e.g., the construction of wind turbines.

## 6.6.2. Climate liability

The goal of so-called private climate lawsuits is to hold private CO<sub>2</sub> emitters liable for their climate-damaging behavior. The private plaintiffs base their claims on their legal rights to life, health and property, which are also guaranteed as human rights. Under the current legal system, such actions are admissible as actions for injunctive relief and damages (Wagner, 2023). However, the success of such actions depends on the requirement of causality (Burtscher and Spitzer, 2017). The action for damages also needs an allegation of unlawful conduct on the part of the CO<sub>2</sub> emitter (Perner and Spitzer, 2021). To facilitate litigation by private injured parties, the previous obstacles to actions for damages and injunctions against climate-damaging conduct need to be overcome by presuming causality, allocating responsibility among polluters in the case of proportional liability, and establishing a climate-related right of action for associations (Wagner, 2018, 2019, 2023; Burtscher and Schindl, 2022; Wallner, 2023; Rabl, 2024).

## 6.7. Budget, green public finances, investments and cost of (in)action

## 6.7.1. Investments for climate mitigation and adaptation

So far, only few studies have addressed the necessary (additional) investment costs for the transformation towards a carbon-neutral society (Gupta et al., 2014; Kreibiehl et al., 2022), and most of them focused exclusively on energy investment requirements (McCollum et al., 2018). Nevertheless, there is agreement that large investments (i.e., capital expenditures – CAPEX) are required to meet the 2030, 2040 and 2050 climate mitigation goals (*limited evidence, high agreement*). Notably, while this will require a substan-

<sup>&</sup>lt;sup>14</sup> A state objective guides and binds the legislative and the executive, but it does not establish individual rights.

tial increase in additional investments, a large share will come from the reallocation of investments, i.e., away from fossil fuels towards renewables (Gupta et al., 2014; Riahi et al., 2022). Investments for climate mitigation are needed in all economic sectors and will need to target both infrastructure (e.g., the electricity grid, district heating, buildings, railways, bicycle lanes, and industrial installations) and technologies (e.g., renewable energy sources such as wind and photovoltaics, heat pumps, electric vehicles, and insulation), as well as R&D and education (Gupta et al., 2014; Kreibiehl et al., 2022). One of the most recent studies on additional investments for decarbonization in Europe, published by Institut Rousseau (2024), estimates that additional annual investments amount to about 2.3 % of European annual GDP until 2050; one important provision is that current fossil investments must be shifted to decarbonization.

Based on the Transition 2040 scenario, sectoral investment costs for climate neutrality and additional cost compared to already required legislative measures and replacement investments were estimated. The results reflect the high investments required in the energy, building and transport infrastructure, from which climate measures cannot be separated. However, the additional investments for climate neutrality are much lower than the total ones (Krutzler et al., 2023; Umweltbundesamt, 2023c). The estimated costs up to 2030 to achieve the Fit for 55 goals for Austria are reported in the National Climate and Energy Plan (NECP) (BMK, 2024b). The estimated costs up to 2040 and 2050 for the WAM NECP are reported in Krutzler et al. (2024).

For Austria, based on the Transition 2040 scenario, there is a recent estimate of the costs of achieving carbon neutrality (Weyerstraß et al., 2024). Additional costs (compared to a baseline scenario without additional measures, i.e., only with existing measures and policies) may amount to EUR<sub>2023</sub> 6.2-10.9 billion/yr, for the period 2024 to 2040 (see Figure 6.1). This corresponds to about 1.3-2.3 %/yr of the Austrian GDP of 2023 (Weyerstraß et al., 2024). A previous study estimated the total investment potential and gap from 2022 until 2030 on the path to climate neutrality at EUR 169.7 billion (Miess et al., 2022). This corresponds to additional annual investments until 2030 of about 3.6-4.8 % of the Austrian GDP, inducing annual value-added effects between 2-2.7 % of GDP as well as creating and securing between 60,000 and 80,000 jobs annually. This estimate, based on transformation assumptions for all energy sectors, considers the largest investments until 2030 in the energy, building and mobility sectors. Industry enters the transformation process with comparably low investments, which means that most investments in the industry sector will follow after 2030 and towards 2040 and 2050 (Miess et al., 2022). For all sectors and the energy system it is important that climate-neutral and renewable energy becomes cheaper than fossil energy (robust evidence, medium agreement).



Figure 6.1 Expected investment costs for climate neutrality 2040 per sector (Source: Umweltbundesamt, 2023, in Weyerstraß et al., 2024).

A substantial share of additional investments will be made by the governments at different levels, or by public enterprises. Bröthaler et al. (2023) find that, until 2030, the Austrian government can make direct green investments between approximately  $EUR_{2023}$  80 billion (conversion scenario) and  $EUR_{2023}$  138 billion (expansion scenario – in line with the Transition 2040 scenario, especially for the energy sector) through specific decarbonization of the public capital stock and thus directly contribute to achieving climate neutrality in Austria. Compared to a baseline scenario, the government, including public enterprises, could directly make additional annual investments of between  $EUR_{2023}$  5.4 billion (1 % of GDP) and approximately  $EUR_{2023}$  12.6 billion (2.4 % of GDP) to contribute to a carbon-neutral Austrian economy and infrastructure.

A recent study by Schützenhofer et al. (2024) finds that the transformation of Austrian industries (iron and steel, mining, quarrying and glass, chemical and petrochemical products, paper and printing, wood products, food products, construction, vehicle construction) to meet Austria's climate targets in 2040 requires a total investment between  $EUR_{2023}$  17.4 and 24.4 billion for the period 2025–2040. Investment volumes depend on the assumed scenarios. A focus on circular economy and thus secondary production requires the lowest investment volumes, while a focus on innovation, e.g., best-available and break-through technologies, requires the highest investment volumes.

Investments for climate adaptation will also be required to minimize damages from future climate change impacts that we are already committed to. For example, infrastructure will need to be made more resilient to extreme weather events (e.g., adjusting flood level protection levels) and suitable technologies implemented or developed to cope with ongoing climatic change (e.g., irrigation systems to better deal with drought spells, sowing of crop varieties and trees that are better suited to future climatic conditions). Finally, divesting from stranded assets (e.g., crude and refined oil or natural gas production) is a challenging but necessary requirement for achieving climate mitigations goals.

The macroeconomic effects of green investment are also of interest (see the emerging literature on green multipliers; for an overview see Köppl and Schratzenstaller (2022)).

## 6.7.2. Cost of (in)action

Climate inaction entails costs and future financial risks for private actors and public budgets as well as for the economy as a whole. The total costs of climate inaction already incurred today as well as future (budgetary) risks cannot be precisely and comprehensively quantified for Austria due to missing data (particularly for the subnational levels), uncertainties regarding future international developments (e.g., regarding climate-induced international migration; see Section 6.8.2) and the further progress of global warming, and because of measurement and valuation issues, particularly related to non-economic aspects (e.g., with regard to the recreational value of forests, biodiversity, or quality of life) (Parrado et al., 2021; Schiman-Vukan, 2022; Zenios, 2022; Klusak et al., 2023). Although data and estimates are incomplete and sparse, available analyses suggest that costs of climate inaction for Austria are already substantial today and are expected to increase in the coming decades (Steininger et al., 2020; Köppl and Schratzenstaller, 2024a) (limited evidence, high agreement). Obviously, only a part of these costs and future (budgetary) risks can be mitigated through policy measures by the Austrian government.

Climate change related damages, caused by extreme weather events and climate conditions as well as negative consequences of climate change for energy supply and public health, are estimated at currently EUR<sub>2019</sub> 2 billion on average per year and are expected to increase to at least EUR<sub>2019</sub>2.5-5.2 billion/yr by 2030 and to  $\mathrm{EUR}_{2019}$  4.3–10.8 billion/yr by 2050 (Steininger et al., 2020). The European Environment Agency (EEA) estimates cumulative damages through extreme climate and weather conditions in Austria for the period 1980-2021 at EUR<sub>2019</sub> 12.4 billion/yr. As of today, property insurance risks through weather hazards are estimated at 0.25 % of GDP by the Swiss Re Institute (Bannerjee et al., 2024). According to simulations by Bachner and Bednar-Friedl (2019), under a medium scenario of 2°C global warming by 2100, Austrian GDP will be reduced by 0.2 % annually until 2050. Gugele et al. (2022) and Schiman-Vukan (2022) estimate that climate change will reduce productivity by 0.05 % annually until 2050. Austria's overall - i.e., public and private - contributions to international climate finance reached EUR 398 million on average for 2021 and 2022 (BMK, 2022). Another cost factor for the overall economy are net imports of fossil energy, which amounted to a yearly average of EUR 9 billion in the period 2010–2021 and jumped to EUR 10.7 billion in 2023, illustrating the vulnerability of the Austrian economy due to its high dependency on fossil fuel imports. Moreover, privately as well as publicly owned fossil firms are exposed to substantial financial risks through stranded assets, for which no estimations exist.

Existing estimates point at considerable direct and indirect climate-related costs and future risks for public budgets (Köppl and Schratzenstaller, 2024a) (medium evidence, high agreement). Bachner and Bednar-Friedl (2019) estimate annual climate adaptation expenditures by the federal government of EUR 1.1 billion for 2014-2020, which in a medium scenario of global warming will grow to a EUR 1.68 billion/ yr from 2021-2030 and to EUR 2.338 billion from 2031-2050. Of the cumulative damages related to extreme climate and weather events estimated by the EEA for 1980-2021, EUR 9.4 billion (75.8 %) were not insured, so that it can be assumed that at least part of them would have had to be compensated by the government. In this context, it is highly relevant that the Austrian public sector focuses on ex-post compensation payments for private actors or subnational jurisdictions affected by damages through extreme weather events, instead of providing incentives for more formalized insurance arrangements, which could lead to lower and more predictable ex-post compensation payments for the public purse (Sinabell and Url, 2006; Unterberger et al., 2019). Bachner and Bednar-Friedl (2019) estimate an increase of public disaster management expenditures by 184 % (EUR<sub>2019</sub> 547 million) until 2050. Based on the projection of GHG emissions by Gugele et al. (2022), Austrian emissions will be reduced by 30 % compared to 2005. On the new burden-sharing reduction goal of 48 %, Schiman-Vukan (2022) estimates overall expenditures for emission certificates of EUR<sub>2022</sub> 4.7 billion for the period 2021-2030 and an annual average of 0.2 % of GDP after 2031. Average public contributions to international climate finance reached EUR<sub>2022</sub> 327 million on average for 2021 and 2022 (BMK, 2022). Assuming the average yearly growth rate of these contributions between 2013 and 2020, which amounted to 10.2 %, Austria's public contributions will increase to EUR<sub>2023</sub> 658 million by 2030 (Köppl and Schratzenstaller, 2024b). Climate counterproductive permanent public subsidies are substantial (limited evidence, high agreement). Kletzan-Slamanig et al. (2022) estimate climate counterproductive public subsidies at EUR<sub>2019</sub> 4.1 to 5.7 billion annually on average during the past few years. Klusak et al. (2023) estimate that deficits in climate resilience and vulnerability to climate change will decrease the rating of the Republic of Austria by 1.16 notches by 2100, which would increase interest rates and therefore spending related to public debt by EUR<sub>2023</sub> 204 to 296 million/yr in a medium scenario limiting global warming to 2°C. The estimated climate change induced decrease of GDP by 0.2 %/yr until 2050 entails a yearly reduction of tax revenues by 0.3 % and an increase of unemployment payments by EUR<sub>2018</sub> 618 million (10.6 %) according to simulations by Bachner and Bednar-Friedl (2019). Not least, migration into Austria induced by climate change will put additional strain on public budgets; however, estimates for the potential additional public spending are not available (ACA, 2021).

Part of these (future) costs of climate inaction and climate change can be influenced by Austrian policy. The expansion of renewables reduces the dependence on and the cost of fossil fuel imports. By reaching the Austrian climate goals, public expenditures for emission certificates can be avoided; many climate-counterproductive subsidies can be influenced by the Austrian government; and by making the Austrian economy more climate resilient, an increase of interest rates for public loans can be avoided.

## 6.7.3. Greening public finances in Austria

The public sector plays an important role in advancing the green transformation (see Figure 6.2) (*high confidence*). In Austria, a federalist country, the relevant governmental actors are the federal government, the federal provinces, and the municipalities.

At the federal level, various implementation and monitoring mechanisms supporting the greening of public finances have been adopted recently. In 2021, the Action Plan Sustainable Public Procurement of 2010 was updated and reinforced to reorient public procurement, which accounts for about 18 % of GDP in Austria on average in the period 2015-2020 (Klien et al., 2023) and for about 19 million tons (5.6 million tons) of GHG emissions worldwide (in Austria), towards the SDGs. The Action Plan complements and reinforces the Federal Public Procurement Law, which anchors the best bidder principle and thus allows for the explicit application of sustainability-oriented selection criteria. A comprehensive evaluation of the effectiveness of the Action Plan is lacking. For public procurement at the federal level in six crucial areas, Klien and Berger (2023) estimate that the Action Plan will reduce GHG emissions by 2030 by more than 100,000 tCO<sub>2</sub>eq in the first years and 43,000 tCO<sub>2</sub>eq in 2030.

Starting in 2022, following the recommendations of international organizations and the examples of other OECD and EU countries, respectively (Bova, 2021; OECD, 2021b), the BMF launched several initiatives related to green budgeting (BMF, 2023). As one element, green spending reviews, which are part of the reforms included in the National Recovery and Resilience Plan, have been conducted. Federal expenditures with a positive climate impact are identified and quantified within yearly federal budgetary plans, whereby this green spending review focuses on mitigation spending,
thus neglecting expenditures for adaptation. Green as well as environmentally harmful tax expenditures are identified, however, without quantifying their budgetary and environmental impacts. At the subnational levels, green budgeting initiatives play a very limited role. An exception is the City of Vienna, which is currently setting up a climate budget. Moreover, the new Fiscal Equalization Agreement 2024-2028 foresees the implementation of green budgeting pilot projects in two federal provinces. An alignment of green budgeting with performance budgeting at the federal level and existing priority budgeting approaches (gender budgeting in particular) are lacking so far. Other priority budgeting approaches and in particular SDG budgeting do not play any role yet. Moreover, medium- and long-term budget planning and projections, which are generally underused in Austria, largely ignore climate-related aspects.

Environmental taxes (as defined by the OECD/Eurostat) have been increasing over time in Austria, although they remain considerably below the EU average (Köppl et al., 2023a). While tax increases in this area were initially motivated primarily by revenue considerations, environmental objectives have increasingly come to the fore in the last decades. With the recent steps to green the tax system, and particularly the introduction of a CO<sub>2</sub> price in 2022 (Baumgartner et al., 2021), Austria is joining the growing group of countries that have implemented a CO<sub>2</sub> price as an instrument of climate protection, albeit at a very moderate level and based on a rather unambitious trajectory (the CO<sub>2</sub> price is set to increase from EUR 30 in 2022 to EUR 55 in 2025) (Köppl et al., 2021). Recent initiatives to green the Austrian tax system through the implementation of new or the extension of existing green taxes and tax expenditures have focused almost exclusively on climate objectives, while further environmental objectives, in particular reducing resource use (including soil), protecting biodiversity, and supporting the circular economy, play only a minor role. Moreover, the potential of green levies at the subnational levels is not fully utilized (Kletzan-Slamanig and Schratzenstaller, 2022), particularly as instruments to contain land use (Arnold et al., 2023; Bröthaler et al., 2024) (limited evidence, high agreement).

Another revenue category that has recently gained importance is public green securities. The Republic of Austria's Green Framework, adopted in 2021, aims to raise green funds to finance green expenditures in eight categories: (i) Clean transportation, (ii) renewable energy, (iii) energy efficiency, (iv) pollution prevention and control, (v) environmentally sustainable management of living natural resources and land use, (vi) terrestrial and aquatic biodiversity, (vii) sustainable water and wastewater management, (viii) climate change adaptation (OeBFA, 2022). In 2022, Austria issued EUR 5.1 billion in green securities, followed by EUR 5.5 billion in 2023. According to the associated performance and impact assessment, the projects and infrastructure financed through green securities issued in 2023 lead to an annual GHG emissions reduction or avoidance of 4.4 million t (OeBFA, 2024).

A comprehensive greening of intragovernmental fiscal relations is still lacking (Kletzan-Slamanig et al., 2023), and the new Fiscal Equalization Act 2024–2028 implies only limited progress in this respect (Bittschi et al., 2024; Mitterer, 2024) (*limited evidence, high agreement*). Ecological fiscal transfers, which can be found in a growing number of countries worldwide (Busch et al., 2021), are used unsystematically in selected areas only. Additionally, there is a lack of a comprehensive multi-level climate governance framework aimed at coordinating particularly environmentally friendly subsidies and the reform of climate-counterproductive subsidies, green public investments, commitments to meet Austria's climate targets, and implementation and monitoring mechanisms across all levels of government.

Regarding European/international commitments and initiatives, three areas are of particular relevance. First, 59 % of the measures included in the Austrian Resilience and Recovery Plan (RRF), based on which funds from the European Resilience and Recovery Facility are granted, are dedicated to climate objectives, which by far exceeds the minimum share of 37 % and the share of 40 % for the RRF. Second, against the background of the current international discussions and negotiations on the contributions of industrialized countries to international climate finance, Austria's respective contribution is of interest. In 2015 Austria committed itself to make at least EUR 0.5 billion available by 2020 in contributions to international climate finance; with a total of EUR 1.365 billion, the promised amount was exceeded by far (BMF, 2023). Third, the national  $CO_2$  pricing system is to be integrated into the EU ETS, which will be revised and extended by 2027 through a second EU ETS for transport and buildings (see Chapter Box 6.1).

In general, some progress has been made in greening of public budgets, however, substantial gaps remain (*limited evidence, high agreement*). Significant data gaps on the current status of the greening of public finances exist for all governmental levels, with the largest gaps at the subnational levels (Köppl et al., 2023a). Recently, some important steps and initiatives have been taken in several relevant areas, in par-



Figure 6.2 Areas and mechanisms relevant for greening public finances (Source: Modified from Köppl et al., 2023b).

ticular regarding climate expenditure, implementation and monitoring mechanisms, as well as the greening of public procurement, the tax system, the fiscal equalization system and public debt. Progress has been strongest at the federal level, with subnational levels lagging behind. There has been little progress in reforming the considerable volume of environmentally harmful subsidies, which have been significantly increased by temporary support measures to alleviate the social and economic consequences of the recent energy price crisis (Kettner et al., 2023; Kletzan-Slamanig et al., 2023). As Egger et al. (2024) show for the various areas of the welfare state, there is considerable potential for making social infrastructure and provisions more climate-friendly. Moreover, the current perspective on the greening of public finances is still largely input-oriented, while a systematic monitoring and evaluation of budgetary climate mitigation and adaptation measures is missing and existing mechanisms - in particular the performance-informed budgeting at the federal level - are underused (Köppl et al., 2023a). The negative consequences of climate change are already placing a substantial burden on public budgets (see Section 6.7.2). Future budgetary risks resulting from climate change are largely neglected in budgetary projections and medium-term budget planning, although they are estimated to be significant.

#### 6.8. Societal aspects

#### 6.8.1. Just transition

The just transition concept was first proposed in the 1970s by trade union movements to address tensions between social and environmental justice caused by the impacts of a shift to cleaner forms of production on the labor market (Henry et al., 2020; Stevis and Felli, 2020; Bainton et al., 2021). While maintaining the focus on labor markets and paid work as the central arena for securing livelihoods and welfare, the concept has greatly expanded, with NGOs, local authorities, corporations, governments and also scholars promoting multiple and contested interpretations (McCauley and Heffron, 2018; Roberts et al., 2018; Atteridge and Strambo, 2020; Wang and Lo, 2021; Newell et al., 2022; Kyriazi and Miró, 2023) (high confidence). These interpretations range from narrower visions of a socio-technical transition to efforts aimed at building a comprehensive social-ecological framework for justice. Key questions include who deserves justice, the balance between social and ecological justice, the role of the state, and the links between the Global North and South (White, 2019;

Tomassetti, 2020; Kalt, 2021; Pichler et al., 2021b), see also Cross-Chapter Box 7.

This rich foundation is reflected in the IPCC's (Pathak et al., 2022) expansive definition in its recent report, which describes just transition as "a set of principles, processes and practices aimed at ensuring that no people, workers, places, sectors, countries or regions are left behind in the move from a high-carbon to a low-carbon economy". The flexible use of the just transition concept is also visible in Austria, where stakeholders use the term differently, to promote socio-technical investments in SMEs in high-emission regions (ÖROK, 2022), and as a general call for basic public services, for co-determination and for the democratization of work (Initiative Wege aus der Krise, 2019; ÖGB, 2021).

The importance of a just transition is recognized for securing public acceptance, support for climate policy, and overcoming resistance to rapid decarbonization (Meckling et al., 2015; Evensen et al., 2018; Atteridge and Strambo, 2020; Groh and v. Möllendorff, 2020; Bolet et al., 2023) (high confidence). Conversely, high levels of existing inequality also hinder support for green policies (Vona, 2023). Besides their impact on fossil-intensive economic sectors and jobs, policies aimed at reducing CO<sub>2</sub> emissions can have direct and indirect negative impacts on vulnerable groups of people or households; they can also exacerbate existing inequalities or hinder social policy objectives (Gough et al., 2008; Markkanen and Anger-Kraavi, 2019; Lamb et al., 2020b) (high confidence). With such policies potentially having more pronounced and multi-dimensional regressive effects than previously thought, targeted place-based just transition policies for particularly vulnerable communities can be most impactful, especially when sensitive to political acceptability concerns (Vona, 2023; Weller et al., 2024). Just transition policies must consider structural differences (such as varying needs) and vulnerabilities such as low household income, old age and other obstacles to social inclusion (Atteridge and Strambo, 2020; Lamb et al., 2020b; Tikkakoski et al., 2024).

Little peer-reviewed research exists on labor market impacts of just transitions in Austria. Reports and gray literature use the term mainly to quantify sectoral-regional impacts of sustainability transitions. Decarbonization impacts extend beyond the immediate sectors and the overall employment effects are uncertain, with projections of employment impacts in Austria ranging from slightly positive to very negative outcomes. The studies agree on the most affected sectors but reach different conclusions due to methodologies and positions on the socio-technical transition potential. There is a high degree of agreement that the automotive and metal industries, mineral and chemical processing, pharmaceuticals, and the glass and ceramics industries will be strongly affected (Gabelberger et al., 2020; Großmann et al., 2020; Hoffmann and Spash, 2021; Keil, 2021; Meinhart et al., 2022) (*limited evidence, high agreement*). Research also highlights various federal provinces, such as Lower Austria, Carinthia, Upper Austria, and especially Styria, as significant areas with high GHG emissions and a substantial industrial workforce, with a potential regional mismatch of jobs (Gabelberger et al., 2020; Großmann et al., 2020; Hoffmann and Spash, 2021; Keil, 2021; Meinhart et al., 2022).

There is disagreement on the recovery potential of certain sectors (e.g., pulp and paper processing) and on overall labor market effects (Gabelberger et al., 2020; Großmann et al., 2020; Meinhart et al., 2022). The Austrian government's 'Just Transition' report (ÖROK, 2022) emphasizes that medium-term projections and forecasts have largely not considered the significantly increased adjustment pressure on different industries to meet the new target requirements of the Paris Agreement, with low-skilled workers being severely affected, especially support staff in highly affected sectors. Skills mismatches are therefore likely to worsen. Regardless of the overall expected level of employment, there is consensus on the need for retraining, upskilling, and addressing regional disparities (Gabelberger et al., 2020; Großmann et al., 2020; Hoffmann and Spash, 2021; Keil, 2021; Meinhart et al., 2022). Just transition potentially entails broader changes, including restructuring work and production, and providing basic services beyond paid labor (Kalt, 2021; Abram et al., 2022).

Regarding the just transition requirements on the labor market, Austria has prepared a national action plan on training and re-qualification (*Just Transition – Aktionsplan Aus- und Weiterbildung*). This action plan focuses on skill requirements for employees, job entrants and job seekers, defining education and training measures in the energy and heating sector. An Austrian Labor Market Service policy review (Neier et al., 2024) for the Austrian Chamber of Labor underscores Austria's just transition-relevant labor market policies, such as short-term work assistance, (re)training allowance, partial retirement, and educational leave. However, it reveals challenges, including undefined 'green jobs' and skills, insufficient information on environmentally friendly occupations, and lack of tailored programs to promote sustainable occupations, hampering a broader just transition.

A study commissioned by the Ministry of Social Affairs (Seebauer et al., 2021), which screened 300 climate change mitigation measures at the federal and regional level in Austria, highlights important categories of measures that have been implemented so far, comprising levies on energy,  $CO_2$  pricing, the introduction of energy efficiency standards, increased investment in public transport, energy consultancy and a broad range of subsidies and incentives (for, among others, energy efficiency retrofits, the thermal renovation of buildings, solar PV installations, greening of buildings, e-cars and e-mobility).<sup>15</sup>

There is limited empirical evidence on the social impacts of other Austrian climate policies, particularly from ex-post evaluations (Schneider, 2023). More generally it has been noted that "sophisticated empirical studies are lacking as scholars seem more interested in creating various just transition analytical frameworks than in applying these frameworks to empirical investigations" (Wang and Lo, 2021). An Austrian pilot project suggests that personalized consultation can assist households, especially those with low incomes, in managing rising energy expenses (Seebauer et al., 2021). Subsidies and incentives for energy efficiency retrofits, thermal building renovation, solar energy, and similar measures, can make important contributions to reduce GHG emissions (high confidence). Because they require initial investments, however, they tend to benefit wealthier households disproportionally, exacerbating pre-existing inequalities (Lamb et al., 2020b; Seebauer et al., 2021). In cases where thermal insulation and refurbishments lead to higher rents, vulnerable households might be affected negatively. An Austrian case study focusing on low-income, energy-poor households shows that refurbishing rented dwellings can lead to rent increases surpassing energy cost savings, forcing low-income and energy-poor households to leave due to unaffordable rent increases (Berger and Höltl, 2019). Existing inequalities and social exclusion may limit the effectiveness of emissions reduction measures, by curbing adoption rates (Schneider, 2023). International examples show possibilities for designing measures to target low-income households, for example with interest-free loans and progressive subsidies (Lamb et al., 2020b).

At the European level, the European Green Deal includes funding for just transition policies through the Just Transition Fund (JTF) and the Social Climate Fund (SCF). The JTF aids employment, regions and industries during the green transition, supporting fossil-fuel dependent areas. The SCF compensates vulnerable households affected by higher fuel prices from ETS 2 (Kyriazi and Miró, 2023). The JTF allocates EUR 17.5 billion from 2021-2027, with Austria receiving EUR 136 million for Upper Austria, Carinthia, Lower Austria, and Styria, based on the Territorial Just Transition Plan (TJTP) (C/2021/4872; ÖROK, 2022). SCF should provide up to EUR 65 billion (2026-2032) for income support, energy efficient renovations and sustainable transport, with a maximum of EUR 643.5 million going to Austria (C/2021/4872; Wilson et al., 2023). These European schemes can contribute to a just transition. However, scholars have raised concerns about the governance and general setup of the JTF, which might lead to implementation problems, risk the propagation of already present injustice, and create lock-in effects, which reduce incentives for transformation, for example through allowing financing of gas projects (Moesker and Pesch, 2022; Kyriazi and Miró, 2023). The Austrian Chamber of Labor has acknowledged the contribution of the TJTP in Austria's transformation policy, while criticizing its limited scope and governance problems (Soder and Templ, 2022). Details on the measures associated with the SCF are still to be defined, as Member States are required to submit national Social Climate Plans, in which climate action and social compensation measures are set out.

#### 6.8.2. Migration

#### Climate change and migration linkages

Environmental factors influence migration in complex and often non-linear ways. The strength and direction of the relationship is shaped by economic, socio-political, and demographic conditions which moderate the extent to which households are exposed and vulnerable to climate change impacts (Bardsley and Hugo, 2010; McLeman, 2018) (*high confidence*).

Climate change affects migration through multiple channels, including through its impacts on agricultural productivity (Feng et al., 2010; Cai et al., 2016), food and water security, threats to (traditional) livelihoods and opportunities for income generation (Kaczan and Orgill-Meyer, 2020), sociopolitical stability and conflict (Schleussner et al., 2016; Abel et al., 2019), and health (Hunter and Simon, 2019; Schwerdtle et al., 2020). These channels are often closely interlinked, challenging the identification and forecasting of migration impacts (Hoffmann et al., 2020) (*medium confidence*).

<sup>&</sup>lt;sup>15</sup> For evidence on the social impact of carbon pricing, see Section 6.5.1, particularly the subsection on compensation instruments.

In contexts where environmental factors influence migration, they typically have a stronger impact on internal as opposed to international migration (*medium evidence, high agreement*). In many cases, migrants move towards urban centers, contributing to rapid urbanization processes in some regions (Barrios et al., 2006; Henderson et al., 2017; Adger et al., 2020) (*medium confidence*).

Recent literature provides evidence that impacts of climate change together with non-climatic drivers can increase mobility constraints, potentially trapping populations in hazardous places (Black et al., 2013; Nawrotzki and DeWaard, 2018; Zickgraf, 2019). This may result in a vicious cycle of increased exposure, vulnerability, and limited possibilities to escape from the hazards (*medium evidence, high agreement*).

Migration can represent a suitable means for migrants and their communities to adapt to climate change impacts (Black et al., 2011, 2013) (*medium evidence, high agreement*), but it also comes with significant challenges and risks for migrants and their communities. Migration can be maladaptive if it does not lead to long-term benefits and improved conditions (Gemenne and Blocher, 2017; Jacobson et al., 2019; Vinke et al., 2020) (*medium evidence, high agreement*).

#### Climate impacts on migration to and within Europe

There is mixed evidence that environmental factors influence migration to Europe (Mulligan et al., 2014; Hoffmann et al., 2020; Beine and Jeusette, 2021; Šedová et al., 2021) (*limited evidence, medium agreement*). Studies show that temperature fluctuations and increases can lead to increased asylum applications in the European Union (Missirian and Schlenker, 2017) and raise migration to OECD countries (Cai et al., 2016). At the same time, droughts in Africa were found to decrease irregular migration to the European Union, especially from agrarian countries, suggesting more complex relationships (Cottier and Salehyan, 2021).

Limited evidence exists on the role of environmental factors for migration within Europe, which has been considered primarily as a destination region for climate migrants (Obokata et al., 2014; Piguet et al., 2018). Only recently, European Union institutions have started to engage more closely with the topic (Noonan and Rusu, 2022). The lack of empirical evidence on climate-induced migration within Europe constitutes a major gap in the scientific literature with implications for policy and planning processes (*limited evidence, high agreement*).

#### Climate impacts on migration to and within Austria

There is no evidence about the impacts of climate change on migration flows within Austria. Instead, Austria is primarily considered as potential migration destination. A recent study examined the immigration acceptance among students in an Austrian university, showing a similarly high acceptance level for migrants who move because of climate change compared to those who move because of conflict (Henning et al., 2022) (*limited evidence, high agreement*).

There is evidence that environmental degradation and impacts of disasters in countries of origin are being taken into consideration by Austrian courts when making decisions about the granting of subsidiary protection. Country-of-origin information about environmental degradation and disaster impacts provided to courts varies in quantity and quality depending on geographical region (Mayrhofer and Ammer, 2022) (*limited evidence, high agreement*).

Planned relocation (otherwise also referred to as 'managed retreat'), has been undertaken in Austria as a flood management policy, however this is seldom joined up to the migration literature. In this context, planned relocation has been defined as "a directed measure that is initiated, overseen and financed by public authorities at different political level, in which a community of private households moves from a risk to a non-risk location where they resettle permanently" (Thaler et al., 2020, p. 2). In planned relocation that has taken place along the Danube in Upper and Lower Austria, deficiencies have been found in knowledge transfer between national, regional, and local authorities as well as citizen engagement (Thaler et al., 2020), and programs have been organized in a top-down manner. Whether households choose to take part in relocation programs depends on several factors: Rational economic calculations, fear of future flood events, place identity, belief in efficiency of future flood preparedness, and status as a long-term resident. Decisions on whether to relocate are taken from an intergenerational perspective, with children's opportunities being crucial (Seebauer and Winkler, 2020a). However, keeping relocation programs open for a long time has been shown to increase uptake (Thaler et al., 2020). Furthermore, the planned relocations failed to address social inequalities and did not focus enough on vulnerable populations (Thaler, 2021). Households who choose not to participate in relocation problems face a different set of challenges of living in now de-populated areas (Seebauer and Winkler, 2020b) (limited evidence, high agreement).

# Political discourses on climate change and migration in Austria

A great deal of literature points to the dangers of the securitization of the discourse on climate change and migration, which is not likely to lead to increased efforts for climate mitigation as often intended (Bettini et al., 2017) (*medium evidence, high agreement*). Within Austria, the links between climate change and migration are often dismissed as irrelevant until such a point where international migration towards Austria becomes visible. This is partly due to the restrictiveness of Austria's migration politics in general, making civil society actors reluctant to engage in the area of work (Nash, 2023) (*limited evidence, high agreement*). Among parliamentarians at the nation state-level, differing discourses on climate change and migration can be traced back to differing understandings of responsibility and conceptions of national interests (Nash, 2023).

#### 6.8.3. Climate Change Education

Researchers worldwide emphasize and confirm the great potential of Climate Change Education (CCE) in actively promoting a just transition, covering mitigation, adaptation, and transformation education (Anderson, 2012; Lutz et al., 2014; Lee et al., 2015; Otto et al., 2020; Reid et al., 2021; Reimers, 2021). The immense importance of education has also been universally acknowledged by highlighting the special role of SDG 4 'Quality Education' as a key instrument for achieving all SDGs, which is particularly true for Climate Change Education and SDG 13, 'Climate Action'. By identifying CCE as a 'social tipping element', Otto et al. (2020) highlight the prominent role of education, as it opens up the possibility to help activate social-tipping dynamics that can lead to a sufficiently fast reduction in GHG emissions. Not only can education in its various forms (formal, non-formal, livelong, etc.) and facets (e.g., political literacy) (Kranz et al., 2022) support and amplify norms and values, but also inspire behavior change among individuals and lead to social tipping processes, especially when the new generation enters the job market and public decision-making bodies (Otto et al., 2020; Schrot et al., 2021). Hence, by educating key actors in education such as university teachers, teacher educators, in-service teachers, pre-service teachers, environmental educators, and university and school students, they can act as multipliers for climate change knowledge, awareness, and action. As a consequence, they help raise societal acceptance of climate policy measures (e.g., Schelly et al., 2012; Tobler et al., 2012; Andersen, 2018; Stevenson et al., 2018; Brennan, 2019; Kuthe et al., 2019; Thaller and Brudermann, 2020). Beyond that, in order to push climate action right now, it is as important to educate adult multipliers in all societal sectors such as work councils (Arbeiterkammer Wien, 2022), farmers (HBLFA Raumberg-Gumpenstein, 2023), administrative personnel (AdaptBehaviour, 2017), and the general public through lifelong learning approaches (Schrot et al., 2019, 2021).

School and university students and their respective teachers are seen as key 'change agents', defined as strategic multipliers towards climate-friendly competence and action (Skamp et al., 2013; Stevenson et al., 2017; Andersen, 2018; Kuthe et al., 2019; Möller et al., 2021; Winter et al., 2022). Studies show that young people can influence the energy-saving behaviors or climate change awareness of their families and friends in positive ways (Hiramatsu et al., 2014; Andersen, 2018; Brennan, 2019; Parth et al., 2020, 2024; Keller et al., 2022). On the larger scale, the recent FFF movement demonstrates that school students did not only shape the visibility of the topic but also influenced political decisions. It is estimated that within just half a year the movement grew to 1.5 million students in 125 countries, including Austria (Wahlström et al., 2019; Nash and Steurer, 2021). The effects of educational campaigns can be strengthened by participation in FFF (Deisenrieder et al., 2020), but also by a supportive family and community context as well as by media campaigns (Suranovic, 2013; Kubisch et al., 2020). For Austrian students, this was shown to be especially relevant for action-related components of climate change awareness or an enhanced feeling of self-efficacy that might be triggered by perceived collective efficacy (Deisenrieder et al., 2020).

However, CCE, in the framework of Education for Sustainable Development (ESD), is a complex socio-scientific issue that requires the acquisition of 'Climate Literacy', a framework whose key component is an understanding of how humans influence the climate and vice versa (Wise, 2010; Leve et al., 2023). There is growing evidence that teaching and learning CCE in order to reach climate literacy, climate change awareness and climate action, is highly demanding as it has specific challenges: (i) A highly complex and interdisciplinary science with multi-layered interrelationships of effects that complicate understanding, (ii) ethical, political, psychological, and social dimensions, (iii) the constant updating of scientific findings, the still open development of consequences and solutions, and (iv) the resulting public debate, which is controversial and polarized, and can lead to a misinformation about underlying scientific facts (Dawson,

2015; Cook et al., 2018; Rahmstorf and Schellnhuber, 2018; Horn and Bergthaller, 2019; Möller et al., 2021; Winter et al., 2022; Schubatzky and Haagen-Schützenhöfer, 2023). As achieving CCE goals is an extremely complex challenge, close consideration of research results about CCE is vital. These suggest that inter- and transdisciplinary education (and research) settings in combination with moderate constructivist learning settings, like in the Austria-wide CCE research and education program 'makingAchange', promise high learning effects (Rieckmann and Bormann, 2020; Kohl et al., 2022; Bohunovsky and Keller, 2023; Liebhaber et al., 2023; Parth et al., 2024; Rauch et al., 2024; Schickl et al., 2024).

Taking into account the above-mentioned demands and challenges, there is growing evidence that the formal Austrian education system at school and at higher education levels (e.g., university), with its current objectives and structures, does not contribute to a sustainable future and climate-friendly living to the extent necessary, and limits the promotion of action-oriented and participatory CCE (Winter et al., 2022; Bohunovsky and Keller, 2023). In order to achieve a fundamental paradigm shift in Austrian education structures (e.g., whole-institution approaches) and systems towards fostering competences for climate-friendly living and sustainable development, there is high agreement that a transdisciplinary development and implementation of comprehensive CCE concepts at all levels of formal and non-formal (e.g., communities/municipalities, museums, libraries, etc.) education must be promoted and that the contribution of all actors and stakeholders to CCE needs to be fundamentally transformed and intensified strongly and promptly (Martens et al., 2010; Leiringer and Cardellino, 2011; Ryan, 2011; Saltmarsh and Hartley, 2011; Rauch and Steiner, 2013; Sachs et al., 2019; Hübner et al., 2020; Winter et al., 2022; Bohunovsky and Keller, 2023). As a first step, the structural anchoring of climate change contents (physical science, mitigation, adaptation, societal transformation) and CCE concepts in the curricula of all school and university subjects is key, especially in those subjects whose graduates will have a high impact on education (teachers) or public decision-making bodies (legal practitioners, business people) in the future (Otto et al., 2020; Bohunovsky and Keller, 2023). In addition, CCE should be established in the curricula of vocational education and in-service training, especially for teachers (Keller and Rauch, 2021; Bohunovsky and Keller, 2023). Vocational education for sustainable development in Germany (Michaelis and Berding, 2021; Kaiser and Schwarz, 2022; Bohunovsky and Keller, 2023), successful CCE in-service teacher training (Kubisch et al., 2020; Möller et al., 2021) as well as ESD certificate programs for university teachers (Hübner et al., 2020) in Austria can serve as a blueprint.

Systematic research for Austrian school and university curricula is lacking, but in the USA the probability that a student takes at least one climate change class via the core curriculum at the top 100 universities and liberal-arts colleges is estimated at 0.17 % (Hess and Collins, 2018; Hess and Maki, 2019). In a representative study, the vast majority of German in-service teachers indicate that ESD has 'never' (69 %) or 'seldomly' (22 %) been the topic in their teacher training (Grund and Brock, 2018), assuming the same or worse for CCE. Hence it is not surprising that the majority of pre-service teachers as well as in-service teachers lack foundational content knowledge in the field of CCE, often have the same rudimentary or scientifically incorrect understandings of key climate change concepts as their students, and fail to accurately present the scientific consensus on climate change in their lessons (e.g., Papadimitriou, 2004; Boon, 2010; Shepardson et al., 2011; Lambert and Bleicher, 2013; Niebert and Gropengiesser, 2013; Plutzer et al., 2016; Namdar, 2018; Competente, 2019). School textbooks, also in Austria, not only rarely take these misconceptions into account, but often contain scientifically incorrect or outdated content on climate change themselves (e.g., Choi et al., 2010; Kuthe et al., 2019), which can further reinforce erroneous conceptions. Even though some current Austrian school and university curricula suggest taking CCE into account, for example through cross-curricular approaches or in specific subjects such as Biology, Geography or Physics (BGBl. II Nr. 185/2012; BMBF, 2014; Universität Wien, 2016), and even though Austrian teachers report that CCE is very important and that they are willing to teach it, they often do not feel sufficiently supported by their university and in-service education to actualize their transformative potential as change agents (for details, see Winter et al., 2022). They also believe that they will not have enough resources or teaching time to conduct CCE in the future (Winter et al., 2023). In general, it remains unclear whether teacher training programs include not only the promotion of scientific content knowledge about climate change, but also topic-specific pedagogical content knowledge (i.e., didactic knowledge) to address climate change as a socio-scientific issue (Dawson, 2015; Clausen, 2018; Li et al., 2021; Möller et al., 2021). Also, research shows that teachers' political beliefs and values are the greatest predictor of rejection of climate science knowledge, and relevant for the willingness to teach CCE (e.g., Plutzer et al., 2016; Stevenson et al., 2016; Nation

and Feldman, 2021). In terms of CCE teacher professionalization, emphasis should therefore also be placed on affective aspects such as attitudes and values in students and teachers alike (Oberauer et al., 2023).

Currently, inter- and transdisciplinary cooperations, which are pivotal for ESD and especially CCE, are disadvantaged by the Austrian school and university system, which is dominated by discipline-centered structures (Engartner, 2019; Winter et al., 2022). There is evidence that the creation of specific structures for inter- and transdisciplinarity in education is necessary, such as the establishment of appropriate institutes, research centers, professorships ('Brückenprofessuren'), study programs, textbooks, journals, societies, research networks (for an overview, see Bohunovsky and Keller, 2023). To achieve climate-friendly and sustainable living, research literature suggests that CCE should generate system knowledge, target knowledge, and transformation knowledge (Schneidewind and Singer-Brodowski, 2014) and be open to the production of new types of knowledge and competencies (Norström et al., 2020; Chambers et al., 2021).

If the scientific evidence base on the effects of (novel) CCE approaches in the Austrian education system is to be increased, there is high agreement that accompanying research about CCE and evaluation of CCE programs is pivotal, not only in the relevant subjects but also in an inter- and transdisciplinary way (*high confidence*).

### 6.8.4. Media representation of climate change in Austria

In principle, the climate crisis, as a long-term, global, highly complex process can be only partially directly perceived by individual senses. As a result, most people's knowledge of climate change, and the environmental and weather changes it triggers, comes from (news) media (Lörcher, 2019).<sup>16</sup> Media also informs about causes and effects of climate change (e.g., Bolsen and Shapiro, 2018). However, it is also socially constitutive in the sense that communicative negotiation processes discursively construct and frame which social, political and economic adaptation measures are necessary and feasible to prevent or mitigate climate change, as well as which social groups and institutions are responsible (Nisbet, 2009; Reisigl, 2021). In addition, media is a central platform

for various actors to present, interpret and shape debates around climate change and its governance (Tindall et al., 2018).<sup>17</sup> There is only very little research on the media representation of climate change in Austria (Rhomberg, 2016; Theine and Regen, 2023), however, studies from other (German-speaking) countries and comparative cross-country evidence exists.

In Austria, media attention to climate change increased since the mid-2000s both in terms of content (Kathrein, 2015; Narodoslawsky, 2020) and organizationally, as some media outlets have established special beats, pages or special editions on climate topics (Theine and Regen, 2023) (medium confidence). At the same time, long-term analyses show that media attention - despite the increase - is still at a low to medium level (Theine and Regen, 2023, based on; Boykoff et al., 2023, 2022) (limited evidence, high agreement). In addition, the basic trend of increasing media attention is highly dependent on geographic, social and media-specific conditions (Boykoff, 2013; Schmidt et al., 2013). Regarding the media-specific conditions, media representation of climate change is shaped by different media systems, structural determinants of media organizations as well as wider cultural and political factors (Eskjær, 2013; Theine and Regen, 2023). A country's exposure to the impacts of the climate crisis seems to be only a weakly positive determinant of increasing media coverage (Barkemeyer et al., 2017). Media attention is driven by shorter-term extreme weather phenomena, high stakes political events like the annual World Climate Conferences, protests (particularly by FFF) and less so by longterm climatic changes (Pianta and Sisco, 2020; von Zabern and Tulloch, 2021) (limited evidence, high agreement).

Increasing media coverage does not necessarily entail scientifically accurate coverage of the climate crisis and transformative discourses. Internationally, several media-specific processes and structural features have been identified that promote misinformation and false balance. Such processes and structural features are: (i) The ideological stance of the media, (ii) the proximity to far-right political milieus and think tanks (McKnight, 2010b, 2010a; Forchtner et al., 2018), (iii) the unique access of fossil fuel, automotive and other high-emitting industries to influence mainstream media representations via advertisements and source power

<sup>&</sup>lt;sup>16</sup> The subchapter excludes research on social media and climate change for the following reasons: There is very little research on the Austrian case, existing reviews of international literature (e.g., Pearce et al., 2019) are somewhat applicable to Austria due to shared global dynamics in social media use.

<sup>&</sup>lt;sup>17</sup> Media and journalism are in a complex relationship with societal actors and state and governance institutions. Ideally, they fulfill a watchdog-role vis-a-vis other power actors, but they clearly have some impact on forming public knowledge and public opinion, and are typically located outside of official state governance (at least in most democratic countries).

(Bacon and Nash, 2012; Beattie, 2020; Schäfer and Painter, 2021), as well as (iv) the impact of the media crisis through reduced resources for (investigative and specialized) reporting (Gibson, 2016; Schäfer and Painter, 2021).

For Austria, the high degree of tabloidization and the affiliation of some media outlets with the ÖVP, the FPÖ and far-right movements likely favors climate change-skeptical positions and climate delay (Magin and Stark, 2015; Forchtner, 2019) (*limited evidence, high agreement*). On the other hand, evidence from Germany, where high trust in

public service broadcasting parallels high expectations regarding broad and serious coverage of the climate crisis and of extreme weather events, might also hold true for the Austrian context (Neverlaa and Hoppe, 2023; Reif et al., 2024). One discourse analysis of the public broadcaster's ORF2 program 'konkret' suggests that its coverage engages with climate change in the appropriate manner but also suggests that the focus of the reporting is strongly driven by economic and consumer-oriented logics (Sedlaczek, 2017) (*limited evidence, high agreement*).

#### REFERENCES

- Abel, G. J., Brottrager, M., Crespo Cuaresma, J., and Muttarak, R. (2019). Climate, conflict and forced migration. *Global Environmental Change* 54, 239–249. <u>https://doi.org/10.1016/j.gloenvcha.2018.12.003</u>
- Abou-Chadi, T., Jansen, J., Kollberg, M., and Redeker, N. (2024).
  Debunking the Backlash. Uncovering European Voters' Climate Preferences. Berlin: Hertie School; Jacques Delors Centre. Available at: <a href="https://www.delorscentre.eu/fileadmin/2\_Research/1\_About\_our\_research/2\_Research\_centres/6\_Jacques\_Delors\_Centre/Publications/20240307\_Debunking\_the\_Backlash\_Abou-Chadi\_Janssen\_Kollberg\_Redeker.pdf">https://www.delorscentre.eu/fileadmin/2\_Research/1\_About\_our\_research/2\_Research\_centres/6\_Jacques\_Delors\_Centre/Publications/20240307\_Debunking\_the\_Backlash\_Abou-Chadi\_Janssen\_Kollberg\_Redeker.pdf</a> (Accessed May 8, 2024).
- Abram, N., Gattuso, J. P., Prakash, A., Cheng, L., Chidichimo, M. P., Crate, S., et al. (2022). "Framing and Context of the Report", in *IPCC Special Report on the Ocean and Cryosphere in a Changing Climate*, eds. H.-O. Pörtner, D. C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press). Available at: <u>https://ri.conicet.gov.ar/handle/11336/193688</u> (Accessed June 11, 2024).
- Abstiens, K., Gangl, K., Karmasin, S., Kimmich, C., Kirchler, E., Spitzer, F., et al. (2021). Die Klimawandel-Landkarte Österreichs: Treibende Kräfte und nächste Schritte. Vienna: Institute for Advanced Studies. Available at: <u>https://irihs.ihs.ac.at/id/ eprint/5895/</u> (Accessed July 19, 2023).
- ACA (2021). Klimaschutz in Österreich Maßnahmen und Zielerreichung 2020. Vienna: Court of Audit Austria (ACA). Available at: <u>https://www.rechnungshof.gv.at/rh/home/home/ Bund\_2021\_16\_Klimaschutz\_in\_Oesterreich.pdf</u> (Accessed September 15, 2023).
- Acemoglu, D., Aghion, P., Bursztyn, L., and Hemous, D. (2012). The Environment and Directed Technical Change. American Economic Review 102, 131–166. <u>https://doi.org/10.1257/</u> aer.102.1.131
- Ackerschott, A., Kohlhase, E., Vollmer, A., Hörisch, J., and von Wehrden, H. (2023). Steering of land use in the context of sustainable development: A systematic review of economic instruments. *Land Use Policy* 129, 106620. <u>https://doi.org/10.1016/j.</u> landusepol.2023.106620
- AdaptBehaviour (2017). Environmental psychology knowledge for administrative decision makers-improving behaviour change effects of climate adaptation and mitigation measures. Vienna: JOANNEUM RESEARCH Forschungsgesellschaft mbH. Available at: <u>https://projektdatenbank.net/finalreport/4188/</u> (Accessed February 5, 2025).
- Adelle, C., and Russel, D. (2013). Climate Policy Integration: a Case of Déjà Vu? *Environmental Policy and Governance* 23, 1–12. <u>https://doi.org/10.1002/eet.1601</u>
- Adger, W. N., Crépin, A.-S., Folke, C., Ospina, D., Chapin, F. S., Segerson, K., et al. (2020). Urbanization, Migration, and Adaptation to Climate Change. *One Earth* 3, 396–399. <u>https://doi.org/10.1016/j.oneear.2020.09.016</u>
- Agrawala, S., Fankhauser, S., Hanrahan, D., Pope, G., Skees, J., Stephens, C., et al. (2008). Economic and Policy Instruments to Promote Adaptation. Paris: OECD Publishing. <u>https://doi. org/10.1787/9789264046214-5-en</u>
- Ahn, B., Friesenecker, M., Kazepov, Y., and Brandl, J. (2023). How Context Matters: Challenges of Localizing Participatory Bud-

geting for Climate Change Adaptation in Vienna. Urban Planning 8, 399–413. https://doi.org/10.17645/up.v8i1.6067

- Alge, T., Kroiss, F., and Schmidhuber, B. (2019). "20. Strategische Umweltprüfung (SUP)", in Umweltrecht: Handbuch, eds. D. Ennöckl, N. Raschauer, and W. Wessely (Vienna: Facultas). Available at: <u>https://360.lexisnexis.at/d/fachbuch/20\_strategische\_ umweltprufung\_sup/h\_93050\_001300\_93050\_Ennoeckl\_Umweltrec\_2f16b3a44a?origin=cm&podorigin=</u> (Accessed October 17, 2024).
- Altenburger, D., Berger, W., and Nigmatullin, E. (2024). Detailfragen der UVP-G-Novelle 2023. *RdU-UT* 2. Available at: <u>https://</u> <u>rdb.manz.at/document/rdb.tso.LIrduut20240203#doc-B</u> (Accessed October 17, 2024).
- Andersen, P. (2018). Children as intergenerational environmental change agents: using a negotiated Protocol to foster environmentally responsible behaviour in the family home. *Environmental Education Research* 24, 1076–1076. <u>https://doi.org/10.</u> <u>1080/13504622.2017.1392487</u>
- Anderson, A. (2012). Climate Change Education for Mitigation and Adaptation. Journal of Education for Sustainable Development 6, 191–206. <u>https://doi.org/10.1177/0973408212475199</u>
- Andor, M. A., Gerster, A., Peters, J., and Schmidt, C. M. (2020). Social Norms and Energy Conservation Beyond the US. *Journal of Environmental Economics and Management* 103, 102351. <u>https://doi.org/10.1016/j.jeem.2020.102351</u>
- APCC (2023). APCC Special Report: Strukturen für ein klimafreundliches Leben (APCC SR Klimafreundliches Leben)., eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. Berlin, Heidelberg: Springer Spektrum. <u>https://doi.org/10. 1007/978-3-662-66497-1</u>
- Arbeiterkammer Wien (2022). Arbeitswelt im (Klima-)Wandel. Available at: <u>https://wien.arbeiterkammer.at/service/veranstal-tungen/rueckblicke/Arbeitswelt\_im\_Wandel.html</u> (Accessed June 17, 2024).
- Arkolakis, C., and Walsh, C. (2023). Clean Growth. NBER Working Paper No. w31615. Available at: <u>https://papers.ssrn.com/abstract=4553618</u> (Accessed September 16, 2023).
- Arnecke, M. C. (2024). Contentious Policymaking on Climate Change Mitigation in Austria after 2019. An Actor Analysis. Social Ecology Working Papers 200. Available at: <u>https://boku.ac.at/fileadmin/data/H03000/H73000/H73700/Publikationen/ Working\_Papers/WP\_200\_Web\_Arnecke.pdf</u> (Accessed December 20, 2024).
- Arnold, E., Falkner, K., Schratzenstaller, M., and Sinabell, F. (2023). Auswirkungen des Flächenverbrauchs für die Versorgungssicherheit und steuerliche Instrumente zu dessen Eindämmung. Vienna: WIFO. Available at: <u>https://www.wifo.ac.at/publication/72800/</u> (Accessed May 8, 2024).
- Arrighi, G., and Silver, B. J. (2001). Capitalism and world (dis) order. Review of International Studies 27, 257–279. <u>https://doi.org/10.1017/S0260210501008117</u>
- Atteridge, A., and Strambo, C. (2020). Seven principles to realize a just transition to a low-carbon economy. SEI policy report. Stockholm: Environment Institute, Stockholm. Available at: <u>https://www.sei.org/publications/seven-principles-to-realize-a-just-transition-to-a-low-carbon-economy/</u> (Accessed September 15, 2023).
- Auel, K., and Schmidt, P. (2023). "Austria: Weathering the Storm and Greening the Economy?", in *Climate Change and the Future* of Europe: Views from the Capitals, eds. M. Kaeding, J. Pollak,

and P. Schmidt (Cham: Springer International Publishing), 3–6. https://doi.org/10.1007/978-3-031-23328-9\_1

- Austrian Parliament (2016). Nationalrat stimmt Ratifikation des Weltklimavertrags von Paris zu. FPÖ weiter ablehnend, Umsetzung stellt Parlament und Regierung vor große Herausforderungen. Vienna. Available at: <u>https://www.parlament.gv.at/</u> <u>aktuelles/pk/jahr\_2016/pk0840</u> (Accessed September 7, 2023).
- Babić, M., and Dixon, A. D. (2023). Decarbonising states as owners. New Political Economy 28, 608–627. <u>https://doi.org/10. 1080/13563467.2022.2149722</u>
- Bachner, G., and Bednar-Friedl, B. (2019). The Effects of Climate Change Impacts on Public Budgets and Implications of Fiscal Counterbalancing Instruments. *Environ Model Assess* 24, 121– 142. <u>https://doi.org/10.1007/s10666-018-9617-3</u>
- Bäckstrand, K., and Kronsell, A. eds. (2017). Rethinking the Green State: Environmental governance towards climate and sustainability transitions. London: Routledge.
- Bacon, W., and Nash, C. (2012). PLAYING THE MEDIA GAME: The relative (in)visibility of coal industry interests in media reporting of coal as a climate change issue in Australia. *Journalism Studies* 13, 243–258. <u>https://doi.org/10.1080/1461670X.</u> 2011.646401
- Bainton, N., Kemp, D., Lèbre, E., Owen, J. R., and Marston, G. (2021). The energy-extractives nexus and the just transition. *Sustainable Development* 29, 624–634. <u>https://doi.org/10.1002/sd.2163</u>
- Bannerjee, C., Bevere, L., Garbers, H., and Saner, P. (2024). Changing climates: the heat is (still) on. Hazard intensification set to compound economic losses. Zurich: Swiss Re Institute. Available at: <u>https://www.swissre.com/dam/jcr:cdbae8ed-24d0-4ec8ad49-14f16846e556/2024-02-28-sri-expertise-publicationchanging-climates-heat-still-on.pdf</u> (Accessed May 7, 2024).
- Bardsley, D. K., and Hugo, G. J. (2010). Migration and climate change: examining thresholds of change to guide effective adaptation decision-making. *Popul Environ* 32, 238–262. <u>https://doi.org/10.1007/s11111-010-0126-9</u>
- Barkemeyer, R., Figge, F., Hoepner, A., Holt, D., Kraak, J. M., and Yu, P.-S. (2017). Media coverage of climate change: An international comparison. *Environment and Planning C: Politics and Space* 35, 1029–1054. <u>https://doi.org/10.1177/0263774X16680818</u>
- Bärnthaler, R. (2023). Towards eco-social politics: a case study of transformative strategies to overcome forms-of-life crises. *Environmental Politics* 33, 92–113. <u>https://doi.org/10.1080/09644</u> 016.2023.2180910
- Bärnthaler, R., and Gough, I. (2023). Provisioning for sufficiency: envisaging production corridors. Sustainability: Science, Practice and Policy 19, 2218690. <u>https://doi.org/10.1080/15487733.</u> 2023.2218690
- Bärnthaler, R., Novy, A., and Plank, L. (2021). The Foundational Economy as a Cornerstone for a Social–Ecological Transformation. *Sustainability* 13, 10460. <u>https://doi.org/10.3390/ su131810460</u>
- Barrios, S., Bertinelli, L., and Strobl, E. (2006). Climatic change and rural–urban migration: The case of sub-Saharan Africa. *Journal of Urban Economics* 60, 357–371. <u>https://doi.org/10.1016/j. jue.2006.04.005</u>
- Bauknecht, D., and Kubeczko, K. (2024). Regulatory experiments and real-world labs: A fruitful combination for sustainability. *GAIA – Ecological Perspectives for Science and Society* 33, 44–50. <u>https://doi.org/10.14512/gaia.33.S1.7</u>

- Baumgartner, C., and Niederhuber, M. (2023). "Klimaschutz und UVP", in *Klimaschutzrecht*, ed. D. Ennöckl (Verlag Österreich), 267–295. https://doi.org/10.33196/9783704692610-111
- Baumgartner, J., Felbermayr, G., Kettner-Marx, C., Köppl, A., Kletzan-Slamanig, D., Loretz, S., et al. (2022). Stark steigende Energiepreise – Optionen für eine Entlastung von Haushalten und Unternehmen. Vienna: Austrian Institute of Economic Research (WIFO). Available at: <u>https://www.wifo.ac.at/jart/prj3/ wifo/main.jart?content-id=1454619331110&publikation\_id=</u> 69453&detail-view=yes (Accessed June 13, 2023).
- Baumgartner, J., Serguei, K., Fink, M., and Schratzenstaller, M. (2021). Steuerreform 2022/2024 – Gesamtwirtschaftliche Wirkungen. Vienna: Austrian Institute of Economic Research (WIFO). Available at: <u>https://www.wifo.ac.at/jart/prj3/wifo/resources/ person\_dokument/person\_dokument.jart?publikationsid= 69242&mime\_type=application/pdf</u> (Accessed September 13, 2023).
- Bayer, P., and Aklin, M. (2020). The European Union Emissions Trading System reduced CO 2 emissions despite low prices. *Proc. Natl. Acad. Sci. U.S.A.* 117, 8804–8812. <u>https://doi.org/10.1073/pnas.1918128117</u>
- Beattie, G. (2020). Advertising and media capture: The case of climate change. *Journal of Public Economics* 188, 104219. <u>https:// doi.org/10.1016/j.jpubeco.2020.104219</u>
- Beckert, J. (2024). Verkaufte Zukunft: Warum der Kampf gegen den Klimawandel zu scheitern droht. Berlin: Suhrkamp.
- Beine, M., and Jeusette, L. (2021). A meta-analysis of the literature on climate change and migration. *Journal of Demographic Economics* 87, 293–344. <u>https://doi.org/10.1017/dem.2019.22</u>
- Bennear, L. S., and Stavins, R. N. (2007). Second-best theory and the use of multiple policy instruments. *Environ Resource Econ* 37, 111–129. <u>https://doi.org/10.1007/s10640-007-9110-y</u>
- Berger, T., and Höltl, A. (2019). Thermal insulation of rental residential housing: Do energy poor households benefit? A case study in Krems, Austria. *Energy Policy* 127, 341–349. <u>https://doi.org/10.1016/j.enpol.2018.12.018</u>
- Bergthaller, H. (2017). "Fossil freedoms: the politics of emancipation and the end of oil", in *The Routledge Companion to the Environmental Humanities*, eds. U. K. Heise, J. Christensen, and M. Niemann (London; New York: Routledge).
- Bertel, M. (2023). Klimaschutz und Klimawandelanpassung Verfassungsrechtliche Aspekte mit Bezug zur Mobilität in Österreich. ZVR 1a, 42.
- Bertel, M. (2024). "Bodennutzung und Klimawandelanpassung", in Bodenschutz im Zeichen der Klimakrise: Rechtliche Rahmenbedingungen, politische Herausforderungen und Konfliktpotentiale, eds. G. Schnedl, O. C. Ruppel, and M. Hofer (Wien: Verlag Österreich). https://doi.org/10.33196/9783704694454
- Bertoldi, P., Economidou, M., Palermo, V., Boza-Kiss, B., and Todeschi, V. (2021). How to finance energy renovation of residential buildings: Review of current and emerging financing instruments in the EU. WIREs Energy and Environment 10. <u>https:// doi.org/10.1002/wene.384</u>
- Bettini, G., Nash, S. L., and Gioli, G. (2017). One step forward, two steps back? The fading contours of (in)justice in competing discourses on climate migration. *Geogr J* 183, 348–358. <u>https://doi.org/10.1111/geoj.12192</u>
- BGBl. I Nr. 8/2024 (2024). Bundesgesetz über die erneuerbare Wärmebereitstellung in neuen Baulichkeiten (Erneuerbare-Wärme-Gesetz – EWG). Available at: <u>https://www.ris.bka.</u> <u>gv.at/eli/bgbl/I/2024/8</u> (Accessed May 10, 2024).

AAR2

- BGBl. I Nr. 10/2022 (2022). Bundesgesetz über einen nationalen Zertifikatehandel für Treibhausgasemissionen (Nationales Emissionszertifikatehandelsgesetz 2022 – NEHG 2022). Available at: <u>https://www.ris.bka.gv.at/GeltendeFassung.wxe?Abfrage=Bundesnormen&Gesetzesnummer=20011818</u> (Accessed September 14, 2023).
- BGBl. I Nr. 11/2022 (2022). Klimabonusgesetz. Available at: https://www.ris.bka.gv.at/NormDokument.wxe?Abfrage=Bundesnormen&Gesetzesnummer=20011819&FassungVom= 2023-04-12&Artikel=&Paragraf=0&Anlage=&Uebergangsrecht=&ShowPrintPreview=True
- BGBl. I Nr. 72/2014 (2014). Bundes-Energieeffizienzgesetz § 38. Available at: <u>https://www.ris.bka.gv.at/NormDokument.wxe?</u> <u>Abfrage=Bundesnormen&Gesetzesnummer=20008914&FassungVom=2024-03-28&Artikel=&Paragraf=38&Anlage=&Uebergangsrecht=</u> (Accessed October 17, 2024).
- BGBI. I Nr. 111/2013 (2013). Nachhaltigkeit, Tierschutz, umfassender Umweltschutz, Sicherstellung der Wasser- und Lebensmittelversorgung und Forschung. Available at: <u>https://www.ris.</u> <u>bka.gv.at/eli/bgbl/i/2013/111/P4/NOR40153082</u>
- BGBl. II Nr. 185/2012 (2012). Verordnung der Bundesministerin für Unterricht, Kunst und Kultur über die Lehrpläne der Mittelschulen; Bekanntmachung der Lehrpläne für den Religionsunterricht.
- BGBI. Nr. 375/1992 (1992). Bundesgesetz, mit dem Maßnahmen zur Sicherung der Ernährung sowie zur Erhaltung einer flächendeckenden, leistungsfähigen, bäuerlichen Landwirtschaft getroffen werden (Landwirtschaftsgesetz 1992 – LWG). Available at: <u>https://ris.bka.gv.at/GeltendeFassung.wxe?Abfrage=Bundesnormen&Gesetzesnummer=10010681</u> (Accessed October 22, 204AD).
- Biermann, F., Abbott, K., Andresen, S., Bäckstrand, K., Bernstein, S., Betsill, M. M., et al. (2012). Transforming governance and institutions for global sustainability: key insights from the Earth System Governance Project. *Current Opinion in Environmental Sustainability* 4, 51–60. <u>https://doi.org/10.1016/j.cos-</u> ust.2012.01.014
- Binder, C., and Ritter, V. (2023). "Klimaschutz und Völkerrecht", in *Klimaschutzrecht*, ed. D. Ennöckl (Wien: Verlag Österreich), 1–38. <u>https://doi.org/10.33196/9783704692610-101</u>
- Bittschi, B., Famira-Mühlberger, U., Kletzan-Slamanig, D., Klien, M., Pitlik, H., and Schratzenstaller, M. (2024). Finanzausgleich 2024 bis 2028. Erste Schritte zu einer Wirkungsorientierung. WIFO. Available at: <u>https://www.wifo.ac.at/wp-content/uploads/upload-3777/mb\_2024\_01\_04\_finanzausgleich\_.pdf</u> (Accessed May 7, 2024).
- Black, R., Arnell, N. W., Adger, W. N., Thomas, D., and Geddes, A. (2013). Migration, immobility and displacement outcomes following extreme events. *Environmental Science & Policy* 27, S32–S43. https://doi.org/10.1016/j.envsci.2012.09.001
- Black, R., Bennett, S. R. G., Thomas, S. M., and Beddington, J. R. (2011). Migration as adaptation. *Nature* 478, 447–449. <u>https:// doi.org/10.1038/478477a</u>
- Blanchard, O. J., Gollier, C., and Tirole, J. (2022). The Portfolio of Economic Policies Needed to Fight Climate Change. SSRN Journal. <u>https://doi.org/10.2139/ssrn.4276886</u>
- Blühdorn, I. (2020). "Die Gesellschaft der Nicht-Nachhaltigkeit Skizze einer umweltsoziologischen Gegenwartsdiagnose", in Nachhaltige Nicht-Nachhaltigkeit. Warum die ökologische Transformation der Gesellschaft nicht stattfindet, eds. I. Blühdorn, F.

Butzlaff, M. Deflorian, D. Hausknost, and M. Mock (transcript Verlag), 83–160. https://doi.org/10.1515/9783839454428-006

- Blühdorn, I. (2022). Liberation and limitation: Emancipatory politics, socio-ecological transformation and the grammar of the autocratic-authoritarian turn. *European Journal of Social Theory* 25, 26–52. <u>https://doi.org/10.1177/13684310211027088</u>
- BMBF (2014). Grundsatzerlass Umweltbildung für nachhaltige Entwicklung. Vienna: Federal Ministry of Education and Women. Available at: <u>https://rundschreiben.bmbwf.gv.at/download/2014\_20.pdf</u> (Accessed July 15, 2024).
- BMF (2023). Klima- und Umweltschutz. Übersicht gemäß § 42 Abs. 4 BHG 2013. Vienna: Federal Ministry of Finances Austria. Available at: <u>https://www.bmf.gv.at/dam/jcr:445a5d22-7622-438c-b404-be583b8d3efa/Beilage\_Klima-\_und\_Umwelt-schutz\_2023.pdf</u> (Accessed September 15, 2023).
- BMK (2022). Bericht zur internationalen Klimafinanzierung. Berichtjahr 2021. Vienna: Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie. Available at: <u>https://www.bmk.gv.at/dam/jcr:48919182-7815-4169a3cd-40966595121e/Klimafinanzierungsbericht\_2021.pdf</u> (Accessed September 15, 2023).
- BMK (2024a). Austrian Strategy for Adaptation to Climate Change. Vienna: Federal Ministry of Climate Action, Environment, Energy, Mobility, Innovation and Technology. Available at: <u>https:// www.bmk.gv.at/en/topics/climate-environment/climate-protection/austrian-strategy-adaptaion.html</u> (Accessed August 23, 2024).
- BMK (2024b). Integrierter nationaler Energie- und Klimaplan für Österreich – Periode 2021–2030. Vienna: Ministry of Climate Action, Environment, Energy, Mobility, Innovation and Technology (BMK). Available at: <u>https://www.bmk.gv.at/dam/ jcr:6c55ea04-e4b8-499f-ac3b-9d8786147cee/NEKP\_Aktualisierung\_2023\_2024\_final.pdf</u> (Accessed November 4, 2024).
- BML (2024). Austria Rural Development Programme (National) Version 12. Wien: Bundesministerium Land- und Forstwirtschaft, Regionen und Wasserwirtschaft, Wien. Available at: <u>https://info.bml.gv.at/dam/jcr:20449823-0c25-4a55-98bb-82c8f931518b/Programm%20LE14-20\_Version%2012.pdf</u>
- BMWK (2023). Grünbuch Reallabore. Konsultation für ein Reallabore-Gesetz und ergänzende Maßnahmen. Berlin: German Federal Ministry of Finance. Available at: <u>https://www.bmwk.</u> <u>de/Redaktion/DE/Downloads/G/gruenbuch-reallabore.pdf?</u> <u>blob=publicationFile&v=10</u>
- Bodirsky, B. L., Chen, D. M.-C., Weindl, I., Soergel, B., Beier, F., Molina Bacca, E. J., et al. (2022). Integrating degrowth and efficiency perspectives enables an emission-neutral food system by 2100. *Nat Food* 3, 341–348. <u>https://doi.org/10.1038/s43016-022-00500-3</u>
- Bohnenberger, K. (2022). Greening work: labor market policies for the environment. *Empirica* 49, 347–368. <u>https://doi.org/ 10.1007/s10663-021-09530-9</u>
- Bohunovsky, L., and Keller, L. (2023). "Kapitel 21. Bildung und Wissenschaft für ein klimafreundliches Leben", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 567–589. <u>https://doi.org/10.1007/978-3-662-66497-1\_25</u>
- Bolet, D., Green, F., and González-Eguino, M. (2023). How to Get Coal Country to Vote for Climate Policy: The Effect of a "Just Transition Agreement" on Spanish Election Results. American

*Political Science Review* 118, 1–16. <u>https://doi.org/10.1017/</u> S0003055423001235

- Bolsen, T., and Shapiro, M. A. (2018). The US News Media, Polarization on Climate Change, and Pathways to Effective Communication. *Environmental Communication* 12, 149–163. <u>https:// doi.org/10.1080/17524032.2017.1397039</u>
- Boon, H. J. (2010). Climate change? Who knows? A comparison of secondary students and pre-service teachers. *Australian Journal of Teacher Education* 35, 104–120. <u>https://doi.org/10.14221/</u> ajte.2010v35n1.9
- Bourgeois, C., Giraudet, L.-G., and Quirion, P. (2021). Lump-sum vs. energy-efficiency subsidy recycling of carbon tax revenue in the residential sector: A French assessment. *Ecological Economics* 184, 107006. <u>https://doi.org/10.1016/j.ecolecon.2021.107006</u>
- Bova, E. (2021). Green Budgeting Practices in the EU: A First Review. Brussels: European Commission. Available at: <u>https://economy-finance.ec.europa.eu/system/files/2021-05/dp140\_en.pdf</u> (Accessed September 13, 2023).
- Bovera, F., and Lo Schiavo, L. (2022). From energy communities to sector coupling: a taxonomy for regulatory experimentation in the age of the European Green Deal. *Energy Policy* 171, 113299. <u>https://doi.org/10.1016/j.enpol.2022.113299</u>
- Boykoff, M., Daly, M., Reyes, R. F., McAllister, L., McNatt, M., Nacu-Schmidt, A., et al. (2023). World Newspaper Coverage of Climate Change or Global Warming, 2004–2023 – July 2023. <u>https://doi.org/10.25810/4c3b-b819.64</u>
- Boykoff, M., Fernández-Reyes, R., Kjerulf Petersen, L., McAllister, L., Doi, K., Simonsen, A. H., et al. (2022). World Newspaper Coverage of Climate Change or Global Warming, 2004–2022 – October 2022. <u>https://doi.org/10.25810/4C3B-B819.55</u>
- Boykoff, M. T. (2013). Who Speaks for the Climate? Making Sense of Media Reporting on Climate Change., 1. publ., 3. print. Cambridge: Cambridge Univ. Press.
- Brad, A., Schneider, E., Dorninger, C., Haas, W., Wiedenhofer, D., and Gingrich, S. (2025). Existing demand-side climate change mitigation policies neglect avoid options. <u>http://dx.doi.org/10.21203/rs.3.rs-5998199/v1</u>
- Brad, A., Schneider, E., Maneka, D., Hirt, C., and Gingrich, S. (2024). The politics of carbon management in Austria: Emerging fault lines on carbon capture, storage, utilization and removal. *Energy Research & Social Science* 116, 103697. <u>https:// doi.org/10.1016/j.erss.2024.103697</u>
- Brand, U. (2016). How to Get Out of the Multiple Crisis? Contours of a Critical Theory of Social-Ecological Transformation. *Environmental Values* 25, 503–525. <u>https://doi.org/10.3197/0963271</u> <u>16X14703858759017</u>
- Brand, U., and Niedermoser, K. eds. (2017). Gewerkschaften und die Gestaltung einer sozial-ökologischen Gesellschaft. Wien: Verlag des Österreichischen Gewerkschaftsbundes GmbH.
- Brand, U., and Niedermoser, M. K. (2019). The role of trade unions in social-ecological transformation: Overcoming the impasse of the current growth model and the imperial mode of living. *Journal of Cleaner Production* 225, 173–180. <u>https://doi.org/10.1016/j.jclepro.2019.03.284</u>
- Brand, U., and Pawloff, A. (2014). Selectivities at Work: Climate Concerns in the Midst of Corporatist Interests. The Case of Austria. *Journal of Environmental Protection* 5, 780–795. <u>https:// doi.org/10.4236/jep.2014.59080</u>
- Brand, U., and Wissen, M. (2021). The Imperial Mode of Living: Everyday Life and the Ecological Crisis of Capitalism. Brooklyn: Verso Books.

- Brand, U., and Wissen, M. (2024). Kapitalismus am Limit: öko-imperiale Spannungen, umkämpfte Krisenpolitik und solidarische Perspektiven. München: Oekom Verlag.
- Bräuninger, M., Butzengeiger-Geyer, S., Dlugolecki, A., Hochrainer-Stigler, S., Köhler, M., Linnerooth-Bayer, J., et al. (2011). Application of economic instruments for adaptation to climate change. Final report. Hamburg: Perspectives GmbH.
- Braunschweiger, D., Pütz, M., Heidmann, F., and Bludau, M.-J. (2018). Mapping governance of adaptation to climate change in Switzerland. *Regional Studies, Regional Science* 5, 398–401. <u>https://doi.org/10.1080/21681376.2018.1549507</u>
- Brennan, M. (2019). Changing Teaching and Teacher Education in the "Anthropocene." On Education. Journal for Research and Debate 4. <u>https://doi.org/10.17899/on\_ed.2019.4.6</u>
- Bröthaler, J., Dillinger, T., Getzner, M., Kanonier, A., Grinzinger, E., and Chamraci, M. (2024). Klimaorientierte und ressourcenschonende Raumentwicklung und Finanzausgleich. Zur Raumwirksamkeit des Finanzausgleichs unter besonderer Berücksichtigung des ÖREK 2030, des Klimaschutzes und des sparsamen Bodenverbrauchs. Endbericht. Vienna: Vienna University of Technology. Available at: <u>https://repositum.tuwien.at/</u> <u>handle/20.500.12708/190477</u> (Accessed May 7, 2024).
- Bröthaler, J., Getzner, M., Müller, H. L., Plank, L., Miess, M., Niedertscheider, M., et al. (2023). Öffentliche Investitionen für den Klimaschutz in Österreich: Potenziale des öffentlichen Vermögens. Kammer für Arbeiter und Angestellte für Wien. Available at: <a href="https://repositum.tuwien.at/handle/20.500.12708/187853">https://repositum.tuwien.at/handle/20.500.12708/187853</a> (Accessed September 6, 2023).
- Brüggemann, M., Neverla, I., Hoppe, I., and Walter, S. (2018). "Klimawandel in den Medien", in *Hamburger Klimabericht – Wissen* über Klima, Klimawandel und Auswirkungen in Hamburg und Norddeutschland, eds. H. von Storch, I. Meinke, and M. Claußen (Berlin, Heidelberg: Springer Spektrum), 243–254. <u>https://doi.org/10.1007/978-3-662-55379-4\_12</u>
- Brulle, R. J., Roberts, J. T., and Spencer, M. C. eds. (2024). *Climate Obstruction across Europe*. Oxford, New York: Oxford University Press.
- Buch-Hansen, H., and Carstensen, M. B. (2021). Paradigms and the political economy of ecopolitical projects: Green growth and degrowth compared. *Competition & Change* 25, 308–327. <u>https://doi.org/10.1177/1024529420987528</u>
- Büchs, M., Cass, N., Mullen, C., Lucas, K., and Ivanova, D. (2023). Emissions savings from equitable energy demand reduction. *Nat Energy* 8, 758–769. <u>https://doi.org/10.1038/s41560-023-01283-y</u>
- Büchs, M., Ivanova, D., and Schnepf, S. V. (2021). Fairness, effectiveness, and needs satisfaction: new options for designing climate policies. *Environ. Res. Lett.* 16, 124026. <u>https://doi.org/10. 1088/1748-9326/ac2cb1</u>
- Budgetdienst des Österreichischen Parlaments (2022). Verteilungswirkung der drei Maßnahmenpakete zum Teuerungsausgleich. Available at: <u>https://www.parlament.gv.at/fachinfos/</u> <u>budgetdienst/Verteilungswirkung-der-drei-Massnahmenpakete-zum-Teuerungsausgleich</u> (Accessed June 14, 2023).
- Burtscher, B., and Schindl, D. (2022). Klimaklagen: eine Zeitenwende? ÖJZ 13. Available at: <u>https://rdb.manz.at/document/</u> <u>rdb.tso.LIoejz20221302</u> (Accessed October 17, 2024).
- Burtscher, B., and Spitzer, M. (2017). Haftung für Klimaschäden. ÖJZ 21. Available at: <u>https://rdb.manz.at/document/rdb.tso.</u> <u>LIoejz20172102</u> (Accessed October 17, 2024).

- Busch, J., Ring, I., Akullo, M., Amarjargal, O., Borie, M., Cassola, R. S., et al. (2021). A global review of ecological fiscal transfers. *Nat Sustain* 4, 756–765. <u>https://doi.org/10.1038/s41893-021-00728-0</u>
- Buzogány, A., Ehs, T., Plöchl, J., and Scherhaufer, P. (2022). Evaluation Report of the Austrian Climate Citizens' Assembly: Assessment of input, process, and output. Vienna: University of Natural Resources and Life Sciences (BOKU). Available at: <a href="https://boku.ac.at/fileadmin/data/H03000/H73000/H73200/InFER\_Discussion\_Papers/Final\_Report\_Austrian\_Climate-Citizens-Assembly\_BOKU-part.pdf">https://boku.ac.at/fileadmin/data/H03000/H73000/H73200/InFER\_Discussion\_Papers/Final\_Report\_Austrian\_Climate-Citizens-Assembly\_BOKU-part.pdf</a> (Accessed June 16, 2023).
- Buzogány, A., and Mikecz, D. (2019). "Country Report: Austria", in Protest for a Future: Composition, Mobilization and Motives of the Participants in Fridays for Future Climate Protests on 15 March, 2019 in 13 European Cities, eds. M. Wahlström, P. Kocyba, M. de Vydt, and J. De Moor, 90–98. Available at: <u>https:// protestinstitut.eu/wp-content/uploads/2019/06/20190625\_Protest-for-a-future\_GCS-Descriptive-Report\_ipb.pdf</u> (Accessed May 7, 2024).
- Buzogány, A., and Mohamad-Klotzbach, C. (2022). "Environmental Populism", in *The Palgrave Handbook of Populism*, ed. M. Oswald (Cham: Palgrave Macmillan), 321–340. <u>https://doi.org/10.1007/978-3-030-80803-7\_19</u>
- Buzogány, A., and Scherhaufer, P. (2018). Austrian Greens: from pyrrhic presidential victory to parliamentary exit. *Environmen*tal Politics 27, 566–571. <u>https://doi.org/10.1080/09644016.2018.</u> <u>1438793</u>
- Buzogány, A., and Scherhaufer, P. (2022). Framing different energy futures? Comparing Fridays for Future and Extinction Rebellion in Germany. *Futures* 137, 102904. <u>https://doi.org/10.1016/j.futures.2022.102904</u>
- C/2021/4872 (2021). Commission Implementing Decision (EU) 2021/1129 of 5 July 2021 setting out the annual breakdown of available allocations under the Just Transition Fund per Member State (notified under document C(2021) 4872). Available at: <a href="https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=uriserv:OJ.L\_2021.244.01.0004.01.ENG">https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=uriserv:OJ.L\_2021.244.01.0004.01.ENG</a> (Accessed September 13, 2023).
- Cahen-Fourot, L. (2021). Central banking for a social-ecological transformation. *Ecological Economic Papers No.* 40. <u>https://doi.org/10.57938/cda7db4d-0c62-4aad-85a1-0ee7cafca66f</u>
- Cai, R., Feng, S., Oppenheimer, M., and Pytlikova, M. (2016). Climate variability and international migration: The importance of the agricultural linkage. *Journal of Environmental Economics* and Management 79, 135–151. <u>https://doi.org/10.1016/j.jeem.</u> 2016.06.005
- Cappelli, F. (2023). Investigating the origins of differentiated vulnerabilities to climate change through the lenses of the Capability Approach. *Econ Polit* 40, 1051–1074. <u>https://doi.org/10.1007/</u> s40888-023-00300-3
- Cattaneo, C., Kallis, G., Demaria, F., Zografos, C., Sekulova, F., D'Alisa, G., et al. (2022). A degrowth approach to urban mobility options: just, desirable and practical options. *Local Environment* 27, 459–486. <u>https://doi.org/10.1080/13549839.2022.</u> 2025769
- Cejka, S. (2024). Die Energie- und Mobilitätswende in den Bauordnungsnovellen 2023. *Recht der Umwelt (RdU)* 37. Available at: <u>https://rdb.manz.at/document/rdb.tso.LIrdu20240204</u> (Accessed June 7, 2024).
- Ćetković, S., and Hagemann, C. (2020). Changing climate for populists? Examining the influence of radical-right political parties

on low-carbon energy transitions in Western Europe. *Energy Research & Social Science* 66, 101571. <u>https://doi.org/10.1016/j.</u> erss.2020.101571

- Chambers, J. M., Wyborn, C., Ryan, M. E., Reid, R. S., Riechers, M., Serban, A., et al. (2021). Six modes of co-production for sustainability. *Nature Sustainability* 4, 983–996. <u>https://doi.org/ 10.1038/s41893-021-00755-x</u>
- Choi, S., Niyogi, D., Shepardson, D. P., and Charusombat, U. (2010).
   Do Earth and Environmental Science Textbooks Promote Middle and High School Students' Conceptual Development About Climate Change? *Bulletin of the American Meteorological Society* 91, 889–898. <a href="https://doi.org/10.1175/2009bams2625.1">https://doi.org/10.1175/2009bams2625.1</a>
- Christophers, B. (2022). Fossilised Capital: Price and Profit in the Energy Transition. *New Political Economy* 27, 146–159. <u>https:// doi.org/10.1080/13563467.2021.1926957</u>
- Clar, C., Omann, I., and Scherhaufer, P. (2023). Der österreichische Klimarat – ein Beitrag zur Weiterentwicklung von Demokratie und Politik? SWS-Rundschau 63, 259–275. Available at: <u>https:// nbn-resolving.org/urn:nbn:de:0168-ssoar-90796-5</u>
- Clar, C., and Steurer, R. (2014). Mainstreaming climate change adaptation in a federal state setting: Policy changes in the flood protection and tourism sectors in Austria? Österreichische Zeitschrift für Politikwissenschaft 43, 23–47. <u>https://doi.org/10. 15203/ozp.84.vol43iss1</u>
- Clar, C., and Steurer, R. (2019). Climate change adaptation at different levels of government: Characteristics and conditions of policy change. *Natural Resources Forum* 43, 121–131. <u>https:// doi.org/10.1111/1477-8947.12168</u>
- Clausen, S. W. (2018). Exploring the pedagogical content knowledge of Danish geography teachers: teaching weather formation and climate change. *International Research in Geographical and Environmental Education* 27, 267–280. <u>https://doi.org/10.1080/</u> 10382046.2017.1349376
- COM/2021/82 final (2021). COMMUNICATION FROM THE COMMISSION TO THE EUROPEAN PARLIAMENT, THE COUNCIL, THE EUROPEAN ECONOMIC AND SOCIAL COMMITTEE AND THE COMMITTEE OF THE REGIONS Forging a climate-resilient Europe – the new EU Strategy on Adaptation to Climate Change. Available at: <u>https://eur-lex.europa. eu/legal-content/EN/TXT/?uri=COM%3A2021%3A82%3AFIN</u> (Accessed October 17, 2024).
- COM/2021/555 final (2021). Proposal for a REGULATION OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL amending Regulation (EU) 2018/842 on binding annual greenhouse gas emission reductions by Member States from 2021 to 2030 contributing to climate action to meet commitments under the Paris Agreement. Available at: <u>https://eur-lex.europa.eu/ legal-content/EN/TXT/?uri=COM%3A2021%3A555%3AFIN</u>
- COM/2021/563 final (2021). Proposal for a COUNCIL DIREC-TIVE restructuring the Union framework for the taxation of energy products and electricity (recast). Available at: <u>https://eurlex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A52021</u> <u>PC0563</u> (Accessed October 17, 2024).
- COM/2021/802 final (2021). Proposal for a DIRECTIVE OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL on the energy performance of buildings (recast). Available at: <u>https:// eur-lex.europa.eu/legal-content/EN/TXT/?uri=celex:52021</u> <u>PC0802</u> (Accessed September 26, 2023).
- COM/2021/805 (2021). Proposal for a REGULATION OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL on methane emissions reduction in the energy sector and amend-

ing Regulation (EU) 2019/942. Available at: <u>https://eur-lex.euro-</u> pa.eu/legal-content/EN/TXT/?uri=COM%3A2021%3A805%3 <u>AFIN</u>

- Competente, R. J. T. (2019). Pre-Service Teachers' Inclusion of Climate Change Education. *International Journal of Evaluation and Research in Education* 8, 119–126. <u>https://doi.org/10.11591/</u> <u>ijere.v8.i1.pp119-126</u>
- Cook, J., Ellerton, P., and Kinkead, D. (2018). Deconstructing climate misinformation to identify reasoning errors. *Environmental Research Letters* 13, 024018. <u>https://doi.org/10.1088/1748-9326/aaa49f</u>
- Costanza, R., Kubiszewski, I., Giovannini, E., Lovins, H., McGlade, J., Pickett, K. E., et al. (2014). Development: Time to leave GDP behind. *Nature* 505, 283–285. https://doi.org/10.1038/505283a
- Cottier, F., and Salehyan, I. (2021). Climate variability and irregular migration to the European Union. *Global Environmental Change* 69, 102275. <u>https://doi.org/10.1016/j.gloenvcha.2021.102275</u>
- Council Directive 92/43/EEC (1992). Council Directive 92/43/EEC of 21 May 1992 on the conservation of natural habitats and of wild fauna and flora. Available at: <u>https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A01992L0043-20130701</u>
- Council Regulation (EU) 2022/2577 (2022). Council Regulation (EU) 2022/2577 of 22 December 2022 laying down a framework to accelerate the deployment of renewable energy. Available at: <u>https://eur-lex.europa.eu/eli/reg/2022/2577/oj</u> (Accessed September 6, 2023).
- Creutzig, F., Roy, J., Devine-Wright, P., Díaz-José, J., Geels, F. W., Grubler, A., et al. (2022). "Demand, services and social aspects of mitigation Supplementary Material", in *IPCC*, 2022: Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, eds. P. R. Shukla, J. Skea, R. Slade, A. A. Khourdajie, R. van Diemen, D. McCollum, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press). https://doi.org/10.1017/9781009157926.007
- Cunningham, K., Hix, S., Dennison, S., and Learmonth, I. (2024). A sharp right turn: A forecast for the 2024 European Parliament elections. European Council of Foreign Relations. Available at: <u>https://ecfr.eu/publication/a-sharp-right-turn-a-forecast-forthe-2024-european-parliament-elections/</u> (Accessed October 17, 2024).
- Daggett, C. (2018). Petro-masculinity: Fossil Fuels and Authoritarian Desire. *Millennium* 47, 25–44. <u>https://doi.org/10.1177/</u> 0305829818775817
- D'Alessandro, S., Cieplinski, A., Distefano, T., and Dittmer, K. (2020). Feasible alternatives to green growth. *Nature Sustain-ability* 3, 329–335. <u>https://doi.org/10.1038/s41893-020-0484-y</u>
- Daly, H. E. (1973). *Toward a steady-state economy*. San Francisco: W. H. Freeman.
- Daniel, A., and Deutschmann, A. (2020). "Austria", in Protest for a future II Composition, mobilization and motives of the participants in Fridays For Future climate protests on 20-27 September, 2019, in 19 cities around the world, eds. J. De Moor, K. Uba, M. Wahlström, M. Wennerhag, and M. De Vydt. Available at: <u>https://ie.univie.ac.at/fileadmin/user\_upload/p\_ie/FORSCHUNG/FoWe\_Protest/Protest\_for\_a\_Future\_II\_-\_2020-02-24.pdf</u> (Accessed June 21, 2023).
- David, M. (2017). Moving beyond the heuristic of creative destruction: Targeting exnovation with policy mixes for energy transitions. *Energy Research & Social Science* 33, 138–146. <u>https://doi.org/10.1016/j.erss.2017.09.023</u>

- Davidson, D. J. (2019). Exnovating for a renewable energy transition. *Nat Energy* 4, 254–256. <u>https://doi.org/10.1038/s41560-019-0369-3</u>
- Dawson, V. (2015). Western Australian High School Students' Understandings about the Socioscientific Issue of Climate Change. International Journal of Science Education 37, 1024–1043. <u>https://doi.org/10.1080/09500693.2015.1015181</u>
- Decision (EU) 2015/1814 (2015). Decision (EU) 2015/1814 of the European Parliament and of the Council of 6 October 2015 concerning the establishment and operation of a market stability reserve for the Union greenhouse gas emission trading scheme and amending Directive 2003/87/EC. Available at: <u>https://eurlex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:32015</u> D1814 (Accessed September 14, 2023).
- Dederer, H.-G. (2021). Die Governance-Verordnung der Union. Nachhaltigkeitsrecht 1. https://doi.org/10.33196/nr202101002501
- Deisenrieder, V., Kubisch, S., Keller, L., and Stötter, J. (2020). Bridging the Action Gap by Democratizing Climate Change Education—The Case of k.i.d.Z.21 in the Context of Fridays for Future. *Sustainability* 12, 1748. <u>https://doi.org/10.3390/su12051748</u>
- Di Giulio, A., and Fuchs, D. (2014). Sustainable Consumption Corridors: Concept, Objections, and Responses. GAIA – Ecological Perspectives for Science and Society 23, 184–192. <u>https://doi. org/10.14512/gaia.23.S1.6</u>
- Directive 2003/87/EC (2003). Directive 2003/87/EC of the European Parliament and of the Council of 13 October 2003 Establishing a Scheme for Greenhouse Gas Emission Allowance Trading Within the Community and Amending Council Directive 96/61/Ec. Available at: <u>https://eur-lex.europa.eu/legal-content/</u> <u>EN/TXT/PDF/?uri=CELEX:32003L0087</u> (Accessed September 14, 2023).
- Directive 2008/101/EC (2008). Directive 2008/101/EC of the European Parliament and of the Council of 19 November 2008 amending Directive 2003/87/EC so as to include aviation activities in the scheme for greenhouse gas emission allowance trading within the Community. Available at: <a href="https://eurlex.europa.eu/legal-content/EN/TXT/?uri=celex%3A32008L0101">https://eurlex.europa.eu/legal-content/EN/TXT/?uri=celex%3A32008L0101</a>
- Directive (EU) 2023/959 (2023). Directive (EU) 2023/959 of the European Parliament and of the Council of 10 May 2023 amending Directive 2003/87/EC establishing a system for greenhouse gas emission allowance trading within the Union and Decision (EU) 2015/1814 concerning the establishment and operation of a market stability reserve for the Union greenhouse gas emission trading system (Text with EEA relevance). Available at: https://eur-lex.europa.eu/legal-content/EN/TXT/?toc=OJ% 3AL%3A2023%3A130%3ATOC&uri=uriserv%3AOJ.L\_.2023. 130.01.0134.01.ENG (Accessed September 14, 2023).
- Directive (EU) 2023/2413 (2023). Directive (EU) 2023/2413 of the European Parliament and of the Council of 18 October 2023 amending Directive (EU) 2018/2001, Regulation (EU) 2018/1999 and Directive 98/70/EC as regards the promotion of energy from renewable sources, and repealing Council Directive (EU) 2015/652. Available at: <u>https://eur-lex.europa.eu/eli/ dir/2023/2413/oj</u>
- Directive (EU) 2024/1275 (2024). Directive (EU) 2024/1275 of the European Parliament and of the Council of 24 April 2024 on the energy performance of buildings (recast) (Text with EEA relevance). Available at: <u>https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A32024L1275</u> (Accessed October 19, 2024).

- Directive (EU) 2024/1788 (2024). Directive (EU) 2024/1788 of the European Parliament and of the Council of 13 June 2024 on common rules for the internal markets for renewable gas, natural gas and hydrogen, amending Directive (EU) 2023/1791 and repealing Directive 2009/73/EC (recast) (Text with EEA relevance). Available at: https://eur-lex.europa.eu/eli/dir/2024/1788 (Accessed October 22, 2024).
- Dobruszkes, F., Mattioli, G., and Mathieu, L. (2022). Banning super short-haul flights: Environmental evidence or political turbulence? *Journal of Transport Geography* 104, 103457. <u>https://doi. org/10.1016/j.jtrangeo.2022.103457</u>
- Dolezal, M. (2016). "The Greens in Austria and Switzerland: Two successful opposition parties", in *Green Parties in Europe*, (Routledge).
- Dorninger, C., Hornborg, A., Abson, D. J., von Wehrden, H., Schaffartzik, A., Giljum, S., et al. (2021). Global patterns of ecologically unequal exchange: Implications for sustainability in the 21<sup>st</sup> century. *Ecological Economics* 179, 106824. <u>https://doi.org/10. 1016/j.ecolecon.2020.106824</u>
- Dörre, K., Liebig, S., Lucht, K., and Sittel, J. (2024). Klasse gegen Klima? Transformationskonflikte in der Autoindustrie. *Berlin* J Soziol 34, 9–46. <u>https://doi.org/10.1007/s11609-023-00514-z</u>
- Dow, K., Berkhout, F., Preston, B., Klein, R., Midgley, G., and Shaw, M. (2013). Limits to adaptation. *Nature Climate Change* 3, 305– 307. <u>https://doi.org/10.1038/nclimate1847</u>
- Dubo, T., Palomo, I., Camacho, L. L., Locatelli, B., Cugniet, A., Racinais, N., et al. (2022). Nature-based solutions for climate change adaptation are not located where they are most needed across the Alps. *Regional Environmental Change* 23, 12. <u>https:// doi.org/10.1007/s10113-022-01998-w</u>
- Dubois, G., Sovacool, B., Aall, C., Nilsson, M., Barbier, C., Herrmann, A., et al. (2019). It starts at home? Climate policies targeting household consumption and behavioral decisions are key to low-carbon futures. *Energy Research & Social Science* 52, 144–158. https://doi.org/10.1016/j.erss.2019.02.001
- Eberl, J.-M., Huber, L. M., and Plescia, C. (2020). A tale of firsts: the 2019 Austrian snap election. *West European Politics* 43, 1350–1363. <u>https://doi.org/10.1080/01402382.2020.1717836</u>
- Eckersley, R. (2021). Greening states and societies: from transitions to great transformations. *Environmental Politics* 30, 245–265. https://doi.org/10.1080/09644016.2020.1810890
- Eckhardt, G. (2022). Umwelt- und Klimaschutz in der Landwirtschaft – Zur Reform der GAP 2021. Nachhaltigkeitsrecht 2, 286– 295. <u>https://doi.org/10.33196/nr202203028601</u>
- Eder, J., and Schneider, E. (2018). Progressive Industrial Policy A Remedy for Europe!? *Journal für Entwicklungspolitik* 34, 108– 142. https://doi.org/10.20446/JEP-2414-3197-34-3-108
- EEA (2021a). The contribution of national advisory bodies to climate policy in Europe. *European Environment Agency*. Available at: <u>https://www.eea.europa.eu/publications/the-contribution-of-national-advisory/the-contribution-of-national-advisory</u> (Accessed August 20, 2024).
- EEA (2021b). Overview of reported national greenhouse gas policies and measures in Europe in 2021. Mol, Belgium: European Environmental Agency. Available at: <u>https://www.eionet.</u> <u>europa.eu/etcs/etc-cme/products/etc-cme-reports/etc-cme-report-5-2021-overview-of-reported-national-greenhouse-gaspolicies-and-measures-in-europe-in-2021/@@download/file/ <u>ETC-CME\_EIONET\_report\_5-2021.pdf</u> (Accessed September 7, 2023).</u>

- EEA (2024). Economic losses from weather- and climate-related extremes in Europe. Brussels: European Environment Agency. Available at: <u>https://www.eea.europa.eu/en/analysis/indicators/</u> <u>economic-losses-from-climate-related</u> (Accessed February 6, 2025).
- EEA Joint Committee (2007). DECISION OF THE EEA JOINT COMMITTEE No 146/2007 of 26 October 2007. Available at: <u>https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CE-</u> LEX:22007D0146&from=RO (Accessed September 14, 2023).
- Egger, A., Liebeswar, C., Mayer, W., Bock-Schappelwein, J., Falkner, K., Famira-Mühlberger, U., et al. (2024). "Ökosozialstaat – Handlungsfelder eines ökologisch nachhaltigen Sozialstaats", in Sozialbericht 2024, Band II: Sozialpolitische Analysen, ed.Bundesministerium für Soziales, Gesundheit, Pflege und Konsumentenschutz (Vienna: BMSGPK), 99–127. Available at: <u>https://forschungsnetzwerk.ams.at/dam/jcr:1258405b-1682-454b-968d-a82a287831fb/BMSGPK\_Sozialbericht2024\_Band-II\_Analysen.pdf</u>
- Eisenmenger, N., Plank, B., Milota, E., and Gierlinger, S. (2020). Ressourcennutzung in Österreich 2020. Band 3. Wien: Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie (BMK). Available at: <u>https://www. bmk.gv.at/themen/klima\_umwelt/nachhaltigkeit/ressourceneffizienz/publikationen/bericht2020.html</u> (Accessed January 17, 2025).
- Eisner, A., Kulmer, V., and Kortschak, D. (2021). Distributional effects of carbon pricing when considering household heterogeneity: An EASI application for Austria. *Energy Policy* 156, 112478. <u>https://doi.org/10.1016/j.enpol.2021.112478</u>
- Ekberg, K., Forchtner, B., Hultman, M., and Jylhä, K. M. (2022). Climate Obstruction: How Denial, Delay and Inaction are Heating the Planet., 1st Edn. London: Routledge. Available at: <u>https:// www.taylorfrancis.com/books/9781003181132</u> (Accessed May 2, 2024).
- Elgouacem, A. (2020). Designing fossil fuel subsidy reforms in OECD and G20 countries: A robust sequential approach methodology. OECD Environment Working Papers. Paris: OECD Publishing. <u>https://doi.org/10.1787/d888f461-en</u>
- Ellerman, A. D., Convery, F. J., and De Perthuis, C. (2010). Pricing Carbon: The European Union Emissions Trading Scheme. Cambridge: Cambridge University Press. Available at: <u>https://cadmus.eui.eu/handle/1814/15503</u> (Accessed May 7, 2024).
- Engartner, T. (2019). Die integrative Kraft sozioökonomischer Bildung-oder: Herausforderungen für die sozialwissenschaftliche Kontextualisierung wirtschaftlicher Phänomene und ökonomischer Logiken. *GW-Unterricht* 153, 20–26. <u>https://doi.org/10.</u> <u>1553/gw-unterricht153s20</u>
- Ennöckl, D. (2020). Wie kann das Recht das Klima schützen? Österreichische Jurist:innenzeitung 7, 302.
- Ennöckl, D. (2023a). "Klimaschutz und Verfassung", in Klimaschutzrecht, ed. D. Ennöckl (Verlag Österreich), 75–103. <u>https:// doi.org/10.33196/9783704692610-103</u>
- Ennöckl, D. (2023b). "Klimaschutzgesetz Österreich", in Klimaschutzrecht, ed. D. Ennöckl (Verlag Österreich), 105–120. https://doi.org/10.33196/9783704692610-104
- Ennöckl, D., Handig, N., Polzer, N., Rahtmayer, F., and Vouk, L. (2024). KlimaSeniorinnen erkämpfen Recht auf Klimaschutz vor dem EGMR. *ÖJZ* 10. Available at: <u>https://rdb.manz.at/document/rdb.tso.Lloejz20241018#doc-a\_1\_2\_1</u> (Accessed October 17, 2024).

- Ennöckl, D., and Sander, P. (2023). Die Berücksichtigung von Aspekten des Klimaschutzes im Rahmen der Umweltverträglichkeitsprüfung. ÖZW 2, 38.
- Ennser, B. ed. (2022). *EAG Erneuerbaren-Ausbau-Gesetz*. Wien: Verlag Österreich.
- Eskjær, M. (2013). The Regional Dimension: How Regional Media Systems Condition Global Climate-Change Communication. Journal of International and Intercultural Communication 6, 61–81. <u>https://doi.org/10.1080/17513057.2012.748933</u>
- European Commission, and Swiss Confederation (2017). AGREE-MENT between the European Union and the Swiss Confederation on the linking of their greenhouse gas emissions trading systems. Available at: <u>https://eur-lex.europa.eu/legal-content/</u> <u>EN/TXT/PDF/?uri=CELEX:22017A1207(01)</u> (Accessed September 14, 2023).
- Evans, N., and Duwe, M. (2021). Climate governance systems in Europe: the role of national advisory bodies. Berlin, Paris: Ecologic Institute, IDDRI. Available at: <u>https://www.ecologic.eu/sites/ default/files/publication/2021/Evans-Duwe-Climate-governance-in-Europe-the-role-of-national-advisory-bodies-2021-Ecologic-Institute.pdf</u> (Accessed May 24, 2024).
- Evensen, D., Demski, C., Becker, S., and Pidgeon, N. (2018). The relationship between justice and acceptance of energy transition costs in the UK. *Applied Energy* 222, 451–459. <u>https://doi. org/10.1016/j.apenergy.2018.03.165</u>
- Eversberg, D. (2018). Innerimperiale Kämpfe: Drei Thesen zum Verhältnis zwischen autoritärem Nationalismus und imperialer Lebensweise. PROKLA. Zeitschrift für kritische Sozialwissenschaft 48, 43–54. https://doi.org/10.32387/prokla.v48i190.31
- Eversberg, D., Fritz, M., Faber, L. von, and Schmelzer, M. (2024). Der neue sozial-ökologische Klassenkonflikt: Mentalitäts- und Interessengegensätze im Streit um Transformation. Frankfurt New York: Campus Verlag.
- Ewald, J., Sterner, T., and Sterner, E. (2022). Understanding the resistance to carbon taxes: Drivers and barriers among the general public and fuel-tax protesters. *Resource and Energy Economics* 70, 101331. <u>https://doi.org/10.1016/j.reseneeco.2022.101331</u>
- Fanning, A. L., O'Neill, D. W., and Büchs, M. (2020). Provisioning systems for a good life within planetary boundaries. *Global Environmental Change* 64, 102135. <u>https://doi.org/10.1016/j.gloenvcha.2020.102135</u>
- Fedele, G., Donatti, C. I., Harvey, C. A., Hannah, L., and Hole, D. G. (2019). Transformative adaptation to climate change for sustainable social-ecological systems. *Environmental Science & Policy* 101, 116–125. <u>https://doi.org/10.1016/j.envsci.2019.07.001</u>
- Feichtinger, J., Stickler, T., Schuch, K., and Lexer, W. (2021). Sustainable development and climate change mitigation at the rural municipal level in Austria: Tracing policy diffusion, process dynamics and political change. GAIA – Ecological Perspectives for Science and Society 30, 189–197. <u>https://doi.org/10.14512/ gaia.30.3.12</u>
- Fekete, A., Fuchs, S., Garschagen, M., Hutter, G., Klepp, S., Lüder, C., et al. (2022). Adjustment or transformation? Disaster risk intervention examples from Austria, Indonesia, Kiribati and South Africa. *Land Use Policy* 120, 106230. <u>https://doi.org/ 10.1016/j.landusepol.2022.106230</u>
- Feng, S., Krueger, A. B., and Oppenheimer, M. (2010). Linkages among climate change, crop yields and Mexico–US cross-border migration. *Proc. Natl. Acad. Sci. U.S.A.* 107, 14257–14262. <u>https://doi.org/10.1073/pnas.1002632107</u>

- Feola, G. (2020). Capitalism in sustainability transitions research: Time for a critical turn? *Environmental Innovation and Societal Transitions* 35, 241–250. <u>https://doi.org/10.1016/j.eist.2019.</u> 02.005
- Fidler, P. (2023). Elektromobilität nach der WEG-Nov 2022. ZVR 4, 188.
- Filatova, T. (2014). Market-based instruments for flood risk management: A review of theory, practice and perspectives for climate adaptation policy. *Environmental Science & Policy* 37, 227–242. https://doi.org/10.1016/j.envsci.2013.09.005
- Fitz, J. (2023a). Rechtsvergleichende Ideen für die Mobilität der Zukunft. ZVR 4, 183.
- Fitz, J. (2023b). "Klimaklagen", in *Klimaschutzrecht*, ed. D. Ennöckl (Vienna: Verlag Österreich), 517–540.
- Foldal, C. B., Kasper, M., Ecker, E., and Zechmeister-Boltenstein, S. (2019). Evaluierung verschiedener ÖPUL Maßnahmen in Hinblick auf die Reduktion von Treibhausgasemissionen, insbesondere Lachgas. Wien: University of Life Sciences Vienna (BOKU). Available at: <u>https://info.bml.gv.at/dam/jcr:41ac1615a50b-4945-b144-969d0137353d/Endbericht%20Lachgas%20 Final\_%2002\_12\_2019.pdf</u>
- Forchtner, B. (2019). Climate change and the far right. WIREs Clim Change 10. https://doi.org/10.1002/wcc.604
- Forchtner, B. (2020). "Articulations of climate change by the Austrian far right A discourse-historical perspective on what is 'allegedly manmade," in *Europe at the Crossroads: Confronting Populist, Nationalist, and Global Challenges*, eds. P. Bevelander and R. Wodak (Nordic Academic Press), 159–190. <u>https://doi.org/10.2307/jj.919494</u>
- Forchtner, B., Kroneder, A., and Wetzel, D. (2018). Being Skeptical? Exploring Far-Right Climate-Change Communication in Germany. *Environmental Communication* 12, 589–604. <u>https://doi.org/10.1080/17524032.2018.1470546</u>
- Fouquet, R. ed. (2019). Handbook on green growth. Cheltenham, UK Northampton, MA, USA: Edward Elgar Publishing. <u>https:// doi.org/10.4337/9781788110686</u>
- Fournier, V. (2008). Escaping from the economy: the politics of degrowth. *International journal of sociology and social policy* 28, 528–545. https://doi.org/10.1108/01443330810915233
- Frenz, W. (2022). "EU-Klimagesetz", in Klimaschutzrecht: EU-Klimagesetz, KSG Bund und NRW, BEHG, Steuerrecht, Querschnittsthemen, ed. W. Frenz (Berlin: Erich Schmidt Verlag GmbH & Co. KG), 427–524. <u>https://doi.org/10.37307/b.978-3-503-20687-2.03</u>
- Freudenschuß, A., Sedy, K., Zethner, G., and Spiegel, H. (2010). Arbeiten zur Evaluierung von ÖPUL-Maßnahmen hinsichtlich ihrer Klimawirksamkeit – Schwerpunkt agrarische Bewirtschaftung. Vienna: Federal Environment Agency. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/publikationen/rep0290.pdf</u>
- Fritz, M., and Eversberg, D. (2024). Mentalities, classes and the four lines of conflict in the social-ecological transformation. *Eur Polit Sci* 23, 39–55. <u>https://doi.org/10.1057/s41304-023-00457-2</u>
- Gabelberger, F., Kettner-Marx, C., Peneder, M., and Streicher, G. (2020). Landkarte der "(De-)Karbonisierung" für den produzierenden Bereich in Österreich. Vienna: Austrian Institute of Economic Research (WIFO). Available at: <u>https://emedien.</u> <u>arbeiterkammer.at/viewer/api/v1/records/AC16063422/files/</u> <u>source/AC16063422.pdf</u> (Accessed September 13, 2023).
- Gangale, F., Chondrogiannis, S., and Shortall, R. (2023). Making energy regulation fit for purpose: State of play of regulato-

ry experimentation in the EU, EUR 31438 EN. Luxembourg: Publications Office of the European Union. <u>https://dx.doi.</u> org/10.2760/32253

- Gärner, C., Pinetz, E., Schaffer, E., Tillian, F., and Ziniel, T. (2023). Rechtsprechung des Verfassungsgerichtshofs. *ecolex* 11, 986. Available at: <u>https://rdb.manz.at/document/rdb.tso.LIecolex</u> 20240546
- Gavenda, M., and Umit, R. (2016). The 2016 Austrian Presidential Election: A Tale of Three Divides. *Regional & Federal Studies* 26, 419–432. https://doi.org/10.1080/13597566.2016.1206528
- GCEC (2014). Better Growth Better Climate: The New Climate Economy Report. The Synthesis Report. Washington D.C.: New Climate Economy. Available at: <u>https://sustainabledevelopment.</u> <u>un.org/content/documents/1595TheNewClimateEconomyReport.pdf</u> (Accessed June 17, 2024).
- Geels, F. W. (2014). Regime Resistance against Low-Carbon Transitions: Introducing Politics and Power into the Multi-Level Perspective. *Theory, Culture & Society* 31, 21–40. <u>https://doi.org/10.1177/0263276414531627</u>
- Gemenne, F., and Blocher, J. (2017). How can migration serve adaptation to climate change? Challenges to fleshing out a policy ideal. *Geogr J* 183, 336–347. <u>https://doi.org/10.1111/geoj.12205</u>
- George, J. F., Werner, S., Preuß, S., Winkler, J., Held, A., and Ragwitz, M. (2023). The landlord-tenant dilemma: Distributional effects of carbon prices, redistribution and building modernisation policies in the German heating transition. *Applied Energy* 339, 120783. <u>https://doi.org/10.1016/j.apenergy.2023.120783</u>
- Gibson, T. A. (2016). "Economic, Technological, and Organizational Factors Influencing News Coverage of Climate Change", in Oxford Research Encyclopedia of Climate Science, (Oxford: Oxford University Press). <u>https://doi.org/10.1093/acrefore/9780190228620.013.355</u>
- Giest, S. (2020). Do nudgers need budging? A comparative analysis of European smart meter implementation. Government Information Quarterly 37, 101498. <u>https://doi.org/10.1016/j.giq.2020.101498</u>
- Giljum, S., Wieland, H., Lutter, S., Eisenmenger, N., Schandl, H., and Owen, A. (2019). The impacts of data deviations between MRIO models on material footprints: A comparison of EXIOBASE, Eora, and ICIO. *Journal of Industrial Ecology* 23, 946–958. https://doi.org/10.1111/jiec.12833
- Gillingham, K., Rapson, D., and Wagner, G. (2016). The Rebound Effect and Energy Efficiency Policy. *Review of Environmental Economics and Policy* 10, 68–88. <u>https://doi.org/10.1093/reep/ rev017</u>
- Goers, S., and Schneider, F. (2019). Austria's Path to a Climate-Friendly Society and Economy—Contributions of an Environmental Tax Reform. *Modern Economy* 10, 1369–1384. <u>https://doi.org/10.4236/me.2019.105092</u>
- Görg, C., Brand, U., Haberl, H., Hummel, D., Jahn, T., and Liehr, S. (2017). Challenges for Social-Ecological Transformations: Contributions from Social and Political Ecology. *Sustainability* 9, 1045. <u>https://doi.org/10.3390/su9071045</u>
- Gough, I. (2015). Climate change and sustainable welfare: the centrality of human needs. Cambridge Journal of Economics 39, 1191–1214. <u>https://doi.org/10.1093/cje/bev039</u>
- Gough, I., Meadowcroft, J., Dryzek, J., Gerhards, J., Lengfeld, H., Markandya, A., et al. (2008). JESP symposium: Climate change and social policy. *Journal of European Social Policy* 18, 325–344. https://doi.org/10.1177/0958928708094890

- Graber, F. (2023). "Klimaschutz und Mobilitätsinfrastruktur", in *Klimaschutzrecht*, ed. D. Ennöckl (Vienna: Verlag Österreich), 423–469. <u>https://doi.org/10.33196/9783704692610-116</u>
- Granier, A., Reichstein, M., Bréda, N., Janssens, I. A., Falge, E., Ciais, P., et al. (2007). Evidence for soil water control on carbon and water dynamics in European forests during the extremely dry year: 2003. Agricultural and Forest Meteorology 143, 123– 145. <u>https://doi.org/10.1016/j.agrformet.2006.12.004</u>
- Green, F. (2018). The logic of fossil fuel bans. *Nature Clim Change* 8, 449-451. <u>https://doi.org/10.1038/s41558-018-0172-3</u>
- Green, J. F. (2021). Does carbon pricing reduce emissions? A review of ex-post analyses. *Environ. Res. Lett.* 16, 043004. <u>https://doi.org/10.1088/1748-9326/abdae9</u>
- Greenpeace International (2021). Destruction: Certified. Greenpeace International. Available at: <u>https://www.greenpeace.org/</u> <u>international/publication/46812/destruction-certified/</u> (Accessed July 14, 2023).
- Groh, E. D., and v. Möllendorff, C. (2020). What shapes the support of renewable energy expansion? Public attitudes between policy goals and risk, time, and social preferences. *Energy Policy* 137, 111171. <u>https://doi.org/10.1016/j.enpol.2019.111171</u>
- Großmann, A., Stocker, A., and Wolter, M. I. (2019). Evaluation von Klimaschutzmaßnahmen mit dem Modell e3.at meetPASS: meeting the Paris Agreement and Supporting Sustainability Working Paper No. 5. <u>https://doi.org/10.13140/RG.2.2.32733.</u> 44000
- Großmann, A., Wolter, M. I., Hitnerberger, F., and Püls, L. (2020). Die Auswirkungen von klimapolitischen Maßnahmen auf den österreichischen Arbeitsmarkt. ExpertInnenbericht. Osnabrück, Vienna. Available at: <u>https://www.gws-os.com/en/publications/all-publications/detail/2020-die-auswirkungen-vonklimapolitischen-massnahmen-auf-den-oesterreichischenarbeitsmarkt</u> (Accessed September 13, 2023).
- Grubb, M., Crawford-Brown, D., Neuhoff, K., Schanes, K., Hawkins, S., and Poncia, A. (2020). Consumption-oriented policy instruments for fostering greenhouse gas mitigation. *Climate Policy* 20, S58–S73. <u>https://doi.org/10.1080/14693062.2020.17</u> <u>30151</u>
- Grubler, A., Wilson, C., Bento, N., Boza-Kiss, B., Krey, V., McCollum, D. L., et al. (2018). A low energy demand scenario for meeting the 1.5 °C target and sustainable development goals without negative emission technologies. *Nature Energy* 3, 515– 527. <u>https://doi.org/10.1038/s41560-018-0172-6</u>
- Gruère, G. (2013). A Characterisation of Environmental Labelling and Information Schemes. OECD Environment Working Papers, No. 62. Paris: OECD Publishing. <u>https://doi.org/10. 1787/5k3z11hpdgq2-en</u>
- Grund, J., and Brock, A. (2018). Bildung für nachhaltige Entwicklung in Lehr-Lernsettings–Quantitative Studie des nationalen Monitorings–Befragung junger Menschen. Executive Summary. Berlin. Available at: <u>https://www.bne-portal.de/bne/ shareddocs/downloads/files/nationales-monitoring\_quantitative-studie\_junge-menschen.pdf?\_\_blob=publicationFile&v=2</u>
- Gugele, B., Krutzler, T., Miess, M., and Vogel, J. (2022). Pilotprojekt: Integration eines Klimamoduls in die langfristige Budgetprognose. Endbericht. Wien: Umweltbundesamt (Environment Agency Austria). Available at: <u>https://www.umweltbundesamt.</u> <u>at/fileadmin/site/publikationen/rep0837.pdf</u> (Accessed January 20, 2023).
- Gugler, K., Haxhimusa, A., and Liebensteiner, M. (2021). Effectiveness of climate policies: Carbon pricing vs. subsidizing re-

newables. Journal of Environmental Economics and Management 106, 102405. https://doi.org/10.1016/j.jeem.2020.102405

- Gupta, S., Harnisch, D., Barua, L., Chingambo, P., Frabkel, R. J., Garrido Vázquez, L., et al. (2014). "Cross-cutting Investment and Finance Issues", in *Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Inter- governmental Panel on Climate Change*, eds. O. Edenhofer, R. Pichs-Madruga, Y. Sokona, E. Farahani, S. Kadner, K. Seyboth, et al. (Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press).
- Haberl, H., Wiedenhofer, D., Virág, D., Kalt, G., Plank, B., Brockway, P., et al. (2020). A systematic review of the evidence on decoupling of GDP, resource use and GHG emissions, part II: synthesizing the insights. *Environ. Res. Lett.* 15, 065003. <u>https:// doi.org/10.1088/1748-9326/ab842a</u>
- Haites, E. (2018). Carbon taxes and greenhouse gas emissions trading systems: what have we learned? *Climate Policy* 18, 955–966. <u>https://doi.org/10.1080/14693062.2018.1492897</u>
- Handig, N. (2023). Chamäleon Umweltorganisation: Spektrum ihrer Rechtsstellung in Umweltverfahren. *RdU* 85, 147.
- Handig, N., and Rathmayer, F. (2024). RED III und Genehmigungsverfahren. *RdW* 70, 97.
- Handig, N., and Stangl, F. (2023). "Klimaschutz und Unionsrecht", in *Klimaschutzrecht*, ed. D. Ennöckl (Vienna: Verlag Österreich), 39–73. https://doi.org/10.33196/9783704692610-102
- Handstanger, M. (2022). "Klimaschutz und Verfassung", in Jahrbuch des österreichischen und europäischen Umweltrechts 2022: Klimaschutz im Recht, ed. Institut für Umweltrecht (Wien: MANZ'sche Verlags- und Universitätsbuchhandlung). <u>https:// doi.org/10.5771/9783214250065</u>
- Hanger-Kopp, S., Thaler, T., Seebauer, S., Schinko, T., and Clar, C. (2022). Defining and operationalizing path dependency for the development and monitoring of adaptation pathways. *Global Environmental Change* 72, 102425. <u>https://doi.org/10.1016/j.gloenvcha.2021.102425</u>
- Hardt, L., Barrett, J., Taylor, P. G., and Foxon, T. J. (2020). Structural Change for a Post-Growth Economy: Investigating the Relationship between Embodied Energy Intensity and Labour Productivity. *Sustainability* 12, 962. <u>https://doi.org/10.3390/ su12030962</u>
- Hardt, L., Barrett, J., Taylor, P. G., and Foxon, T. J. (2021). What structural change is needed for a post-growth economy: A framework of analysis and empirical evidence. *Ecological Economics* 179, 106845. <u>https://doi.org/10.1016/j.ecolecon.2020.</u> <u>106845</u>
- Harvey, D. (2003). The New Imperialism. Oxford: Oxford University Press. <u>https://doi.org/10.1093/oso/9780199264315.001.0001</u>
- Hausknost, D. (2014). Decision, choice, solution: 'agentic deadlock' in environmental politics. *Environmental Politics* 23, 357–375. https://doi.org/10.1080/09644016.2013.874138
- Hausknost, D. (2020). The environmental state and the glass ceiling of transformation. *Environmental Politics* 29, 17–37. <u>https:// doi.org/10.1080/09644016.2019.1680062</u>
- Häusler, K., and Schlenk, M. (2021). Nachhaltiger Wohnbau: Vorgaben des Raumordnungs- und Baurechts. *Immolex* 11, 366– 370.
- Hawken, P., Lovins, A., and Lovins, L. H. (2000). Natural Capitalism: Creating the next Industrial Revolution., 1. paperback ed., [Nachdr.]. New York: Little, Brown and Co.
- HBLFA Raumberg-Gumpenstein (2023). Klimawandel-Anpassung – Bildungs- und Beratungstools. Available at: https://

raumberg-gumpenstein.at/forschung/forschung-aktuelles/ klimawandel-anpassung-bildungs-und-beratungstools.html #downloadbereich-und-forschungsdokumentation (Accessed May 8, 2024).

- Henderson, J. V., Storeygard, A., and Deichmann, U. (2017). Has climate change driven urbanization in Africa? *Journal of Devel*opment Economics 124, 60–82. <u>https://doi.org/10.1016/j.jdeveco.2016.09.001</u>
- Henning, K., Steimanis, I., and Vollan, B. (2022). (Climate) Migrants welcome? Evidence from a survey experiment in Austria. *Reg Environ Change* 22, 108. <u>https://doi.org/10.1007/s10113-022-01955-7</u>
- Henry, M. S., Bazilian, M. D., and Markuson, C. (2020). Just transitions: Histories and futures in a post-COVID world. *Energy Research & Social Science* 68, 101668. <u>https://doi.org/10.1016/j.</u> erss.2020.101668
- Hermann, A. T., Pregernig, M., Hogl, K., and Bauer, A. (2015). Cultural Imprints on Scientific Policy Advice: Climate science–policy interactions within Austrian neo-corporatism. *Environmental Policy and Governance* 25, 343–355. <u>https://doi.org/10.1002/ eet.1674</u>
- Herring, R., Sandeman, K., and Zarsky, L. (2024). Decarbonization, critical minerals, and tribal sovereignty: Pathways towards conflict transformation. *Energy Research & Social Science* 113, 103561. <u>https://doi.org/10.1016/j.erss.2024.103561</u>
- Hess, D. J., and Collins, B. M. (2018). Climate change and higher education: Assessing factors that affect curriculum requirements. *Journal of Cleaner Production* 170, 1451–1458. <u>https:// doi.org/10.1016/j.jclepro.2017.09.215</u>
- Hess, D. J., and Maki, A. (2019). Climate change belief, sustainability education, and political values: Assessing the need for higher-education curriculum reform. *Journal of Cleaner Production* 228, 1157–1166. <u>https://doi.org/10.1016/j.jclepro.2019.04.291</u>
- Heutel, G. (2015). Optimal policy instruments for externality-producing durable goods under present bias. *Journal of Environmental Economics and Management* 72, 54–70. <u>https://doi.org/10.1016/j.jeem.2015.04.002</u>
- Heyen, D. A., Hermwille, L., and Wehnert, T. (2017). Out of the Comfort Zone! Governing the Exnovation of Unsustainable Technologies and Practices. GAIA – Ecological Perspectives for Science and Society 26, 326–331. <u>https://doi.org/10.14512/gaia.</u> <u>26.4.9</u>
- Heyl, K., Ekardt, F., Sund, L., and Roos, P. (2022). Potentials and Limitations of Subsidies in Sustainability Governance: The Example of Agriculture. Sustainability 14, 15859. <u>https://doi.org/10.3390/su142315859</u>
- Hickel, J. (2017). *The Divide: A Brief Guide to Global Inequality and its Solutions*. London: William Heinemann.
- Hickel, J., and Hallegatte, S. (2022). Can we live within environmental limits and still reduce poverty? Degrowth or decoupling? *Development Policy Review* 40, e12584. <u>https://doi.org/10.1111/ dpr.12584</u>
- Hickel, J., Kallis, G., Jackson, T., O'Neill, D. W., Schor, J. B., Steinberger, J. K., et al. (2022). Degrowth can work — here's how science can help. *Nature* 612, 400–403. <u>https://doi.org/10.1038/</u> d41586-022-04412-x
- Hiramatsu, A., Kurisu, K., Nakamura, H., Teraki, S., and Hanaki, K. (2014). Spillover Effect on Families Derived from Environmental Education for Children. *Low Carbon Economy* 5, 40–50. <u>https://doi.org/10.4236/lce.2014.52005</u>

- Hirth, S., Kreinin, H., Fuchs, D., Blossey, N., Mamut, P., Philipp, J., et al. (2023). Barriers and enablers of 1.5° lifestyles: Shallow and deep structural factors shaping the potential for sustainable consumption. *Front. Sustain.* 4. <u>https://doi.org/10.3389/ frsus.2023.1014662</u>
- Hofbauer, J., Gerold, S., Klaus, D., and Wukovitsch, F. (2023). "Kapitel 7. Erwerbsarbeit", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer, Spektrum), 285–307. <u>https://doi.org/10.1007/978-3-662-66497-1\_11</u>
- Hofer, M. (2023a). "Klimaschutz und Anlagenrecht (Non-ETS)", in *Klimaschutzrecht*, ed. D. Ennöckl (Vienna: Verlag Österreich), 249–266. <u>https://doi.org/10.33196/9783704692610-110</u>
- Hofer, M. (2023b). Klimaschutz in der Umweltverträglichkeitsprüfung. Nachhaltigkeitsrecht: Zeitschrift für das Recht der nachhaltigen Entwicklung. Heft 4 3, 402–408.
- Hofer, M. (2024a). "7. Internationales und Europäisches Klimaschutzrecht – Hoffnung oder Hemmschuh im Kampf gegen die Erderwärmung?", in *Nachhaltigkeit im Spiegel des Rechts*, eds. M. Kramme and E. Ponholzer (Vienna: Verlag Österreich), 101–113. <u>https://doi.org/10.33196/9783704693631-107</u>
- Hofer, M. (2024b). Klimaschutz als Grundrecht: EGMR erkennt positive Klimaschutzpflichten der Vertragsstaaten an. *RdU* 87, 136.
- Hoffmann, M., and Spash, C. L. (2021). The impacts of climate change mitigation on work for the Austrian economy. Institute for Multilevel Governance and Development. SRE – Discussion Papers No. 10/2021. Vienna: WU Vienna University of Economics and Business. Available at: <u>https://research.wu.ac.at/ws/ files/19015690/sre-disc-2021\_10.pdf</u> (Accessed September 15, 2023).
- Hoffmann, R., Dimitrova, A., Muttarak, R., Crespo Cuaresma, J., and Peisker, J. (2020). A meta-analysis of country-level studies on environmental change and migration. *Nat. Clim. Chang.* 10, 904–912. <u>https://doi.org/10.1038/s41558-020-0898-6</u>
- Hollaus, B. (2024). Das Urteil des EGMR im Fall KlimaSeniorinnen und seine Implikationen für den europäischen Grundrechtsschutz. *JBl – Juristische Blätter* 8, 485.
- Holzner, J. (2008). Wenn der Klimawandel zum Thema wird. Betrachtung der Berichterstattung ausgewählter österreichischer Printmedien 2001–2007. Graz: Karl-Franzens-Universität Graz.
- Horn, E., and Bergthaller, H. (2019). *The Anthropocene: Key Issues for the Humanities*. London: Routledge.
- Hübner, R., Weber, M., Lindenthal, T., and Rauch, F. (2020). Für Nachhaltigkeit bilden? Bildung für Nachhaltige Entwicklung für Hochschullehrende an Universitäten in Österreich. GAIA-Ecological Perspectives for Science and Society 29, 70–72. <u>http://dx. doi.org/10.14512/gaia.29.1.17</u>
- Hugeneck, M. (2023). Die 33. StVO-Novelle 2022 für eine neue Geh- und Radkultur – Staffel wandert zum Vollzug. *RdU-UT* 9, 24.
- Hunter, L. M., and Simon, D. H. (2019). "Climate change, migration and health", in A research agenda for migration and health, eds. K. B. Newbold and K. Wilson (Cheltenham, UK Northampton, MA, USA: Edward Elgar Publishing), 50–66. <u>https://doi.org/10.4337/9781786438362.00008</u>
- IEA (2020). Iron and Steel Technology Roadmap. Paris: International Energy Agency (IEA). Available at: <u>https://www.iea.org/</u> reports/iron-and-steel-technology-roadmap

- In, S. Y., and Schumacher, K. (2021). "Carbonwashing: ESG Data Greenwashing in a Post-Paris World," in *Settling Climate Accounts*, eds. T. Heller and A. Seiger (Cham: Palgrave Macmillan), 39–58. <u>https://doi.org/10.1007/978-3-030-83650-4\_3</u>
- Initiative Wege aus der Krise (2019). Just Transition Klimaschutz demokratisch gestalten! Vienna: Initiative Wege aus der Krise. Available at: <u>https://www.wege-aus-der-krise.at/images/Just\_Transition\_final.pdf</u> (Accessed September 13, 2023).
- Institut Rousseau (2024). Road to Net Zero Bridging the Green Investment Gap. Available at: <u>https://drive.google.com/file/d/1c</u> <u>DQmQB0iezwthxiaoM83VxbllhrNz-0O/view</u> (Accessed May 8, 2023).
- Jackson, T., and Victor, P. A. (2020). The Transition to a Sustainable Prosperity-A Stock-Flow-Consistent Ecological Macroeconomic Model for Canada. *Ecological Economics* 177, 106787. <u>https:// doi.org/10.1016/j.ecolecon.2020.106787</u>
- Jacobson, C., Crevello, S., Chea, C., and Jarihani, B. (2019). When is migration a maladaptive response to climate change? *Reg Environ Change* 19, 101–112. <u>https://doi.org/10.1007/s10113-018-</u> <u>1387-6</u>
- Jarre, M., Noussan, M., and Campisi, E. (2024). Avoid–Shift–Improve: Are Demand Reduction Strategies Under-Represented in Current Energy Policies? *Energies* 17, 4955. <u>https://doi. org/10.3390/en17194955</u>
- Jessop, B. (2007). State power: a strategic-relational approach. Cambridge; Malden, MA: Polity Press.
- Johnstone, P., and Newell, P. (2018). Sustainability transitions and the state. *Environmental Innovation and Societal Transitions* 27, 72–82. <u>https://doi.org/10.1016/j.eist.2017.10.006</u>
- Jones, R. N., Patwardhan, A., Cohen, S., Dessai, S., Lammel, A., Lempert, R., et al. (2014). "Foundations for Decision Making," in Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change, eds. C. B. Field, V. Barros, D. J. Dokken, K. J. Mach, M. D. Mastrandrea, T. E. Bilir, et al. (Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press), 195–228. <u>https://doi.org/10.1017/CBO9781</u> 107415379.007
- Jordan, A., and Lenschow, A. (2010). Environmental policy integration: a state of the art review. *Environmental Policy and Gov*ernance 20, 147–158. <u>https://doi.org/10.1002/eet.539</u>
- Jordan, A., Wurzel, R. K. W., and Zito, A. R. (2003). "New" Instruments of Environmental Governance: Patterns and Pathways of Change. *Environmental Politics* 12, 1–24. <u>https://doi.org/10. 1080/714000665</u>
- Jordan, A., Wurzel, R. K. W., and Zito, A. R. (2013). Still the century of 'new' environmental policy instruments? Exploring patterns of innovation and continuity. *Environmental Politics* 22, 155– 173. <u>https://doi.org/10.1080/09644016.2013.755839</u>
- Kaczan, D. J., and Orgill-Meyer, J. (2020). The impact of climate change on migration: a synthesis of recent empirical insights. *Climatic Change* 158, 281–300. <u>https://doi.org/10.1007/s10584-019-02560-0</u>
- Kaiser, F., and Schwarz, H. (2022). "Kritische Reflexionen zur Genese und aktuellen Verankerung der Nachhaltigkeit in den Mindeststandards der Ausbildungsordnungen", in Berufsbildung für eine nachhaltige Entwicklung. Umsetzungsbarrieren und interdisziplinäre Forschungsfragen. Bielefeld: wbv Publikation, (Bielefeld: wbv Publikation), 133–152.

- Kallis, G., Kostakis, V., Lange, S., Muraca, B., Paulson, S., and Schmelzer, M. (2018). Research On Degrowth. *Annual Review* of Environment and Resources 43, 291–316. <u>https://doi.org/10. 1146/annurev-environ-102017-025941</u>
- Kalt, T. (2021). Jobs vs. climate justice? Contentious narratives of labor and climate movements in the coal transition in Germany. *Environmental Politics* 30, 1135–1154. <u>https://doi.org/10.1080/ 09644016.2021.1892979</u>
- Kalt, T. (2024). Transition conflicts: A Gramscian political ecology perspective on the contested nature of sustainability transitions. *Environmental Innovation and Societal Transitions* 50, 100812. <u>https://doi.org/10.1016/j.eist.2024.100812</u>
- Kapeller, J., Leitch, S., and Wildauer, R. (2023). Can a European wealth tax close the green investment gap? *Ecological Economics* 209, 107849. <u>https://doi.org/10.1016/j.ecolecon.2023.107849</u>
- Karimi-Schmidt, Y. (2019). Klimaschutz aus völkerrechtlicher Sicht unter besonderer Berücksichtigung der zivilen Luftfahrt. Zeitschrift für öffentliches Recht 74, 35–68. <u>https://doi.org/10. 33196/zoer201901003501</u>
- Kärnä, A., Karlsson, J., and Engberg, E. (2020). Political Failures in Innovation Policy: A Cautionary Note. IFN Working Paper, No. 1334. Stockholm: Research Institute of Industrial Economics (IFN). Available at: <u>http://www.diva-portal.org/smash/get/</u> diva2:1529420/FULLTEXT01.pdf (Accessed October 22, 2024).
- Kathrein, S. (2015). Medienpräsenz von Umweltproblemen in Österreich. Graz: Karl-Franzens-Universität Graz. Available at: <u>https://unipub.uni-graz.at/obvugrhs/content/titleinfo/309919/</u> <u>full.pdf</u> (Accessed September 13, 2023).
- Keil, A. K. (2021). Just Transition strategies for the Austrian and German automotive industry in the course of vehicle electrification. Vienna: Austrian Chamber of Labour. Available at: <u>https:// emedien.arbeiterkammer.at/viewer/image/AC16154784/49/ LOG\_0041/</u> (Accessed September 15, 2023).
- Keller, L., and Rauch, F. (2021). "SDG\_04 Hochwertige Bildung", in Optionenbericht: Österreichs Handlungsoptionen für die Um-setzung der UN-Agenda 2030 für eine lebenswerte Zukunft. UniNETZ – Universitäten und Nachhaltige Entwicklungsziele., ed. UniNetz – Allianz Nachhaltige Universitäten in Österreich (Vienna: UniNetz). Available at: <u>https://www.uninetz.at/optionenbericht\_downloads/SDG\_04.pdf</u> (Accessed September 13, 2023).
- Keller, L., Riede, M., Link, S., Hüfner, K., and Stötter, J. (2022). Can Education Save Money, Energy, and the Climate?—Assessing the Potential Impacts of Climate Change Education on Energy Literacy and Energy Consumption in the Light of the EU Energy Efficiency Directive and the Austrian Energy Efficiency Act. *Energies* 15, 1118. <u>https://doi.org/10.3390/en15031118</u>
- Kerschner, C., Wächter, P., Nierling, L., and Ehlers, M.-H. (2018). Degrowth and Technology: Towards feasible, viable, appropriate and convivial imaginaries. *Journal of Cleaner Production* 197, 1619–1636. <u>https://doi.org/10.1016/j.jclepro.2018.07.147</u>
- Kettner, C., and Kletzan-Slamanig, D. (2018). Climate Policy Integration on the National and Regional Level: A Case Study for Austria and Styria. *International Journal of Energy Economics* and Policy 8, 259–269.
- Kettner, C., Leoni, T., Köberl, J., Kortschak, D., Kirchner, M., Sommer, M., et al. (2024). Modelling the economy-wide effects of unilateral CO2 pricing under different revenue recycling schemes in Austria – Searching for a triple dividend. *Energy Economics* 137, 107783. <u>https://doi.org/10.1016/j.eneco.2024.107783</u>

- Kettner, C., Schratzenstaller, M., and Sutrich, A. (2023). Österreichs Anti-Teuerungsmaßnahmen 2022 bis 2026. Treffsicherheit und ökologische Aspekte. Vienna: Austrian Institute of Economic Research (WIFO). Available at: <u>https://www.wifo.ac.at/</u> publication/48888/ (Accessed June 14, 2023).
- Kettner-Marx, C., Loretz, S., and Schratzenstaller, M. (2021). Steuerreform 2022/2024 – Maßnahmenüberblick und erste Einschätzung. Wien: WIFO. Available at: <u>https://www.wifo.ac.at/ jart/prj3/wifo/resources/person\_dokument/person\_dokument.</u> jart?publikationsid=69189&mime\_type=application/pdf (Accessed September 8, 2023).
- Keyßer, L. T., and Lenzen, M. (2021). 1.5 °C degrowth scenarios suggest the need for new mitigation pathways. *Nature Communications* 12, 2676. <u>https://doi.org/10.1038/s41467-021-22884-9</u>
- Kikstra, J., Daioglou, V., Min, J., Sferra, F., Soergel, B., Kriegler, E., et al. (2024a). Closing decent living gaps in energy and emissions scenarios: introducing DESIRE [preprint v1.0]. <u>https://doi.org/10.13140/RG.2.2.27951.14241</u>
- Kikstra, J., Li, M., Brockway, P., Hickel, J., Keyßer, L., Malik, A., et al. (2023). Towards degrowth in integrated assessment models [preprint v1.1]. https://doi.org/10.13140/RG.2.2.20355.68647
- Kikstra, J. S., Li, M., Brockway, P. E., Hickel, J., Keysser, L., Malik, A., et al. (2024b). Downscaling down under: towards degrowth in integrated assessment models. *Economic Systems Research* 36, 576–606. <u>https://doi.org/10.1080/09535314.2023.2301443</u>
- Kikstra, J. S., Mastrucci, A., Min, J., Riahi, K., and Rao, N. D. (2021). Decent living gaps and energy needs around the world. *Environ. Res. Lett.* 16, 095006. <u>https://doi.org/10.1088/1748-9326/ac1c27</u>
- King, L., Savin, I., and Drews, S. (2023). Shades of Green Growth Scepticism Among Climate Policy Researchers. SSRN Journal. <u>https://doi.org/10.2139/ssrn.4516544</u>
- Kirchengast, G., Kromp-Kolb, H., Steininger, K., Stagl, S., Kirchner, M., Ambach, C., et al. (2019a). Referenzplan als Grundlage für einen wissenschaftlich fundierten und mit den Pariser Klimazielen in Einklang stehenden Nationalen Energie- und Klimaplan für Österreich (Ref-NEKP) – Gesamtband. Wien-Graz, Wien Österreich: CCCA, Verlag der ÖAW. Available at: <u>https:// ccca.ac.at/fileadmin/00\_DokumenteHauptmenue/03\_Aktivitaeten/UniNEtZ\_SDG13/RefNEKP/Ref-NEKP\_Gesamtband\_ Nov2019\_VerlOeAW.pdf</u> (Accessed September 7, 2023).
- Kirchengast, G., Kromp-Kolb, H., and Steininger, K. (2019b). Ref-NEKP Team Ergebnisinformation – Wissenschaftliche Bewertung Parteipositionen. Wien. Available at: <u>https://ccca.ac.at/fileadmin/00\_DokumenteHauptmenue/02\_Klimawissen/Papiere/RefNEKP-Parteien\_Bewertungstabellen-und-Backinfos\_25 Sep2019.pdf</u>
- Kirchmair, L., and Krempelmeier, S. (2023). Das Klimaschutzprinzip im BVG Nachhaltigkeit: Ein schlafender Riese: Zur Verfassungswidrigkeit klimaschädlicher Gesetze in Österreich. *Journal für Rechtspolitik* 31, 74–92. <u>https://doi.org/10.33196/jrp202301007401</u>
- Kirchner, M., Sommer, M., Kratena, K., Kletzan-Slamanig, D., and Kettner-Marx, C. (2019). CO<sub>2</sub> taxes, equity and the double dividend – Macroeconomic model simulations for Austria. *Energy Policy* 126, 295–314. <u>https://doi.org/10.1016/j.enpol.2018.11.030</u>
- Kivimaa, P., and Kern, F. (2016). Creative destruction or mere niche support? Innovation policy mixes for sustainability transitions. *Research Policy* 45, 205–217. <u>https://doi.org/10.1016/j. respol.2015.09.008</u>

- Klenert, D., Mattauch, L., Combet, E., Edenhofer, O., Hepburn, C., Rafaty, R., et al. (2018). Making carbon pricing work for citizens. *Nature Clim Change* 8, 669–677. <u>https://doi.org/10.1038/</u> <u>s41558-018-0201-2</u>
- Kletzan-Slamanig, D., Köppl, A., Pitlik, H., and Schratzenstaller, M. (2023). Der Finanzausgleich als Hebel zur Umsetzung der österreichischen Klimaziele. Vienna: Austrian Institute of Economic Research (WIFO). Available at: <u>https://www.wifo.ac.at/ jart/prj3/wifo/resources/person\_dokument/person\_dokument. jart?publikationsid=70785&mime\_type=application/pdf</u> (Accessed September 15, 2023).
- Kletzan-Slamanig, D., Köppl, A., Sinabell, F., Kirchmayr, S., Müller, S., Rimböck, A., et al. (2022). Analyse klimakontraproduktiver Subventionen in Österreich. Vienna: Österreichisches Institut für Wirtschaftsforschung (WIFO). Available at: <u>https://www. wifo.ac.at/jart/prj3/wifo/resources/person\_dokument/person\_ dokument.jart?publikationsid=69687&mime\_type=application/pdf</u> (Accessed September 15, 2023).
- Kletzan-Slamanig, D., and Schratzenstaller, M. (2022). "Ökologisierung des Abgabensystems im Föderalstaat", in *Handbuch der* österreichischen Finanzverfassung, eds. P. Bußjäger and M. Eller (Wien, Hamburg: new academic press), 221–244. Available at: <u>https://www.wifo.ac.at/publication/pid/25091936</u>
- Klien, M., and Berger, C. (2023). WIFO-Studie: "Die Rolle des öffentlichen Vergabewesens für eine klimaneutrale Produktions- und Lebensweise." Vienna: WIFO. Available at: <u>https://</u> www.wifo.ac.at/publication/257000/ (Accessed May 8, 2024).
- Klien, M., Böheim, M., and Streicher, G. (2023). Die Rolle des öffentlichen Vergabewesens für eine klimaneutrale Produktions- und Lebensweise. Vienna: WIFO. Available at: <u>https://</u> <u>www.wifo.ac.at/publication/48286/</u> (Accessed August 22, 2024).
- Klintman, M. (2016). A Review of Public Policies relating to the Use of Environmental Labelling and Information Schemes (ELIS). <u>https://doi.org/10.1787/5jm0p34bk7hb-en</u>
- Klusak, P., Agarwala, M., Burke, M., Kraemer, M., and Mohaddes, K. (2023). Rising Temperatures, Falling Ratings: The Effect of Climate Change on Sovereign Creditworthiness. *Management Science*. <u>https://doi.org/10.1287/mnsc.2023.4869</u>
- Knittel, N., and Bednar-Friedl, B. (2016). The role of public authorities in Austria's national adaptation strategy. PACINAS Working Paper #01. Graz: University of Graz. Available at: <a href="http://an-passung.ccca.at/pacinas/wp-content/uploads/sites/3/2015/01/">http://an-passung.ccca.at/pacinas/wp-content/uploads/sites/3/2015/01/</a>
   PublicAuthoritiesintheNAS\_Knittel\_BednarFriedl.pdf (Accessed June 6, 2023).
- Koch, M. (2022). Social Policy Without Growth: Moving Towards Sustainable Welfare States. Social Policy and Society 21, 447– 459. <u>https://doi.org/10.1017/S1474746421000361</u>
- Koch, N., Naumann, L., Pretis, F., Ritter, N., and Schwarz, M. (2022). Attributing agnostically detected large reductions in road CO2 emissions to policy mixes. *Nat Energy* 7, 844–853. <u>https://doi.org/10.1038/s41560-022-01095-6</u>
- Kohl, K., Hopkins, C., Barth, M., Michelsen, G., Dlouhá, J., Razak, D. A., et al. (2022). A whole-institution approach towards sustainability: a crucial aspect of higher education's individual and collective engagement with the SDGs and beyond. *International Journal of Sustainability in Higher Education* 23, 218–236. https://doi.org/10.1108/IJSHE-10-2020-0398
- Köppl, A., Schratzenstaller, M., and Schleicher, S. (2023a). Der Beitrag des österreichischen Abgabensystems zur sozio-ökologischen Transformation. WIFO Research Briefs 3. Available at: https://www.wifo.ac.at/jart/prj3/wifo/resources/person\_doku-

ment/person\_dokument.jart?publikationsid=70654&mime\_ type=application/pdf (Accessed September 13, 2023).

- Köppl, A., Schleicher, S., and Schratzenstaller, M. (2023b). Klima- und umweltrelevante öffentliche Ausgaben in Österreich. Vienna: Österreichisches Institut für Wirtschaftsforschung (WIFO). Available at: <u>https://www.wifo.ac.at/jart/prj3/wifo/ resources/person\_dokument/person\_dokument.jart?publikationsid=70536&mime\_type=application/pdf</u> (Accessed September 15, 2023).
- Köppl, A., and Schratzenstaller, M. (2022). Macroeconomic Effects of Green Recovery Programmes Conceptual Framing and a Review of the Empirical Literature. WIFO Working Papers 646. Available at: <u>https://www.wifo.ac.at/jart/prj3/wifo/resources/ person\_dokument/person\_dokument.jart?publikationsid= 69620&mime\_type=application/pdf</u> (Accessed September 15, 2023).
- Köppl, A., and Schratzenstaller, M. (2023). Carbon taxation: A review of the empirical literature. *Journal of Economic Surveys* 37, 1353–1388. <u>https://doi.org/10.1111/joes.12531</u>
- Köppl, A., and Schratzenstaller, M. (2024a). Macroeconomic effects of green recovery programs. *Eurasian Econ Rev* 14, 61–86. https://doi.org/10.1007/s40822-023-00250-y
- Köppl, A., and Schratzenstaller, M. (2024b). Budgetäre Kosten und Risiken durch klimapolitisches Nichthandeln und Klimarisiken. Vienna: WIFO. Available at: <u>https://www.wifo.ac.at/publication/49048/</u> (Accessed May 8, 2024).
- Köppl, A., Schratzenstaller, M., and Schleicher, S. (2021). CO2 -Bepreisung in der Steuerreform 2022/2024. WIFO Research Briefs 13. Available at: <u>https://www.wifo.ac.at/jart/prj3/wifo/resources/person\_dokument/person\_dokument.jart?publikations-id=69168&mime\_type=application/pdf</u> (Accessed September 13, 2023).
- Kozicka, M., Havlík, P., Valin, H., Wollenberg, E., Deppermann, A., Leclère, D., et al. (2023). Feeding climate and biodiversity goals with novel plant-based meat and milk alternatives. *Nat Commun* 14, 5316. <u>https://doi.org/10.1038/s41467-023-40899-2</u>
- Kranz, J., Schwichow, M., Breitenmoser, P., and Niebert, K. (2022). The (Un)political Perspective on Climate Change in Education—A Systematic Review. *Sustainability* 14, 4194. <u>https://doi. org/10.3390/su14074194</u>
- Krasznai, R. (2023). Von der Straße auf die Schiene der Abfalltransport als Pilotprojekt der Schwerverkehrslenkung. Recht der Umwelt (RdU) 2a, 23.
- Kratena, K., and Schleicher, S. (1999). Impact of Carbon Dioxide Emissions Reduction on the Austrian Economy. *Economic Sys*tems Research, Taylor & Francis Journals 11, 245–261. <u>https://</u> doi.org/10.1080/09535319900000017
- Kraxner, F., Zollitsch, W., Kottusch, C., Bruckman, V. J., Glatzel, S., Hood-Nowotny, R., et al. (2024). "Kapitel 5. Mitigation des Klimawandels", in APCC Special Report: Landnutzung und Klimawandel in Österreich (APCC SR Land), eds. R. Jandl, U. Tappeiner, C. B. Foldal, and K.-H. Erb (Berlin, Heidelberg: Springer Spektrum), 275–338. <u>https://doi.org/10.1007/978-3-662-67864-0\_7</u>
- Kreibiehl, S., Yong Jung, T., Battiston, S., Carvajal, P. E., Clapp, C., Dasgupta, D., et al. (2022). Investment and finance. Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. <u>https://doi.org/10.1017/9781009157926.017</u>

- Krenmayr, N., Maneka, D., Pichler, M., Brand, U., Wissen, M., and Högelsberger, H. (2020). Industriepolitik zwischen Modernisierung und Transformation – Das Beispiel der österreichischen Autoindustrie. *Kurswechsel*, 62–74. Available at: <u>http://www. beigewum.at/wp-content/uploads/KuWe\_4\_20\_Krenmayr-ManekaPichlerBrandHoegelsbergerWissen.pdf</u>
- Krogstrup, S., and Oman, W. (2019). Macroeconomic and Financial Policies for Climate Change Mitigation: A Review of the Literature. *IMF Working Papers* 2019, 58. <u>https://doi. org/10.5089/9781513511955.001</u>
- Krüger, T., and Pellicer-Sifres, V. (2020). From innovations to exnovations. Conflicts, (De-)Politicization processes, and power relations are key in analysing the ecological crisis. *Innovation: The European Journal of Social Science Research* 33, 115–123. <u>https://doi.org/10.1080/13511610.2020.1733936</u>
- Krutzler, T., Anderl, M., Böhmer, S., Freisinger, E., Gössl, M., Gugele, B., et al. (2024). Energie und Treibhausgasszenarien zum Nationalen Energie- und Klimaplan 2024. WEM 2024 und WAM 2024 mit Zeitreihen von 2020 bis 2050. Wien: Federal Environment Agency. Available at: <u>https://www.umweltbundesamt.at/studien-reports/publikationsdetail?pub\_id=2572&cHash=af7cf 17493bedac0862bbdff99a9485c</u> (Accessed February 6, 2025).
- Krutzler, T., Schindler, I., and Wasserbaur, R. (2023). Energie- und Treibhausgas- Szenario Transition 2040. Bericht für das Szenario Transition 2040 mit einer Zeitreihe von 2020 bis 2050. Vienna: Environment Agency Austria (EAA). Available at: <u>https:// www.umweltbundesamt.at/fileadmin/site/publikationen/rep</u> 0880.pdf (Accessed May 8, 2024).
- Kubat, A., and Tokić, A. (2024). "Sustainability and Competition Law in Austria", in Sustainability Objectives in Competition and Intellectual Property Law. LIDC Contributions on Antitrust Law, Intellectual Property and Unfair Competition, eds. P. Këllezi, P. Kobel, and B. Kilpatrick (Cham: Springer Nature Switzerland), 27–50. https://doi.org/10.1007/978-3-031-44869-0\_2
- Kubisch, S., Parth, S., Deisenrieder, V., Oberauer, K., Stötter, J., and Keller, L. (2020). From Transdisciplinary Research to Transdisciplinary Education—The Role of Schools in Contributing to Community Well-Being and Sustainable Development. *Sustainability* 13, 306. <u>https://doi.org/10.3390/su13010306</u>
- Kuthe, A., Keller, L., Körfgen, A., Stötter, H., Oberrauch, A., and Höferl, K.-M. (2019). How many young generations are there? – A typology of teenagers' climate change awareness in Germany and Austria. *The Journal of Environmental Education* 50, 172– 182. <u>https://doi.org/10.1080/00958964.2019.1598927</u>
- Kyriazi, A., and Miró, J. (2023). Towards a socially fair green transition in the EU? An analysis of the Just Transition Fund using the Multiple Streams Framework. *Comp Eur Polit* 21, 112–132. https://doi.org/10.1057/s41295-022-00304-6
- Lachmayer, K., and Müller, C. (2023). Sozio-ökologisches Transformationsrecht: Öffentlich-rechtliche Überlegungen zu einer öko-sozialen Transformation in Österreich. *juridikum* 2023, 316–325. <u>https://doi.org/10.33196/juridikum202303031601</u>
- Lamb, W. F., Mattioli, G., Levi, S., Roberts, J. T., Capstick, S., Creutzig, F., et al. (2020a). Discourses of climate delay. *Global Sustainability* 3, e17. <u>https://doi.org/10.1017/sus.2020.13</u>
- Lamb, W. F., Antal, M., Bohnenberger, K., Brand-Correa, L. I., Müller-Hansen, F., Jakob, M., et al. (2020b). What are the social outcomes of climate policies? A systematic map and review of the ex-post literature. *Environ. Res. Lett.* 15, 113006. <u>https://doi. org/10.1088/1748-9326/abc11f</u>

- Lamb, W. F., and Steinberger, J. K. (2017). Human well-being and climate change mitigation. WIREs Climate Change 8, e485. https://doi.org/10.1002/wcc.485
- Lambert, J. L., and Bleicher, R. E. (2013). Climate Change in the Preservice Teacher's Mind. *Journal of Science Teacher Education* 24, 999–1022. <u>https://doi.org/10.1007/s10972-013-9344-1</u>
- Latouche, S. (2009). *Farewell to growth*. Cambridge, UK: Polity Press.
- Lauber, V. (1997). "Austria: a latecomer which became a pioneer", in *European environmental policy. The pioneers.*, eds. S. M. Andersen and D. Liefferink (Manchester and New York: Manchester University Press), 81–118. Available at: <u>https://books.google.at/books?hl=de&lr=&id=m3q7AAAIAAJ&oi=fnd&pg=</u> <u>PR8&dq=austria+discursive+production+%22environmental+</u> policy%22&ots=vlecdi9oCV&sig=0EC1z\_HjlvIHbOpGT3g QG2hWpLo#v=onepage&q&f=false
- Lee, T. M., Markowitz, E. M., Howe, P. D., Ko, C.-Y., and Leiserowitz, A. A. (2015). Predictors of public climate change awareness and risk perception around the world. *Nature Clim Change* 5, 1014–1020. https://doi.org/10.1038/nclimate2728
- Leiringer, R., and Cardellino, P. (2011). Schools for the twenty-first century: School design and educational transformation. *British Educational Research Journal* 37, 915–934. <u>http://dx.doi.org/10.</u> <u>1080/01411926.2010.508512</u>
- Leitner, M., Babcicky, P., Schinko, T., and Glas, N. (2020). The status of climate risk management in Austria. Assessing the governance landscape and proposing ways forward for comprehensively managing flood and drought risk. *Climate Risk Management* 30, 100246. <u>https://doi.org/10.1016/j.crm.2020.100246</u>
- Lemken, D., Zühlsdorf, A., and Spiller, A. (2021). Improving Consumers' Understanding and Use of Carbon Footprint Labels on Food: Proposal for a Climate Score Label. *EuroChoices* 20, 23–29. <u>https://doi.org/10.1111/1746-692X.12321</u>
- Lenzen, M., Keyβer, L., and Hickel, J. (2022). Degrowth scenarios for emissions neutrality. *Nat Food* 3, 308–309. <u>https://doi.</u> org/10.1038/s43016-022-00516-9
- Leoni, D., Jackson, A., and Jackson, T. (2023). Post Growth and the North-South Divide: a post-Keynesian stock-flow consistent analysis. CUSP Working Paper No. 38. Guildford: Centre for the Understanding of Sustainable Prosperity.
- Lesk, C., Csala, D., Hasse, R., Sgouridis, S., Levesque, A., Mach, K. J., et al. (2022). Mitigation and adaptation emissions embedded in the broader climate transition. *Proceedings of the National Academy of Sciences* 119, e2123486119. <u>https://doi.org/10.1073/</u> pnas.2123486119
- Lessenich, S. (2019). Living Well at Others' Expense: The Hidden Costs of Western Prosperity. Medford, MA: Polity Press.
- Leve, A.-K., Michel, H., and Harms, U. (2023). Implementing climate literacy in schools — what to teach our teachers? *Climatic Change* 176, 134. <u>https://doi.org/10.1007/s10584-023-03607-z</u>
- LGBl. Nr. 46/2005 (2005). Gesetz über die Neuregelung der Elektrizitätswirtschaft (Wiener Elektrizitätswirtschaftsgesetz 2005 – WElWG 2005). Available at: <u>https://www.ris.bka.gv.at/GeltendeFassung.wxe?Abfrage=LrW&Gesetzesnummer=20000048</u> (Accessed October 22, 2024).
- LGBl. Nr. 49/2019 (2019). Gesetz vom 4. Juli 2019 über die Raumplanung im Burgenland 2019 (Burgenländisches Raumplanungsgesetz 2019 – Bgld. RPG 2019). Available at: <u>https://www. ris.bka.gv.at/GeltendeFassung.wxe?Abfrage=LrBgld&Gesetzesnummer=20001224</u> (Accessed October 22, 2024).

- Li, C. J., Monroe, M. C., Oxarart, A., and Ritchie, T. (2021). Building teachers' self-efficacy in teaching about climate change through educative curriculum and professional development. *Applied Environmental Education & Communication* 20, 34–48. <u>https://doi.org/10.1080/1533015X.2019.1617806</u>
- Li, M., Keyßer, L., Kikstra, J. S., Hickel, J., Brockway, P. E., Dai, N., et al. (2023). Integrated assessment modelling of degrowth scenarios for Australia. *Economic Systems Research* 36, 545–575. <u>https://doi.org/10.1080/09535314.2023.2245544</u>
- Liebhaber, N., Ramjan, C., Frick, M., Mannion, G., and Keller, L. (2023). Transformative climate change education and the school caretaker: a more-than-human analysis with young people. *Environmental Education Research* 30, 2057–2072. <u>https:// doi.org/10.1080/13504622.2023.2286936</u>
- Liebig, S. (2019). "Arbeitszeitverkürzung für eine nachhaltigere Wirtschaft?", in Große Transformation? Zur Zukunft moderner Gesellschaften: Sonderband des Berliner Journals für Soziologie, eds. K. Dörre, H. Rosa, K. Becker, S. Bose, and B. Seyd (Wiesbaden: Springer VS), 211–228. <u>https://doi.org/10.1007/978-3-658-25947-1\_11</u>
- Lörcher, I. (2019). "Wie kommt der Klimawandel in die Köpfe?", in Klimawandel im Kopf: Studien zur Wirkung, Aneignung und Online-Kommunikation, eds. I. Neverla, M. Taddicken, I. Lörcher, and I. Hoppe (Wiesbaden: Springer VS), 53–76. <u>https://doi.org/10.1007/978-3-658-22145-4\_3</u>
- Lutz, W., Muttarak, R., and Striessnig, E. (2014). Universal education is key to enhanced climate adaptation. *Science* 346, 1061– 1062. <u>https://doi.org/10.1126/science.1257975</u>
- Mach, K. J., and Siders, A. R. (2021). Reframing strategic, managed retreat for transformative climate adaptation. *Science* 372, 1294–1299. <u>https://doi.org/10.1126/science.abh1894</u>
- Madner, V., and Schulev-Steindl, E. (2017). Dritte Piste Klimaschutz als Willkür? ZÖR 3, 589.
- Magin, M., and Stark, B. (2015). Explaining National Differences of Tabloidisation Between Germany and Austria: Structure, conduct and performance. *Journalism Studies* 16, 577–595. <u>https:// doi.org/10.1080/1461670X.2014.921398</u>
- Malm, A. (2013). The Origins of Fossil Capital: From Water to Steam in the British Cotton Industry. *Hist Mater* 21, 15–68. <u>https://doi.org/10.1163/1569206X-12341279</u>
- Malm, A. (2021). White skin, black fuel: on the danger of fossil fascism. London; New York: Verso.
- Maniates, M. (2010). "Editing Out Unsustainable Behavior", in State of the World 2010: Transforming Cultures from Consumerism to Sustainability, ed. Worldwatch Institute (London: Routledge), 119–126.
- Markkanen, S., and Anger-Kraavi, A. (2019). Social impacts of climate change mitigation policies and their implications for inequality. *Climate Policy* 19, 827–844. <u>https://doi.org/10.1080/1</u> <u>4693062.2019.1596873</u>
- Markusson, N., McLaren, D., and Tyfield, D. (2018). Towards a cultural political economy of mitigation deterrence by negative emissions technologies (NETs). *Global Sustainability* 1, e10. https://doi.org/10.1017/sus.2018.10
- Marquardt, J. (2017). Conceptualizing power in multi-level climate governance. *Journal of Cleaner Production* 154, 167–175. <u>https://doi.org/10.1016/j.jclepro.2017.03.176</u>
- Martens, K., Nagel, A. K., Windzio, M., and Weymann, A. eds. (2010). *Transformation of education policy*. London: Palgrave Macmillan.

- Max-Neef, M. (1991). Human Scale Development: Conception, Application and Further Reflections. New York: Zed Books Ltd.
- Mayer, J., Dugan, A., Bachner, G., and Steininger, K. W. (2021). Is carbon pricing regressive? Insights from a recursive-dynamic CGE analysis with heterogeneous households for Austria. *Energy Economics* 104, 105661. <u>https://doi.org/10.1016/j.eneco.2021.105661</u>
- Mayr, S., Hollaus, B., and Madner, V. (2021). Palm oil, the RED II and WTO law: EU sustainable biofuel policy tangled up in green? *Rev Euro Comp Intl Enviro* 30, 233–248. <u>https://doi.org/10.1111/reel.12386</u>
- Mayrhofer, M., and Ammer, M. (2022). Climate mobility to Europe: The case of disaster displacement in Austrian asylum procedures. *Frontiers in Climate* 4. <u>https://doi.org/10.3389/fclim.2022.990558</u>
- Mazzucato, M. (2014). *The Entrepreneurial State: Debunking Public* vs. *Private Sector Myths.*, Rev. ed. London: Anthem Press.
- McCauley, D., and Heffron, R. (2018). Just transition: Integrating climate, energy and environmental justice. *Energy Policy* 119, 1–7. <u>https://doi.org/10.1016/j.enpol.2018.04.014</u>
- McCollum, D. L., Zhou, W., Bertram, C., De Boer, H.-S., Bosetti, V., Busch, S., et al. (2018). Energy investment needs for fulfilling the Paris Agreement and achieving the Sustainable Development Goals. *Nat Energy* 3, 589–599. <u>https://doi.org/10.1038/ s41560-018-0179-z</u>
- McKnight, D. (2010a). A change in the climate? The journalism of opinion at News Corporation. *Journalism* 11, 693–706. <u>https://doi.org/10.1177/1464884910379704</u>
- McKnight, D. (2010b). Rupert Murdoch's News Corporation: A Media Institution with A Mission. *Historical Journal of Film*, *Radio and Television* 30, 303–316. <u>https://doi.org/10.1080/014</u> <u>39685.2010.505021</u>
- McLeman, R. (2018). Thresholds in climate migration. *Popul Environ* 39, 319–338. <u>https://doi.org/10.1007/s11111-017-0290-2</u>
- Meadows, D. H., Meadows, D. L., and Randers, J. (1992). *Beyond the limits: global collapse or a sustainable future*. Earthscan Publications Ltd.
- Meadows, D. H., Meadows, D. L., Randers, J., and Behrens, W. H. (1972). *The Limits To Growth*. New York, United States of America: Universe Books.
- Mechler, R., Bouwer, L. M., Linnerooth-Bayer, J., Hochrainer-Stigler, S., Aerts, J. C. J. H., Surminski, S., et al. (2014). Managing unnatural disaster risk from climate extremes. *Nature Clim Change* 4, 235–237. <u>https://doi.org/10.1038/nclimate2137</u>
- Mechler, R., Singh, C., Ebi, K., Djalante, R., Thomas, A., James, R., et al. (2020). Loss and Damage and limits to adaptation: recent IPCC insights and implications for climate science and policy. *Sustain Sci* 15, 1245–1251. <u>https://doi.org/10.1007/s11625-020-00807-9</u>
- Meckling, J., Kelsey, N., Biber, E., and Zysman, J. (2015). Winning coalitions for climate policy. *Science* 349, 1170–1171. <u>https:// doi.org/10.1126/science.aab1336</u>
- Meckling, J., and Nahm, J. (2022). Strategic State Capacity: How States Counter Opposition to Climate Policy. *Comparative Political Studies* 55, 493–523. <u>https://doi.org/10.1177/</u> 00104140211024308
- Meinhart, B., Gabelberger, F., Sinabell, F., and Streicher, G. (2022). Transformation und "Just Transition" in Österreich. Wien: Österreichisches Institut für Wirtschaftsforschung (WIFO). Available at: <u>https://www.wifo.ac.at/jart/prj3/wifo/resources/per-</u>

son\_dokument/person\_dokument.jart?publikationsid=68029 &mime\_type=application/pdf

- Meis-Harris, J., Klemm, C., Kaufman, S., Curtis, J., Borg, K., and Bragge, P. (2021). What is the role of eco-labels for a circular economy? A rapid review of the literature. *Journal of Cleaner Production* 306, 127134. <u>https://doi.org/10.1016/j.jclepro.2021.</u> <u>127134</u>
- Melidis, M., and Russel, D. J. (2020). Environmental policy implementation during the economic crisis: an analysis of European member state 'leader-laggard' dynamics. *Journal of Environmental Policy & Planning* 22, 198–210. <u>https://doi.org/10.1080/152</u> <u>3908X.2020.1719051</u>
- Merli, F. (2017). Ein seltsamer Fall von Willkür: Die VfGH-Entscheidung zur dritten Piste des Flughafens Wien. *wbl* 12, 682.
- Metcalf, G. E., and Weisbach, D. (2009). The Design of a Carbon Tax. Harv. Envtl. L. Rev. 33, 499. <u>https://dx.doi.org/10.2139/</u> <u>ssrn.1324854</u>
- Michaelis, C., and Berding, F. eds. (2021). Berufsbildung für nachhaltige Entwicklung: Umsetzungsbarrieren und interdisziplinäre Forschungsfragen. Bielefeld: wbv Media GmbH & Company KG.
- Michaelowa, A. (1998). Climate policy and interest Groups—A Public choice analysis. *Intereconomics* 33, 251–259. <u>https://doi.org/10.1007/BF02929886</u>
- Miess, M., Böhmer, S., Frei, E., Glas, N., Krutzler, T., Lichtblau, G., et al. (2022). Analysis of the investment potential until 2030 on the pathway to climate neutrality (Analyse des Investitionspotenzials bis 2030 auf dem Weg zur Klimaneutralität). Vienna. Available at: Data in: <u>https://www.wko.at/oe/bank-versicherung/folien-pk-investitionskosten-transformation.pdf</u> (Accessed May 16, 2024).
- Millward-Hopkins, J. (2022). Inequality can double the energy required to secure universal decent living. *Nat Commun* 13, 5028. <u>https://doi.org/10.1038/s41467-022-32729-8</u>
- Millward-Hopkins, J., Steinberger, J. K., Rao, N. D., and Oswald, Y. (2020). Providing decent living with minimum energy: A global scenario. *Global Environmental Change* 65, 102168. <u>https://doi.org/10.1016/j.gloenvcha.2020.102168</u>
- Missirian, A., and Schlenker, W. (2017). Asylum applications respond to temperature fluctuations. *Science* 358, 1610–1614. <u>https://doi.org/10.1126/science.aa00432</u>
- Mitchell, T. (2023). *Carbon democracy: political power in the age of oil.*, Revised paperback edition. London New York: Verso.
- Mitterer, K. (2024). "Klimaschutz und Klimawandelanpassung im Finanzausgleich", in *Finanzausgleich 2024: ein Handbuch: mit Kommentar zum FAG 2024*, eds. H. Bauer, P. Biwald, and K. Mitterer (Wien: NWV), 501–516.
- Mock, M. (2023). Making and breaking links: the transformative potential of shared mobility from a practice theories perspective. *Mobilities* 18, 374–390. <u>https://doi.org/10.1080/17450101.2022.</u> 2142066
- Moesker, K., and Pesch, U. (2022). The just transition fund Did the European Union learn from Europe's past transition experiences? *Energy Research & Social Science* 91, 102750. <u>https://doi.org/10.1016/j.erss.2022.102750</u>
- Möller, A., Kranz, J., Pürstinger, A., and Winter, V. (2021). "Professionsverantwortung in der Klimakrise: Klimawandel unterrichten. Befähigung Lehramtsstudierender zur Klimabildung als wichtiger Beitrag zum Erreichen der SDGs", in *Lehrkräftebildung neu gedacht*, eds. M. Kubsch, S. Sorge, J. Arnold, and N. Graulich (Münster Germany: Waxmann), 208–217.

- Morgan, T. (2020). "Challenges and Potentials for Socio-Ecological Transformation: Considering Structural Aspects of Change", in *Ireland and the Climate Crisis*, eds. D. Robbins, D. Torney, and P. Brereton (Cham: Palgrave Macmillan), 149–168. <u>https://doi.org/10.1007/978-3-030-47587-1\_9</u>
- Moshiri, S., and Aliyev, K. (2017). Rebound effect of efficiency improvement in passenger cars on gasoline consumption in Canada. *Ecological Economics* 131, 330–341. <u>https://doi.org/10.1016/j.ecolecon.2016.09.018</u>
- Moyer, J. D. (2023). Modeling transformational policy pathways on low growth and negative growth scenarios to assess impacts on socioeconomic development and carbon emissions. *Sci Rep* 13, 15996. <u>https://doi.org/10.1038/s41598-023-42782-y</u>
- Muller, N. Z., Mendelsohn, R., and Nordhaus, W. (2011). Environmental Accounting for Pollution in the United States Economy. *American Economic Review* 101, 1649–1675. <u>https://doi.org/10. 1257/aer.101.5.1649</u>
- Mulligan, M., Burke, S., and Douglas, C. (2014). "Environmental Change and Migration Between Europe and its Neighbours", in *People on the Move in a Changing Climate. Global Migration Issues, vol 2.*, eds. E. Piguet and F. Laczko (Dordrecht: Springer).
- Namdar, B. (2018). Teaching global climate change to pre-service middle school teachers through inquiry activities. *Research in Science & Technological Education* 36, 440–462. <u>https://doi.org/</u> 10.1080/02635143.2017.1420643
- Narodoslawsky, B. (2020). Inside Fridays for Future: die faszinierende Geschichte der Klimabewegung in Österreich. Wien: Falter Verlag.
- Nash, S. L. (2023). "The View from the Fortress: European Governance Perspectives on Climate Change and Migration", in *Climate Migration: Critical Perspectives for Law, Policy, and Research*, eds. C. Nicholson and B. Mayer (Oxford: Hart Publishing). <u>https://doi.org/10.5040/9781509961771.ch-006</u>
- Nash, S. L., and Steurer, R. (2019). Taking stock of Climate Change Acts in Europe: living policy processes or symbolic gestures? *Climate Policy* 19, 1052–1065. <u>https://doi.org/10.1080/146930</u> 62.2019.1623164
- Nash, S. L., and Steurer, R. (2021). Climate Change Acts in Scotland, Austria, Denmark and Sweden: the role of discourse and deliberation. *Climate Policy* 21, 1120–1131. <u>https://doi.org/10. 1080/14693062.2021.1962235</u>
- Nash, S., and Steurer, R. (2023). "Klimapolitik", in *Das Politische System Österreichs*, eds. K. Praprotnik and F. Perlot (Wien/Köln: Böhlau Verlag), 495–521.
- Nation, M. T., and Feldman, A. (2021). Environmental Education in the Secondary Science Classroom: How Teachers' Beliefs Influence Their Instruction of Climate Change. *Journal of Science Teacher Education* 32, 481–499. <u>https://doi.org/10.1080/10465</u> 60x.2020.1854968
- Nawrotzki, R. J., and DeWaard, J. (2018). Putting trapped populations into place: climate change and inter-district migration flows in Zambia. *Reg Environ Change* 18, 533–546. <u>https://doi.org/10.1007/s10113-017-1224-3</u>
- Neier, T., Kreinin, H., Gerold, S., Heyne, S., Laa, E., and Bohnenberger, K. (2024). Navigating labor-market transitions: an eco-social policy toolbox for public employment services. Sustainability: Science, Practice and Policy 20, 2386799. <u>https://doi.org/10.1080/15487733.2024.2386799</u>
- Neverlaa, I., and Hoppe, I. (2023). Klimawandel und Biodiversität im deutschen Fernsehen. Was zeigt das Fernsehen? Was wollen die ZuschauerInnen? Available at: <u>https://www.malisastiftung.</u>

org/studien/klimawandel-und-biodiversitt-was-zeigt-das-fernsehen-was-wollen-die-zuschauerinnen (Accessed October 17, 2023).

- Newell, P. (2015). "The Politics Of Green Transformations In Capitalism", in *The Politics of Green Transformations*, eds. I. Scoones, M. Leach, and P. Newell (London: Routledge).
- Newell, P. (2019). *Trasformismo* or transformation? The global political economy of energy transitions. *Review of International Political Economy* 26, 25–48. <u>https://doi.org/10.1080/09692</u> 290.2018.1511448
- Newell, P. J., Geels, F. W., and Sovacool, B. K. (2022). Navigating tensions between rapid and just low-carbon transitions. *Environ. Res. Lett.* 17, 041006. <u>https://doi.org/10.1088/1748-9326/ac622a</u>
- Niebert, K., and Gropengiesser, H. (2013). Understanding and communicating climate change in metaphors. *Environmental Education Research* 19, 282–302. <u>https://doi.org/10.1080/135</u> 04622.2012.690855
- Niedertscheider, M., Haas, W., and Görg, C. (2018). Austrian climate policies and GHG-emissions since 1990: What is the role of climate policy integration? *Environmental Science & Policy* 81, 10–17. https://doi.org/10.1016/j.envsci.2017.12.007
- Nieto, J., Carpintero, Ó., Miguel, L. J., and de Blas, I. (2020). Macroeconomic modelling under energy constraints: Global low carbon transition scenarios. *Energy Policy* 137, 111090. <u>https://doi.org/10.1016/j.enpol.2019.111090</u>
- Niinimäki, K., Peters, G., Dahlbo, H., Perry, P., Rissanen, T., and Gwilt, A. (2020). The environmental price of fast fashion. *Nat Rev Earth Environ* 1, 189–200. <u>https://doi.org/10.1038/s43017-020-0039-9</u>
- Nisbet, M. C. (2009). Communicating Climate Change: Why Frames Matter for Public Engagement. *Environment: Science* and Policy for Sustainable Development 51, 12–23. <u>https://doi.org/10.3200/ENVT.51.2.12-23</u>
- Noonan, E., and Rusu, A. (2022). The future of climate migration. At a Glance. Brussels: European Parliamentary Research Service. Available at: <u>https://www.europarl.europa.eu/thinktank/</u><u>en/document/EPRS\_ATA(2022)729334</u> (Accessed September 13, 2023).
- Nordbeck, R., Steurer, R., and Löschner, L. (2019). The future orientation of Austria's flood policies: from flood control to anticipatory flood risk management. *Journal of Environmental Planning and Management* 62, 1864–1885. <u>https://doi.org/10.1080/ 09640568.2018.1515731</u>
- Nordhaus, W. D., Stavins, R. N., and Weitzman, M. L. (1992). Lethal model 2: The limits to growth revisited. *Brookings papers on economic activity* 1992, 1–59.
- Norström, A. V., Cvitanovic, C., Löf, M. F., West, S., Wyborn, C., Balvanera, P., et al. (2020). Principles for knowledge co-production in sustainability research. *Nat Sustain* 3, 182–190. <u>https:// doi.org/10.1038/s41893-019-0448-2</u>
- Novy, A., Haderer, M., Kubeczko, K., Aigner, E., Bärnthaler, R., Brand, U., et al. (2023a). "Kapitel 2. Perspektiven zur Analyse und Gestaltung von Strukturen klimafreundlichen Lebens", in *APCC Special Report: Strukturen für ein klimafreundliches Leben*, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 195–213. https://doi.org/10.1007/978-3-662-66497-1\_6
- Novy, A., Kubeczko, K., Haderer, M., Bärnthaler, R., Brand, U., Brudermann, T., et al. (2023b). "Kapitel 24. Theorien des Wandels und der Gestaltung von Strukturen", in APCC Special Re-

port: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 651–652. Available at: https://doi.org/10.1007/978-3-662-66497-1\_28 (Accessed September 16, 2023).

- Oberauer, K., Schickl, M., Zint, M., Liebhaber, N., Deisenrieder, V., Kubisch, S., et al. (2023). The impact of teenagers' emotions on their complexity thinking competence related to climate change and its consequences on their future: looking at complex interconnections and implications in climate change education. *Sustain Sci* 18, 907–931. <u>https://doi.org/10.1007/s11625-022-01222-y</u>
- Obokata, R., Veronis, L., and McLeman, R. (2014). Empirical research on international environmental migration: a systematic review. *Popul Environ* 36, 111–135. <u>https://doi.org/10.1007/ s11111-014-0210-7</u>
- O'Connor, J. (1991). On the two contradictions of capitalism. Capitalism Nature Socialism 2, 107–109. <u>https://doi.org/10.1080/</u> 10455759109358463
- OeBFA (2022). Republic of Austria Green Investor Report 2022. Vienna: Austrian Treasury (OeBFA). Available at: <u>https://www.oebfa.at/en/presse/presseuebersicht/2023/green-investor-re-port-2022.html</u> (Accessed May 10, 2024).
- OeBFA (2024). Republic of Austria Green Investor Report 2023. Vienna: Austrian Treasury (OeBFA). Available at: <u>https://www.oebfa.at/presse/presseuebersicht/2024/green-investor-report-2023.html</u> (Accessed October 17, 204AD).
- OECD (2005). Environmentally Harmful Subsidies: Challenges for Reform. Paris: OECD Publishing. Available at: <u>https://www.oecd-ilibrary.org/agriculture-and-food/environmentally-harmful-subsidies\_9789264012059-en</u> (Accessed October 22, 2024).
- OECD (2021a). OECD Economic Surveys: Austria. OECD. Available at: <u>https://www.oecd.org/en/publications/oecd-econom-ic-surveys-austria-2021\_eaf9ec79-en.html</u> (Accessed January 27, 2025).
- OECD (2021b). Green Budgeting in OECD Countries. Paris: OECD Publishing. https://doi.org/10.1787/acf5d047-en
- ÖGB (2021). Klimapolitik aus ArbeitnehmerInnen- Perspektive. Vienna: Austrian Trade Union Federation (ÖGB). Available at: <u>https://www.oegb.at/content/dam/oegb/downloads/forderung-spapiere/Klima-Positionspapier%20des%20%C3%96GB\_Klimapolitik%20aus%20ArbeitnehmerInnen-Perspektive.pdf</u>
- Oki, T. (2021). European fuel economy policy for new passenger cars: a historical comparative analysis of discourses and change factors. *Int Environ Agreements* 21, 165–181. <u>https://doi.org/10.1007/s10784-020-09510-7</u>
- Olesen, G. B., and Vikkelsø, A. (2024). Sufficiency in European Climate Policies, four country NECPs analysed [version 1; peer review: 1 approved with reservations]. *Open Res Europe* 4, 208. <u>https://doi.org/10.12688/openreseurope.18297.1</u>
- ÖROK (2022). Territorialer Plan für einen gerechten Übergang Österreich 2021–2027. Vienna: Österreichische Raumordnungskonferenz (ÖROK). Available at: <u>https://www.oerok.gv.at/</u><u>fileadmin/user\_upload/Bilder/3.Reiter-Regionalpolitik/EU-Fonds\_2021-2027/Fonds/2022-08-03\_JTP\_final.pdf</u> (Accessed September 17, 2023).
- Ortiz, R. J. (2020). Oil-Fueled Accumulation in Late Capitalism: Energy, Uneven Development, and Climate Crisis. *Critical Historical Studies* 7, 205–240. <u>https://doi.org/10.1086/710799</u>

- Osberg, L., and Smeeding, T. (2006). "Fair" Inequality? Attitudes toward Pay Differentials: The United States in Comparative Perspective. *Am Sociol Rev* 71, 450–473. <u>https://doi.org/10.1177/000312240607100305</u>
- Otero, I., Farrell, K. N., Pueyo, S., Kallis, G., Kehoe, L., Haberl, H., et al. (2020). Biodiversity policy beyond economic growth. *Con*servation Letters 13, e12713. <u>https://doi.org/10.1111/conl.12713</u>
- Otto, I. M., Donges, J. F., Cremades, R., Bhowmik, A., Hewitt, R. J., Lucht, W., et al. (2020). Social tipping dynamics for stabilizing Earth's climate by 2050. *Proc. Natl. Acad. Sci. U.S.A.* 117, 2354–2365. <u>https://doi.org/10.1073/pnas.1900577117</u>
- Papadimitriou, V. (2004). Prospective Primary Teachers' Understanding of Climate Change, Greenhouse Effect, and Ozone Layer Depletion. *Journal of Science Education and Technology* 13, 299–307. https://doi.org/10.1023/b:jost.0000031268.72848.6d
- Parrado, R., Bosello, F., and Standardi, G. (2021). Macroeconomic assessment of Climate Change Impacts (Version 1) [Data set]. Zenodo. <u>https://doi.org/10.5281/zenodo.5546248</u>
- Parth, S., Schickl, M., Keller, L., and Stoetter, J. (2020). Quality Child–Parent Relationships and Their Impact on Intergenerational Learning and Multiplier Effects in Climate Change Education. Are We Bridging the Knowledge–Action Gap? *Sustainability* 12, 7030. <u>https://doi.org/10.3390/su12177030</u>
- Parth, S., Schickl, M., Oberauer, K., Kubisch, S., Deisenrieder, V., Liebhaber, N., et al. (2024). Teenagers performing research on climate change education in a fully integrated design-based research setting. *International Journal of Science Education* 46, 978–1000. https://doi.org/10.1080/09500693.2023.2268295
- Paterson, M. (2020). Climate change and international political economy: between collapse and transformation. *Review of International Political Economy* 28, 394–405. <u>https://doi.org/10.</u> <u>1080/09692290.2020.1830829</u>
- Paterson, M., and P-Laberge, X. (2018). Political economies of climate change. WIREs Climate Change 9, e506. <u>https://doi.org/ 10.1002/wcc.506</u>
- Paterson, M., Wilshire, S., and Tobin, P. (2024). The Rise of Anti-Net Zero Populism in the UK: Comparing Rhetorical Strategies for Climate Policy Dismantling. *Journal of Comparative Policy Analysis: Research and Practice* 26, 332–350. <u>https://doi.org/10.1080/13876988.2023.2242799</u>
- Pathak, M., Slade, R., Shukla, P. R., Skea, J., Pichs-Madruga, R., and Ürge-Vorsatz, D. (2022). "Technical Summary", in Climate Change 2022 – Mitigation of Climate Change: Working Group III Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, eds. P. R. Shukla, J. Skea, R. Slade, R. Al Khourdajie, R. van Diemen, D. McCollum, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press), 51–148. https://doi.org/10.1017/9781009157926.002
- Patterson, J. J. (2023). Backlash to Climate Policy. Global Environmental Politics 23, 68–90. <u>https://doi.org/10.1162/glep\_a\_00684</u>
- Pearce, W., Niederer, S., Özkula, S. M., and Sánchez Querubín, N. (2019). The social media life of climate change: Platforms, publics, and future imaginaries. WIREs Climate Change 10, e569. <u>https://doi.org/10.1002/wcc.569</u>
- Pearson, M. (2024). Back to the Stone Age: Europe's Mainstream Right and Climate Change. *The Political Quarterly* 95, 123–130. https://doi.org/10.1111/1467-923X.13316
- Pearson, M., and Rüdig, W. (2020). The Greens in the 2019 European elections. *Environmental Politics* 29, 336–343. <u>https://doi.org/10.1080/09644016.2019.1709252</u>

- Perner, S., and Spitzer, M. (2021). Royal Dutch Shell Klimaklagen auf dem Weg ins Privatrecht. *ÖJZ* 12, 591.
- Perreault, T., Bridge, G., and McCarthy, J. eds. (2015). The Routledge Handbook of Political Ecology. London: Routledge. <u>https://</u> doi.org/10.4324/9781315759289
- Peters, H.-R. (2015). *Wirtschaftspolitik*. Walter de Gruyter GmbH & Co KG.
- Pianta, S., and Sisco, M. R. (2020). A hot topic in hot times: how media coverage of climate change is affected by temperature abnormalities. *Environ. Res. Lett.* 15, 114038. <u>https://doi.org/10.1088/1748-9326/abb732</u>
- Pichler, M. (2023). Political dimensions of social-ecological transformations: polity, politics, policy. Sustainability: Science, Practice and Policy 19, 2222612. <u>https://doi.org/10.1080/15487733.</u> 2023.2222612
- Pichler, M., Brand, U., and Görg, C. (2020). The double materiality of democracy in capitalist societies: challenges for social-ecological transformations. *Environmental Politics* 29, 193–213. <u>https://doi.org/10.1080/09644016.2018.1547260</u>
- Pichler, M., Krenmayr, N., Maneka, D., Brand, U., Högelsberger, H., and Wissen, M. (2021a). Beyond the jobs-versus-environment dilemma? Contested social-ecological transformations in the automotive industry. *Energy Research & Social Science* 79, 102180. <u>https://doi.org/10.1016/j.erss.2021.102180</u>
- Pichler, M., Krenmayr, N., Schneider, E., and Brand, U. (2021b). EU industrial policy: Between modernization and transformation of the automotive industry. *Environmental Innovation* and Societal Transitions 38, 140–152. <u>https://doi.org/10.1016/j.eist.2020.12.002</u>
- Pierre, J., and Peters, B. G. (2020). *Governance, politics, and the state.*, 2<sup>nd</sup> ed. London: Red globe press.
- Pigou, A. C. (2013). The Economics of Welfare., Fourth edition. Houndmills, Basingstoke, Hampshire; New York: Palgrave Macmillan. Available at: <u>https://link.springer.com/book/97802</u> <u>30249318</u>
- Piguet, E., Kaenzig, R., and Guélat, J. (2018). The uneven geography of research on "environmental migration." *Popul Environ* 39, 357–383. <u>https://doi.org/10.1007/s11111-018-0296-4</u>
- Pindyck, R. S., and Rubinfeld, D. L. (2018). *Microeconomics.*, Ninth edition. NY: Pearson.
- Plank, B., Streeck, J., Virág, D., Krausmann, F., Haberl, H., and Wiedenhofer, D. (2022). Compilation of an economy-wide material flow database for 14 stock-building materials in 177 countries from 1900 to 2016. *Methods X* 9, 101654. <u>https://doi.org/10.1016/j.mex.2022.101654</u>
- Plank, C., Haas, W., Schreuer, A., Irshaid, J., Barben, D., and Görg, C. (2021). Climate policy integration viewed through the stakeholders' eyes: A co-production of knowledge in social-ecological transformation research. *Environmental Policy and Governance* 31, 387–399. https://doi.org/10.1002/eet.1938
- Plehwe, D., Neujeffski, M., and Haas, T. (2024). "Climate Obstruction in the European Union: Business Coalitions and the Technocracy of Delay", in *Climate Obstruction across Europe*, eds. R. J. Brulle, J. T. Roberts, and M. C. Spencer (Oxford University Press), 320–346. <u>https://doi.org/10.1093/oso/97801977</u> <u>62042.003.0013</u>
- Plutzer, E., McCaffrey, M., Hannah, A. L., Rosenau, J., Berbeco, M., and Reid, A. H. (2016). Climate confusion among US teachers. *Science* 351, 664–665. <u>https://doi.org/10.1126/science.aab3907</u>
- Pories, S., and Pfeiffer, M. (2024). "Bodenschutz im Zeichen der Klima- und Biodiversitätskrise", in *Bodenschutz im Zeichen*

*der Klimakrise*, eds. G. Schnedl, O. C. Ruppel, and M. Hofer (Verlag Österreich), 3–27. <u>https://doi.org/10.33196/978370469</u> <u>4454-101</u>

- Praprotnik, K., Ingruber, D., Nash, S., and Rodenko, R. (2022). Evaluation Report of the Austrian 'Klimarat' UWK, Assessment of the Perspectives of the Members and the Public. University for Continuing Education Krems. Available at: <u>https:// www.donau-uni.ac.at/dam/jcr:f8f52750-1594-485e-9f4f-89efc0ad2d32/SACCA\_final%20report\_UWKpart.pdf</u> (Accessed June 16, 2023).
- Preston, B. L., Dow, K., and Berkhout, F. (2013). The Climate Adaptation Frontier. *Sustainability* 5, 1011–1035. <u>https://doi.org/10.3390/su5031011</u>
- Probst, T., Hohmann, R., Pütz, M., Braunschweiger, D., and Kuhn-Belaid, R. (2019). Climate Adaptation Governance in the Alpine Space. Transnational Synthesis Report (WP1). Zürich: European Regional Development Fund. Available at: <u>https:// www.alpine-space.eu/wp-content/uploads/2021/10/37-1-goapply-Climate-Adaptation-Governance-in-the-Alpine-Space-Transnational-Synthesis-Report-output.pdf</u> (Accessed September 9, 2023).
- Pürgy, E., and Hofer, T. E. (2019). "Verkehrsrecht", in *Öffentliches Wirtschaftsrecht*, eds. M. Holoubek and M. Potacs (Vienna: Verlag Österreich). <u>https://doi.org/10.33196/9783704684141</u>
- Pye, S., Dobbins, A., Baffert, C., Brajković, J., Deane, P., and De Miglio, R. (2017). "Chapter 30 – Energy Poverty Across the EU: Analysis of Policies and Measures", in *Europe's Energy Transition*, eds. M. Welsch, S. Pye, D. Keles, A. Faure-Schuyer, A. Dobbins, A. Shivakumar, et al. (Academic Press), 261–280. <u>https:// doi.org/10.1016/B978-0-12-809806-6.00030-4</u>
- Rabl, T. (2024). Anything goes?! ecolex 1. Available at: <u>https://rdb.manz.at/document/rdb.tso.Llecolex20240101</u> (Accessed October 17, 2024).
- Rahmstorf, S., and Schellnhuber, H. J. (2018). *Der Klimawandel: Diagnose, Prognose, Therapie*. München: CH Beck.
- Raimund, W. (2023). Rebound-Effekte in der Mobilität. Vienna: Austrian Environmental Agency. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/publikationen/dp178.pdf</u> (Accessed August 18, 2023).
- Rao, N. D., and Min, J. (2018). Decent Living Standards: Material Prerequisites for Human Wellbeing. *Soc Indic Res* 138, 225–244. https://doi.org/10.1007/s11205-017-1650-0
- Rauch, F., Glettler, C., Steiner, R., and Dulle, M. (2024). "Environmental and Sustainability Education in Austria", in *World Review*, eds. M. Rieckmann and R. T. Muñoz (Boca Raton: CRC Press).
- Rauch, F., and Steiner, R. (2013). Competences for education for sustainable development in teacher education. *CEPS journal* 3, 9–24. <u>https://doi.org/10.25656/01:7663</u>
- Raworth, K. (2017). A Doughnut for the Anthropocene: humanity's compass in the 21<sup>st</sup> century. *The Lancet Planetary Health* 1, e48–e49. <u>https://doi.org/10.1016/S2542-5196(17)30028-1</u>
- Regulation (EU) 2018/1999 (2018). Regulation (EU) 2018/1999 of the European Parliament and of the Council of 11 December 2018 on the Governance of the Energy Union and Climate Action, amending Regulations (EC) No 663/2009 and (EC) No 715/2009 of the European Parliament and of the Council, Directives 94/22/EC, 98/70/EC, 2009/31/EC, 2009/73/EC, 2010/31/EU, 2012/27/EU and 2013/30/EU of the European Parliament and of the Council, Council Directives 2009/119/EC and (EU) 2015/652 and repealing Regulation (EU) No 525/2013 of the

European Parliament and of the Council. Available at: <u>https://eur-lex.europa.eu/legal-content/EN/TXT/HTML/?uri=CE-LEX:32018R1999</u>

- Regulation (EU) 2021/1119 (2021). Regulation (EU) 2021/1119 of the european parliament and of the council. Available at: <u>https:// eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:</u> <u>32021R1119</u> (Accessed September 6, 2023).
- Regulation (EU) 2023/839 (2023). REGULATION (EU) 2023/839 OF THE EUROPEAN PARLIAMENT AND OF THE COUN-CIL of 19 April 2023 amending Regulation (EU) 2018/841 as regards the scope, simplifying the reporting and compliance rules, and setting out the targets of the Member States for 2030, and Regulation (EU) 2018/1999 as regards improvement in monitoring, reporting, tracking of progress and review. Available at: <u>https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/</u> ?uri=CELEX:32023R0839&qid=1694704932671 (Accessed September 14, 2023).
- Regulation (EU) 2023/851 (2023). Regulation (EU) 2023/851 of the European Parliament and of the Council of 19 April 2023 amending Regulation (EU) 2019/631 as regards strengthening the CO2 emission performance standards for new passenger cars and new light commercial vehicles in line with the Union's increased climate ambition (Text with EEA relevance). Available at: <u>https://eur-lex.europa.eu/eli/reg/2023/851</u> (Accessed September 14, 2023).
- Regulation (EU) 2023/857 (2023). Regulation (EU) 2023/857 of the European Parliament and of the Council of 19 April 2023 amending Regulation (EU) 2018/842 on binding annual greenhouse gas emission reductions by Member States from 2021 to 2030 contributing to climate action to meet commitments under the Paris Agreement, and Regulation (EU) 2018/1999 (Text with EEA relevance). Available at: <u>http://data.europa.eu/eli/ reg/2023/857/oj/eng</u> (Accessed May 15, 2024).
- Regulation (EU) 2023/955 (2023). Regulation (EU) 2023/955 of the European Parliament and of the Council of 10 May 2023 establishing a Social Climate Fund and amending Regulation (EU) 2021/1060. Available at: <u>https://eur-lex.europa.eu/eli/ reg/2023/955</u> (Accessed September 14, 2023).
- Regulation (EU) 2023/956 (2023). Regulation (EU) 2023/956 OF THE EUROPEAN PARLIAMENT AND OF THE COUN-CIL of 10 May 2023 establishing a carbon border adjustment mechanism (Text with EEA relevance). Available at: <u>https:// eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:</u> <u>32023R0956&qid=1694704717908</u> (Accessed September 14, 2023).
- Regulation (EU) 2023/1319 (2023). COMMISSION IMPLE-MENTING DECISION (EU) 2023/1319 of 28 June 2023 amending Implementing Decision (EU) 2020/2126 to revise Member States' annual emission allocations for the period from 2023 to 2030 (Text with EEA relevance). Available at: <u>https:// eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:</u> <u>32023D1319&qid=1694704818912</u> (Accessed September 14, 2023).
- Regulation (EU) 2023/1804 (2023). Regulation (EU) 2023/1804 of the European Parliament and of the Council of 13 September 2023 on the deployment of alternative fuels infrastructure, and repealing Directive 2014/94/EU (Text with EEA relevance). Available at: https://eur-lex.europa.eu/eli/reg/2023/1804/oj
- Regulation (EU) 2023/1805 (2023). Regulation (EU) 2023/1805 of the European Parliament and of the Council of 13 September

2023 on the use of renewable and low-carbon fuels in maritime transport, and amending Directive 2009/16/EC.

- Regulation (EU) 2023/2405 (2023). Regulation (EU) 2023/2405 of the European Parliament and of the Council of 18 October 2023 on ensuring a level playing field for sustainable air transport (ReFuelEU Aviation). Available at: <u>https://eur-lex.europa.eu/ eli/reg/2023/2405/oj</u>
- Regulation (EU) 2024/1787 (2024). Regulation (EU) 2024/1787 of the European Parliament and of the Council of 13 June 2024 on the reduction of methane emissions in the energy sector and amending Regulation (EU) 2019/942 (Text with EEA relevance). Available at: <u>https://eur-lex.europa.eu/eli/reg/2024/1787/oj</u> (Accessed October 19, 2024).
- Reid, A., Dillon, J., Ardoin, N., and Ferreira, J.-A. (2021). Scientists' warnings and the need to reimagine, recreate, and restore environmental education. *Environmental Education Research* 27, 783–795. <u>https://doi.org/10.1080/13504622.2021.1937577</u>
- Reif, A., Guenther, L., Tschötschel, R., and Brüggemann, M. (2024). Wandel der Einstellungen und Kommunikation zu Klimawandel und Klimapolitik von 2015 bis 2023 Rückschlag für den Klimaschutz. *Media Perspektiven* 1. Available at: <u>https://www. ard-media.de/fileadmin/user\_upload/media-perspektiven/pdf/ 2024/MP\_14\_2024\_Rueckschlag\_fuer\_den\_Klimaschutz.pdf</u> (Accessed October 17, 2024).
- Reimers, F. M. (2021). "The Role of Universities Building an Ecosystem of Climate Change Education", in *Education and Climate Change*, ed. F. M. Reimers (Springer, Cham), 1–44.
- Reisigl, M. (2021). "Narrative?" I can't hear that anymore? A linguistic critique of an overstretched umbrella term in cultural and social science studies, discussed with the example of the discourse on climate change. Critical Discourse Studies 18, 368– 386. <u>https://doi.org/10.1080/17405904.2020.1822897</u>
- Renström, T. I., Spataro, L., and Marsiliani, L. (2021). Can subsidies rather than pollution taxes break the trade-off between economic output and environmental protection? *Energy Economics* 95, 105084. <u>https://doi.org/10.1016/j.eneco.2020.105084</u>
- Reynaert, M. (2021). Abatement Strategies and the Cost of Environmental Regulation: Emission Standards on the European Car Market. *The Review of Economic Studies* 88, 454–488. https://doi.org/10.1093/restud/rdaa058
- Rhomberg, M. (2016). "Climate Change Communication in Austria", in Oxford Research Encyclopedia of Climate Science, (Oxford: Oxford University Press). <u>https://doi.org/10.1093/acrefore/9780190228620.013.449</u>
- Riahi, K., Schaeffer, R., Arango, J., Calvin, K., Guivarch, C., Hasegawa, T., et al. (2022). Mitigation pathways compatible with long-term goals. *IPCC*, 2022: Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. <u>https://doi.org/10.1017/9781009157926.005</u>
- Rieckmann, M., and Bormann, I. (2020). *Higher Education Institutions and Sustainable Development*. Basel: MDPI.
- Rinscheid, A., Rosenbloom, D., Markard, J., and Turnheim, B. (2021). From terminating to transforming: The role of phaseout in sustainability transitions. *Environmental Innovation and Societal Transitions* 41, 27–31. <u>https://doi.org/10.1016/j.eist.</u> <u>2021.10.019</u>
- Roberts, C., Geels, F. W., Lockwood, M., Newell, P., Schmitz, H., Turnheim, B., et al. (2018). The politics of accelerating low-carbon transitions: Towards a new research agenda. *Energy Re-*

search & Social Science 44, 304-311. <u>https://doi.org/10.1016/j.</u> erss.2018.06.001

- Robeyns, I., Buskens, V., van de Rijt, A., Vergeldt, N., and van der Lippe, T. (2021). How Rich is Too Rich? Measuring the Riches Line. Soc Indic Res 154, 115–143. <u>https://doi.org/10.1007/ s11205-020-02552-z</u>
- Rodrik, D. (2014). Green industrial policy. Oxford Review of Economic Policy 30, 469–491. <u>https://doi.org/10.1093/oxrep/gru025</u>
- Rosenbloom, D., Markard, J., Geels, F. W., and Fuenfschilling, L. (2020). Why carbon pricing is not sufficient to mitigate climate change—and how "sustainability transition policy" can help. *Proceedings of the National Academy of Sciences* 117, 8664–8668. <u>https://doi.org/10.1073/pnas.2004093117</u>
- Rucht, D. (2023). Die Letze Generation: Eine kritische Zwischenbilanz. Forschungsjournal Soziale Bewegungen 36, 186–204. <u>https://doi.org/10.1515/fjsb-2023-0018</u>
- Ruser, A., and Machin, A. (2019). Nationalising the Climate: Is the European Far Right Turning Green? *European Green Journal*. Available at: <u>https://www.greeneuropeanjournal.eu/nationalis-ing-the-climate-is-the-european-far-right-turning-green/</u> (Accessed September 7, 2023).
- Ryan, J. ed. (2011). China's Higher Education Reform and Internationalisation. London: Routledge.
- Sachs, J. D., Schmidt-Traub, G., Mazzucato, M., Messner, D., Nakicenovic, N., and Rockström, J. (2019). Six Transformations to achieve the Sustainable Development Goals. *Nat Sustain* 2, 805–814. https://doi.org/10.1038/s41893-019-0352-9
- Saitō, K. (2016). Natur gegen Kapital: Marx' Ökologie in seiner unvollendeten Kritik des Kapitalismus. Frankfurt am Main, New York: Campus Verlag.
- Saltmarsh, J., and Hartley, M. (2011). "To serve a larger purpose": Engagement for Democracy and the Transformation of Higher Education. Philadelphia: Temple University Press.
- Sander, P., and Schlatter, B. (2014). "Das Bundesverfassungsgesetz über die Nachhaltigkeit, den Tierschutz, den umfassenden Umweltschutz, die Sicherstellung der Wasser- und Lebensmittelversorgung und die Forschung", in Jahrbuch Öffentliches Recht, (Vienna), 235.
- Schäfer, M. S., and Painter, J. (2021). Climate journalism in a changing media ecosystem: Assessing the production of climate change-related news around the world. WIREs Clim Change 12. https://doi.org/10.1002/wcc.675
- Scheffran, J. (2023). Limits to the Anthropocene: geopolitical conflict or cooperative governance? *Front. Polit. Sci.* 5. <u>https://doi.org/10.3389/fpos.2023.1190610</u>
- Schelly, C., Cross, J. E., Franzen, W., Hall, P., and Reeve, S. (2012). How to Go Green: Creating a Conservation Culture in a Public High School Through Education, Modeling, and Communication. *The Journal of Environmental Education* 43, 143–161. https://doi.org/10.1080/00958964.2011.631611
- Scherhaufer, P., and Clar, C. (2021). "Klimapolitik auf Österreichisch: 'Ja, aber ...," in *Klimasoziale Politik: eine gerechte und emissionsfreie Gesellschaft gestalten*, eds. Die Armutskonferenz, ATTAC, and Beirat für Gesellschafts-, Wirtschafts- und Umweltpolitische Alternativen (Wien: bahoe books), 31–40.
- Schickl, M., Oberauer, K., Stötter, J., Kromp-Kolb, H., and Keller, L. (2024). makingAchange – eine der bislang umfassendsten Klimawandelbildungsinitiativen Österreichs. Erkenntnisse zu Wirkungsweisen und einer dringend notwendigen Transformation in der Klimawandelbildung. GW Unterricht 173, 7–19. https://doi.org/10.1553/gw-unterricht173s7

- Schiman-Vukan, S. (2022). Langfristige Perspektiven der öffentlichen Finanzen in Österreich (Long-term Perspectives of Public Finances in Austria). Vienna: Österreichisches Institut für Wirtschaftsforschung. Available at: <u>https://www.wifo.ac.at/ jart/prj3/wifo/resources/person\_dokument/person\_dokument.</u> jart?publikationsid=70395&mime\_type=application/pdf (Accessed September 13, 2023).
- Schinko, T., Karabaczek, V., Kienberger, S., and Menk, L. (2022). Grenzen der Anpassung in Österreich? TransLoss Policy Brief. Laxenburg & Salzburg: Paris-Lodron-Universität Salzburg. Available at: <u>https://pure.iiasa.ac.at/id/eprint/18008/</u> (Accessed October 21, 2024).
- Schinko, T., Karabaczek, V., Menk, L., and Kienberger, S. (2024). Identifying constraints and limits to climate change adaptation in Austria under deep uncertainty. *Front. Clim.* 6, 1303767. <u>https://doi.org/10.3389/fclim.2024.1303767</u>
- Schinko, T., Mechler, R., and Hochrainer-Stigler, S. (2017). A methodological framework to operationalize climate risk management: managing sovereign climate-related extreme event risk in Austria. *Mitig Adapt Strateg Glob Change* 22, 1063–1086. https://doi.org/10.1007/s11027-016-9713-0
- Schleussner, C.-F., Donges, J. F., Donner, R. V., and Schellnhuber, H. J. (2016). Armed-conflict risks enhanced by climate-related disasters in ethnically fractionalized countries. *Proc. Natl. Acad. Sci. U.S.A.* 113, 9216–9221. <u>https://doi.org/10.1073/pnas.</u> <u>1601611113</u>
- Schmidt, A., Ivanova, A., and Schäfer, M. S. (2013). Media attention for climate change around the world: A comparative analysis of newspaper coverage in 27 countries. *Global Environmental Change* 23, 1233–1248. <u>https://doi.org/10.1016/j.gloenvcha.</u> 2013.07.020
- Schneider, E., Brad, A., Brand, U., Krams, M., and Lenikus, V. (2023). "Historical-materialist policy analysis of climate change policies", in *Handbook on Critical Political Economy and Public Policy*, (Cheltenham, UK: Edward Elgar Publishing), 110– 126. Available at: <u>https://www.elgaronline.com/edcollchap/ book/9781800373785/book-part-9781800373785-17.xml</u> (Accessed June 16, 2023).
- Schneider, F. G., Tichler, R., and Steinmüller, H. (2010). Effekte der Einführung einer CO2-Steuer in Österreich im Jahr 2010. *Energie Information*. Available at: <u>https://www.researchgate.net/</u> profile/Robert-Tichler/publication/272466239\_Effekte\_der\_ <u>Einfuhrung\_einer\_CO2-Steuer\_in\_Osterreich\_im\_Jahr\_2010/</u> <u>links/54e4b4640cf29865c334dd3c/Effekte-der-Einfuehrungeiner-CO2-Steuer-in-Oesterreich-im-Jahr-2010.pdf</u> (Accessed June 17, 2024).
- Schneider, U. (2023). "Kapitel 18. Sozialstaat und Klimawandel", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 499– 528. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1</u> 22 (Accessed October 4, 2023).
- Schneidewind, U., and Singer-Brodowski, M. (2014). Transformative Wissenschaft. Klimawandel im deutschen Wissenschafts-und Hochschulsystem., 2nd Edn. Marburg: Metropolis-Verl.
- Schrot, O. G., Keller, L., Peduzzi, D., Riede, M., Kuthe, A., and Ludwig, D. (2019). "Teenagers Expand Their Conceptions of Climate Change Adaptation Through Research-Education Cooperation", in *Climate Change and the Role of Education*, eds. W. Leal Filho and S. L. Hemstock (Cham: Springer International

Publishing), 525–547. <u>https://doi.org/10.1007/978-3-030-328</u> 98-6\_29

- Schrot, O. G., Peduzzi, D., Ludwig, D., Riede, M., and Keller, L. (2021). Is it possible to build adolescents' cognitive adaptive capacity through climate change education? Insights into a two-year long educational programme in North Tyrol (Austria) and South Tyrol (Italy). *Climate Risk Management* 33, 100327. <u>https://doi.org/10.1016/j.crm.2021.100327</u>
- Schubatzky, T., and Haagen-Schützenhöfer, C. (2023). "Inoculating Adolescents Against Climate Change Misinformation", in *Fostering Scientific Citizenship in an Uncertain World: Selected Papers from the ESERA 2021 Conference*, eds. G. S. Carvalho, A. S. Afonso, and Z. Anastácio (Cham: Springer International Publishing), 275–292. <u>https://doi.org/10.1007/978-3-031-32225-9\_17</u>
- Schulev-Steindl, E., Hofer, M., and Franke, L. (2020). Evaluierung des Klimaschutzgesetzes. Graz: ClimLaw: Graz – Forschungszentrum für Klimaschutzrecht. Available at: <u>https://www.bmk. gv.at/dam/jcr:0e6aead9-19f5-4004-9764-4309b089196d/KSG\_ Evaluierung\_ClimLawGraz\_ua.pdf</u> (Accessed September 6, 2023).
- Schulev-Steindl, E., Romirer, C., and Liebenberger, L. (2021). Mobilitätswende: Klimaschutz im Verkehr auf dem rechtlichen Prüfstand (Teil I). *Recht der Umwelt (RdU)* 6, 237.
- Schützenhofer, C., Alton, V., Gahleitner, B., Köttner, S., Kubeczko, K., Leitner, K.-H., et al. (2024). transform.industry Transformationspfade und FTI-Fahrplan für eine klimaneutrale Industrie 2040 in Österreich. Wien: Klima und Energiefonds der österreichischen Bundesregierung. Available at: <u>https://energieforschung.at/wp-content/uploads/sites/11/2022/04/Transformindustry\_Endbericht\_RZ\_BF.pdf</u> (Accessed February 5, 2025).
- Schwerdtle, P. N., McMichael, C., Mank, I., Sauerborn, R., Danquah, I., and Bowen, K. J. (2020). Health and migration in the context of a changing climate: a systematic literature assessment. *Environ. Res. Lett.* 15, 103006. <u>https://doi.org/10.1088/1748-9326/ab9ece</u>
- Schwörer, J. (2024). Mainstream parties and global warming: What determines parties' engagement in climate protection? *European Journal of Political Research* 63, 303–325. <u>https://doi.org/10.1111/1475-6765.12602</u>
- Scoones, I., Leach, M., and Newell, P. (2015). The Politics of Green Transformations. London: Routledge. <u>https://doi.org/10.4324/</u> 9781315747378
- Sedlaczek, A. S. (2017). The field-specific representation of climate change in factual television: a multimodal critical discourse analysis. *Critical Discourse Studies* 14, 480–496. <u>https://doi.org/ 10.1080/17405904.2017.1352003</u>
- Šedová, B., Čizmaziová, L., and Cook, A. (2021). A meta-analysis of climate migration literature. CEPA Discussion Papers 29. https://doi.org/10.25932/publishup-49982
- Seebauer, S. (2021). How to make building renovation work for low-income renters: Preferences for distributional principles and procedural options in Austria. *Energy Research & Social Science* 82, 102270. https://doi.org/10.1016/j.erss.2021.102270
- Seebauer, S., Friesenecker, M., and Eisfeld, K. (2019). Integrating climate and social housing policy to alleviate energy poverty: an analysis of targets and instruments in Austria. *Energy Sources, Part B: Economics, Planning, and Policy* 14, 304–326. <u>https://doi. org/10.1080/15567249.2019.1693665</u>
- Seebauer, S., Lückl, A., Köberl, J., and Kulmer, V. (2021). Soziale Folgen des Klimawandels in Österreich. Vienna: BMSGPK

Federal Ministry of Social Affairs. Available at: <u>https://www.sozialministerium.at/Themen/Soziales/Soziale-Themen/Soziales-und-Klimawandel.html</u> (Accessed January 17, 2025).

- Seebauer, S., Thaler, T., Hanger-Kopp, S., and Schinko, T. (2023). How path dependency manifests in flood risk management: observations from four decades in the Ennstal and Aist catchments in Austria. *Reg Environ Change* 23, 31. <u>https://doi.org/10.1007/ s10113-023-02029-y</u>
- Seebauer, S., and Winkler, C. (2020a). Should I stay or should I go? Factors in household decisions for or against relocation from a flood risk area. *Global Environmental Change* 60, 102018. <u>https://doi.org/10.1016/j.gloenvcha.2019.102018</u>
- Seebauer, S., and Winkler, C. (2020b). Correction to: Coping strategies and trajectories of life satisfaction among households in a voluntary planned program of relocation from a flood-risk area. *Climatic Change* 162, 2241. <u>https://doi.org/10.1007/s10584-020-02809-z</u>
- Segerson, K., Polasky, S., Scheffer, M., Sumaila, U. R., Cárdenas, J. C., Nyborg, K., et al. (2024). A cautious approach to subsidies for environmental sustainability. *Science* 386, 28–30. <u>https://doi.org/10.1126/science.ado2615</u>
- Selk, V., and Kemmerzell, J. (2022). Retrogradism in context. Varieties of right-wing populist climate politics. *Environmental Politics* 31, 755–776. <u>https://doi.org/10.1080/09644016.2021</u>. 1999150
- Sen, A. (2012). "Development as capability expansion", in *The community development reader*, eds. J. DeFilippis and S. Saegert, 305.
- Sers, M. R. (2022). Ecological macroeconomic assessment of meeting a carbon budget without negative emissions. *Global Sustainability* 5, e6. <u>https://doi.org/10.1017/sus.2022.2</u>
- Setzer, J., and Bangalore, M. (2017). "Regulating climate change in the courts", in *Trends in Climate Change Legislation*, eds. A. Averchenkova, S. Fankhauser, and M. Nachmany (Cheltenham: Edward Elgar Publishing). Available at: <u>https://econpapers. repec.org/bookchap/elgeebook/17582.htm</u> (Accessed September 7, 2023).
- Shehata, A., and Hopmann, D. N. (2012). FRAMING CLIMATE CHANGE: A study of US and Swedish press coverage of global warming. *Journalism Studies* 13, 175–192. <u>https://doi.org/10. 1080/1461670X.2011.646396</u>
- Shepardson, D. P., Niyogi, D., Choi, S., and Charusombat, U. (2011). Students' conceptions about the greenhouse effect, global warming, and climate change. *Climatic Change* 104, 481–507. <u>https://doi.org/10.1007/s10584-009-9786-9</u>
- Shmelev, S. E., and Speck, S. U. (2018). Green fiscal reform in Sweden: Econometric assessment of the carbon and energy taxation scheme. *Renewable and Sustainable Energy Reviews* 90, 969– 981. <u>https://doi.org/10.1016/j.rser.2018.03.032</u>
- Silvester, B. R., and Fisker, J. K. (2023). A relational approach to the role of the state in societal transitions and transformations towards sustainability. *Environmental Innovation and Societal Transitions* 47, 100717. <u>https://doi.org/10.1016/j.eist.2023.</u> <u>100717</u>
- Simsa, R. (2024). Analyse der Situation österreichischer zivilgesellschaftlicher Akteur:innen im Umweltbereich. Global 2000. Available at: <u>https://www.global2000.at/sites/global/files/cpr\_studie\_1.10.24.pdf</u> (Accessed October 17, 2024).
- Sinabell, F., Bock-Schappelwein J, Firgo, M., Friesenbichler, K. S., Piribauer, P., Streicher, G., et al. (2019). Eine Zwischenbilanz zu den Wirkungen des Programms der Ländlichen Entwicklung

2014–2020. Wien: Österreichisches Institut für Wirtschaftsforschung, Universität für Bodenkultur Wien, Statistik Austria. Available at: https://www.wifo.ac.at/publication/pid/4143790

- Sinabell, F., and Url, T. (2006). Versicherungslösungen für Naturgefahren. Versicherungsrundschau, 177–178. Available at: <u>https://</u> www.wifo.ac.at/publication/pid/12827789
- Six, E., and Lechinger, V. (2021). Die soziale Gestaltung einer ökologischen Steuerreform? : Das Beste aus mehreren Welten. Wirtschaft und Gesellschaft 47, 171–196. <u>https://doi.org/10.59288/wug472.63</u>
- Skamp, K., Boyes, E., and Stanisstreet, M. (2013). Beliefs and Willingness to Act About Global Warming: Where to Focus Science Pedagogy? *Science Education* 97, 191–217. <u>https://doi.org/</u> 10.1002/sce.21050
- Skovgaard, J., and van Asselt, H. (2018). "The Politics of Fossil Fuel Subsidies and Their Reform: An Introduction", in *The Politics* of Fossil Fuel Subsidies and their Reform, eds. J. Skovgaard and H. van Asselt (Cambridge: Cambridge University Press), 3–20. https://doi.org/10.1017/9781108241946.003
- Slameršak, A., Kallis, G., and O'Neill, D. W. (2022). Energy requirements and carbon emissions for a low-carbon energy transition. *Nat Commun* 13, 6932. <u>https://doi.org/10.1038/s41467-022-33976-5</u>
- Smetschka, B., and Wiedenhofer, D. (2023). "Kapitel 9. Freizeit und Urlaub", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 329–344. <u>https://doi.org/10.1007/978-3-662-66497-1\_13</u>
- Soder, M., Niedermoser, K., and Theine, H. (2018). Beyond growth: new alliances for socio-ecological transformation in Austria. *Globalizations* 15, 520–535. <u>https://doi.org/10.1080/14747731.</u> 2018.1454680
- Soder, M., and Templ, N. (2022). Österreichs Just Transition Plan: Wegweiser in eine faire klimaneutrale Zukunft? Vienna: Austrian Chamber of Labour. Available at: <u>https://wien.arbeiterkammer.at/interessenvertretung/eu/infobrief/IB22\_3/Soder\_ Templ\_O-sterreichsJustTransitionPlan\_AkInfo3\_22\_S19-24.</u> pdf (Accessed October 17, 2024).
- Sorrell, S., Gatersleben, B., and Druckman, A. (2020). The limits of energy sufficiency: A review of the evidence for rebound effects and negative spillovers from behavioural change. *Energy Research & Social Science* 64, 101439. <u>https://doi.org/10.1016/j.</u> <u>erss.2020.101439</u>
- Sovacool, B. K., Hook, A., Martiskainen, M., Brock, A., and Turnheim, B. (2020). The decarbonisation divide: Contextualizing landscapes of low-carbon exploitation and toxicity in Africa. *Global Environmental Change* 60, 102028. <u>https://doi.org/10.1016/j.gloenvcha.2019.102028</u>
- Stangl, F. (2024). "RED III": Die Vorgaben der neuen Erneuerbare-Energie-Richtlinie und ihre Auswirkungen auf die Energiewende. Nachhaltigkeitsrecht 4, 20–30. <u>https://doi.org/10. 33196/nr202401002001</u>
- Statistik Austria (2024). Material flow accounts. Available at: <u>https://www.statistik.at/en/statistics/energy-and-environment/</u> <u>environment/material-flow-accounts</u> (Accessed May 8, 2024).
- Steininger, K., Bednar-Friedl, B., Knittel, N., Kirchengast, G., Nabernegg, S., Williges, K., et al. (2020). Klimapolitik in Österreich: Innovationschance Coronakrise und die Kosten des Nicht-Handelns. Graz: Wegener Center Verlag. <u>https://doi.org/ 10.25364/23.2020.1</u>

- Steininger, K. W., Riahi, K., Stagl, S., Kromp-Kolb, H., Kirchengast, G., Rosenfeld, D. C., et al. (2024). Nationaler Energie- und Klimaplan (NEKP) für Österreich – Wissenschaftliche Bewertung der in der Konsultation 2023 vorgeschlagenen Maßnahmen. Wien: Climate Change Centre Austria (CCCA). Available at: https://ccca.ac.at/fileadmin/00\_DokumenteHauptmenue/02\_ Klimawissen/RefNEKP/Bericht/NEKP\_Wissenschaftliche\_Bewertung\_der\_Massnahmen\_der\_Stellungnahmen\_Februar 2024.pdf (Accessed May 10, 2024).
- Steurer, R., Buzogány, A., Scherhaufer, P., Clar, C., and Nash, S. (2022). "Governance und politische Beteiligung", in APCC Special Report: Strukturen für ein klimafreundliches Leben (APCC SR Klima- freundliches Leben), eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. Steininger, et al. (Berlin/Heidelberg: Springer Spektrum). Available at: <u>https://papers.ssrn.com/sol3/ papers.cfm?abstract\_id=4225548</u>
- Steurer, R., and Clar, C. (2015). Is decentralisation always good for climate change mitigation? How federalism has complicated the greening of building policies in Austria. *Policy Sciences* 48, 85–107. https://doi.org/10.1007/s11077-014-9206-5
- Steurer, R., and Clar, C. (2018). The ambiguity of federalism in climate policy-making: how the political system in Austria hinders mitigation and facilitates adaptation. *Journal of Environmental Policy & Planning* 20, 252–265. <u>https://doi.org/10.1080/</u> 1523908X.2017.1411253
- Steurer, R., Clar, C., and Casado-Asensio, J. (2020). Climate change mitigation in Austria and Switzerland: The pitfalls of federalism in greening decentralized building policies. *Natural Resources Forum* 44, 89–108. https://doi.org/10.1111/1477-8947.12166
- Stevenson, K. T., Peterson, M. N., and Bondell, H. D. (2018). Developing a model of climate change behavior among adolescents. *Climatic Change* 151, 589–603. <u>https://doi.org/10.1007/s10584-018-2313-0</u>
- Stevenson, K. T., Peterson, M. N., and Bradshaw, A. (2016). How Climate Change Beliefs among U.S. Teachers Do and Do Not Translate to Students. *PLOS ONE* 11, e0161462. <u>https://doi.org/ 10.1371/journal.pone.0161462</u>
- Stevenson, R. B., Nicholls, J., and Whitehouse, H. (2017). What Is Climate Change Education? *Curriculum Perspectives* 37, 67–71. <u>https://doi.org/10.1007/s41297-017-0015-9</u>
- Stevis, D., and Felli, R. (2020). Planetary just transition? How inclusive and how just? *Earth System Governance* 6, 100065. <u>https:// doi.org/10.1016/j.esg.2020.100065</u>
- Stiglitz, J. E., Sen, A., and Fitoussi, J.-P. (2009). Report by the Commission on the Measurement of Economic Performance and Social Progress. The Commission Paris. Available at: <u>https://ec.europa.eu/eurostat/documents/8131721/8131772/Stiglitz-Sen-</u> <u>Fitoussi-Commission-report.pdf</u>
- Stirling, A. (2019). How deep is incumbency? A 'configuring fields' approach to redistributing and reorienting power in socio-material change. *Energy Research & Social Science* 58, 101239. <u>https://doi.org/10.1016/j.erss.2019.101239</u>
- Stöglehner, G. ed. (2023a). Grundlagen der Raumplanung. 1: Theorien, Methoden, Instrumente., 2. akt. Auflage. Wien: facultas.
- Stöglehner, G. (2023b). "Klimaschutz in Raumordnung und Strategischer Umweltprüfung", in *Klimaschutzrecht*, ed. D. Ennöckl (Verlag Österreich), 403–422. <u>https://doi.org/10.33196/</u> <u>9783704692610-115</u>
- Storr, S. (2024). Energieraumplanung auf Landesebene Herausforderungen und Steuerungspotenziale. BBL 4, 134.

- Straßheim, H. (2020). The Rise and Spread of Behavioral Public Policy: An Opportunity for Critical Research and Self-Reflection. *International Review of Public Policy* 2, 115–128. <u>https://</u> doi.org/10.4000/irpp.897
- Suchanek, D., Fliedl, M., and Ruiss, M. (2023). "Bundes-Energieeffizienzgesetz", in *Klimaschutzrecht*, ed. D. Ennöckl (Wien: Verlag Österreich), 343–374. <u>https://doi.org/10.33196/</u> <u>9783704692610-113</u>
- Sugiyama, M., Wilson, C., Wiedenhofer, D., Boza-Kiss, B., Cao, T., Chatterjee, J. S., et al. (2024). High with low: Harnessing the power of demand-side solutions for high wellbeing with low energy and material demand. *Joule* 8, 1–6. <u>https://doi.org/ 10.1016/j.joule.2023.12.014</u>
- Suranovic, S. (2013). Fossil fuel addiction and the implications for climate change policy. *Global environmental change* 23, 598– 608. <u>https://doi.org/10.1016/j.gloenvcha.2013.02.006</u>
- Tálos, E., and Hinterseer, T. (2019). Sozialpartnerschaft: ein zentraler politischer Gestaltungsfaktor der Zweiten Republik am Ende?., 1. Auflage. Innsbruck: Studien Verlag.
- Tebecis, T. (2023). Have climate policies been effective in Austria? A reverse causal analysis. WU Vienna University of Economics and Business. Department of Economics Working Paper Series. <u>https://doi.org/10.57938/92adb3ea-18cd-4d05-8d09-223</u> bea611536
- Thaler, R. H., and Sunstein, C. R. (2009). *Nudge: improving decisions about health, wealth and happiness.*, Revised edition. London: Penguin Books.
- Thaler, T. (2021). Just retreat—how different countries deal with it: examples from Austria and England. *J Environ Stud Sci* 11, 412–419. https://doi.org/10.1007/s13412-021-00694-1
- Thaler, T., Hanger-Kopp, S., Schinko, T., and Nordbeck, R. (2023). Addressing path dependencies in decision-making processes for operationalizing compound climate-risk management. *iScience* 26, 107073. <u>https://doi.org/10.1016/j.isci.2023.107073</u>
- Thaler, T., and Seebauer, S. (2019). Bottom-up citizen initiatives in natural hazard management: Why they appear and what they can do? *Environmental Science & Policy* 94, 101–111. <u>https:// doi.org/10.1016/j.envsci.2018.12.012</u>
- Thaler, T., Seebauer, S., and Schindelegger, A. (2020). Patience, persistence and pre-signals: Policy dynamics of planned relocation in Austria. *Global Environmental Change* 63, 102122. <u>https:// doi.org/10.1016/j.gloenvcha.2020.102122</u>
- Thaller, A., and Brudermann, T. (2020). "You know nothing, John Doe" – Judgmental overconfidence in lay climate knowledge. *Journal of Environmental Psychology* 69, 101427. <u>https://doi.org/</u>10.1016/j.jenvp.2020.101427
- Theine, H., Humer, S., Moser, M., and Schnetzer, M. (2022). Emissions inequality: Disparities in income, expenditure, and the carbon footprint in Austria. *Ecological Economics* 197, 107435. https://doi.org/10.1016/j.ecolecon.2022.107435
- Theine, H., and Regen, L. (2023). "Mediendiskurse und -strukturen", in APCC Special Report: Strukturen für ein klimafreundliches Leben (APCC SR Klimafreundliches Leben), eds. C. Görg, V. Madner, K. Steininger, and E. Aigner (Berlin/Heidelberg: Springer Spektrum). Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_24</u>
- Tiefenthaler, M. (2022). Verbot der geologischen Speicherung von CO2 Carbon Capture and Storage in Österreich. *Recht der Umwelt (RdU)* 5, 78.
- Tikkakoski, P., Aulake, M., and Paloniemi, R. (2024). Towards just transition: Tackling inequity and structural causes of vulnera-

bility in key environment, health and climate related policies in Finland. *Environmental Science & Policy* 156, 103736. <u>https://</u>doi.org/10.1016/j.envsci.2024.103736

- Tindall, D. B., Stoddart, M. C. J., and Callison, C. (2018). "The Relationships Between Climate Change News Coverage, Policy Debate, and Societal Decisions", in Oxford Research Encyclopedia of Climate Science, (Oxford: Oxford University Press). https://doi.org/10.1093/acrefore/9780190228620.013.370
- Tobin, P. (2017). Leaders and Laggards: Climate Policy Ambition in Developed States. *Global Environmental Politics* 17, 28–47. <u>https://doi.org/10.1162/GLEP\_a\_00433</u>
- Tobler, C., Visschers, V. H. M., and Siegrist, M. (2012). Consumers' knowledge about climate change. *Climatic Change* 114, 189– 209. <u>https://doi.org/10.1007/s10584-011-0393-1</u>
- Tomassetti, P. (2020). From Treadmill of Production to Just Transition and Beyond. *European Journal of Industrial Relations* 26, 439–457. <u>https://doi.org/10.1177/0959680120951701</u>
- Tosun, J., and Debus, M. (2022). Right-wing populist parties and environmental politics: insights from the Austrian Freedom Party's support for the glyphosate ban. *Environmental Politics* 30, 224–244. <u>https://doi.org/10.1080/09644016.2020.1813997</u>
- Turner, B., Devisscher, T., Chabaneix, N., Woroniecki, S., Messier, C., and Seddon, N. (2022). The Role of Nature-Based Solutions in Supporting Social-Ecological Resilience for Climate Change Adaptation. Annu. Rev. Environ. Resour. 47, 123–148. <u>https:// doi.org/10.1146/annurev-environ-012220-010017</u>
- Turnheim, B., and Sovacool, B. K. (2020). Forever stuck in old ways? Pluralising incumbencies in sustainability transitions. *Environmental Innovation and Societal Transitions* 35, 180–184. <u>https://doi.org/10.1016/j.eist.2019.10.012</u>
- Üblagger, M. (2024). "Klimaschutz als Abwägungsinteresse in verwaltungsrechtlichen Abwägungsentscheidungen", in Nachhaltig in die Zukunft: Das öffentliche Recht im Zeichen der Veränderung, eds. S. Bartl, E. Falch, J. Kaschka, F. Klebelsberg, M. Klema, P. Lechner, et al. (Wien: Jan Sramek Verlag).
- Umweltbundesamt (2023a). Klimaschutzbericht 2023. Vienna: Environment Agency Austria (EAA). Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/publikationen/rep0871bfz.pdf</u> (Accessed September 8, 2023).
- Umweltbundesamt (2023b). Szenario "With additional Measures"
   Ergebnisse & Daten. Wien: Environment Agency Austria (EAA). Available at: <u>https://www.umweltbundesamt.at/filead-min/site/aktuelles/2023/pk\_wam-szenarien230704.pdf</u> (Accessed September 8, 2023).
- Umweltbundesamt (2023c). Energie- und Treibhausgasszenarien 2023: WEM, WAM und Transition mit Zeitreihen von 2020 bis 2050. Wien: Umweltbundesamt GmbH. Available at: <u>https:// www.umweltbundesamt.at/fileadmin/site/publikationen/rep</u> 0882.pdf
- Umweltbundesamt (2024). Klimaschutzbericht 2024. Wien: Environment Agency Austria (EAA). Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/publikationen/rep0913.pdf</u> (Accessed October 17, 2024).
- UNEP IRP (2023). Global Material Flows Database. Available at: <u>https://www.resourcepanel.org/global-material-flows-database</u> (Accessed September 16, 2023).
- Universität Wien (2016). Teilcurriculum für das Unterrichtsfach Biologie und Umweltkunde im Rahmen des Bachelorstudiums zur Erlangung eines Lehramts im Bereich der Sekundarstufe (Allgemeinbildung) im Verbund Nord-Ost. Available
  - at: https://senat.univie.ac.at/fileadmin/user\_upload/s\_senat/

konsolidiert\_Lehramt/Teilcurriculum\_Biologie\_und\_Umweltkunde\_BA\_Lehramt.pdf (Accessed July 15, 2024).

- Unruh, G. C. (2002). Escaping carbon lock-in. *Energy Policy* 30, 317–325. <u>https://doi.org/10.1016/S0301-4215(01)00098-2</u>
- Unterberger, C., Hudson, P., Wouter Botzen, W. J., Schroeer, K., and Steininger, K. W. (2019). Future Public Sector Flood Risk and Risk Sharing Arrangements: An Assessment for Austria. *Ecological Economics* 156, 153–163. <u>https://doi.org/10.1016/j. ecolecon.2018.09.019</u>
- van den Bergh, J., Castro, J., Drews, S., Exadaktylos, F., Foramitti, J., Klein, F., et al. (2021). Designing an effective climate-policy mix: accounting for instrument synergy. *Climate Policy* 21, 745–764. <u>https://doi.org/10.1080/14693062.2021.1907276</u>
- van der Ven, H. (2016). Power and Authority in Global Climate Governance. Global Environmental Politics 16, 130–135. <u>https:// doi.org/10.1162/GLEP\_r\_00381</u>
- Vélez-Henao, J. A., and Pauliuk, S. (2023). Material Requirements of Decent Living Standards. *Environ. Sci. Technol.* 57, 14206– 14217. <u>https://doi.org/10.1021/acs.est.3c03957</u>
- Victor, P. A. (2012). Growth, degrowth and climate change: A scenario analysis. *Ecological Economics* 84, 206–212. <u>https://doi.org/10.1016/j.ecolecon.2011.04.013</u>
- Vinke, K., Bergmann, J., Blocher, J., Upadhyay, H., and Hoffmann, R. (2020). Migration as Adaptation? *Migration Studies* 8, 626– 634. <u>https://doi.org/10.1093/migration/mnaa029</u>
- Vogel, J., and Hickel, J. (2023). Is green growth happening? An empirical analysis of achieved versus Paris-compliant CO<sub>2</sub>-GDP decoupling in high-income countries. *The Lancet Planetary Health* 7, e759–e769. <u>https://doi.org/10.1016/S2542-5196(23)</u> 00174-2
- Vogt, M. ed. (2019). Sustainability certification schemes in the agricultural and natural resource: sectors outcomes for society and the environment. Abingdon, Oxon; New York, NY: Routledge.
- von Zabern, L., and Tulloch, C. D. (2021). Rebel with a cause: the framing of climate change and intergenerational justice in the German press treatment of the Fridays for Future protests. *Media, Culture & Society* 43, 23–47. <u>https://doi.org/10. 1177/0163443720960923</u>
- Vona, F. (2023). Managing the distributional effects of climate policies: A narrow path to a just transition. *Ecological Economics* 205, 107689. <u>https://doi.org/10.1016/j.ecolecon.2022.107689</u>
- Voss, K. (2019). "The Ecological Component of the Ideology and Legislative Activity of the Freedom Party of Austria", in *The Far Right and the Environment*, ed. B. Forchtner (Routledge), 153– 183. <u>https://doi.org/10.4324/9781351104043-10</u>
- Vu, H. T., Blomberg, M., Seo, H., Liu, Y., Shayesteh, F., and Do, H. V. (2021). Social Media and Environmental Activism: Framing Climate Change on Facebook by Global NGOs. *Science Communication* 43, 91–115. <u>https://doi.org/10.1177/1075547020971644</u>
- Wagner, E. (2018). "Klimaschutz mit den Mitteln des Privatrechts? Präventive privatrechtliche Instrumente: Klimaschutzklagen", in *Klimaschutzrecht zwischen Wunsch und Wirklichkeit*, ed. G. Kirchengast (Wien: Böhlau Verlag), 217–234. <u>https://doi.org/ 10.7767/9783205206064.217</u>
- Wagner, E. (2019). Die Notwendigkeit einer Verbandsklage im Klimaschutzrecht. Zeitschrift f
  ür Europ
  äisches Umwelt- und Planungsrecht 17, 185–193.
- Wagner, E. (2023). "Klimaschutz und Zivilrecht", in Klimaschutzrecht, ed. D. Ennöckl (Wien: Verlag Österreich), 541–575. <u>https://doi.org/10.33196/9783704692610-120</u>

- Wagner, E., and Ecker, D. (2013). "Realisierungsvorsorge mittels Ausgleichsflächen – quo vadis?", in *Interdisziplinäre Rechtswissenschaft – Schutzansprüche und Schutzaufgaben im Recht: Festschrift für Ferdinand Kerschner zum 60. Geburtstag*, eds. E. Wagner and W. Bergthaler (Wien: Verlag Österreich).
- Wagner, E., and Ecker, D. (2019). Naturverträglichkeitsprüfung: systematische Aufarbeitung der Prüfung nach Art 6 der FFH-RL. Wien: Jan Sramek Verlag.
- Wagner, E., Ecker, D., Hartl, A., and Burgstaller, L. (2021). "Defizite und Chancen im österreichischen Klima- und Biodiversitätsschutz – Handlungsfelder im Recht zur Erreichung der Agenda 2030", in Adaptation to climate change from the perspective of private and public (international, european and national) law, eds. M. Damohorský, F. Kerschner, V. Stejskal, and E. Wagner (Linz).
- Wagner, E., and Jandl, C. (2023). Einführung in das Naturgefahrenrecht., 2. Auflage 2023, ed. E. Wagner. Linz: Trauner Verlag.
- Wahlström, M., Sommer, M., Kocyba, P., De Vydt, M., De Moor, J., Davies, S., et al. (2019). Protest for a future: Composition, mobilization and motives of the participants in Fridays For Future climate protests on 15 March, 2019 in 13 European cities.
- Wallner, B. (2023). Hot climate cold cases? Plädoyer für einen schadenersatzrechtlichen Zugang bei "Klimaklagen." *RdU* 5, 194.
- Wallner, J. (2022). "The first Austrian climate lawsuit", in *Climate Change, Responsibility and Liability*, eds. E. Schulev-Steindl, M. Hinteregger, G. Kirchengast, L. H. Meyer, O. C. Ruppel, G. Schnedl, et al. (Nomos Verlagsgesellschaft mbH & Co. KG), 361–380. https://doi.org/10.5771/9783748930990-361
- Wang, S., Hausfather, Z., Davis, S., Lloyd, J., Olson, E. B., Liebermann, L., et al. (2023). Future demand for electricity generation materials under different climate mitigation scenarios. *Joule* 7, 309–332. https://doi.org/10.1016/j.joule.2023.01.001
- Wang, X., and Lo, K. (2021). Just transition: A conceptual review. Energy Research & Social Science 82, 102291. <u>https://doi.org/10.1016/j.erss.2021.102291</u>
- Weller, S., Beer, A., and Porter, J. (2024). Place-based just transition: domains, components and costs. *Contemporary Social Science* 19, 355–374. <u>https://doi.org/10.1080/21582041.2024.2333272</u>
- Welzer, H. (2011). Mental Infrastructures How Growth Entered the World and Our Souls., ed.Heinrich Böll Stiftung. Berlin. Available at: <u>https://www.boell.de/sites/default/files/endf\_mental\_infrastructures.pdf</u>
- Werners, S. E., Wise, R. M., Butler, J. R. A., Totin, E., and Vincent, K. (2021). Adaptation pathways: A review of approaches and a learning framework. *Environmental Science & Policy* 116, 266– 275. <u>https://doi.org/10.1016/j.envsci.2020.11.003</u>
- Weyerstraß, K., Getzner, M., Gugele, B., Plank, K., Plank, L., Schieder, W., et al. (2024). Gesamtwirtschaftlicher Investitionsbedarf in Österreich zur Erreichung der Klimaziele. Vienna. Available at: <u>https://wien.arbeiterkammer.at/interessenvertretung/wirtschaftswissenschaften/Studie\_Investitionsbedarf\_20240910.pdf</u> (Accessed February 4, 2025).
- White, D. (2019). "Critical design, hybrid labor, just transitions: Moving beyond technocratic ecomodernisms and the it's-toolate-o-cene", in *Rethinking the Environment for the Anthropo*-

*cene*, eds. M. Arias-Maldonado and Z. Trachtenberg (London: Routledge).

- Wilson, A., Widuto, A., and Pari, M. (2023). Social climate fund: "Fit for 55" package. EU Legislation in Progress Briefing. EPRS (European Parliamentary Research Service). Available at: <u>https:// www.europarl.europa.eu/RegData/etudes/BRIE/2021/698777/ EPRS\_BRI(2021)698777\_EN.pdf</u>
- Winter, V., Büssing, A., Gericke, N., and Möller, A. (2023). Pre-Service Biology Teacher Beliefs about Climate Change Education.
- Winter, V., Kranz, J., and Möller, A. (2022). Climate Change Education Challenges from Two Different Perspectives of Change Agents: Perceptions of School Students and Pre-Service Teachers. Sustainability 14, 6081. <u>https://doi.org/10.3390/su14106081</u>
- Wise, S. B. (2010). Climate Change in the Classroom: Patterns, Motivations, and Barriers to Instruction Among Colorado Science Teachers. *Journal of Geoscience Education* 58, 297–309. <u>https:// doi.org/10.5408/1.3559695</u>
- Wolf, A. (2021). Handelspolitik als Klimaschutzinstrument? Wirtschaftsdienst 2021, 342–346. <u>https://doi.org/10.1007/s10</u> 273-021-2914-5
- World Bank (2023a). Creating an Enabling Environment for Private Sector Climate Action. An Evaluation of World Bank Group Support, Fiscal Years 2013–22. Washington D.C.: World Bank. Available at: <u>https://ieg.worldbankgroup.org/sites/default/files/ Data/Evaluation/files/Private-Sector-Climate-Action.pdf</u> (Accessed June 17, 2024).
- World Bank (2023b). State and Trends of Carbon Pricing 2023. Washington DC: World Bank. Available at: <u>https://www.ecolog-ic.eu/sites/default/files/publication/2023/World%20Bank%20</u> <u>State%20and%20Trends%20of%20Carbon%20Pricing%20</u> 2023.pdf (Accessed September 14, 2023).
- World Bank (2024). State and Trends of Carbon Pricing 2024. Washington, DC: World Bank. Available at: <u>http://hdl.handle.net/10986/41544</u> (Accessed October 28, 2024).
- Wurzel, R. K. W., Zito, A. R., and Jordan, A. J. (2013). Environmental Governance in Europe: A Comparative Analysis of New Environmental Policy Instruments. Cheltenham: Edward Elgar.
- Zell-Ziegler, C., Thema, J., Best, B., Wiese, F., Lage, J., Schmidt, A., et al. (2021). Enough? The role of sufficiency in European energy and climate plans. *Energy Policy* 157, 112483. <u>https://doi. org/10.1016/j.enpol.2021.112483</u>
- Zenios, S. A. (2022). The risks from climate change to sovereign debt. Climatic Change 172, 30. <u>https://doi.org/10.1007/s10584-022-03373-4</u>
- Zickgraf (2019). Keeping People in Place: Political Factors of (Im) mobility and Climate Change. *Social Sciences* 8, 228. <u>https://doi.org/10.3390/socsci8080228</u>
- Zscheischler, J., Westra, S., van den Hurk, B. J. J. M., Seneviratne, S. I., Ward, P. J., Pitman, A., et al. (2018). Future climate risk from compound events. *Nature Clim Change* 8, 469–477. <u>https://doi. org/10.1038/s41558-018-0156-3</u>
- zu Ermgassen, S. O. S. E., Drewniok, M. P., Bull, J. W., Corlet Walker, C. M., Mancini, M., Ryan-Collins, J., et al. (2022). A home for all within planetary boundaries: Pathways for meeting England's housing needs without transgressing national climate and biodiversity goals. *Ecological Economics* 201, 107562. <u>https://doi. org/10.1016/j.ecolecon.2022.107562</u>





Chapter 7

# The Austrian Alps as multi-dimensional focal area

# Second Austrian Assessment Report on Climate Change | AAR2

# Chapter 7 The Austrian Alps as multi-dimensional focal area

This publication was subject to an anonymous international peer review process guided by several review editors, the AAR2 co-chairs and the APCC Steering Committee.

The views or opinions expressed in this publication are not necessarily those of the institutions with whose support this work was carried out. The publisher and the supporting institutions do not guarantee the accuracy or continued availability of urls to external websites or third-party websites mentioned in this publication and accept no responsibility for the accuracy or appropriateness of any content contained therein.

#### **Recommended citation**

S. Fuchs, W. Schöner, R. Steiger, et al. (2025): The Austrian Alps as multi-dimensional focal area. In "Second Austrian Assessment Report on Climate Change (AAR2) of the Austrian Panel on Climate Change (APCC)". [D. Huppmann, M. Keiler, K. Riahi, H. Rieder (eds.)]. Austrian Academy of Sciences Press, Vienna, Austria | doi: <u>doi.org/10.1553/aar2-ch7</u>

You can download the recommended citation in the formats Endnote (.ris) and BibTex (.bib) at <u>aar2.ccca.ac.at/citation</u>.

Vienna, June 2025 ISBN: 978-3-7001-5110-4 | <u>austriaca.at/5110-4</u> (full report)

#### Production

Publisher: Austrian Academy of Sciences Press, Vienna Typesetting: Crossdesign Werbeagentur GmbH, Graz Cover: paulaidaanton | Agentur für digiloges Design Graphic design Figure 7.3: paulaidaanton | Agentur für digiloges Design Al-based tools such as DeepL were used to support the linguistic revision.

Contact address for product safety: Austrian Academy of Sciences Press Dr. Ignaz Seipel-Platz 2 A-1010 Wien Tel.: +43-1-51581-3420 E-Mail: verlag@oeaw.ac.at

#### Funding acknowledgement

The AAR2 was funded by the Climate and Energy Fund under the 14th Austrian Climate Research Program (ACRP).

#### **Copyright and license**

© 2025 Austrian Panel on Climate Change (APCC) This report is published under a Creative Commons CC-BY 4.0-License. <u>creativecommons.org/licenses/by/4.0</u>





The AAR2 is a report by the Austrian Panel on Climate Change (APCC). https://aar2.ccca.ac.at/en
# **Coordinating Lead Authors**

Sven Fuchs	BOKU University
Wolfgang Schöner	University of Graz
Robert Steiger	University of Innsbruck   UIBK

# **Lead Authors**

Bernhard Gems	University of Innsbruck   UIBK
Robert Jandl	Austrian Research Centre for Forests   BFW
Markus Mailer	University of Innsbruck   UIBK
Jan-Christoph Otto	Paris Lodron University of Salzburg   PLUS
Maria Papathoma-Köhle	BOKU University
Harald Pauli	Austrian Academy of Sciences   ÖAW
Ulrike Pröbstl-Haider	BOKU University
Mathias Rotach	University of Innsbruck   UIBK
Barbara Schneider-Muntau	University of Innsbruck   UIBK
Karsten Schulz	BOKU University
Walter Seher	BOKU University
Contributing Authors	
Andreas Bohner	HBLFA Raumberg-Gumpenstein
Franziska Schöniger	Austrian Institute of Technology GmbH   AIT & BOKU University
Michaela Teich	Austrian Research Centre for Forests   BFW

**BOKU University** 

# **Chapter Scientist**

Maria Papathoma-Köhle

# **Review Editors**

Carolina Adler Christian Rixen Mountain Research Initiative & University of Bern Swiss Federal Institute for Snow and Avalanche Research | SLF

# **Liaisons from AAR2 Technical Support Unit**

Anika Donner	University of Innsbruck   UIBK
Kay Helfricht	University of Innsbruck   UIBK

# **TABLE OF CONTENTS**

Exec	cecutive summary		461		
7.1.	Chapt	er introduction	463		
7.2. Delineation and differentiation of the					
	Austri	an Alps	464		
	7.2.1.	Natural boundaries	464		
	7.2.2.	Socioeconomic settings	465		
7.3.	Manif	estation of climate change in the			
	Austri	an Alps	466		
	7.3.1.	Past climate change in the Austrian Alps.	466		
	7.3.2.	Future climate change in the			
		Austrian Alps	467		
7.4. Impacts of climate change on the					
	Austri	an Alps	469		
	7.4.1.	The Alps as natural environment	469		
		The hydrosphere and cryosphere	469		
		The (terrestrial) biosphere	472		
		Natural hazards	475		

7.4.2.	The Alps as a provider of services, goods	
	and livelihoods	479
	Forestry and agriculture	479
	Energy	482
	Tourism	485
7.4.3.	The Alps as home to communities	488
	Demography and settlements	488
	Mobility and transport infrastructure	490
7.5. Needs for resilient mountain regions in		
Austria		493
7.5.1.	Monitoring of natural processes/hazards	
	and provision of climate services	493
7.5.2.	Transformation	494
7.5.3.	Non-structural adaptation measures and	
	nature-based solutions	495
7.5.4.	Structural adaptation measures	495
References		
Glossary	<u>aar2.ccca.ac.at/glo</u>	ssary
List of abbreviations <u>aar2.ccca.ac.at/abbreviations</u>		
Full report.	<u>aar2.ccca.ac.at/re</u>	port

#### **EXECUTIVE SUMMARY**

By 2024, the temperature in the Austrian Alps had increased by 3.1°C compared to pre-industrial times (*high confidence*). It is projected to keep increasing in the future, conditional on future emissions (*high confidence*). While past and future temperature increases within the Austrian Alps are spatially relatively homogeneous, changes in precipitation vary from region to region (strongest increase of 10 % for median precipitation in the north in winter and almost no change in summer for GWL 4.0°C) (*medium confidence*). {7.3.1, 7.3.2}

In the past (instrumental period), warming in the Austrian Alps was almost identical to the rest of Austria (high confidence); for the future, model simulations show a slight elevation dependent warming (medium confidence). In an Austria-wide comparison, the role of the Alps as sentinels of climate change is thus hardly evident in their specific warming trend, but rather in the change in specific impacts such as the strong decline of the cryosphere in response to warming. The following changes in the cryosphere characteristics of the Alpine landscape are directly related to climate change: (i) Decreasing snowfall and snow on the ground (about 3 cm/decade decrease in mean snow depth); (ii) About 7 days/decade decrease in snow-cover days since 1961 and further decline by 10-15 snow-cover days under GWL 2.0°C/60-80 snow-cover days under GWL 4.0°C) (robust evidence, medium agreement); (iii) Glacier retreat (40 % of glacier area in Austria were lost between 1969 and 2015 and glaciers will completely disappear under GWL 4.0°C) (high confidence); (iv) Increase in permafrost temperature (e.g., +0.1°C/year at 3 m depth since 2016) and widespread permafrost degradation (low confidence). These changes will further exacerbate alpine and downstream droughts, challenge tourism and alter the ecosystem services of the Alps.  $\{7.3.1, 7.3.2, 7.4.1, 7.4.2\}$ 

The role of the Alps as a water tower for Austria and even more for Central Europe is evident from the climatic water balance (which is positive by average of 466 mm for the period 1981–2010) (*high confidence*) and from the comparison of the runoff contribution to the area-related mountain share of individual river catchment areas. Such a comparison shows that the runoff contribution clearly exceeds the area share (33 % to 17 % for the entire Danube catchment, 38 % to 21 % for the entire Rhine catchment) (*high confidence*). As a result of future climate change, a positive shift in the climatic water balance in Austria is anticipated ( $+107\pm56$  mm) under GWL 2.0°C. The role as a water tower will increase on average, but the increasing risk of droughts will lead to a stronger negative climatic water balance for subperiods within a year (*medium confidence*). {7.4.1}

In recent decades and in the future, the hydrological cycle in the Austrian Alps has been and will be characterized by changes in snow depth, earlier and increased snowmelt (*high confidence*) and an increase in extreme precipitation events on the time scale of hours (thunderstorms) in summer (*robust evidence, medium agreement*). As a result, summer floods in small catchments in the Alps have increased and will continue to do so in the future, while floods caused by heavy spring snowmelt are increasingly shifting to winter (*high confidence*). Increasing glacier loss exacerbates low flow conditions in glacierized catchment areas in the Alps in summer, however, with a decreasing contribution in the future due to the disappearing glaciers (*high confidence*). {7.4.1}

There is evidence that some natural hazards in the Alpine region will become more frequent and more intense in the future, such as wildfire and fluvial flooding (high confidence), torrential flooding (medium confidence) and pluvial flooding (low confidence). For fluvial floods, the trend is more pronounced in smaller catchments than in larger ones. A clear seasonal shift from winter and spring fluvial floods to summer floods is detectable (high confidence). Expected changes in the components of the hydrological cycle as the main driver of slope failure will facilitate the occurrence of landslides. This will affect alpine communities by causing major economic losses, specifically as the built environment and land use expand. Furthermore, it will affect the transport infrastructure and services, and thus the mobility of the population and the accessibility of mountain communities, and it may interrupt pan-European transport routes. {7.4.1, 7.4.3

There is evidence of an upward shift of plant and animal species due to warmer conditions (*high confidence*). Therefore, cold environments above the tree line will shrink, leading to a loss of biodiversity. Large-ranged, warm-demanding species will replace short-ranged, cold-adapted species, some of which are endemic and restricted to the Austrian Alps (*medium evidence, high agreement*). There is an increasing risk of mismatching of interactions between groups of organisms, e.g., plants and pollinators, and of physiological stress of high-elevation species due to warming (*medium confidence*). {7.4.1}

Climate change has a negative impact on the condition and availability of transportation infrastructure in Alpine regions (high confidence). An increasing frequency of high temperatures can cause physical deterioration of traffic infrastructure, including railway tracks and road tarmac. This phenomenon is particularly prevalent in mountainous regions especially in tight bends, leading to (thermal) overload of the infrastructure. This is exacerbated by the increase in heavy goods traffic, growth of tourism and local travel. Congestion increases due to the limited availability of alternative routes in mountainous areas and emissions increase significantly as a result. Adaptation can be achieved by shifting transit and holiday traffic to rail by, e.g., robust accompanying planning and policy measures for the Brenner Base Tunnel, making public transport more attractive, improving services for carless travel, and reducing traffic-related pollutant emissions through alternative drive systems. {7.4.3}

Winter tourism based on snow and ice sports will be severely affected by shortened ski-seasons and deteriorating snow-making conditions (*high confidence*). Without adaptation and transition to snow-independent tourism, this could lead to significant economic losses and unemployment in regions with high climate-related risks (*medium evidence, high agreement*). Job losses in regions dependent on winter tourism could increase commuting and out-migration. {7.4.2} Summer tourism may benefit from more pleasant temperatures compared to the lowlands and urban areas, but risks and costs from natural hazards for hikers and the hiking infrastructure are likely to increase (*medium evidence*, *high agreement*). The maintenance of hiking and climbing infrastructure is currently mainly provided by alpine NGOs. If increasing costs for maintenance cannot be handled by NGOs anymore, other stakeholders such as destination management organizations or regional governments are required to step in to prevent a decline of summer tourism's high quality and reliability. {7.4.2}

Rising air temperatures and more frequent and severe drought periods will lead to shifts in tree species distribution and forest composition, which may have negative consequences for the protective function of forests at elevations <1,000 m a.s.l. This will happen in parallel with other natural forest disturbances caused by wind, fire, pests and insects (e.g., bark beetle) leading to a decrease in the protective function of forests at higher elevations (high confidence). Tangible and intangible losses to infrastructure, buildings and human lives may result because of increasing hazard potential downslope of a degenerated protection forest. Furthermore, secondary hazards may develop in affected areas (such as snow avalanches and debris flows), and soil loss and increasing surface runoff may be observed. Post-disturbance management decisions can have an important impact on the protective effect of forests. If left in the stand, deadwood could maintain the protective function of forests after windthrow and bark beetle disturbances, especially during the first 15 years. To better adapt to these threats, uneven and multi-layered stands with trees of all sizes and age classes, and a minimum canopy cover of about 40 %, only small openings and a sufficient presence of natural regeneration should be considered. {7.4.2}

# 7.1. Chapter introduction

The European Alps are the largest mountain range in Europe, covering more than 190,000 km<sup>2</sup> and are home to 14 million people, more than 30,000 animal species and 13,000 plant species (Permanent Secretariat of the Alpine Convention, 2019). Eight countries share the Alpine area and according to the Alpine Convention, Austria is the country with the largest share of the Alpine area (28.7 %) (Salto, 2024). The European Alps are often referred to as the 'water towers' of Europe as they supply four large European river basins (Danube, Rhine, Rhone and Po) with water (Permanent Secretariat of the Alpine Convention, 2009). Consequently, the European water balance is dependent on the Alpine water storage volume. Moreover, the European Alps contain unique landscapes, diverse cultural heritage, ecosystem services, they are an important tourist destination, and a source of livelihoods. Since a significant number of studies suggest that the European Alps are disproportionately affected by climate change, the challenge is to strike a balance between promoting economic development and, at the same time, preserving environmental quality (Alpine Convention, 2015).

Chapter 7 focuses on the Austrian Alps (hereafter 'the Alps') and the impacts of climate change on them, as this region not only offers special lifestyle opportunities for its

inhabitants but is also of great importance for Austria as a whole. Thus, the Alps are examined on three levels: As a natural or near-natural environment, as a provider of services, goods and livelihoods, and as a home to communities. The following two hypotheses form the backbone of the assessment in Chapter 7 and are explored throughout.

- Climate change in the Alps shows a different intensity compared to the foreland, as suggested by various processes (e.g., albedo feedback, water vapor effect, cloud formation). These processes will continue to have an impact in the future, making the Alps particularly sensitive to climate change.
- Due to their special socioeconomic features, their dependence on certain sectors, and the national and European significance of the region, the Alps have special vulnerabilities but also capacities to adapt to and mitigate these changes compared to the rest of the country.

This chapter aims to describe the manifestations of climate change in the Alps and its impacts on natural hazards, biodiversity, demography, settlements, mobility, and ecosystem services, and services and goods (forestry, agriculture, tourism, energy). Existing mitigation and adaptation options are presented and the needs for further adaptation and mitigation are outlined. First, the area of interest of Chapter 7



Figure 7.1 Schematic and simplified North-South cross-section through the Austrian Alps at approximately Salzburg with information on altitudinal vegetation belts, annual mean air temperature and mean annual precipitation. The climatic snowline and tree line are higher in the Inner Alps, because of lower precipitation and higher solar insolation.

(7.2.1) together with the demographic and spatial setting (7.2.2) are presented. The manifestation of climate change in the Alps is then shown (7.3). The Alps are then addressed as a natural environment (7.4.1), as a provider of services and goods (7.4.2), and as a home to communities (7.4.3). Based on these insights, adaptation and mitigation needs are identified and discussed in Section 7.5.

# 7.2. Delineation and differentiation of the Austrian Alps

This subchapter defines the area of interest (presents the criteria for such a definition) and describes its special characteristics in the context of climate change.

# 7.2.1. Natural boundaries

Austria has the largest share of the European Alps, as defined by the Alpine Convention, covering more than 65 % of the country. Regional subdivisions of the Austrian Alps, such as North, Central and South Alps further subdivide the Alps, but more important in the context of climate change is the vertical vegetation zonation such as the montane, alpine, subnival and nival zones as well as the tree line ecotone, and the expected changes due to the effects of climate change (Figure 7.1).

Natural boundaries can be used to delineate the Austrian Alps from the rest of the country. Delineation can be done using natural boundaries such as vegetation or climate boundaries, gradient thresholds, features, characteristics, topography, etc., but none of these seem to be universally applicable (and climate alone will not be sufficient to delineate the Austrian Alps). For an in-depth discussion see, for example, studies by Kapos et al. (2000), Elsen and Tingley (2015), Körner et al. (2021) and Price et al. (2022). Mountain topography has a high degree of complexity that affects climatic conditions. This includes elevation, slope steepness and slope direction, resulting in (steep) gradients of climate variables in both vertical and horizontal directions. Gradients can be observed in terms of, e.g., temperature, precipitation and solar radiation. Additional effects of mountain



Sources: data.gv.at, data-synergis.opendata.arcgis.com, www.atlas.alpconv.org (11.11.2024)

Figure 7.2 Delineation of the Austrian Alps according to usage in Chapter 7.

topography such as shading, clouds, surface material (bedrock, bare rock, vegetation of various types), local wind, and snow drift add to the observed variability. The lower human influence, compared to the Central European lowlands, together with the compression of thermal life zones along the elevation gradient, the diversity of microclimates and geology make the Alps an outstanding biodiversity hotspot (Smyčka et al., 2017). With about 4,500 species of vascular plants, they host about 40 % of the continent's native flora, 750–800 of which are alpine, including 270 endemic species (Ozenda and Borel, 2003).

In this chapter, the widely used delimitation according to the Alpine Convention is used (Figure 7.2).

#### 7.2.2. Socioeconomic settings

Although the Alps cover 65 % of the country, only 39 % of the population lives in the region. However, it needs to be considered that in most Alpine municipalities the potentially suitable area for settlements, traffic infrastructure and agriculture is less than 25 % of the total area (Permanent Settlement Area, DSR - Dauersiedlungsraum) (Wonka, 2008). This limitation has a significant impact on the development of settlements in Alpine areas. The DSR is significantly lower in Alpine federal provinces and municipalities than in non-alpine areas. In Tyrol, the DSR amounts to 12 % of the total state area of the federal province resulting in a concentration of settlement activities, major transportation routes and agriculture in a very limited space (Österreichische Raumordnungskonferenz, 2023a). Related to the DSR, population density in many municipalities in the main valley areas exceeds 250 inhabitants/km<sup>2</sup> (Österreichische Raumordnungskonferenz, 2020). According to the urban-rural typology of the European Union, a population density of more than 300 inhabitants/km<sup>2</sup> is considered urban (European Union, 2010). Thus, agglomeration effects are not limited to alpine cities but also occur in ribbon-like settlements in the main valley areas. Given the significant demand for housing and commercial space, the limited amount of land available for development also affects housing density, the proportion of building land and subsequently the amount of land take and soil sealing (Österreichische Raumordnungskonferenz, 2023b). However, both the amount of building land per capita and the amount of sealed area per capita, are significantly lower in Alpine federal provinces and districts (Österreichische Raumordnungskonferenz, 2016; Umweltbundesamt, 2023a), indicating a more efficient use of building land there (Österreichische Raumordnungskonferenz, 2021). This is attributed to the comparably higher building land prices in many Alpine regions, which show significant peaks in urban centers and distinct tourist areas (Statistik Austria, 2023a). In addition to the influence of limited development space, high land prices in Alpine valleys can also be regarded as the result of increasing land competition there.

The Austrian Alps are characterized by different settlement structures. While most of them can also be found outside the Alps, ribbon-like settlement structures with high population densities in the main valley areas (e.g., Rhine valley, Innsbruck urban region) (Schindelegger et al., 2022), scattered settlements on slopes and hillsides (permanent and recreational) as well as tourist resorts with urban characteristics, can be regarded specific to Alpine regions. Major differences compared to settlement development outside the Alps are caused by specific framework conditions including the limited area suitable for development, topographic barriers to infrastructure provision, the potential exposure to various natural hazards (Fuchs et al., 2015b; Meyer and Job, 2022) and of local significance high development pressure from tourism. These conditions also have a significant impact on the transport systems and, thus, the mobility patterns in the Alpine areas compared to the rest of Austria (Tomschy et al., 2016).

Demographic changes, including changes in population and households, reflect existing structural and economic disparities among Alpine regions. Furthermore, demographic changes are a major driver of settlement and infrastructure development. Available data on population development for the period 2011-2021 (Österreichische Raumordnungskonferenz, 2023c), indicate a divide in the Austrian Alps with a population increase in the western part (i.e., Vorarlberg, Tyrol except for East Tyrol, Salzburg except for Lungau) and population decline in the eastern and southern part. Population data at the municipal level for 2011-2021 indicate population growth in all urban regions and, for the western part of the Austrian Alps, also in the main valleys. In the western part of the Austrian Alps, municipalities in side valleys and peripheral locations show a smaller increase and in some cases a decrease in population. In the eastern part of the Austrian Alps, population growth is mainly limited to urban and regional centers. These regional migration trends from side valleys to main valleys and from rural regions to urban and regional centers have also been observed in previous decades (Österreichische Raumordnungskonferenz, 2023c).

Population growth in Alpine regions affects the growth in all settlement types, with a greater impact on the expansion of building land in ribbon-like structures. A decrease in population results in vacant buildings rather than a reduction in built-up areas.

Urban influences outside of cities and ribbon-like settlement structures can be found in distinct tourism locations where urbanization processes occur selectively, on a small scale and far from existing urban areas (e.g., Sölden, St. Anton am Arlberg), a phenomenon referred to as micro-urbanization (Chai and Seto, 2019).

These demographic and spatial characteristics of alpine areas together with natural features suggest that climate change may have different impacts that require alpine-specific adaptation and that the potential for mitigation may also differ from non-mountainous areas in Austria. These aspects are analyzed and discussed in the following sections.

An economic activity of above-average importance in the Austrian Alps compared to the rest of Austria is tourism. As landscape attractiveness and perceived naturalness are important resources for tourism (Romeo et al., 2021), the Austrian Alps offer good natural conditions for this economic activity. Abundant natural resources combined with limited economic alternatives in rural areas have led to the importance of tourism in the Austrian Alps. In Tyrol, the most tourism-intensive federal province of Austria, the contribution of tourism to the gross regional domestic product is 14.9 % if only direct effects are considered, or 19.7 % if indirect effects are also included (Fritz et al., 2021). Due to the importance of climate-sensitive natural resources, climate change is considered a major challenge for mountain tourism (Steiger et al., 2022) (*high confidence*).

The labor market in the Alps differs when it comes to employment opportunities in tourism. The alpine tourism labor market is characterized by high seasonality and many part-time jobs. Therefore, finding and keeping qualified workers is an evident problem in the hospitality industry in the Alpine region (Heimerl et al., 2020). While the demand is increasing, the seasonal labor force, which is needed for winter tourism, is decreasing (Humer and Spiegelfeld, 2020). Unattractive working conditions, difficulty in finding housing in ski resorts, labor law issues and climate change, leading to uncertainty and a possible shorter season are mentioned as the main reason for this development (ORF, 2023). Furthermore, the dismissal of seasonal workers at the beginning of the pandemic may have damaged the image of workplaces in tourism (Mayer et al., 2021). To counteract the shortage of staff, improving the working atmosphere, the attractiveness of the job and the payment were found to increase occupational commitment in response to staff shortages (Schwaiger and Zehrer, 2022).

# 7.3. Manifestation of climate change in the Austrian Alps

The manifestation of climate change in the Austrian Alps is presented in the following sections starting with a description of past climate change, followed by a subchapter focusing on the future climate change in the area.

## 7.3.1. Past climate change in the Austrian Alps

Past climate in the Austrian Alps has been described by observations since about 1770 (see Section 1.2.1), but observation networks at higher alpine sites are less dense and have been established more recently (e.g., Hochobir, 2,140 m a.s.l., in 1851). As a result of the ZAMG HISTALP-initiative (Geo-Sphere Austria, 2017) and later projects, the description of past climate in the Austrian Alps is based on an outstanding database and related analyses.

The **air temperature** in the Alpine region has increased relatively homogeneously in space by 2°C since pre-industrial times (3.1°C for year 2024, see Section 1.2.1), which is about twice as much as on the global scale (Auer et al., 2014; Olefs et al., 2021). Individual months have shown weak signals of an altitude dependence of the warming in the Alps, which, however, averaged out when analyzed on an annual basis (Auer et al., 2014; Tudoroiu et al., 2016).

In contrast to temperature change, precipitation change varied both seasonally and spatially in the Alpine region (also within Austria) (Auer et al., 2014; Olefs et al., 2021). A dichotomy (e.g., increase of precipitation in the north, decrease in the south in winter) (Auer et al., 2007) reflects the effect of the Alps as a barrier on atmospheric circulation and thus on precipitation. The accuracy of precipitation measurements decreases significantly with increasing altitude. A high-resolution climate model experiment (Ménégoz et al., 2020) forced by ERA-20C for the period 1903-2010, confirmed the precipitation increase in the northern Alps in winter and suggested accelerated increase at higher elevations, whereas drying in summer was decreased or even not significant at higher elevations. Overall precipitation trends in Austria vary seasonally and spatially, with a general wetting trend in the north in winter and only a weak trend or a general decreasing trend in the north and south in summer (robust evidence). Moreover, the wetting intensity increases

with altitude in the north in winter and decreases with altitude in summer (*low confidence*). Generally, the confidence in spatio-temporal trends of precipitation decreases significantly with altitude.

The increase in air temperature has led to a vertical shift of the zero degree line (temporal trends transferred from Switzerland, north: 133±40 m/decade, south: 118±37 m/decade, for the period 1984-2018) (Scherrer et al., 2021), as well as the snow line by 42±26 m/decade in the north and 88±44 m/decade in the south (Hu et al., 2020). In an earlier work, Hantel et al. (2012) estimated a temperature sensitivity of the median snowline of 166 m/°C for 1961-2010. This has significantly reduced the proportion of snow precipitation at high elevations in the Alps, especially in the summer. The associated effects on the energy balance at the Earth's surface (albedo lowering) have led to a significant increase in the snow and ice melt (snow cover, glaciers, permafrost) in the Austrian Alps (Olefs et al., 2021; Schöner et al., 2019) (for glaciers and permafrost, see also Section 7.4.1). However, the dependence of the winter snow cover was also strongly influenced by weather conditions and precipitation amounts. This means that the change in mean snow depth in Austria varied spatially. A significant decrease in the south (-12 cm/ decade) contrasted with only a slight decrease in the northeast resulting in an overall trend for Austria of -3 cm/decade for 1961-2020. Mean snow depth trends at low elevations have been driven by changes in air temperature, while those at the highest elevation sites of the Austrian Alps have been driven by precipitation changes (Schöner et al., 2019).

In summary, regarding the snow in the Alps, temperature increase has led to an increase of zero-degree line altitude by about 130 m/decade in the northern part of the Alps and by about 120 m/decade in the southern part of the Alps (derived for 1984–2018) (*medium evidence, low agreement*). Compared to the zero-degree line elevation the increasing trend of the snowline was much weaker for 1984–2018, with about 40 m/decade in the north and 90 m/decade in the south (*medium evidence, low agreement*). Snow depth has generally decreased in the Austrian Alps over the last 50 years but with clear spatial patterns (strongest decreasing trend in the south and non-significant trend in the northeast (*robust evidence, medium agreement*).

High-quality measurements of radiation in Austria have only been available since 2012 (ARAD Measurement Network) (Olefs et al., 2016). Measurements of sunshine duration (as a proxy for global radiation dating back to the 1880s in Austria) showed a clear increase since about 1980, especially in summer (see Section 1.2.1). Altitude-dependent effects are difficult to interpret due to data uncertainty and small differences in measurements. A recent study by Correa et al. (2022) detected a west-east trend in surface solar radiation in the Austrian-Swiss Alps. The trend was positive/negative in the east/west for the period 1980–1995, but negative/positive for the east/west for the more recent period of 1996–2015 (see also Section 1.2.1). According to Correa et al. (2022), trends of surface solar radiation at low elevations were likely driven by clear sky forcing, while at high-elevation sites cloud optical properties and surface albedo seemed to be the main drivers of the trends.

# 7.3.2. Future climate change in the Austrian Alps

Given the rugged terrain and correspondingly steep horizontal gradients, even the 'high-resolution' CORDEX simulations (see below) are too coarse to adequately resolve the relevant atmospheric processes over mountainous terrain like the Alps (Rotach et al., 2022). Climate simulations are therefore more challenging and have potentially greater uncertainties than those over flat terrain (Rotach et al., 2022).

The most comprehensive information on the possible development of climate variables comes from so-called ensemble simulations. For the Alpine region they stem from EURO-CORDEX (Jacob et al., 2020), an initiative under the umbrella of the World Climate Research Program of the WMO, in which coordinated numerical experiments are carried out. The coordination concerns the emission scenarios, the grid spacing, the time span, etc. so that the results are comparable. The latest EURO-CORDEX results are available for 3 scenarios (RCP2.6/4.5/8.5) and for medium resolution (about 50 km grid spacing) and high resolution (about 12.5 km grid spacing). For the Alpine region, an update and extension of the earlier work of Gobiet et al. (2014) and Smiatek et al. (2016) is presented by Kotlarski et al. (2023), with a special focus on characteristic areas (ALPS NW, ALPS NE, ALPS S - and all three considered together as ALPS). The Austrian Alps are best represented by ALPS NE, so the numbers given below refer to the results for this region, where available. In this study, 24 RCM simulations of high resolution for scenarios RCP2.6/4.5/8.5 are used and evaluated for the 'climate period 2070-2099' (as 'end of the 21st century'). Temperature, precipitation and snow cover (i.e., snow water equivalent, SWE) are considered. The results of these studies are translated to Global Warming Levels (GWLs) below.

Within the CORDEX framework, a Flagship Pilot Study (CORDEX-FPS) with convection-permitting (resolving/ allowing) or 'kilometer-scale' dynamical downscaling experiments (from about 12.5 km to a few km grid spacing) has been established (Coppola et al., 2020), which also has the 'greater Alpine region' as a focus. Due to the immense requirements of computing power, the resulting ensemble simulations typically consist of fewer members and are performed as 'time-slices' for the reference period ('current climate' 1996–2005), mid-century (2041–2050) and end-of-century (2090–2099) 10-year periods. Initial analyses focus on the validation (Ban et al., 2020, 2021, for precipitation; Soares et al., 2024, for temperature) of the current climate. Pichelli et al. (2021) is the first available study using the CORDEX-FPS to assess future climate change signals for precipitation in the Alpine region.

Single-model future climate simulations for the Alpine region at the km scale are furthermore available and typically focus on specific variables (Ban et al., 2015, 2020 – precipitation; Lüthi et al., 2019 – snow cover; Vergara-Temprado et al., 2021 – sub-hourly precipitation; Peleg et al., 2022 – spatial storm structure).

AAR2

In summary, the general characteristics of the future climate in the Austrian Alps are as follows:

Regarding the **temperature**, a consistent increase is expected in all regions, in all seasons (all models) (*high con-fidence*) (Figure 7.3). The magnitude strongly depends on the scenario. Temperature changes are elevation dependent: Simulation studies consistently find a stronger heating signal at higher elevations – but the details (seasonal behavior) depend on model choice and resolution (Lüthi et al., 2019) (*medium evidence, low agreement*).

Concerning **precipitation**, the mean annual precipitation amount does not show a clear signal because it is the result of a seasonal redistribution (Figure 7.4): A decrease in summer precipitation and an increase in winter precipitation (*medium evidence, high agreement*). The magnitude of the (seasonal) changes is strongly dependent on the forcing (strongest for RCP8.5, which corresponds to a GWL of 3.8°C for the shown period 2070–90, shown in Figure 7.4) and strong signals are only emerging in the second half of the century



**Figure 7.3** Synthesis for key variables of altitude-dependent climate change in the Austrian Alps for the future compared to the 1961–1990 reference (1984–2018 for the snow line). The altitude dependence of the air temperature for January and July is taken from Hiebl et al. (2009), with the variation within the Alps shown by the shaded areas in light gray. The median temperature increases for GWL 2.0°C and GWL 4.0°C were derived from Kotlarski et al. (2023), based on the EURO-CORDEX cmlP5 ensemble for the Alps. Days with snow cover (>1 cm) for 1961–1990 and for GWL 4.0°C are taken from Chapter Box 1.1 of this report (derived from the FUSE-AT project). The snow line estimates (period April to June) are satellite data analyzed by Hu et al. (2020), using the values for the Salzach catchment and the temperature sensitivity of the snow line from Hantel et al. (2012) to project the snow line for GWL 4.0°C. The equilibrium line altitude (ELA) of glaciers in the Alps for 1961–1990 and for GWL 4.0°C is based on the study by Žebre et al. (2021).



**Figure 7.4** Future precipitation changes in the Alps. The figure shows change of seasonal mean precipitation [%] between 1981–2010 and 2070–2099 in the three sub-domains and for the three emission scenarios RCP 2.6, 4.5 and 8.5. RCP 4.5 is equivalent to a GWL 2.2°C, RCP 8.5 to a GWL 3.8°C. Small dots refer to the individual simulations of the full model ensemble (EUR-11 and EUR-44, based on the EURO-CORDEX cmlP5 ensemble), bold dots to the multi-model ensemble mean. Colored bars indicate the ensemble uncertainty range as given by the lower (p5) and upper estimate (p95). The numbers below the panels refer to the percentage of simulations (rounded to full numbers) in the respective ensemble that shows a statistically significant change of seasonal mean precipitation (two-sided unpaired t test, p value of 0.05) (figure from Kotlarski et al., 2023).

(*medium confidence*). The winter increase is accompanied by a shift from solid to liquid, thus resulting in a reduction of 20–40 % in seasonal (Sept.-May) snow amount (Frei et al., 2018) (*limited evidence, medium agreement*). Summer precipitation is characterized by a decrease in the precipitation frequency and number of precipitation 'events', while the intensity of extreme events increases (Ban et al., 2020, 2021) (*medium confidence*).

Regarding the snow cover (measured as snow depth), there is an overall consistent decrease in seasonal snow cover, that is strongly scenario-dependent (strongest in RCP 8.5 of course) (high confidence). Future changes in snow cover are expected to be strongly elevation-dependent (high confidence). Changes for the Austrian Alps are assessed by the Fuse-AT project (Gobiet, 2021). Snow cover days (SCD) in Austria will continue to decrease by 10-15 days under GWL 2.0°C and by 60 to 80 days under GWL 4.0°C. A case study for Obergurgl (Ötztal) by Kotlarski et al. (2023), shows that in the scenario with the highest emissions, the reduction in the number of snow days at 1,920 m a.s.l. (Obergurgl) leads to a reduction in season length: -40 % by 2050, -80 % by 2100; in the ski area (i.e., at 2,500 m a.s.l.), the changes are -10 % by 2050, -35 % by 2100 (not including artificial snow production).

# 7.4. Impacts of climate change on the Austrian Alps

## 7.4.1. The Alps as natural environment

### The hydrosphere and cryosphere

A detailed assessment of the impacts of climate change on the hydrosphere, including runoff generation processes, magnitude, seasonality, and frequency of fluxes in streams, is presented in detail for Austria in Section 1.4. The most important impacts are briefly summarized here, where possible with a special focus on the Alpine region.

The role of the Alps as a water tower for downstream areas, even far away from the Alps, has been discussed for a long time, but it is only in the last decades that it has been thoroughly documented by quantitative studies (Viviroli et al., 2007; Rounce et al., 2023). For Austria and even more for Central Europe, the influence of the Alps on the water cycle is evident from the climatic water balance, which is positive with an average of 466 mm for the period 1981–2010 (Haslinger et al., 2023) (*high confidence*), and from the comparison of the runoff contribution to the area-related mountain share of individual river basins. Such a comparison shows that the runoff contribution exceeds the area share (33 to 17 % for the entire Danube catchment, 38 to 21 % for the entire Rhine catchment) (Messerli et al., 2004) (*high confidence*).

Hydrological regimes of mountain regions are expected to be strongly affected by climate change by 2050 and further altered by the end of the 21st century, as also shown by several studies for the Austrian Alps (Eitzinger et al., 2014; Nachtnebel et al., 2014; Laaha et al., 2016; Blöschl et al., 2018; Pistotnik et al., 2020; Schöner and Haslinger, 2020; Hanus et al., 2021). Generally, future surface water availability is characterized by both a more positive water balance and an increase in temporary droughts. Changes in precipitation and snowmelt are the main drivers of this future water balance alteration in Austria (Haslinger et al., 2022). Increasing air temperature leads to an increase in evaporation, exacerbating the summer low-flow situation. A more rapid decline in runoff is expected in the spring, resulting in an earlier start of the summer low-flow period. The winter months are characterized by an upward shift of the snow line elevation and a weakening of frost processes (high confidence), resulting in a higher low-flow discharge. Snowmelt is shifted towards winter, changing the annual runoff pattern (Blöschl et al., 2011a). Trend projections of runoff under different climate scenarios consistently show increasing low flows (Q95-values) of about 10-30 % by 2021-2050 (Blöschl et al., 2011a; Eitzinger et al., 2014; Nachtnebel et al., 2014; Laaha et al., 2016; Pistotnik et al., 2020; Schöner and Haslinger, 2020) (high confidence) and a decrease of the mean summer runoff of up to about 10 % by 2050 (Blöschl et al., 2011a, 2011b, 2018) (limited evidence, high agreement). These changes are expected to intensify by the end of the 21st century (Blöschl et al., 2011a, 2011b, 2018) (medium evidence, high agreement). Glaciers can significantly dampen the low-flow discharges during summer in the Alpine basins of rivers such as the Salzach, Inn or Ill (Koboltschnig et al., 2008; Koboltschnig and Schöner, 2011; Stahl et al., 2022) but with a decreasing meltwater contribution in the future, as many glaciers in the Austrian Alps have already passed their peak (Wimberly et al., 2024).

Recent updates of mean flow conditions for Austria, which represent the average water availability in the catchment as the difference between precipitation and evapotranspiration, have been provided by Duethmann et al. (2020). They found an increase in annual precipitation totals of approximately 120 mm over the last 35 years. This increase offsets a rise in evapotranspiration of approximately 110 mm over the same period, resulting in nearly unchanged mean runoff conditions. In particular, an increasing trend in mean flow conditions has been observed in the Alps. Regarding future conditions, Blöschl et al. (2017) and Haslinger et al. (2023) reaffirm the water balance assessment of the previous AAR14 report as largely valid. They anticipate an increase in winter runoff in the Alps due to a shift in snowmelt to winter and a higher proportion of liquid precipitation.

Flood events, triggered by significant and/or intense precipitation, combined with antecedent snow and wet conditions, as well as complex runoff generation processes, have increased significantly in 21 % of Austrian catchments for the period 1978–2020, especially in small catchments north of the Alpine ridge (Blöschl et al., 2017; Laaha et al., 2025). The trend averages +11.8 % in 43 years (+2.7 %/decade), with more pronounced values in the winter period (5.2 %/ decade; see Section 1.4.1).

Given the physical reasons for a 7 % increase in the intensity of heavy precipitation events per degree Celsius, the current positive trends in hourly precipitation totals or extreme precipitation (98th percentile) are likely to persist (see Section 1.2.1). While some of the heavy precipitation-related weather patterns are expected to become less frequent in the coming decades, they may produce heavier amounts when they do occur (Blöschl et al., 2017). However, all projections of heavy precipitation are highly uncertain due to limitations in the model representation of convective processes. Blöschl et al. (2017) summarize the projections of climate change on future floods as follows: (i) Seasonal changes in precipitation will hardly impact HQ100 except in areas with increasing intensity of convective summer events; (ii) An increase in intense convective events may potentially increase floods throughout Austria; (iii) There will be a minimal effect on floods from a rise in the snow line; (iv) Earlier snowmelt and the resulting lower summer runoff coefficient, as well as higher evapotranspiration, will slightly reduce flooding; however, (v) floods (HQ100) along the northern rim of the Alps show hardly any change. Systematic investigations of pluvial floods over Austria or the Alpine region are currently limited and certainly deserve further attention.

Low flow generation is primarily attributed to freezing and temporary snow storage during the winter months, contrasting with summer precipitation deficits in the lowlands below 900 m a.s.l. (Laaha and Blöschl, 2006). During summer heatwaves, the water sectors can suffer significant impacts, while potential threats to ecosystems exist, e.g., due to high water temperatures during low flow conditions (see Section 1.4.1). In recent years, low flow trends have become more pronounced, particularly in Alpine catchments. Drying trends, characterized by decreasing Q95-values (the 5th percentile of daily runoff), have been observed in the lowlands. Conversely, in alpine catchments above 900 m a.s.l., wetting trends with increasing low flow conditions prevail (Laaha et al., 2016) (Section 1.4.1). Moreover, the magnitude of mean low flows has intensified, with a decrease of 3.9 % per decade in lowland areas and an increase of 5.3 % in high-elevation catchments for the period 1977–2018 (Laaha et al., 2016; Blöschl et al., 2017). While future low flow scenarios for Austria yield different results, a significant increase of 10–15 % is projected for the Austrian Alps.

Changes in Austrian glaciers have been continuously monitored for more than a century documenting climate-induced glacier change in one of the longest records worldwide. Recent projections of future glacier change indicate a loss of 80-100 % of glaciers in the Central European Alps under warming scenarios between +1.5°C and +4°C (Rounce et al., 2023). Most of the ice loss will occur by 2050 and ice melt will increase significantly at elevations below 3,200 m a.s.l. (Zekollari et al., 2019; Sommer et al., 2020; Cook et al., 2023) (high confidence). Glaciers in Austria are located at lower elevations compared to the Western Alps (Switzerland, France) (Fischer et al., 2015; Sommer et al., 2020) indicating that their disappearance is expected to occur earlier than in the Western Alps (high confidence). Debris cover on glaciers can significantly delay the ice melt at individual glaciers and impact local changes in glacier runoff. Currently, debris cover on glaciers in Austria is increasing (Fleischer et al., 2021) (see Section 1.3.2).

The melting and disappearance of glaciers affect mountains in a variety of ways ranging from landscape changes, river runoff, slope stability, and biodiversity to tourism. Effects will occur at different temporal and spatial scales, and some impacts will show a transition towards new conditions once the glacier has melted. On-site impacts of glacier retreat locally modify previously ice-covered ground at high elevations. These impacts include the reworking of proglacial sediments and an increased sediment supply into rivers (Lane et al., 2017), the establishment of biodiversity through the colonization of ice-free ground, and the formation of new glacier lakes. The number of glacier lakes above 1,700 m a.s.l. has increased over the past century with formation rates increasing over the past 20 years (Buckel et al., 2018). This trend is expected to continue with lakes forming at higher elevations (Otto et al., 2022) (medium evidence, low agreement).

Glacierized catchments in Austria are considered to have passed the point of peak water beyond which runoff decreases due to decreasing glacier volume. For the major rivers draining the Alps (Danube, Rhine, Rhone, Po), peak water was reached in 2006 and glacier melt contribution to runoff has decreased since then (Huss and Hock, 2018). Ongoing glacier melt will lead to a seasonal shift of the glacier runoff peak from late summer to early summer towards the middle of the 21st century (Hanus et al., 2021). For example, the runoff contribution from glaciers in the Danube catchment is estimated to decrease by up to 60 % in August, September and October in 2090 (Huss and Hock, 2018).

Steep bedrock slopes and cirque headwalls may experience an increase in slope instabilities, such as rockfalls, as a result from the debuttressing of slopes and changing temperature conditions following glacier melt, as has been reported in a case study at Kitzsteinhorn (Hartmeyer et al., 2020). These slope instabilities may affect high alpine infrastructure and are likely to increase as glaciers continue to melt (*medium evidence, low agreement*) (Cross-Chapter Box 1).

Locally, rockfall events associated with glacier melt are interfering with changes in permafrost conditions. The distribution and temperature conditions of permafrost in Austria are largely unknown and limited to restricted areas (Schrott et al., 2012; Kellerer-Pirklbauer, 2019; Rode et al., 2020). Due to the diverse landscape patterns of the Austrian Alps, permafrost occurrence is highly fragmented and not exclusively limited to the upper altitudinal zones of the mountains. In the Hohe Tauern Range, permafrost occurs at elevations above 2,500-3,000 m a.s.l. on the northern and southern slopes (Schrott et al., 2012). Isolated patches of permafrost are reported from extreme locations at altitudes around 1,000 m a.s.l. (Stiegler et al., 2014). Borehole temperatures at permafrost sites throughout the European Alps show an increase in permafrost temperatures even at depths of 20 m (Biskaborn et al., 2019; Haberkorn et al., 2021). Existing borehole data from Austria report permafrost temperatures around -2 to -1°C at about 10 m depth (Schöner et al., 2012; Hartmeyer et al., 2020; Greilinger et al., 2023; Hartmeyer, 2023) and a near-surface increase of 0.1°C/yr (Hansche et al., 2023; Hartmeyer and Otto, 2024). Considering the ongoing rise of air temperatures, the permafrost area is likely to decline significantly (Chadburn et al., 2017; Magnin et al., 2017). Bedrock temperatures in Austria are likely to increase towards zero degrees in the coming decades (medium evidence, high agreement). In unfavorable locations, bedrock temperatures close to zero degrees

may lead to reduced slope stability, which favors increased frequency of rockfall processes and increased hazard potential at higher elevations (Haeberli et al., 2017) (*medium evidence, high agreement*). As glaciers continue to melt, water storage and discharge from permafrost landforms such as rock glaciers will become increasingly important. Rock glaciers in Austria are assumed to contain up to 2 km<sup>3</sup> of freshwater and are capable of storing additional resources for several months (Wagner et al., 2021). Rock glaciers are assumed to be less affected by warming due to the protective effect of the rock above the ice content (see Section 1.3.3 and Cross-Chapter Box 1).

# The (terrestrial) biosphere

Within the Austrian Alps, two main zones can be distinguished with regard to the predominant vegetation types and human land use: (i) The valley bottom and the adjacent cultivated grassland and forest area upslope to the climatic forest line, comprising the montane belt and the lower part of the subalpine belt (addressed in Section 2.2); (ii) The high mountain area from the tree line ecotone to the highest elevations, comprising the upper part of the subalpine, the alpine, subnival and nival zones (see below). For consistency reasons, only the area above the timberline is considered in this section, whereas forests and agricultural areas are dealt with in Section 7.4.2.

Due to their difficult accessibility and low temperatures, the Alps remain the only large more or less contiguous area of Austria, where ecosystems have not been completely or predominantly transformed into landscapes of intensive agricultural production, forest plantations, settlements, transport, commercial and industrial infrastructure. As the Alps cover more than 65 % of Austria, the country contains larger proportions of forested areas, mountain valleys, interspersed traditional cultural landscapes and naturally treeless alpine ecosystems than anywhere else in central Europe.

# *Effects of climate change above the tree line (high mountain area)*

The area above the tree line comprises the majority of unmanaged land in Austria, where settlements and traditional land use practices cease and give way to naturally treeless alpine landscapes. However, the area above the tree line in the Austrian Alps is only partly unmanaged and partly used for tourism and low-intensity livestock grazing, mainly in the lower alpine zone, but occasionally extending to above 3,000 m a.s.l. (Grabherr and Ringler, 2018). For the Alps as a whole, 29 % of its mountainous area lies above the elevation of 2,000 m and 10 % above 2,500 m a.s.l. (Körner, 2007). A similar proportion can be assumed for the Austrian Alps. On the European level, alpine ecosystems cover only 3 % of the continent, however, they are exceptionally rich in biodiversity. About 20 % of the continent's native vascular plant species have been reported to have their center of distribution in this area (Väre et al., 2003). In contrast to the Scandinavian mountains, which contain the largest proportion of alpine land, the Alps and the European mountains further south, host the largest proportion of endemic mountain species (Pauli et al., 2012). Austria, therefore, bears a great responsibility for the preservation of biodiversity in Europe's high mountain biome. Even if the immediate urgency of nature conservation measures in the high Alps is currently less pressing than in the small, fragmented and highly endangered remnant areas of the lowlands, the issue of nature conservation is a priority in alpine areas - particularly in the context of climate change. This is because climate change effects are pervasive and affect the most remote parts of the Alps. In addition, mountain areas such as the Austrian Alps will rapidly gain in importance as refugia for natural biota and in attractiveness for humans, who are increasingly suffering from a hotter and probably drier environment in the surrounding lowlands (Beniston et al., 2018). The low temperature regime, including a short growing season, is the strongest ecological factor determining ecosystems in the Alps and is the main ecological filter preventing the growth of woody plants (shrubs, trees) in the alpine zone (Körner, 2021). In addition, the amount of precipitation, especially the duration of snow cover, influences the position of the tree line. In the outer ranges, such as the northern, northeastern and southern Limestone Alps, the tree line is several hundred meters lower than in the central Alps due to higher rainfall and snow loads as well as fewer hours of sunshine (Figure 7.1 and Cross-Chapter Box 1). Climate change leads to changes in the distribution patterns of species, of their interactions and community compositions. Differences in species' abilities to cope with rapid climate change, such as different rates of movement, and capabilities to newly establish and compete with new neighbors, are expected to result in substantial biodiversity losses (Alexander et al., 2015; Lenoir and Svenning, 2015) (see also Cross-Chapter Box 1). Climate warming affects all parts of the Austrian Alps, regardless of their remoteness, by shifting the low temperature filter uphill and from southern to northern

slopes. This has several implications for biota living in, and in many cases confined to, mountains:

(i) An upward shift of vascular plant species leads to increased species numbers, but also to competitive displacement of dwarf-growing light-demanding alpine plants by taller plants from lower elevations. Accelerating rates of plant species colonization at summit habitats at high elevations have been found to be highly synchronized with rising temperatures. This pattern was consistent across temperate to northern Europe, including the Alps and the Arctic (Steinbauer et al., 2018) (high confidence). The warmer slopes (i.e., the southern and eastern sides of the mountains) showed the highest number of vascular plant colonizations (Winkler et al., 2016) (medium evidence, high agreement) and in the Eastern Alps movement rates of 17-40 m of upward shift per decade have been reported (Vitasse et al., 2021). Besides the previously observed upslope shifts of upper range margins of species (Pauli et al., 2012), a more recent study showed that the lower range margins of vascular plants and the optimal ranges of plant species have been moving upslope, at least as fast as the upper range margins in the Eastern Alps (Rumpf et al., 2018) (medium confidence). Species of other organism groups have also been found to shift upward (limited evidence, high agreement) such as butterflies, albeit with concurrent declines in abundance (Kerner et al., 2023), and fruiting bodies of fungi (Diez et al., 2020). See Section 1.6.4 for further references on the upward shift of different taxonomic groups in Austria.

(ii) Changes in species abundances are generally a faster response than actual range shifts, however, detailed historical data for comparison are scarce. An increasing abundance of alpine vegetation has been detected by remote sensing techniques, showing a greening (i.e., productivity gain) in 77 % of the European Alps above the tree line, while <1 % showed a browning (productivity loss) (Rumpf et al., 2022). Increasing greening can be caused by the promotion of more warmth-demanding and more vigorously growing plants at the expense of the cold-adapted species. This process, first described for alpine areas as thermophilization by Gottfried et al. (2012), has been repeatedly confirmed in the Austrian Alps (Lamprecht et al., 2018) (high confidence). Plant species adapted to very cold conditions (cryophytes) continued to decline in abundance (Lamprecht et al., 2018) (medium evidence, high agreement). This dieback may have been caused by overly warm conditions, leading to direct physiological disadvantages for the cryophytes due to their inability to down-regulate dark respiration under warmer conditions, leading to a rapid loss of carbohydrates (Larigauderie and Körner, 1995; Steinbauer et al., 2020) (*limited evidence, medium agreement*).

(iii) Further, a shift to longer growing seasons will change the duration of phenological development stages (Beniston et al., 2018). Leafing and flowering have occurred an average of 2.4-2.8 days/decade earlier over the past four decades (Vitasse et al., 2021), which can lead to both exposure to late frost events and mismatches in plant-pollinators interactions (medium evidence, high agreement). Earlier snowmelt is expected to have a strong influence on soil microbial life, because snowmelt triggers an abrupt transition in the composition of soil microbial communities, as has been confirmed by experimental snow cover manipulations in the Tyrolean Alps (Broadbent et al., 2021). This is closely linked to shifts in soil microbial functioning and biogeochemical pools and fluxes. Further, earlier snowmelt due to climate change could disrupt the synchronized seasonal dynamics of nutrient exchange between plant and soil microbiota, i.e., microbial organisms depend on carbon supply, plants on labile forms of nitrogen, which show strong seasonal variations (Bardgett et al., 2005; Bardgett and van der Putten, 2014) (limited evidence, high agreement). Some vertebrate species, such as the mountain hare (Lepus timidus) and the rock ptarmigan (Lagopus muta), undergo seasonal color change from brown to white for reasons of camouflage, which is mainly controlled by the photoperiod. Earlier snowmelt can, therefore, be detrimental for these species, as they become easily detectable prey (Zimova et al., 2018).

There is some debate regarding the extent and immediacy of the threat to high mountain species from climate change. Some authors argue that a rugged topography and the resulting variation in microclimate provide a buffer against the threatening effects of climate change, as plant and animal species may be able to find suitably cold refugia in nearby upslope or less sun-exposed locations (Körner and Hiltbrunner, 2021). Others point out that the high topographic variation and elevation differences in mountains have favored orographic isolation and thus the generation of many small-ranged endemic species (Dirnböck et al., 2011). In the Austrian Alps, endemic plant and animal species are concentrated in the subalpine to alpine zone of marginal mountain ranges, such as the northeastern Alps, which remained unglaciated during the Pleistocene, but where potential refugia are very limited due to lower elevations compared to the Central Alps (Semenchuk et al., 2021). Increasing warming-induced fragmentation and shrinkage of alpine areas and the associated competitive displacement of alpine plants by taller and more vigorous plants from lower

elevations, are therefore expected to accelerate extinction processes (Wessely et al., 2017). The generally slow-growing, but often long-lived, alpine plant species may persist in their increasingly climatically unsuitable habitats, but with declining populations until they reach their lifespan (Cotto et al., 2017), i.e., when their extinction debt is paid off (Dullinger et al., 2012; Alexander et al., 2018). In the case of the high central part of the Alps, from the Hohe Tauern westward, cold-enough refugia will be more abundant, however only in scattered patches where soil substrates are available (Kulonen et al., 2018). Moreover, even if such refugia did exist in nearby locations, the pronounced topographically determined variation in meso- and microclimates could form abrupt climatic barriers to the colonization of new habitats, e.g., toward a nearby north-exposed slope (Dobrowski and Parks, 2016). In this context, a recent study from the Stubai Alps suggests that the mesoscale topography and the overall elevation gradient are stronger determinants of plant distribution than the microscale (Chytrý et al., 2024), which does not support the hypothesis of climate change buffering effects through topographic variation. The role of the Alps as a biodiversity refugium and/or as a trap is an understudied topic (medium evidence, low agreement).

There is *medium evidence* but *high agreement* that shrub and forest vegetation will advance to currently treeless zones and thus outcompete dwarf-growing alpine species (Lenoir and Svenning, 2015). Yet, in the Alps, only a small percentage of forest advance could be attributed to responses to climate warming, rather than to land abandonment because actual tree lines are positioned well below their climatic limits (Gehrig-Fasel et al., 2007). Similarly, the recent expansion of shrubs in subalpine areas, mostly of green alder (*Alnus alnobetula*), was mainly explained as a consequence of land abandonment (Caviezel et al., 2017; Hohensinner et al., 2021).

The situation is different in the alpine zone. Repeatedly confirmed warming-related species enrichments and upward shifts of alpine plants (Matteodo et al., 2013; Winkler et al., 2016; Steinbauer et al., 2020; Nicklas et al., 2021; Vitasse et al., 2021) (*high confidence*) are of immediate relevance for nature conservation. Especially the alpine areas at the margins of the Alps, where many endemic species live, are of great concern because their fragmented alpine areas are already very limited in space. Due to the conical shape of mountains, shifts of the lower range margins of plant species to higher elevations, as detected in data from across the Eastern Alps (Rumpf et al., 2018) and confirmed for several other organisms (Rumpf et al., 2019), can lead to rapid declines and species extirpation (*medium confidence*).

Two other ecological factors may amplify the vulnerability of alpine biota to climate change. First, rising temperatures may lead to drier conditions even with unchanged precipitation, due to increased evapotranspiration. This would expose alpine plants that grow, for example, on the humid northeastern margin of the Alps, to a climate to which they are not adapted. Experimental studies in the Alps have shown that alpine plants are more affected by drought than their relatives at lower elevations (De Boeck et al., 2016; Rosbakh et al., 2017) (medium confidence). The second factor is the increasing availability of soluble nitrogen, which may favor colonization by more warmth-demanding plants from lower elevations, resulting in a competitive displacement of alpine species. An increase in widespread nitrophilous plants and a concurrent loss of small-ranged species has been observed in protected areas in lowland meadows, forest understorey vegetation as well as on alpine summits (Staude et al., 2022). While soluble nitrogen deposition from industrial agriculture is expected to be the primary cause at lower elevations, a combination of warming, leading to increased nitrogen availability and nitrogen deposition may explain the situation in alpine areas. Many parts of the Alps currently receive many times the pre-industrial rates of soluble nitrogen deposition, even in areas above the tree line (Körner et al., 2021), and alpine habitats are typically oligotrophic and, thus, sensitive to nutrient additions (medium confidence).

# Effects of climate change on conservation areas

The Austrian Alps host several protected areas including three national parks, three UNESCO biosphere reserves, two UNESCO geoparks and several nature parks, nature reserves (ALPARC, 2019a) and Natura 2000 areas. They are in a rather difficult situation, especially the smaller ones with limited vertical extent, in the face of the effects of anthropogenic warming and nutrient deposition. Refugial areas on the margins of the Alps are already very small, and regions where receding glaciers are opening up land for new colonization (Fischer et al., 2019) are usually far away from the areas of endemic species. Besides, the future area of glacier forelands will be too small (see Sections 1.3.2 and 1.3.3), with their lower parts located in alpine climates, and will therefore be rapidly colonized (Fickert and Grüninger, 2018). Progressive succession will lead to closed vegetation (Fickert, 2020), leaving no space for cold-adapted species (*medium evidence, high agreement*).

Nevertheless, the role of mountain protected areas will multiply in importance, especially if they are extended over large elevation gradients and connected to corridor areas (Elsen et al., 2018). In this way, the potential of the Austrian Alps as a top priority biodiversity refugium can be significantly enhanced. An intact biosphere in the Alps will be decisive for the sustainable functioning of its ecosystems, such as water supply, slope stabilization, biodiversity refugium, and recreation. As warming and drought influence human activities such as tourist infrastructure, hydroelectric, wind, and solar power development, careful consideration of their impacts in the context of the biodiversity crisis is essential. Traditional grazing with low livestock numbers is important for the conservation of species-rich pastures in the montane zone and may slow down uphill migration of tall and competitive plants in the lower alpine zone, but additional inputs of manure or other fertilizers can rapidly lead to species impoverishment (Grabherr and Ringler, 2018). Under changing climatic conditions, protected areas in the Alps need to be more flexible, adapted to the local situation including the local population, and large enough to effectively protect biodiversity (ALPARC, 2019b; Job et al., 2022). Maintaining or rewilding alpine ecosystems and restoring parts of the forest belt (Pereira and Navarro, 2015) will become particularly important in the course of an amplifying climate crisis.

## Natural hazards

The natural hazards described below are in line with Table 1.A.3 and Section 1.8. According to this table, rockfall and rock avalanches are particularly affected by climate change in high alpine regions. Furthermore, according to this table, all the hazards described below show an upward trend, except for snow avalanches, which show a diverse trend depending on the elevation.

### River and torrential flooding

In the coming decades, climate change-induced changes in river flooding in Austria are expected to be relatively small compared to the observed natural variability (Blöschl et al., 2018; Blöschl, 2020) (*high confidence*) (Section 1.4.1). Large-scale (stratiform) precipitation events (especially due to Vb cyclones) (van Bebber, 1891; Messmer et al., 2015), which lead to large-scale flooding events, are not expected to increase in frequency (Hofstätter et al., 2015; Blöschl et al., 2018; Blöschl, 2020), while there are indications that summer convective precipitation is expected to increase over the Alps (Giorgi et al., 2016). The intensity of short heavy precipitation events is expected to increase in summer (Ban et al., 2015; Chimani et al., 2015; Hofstätter et al., 2015; Formayer and Fritz, 2016; Blöschl et al., 2018; Blöschl, 2020; Hanus et al., 2021) (medium confidence). The existing trend towards increasing magnitudes of small to moderate floods outside the summer season (especially of winter floods; Section 1.4.1) as well as the lengthening of the season of summer low-pressure systems with particularly high precipitation, is expected to increase in the Austrian mountain regions in the future (medium confidence), as it is expected in the Swiss mountain regions and on a European scale (Beniston, 2006; Blöschl et al., 2011a, 2011b; Hall et al., 2014; Nachtnebel et al., 2014; Beniston et al., 2018; Brönnimann et al., 2018; Pistotnik et al., 2020; Hanus et al., 2021; Schlögl et al., 2021; Laaha et al., 2025). Equally, extreme floods (return periods >10 years) are expected to occur in the summer and increase in intensity (Brönnimann et al., 2018) (low confidence).

Concerning torrential floods, studies on the Austrian Alps and comparable Swiss studies (Stoffel et al., 2014; Heiser et al., 2019; Prenner et al., 2019; Schlögl et al., 2021), do not show a clear trend for damage-causing processes despite increasing exposure of elements at risk (Fuchs et al., 2015b), partly due to a compensatory effect of an increasing number of technical protection measures. At medium elevations, rain-on-snow events may become more frequent by the end of the 21st century (Stoffel et al., 2014; Stoffel and Corona, 2018; Prenner et al., 2019), extending the debris flow and flood season into the March-December period (Stoffel et al., 2014; Prenner et al., 2019; Hanus et al., 2021). The likelihood of critical triggering conditions for debris flows may vary regionally and seasonally, extending to earlier in the year (Stoffel et al., 2014; Hirschberg et al., 2021; Kaitna, 2022) (limited evidence, medium agreement). Early season debris flows are expected to remain small, given the current climate and sediment potential, and to increase in size as rain-on-snow events become increasingly relevant. Late summer season events are expected to become more critical due to the increasing sediment potential (Stoffel et al., 2014) (low confidence).

From a local to regional perspective, the sediment potential is expected to increase at high altitudes where (i) glacier retreat, (ii) permafrost degradation and (iii) increased physical weathering occur. At medium elevations, sediment potential may decrease due to weaker physical weathering, potentially causing decreased debris flow activity in sediment-limited catchments. The decrease in summer precipitation may increase sediment accumulation, potentially increasing the magnitude of debris flows (Lugon and Stoffel, 2010; Blöschl et al., 2011a; Sattler et al., 2011; Stoffel et al., 2011; Eitzinger et al., 2014; Gems et al., 2020; Kaitna et al., 2020; Hanus et al., 2021; Hirschberg et al., 2021; Kaitna, 2022; Maraun and Jury, 2022) (*high confidence*). In larger catchments, the expected change in sediment potential is within the natural variation (Blöschl et al., 2011a) (*medium evidence, low agreement*).

Historical and future human-induced land use changes in mountain catchments are expected to have a major impact on flood generation processes. With the knowledge from currently available studies, a robust assessment of the contributions of land use and climate change to the alteration of the flood generation processes is not possible (Habersack et al., 2011) (*low confidence*).

#### Landslides

Evidence of the impact of climate change on landslide activity in Austria is mainly site-dependent, especially with respect to the altitude. Due to changes in permafrost conditions, increased glacier melt and destabilization of rock glaciers, local accelerations of slope movements are being recorded (Schoeneich et al., 2015; Hartmeyer et al., 2020; Hartl et al., 2023) (high confidence) (see also Section 1.7). In the lower regions, especially those not affected by permafrost and glacier retreat, there is no evidence yet of an increase in either the frequency or the magnitude of slide and fall events (Sass and Oberlechner, 2012) (limited evidence, medium agreement). Rock, earth and debris slide and fall occurrence, and velocities are, however, strongly dependent on groundwater flow, which has been numerically demonstrated (Zieher et al., 2017a; Schneider-Muntau, 2020; Schneider-Muntau et al., 2022) (high confidence). A direct correlation between precipitation and slide or fall occurrence has been often postulated (Walter et al., 2011; Zieher et al., 2017b, 2023; Stumvoll et al., 2020), although this correlation is difficult to assess and is only based on case studies due to the different duration of precipitation to groundwater flow (Gassner et al., 2015; Zieher et al., 2023) and the still unclear influence of long-term precipitation events versus short-term but extreme events (Offenthaler et al., 2020) (medium confidence) (see also Section 1.7). Recent research indicates that there may also be a significant difference between the influence of precipitation and snowmelt, with snowmelt appearing to have greater effect on creep velocity (Holzmann and Perzlmaier, 2022). However, a robust correlation between snowmelt, precipitation, and groundwater levels, which would fill the current gap in understanding the effects of climate change on slide and fall occurrence, has not yet been found.

#### Snow avalanches

The relationship between avalanche activity and climate change can be direct (effect of changes in snow and meteorological drivers on avalanche release and propagation on short timescales), or indirect, as a consequence of changes in land cover due to climate change. For example, the warming-induced rise of the tree line affects avalanche release (Eckert et al., 2024). With ongoing climate change, the types of avalanches as well as their frequency may change. There is high agreement with medium evidence that temperature increase will cause changes in the frequency, intensity, and types of snowfall (Hock et al., 2019). Depending on elevation, this will affect the quantity and quality of snow cover, leading to changes in the magnitude and frequency of avalanches. While it seems clear how climate change will affect mountain snow cover at lower elevations, changes above the tree line (1,800-2,200 m a.s.l. in the European Alps) are less well studied (Strapazzon et al., 2021) (limited evidence, high agreement). The effects of climate change on snow avalanches, therefore, remain vague, especially since most avalanche release zones in the Austrian Alps are located above the tree line. A common view, expressed with medium confidence in a recent IPCC special report (Hock et al., 2019), is that the number of avalanches and runout distances will decrease at lower elevations. Due to warmer temperatures, the snow volume may respond with a reduction of 90-50 % at mid-elevation sites (1,000-2,000 m a.s.l.) and 35 % at high-elevation sites over the next few decades (Keiler et al., 2010). Future snow avalanche hazards will depend on interactions between increasing air temperatures and, possibly, increasing precipitation intensities (Reuter et al., 2020). An increase in air temperature will raise the snowfall line, leading to more liquid precipitation and less solid precipitation (Blöschl et al., 2018) (medium evidence, high agreement). Furthermore, the duration of snow cover will be greatly reduced, mainly due to earlier spring snowmelt.

The number of studies focusing on the effect of future environmental change on the frequency and magnitude of snow avalanches is limited (Fuchs et al., 2015a). However, a few papers provide insights into the climatic control of snow avalanches (Eckert et al., 2010). For the period between 1950 and 2017, Eckert et al. (2024) reported a 19 % decrease in the number of avalanches in the French Alps, with a significant correlation with increases in mean winter temperature (R= -0.42, p=0.002) and decreases in snow depth at 2,400 m a.s.l. There is no doubt, that changes in temperature, precipitation (amount and solid/liquid share) and wind characteristics influence the structure and stratigraphy of the snowpack and, consequently, the release and properties of snow avalanches. On the one hand, this leads to a decrease in snow cover duration at lower altitudes (high confidence), and on the other hand, more precipitation - due to warming - is expected to negatively influence snow cover stability (medium evidence, high agreement) (see also Glade et al., 2014). The study by Eckert et al. (2010) focused on changing annual avalanche runouts in the French Alps and correlated them to climate variability using an advanced statistical framework. The results indicate no change in the mean avalanche runout altitude during the last 60 years, despite an increase in temperature. In contrast, it is reported that the release zones of avalanches migrated upslope during the period of 1850–1920, together with a more than sevenfold reduction in the annual number of avalanches, a severe shrinkage of avalanche size, and shorter avalanche seasons as well as a reduction in the extent of avalanche-prone terrain (Giacona et al., 2021). Corresponding to the high variability of snow depth and snow cover in mountain areas, possible effects on snow avalanche activity will cover a wide range from a decrease or an increase in occurrence to a shift from dry to wet snow avalanches (limited evidence, high agreement). Results from downscaled climate models under climate change scenarios coupled with snow models, calculated that spontaneous avalanche activity in the French Alps will decrease by 20 % and 30 % by the mid and late twenty-first century, respectively, relative to a 1960-1990 baseline. However, high winter avalanche activity is expected to increase in the mid and late twenty-first century relative to the same baseline, due to an earlier transition from dry to wet snow conditions (Castebrunet et al., 2014). This is consistent with studies by Zgheib et al. (2022) who reported a general increase in the annual avalanche frequency from 1946 to 2009 in the French Alps, and by Pielmeier et al. (2013) and Mayer et al. (2024a) for the Swiss Alps.

For the French Alps, it has been reported that despite the anticipated stability of precipitation sums, the interaction between temperature increase and topography will constrain the evolution of snow-related variables on all considered spatio-temporal scales. This will result in a decrease of the dry snowpack and an increase of the wet snowpack. Wet snow conditions are projected to occur at high elevations earlier in the season (Castebrunet et al., 2014). In AAR14 it was reported that a decrease in avalanche activity was expected, however, this decrease was not supported by data or other evidence (APCC, 2014). Following Castebrunet et al. (2014), a general decrease in mean (20-30 %) and interannual variability of avalanche activity is projected. These changes are relatively strong compared to changes in snow and meteorological variables. The decrease is amplified in spring and at lower elevations. In contrast, an increase in avalanche activity is expected at high altitudes in winter due to conditions favorable for wet snow avalanches early in the season (Castebrunet et al., 2014) (limited evidence, high

#### Glacial Lake Outburst Floods (GLOFs)

agreement).

The release of previously covered bedrock after glacier retreat leads to a debuttressing of slopes and the establishment of different thermal conditions in the rock. An increase in rockfall processes is reported for deglaciated bedrock slopes in an Austrian case study (Kitzsteinhorn) (Hartmeyer et al., 2020). Glacier retreat may also cause the formation of glacial lakes and landslides, in some cases with the potential for glacial lake outburst floods (GLOFs). GLOFs result from dam bursts or overtopping due to rockfall into the lakes. The most recent documented catastrophic GLOFs in the Austrian Alps occurred in the 17th-19th centuries with a maximum in the 1860s, when the surging Vernagtferner glacier repeatedly dammed lakes in the Ötz Valley in Tyrol (Hoinkes, 1969; Braun, 1995), producing several destructive GLOFs (Aulitzky et al., 1994; Embleton-Hamann, 2007). Even though the number of glacial lakes increased during the 20th century, none of the existing lakes are currently considered hazardous (Mergili et al., 2012; Buckel et al., 2018; Fuchs et al., 2022) (medium confidence). According to modeling studies, the number of lakes in Austria is expected to increase and changes in the hazard potential are possible, especially if future lakes form at higher elevations closer to steep bedrock slopes (Otto et al., 2022) (medium evidence, low agreement). Further studies on GLOF hazards in the Austrian Alps are currently lacking. Historical outlines of GLOFs are reported in Nicolussi (2013) and Richter (1889, 1892).

#### Wildfires

In the European context, wildfires are considered an emerging risk (see Cross-Chapter Box 1). The burnt area in Europe in 2021 is about 2.5 times higher than the area burnt between 2008 and 2020 (IFRC, 2021). It is particularly alarming that significant wildfires have recently occurred not only in countries of the European South but also in countries with a short record of wildfire events (e.g., Sweden, Austria). Wildfires in Austria are mainly human-induced (85 %) and related to tourism and recreational activities (Müller et al., 2020). Consequently, the wildfire risk in Austria is expected to change partly due to the expected increase of fire weather days (Müller et al., 2020) and partly due to the continuous development of the wildland-urban interface (Papathoma-Köhle et al., 2025) (see also Cross-Chapter Box 1, Sections 1.6.2 and 2.2.2). In more detail, fire weather is the combination of certain conditions of surface temperature, humidity and wind (Clarke et al., 2022). Based on these characteristics, fire indices are developed and used to predict wildfire danger. According to Arpaci et al. (2013), the Canadian Fire Weather Index (FWI) is the best performing FWI In Austria. Models developed during the FIRIA (Fire Risk and Vulnerability of Austrian Forests under the Impact of Climate Change) project (2011-2014), showed that not only the number of fire weather days in Tyrol will increase by more than 40 days by 2100 (2014-2100), but also that areas that have not experienced wildfires in the past will be at increased risk of wildfires in the future. Moreover, the fire seasons are expected to expand and shift throughout the year (Müller et al., 2020). Relatively recent wildfire events in Austria (e.g., Absam 2014, Lurnfeld 2015, Rax 2022), have shown that crown fires in combination with strong winds can challenge suppression efforts and can be highly dangerous to people, infrastructure and property, in addition to affecting the forests themselves. The term 'forest fire' which refers to wildfires affecting forests excluding grassland fires or other types of wildfires, is used elsewhere in this report (see Sections 1.6.2 and 2.2.2) since this is a very common type of wildfire in Austria. There is no study addressing wildfire risk only for the Austrian Alps, but it can be assumed that since the topography is one of the factors that influence the initiation and extent of a wildfire in mountainous areas (Airey-Lauvaux et al., 2022), and that the number of fire weather days are going to increase, the wildfire hazard will also increase, threatening settlements in the wildland-urban interface and ecosystems. On the other hand, significant changes are expected in the exposure of people and ecosystems. In more detail, the wildland-urban interface and intermix (WUI) in Austria, where people live very close to or within the forest, occupies 13.6 % of the country (Bar-Massada et al., 2023). Additionally, the number of people living in the WUI in Europe is expected to increase according to different climate change scenarios. For the 3°C global warming scenario (GWL 3°C), an increase of 24 % of the population living in the WUI and exposed to high to extreme fire danger at least 10 days per year is expected. This percentage is reduced to 7.9 % for 1.5°C global warming (GWL 1.5°C) (Feyen et al., 2020). According to Feyen et al. (2020), the vulnerability of the ecosystems in the WUI is also expected to increase due to the northward shift of European ecological areas.

#### Multi-hazards (compound events, hazard cascades)

Many hazards occur, coincidentally or not, simultaneously or sequentially (see also Section 1.8). In general, preparedness and adaptation to multi hazard events are a prerequisite for a resilient society (UNDRR, 2015). Risk assessment for single hazards should be extended to include secondary effects (floods, landslides) or processes occurring simultaneously (wildfires, heatwaves, droughts) (Terzi et al., 2019). There are a few studies focusing exclusively on the influence of climate change on multi-hazards (occurrence, frequency, magnitude) for the Austrian Alps, although multi-hazard events have occurred frequently in the region in the past (Hübl et al., 2005; Eder et al., 2006; Pfurtscheller, 2014). Stoffel and Corona (2018), focusing on the wider Alpine space, argue that future winters will be characterized by multiple events triggered by changes in precipitation and temperature. However, the interactions between different hazardous processes in the Austrian Alps have often been the topic of research studies. For example, Sass et al. (2012) investigated the effects of wildfires on other geomorphological processes in Tyrol, and Kaitna et al. (2023) investigated the influence of hydrometeorological triggers on the occurrence of debris flows in the Austrian Alps. Some studies focused on specific elements at risk, such as the road network (Oberndorfer et al., 2020), while others focused on multi-hazard exposure mapping in Austria (Fuchs et al., 2015b; Nachappa et al., 2020), or in the Alps, including part of the Austrian Alps (Pittore et al., 2023). Terzi et al. (2019) have developed a framework for climate change and related multi-hazard for mountain areas. Nevertheless, it is not clear from the literature whether mountain areas are more

susceptible to multi-hazards and if climate change will affect multi-hazards differently in mountain areas than in the rest of the country.

#### Implications for vulnerability and risk

In summary, changes in the frequency and intensity of natural hazards in the Austrian Alps must be considered in order to adapt to climate change. The risk to natural hazards is subject to change not only due to changes in the frequency and intensity of natural hazards in the Alps but also due to increased exposure (Fuchs et al., 2015b) and vulnerability dynamics, according to a number of publications (Papathoma-Köhle et al., 2021, 2022a, 2022b) (see also Sections 7.4.2 and 7.4.3). Specifically, around 5 % of the building stock in Austria is exposed to torrential flooding, with hotels and guest houses as well as agricultural buildings showing an above-average exposure. The total number of properties exposed to torrential flooding has risen by a factor of six, from 18,797 to 111,673 buildings, over the last century. Considering these temporal dynamics, it can be assumed that if construction activity continues in the same pattern as in the last 100 years, the exposed buildings will increase by 2 % per year (Fuchs et al., 2015b, 2017).

# 7.4.2. The Alps as a provider of services, goods and livelihoods

# Forestry and agriculture

#### Effects of climate change on forests

The Austrian Alpine area as delineated by the Alpine Convention is largely dominated by forests (Figure 7.5). Forests are divided into productive forests, recreational forests, and protective forests (BML, 2022). Timber production drives the forestry sector in Austria, which is mainly dependent on coniferous species that cover 53 % of the forested area, dominated by Norway spruce (42 %). Single-species spruce forests, established during the economic crisis following World War II, often exceeded ecological limits, resulting in 'secondary spruce forests' (BFW, 2022). Forestry is an important backbone of the regional economy in the mountain regions of Austria. It supports the local value chain of the bio-economy and provides energy in rural areas (Jandl, 2023). Due to climate change, a wide range of possible changes in forestry are expected, e.g., a change in tree species requires a change in wood processing technology (Pramreiter et al., 2023). However, recent research shows that the high proportion of



Sources: data.gv.at, data-synergis.opendata.arcgis.com, basemap.at (11.11.2024)

Figure 7.5 Protective forest in Austria.

private forest owners makes it difficult to estimate the implementation of necessary adaptation measures (Mostegl et al., 2019; Pröbstl-Haider et al., 2020a) (*limited evidence, high agreement*) (see Section 1.6.2). Effects of climate change on forestry and agriculture in Austria are discussed in Section 2.2. Here, the focus is on forests and agriculture in mountain areas. With regard to forestry, the focus in mountain areas is on protective forests.

Protective forests have the primary function of protecting people or assets from the effects of natural hazards (Brang et al., 2001). They are common in mountainous areas, and less common in low elevations (Figure 7.5) (Makino and Rudolf-Miklau, 2021). Strategic management concepts are implemented to sustainably maintain a desired forest structure that provides protection against gravitational and hydrological hazards (Perzl, 2014; Perzl et al., 2021; Teich et al., 2022; Moos et al., 2023; Gaube et al., 2024). Protective forests are uneven and multi-layered stands with trees of all sizes and age classes, a minimum canopy cover of about 40 %, only small openings and a sufficient presence of natural regeneration (Frehner et al., 2005; Perzl, 2008). However, species composition and specific silvicultural objectives for protective forest management are highly dependent on the natural hazard process while the assessment of the protective effects is still subject to uncertainties (Perzl et al., 2021).

Global warming poses several challenges for mountain forests. However, at high elevations (montane and subalpine forests), the pressures from climate change are not as manifest as in valleys. Temporarily, climate change even promotes productivity and the upward shift of the tree line. Over several decades, the positive effects of warming will be outweighed by the negative effects that reduce forest productivity (Bebi et al., 2009; Lexer et al., 2015; Irauschek et al., 2017a, 2017b; Ledermann et al., 2022; Gaube et al., 2024). In addition, mountain forests will be more frequently and severely affected by natural disturbances (Seidl et al., 2017). In a warmer world, pests and pathogens will become an increasing challenge (Netherer et al., 2015; Hlásny et al., 2019; Hoch et al., 2019; Steyrer et al., 2023). For example, epidemic bark beetle outbreaks are becoming more common as bark beetle habitat shifts upward due to rising air temperatures, causing increasing problems in high-elevation forests (Jakoby et al., 2019; Hallas et al., 2024) (high confidence).

Rising air temperatures and more frequent and severe droughts will lead to shifts in tree species distribution and forest composition (Hanewinkel et al., 2013; Mauri et al., 2022; Amt der Steiermärkischen Landesregierung, 2023a, 2023b). The dynamics of natural forest disturbances such as wind, fire, pests and insects are accelerated by climate change such as the ongoing unprecedented bark beetle outbreak in protective forests in Austria and other countries in Central Europe, which may result in a temporary or irreversible loss of their protective effects (Seidl et al., 2017; Albrich et al., 2020; Senf and Seidl, 2020; Moos et al., 2023) (high confidence). However, post-disturbance management decisions have an important impact on the protection. For example, remaining deadwood after windthrow and bark beetle disturbances, can have a positive effect on rockfall and avalanche protection, especially during the first 15 years (Wohlgemuth et al., 2017; Teich et al., 2019; Caduff et al., 2022). However, a careful balance between salvage logging to reduce bark beetle infestations, and salvage logging to maintain the protective function is of paramount importance and depends heavily on the local conditions as well as the severity and the stage of the disturbance.

### Mitigation

The role of forests in mitigating climate change has been analyzed in the Special Report Land Use (Jandl et al., 2024), including the carbon storage in the tree biomass and in soils, and its impact on the Austrian greenhouse gas (GHG) emissions budget (Gingrich et al., 2024). Since 1990, forests have compensated between 4 and 35 % (on average 15 %) of the GHG emissions of other sectors (Umweltbundesamt, 2023b). The mitigation role of forest management has been investigated in simulation projects (Braun et al., 2016; Weiss et al., 2020; Kraxner et al., 2024). The publications cover the entire Austrian forest and not only mountain forests. The simulations suggest that the climate change mitigation effect of forests is temporary. With sustainably managed forests in a regime of moderate climate change, the carbon pools in soils, biomass, and wood products can be maintained. Unabated warming trends threaten the sink capacity of forests. The scientific discourse on the future role of forests and forest management in mitigating climate change is compromised by opinionated views on the desirable intensity of forest management (robust evidence, medium agreement).

# Adaptation

Adaptation measures for forests in non-mountainous areas in Austria are discussed in Section 2.3. Here, the focus is on adaptation for forests in the Alpine region. Some measures to adapt forests to climate change are controversial. One aspect is to reduce the population of ungulates (red deer, roe deer, chamois), as natural regeneration of various tree species is low due to browsing damage (Reimoser, 2003; Mayer et al., 2017). A decade-long dialog between foresters and hunters has not resolved the issue (BML, 2022) (robust evidence, low agreement). Another controversy is the choice of tree species. Foresters assert the need to establish stands of non-native tree species in locations where they are promising alternatives to native species that are struggling, or for use in locations where native tree species or provenances are not competitive in a future climate (Hanewinkel et al., 2013; Baumgarten et al., 2024; Chakraborty et al., 2024) (see also Section 4.3). The option of introducing non-native tree species is heavily debated (Bauhus et al., 2021) (robust evidence, low agreement). A fallback position for unstable and particularly vulnerable forests is to reduce the rotation cycle to avoid premature stand collapse. Common ground is reached when mixed-species stands are to replace single-species stands. From an adaptation perspective, this strategy simply distributes the externally applied risk agents (fire, storm, bark beetle, endemic pathogens) among tree species with different traits. The expectation is that a disturbance event will not destroy the entire forest stand, but only some elements that are particularly vulnerable with respect to the relevant disturbance agent (Pluess et al., 2016; Pretzsch et al., 2017; Yousefpour and Gray, 2022) (high confidence).

Recommendations for the management of protective forests to cope with the impacts of climate change on their protective effects are site-specific. Studies examining different management interventions such as thinning and regeneration cuts in combination with different climate scenarios show contrasting impacts on the protective effects of forests against natural hazards, also suggesting strong local context dependencies (Moos et al., 2023).

#### Vulnerability

Disturbance-induced tree mortality and regeneration failure may seriously compromise the provision of protective services against avalanches, rockfall and landslides (Temperli et al., 2017). The exposure of communities, the built environment and infrastructure to natural hazards will eventually increase. Although protection forests may not be more vulnerable to climate change than other forests in Austria, alpine communities that depend on the protection of these forests are, therefore, more vulnerable than other communities in the country that are not protected by nature-based solutions that may be affected by climate change.

#### Effects of climate change on agriculture

Overall, the agricultural sector is characterized by a rather extensive production compared to the lowlands. The dominant form of agriculture in mountain areas is grassland. In Austria the grassland area has decreased from 1,812,380 ha (1990) to 1,611,521 ha (2022). During these 32 years, 85,927 ha were converted to forest, 54,972 ha to cropland, and 27,995 ha to settlements (Umweltbundesamt, 2024) (see also Section 2.2.1). As far as grassland and alpine/mountain pastures are concerned, climate change would allow a shift towards cropland. Economically, such a change in land-use reduces the importance of dairy production. Ecologically, a decline in soil carbon stocks and in plant diversity is expected (Bohner et al., 2017; Bohner, 2023). The boldest changes in agriculture are shifts in production areas for climatic reasons. The current main agricultural production area is located in the northeast of Austria. This area is expected to be subject to more frequent and intense drought events (Haslinger et al., 2019, 2022). Accordingly, the main agricultural area is expected to shift westward and upward where climatic conditions are more conducive to crop production (Haslmayr et al., 2018). However, future site conditions are highly uncertain. Longer and more frequent droughts will reduce the yield of grasslands and the quality of fodder (Bohner, 2022, 2023). There is greater certainty about future pest and pathogen pressure and the adverse effects of extreme climate events.

## Mitigation

Climate change mitigation in agriculture has some peculiarities. Carbon capture in the biomass will be of little relevance. A prominent role is given to consumer demands, or social transformation (APCC, 2023) (see also Sections 2.3.2, 2.3.4 and 2.A.9). Currently, a meat-producing farm uses more than 5 ha for corn production to supply feed. Additional feed can be purchased from other countries at an ecological cost (soy imports). The ecological footprint is therefore significant.

Grassland soils have higher soil carbon stocks than cropland soils (Baumgarten et al., 2024; Umweltbundesamt, 2024). Many realistic mitigation strategies aim to maintain rather than increase current soil carbon stocks (Zosso et al., 2023). Further mitigation strategies include the integration of landscape elements that increase the plant species diversity and the implementation of agroforestry. However, the increase in soil carbon stocks after the reforestation of extensively used grassland is low (Hiltbrunner et al., 2013; De Stefano and Jacobson, 2017).

#### Adaptation

Climate change will require agriculture to adapt. In some areas, technological adaptation will be required, such as irrigation of crops, in others different crop varieties and the introduction of non-traditional crops may be the solution. Soil protection and sustainable forms of soil management are well studied. The carbon sequestration potential of soils is not well understood. Soil processes leading to an increase in soil carbon stocks are usually slow. Many strategies have already been implemented in Austria. Further increases in soil carbon stocks are expected to be small (Tiefenbacher et al., 2021; Wenzel et al., 2022) (*medium evidence, high agreement*).

In Austria, permanent grassland soils are important for organic carbon (C) storage due to their high soil organic matter (SOM) stocks and large areas. Especially, subsoils have a high potential for long-term soil carbon storage (SCS). In permanent grassland, soil organic carbon stocks are controlled by climate, soil properties (particularly soil moisture status and soil thickness), vegetation type (root biomass) and management intensity. Plant roots, microbial necromass and C-rich organic fertilizers (cattle manure, composted cattle manure) are most important for soil organic matter accumulation, which is generally a slow process. In Austrian permanent grassland, soil organic carbon storage is highest at an intermediate management intensity (2-4 mowings per year, annual application of cattle manure or composted cattle manure). Management practices that increase annual soil organic matter turnover are more important than those that increase the soil organic matter stocks.

In permanent grassland, the following measures and strategies are important to maintain or increase soil organic carbon stocks (Bohner, 2023; Meyer et al., 2024):

- No conservation of permanent grassland to cropland;
- Conversion of temporary grassland to permanent grassland (no plowing of grassland);
- Maintaining moderate management intensity (2–4 cuts per year), avoiding further intensification in currently intensively managed grasslands;
- Moderate grazing intensity (no long-term overgrazing);
- Increasing root growth, root biomass and rooting depth in soils by reducing management intensity in intensively managed grasslands (particularly intensively managed pastures);

- Annual application of farmyard manure (cattle manure or composted cattle manure);
- Favoring plant species with large and deep root systems;
- Maintaining or increasing the groundwater table in grassland soils affected by groundwater (particularly clay and peat soils).

#### Vulnerability

The vulnerability of individual crops depends mainly on their water requirements. On the one hand, in droughtprone karst landscapes, crops with high water requirements are particularly vulnerable. On the other hand, sites with high water tables are less susceptible to drought and therefore more suitable for crops with higher water requirements. Providing water through irrigation is not realistic due to high investment costs and conflicting water interests of different groups. There are two additional issues related to the vulnerability of agriculture. First, the water supply for livestock in mountain areas, which may also be reduced in the future due to climate change, makes them particularly vulnerable compared to livestock in other parts of Austria. Second, the dependence of mountain farms on forests, which are also affected by climate change, makes mountain farming more vulnerable than in other areas of the country where this dependence does not exist. Conversion of land from agricultural use to alpine tourism to mitigate losses has been proposed as a solution but has been rejected by farmers (Wanner et al., 2021).

## Energy

This section describes the specific role of the Alps in the energy system. For more information on the overall energy system and its transformation, see Section 4.5. The Alpine region plays an important role in Austria's energy supply, mainly through electricity generation from hydropower reservoirs and pumped storage, run-of-river hydropower and wind power plants in exposed locations.

Solar power is currently less developed at higher altitudes, although there are potential benefits for the overall energy system from increased winter solar generation in the mountains compared to lower altitudes (Kahl et al., 2019; Đukan et al., 2024) (see Section 4.5.2). Hydropower pumped storage and reservoir power plants are located exclusively in Alpine regions due to the topographic requirements of storage power plants. Run-of-river hydropower plants are scattered throughout the country. However, the Alpine regions have a cascading effect on almost all hydro run-of-river hydropower plants in Austria due to the high water inflow resulting from the topographic setting, which determines their importance for the local and downstream hydropower utilization (Fatichi et al., 2015). In 2019, about 9 % of the installed wind power capacity in Austria was located in the Austrian Alps, with two-thirds of them located above 1,400 m a.s.l. (Austrian Wind Energy Association, 2019; APG, 2024). The majority of Austrian Alpine wind power plants are located in Styria (Austrian Wind Energy Association, 2019).

The literature base on renewable energy potentials for the Austrian Alps is limited. At the NUTS2 level, data availability is higher, with an approximation of the Alpine region in Austria using the states of Carinthia, Styria, Salzburg, Tyrol, and Vorarlberg. These federal provinces account for about 19-46 % of the technical wind potential (European Commission and Joint Research Centre (JRC), 2021; Kakoulaki et al., 2021), 39 % of the technical solar power potential, 40 % of the technical biomass potential (European Commission and Joint Research Centre (JRC), 2021), and 64 % of the technical hydropower potential, with the electricity demand of this area only accounting for around 37 % of Austria (2019) (Kakoulaki et al., 2021). In addition to the flat agricultural lowlands in northeastern Austria, remote locations in the Alps host the majority of Austria's wind potential (Brudermann et al., 2019). Generally, most of the wind potential in the Alpine federal provinces is located in non-alpine areas. However, if only the alpine areas in those federal provinces are considered, exposed sites offer the best wind potential (Ciolli et al., 2015). One third of Austria's theoretical Alpine wind power potential is located in protected areas, and the potential outside of protected areas is mainly at relatively low altitudes (Balest et al., 2015).

The sustainable potential for renewable energy in the Alps is limited by the consideration of conflicting interests such as nature conservation (Balest et al., 2015) and economic activities in tourism (Brudermann et al., 2019). The potential wind power plant sites in the Alps are sparsely populated or uninhabited, so conflicts with residential structures are less pronounced than in the lowlands (Brudermann et al., 2019). However, wind turbines at high altitudes mean increased visual intrusion at longer distances, which can lead to reduced acceptance (Kraxner et al., 2015) and conflicts with nature conservation management strategies and recreational activities such as hiking, skiing or cycling, for locals and tourists (Brudermann et al., 2019). A quarter of the total theoretical potential for wind and solar power generation in the Alps in general is located in Natura 2000 areas (Balest et al., 2015). The wide range of protected areas means different administrative and management regulations, with different levels of allowance for renewable energy expansion, which must fit within the management objectives of the protected areas (Kraxner et al., 2015). These facts highlight the potential conflicts that may arise from the expansion of renewable energy production in the Alpine region and that need to be considered in climate change mitigation and adaptation measures.

Electricity from the Alpine area is largely supplied to the neighboring urban areas due to the higher concentration of electricity demand (Kakoulaki et al., 2021). The energy demand in Alpine regions differs from other parts of Austria in that tourism is more important and snow management plays a significant role in the energy demand of certain regions (see Section 7.4.2). This has implications for the energy demand profile, such as seasonality or concentrated peak demand corresponding to the ski season. From a tourism perspective, there is a growing demand for energy for tourism infrastructure, especially for ski resorts and large wellness facilities, as well as for charging electric vehicles of visitors/tourists at the destination. However, the acceptance by tourists, who expect pristine tourism landscapes, is rather low, especially when alternative destinations are available (Broekel and Alfken, 2015). They would prefer small-scale solutions using existing infrastructure (Jiricka-Pürrer et al., 2019). Therefore, the trade-off for tourism destinations is rather difficult.

## Effects of climate change

A potential impact of climate change on energy supply from the Alpine region could occur through reduced precipitation as snow, recurrent droughts, and water scarcity (see Section 7.3). The effects of climate change are expected to lead to a change in the annual production of renewable energy and to changes in seasonality (see Section 4.6.2). The effects of climate change on biomass, which is mainly relevant for the heating sector (see Sections 3.3 and 4.5.2), are described in Section 7.4.2.

# Mitigation

Transforming the energy system towards a climate-neutral system is a key requirement for mitigating climate change. Due to the current potential for renewable energy production and storage in the Alps, this region can make a significant contribution to these efforts. The aim of the Alpine Convention is to make a long-term contribution to meeting Europe's energy needs while limiting the negative effects of power plants on the environment and the landscape (European Union, 2005). Specific challenges for renewable energy expansion and operation in Alpine areas are steep slopes and a lack of connectivity to electricity and road networks (Kraxner et al., 2015).

Alpine regions can support the renewable energy transition, especially by providing flexibility to the overall electricity system through hydropower (pumped storage), which helps integrating high shares of highly volatile renewable energy sources into the energy system (see Section 4.6.2). Austria's topographical situation is an advantage in this case, as hydropower (pumped storage) can support the security of the energy supply in Austria as well as in the European interconnected electricity system.

For mutual benefits, it is important to consider the compatibility of biodiversity conservation and climate change mitigation through renewable energy use which has been demonstrated in best practice examples (Brudermann et al., 2019). It is unclear how sensitive the attractiveness of tourist destinations is to the installation of large-scale energy production facilities. Surveys show high levels of opposition, especially to wind turbines and large biomass or photovoltaic plants. Existing research suggests a trade-off between sustainable energy supply and tourism product development (Jiricka-Pürrer et al., 2019) (*limited evidence, high agreement*). Whether and under what conditions guests' attitudes toward sustainability will change in the long term is an open research question.

#### Adaptation

Climate change affects energy supply and demand and poses risks to infrastructure, making it important to consider these factors in adaptation of the alpine energy system (see Cross-Chapter Box 1). Possible hazards for Alpine energy infrastructure include glacier retreat, permafrost reduction, heavy precipitation and thunderstorms (Pröbstl-Haider et al., 2020b). In addition to the general trends of climate impacts on energy systems (see Section 4.6.2), the energy infrastructure in the Alps is confronted with rockfalls and rock and snow avalanches. Melting glaciers have a dual impact on local renewable energy production. First, the challenge of steep slopes for alpine (energy) infrastructure may be exacerbated by slope instability due to glacier melt (see Section 7.4.1). Second, local hydropower generation will be altered in terms of total production and seasonality which will then have cascading downstream effects on the entire energy system (Wagner et al., 2017; Schöniger et al., 2023; Wechsler et al., 2023). Adaptation strategies can increase the adaptive capacity of water governance and management systems (Maran et al., 2014). Hydropower plant design discharge optimization can partially compensate for climate change-related power generation losses, with greater design optimization potential in summer than in winter (Wechsler et al., 2023). However, the overall adaptive capacity of alpine hydropower systems to climate change is assessed as being rather low (Anghileri et al., 2018; Soomro et al., 2024), especially in catchments dominated by ice melt (Anghileri et al., 2018).

#### Vulnerability

Concerning hydropower, there is *low agreement* in literature as to whether climate change impacts lead to an increase or decrease of total generation, with a strong dependence on the climate scenarios chosen and significant geographical differences (see Section 4.6.2). The majority of currently available studies project a small decrease in the national mean annual hydropower potential by the end of the 21st century (for different GWLs) (Kranzl et al., 2010; Blöschl et al., 2011a; Eitzinger et al., 2014), with more recent climate models and scenarios showing a slight increase (Formayer et al., 2023) (medium evidence, low agreement). Wagner et al. (2017) predict a range of average annual changes in electricity generation in the Alps between +5 and -8 % for the period 2031-2050 in different scenarios/GWLs compared to no climate change impact. Wechsler et al. (2023) find a decrease of annual run-of-river generation in the Swiss Alps of about 2-7 % by the end of the century. The impact of climate change on run-of-river hydropower production depends strongly on the elevation of the hydropower plants with production increasing for higher elevated power plants (Wechsler et al., 2023) (limited evidence, medium agreement).

There is a *high agreement* in the literature on a seasonal shift of hydropower production from the summer to the winter season with increasing climate change (see Section 4.6.2) due to changes in precipitation/snowfall and earlier runoff due to warmer spring temperatures (see Section 1.4.1) (*medium evidence, high agreement*). This pattern is even more pronounced for the Alpine region than to Austria as a whole (see Figure 7.6). The projected seasonal shift in hydropower generation corresponds to the projected increase in mean and low winter flows in the Alps (see Section 1.4.1). The projected decrease of hydropower potential in summer and increase in winter has been shown for the entire Alpine region (Wagner et al., 2017), the Austrian Alps (Blöschl et al., 2011a; Formayer et al., 2023; Schöniger et al., 2023), and the Swiss Alps (Finger et al., 2012; Wechsler et al., 2023). A decrease in summer hydropower potential in mountain regions by about 2–22 % is expected (Blöschl et al., 2011a; Eitzinger et al., 2014; Wechsler et al., 2023) (*limited evidence, medium agreement*). For specific periods, a decrease of up to 70 % was found in the summer months of July and August for warm-dry periods by 2050 compared to 1961–1990 (Wagner et al., 2017). For Alpine winter runof-river generation, an increase of 4–9 % by the end of the century is reported (Wechsler et al., 2023).

Figure 7.6 compares the hydropower generation from run-of river-power plants in the Austrian Alps and all of Austria for the periods 1971–2005 and the 20-year period up to GWL 3.0°C, according to the considered climate scenario (Formayer et al., 2023; Schöniger et al., 2023). The monthly mean production for the periods 1971–2005 and GWL 3.0°C is compared with the mean generation of the entire period 1971–2005. Although weaker and stronger GWLs are not shown in the figure, they do show the same trend. (Pumped) storage power plants are expected to face these developments robustly, as their storage capacities buffer runoff fluctuations (Haas et al., 2008; Kranzl et al., 2010; Maran et al., 2014). However, the impact of glacier melt on the operations of reservoirs is still an open research question. For run-of-river hydropower power plants, climate-induced changes in hydrological regimes may require operational impairments and increased attention to structural and operational impacts of flooding (Habersack et al., 2011) (*low confidence*). Potential changes in sediment potential and transport dynamics further exacerbate the already existing challenges related to sediment management strategies at hydropower reservoirs (Habersack et al., 2011) (*low confidence*).

# Tourism

## Effects of climate change and vulnerability

Alpine regions account for 53 % of overnight stays in Austria during the summer season and 70 % during the winter season (Burton et al., 2024), making them the most important sub-sector of tourism in Austria. At the same time, tourism contributes a significant share to regional domestic products,



Figure 7.6 Changes in the mean run-of-river hydropower generation for the Austrian Alps and whole Austria for 1971–2005 and GWL 3.0°C (Source: Based on SECURES project runoff calculated by E-hype driven by RCP 8.5 RCM RACMO22E (KNMI, Netherlands) and GCM EC-EARTH (ICHEC, Ireland), see Formayer et al. (2023) and Schöniger et al. (2023)).

e.g., 19.7 % including indirect effects in the federal province of Tyrol (Fritz et al., 2021), and is the backbone of the regional economy in many rural Alpine municipalities (Steiger and Scott, 2020). As most tourism products in alpine tourism are dependent on natural resources (Steiger et al., 2020), climate change can have far-reaching consequences. While direct impacts refer to a change in the natural resources necessary for specific activities and tourism types (e.g., snow for skiing), indirect impacts refer to changes in landscape aesthetics (e.g., due to land use changes, shifting tree line, loss of glaciers) (Pröbstl-Haider et al., 2021). Snow- and ice-based forms of tourism are projected to face a higher risk of discontinued operation (Steiger et al., 2021a). However, the remaining regions that still provide reliable snow and ice conditions (whether due to natural or artificial snow and ice) could gain a competitive advantage as these resources become scarcer, which could even result in increasing tourism demand and pressure on tourism development (Steiger and Scott, 2020; Steiger et al., 2021a) (high confidence). In summer, the Alpine region is likely to benefit from the heat islands in the pre-Alpine area and urban areas, potentially stimulating summer tourism (Weber et al., 2017). In addition, summer tourism could benefit from warmer and more stable weather conditions, especially during periods of low demand such as the spring and fall months (Pröbstl-Haider et al., 2021). However, many summer tourism activities such as climbing, fishing, kayaking, hiking, and all types of aviation sports are affected by an increasing number of hot days and changes in the flow regime of alpine rivers (Pröbstl-Haider et al., 2021). Recent research reports an increasing safety risk on alpine trails due to thawing permafrost, heavy rainfall and thunderstorms (Pröbstl-Haider et al., 2020b) (limited evidence, high agreement). Human health problems (such as heat stress, allergies from neophytes, and deterioration of the water quality in lakes) are likely to increase in the Alps, but will occur less frequently than in the lowlands, representing a comparative advantage (Pröbstl-Haider et al., 2021).

There is a high likelihood that the costs for restoring and maintaining the extensive hiking and climbing infrastructure will increase (Pröbstl-Haider et al., 2021). The maintenance of this infrastructure is currently mainly in the hands of alpine NGOs. The demographic change on the one hand, and the increasing need for restoration and maintenance of the Austrian trail system in the mountains on the other hand, will require new adaptation strategies with significant financial support. Otherwise, the trail network which is the backbone of any kind of summer tourism will lose its current high quality and reliability. Exposure to dangerous situations could increase in the absence of professional monitoring (Pröbstl-Haider et al., 2016). For summer ski tourism, a strong decline between 2002 and 2019 was observed (-48.3 % in the summer half-year) (Mayer and Abegg, 2024). However, climate change impacts on glaciers and summer snow only explain part of this development, as societal trends (declining interest in summer skiing) also play an important role (Mayer and Abegg, 2024).

#### Mitigation

The biggest lever for reducing GHG emissions from tourism is mobility (see Section 4.3.3). Although rough estimates for tourism-related GHG emissions exist at the national level (Neger et al., 2021), there is a lack of more detailed monitoring of emissions at the subnational level that takes into account destination-specific factors such as the place of origin of tourists, the structure of the accommodation sector and the main tourist activities (high vs. low energy consumption). Opportunities to reduce GHG emissions in the tourism sector in the Alps differ from those in non-alpine areas in several ways:

Public transport to tourist destinations has been enhanced in several mountain regions in recent years, for example with night train connections from major source markets, such as Germany. While the German Railway (DB) discontinued its night train service in 2016, the Austrian Federal Railways (ÖBB) has very successfully taken over some of DB's connections and has since expanded them to other European markets; more high-speed train connections and more frequent public transport at the destination level (Gühnemann et al., 2021). However, in order to achieve a relevant shift from the car to public transport, the tourism sector depends on the tourists' willingness to consider alternatives to their usual mobility choices (for details, see Section 7.4.3). The greatest potential for reducing emissions in the lodging sector lies in improving the building structure (insulation), energy efficiency measures (room temperature, hot water use) and lighting (Energieinstitut der Wirtschaft GmbH, 2012). In Austria, the energy demand for heating buildings will decrease while energy demand for cooling buildings will increase as climate change continues (Prettenthaler et al., 2008; Dowling, 2013; Berger et al., 2014a). The net balance is estimated to be zero (Berger et al., 2014b) or even negative (i.e., decreasing energy demand) (Prettenthaler et al., 2008). Taking this into account, it can be concluded that the energy demand for the accommodation sector in Austria's mountain regions will rather decrease, since in many destinations

there will be no need to cool buildings due to the higher altitude and therefore lower temperatures, and less heating of buildings will be required.

Providers of infrastructure for tourism activities can reduce GHG emissions by purchasing and/or producing green energy, increasing the share of lower emission fuels in the fleet, or promoting public transportation to/from the attraction. Ski resorts are one of the main providers of infrastructure for tourism activities in mountain regions. Ski resorts can purchase green electricity, and produce their own green electricity with wind, photovoltaics and hydropower facilities (the latter sometimes in combination with snowmaking infrastructure) (Steiger et al., 2021a). Certification (ISO, EMAS, Österreichisches Umweltzeichen), sustainability and energy concepts, support the transition in winter tourism (Pröbstl-Haider et al., 2018). A ski resort in Salzburg (Snow Space Salzburg Bergbahnen) is testing e-fuel snow groomers which have the potential to reduce emissions by up to 90 % (Umwelt Journal, 2023). The same ski resort is incentivizing public transportation by offering free transportation within the federal province when a lift ticket is purchased online (Snow Space Salzburg, 2024).

Significant potential for GHG emission reduction also lies in the composition of the countries of origin of the visitors. Although far-away source markets such as China account for only a small share of overnight stays (2.2 %), their share of tourism-related emissions is disproportionately high (18.5 %) (Neger et al., 2021). The number of overnight stays from long-haul source countries increased before the pandemic, and is expected to increase in the coming years (Lehner-Telič, 2024). Therefore, demarketing strategies for long-haul source markets could slow down demand growth and associated GHG emissions.

#### Adaptation

Snowmaking in ski resorts is commonly regarded as a climate change adaptation strategy although Steiger and Mayer (2008) showed that the underlying reasons for investing in snowmaking are multifaceted. Nevertheless, snowmaking provides an opportunity to sustain ski operations in a warming climate (Steiger et al., 2022). The amount of natural snowfall and the duration of snow cover have been decreasing in recent decades and will continue to do so in the future (see Section 1.3.1). For Austrian ski resorts, Steiger and Scott (2020) show that the season length with natural snow will decrease from 80 days in the reference period (1981–2010) to 63 days (GWL 1.8°C), 44 days (GWL 2.6°C) and 19 days (GWL 4.3°C). With snowmaking, ski seasons would be 125 days long in the reference period, 120 days (GWL 1.8°C), 111 days (GWL 2.6°C) and 84 days (GWL 4.3°C). More relevant than the average shortening of the ski season, however, is the share of ski resorts with reliable snow conditions. While in the reference period (1981-2010), 99 % of the ski resorts in Austria can be considered technically snow reliable (i.e., considering modern snowmaking systems), this share decreases to 93 % (GWL 1.8°C), 80 % (GWL 2.6°C) and 31 % (GWL 4.3°C). However, despite shortened ski seasons, a substantial increase in water for snowmaking is required: 33 % (GWL 1.8°C), 60 % (GWL 2.6°C) and 73 % (GWL 4.3°C) (Steiger and Scott, 2020). Some Austrian ski resorts will therefore have to find alternative solutions for increasingly difficult winter operations, as it may no longer be technically possible to provide sufficient snow cover in some locations.

Another adaptation measure is to consolidate the existing ski area by concentrating ski operations and related infrastructure to reduce costs and operate more efficiently. In a ski area in Lower Austria, a part of the ski area was closed, and the snowmaking infrastructure was transferred to the remaining ski area, increasing the snowmaking capacity of that part (Scherrer and Begert, 2019; Redl, 2024).

In addition, a decrease in the size of large glaciers, as well as the disappearance of Austria's small and medium-sized glaciers is expected (Helfricht et al., 2019; François et al., 2023) (see Section 1.3.2). In the short to medium term, shrinking glaciers may provide some potential for increased demand, so-called 'last-chance tourism'. However, as noted by Mayer et al. (2024b), this may not be an appropriate longterm strategy. Adaptation options for summer skiing include the cessation of summer skiing operations and shifting to excursionists and technical adaptation measures such as geotextiles to reduce snow and ice melt, and depot snowmaking to ensure the opening of the ski season in the snowscarce fall months (Mayer and Abegg, 2024).

Broadening the winter offer to include activities that are also possible without snow, will become increasingly important. It can also be used to address leisure trends. The proportion of winter-holidaymakers whose main activity is winter hiking increased from 3 to 13 % between 2012 and 2018 (Steiger et al., 2021a). In principle, a range of summer activities can be extended to the (snow-free) winter. For example, during the extremely snow-poor Christmas holidays of 2022/23, mountain bike trails, summer toboggan runs, and motor skills parks were reopened in some Lower Austrian ski resorts (NÖN-Redaktion, 2023). Strengthening the summer season can reduce dependence on the winter operations. Many destinations are already successful in this respect, as shown by the stronger growth in overnight stays in recent summer seasons. Escaping the heat may provide some potential for summer tourism in alpine areas (Juschten et al., 2019). However, daily spending by guests in winter (EUR 205 on average) is still significantly higher than in summer (EUR 163) (Österreich Werbung, 2023). Although the number of tourists could be increased, social and ecological impacts could be exceeded and should therefore be considered in adaptation strategies.

#### Vulnerability

Compared to urban tourism, the vulnerability of alpine tourism is significantly higher, as many destinations are still dependent on winter tourism. However, the importance of summer tourism has increased in mountainous areas reducing the dependence on snow over time (Pröbstl-Haider et al., 2021). Currently, the additional income in summer does not compensate for the losses in winter, as tourist spending is 25 % higher in winter than in summer and therefore the total value added is likely to be different as well (Burton et al., 2024). For summer tourism, increasing risks from natural hazards (e.g., by thawing permafrost and glaciers) and maintenance of the hiking infrastructure are significant challenges.

Family firms, which make up the majority of tourism businesses in the Alps, have to cope with competitive disadvantages that include poor economies of scale and scope, minimal potential for diversification and innovation and limited access to capital markets. Therefore, this structure is perceived as very fragile and less resilient due to the small company size, low growth rates, weak internationalization, relatively low market entry barriers (allowing fewer professional entrepreneurs to enter the market), limited potential for diversification, high debt-to-capital-ratios and limited qualification levels (Zehrer and Mössenlechner, 2009; Pikkemaat and Zehrer, 2016). This structural weakness could lead to significant economic impacts, if many of these family businesses were to go out of business due to climate change. Economic vulnerability is therefore higher in the Alpine region than in the lowlands of Austria. In the Alps, alternative employment opportunities outside of tourism and agriculture are limited.

#### AAR2

#### 7.4.3. The Alps as home to communities

# Demography and settlements

#### Effects of climate change

Changes in demography and land use are mainly driven by socio-economic factors. Currently there is no evidence that climate change significantly influences demographic changes as well as allocation and distribution of settlements and infrastructure in the Austrian Alps. Conversely, it is evident that demography and existing settlement structures influence GHG emissions mainly through the associated energy and mobility demand (Stöglehner et al., 2016a; Stöglehner and Abart-Heriszt, 2022; Haberl et al., 2023) (medium evidence, high agreement) (see Section 3.2.2). Furthermore, demography, settlement and built infrastructure influence the potential exposure to climate-related impacts, such as heat waves (Oleson et al., 2015; GeoSphere Austria, 2022) and various natural hazards (Cammerer et al., 2013; Fuchs et al., 2015b; Löschner et al., 2017; Junger et al., 2022) (high confidence).

The regional divide in population development in the Austrian Alps (see Section 7.2.2) will continue to increase, as will the unbalanced intra-regional population development between the main valley areas on the one hand and the side valleys and peripheral locations on the other hand. Population forecasts at the district level until 2050 largely confirm and extend these trends for the coming decades (Österreichische Raumordnungskonferenz, 2023c). Household forecasts at the district level until 2030 indicate an increase in households in the vast majority of the Austrian Alpine districts (Österreichische Raumordnungskonferenz, 2024). The development of households is more strongly correlated with settlement development (resulting in further land take) than with population development. A decrease in average household size is projected for all Alpine districts (Österreichische Raumordnungskonferenz, 2024).

Based on these demographic trends, an increase in building, land use and infrastructure can be expected in growth areas, particularly in the already densely populated urban centers and ribbon-like settlement structures in the valleys and, depending on the development of tourism, also in tourism resorts with distinct urban characteristics. Whether the expansion of built-up land also takes place in scattered settlement areas with low densities, depends primarily on spatial planning policy (e.g., the willingness to regionally restrict settlement development to central locations), since a decline in the population and a slight increase in the number of households do not necessarily lead to a stagnation in the expansion of build-up land. There is no confirmed evidence whether and to what extent climate change is currently influencing these demographic and land use trends. Prolonged heat waves and their impacts on densely built-up urban areas (predominantly outside the Alps) could increase the attractiveness of currently sparsely populated Alpine areas as residential locations in the future, leading to an increase in population and building land uses there (Membretti et al., 2023). A substantial reduction in the intensity of winter tourism due to rising snowlines could lead to a population decline in Alpine tourist resorts (Steiger et al., 2021b).

Land competition in Alpine valleys is likely to intensify (see Section 3.2.2). This will put additional pressure on open spaces and the various ecosystem services they provide. Whether increasing settlement development in the valley areas will lead to an increase in GHG emissions, depends on how additional development is coordinated through spatial planning and the extent to which increasing mobility demands can be met by public transport. Increasing land use and population densities influence exposure to climate-related impacts (*medium evidence, high agreement*).

Comparable to the whole country, the share of older people (aged 65 and over) in the Alpine region is projected to increase (Österreichische Raumordnungskonferenz, 2023c). This development is most pronounced in Alpine regions, which are confronted with a significant population decline. It eventually leads to increased vulnerability to extreme weather events, particularly to heat waves (Hutter et al., 2007; Haas et al., 2014) (*high confidence*).

Finally, it should be noted that socio-economic development is based on complex interrelations. Related trends are therefore non-linear. Single, even minor causes may have largely unpredictable far-reaching consequences.

#### Mitigation

Density, together with the implementation of other spatial planning principles such as compactness and mix of spatial functions, urban redevelopment rather than greenfield development, and coordination of development with public mobility services (Jabareen, 2006), are crucial for energy demand, renewable energy supply (e.g., by district heating systems) and public transport systems. In contrast, low-density development and urban sprawl contribute substantially to an increase in energy consumption and motorized private transport and, thus, to an increase in GHG emissions (Stöglehner et al., 2016b; Stöglehner and Abart-Heriszt, 2022) (*high confidence*) (see Section 3.2.2).

Some spatial framework conditions in Alpine areas, particularly in densely populated urban centers and ribbon-like settlement structures in the valleys, potentially provide favorable conditions for climate change mitigation. Limited development space and higher land prices provide incentives for higher density housing and, thus, land-saving development (Österreichische Raumordnungskonferenz, 2021, 2024). Topography, limited development space and accessibility reasons (large transport infrastructure in valleys) lead to a concentration of commercial and industrial development in the valley areas. Both densification effects can be regarded as supportive of efforts to reduce GHG emissions (*limited evidence, high agreement*).

Without appropriate regional planning regulation, the development of peripheral settlements and scattered housing structures as well as decentralized tourism projects (e.g., chalets, holiday residences) carries the risk of low-density development and urban sprawl (Meyer and Job, 2022), which contributes to land take, requires substantial infrastructure efforts and increases motorized individual transport and related GHG emissions. Urban redevelopment and concentration of settlement development in local and regional development centers on the one hand, and limiting development in scattered housing areas on the other, are spatial planning strategies that support climate change mitigation (Stöglehner and Abart-Heriszt, 2022) (high confidence). The unbalanced population development trends in the Austrian Alps provide opportunities to concentrate development in locations that are favorable in terms of energy demand, renewable energy supply and public transportation. Related trade-offs with rural development strategies should be considered.

#### Adaptation

Aiming at higher compactness in settlement development is regarded a key issue for climate change mitigation, however, trade-offs with the provision of sufficient open spaces (including those required for climate change adaptation) and potential increased exposure to heat stress and natural hazards need to be considered (see Section 3.2.3). Population development trends in the Austrian Alps are likely to result in higher settlement densities in urban centers and ribbon-like settlement structures in the valley areas. Additionally, limited development space and high land prices put pressure on open spaces making it more difficult to implement nature-based options for adaptation to climate change (Schindelegger et al., 2022).

The need for adaptation to heat stress is likely to increase in densely built-up areas, exacerbated by an ageing population. However, urban heat islands not only depend on building density and the degree of sealing, but also on topography and climate. In particular, the supply of cold air from forests, agricultural areas, urban green spaces, and other fresh air corridors is important. In alpine areas, warm nights are less frequent than in the lowlands due to the nocturnal flow of cold air down the slopes into the valleys (Gerersdorfer et al., 2009). As Schöner and Haslinger (2020) point out, temperature extremes will play a comparatively minor role at altitudes above 800 m a.s.l. (Schöner and Haslinger, 2020). At altitudes below 800 m, temperature extremes have increased in recent decades and will continue to increase in the future. The altitude range below 800 m includes all urban centers and densely built-up ribbon-like settlement structures in the Austrian Alps (medium evidence, high agreement).

Whether settlement development and, especially densification, affects the need for adaptation to natural hazards depends on the location of the development and whether current and potential future hazard zones can be avoided. Due to the topography, limited development space and accessibility reasons, building and infrastructure densities and, thus, potential exposure to natural hazards are high in Alpine valley areas (Fuchs et al., 2015b). This is particularly the case for areas with a lower hazard intensity where development is not restricted by spatial planning regulations (Nachtnebel and Apperl, 2015; Löschner et al., 2017) (high confidence). With regard to river floods, for example, a damage potential of EUR 357 million was calculated for a 20 km stretch in the Lower Inn Valley in Tyrol in the event of a 100year flood. In the case of a 300-year flood, the corresponding damage potential is EUR 736 million (Bundeswasserbauverwaltung Tirol, 2016; Blöschl, 2020). If flood hazards are exacerbated by the effects of climate change, this would result in further adaptation needs, particularly in urban centers and ribbon-like settlement structures. Stagnation or population decline usually do not result in a reduction in built-up areas. It is therefore likely that adaptation needs regarding natural hazards are also relevant for other Alpine settlements. In particular, the tradition of scattered housing outside the valleys has led to an extensive rural road network connecting (former) single farms to the main transport infrastructure in the valleys, which is considerably exposed to natural hazards such flash floods or landslides (limited evidence, high agreement).

Alpine settlements and their development, including associated infrastructure, are highly dependent on the functioning of protective forests. Climate change impacts challenge their stability, mainly by windthrow, infestation by bark beetles or wildfires (Vacik et al., 2020). These impacts are likely to influence long-term hazard exposure of settlements and infrastructure (Schindelegger et al., 2022) and may result in further adaptation needs (*limited evidence, high agreement*).

## Vulnerability

There is no confirmed evidence that climate change impacts are currently influencing the number and distribution of people and building land uses in the Austrian Alps. Related changes are solely triggered by socio-economic factors. Population and household projections indicate a further increase in building land uses (settlements and infrastructure), particularly in the main valley areas of the western Austrian Alps, which will most likely lead to increased land competition and further pressure on existing open spaces. This is likely to lead to increased exposure to natural hazards and reduced scope for adaptation to climate change. These effects are less pronounced in most of the valleys of the eastern Austrian Alps as well as in areas with low settlement densities. Spatial and temporal variability leads to differentiated vulnerability patterns.

# Mobility and transport infrastructure

## Effects of climate change

Mobility in the Alpine region is characterized by the overlapping of the everyday and leisure mobility of the (local) population, the arrival and departure of tourists and their onsite mobility (including day trips from neighboring (urban) regions) and tourists passing through (passenger transit). Likewise, the goods traffic of the local economy and supply overlaps with long-distance freight transit traffic. The following figures demonstrate the extent of this overlap. In Tyrol in 2022, the prominently discussed truck transit accounted for about 2.56 million trips per year (Land Tirol, 2023). In the tourism sector, there were about 11 million tourist arrivals, leading to 22 million arrivals and departures in that year (Statistik Austria, 2023b), while the local population made about 1.25 million car trips per day (Köll and Bader, 2022; Land Tirol, 2023). Due to the topographical conditions in the valley, the transport network does not offer any alternatives so that this overlapping takes place on the few main roads and leads to corresponding (temporary and seasonal)

congestion and negative effects, e.g., through noise and emissions, on the population in the densely populated valleys, but also in the adjacent hillside locations (*high confidence*).

The spatial pattern of the Alpine region also leads to special characteristics of everyday and leisure mobility. Due to the high density in the central valleys, the shares of walking and cycling are comparatively high (Tomschy et al., 2016). There is also an expectation that climate change will increase the popularity of cycling, which is currently very seasonal. However, there are no reliable findings on this in the literature. On the other hand, long supply and commuting routes in peripheral valley locations are (or have to be) covered by motorized transport. This also leads to high traffic loads and congestion at the valley entrances (e.g., Zillertal Brettfall Tunnel, outer Ötztal). The valley locations have comparatively good, linear public transport connections. However, the quality of service is poor in peripheral locations, where regular service is not economically justifiable (medium evidence, high agreement).

Tourist mobility in the Alpine region also has special characteristics. Arrivals and departures are predominantly by car (over 80 % in winter and summer), the share of rail is relatively low at approximately 6 % (e.g., for Tyrol) (Bursa et al., 2022). Relevant winter air traffic comes from more distant source markets to regional airports (Innsbruck, Salzburg, Klagenfurt) and nearby major airports (Munich, Zurich). Although the tourism sector is of great importance for the national economy with a share of 7.6 % of the gross domestic product (in 2019 including direct and indirect effects) and even higher shares in the Alpine regions (e.g., Tyrol 16.9 %, Salzburg 13.7 %; according to Statistik Austria (2021)), it is largely unknown how much emissions the Austrian tourism sector causes. Neger et al. (2021) attribute this mainly to the fact that most products and services consumed by tourists are not exclusively related to tourism. They provide an estimate based on available data with a destination-based accounting approach that includes not only emissions within Austria but also those caused by international guests traveling abroad. They estimate the total emissions of the tourism sector in Austria at 4.31 MtCO<sub>2</sub>eq, which is a share of 4.2 % of the total emissions in Austria. Of this, 92.2 % of CO<sub>2</sub>eq is attributable to travel. As they note, this high share is mainly due to air travel by international guests. In alpine tourism, however, this share is significantly lower, with more than 80 % of the guests arriving by car. Accordingly, in a case study of a Tyrolean destination (Alpbach), Unger et al. (2016) determined the share of transport in tourism emissions to be 54 %. Studies show that holidaymakers are well aware of the high  $CO_2$  emissions associated with vacations but that they indicate a low willingness to change their behavior significantly in this regard (less frequent but longer vacations, closer destinations, no air travel), thus creating an awareness–behavior gap (Mailer et al., 2019).

The transport infrastructure is characterized, on the one hand, by the fact that there are only a few efficient routes or connections in the main valleys (transit areas). When the motorways become congested, traffic increasingly tries to use the less efficient parallel rural roads, causing high levels of congestion in the settlement areas. Recently, this has been countered by banning motorway exits forcing passing vehicles to stay on the motorways. Due to the topography, railways are only present in a few side valleys. On the other hand, transport infrastructures in Alpine regions are particularly exposed to natural hazards (rockfall, debris flows, snow avalanches), which require protective structures (it should be noted, that roads and railways are already more expensive to build and maintain due to the higher demand for artificial structures such as bridges and tunnels) and there is still the risk of temporary closures that restrict the accessibility of valleys. In addition, there is increased operational effort (winter service, maintenance of railway lines with narrow bends, etc.) (high confidence).

Population growth is leading to an increase in everyday and leisure mobility. Tourism also continues to focus on growth in the number of guests and overnight stays. Attempts to promote rail travel have not yet yielded the desired results. Shorter stays are also leading to a disproportionate increase in arrivals and departures.

While in the 1970s, guests in Austria spent on average just under 7 nights (for guests from abroad), or 6 nights (for domestic guests) in holiday accommodation, this number has now been reduced to just under 3.5 or 2.8 overnight stays (BMLRT, 2020), in line with international trends in lengthof-stay (Gössling et al., 2018). Correspondingly, the number of arrivals at accommodation establishments in Austria tripled as the number of overnight stays increased by 50 % (Statistik Austria, 2019a, 2019b). This means that the intensity of travel has increased and that there is a greater overlap of arrival and departure traffic with local traffic, even on weekdays. Travelers' demands for flexibility and speed of the available means of transportation have increased significantly, in order to keep the travel time as short as possible and to spend as much time as possible at the destination. This favors the private car as the mode of transport of choice and poses major challenges for destinations that rely on climate-friendly travel (Gronau and Kagermeier, 2007).

By 2019, the number of guests from more distant growth markets had been increasing, which is also reflected in the growing number of passengers at airports near tourist destinations (e.g., Innsbruck). These markets have an above-average share of arrivals and departures by air, which also has an increasing impact on GHG emissions (Zech et al., 2013; Peters et al., 2016) (*high confidence*).

Restricting freight transit traffic is contrary to the demands of neighboring countries and the principle of free movement of goods and people in the EU. Even with great efforts (Brenner Base Tunnel), with current measures, a shift to rail seems to be limited to absorbing not much more than the increase in freight traffic. A higher shift can only be reached with additional powerful policy measures (BCP – Brenner Corridor Platform, 2021). Efforts to increase tourist traffic in the southern neighboring countries such as Italy/ South Tyrol, Slovenia and Croatia will result in an increase in passenger transit (*high confidence*).

The impact of climate change on tourism has already been addressed in Section 4.3.3. This has particular implications for Alpine regions. Changing climatic conditions can alter travel and thus mobility behavior of tourists. For example, a lack of snow in some areas and a shorter winter season, can lead to higher numbers of visitors in a shorter timeframe, to tourists traveling to more snow-reliable but also more distant winter destinations in the inner Alps, or to tourists switching to other non-snow holiday destinations (e.g., the Canary Islands), which involves a switch of transport mode (from car to air) and increases travel distance. Changes in travel behavior and thus mobility behavior can also be seen in the summer season. Hot summers, in particular, drive city dwellers to the surrounding areas and nearby alpine destinations, and short summer breaks are becoming increasingly important (Cavallaro et al., 2017; Thurm et al., 2017; Juschten et al., 2019; Juschten and Hössinger, 2021) (high confidence).

Climate change also increases the potential for some natural hazard types to affect infrastructure (see Section 7.4.1), thus, increasing the vulnerability of transport relationships (accessibility and mobility). Increasing traffic flows and dependence on motorized (long-distance) transport will also increase this effect. In alpine regions, the expected rise in temperature is likely to have ambivalent effects on the infrastructure. On the one hand, roads will require less effort in terms of frost resistance and spring thaw. On the other hand, higher surface temperatures require higher quality materials. The same applies to railways, where the mountain routes are characterized by narrow curves, and where rising temperatures have a stressing effect and increase the risk of buckling, while less snow leads to less maintenance and operational effort and related emissions (snow removal, prevention of icing) (*medium evidence, high agreement*).

The supply of transport services can be affected by potential climate impacts. In particular, extreme weather events such as heat waves, extreme rainfall, and storms can severely restrict traffic flow and disrupt operations. But also associated hazards, triggered by events such as landslides and floods, can lead to significant damage to the transport infrastructure and disrupt traffic flows and logistics (Stahel et al., 2013; Cavallaro et al., 2017). In this context, adaptation measures are needed to strengthen the resilience of the transport systems through improved traffic management in the event of a crisis, as well as to adapt the infrastructure to ensure (not only, but in particular) the accessibility of tourist destinations (Gühnemann et al., 2021) (*high confidence*).

### Mitigation

Avoid/shift/improve strategies and the potential of virtual mobility (see Sections 3.4.2, 3.4.3, 3.4.4) have to be considered regarding the special conditions in the Alpine region. For decarbonization, e.g., through alternative drives, there are special requirements due to the topography and the (winter) climate regarding e-mobility and the charging infrastructure due to the massive tourist travel flows and the need for on-trip charging (*medium evidence, high agreement*).

A significant reduction of travel-related GHG emissions in tourism cannot be solely achieved with technological solutions (electric vehicles) or trends (sharing); it also requires shifts in transportation patterns, such as increased use of rail and public transport over air and private road travel. Similarly, rail travel can be supported by improving on-site mobility and services along the entire travel chain (from pre-booking information to return). Due to the mentioned awareness and behavioral gap among tourists, this calls for measures that not only offer incentives (fast and direct train and bus connections, climate-friendly local transport, luggage services, attractive all-inclusive packages, etc.), but also focus on closer rather than distant home markets, encourage longer stays, and offer awareness campaigns for local stakeholders and tourists, as well as push and pull strategies (high confidence). Several case studies show significant effects of improving public transport services (e.g., free arrival by train) (Blättler et al., 2024) or local mobility services at the destination such as bus services and mobility hubs (BMNT, 2017; Bursa et al., 2024). The awareness created by

COVID-19 and the climate crisis can be seen as an opportunity to take appropriate measures (Gühnemann et al., 2021) even if current developments clearly indicate that there is a risk of falling back into old patterns.

#### Adaptation

There will be a need to increase resilience by upgrading and protecting transportation infrastructure against natural hazards, but also in the context of spatial planning that strengthens local supply (amenities) and thus local mobility. This is also important to ensure escape routes and supplies in the event of natural hazards. Developing emergency plans and strategies can help enhance preparedness and response efforts. When protecting the infrastructure, it is important to remember that the constructional measures themselves have a climate impact. These rebound effects can be minimized through careful selection of construction methods and materials (*medium evidence, high agreement*).

In order to keep public transport as an attractive offer both for arrivals and departures as well as for local mobility, the following measures from the adaptation strategy (BMNT, 2017) are also relevant in response to increasing heat days: (i) Ensuring thermal comfort by reducing the thermal loads in transport stations and their surroundings; (ii) Reducing possible thermal loads for passengers and staff in public transport through appropriate air conditioning (*limited evidence, high agreement*).

#### Vulnerability

In the Alpine regions, natural hazards can restrict the mobility of the population and the accessibility of places for daily needs, as there are often no alternative routes due to the limited space. The remoteness and lack of redundancy of the road network make the Alpine region particularly vulnerable. On the other hand, the expansion of protective measures for transport routes is associated with significant  $CO_2$  emissions (*high confidence*).

# 7.5. Needs for resilient mountain regions in Austria

In mountain areas in general, most existing adaptation measures (structural and non-structural) are reactive rather than proactive to climate change and are often part of a post-disaster recovery strategy (McDowell et al., 2021) (*ro*- bust evidence, medium agreement). Although many adaptation measures have been adopted in Austria, specifically for mountain regions, there are still gaps in effective adaptation. To close these gaps, transformative adaptation in different sectors is needed in mountain regions (McDowell et al., 2021) (see Section 7.5.2). However, as suggested by Salim et al. (2021), coping and incremental approaches are also needed, in addition to transformative adaptation approaches. For this reason, the section on adaptation and mitigation needs in the Austrian Alps is structured as follows: Monitoring (7.5.1), (socio-economic) transformation (7.5.2), non-structural protection measures and nature-based solutions (7.5.3), and structural measures (7.5.4). The adaptation options for mountain areas in Austria are presented in Table 7.1, according to the ASI principle (Avoid, Shift, Improve; see Cross-Chapter Box 4).

# 7.5.1. Monitoring of natural processes/hazards and provision of climate services

Continued and improved monitoring of environmental conditions and processes (e.g., climate, hydrology, glacier and permafrost changes, natural hazards) may serve as a basis for information on climate adaptation. Climate model scenarios for the Alps are still subject to high uncertainties (Section 7.3.2), especially for precipitation. Climate services of the same quality for mountain regions, as for the rest of Austria, must be advanced in regional climate modeling for the Alpine region (e.g., convection-permitting models). Monitoring of runoff and sediment dynamics of alpine rivers needs to be further improved, especially in torrent catchments that have not been monitored to date and pose a significant hazard potential and risk to the built environment (this includes also monitoring of soil properties, e.g., water storage capacity and carbon stock). Monitoring of hazardous conditions in tourist areas (e.g., hiking trails) could be used to better inform tourism stakeholders and tourists about potential risks. Moreover, monitoring and management of invasive species (Permanent Secretariat of the Alpine Convention, 2019) can support the conservation of Europe's high mountain biome. In terms of biodiversity, monitoring of species shifts, and changes in abundance and growing seasons could be beneficial. Additionally, it is important to monitor biodiversity changes of different organism groups (e.g., plants but also animal groups and soil organisms). Monitoring and mitigation of natural forest disturbances (wind, fire, pests) may also support adaptation to climate change. Monitoring of permafrost and erosion and their short- and long-term

effects (Permanent Secretariat of the Alpine Convention, 2019) is also important for the assessment of changing risks to infrastructure in high alpine terrain (huts, alpine routes, ski lifts). Generally, an improved and denser monitoring network for hazard triggers and hazard processes could improve monitoring results.

According to the WMO, early warning systems (EWS) are a top priority for climate change adaptation and could be a valuable tool for disaster risk reduction in Austria's alpine region. According to Pappenberger et al. (2015), there are significant monetary benefits related to early flood warnings in Europe. Specifically, for every EUR 1 invested in early flood warning, there is a benefit of EUR 400. According to the same source, the avoided damage due to early flood warning may reach 32.85 % of the total damage. While the drivers of natural processes such as temperature and precipitation have been affected by climate change, there is still limited evidence that this has affected the number of natural hazards occurring in mountain areas. However, there is high confidence that the exposure in mountain areas has increased (Fuchs et al., 2015b, 2017). In view of the particularly strong changes in the frequency and intensity of hazard processes due to climate change, it may be beneficial to reconsider the current delimitation criteria for hazard zones (avalanches, debris flows, floods).

### 7.5.2. Transformation

Several mountain areas in Austria are characterized by climate change-related impacts (e.g., reduced snow cover), land use changes as well as other societal and structural problems, and require a significant change based on holistic analyses and awareness of the interrelated challenges. In other words, transformation is required in several sectors.

More specifically, as far as **agriculture** is concerned, adaptation measures in mountain areas may be related to transformation. Some of them are not directly related to climate change (hidden adaptation, e.g., organic farming), but to other challenges (e.g., changing agricultural policies and market conditions) (Grüneis et al., 2016). However, some hidden adaptation actions, which include measures such as long-term learning processes, self-organization, local knowledge, and adaptive governance, may be more effective. Transformation could include the cultivation of non-traditional crops, the transformation of consumer demands (consumption of local products) as well as the energy efficiency of farms (Permanent Secretariat of the Alpine Convention, 2019). In addition, shifting agricultural land upwards and improving soil properties could also contribute to adaptation (see Section 7.4.2). With regard to forests, the conversion of forest ecosystems to include endemic species that have been adapted to climate change (Permanent Secretariat of the Alpine Convention, 2019) as well as a shift from single tree types to multiple tree type forest could support adaptation to climate change (see Section 7.4.2). Nevertheless, the choice of tree species is often highly controversial. Regarding tourism, transformative adaptation measures may include alternative hiking routes and season shifting with mixed results (Adler et al., 2022) but also the promotion of activities that can be carried out without snow, the extension of summer activities in winter and the strengthening of the summer season (see Section 7.4.2). Finally, the expectation that the Alps will benefit from a shift of tourism from the Mediterranean to the Alps, due to high temperatures, is not supported by the existing literature, although the modeling of tourism demand in Europe showed that it is expected to decrease (under different climate change scenarios) for southern Europe while tourism demand for northern Europe is expected to increase (Matei et al., 2023). Further, since tourism requires energy, the transition to green fuels and investments in hydrogen drives, sustainability and energy concepts could support adaptation. Decision-making tools have been developed for other locations in the European Alps to support ski resorts in the assessment of their energy demand and provide them with information on available renewable energy resources (Polderman et al., 2020). Moreover, a transition in the accommodation sector (e.g., improving infrastructure efficiency, behavioral changes, greening energy production and carbon off-setting) would be beneficial since it is a substantial contributor (10 %) to total GHG emissions from the tourism sector. Improving infrastructure efficiency alone can reduce 15-20 % of emissions from the accommodation sector (EY Parthenon et al., 2021). However, financing can be a challenge, especially in small resorts. In addition, minimizing the carbon footprint of alpine hotels and gastronomy (regional products) should be an aim (Permanent Secretariat of the Alpine Convention, 2019). Ski areas can purchase green electricity to reduce CO<sub>2</sub> emissions from lift operation and snowmaking. For slope grooming, emissions can be effectively reduced by using alternative forms of propulsion (e-drive) or fuels (e.g., HVO) (Steiger et al. 2021a). In the energy sector, alongside efforts to improve energy efficiency and savings across all areas, there is potential for expanding renewable energy and decentralizing its production (Permanent Secretariat of the Alpine Convention, 2019). The built environment in the Austrian Alps also needs to be trans-
formed to adapt to the effects of climate change, including the avoidance of new development in hazard-prone areas and the adaptation of the existing building stock. Regarding mitigation, spatial planning can be used to support the densification of existing settlement areas and guide new developments toward urban and regional centers (Section 7.4.3). In the mobility sector, mitigation measures could include the promotion of rail travel, on-site mobility with public transport and active transportation modes, climate-friendly local transport vehicles, a shift to short and medium distance source markets, and the provision of alternative or public transport options for tourists to and within Alpine destinations (Permanent Secretariat of the Alpine Convention, 2019). Adaptation could include a further expansion of information and early warning systems related to tourism and mobility, a functioning transport system, review/adaptation of legal standards for the construction and operation of transport infrastructure considering climate change, and ensuring thermal comfort within public transport and its surroundings (e.g., stations and stops) (Gühnemann et al., 2021).

# 7.5.3. Non-structural adaptation measures and nature-based solutions

River renaturation is crucial but there is also a need to focus on the decision-making context of nature-based solutions in the Alps, taking into account local climate projections (Dubo et al., 2022). With regard to biodiversity and ecology, more ecological connectivity projects are needed (Permanent Secretariat of the Alpine Convention, 2019). Examples of such projects already exist in Austria (e.g., project PORTAL, 'Pathways of Transformation in the Alps') (Observatoire des Sciences de l'Univers de Grenoble, 2024). Additional adaptation needs may include landscape preservation, avoidance, reduction and monitoring of soil sealing, improvement of soil quality and enhancement of the protective function of forests (Permanent Secretariat of the Alpine Convention, 2019). All of these should be supported by the appropriate legal frameworks. In protective forests, interventions such as thinning and regeneration cutting may support the coping capacity of forests (see Section 7.4.2).

To ensure effective implementation of adaptation measures, cost-benefit analyses, monitoring of adaptation plans, governance and participation as well as financing are needed, as is shown in the pilot program 'Climate Change Adaptation Model Regions for Austria – KLAR! (Klimawandel-Anpassungsmodellregionen)', which is funded by the Austrian Climate and Energy Fund. Transnational/transboundary cooperation could be emphasized, as well as communication among regions, countries, scientists, practitioners, authorities, and the public. Implementation requires giving high priority to adaptation, climate-proofing in all activities, initiating pilot actions, promoting international information exchange among mountain regions, monitoring of all actions, and enabling communication and cooperation among stakeholders (Scott et al., 2022). An example of this type of collaboration is the AlpEs ('Alpine Ecosystem Services') project, which aims to map, maintain and manage ecosystem services in several Alpine countries including Austria (Interreg Alpine Space, 2021). Finally, involving people in the adaptation decision-making process and addressing multiple rather than single risks (Terzi et al., 2019) lead to more robust adaptation strategies (Owen, 2020).

### 7.5.4. Structural adaptation measures

Although there is a trend to move from structural to non-structural measures, structural protective measures for the alpine infrastructure and settlements are still necessary. Grüneis et al. (2016) suggested that the focus is on technical adaptation measures, which may also be vulnerable to climate-related extreme events (e.g., power failure). Local structural protection mitigates hazards at low cost (Papathoma-Köhle et al., 2019) and building codes may help adapt buildings to climate change, but also improve the Build Back Better (BBB) after disaster recovery. Insurance incentives and improved risk transfer support recovery.

Climate-proofing hydropower is crucial. Future changes in flow and sediment dynamics may require a reconsideration of operational policies for existing plants, including adaptive water targets and sediment management. Cost-benefit analyses for renewals and new plants should take into account long-term effects of climate change on flow and runoff (Permanent Secretariat of the Alpine Convention, 2019).

Finally, maladaptation (adaptation measures that increase rather than reduce vulnerability) should be avoided. In this respect, cross-sectoral policies may be the solution as adaptation measures in one sector may increase the vulnerability of another sector (Alpine Convention Guidelines). For example, it is not clear whether artificial snowmaking in alpine ski resorts is a maladaptation (Roncucci, 2020), but there is potential to improve its performance, which will determine whether it can be considered a maladaptive or successful adaptation (Scott et al., 2022). According to Schinko et al. (2013), artificial snow can reduce up to 75 % of the economic impacts of climate change on winter tourism, but at the same time it increases production costs.

Thematic area	Avoid	Shift	Improve					
Climate change	Further $CO_2$ emissions through mobility and tourism in the Alps	Transitions in many sectors (e.g., mobility, tourism) to reduce further CO <sub>2</sub> emissions	Improve climate scenarios for the alpine space					
			Monitoring of atmospheric pro- cesses in the Alps					
Natural hazards			Monitoring of hazards and trigger- ing processes					
	Building in hazardous areas (con- sidering variations due to climate change)	Reduce the vulnerability of the built environment and infrastruc- ture to natural hazards by adapted design and local adaptation	Risk transfer options (insurance) and incentives for local adaptation measures -Existing hazard map- ping to include climate change					
		River renaturation to reduce flood events	Local and regional land use plan- ning					
Biodiversity	Impacts of human land use (ag- riculture, energy production and tourism development) on biodi- versity	Improvement of spatial planning	Monitoring of biodiversity changes in the Alps Collaboration with nature conser- vation authorities to design man- agement strategies					
			Public awareness about the risk of losses Preservation of larger and bet- ter-connected areas to mitigate losses of alpine ecosystems					
Forestry and agriculture	New development that could have negative effects on protective forests	Non-native species more resilient species to climate change (debat- able) From single species strands to mixed species strands	Preservation and restoration of protective forest Financial support for the managers of protective forests					
		Introduction of non-traditional crops Shift in consumer demands and preferences	Biological soil properties by nitro- gen-fixing plants, the addition of organic material, and water con- servation by a cutting height of at least 8 cm to tackle drought in the agricultural sector Irrigation of crops					
Buildings, infrastructure and mobility	New development of energy ineffi- cient buildings and infrastructure Reduce the need to travel (e.g., digitalization)	Shift from private car to public transport or non-motorized ve- hicles Adapt building stock to withstand natural hazards	Insulation of buildings (also rele- vant for the thematic area 'energy') Redundancy in the road network Improve the energy efficiency of vehicles					
Energy	Impact from natural hazards on energy infrastructure	More green sources of energy (also relevant for the 'tourism' thematic area)	Acceptance of renewable sources of energy from the tourism sector					
Tourism	Old practices that encourage further CO <sub>2</sub> emissions (e.g., non- green electricity for powering ski lifts and snow making machines, diesel-powered slope preparation) Destruction of carbon sinks for new touristic development	Find employment alternatives through the economic strengthen- ing of regional centers Identification and development of alternative tourism products Promote sustainable destinations Promote alternative transport modes (see transport)	Make existing infrastructure ener- gy efficient (e.g., energy efficient hotels)					

 Table 7.1
 Adaptation options for the Austrian Alps presented considering the ASI principle (Avoid, Shift, Improve)

#### REFERENCES

- Adler, C., Wester, P., Bhatt, I., Huggel, C., Insarov, G. E., Morecroft, M. D., et al. (2022). Cross-Chapter Paper 5: Mountains. Climate Change 2022: Impacts, adaptation and vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, 2273–2318. <u>https://</u> doi.org/10.1017/9781009325844.022
- Airey-Lauvaux, C., Pierce, A. D., Skinner, C. N., and Taylor, A. H. (2022). Changes in fire behavior caused by fire exclusion and fuel build-up vary with topography in California montane forests, USA. *Journal of Environmental Management* 304, 114255. <u>https://doi.org/10.1016/j.jenvman.2021.114255</u>
- Albrich, K., Rammer, W., and Seidl, R. (2020). Climate change causes critical transitions and irreversible alterations of mountain forests. *Global Change Biology* 26, 4013–4027. <u>https://doi. org/10.1111/gcb.15118</u>
- Alexander, J. M., Chalmandrier, L., Lenoir, J., Burgess, T. I., Essl, F., Haider, S., et al. (2018). Lags in the response of mountain plant communities to climate change. *Global Change Biology* 24, 563–579. <u>https://doi.org/10.1111/gcb.13976</u>
- Alexander, J. M., Diez, J. M., and Levine, J. M. (2015). Novel competitors shape species' responses to climate change. *Nature* 525, 515–518. https://doi.org/10.1038/nature14952
- ALPARC (2019a). Die Schutzgebiete. ALPARC Das Netzwerk Alpiner Schutzgebiete. Available at: <u>https://alparc.org/de/the-protected-areas</u> (Accessed October 31, 2024).
- ALPARC (2019b). Alpine Parks 2023. ALPARC Das Netzwerk Alpiner Schutzgebiete. Available at: <u>https://alparc.org/de/alpineparks-2030</u> (Accessed October 31, 2024).
- Alpine Convention (2015). Guidelines for Climate Change Adaptation at the local level in the Alps., eds. A. Ballarin-Dent, L. Cetara, and M. Teresa Idone. Available at: <u>https://www.alpconv. org/fileadmin/user\_upload/Publications/AS/AS7\_EN.pdf</u> (Accessed November 5, 2024).
- Amt der Steiermärkischen Landesregierung ed. (2023a). Dynamische Waldtypisierung – Anpassungsoptionen an den Klimawandel und Beschreibung der Waldgruppen. Graz: Abteilung 10 Land- und Forstwirtschaft, Landesforstdirektion. Available at: <u>https://www. agrar.steiermark.at/cms/dokumente/12733633\_151504582/</u> aec867d0/FORSITE\_Band\_2\_080823.pdf
- Amt der Steiermärkischen Landesregierung ed. (2023b). Dynamische Waldtypisierung – Methodische Grundlagen und Beschreibung der Waldstandortseinheiten. Graz: Abteilung 10 Land- und Forstwirtschaft, Landesforstdirektion. Available at: <u>https://www. agrar.steiermark.at/cms/dokumente/12733633\_151504582/ daf513e5/FORSITE\_Band1\_030823.pdf</u>
- Anghileri, D., Botter, M., Castelletti, A., Weigt, H., and Burlando, P. (2018). A comparative assessment of the impact of climate change and energy policies on alpine hydropower. *Water Resources Research* 54, 9144–9161. <u>https://doi.org/10.1029/2017WR022289</u>
- APCC (2014). Österreichischer Sachstandsbericht Klimawandel 2014 (AAR14)., ed. Austrian Panel on Climate Change (APCC).
   Wien, Österreich: Verlag der Österreichischen Akademie der Wissenschaften.
- APCC (2023). APCC Special Report: Strukturen für ein klimafreundliches Leben (APCC SR Klimafreundliches Leben)., eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. Berlin, Heidelberg: Springer Spektrum. <u>https://</u> doi.org/10.1007/978-3-662-66497-1

- APG (2024). Market transparency. Installed power plant capacity. Available at: <u>https://markttransparenz.apg.at/en/markt/Markt-transparenz/erzeugung/installierte-leistung</u> (Accessed November 5, 2024).
- Arpaci, A., Eastaugh, C. S., and Vacik, H. (2013). Selecting the best performing fire weather indices for Austrian ecoregions. *Theoretical and Applied Climatology* 114, 393–406. <u>https://doi.org/10.1007/s00704-013-0839-7</u>
- Auer, I., Böhm, R., Jurkovic, A., Lipa, W., Orlik, A., Potzmann, R., et al. (2007). HISTALP—historical instrumental climatological surface time series of the Greater Alpine Region. *International Journal of Climatology* 27, 17–46. <u>https://doi.org/10.1002/</u> joc.1377
- Auer, I., Foelsche, U., Böhm, R., Chimani, L., Haimberger, H., Kerschner, K. A., et al. (2014). "Vergangene Klimaänderung in Österreich", in Österreichischer Sachstandbericht Klimawandel 214 (AAR14). Austrian Panel on Climate Change (APCC), (Wien, Österreich: Verlag der Österreichischen Akademie der Wissenschaften), 227–300. <u>https://dor.org/10.1553/aar14</u>
- Aulitzky, H., Heuberger, H., and Patzelt, G. (1994). Mountain Hazard Geomorphology of Tyrol and Vorarlberg, Austria. *Mountain Research and Development* 14, 273–305. <u>https://doi.org/10.</u> 2307/3673725
- Austrian Wind Energy Association (2019). Windkraft im Alpenkonventionsgebiet. Available at: <u>https://www.igwindkraft.at/?x-</u> <u>mlval\_ID\_KEY[0]=1332</u> (Accessed December 12, 2024).
- Balest, J., Bertin, S., Ciolli, M., Garegnani, G., Grilli, G., Haimerl, G., et al. (2015). "Renewable energy exploitation and ecosystem services: Alpine region and pilot area analysis", in *Sustainable Renewable Energy Planning in the Alps. A handbook for experts & decision makers*, (recharge.green project), 70–118. Available at: <a href="https://pure.iiasa.ac.at/id/eprint/11565/1/Handbook-Recharge-green\_final\_web.pdf">https://pure.iiasa.ac.at/id/eprint/11565/1/Handbook-Recharge-green\_final\_web.pdf</a>
- Ban, N., Caillaud, C., Coppola, E., Pichelli, E., Sobolowski, S., Adinolfi, M., et al. (2021). The first multi-model ensemble of regional climate simulations at kilometer-scale resolution, part I: evaluation of precipitation. *Climate Dynamics* 57, 275–302. <u>https://doi.org/10.1007/s00382-021-05708-w</u>
- Ban, N., Rajczak, J., Schmidli, J., and Schär, C. (2020). Analysis of Alpine precipitation extremes using generalized extreme value theory in convection-resolving climate simulations. *Climate Dynamics* 55, 61–75. <u>https://doi.org/10.1007/s00382-018-4339-4</u>
- Ban, N., Schmidli, J., and Schär, C. (2015). Heavy precipitation in a changing climate: Does short-term summer precipitation increase faster? *Geophysical Research Letters* 42, 1165–1172. <u>https://doi.org/10.1002/2014GL062588</u>
- Bardgett, R. D., Bowman, W. D., Kaufmann, R., and Schmidt, S. K. (2005). A temporal approach to linking aboveground and belowground ecology. *Trends in Ecology & Evolution* 20, 634–641. <u>https://doi.org/10.1016/j.tree.2005.08.005</u>
- Bardgett, R. D., and van der Putten, W. H. (2014). Belowground biodiversity and ecosystem functioning. *Nature* 515, 505–511. <u>https://doi.org/10.1038/nature13855</u>
- Bar-Massada, A., Alcasena, F., Schug, F., and Radeloff, V. C. (2023). The wildland – urban interface in Europe: Spatial patterns and associations with socioeconomic and demographic variables. *Landscape and Urban Planning* 235, 104759. <u>https://doi.org/10.1016/j.landurbplan.2023.104759</u>
- Bauhus, J., Dieter, M., Farwig, N., Hafner, A., Kätzel, R., Kleinschmit, B., et al. (2021). Die Anpassung von Wäldern und Waldwirtschaft an den Klimawandel. Berichte über Landwirtschaft.

Zeitschrift für Agrarpolitik und Landwirtschaft Sonderheft Nr. 233. Available at: <u>https://buel.bmel.de/index.php/buel/article/</u> view/386/590

- Baumgarten, A., Lapin, K., Schüler, S., Freudenschuss, A., Grüneis, H., Konrad, H., et al. (2024). "Kapitel 4. Anpassungsoptionen in der Landnutzung an den Klimawandel", in APCC Special Report: Landnutzung und Klimawandel in Österreich, eds. R. Jandl, U. Tappeiner, C. B. Foldal, and K. Erb (Berlin, Heidelberg: Springer Spektrum), 217–274. <u>https://doi.org/10.1007/978-3-662-67864-0\_6</u>
- BCP Brenner Corridor Platform (2021). Freight traffic study executive summary. Available at: <u>https://www.bcplatform.eu/korridorstudien/</u>
- Bebi, P., Kulakowski, D., and Rixen, C. (2009). Snow avalanche disturbances in forest ecosystems—State of research and implications for management. *Forest Ecology and Management* 257, 1883–1892. <u>https://doi.org/10.1016/j.foreco.2009.01.050</u>
- Beniston, M. (2006). August 2005 intense rainfall event in Switzerland: Not necessarily an analog for strong convective events in a greenhouse climate. *Geophysical Research Letters* 33, L05701. <u>https://doi.org/10.1029/2005GL025573</u>
- Beniston, M., Farinotti, D., Stoffel, M., Andrassen, L. M., Coppola, E., Eckert, N., et al. (2018). The European mountain cryosphere: a review of its current state, trends, and future challenges. *The Cryosphere* 12, 759–794. https://doi.org/10.5194/tc-12-759-2018
- Berger, T., Amann, C., Formayer, H., Korjenic, A., Pospichal, B., Neururer, C., et al. (2014a). Impacts of urban location and climate change upon energy demand of office buildings in Vienna, Austria. *Building and Environment* 81, 258–269. <u>https://doi. org/10.1016/j.buildenv.2014.07.007</u>
- Berger, T., Amann, C., Formayer, H., Korjenic, A., Pospischal, B., Neururer, C., et al. (2014b). Impacts of climate change upon cooling and heating energy demand of office buildings in Vienna, Austria. *Energy and Buildings* 80, 517–530. <u>https://doi. org/10.1016/j.enbuild.2014.03.084</u>
- BFW (2022). Österreichische Waldinventur. Waldinformationen aus erster Hand. Umfassend. Kompetent. Aktuell. Available at: <u>https://waldinventur.at</u> (Accessed October 30, 2024).
- Biskaborn, B. K., Smith, S. L., Noetzli, J., Matthes, H., Vieira, G., Streletskiy, D. A., et al. (2019). Permafrost is warming at a global scale. *Nature Communications* 10, 264. <u>https://doi.org/10.1038/</u> s41467-018-08240-4
- Blättler, K., Wallimann, H., and von Arx, W. (2024). Free public transport to the destination: A causal analysis of tourists' travel mode choice. *Transportation Research Part A: Policy and Practice* 187, 104166. <u>https://doi.org/10.1016/j.tra.2024.104166</u>
- Blöschl, G. (2020). "Hochwasser", in ExtremA 2019. Aktueller Wissensstand zu Extremereignissen alpiner Naturgefahren in Österreich, eds. T. Glade, M. Mergili, and K. Sattler (Vienna University Press), 229–246.
- Blöschl, G., Blaschke, A. P., Haslinger, K., Hofstätter, M., Parajka, J., Salinas, J., et al. (2018). Auswirkungen der Klimaänderung auf Österreichs Wasserwirtschaft – ein aktualisierter Statusbericht. Österreichische Wasser- und Abfallwirtschaft 70, 462–473. https://doi.org/10.1007/s00506-018-0498-0
- Blöschl, G., Parajka, J., Blaschke, A. P., Hofstätter, M., Haslinger, K., and Schöner, W. (2017). *Klimawandel in der Wasserwirtschaft*. Wien, Österreich: Bundesministerium für Land- und Forstwirtschaft, Umwelt und Wasserwirtschaft. Available at: <u>https://info.bml.gv.at/dam/jcr:e137df9c-4431-458b-a84e-e97c</u> <u>31385648/Klimawandel\_2017\_final29112017fv4.pdf</u>

- Blöschl, G., Schöner, W., Kroiß, H., P. Blaschke, A., Böhm, R., Haslinger, K., et al. (2011a). Anpassungsstrategien an den Klimawandel für Österreichs Wasserwirtschaft – Ziele und Schlussfolgerungen der Studie für Bund und Länder. Österreichische Wasser- und Abfallwirtschaft 63, 1–10. <u>https://doi.org/10.1007/s00506-010-0274-2</u>
- Blöschl, G., Viglione, A., Merz, R., Parajka, J., Salinas, J. L., and Schöner, W. (2011b). Auswirkungen des Klimawandels auf Hochwasser und Niederwasser. Österreichische Wasser- und Abfallwirtschaft 63, 21–30. <u>https://doi.org/10.1007/s00506-010-0269-z</u>
- BML (2022). Wildschadensbericht 2021. Vienna: BML. Available at: <u>https://info.bml.gv.at/themen/wald/wald-in-oesterreich/ wald-wild-und-jagd/Wildschadensbericht/wildschadensbericht-2021.html (Accessed November 5, 2024).</u>
- BMLRT (2020). *Tourismus Österreich 2019*. Vienna, Austria. Available at: <u>https://www.bmaw.gv.at/dam/jcr:4d7bd6b1-31e3-4c64-</u> <u>a9cb-f83ffcb7b90f/TOURISMUSBERICHT%202019\_final.pdf</u> (Accessed November 5, 2024).
- BMNT (2017). Plan T Masterplan für Tourismus. Available at: <u>https://www.bmaw.gv.at/Themen/Tourismus/plan-t.html</u> (Accessed November 5, 2024).
- Bohner, A. (2022). "Klimawandelanpassungsstrategien im Dauergrünland", in 8. Umweltökologisches Symposium Klimaschutz und Klimaanpassung, ed. Höhere Bundeslehr-und Forschungsanstalt für Landwirtschaft Raumberg-Gumpenstein (Irdning-Donnersbachtal: Höhere Bundeslehr- und Forschungsanstalt für Landwirtschaft Raumberg-Gumpenstein), 57–62. Available at: <u>https://raumberg-gumpenstein.at/jdownloads/Tagungen/Umweltoekologisches\_Symposium/Umweltoekologisches\_Symposium\_2022/2u\_2022\_bohner.pdf</u> (Acccessed December 19, 2024).
- Bohner, A. (2023). Strategien und Massnahmen zur Anpassung an den Klimawandel im Dauergrünland. Zeitschrift Wasserland Steiermark, 12–15. Available at: <u>https://raumberg-gumpenstein.</u> <u>at/forschung/infothek/downloads/download.html?path=For-</u> <u>schungsberichte%252F2\_2023\_Artikel\_Wasserland\_HP.pdf</u>
- Bohner, A., Foldal, C., and Jandl, R. (2017). Kohlenstoffvorräte und Sequestrierungspotential von Dauergrünlandböden. Arbeitsgemeinschaft für Lebensmittel-, Veterinär- und Agrarwesen 72, 298–300.
- Brang, P., Schönenberger, W., Ott, E., and Gardner, B. (2001). "Forests as Protection from Natural Hazards", in *The Forests Handbook: Applying Forest Science for Sustainable Management*, ed. J. Evans (Oxford: Blackwell Science), 53–81.
- Braun, L. N. (1995). Les résultats principaux de 100 ans de recherche au Vernagtferner, Oetztal (Autriche). La Houille Blanche 81, 109–110. <u>https://doi.org/10.1051/lhb/1995054</u>
- Braun, M., Fritz, D., Weiss, P., Braschel, N., Büchsenmeister, R., Freudenschuß, A., et al. (2016). A holistic assessment of greenhouse gas dynamics from forests to the effects of wood products use in Austria. *Carbon Management* 7, 271–283. <u>https://doi.org/ 10.1080/17583004.2016.1230990</u>
- Broadbent, A. A. D., Snell, H. S. K., Michas, A., Pritchard, W. J., Newbold, L., Cordero, I., et al. (2021). Climate change alters temporal dynamics of alpine soil microbial functioning and biogeochemical cycling via earlier snowmelt. *The ISME Journal* 15, 2264–2275. https://doi.org/10.1038/s41396-021-00922-0
- Broekel, T., and Alfken, C. (2015). Gone with the wind? The impact of wind turbines on tourism demand. *Energy Policy* 86, 506– 519. <u>https://doi.org/10.1016/j.enpol.2015.08.005</u>

- Brönnimann, S., Rajczak, J., Fischer, E. M., Raible, C. C., Rohrer, M., and Schär, C. (2018). Changing seasonality of moderate and extreme precipitation events in the Alps. *Natural Hazards and Earth System Sciences* 18, 2047–2056. <u>https://doi.org/10.5194/</u> nhess-18-2047-2018
- Brudermann, T., Zaman, R., and Posch, A. (2019). Not in my hiking trail? Acceptance of wind farms in the Austrian Alps. *Clean Technologies and Environmental Policy* 21, 1603–1616. <u>https:// doi.org/10.1007/s10098-019-01734-9</u>
- Buckel, J., Otto, J. C., Prasicek, G., and Keuschnig, M. (2018). Glacial lakes in Austria – Distribution and formation since the Little Ice Age. *Global and Planetary Change* 164, 39–51. <u>https://doi. org/10.1016/j.gloplacha.2018.03.003</u>
- Bundeswasserbauverwaltung Tirol (2016). Hochwasserschutz Tiroler Unterinntal. Available at: <u>https://wasserverband-uui.at/</u> <u>wp-content/uploads/hws-uui\_gp\_a01\_100\_zusfasbericht\_16</u> <u>1230.pdf</u>
- Bursa, B., Mailer, M., and Axhausen, K. W. (2022). Travel behavior on vacation: transport mode choice of tourists at destinations. *Transportation Research Part A: Policy and Practice* 166, 234– 261. <u>https://doi.org/10.1016/j.tra.2022.09.018</u>
- Bursa, B., Tappeiner, G., Vicoli, S., Mölk, F., and Mailer, M. (2024). Drive Them to Rail—How Can Tourist Destinations Increase the Market Share of Rail Travel? A Discrete Choice Experiment. *International Journal of Tourism Research* 26, e2799. <u>https://doi.org/10.1002/jtr.2799</u>
- Burton, A., Fritz, O., and Steiger, R. (2024). Herausforderungen des alpinen Wintertourismus in Österreich. WIFO-Monatsberichte 97, 165–176. <u>https://www.wifo.ac.at/publication/pid/51444044</u>
- Caduff, M. E., Brožová, N., Kupferschmid, A. D., Krumm, F., and Bebi, P. (2022). How large-scale bark beetle infestations influence the protective effects of forest stands against avalanches: A case study in the Swiss Alps. *Forest Ecology and Management* 514, 120201. <u>https://doi.org/10.1016/j.foreco.2022.120201</u>
- Cammerer, H., Thieken, A. H., and Verburg, P. H. (2013). Spatio-temporal dynamics in the flood exposure due to land use changes in the Alpine Lech Valley in Tyrol (Austria). *Nat Hazards* 68, 1243–1270. <u>https://doi.org/10.1007/s11069-012-0280-8</u>
- Castebrunet, H., Eckert, N., Giraud, G., Durand, Y., and Morin, S. (2014). Projected changes of snow conditions and avalanche activity in a warming climate: the French Alps over the 2020–2050 and 2070–2100 periods. *The Cryosphere* 8, 1673–1697. <u>https:// doi.org/10.5194/tc-8-1673-2014</u>
- Cavallaro, F., Ciari, F., Nocera, S., Prettenthaler, F., and Scuttari, A. (2017). The impacts of climate change on tourist mobility in mountain areas. *Journal of Sustainable Tourism* 25, 1063–1083. <u>https://doi.org/10.1080/09669582.2016.1253092</u>
- Caviezel, C., Hunziker, M., and Kuhn, N. J. (2017). Green alder encroachment in the European Alps: The need for analyzing the spread of a native-invasive species across spatial data. *Catena* 159, 149–158. <u>https://doi.org/10.1016/j.catena.2017.08.006</u>
- Chadburn, S. E., Burke, E. J., Cox, P. M., Friedlingstein, P., Hugelius, G., and Westermann, S. (2017). An observation-based constraint on permafrost loss as a function of global warming. *Nature Climate Change* 7, 340–344. <u>https://doi.org/10.1038/ nclimate3262</u>
- Chai, B., and Seto, K. C. (2019). Conceptualizing and characterizing micro-urbanization: A new perspective applied to Africa. *Landscape and Urban Planning* 190. <u>https://doi.org/10.1016/j.</u> <u>landurbplan.2019.103595</u>

- Chakraborty, D., Ciceu, A., Ballian, D., Benito Garzón, M., Bolte, A., Bozic, G., et al. (2024). Assisted tree migration can preserve the European forest carbon sink under climate change. *Nature Climate Change* 14, 845–852. <u>https://doi.org/10.1038/s41558-024-02080-5</u>
- Chimani, B., Heinrich, G., Hofstätter, M., Kerschbaumer, M., Kienberger, S., Leuprecht, A., et al. (2015). ÖKS15: Klimaszenarien für Österreich. Daten – Methoden – Klimaanalyse. Endbericht, ed. GeoSphere Austria. Wien: CCCA. <u>https://doi.org/10.60669/ wgxm-4927</u>
- Chytrý, K., Helm, N., Hülber, K., Moser, D., Wessely, J., Hausharter, J., et al. (2024). Limited impact of microtopography on alpine plant distribution. *Ecography*, e06744. <u>https://doi.org/10.1111/ecog.06744</u>
- Ciolli, M., Garegnani, G., Hastik, R., Kraxner, F., Kuenzer, N., Miotello, F., et al. (2015). Energy & nature in the Alps: a balancing act. Perpetual calendar. Vienna: recharge.green Project. Available at: <u>https://pure.iiasa.ac.at/12117</u>
- Clarke, B., Otto, F., Stuart-Smith, R., and Harrington, L. (2022). Extreme weather impacts of climate change: an attribution perspective. *Environmental Research: Climate* 1, 012001. <u>https:// doi.org/10.1088/2752-5295/ac6e7d</u>
- Cook, S. J., Jouvet, G., Millan, R., Rabatel, A., Zekollari, H., and Dussaillant, I. (2023). Committed ice loss in the European Alps until 2050 using a deep-learning-aided 3D ice-flow model with data assimilation. *Geophysical Research Letters* 50, e2023GL105029. <u>https://doi.org/10.1029/2023GL105029</u>
- Coppola, E., Sobolowski, S., Pichelli, E., Raffaele, F., Ahrens, B., Anders, I., et al. (2020). A first-of-its-kind multi-model convection permitting ensemble for investigating convective phenomena over Europe and the Mediterranean. *Climate Dynamics* 55, 3–34. <u>https://doi.org/10.1007/s00382-018-4521-8</u>
- Correa, L. F., Folini, D., Chtirkova, B., and Wild, M. (2022). Causes for decadal trends in Surface Solar Radiation in the Alpine region. <u>https://doi.org/10.22541/essoar.167160613.35436973/v1</u>
- Cotto, O., Wessely, J., Georges, D., Klonner, G., Schmid, M., Dullinger, S., et al. (2017). A dynamic eco-evolutionary model predicts slow response of alpine plants to climate warming. *Nature Communications* 8, 15399. <u>https://doi.org/10.1038/ncomms15399</u>
- De Boeck, H. J., Bassin, S., Verlinden, M., Zeiter, M., and Hiltbrunner, E. (2016). Simulated heat waves affected alpine grassland only in combination with drought. *New Phytologist* 209, 531– 541. <u>https://doi.org/10.1111/nph.13601</u>
- De Stefano, A., and Jacobson, M. G. (2017). Soil carbon sequestration in agroforestry systems: a meta-analysis. Agroforestry Systems 92, 285–299. <u>https://doi.org/10.1007/s10457-017-0147-9</u>
- Diez, J., Kauserud, H., Andrew, C., Heegaard, E., Krisai-Greilhuber, I., Senn-Irlet, B., et al. (2020). Altitudinal upwards shifts in fungal fruiting in the Alps. *Proceedings of the Royal Society B-Biological Sciences* 287. <u>https://doi.org/10.1098/rspb.2019.2348</u>
- Dirnböck, T., Essl, F., and Rabitsch, W. (2011). Disproportional risk for habitat loss of high-altitude endemic species under climate change. *Global Change Biology* 17, 990–996. <u>https://doi. org/10.1111/j.1365-2486.2010.02266.x</u>
- Dobrowski, S. Z., and Parks, S. A. (2016). Climate change velocity underestimates climate change exposure in mountainous regions. *Nature Communications* 7, 12349. <u>https://doi.org/10.</u> <u>1038/ncomms12349</u>

- Dowling, P. (2013). The impact of climate change on the European energy system. *Energy Policy* 60, 406–417. <u>https://doi.org/10.1016/j.enpol.2013.05.093</u>
- Dubo, T., Palomo, I., Camacho, L. L., Locatelli, B., Cugniet, A., Racinais, N., et al. (2022). Nature-based solutions for climate change adaptation are not located where they are most needed across the Alps. *Regional Environmental Change* 23, 12. <u>https:// doi.org/10.1007/s10113-022-01998-w</u>
- Duethmann, D., Blöschl, G., and Parajka, J. (2020). Why does a conceptual hydrological model fail to correctly predict discharge changes in response to climate change? *Hydrology and Earth System Sciences* 24, 3493–3511. <u>https://doi.org/10.5194/hess-24-3493-2020</u>
- Đukan, M., Gut, D., Gumber, A., and Steffen, B. (2024). Harnessing solar power in the Alps: A study on the financial viability of mountain PV systems. *Applied Energy* 375, 124019. <u>https://doi. org/10.1016/j.apenergy.2024.124019</u>
- Dullinger, S., Gattringer, A., Thuiller, W., Moser, D., Zimmermann, N. E., Guisan, A., et al. (2012). Extinction debt of high-mountain plants under twenty-first-century climate change. *Nature Climate Change* 2, 619–622. <u>https://doi.org/10.1038/NCLI-MATE1514</u>
- Eckert, N., Baya, H., and Deschatres, M. (2010). Assessing the Response of Snow Avalanche Runout Altitudes to Climate Fluctuations Using Hierarchical Modeling: Application to 61 Winters of Data in France. *Journal of Climate* 23, 3157–3180. <u>https://doi. org/10.1175/2010JCLI3312.1</u>
- Eckert, N., Corona, C., Giacona, F., Gaume, J., Mayer, S., van Herwijnen, A., et al. (2024). Climate change impacts on snow avalanche activity and related risks. *Nature Reviews Earth & Environment* 5, 369–389. <u>https://doi.org/10.1038/s43017-024-00540-2</u>
- Eder, S., Poscher, G., and Prager, C. (2006). Risk Analyses and Risk Management – Slope Instabilities in Alpine Environments., in *Geohazards*, (Lillehammer, Norway). Available at: <u>https://dc.en-gconfintl.org/geohazards/15/</u> (Accessed November 5, 2024).
- Eitzinger, J., Haberl, H., Amon, B., Blamauer, B., Essl, F., Gaube, V., et al. (2014). "Land- und Forstwirtschaft, Wasser, Ökosysteme und Biodiversität", in Österreichischer Sachstandsbericht Klimawandel 2014 (AAR14), Austrian Panel on Climate Change (APCC), ed. Austrian Panel on Climate Change (APCC) (Vienna: Verlag der Österreichischen Akademie der Wissenschaften), 771–856.
- Elsen, P. R., Monahan, W. B., and Merenlender, A. M. (2018). Global patterns of protection of elevational gradients in mountain ranges. *Proceedings of the National Academy of Sciences of the* USA 115, 6004–6009. https://doi.org/10.1073/pnas.1720141115
- Elsen, P. R., and Tingley, M. W. (2015). Global mountain topography and the fate of montane species under climate change. *Nature Climate Change* 5, 772–776. <u>https://doi.org/10.1038/</u> nclimate2656
- Embleton-Hamann, C. (2007). "Geomorphological hazards in Austria", in *Geomorphology for the future*, eds. M. Keiler, C. Embleton-Hamann, J. Stötter, and A. Kellerer-Pirklbauer (Innsbruck: Innsbruck University Press), 33–56.
- Energieinstitut der Wirtschaft GmbH (2012). Energiekennzahlen in Dienstleistungsgebäuden. Energieinstitut der Wirtschaft GmbH. Available at: <u>https://www.energieinstitut.net/de/system/files/0903\_final\_dienstleistungsgebaude\_20120530.pdf</u> (Accessed June 7, 2024).

- European Commission and Joint Research Centre (JRC) (2021). ENSPRESO – INTEGRATED DATA. Available at: <u>http://data.</u> <u>europa.eu/89h/88d1c405-0448-4c9e-b565-3c30c9b167f7</u> (Accessed May 15, 2024).
- European Union (2005). Protocol on the implementation of the Alpine Convention of 1991 in the field of energy. Energy Protocol. Official journal of the European Union. *Official Journal of the European Union* L337, 36–46.
- European Union (2010). "A revised urban-rural typology", in *Eurostat regional yearbook 2010*, (Luxemburg: Publications Office of the European Union), 239–253. Available at: <u>https://ec.europa.eu/eurostat/documents/3217494/5726181/KS-HA-10-001-15-EN.PDF/5499ee07-b61e-4615-9631-ed76e2a31f81?version=1.0</u>
- EY Parthenon, Booking.com, and OC&C Strategy consultants (2021). Global accommodation sector: The road to net zero emissions. Available at: <u>https://app.box.com/s/4ffylkg4jetbx98</u> infh7wowd6l89m0hc (Accessed December 12, 2024).
- Fatichi, S., Rimkus, S., Burlando, P., Bordoy, R., and Molnar, P. (2015). High-resolution distributed analysis of climate and anthropogenic changes on the hydrology of an Alpine catchment. *Journal of Hydrology* 525, 362–382. <u>https://doi.org/10.1016/j.jhydrol.2015.03.036</u>
- Feyen, L., Ciscar, J., Gosling, S., Ibarreta, D., Soria, A., Ibarreta, D., et al. (2020). Climate change impacts and adaptation in Europe: JRC PESETA IV final report. Luxembourg: Publications Office of the European Union. <u>https://doi.org/10.2760/171121</u>, JRC119178
- Fickert, T. (2020). Common patterns and diverging trajectories in primary succession of plants in Eastern Alpine Glacier Forelands. *Diversity* 12, 191. <u>https://doi.org/10.3390/d12050191</u>
- Fickert, T., and Grüninger, F. (2018). High-speed colonization of bare ground – Permanent plot studies on primary succession of plants in recently deglaciated glacier forelands. *Land Degradation & Development* 29, 2668–2680. <u>https://doi.org/10.1002/</u> ldr.3063
- Finger, D., Heinrich, G., Gobiet, A., and Bauder, A. (2012). Projections of future water resources and their uncertainty in a glacierized catchment in the Swiss Alps and the subsequent effects on hydropower production during the 21<sup>st</sup> century. *Water Resources Research* 48, 2011WR010733. <u>https://doi.org/10.1029/2011WR010733</u>
- Fischer, A., Fickert, T., Schweizer, G., Patzelt, G., and Gross, G. (2019). Vegetation dynamics in Alpine glacier forelands tackled from space. *Scientific Reports* 9, 13918. <u>https://doi.org/10.1038/ s41598-019-50273-2</u>
- Fischer, A., Seiser, B., Stocker Waldhuber, M., Mitterer, C., and Abermann, J. (2015). Tracing glacier changes in Austria from the Little Ice Age to the present using a lidar-based high-resolution glacier inventory in Austria. *The Cryosphere* 9, 753–766. <u>https://doi.org/10.5194/tc-9-753-2015</u>
- Fleischer, F., Otto, J. C., Junker, R. R., and Hölbling, D. (2021). Evolution of debris cover on glaciers of the Eastern Alps, Austria, between 1996 and 2015. *Earth Surface Processes and Landforms* 46, 1673–1691. https://doi.org/10.1002/esp.5065
- Formayer, H., and Fritz, A. (2016). Temperature dependency of hourly precipitation intensities – surface versus cloud layer temperature. *International Journal of Climatology* 37, 1–10. <u>https:// doi.org/10.1002/joc.4678</u>
- Formayer, H., Nadeem, I., Leidinger, D., Maier, P., Schöniger, F., Suna, D., et al. (2023). SECURES-Met: A European meteorological data set suitable for electricity modelling applications.

Scientific Data 10, 590. <u>https://doi.org/10.1038/s41597-023-</u>02494-4

- François, H., Samacoïts, R., Bird, D. N., Köberl, J., Prettenthaler, F., and Morin, S. (2023). Climate change exacerbates snow-water-energy challenges for European ski tourism. *Nature Climate Change* 13, 935–942. <u>https://doi.org/10.1038/s41558-023-017</u> <u>59-5</u>
- Frehner, M., Wasser, B., and Schwitter, R. (2005). NaiS. Nachhaltigkeit und Erfolgskontrolle im Schutzwald. Wegleitung für Pflegemassnahmen in Wäldern mit Schutzfunktion. Bern, Switzerland: Bundesamt für Umwelt, Wald und Landschaft (BUWAL). Available at: <u>https://www.bafu.admin.ch/dam/ bafu/de/dokumente/naturgefahren/uv-umwelt-vollzug/nachhaltigkeit-und-erfolgskontrolle-im-schutzwald-nais.pdf.download.pdf/UV-2409-D\_NaiS.pdf</u>
- Frei, P., Kotlarski, S., Liniger, M. A., and Schär, C. (2018). Future snowfall in the Alps: projections based on the EURO-CORDEX regional climate models. *The Cryosphere* 12, 1–24. <u>https://doi.org/10.5194/tc-12-1-2018</u>
- Fritz, O., Burton, A., Ehn-Fragner, S., Streicher, G., Laimer, P., Orsolic, I., et al. (2021). Ein Regionales Tourismus-Satellitenkonto für Tirol. Wien: Österreichisches Institut für Wirtschaftsforschung, Statistik Austria. Availabe at: <u>https://www.wifo.ac.at/</u> <u>publication/pid/4807555</u>
- Fuchs, S., Keiler, M., and Sokratov, S. (2015a). "Snow and avalanches", in *The high-mountain cryosphere: environmental changes* and human risks, eds. C. Huggel, M. Carey, J. J. Clague, and A. Kääb (Cambridge: Cambridge University Press), 50–70.
- Fuchs, S., Keiler, M., and Zischg, A. (2015b). A spatiotemporal multi-hazard exposure assessment based on property data. *Nat. Hazards Earth Syst. Sci.* 15, 2127–2142. <u>https://doi.org/10.5194/</u> <u>nhess-15-2127-2015</u>
- Fuchs, S., Röthlisberger, V., Thaler, T., Zischg, A., and Keiler, M. (2017). Natural Hazard Management from a Coevolutionary Perspective: Exposure and Policy Response in the European Alps. Annals of the American Association of Geographers 107, 382–392. https://doi.org/10.1080/24694452.2016.1235494
- Fuchs, S., Wenk, M., and Keiler, M. (2022). "Geomorphic Hazards in Austria", in *Landscapes and Landforms of Austria*, ed. C. Embleton-Hamann (Cham: Springer International Publishing), 105–117. https://doi.org/10.1007/978-3-030-92815-5\_6
- Gassner, C., Promper, C., Beguería, S., and Glade, T. (2015). "Climate Change Impact for Spatial Landslide Susceptibility", in *Engineering Geology for Society and Territory Volume 1*, eds. G. Lollino, A. Manconi, J. Clague, W. Shan, and M. Chiarle (Cham: Springer), 429–433. <u>https://doi.org/10.1007/978-3-319-09300-0\_82</u>
- Gaube, V., Schneider, S., Thaler, T., Borsky, S., Kirchner, M., Loibl, W., et al. (2024). "Kapitel 3. Sozioökonomische und klimatische Treiber der Änderung der Landnutzung in Österreich", in APCC Special Report: Landnutzung und Klimawandel in Österreich, eds. R. Jandl, U. Tappeiner, C. B. Foldal, and K. Erb (Berlin, Heidelberg: Springer Spektrum), 163–216. <u>https://doi.org/10.1007/978-3-662-67864-0\_5</u>
- Gehrig-Fasel, J., Guisan, A., and Zimmermann, N. E. (2007). Tree line shifts in the Swiss Alps: Climate change or land abandonment? *Journal of Vegetation Science* 18, 571–582. <u>https://doi.org/10.1111/j.1654-1103.2007.tb02571.x</u>
- Gems, B., Kammerlander, J., Moser, M., and Aufleger, M. (2020). "Fluviatile Feststoffereignisse", in *ExtremA 2019. Aktueller Wissensstand zu Extremereignissen alpiner Naturgefahren in Öster-*

- GeoSphere Austria (2017). HISTALP: Historical Instrumental Climatological Surface Time Series Of The Greater Alpine Region. Available at: <u>www.zamg.ac.at/histalp</u> (Accessed October 30, 2024).
- GeoSphere Austria (2022). Änderung der Landnutzung kann Hitzeinseln in Städten verstärken oder mildern. Available at: <u>https://www.zamg.ac.at/cms/de/klima/news/aenderung-der-landnutzung-kann-hitzeinseln-in-staedten-verstaerken-oder-mildern</u> (Accessed November 5, 2024).
- Gerersdorfer, T., Frank, A., Formayer, H., Haas, P., and Moshammer, H. (2009). Untersuchung zur nächtlichen Abkühlung in einem sich ändernden Klima (Endbericht StartClim2005.A1b, November 2006). BOKU-Met Report 8. Available at: <u>http://www.boku.ac.at/met/report/BOKU-Met\_Report\_08\_online.pdf</u>
- Giacona, F., Eckert, N., Corona, C., Mainieri, R., Morin, S., Stoffel, M., et al. (2021). Upslope migration of snow avalanches in a warming climate. *Proceedings of the National Academy of Scienc*es 118, e2107306118. <u>https://doi.org/10.1073/pnas.2107306118</u>
- Gingrich, S., Weiss, P., Wenzel, W., Anderl, M., Bruckner, M., Díaz-Pinés, E., et al. (2024). "Kapitel 2. Auswirkungen der Landnutzung und -bewirtschaftung sowie naturnaher Ökosysteme auf den Klimawandel: Biophysikalische Effekte, Treibhausgasemissionen und Kohlenstoffspeicher", in *APCC Special Report: Landnutzung und Klimawandel in Österreich*, eds. R. Jandl, U. Tappeiner, C. B. Foldal, and K. Erb (Berlin, Heidelberg: Springer Spektrum), 107–162. https://doi.org/10.1007/978-3-662-67864-0\_4
- Giorgi, F., Torma, C., Coppola, E., Ban, N., Schär, C., and Somot, S. (2016). Enhanced summer convective rainfall at alpine high elevations in response to climate warming. *Nature Geoscience* 9, 584–589. https://doi.org/10.1038/ngeo2761
- Glade, T., Bell, R., Dobesberger P, Embleton-Hamann, C., Fromm, R., Fuchs, S., et al. (2014). "Der Einfluss des Klimawandels auf die Reliefsphäre", in Österreichischer Sachstandsbericht Klimawandel 2014 (AAR14), ed. Austrian Panel on Climate Change (APPCC) (Vienna: Verlag der Österreichischen Akademie der Wissenschaften), 557–600.
- Gobiet, A. (2021). Future Snow Cover Evolution in Austria. Available at: <u>https://fuse-at.cccaa.cat/</u> (Accessed October 30, 2024).
- Gobiet, A., Kotlarski, S., Beniston, M., Heinrich, G., Rajczak, J., and Stoffel, M. (2014). 21<sup>st</sup> century climate change in the European Alps – A review. *The Science of the Total Environment* 493, 1138–1151. <u>https://doi.org/10.1016/j.scitotenv.2013.07.050</u>
- Gössling, S., Scott, D., and Hall, C. M. (2018). Global trends in length of stay: implications for destination management and climate change. *Journal of Sustainable Tourism* 26, 2087–2101. https://doi.org/10.1080/09669582.2018.1529771
- Gottfried, M., Pauli, H., Futschik, A., Akhalkatsi, M., Barancok, P., Benito Alonso, J. L., et al. (2012). Continent-wide response of mountain vegetation to climate change. *Nature Climate Change* 2, 111–115. <u>https://doi.org/10.1038/nclimate1329</u>
- Grabherr, G., and Ringler, A. (2018). Grünland der Alpen: Typen, Erhaltungsprobleme und Entwicklungsperspektiven. Jahrbuch des Vereins zum Schutz der Bergwelt (München) 83, 117–164.
  Available at: <u>https://www.zobodat.at/pdf/Jb-Verein-Schutz-Bergwelt\_83\_2018\_0117-0164.pdf</u>
- Greilinger, M., Reisenhofer, S., and Riedl, C. (2023). Hoher Sonnblick. KryoMon.AT – Kryosphären Monitoring Österreich 2021/22. Kryosphärenbericht Nr. 1. https://doi.org/10.25364/402.2023.1

- Gronau, W., and Kagermeier, A. (2007). Key factors for successful leisure and tourism public transport provision. *Journal of Transport Geography* 15, 127–135. <u>https://doi.org/10.1016/j.</u> jtrangeo.2006.12.008
- Grüneis, H., Penker, M., and Höferl, K.-M. (2016). The full spectrum of climate change adaptation: testing an analytical framework in Tyrolean mountain agriculture (Austria). *SpringerPlus* 5, 1848. <u>https://doi.org/10.1186/s40064-016-3542-1</u>
- Gühnemann, A., Kurzweil, A., and Mailer, M. (2021). Tourism mobility and climate change – a review of the situation in Austria. *Journal of Outdoor Recreation and Tourism* 34, 100382. <u>https:// doi.org/10.1016/j.jort.2021.100382</u>
- Haas, W., Weisz, U., Balas, M., McCallum, S., Lexer, W., Pazdernik, K., et al. (2008). Identifikation von Handlungsempfehlungen zur Anpassung an den Klimawandel in Österreich, 1. Phase. Vienna: AustroClim. On behalf of the BMLFUW.
- Haas, W., Weisz, U., Maier, P., Scholz, F., Themessl, M., Wolf, A., et al. (2014). Auswirkungen des Klimawandels auf die Gesundheit des Menschen. Available at: <u>https://ccca.ac.at/fileadmin/00</u> <u>DokumenteHauptmenue/02\_Klimawissen/FactSheets/6\_gesundheit\_v4\_02112015.pdf</u>
- Haberkorn, A., Kenner, R., Noetzli, J., and Phillips, M. (2021). Changes in ground temperature and dynamics in mountain permafrost in the Swiss Alps. *Frontiers in Earth Science* 9, 626686. <u>https://doi.org/10.3389/feart.2021.626686</u>
- Haberl, H., Löw, M., Perez-Laborda, A., Matej, S., Plank, B., Wiedenhofer, D., et al. (2023). Built structures influence patterns of energy demand and CO2 emissions across countries. *Nat Commun* 14, 3898. https://doi.org/10.1038/s41467-023-39728-3
- Habersack, H., Wagner, B., Hauer, C., Jäger, E., Krapesch, G., Strahlhofer, L., et al. (2011). DSS\_KLIM: EN: Entwicklung eines Decision Support Systems zur Beurteilung der Wechselwirkungen zwischen Klimawandel, Energie aus Wasserkraft ung Ökologie (Endbericht). Studie im Auftrag der Kommunalkredit Austria AG, Klima- und Enegiefonds. Vienna, Austria.
- Haeberli, W., Schaub, Y., and Huggel, C. (2017). Increasing risks related to landslides from degrading permafrost into new lakes in de-glaciating mountain ranges. *Geomorphology* 293, 405–417. <u>https://doi.org/10.1016/j.geomorph.2016.02.009</u>
- Hall, J., Arheimer, B., Borga, M., Brázdil, R., Claps, P., Kiss, A., et al. (2014). Understanding flood regime changes in Europe: a stateof-the-art assessment. *Hydrology and Earth System Sciences* 18, 2735–2772. <u>https://doi.org/10.5194/hess-18-2735-2014</u>
- Hallas, T., Steyrer, G., Laaha, G., and Hoch, G. (2024). Two unprecedented outbreaks of the European spruce bark beetle, Ips typographus L. (Col., Scolytinae) in Austria since 2015: Different causes and different impacts on forests. *Central European Forest*ry Journal 70, 263–274. <u>https://doi.org/10.2478/forj-2024-0014</u>
- Hanewinkel, M., Cullmann, D. A., Schelhaas, M.-J., Nabuurs, G.-J., and Zimmermann, N. E. (2013). Climate change may cause severe loss in the economic value of European forest land. *Nature Climate Change* 3, 203–207. <u>https://doi.org/10.1038/nclimate1687</u>
- Hansche, I., Fischer, A., Greilinger, M., Hartl, L., Hartmeyer, I., Helfricht, K., et al. (2023). KryoMon.AT – Kryosphären Monitoring Österreich: 2021/22 Kryosphärenbericht Nr. 1. Graz: Bundesministerium Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie. <u>https://doi.org/10.25364/402.2023.1</u>
- Hantel, M., Maurer, C., and Mayer, D. (2012). The snowline climate of the Alps 1961–2010. *Theoretical and Applied Climatology* 110, 517–537. <u>https://doi.org/10.1007/s00704-012-0688-9</u>

- Hanus, S., Hrachowitz, M., Zekollari, H., Schoups, G., Vizcaino, M., and Kaitna, R. (2021). Future changes in annual, seasonal and monthly runoff signatures in contrasting Alpine catchments in Austria. *Hydrol. Earth Syst. Sci.* 25, 3429–3453. <u>https:// doi.org/10.5194/hess-25-3429-2021</u>
- Hartl, L., Zieher, T., Bremer, M., Stocker-Waldhuber, M., Zahs, V., Höfle, B., et al. (2023). Multi-sensor monitoring and data integration reveal cyclical destabilization of the Äußeres Hochebenkar rock glacier. *Earth Surface Dynamics* 11, 117–147. <u>https:// doi.org/10.5194/esurf-11-117-2023</u>
- Hartmeyer, I. (2023). "Kitzsteinhorn", in KryoMon.AT Kryosphären Monitoring Österreich 2021/22. Kryosphärenbericht Nr. 1, ed. W. Schöner, 136–138. Available at: <u>https://unipub.unigraz.at/obvugrveroeff/content/titleinfo/9405860</u>
- Hartmeyer, I., Delleske, R., Keuschnig, M., Krautblatter, M., Lang, A., Otto, J.-C., et al. (2020). Current glacier recession causes significant rockfall increase: the immediate paraglacial response of deglaciating cirque walls. *Earth Surface Dynamics* 8, 729–751. <u>https://doi.org/10.5194/esurf-8-729-2020</u>
- Hartmeyer, I., and Otto, J.-C. (2024). Rockfall, glacier recession, and permafrost degradation: long-term monitoring of climate change impacts at the Open-Air-Lab Kitzsteinhorn, Hohe Tauern. DEUQUA Special Publications 5, 3–12. <u>https://doi.org/10. 5194/deuquasp-5-3-2024</u>
- Haslinger, K., Hofstätter, M., Kroisleitner, C., Schöner, W., Laaha, G., Holawe, F., et al. (2019). Disentangling Drivers of Meteorological Droughts in the European Greater Alpine Region during the Last Two Centuries. *Journal of Geophysical Research: Atmospheres* 124, 12404–12425. <u>https://doi.org/10.1029/ 2018JD029527</u>
- Haslinger, K., Schöner, W., Abermann, J., Laaha, G., Andre, K., Olefs, M., et al. (2022). Contradictory signal in future surface water availability in Austria: increase on average vs. higher probability of droughts, EGUsphere [preprint]. Available at: <u>https://egusphere.copernicus.org/preprints/2022/egusphere-2022-191/</u> (Accessed July 24, 2023).
- Haslinger, K., Schöner, W., Abermann, J., Laaha, G., Andre, K., Olefs, M., et al. (2023). Apparent contradiction in the projected climatic water balance for Austria: wetter conditions on average versus higher probability of meteorological droughts. *Natural Hazards and Earth System Sciences* 23, 2749–2768. <u>https://doi. org/10.5194/nhess-23-2749-2023</u>
- Haslmayr, H.-P., Baumgarten, A., Schwarz, M., Huber, S., Prokop, G., Sedy, K., et al. (2018). BEAT – Bodenbedarf für die Ernährungssicherung in Österreich. Bundesministerium für Nachhaltigkeit und Tourismus. Available at: <u>https://dafne.at/content/report\_</u> <u>release/aa85879d-af0f-4273-a1e2-b7f1d7178d41\_0.pdf</u>
- Heimerl, P., Haid, M., Benedikt, L., and Scholl-Grissemann, U. (2020). Factors Influencing Job Satisfaction in Hospitality Industry. SAGE Open 10. <u>https://doi.org/10.1177/2158244020982998</u>
- Heiser, M., Hübl, J., and Scheidl, C. (2019). Completeness analyses of the Austrian torrential event catalog. *Landslides* 16, 2115– 2126. <u>https://doi.org/10.1007/s10346-019-01218-3</u>
- Helfricht, K., Huss, M., Fischer, A., and Otto, J.-C. (2019). Calibrated Ice Thickness Estimate for All Glaciers in Austria. *Frontiers* in Earth Science 7, 68. <u>https://doi.org/10.3389/feart.2019.00068</u>
- Hiebl, J., Hartmeyer, I., Böhm, R., Schöner, W., Maugeri, M., Lentini, G., et al. (2009). A high-resolution 1961–1990 monthly temperature climatology for the Greater Alpine Region. *Meteorologische Zeitschrift* 18, 507–530. <u>https://doi.org/10.1127/0941-2948/2009/0403</u>

- Hiltbrunner, D., Zimmermann, S., and Hagedorn, F. (2013). Afforestation with Norway spruce on a subalpine pasture alters carbon dynamics but only moderately affects soil carbon storage. *Biogeochemistry* 115, 251–266. <u>https://doi.org/10.1007/s10533-013-9832-6</u>
- Hirschberg, J., Fatichi, S., Bennett, G. L., McArdell, B. W., Peleg, N., Lane, S. N., et al. (2021). Climate Change Impacts on Sediment Yield and Debris-Flow Activity in an Alpine Catchment. *Journal of Geophysical Research: Earth Surface* 126, e2020JF005739. <u>https://doi.org/10.1029/2020JF005739</u>
- Hlásny, T., Krokene, P., Liebhold, A., Montagné-Huck, C., Müller, J., Qin, H., et al. (2019). Living with bark beetles: impacts, outlook and management options. European Forest Institute. <u>https://doi.org/10.36333/fs08</u>
- Hoch, G., Schopf, A., and Weizer, G. eds. (2019). *Der Buchdrucker. Biologie, Ökologie, Management.*, 2nd Edn. Wien: BFW.
- Hock, R., Rasul, G., Adler, C., Cáceres, B., Gruber, S., Hirabayashi, Y., et al. (2019). "High Mountain Areas", in *IPCC Special Report* on the Ocean and Cryosphere in a Changing Climate, eds. H.-O. Pörtner, D. C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press), 131–202.
- Hofstätter, M., Jacobeit, J., Homann, M., Lexer, A., Chimani, B., Philipp, A., et al. (2015). WETRAX – Weather patterns, cyclone tracks and related precipitation extremes. Großflächige Starkniederschläge im Klimawandel in Mitteleuropa. Augsburg: Institut für Geographie, Universität Augsburg.
- Hohensinner, S., Atzler, U., Fischer, A., Schwaizer, G., and Helfricht, K. (2021). Tracing the Long-Term Evolution of Land Cover in an Alpine Valley 1820–2015 in the Light of Climate, Glacier and Land Use Changes. *Frontiers in Environmental Science* 9, 683397. <u>https://doi.org/10.3389/fenvs.2021.683397</u>
- Hoinkes, H. C. (1969). Surges of the Vernagtferner in the Ötztal Alps since 1599. *Canadian Journal of Earth Sciences* 6, 853–861. https://doi.org/10.1139/e69-086
- Holzmann, M., and Perzlmaier, S. (2022). Impact of an exceptionally rapid snow melt on the deformation characteristics of a rockslide interacting with a reservoir. *Geomechanics and Tunnelling* 15, 617–625. <u>https://doi.org/10.1002/geot.202200042</u>
- Hu, Z., Dietz, A., Zhao, A., Uereyen, S., Zhang, H., Wang, M., et al. (2020). Snow Moving to Higher Elevations: Analyzing Three Decades of Snowline Dynamics in the Alps. *Geophysical Research Letters* 47, e2019GL085742. <u>https://doi.org/10.1029/2019</u> GL085742
- Hübl, J., Ganahl, E., Bacher, M., Chiari, M., Holub, M., Kaitna, R., et al. (2005). Dokumentation der Wildbachereignisse von 22./23.
  August 2005 in Tirol. Band 1: Generelle Aufnahme (5W-Standard); IAN Report 109 Band 1. Wien (unveröffentlicht).
- Humer, L., and Spiegelfeld, A. (2020). Seasonal workers from third countries in Austria. Vienna: Emn-Austria Study, International Organization for Migration (IOM). Available at: <u>https://www. emn.at/wp-content/uploads/2020/11/emn-national-report-2020-seasonal-workers-in-at.pdf</u>
- Huss, M., and Hock, R. (2018). Global-scale hydrological response to future glacier mass loss. *Nature Climate Change* 8, 135–140. <u>https://doi.org/10.1038/s41558-017-0049-x</u>
- Hutter, H.-P., Moshammer, H., Wallner, P., Leitner, B., and Kundi, M. (2007). Heatwaves in Vienna: effects on mortality. Wiener klinische Wochenschrift 119, 223–227. <u>https://doi.org/10.1007/ s00508-006-0742-7</u>

- IFRC (2021). Europe-Wildfires. Information Bulletin 2. Available at: https://go-api.ifrc.org/publicfile/download?path=/docs/Appeals/21/&name=IB\_Europe\_Wildfires13082021.pdf
- Interreg Alpine Space (2021). AlpES: Alpine Ecosystem Services mapping, maintenance and management. Available at: <u>https:// www.alpine-space.eu/project/alpes/</u> (Accessed October 28, 2024).
- Irauschek, F., Rammer, W., and Lexer, M. J. (2017a). Can current management maintain forest landscape multifunctionality in the Eastern Alps in Austria under climate change? *Regional Environmental Change* 17, 33–48. <u>https://doi.org/10.1007/s10113-015-0908-9</u>
- Irauschek, F., Rammer, W., and Lexer, M. J. (2017b). Evaluating multifunctionality and adaptive capacity of mountain forest management alternatives under climate change in the Eastern Alps. *European Journal of Forest Research* 136, 1051–1069. https://doi.org/10.1007/s10342-017-1051-6
- Jabareen, Y. R. (2006). Sustainable Urban Forms: Their Typologies, Models, and Concepts. Journal of Planning Education and Research 26, 38–52. <u>https://doi.org/10.1177/0739456X05285119</u>
- Jacob, D., Teichmann, C., Sobolowski, S., Katragkou, E., Anders, I., Belda, M., et al. (2020). Regional climate downscaling over Europe: perspectives from the EURO-CORDEX community. *Regional Environmental Change* 20, 51. <u>https://doi.org/10.1007/</u> s10113-020-01606-9
- Jakoby, O., Lischke, H., and Wermelinger, B. (2019). Climate change alters elevational phenology patterns of the European spruce bark beetle (*Ips typographus*). *Global Change Biology* 25, 4048–4063. https://doi.org/10.1111/gcb.14766
- Jandl, R. (2023). Renewable energy potentials: Woody biomass., in Energy Transition for a sustainable society, eds. V. J. Bruckman and W. Winiwarter (Österreichische Akademie der Wissenschaften), 37–39. <u>https://doi.org/10.1553/KIOESOP\_013</u>
- Jandl, R., Tappeiner, U., Foldal, C. B., and Erb, K.-H. eds. (2024). APCC Special Report: Landnutzung und Klimawandel in Österreich. Berlin, Heidelberg: Springer Spektrum. <u>https://doi.org/10. 1007/978-3-662-67864-0</u>
- Jiricka-Pürrer, A., Schmied, J., and Pröbstl-Haider, U. (2019). Preferences for renewable energy sources among tourists in the European Alps. CABI, 400–424. <u>https://doi.org/10.1079/</u> 9781786395207.0400
- Job, H., Meyer, C., Coronado, O., Koblar, S., Laner, P., Omizzolo, A., et al. (2022). Open Spaces in the European Alps—GIS-Based Analysis and Implications for Spatial Planning from a Transnational Perspective. *Land* 11, 1605. <u>https://doi.org/10.3390/</u> <u>land11091605</u>
- Junger, L., Hohensinner, S., Schroll, K., Wagner, K., and Seher, W. (2022). Land Use in Flood-Prone Areas and Its Significance for Flood Risk Management—A Case Study of Alpine Regions in Austria. *Land* 11, 392. <u>https://doi.org/10.3390/land11030392</u>
- Juschten, M., Brandenburg, C., Hössinger, R., Liebl, U., Offenzeller, M., Prutsch, A., et al. (2019). Out of the City Heat—Way to Less or More Sustainable Futures? *Sustainability* 11, 214. <u>https://doi.org/10.3390/su11010214</u>
- Juschten, M., and Hössinger, R. (2021). Out of the city but how and where? A mode-destination choice model for urban-rural tourism trips in Austria. *Current Issues in Tourism* 24, 1465– 1481. <u>https://doi.org/10.1080/13683500.2020.1783645</u>
- Kahl, A., Dujardin, J., and Lehning, M. (2019). The bright side of PV production in snow-covered mountains. *Proceedings of* the National Academy of Sciences 116, 1162–1167. <u>https://doi.org/10.1073/pnas.1720808116</u>

- Kaitna, R. (2022). Einfluss des Klimawandels auf Wildbacheinzugsgebiete mit speziellem Fokus auf Muren. Wildbach- und Lawinenverbau 190, 50–56.
- Kaitna, R., Prenner, D., and Hübl, J. (2020). "Muren", in *ExtremA:* Aktueller Wissensstand zu Extremereignissen alpiner Naturgefahren in Österreich, eds. T. Glade, M. Mergili, and K. Sattler (Vienna University Press), 489–510.
- Kaitna, R., Prenner, D., Switanek, M., Maraun, D., Stoffel, M., and Hrachowitz, M. (2023). Changes of hydro-meteorological trigger conditions for debris flows in a future alpine climate. *Science* of The Total Environment 872, 162227. <u>https://doi.org/10.1016/j. scitotenv.2023.162227</u>
- Kakoulaki, G., Kougias, I., Taylor, N., Dolci, F., Moya, J., and Jäger-Waldau, A. (2021). Green hydrogen in Europe – A regional assessment: Substituting existing production with electrolysis powered by renewables. *Energy Conversion and Management* 228, 113649. <u>https://doi.org/10.1016/j.enconman.2020.113649</u>
- Kapos, V., Rhind, J., Edwards, M., Price, M. F., and Ravilious, C. (2000). Developing a map of the world's mountain forests. *IUFRO Research Series*, 4–19. <u>https://doi.org/10.1079/978085</u> 1994468.0004
- Keiler, M., Knight, J., and Harrison, S. (2010). Climate change and geomorphological hazards in the eastern European Alps. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences* 368, 2461–2479. <u>https://doi.org/10.1098/rsta.2010.0047</u>
- Kellerer-Pirklbauer, A. (2019). Long-term monitoring of sporadic permafrost at the eastern margin of the European Alps (Hochreichart, Seckauer Tauern range, Austria). *Permafrost and Peri*glacial Processes 30, 260–277. <u>https://doi.org/10.1002/ppp.2021</u>
- Kerner, J. M., Krauss, J., Maihoff, F., Bofinger, L., and Classen, A. (2023). Alpine butterflies want to fly high: Species and communities shift upwards faster than their host plants. *Ecology* 104, e3848. <u>https://doi.org/10.1002/ecy.3848</u>
- Koboltschnig, G. R., and Schöner, W. (2011). The relevance of glacier melt in the water cycle of the Alps: the example of Austria. *Hydrology and Earth System Sciences* 15, 2039–2048. <u>https://doi.org/10.5194/hess-15-2039-2011</u>
- Koboltschnig, G. R., Schöner, W., Zappa, M., Kroisleitner, C., and Holzmann, H. (2008). Runoff modelling of the glacierized Alpine Upper Salzach basin (Austria): multi-criteria result validation. *Hydrological Processes* 22, 3950–3964. <u>https://doi.org/10. 1002/hyp.7112</u>
- Köll, H., and Bader, M. (2022). Mobilitätserhebung Tirol Frühling 2022. Reith bei Seefeld: PLANOPTIMO. Available at: <u>https:// www.tirol.gv.at/fileadmin/themen/verkehr/verkehrsplanung/</u> <u>Dateien/Bericht\_2022\_-\_final.pdf</u>
- Körner, C. (2007). The use of 'altitude' in ecological research. Trends in Ecology & Evolution 22, 569–574. <u>https://doi.org/10.1016/j.</u> <u>tree.2007.09.006</u>
- Körner, C. (2021). Alpine Plant Life: Functional Plant Ecology of High Mountain Ecosystems. Cham: Springer Nature Switzerland AG. <u>https://doi.org/10.1007/978-3-030-59538-8</u>
- Körner, C., and Hiltbrunner, E. (2021). Why is the Alpine Flora Comparatively Robust against Climatic Warming? *Diversity* 13, 383. <u>https://doi.org/10.3390/d13080383</u>
- Körner, C., Urbach, D., and Paulsen, J. (2021). Mountain definitions and their consequences. *Alpine Botany* 131, 213–217. https://doi.org/10.1007/s00035-021-00265-8

- Kotlarski, S., Gobiet, A., Morin, S., Olefs, M., Rajczak, J., and Samacoïts, R. (2023). 21<sup>st</sup> Century alpine climate change. *Climate Dynamics* 60, 65–86. https://doi.org/10.1007/s00382-022-06303-3
- Kranzl, L., Haas, R., Kalt, G., Müller, A., Nakicenovic, N., Redl, C., et al. (2010). KlimAdapt. Ableitung von prioritären Maßnahmen zur Adaption des Energiesystems an den Klimawandel, Energie-der-Zukunft Report. Energy Economics Group, TU Vienna, BOKU Vienna.
- Kraxner, F., Leduc, S., Hernán Serrano León, Jessica Balest, Garegnani, G., Grilli, G., et al. (2015). *Recommendations and lessons learned for a renewable energy strategy in the Alps.* recharge. green Project. Available at: <u>https://pure.iiasa.ac.at/11618</u>
- Kraxner, F., Zollitsch, W., Kottusch, C., Bruckman, V. J., Glatzel, S., Hood-Nowotny, R., et al. (2024). "Kapitel 5. Mitigation des Klimawandels", in APCC Special Report: Landnutzung und Klimawandel in Österreich (APCC SR Land), eds. R. Jandl, U. Tappeiner, C. B. Foldal, and K.-H. Erb (Berlin, Heidelberg: Springer Spektrum), 275–338. <u>https://doi.org/10.1007/978-3-662-67864-0\_7</u>
- Kulonen, A., Imboden, R. A., Rixen, C., Maier, S. B., and Wipf, S. (2018). Enough space in a warmer world? Microhabitat diversity and small-scale distribution of alpine plants on mountain summits. *Diversity and Distributions* 24, 252–261. <u>https://doi. org/10.1111/ddi.12673</u>
- Laaha, G., and Blöschl, G. (2006). Seasonality indices for regionalizing low flows. *Hydrological Processes* 20, 3851–3878. <u>https:// doi.org/10.1002/hyp.6161</u>
- Laaha, G., Laimighofer, J., Parajka, J., Bertola, M., and Blöschl, G. (2025). Abflusstrends in Österreichs Fließgewässern 1977– 2020. Österr Wasser- und Abfallw. <u>https://doi.org/10.1007/ s00506-024-01106-8</u>
- Laaha, G., Parajka, J., Viglione, A., Koffler, D., Haslinger, K., Schöner, W., et al. (2016). A three-pillar approach to assessing climate impacts on low flows. *Hydrol. Earth Syst. Sci.* 20, 3967– 3985. <u>https://doi.org/10.5194/hess-20-3967-2016</u>
- Lamprecht, A., Semenchuk, P. R., Steinbauer, K., Winkler, M., and Pauli, H. (2018). Climate change leads to accelerated transformation of high-elevation vegetation in the central Alps. *New Phytologist* 220, 447–459. https://doi.org/10.1111/nph.15290
- Land Tirol (2023). Verkehr in Tirol. Bericht 2022. Innsbruck: Amt der Tiroler Landesregierung Abteilung Mobilitätsplanung. Available at: <u>https://www.tirol.gv.at/fileadmin/themen/verkehr/ verkehrsplanung/Dateien/VB\_2022\_Download\_aktualisiert24.</u> <u>pdf</u>
- Lane, S. N., Bakker, M., Gabbud, C., Micheletti, N., and Saugy, J.-N. (2017). Sediment export, transient landscape response and catchment-scale connectivity following rapid climate warming and Alpine glacier recession. *Geomorphology* 277, 210–227. https://doi.org/10.1016/j.geomorph.2016.02.015
- Larigauderie, A., and Körner, C. (1995). Acclimation of Leaf Dark Respiration to Temperature in Alpine and Lowland Plant Species. Annals of Botany 76, 245–252. <u>https://doi.org/10.1006/ anbo.1995.1093</u>
- Ledermann, T., Braun, M., Kindermann, G., Jandl, R., Ludvig, A., Schadauer, K., et al. (2022). Effects of Silvicultural Adaptation Measures on Carbon Stock of Austrian Forests. *Forests* 13, 565. <u>https://doi.org/10.3390/f13040565</u>
- Lehner-Telič, E. (2024). China: Tourismusausblick. Austria Tourism. Available at: <u>https://www.austriatourism.com/tourismusforschung/tourismusausblick/china-tourismusausblick/</u> (Accessed October 28, 2024).

- Lenoir, J., and Svenning, J. C. (2015). Climate-related range shifts – a global multidimensional synthesis and new research directions. *Ecography* 38, 15–28. <u>https://doi.org/10.1111/ecog.00967</u>
- Lexer, M. J., Jandl, R., Nabernegg, S., and Bednar-Friedl, B. (2015). "Forestry", in *Economic Evaluation of Climate Change Impacts*, eds. K. W. Steininger, M. König, B. Bednar-Friedl, L. Kranzl, W. Loibl, and F. Prettenthaler (Cham: Springer), 145–165.
- Löschner, L., Herrnegger, M., Apperl, B., Senoner, T., Seher, W., and Nachtnebel, H. P. (2017). Flood risk, climate change and settlement development: a micro-scale assessment of Austrian municipalities. *Reg Environ Change* 17, 311–322. <u>https://doi. org/10.1007/s10113-016-1009-0</u>
- Lugon, R., and Stoffel, M. (2010). Rock-glacier dynamics and magnitude-frequency relations of debris flows in a high-elevation watershed: Ritigraben, Swiss Alps. *Global and Planetary Change* 73, 202–210. <u>https://doi.org/10.1016/j.gloplacha.2010.06.004</u>
- Lüthi, S., Ban, N., Kotlarski, S., Steger, C. R., Jonas, T., and Schär, C. (2019). Projections of Alpine Snow-Cover in a High-Resolution Climate Simulation. *Atmosphere* 10, 463. <u>https://doi. org/10.3390/atmos10080463</u>
- Magnin, F., Josnin, J.-Y., Ravanel, L., Pergaud, J., Pohl, B., and Deline, P. (2017). Modelling rock wall permafrost degradation in the Mont Blanc massif from the LIA to the end of the 21<sup>st</sup> century. *The Cryosphere* 11, 1813–1834. <u>https://doi.org/10.5194/</u> tc-11-1813-2017
- Mailer, M., Abegg, B., Jänicke, L., and Bursa, B. (2019). Mobilitätsbedingte Klimawirkung einer alpinen Tourismusdestination: CO2-Bilanz und Einschätzung durch Touristen, Bewohner und Beschäftigte. Zeitschrift für Tourismuswissenschaft 11, 211– 236. <u>https://doi.org/10.1515/tw-2019-0013</u>
- Makino, Y., and Rudolf-Miklau, F. (2021). The protective functions of forests in a changing climate – European experience. Forestry Working Paper No. 26. Rome: FAO and Austrian Federal Ministry for Agriculture, Regions and Tourism. <u>https://doi.org/10.4060/cb4464en</u>
- Maran, S., Volonterio, M., and Gaudard, L. (2014). Climate change impacts on hydropower in an alpine catchment. *Environmental Science & Policy* 43, 15–25. <u>https://doi.org/10.1016/j.envsci.2013.12.001</u>
- Maraun, D., and Jury, M. (2022). Der Klimawandel und Naturgefahren im Alpenraum. Wildbach- und Lawinenverbau 190, 38–49.
- Matei, N. A., García-León, D., Dosio, A., Batista e Silva, F., Barranco, R., and Císcar Martínez, J. (2023). *Regional impact of climate change on European tourism demand*. Luxembourg: Publications Office of the European Union. <u>https://doi.org/10.2760/899611</u>
- Matteodo, M., Wipf, S., Stockli, V., Rixen, C., and Vittoz, P. (2013). Elevation gradient of successful plant traits for colonizing alpine summits under climate change. *Environmental Research Letters* 8, 024043. <u>https://doi.org/10.1088/1748-9326/8/2/024043</u>
- Mauri, A., Girardello, M., Strona, G., Beck, P. S. A., Forzieri, G., Caudullo, G., et al. (2022). EU-Trees4F, a dataset on the future distribution of European tree species. *Scientific Data* 9, 37. <u>https://doi.org/10.1038/s41597-022-01128-5</u>
- Mayer, M., and Abegg, B. (2024). Development of summer skiing days in Austrian glacier ski areas in the first two decades of the twenty-first century. *International Journal of Biometeorology* 68, 547–564. https://doi.org/10.1007/s00484-022-02371-6
- Mayer, M., Bichler, B. F., Pikkemaat, B., and Peters, M. (2021). Media discourses about a superspreader destination: How mismanagement of Covid-19 triggers debates about sustainabili-

AAR2

Mayer, M., Matthews, B., Rosinger, C., Sandén, H., Godbold, D. L., and Katzensteiner, K. (2017). Tree regeneration retards decomposition in a temperate mountain soil after forest gap disturbance. Soil Biology and Biochemistry 115, 490–498. <u>https://doi. org/10.1016/j.soilbio.2017.09.010</u>

org/10.1016/j.annals.2021.103278

- Mayer, S., Hendrick, M., Michel, A., Richter, B., Schweizer, J., Wernli, H., et al. (2024a). Impact of climate change on snow avalanche activity in the Swiss Alps. *The Cryosphere* 2024, 5495– 5517. <u>https://doi.org/10.5194/tc-18-5495-2024</u>
- Mayer, M., Sacher, P., Böhme, M., Eckert, M., and Salim, E. (2024b). Gletschertourismus ohne Eis? Die Transformation und Positionierung von alpinen Gletschertourismus-Standorten in Zeiten des Klimawandels. Zeitschrift für Tourismuswissenschaft 16, 15–42. https://doi.org/10.1515/tw-2024-0006
- McDowell, G., Stevens, M., Lesnikowski, A., Huggel, C., Harden, A., DiBella, J., et al. (2021). Closing the Adaptation Gap in Mountains. *Mountain Research and Development* 41, A1–A10. https://doi.org/10.1659/MRD-JOURNAL-D-21-00033.1
- Membretti, A., Tartari, G., Barbera, F., Modica, M., Keeling, S., Tomniuk, V., et al. (2023). Migrazioni climatiche e mobilità interna nella metromontagna padana. Fondazione Cariplo. Rapporto Finale del Progetto MICLIMI.
- Ménégoz, M., Valla, E., Jourdain, N. C., Blanchet, J., Beaumet, J., Wilhelm, B., et al. (2020). Contrasting seasonal changes in total and intense precipitation in the European Alps from 1903 to 2010. *Hydrology and Earth System Sciences* 24, 5355–5377. <u>https://doi.org/10.5194/hess-24-5355-2020</u>
- Mergili, M., Kopf, C., Müllebner, B., and Schneider, J. F. (2012). Changes of the cryosphere and related geohazards in the high-mountain areas of tajikistan and austria: a comparison. *Geografiska Annaler: Series A, Physical Geography* 94, 79–96. <u>https://doi.org/10.1111/j.1468-0459.2011.00450.x</u>
- Messerli, B., Viviroli, D., and Weingartner, R. (2004). Mountains of the world: vulnerable water towers for the 21<sup>st</sup> Century. Ambio Special report No 13, 29–34. <u>https://doi.org/10.5167/uzh-110516</u>
- Messmer, M., Gómez-Navarro, J. J., and Raible, C. C. (2015). Climatology of Vb cyclones, physical mechanisms and their impact on extreme precipitation over Central Europe. *Earth System Dynamics* 6, 541–553. <u>https://doi.org/10.5194/esd-6-541-2015</u>
- Meyer, C., and Job, H. (2022). "Rahmenbedingungen und Herausforderungen des Alpenraums", in *Handbuch Alpine Raumordnung. Ein Raumentwicklungskonzept für den Alpenraum*, ed. CIPRA Österreich im Umweltdachverband (Wien), 16–22.
- Meyer, I., Sinabell, F., Streicher, G., Bohner, A., and Spiegel, H. (2024). "Carbon Sequestration in Austrian Soils: Environmental Significance and Economic Effects of Selected Measures", in *Biodiversity and Climate. Tackling Global Footprints.*, eds. E. Civel, C. de Perthuis, J. E. Milne, M. S. Andersen, and H. Ashiabo (Cheltenham, UK: Edward Elgar Publishing), 74–87.
- Moos, C., Stritih, A., Teich, M., and Bottero, A. (2023). Mountain protective forests under threat? an in-depth review of global change impacts on their protective effect against natural hazards. *Frontiers in Forests and Global Change* 6, 1223934. <u>https:// doi.org/10.3389/ffgc.2023.1223934</u>
- Mostegl, N. M., Pröbstl-Haider, U., Jandl, R., and Haider, W. (2019). Targeting climate change adaptation strategies to small-scale private forest owners. *Forest Policy and Economics* 99, 83–99. https://doi.org/10.1016/j.forpol.2017.10.001

- Müller, M. M., Vilardell, L. V., Vacik, H., Mayer, C., Mayr, S., Carrega, P., et al. (2020). Forest fires in the Alps State of knowledge, future challenges and options for an integrated fire management White Paper for policy makers. <u>https://doi.org/10.13140/RG.2.2.15609.42081</u>
- Nachappa, T. G., Ghorbanzadeh, O., Gholamnia, K., and Blaschke, T. (2020). Multi-hazard Exposure Mapping Using Machine Learning for the State of Salzburg, Austria. *Remote Sensing* 12, 2757. <u>https://doi.org/10.3390/rs12172757</u>
- Nachtnebel, H. P., and Apperl, B. (2015). Beurteilung des Hochwasser-Schadenspotenzials unter dynamischen Bedingungen. Österreichische Wasser- und Abfallwirtschaft 67, 120–130. <u>https://</u> doi.org/10.1007/s00506-015-0220-4
- Nachtnebel, H. P., Dokulil, M., Kuhn, M., Loiskandl, W., Sailer, R., and Schöner, W. (2014). "Der Einfluss des Klimawandels auf die Hydrosphäre", in Österreichischer Sachstandsbericht Klimawandel 2014 (AAR14), ed. Austrian Panel on Climate Change (APCC) (Vienna: Verlag der Österreichischen Akademie der Wissenschaften), 411–465.
- Neger, C., Prettenthaler, F., Gössling, S., and Damm, A. (2021). Reprint of: Carbon intensity of tourism in Austria: Estimates and policy implications. *Journal of Outdoor Recreation and Tourism* 34, 100434. <u>https://doi.org/10.1016/j.jort.2021.100434</u>
- Netherer, S., Matthews, B., Katzensteiner, K., Blackwell, E., Henschke, P., Hietz, P., et al. (2015). Do water-limiting conditions predispose Norway spruce to bark beetle attack? *New Phytologist* 205, 1128–1141. <u>https://doi.org/10.1111/nph.13166</u>
- Nicklas, L., Walde, J., Wipf, S., Lamprecht, A., Mallaun, M., Rixen, C., et al. (2021). Climate Change Affects Vegetation Differently on Siliceous and Calcareous Summits of the European Alps. Frontiers in Ecology and Evolution 9, 642309. <u>https://doi. org/10.3389/fevo.2021.642309</u>
- Nicolussi, K. (2013). "Zur Geschichte des Vernagtferners Gletschervorstöße und Seeausbrüche im vergangenen Jahrtausend", in *Klima, Wetter, Gletscher im Wandel*, eds. E. Koch and B. Erschbamer (Innsbruck: Innsbruck University Press), 69–94.
- NÖN-Redaktion (2023). Skigebiete in Niederösterreich bieten alternative Angebote. NÖN – Niederösterreichische Nachrichten. Available at: <u>https://www.noen.at/freizeit/ausflug-reise/schneemangel-skigebiete-in-niederoesterreich-bieten-alternative-angebote-niederoesterreich-jochen-danninger-oetscherlifte-semmering-skigebiete-hochkar-349258184</u> (Accessed March 25, 2024).
- Oberndorfer, S., Sander, P., and Fuchs, S. (2020). Multi-hazard risk assessment for roads: probabilistic versus deterministic approaches. *Natural Hazards and Earth System Sciences* 20, 3135– 3160. <u>https://doi.org/10.5194/nhess-20-3135-2020</u>
- Observatoire des Sciences de l'Univers de Grenoble (2024). POR-TAL: A research platform to showcase climate initiatives in the Alps. Available at: <u>https://portal.osug.fr/</u> (Accessed October 30, 2024).
- Offenthaler, I., Felderer, A., Formayer, H., Glas, N., Leidinger, D., Leopold, P., et al. (2020). Threshold or Limit? Precipitation Dependency of Austrian Landslides, an Ongoing Challenge for Hazard Mapping under Climate Change. *Sustainability* 12, 6182. <u>https://doi.org/10.3390/su12156182</u>
- Olefs, M., Baumgartner, D. J., Obleitner, F., Bichler, C., Foelsche, U., Pietsch, H., et al. (2016). The Austrian radiation monitoring network ARAD – best practice and added value. *Atmospheric Measurement Techniques* 9, 1513–1531. <u>https://doi.org/10.5194/</u> amt-9-1513-2016

- Olefs, M., Formayer, H., Gobiet, A., Marke, T., Schöner, W., and Revesz, M. (2021). Past and future changes of the Austrian climate – Importance for tourism. *Journal of Outdoor Recreation* and *Tourism* 34, 100395. <u>https://doi.org/10.1016/j.jort.2021.</u> 100395
- Oleson, K. W., Monaghan, A., Wilhelmi, O., Barlage, M., Brunsell, N., Feddema, J., et al. (2015). Interactions between urbanization, heat stress, and climate change. *Climatic Change* 129, 525– 541. <u>https://doi.org/10.1007/s10584-013-0936-8</u>
- ORF (2023). Viele Gäste, hohe Preise, wenig Personal. *News ORF.* at. Available at: <u>https://orf.at/stories/3341414</u> (Accessed October 31, 2024).
- Österreich Werbung ed. (2023). Tourismus in Österreich 2022/2023. Available at: <u>https://www.austriatourism.com/file-admin/user\_upload/Media\_Library/Downloads/Tourismus-forschung/2023G\_Factsheet\_Tourismus\_in\_OE\_2022-2023\_extern\_Stand\_05.07.2023.pdf</u>
- Österreichische Raumordnungskonferenz (2016). Flächenwidmung – Bauland. ÖROK-Atlas. Available at: <u>https://www. oerok-atlas.at/#indicator/72</u> (Accessed October 30, 2024).
- Österreichische Raumordnungskonferenz (2020). Einwohnerdichte. Available at: <u>https://www.oerok-atlas.at/#indicator/91</u> (Accessed October 30, 2024).
- Österreichische Raumordnungskonferenz (2021). 16. Raumordnungsbericht. Analysen und Berichte zur räumlichen Entwicklung Österreichs 2018–2020. Vienna: ÖROK Eigenverlag. Available at: <u>https://www.oerok.gv.at/fileadmin/user\_upload/</u> <u>Bilder/2.Reiter-Raum\_u.\_Region/2.Daten\_und\_Grundlagen/</u> <u>ROB/16\_ROB\_Web.pdf</u> (Accessed November 5, 2024).
- Österreichische Raumordnungskonferenz (2023a). Dauersiedlungsraum. ÖROK-Atlas. Available at: <u>https://www.oerok-atlas.</u> <u>at/#indicator/74</u> (Accessed October 30, 2024).
- Österreichische Raumordnungskonferenz (2023b). Bodenversieglung in Österreich. ÖROK-Atlas. Available at: <u>https://www. oerok-atlas.at/#indicator/61</u> (Accessed October 30, 2024).
- Österreichische Raumordnungskonferenz (2023c). Bevölkerungsentwicklung – 10 Jahre. ÖROK-Atlas. Available at: <u>https://</u> www.oerok-atlas.at/#indicator/12 (Accessed October 30, 2024).
- Österreichische Raumordnungskonferenz (2024). ÖROK-Haushaltsprognose 2022–2051. ÖROK-Atlas. Available at: <u>https://</u> <u>www.oerok-atlas.at/#indicator/79</u> (Accessed October 30, 2024).
- Otto, J.-C., Helfricht, K., Prasicek, G., Binder, D., and Keuschnig, M. (2022). Testing the performance of ice thickness models to estimate the formation of potential future glacial lakes in Austria. *Earth Surface Processes and Landforms* 47, 723–741. <u>https:// doi.org/10.1002/esp.5266</u>
- Owen, G. (2020). What makes climate change adaptation effective? A systematic review of the literature. *Global Environmental Change* 62, 102071. <u>https://doi.org/10.1016/j.gloenvcha.2020.102071</u>
- Ozenda, P., and Borel, J.-L. (2003). "The alpine vegetation of the Alps", in *Alpine biodiversity of Europe*, eds. L. Nagy, G. Grabherr, C. Körner, and D. B. A. Thompson (Berlin: Springer), 12.
- Papathoma-Köhle, M., Hausharter, D., Schlögl, M., and Fuchs, S. (2025). Towards a wildfire vulnerability index using expert judgement. *International Journal of Wildland Fire* 34. Available at: <u>https://doi.org/10.1071/WF24114</u>
- Papathoma-Köhle, M., Schlögl, M., Dosser, L., Roesch, F., Borga, M., Erlicher, M., et al. (2022a). Physical vulnerability to dynamic flooding: Vulnerability curves and vulnerability indices. *Journal of Hydrology* 607, 127501. <u>https://doi.org/10.1016/j.jhydrol.2022.127501</u>

- Papathoma-Köhle, M., Schlögl, M., and Fuchs, S. (2019). Vulnerability indicators for natural hazards: an innovative selection and weighting approach. *Scientific Reports* 9, 15026. <u>https://doi. org/10.1038/s41598-019-50257-2</u>
- Papathoma-Köhle, M., Schlögl, M., Garlichs, C., Diakakis, M., Mavroulis, S., and Fuchs, S. (2022b). A wildfire vulnerability index for buildings. *Scientific Reports* 12, 6378. <u>https://doi.org/ 10.1038/s41598-022-10479-3</u>
- Papathoma-Köhle, M., Thaler, T., and Fuchs, S. (2021). An institutional approach to vulnerability: evidence from natural hazard management in Europe. *Environmental Research Letters* 16, 044056. <u>https://doi.org/10.1088/1748-9326/abe88c</u>
- Pappenberger, F., Cloke, H. L., Parker, D. J., Wetterhall, F., Richardson, D. S., and Thielen, J. (2015). The monetary benefit of early flood warnings in Europe. *Environmental Science & Policy* 51, 278–291. <u>https://doi.org/10.1016/j.envsci.2015.04.016</u>
- Pauli, H., Gottfried, M., Dullinger, S., Abdaladze, O., Akhalkatsi, M., Benito-Alonso, J. L., et al. (2012). Recent Plant Diversity Changes on Europe's Mountain Summits. *Science* 336, 353–355. <u>https://doi.org/10.1126/science.1219033</u>
- Peleg, N., Ban, N., Gibson, M. J., Chen, A. S., Paschalis, A., Burlando, P., et al. (2022). Mapping storm spatial profiles for flood impact assessments. *Advances in Water Resources* 166, 104258. <u>https://doi.org/10.1016/j.advwatres.2022.104258</u>
- Pereira, H. M., and Navarro, L. M. eds. (2015). Rewilding European landscapes. Cham: Springer Cham. Available at: <u>https://link.springer.com/content/pdf/10.1007/978-3-319-12039-3.pdf</u>
- Permanent Secretariat of the Alpine Convention ed. (2009). Water and water management issues. Report on the State of the Alps. Available at: <u>https://www.alpconv.org/fileadmin/user\_upload/</u> <u>Publications/RSA/RSA2\_long\_EN.pdf</u> (Accessed November 5, 2024).
- Permanent Secretariat of the Alpine Convention ed. (2019). Climate-neutral and climate-resilient Alps 2050. Declaration of Innsbruck Alpine Climate Target System 2050 7<sup>th</sup> report on the state of the Alps "Natural Hazard Risk Governance." Alpine Convention. Available at: <u>https://www.alpconv.org/fileadmin/user\_upload/Publications/Climate2050\_EN.pdf</u>
- Perzl, F. (2008). Ein Minimalstandard für die Dokumentation der Schutzwirkungen des Waldes im Rahmen der Österreichischen "Initiative Schutz durch Wald", (Dornbirn, Austria), 551–562.
- Perzl, F. (2014). "Der Objektschutzwald. Bedeutung und Herausforderung", in *BFW.Praxisinformation*. *Naturgefahren und Schutzwald*, (Wien), 20–24.
- Perzl, F., Bono, A., Garbarino, M., and Motta, R. (2021). "Protective Effects of Forests against Gravitational Natural Hazards", in *Protective Forests as Ecosystem-based Solution for Disaster Risk Reduction (Eco-DRR)*, eds. M. Teich, C. Accastello, F. Perzl, and Kleemayr,K. (Rijeka: IntechOpen). <u>https://doi.org/10.5772/</u> intechopen.99506
- Peters, K., Langston, L., Tanner, T., and Bahadur, A. (2016). "Resilience" across the post-2015 frameworks: towards coherence? *Overseas Development Institute*. Available at: <u>https://odi.org/en/ publications/resilience-across-the-post-2015-frameworks-towards-coherence/</u>
- Pfurtscheller, C. (2014). Regional economic impacts of natural hazards – the case of the 2005 Alpine flood event in Tyrol (Austria). *Natural Hazards and Earth System Sciences* 14, 359–378. <u>https:// doi.org/10.5194/nhess-14-359-2014</u>
- Pichelli, E., Coppola, E., Sobolowski, S., Ban, N., Giorgi, F., Stocchi, P., et al. (2021). The first multi-model ensemble of regional cli-

mate simulations at kilometer-scale resolution part 2: historical and future simulations of precipitation. *Climate Dynamics* 56, 3581–3602. <u>https://doi.org/10.1007/s00382-021-05657-4</u>

- Pielmeier, C., Techel, F., Marty, C., and Stucki, T. (2013). "Wet snow avalanche activity in the Swiss Alps – trend analysis for mid-winter season", in *Proceedings of ISSW International Snow Science Workshop*, eds. F. Naaim-Bouvet, Y. Durand, and R. Lambert (Genoble – Chamonix Mont-Blanc, France: ISSW), 1240–1246.
- Pikkemaat, B., and Zehrer, A. (2016). Innovation and service experiences in small tourism family firms. *International Journal of Culture, Tourism and Hospitality Research* 10, 343–360. <u>https://</u> doi.org/10.1108/IJCTHR-06-2016-0064
- Pistotnik, G., Hofstätter, M., and Lexer, A. (2020). "Starkniederschlag und Hagel", in *ExtremA 2019. Aktueller Wissensstand zu Extremereignissen alpiner Naturgefahren in Österreich*, eds. T. Glade, M. Mergili, and K. Sattler (Göttingen: Vienna University Press), 141–172.
- Pittore, M., Campalani, P., Renner, K., Plörer, M., and Tagliavini, F. (2023). Border-independent multi-functional, multi-hazard exposure modelling in Alpine regions. *Natural Hazards* 119, 837–858. <u>https://doi.org/10.1007/s11069-023-06134-3</u>
- Pluess, A., Brang, P., and Augustin, S. eds. (2016). Wald im Klimawandel: Grundlage f
  ür Adaptationsstrategien. Bern, Stuttgart, Wien: Haupt.
- Polderman, A., Haller, A., Viesi, D., Tabin, X., Sala, S., Giorgi, A., et al. (2020). How Can Ski Resorts Get Smart? Transdisciplinary Approaches to Sustainable Winter Tourism in the European Alps. Sustainability 12, 5593. <u>https://doi.org/10.3390/su</u> 12145593
- Pramreiter, M., Nenning, T., Malzl, L., and Konnerth, J. (2023). A plea for the efficient use of wood in construction. *Nature Re*views Materials 8, 217–218. <u>https://doi.org/10.1038/s41578-023-00534-4</u>
- Prenner, D., Hrachowitz, M., and Kaitna, R. (2019). Trigger characteristics of torrential flows from high to low alpine regions in Austria. *Science of the Total Environment* 658, 958–972. <u>https:// doi.org/10.1016/j.scitotenv.2018.12.206</u>
- Prettenthaler, F., Töglhofer, C., Habsburg-Lothringen, C., and Türk, A. (2008). "Klimabedingte Änderung des Heiz- und Kühlenergiebedarfs für Österreich", in *Studien zum Klimawandel in Österreich – Band 2: Heizen und Kühlen im Klimawandel – Teil 1*, eds. F. Prettenthaler and A. Gobiet (Vienna: Verlag der Österreichischen Akademie der Wissenschaften (ÖAW)), 29–49.
- Pretzsch, H., Forrester, D. I., and Bauhus, J. eds. (2017). Mixed-Species Forests: Ecology and Management. Berlin, Heidelberg: Springer. <u>https://doi.org/10.1007/978-3-662-54553-9</u>
- Price, M. F., Gurgiser, W., Juen, I., Adler, C., Wymann von Dach, S., Kaser, G., et al. (2022). The International Mountain Conference, Innsbruck, Austria, September 2019 (IMC2019): A Synthesis with recommendations for research. *Mountain Research and Development* 42, A1–A16. <u>https://doi.org/10.1659/ MRD-JOURNAL-D-21-00027.1</u>
- Pröbstl-Haider, U., Dabrowska, K., and Haider, W. (2016). Risk perception and preferences of mountain tourists in light of glacial retreat and permafrost degradation in the Austrian Alps. *Journal of Outdoor Recreation and Tourism* 13, 66–78. <u>https:// doi.org/10.1016/j.jort.2016.02.002</u>
- Pröbstl-Haider, U., Hödl, C., Ginner, K., and Borgwardt, F. (2021). Climate change: Impacts on outdoor activities in the summer

and shoulder seasons. Journal of Outdoor Recreation and Tourism 34, 100344. https://doi.org/10.1016/j.jort.2020.100344

- Pröbstl-Haider, U., Lund-Durlacher, D., Antonschmidt, H., and Hödl, C. (2018). Mountain bike tourism in Austria and the Alpine Region – towards a sustainable model for multi-stakeholder product development. *Journal of Sustainable Tourism* 26, 567–582. https://doi.org/10.1080/09669582.2017.1361428
- Pröbstl-Haider, U., Mostegl, N. M., and Haider, W. (2020a). Smallscale private forest ownership: Understanding female and male forest owners' climate change adaptation behaviour. *Forest Policy and Economics* 112, 102111. <u>https://doi.org/10.1016/j.forpol.2020.102111</u>
- Pröbstl-Haider, U., Lund-Durlacher, D., Olefs, M., and Prettenthaler, F. eds. (2020b). *Tourismus und Klimawandel*. Berlin, Heidelberg: Springer Spektrum Redl, M. (2024). *Die Zukunft der Skigebiete: Das weiße Gold wird grün!*. Wien: story.one Publishing.
- Reimoser, F. (2003). Steering the impacts of ungulates on temperate forests. *Journal of Nature Conservation* 10, 243–252. <u>https://doi.org/10.1078/1617-1381-00024</u>
- Reuter, B., Mitterer, C., and Bellaire, S. (2020). "WARNSIGNAL KLIMA: Die Lawinengefahr im Klimawandel", in *Warnsignal Klima: Hochgebirge im Wandel*, eds. J. L. Lozán, S.-W. Breckle, H. Escher-Vetter, H. Grassl, D. Kasang, F. Paul, et al. (Hamburg: Verlag Wissenschaftliche Auswertungen in Kooperation mit GEO Magazin-Hamburg), 316–323.
- Richter, E. (1889). Der Gletscherausbruch im Martellthal und seine Wiederkehr. *Mittheilungen des Deutschen und Oesterreichischen Alpenvereins* 1889, 231–233.
- Richter, E. (1892). Urkunden über die Ausbrüche des Vernagt- und Gurglergletschers im 17. und 18. Jahrhundert. Forschungen zur deutschen Landes- und Volkskunde 6, 345–440.
- Rode, M., Sass, O., Kellerer-Pirklbauer, A., Schnepfleitner, H., and Gitschthaler, C. (2020). Permafrost distribution and conditions at the headwalls of two receding glaciers (Schladming and Hallstatt glaciers) in the Dachstein Massif, Northern Calcareous Alps, Austria. *The Cryosphere* 14, 1173–1186. <u>https://doi.org/10.5194/tc-14-1173-2020</u>
- Romeo, R., Russo, L., Parisi, F., Notarianni, M., Manuelli, S., Carvao, S., et al. (2021). Mountain tourism Towards a more sustainable path. Rome: FAO. <u>https://doi.org/10.4060/cb7884en</u>
- Roncucci, R. (2020). Artificial snow may help the Alps now, but it comes at a cost to the future. *Global Center on Adaptation*. Available at: <u>https://gca.org/artificial-snow-may-help-the-alpsnow-but-it-comes-at-a-cost-to-the-future/</u> (Accessed October 30, 2024).
- Rosbakh, S., Leingartner, A., Hoiss, B., Krauss, J., Steffan-Dewenter, I., and Poschlod, P. (2017). Contrasting Effects of Extreme Drought and Snowmelt Patterns on Mountain Plants along an Elevation Gradient. *Frontiers in Plant Science* 8, 1478. <u>https:// doi.org/10.3389/fpls.2017.01478</u>
- Rotach, M. W., Serafin, S., Ward, H. C., Arpagaus, M., Colfescu, I., Cuxart, J., et al. (2022). A Collaborative Effort to Better Understand, Measure, and Model Atmospheric Exchange Processes over Mountains. *Bulletin of the American Meteorological Society* 103, E1282–E1295. https://doi.org/10.1175/BAMS-D-21-0232.1
- Rounce, D. R., Hock, R., Maussion, F., Hugonnet, R., Kochtitzky, W., Huss, M., et al. (2023). Global glacier change in the 21<sup>st</sup> century: Every increase in temperature matters. *Science* 379, 78–83. <u>https://doi.org/10.1126/science.abo1324</u>
- Rumpf, S. B., Gravey, M., Bronnimann, O., Luoto, M., Cianfrani, C., Mariethoz, G., et al. (2022). From white to green: Snow cov-

er loss and increased vegetation productivity in the European Alps. *Science* 376, 1119–1122. <u>https://doi.org/10.1126/science.abn6697</u>

- Rumpf, S. B., Hülber, K., Klonner, G., Moser, D., Schuetz, M., Wessely, J., et al. (2018). Range dynamics of mountain plants decrease with elevation. *Proceedings of the National Academy of Sciences of the USA* 115, 1848–1853. <u>https://doi.org/10.1073/pnas.1713936115</u>
- Rumpf, S. B., Hülber, K., Zimmermann, N. E., and Dullinger, S. (2019). Elevational rear edges shifted at least as much as leading edges over the last century. *Global Ecology and Biogeography* 28, 533–543. https://doi.org/10.1111/geb.12865
- Salim, E., Ravanel, L., Bourdeau, P., and Deline, P. (2021). Glacier tourism and climate change: effects, adaptations, and perspectives in the Alps. *Regional Environmental Change* 21, 120. <u>https://doi.org/10.1007/s10113-021-01849-0</u>
- Salto (2024). Die Nationalität der Alpen. Salto Environment. Available at: <u>https://salto.bz/en/article/13082021/die-nationali-taet-der-alpen</u> (Accessed November 5, 2024).
- Sass, O., Haas, F., Schimmer, C., Heel, M., Bremer, M., Stöger, F., et al. (2012). Impact of forest fires on geomorphic processes in the tyrolean limestone alps. *Geografiska Annaler: Series A*, *Physical Geography* 94, 117–133. <u>https://doi.org/10.1111/j.1468-0459.2012.00452.x</u>
- Sass, O., and Oberlechner, M. (2012). Is climate change causing increased rockfall frequency in Austria? *Natural Hazards and Earth System Sciences* 12, 3209–3216. <u>https://doi.org/10.5194/</u> <u>nhess-12-3209-2012</u>
- Sattler, K., Keiler, M., Zischg, A., and Schrott, L. (2011). On the Connection between Debris Flow Activity and Permafrost Degradation: A Case Study from the Schnalstal, South Tyrolean Alps, Italy. *Permafrost and Periglacial Processes* 22, 254–265. https://doi.org/10.1002/ppp.730
- Scherrer, S. C., and Begert, M. (2019). Effects of large-scale atmospheric flow and sunshine duration on the evolution of minimum and maximum temperature in Switzerland. *Theoretical* and Applied Climatology 138, 227–235. <u>https://doi.org/10.1007/</u> s00704-019-02823-x
- Scherrer, S. C., Gubler, S., Wehrli, K., Fischer, A. M., and Kotlarski, S. (2021). The Swiss Alpine zero degree line: Methods, past evolution and sensitivities. *International Journal of Climatology* 41, 6785–6804. https://doi.org/10.1002/joc.7228
- Schindelegger, A., Steinbrunner, B., and Ertl, M. (2022). Climate-Resilient Spatial Planning in the Alps. EUSALP Action Group 8. Available at: <u>https://risknat.org/wp-content/uploads/ 2022/08/clisp-alp\_report\_final.pdf</u>
- Schinko, T., Köberl, J., Prettenthaler, F., Bednar-Friedl, B., Töglhofer, C., Heinrich, G., et al. (2013). Climate Change and the Austrian Tourism Sector: Impacts, Adaptation and Macroeconomic Spillover Effects. *EconPapers*. Available at: <u>https://Econ-Papers.repec.org/RePEc:ekd:004912:5601</u> (Accessed November 5, 2024).
- Schlögl, M., Fuchs, S., Scheidl, C., and Heiser, M. (2021). Trends in torrential flooding in the Austrian Alps: A combination of climate change, exposure dynamics, and mitigation measures. *Climate Risk Management* 32, 100294. <u>https://doi.org/10.1016/j. crm.2021.100294</u>
- Schneider-Muntau, B. (2020). Modelling of the interaction between structures and creeping slopes. *Geomechanics and Tunnelling* 13, 612–619. <u>https://doi.org/10.1002/geot.202000032</u>

- Schneider-Muntau, B., Dai, X., and Fellin, W. (2022). Sensitivity analyses of the different influencing factors on numerical investigations of landslides. *Geomechanics and Tunnelling* 15, 582–595. <u>https://doi.org/10.1002/geot.202200014</u>
- Schoeneich, P., Bodin, X., Echelard, T., Kaufmann, V., Kellerer-Pirklbauer, A., Krysiecki, J.-M., et al. (2015). "Velocity Changes of Rock Glaciers and Induced Hazards", in *Engineering Geology for Society and Territory – Volume 1*, eds. G. Lollino, A. Manconi, J. Clague, W. Shan, and M. Chiarle (Cham: Springer), 223–227.
- Schöner, W., Boeckli, L., Hausmann, H., Otto, J. C., Reisenhofer, S., Riedl, C., et al. (2012). Spatial patterns of permafrost at Hoher Sonnblick (Austrian Alps) – Extensive field-measurements and modelling approaches. *Austrian Journal of Earth Sciences* 105, 154–168. http://hdl.handle.net/20.500.12708/164874
- Schöner, W., and Haslinger, K. (2020). "Extremtemperaturen und Trockenheit", in ExtremA 2019. Aktueller Wissensstand zu Extremereignissen alpiner Naturgefahren in Österreich, eds. T. Glade, M. Mergili, and K. Sattler (Vienna University Press), 78–97.
- Schöner, W., Koch, R., Matulla, C., Marty, C., and Tilg, A.-M. (2019). Spatiotemporal patterns of snow depth within the Swiss-Austrian Alps for the past half century (1961 to 2012) and linkages to climate change. *International Journal of Climatology* 39, 1589–1603. https://doi.org/10.1002/joc.5902
- Schöniger, F., Resch, G., Suna, D., Formayer, H., Pardo-Garcia, N., Hasengst, F., et al. (2023). Securing Austria's Electricity Supply in times of Climate Change; Final report of the ACRP12 project SECURES (2020–2023). Available at: <u>https://www.secures.</u> <u>at/fileadmin/cmc/Final\_Report\_SECURES.pdf</u> (Accessed May 6, 2024).
- Schrott, L., Otto, J. C., and Keller, F. (2012). Modelling alpine permafrost distribution in the Hohe Tauern region, Austria. Austrian Journal of Earth Sciences 105, 169–183. Available at: https:// api.semanticscholar.org/CorpusID:218478088
- Schwaiger, K., and Zehrer, A. (2022). The relationship between employer image and employee commitment in family-run hospitality firms. *International Journal of Culture, Tourism* and Hospitality Research 16, 352–369. <u>https://doi.org/10.1108/</u> IJCTHR-10-2020-0238
- Scott, D., Knowles, N., and Steiger, R. (2022). Is snowmaking climate change maladaptation? *Journal of Sustainable Tourism*, 282–303. <u>https://doi.org/10.1080/09669582.2022.2137729</u>
- Seidl, R., Thom, D., Kautz, M., Martin-Benito, D., Peltoniemi, M., Vacchiano, G., et al. (2017). Forest disturbances under climate change. *Nature Climate Change* 7, 395–402. <u>https://doi. org/10.1038/nclimate3303</u>
- Semenchuk, P., Moser, D., Essl, F., Schindler, S., Wessely, J., Gattringer, A., et al. (2021). Future Representation of Species' Climatic Niches in Protected Areas: A Case Study With Austrian Endemics. *Frontiers in Ecology and Evolution* 9, 685753. <u>https:// doi.org/10.3389/fevo.2021.685753</u>
- Senf, C., and Seidl, R. (2020). Mapping the forest disturbance regimes of Europe. *Nature Sustainability* 4, 63–70. <u>https://doi.org/10.1038/s41893-020-00609-y</u>
- Smiatek, G., Kunstmann, H., and Senatore, A. (2016). EURO-COR-DEX regional climate model analysis for the Greater Alpine Region: Performance and expected future change. *Journal of Geophysical Research: Atmospheres* 121, 7710–7728. <u>https://doi.org/10.1002/2015JD024727</u>
- Smyčka, J., Roquet, C., Renaud, J., Thuiller, W., Zimmermann, N. E., and Lavergne, S. (2017). Disentangling drivers of plant ende-

mism and diversification in the European Alps – a phylogenetic and spatially explicit approach. *Perspectives in Plant Ecology Evolution and Systematics* 28, 19–27. <u>https://doi.org/10.1016/j.</u> ppees.2017.06.004

- Snow Space Salzburg (2024). Anreise nach Flachau, Wagrain & St. Johann. Available at: <u>https://www.snow-space.com/anreise</u> (Accessed October 28, 2024).
- Soares, P. M. M., Careto, J. A. M., Cardoso, R. M., Goergen, K., Katragkou, E., Sobolowski, S., et al. (2024). The added value of km-scale simulations to describe temperature over complex orography: the CORDEX FPS-Convection multi-model ensemble runs over the Alps. *Climate Dynamics* 62, 4491–4514. https://doi.org/10.1007/s00382-022-06593-7
- Sommer, C., Malz, P., Seehaus, T. C., Lippl, S., Zemp, M., and Braun, M. H. (2020). Rapid glacier retreat and downwasting throughout the European Alps in the early 21<sup>st</sup> century. *Nature Communications* 11, 3209. <u>https://doi.org/10.1038/s41467-020-16818-0</u>
- Soomro, Seh., Soomro, A. R., Batool, S., Guo, J., Li, Y., Bai, Y., et al. (2024). How does the climate change effect on hydropower potential, freshwater fisheries, and hydrological response of snow on water availability? *Applied Water Science* 14, 65. <u>https://doi. org/10.1007/s13201-023-02070-6</u>
- Stahel, A., Ciari, F., and Axhausen, K. W. (2013). Modeling impacts of weather conditions in agent-based transport microsimulations., (Washington DC, USA: Transportation Reserach Board), 14–2668. <u>https://doi.org/10.3929/ethz-b-000070384</u>
- Stahl, K., Weiler, M., van Tiel, M., Kohn, I., Hänsler, A., Freudiger, D., et al. (2022). Impact of climate change on the rain, snow and glacier melt components of streamflow of the river Rhine and its tributaries. Lelystad: International Commission for the Hydrology of the Rhine basin (CHR). Available at: <u>https://www. chr-khr.org/sites/default/files/chrpublications/ASG-II\_Synthese\_EN\_mit-Links\_korr-Tab-2.pdf</u>
- Statistik Austria (2019a). Tourismus in Österreich Ergebnisse der Beherbergungsstatistik. Vienna. Available at: <u>https://www. statistik.at/fileadmin/publications/Tourismus\_in\_OEsterreich\_</u> 2018.pdf (Accessed April 10, 2023).
- Statistik Austria (2019b). Urlaubs- und Geschäftsreisen Kalenderjahr 2018. Wien, Österreich. Available at: <u>https://www.statistik.</u> <u>at/statistiken/tourismus-und-verkehr/tourismus/reiseverhalten</u> (Accessed April 10, 2023).
- Statistik Austria (2021). Regionalwirtschaftliche Bedeutung des Tourismus in acht Bundesländer 2018. Available at: <u>https:// www.statistik.at/statistiken/tourismus-und-verkehr/tourismus/ tourismus-satellitenkonto/regionale-tourismus-satellitenkonten</u> (Accessed April 20, 2024).
- Statistik Austria (2023a). Immobilien-Durchschnittspreise. Available at: <u>https://www.statistik.at/statistiken/volkswirtschaft-</u> <u>und-oeffentliche-finanzen/preise-und-preisindizes/immobil-</u> <u>ien-durchschnittspreise</u> (Accessed November 5, 2024).
- Statistik Austria (2023b). Tourismus in Österreich 2022 Ergebnisse der Beherbergungsstatistik. Vienna. Available at: <u>https://www. statistik.at/statistiken/tourismus-und-verkehr/tourismus/beherbergung/ankuenfte-naechtigungen</u> (Accessed June 1, 2024).
- Staude, I. R., Pereira, H. M., Daskalova, G. N., Bernhardt-Römermann, M., Diekmann, M., Pauli, H., et al. (2022). Directional turnover towards larger-ranged plants over time and across habitats. *Ecology Letters* 25, 466–482. <u>https://doi.org/10.1111/ ele.13937</u>
- Steiger, R., Damm, A., Prettenthaler, F., and Pröbstl-Haider, U. (2021a). Climate change and winter outdoor activities in Aus-

tria. Journal of Outdoor Recreation and Tourism 34, 100330. https://doi.org/10.1016/j.jort.2020.100330

- Steiger, R., Knowles, N., Pöll, K., and Rutty, M. (2022). Impacts of climate change on mountain tourism: a review. *Journal of Sustainable Tourism* 32, 1984–2017. <u>https://doi.org/10.1080/09669</u> 582.2022.2112204
- Steiger, R., and Mayer, M. (2008). Snowmaking and Climate Change: Future Options for Snow Production in Tyrolean Ski Resorts. *Mountain Research and Development* 28, 292–298. <u>https://doi.org/10.1659/mrd.0978</u>
- Steiger, R., Posch, E., Tappeiner, G., and Walde, J. (2020). The impact of climate change on demand of ski tourism a simulation study based on stated preferences. *Ecological Economics* 170, 106589. https://doi.org/10.1016/j.ecolecon.2019.106589
- Steiger, R., Pröbstl-Haider, U., Prettenthaler, F., Damm, A., Falk, M., and Neger, C. (2021b). "Outdooraktivitäten und damit zusammenhängende Einrichtungen im Winter", in *Tourismus und Klimawandel*, eds. U. Pröbstl-Haider, D. Lund-Durlacher, M. Olefs, and F. Prettenthaler (Berlin, Heidelberg: Springer Spektrum), 109–122. https://doi.org/10.1007/978-3-662-61522-5\_6
- Steiger, R., and Scott, D. (2020). Ski tourism in a warmer world: Increased adaptation and regional economic impacts in Austria. *Tourism Management* 77, 104032. <u>https://doi.org/10.1016/j. tourman.2019.104032</u>
- Steinbauer, K., Lamprecht, A., Semenchuk, P., Winkler, M., and Pauli, H. (2020). Dieback and expansions: species-specific responses during 20 years of amplified warming in the high Alps. *Alpine Botany* 130, 1–11. <u>https://doi.org/10.1007/s00035-019-00230-6</u>
- Steinbauer, M. J., Grytnes, J.-A., Jurasinski, G., Kulonen, A., Lenoir, J., Pauli, H., et al. (2018). Accelerated increase in plant species richness on mountain summits is linked to warming. *Nature* 556, 231–234. <u>https://doi.org/10.1038/s41586-018-0005-6</u>
- Steyrer, G., Hinterstoisser, W., Perny, B., Schwanda, K., Tatzber, M., and Hoch, G. (2023). Waldschutzsituation 2022. Forstzeitung 4. Available at: <u>https://www.forstzeitung.at/markt/2023/04/wald-schutzsituation-2022.html</u> (Accessed January 21, 2025).
- Stiegler, C., Rode, M., Sass, O., and Otto, J. C. (2014). An Undercooled Scree Slope Detected by Geophysical Investigations in Sporadic Permafrost below 1000 M ASL, Central Austria. *Permafrost and Periglacial Processes* 25, 194–207. <u>https://doi.org/ 10.1002/ppp.1813</u>
- Stoffel, M., Bollschweiler, M., and Beniston, M. (2011). Rainfall characteristics for periglacial debris flows in the Swiss Alps: past incidences – potential future evolutions. *Climatic Change* 105, 263–280. <u>https://doi.org/10.1007/s10584-011-0036-6</u>
- Stoffel, M., and Corona, C. (2018). Future winters glimpsed in the Alps. Nature Geoscience 11, 458–460. <u>https://doi.org/10.1038/ s41561-018-0177-6</u>
- Stoffel, M., Mendlik, T., Schneuwly-Bollschweiler, M., and Gobiet, A. (2014). Possible impacts of climate change on debris flow activity in the Swiss Alps. *Climatic Change* 122, 141–155. <u>https:// doi.org/10.1007/s10584-013-0993-z</u>
- Stöglehner, G., and Abart-Heriszt, L. (2022). Integrated spatial and energy planning in Styria – A role model for local and regional energy transition and climate protection policies. *Renewable and Sustainable Energy Reviews* 165, 112587. <u>https://doi.org/10.1016/j.rser.2022.112587</u>
- Stöglehner, G., Neugebauer, G., Erker, S., and Narodoslawsky, M. eds. (2016a). Integrated Spatial and Energy Planning. Supporting Climate Protection and the Energy Turn with Means of Spatial

Stöglehner, G., Narodoslawsky, M., Erker, S., and Neugebauer, G. (2016b). "System Interrelations Between Spatial Structures, Energy Demand, and Energy Supply", in *Integrated Spatial and Energy Planning. Supporting Climate Protection and the Energy Turn with Means of Spatial Planning*, eds. G. Stöglehner, M. Narodoslawsky, S. Erker, and G. Neugebauer (Cham: Springer), 11–34.

31870-7

- Strapazzon, G., Schweizer, J., Chiambretti, I., Brodmann Maeder, M., Brugger, H., and Zafren, K. (2021). Effects of Climate Change on Avalanche Accidents and Survival. *Frontiers in Physiology* 12, 639433. https://doi.org/10.3389/fphys.2021.639433
- Stumvoll, M. J., Canli, E., Engels, A., Thiebes, B., Groiss, B., Glade, T., et al. (2020). The "Salcher" landslide observatory – experimental long-term monitoring in the Flysch Zone of Lower Austria. Bulletin of Engineering Geology and the Environment 79, 1831–1848. https://doi.org/10.1007/s10064-019-01632-w
- Teich, M., Accastello, C., Perzl, F., and Berger, F. (2022). "Protective Forests for Ecosystem-based Disaster Risk Reduction (Eco-DRR) in the Alpine Space", in *Protective Forests as Ecosystem-based Solution for Disaster Risk Reduction (Eco-DRR)*, eds. M. Teich, C. Accastello, F. Perzl, and K. Kleemayr (Rijeka: IntechOpen). <u>https://doi.org/10.5772/intechopen.99505</u>
- Teich, M., Giunta, A. D., Hagenmuller, P., Bebi, P., Schneebeli, M., and Jenkins, M. J. (2019). Effects of bark beetle attacks on forest snowpack and avalanche formation – implications for protection forest management. *Forest Ecology and Management* 438, 186–203. https://doi.org/10.1016/j.foreco.2019.01.052
- Temperli, C., Stadelmann, G., Thürig, E., and Brang, P. (2017). Silvicultural strategies for increased timber harvesting in a Central European mountain landscape. *European Journal of Forest Re*search 136, 493–509. <u>https://doi.org/10.1007/s10342-017-1048-1</u>
- Terzi, S., Torresan, S., Schneiderbauer, S., Critto, A., Zebisch, M., and Marcomini, A. (2019). Multi-risk assessment in mountain regions: A review of modelling approaches for climate change adaptation. *Journal of Environmental Management* 232, 759– 771. <u>https://doi.org/10.1016/j.jenvman.2018.11.100</u>
- Thurm, B., Vielle, M., and Vöhringer, F. (2017). Impacts of climate change for Swiss winter and summer tourism: a general equilibrium analysis., in *Proceedings of the EAERE*, (Athens, Greece 18 June-1 July 2017), 1–31. Available at: <u>https://infoscience.epfl.ch/server/api/core/bitstreams/1548a1d3-f7bc-4315-9d42-a8a5014 cc61f/content</u>
- Tiefenbacher, A., Sandén, T., Haslmayr, H.-P., Miloczki, J., Wenzel, W., and Spiegel, H. (2021). Optimizing Carbon Sequestration in Croplands: A Synthesis. *Agronomy* 11, 882. <u>https://doi. org/10.3390/agronomy11050882</u>
- Tomschy, R., Herry, M., Sammer, G., Klementschitz, R., Riegler, S., Follmer, G., et al. (2016). Österreich unterwegs 2013/2014. Ergebnisbericht zur österreichweiten Mobilitätserhebung "Österreich unterwegs 2013/2014". Vienna: Bundesministerium für Verkehr, Innovation und Technologie. Available at: <u>https://www.bmk.gv.at/dam/jcr:fbe20298-a4cf-46d9-bbee-01ad771a7fda/oeu\_2013-2014\_Ergebnisbericht.pdf</u> (Accessed February 6, 2025).
- Tudoroiu, M., Eccel, E., Gioli, B., Gianelle, D., Schume, H., Genesio, L., et al. (2016). Negative elevation-dependent warming trend in the Eastern Alps. *Environmental Research Letters* 11, 044021. https://doi.org/10.1088/1748-9326/11/4/044021

- Umwelt Journal (2023). Erste wasserstoffbetriebene Pistenraupe in Flachau. Umwelt Journal Fachmaganzin für Umwelttechnik, Energie, Abfallwirtschaft, Klima, Mobilität und Green Finance. Available at: <u>https://www.umwelt-journal.at/erste-wasserstoffbetriebene-pistenraupe-in-flachau/</u>
- Umweltbundesamt (2023a). Flächeninanspruchnahme in Österreich 2020. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/themen/boden/flaecheninanspruchnahme\_2020.</u> <u>pdf</u> (Accessed November 5, 2024).
- Umweltbundesamt (2023b). Austria's Annual Greenhouse Gas Inventory 1990–2021: Submission under Regulation (EU) no 2018/1999. Vienna: Umweltbundesamt. Available at: <u>https:// www.umweltbundesamt.at/fileadmin/site/publikationen/rep</u> 0841.pdf
- Umweltbundesamt (2024). Austria's National Inventory Report 2024. Vienna: Umweltbundesamt. Available at: <u>https://www. umweltbundesamt.at/studien-reports/publikationsdetail?pub\_</u> id=2527&cHash=4f93d43917ca9a3ce6e2ba4c1191ca1a
- UNDRR (2015). Sendai Framework for Disaster Risk Reduction 2015–2030. Geneva, Switzerland. Available at: <u>https://www. undrr.org/publication/sendai-framework-disaster-risk-reduction-2015-2030</u> (Accessed January 22, 2025).
- Unger, R., Abegg, B., Mailer, M., and Stampfl, P. (2016). Energy consumption and greenhouse gas emissions resulting from tourism travel in an alpine setting. *Mountain Research and Development* 36, 475–483. <u>https://doi.org/10.1659/MRD-JOUR-NAL-D-16-00058.1</u>
- Vacik, H., Müller, M., Degenhart, J., and Sass, O. (2020). "Auswirkungen von Waldbränden auf die Schutzfunktionalität alpiner Wälder", in *ExtremA 2019. Aktueller Wissensstand zu Extremereignissen alpiner Naturgefahren in Österreich*, ed. T. Glade (Göttingen: Vienna University Press), 173–201.
- van Bebber, W. J. (1891). Die Zugstrassen der barometrischen Minima nach den Bahnenkarten der deutschen Seewarte für den Zeitraum 1875–1890. Meteorologische Zeitschrift 8, 361–366.
- Väre, H., Lampinen, R., Humphries, C., and Williams, P. (2003). "Taxonomic Diversity of Vascular Plants in the European Alpine Areas", in *Alpine Biodiversity in Europe – A Europe-wide Assessment of Biological Richness and Change*, eds. L. Nagy, G. Grabherr, C. Körner, and D. B. A. Thompson (Berlin: Springer), 133–148. Available at: 10.1007/978-3-642-18967-8\_5
- Vergara-Temprado, J., Ban, N., and Schär, C. (2021). Extreme Sub-Hourly Precipitation Intensities Scale Close to the Clausius-Clapeyron Rate Over Europe. *Geophysical Research Letters* 48, e2020GL089506. <u>https://doi.org/10.1029/2020GL089506</u>
- Vitasse, Y., Ursenbacher, S., Klein, G., Bohnenstengel, T., Chittaro, Y., Delestrade, A., et al. (2021). Phenological and elevational shifts of plants, animals and fungi under climate change in the European Alps. *Biological Reviews* 96, 1816–1835. <u>https://doi. org/10.1111/brv.12727</u>
- Viviroli, D., Dürr, H. H., Messerli, B., Meybeck, M., and Weingartner, R. (2007). Mountains of the world, water towers for humanity: Typology, mapping, and global significance. *Water Resources Research* 43. <u>https://doi.org/10.1029/2006WR005653</u>
- Wagner, T., Seelig, S., Helfricht, K., Fischer, A., Avian, M., Krainer, K., et al. (2021). Assessment of liquid and solid water storage in rock glaciers versus glacier ice in the Austrian Alps. *Science* of *The Total Environment* 800, 149593. <u>https://doi.org/10.1016/j. scitotenv.2021.149593</u>
- Wagner, T., Themeßl, M., Schüppel, A., Gobiet, A., Stigler, H., and Birk, S. (2017). Impacts of climate change on stream flow and

hydro power generation in the Alpine region. *Environmental Earth Sciences* 76, 4. <u>https://doi.org/10.1007/s12665-016-6318-6</u>

- Walter, M., Walser, M., and Joswig, M. (2011). Mapping Rainfall-Triggered Slidequakes and Seismic Landslide-Volume Estimation at Heumoes slope. *Vadose Zone Journal* 10, 487–495. <u>https://doi.org/10.2136/vzj2009.0147</u>
- Wanner, A., Pröbstl-Haider, U., and Feilhammer, M. (2021). The future of Alpine pastures – Agricultural or tourism development? Experiences from the German Alps. *Journal of Outdoor Recreation and Tourism* 35, 100405. <u>https://doi.org/10.1016/j.</u> jort.2021.100405
- Weber, F., Roßmeier, A., Jenal, C., and Kühne, O. (2017). "Landschaftswandel als Konflikt", in Landschaftsästhetik und Landschaftswandel. RaumFragen: Stadt Region Landschaft, eds. O. Kühne, H. Megerle, and F. Weber (Wiesbaden: Springer VS), 215–244. https://doi.org/10.1007/978-3-658-15848-4\_13
- Wechsler, T., Stähli, M., Jorde, K., Zappa, M., and Schaefli, B. (2023). The future of alpine run-of-river hydropower production: climate change, environmental flow requirements, and technical production potential. *Science of The Total Environment* 890, 163934. <u>https://doi.org/10.1016/j.scitotenv.2023.163934</u>
- Weiss, P., Braun, M., Fritz, D., Gschwantner, T., Hesser, F., Jandl, R., et al. (2020). Endbericht zum Projekt CareforParis. Klima- und Energiefonds Wien. Vienna: Umweltbundesamt. Available at: <u>https://www.waldgeschichten.com/wp-content/ uploads/2021/11/B670274-ACRP9-CareforParis-KR16AC0K</u> <u>13154-EB.pdf</u>
- Wenzel, W. W., Duboc, O., Golestanifard, A., Holzinger, C., Mayr, K., Reiter, J., et al. (2022). Soil and land use factors control organic carbon status and accumulation in agricultural soils of Lower Austria. *Geoderma* 409, 115595. <u>https://doi.org/10.1016/j.geoderma.2021.115595</u>
- Wessely, J., Hülber, K., Gattringer, A., Kuttner, M., Moser, D., Rabitsch, W., et al. (2017). Habitat-based conservation strategies cannot compensate for climate-change-induced range loss. *Nature Climate Change* 7, 823–827. <u>https://doi.org/10.1038/nclimate3414</u>
- Wimberly, F., Ultee, L., Schuster, L., Huss, M., Rounce, D. R., Maussion, F., et al. (2024). Inter-model differences in 21<sup>st</sup> Century Glacier Runoff for the World's Major River Basins. *EGUsphere[preprint]* 2024, 1–27. <u>https://doi.org/10.5194/egusphere-2024-1778</u>
- Winkler, M., Lamprecht, A., Steinbauer, K., Hülber, K., Theurillat, J.-P., Breiner, F., et al. (2016). The rich sides of mountain summits – a pan-European view on aspect preferences of alpine plants. *Journal of Biogeography* 43, 2261–2273. <u>https://doi. org/10.1111/jbi.12835</u>
- Wohlgemuth, T., Schwitter, R., Bebi, P., Sutter, F., and Brang, P. (2017). Post-windthrow management in protection forests of the Swiss Alps. *European Journal of Forest Research* 136, 1029– 1040. <u>https://doi.org/10.1007/s10342-017-1031-x</u>
- Wonka, E. (2008). Neuabgrenzung des Dauersiedlungsraums. *Statistische Nachrichten*, 432–442.
- Yousefpour, R., and Gray, D. R. (2022). Managing forest risks in uncertain times of climate change. *Annals of Forest Science* 79, 16. <u>https://doi.org/10.1186/s13595-022-01135-x</u>
- Žebre, M., Colucci, R. R., Giorgi, F., Glasser, N. F., Racoviteanu, A. E., and Del Gobbo, C. (2021). 200 years of equilibrium-line altitude variability across the European Alps (1901–2100). *Climate Dynamics* 56, 1183–1201. <u>https://doi.org/10.1007/s00382-020-05525-7</u>

- Zech, S., Dangschat, J. S., Dillinger, A., Feilmayr, W., Hauger, G., Kogler, R., et al. (2013). *Tourismusmobilität 2030.*, ed. Bundesministerium für Wirtschaft, Familie und Jugend (BMWFJ). Vienna. Available at: <u>https://www.wko.at/oe/tourismus-freizeitwirtschaft/hp-tourismusmobilitaet2030-langfassung-25.11.pdf</u> (Accessed January 22, 2025).
- Zehrer, A., and Mössenlechner, C. (2009). Key Competencies of Tourism Graduates: The Employers' Point of View. Journal of Teaching in Travel & Tourism 9, 266–287. <u>https://doi.org/10. 1080/15313220903445215</u>
- Zekollari, H., Huss, M., and Farinotti, D. (2019). Modelling the future evolution of glaciers in the European Alps under the EU-RO-CORDEX RCM ensemble. *The Cryosphere* 13, 1125–1146. <u>https://doi.org/10.5194/tc-13-1125-2019</u>
- Zgheib, T., Giacona, F., Granet-Abisset, A. M., Morin, S., Lavigne, A., and Eckert, N. (2022). Spatio-temporal variability of avalanche risk in the French Alps. *Regional Environmental Change* 22, 8. <u>https://doi.org/10.1007/s10113-021-01838-3</u>
- Zieher, T., Gallotti, G., Rianna, G., Reder, A., and Pfeiffer, J. (2023). Exploring the effects of climate change on the water balance of a continuously moving deep-seated landslide. *Natural Hazards* 115, 357–387. <u>https://doi.org/10.1007/s11069-022-05558-7</u>

- Zieher, T., Rutzinger, M., Schneider-Muntau, B., Perzl, F., Leidinger, D., Formayer, H., et al. (2017a). Sensitivity analysis and calibration of a dynamic physically based slope stability model. *Natural Hazards and Earth System Sciences* 17, 971–992. <u>https:// doi.org/10.5194/nhess-17-971-2017</u>
- Zieher, T., Schneider-Muntau, B., and Mergili, M. (2017b). Are real-world shallow landslides reproducible by physically-based models? Four test cases in the Laternser valley, Vorarlberg (Austria). Landslides 14, 2009–2023. <u>https://doi.org/10.1007/ s10346-017-0840-9</u>
- Zimova, M., Hackländer, K., Good, J. M., Melo-Ferreira, J., Alves, P. C., and Mills, L. S. (2018). Function and underlying mechanisms of seasonal colour moulting in mammals and birds: what keeps them changing in a warming world? *Biological Reviews* 93, 1478–1498. https://doi.org/10.1111/brv.12405
- Zosso, C. U., Ofiti, N. O. E., Torn, M. S., Wiesenberg, G. L. B., and Schmidt, M. W. I. (2023). Rapid loss of complex polymers and pyrogenic carbon in subsoils under whole-soil warming. *Nature Geoscience* 16, 344–348. <u>https://doi.org/10.1038/s41561-023-01142-1</u>





# Chapter 8 Transformation pathways



# Second Austrian Assessment Report on Climate Change | AAR2

# Chapter 8 Transformation pathways

This publication was subject to an anonymous international peer review process guided by several review editors, the AAR2 co-chairs and the APCC Steering Committee.

The views or opinions expressed in this publication are not necessarily those of the institutions with whose support this work was carried out. The publisher and the supporting institutions do not guarantee the accuracy or continued availability of urls to external websites or third-party websites mentioned in this publication and accept no responsibility for the accuracy or appropriateness of any content contained therein.

#### **Recommended citation**

A. Novy, G. Resch, N. Spittler, et al. (2025): Transformation pathways.
In "Second Austrian Assessment Report on Climate Change (AAR2) of the Austrian Panel on Climate Change (APCC)".
[D. Huppmann, M. Keiler, K. Riahi, H. Rieder (eds.)]. Austrian Academy of Sciences Press, Vienna, Austria | doi: doi.org/10.1553/aar2-ch8

You can download the recommended citation in the formats Endnote (.ris) and BibTex (.bib) at <u>aar2.ccca.ac.at/citation</u>.

Vienna, June 2025 ISBN: 978-3-7001-5110-4 | <u>austriaca.at/5110-4</u> (full report)

#### Production

Publisher: Austrian Academy of Sciences Press, Vienna Typesetting: Crossdesign Werbeagentur GmbH, Graz Cover: paulaidaanton | Agentur für digiloges Design Graphic design Figures 8.4, 8.5: paulaidaanton | Agentur für digiloges Design Al-based tools such as DeepL were used to support the linguistic revision.

Contact address for product safety: Austrian Academy of Sciences Press Dr. Ignaz Seipel-Platz 2 A-1010 Wien Tel.: +43-1-51581-3420 E-Mail: verlag@oeaw.ac.at

#### Funding acknowledgement

The AAR2 was funded by the Climate and Energy Fund under the 14th Austrian Climate Research Program (ACRP).

#### **Copyright and license**

© 2025 Austrian Panel on Climate Change (APCC) This report is published under a Creative Commons CC-BY 4.0-License. <u>creativecommons.org/licenses/by/4.0</u>





The AAR2 is a report by the Austrian Panel on Climate Change (APCC). https://aar2.ccca.ac.at/en

# **Coordinating Lead Authors**

Andreas Novy	Vienna University of Economics and Business   WU
Gustav Resch	Austrian Institute of Technology GmbH   AIT
Nathalie Spittler	BOKU University

# Lead Authors

Ernest Aigner	Leuphana University Lüneburg
Richard Bärnthaler	University of Leeds
Birgit Bednar-Friedl	University of Graz
Tara Esterl	Austrian Institute of Technology GmbH   AIT
Veronika Gaube	BOKU University
Michael Getzner	Technical University Vienna   TU Wien
Simon Grabow	Vienna University of Economics and Business   WU
Klaus Kubeczko	Austrian Institute of Technology GmbH   AIT
Wolfgang Lexer	Environment Agency Austria   UBA
Michael Miess	King Abdullah University of Science and Technology & Environment Agency Austria
	UBA
Irene Monasterolo	Utrecht University & Centre for Economic Policy Research & Vienna University of
	Economics and Business   WU
Asjad Naqvi	Austrian Institute of Economic Research   WIFO
Ece Özer	Technical University Vienna   TU Wien
Elke Pirgmaier	University of Lausanne
Sigrid Stagl	Vienna University of Economics and Business   WU
Keith Andrew Williges	University of Graz

# **Contributing Authors**

Bertram Barth	INTEGRAL institute
Esther Blanco	University of Innsbruck   UIBK
Karin Fischer	Johannes Kepler University Linz   JKU
Elisabeth Gsottbauer	University of Innsbruck   UIBK
Jakob Kapeller	University of Duisburg-Essen
Michael Kirchler	University of Innsbruck   UIBK
Mathias Kirchner	BOKU University
Daniela Kletzan-Slamanig	Austrian Institute of Economic Research   WIFO
Teresa Lackner	University of Graz
Lukas Liebmann	Austrian Institute of Technology GmbH   AIT
Verena Madner	Vienna University of Economics and Business   WU
Werner Raza	Austrian Foundation for Development Research   ÖFSE
Chapter Scientist	
Simon Grabow	Vienna University of Economics and Business   WU
<b>Review Editors</b>	
Celine Guivarch	Centre international de recherche sur l'environnement et le développement   CIRED
Gunnar Luderer	Potsdam Institute for Climate Impact Research   PIK

## Liaisons from AAR2 Technical Support Unit

Karl Scheifinger	International Institute for Applied Systems Analysis   IIASA
Jan Steinhauser	International Institute for Applied Systems Analysis   IIASA

## **TABLE OF CONTENTS**

Exec	utive S	Summary	517
8.1.	Introd	luction	520
8.2.	Bioph	ysical limits and carbon budgets	520
	8.2.1.	Biophysical limits with respect to	
		planetary boundaries	520
	8.2.2.	Carbon budget analysis	522
8.3.	Synth	esis and assessment of mitigation	
	optior	าร	524
	8.3.1.	Mitigation options and the Sustainable	
		Development Goals	524
	8.3.2.	Costs and potentials of mitigation	
		options	527
8.4.	Assess	sment of Austrian mitigation scenarios	529
	8.4.1.	Overview of Austrian mitigation	
		scenarios	529
	8.4.2.	Assessment of Austrian mitigation	
		scenarios: GHG emissions and energy	F 2 1
		balance	531
		Total GHG emissions	531
		Energy-related CO <sub>2</sub> emissions	532
		Energy balance	533
	8.4.3.	Spotlight on key energy carriers	535
		Electricity sector	535
	~ • • •	Hydrogen and renewable synthetic fuels	537
	8.4.4.	Economic impacts	538
		The EAA's WEM, WAM, Transition 2040	
		of Climate Protection (BMK) and ELL	
		monitoring mechanism	538
		Base and activity scenarios for long term	550
		budget forecast for the Austrian Ministry	
		of Finance (EAA-BMF scenarios)	540
		Investment needs	541
	8.4.5.	Scenario assessment with regard to	
		sustainable development and available	
		carbon budgets	542
		Austrian mitigation scenarios and the	
		Sustainable Development Goals	542
		Austrian mitigation scenarios and	
		carbon budgets under varying fairness	
		criteria	543

8.5.	Politico-economic and socio-cultural								
	feasibility challenges for climate-friendly								
	transf	ormation pathways	543						
	8.5.1.	Global governance	544						
		Inhibitors of climate-friendly living	544						
		Enablers of climate-friendly living	545						
	8.5.2.	Democracy, law and science	545						
		Inhibitors of climate-friendly living	545						
		Enablers of climate-friendly living	546						
	8.5.3.	Politico-economic framework conditions	548						
		Inhibitors of climate-friendly living	548						
		Enablers of climate-friendly living	549						
	8.5.4.	Social and socio-cultural framework							
		conditions	550						
		Inhibitors of climate-friendly living	550						
		Enablers of climate-friendly living	551						
	8.5.5.	Adaptation	552						
		Inhibitors of climate-friendly living	552						
		Enablers of climate-friendly living	553						
	Chapt	er Box 8.1							
	Persp	ectives, multi-perspectivity, narratives							
	and q	ualitative transformation pathways	554						
8.6.	Clima	te-friendly transformation pathways	556						
8.6.	<b>Clima</b> 8.6.1.	<b>te-friendly transformation pathways</b> Planning climate-friendly	556						
8.6.	<b>Clima</b> 8.6.1.	<b>te-friendly transformation pathways</b> Planning climate-friendly transformations	<b>556</b>						
8.6.	Clima 8.6.1. 8.6.2.	<b>te-friendly transformation pathways</b> Planning climate-friendly transformations Advancing from an 'eco-modernist' to a	<b>556</b> 556						
8.6.	<b>Clima</b> 8.6.1. 8.6.2.	<b>te-friendly transformation pathways</b> Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway	<b>556</b> 556 558						
8.6.	Clima 8.6.1. 8.6.2.	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco-	<b>556</b> 556 558						
8.6.	<b>Clima</b> 8.6.1. 8.6.2.	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway	<b>556</b> 556 558 558						
8.6.	<b>Clima</b> 8.6.1. 8.6.2.	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible	<b>556</b> 558 558						
8.6.	<b>Clima</b> 8.6.1. 8.6.2.	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway	<b>556</b> 558 558 558						
8.6.	Clima 8.6.1. 8.6.2. 8.6.3.	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Advancing from a 'system change' to a	<b>556</b> 558 558 558						
8.6.	Clima 8.6.1. 8.6.2. 8.6.3.	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Advancing from a 'system change' to a climate-friendly transformation pathway	<b>556</b> 558 558 558 558 558						
8.6.	Clima 8.6.1. 8.6.2. 8.6.3.	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Advancing from a 'system change' to a climate-friendly transformation pathway Underlying perspectives of a 'system	<b>556</b> 558 558 558 558 558						
8.6.	Clima 8.6.1. 8.6.2. 8.6.3.	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Advancing from a 'system change' to a climate-friendly transformation pathway Underlying perspectives of a 'system change' transformation pathway	<b>556</b> 558 558 558 558 558 561						
8.6.	Clima 8.6.1. 8.6.2. 8.6.3.	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Advancing from a 'system change' to a climate-friendly transformation pathway Underlying perspectives of a 'system change' transformation pathway Advancing to an effective and feasible	<ul> <li>556</li> <li>558</li> <li>558</li> <li>558</li> <li>561</li> <li>561</li> </ul>						
8.6.	Clima 8.6.1. 8.6.2. 8.6.3.	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Advancing from a 'system change' to a climate-friendly transformation pathway Underlying perspectives of a 'system change' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway	<ul> <li><b>556</b></li> <li><b>558</b></li> <li><b>558</b></li> <li><b>558</b></li> <li><b>561</b></li> <li><b>561</b></li> <li><b>561</b></li> </ul>						
8.6.	Clima 8.6.1. 8.6.2. 8.6.3.	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Advancing from a 'system change' to a climate-friendly transformation pathway Underlying perspectives of a 'system change' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Chapter Box 8   Green Finance	<ul> <li>556</li> <li>558</li> <li>558</li> <li>558</li> <li>561</li> <li>561</li> <li>561</li> <li>561</li> <li>561</li> <li>561</li> <li>565</li> </ul>						
8.6.	Clima 8.6.1. 8.6.2. 8.6.3. Cross- Synth	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Advancing from a 'system change' to a climate-friendly transformation pathway Underlying perspectives of a 'system change' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Chapter Box 8   Green Finance	<ul> <li>556</li> <li>558</li> <li>558</li> <li>558</li> <li>561</li> <li>561</li> <li>561</li> <li>561</li> <li>565</li> </ul>						
8.6.	Clima 8.6.1. 8.6.2. 8.6.3. Cross- Synth transf	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Advancing from a 'system change' to a climate-friendly transformation pathway Underlying perspectives of a 'system change' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Chapter Box 8   Green Finance esis of policy-relevant insights of formation pathways	<ul> <li>556</li> <li>558</li> <li>558</li> <li>558</li> <li>561</li> <li>561</li> <li>561</li> <li>565</li> <li>567</li> </ul>						
8.6. 8.7. Refe	Clima 8.6.1. 8.6.2. 8.6.3. Cross- Synth transf erences	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Advancing from a 'system change' to a climate-friendly transformation pathway Underlying perspectives of a 'system change' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Chapter Box 8   Green Finance esis of policy-relevant insights of formation pathways	<ul> <li>556</li> <li>558</li> <li>558</li> <li>558</li> <li>561</li> <li>561</li> <li>561</li> <li>561</li> <li>565</li> <li>567</li> <li>570</li> </ul>						
8.6. 8.7. Refe	Clima 8.6.1. 8.6.2. 8.6.3. 8.6.3. Cross- Synth transf erences	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Advancing from a 'system change' to a climate-friendly transformation pathway Underlying perspectives of a 'system change' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Chapter Box 8   Green Finance esis of policy-relevant insights of formation pathways s 	5556 5558 5558 5558 5558 561 561 561 565 567 570 5570 5570						
8.6. 8.7. Refe	Clima 8.6.1. 8.6.2. 8.6.3. 8.6.3. Cross- Synth transf erences ssary of abbr	te-friendly transformation pathways Planning climate-friendly transformations Advancing from an 'eco-modernist' to a climate-friendly transformation pathway Underlying perspective of an 'eco- modernist' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Advancing from a 'system change' to a climate-friendly transformation pathway Underlying perspectives of a 'system change' transformation pathway Advancing to an effective and feasible climate-friendly transformation pathway Chapter Box 8   Green Finance esis of policy-relevant insights of formation pathways s 	5556 5558 5558 5558 5558 561 561 561 561 565 567 570 5307 5307						

Full report.....aar2.ccca.ac.at/report

#### **EXECUTIVE SUMMARY**

Climate-friendly transformation pathways are trajectories for the pursuit of a good life for all within planetary boundaries and integrate mitigation and adaptation strategies as well as efficiency and sufficiency measures. These pathways draw on narratives that highlight the everyday benefits of climate action and the costs of inaction (robust evidence, medium agreement). As such, climate-friendly transformation pathways integrate the reduction of both direct and indirect emissions to achieve climate neutrality, enhance resilience to climate change impacts through adaptation, and ensure decent living standards. Science-based, plan-led climate policy that underpins these pathways combines bundles of measures that merge insights from a technology-centered eco-modernist, and a sufficiency-oriented system change perspective, with their varying problem diagnoses and policy approaches. Such an integrated approach tends to be more effective than singular approaches due to mobilizing a greater range of available policy tools. Strategic autonomy as well as reduced vulnerabilities and fossil fuel import dependencies can be achieved by lowering demand through sufficiency and avoid-measures. For a long-term transition towards climate-friendly living, it is essential to limit the power and influence of incumbent interests that have diluted regulations. In the short and medium term, improving everyday living conditions by facilitating adaptation to change is crucial (e.g., living cost reducing measures like the climate bonus, repair bonus, and climate ticket). Important are initiatives with co-benefits. Examples are promoting human and planetary health, e.g., by supporting dense settlement structures that provide local access to retail, care, and leisure facilities while also reducing soil-sealing and car-dependency (high confi*dence*). {8.6.1, 8.6.2, 8.6.3, 8.7, Table 8.3}

Climate-friendly transformation pathways strike a balance between incremental and disruptive changes (*robust evidence, medium agreement*). Relying solely on incremental measures, especially those that focus narrowly on technology and efficiency, constitutes a form of climate delay, as such measures alone are insufficient to achieve climate targets. Climate-friendly transformation pathways require policy-mixes that integrate incremental and disruptive measures and combine broadly accepted actions (e.g., green spaces and public transport expansion) with less popular approaches (e.g., dismantling soil-sealing structures like asphalt parking spaces) (*robust evidence, medium agreement*). {8.6.1, 8.6.2, 8.6.3, 8.7, Table 8.3} Austria's remaining carbon budget to stay within global targets of 1.5°C or 2°C is nearly (and by some estimates already) depleted, especially when responsibility for historical emissions and other fairness concerns are considered (*medium evidence, high agreement*). Meeting international obligations to stay within such limits requires Austria to rapidly decarbonize its economy and implement all measures aligned with achieving carbon neutrality by 2040. A depleted carbon budget necessitates costly participation in international assistance and compensation schemes. {8.2.2, 8.4.5, Figure 8.18}

Mitigating climate change can contribute towards the achievement of the Sustainable Development Goals (SDGs), though different mitigation options vary in terms of their synergy and trade-off potential. Demand-side reduction options appear more beneficial for the achievement of SDGs compared to supply-side options (*medium evidence, high agreement*). Across all assessed mitigation scenarios, synergy potentials outweigh trade-offs. However, the extent to which synergies or trade-offs are realized, both within Austria and beyond, depends on the specific measures adopted and their implementation strategies (*high confidence*). {8.3.1, 8.4.5, Figure 8.4, Figure 8.17}

Costs and potentials of mitigation options indicate that electrification, paired with the decarbonization of the electricity supply, as well as demand-side reduction measures are key for Austria to achieve carbon neutrality in 2040 (*medium evidence, high agreement*). Among the assessed mitigation options, electrification of the transport, industry, and building sectors offers a significant greenhouse gas (GHG) reduction potential, provided it is accompanied by decarbonizing the electricity sector through renewable energy sources (*high confidence*). Reducing final energy demand and directly cutting GHG emissions are also necessary, though their potential has only been partially assessed for Austria (*high confidence*). {8.3.2, Figure 8.5}

From a technological and economic standpoint, achieving climate neutrality in the Austrian energy system by 2040 is feasible if unprecedented rates of change in the evolution of the Austrian energy system are realized – along with significant transformations in the associated institutional, regulatory, and legal frameworks (*high confidence*). Current and planned policies fail to achieve carbon neutrality

by 2040. Reducing final energy demand through efficiency improvements and changes in service demand, is thus indispensable for reaching climate neutrality in Austria. The extent of the required demand reduction varies across assessed scenarios, ranging from -8 to -36 % by 2040 compared to 2021 levels. Strong differences are thereby applicable in final energy demand for transport and industry. {8.4.2, Figure 8.6, Figure 8.9, Figure 8.10}

All assessed scenarios that achieve climate neutrality of the energy system share key technological characteristics - but show differences in the technology uptake by 2040 and beyond (medium evidence, high agreement). A common feature across scenarios is increased electrification, ranging from 29 to 76 % by 2040 compared to 2021, alongside a high share of renewables in gross domestic consumption. Wind and solar PV form the backbone of the electricity supply in all scenarios for meeting the growing demand. Biomass and hydropower are additional key technology options, though their expansion potential is limited. Renewable gases, such as hydrogen, biomethane, and synthetic fuels, play a role in decarbonizing industrial processes, certain transport modes (air and ship), and, to a limited extent, the electricity sector. Their deployment by 2040 varies across assessed scenarios, ranging from 106 to 322 PJ/yr. Carbon Capture, Utilization, and Storage (CCUS) appears necessary to counterbalance residual emissions, specifically in industry, due to the absence of other viable technological mitigation options. Alternatively, sufficiency measures may also help mitigating residual emissions in other sectors like agriculture. According to the assessed scenarios, Austria's energy supply continues to rely on imports, particularly for hydrogen, ranging from 8 to 36 % of gross domestic consumption by 2040. This marks a significant decline from today's import dependency, which ranges from 52 to 75 %. {8.4.3, Figure 8.11, Figure 8.12}.

Achieving carbon neutrality by 2040 requires additional annual investments ranging from 1.1 to 1.9 % of GDP or a total of  $EUR_{2023}$  6.2 to 10.9 billion. These investments are below the expected monetized annual costs of inaction of  $EUR_{2023}$  7.3 to 14.6 billion, especially when considering the multiple climate change impacts that have so far not been expressed in monetary terms (e.g., injuries and deaths due to extreme weather events). The required investments are furthermore expected to have positive impacts on employment, income and wealth distribution, as well as on overall economic welfare (*medium confidence*). The

limited number of estimates available on additional investment needs in energy supply, infrastructure, storage, and other sectors (buildings, transport, industry) suggest that these investments can serve as drivers of production and employment. {8.4.4, Figure 8.14, Figure 8.15, Figure 8.16, Table 8.1}

Immediate actions are needed to overcome the inertia in key sectors and to achieve both 2030 and 2040 climate policy targets (high confidence). Current and planned policies in Austria are not sufficient to achieve the 2030 climate policy targets imposed by EU regulation. The recently amended EU Effort Sharing Regulation, comprising smaller GHG emitters that are not controlled by the EU Emission Trading Scheme, requires urgent action to reduce GHG emissions specifically in the transport and building sector. The urgent implementation of far-reaching mitigation measures is also necessary for achieving Austria's long-term GHG commitments: All assessed mitigation scenarios show similar trends concerning demand reduction and the uptake of renewable energies until 2030. This helps to overcome the inertia in the energy system and other provisioning systems (mobility, housing, food, etc.). {8.4.2, 8.4.3}

If political goals, such as increasing geopolitical security and short-term competitiveness, fail to account for climate impacts and equity concerns, Austria will not meet its national climate targets or fulfill international obligations (*high confidence*). Integrating climate policies that transform the provisioning of mobility, housing, food, health, care and leisure within broader sustainable development policies, can help overcoming the policy silos of mitigation and adaptation. Enlarging the policy space is important to better plan transformations. Innovative, democratic, and multi-level governance models can support climate policy integration to better scale up innovations, such as renewable energy expansion, and foster exnovations, such as phasing out combustion engines (*medium confidence*). {8.5.1, 8.6.1}

Democratic policymaking can increase the probability of meeting long-term climate targets by integrating individual measures that currently lack societal support into policy-mixes that lead to benefits for everyday life (*medium confidence*). Such a strategy can become feasible in a democratic setting if disadvantages for actors (e.g., bureaucratic costs and inconveniences) are minimized, as societal and political acceptance of climate measures often increases after implementation – thus legitimizing ambitious climate actions ex post. Democratic policymaking can also choose to align with prevailing societal preferences and obtain political support by reducing public investments and avoiding regulations, thereby, however, implicitly accepting the likelihood of unmet climate targets. {8.6.1, 8.7}

Building on the previous chapters, which primarily explored the broader physical manifestations of climate change in Austria (Chapter 1) and sector-specific impacts and mitigation potentials (Chapters 2-6), Chapter 8 adopts a systemic and integrative approach to assess transformation pathways that align with national and European climate targets (Section 6.3). It comprehensively assesses the Austrian scenario landscape and embeds national mitigation efforts in broader biophysical, socio-economic, and socio-cultural contexts. Mitigation targets are expanded to align with the overarching goal of climate-friendly living, which is defined by the APCC (2023) as a good life for all within planetary boundaries. This concept encompasses both the mitigation of emissions to achieve climate neutrality by 2040 and the adaptation to climate impacts to promote climate resilience. It thereby encompasses changes in the way of living, working and producing.

Section 8.2 departs from an overview of the planetary boundaries, focusing on the crises of biodiversity (Section 8.2.1) and climate change (Section 8.2.2). It provides a carbon budget analysis, examining Austria's remaining carbon budget for limiting global average temperature rise to 1.5°C and 2°C. Section 8.3 integrates, synthesizes, and assesses mitigation options in terms of their broader implications for achieving the Sustainable Development Goals (SDGs) (Section 8.3.1) as well as potentials and costs (Section 8.3.2). Section 8.4 reviews existing mitigation scenarios for Austria and assesses them based on (i) key scenario properties, such as emission trajectories, energy demand and investment needs (Sections 8.4.2, 8.4.3, 8.4.4), and (ii) the scenarios' interactions with and contributions to the SDGs, as well as their alignment with carbon budgets under different fairness criteria (Section 8.4.5). Section 8.5 focuses on the political, economic, and socio-cultural conditions that are critical for realizing climate friendly transformation pathways, as these factors can either hinder or enable effective climate action. In what follows, Section 8.6 extends the assessment by exploring the socio-cultural and politico-economic feasibility of transformation pathways, with the aim of widening the policy space for transitioning towards a climate-friendly future. It critically assesses two ideal-typical pathways, each based on different limited perspectives, and examines their potential for transitioning to a climate-friendly transformation pathway that meets both mitigation and adaptation targets. Finally, Section 8.7 links policy-relevant insights from the assessment of scenarios and transformation pathways,

thereby illustrating how policymakers can use multi-perspectivity to improve their respective value- and interest-based decisions.

#### 8.2. Biophysical limits and carbon budgets

The planetary boundaries framework draws upon Earth system science and identifies processes (such as CO<sub>2</sub> emissions, nitrogen and the land-use system) that are critical for maintaining the stability and resilience of the Earth system (Steffen et al., 2015). Section 8.2 therefore examines the biophysical framework of climate measures, focusing on two crucial policy fields that are at risk of trespassing planetary (systemic) boundaries. First, it undertakes an analysis of biodiversity loss in Austria. Second, it estimates the total emissions for Austria's remaining carbon budget to stay within 1.5°C and 2°C global average temperature rise. This is in line with the growing emphasis on distributing the global carbon budget among subsidiary entities, such as countries, to provide a basic benchmark. Based on this, more sophisticated analyses can be formed to specify feasible and detailed pathways. In fact, it is increasingly recognized that not the public commitment to a net zero emissions target year (e.g., 2040 for Austria) but the cumulative emissions up to that target constitute the greater challenge (CCCA, 2022). Biophysical processes are furthermore closely linked to socio-economic ones, as both affect the global opportunity structure and the biophysical boundary conditions.

# 8.2.1. Biophysical limits with respect to planetary boundaries

The concept of planetary boundaries is based on the finite nature of natural resources and sinks for wastes and emissions. Exceeding even some of these boundaries can have devastating consequences. For example, agricultural biomass production can lead to soil erosion and degradation, phosphorus and nitrogen runoff from overuse of fertilizers, and exceeding the boundaries of biogeochemical cycles and biodiversity loss. Therefore, economic structures and ways of living that enable society to live well within planetary boundaries are needed (Raworth, 2017).

Globally defined planetary boundaries can be specified at the level of individual countries and then compared with country-specific resource use footprint indicators (O'Neill et al., 2018). The analysis shows that Austrian consumption levels far exceed what is tolerable for the country's environment. The high  $CO_2$  emission footprint extends far beyond the planetary boundary of climate change. Similarly, the recorded levels for phosphorus and nitrogen, which are used in large quantities in agriculture, exceed the defined pollution thresholds. Austria's material and ecological footprint exceed the critical thresholds of the planet three- or fourfold. In the case of land use, Austria is only slightly outside the acceptable limits, and only water use remains within the boundaries, primarily due to the large available water reserves in Austria. Translating planetary boundaries into targets that are relevant for policy and decision-making requires addressing their biophysical, socio-economic and ethical dimensions. This requires an ongoing, iterative dialogue and deeper collaboration between scientists and policy makers (Hoff et al., 2017).

One important social driver jeopardizing the feasibility of achieving climate targets and the objective of staying within planetary boundaries is consumption and the associated resource use. In Austria, consumption levels and the resulting resource utilization of production and consumption processes have an adverse impact on biogeochemical cycles and biodiversity patterns of global ecosystems (BMK, 2020). In relation to the planetary boundaries, scientific and political endeavors increasingly focus on enhancing capabilities for estimating the global impacts of national resource use



Figure 8.1 Austria and the planetary boundaries (Fanning et al., 2022).

and to define acceptable limits at this scale (Rockström et al., 2009; Steffen et al., 2015). Fanning et al. (2022) analyze the historical dynamics of social and biophysical indicators across more than 140 countries from 1992 to 2015. Social indicators include, for example, life expectancy, nutrition, access to energy, and education. The ecological indicators downscale the three planetary boundaries of climate change, biogeochemical flows and land use change by applying different environmental footprint indicators (e.g.,  $CO_2$  emissions). The study demonstrates that **Austria is among the affluent countries with low social shortfall but high ecological overshoot** (Figure 8.1). In addition, it shows that the magnitude of the ecological overshoot in Austria has been increasing over the period 1992–2015, while social indicators have only hardly improved.

A plethora of measures exist to protect, sustainably manage and restore natural and modified ecosystems, many of which demonstrate co-benefits for climate change adaptation and mitigation, as well as biodiversity conservation (Smith et al., 2022). In addition to the Paris Agreement (Section 6.3.1), the EU is obliged to comply with the Kunming-Montreal Global Biodiversity framework, which was adopted at the Convention on Biological Diversity at the 15th Conference of the Parties in 2022. The most prominent targets include the restoration of 30 % of degraded ecosystems globally by 2030. Restoration refers to the process of halting the deterioration of ecosystems or reversing the effects of land degradation through activities such as reforestation, soil conservation and the protection of natural processes. The objective of these actions is to enhance biodiversity, recover ecosystem services and reduce the impact of climate change (Scholes et al., 2018). Austria is characterized by a considerable high diversity of landscapes, encompassing the Pannonian plains in the east and high alpine regions in the west, including wetlands and forest areas (Section 1.1). Consequently, it is one of the most species-rich countries in Central Europe. Human intervention, however, has been a significant factor in shaping Austria's biodiversity throughout history, particularly through agriculture, forestry, hunting, and fishing (Section 1.6.4). Among the main causes of biodiversity loss are habitat destruction, degradation, and fragmentation, in particular the sealing and fragmentation of landscapes due to human settlement and transport infrastructure expansion. Additional threats include the abandonment of traditional land use practices and the intensification of land use. According to BMK (2022), approximately one-third to half of all species are critically endangered.

#### 8.2.2. Carbon budget analysis

The most directly relevant planetary boundary being transgressed regarding climate mitigation is the exceedance of the remaining global carbon budget. Such budgets are based on concepts from the physical sciences. The allocation of a global budget to regional levels, such as countries or sub-national regions, however, is a process driven by normative considerations (e.g., Friedlingstein et al., 2014; Pan et al., 2015; Rogelj et al., 2016; Millar et al., 2017; Xu and Ramanathan, 2017; Dooley et al., 2021; CCCA, 2022). Dividing budgets among subsidiary entities can be driven by any one, or a combination of, equity and fairness arguments (Caney, 2009; van den Berg et al., 2020). While a specific budget (or uncertainty range) may be feasible to determine at a global level, estimates of the remaining budget for Austria vary more widely and may change substantially due to different forms of emission accounting and allocation mechanisms at play (Steininger et al., 2022; Williges et al., 2022).

The means of accounting for emissions can lead to differing carbon budget allocations and emission reduction requirements. Several potential approaches exist, which can have different implications for countries when discussing the effects of their actions (i.e., production processes or consumption habits) on global mitigation efforts. The most common approaches are production- and consumption-based accounting. However, emissions could be attributed to the country that extracted the fossil fuels leading to the emissions (extraction-based), or to the countries that add value to a production process (income-based) recognizing that, e.g., labor and capital benefit from production-related emissions by earning income. Each approach implies a different perspective on the economy, agency, and responsibility for emissions (Steininger et al., 2016; Aigner et al., 2023e).

The primary framework used with respect to carbon budgets is production-based accounting. It follows a sector-based approach to measure emissions arising from production and consumption processes within the respective geographical borders and allocates them to the original producers such as industries, private households and public agents (e.g., 'source categories') (UNFCCC, 2009). Alternatively, consumption-based accounting measures emissions arising from a country's final demand (e.g., household consumption, government expenditures and capital accumulation through investment) along international supply chains. These emissions can be allocated to different consumption categories, thereby excluding territorial emissions from the production of goods and services that are exported. Since the production of emissions and the consumption of related goods and services occur in different regions, the two accounting frameworks tend to show different pictures of a country's GHG emissions (Steininger et al., 2018). Generally, **countries of the global North or high-income countries are net-carbon importers**. Consequently, their production-based are lower than consumption-based emissions. When using production-based distributions, these countries then have comparatively lower emissions levels (and thus, higher relative shares of a carbon budget) as compared to consumption-based approaches.

The equity principles underlying budget allocation to countries can also have a substantial impact on the mitigation effort required. Three broad equity aspects are relevant: (i) capability, (ii) responsibility and (iii) need (or equality) (Höhne et al., 2014; Williges et al., 2022). Capability considerations influence how a global budget is allocated by placing higher mitigation burden on those countries with the highest ability to pay or abate (Shue, 1999; Caney, 2009). Responsibility focuses on the 'polluter pays' principle by allocating higher mitigation burden to high-emitting countries (Gosseries, 2004). Finally, considerations of satisfaction of need focus on equality of opportunity, reduction of poverty, and focusing on those most vulnerable to climate change (Gough, 2015). Typically, budget allocation mechanisms use a combination of equity considerations, incorporating aspects of equality or need, historical responsibility and capability (Höhne et al., 2014; van den Berg et al., 2020; Steininger et al., 2022; Williges et al., 2022). Additional approaches to effort-sharing focus on equal marginal abatement costs, i.e., allocations arising from a global carbon price, or the idea of equal cumulative per capita emissions. In contrast to equality or equal cumulative per capita emissions, arguments have been made for convergence to equal per capita emissions at a future point in time (e.g., 'grandfathering'), which give some normative weight to the current unequal rates of emissions, which persists at a diminishing rate into the future (Williges et al., 2022).

Beyond the basic choice of equity principles employed, the manner in which these principles are interpreted and implemented can also have a large effect on the resulting budget (Steininger et al., 2022). This implies that there is not a single available carbon budget for a given temperature target for Austria. Rather, any result will be the product of a set of normative decisions as to what is deemed important when allocating a budget. Estimates for Austria do allow for drawing conclusions as to the likely range of the carbon budget left to achieve climate goals. Additionally, trends can be observed regarding how normative considerations and emission accounting choices affect the resulting budget.

Figure 8.2a illustrates historical Austrian emissions from 1995 onwards, differentiating between production- and consumption-based emissions to 2022, with 2022 to 2050 illustrating the range of possible emission pathways compatible with national carbon budgets derived from the literature. Such budgets correspond to either a 1.5 or 2°C temperature stabilization target, and take into account a range of historical responsibility, equality, and need considerations. As panel (a) shows, consumption-based emissions consistently exceed production-based, typically around 30 Mt per year. Given that most carbon budgets follow a production-based approach, considerations of consumption emissions could narrow future equity-compatible pathways even further. As it is, most budgets targeting 1.5 or 2°C would require net negative emissions at some point between now and 2050. A majority of budgets for the 1.5°C target would require overall negative budgets for the entire time period. As can be seen, the current WEM and WAM scenarios would not correspond to what the literature considers a fair allocation of a global carbon budget.

As noted previously, while a single budget estimate for Austria is incompatible with the normative concepts and their implementation when dividing the global budget, a range of estimates can be derived. Figure 8.2b depicts this range of estimated budget amounts in tons of  $CO_2$  for two





temperature targets: A 1.5°C target and a well-below 2°C target (66 % probability/no overshoot), with estimates derived from the literature on the allocation of global or regional (e.g., EU) budgets. Median values are represented by solid lines, while boxes represent values between the 25th and 75th percentile range, with whiskers indicating values within 1.5 times the interquartile range. Other values are depicted as points. Budgets are differentiated by the underlying equity considerations driving the allocation mechanism, in line with the categories mentioned above.

A report by the CCCA (2022) estimates the remaining carbon budget for Austria from 2022 to 2050 to be 240 Mt for a 1.5°C target (with 66 % probability) and 280 Mt for a 1.5°C target with overshoot (again with 66 % probability). This aligns with an equality approach to allocation, but it should be noted that this represents just one type of distribution mechanism. As Figure 8.2b shows, results can vary widely depending on the underlying equity considerations. Approaches emphasizing responsibility (e.g., historical emissions considerations) or capability interpretations such as increased effort for high GDP countries lead to extremely low budgets – in the case of historical emissions, budgets are often negative. Conversely, per capita convergence (or 'status-quo') approaches starting from today's levels of emissions lead to much higher allocations.

Given that Austria's CO<sub>2</sub> emissions in 2022 were roughly 57 Mt and future scenarios do not anticipate a steep immediate decline, it is highly likely the remaining budget will be rapidly depleted, if it is not already (due to the large number of negative budgets). As these budgets correspond to agreed-upon targets of well below 2°C of global change, they imply a need for concerted effort to implement measures aligned with reaching net zero emissions as soon as possible to meet international obligations (e.g., the Paris Agreement).

Going beyond the national level, recent work emphasizes the relevance of carbon budgets and accounting for sectors (Steininger et al., 2020b) and sub-national authorities such as cities and federal provinces (Salon et al., 2010; Anderson et al., 2018; CCCA, 2022; Hale et al., 2022; Kuriakose et al., 2022) but evidence for Austria is scarce. Such budgets and accounts can provide planning guidance for sub-national authorities and actor groups when designing, implementing, monitoring and adjusting local mitigation policies. The distinction between production- and consumption-based emissions gains in importance the smaller the regional scale. Hence, the (fair) allocation of carbon budgets across federal states is a complex task and needs to differentiate between production- and consumption-based accounting (CCCA, 2022). Research and debate as to the proper approach to downscaling national budgets and efficacy of, e.g., production- vs. consumption-based emissions in representing or obscuring globally interdependent value chains, multi-level production structures and the political or institutional mandates states and subnational regions is ongoing and makes determination of sub-national budgets a complex task with no exact results.

# 8.3. Synthesis and assessment of mitigation options

# 8.3.1. Mitigation options and the Sustainable Development Goals

Section 8.3.1 assesses how climate change mitigation options interact with sustainable development. It first evaluates the synergy and trade-off potentials between mitigation options considered in the quantitative scenarios of Section 8.4 and the Sustainable Development Goals (SDGs). It then provides an overview of the main interactions between key climate adaptations strategies and the SDGs.

To identify climate-friendly transformation pathways (Section 8.6), it is essential to take a systems perspective that goes beyond the mere decarbonization of the economy by including the SDGs. These goals were established in 2015, the same year that the Paris Agreement was signed, and came into effect in January 2016. The 17 SDGs encompass 169 targets related to sustainable development in the environmental, social and economic domain, with progress being measured by 169 indicators at the national, European and international level. The reported indicators on the different levels are defined by the statistics available on those levels. Following UN level indicators, Austria currently has an SDG Index Score of 82.3, ranking 5th out of the 166 countries, while Finland ranks highest with a score of 86.76, and South Sudan ranks lowest with a score of 38.68 (Sachs et al., 2023). Considering EU level indicators, Austria scores 77.7, placing it 4th of 34 countries. While Finland again ranks highest with a score of 80.64, Turkey ranks lowest with a score of 57.14 (Sachs et al., 2023).

Targets related to climate change mitigation and adaptation are mostly represented in SDG13: Climate Action. However, other SDGs also address climate-relevant aspects, such as emission reductions. Some SDGs and their respective targets directly relate to measures to reduce climate impacts (e.g., SDG7: Affordable and Clean Energy, SDG 12: Cleaner Production and Consumption), while others are only indirectly linked to climate change mitigation and adaptation (e.g., SDG6: Clean Water and Sanitation for All, SDG12: Life on Land). A growing body of literature assesses the interaction of achieving individual SDGs (Bennich et al., 2020), as many synergies and trade-offs result from strategies that aim to achieve specific goals or a subset of targets (Horvath et al., 2022). In Austria, a number of scientific options for achieving the SDGs were developed and their interactions assessed (Allianz Nachhaltige Universitäten in Österreich, 2021).

The latest IPCC reports (IPCC, 2022b; Pörtner et al., 2022) assessed the synergy and trade-off potentials between

climate change mitigation options and the SDGs. The results from the global assessment by Roy et al. (2018) were used as a starting point for the Austria-specific assessment (for a more detailed description of the methods applied, see 8.A.1). The Austrian analysis, however, only considers interactions relevant for domestic SDG attainment. Therefore, SDG 14: Life below Water, SDG 16: Peace and Justice Strong Institutions, and SDG 17: Partnerships for Goals were excluded from the assessment (for discussion of spillover effects, see Section 8.4.5). Overall, the final results (Figure 8.3) align with those of the IPCC SR1.5 (Roy et al., 2018) in terms of synergy or trade-off potentials, but differ with regards to the strength of their interaction potential. Confidence in these interactions is medium to high (light and dark blue),

			Relation with Sustainable Development Goals													
			social enviror							ronme	imental economic					
Climate cl	nange mitigation me	asures and its interaction with SDGs	1	2	3	4	5	10	6	12	15	7	8	9	11	
	Industry	Accelerating energy efficiency improvement			+	+		+	x	+		+	x	+	+	
		Low-carbon fuel switch			+	+			X	+	X	x	+	+	+	
		Decarbonisation/CCS/CCU			1				-	+		x	+	+		
D	Buildings	Behaviorial response	+		+					+		+	+	+	+	
E M		Accelerating energy efficiency improvement	x		+					+	+	+	x	+	+	
A N		Improved access & fuel switch to modern low-carbon energy	+		+					+		+	+	+	+	
D	Transport	Behavioural response	x	+	x	+	+	+	+	+	+	+	x	+	+	
		Accelerating energy efficiency improvement	x		+				x	+	x	+	x	x	+	
		Improved access & fuel switch to modern low-carbon energy	x		+			+	+	+		+	x	+	+	
S U	Replacing coal	Non-biomass renewables solar, wind, hydro	+		+		+	x	x	+	x	+	+	x	+	
P		Increased use of biomass	x		+				x	+	х	+	+			
P L		Nuclear/Advanced Nuclear			_				x		_	+	+	_		
Y	Advanced coal	CCS: Fossil			_				_			+	_	+		
	Agriculture & Livestock	Behaviourial response: Sustainable healthy diets and reduced food waste	_	+	+				+	+	x	+	x	+		
O T		Land based greenhouse gas reduction and soil carbon sequestration		+		x			x	+	x	+	x	x		
I H E R		Greenhouse gas reduction from improved livestock production and manure management systems	+	+					x	+	x	+	+	+		
	Forest	Reduced deforestation, REDD+	+	-					x	+	+	x		x		
		Afforestation and reforestation		x	+	+			+		+	x			+	
		Behaviourial response (responsible sourcing)							+	+	+	+	+	+	+	
			Туре	of rel	ation:						Confi	idence	level			
			+ Synergies							High confidence						

-

х

Trade-offs

Both synergies and trade-offs

Blanks represent no interaction or assessment



Medium confidence

Low confidence

with a few low confidence (gray) cases. Nonetheless, further research is needed to understand whether missing interactions are a result of a lack of direct interactions or due to research gaps. The assessment reveals that most mitigation options have synergy potentials (+) or synergy and trade-off potentials (x) with the SDGs, while only a few mitigation options demonstrate trade-off potentials (-). However, it is important to emphasize that the realization of synergy or trade-off potentials depends on the specific means of implementation. Thoughtful policy design - such as in the case of carbon tax reforms - can maximize synergies and reduce trade-offs, enhancing climate-friendly living (Klenert et al., 2018; Großmann et al., 2019; Kirchner et al., 2019; Mayer et al., 2021). Complementing climate change mitigation strategies that show trade-off potentials with additional measures thus bears the potential to further reduce these trade-off potentials.

In line with the findings of Roy et al. (2018) on the international level, synergy potentials consistently outweigh trade-off potentials across all mitigation options. For energy system related (i.e., combined supply and demand side in Figure 8.3) mitigation options, demand side mitigation options show greater synergy potential than supply side options across the environmental, social and economic dimension (Figure 8.4). Across all mitigation categories, the potential synergies are greatest for SDGs within the economic domain, largely due to the strong synergetic potential of demand-focused mitigation options. Land-use-related mitigation options, in contrast, show the greatest synergetic potentials with environmental SDGs and generally have lower trade-off potentials than supply side options. Trade-off potentials remain minimal across all categories and most SDGs do not show any trade-off potential at all. While land-based mitigation options have lower overall interaction potential than energy system solutions, they still yield more synergy potentials than trade-offs.

Beyond the conventional categorization of supply, demand and land mitigation options, mitigation options can also be categorized according to the related ideal typical transformation pathways (Section 8.6). While some miti-



Figure 8.4 Indicative linkages between aggregated mitigation options and SDGs. Economic SDGs include SDG7, 8, 9 and 11. Environmental SDGs include SDG6, 12, 15. Social SDGs include SDG1, 2, 3, 4, 5 and 10. The bar length indicates respective synergy/trade-off potentials according to the Nilsson scale. Figure based on IPCC (2022c).

gation options relate to ecomodernism by improving components of the current system, others aim at degrowth by transforming existing system structures. Mitigation options that are more optimizing have higher synergy potentials with economic SDGs and mitigation options that aim at transforming the system have higher synergy potentials with social SDGs.

Adaptation measures are equally important to climate-friendly living. A limited number of key adaptation options were selected to investigate synergy and trade-off potentials. These point towards climate change adaptation options showing high synergetic potentials with regards to the SDGs across all domains (environmental, social, economic). Especially SDG11: Sustainable Cities and Communities, demonstrates high synergetic potential with adaptation options that address infrastructure, such as resilient water management, 'sponge cities', and information and warning systems. These adaptation strategies, however, can also support SDG 4: Quality Education, SDG 5: Gender Equality, SDG 6: Clean Water and Sanitation, SDG 7: Affordable and Clean Energy, SDG 9: Industry, Innovation and Infrastructure, and SDG 15: Life on Land. Resilient water management strategies for example do not only directly affect water management but also aim to make cities and human settlements inclusive, safe, resilient, and sustainable, by mitigating flood risks, enhancing green spaces through blue-green infrastructure, and ensuring a sustainable urban environment that can adapt to climate change challenges (Mikovits et al., 2017; Liu and Jensen, 2018).

### 8.3.2. Costs and potentials of mitigation options

Assessed costs and potentials of mitigation options, depicted in Figure 8.5, highlight that electrification accompanied by decarbonizing electricity supply and demand reduction measures are of key relevance for Austria to achieve carbon neutrality by 2040. Various mitigation options are available within each sector. Among those, electrification in transport, industry, and buildings shows a significant GHG saving potential and needs to go hand in hand with decarbonizing the electricity sector via renewables. While reducing final energy demand and cutting GHG emissions directly are also necessary, their potential has only been partly assessed for Austria.

The electricity sector (Section 4.5) exhibits the highest net emission reduction potential, with five different options ranging from 3.4-7.3 MtCO<sub>2</sub>eq per year at the lower bound, and from 4.8-29.4 MtCO<sub>2</sub>eq per year at the upper bound. Achieving these emission reductions, however, can only be achieved if sector-coupling between electricity and heating and cooling as well as with the transport sector (i.e., e-mobility) is pursued in the forthcoming years. The cost assessment indicates that the two options with the highest mitigation potential, namely wind and ground-mounted PV systems, can be implemented at relatively low costs. Specifically, ground-mounted PV systems can achieve mitigation potentials of 1.6-4.2 MtCO<sub>2</sub>eq per year at costs below the reference level, while additional mitigation potentials of 3.1-8.3 MtCO<sub>2</sub>eq per year are available at below EUR<sub>2023</sub> 50/ tCO<sub>2</sub>eq. Between 1.9-8.3 CO<sub>2</sub>eq per year can be realized at a cost range between EUR<sub>2023</sub> 50 and 200 per tCO<sub>2</sub>eq. Finally, an emission reduction potential of 0.7-2.3 MtCO<sub>2</sub>eq per year is available at more than EUR<sub>2023</sub> 200 per tCO<sub>2</sub>eq. Wind energy, in contrast, presents zero-cost (i.e., lower than reference) mitigation potentials of 1-5.3 MtCO<sub>2</sub>eq per year, with additional potentials of 1.9-10.6 MtCO<sub>2</sub>eq per year being available at costs between EUR<sub>2023</sub> 0-50 per tCO<sub>2</sub>eq. Costs of between EUR<sub>2023</sub> 50-200 per tCO<sub>2</sub>eq are required to realize additional wind potentials of 2.6-10.6 MtCO<sub>2</sub>eq per year. Finally, potentials in the range from 0.6-2.9 MtCO<sub>2</sub>eq are realizable at costs approximately above EUR<sub>2023</sub> 200 per tCO<sub>2</sub>eq. Apart from wind and ground-mounted PV systems, other options for decarbonizing electricity supply include rooftop PV, hydropower and biomethane. Despite offering lower potentials compared to ground-mounted PV and wind, making use of these sources appears of relevance from a system perspective.

The industry sector showcases the second highest potential. Here again five different options were identified, with individual potentials ranging from 1.5-8.5 MtCO<sub>2</sub>eq per year at the lower bound, and 1.7-11.6 MtCO<sub>2</sub>eq per year at the upper bound. Within this sector, steelmaking processes, particularly those involving a switch to methane- (CH<sub>4</sub>) and hydrogen- (H<sub>2</sub>) based direct reduction, show the highest mitigation potentials, ranging from 5.9-11.9 MtCO<sub>2</sub>eq per year for methane- and 4.3-11.6 MtCO<sub>2</sub>eq per year for hydrogen-based reductions. These options are applicable at costs between approximately EUR<sub>2023</sub> 50-200 per tCO<sub>2</sub>eq (CH<sub>4</sub>-based direct reduction) or above EUR<sub>2023</sub> 200 per tCO<sub>2</sub>eq (H<sub>2</sub>-based direct reduction), respectively. Assessed mitigation options for other industry subsectors include fuel and process switching in chemical industries, CCS via oxyfuel technology in clinker for cement production, and high-temperature heat pumps in paper industry. These alternatives demonstrate a similar order of magnitude as those for decarbonizing steel production processes.



**Figure 8.5** This panel illustrates the upper and lower bound of the assessed potentials and related costs of individual mitigation options. These options refer, with respect to expressed potentials, to the time horizon 2040 and beyond (in comparison to the status quo as of 2021), indicating the cumulative mitigation potential until then, whereas related costs reflect the current perspective, including a forward-looking component. Expressed costs represent additional costs in comparison to a conventional (fossil-based) reference option that reflects current practice in the respective field. Underlying data generally relies on literature, partly complemented by expert judgments (for further details, see 8.A.2). The uncertainty and, consequently, large bandwidth applicable in the data on both potentials and costs caused the expression as lower and upper bounds. Furthermore, due to the methodology used to assess costs and potentials, the mitigation potentials of different options should not be added up because of possible interactions between individual options.

The transport sector offers five distinct mitigation options, with individual potentials ranging from 0.8-8.5 MtCO<sub>2</sub>eq per year. According to the assessment, technological improvements in cars and trucks, implying a switch to e-mobility, offer the highest mitigation potentials. Raising fuel taxes and, in turn, phasing out fuel exports represents another promising option offering a comparatively high mitigation potential. Large parts of all previously mentioned potentials come at low costs, some even below zero. Other mitigation options in the transport sector include behavioral adaptation and improvements in freight transport framework conditions. For the building sector, three different mitigation options were assessed, including a switch to renewable energies in both decentral and district heat supply, for example via the use of electric heat pumps, and renovating building envelopes. Individual net emission reduction potentials range from 3.7-6.9 MtCO<sub>2</sub>eq per year at the lower bound, and from 3.7-10.2 MtCO<sub>2</sub>eq per year at the upper bound, respectively. In terms of costs, a major part of these potentials can be mitigated at costs below the reference level, specifically when adopting a more optimistic view on related costs and expenditures (upper bound).

Mitigation options in the sectors of agriculture, forestry and other land use (AFOLU) depend upon future scenarios. Ten different options were assessed, with individual mitigation potentials ranging from 0-7.0 MtCO<sub>2</sub>eq per year at the lower bound, and from 0.1-7.4 MtCO<sub>2</sub>eq per year at the upper bound. The two options with the highest mitigation potentials are: First, a shift towards healthy, plantbased (lancet) diets which, in turn, enables afforestation and renaturation, and; second, reducing overall wood use and harvest. Cost assessments furthermore indicate that mitigating AFOLU emissions can be expensive, with reducing GHG emissions from agricultural soils and carbon sequestration, agricultural inputs, livestock farming or related to the use and harvesting of wood being only available at above EUR<sub>2023</sub> 200 per MtCO<sub>2</sub>eq. Against the background of Austrian mitigation scenarios failing to reach climate targets by 2040, partly due to residual emissions in the agricultural sector (Section 8.4), reducing AFOLU emissions will be important, despite relatively high marginal abatement costs.

The quantitative assessment of mitigation options in terms of their net emission reduction potentials and associated costs focuses only on those for which data is available for the Austrian context. The mitigation options assessed in Figure 8.5 therefore represent a non-exhaustive list. For example, the list lacks certain shift and avoid measures of demand-side sectors, such as reducing floor-space in the building sector.

# 8.4. Assessment of Austrian mitigation scenarios

#### 8.4.1. Overview of Austrian mitigation scenarios

This section provides an overview of existing mitigation scenarios for Austria, assessing relevant scenarios according to key scenario properties. It begins with an overview on the available scenario literature on climate neutrality in Austria, introducing key studies included in the corresponding comparative assessment.

At present there is only a limited set of studies and accompanying scenarios available that take a holistic angle, assessing the transformation towards climate neutrality for the whole Austrian economy. To extend the scenario literature, this assessment also considers scenarios that include the whole energy sector and related industrial activities. Doing so allows for covering the activities responsible for the majority of GHG emissions at present, and for assessing the derived decarbonization scenarios in a comparative manner. While the number of studies and scenarios is larger, our assessment focuses on recently developed key scenarios for which the underlying data has been provided. In addition to these quantitative analyses having taken a holistic angle, there is a large variety of partial scenarios dedicated to specific aspects of the energy system, including the electricity sector and/or other parts of the energy domain. A more limited set of scenarios is available for sectors like industry, agriculture, buildings, and mobility.

To assess Austrian climate targets, including the 2030 targets and the current (as of September 2024) government target of climate neutrality by 2040, and in line with reporting as part of the monitoring mechanism laid down in EU Energy Union legislation, the Environment Agency Austria (EAA) coordinates a scenario process on integrated energy (Umweltbundesamt, 2023d) and emission scenarios for Austria<sup>1</sup> (Umweltbundesamt, 2023d). This process was commissioned and financed by the Austrian Ministry for Climate Protection (BMK). It involved several scientific partners, among others TU Wien, TU Graz, and e-think, and

<sup>&</sup>lt;sup>1</sup> This process is part of the monitoring mechanism laid down in EU Energy Union legislation (Regulation 2018/1999).

a broad range of stakeholders such as representatives from ministries and other governmental bodies, social partners, representatives from industry and the civil society. The EAA uses the macroeconomic-energy-emission model MIO-ES (Kratena and Scharner, 2020) as one of the main analytical tools for all its scenarios. This model is based on a hybrid IO-structure and is constantly developed and maintained by the EAA.

The monitoring mechanism conducted by the EAA for the EU and the BMK includes the construction of a With Existing Measures (WEM), a With Additional Measures (WAM), and a Transition scenario (Umweltbundesamt, 2023c). A WEM scenario (Umweltbundesamt, 2023e) is a baseline or Business-As-Usual (BAU) scenario that depicts the continuation of the status quo of economic, energy and emission trajectories, and includes all policies that are part of current government legislation. A WAM scenario (Umweltbundesamt, 2023c), in contrast, describes a more ambitious scenario, where additional expert-defined climate policy measures are modeled that are likely to be included in governmental action plans. The WAM scenario typically serves as the reference for the Austrian National Climate and Energy Plan (NECP). A Transition scenario, such as the Austrian Transition 2040 (Umweltbundesamt, 2024), shows options for how climate targets can be reached under coherent, consistent, and feasible assumptions. All changes in emissions in this scenario are linked to various policy measures developed collaboratively with a broad range of stakeholders and aligned with existing legal frameworks.

The scenarios by the EAA serve as Austrian reference scenarios for two main reasons: (1) They are elaborated by a large number of experts from various institutions in collaboration and close coordination with above-mentioned key stakeholders in Austria, including input from participatory frameworks such as the Austrian Climate Council ('Klimarat') in case of the Transition 2040 scenario (Umweltbundesamt, 2024). (2) WEM and WAM scenarios are integral to the official monitoring mechanism of the EU Energy Union. The assessment of GHG emission includes two variants for WEM and WAM<sup>2</sup>: First, the 2023 variants, serving as basis for the detailed comparative analysis between WEM, WAM and Transition (Umweltbundesamt, 2024), and; second, the 2024 updates, which reflect the latest progress on achieved GHG emission reductions (as of 2022) and serving as basis for the latest available update of the Austrian NECP (as of 2024) (BMK, 2024). Additionally, a collaboration between EAA and the Austrian Ministry of Finance (BMF) resulted in holistic scenarios analogous to the WEM and the Transition 2040 scenarios, referred to as 'Base' and 'Activity' scenario. These scenarios were developed as part of a pilot project to integrate climate policy and emission scenarios into the long-term budget forecast (Gugele et al., 2022). The scenarios 'Transition 2040' for the BMK (Umweltbundesamt, 2024) and the 'Activity' scenario for the BMF (Gugele et al., 2022) aim to demonstrate how climate neutrality by 2040 could be targeted (but not necessarily reached), considering integrated economic, energy, and emission pathways. However, due to their overlap with the aforementioned EAA scenarios and a lack of accessible data, these scenarios are not considered in the subsequent comparative scenario assessment.

The recent ACRP project NetZero20403 (Schmidt et al., 2025) - a collaboration between the Austrian Energy Agency (AEA), BOKU and IIASA - aims to establish a set of comprehensive and consistent alternative emission pathways that can ensure reaching Austria's 2040 climate target. It considers the energy system, energy imports and energy demand by the agricultural, household, industry, service and transport sector. The four scenarios developed can be distinguished according to the degree of energy imports and total energy requirements: (1) Scenario A describes a combination of energy supply and demand measures that lead towards energy sufficient lifestyles and a rapid expansion in renewable energy sources; (2) Scenario B, in contrast, outlines a world characterized by a continuous increase in energy demand and international energy treaties enabling the import of carbon-neutral energy; (3) In Scenario C, the rapid expansion of domestic renewable energy sources leads to a situation of relative energy independence, while resource and energy consumption continuously increase due to rebound effects; (4) Finally, Scenario D is characterized by a shift towards energy-sufficient behavioral practices and a lack of renewable energy expansion being offset by increased levels of energy imports. While all scenarios achieve the goal of a climate neutral energy system in 2040, they do not address non-energy-related emissions from agriculture and Land Use, Land Use-Change and Forestry (LULUCF).

<sup>&</sup>lt;sup>2</sup> Accessible detailed data for the latest (2024) update of WEM and WAM is limited to GHG emission balances. Thus, these variants could not be considered for energy-related comparisons or the assessment of economic impacts.

netzero2040.at/szenarien-1
Four additional scenarios were modelled by the ACRP Project INTEGRATE<sup>4</sup> (Steininger et al., 2024) to develop cross-sectoral integrated pathways that achieve the 2040 goal of climate neutrality including the building, transport, energy and industry sector. Two high demand scenarios are used as reference differentiating between limited and unlimited carbon neutral energy imports. Additionally, two INTE-GRATE scenarios are characterized by low energy demand, with differences between them being based on the availability of energy imports. INTEGRATE furthermore calculates the overall economic effects, such as GDP, income, sectoral effects and distributional effects. However, due to the lack of available data, the economic effects of INTEGRATE scenarios could not be integrated in the cross-scenario assessment.

# 8.4.2. Assessment of Austrian mitigation scenarios: GHG emissions and energy balance

# Total GHG emissions

At present, only the EAA scenarios can be classified as holistic, including energy and total GHG emission projections for LULUCF. These scenarios comprehensively cover energy-related emission sectors – i.e., energy, buildings, mobility, and industry – according to a consistent model-based methodology using the MIO-ES model (Kratena and Scharner, 2020). However, emission projections for agriculture and processes are usually based on different models, studies or expert judgements.

To put EAA scenarios on the future development of total GHG emissions in Austria into perspective, Figure 8.6a includes the historic development from 1990 in addition to future trends.<sup>5</sup> As visualized, total GHG emissions increased in the early years until 2005, mainly related to a strong increase in transport-related, and a comparatively moderate rise in industry-related activities and emissions. Since 2005, a reduction of GHG emissions could be achieved, with a reported 11 % decline in 2022 compared to 2005 (Umweltbundesamt, 2023b). Mainly responsible for the downward trend are the uptake of renewables in energy supply and energy efficiency improvements in buildings, leading to a decline in GHG emissions in energy supply and demand overall. Taking a look into the future, the WEM 2023 and 2024 scenarios (Umweltbundesamt, 2023e) include all mitigation and climate-related policy measures implemented before January 2022. The 2023 and 2024 variants of WEM as well as WAM differ in terms of their starting point for the prospective modeling. While the 2023 variants build on 2021 statistics, the 2024 variants acknowledge the recently achieved progress (as of 2022) of GHG emission reductions. Due to WEM's nature as a business-as-usual scenario, relying on currently implemented policies, its mitigation efforts are limited: The WEM 2023 (2024) scenario predicts a decrease in total GHG emissions by 32 (34) % by 2050 relative to 2005, reaching GHG emissions of 52.4 (50.6) MtCO<sub>2</sub>eq in 2050. In comparison, the WAM 2023 (2024) scenarios project a reduction of 32 (34) % already by 2030 (relative to 2005), indicating that planned measures would be effective in reducing GHG emissions also in the near future. By 2050, WAM 2023 (2024) achieves further emission reductions by 69 (71) % relative to 2005 to reach a level of 23.6 (22.4) MtCO<sub>2</sub>eq. The EAA Transition 2040 scenario, in contrast, imposes more radical mitigation measures than currently adopted or planned, thus achieving stronger GHG reductions both in the short (2030) and long term (2040 and beyond) (Umweltbundesamt, 2023c, 2023e). Overall, total GHG emission could then be reduced by 54 and 92 % by 2050 compared to 2005 levels. Although significant, the Transition 2040 scenario does not achieve the emission reductions required to meet the Austrian policy target of achieving carbon neutrality in 2040 due to residual emissions of 11.0 MtCO<sub>2</sub>eq that originate mainly from hard-to-abate industries and agriculture. Negative emission potentials from land-based carbon dioxide removal (CDR) related to LULUCF (-4.6 MtCO2eq) do not suffice to coun-

<sup>&</sup>lt;sup>4</sup> wegcwp.uni-graz.at/integrate/aims/

An update of the Austrian greenhouse gas inventory was published in January 2025 (Umweltbundesamt, 2025), after the literature cut-off deadline of this report. Due to a modification of the accounting method for GHG emissions related to land use, land-use change and forestry (LULUCF), historical emissions were altered significantly. This change was not considered in this chapter, because all prospective scenarios assessed in this section were based on former historic GHG inventories and underlying data.

For comparative reasons, Figure 8.6a includes net emissions according to the latest Austrian GHG inventory as published in January 2025 (Umweltbundesamt, 2025) – cf. the dotted black line in Figure 8.6a indicating past total GHG emissions until 2020. Latest data for 2023 (not shown in this panel) shows that the LULUCF sector has turned from an emission sink to a

source in recent years. The change in methodology causes an increase of total GHG emission including LULUCF in 2020 by 4.3 MtCO<sub>2</sub>eq compared to the historical data according to the previous GHG inventory (Umweltbundesamt, 2023a) which was used to develop EAA's Transition scenario.



**Figure 8.6** (a) Breakdown of historic and future total GHG emissions trends in Austria according to EAA scenarios, with sectoral details for the EAA Transition 2040 scenario (left); (b) Comparison of energy-related CO<sub>2</sub> emission trends according to all assessed scenarios (right).

terbalance these residual emissions, requiring additional measures like Carbon Capture, Utilization, and Storage (CCUS) to counterbalance residual emissions, as no other technological mitigation options exist yet.

In the context of the 2030 climate policy targets imposed by EU regulation, the Effort Sharing Regulation (ESR) in accordance with the recent amendment as of 2023 (Regulation (EU) 2023/857), sets Austria a 48 % GHG emission reduction target (compared to 2005) for all non-ETS sectors<sup>6</sup> by 2030, with the transport and building sector being the largest ESR emitters at present. Predicted emission reductions by 38 (42) % in 2030 under the WAM 2023 (2024) scenario suggest that the strengthened ESR target will not be met. The Transition 2040 scenario, in contrast, predicts GHG emission reductions of 59 % by 2030, indicating that Austria could meet the legally binding EU target if more ambitious and far-reaching climate mitigation measures are implemented immediately.

# Energy-related CO<sub>2</sub> emissions

The energy sector is responsible for more than two thirds of Austria's total GHG emissions at present, with varying contributions from year to year<sup>7</sup>. Since not all assessed studies report on GHG emissions, energy-related CO<sub>2</sub> emissions serve as common basis for comparing the decarbonization ambition and underlying trajectory of assessed scenarios and studies. Figure 8.6b shows the future development of total energy-related CO<sub>2</sub> emissions for all assessed scenarios, including estimates the INTEGRATE and NetZero2040 in addition to EAA scenarios. According to these projections, minor differences appear in the emission 'endpoints' by 2040 or 2050. NetZero2040 scenarios are the most ambitious, reducing  $CO_2$  emissions to 0.3 Mt already by 2040. INTEGRATE scenarios, in contrast, predict 0.7 MtCO<sub>2</sub> by 2050. The EAA scenario Transition 2040 shows a CO<sub>2</sub> emission reduction to 2.8 Mt by 2040, further declining to 2.2 Mt in 2050. Apart from differences in terms of endpoints and underlying mitigation ambition, there are both differences and similarities in the starting points and underlying trajectories. All studies and the corresponding scenarios acknowledge the upward trend in emissions from 2020-2021 as consequence of the global disturbances in economic activities during the COVID pandemic. For all scenarios and studies with strong GHG mitigation (i.e., Transition 2040, NetZero2040 and INTEGRATE), CO<sub>2</sub> emission show a downward trend towards 2030 starting in 2021, aligning with latest statistics for 2022. Beyond 2030, NetZero2040 scenarios maintain a strong GHG mitigation ambition until 2035, thereafter slowing down their pace for the final years towards 2040. The Transition 2040 scenario follows a similar trend but maintains a more constant GHG reduction path between 2030 and 2040. INTEGRATE scenarios do not provide 2040 data but demonstrate the achievement of climate neutrality by 2050. However, as stated above, none of the assessed scenarios demonstrates how climate neutrality could be achieved in Austria's entire economy by 2040 and beyond.

<sup>&</sup>lt;sup>6</sup> ETS sectors are already under direct control by EU climate policy.

<sup>&</sup>lt;sup>7</sup> For example, in 2021 energy-related  $CO_2$  emissions held a share of 76 % on Austria's total GHG emissions whereas in 2022 that share declined to 69 %.



Figure 8.7 (a) Comparison of gross domestic consumption trends according to all assessed scenarios (left). Note that NetZero2040 and INTEGRATE scenarios include non-energetic use of fossil fuels; (b) Breakdown of net imports by fuel in 2040 according to all assessed scenarios (right).

# **Energy balance**

This subsection provides an overview of trends in overall energy demand and supply across the assessed scenario landscape. Gross domestic consumption acts as the starting point for the comparative assessment, followed by a discussion of trends in final energy demand.

#### Gross domestic consumption and the role of imports

Gross domestic consumption shows a downward trend in all scenarios aiming for decarbonization, as depicted in Figure 8.7a.<sup>8</sup> By 2040, reductions in gross domestic consumption ranges between 25–40 % compared to 2021, with only slight differences among the corresponding scenarios. In contrast, the WEM 2023 and WAM 2023 scenarios, reflecting existing and planned policy measures, indicate different trajectories: In WAM 2023, demand decreases by 12 % by 2040 (compared to 2021), whereas WEM 2023 predicts that demand initially increases until 2030 and then decreases, resulting in an overall demand reduction of only 4 % by 2040 (compared to 2021).

In addition to reductions in energy demand, important changes are expected on the supply side. Decarbonization implies an increase in renewable energies and, in turn, a phase-out of fossil fuels (Luderer et al., 2022; Umweltbundesamt, 2023c). This consequently affects the import dependency of Austria's energy supply. According to assessed literature (e.g., European Commission and Joint Research Centre (JRC), 2021) (Section 4.5.3), renewable energy sources like wind and solar PV or ambient heat (heat pumps) are economically viable and available at large scale domestically. They can furthermore be complemented by historically well-established renewable sources like biomass and hydropower for Austria to meet its respective energy needs, especially in the electricity sector and for heating and cooling. However, renewable gases like hydrogen and renewable fuels of non-biologic origin (RFNBO) may partly be imported due to competitive cost advantages abroad (Umweltbundesamt, 2023c; Steininger et al., 2024; Schmidt et al., 2025). In this context, Figure 8.7b compares net imports by fuel in absolute terms in 2040 and shows Austria's import dependency in relative terms. Today, Austria's energy import dependency (i.e., total amount of energy imports as share of gross domestic consumption) varies between 52 % (2020) and 75 % (2022) (BMK, 2023), with differences caused by stock management practices as well as weather-driven changes in domestic demand and supply. Most of these imports are fossil fuels like oil, gas, and coal, with a smaller share being attributed to electricity, where Austria is net importer at present. There is strong agreement in the assessed scenario literature about a decline in import dependency and a change in the underlying fuel mix. Thus, the available scenario literature agrees upon fossil fuel imports being largely replaced by hydrogen and other RFNBO by 2040. In addition, Austria is expected to become a net exporter of electricity or at least achieve an even balance, driven by the strong uptake of wind and solar PV. However, the extent of this change in import dependency varies among assessed decarbonization scenarios. By 2040, the Transition 2040 scenario projects the lowest import dependency (e.g., 8 %

<sup>&</sup>lt;sup>8</sup> Due to a lack of INTEGRATE data on gross domestic consumption, they have been excluded from this comparison.



Figure 8.8 Comparison of final energy demand trends according to all assessed scenarios.

in relation to gross domestic consumption), while the Net-Zero2040 scenarios show higher import ratios, ranging from 24–36 %, depending on the underlying assumptions and the respective scenario conception (low or high imports).

#### Final energy demand

Similar to gross domestic consumption, the scenario literature agrees that reducing final energy demand is essential for effectively combating climate change (Figure 8.8). There are, however, strong differences in the amount of degrowth required across analyzed decarbonization scenarios. By 2040, reductions in final energy demand relative to 2021 levels range widely, with -8 % in the INTEGRATE high demand scenarios and -36 % in the Transition 2040 scenario. NetZero2040 scenarios fall within the expressed degrowth corridor, with differences being driven by varying scenario conceptions and underlying assumptions (e.g., high versus low demand scenarios). These scenarios also show distinct trends, with a comparatively strong decline in demand until 2030, followed by a demand increase in the period after 2030. These differences reflect variations in the emphasis placed on and the mix of underlying demand-side policy measures. The Transition 2040 scenario, for instance, places strong emphasis on demand-side measures as driver for final energy demand reductions.

Complementary to the dynamic evolution of final energy demand, Figure 8.9 explores the underlying decomposition of final energy demand for 2040 by sector (Figure 8.9a) as well as by fuel (Figure 8.9b). The sectoral breakdown provides an indication of the emphasis on demand-side measures and highlights underlying macroeconomic and technological trends:

- Transport: The assessed scenario literature outlines that final energy demand in the transport sector will decline as decarbonization progresses. The observed (de-)growth rates from 2021 to 2040 vary between -35 % (INTEGRATE high demand scenarios) and -73 % (NetZero2040 low demand scenarios), with EAA's Transition 2040 scenario achieving a 59 % decrease. Key drivers are technological improvements (e.g., switch to e-mobility) and anticipated behavioral changes, such as increased use of public transport and active climate-friendly mobility.
- Industry: There is no consensus in the available scenario literature regarding the future development of industrial final energy demand. While EAA's Transition 2040 scenario assumes a demand decline by 16 % by 2040 relative to 2021, the INTEGRATE scenarios project a tremendous increase of 53–66 % but their modeling also builds on a significantly higher industrial demand for the status quo (i.e., more than 40 % above data reported in Statistik Austria) (Luderer et al., 2022). The NetZero2040 scenarios also assume an increase in industrial final energy demand but at a comparatively smaller rate, ranging from 8–46 %.
- Other sectors: Final energy demand of other sectors including residential, service and agriculture or, according to a different classification, buildings and (electrical) appliances – is expected to decrease according to the assessed literature. Scenarios differ, however, in terms of the extent of decline. Growth rates, comparing 2040 with 2021, vary between -1 % (NetZero high demand scenarios) and -56 % (INTEGRATE low demand scenarios), while the Transition 2040 assumes demand reductions of 30 %, lying in between the two extremes.

In addition to the sectoral breakdown of 2040 final energy demand, the decomposition by fuel reveals important technological trends:

- First, the assessed literature agrees that **phasing-out fos**sil fuels represents a prerequisite for climate neutrality. However, INTEGRATE scenarios retain oil and natural gas usage by 2040, reflecting their goal of achieving climate neutrality ten years later than other studies, namely by 2050 instead of 2040.
- Second, electrification is a predominant decarbonization strategy across all scenarios, helping to decarbonize fossil-fuel intensive sectors like transport, buildings or industry. Electricity demand will consequently grow



Figure 8.9 (a) Breakdown of final energy demand by sector in 2040 according to all assessed scenarios (left); (b) Breakdown of final energy demand by fuel in 2040 according to all assessed scenarios (right).

between 2021 and 2040, varying between 29 % (NetZero2040 low-demand/low-import scenario) and 76 % (IN-TEGRATE high-demand scenario), whereas the Transition 2040 scenario expects electricity demand to grow by 47 %.

- Third, green hydrogen and RFNBO are key pillars for decarbonizing Austria's energy sector and economy. The extent to which this - from today's perspective - expectably costly option (Luderer et al., 2022) is taken up differs across the assessed scenarios. Since the use of these fuel options is not yet visible in current energy statistics (as of 2021), no growth rate for the corresponding 2040 use pattern can be expressed. Instead, the EAA's WAM scenario serves as an alternative anchor point for the scenario comparison. The Transition 2040 scenario expects hydrogen and RFNBO use to double compared to WAM levels. An even higher uptake is observable in the INTEGRATE scenarios, where 3 to almost 6 times higher amounts are used in 2040. NetZeto2040 shows the strongest uptake of hydrogen and RFNBO, ranging from a 7to a 10-fold increase compared to WAM.
- Fourth, biomass, in various forms (solid, liquid and gaseous), serves as another pillar for a decarbonized energy system, especially for uses that cannot be electrified through abiotic renewable energies and for balancing the volatility of solar and wind. The use of biomass for energy, however, induces trade-offs, such as reducing carbon sequestration in ecosystems and in societal structures and affecting biodiversity (Sections 2.3.2, 4.5.3). In the Transition 2040 scenario, the total direct use of biomass remains stable between 2021 and 2040, although its applications shift from heat fuels in buildings towards

fuel- and material-use in industry (which is however not accounted in the energy balance) (Umweltbundesamt, 2024). Other scenarios indicate either comparatively small increases of up to 23 % (INTEGRATE high demand scenarios), or slight decreases of 17 % (NetZero2040 low demand scenarios) when comparing 2040 results with the status quo (2021).

 Fifth, district heating (and cooling) is another key technology option for decarbonizing heating and cooling supply in urban or peri-urban areas. The Transition 2040 scenario assumes a 13 % increase by 2040 relative to 2021. The NetZero2040 scenarios align with this trend, predicting increases in district heat demand between 15 % to 30 %. Contrarily, the INTEGRATE scenarios predict a decline in district heat demand of 17 % to 32 % when comparing 2040 with the status quo (2021). Note that this can be partly explained by INTEGRATE reporting a lower uptake of district heat for the status quo than other scenarios.

### 8.4.3. Spotlight on key energy carriers

Complementary to the above, this section examines two key energy carriers/sectors that are crucial for achieving climate neutrality.

# Electricity sector

According to existing literature, the electricity sector will become the key energy sector in Austria, as well as at the European and global scale (Del Granado et al., 2020; Resch et al., 2022). As sector coupling and efforts to decarbonize electricity supply continue, electrification may help in de-



Figure 8.10 Comparison of final electricity demand trends according to all assessed scenarios.

carbonizing fossil-fuel intensive sectors such as transport, buildings and industry. Consequently, there is a strong consensus across the assessed scenarios that electricity demand will grow in the forthcoming years. This trend is observable in Figure 8.10, which illustrates the development of final electricity demand in Austria up to 2040. There are, however, differences in projected demand growth across the assessed scenarios, partly driven by varying scenario conceptions and corresponding assumptions. The comparison of expected demand increases between 2021 and 2040 reveals a wide range, with projections varying between 29 % (NetZero2040 low-demand/low-import scenario) and 76 % (INTEGRATE high-demand scenario). In addition to the evolution of final electricity demand, Figure 8.11 illustrates the underlying decomposition of demand and supply patterns in the assessed scenarios.<sup>9</sup> More precisely, Figure 8.11a provides a breakdown of gross electricity demand in 2040 by sector or category, while Figure 8.11b illustrates the composition of domestic electricity supply in 2040, indicating fuel types for scenarios that provide this level of detail.

Key trends on the demand side include the following: First, the projected uptake of e-mobility increases the electricity demand in the transport sector significantly, with projections rising from currently (2021) 12 PJ to 56-90 PJ in 2040. Second, depending on the scenario, the industry sector will become either the largest or second largest consumer of electricity. For the demand category 'others', encompassing the residential, service and agriculture sector, no clear trend is observable. Demand in these sectors may either decline by 9 PJ (Transition 2040 scenario) or increase by 10-21 PJ (NetZero2040 scenarios) by 2040 relative to 2021. Additionally, there is consensus that a new demand category may emerge, with renewable electricity being used for producing green hydrogen domestically in the range from 24-45 PJ (NetZero2040 scenarios) to 57 PJ (Transition 2040 scenario). No clear trend can be identified from the scenario comparison for losses and own electricity consumption within the energy sector. There is, however, a broad set of complementary sector-specific literature available that indi-

<sup>&</sup>lt;sup>9</sup> Due to a lack of corresponding data, INTEGRATE scenarios were excluded from this comparison.



Comparison of electricity supply by fuel in 2040 according to assessed scenarios



Figure 8.11 (a) Breakdown of gross electricity demand by sector in 2040 according to assessed scenarios (left); (b) Breakdown of domestic electricity supply by fuel in 2040 according to assessed scenarios (right).



**Figure 8.12** Comparison of final  $H_2$  and RFNBO demand trends according to all assessed scenarios. Note that the INTEGRATE low-demand/(un-)limited-imports scenarios project the same final  $H_2$  and RFNBO demand as the WAM 2023 scenario.

cates increases both in transport losses and own consumption because of ongoing electrification (Section 4.5.2).

On the supply side, the scenarios assume a strong uptake of renewable energies in the period up to 2040 (Figure 8.11b). Complementary to hydropower, which is Austria's traditional renewable source for electricity supply, wind and solar PV are expected to act as key pillars for meeting growing electricity demand. This is supported by the available literature, although there are significant differences regarding the scale of wind and solar PV deployment by 2040. The strong uptake of weather-dependent renewable energies and preferred technology selection influence infrastructural needs (e.g., grids, storages) and required demand-side solutions. A detailed discussion of these aspects is provided by Chapter 4 (Section 4.5).

### Hydrogen and renewable synthetic fuels

Green hydrogen and renewable synthetic fuels are another key solution for Austria's efforts to decarbonize its energy sector and the whole economy. However, as stated above, the available scenario literature provides no clear answer on the required amounts of these – from today's perspective – relatively costly options (Luderer et al., 2022). Figure 8.12 illustrates the projected final demand for H<sub>2</sub> and RFNBO up to 2040 across the assessed scenarios. At the lower end, the Transition 2040 scenario projects moderate levels of final energy demand for hydrogen and RFNBO of 59 PJ by 2040. At the upper end, the NetZero2040 high-import/high-demand scenario shows significantly higher final demand of 267 PJ by 2040, with other NetZero2040 scenarios approaching but not reaching this level of demand.

Figure 8.13 illustrates the decomposition of demand and supply patterns for the assessed scenario literature. More precisely, Figure 8.13a displays the sectoral breakdown of the overall demand for  $H_2$  and RFNBO in 2040, including – in contrast to Figure 8.12 – the transformation input to electricity and district heating and cooling. Complementing this, Figure 8.13b demonstrates how this demand can be met, either via domestic production or imports.

On the demand side, the available scenario literature suggests that the industry and the transport sector will require



Figure 8.13 (a) Breakdown of  $H_2$  and RFNBO demand by sector in 2040 according to assessed scenarios (left); (b) Breakdown of  $H_2$  and RFNBO supply in 2040 according to assessed scenarios (right).

hydrogen and/or RFNBO for decarbonization. By 2040, industry demand is expected to be two to three times higher compared to that in transport. However, the projected overall demand for hydrogen and RFNBO varies significantly across scenarios, with NetZero2040 scenarios indicating a three to four times higher demand (236-322 PJ), driven primarily by industry and transport, in comparison to the Transition 2040 scenario (106 PJ). Discrepancies among available literature are furthermore observable with respect to the demand category 'others', which includes the residential, service and agriculture sector. While the Transition 2040 scenario does not foresee any uptake of hydrogen or RFNBO, the NetZero2040 scenarios project demand increases comparable to that of the transport sector. They agree, however, regarding the use of hydrogen for the electricity sector and district heat supply, especially for peak loads during times of low renewable infeed.

On the supply side, assessed scenarios agree that imports will meet most of the hydrogen and RFNBO demand, though the extent of imports required varies among scenarios, corresponding to the overall projected demand for types of fuels.

#### 8.4.4. Economic impacts

This section assesses and compares existing holistic scenarios for Austria that include macroeconomic effects in terms of their economic impacts. In general, there is limited Austrian-specific literature that combines holistic energetic-emission scenario analysis with the assessment of economic impacts of scenario measures (Gugele et al., 2022; Umweltbundesamt, 2023c, 2023e, 2024).

# The EAA's WEM, WAM, Transition 2040 scenarios for the Austrian Ministry of Climate Protection (BMK) and EU monitoring mechanism

In order to estimate the macroeconomic consequences of EAA's mitigation scenarios (Umweltbundesamt, 2023c, 2023e, 2024) in terms of employment, value creation, private consumption and income distribution, the MIO-ES model (Kratena and Scharner, 2020) was employed. The results of the WAM 2023 scenario are compared with those of the Transition 2040 scenario. The WEM 2023 scenario serves as the reference scenario or base scenario for both cases. The



Figure 8.14 Changes in macroeconomic variables in the WAM 2023 and Transition 2040 scenarios compared to the WEM 2023 reference scenario (Umweltbundesamt, 2023c).

scenarios are based on assumptions regarding energy and  $CO_2$  prices, investments, and sectoral measures.

In all three scenarios, employment and value creation increases between 2023 and 2050. However, investment requirements for the climate-relevant measures included in the WAM 2023 and Transition 2040 scenario result in higher economic growth rates than in the WEM 2023 scenario (Figure 8.14). The WAM 2023 scenario indicates that, on average, approximately 37,000 additional full-time equivalent jobs will be created per year between 2023 and 2040 compared to the baseline WEM 2023. Transition 2040, on the other hand, suggests that there will be around 50,000 additional full-time equivalent jobs created per year over the same period. Thus, average employment figures for the period 2023-2040 are 0.85 % (WAM 2023 scenario) and 1.15 % (Transition 2040 scenario) higher than in the WEM 2023 scenario. The average unemployment rate in the WEM 2023 scenario is 4.9 % between 2023-2040, while it is slightly lower in the WAM 2023 scenario at 4.3 % and in the Transition 2040 scenario at 4.2 %. WAM 2023 and Transition 2040 furthermore outperform the WEM scenario in terms of value added.

The drivers of value creation and employment in the scenario WAM 2023 and Transition 2040 are investments in the areas of energy supply, transport, buildings, and industry that are included in the modeled sectoral measures. On average, the investment level is 4.8 % higher in the WAM 2023 scenario and 6.4 % higher in the Transition scenario than in the WEM 2023 reference scenario. As a result of the additional production and employment stimulated by the investments, private consumption also increases.

Considering industry-level effects, the additional investments in the WAM 2023 and Transition 2040 scenario yield positive outcomes with respect to value creation and employment, among others in the construction sector and the economic sectors upstream. These sectors are stimulated due to the infrastructure expansion of public transport, cycling infrastructure, the building sector (renovation and boiler replacement) as well as renewable electricity generation, electricity storage and transmission. Notably, construction is a labor-intensive industry, requiring more workers per unit of production unit compared to other industries, which further supports domestic jobs creation.

In terms of the distributional effects, the impact assessment presents a positive outlook, indicating that the disposable household income of all income groups remains above the levels projected in the baseline WEM 2023 scenario well into the 2030s (Figure 8.14). Thereby, disposable income growth is more pronounced among the lower-income than among upper-income quintiles. This can be attributed to low-income earners deriving greater benefits from the increase in employment than their more affluent counterparts. Additionally, reimbursement measures like the climate bonus ('Klimabonus'), which slightly benefits lower-income households, and income-based subsidies, such as the government's 'Clean Heating for All' initiative to address energy poverty, have a dampening effect because they reduce the energy costs of low-income households. From the mid-2030s onwards, electricity prices in the Transition 2040 scenario are projected to be approximately 10 % higher than in the WEM 2023 and WAM 2023 scenario. This increase may place added financial pressure on the low-income households. However, in line with the principles of a Just Transition, targeted subsidies are provided to compensate negative side effects of the transformation for the lower-income groups.



Figure 8.15 Changes in disposable income by quintile in the WAM 2023 and Transition scenarios compared to the WEM 2023 scenario (Umweltbundesamt, 2023c).

Base and activity scenarios for long term budget forecast for the Austrian Ministry of Finance (EAA-BMF scenarios)

The economic impacts of the WEM 2023, WAM 2023, Transition 2040 (Section 8.4.4) scenarios are comparable to those of the scenarios conducted by the EAA for the Austrian Ministry of Finance (EAA-BMF scenarios) (Gugele et al., 2022). In the latter scenarios, the economy in the Activity scenario demonstrates greater annual and cumulative growth than in the Base scenario. The additional cumulative value added in the Activity scenario amounts to approximately 1.3 % in 2050, compared to the Base scenario. In addition to investment requirements, the expansion of renewable energy infrastructure in Austria reduces the import requirements for fossil energy sources, which lowers expenditures on fossil energy imports. Finally, government subsidies, some of which are distributed directly to households as transfers and some of which trigger investments in climate protection measures, have a direct and indirect growth-promoting effect.

Figure 8.16 shows the cumulative macroeconomic effects (the levels of variables in the two scenarios are compared over the simulation horizon) as compared between Base and Activity scenario. In the Activity scenario, the investments required for the transformation increase by approximately 8 % compared to the Base scenario until 2035, which represents the peak of investments in decarbonization. Thereafter, investments stabilize at a higher level. From 2040 onwards, the combination of low inflation and lower fossil imports, along with the expansion of renewable energy sources, contributes to economic growth despite a decline in investments. The positive growth effects, reduced import costs for fossil fuels and new taxes, such as increased CO<sub>2</sub>-taxes compensating for falling tax revenues from fossil sources, result in an overall government deficit and debt-to-GDP ratio that remains almost the same in the Activity scenario as compared to the Base scenario (Gugele et al., 2022).

It should be emphasized that GDP growth effects are contingent upon the rapid expansion of renewable electricity generation capacity. In particular, the favorable development of electricity prices from 2040 onwards, determined by model assumptions, represents as significant driving force for additional economic growth. Given that electricity will be by far the most important energy source in Austria from 2040 onwards, electricity prices are expected to have a major impact on the national economy. In addition, two areas warrant particular attention with regard to investment. On the one hand, investments in the expansion of renewable electricity production. Initially, this entails the implementation of the Austrian directive on the expansion of renewable electricity ('Erneuerbaren Ausbau Gesetz'). Thereafter, investments must be maintained at a high level or increased until 2040. From 2040 onwards, investment in renewable electricity production declines as the economy has largely converted to a primarily electricity-based energy supply. On the other hand, the Activity scenario highlights investments in the building sector (implementation of the 'Heating Fu-



Figure 8.16 Overview of macroeconomic effects – cumulative change in the Activity scenario compared to the Base scenario (Gugele et al., 2022).

Emissions sector	Total investments (mean)	Additional investments (lower bound)	Additional investments (upper bound)	Additional annual investments (lower bound)	Additional annual investments (upper bound)
Industrial production	21.4	8.4	15.0	0.5	1.0
Energy production	160.5	37.4	80.7	2.3	5.1
Transport/mobility	323.0	43.8	63.5	2.7	4.0
Buildings	360.6	10.0	14.6	0.7	0.9
Total	865.6	102.5	173.7	6.2	10.9

Table 8.1 Total and additional investments in the Austrian economy (transition scenario, 2024–2040/2050; EUR<sub>2023</sub> billion) (Weyerstraß et al., 2024).

ture for Buildings 2050' scenario), which will be highest in the first half of the 2030s. Thereafter, investment declines from 2035 (exit from oil heating systems completed) and 2040 (exit from gas heating systems completed). After 2040, investments remain for building renovations.

In 2040, the levels of employment and private consumption in the Activity scenario are about 1 % below the Base scenario. This can be attributed to the interaction of several factors. First, the relatively stronger increase in energy prices in the Activity scenario (CO<sub>2</sub> price, electricity prices) induces higher wages, which moderately retard employment growth due to labor being substituted by capital in the production process. Second, higher energy prices dampen private consumption, especially among low-income deciles. Third, the high capital intensity of the work required for the transformation contributes to a shift in the production structure from labor to capital. While some of the measures, such as the thermal refurbishment of buildings, have a high share of labor input, others demonstrate high capital input requirements (e.g., the expansion of wind turbines for renewable electricity production). Rising electricity prices are a further important driver of consumption prices. This can be attributed to wage-price spirals implemented in MIO-ES, whereby all energy prices also co-determine wages and thus employment as well as consumption.

# Investment needs

The most recent estimation of additional investment requirements for the Transition 2040 scenario that almost achieves climate neutrality by 2040 differentiates between total investments and additional investments. The sectoral coverage encompasses industrial production (in particular, the steel, cement and chemical industries), energy production (wind, PV, biomass, hydrogen, energy networks [electricity, hydrogen]), buildings (energetic-thermal retrofitting), and transportation (public transport, E-mobility). Additional investments are defined as investments in addition to the usual sector-specific investment cycles. The investment in an electric blast furnace, for instance, would only be considered additional if the investment costs exceed that of a coal-fired blast furnace. In contrast, if the investment were cost-neutral and aligned with the investment cycle of the steel-producing company (i.e., if a new blast furnace were necessary regardless of climate policies), additional investment costs would be minimal and potentially negative if carbon-neutral investments were more cost-effective than fossil ones.

Table 8.1 provides a summary of total and additional investments for the Transition 2040 scenario for the period 2022–2050. Further details are documented in Weyerstraß et al. (2024).

The estimates displayed in Table 8.1 are based on a broad range of assumptions, including those pertaining to investment cycles, carbon prices, technological planning and choices. Furthermore, some investments extend into the period of 2040–2050 (e.g., industrial production). Investments and policies in other sectors (e.g., land use, agriculture, forestry) are not included in the table. It is noteworthy that additional investments for climate neutrality in 2040, according to the underlying scenario, amount to approximately 14–17 % of total investments.

In relation to GDP, additional investment requirements amount to approximately to EUR<sub>2023</sub> 6.2 to 10.9 billion (roughly 1.1 to 1.9 % of GDP in relation to cumulative forecast GDP for 2024 to 2040).While only one recent Austrian estimation study is available on the total and additional investments necessary to achieve emission reductions outlined in the Transition 2040 scenario, the range of the GDP share of investments is broadly in the same order of magnitude as that of similar Austrian studies (e.g., Stern, 2007; Miess et al., 2022; Bröthaler et al., 2023). One of the most recent international studies on additional investments for decarbonization in Europe, published by Institut Rousseau (2024), estimates that additional annual investments amount to approximately 2.3 % of European annual GDP until 2050. It should be noted that an important prerequisite for all estimates is that current fossil investments must be shifted towards decarbonization.

# 8.4.5. Scenario assessment with regard to sustainable development and available carbon budgets

# Austrian mitigation scenarios and the Sustainable Development Goals

This section assesses and compares Austrian mitigation scenarios based on their contribution to the SDGs. Due to the fact that only two scenarios include mitigation options beyond the energy system (Section 8.4.2), analysis is limited to energy-related supply and demand options. Due to data limitations regarding the relevant proxies, only the NetZero2040 scenarios and the EAA's WAM and Transition 2040 scenario were evaluated. To ensure comparability, the assessment is restricted to energy system mitigation options (for a detailed description of the methodology, see 8.A.1). As discussed in Section 8.3.1, synergy potentials between individual mitigation options and SDGs generally outweigh trade-off potentials. As shown in Figure 8.17, the scenario analysis also reflects this, with the cumulative synergy potential of modeled mitigation options exceeding trade-off potential. However, the mitigation options are not all implemented to the same extent in the different scenarios. Dark green or red interactions thereby indicate the highest and light green or red the lowest synergetic and trade-off potential compared to other scenarios.

The quantitative assessment reveals that the Transition 2024 scenario has the highest trade-off potential, while the

NetZero2040 scenarios with low demand show the highest synergy potentials. This reflects the high synergy potentials identified for demand side mitigation options (Section 8.3.1). How considered mitigation options are implemented is key realizing synergy or trade-off potentials. The Net-Zero2040 high-import /low-demand scores high for SDG synergy potentials, primarily due to externalized negative spillovers.

As the current assessment only indicates the synergy and trade-off potentials that exist within Austria, it is important to mention that any scenario with high import and export levels may have negative impacts on global SDG attainment due to spillover effects. The expansion of renewable energy infrastructure, for instance, depends on the availability of critical materials and could introduce trade-off if those materials are sourced from countries with low environmental and employment standards. Hence, although domestic synergy potentials may be high, trade-off potentials might be higher if spillovers beyond Austria are considered. As spillovers are context specific, they were not assessed within this framework but need to be considered and assessed when designing mitigation strategies. The Spillover Index tracks these impacts along the three dimensions of (a) environmental spillover, (b) social impacts embodied into trade, economic activities and finance, and (c) security. Thereby, a higher score reflects more positive spillover effects from Austria to the rest of the world with regards to the SDGs. Currently, Austria scores 59.9, ranking as low as 152 of 166 countries (Sachs et al., 2023) and 64 on the European index, placing 19th of 31 countries (Lafortune et al., 2024).

As adaptation options were not included in the assessed scenarios, they could not be assessed here but should be considered for climate-friendly transformation pathways.



Figure 8.17 Relative implications of different Austrian mitigation scenarios for the SDGs. Shading indicates strength of synergy, with darker shading representing the highest and lighter shading the lowest interaction potential.



**Figure 8.18** Projected emissions from specified scenarios, for the period 2022 – 2050, and respective adherence to equity-based carbon budgets for the same period.

# Austrian mitigation scenarios and carbon budgets under varying fairness criteria

As illustrated in Section 8.2.2, most distributions of a remaining global carbon budget to Austria for the period 2022-2050 are negative, implying net carbon removals for the entire period. As negative emission technologies, e.g., carbon capture and storage, are yet not technically or politically feasible in Austria, this would imply that most budget allocations targeted at 1.5°C or well below 2°C of temperature rise are unachievable for the country without significant changes in the political landscape. For both 1.5°C and 2°C targets, the only fairness principle leading to allocations still achievable is responsibility - some interpretations of this principle result in budgets of up to ~430 MtCO<sub>2</sub> (1.5°C target) or ~626 MtCO<sub>2</sub> (2°C target) (CCCA, 2022) (third quartile values of responsibility budget estimates from Figure 8.2b). Both the WEM and WAM scenarios from the EAA would exceed these thresholds, with the WAM scenario approaching the 2°C limit. The Transition 2040 scenario, in contrast, would be compatible with a 2°C budget in some manner recognizing responsibility for past emissions. However, allocations based on status quo approaches, e.g., grandfathering, would allow for significantly higher budgets - but these are not considered to be consistent with considerations of fairness. Status quo approaches targeting 2°C of temperature rise would lead to budgets with more emissions than those projected in both WEM and WAM.

# 8.5. Politico-economic and socio-cultural feasibility challenges for climatefriendly transformation pathways

So far, little attention has been paid in climate research to the political, economic, social, and cultural conditions necessary for successful transformations – both within Austria and in the context of broader developments that affect the country. This section complements the literature on technological feasibility by assessing the social science literature on the politico-economic and socio-cultural feasibility challenges of achieving climate targets. This will contribute to closing the implementation gap resulting from the difficulty to implement effective measures due to societal and political resistance.

Given the broad agreement among researchers in favor of broader climate targets, the APCC (2023) introduces the concept of 'climate-friendly living', which seeks to foster a good life for all within planetary boundaries. In line with this definition, climate-friendly transformation pathways aim to reduce both direct and indirect emissions to achieve mitigation targets (climate neutrality), limit vulnerability to climate change through adaptation (climate resilience), and, more broadly, ensure decent living standards (Aigner et al., 2023d). This section assesses factors that potentially hinder or enable climate-friendly transformation pathways, which in turn affect the likelihood of achieving climate targets.

Currently, there is considerable societal acceptance for democratic climate governance, with high levels of acceptance for more ambitious climate actions in Austria (Margotti, 2024), in Europe (Abou-Chadi et al., 2024) and globally; 73 % would be even willing to contribute 1 % of their income (Andre et al., 2024). In Austria, 86 % of respondents indicated that they believe democracy to be the best form of political organization (Zandonella, 2021). Additionally, 73 % expressed trust in science (ÖAW, 2023) and 75 % of people expressed concerns that necessary measures to combat climate change have not been implemented in time (INTEGRAL, 2022).

Notwithstanding this support, effective climate policymaking is confronted with various forms of climate-action-delaying practices. Although the outright denial of (the effects of) climate change has been marginalized in public debate, discourses of climate delay – defined by Cass et al. as "arguments in public discourse for climate inaction by governments, individuals, and other actors" (2023, p. 2) – have increasingly influenced climate policies. These discourses tend to redirect responsibility, ignore the costs of inaction,

Meta-level policy fields	Inhibitors of climate-friendly transformations	Enablers of climate-friendly transformations		
Global governance	<ul> <li>Contested global climate governance</li> <li>Market-biased governance limiting policy space</li> <li>Precedence of national security policies over cooperative governance</li> </ul>	<ul> <li>Effective democratic policy space</li> <li>Increasing strategic autonomy and reducing import dependency through sufficiency and avoid-measures</li> </ul>		
Democracy, law and science	<ul> <li>Democratic climate-action-delaying policies</li> <li>Science skepticism and culture war-rhetoric</li> </ul>	<ul> <li>Linking representative with deliberative and participatory forms of democracy, and scientific expertise</li> <li>Emphasizing benefits of climate action alongside costs of inaction</li> <li>Climate litigation and lawsuits</li> </ul>		
Politico-economic framework conditions	<ul> <li>Macro- and micro-economic growth imperatives</li> <li>Power of incumbent business interests</li> <li>Dominance of de-risking policies</li> <li>Contested European Green Deal</li> </ul>	<ul> <li>Coordinated economic planning to foster innovation and exnovation</li> <li>Limiting incumbent business interests</li> <li>Fiscal, monetary and industrial policies combining a 'sticks and carrots' policy mix</li> </ul>		
Socio-cultural framework conditions	<ul> <li>Status quo bias (defending unsustainable and inegalitarian consumption patterns)</li> <li>Awareness-action gap</li> <li>Advertisement promoting fossil fuel-based overconsumption</li> </ul>	<ul> <li>Eco-social policies to reduce inequalities and enhance well-being for all</li> <li>Broader alliances based on benefits for everyday life</li> </ul>		
Adaptation	<ul> <li>Implementation gap due to missing sense of urgency, insufficient political leadership, and lack of dedicated budgets</li> <li>Limits to adaptation shrink the action space for policy</li> </ul>	<ul> <li>Iterative climate risk management can leverage the scale, scope and speed of implementation</li> <li>Transformative adaptation can substitute incremental adaptation if it becomes unable to limit residual damages to acceptable levels</li> </ul>		

Table 8.2 Overview on inhibitors and enablers of climate-friendly transformations.

advocate for non-transformative solutions (Lamb et al., 2020), or justify high-carbon lifestyles by referring to luck and merit rather than privilege (Cass et al., 2023).

# 8.5.1. Global governance

### Inhibitors of climate-friendly living

A key inhibitor for climate-friendly living is the ineffectiveness of global climate governance. Veto coalitions, formed by fossil fuel lobbies, continue to block progress and hinder the implementation of existing commitments (Hollaus et al., 2023; Romanello et al., 2023; Zeller, 2023; Bärnthaler et al., 2024) (Sections 6.2, 6.4.2). COP28, held in Dubai as part of a series of three consecutive COPs organized by oil-exporting countries, saw record-high participation of fossil lobbyists. This resulted in delays to climate action, tolerated by key political players (Global Witness, 2022; Green et al., 2022; Romanello et al., 2023; Zeller, 2023; Bärnthaler et al., 2024). While some nations in the Global South, particularly the BRICS countries, have gained influence in climate negotiations (Skjærseth et al., 2021), many low-income countries have experienced a decline in negotiation power due to increasing debt (UNCTAD, 2023). Actors from the Global North continue to demonstrate reluctance in terms of concretizing and implementing the principle of 'common but differentiated responsibilities' (UN General Assembly, 1994; Fisher, 2015). Despite numerous commitments, including those made by the EU, compliance with climate financing obligations remains to lack behind (Fanning and Hickel, 2023), as does the pledged support for loss and damage compensation (Baumann, 2024). COP28 furthermore reinforced discourses that delay climate-action, focusing on gradual decarbonization such as 'transitioning away from fossil fuels' (Wise, 2023) and 'phasing out of fossil fuels' (Engels et al., 2023b). These discourses support maintaining carbon-based infrastructures (Durrant et al., 2023; Hansen, 2023), while emphasizing Carbon Capture and Storage (CCS) and Carbon Capture and Utilization (CCU) technologies (Pichler et al., 2021; Global Witness, 2022; Green et al., 2022; Romanello et al., 2023; Zeller, 2023; Bärnthaler et al., 2024). The emphasis on CCS and CCU, along with the broader structural preference for techno-economic interests, is reinforced by scientific research that promotes these technologies without adequately addressing their limited potential, associated risks, and high costs (Anderson, 2015; Anderson et al., 2018; Stoddard et al., 2021).

Over the past four decades, neoliberal globalization has significantly shaped global institutional frameworks, actor constellations, and power relations (Jessop, 2004; Lorek and Fuchs, 2013; Brand et al., 2020). Market-biased governance has limited national public institutions and created global economic institutions, equipped with strict legal remedies, including mandatory dispute settlement mechanisms for international trade (Rodrik, 2011, 2017). While being entitled to enforce rules on economic governance, they lack the power to enforce rules for effective climate actions (Streeck, 2014; Fremstad and Paul, 2022). As a result, sustainable development targets remain voluntary, and mitigation targets are subject to ineffective enforcement mechanisms (Rajamani, 2016; Rajamani and Bodansky, 2019).

Geopolitical rivalries are impeding progress on SDG17: Partnerships for the Goals, and undermine global governance, particularly rules- and rights-based frameworks (Keohane and Victor, 2011; Keohane, 2015), which in turn further weakens UN institutions (Elsig et al., 2016). This has resulted in stagnant foreign direct investment (Reiner and Edlinger, 2023), the emergence of new forms of multi-scalar global production networks (Fischer, 2020; Novy, 2022; Fischer et al., 2023), and increased competition for critical raw materials such as cobalt and lithium (Brand and Wissen, 2024; Tröster et al., 2024). There has furthermore been a precedence of national (energy) security policies over cooperative climate governance (Streck and Terhalle, 2013; Selby and Hoffmann, 2014; Baumgartner et al., 2023; Hitzl, 2023; Pagnone et al., 2023) and the achievement of decarbonization targets (Roy and Schaffartzik, 2021; Baumgartner et al., 2023). Moreover, rising military expenditure frequently diverts funding from other public investments needs (Ikegami and Wang, 2023; Tipping Point North South, 2023; Tian et al., 2024). The significant social-ecological 'externalities' of standing armies, conflicts, and wars (Bonneuil and Fressoz, 2017; Rajaeifar et al., 2022) are thereby frequently overlooked. This dynamic reinforces delays in implementing climate measures, exemplified by the sentiment of 'in principle yes, but not now', and even fosters a 'fossil backlash', as European governments expand their liquefied natural gas (LNG) infrastructures (InfluenceMap, 2023; Zeller, 2023). In Austria, fossil energy imports averaged EUR 9 billion annually from 2010 to 2021, rising to EUR 10.7 billion in 2023, highlighting the vulnerability of the Austrian economy due to its heavy dependence on fossil fuel imports (Section 6.7.2). In parallel, profits of Austrian companies such as Schoeller-Bleckmann Oilfield and OMV boomed in 2022 and were accompanied by increased emissions and close to zero alignment activities (Bukold, 2023; Zeller, 2023).

# Enablers of climate-friendly living

The resurgence of strategic and better planned economic policymaking that relies on more coherent state intervention may facilitate climate-friendly living. This 'return of the state' is underscored by initiatives such as the US Inflation Reduction Act, China's Belt and Road Initiative, the Net Zero Industry Act of the EU, and the increased protection of critical resources in the Global South - all of which may expand policy space at the EU and national level (Aiginger and Rodrik, 2020; van Apeldoorn and de Graaff, 2022; AK Europa, 2023; McNamara, 2023). Forms of re-, near- and friend-shoring are proliferating (Fischer et al., 2023), such as the shift from Russian gas to US-LNG, with imports rising from 73.7 million m<sup>3</sup> (2021) to 122.8 million m<sup>3</sup> (2022) (Zeller, 2023). If grounded in international cooperation and a comprehensive understanding of climate policy as part of sustainable development (Skjærseth et al., 2021; UNCTAD, 2023; Brand and Wissen, 2024), this expanded policy space can help overcome departmental thinking and silo approaches to climate policymaking. Strategic autonomy and reducing reliance on fossil fuel imports can be pursued through sufficiency and avoid-strategies (Hache, 2022; Brizga et al., 2023; Hitzl, 2023; Statistik Austria, 2023), as well as by promoting circular economy approaches (Nature Editorial, 2023). This broader approach to protection from harm and security extends beyond military considerations (Mandelli, 2022; Bohnenberger, 2023; Koch et al., 2023) and integrates climate justice (Brand and Wissen, 2024). Ultimately, these efforts can unlock the synergistic potentials between climate action and the SDGs.

#### 8.5.2. Democracy, law and science

# Inhibitors of climate-friendly living

Although Austria has formally committed to the Paris Climate Agreement (Hollaus et al., 2023; Steurer et al., 2023) (Section 6.3.1), climate politics up until 2019 exhibited notable tendencies toward delaying climate action. Emissions remained consistently above 60 MtCO<sub>2</sub> per year, demonstrating little change from the 1990s through 2020 (Aigner et al., 2023e) (Section 8.2.2). The absence of enforceable implementation rules in the climate law ('Klimaschutzgesetz', Section 6.3.3) indicates the ineffectiveness of climate measures (Nash and Steurer, 2021), largely due to the lack of clearly assigned responsibilities and enforceable rules (Hollaus et al., 2023). Social partners, the federal government, and the provincial governments prioritized shortterm socio-economic gains over long-term mitigation and adaptation efforts (Wieser and Kaufmann, 2023). Following 2019, global civil society mobilization, the European Green Deal (EGD), and the formation of a new Austrian federal government - comprising conservatives and Greens - led to the implementation of climate policies designed to be more effective. These measures included the introduction of a climate bonus ('Klimabonus'), energy communities, and a carbon tax in Austria (APCC, 2023). As a result, there has been notable sectoral policy progress (Section 6.4.2) and reductions in CO<sub>2</sub> emissions. Some social partners, particularly those representing employees (trade unions and the Chamber of Labor), became more amenable to transformative climate policies, establishing institutions like the ÖGB-Klimabüro and drafting transformation plans such as the AK-Umbauplan (AK Wien, 2024) (Section 4.8). However, following the pandemic, the war in Ukraine, and the cost-ofliving crisis, Austria experienced a resurgence of climate-action-delaying policies, also of social partners (Section 6.4.1). Measures with long-term impact - such as binding regulations on land use, biodiversity and CO<sub>2</sub>-pricing - were deferred, jeopardizing the progress on the relevant SDGs, including SDG7: Affordable and Clean Energy and SDG12: Life on Land. Employer-oriented stakeholders called for less ambitious climate measures regarding the phase-out of oil and gas (Arnecke, 2024), missing opportunities to enhance long-term resilience (Hüther et al., 2024). Meanwhile, unions have been calling for an economic stimulus package that includes resource-intensive measures such as subsidies for single-family homes (Arnecke, 2024). As a consequence, opportunities for social-ecological forms of basic provisioning, redistribution, and progressive options to design such policies have been missed (Brand and Niedermoser, 2019; Bärnthaler et al., 2021; Büchs et al., 2021; Keil and Kreinin, 2022; Oswald et al., 2023; Chancel et al., 2024).

In Austria, a significant gap remains between its self-image as a country with high environmental standards and the reality of unsustainable practices across many sectors (Section 6.4). In their study of climate delay discourses in Austria, Frühwald et al. (2024) identify various tactics employed to impede climate action. Political actors dominate the climate debate in the media, representing 34 % of all statements and being the most frequent users of delaying rhetoric. In contrast, civil society contributes most constructive statements, accounting for approximately 25 % of contributions. Among political actors in Austria, six out of ten delaying statements originate from the ÖVP, while the FPÖ is responsible for three. Notably, the only instances of explicit climate change denial in the sample also originate from the FPÖ. The most discussed topics – such as the Climate Protection Act ('Klimaschutzgesetz'), the European Green Deal, and the Renewable Heat Act ('Erneuerbare-Wärme-Gesetz') – are disproportionately subject to delays. This suggests that the defense of the status quo intensifies as pressure for change increases in these areas. Common delaying tactics include emphasizing perceived disadvantages while ignoring the costs of inaction (e.g., in Pearson, 2024).

Discourses of climate delay are also prevalent in emerging forms of reactionary politics, as exemplified by Donald Trump (USA, 2017-2020) and Jair Bolsonaro (Brazil, 2019-2022). These leaders promote anti-liberal, anti-egalitarian and anti-science political agendas (Lockwood, 2018; Novy, 2022; Jaeggi, 2023; Sælen and Aasen, 2023). Their rejection of anti-discrimination policies and hostility towards scientific expertise distinguishes them from other types of climate delaying practices (Kulin et al., 2021; Ekberg et al., 2022; Dannemann, 2023). These actors instrumentalize climate policies as part of a broader culture war, framing climate action as an ideological battleground (Abou-Chadi et al., 2024). Such an approach induces a backlash in climate policymaking by rigidly defending the convenience of certain existing forms of life (Brand and Wissen, 2017; Blühdorn, 2022) (Section 6.2), capitalizing on a 'status quo bias' (Börjesson et al., 2016) and exploiting 'trigger points' to provoke opposition (Mau et al., 2023). Traditional and social media further amplify these delays (Theine et al., 2023), often driven by their reliance on advertising revenue from incumbent political and economic actors. This leads them to prioritize sensationalist and negative news to attract attention (Kalpokas, 2018; McIntyre, 2018), reinforcing the disconnect between newsworthiness and the actual usefulness of the information presented (Perga et al., 2023).

# Enablers of climate-friendly living

Liberal constitutional democracies mediate between individual freedom and social order to balance between sectional interests and freedoms (Kelsen, 2006). While fundamental rights constrain political decision-making, democratic rulemaking can impose restrictions on individual freedoms when they obstruct the freedoms of others (Kelsen, 2006), including those of specific social groups or future generations. Fundamental rights protect individuals and qualified minorities from disproportionate interference, obligating lawmakers to impose only proportional restrictions and to carefully weigh competing rights and freedoms (Kelsen, 1929). In the context of climate change, legislators must consider fundamental rights and the principle of proportionality when imposing climate-related bans and restrictions (e.g., diesel bans or bans on advertisement) that impact economic and property rights. Especially, since courts at various levels (domestic, European and international) have established connections between the harmful effects of climate change and fundamentals rights, as well as other constitutional environmental provisions (Hollaus et al., 2023).

Over the past years, individuals and NGOs have filed numerous lawsuits addressing the lack of climate action. Climate litigation is thus another important enabler of climate-friendly living (Section 6.6). There are several key take-aways from climate cases: From an intertemporal perspective, lawmakers may be required to implement emission reduction measures in a timely manner and establish rules and targets that provide clarity and certainty for societal transformation towards climate-neutrality. This duty is reflected in rulings such as the German Federal Constitutional Court's Climate Protection Order (24.3.2021, BVerfGE 157,30, Cf Human Rights Law Journal HRLJ 299-326 (2021) (Hollaus et al., 2023). Additionally, human rights can impose a duty of care on the state, mandating legislators to protect citizens from the negative effects of climate change. The Dutch Urgenda ruling (ECLI:NL:HR:2019:2007) took a lead in this direction. In a landmark ruling ('Verein KlimaseniorInnen Schweiz and others vs Switzerland', 04/09/2024) the European Court of Human Rights (ECHR) established a fundamental right for individuals to effective protection from serious adverse effects of climate change on their life, health, well-being and quality of life (ECHR, 2024). The ECHR determines that states have a positive obligation to adopt and to apply coherent climate mitigation regulations in a timely manner (Arntz and Krommendijk, 2024). While the ECHR grants national legislators considerable discretion in the choice of climate protection instruments, it declared a science-based, plan-led climate policy a fundamental right (Hollaus, 2024).

Although several courts have identified inadequate climate laws, climate litigation also faces challenges and limitations (Section 6.6). In various jurisdictions, courts have adopted a restrictive interpretation of the right to bring cases to court, leading to the dismissal of several climate cases (e.g., the US Juliana case; the Carvalho decision of the Court of Justice of the European Union or the Children and Youth's Climate Case decision of the Austrian Constitutional Court) (Hollaus et al., 2023; Madner, 2023a). Overall, while climate policies must uphold fundamental rights, lawmakers and administrators retain considerable discretion in their decision-making. In rule-of-law-based societies, constitutions can only govern the transformation processes to a limited extent, leaving many specific decisions regarding the choice of means to tackle climate change - such as the policy framework and the choice of sectoral field - to political deliberation and (simple) majority voting (Hollaus et al., 2023; Madner, 2023a). Accordingly, the ECHR highlighted that "judicial intervention [...] cannot replace or provide any substitute for the action, which must be taken by the legislative and executive branches of government" (ECHR, 2024, para. 411). However, the court also held that "democracy cannot be reduced to the will of the majority of the electorate and elected representatives, in disregard of the requirements of the rule of law" (ECHR, 2024, para. 411).

Linking representative with deliberative (e.g., climate assemblies) and participatory (e.g., at the workplace and in the neighborhood) forms of democracy can contribute to the implementation of more effective climate measures (Duvic-Paoli, 2022; Gough, 2022; Aigner et al., 2023d; Bärnthaler, 2023; Clar et al., 2023; Lage et al., 2023). This is particularly true if institutional mechanisms are established to translate deliberative democratic decision-making into public policy (Clar et al., 2023) and if corporate influence on political decisions is limited (Section 8.5.3). By integrating deliberative and participatory approaches into representative democracy, it becomes more likely to reduce status-quo biases, overcome transformational deficits (Haberl et al., 2020; Hammond, 2020), and bridge awareness-action gaps. Deliberative processes can shift people's fundamental norms and outlooks, fostering socio-cultural transformations (Hammond, 2020). Such a revitalized democracy does not merely reflect existing preferences but seeks to reorient them, promoting social acceptance by implementing policies that yield beneficial eco-social outcomes (Bärnthaler, 2024a) (Cross-Chapter Box 4). However, strengthening participatory forms of democracy needs to be aware of possible biases in participant recruitment, as this may limit democratic legitimacy (Blühdorn, 2024).

Increasing the newsworthiness of climate change research requires changes in mediatization, political communication, and science collaboration (Section 6.8.3) (Perga et al., 2023; Theine et al., 2023). First, to be more effective, news coverage can emphasize the benefits of climate action alongside the costs of inaction (Frühwald et al., 2024). This involves shifting the focus from individual actions toward the need for collective structural changes (Aigner et al., 2023d; Cass et al., 2023). Second, instead of focusing on end-of-thecentury projections rooted in the natural sciences (Pielke and Ritchie, 2021), news can highlight local, present-day consequences of climate change. It can also illustrate how collective actions can contribute to transformative changes that reduce emissions in the long-run while making shortterm progress towards the SDGs by capitalizing on potential synergetic effects (Overland and Sovacool, 2020; Perga et al., 2023; Theine et al., 2023) (Section 8.3.1). Effective climate science communication can be strengthened by integrating (quantitative) empirical research with insights from social sciences about institutional challenges (Biermann et al., 2012; Carey et al., 2014; Victor, 2015; Porak and Reinke, 2024; Spash, 2024). This requires a more contextual approach to climate science that evaluates various place-based alternatives through transdisciplinary research (Görg et al., 2017; Jahn et al., 2020; Plank et al., 2021; Aigner et al., 2023e), recognizing that science-society interaction must be honored in academic careers.

## 8.5.3. Politico-economic framework conditions

# Inhibitors of climate-friendly living

Key politico-economic factors inhibiting climate friendly transformation pathways are the micro and macro-economic growth imperative and related corporate strategies (Section 6.2). The growth imperative is rooted in the political economic structure of capitalist economies and actualized through interconnected mechanisms, including corporate competition, rent extraction, technological unemployment, state dependence on growth, positional consumer behavior, and rebound effects (Stratford, 2020; Wiedmann et al., 2020; Corlet Walker et al., 2021; Koch, 2022) (Cross-Chapter Box 7). They also manifest on a sectoral level, particularly in environmentally harmful industries such as aviation and the automotive industry, as staying competitive requires profit and continuous growth (Mattioli et al., 2020; Keil and Steinberger, 2023; Huwe et al., 2024). Additionally, powerful industries, often tied to carbon-intensive activities, exert political influence to weaken or obstruct climate regulations, entrenching growth-oriented agendas that conflict with systemic changes necessary to mitigate emissions (Newell and Paterson, 2010). In Austria, land use decisions are especially susceptible to these politico-economic influences based on

the growth paradigm (Getzner and Kadi, 2020; Müller et al., 2024; Getzner et al., n.d.).

In this framework, corporate strategies tend to be insufficient for addressing climate change as many companies lack intrinsic motivation to support climate policies unless incentivized by win-win solutions or competitive advantages that benefit them and national industries in the short run (Engels et al., 2023a; Steurer et al., 2023; Wieser and Kaufmann, 2023). Despite a significant increase in financial incentives to boost efficiency (Fischer et al., 2023), sufficiency strategies tend to be underrepresented in climate mitigation (Zell-Ziegler et al., 2021; Lage et al., 2023). Specific timelines for phasing out harmful practices, such as fossil subsidies, for instance, are often missing (Kletzan-Slamanig et al., 2022). The influence of incumbent business interests - particularly the automotive industry, e.g., reinforcing car-dependency (Pichler et al., 2021; Bärnthaler et al., 2024), and the meat-based food industry (Penker et al., 2023) - inhibit and delay progress (Seto et al., 2016; Newell, 2019; Schaffartzik et al., 2021). Additionally, distribution of spending in EU lobbying is heavily skewed, with corporate lobbying far surpassing spending of labor and public-interest groups (Porak, 2023).

The European Climate Act aims to reduce GHG emissions by at least 55 % by 2030 compared to 1990 levels (Section 6.3.2). The European Green Deal promotes decarbonization through initiatives, including the 'Fit for 55' package, the 'Circular economy action plan' (focused on reducing non-renewable throughputs), the 'Biodiversity Strategy 2030', green public procurement, a more ambitious Emission Trading Scheme (ETS), the Carbon Border Adjustment Mechanism (CBAM), the 'Net Zero Industry Act', the EU taxonomy and the CSRD (Corporate Sustainability Reporting Directive) to promote ESG (environmental, social and corporate governance) accounting and investment strategies (Fischer et al., 2023) (Section 6.3.1, Table 6.1). However, EU climate policies tend to prioritize subsidizing new technologies and enabling markets while largely ignoring co-benefits and geopolitical advantages of demand reduction (Section 8.5.1). This can be partly explained by 'push' measures, e.g., bans and regulations, frequently encountering strong public opposition (AK Europa, 2023). The business-friendly 'Green Industrial Plan', for example, passed with minimal opposition (Gerasimcikova, 2023), whereas more ambitious proposals of the 'Farm to Fork Strategy' were significantly weakened by agro-industry lobbying (Corporate Europe Observatory, 2020; Holland and Tansey, 2022). Similarly, the 'EU Biodiversity Strategy 2030'

was delayed and watered down due to pressure from business associations. Fossil corporations also lobbied against key legislative efforts such as the 'Energy Performance of Buildings Directive, 'Energy Efficiency Directive, and 'Hydrogen and Gas Decarbonization Package' (InfluenceMap, 2023). If the influence of political and business actors that oppose or delay climate action is not restrained - and if they even continue to gain influence - European climate policies risk becoming unpredictable, potentially delaying long-term investments and necessary exnovations (e.g., the deliberate phasing out of emission-intensive practices and business models) (Kubeczko et al., 2023). Wealthy fossil-fuel interests prioritize protecting their financial holdings over facilitating the energy transition, although wage earners would face minimal impact from potential pension fund losses due to stranded fossil assets (Colgan et al., 2021; Semieniuk et al., 2022).

Greening public finance is proceeding slowly and tends to focus on mitigation, thus neglecting expenditures for adaptation (Section 6.7.3). Public financing in general is constrained by strict fiscal rules in the EU, which jeopardize crucial public investments. Public climate financing tends to neglect further environmental objectives, in particular reducing resource use (including soil), protecting biodiversity, and supporting the circular economy (Kubeczko and Krisch, 2023; Mang and Caddick, 2023; Miess and Ornetzeder, 2023; Weber and Kubeczko, 2023) (Section 6.7.3). Green finance de-risking policies designed to attract private capital and facilitate green investment contain loopholes. Consequently, 'brown' investments (i.e., those tied to carbon-intensive industries) continue to dominate in absolute terms (Christophers, 2023; UNCTAD, 2023). Globally, de-risking strategies rely on voluntary pledges from a small group of highly polluting companies, known as the Carbon Majors (Carbon Majors, 2024). They primarily comprise fossil fuel, cement, and mining companies, being well aligned with major financial institutions, including the three largest asset management firms BlackRock, Vanguard and State Street (Baines and Hager, 2023). However, the role of green finance in achieving climate targets remain contested (Baines and Hager, 2023; Dziwok and Jäger, 2024; Kalinowski, 2024). Despite the increasing cost-competitiveness of renewable energy compared to fossil fuels, companies continue to invest in the exploitation of fossil energy sources. This can be attributed to private investment decisions being driven by expected return on investment rather than investment costs, with renewables remaining less profitable than their fossil alternatives (Christophers, 2024). The effectiveness of de-risking policies is limited due to weak conditionalities, the retention of veto power by private capital, and the lack of differentiation between investment in essential and non-essential sectors (Gabor, 2023; Gabor and Braun, 2023; Mazzucato and Rodrik, 2023).

# Enablers of climate-friendly living

In a just transition, effective planning of climate, economic, social and land use policies can enhance synergetic potentials with the SDGs and broaden the scope for climate policy action beyond market-driven approaches (Cerniglia et al., 2023; Gabor and Braun, 2023) (Section 6.8.1). The inaction and lack of coordination not only jeopardizes the achievement of climate targets but also threatens making progress on the SDGs, including goals on economic development and human well-being. On a global level, the projected economic costs of climate-related damages supersede the financial resources required to limit global temperature rise to 2°C by a factor of 6 (Kotz et al., 2024). In Austria, the monetized costs of inaction are expected to range between EUR<sub>2023</sub> 7.3 and 14.6 billion annually until 2050 (Steininger et al., 2020a). However, this estimate only accounts for monetized costs thus representing only a fraction of the actual costs of inaction, as so far not all impacts could be quantified in monetary terms. The health sector is anticipated to bear the highest costs, particularly in the form of lost healthy life years. In addition, other health impacts (e.g., injuries and deaths due to extreme events (floods) and increasing numbers of waterborne and foodborne diseases) have so far not been monetized (Steininger et al., 2015). In contrast, the expected annual additional investments for reaching carbon neutrality by 2040 in Austria are estimated to range between EUR<sub>2023</sub> 6.4 and 11.2 billion (Weyerstraß et al., 2024) (Section 8.4.4). Until 2030, annual investments of an additional 3.6-4.8 % of GDP are required to reach climate neutrality, which is expected to generate annual value-added effects of 2-2.7 % of addition GDP and create between 60,000 and 80,000 additional jobs each year (Miess et al., 2022). At the EU level, green public investment can be bolstered through various mechanisms, such as exemptions from fiscal spending restrictions, a 'Green Investment Golden Rule' to permanently exempt green investments from budgetary constraints, the creation of an EU Climate Fund (Pekanov and Schratzenstaller-Altzinger, 2023), or 'public green securities' (Section 6.7.3). A large share of the required additional investments can be mobilized be reallocating funds from fossil fuels to renewables (Section 6.7.1, Cross-Chapter Box 8).

The European Green Deal can be strengthened and inaction limited (Gough, 2017, 2022; Schulze Waltrup, 2023) by coordinated economic planning that adopts a 'sticks and carrots policy mix framework' (Dafermos and Nikolaidi, 2023) that curtails corporate power (Haas and Sander, 2020; Haberl et al., 2020; Fuchs and Dolinga, 2022; Vogel and Hickel, 2023; Bärnthaler et al., 2024). Monetary and financial policies could reduce capital requirements for green loans while increasing them for 'dirty' loans, facilitating a transition of the economic and financial sector (Dafermos and Nikolaidi, 2023; Gabor and Braun, 2023). In addition, credit guidance would enable central banks to steer private sector behavior towards socially and ecologically beneficial sectors. Industrial and labor market policies can guide companies in their decarbonization strategies, as outlined in the EU Net Zero Industry Act (Section 4.7), while creating high-quality jobs and protecting workers affected by restructuring through initiatives like a job guarantee and training programs (Froy et al., 2022; Upham et al., 2022; Schuberth and Soder, 2024). Furthermore, technological innovations must be paired with exnovations (i.e., the deliberate phasing out of outdated and harmful practices) (Lütkenhorst et al., 2014; Arnold et al., 2015; Kivimaa and Kern, 2016; Altenburg and Assmann, 2017; Haberl et al., 2020). Integrating employees and social partners in transition processes not only enhances legitimacy (Hofbauer et al., 2023), but helps prevent information failure (Pichler et al., 2021; Schuberth and Soder, 2024).

Climate-friendly fiscal policies can enhance progressivity through measures, including a wealth tax on the top 1 %, a tax on unrealized capital gains, and an increase of the minimum corporate tax. These initiatives could generate additional revenues of approximately 1.9-2.9 % of EU GDP annually (Guzzardi et al., 2023). Furthermore, eliminating climate harming public subsidies could save Austria between EUR<sub>2023</sub> 4.9-6.9 billion (Kletzan-Slamanig et al., 2022). To support low-income households, direct transfer payments can be implemented alongside initiatives promoting climate-friendly household appliances, subsidizing repairs, and facilitating the transition to efficient heating systems. Expanding access to high-quality public services and infrastructure, as a form of collective consumption, can further improve wellbeing while reducing energy demand (Vogel et al., 2021, 2024; APCC, 2023; Vogel and Hickel, 2023; Die Armutskonferenz and Ökobüro, 2024) (Section 6.5.1).

# 8.5.4. Social and socio-cultural framework conditions

# Inhibitors of climate-friendly living

The current consumerist and inegalitarian mode of living obstructs climate-friendly living. Existing provisioning systems both arise from and reproduce inequalities of class, gender, and ethnicity (Brand and Wissen, 2017; Markkanen and Anger-Kraavi, 2019; Aigner et al., 2023a, 2023b, 2023f). Inequality, on the one hand, is a key driver of climate change, with wealthier households contributing disproportionately to the crisis (Chancel, 2022; Chancel et al., 2022; Theine et al., 2022; Millward-Hopkins and Oswald, 2023). At the same time, affluent households are less affected by the effects of climate change and related policies, while possessing greater adaptive capacity (Seebauer, 2021; Engels et al., 2023a; Essletzbichler et al., 2023; Dang et al., 2024; Gilli et al., 2024). Global emissions are also distributed unequally. In 2022, average annual per capita emissions in the EU27 and Austrian exceeded the world average by 1.5 and 2.2 tCO<sub>2</sub> respectively (Our World in Data, 2023). Despite being sustained at the expense of others (Brand and Wissen, 2021), the Western mode of living is often considered 'non-negotiable' (Blühdorn, 2022; Hausknost, 2023). On the other hand, inequality hampers effective climate action. Wealthy elites, for instance, can obstruct climate policies by undermining societal support and increasing political resistance, thereby eroding the social foundations necessary for collective action (Green and Healy, 2022; Aigner et al., 2023d). If economic and social policies fail to provide adequate social protection, support for climate measures declines, as seen during the cost-of-living crisis (Andreoni, 2017; Carattini et al., 2019; Budolfson et al., 2021; Feindt et al., 2021; Lampl et al., 2024). Short-term relief measures, such as fossil fuel subsidies, often take precedence over long-term structural change. In 2022, most of the implemented compensation measures for rising energy prices had negative effects on climate mitigation (Section 6.5.1). These short-term measures may also fail to successfully target low-income households and children (Aigner et al., 2023a).

Although Austrians are relatively well-informed and concerned about the climate crisis (European Commission and Directorate-General for Communication, 2021), there is widespread emphasis on the negative social impacts of climate policies, which contributes to climate delay (Lamb et al., 2020). Similarly, while societal support for expanding renewable energy is strong (INTEGRAL, 2022), the implementation of related projects near one's own residence is often blocked by the 'Not in my backyard' (NIMBY) effect (Sniatynski, 1980; Segreto et al., 2020; Stagl et al., 2024). Further, currently large fractions of advertisement push and legitimize fossil fuel-based forms of overconsumption (Theine et al., 2023). This contributes to the awareness-action gap, where high levels of climate awareness fail to translate into meaningful action. These tendencies are further reinforced by a 'status quo bias', as acceptance of climate measures often only increases after implementation (Andersson and Nässén, 2016; Börjesson et al., 2016; Thaller et al., 2021; Stagl et al., 2024). Recent psychological research (Burroughs and Rindfleisch, 2002; Christopher et al., 2009; Rees, 2010; Amel et al., 2017) suggests that current underlying perspectives, such as mindsets or worldviews, are key inhibitors of collective societal transformation (Kimmerer, 2013; Lent, 2017, 2021; Bridle, 2023). Shifting perspectives is thus a critical lever for transformational change, alongside changes in infrastructures and societal institutions (Meadows, 1999; Abson et al., 2017).

# Enablers of climate-friendly living

Collectively shaping framework conditions - such as underlying perspectives, infrastructures, institutions, discourses, and social norms - is a prerequisite for enabling changes in individual behavior, attitudes, and preferences (Brudermann, 2022; Aigner et al., 2023d). Since effective change is difficult to achieve at the individual level (Brudermann, 2022), collective action becomes crucial. The promotion of collective forms of provisioning can thus facilitate climate-friendly living (Millward-Hopkins et al., 2020; Vogel et al., 2021). However, shaping these framework conditions can lead to conflicts, as more effective measures tend to face greater system resistance, especially from incumbent actors (Geels, 2014; Bärnthaler et al., 2024). While some conflicts can be minimized with win-win solutions capitalizing on potential co-benefits, avoiding them entirely limits implementation to those popular measures that tend to be less effective (Stagl et al., 2024). Two key conflicts must be collectively addressed. First, distributional conflicts (Jorgenson et al., 2019), which align with SDG10: Reduced Inequalities. Second, disputes over the legitimacy of rules that determine which freedoms are promoted or restricted, and for whom (Novy, 2019; Bärnthaler, 2023).

Resistance and conflict are inherent to any transformation (Fuchs and Lederer, 2007; Avelino, 2017). Democratic climate policy requires institutional frameworks that actively engage with conflict and enable compromise (Aigner et al., 2023c) (Section 6.2). Eco-social policies (e.g., Koch and Mont, 2016; Gough, 2017; Koch, 2018; Büchs, 2021; Cucca et al., 2023; Khan et al., 2023; Koch et al., 2023) can help reconcile the needs for social protection with climate mitigation efforts, turning trade-offs into synergies (Mandelli, 2022; Schneider, 2023; Schulze Waltrup, 2023), unlocking co-benefits (Karlsson et al., 2020), and addressing inequalities (Mandelli, 2022; Bohnenberger, 2023). Consumption corridors (e.g., Di Giulio and Fuchs, 2014; Pirgmaier, 2020; Sahakian et al., 2021; Bärnthaler, 2024b) ensure everyone meets basic needs while preventing individuals "from consuming in quantities or ways that hurt others' chances to do the same" (Fuchs et al., 2021, p. 4). Eco-social policies thus go beyond the traditional 'compensatory' role of the welfare state, emphasizing its transformative and enabling potential, e.g., through the public provision of social-ecological infrastructures (Großer et al., 2020; Bärnthaler et al., 2021, 2023; Bohnenberger, 2023; Novy et al., 2023b) or harnessing the potential of specific forms of climate-social work (Aigner et al., 2023f).

Broader alliances for climate-friendly living can emerge from actions with benefits for everyday life and the SDGs, such as affordable housing and energy, quality care, health, and educational infrastructure, local amenities, time prosperity, and caring social relationships (Armutskonferenz et al., 2021; Bohnenberger, 2023; Bärnthaler, 2023; Schneider, 2023; Dengler et al., 2024). Benefits for everyday life result from focusing on service levels rather than on energy or material input. This facilitates achieving not only net-zero but also distributional objectives with respect to affordability, energy poverty and improvements in terms of health, jobs, and energy security (Cross-Chapter Box 3). Since nature conservation and a connection to nature are highly valued across various social strata, there is potential for alliances based on rewilding, preserving green spaces, revitalizing village-, town- and city-centers, and improving local leisure facilities (Barth and Molina, 2023; Borgstedt, 2023; Schleer et al., 2024). Recognizing nature's plurality of values for humans (Pascual et al., 2021, 2023) can facilitate changes in the local environment, like 'sponge cities' or stopping habitat destruction due to the sealing and fragmentation of landscapes, that, over time, contribute to broader systemic shifts (Meadows, 1999; Macy, 2007; Abson et al., 2017). Individuals with an ecologically grounded worldview can play a vital role in informing and educating others about potential collective actions and climate friendly transformation pathways (Martin et al.,

2016; Kohler et al., 2019; Wyborn et al., 2021; McGreevy et al., 2022).

# 8.5.5. Adaptation

### Inhibitors of climate-friendly living

Adaption efforts in Europe and Austria are lagging in scale, scope, depth, and speed (Bednar-Friedl et al., 2022). With Europe warming faster than any other continent and temperature increases in the Alpine region surpassing the European average, a range of climate risks have already reached critical severity levels and could approach catastrophic levels by the end of this century (European Environment Agency, 2024). So far, however, adaptation efforts and societal preparedness fall short of quickly increasing risk levels, creating an urgent need for more decisive, coordinated and precautionary adaptation action (European Environment Agency, 2024). While there has been progress in assessing climate impacts, raising awareness and policy formulation, implementation continues to lag behind, revealing a persistent policy-action gap, which is most pronounced at municipal levels (Aguiar et al., 2018; European Environment Agency, 2020; CDP Worldwide, 2021; Dodman et al., 2022; Pörtner et al., 2022; Jacobi et al., 2024). The majority of planned or implemented adaptation measures tend to be 'soft' and incremental (Lexer et al., 2020; Berrang-Ford et al., 2021), focusing on awareness-raising, knowledge provision, capacity-building or risk monitoring (European Environmental Agency, 2020; Jacobi et al., 2024) rather than preventive, transformative action (Buschmann et al., 2022). Investment in upgrading structural flood protection, for instance, continues to be preferred over avoiding the expansion of settlements in flood-prone areas.

Although evidence on the feasibility of different adaptation options has grown substantially over the past decade (Berrang-Ford et al., 2021), measuring and evaluating their success in reducing climate risks remains challenging (European Environment Agency, 2022; Leitner et al., 2023). Key barriers to effective implementation include a lack of urgency, political leadership, dedicated budgets, and financing (Adger et al., 2005; Bednar-Friedl et al., 2022). Consequently, the costs of adaptation have to be covered mostly from regular public budgets in competition with funding other public tasks (Balas et al., 2024b). In contrast to a growing number of other European countries (e.g., Germany, Switzerland, Ireland, Portugal, Spain, Greece) (European Environment Agency, 2022; Leitner et al., 2023), federal and state-level adaptation strategies in Austria tend to be non-binding 'soft' policies (Clar and Steurer, 2019), lacking enforceability at subnational and local level. Furthermore, it lacks the integration of climate risks and adaptation goals into sector policies ('mainstreaming') and infrastructure projects ('climate-proofing'), vertical and horizontal coordination mechanisms, and mandatory reporting or revision cycles (Clar and Steurer, 2017, 2019; Leitner et al., 2023). Institutionalized multi-level coordination arrangements are thus rather poorly developed, and horizontal policy integration is predominantly voluntary, collaborative, and dependent on the attractiveness of adaptation solutions to sectors and their self-interest (Lexer and Buschmann, 2018; Clar and Steurer, 2019). Although Austria benefits from a well-developed culture of informal, cooperation-based, 'soft' governance modes (Lexer and Buschmann, 2018; Lexer et al., 2018; Steurer and Clar, 2018), the absence of a binding governance framework, such as Germany's Klimaanpassungsgesetz (KAnG), may jeopardize coherent and accelerated adaptation.

Even with immediate and strong mitigation efforts, the need for adaptation will continue to grow in the short- to medium-term due to the accelerating impacts of climate change. Additionally, adaptation strategies may reach their limits as global temperatures rises. In the case of 3°C of global warming, for instance, certain regions and systems may face adaptation limits (Bednar-Friedl et al., 2022). Vulnerable ecosystems (APCC, 2014), flood-prone areas (Schinko et al., 2022), regions facing water scarcity and falling groundwater levels or (seasonal) drought (Schinko et al., 2022; Balas et al., 2024b), and Alpine locations facing natural hazards and land shortages for active or passive adaptation measures (Balas et al., 2024b) are particularly at risk. Current adaptation strategies and risk management systems are also insufficiently prepared for coping with compound climate impacts and cascading risks by extreme weather events. Such events can trigger domino effects across systems or regions, potentially causing complex, long-lasting or even irreversible consequences (European Environment Agency et al., 2018; European Environment Agency, 2024). Examples include mega-droughts leading to water and food insecurity, disruptions of critical infrastructure, and threats to financial markets and stability (European Environment Agency, 2024).

Many smaller Austrian cities and municipalities, especially those outside the KLAR! climate adaptation program, often lack anticipatory, foresighted adaptation planning based on climate scenarios and assessment of impacts, vulnerabilities, and risks (Lexer et al., 2020; Buschmann et al., 2022). Among others, barriers include limited problem awareness and adaptive capacities, low political relevance in comparison to other issues, scarce resources (in terms of authority, financing, personnel, work time, expertise, access to relevant actor networks) and difficulties in accessing funding, the lack of mayoral support and administrative responsibilities, reliance on a few committed actors, and missing legal requirements making adaptation a voluntary task (Wamsler and Brink, 2014; Rauken et al., 2015; Vogel and Henstra, 2015; Lexer et al., 2018, 2020; Buschmann et al., 2022). As a result, adaptation efforts tend to focus on short-term, pragmatic responses to extreme events, rather than adopting preventive, transformative and long-term adaptation pathways (Lexer et al., 2020; Buschmann et al., 2022).

Austria, like many other countries, continues to face challenges in planning and implementing integrated climate actions that pursue both mitigation and adaptation, while supporting sustainable development. Despite the Austrian National Adaptation Strategy and Action Plan (Balas et al., 2024a, 2024b) prioritizing adaptation measures with co-benefits for mitigation, adaptation and mitigation policies remain mostly implemented in decoupled ways across all levels of governance (Aguiar et al., 2018; Reckien et al., 2018; European Environment Agency, 2020). This disconnect is typical across Europe, where coordinated planning and implementation remains rare (European Environment Agency, 2020). Main barriers include a lack of national requirements, institutional inertia, and entrenched communities in research, administration, and practice that operate within 'departmental thinking' or policy silos - often reinforced by fragmented institutional responsibilities (e.g., Kern et al., 2001; Tews, 2005; He, 2013; Runhaar et al., 2018; Landauer et al., 2019; European Environment Agency, 2020).

### Enablers of climate-friendly living

Austria's adaptation policy so far relies on a 'soft', collaborative governance model supported by broad national participation in developing its first national adaptation strategy (Lexer and Buschmann, 2018; Prutsch et al., 2018). This approach includes 'soft' impacts (e.g., awareness-raising, agenda-setting, communication, motivation, legitimation, orientation framework, knowledge base) (Lexer and Buschmann, 2018; Steurer and Clar, 2018; Probst et al., 2019; Lexer et al., 2020), pro-active climate coordinators (Lexer et al., 2020), and national initiatives like the KLAR! Program for local adaptation support, the Natural Hazard and Climate Change Check<sup>10</sup>, and local adaptation advisory services of the federal states (Lexer and Buschmann, 2018; Lexer et al., 2020). While a 'soft', collaborative governance model can strengthen adaptive capacities in Austria, a binding regulatory framework - enacted via a climate or adaptation law - could be a key driver for more effective and coherent adaptation measures across levels and sectors. Institutionalizing adaptation policy processes and formal framework conditions increase the political relevance of adaptation, strengthens the legitimacy of adaptation actors and bears the potential to complement and strengthen more informal governance modes (European Environment Agency, 2022; Leitner et al., 2023). Legal provisions may comprise mandatory climate risk assessments and adaptation plans for subnational levels, sectoral adaptation plans, cross-level coordination bodies and mechanisms, monitoring and reporting obligations, and financing mechanisms. Stronger coercion, however, requires additional support structures for local adaptation efforts (Jacobi et al., 2024).

To enhance adaptive capacities of municipalities and smaller cities and shift from small-scale, reactive and incremental measures towards integrated solutions, several enabling factors have been identified (Biesbroek et al., 2010; Rauken et al., 2015; Vogel and Henstra, 2015; Bausch and Koziol, 2017; Clar and Steurer, 2017; Lexer and Buschmann, 2018; Lexer et al., 2020; Buschmann et al., 2022; Jacobi et al., 2024). The KLAR! program successfully demonstrated many of these factors, including utilizing 'policy windows' created by extreme weather events to initiate adaptation efforts, securing leadership support from mayors and municipal councils, establishing clear responsibilities for local adaptation, offering public financial incentives with low barriers to access, engaging professional external expertise, participating in inter-municipal networks and transnational local authority networks, creating supportive and enabling governance framework with higher administrative levels, implementing 'soft coercion' through binding funding requirements for good adaptation practices (e.g., climate risk assessments and anticipatory adaptation planning), and integrating adaptation into existing municipal development processes. In particular, committed and pro-active coordinators with institutionalized coordination responsibilities are key facilitators for local adaptation efforts. To support 'change agents' (Kristof, 2010), funding should be directed towards strengthening local and regional coordination capacities (Buschmann et al., 2022), including the funding of

<sup>&</sup>lt;sup>10</sup> naturgefahrenimklimawandel.at

dedicated personnel, promoting cross-department coordination, and ensuring adequate qualification and training. This may also involve exploring new and innovative governance models.

Transformative adaptation can substitute incremental adaptation if the latter becomes unable to limit residual damages to acceptable levels. A key enabler for effective adaptation is developing an iterative climate risk management strategy that supports transformative adaptations by encouraging broad, reflexive, and participatory debates on adequate instruments and solutions (Colloff et al., 2017; Fedele et al., 2019). Integrating disaster risk management that can leverage the scale, scope and speed of implementation of adaptation policies (Reiter et al., 2022). In stakeholder-oriented and transformation-focused approaches (Hanger-Kopp et al., 2022) the emphasis shifts from finding 'optimal' solutions to creating robust and resilient strategies. Public funding should also prioritize prevention and preparedness measures and protective infrastructures, rather than focusing on the ex-post compensation of losses (Schinko et al., 2017). Established 'good practices' in adaptation policy making, as reflected in the Austrian national adaptation strategy, aim to enhance feasibility by prioritizing win-win, no-regret, co-benefit, multiple-benefit, or collateral-benefit adaptation outcomes, while realizing potential co-benefits with mitigation. Adaptation measures offering additional societal benefits (i.e., synergetic potentials with the SDGs) are more likely to gain societal and political support. Successful implementation of adaptation measures hinges on the consideration of the diverse impacts on different social groups and their specific needs, as well as affected provisioning systems (Aigner et al., 2023f). Effective adaptation often requires short-term yet foresighted and transformational action to avoid lock-ins and maladaptation risks that can arise from incremental measures, which fail to achieve long-term climate resilience (Lexer and Novy, 2024). Furthermore, transformative adaptation policies can extend the mission of health-care systems to promote human and planetary health. This includes strengthening preventive care, which remains marginal in terms of total health expenditure (0.8 % in 2015) (APCC, 2018), leveraging the co-benefits of time-prosperity (Dengler et al., 2024) and aligning healthcare provision with transport and spatial planning (Weisz et al., 2020).

# Chapter Box 8.1. Perspectives, multi-perspectivity, narratives and qualitative transformation pathways

Perspectives represent fundamentally different approaches to making sense of the world. They may be conceptualized as paradigms (Kuhn, 1967), thought collectives (Fleck, 1980), or epistemic communities. These perspectives are characterized by a set of shared values, interests, mindsets, assumptions, concepts, theories, and methods. They typically adhere to common research programs, agendas, problem diagnoses, target horizons, and design options. At the same time, perspectives tend to resonate with specific socio-cultural groups, facilitating in-group communication while challenging communication across different perspectives. Perspectivism posits that cognition, research, and policymaking, and their underlying narratives are shaped by frames of reference (von Sass, 2019; Aigner et al., 2023e). **Multi-perspectivity** is therefore crucial for bridging such cognitive gaps and to mediate between social groups with different backgrounds (Thompson et al., 1990; Burroughs and Rindfleisch, 2002; Rees, 2010; Amel et al., 2017; Lent, 2017; Kohler et al., 2019; Lenton et al., 2019; Pascual et al., 2021; Wyborn et al., 2021). Cognitive gaps frequently result in structural misunderstandings and the formation of 'echo chambers' in politics, science, (social) media, and the broader public discourse (Abram, 1997; Dittmar et al., 2014; Cinelli et al., 2021; Pascual et al., 2021; Cinus et al., 2022). Multi-perspectivity seeks to integrate different frames of reference, thereby enhancing mutual understanding and expanding the range of possible actions (Novy et al., 2023b). The APCC Special Report on 'Structures for Climate-Friendly Living' (APCC, 2023) introduces four perspectives for assessing structures for climate-friendly living: "From a **Market perspective**, price signals that encourage climate-friendly consumption and investment decisions are central to climate-friendly living. If there are appropriate framework conditions that regulate markets in a climate-friendly way, then the polluter pays principle and true costs contribute to decarbonisation. ... [Focusing on socio-technical renewal,] (t)he **Innovation perspective** focuses on the impact of different forms of innovation and their application on social and economic practices and thus on the environment, on climate (un)friendly living and economic activity. ... From a **Provisioning perspective**, provisioning systems that facilitate sufficiency and resilience practices and forms of life, and thus make them the new status quo, are central to climate-friendly living. The Provisioning perspective is based on a broad understanding of economics, according to which economics concerns the joint organisation of livelihoods. ... From the **Society-Nature perspective**, knowledge about key drivers of the climate crisis (e.g., human-nature dualisms, capital accumulation, social inequality) is essential for climate-friendly living. Theories in the Society-Nature perspective do not consider the social and (biophysical) nature as independent from each other, but as closely intertwined" (Aigner et al., 2023c, p. 9).

Multi-perspectivity facilitates differentiated decision-making processes and enables the identification of synergies and trade-offs. To operationalize the four perspectives, Haas et al. (2023) use the concept of leverage points (Meadows, 1999) – specific points in a system where a small action can produce profound changes. This framework allows for the classification of actions based on their transformative potential. Shallow leverage points focus on altering parameters and feedback mechanisms, while deep leverage points aim to change the overall design and intent of a system, i.e. the direction of transformations (Abson et al., 2017). Based on this distinction, Haas et al. (2023) distinguish four qualitative transformation pathways relevant to Austria. These pathways vary according to their underlying narratives, key actors, and central characteristics:

"(1) Guard rails for a climate-friendly market economy (pricing of emissions and resource consumption; abolition of climate-damaging subsidies, openness to technology) (2) Climate protection through coordinated technology development (state-coordinated technological innovation policy to increase efficiency) (3) Climate protection as public provision (state-coordinated measures to enable climate-friendly living, e.g., through spatial planning, investment in public transport; legal regulations to restrict climate-damaging practices). (4) Climate-friendly quality of life through social innovation (social reorientation, regional economic cycles and sufficiency)" (Aigner et al., 2023c, p. 9).

In contrast, the IPCC highlights two perspectives with differing assumptions and diverging policy proposals, namely ecomodernism and degrowth:

"Two contrasting schools of thought, called **ecomodernism** and **degrowth** (D'Alisa et al., 2014), offer important bounding narratives for 'green economy' approaches that aim achieve the SDGs and Paris Agreement goals. Ecomodernism aims to decouple GHG emissions and other environmental impacts from GDP growth (WGIII Section 1.4.1; Desrochers and Szurmak, 2020) through three primary strategies: (a) 'green' technological innovation, (b) resource efficiency or productivity improvements and (c) the sustainable intensification of land use in both rural and urban areas (Asafu-Adjaye et al., 2015; Isenhour, 2016)... Degrowth proponents question the feasibility of decoupling at a scale and rate sufficient to meet Paris Agreement goals (Kallis, 2017; Parrique et al., 2019; Gómez-Baggethun, 2020; Hickel and Kallis, 2020). Using precautionary principle-rooted arguments (Latouche, 2001), degrowth aims for intentional decreases in both GDP and coupled GHG emissions (Kallis, 2011) using policy mechanisms such a 'cap and share' framework for distributing emissions permits on an annually declining basis with legislation to prohibit the overshoot of established carbon budgets (Douthwaite, 2012; Kallis et al., 2012). Degrowth thus seeks to minimise reliance on negative emissions technologies, such as the large-scale deployment of BECCS... and aims to generate progress toward achieving the SDGs by prioritising redistribution rather than GDP growth" (Ara Begum et al., 2022, p. 173).

Ecomodernism primarily concentrates on optimizing the current system by *improving* energy and resource efficiency. Degrowth, in contrast, advocates for a radical transformation of current systems, their rules and underlying structures, with the objective of *avoiding* harmful activities (Cross-Chapter Box 7).

Section 8.6 explores transformation pathways to achieve climate-friendly living. In general, transformation pathways present narratives on the transition to a future state, in which broader societal goals are accomplished (IPCC, 2022a). This section assesses the potential for transitioning towards climate-friendly living (APCC, 2023) by departing from two ideal-typical transformation pathways that were derived from the IPCC's distinction between eco-modernism and degrowth (Ara Begum et al., 2022) (Chapter Box 8.1), as well as from different perspectives including the Market and Society-Nature perspective (APCC, 2023; Aigner et al., 2023c).

Section 8.6.1 starts by delineating minimum requirements for planning climate-friendly transformations, thereby addressing the politico-economic and socio-cultural inhibitors and enablers identified in Section 8.5. The four minimum requirements for planning climate-friendly transformation pathways are as follows: (1) Broadening climate policies; (2) Adopting climate-friendly governance models; (3) Expanding the mix of instruments, and; (4) Increasing societal and political support (Novy et al., 2024a).

Section 8.6.2 discusses the 'eco-modernist' transformation pathway, starting with its underlying perspectives. The section then examines the pathway's strengths, weaknesses, and potentials with regard to the four minimum requirements for climate-friendly living. Due to 'eco-modernist' reflecting a single thought collective, advancing towards a more climate friendly transformation pathway requires addressing the limitations of eco-modernist and the Market perspective (APCC, 2023).

Section 8.6.3 explores the 'system change' transformation pathway, following the same proceedings as Section 8.6.2. It is argued that for transitioning to climate-friendly living, more pragmatic solutions that address the shortcomings of degrowth and the Society-Nature perspective (APCC, 2023) are required. Doing so can help garner broader societal and political support for the transition.

# 8.6.1. Planning climate-friendly transformations

As ideal-typical transformation pathways are inherently limited to specific perspectives, they tend to resonate primarily with particular socio-cultural groups (Aigner et al., 2023c; Novy et al., 2024a). This is confirmed by mitigation scenarios currently modeled for Austria (Sections 8.4, 8.7), with scenarios combining multiple perspectives demonstrating greater feasibility in achieving climate targets (Section 8.4). Considering various perspectives can thus contribute to "problem diagnoses, target horizons and design options [to] be understood in a differentiated manner, priorities [to] be set in an informed manner and incompatibilities and synergies [to] be identified" (Novy et al., 2023d, p. 196). Multi-perspectivity in scientific discourse and a willingness to compromise within liberal democracies are therefore crucial to achieving climate goals. (Chapter Box 8.1).

Governing multi-perspectivity requires planning as an initial and essential step for coordinated and goal-oriented action (Aigner et al., 2023d; Durand et al., 2023). Planning is already a standard practice across various climate-relevant policy fields, including spatial, land use, transport, urban and energy planning (Getzner and Kadi, 2020; Svanda and Zech, 2023). The EU-mandated National Energy and Climate Plans (NECPs) are designed to facilitate coordination of national efforts in decarbonization, energy efficiency and security. There is furthermore a growing tendency for climate adaptation policies to move away from optimization and single-goal oriented approaches towards more robust strategies that align with the SDGs (Schinko et al., 2017; Schirpke et al., 2023). Despite these efforts, the federal structures in Austria present a significant obstacle to effective planning (Steurer et al., 2023) (Section 6.4.1). Current actions frequently prioritize reactive responses, such as expost natural disaster management, over proactive measures. Improving the use of existing planning instruments and fostering reflexive, climate change education and participatory learning processes can facilitate climate-policy integration (Plank et al., 2021) (Section 6.8.3).

While improved planning does not guarantee success in achieving climate targets, it makes success more plausible by designing and orchestrating bundles of measures that also align with the SDGs. This allows for realizing potential synergetic effects between the SDGs and mitigation as well as adaptation options. Based on the analysis of political-economic and socio-cultural feasibility challenges (Section 8.5), four minimum requirements for planning climate-friendly transformations can be identified (Novy and Barlow, 2022; Riahi et al., 2022; Haas et al., 2023; Heyen and Wicki, 2024).

First, **broadening climate policies** is an effective policy lever to avoid 'silo' approaches to climate policymaking. Climate policies are more feasible if they are part of sustainable development, thereby broadening their scope and objectives (Haas et al., 2023). Thus, climate-friendly transformations integrate not only adaptation and mitigation policies, but also broader socioeconomic policy fields that benefit every-

Minimum requirements for planning climate-friendly transformations	Key politico-economic and socio-cultural framework conditions for enabling climate-friendly transformations		
Broadening climate policies	<ul> <li>Climate policies as part of sustainable development to avoid 'silo' approaches to climate policymaking (Section 8.5.1, 8.5.2)</li> <li>Empowering democratic policy spaces and limiting incumbent business interests (Section 8.5.1, 8.5.2)</li> </ul>		
Adopting climate-friendly governance models	<ul> <li>Linking representative with deliberative and participatory democracy in implementing enforceable administrative rules and provisions (Section 8.5.2)</li> <li>Acknowledging multi-perspectivity in climate research and accepting conflicts and compromises as unavoidable (Section 8.5.1, 8.5.2)</li> </ul>		
Expanding the mix of measures	<ul> <li>Combining mitigation and adaptation measures (Section 8.5.5)</li> <li>Combining efficiency- and sufficiency-measures (Section 8.5.3)</li> <li>Combining incremental and radical measures (Section 8.5.1)</li> </ul>		
Increasing societal and political support	<ul> <li>Highlight the co-benefits of climate action, such as promoting human and planetary health, as well as costs of inaction (Section 8.5.2, 8.5.4, 8.5.5)</li> <li>Fostering new narratives and unconventional alliances that benefit everyday life (Section 8.5.4)</li> </ul>		

Table 8.3	Minimum	requirements f	or planning	climate-friendly	y transformations.
					,

day life. This makes climate measures less susceptible to instrumentalization by incumbent business interests by using war rhetoric. As climate policy integration is much more a question of political priorities and power than of well-designed policies, expanding democratic policy spaces can facilitate climate policy integration by limiting the power of incumbent interests (Bärnthaler et al., 2024) (Section 6.5.3).

Second, adopting climate-friendly governance models can serve as effective by linking climate policy integration with multi-level governance, ensuring climate issues are mainstreamed across multiple levels and all policy areas (Section 6.5.4). Such models can combine representative, deliberative and participatory forms of democracy, while simultaneously employing enforceable administrative rules (Jessop, 2010; Lorek and Fuchs, 2013). It is a minimum requirement to ensure that enforceable administrative rules and provisions remain compatible with safeguarding individual rights (Hollaus et al., 2023; Madner, 2023b). They furthermore facilitate the identification of strategies of how to address power dynamics and resistance in societal, business, and political contexts (Fuchs and Lederer, 2007; Lorek and Fuchs, 2013). Climate-friendly governance models also enable a balance between top-down and bottom-up approaches (Novy et al., 2022; Durand et al., 2023), integrate civil society initiatives (Moulaert and MacCallum, 2019), and public bureaucracies (Kattel and Mazzucato, 2018; Kattel et al., 2022). By connecting social, socio-technical, and system innovations (Köhler et al., 2019; Weber and Kubeczko, 2023), they can help to build, formulate, and negotiate a novel forward-looking social consensus across diverse societal groups. Key instruments include climate change education (Section 6.8.3), citizen (climate) assemblies (Lage et al., 2023), role-play simulations to promote transdisciplinary social learning (Schinko and Bednar-Friedl, 2022), and participatory foresight processes (Weber and Kubeczko, 2023). Well-planned cooperation between various stakeholders at different levels, based on a common narrative of feasible transformation pathways, is therefore essential.

Third, expanding the mix of measures to include adaptation and mitigation, efficiency and sufficiency, and both incremental and radical measures is essential for overcoming delays in climate-action (Sections 6.5.3, 8.5.2). Currently, climate policy is centering around efficiency improvements that mitigate emissions and incremental adaptations, that focus on addressing single risks and lack anticipatory approaches. Both reinforce the status quo (Section 8.5.4, 8.5.5). Transformative pathways can better plan systemic changes by linking measures for long-term mitigation and transformative adaptation. This not only increases feasibility but brings immediate benefits for everyday life, e.g., nature-based solutions to tackle a broader range of impact domains like 'sponge cities' that green neighborhoods (Section 8.3.1, 8.5.5). Thus, combining incremental and radical measures, e.g., radical (Frenken and Punt, 2023) and disruptive innovations (Kivimaa et al., 2021), becomes an effective policy lever. For instance, by combining efficiency-improving and sufficiency-enhancing measures like investing in EV research while restricting car use in city centers (Gough, 2017; Haas et al., 2023; Heyen and Wicki, 2024). Climate-friendly transformation pathways combine directionality, sustainability and inclusivity, a more holistic view on innovation processes, investment in new infrastructures, new markets, and structural conditions supporting skills and capabilities.

Fourth, **increasing societal support**, i.e., societal acceptance that is articulated in policy space, **and political support**, i.e., political will of decision makers, is decisive for the feasibility of climate actions. This support hinges upon the ability to communicate costs of inaction and the advantages of climate action, such as improved human and planetary health (APCC, 2018; Weisz et al., 2020). Building unconventional alliances and new narratives that improve everyday life by supporting transformative adaptations can garner wider support. Examples include the use of CO<sub>2</sub>-tax revenue recycling to funding redistribution to poorer households (Budolfson et al., 2021), to improving public infrastructures (Hausknost, 2023; Lage et al., 2023; Fritz and Eversberg, 2024) or to funding adaptation measures, all of which bear potentials to enhance progress towards the SDGs.

# 8.6.2. Advancing from an 'eco-modernist' to a climate-friendly transformation pathway

# Underlying perspective of an 'eco-modernist' transformation pathway

The 'eco-modernist' transformation pathway is rooted in an ecomodernist approach (Desrochers and Szurmak, 2020), based on the Market perspective (APCC, 2023) and the transformation pathway 'Climate protection through coordinated technology development' (Chapter Box 8.1). By relying on green growth and technological optimism (Asafu-Adjave et al., 2015; Isenhour, 2016; Symons, 2019; Ara Begum et al., 2022), this approach prioritizes market-based governance (Lorek and Fuchs, 2013; Hausknost and Haas, 2019; Bärnthaler, 2023) to achieve the optimal allocation of scarce resources by maximizing consumer utility (Brand-Correa and Steinberger, 2017; Gough, 2017) and profit (Zeller, 2020). Adequate market design, according to this perspective, is a form of 'soft' governance that focuses on pull measures (e.g., subsidies and nudging) to incentivize actors to reduce emissions (Getzner, 2009; Steinberger et al., 2013; Getzner and Kadi, 2020; Bröthaler et al., 2024), switch to new business models (e.g., sharing-economy) (Wieser and Kaufmann, 2023), and invest in low-emission technologies (Kubeczko et al., 2023). Price signals are expected to increase the relative attractiveness of low-emission products and services, guiding consumers towards climate-friendly lifestyles (e.g., denser living) (Novy et al., 2023c). As CO<sub>2</sub> pricing has distributional impacts on the cost of living and production costs, it is implemented cautiously. Greater emphasis is placed on promoting green growth. Accordingly, socio-economic progress and environmental sustainability can be aligned by decoupling GDP growth from emission and resource use, with renewable energy, bioeconomy, genetic engineering, and CCS offering promising short-term solutions (Nordhaus, 2019; Danaher, 2022).

# Advancing to an effective and feasible climatefriendly transformation pathway

#### Broadening climate policies

A key strength of this transformation pathway is its focus on socio-technical innovations and market price mechanisms, alongside measures of social compensation. It insists on the importance of incremental improvements in global governance, such as the commitment made at COP 28 'to work together to triple the world's installed renewable energy generation capacity to at least 11,000 GW by 2030, taking into consideration different starting points and national circumstances' (United Nations Climate Change, 2023). The European Green Deal exemplifies this approach by aiming towards improvements in European competitiveness while promoting the goal of climate neutrality by 2050 (Zeller, 2021). Thereby, facilitating a 'just transition' to 'leave nobody behind' is central, going beyond sole employment creation (McCauley and Heffron, 2018) (Section 6.7.1). This is supported by funds like the Just Transition Fund (JTF) and the Social Climate Fund (SCF). Among the EU-27, Austria leads in terms of renewable job creation with fair conditions (McCauley et al., 2023). It furthermore offers broad incentives for businesses and households, including income-based support for equipment replacement and compensation payments like the climate bonus ('Klimabonus') and repair bonus ('Reparaturbonus'). Introducing green vouchers can achieve similar distributional effect as compensation payments but may reduce potential rebound effects better (Büchs et al., 2021) (Section 6.5.1).

A key weakness of this pathway is its neglect of insights from the Society-Nature perspective (APCC, 2023; Haderer et al., 2023), particularly ignoring society-nature relations, the growth imperative and its exploitative impacts on nature and parts of the Global South (Cross-Chapter Box 7) (Fischer-Kowalski and Haas, 2014; Görg et al., 2017; Hickel, 2020a). Its incremental approach tends to add sustainable technologies to an unsustainable structure (Section 6.5.4) and prioritizes immediate economic goals – such as GDP growth, industrial competitiveness, and public debt reduction – over broader climate goals. Contrary to recommendations by the European Commission (2023b), it tends to disregard long-term sustainable competitiveness and the need for multilateral cooperation for a global just transition (Pollex and Lenschow, 2018). Increased geopolitical rivalries, for instance, have contributed to European policies prioritizing (energy) security and global access to resources over climate targets (Section 8.5.1). This is further reinforced through corporate lobbying to retain climate-damaging subsidies (Brulle, 2018; Brulle et al., 2020; Kletzan-Slamanig et al., 2022; Bärnthaler et al., 2024), increase technology-open innovation funding and favor private green finance via de-risking policies (Kedward et al., 2024).

A lever of this pathway to contribute to transitioning towards climate-friendly living consists thus in better alignment with the multilateral sustainable development agenda, linking ambitious decarbonization targets with social targets for decent living (Clarke et al., 2022; Otto, 2023; UNEP, 2024).

## Adopting climate-friendly governance models

A key strength of the 'eco-modernist' transformation pathway is its use of well-established policy approaches focusing on incentives and efficiency improvements. Proposed measures in this transformation pathway are relatively easy to implement as they do hardly change existing framework conditions. As these measures result in minimal behavioral changes (e.g., e-fuels in comparison to shifting towards public transport), they tend to lead to weak institutional resistance which increases the feasibility of their implementation (Section 8.5.3).

However, its reliance on market and technology solutions is also a major weakness due to widespread market failures and institutional lock-ins in using existing technologies (Hausknost and Haas, 2019; Schulze Waltrup, 2023). Market-based governance, focusing on voluntary, incentive-based measures, as often promoted by Austrian social partnership and federal decision-making structures, can contribute to climate delay (Section 8.5). The current Science, Technology, Innovation (STI) policy focus tends to prioritize technological efficiency (Weber and Kubeczko, 2023). As an 'eco-modernist' pathway fails to challenge the concentration of power by incumbent business interests, it tends to underestimate the power of reactionary forms of climate delay strategies to inhibit climate policy integration (Bärnthaler et al., 2024) (Section 6.5.3). Incumbent interests tend to instrumentalize climate science in a culture war-rhetoric that deviates public debate to conflictive socio-cultural issues and trigger points (Mau et al., 2023). This often impedes collective learning processes of how to organize necessary transitioning processes (Jaeggi, 2014). If the power of incumbent market actors that benefit from climate delaying tactics remains unchecked, the ability of public decision-makers to use national and international policy spaces remains restricted (Section 8.5.2, 8.5.4).

The full potential of this pathway can be actualized by acknowledging that conflicts and compromises are inherent in pluralist societies and liberal democracies (Mouffe, 2005). Decision-makers are required to balance present and future interests and freedoms, including a duty to protect the public from the consequences of inaction (Hollaus et al., 2023). The feasibility of implementing enforceable administrative rules and provisions can be increased by linking representative with deliberative and participatory democracy. Transformative innovation policies (OECD, 2024), informed by the socio-technical Multi-Level-Policy (MLP) theory and other sustainability transition approaches (Geels, 2011; Köhler et al., 2019) increasingly focus on system innovations (Weber et al., 2008). These policies promote mission-oriented instruments (Kubeczko et al., 2023) and an 'entrepreneurial state' (Kattel and Mazzucato, 2018; Kattel et al., 2022). Organizational frameworks may include 'realigned strategic niche management' (Geels and Raven, 2006), 'technological innovation systems' (Markard and Truffer, 2008; Andersson et al., 2023), 'challenge-oriented regional innovation systems' (Tödtling et al., 2022) or 'regional transformation management' (Lexer and Novy, 2024). Meta-governance involving both private and public actors (e.g., through road-mapping with participatory foresight) can steer research and innovation by means of regulation (e.g., green and innovation-oriented procurement), experimental testing (e.g., with the help of real-world laboratories and regulatory sandboxes) and monitoring and evaluation (Weber and Kubeczko, 2023; Bauknecht and Kubeczko, 2024).

#### Expanding the mix of measures

A key strength of this pathway is that people have experience with proven and tested mitigation and adaptation measures which rely on socio-technical solutions that are reliable in implementing known goals, such as thermal rehabilitations, dams and flooding zones. Reoccurring planning cycles (e.g., reinvestment cycles for public infrastructure, budgetary planning cycles) can enable the integration of climate risk assessments and adaptation funding, such as the European Structural and Investment Funds' 30 % spending target on climate action (European Commission, 2024). Additionally, extreme weather events may increase urgency and political will, creating additional windows of opportunity for adaptation (Section 8.5.5).

A major weakness of this pathway is its reliance on soft governance and a limited set of primarily efficiency-enhancing incremental mitigation measures, along with informational strategies to nudge consumer behavior, autonomous adaptation, and 'piggy-backing' onto existing measures. 'Soft' measures only include voluntary measures and market-based instruments that do not constitute deep interventions in the economic structure of society or in consumer practices (Section 6.4). However, incremental changes alone are insufficient to achieve "fundamental change in society" (Ara Begum et al., 2022, p. 171) (Section 5.3.1). Adaptation policies in this pathway are often reactive, responding expost to issues as they arise (e.g., post-disaster recovery). They typically focus on specific sectors, such as flood risk management, agriculture, or other high-risk domains, and emphasize structural protection measures that defend existing - often maladaptive - development pathways against changing climatic and environmental conditions. This approach tends to neglect cases of overload and increasing residual risk (Hartmann et al., 2021) and limits the scope of adaptation measures, as they do not address broader systemic vulnerabilities across various domains. Ultimately, this may lead to maladaptation, if adaptation in one sector (e.g., irrigation in agriculture) amplifies risks in other sectors (e.g., increasing water stress) (European Court of Auditors, 2024). Furthermore, as private insurance against climate risks is central to this approach, adaptation policies often remain confined to incremental improvements, failing to address underlying vulnerabilities emerging risks adequately (Section 8.5.5).

The potential of this pathway is actualized by implementing a better orchestrated mix of measures in line with multi-perspectivity (Novy et al., 2023e) (Chapter Box 8.1). Lawmakers and public administrations can improve climate policy integration by leveraging their regulatory power to elaborate effective and enforceable bundles of measures, including phasing-out emission-intensive technologies, business models, and practices (Madner, 2023b). Bundles of measures that combine incremental and radical measures can drive change across multiple levels. By experimenting with social and socio-technical innovations across policy areas, such an approach can achieve effectiveness already in the short-run by promoting diverse incremental changes without abandoning the transformative long-term goal to overcome environmental degradation and foster inclusive socio-economic renewal, resilience, and security (European Commission, 2023c; Bauknecht and Kubeczko, 2024; OECD, 2024; ÖROK, 2024).

An additional key lever that can be implemented in the short-run to increase pathway effectiveness is combining efficiency-enhancing with cost-saving sufficiency measures for rapid, low-cost emission reductions (e.g., speed limits). To "disrupt existing developmental trends" (Pathak et al., 2022, p. 72) that impede climate-friendly living, measures that avoid or shift emissions can complement efficiency-enhancing strategies.

#### Increasing societal and political support

A key strength of the 'eco-modernist' transformation pathway is its reliance on existing preferences, values, and activities, such as strong support for green products and technologies (Schleer et al., 2024). It furthermore relies primarily upon pull-measures that tend to encounter broad societal and political support. This pathway mainly resonates with the middle- and upper-class milieus - the Established, Performers and the Cosmopolitan Avant-Garde - who are inclined towards technological innovations as well as voluntary, quality-of-life-enhancing behavioral changes (Barth and Molina, 2023; Borgstedt, 2023; Schleer et al., 2024). This is both a strength and a weakness. It is a strength because it garners political support from powerful actors, but a weakness due to opposition from lower-income groups and the Nostalgic milieu, who are concerned about affordability and access to social infrastructure.

This pathway can reach its full potential by expanding its appeal to other milieus, especially to the middle-classes like the Adaptive-Pragmatic milieu (Barth and Molina, 2023; Borgstedt, 2023). Clear communication of individual benefits (Barth and Molina, 2023; Borgstedt, 2023), including greener neighborhoods and reduced costs of living, could attract allies. Such alliances, however, require narratives to include perspectives that are agnostic of green growth ('a-growth'), such as Innovation perspective (APCC, 2023; Kubeczko et al., 2023) and Green Keynesianism (Gough, 2017; Tienhaara and Robinson, 2022; Schulze Waltrup, 2023). Institutionalizing such alliances might lead to a renegotiated social contract for sustainable and inclusive development (European Commission, 2023a; UNEP and Stockholm Environment Institute, 2023).

# 8.6.3. Advancing from a 'system change' to a climate-friendly transformation pathway

# Underlying perspectives of a 'system change' transformation pathway

The 'system change' transformation pathway is grounded in a degrowth approach (Cross-Chapter Box 7) and the Society-Nature perspective (APCC, 2023; Novy et al., 2023d) that advocates for avoid and shift measures (Cross-Chapter Box 4). It emphasizes that the planet's finite ecosystems cannot support endless growth in material consumption and emissions (Vogel and Hickel, 2023). Consequently, some economic sectors are required to shrink (e.g., the automotive sector), while others need to expand (e.g., the care and repair sector) (Krisch et al., 2020; Mattioli et al., 2020; Pichler et al., 2021; Keil and Steinberger, 2023).

The 'system change' transformation pathway argues that climate-friendly living requires overcoming the 'status quo bias' and proposes sociotechnical and socioeconomic transformations to reduce resource demand and lower emissions (Durand et al., 2023). It envisions an economy oriented towards human flourishing (eudaimonia) (Brand-Correa and Steinberger, 2017) and respecting planetary boundaries (O'Neill et al., 2018) (Section 8.2.1). This can be achieved by shifting economic activity towards providing (i.e., producing, consuming and distributing) the infrastructure, services, and goods necessary for a good life (Polanyi, 1977; Jo and Todorova, 2019; Novy et al., 2023a) (Cross-Chapter Box 4).

Over the past 200 years, the limits of sustainable use of ecosystems have been continuously surpassed (Rockström et al., 2023), with Austria being among the affluent countries with high and increasing ecological overshoot (Section 8.2.1). Rising global GHG emissions and material consumption reveal the persistence of a harmfully expansive global socioeconomic system. As problematized by the Society-Nature perspective (APCC, 2023), structural factors such as the growth imperative, commodification and the modern extractivist and exploitative societal relationship with nature (Haderer et al., 2023) contribute to this development. Empirical evidence furthermore points towards absolute decoupling rates of emissions from economic growth being anywhere near the speed and scale required to achieve even less ambitious climate goals globally (Haberl et al., 2020; Vogel and Hickel, 2023) as well as in Austria (Aigner et al., 2023e).

# Advancing to an effective and feasible climatefriendly transformation pathway

### Broadening climate policies

A key strength of the 'system change' transformation pathway is its integrated and comprehensive approach to the climate crisis by embedding it in the more comprehensive doughnut model (Raworth, 2017; O'Neill et al., 2018) that shows that Austria is not only characterized by a high, and increasing ecological overshoot, but also by hardly improving its social indicators (Section 8.2.1). Investigating society-nature relationships stresses the importance of power and policy priorities (Section 6.5.3), demonstrating that environmental degradation results from politico-economic processes of extractivism and unequal ecological exchange (Gorz, 1993; Brand et al., 2021; Fanning and Hickel, 2023; Fraser, 2023; Otto, 2023; UNEP, 2023) (Section 8.5.1). Transformative adaptation measures, prioritizing structural and systemic changes, can reduce exposure and vulnerability by promoting public investment in nature restoration and stronger social safety nets, complementing voluntary emergency services like fire brigades and neighborhood initiatives. This pathway seeks deep transformations (Meadows, 1999), questioning the targets and design of current socioeconomic systems. Key aims include transforming the dominant mode of living and production (Brand and Wissen, 2021, 2024; Hickel, 2021; Buch-Hansen and Nesterova, 2023), addressing unequal ecological exchange (Hickel, 2020b), ending neoextractivism (Canterbury, 2018) and land grabbing (Yang and He, 2021; Gabor and Braun, 2023). Moreover, given huge inequalities with respect to historical emissions (Section 8.2.2), this pathway advocates for increased international support and compensations from high- to low-emitting countries (Hickel, 2020b) (Section 8.5.1).

While there are scientific proposals and best practices of effective measures for transforming the socioeconomic system in a climate-friendly way (Schneidewind and Zahrnt, 2013; Die Armutskonferenz and Ökobüro, 2024; Sachverständigenrat für Umweltfragen, 2024a), a key weakness of this transformation pathway remains its lack of practical experience in shifting away from the current institutional order. These proposals often focus on changing parameters (e.g., taxes and incentives) and the design of provisioning systems in Austria (Haas et al., 2023). Limited experiences exist in changing the 'intent', i.e., the direction of change, systemic logics, and organizing principles (Meadows, 1999; Abson et al., 2017). Examples include the lack of proposals of how to reduce Austria's dependence on global locational competition and the resultant vulnerabilities of its export economy (Streeck et al., 2016; Kapeller and Hubmann, 2023). Instead, the status quo incentivizes free riding and climate delay, including social and ecological dumping, thereby favoring short-term incumbent business interests (e.g., the automotive industry or fossil industry) (Pichler et al., 2021; Bärnthaler et al., 2024). Additionally, while reduced international trade can facilitate long-term self-sufficiency of the Global South (Hickel, 2021; Gabor and Sylla, 2023), it may have short-term adverse effects on export-dependent economies due to reduced markets for raw material exports (Graebner-Radkowitsch and Strunk, 2023).

The 'system change' pathway can advance towards a climate-friendly transformation by exploring the advantages of sufficiency and peaceful international cooperation. Sufficiency approaches can increase resource independence and thus foster strategic autonomy in times of geopolitical rivalries (Durand et al., 2024) (Sections 6.6.2, 8.5.1). Furthermore, export-oriented Austrian economy benefits from peaceful international cooperation, including accepting the principle of 'common but differentiated responsibilities' (UN General Assembly, 1994; Fisher, 2015) that could sustain a fair global economy and facilitate access to green energy imports.

#### Adapting climate-friendly governance models

A key strength of this pathway is its focus on overcoming the 'status quo' bias that preserves climate-harmful structures or leads to maladaptation. Identifying and addressing the structural causes of climate delay is a decisive starting point for deep transformations (Buch-Hansen et al., 2024; Spash, 2024), directing efforts toward structural changes. Examples include limiting the power of incumbent business interests (Geels, 2014; Bärnthaler et al., 2024) and exnovations (David, 2017; Krüger and Pellicer-Sifres, 2020; Kivimaa et al., 2021). By highlighting how difficult it is to live climate-friendly in Austria (Section 8.5), this pathway emphasizes the importance of collectively changing framework conditions (e.g., laws, institutions, or infrastructures), as well as gender, income and production relations (Shove et al., 2009; Stoddard et al., 2021; Aigner et al., 2023c; Smetschka et al., 2023; Aykut et al., 2024).

A key weakness of this pathway is the lack of implementable transition strategies (Barlow et al., 2022). By means of advertising and lobbying powerful incumbent actors tend to stabilize the status quo (Lorek and Fuchs, 2013; Avelino and Wittmayer, 2016). Often democratically legitimized decision makers, supported by the electorate, block changes and defend unsustainable consumption patterns if they endanger unsustainable routines and the profit interests of power complexes (Baines and Hager, 2023). Integrating elements of representative, participatory, and deliberative democracy can help reduce public opposition, especially if supported by appropriate communication and scientific support (Hausknost et al., 2017; Gough, 2022; Heyen and Wicki, 2024) (Section 8.5.3, 8.5.4).

To advance towards a climate friendly transformation pathway the 'system change' pathway can facilitate governance approaches based on a 'whole of government' approach that combine top-down and bottom-up approaches by linking sectoral, spatial, and macroeconomic planning (Stöhr and Taylor, 1981; Friedmann, 2005; Adaman and Devine, 2017; Gabor, 2020; Durand et al., 2023; Schmelzer and Hofferberth, 2023; Aoki et al., 2024). This facilitates overcoming 'silo approaches' to climate policymaking as well as localism that solely relies on civic bottom-up initiatives (Kazepov et al., 2019; Krähmer, 2022).

The advantage of public measures 'from above' is that public decision makers - such as governments, parliament, courts, or administrations - have the resources (due to taxation and personnel) and the constitutional legitimacy (due to legal prerogatives) to change framework conditions. Becoming 'slower by design, not disaster' (Victor, 2019) requires planning and governance models that ensure societal acceptance (via public participation and providing desired outcomes with benefits for everyday life) as well as the integration of experts in public deliberation (via policy learning processes, transdisciplinary research and climate change education) (Lorek and Fuchs, 2013; Aigner et al., 2023e; Durand et al., 2023; Nohrstedt and Parker, 2024) (Section 6.8.3). Effective strategies require enforceable rules (Jessop, 2010; Hollaus et al., 2023), including a binding adaptation framework. Experience from climate councils shows that citizens often support more disruptive sufficiency measures in deliberative processes, even though these tend to be underrepresented in policymaking (Lage et al., 2023).

The advantage of bottom-up initiatives, in contrast, is that they can connect multiple levels and diverse actors, fostering broader societal support. Such bottom-linked actions can pool resources, represent collective political interests, and facilitate the cooperation with public policy institutions (Moulaert and MacCallum, 2019), including workers in planning processes and abolish workplace hierarchies (Harnecker and Bartolomé, 2019; Sorg, 2023). Municipal enterprises (e.g., 'Stadtwerke') can become central due to their potential to democratize the provision of essential infrastructures, such as energy and mobility (Foundational Economy Collective, 2018).

### Expanding mix of measures

A key strength of this transformation pathway is that it adopts a broader perspective of climate policy integration. This pathway promotes transformative adaptation, advocating for spatial planning reforms that limit land use, soil sealing ('land use sufficiency'), and promote ecosystem-based adaptation through green spaces along with stricter monitoring, evaluation and learning processes. It proposes sufficiency as the guiding principle (i.e., the 'intent' of climate-friendly systems), thereby addressing inequalities in material and energy consumption as the main drivers of climate change (Buch-Hansen and Nielsen, 2020; Spash, 2024). By prioritizing avoid over shift and improve measures (Princen, 2003; Fuchs et al., 2023; Gough, 2023), this approach enables immediate emissions reductions while satisfying human needs. Deeply rooted intersectional inequalities of class, gender, and ethnicity are also key drivers of climate change on a global (Chancel, 2022; Chancel et al., 2022; Millward-Hopkins and Oswald, 2023) and national level (Theine et al., 2022; Essletzbichler et al., 2023). A key weakness of this pathway is that sufficiency measures are instrumentalized as trigger points (Mau et al., 2023) in a culture war rhetoric, framing climate policy as threats to 'normal' life, freedom, and prosperity. Due to the currently low level of acceptance of measures that oppose 'everyday imaginaries' (Kaika et al., 2023) and interfere with the 'imperial mode of living' (Brand and Wissen, 2021), even incremental measures like speed limits face opposition and could lead to the broader rejection of climate policy (Stagl et al., 2024).

An important lever for climate-friendly transformation pathways is embracing multi-perspectivity (Novy et al., 2023d), as this facilitates the effective mediation between conflicting claims and seemingly incompatible worldviews (Meadows, 1999; Christopher et al., 2009; Rees, 2010; Abson et al., 2017; Lent, 2017, 2021; Kohler et al., 2019; Pascual et al., 2021, 2023). This helps to vindicating both incremental and radical change. While small steps tend towards ineffective activism within given framework conditions, radicalism often lacks anchoring in the current institutional framework, which is a necessary starting point for transitioning to a new system (Gorz, 1968, 1993; Novy et al., 2022; Soder, 2024).

Combining efficiency-enhancing with sufficiency-enhancing, and pull- with push-measures offers potential for win-win solutions (Heyen and Wicki, 2024). Expanding PV systems does not conflict with reducing energy demand, nor does increasing EVs conflict with collective forms of provisioning that satisfy needs with lower energy requirements (Creutzig et al., 2018; Vogel et al., 2021) (e.g., by reducing car-dependency) (Mattioli et al., 2020). Examples include the provision of universal basic services (Coote and Percy, 2020), commons (Edelmann et al., 2020; Villamayor-Tomas and García-López, 2021) and social-ecological infrastructures (Bärnthaler et al., 2021; Novy et al., 2023b). Additionally, services and goods can be subsidized for low-income households while expanding foundational 'blue' and 'green' infrastructures in neighborhoods (De Castro Mazarro et al., 2023).

Taking sufficiency as the organizing principle (Princen, 2003; Sachverständigenrat für Umweltfragen, 2024b) and a collective endeavor (Schneidewind and Zahrnt, 2013; Jungell-Michelsson and Heikkurinen, 2022; Lage, 2022) can secure that everyone has 'enough' by means of systemic solutions, e.g., by providing existential services and goods in an accessible and sustainable way, and avoid individualizing and moralizing responsibilities (Sachverständigenrat für Umweltfragen, 2024b). Collective sufficiency is operationalized trough production and consumption corridors, which establish 'minima' for decent living standards and 'maxima' to avoid over-production and -consumption (Pirgmaier, 2020; Bärnthaler and Gough, 2023). Stressing 'enough' as a 'floor' that ensures affordable and sustainable foundational infrastructures, services, and goods for all residents (e.g., housing and energy) can reduce resistance (O'Neill et al., 2018; Millward-Hopkins et al., 2020; Vogel et al., 2021). 'Ceilings', in contrast, can limit overconsumption by a few (Lehtonen and Heikkurinen, 2022; Gough, 2023), e.g., by imposing bans or other regulations. Redistributive measures, such as progressive income and wealth taxes or fees for high energy and resource use (Millward-Hopkins and Oswald, 2023; Chancel et al., 2024), can be justified on fairness grounds and increase financial leeway for public investment (Bröthaler et al., 2023).

## Increasing societal and political support

A key strength of this pathway is its focus on the good life in its multiple dimensions (O'Neill et al., 2018; Brand et al., 2021). First, it focuses on fairness and justice (Bal et al., 2023; Otto, 2023), radicalizing insights from 'just transitions' (Büchs and Koch, 2019; Koch, 2020; Armutskonferenz et al., 2021; Koch and Buch-Hansen, 2021; Hirvilammi et al., 2023; Schneider, 2023). By addressing distributive (e.g., consumption corridors) and relational forms of justice, a 'system change' pathway can ensure an equitable transition to climate-friendly living. It can integrate procedural justice (e.g., in forms of participatory governance), recognitive justice (concerned in particular with groups at risk of falling behind) and, finally, international justice (related to respectful treatment) (Bal et al., 2023). Second, it promotes place-based development initiatives, often focusing on transformative adaptation. This can facilitate equal access to infrastructures and can foster social cohesion through communal cooperation and bottom-up solidarity (Moulaert and MacCallum, 2019; De Castro Mazarro et al., 2023; Kaika et al., 2023). Successful examples include the commons (Edelmann et al., 2020; Villamayor-Tomas and García-López, 2021), the foundational economy (Bärnthaler, 2023; Calafati et al., 2023), the social and solidarity economy (Exner and Kratzwald, 2021) the doughnut economy (Raworth, 2017; Wahlund and Hansen, 2022) and the economy for the common good (Goydke and Koch, 2020). In urban development, improvements in social services, accessible retailing, existing public spaces, urban renaturation, building retrofitting and shared housing can benefit everyday life. The city of Vienna is discussed as a - contested best practice model for inclusive transformation (Mocca et al., 2020; Bärnthaler et al., 2023; Haderer, 2023; Novy et al., 2024b). In regional development, experiments include revitalizing village centers (Madner and Grob, 2019), integrated land-use planning to safeguard ecosystem services, increasing land taxes, and urban and regional contractual development agreements to fund multi-story (social) housing and urban infrastructures (Madner and Grob, 2019; Getzner and Kadi, 2020; Schirpke et al., 2023).

The key weakness of this pathway resides in limited political support, as it challenges the advantages of the status quo, be it incumbent interests of residents, business, or politicians. Restricted political support requires even stronger societal support to pressure for climate-friendly measures. Today, however, effective climate policies are primarily supported by parts of the socio-cultural milieus of Post-Materialists and Progressive Realists, who together represent less than 20 % of the population (Barth and Molina, 2023; Borgstedt, 2023; Schleer et al., 2024). Although these groups are willing to reduce their individual footprint, voluntary efforts alone - without comprehensive and democratically decided rules - do not impact on the practices of other socio-cultural milieus. At the same time, large sections of society feel insecure and unprotected due to labor market flexibilization and eroding social security systems (Schneider, 2023). Fears of economic decline and not having 'enough' tend to overshadow concerns about climate change (Mau et al., 2023) (Section 8.5.4). Lower-class milieus, such as the Traditional milieu, tend to be skeptical about social changes and oppose transformative measures, due to a perceived lack of fairness in policymaking (Barth and Molina, 2023; Fritz and Eversberg, 2023).

A lever for advancing towards climate-friendly living lies in fostering unconventional multi-milieu and multi-class alliances based on fairness, justice and common interests, including short- and long-term benefits for everyday life (Fritz and Eversberg, 2023, 2024). Prioritizing spatial planning measures that reduce risk exposure, can score on societal support for widely popular nature-based solutions and ecosystem-based adaptation in both rural contexts (e.g., natural retention, groundwater renewal areas, cool air corridors) and cities (e.g., blue, green spaces). The Adaptive-Pragmatic milieu as well as parts of traditional lower-class milieus can be attracted to this alliance by measures that reduce costs of living, e.g., subsidizing renewal energy installation or public transport. Societal support can be further enlarged by improving working conditions, a job guarantee, fair wages (Gough, 2017; Schulze Waltrup, 2023) and by providing foundational social and green infrastructures (Foundational Economy Collective, 2018; Millward-Hopkins et al., 2020; De Castro Mazarro et al., 2023; Kaika et al., 2023). Targeting middle- and lower-income groups with explicit eco-social policies to reduce inequalities and enhance well-being (Hirvilammi and Koch, 2020; Koch, 2022) can also help countering skepticism towards democratic institutions in these milieus (Zandonella, 2022). Raising confidence in public institutions in uncertain times by maintaining secure and affordable public provisioning can reduce resentments towards democracy, science, and climate policies (Ter Mors et al., 2012; Bal et al., 2023).

Esther Blanco, Michael Getzner, Elisabeth Gsottbauer, Michael Kirchler, Daniela Kletzan-Slamanig, Michael Miess, Irene Monasterolo, Asjad Naqvi, Andreas Novy

# What is green finance?

Broadly, green finance is seen as a channel that can allow economies to achieve climate goals such as net zero emissions as defined by IPCC targets and Nationally Determined Contributions (NDCs). This transition away from carbon-intensive sectors into 'green' sectors requires redirecting investments while also ensuring growth, sustainable jobs, and competitiveness. Thus, green finance encompasses a broad range of financial activities, policies, and climate objectives, which can vary significantly across institutions, regions, and stakeholders. This can range from financing low-carbon transitions through infrastructure projects such as renewable energy to climate adaptation through support for vulnerable communities. Furthermore, the question whether financing itself should come from the private or public sectors or enabled through market (e.g., carbon taxes) or non-market instruments (e.g., regulations) is widely debated. In practice, a mix of public and private financing, or 'blended finance', and a combination of policies is typically implemented. Otto et al. (2020) highlight that finance is one of the key pillars that can induce positive social tipping dynamics to enable a global transformation to carbon neutrality. This could be achieved by employing different instruments such as green bonds, loans, and climate investment funds. Governments also play a crucial role in this process by creating enabling policies, offering incentives, and providing public funding to attract and de-risk private investments. Given that public spending constitutes a significant share of GDP in EU countries, including Austria, the role of the state in driving green finance and fostering systemic change is substantial.

Climate change also poses significant risks to economies and financial systems, broadly categorized into physical risks, which arise from the direct impacts of extreme weather events and long-term environmental shifts, and transition risks, which stem from the economic and financial adjustments required to shift to a low-carbon economy. While on the one hand, there is ample evidence that the Costs of Inaction (COIN) (see Section 6.7.2) will result in higher damages, on the other hand, rushed policies and rapid structural transitions can result in a costly disorderly' transition (Battiston et al., 2017; Campiglio et al., 2018; Dunz et al., 2021) and are also a rising concern for financial supervisors as they can exacerbate economic and financial risks (Battiston et al., 2017; Mercure et al., 2018). In the context of Austria, rising physical risks, such as increasing flood events, heatwaves, and damages to critical infrastructure, add significant financial challenges as more resources need to be allocated towards adaptation and mitigation measures. While insurance, whether private or public, can help improve resilience of households and businesses, this sector is also likely to face additional challenges as rising physical risks might reduce its capacity to absorb losses, providing additional challenges to policymakers if sectors of the population or sets of economic activities become uninsurable.

Climate policies also need to factor in lock-ins and path dependencies of firms as regulatory risks arising from changing policies may be substantial (van der Ploeg and Rezai, 2020) and may have negative consequences on financial markets (Daumas, 2023). Thus, a stringent regulatory framework for sound climate policies is needed that also ensures financial market stability (Roncoroni et al., 2021), and accounts for forward-looking financial planning to ensure that investments move away from industries that are incompatible with decarbonization targets and the broader socio-ecological transformation (Jones, 2015; Polzin, 2017; Steininger et al., 2022). Recent Euro area financial climate stress test studies show (Alogoskoufis et al., 2021; Gourdel et al., 2022) that an orderly transition to a carbon-neutral economy can achieve early co-benefits by reducing carbon emissions (12 % less in 2040 than in 2020) while a disorderly transition worsens the economic performance and financial stability of the euro area and with physical risks leading to real GDP decreases of 12.5 % in 2050 relative to an orderly transition. A disorderly transition can also introduce new risks for the Austrian economy that is deeply integrated along finance and physical value chains within Europe (Battiston, 2019; Guth et al., 2021; Monasterolo et al., 2022a) yet a recent survey of Austrian finance stakeholders concludes that the integration of green finance in climate policies is not yet sufficient (Kletzan-Slamanig and Köppl, 2021).

# The role of green investments

The development of green financing instruments is closely linked to the investment gap that needs to be bridged in order to reach the climate policy objectives (see 6.7.1). Strategies and instruments are being developed and applied with the aim of redirecting (private or institutional) financial flows towards sustainable and/or climate friendly investments. A precondition for achieving this is developing a regulatory framework for financial markets and the business sector that ensures the proper consideration of climate-related risks (physical, transitional, and regulatory) and the implementation of adequate risk management measures in financial institutions. Beyond finance-specific measures, broad climate-economic strategies also play a critical role in improving the competitiveness of green technologies over incumbent ones. Measures such as removing subsidies to dirty technologies or sectors, carbon pricing, subsidies for green innovation, or similar policies can help level the playing field by shifting the financial flows away from fossil fuel-intensive to renewable technologies. In addition, provisions are required to ensure the environmental and ethical soundness of investments and prevent greenwashing (Kletzan-Slamanig and Köppl, 2022).

Following the Paris Agreement, a series of international initiatives, such as the Central Bank Network for Greening the Financial System (NGFS), have come into effect that also aim to comprehensively disclose climate-related financial risk and to define standards for green finance instruments and products. In the EU, the sustainable finance framework including the Taxonomy on Sustainable Finance, the Green Bond Standard, Corporate Sustainability Reporting Directive, and the European Sustainability Reporting Standards, provide a classification of economic activities that inter alia contribute to climate mitigation and define reporting requirements for companies and financial market participants (Getzner, 2022; Nykvist and Maltais, 2022). These EU Directives are in the process of being transposed into national law and thus will substitute voluntary practices with legally binding requirements, that will also improve data transparency. In Austria, the members of the Green Finance Alliance voluntarily agreed to decarbonize their core business until 2050.

Studies indicate a clear appetite for ESG investments among (retail) investors (Delsen and Lehr, 2019; Seifert et al., 2022; Gutsche et al., 2023). Although 17 % of investment funds held by households are already certified with the Ecolabel (Cetković and Zhan, 2022), the proportion of all other financial products is still only in the low single-digit percentage range (Breitenfellner et al., 2020). Even though Austrian investors are paying more attention to climate risks, the understanding and perception of climate-related financial risk vary across financial actors (Glas et al., 2020; Breitenfellner and Kariem, 2024). The role of disclosure, of policy coherence and social engagement are perceived as a main driver to decrease market uncertainty (Glas et al., 2020) and indeed financial actors regard the provision of specific ratings or labels for green and high-carbon activities as an effective approach to increase transparency (Kletzan-Slamanig and Köppl, 2021). Climate-financial risk assessment is crucial to inform a revision of investors' risk management and portfolio rebalancing, to reduce the risk of stranded assets for financial stability. It is also relevant for firms' investment decisions in the transition via the interest rates. Since climate change policies are forward looking (Battiston, 2019), they also need to account for climate-related uncertainties such as fat-tailed events (Weitzman, 2009; Ackerman, 2017), non-linearity and tipping points in both climate systems (Steffen et al., 2018; Lenton et al., 2019) and financial networks (Battiston et al., 2021), climate scenarios play a key role for climate financial risk assessment. Nevertheless, the current climate scenarios (NGFS) recommended to regulators to perform climate financial risk assessment have important limitations in terms of granularity, coverage, and financial relevance (Monasterolo et al., 2022b; Ranger et al., 2022).

# Challenges of green finance

Taxonomies and labels for 'green finance' vary considerably across countries, institutions, economic sectors, and products. This range of classifications can result in greenwashing with negative effects on consumer welfare (Fatehi et al., 2023), can create reputational risks for companies (Monasterolo and Volz, 2020), and are a serious concern for financial institutions (De Freitas Netto et al., 2020; Schumacher, 2020; Breitenfellner and Kariem, 2023). For Austria, Seifert et al. (2022) find in a survey that additional information on financial and environmental impacts of investments can stimulate participants' willingness to invest sustainably with higher confidence and also shield consumers from greenwashing.
The perceptions and preferences of financial professionals regarding their attitudes towards climate emergency and climate policy are important aspects of whether and how green finance instruments will be used in practice (Gsottbauer et al., 2024). Investors are aware of climate risks to their assets, but they address those risks mainly through more engagement in ESG-related investment instead of divestment from carbon-intensive industries (Krueger et al., 2020), which might be driven by financial concerns and diversification considerations.

Another major challenge is that the field of finance might be unable to appropriately take into account ecological tipping points (FAO, 2021; Eberle et al., 2023; Ripple et al., 2023) and that in general the strategy of expanding the financial system to include (parts of) ecosystems as financial assets might be unsuitable to conserve and restore these endangered ecosystems (Kemp-Benedict and Kartha, 2019; Spash and Hache, 2022). How much of nature in the form of biodiversity, land, emissions, water, resources, etc., should be under the management and control of financial markets as financial assets remains unclear (Sullivan, 2018; Miess, 2023).

## 8.7. Synthesis of policy-relevant insights of transformation pathways

Section 8.7 links policy-relevant insights from the scenario assessment (Section 8.4) and the four minimum requirements for climate-friendly transformation pathways (Section 8.6). The minimum requirements (Section 8.6.1) indicate necessary actions to overcome limitations of ideal-typical transformation pathways that are based on restricted perspectives, such as eco-modernization (Section 8.6.2) and system change (Section 8.6.3). By doing so, they can support broader societal alliances by minimizing societal and political resistance to climate actions. Although Chapter 8 does not propose policy-prescriptive pathways, it provides policy-relevant insights that support policymakers in taking informed and democratically legitimized decisions. Section 8.7 illustrates how adopting a multi-perspectivity approach can help democratic policymakers to improve their respective value- and interest-based decisions, which might otherwise rely on a single perspective.

Taking multiple perspectives into account broadens the range of available policy tools and creates more opportunities to secure societal and political support. Section 8.6 demonstrates that **bundles of measures combining insights from different perspectives tend to be more effective than ideal-typical pathways** focusing solely on specific measures, such as demand reductions or efficiency improvements. This insight is reinforced by scenario modelling, especially by the NetZero2040 scenarios.

Two of the four NetZero2040 scenarios align closely with the two ideal-typical transformation pathways (Section 8.6). The NetZero2040 high-import/high-demand scenario resembles an eco-modernist transformation pathway relying mostly on market- and technology-based solutions. This pathway, however, would require dramatic price increases in energy and carbon pricing along with expansions of wind and solar infrastructure, which could erode societal support. The NetZero2040 low-import/low-demand scenario, in contrast, aligns with a degrowth transformation pathway requiring reductions in per capita housing space and new constructions levels – measures unlikely to encounter societal support in the short-run. While these ideal-typical scenarios appeal to specific socio-cultural milieus, they lack broader societal support. The other two NetZero2040 scenarios also fail to garner societal support, as they do not align with either the eco-modernist or system change perspective. This underscores the importance of integrating diverse perspectives to foster societal alliances for climate action.

Based on the analysis of enablers and inhibitors of climate-friendly transformations (Section 8.4), the NetZero2040 high-import/low-demand and the NetZero2040 low-import/high-demand scenarios are techno-economically feasible but lack socio-cultural and politico-economic viability. The NetZero2040 high-import/low-demand scenario implausibly assumes that sufficiency-oriented lifestyles can go hand in hand with resistance to renewable energies, while the NetZero2040 low-import/high-demand scenario assumes that non-sustainable lifestyles can coexist with the radical expansion of renewable energies. Furthermore, the assumption of high energy imports at affordable prices in NetZero2040 scenarios appears overly optimistic given the current rise in geopolitical and geoeconomic tensions (Section 8.4.5).

The EAA scenarios, elaborated explicitly to support policymakers, highlight the necessity of compromises in democratic climate policymaking. Being closer to the reality of policymakers, they all combine different types of measures – avoid/shift/improve, push and pull, sufficiency and efficiency-enhancing – thereby going beyond the ideal-typical transformation pathways. Especially the WEM and WAM scenarios reflect current political priorities, subordinating climate targets to cost of living and competitiveness concerns. Consequently, climate targets in WEM and WAM scenarios are missed due to delayed and insufficient measures. The Transition2040, in contrast, is designed to align with Austria's goal of reaching net-zero emissions by 2040 but necessitates the deployment of CCUS for balancing out residual emissions in the agricultural sector and hard-to-abate industries.

Although the assessed emission scenarios for Austria demonstrate potential mitigation pathways in line with emission reduction targets, they neglect or only indirectly consider other environmental and social indicators (e.g., land-use impacts of expected bioenergy requirements to meet future energy demand) (Section 8.2.1). These scenarios generally aim at optimizing emission reductions from a techno-economic perspective, thereby overlooking other requirements for climate-friendly living, such as climate adaptation (Section 8.5.5), respecting planetary boundaries (Section 8.2.1) or meeting the SDGs (Section 8.3.1). This narrow focus can result in neglecting potential synergies and tradeoffs with other objectives. The expansion of renewables, for instance, may intensify land-use competition and challenges associated with social acceptance, while sharply increasing CO<sub>2</sub>-prices can exacerbate inequalities and trigger distributional conflicts. A more systemic approach is thus needed to assess scenario effectiveness, incorporating multiple objectives for climate-friendly living moving beyond optimization and single-goal (e.g., net-zero) oriented approaches. Demand reduction strategies, for instance, can maximize synergy potentials with the SDGs (Figure 8.3). Likewise, limiting new constructions (e.g., buildings, roads, etc.) not only lowers emissions but also decreases soil sealing, thereby minimizing biodiversity impacts. Likewise, reducing meat production can cut emissions while improving public health and lowering health care costs, while reducing car dependency can enhance neighborhoods' quality of life (Sections 8.3.1, 8.6.3). However, excessive emphasis on demand reduction may provoke resistance from certain societal groups and stakeholders.

All scenarios share the same path dependencies, reducing the scope for deviation from current trends at least until 2030 and, in many sectors, beyond. This is due to existing lock-ins and already taken or planned investments in the energy and infrastructure system. Achieving the 2040 and 2050 climate targets, however, requires transformation pathways that overcome current path-dependencies, such as the reliance on fossil infrastructures and unsustainable modes of living. This shift is particularly crucial in the mobility sector, where shifting to EV and public transport is essential. While such a transformation towards net-zero is technically and economically feasible, recent policy reactions on geopolitical conflicts still appear biased towards safeguarding fossil-fuel supply. Urgent decisions and related investments taken in this respect, for example the replacement of natural gas from Russia by LNG, and, later on, hydrogen, may create technology-biases and lock-ins that can make necessary emission reductions after 2030 far more challenging and costly (Section 8.5.1).

Additionally, strict fiscal rules may pose an obstacle to meeting climate targets by restraining public investment. (Section 8.5.3). While problematizing these fiscal rules and proposing alternatives (e.g., a 'Green Investment Golden Rule') is one possible political strategy, fiscal austerity also presents opportunities for shifting the focus towards regulatory instruments (e.g., laws, provisions, bans, product liability). The advantage of regulatory instruments, be it speed limits or product provisions, is that these measures can be very effective if correctly implemented (Section 6.5.1). In the current situation, a key advantage is that they can contribute to keep public expenditures at a low level and also reduce transaction costs for market participants, potentially leading to immediate results and increasing the fiscal room of maneuver. In Austrian climate policymaking, pull measures (e.g., grants, incentives, etc.) tend to be privileged, even though they may not align with fiscal discipline.

In rule-of-law-based societies, many specific decisions regarding the choice of means to tackle climate change are left to democratic decisions makers (Section 8.5.2). Based on the assessed minimum requirements (Section 8.6) the implications for democratic policy makers are therefore straightforward yet consequential: they can ignore the minimum requirements, accept given societal preferences, reduce public investments and avoid regulations, thereby implicitly accepting the likelihood of unmet climate targets. Alternatively, they can try to respect the minimum requirements for climate-friendly transformations and implement bundles of measures including policies that may initially lack societal and political support (e.g., tax increases or speed limits) but increase the probability of meeting long-term climate targets. Such a comprehensive and ambitious strategy may indeed be feasible in a democratic setting, if burden sharing is considered as fair and disadvantages for private actors (e.g., bureaucratic costs and inconveniences) are minimized, especially since acceptance of climate measures tends to increase after implementation – thus legitimizing effective climate-actions ex-post.

## REFERENCES

- Abou-Chadi, T., Jansen, J., Kollberg, M., and Redeker, N. (2024). Debunking the Backlash. Uncovering European Voters' Climate Preferences. Berlin: Hertie School; Jacques Delors Centre. Available at: <u>https://www.delorscentre.eu/fileadmin/2\_Research/1\_About\_our\_research/2\_Research\_centres/6\_Jacques\_Delors\_ Centre/Publications/20240307\_Debunking\_the\_Backlash\_ Abou-Chadi\_Janssen\_Kollberg\_Redeker.pdf (Accessed May 8, 2024).</u>
- Abram, D. (1997). *The Spell of the Sensuous: Perception and Language in a More-Than-Human World.*, 20th ed. New York: Penguin Random House.
- Abson, D. J., Fischer, J., Leventon, J., Newig, J., Schomerus, T., Vilsmaier, U., et al. (2017). Leverage points for sustainability transformation. *Ambio* 46, 30–39. <u>https://doi.org/10.1007/s13280-016-0800-y</u>
- Ackerman, F. (2017). Worst-Case Economics: Extreme Events in Climate and Finance. Anthem Press.
- Adaman, F., and Devine, P. (2017). "Democracy, Participation and Social Planning", in *Routledge Handbook of Ecological Economics*, ed. C. L. Spash (Abingdon, Oxon; New York, NY: Routledge), 517–525. Available at: <u>https://www.taylorfrancis.com/</u> <u>books/9781317395102/chapters/10.4324/9781315679747-60</u> (Accessed May 14, 2024).
- Adger, W. N., Arnell, N. W., and Tompkins, E. L. (2005). Successful adaptation to climate change across scales. *Global Environmental Change* 15, 77–86. <u>https://doi.org/10.1016/j.gloenv-</u> cha.2004.12.005
- Aguiar, F. C., Bentz, J., Silva, J. M. N., Fonseca, A. L., Swart, R., Santos, F. D., et al. (2018). Adaptation to climate change at local level in Europe: An overview. *Environmental Science & Policy* 86, 38–63. <u>https://doi.org/10.1016/j.envsci.2018.04.010</u>
- Aiginger, K., and Rodrik, D. (2020). Rebirth of Industrial Policy and an Agenda for the Twenty-First Century. J Ind Compet Trade 20, 189–207. <u>https://doi.org/10.1007/s10842-019-00322-3</u>
- Aigner, E., Brugger, K., Lichtenberger, H., Ranftler, J., and Schmidt, A. E. (2023a). Multiple Belastungen: Analyse von Gesundheit, Wohn- und Lebensbedingungen von Armut betroffener Familien im Winter 2022/2023. Wien: Gesundheit Österreich. Available at: <u>https://jasmin.goeg.at/id/eprint/2769/</u> (Accessed May 9, 2024).
- Aigner, E., Brugger, K., Lichtenberger, H., and Schmidt, A. (2023b). "Armutsgefährdete Kinder in der Klimakrise: Betroffene, Anpassung und soziale Infrastruktur. Endbericht von Start-Clim2022.A", in StartClim2022: Schlüsselmaßnahmen, Messbarkeit und Notfallszenarien. Auftraggeber: BMK, BMWFW, Klima- und Energiefonds, Land Oberösterreich. Available at: https://www.researchgate.net/publication/372657118\_Armutsgefahrdete\_Kinder\_in\_der\_Klimakrise\_Betroffene\_Anpassung\_und\_soziale\_Infrastruktur (Accessed May 9, 2024).
- Aigner, E., Görg, C., Krisch, A., Madner, V., Muhar, A., Novy, A., et al. (2023c). "Technical Summary", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 105–170. Available at: https://doi.org/10.1007/978-3-662-66497-1\_4 (Accessed October 4, 2023).
- Aigner, E., Görg, C., Madner, V., Muhar, A., Posch, A., Steininger, K. W., et al. (2023d). "Summary for Policymakers", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C.

Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 19–33. <u>https://doi.org/10.1007/978-3-662-66497-1\_2</u>

- Aigner, E., Görg, C., Madner, V., Novy, A., Steininger, K. W., Nabernegg, S., et al. (2023e). "Kapitel 1. Einleitung: Strukturen für ein klimafreundliches Leben", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 173–194. <u>https://doi.org/10.1007/978-3-662-66497-1\_5</u>
- Aigner, E., Lichtenberger, H., Ranftler, J., and Schmeißl, S. (2023f). "Es ist wie in einer Sauna": Die Betroffenheit armutsbetroffener Kinder und ihrer Familien durch die Klimakrise und sozialpolitische Antworten. *soziales\_kapital* 27, 79–99.
- AK Europa (2023). Neue Europäische Industriepolitik Fokus: Green Deal Industrial Plan & Net-Zero Industry Act. Wien. Available at: <u>https://www.akeuropa.eu/sites/default/files/2023-11/PP\_GDIP\_DE.pdf</u> (Accessed April 24, 2024).
- AK Wien (2024). Eine Zukunft für die Vielen: Der Plan der Arbeiterkammer für den sozialen und ökologischen Umbau. Vienna: Verlag Arbeiterkammer Wien. Available at: <u>https://wien.</u> <u>arbeiterkammer.at/interessenvertretung/arbeitundsoziales/</u> <u>sozialstaat/AK-Umbauplan\_Langfassung\_2024.pdf</u> (Accessed October 22, 2024).
- Allianz Nachhaltige Universitäten in Österreich ed. (2021). UniNEtZ-Optionenbericht: Österreichs Handlungsoptionen für die Umsetzung der UN-Agenda 2030 für eine lebenswerte Zukunft. UniNEtZ – Universitäten und Nachhaltige Entwicklungsziele. Vienna. Available at: <u>https://www.uninetz.at/optionenbericht</u> (Accessed May 14, 2024).
- Alogoskoufis, S., Dunz, N., Emambakhsh, T., Hennig, T., Kaijser, M., Kouratzoglou, C., et al. (2021). ECB economy-wide climate stress test: Methodology and results. ECB Occasional Paper. <u>https://doi.org/10.2866/460490</u>
- Altenburg, T., and Assmann, C. eds. (2017). Green Industrial Policy: Concept, Policies, Country Experiences. Geneva Bonn: UN Environment; German Development Institute / Deutsches Institut für Entwicklungspolitk (DIE).
- Amel, E., Manning, C., Scott, B., and Koger, S. (2017). Beyond the roots of human inaction: Fostering collective effort toward ecosystem conservation. *Science* 356, 275–279. <u>https://doi.org/</u> 10.1126/science.aal1931
- Anderson, K. (2015). Duality in climate science. Nature Geosci 8, 898–900. <u>https://doi.org/10.1038/ngeo2559</u>
- Anderson, K., Schrage, J., Stoddard, I., Tuckey, A., and Wetterstedt, M. (2018). A Guide for a Fair Implementation of the Paris Agreement within Swedish Municipalities and Regional Governments: Part II of the Carbon Budget Reports Submitted to Swedish Local Governing Bodies in the 2018 Project "Koldiox-idbudgetar 2020–2040." A Report comissioned by Swedish municipalities and regional governments. Climate Change Leadership Node, Uppsala University, Sweden. Available at: <u>https://www.diva-portal.org/smash/get/diva2:1382723/FULLTEXT01.pdf</u> (Accessed September 4, 2023).
- Andersson, D., and Nässén, J. (2016). The Gothenburg congestion charge scheme: A pre–post analysis of commuting behavior and travel satisfaction. *Journal of Transport Geography* 52, 82–89. <u>https://doi.org/10.1016/j.jtrangeo.2016.02.014</u>
- Andersson, J., Hojcková, K., and Sandén, B. A. (2023). On the functional and structural scope of technological innovation systems
  A literature review with conceptual suggestions. *Environmen*-

tal Innovation and Societal Transitions 49, 100786. https://doi. org/10.1016/j.eist.2023.100786

- Andre, P., Boneva, T., Chopra, F., and Falk, A. (2024). Globally representative evidence on the actual and perceived support for climate action. *Nat. Clim. Chang.* 14, 253–259. <u>https://doi. org/10.1038/s41558-024-01925-3</u>
- Andreoni, V. (2017). "Social equity and ecological sustainability through the lens of degrowth", in *Cultures of Sustainability and Wellbeing*, (London: Routledge).
- Aoki, N., Tay, M., and Rawat, S. (2024). Whole-of-government and joined-up government: A systematic literature review. *Public Administration* 102, 733–752. <u>https://doi.org/10.1111/padm.</u> <u>12949</u>
- APCC (2014). Österreichischer Sachstandsbericht Klimawandel 2014 (AAR14)., ed. Austrian Panel on Climate Change (APCC). Wien, Österreich: Verlag der Österreichischen Akademie der Wissenschaften. https://doi.org/10.1553/aar14
- APCC (2018). Österreichischer Special Report Gesundheit, Demographie und Klimawandel (ASR18). Austrian Panel on Climate Change (APCC). Wien, Österreich: Verlag der ÖAW. Available at: <u>https://austriaca.at/8427-0</u> (Accessed September 20, 2022).
- APCC (2023). APCC Special Report: Strukturen für ein klimafreundliches Leben (APCC SR Klimafreundliches Leben)., eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. Berlin, Heidelberg: Springer Spektrum. <u>https://doi.org/10. 1007/978-3-662-66497-1</u>
- Ara Begum, R., Lempert, R., Elham, A., Benjaminsen, T. A., Bernauer, T., Cramer, W., et al. (2022). Chapter 1: Point of Departure and Key Concepts. Climate Change 2022 Impacts, Adaptation and Vulnerability: Working Group II Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, 151–214. https://doi.org/10.1017/9781009325844.003
- Armutskonferenz, ATTAC, and BEIGEWUM eds. (2021). Klimasoziale Politik: Eine gerechte und emissionsfreie Gesellschaft gestalten., 2. Auflage. Wien: bahoe books. Available at: <u>https:// klimasozial.at/wp-content/uploads/2024/06/Klimabuch\_FinaleFahnen\_2Auflage.pdf</u>
- Arnecke, M. C. (2024). Contentious Policymaking on Climate Change Mitigation in Austria after 2019. An Actor Analysis. *Social Ecology Working Papers* 200. Available at: <u>https://boku. ac.at/fileadmin/data/H03000/H73000/H73700/Publikationen/ Working\_Papers/WP\_200\_Web\_Arnecke.pdf</u> (Accessed December 20, 2024).
- Arnold, A., David, M., Hanke, G., and Sonnberger, M. eds. (2015). Innovation – Exnovation: Über Prozesse des Abschaffens und Erneuerns in der Nachhaltigkeitstransformation. Marburg: Metropolis-Verlag.
- Arntz, S., and Krommendijk, J. (2024). Historic and Unprecedented. Climate Justice in Strasbourg, VerfBlog. <u>https://doi.org/</u> 10.59704/bc1514f261bbbbd3
- Asafu-Adjaye, J., Blomquist, L., Brand, S., Brook, B., Defries, R., Ellis, E., et al. (2015). *An ecomodernist manifesto*. University Of Tasmania.
- Avelino, F. (2017). Power in Sustainability Transitions: Analysing power and (dis)empowerment in transformative change towards sustainability. *Environmental Policy and Governance* 27, 505–520. https://doi.org/10.1002/eet.1777
- Avelino, F., and Wittmayer, J. M. (2016). Shifting Power Relations in Sustainability Transitions: A Multi-actor Perspective. *Journal* of Environmental Policy & Planning 18, 628–649. <u>https://doi.org/10.1080/1523908X.2015.1112259</u>

- Aykut, S. C., Hüppauff, L., Frerichs, L., Fünfgeld, A., Walter, Y., Aguirre, F., et al. (2024). Klimawende Ausblick 2024. Gesellschaftliche Treiber der Transformation in Deutschland. Band 1. Klimapolitik, Klimabewegung und Klimaklagen. Universität Hamburg. Available at: <u>https://www.fdr.uni-hamburg.</u> <u>de/record/14271</u> (Accessed May 13, 2024).
- Baines, J., and Hager, S. B. (2023). From passive owners to planet savers? Asset managers, carbon majors and the limits of sustainable finance. *Competition & Change* 27, 449–471. <u>https:// doi.org/10.1177/10245294221130432</u>
- Bal, M., Stok, M., Bombaerts, G., Huijts, N., Schneider, P., Spahn, A., et al. (2023). A fairway to fairness: Toward a richer conceptualization of fairness perceptions for just energy transitions. *Energy Research & Social Science* 103, 103213. <u>https://doi.org/ 10.1016/j.erss.2023.103213</u>
- Balas, M., Buschmann, D., Neumann, J., Offenzeller, M., Völler, S., and Vollgruber, D. (2024a). Die Österreichische Strategie zur Anpassung an den Klimawandel Teil 1 – Kontext. Vienna: Ministry of Climate Action, Environment, Energy, Mobility, Innovation and Technology (BMK). Available at: <u>https://www. bmk.gv.at/dam/jcr:cd2c1a8f-7c71-4db1-8d73-7ddd4d8bc992/ BMK\_NAS\_Teil-1\_Kontext\_UA.pdf</u> (Accessed October 17, 2024).
- Balas, M., Lexer, W., Lindinger, H., Neumann, J., Offenzeller, M., Völler, S., et al. (2024b). Die Österreichische Strategie zur Anpassung an den Klimawandel Teil 2 – Aktionsplan Handlungsempfehlungen für die Umsetzung. Vienna: Ministry of Climate Action, Environment, Energy, Mobility, Innovation and Technology (BMK). Available at: <u>https://www.bmk.gv.at/ dam/jcr:07aab774-27c5-4de2-86d7-9276563b8934/BMK</u> <u>NAS\_Teil-2\_Aktionsplan\_UA.pdf</u> (Accessed October 17, 2024).
- Barlow, N., Regen, L., Cadiou, N., Chertkovskaya, E., Hollweg, M., Plank, C., et al. eds. (2022). Degrowth & Strategy: how to bring about social-ecological transformation. Mayfly Books. Available at: <u>https://mayflybooks.org/degrowth-strategy/</u> (Accessed August 18, 2023).
- Bärnthaler, R. (2023). Towards eco-social politics: a case study of transformative strategies to overcome forms-of-life crises. *Environmental Politics* 33, 92–113. <u>https://doi.org/10.1080/09644</u> 016.2023.2180910
- Bärnthaler, R. (2024a). Problematising degrowth strategising: On the role of compromise, material interests, and coercion. *Ecological Economics* 223, 108255. <u>https://doi.org/10.1016/j.ecolecon.2024.108255</u>
- Bärnthaler, R. (2024b). When enough is enough: Introducing sufficiency corridors to put techno-economism in its place. *Ambio* 53, 960–969. <u>https://doi.org/10.1007/s13280-024-02027-2</u>
- Bärnthaler, R., and Gough, I. (2023). Provisioning for sufficiency: envisaging production corridors. Sustainability: Science, Practice and Policy 19, 2218690. <u>https://doi.org/10.1080/15487733.</u> 2023.2218690
- Bärnthaler, R., Novy, A., Arzberger, L., Krisch, A., and Volmary, H. (2024). The power to transform structures: power complexes and the challenges for realising a wellbeing economy. *Humanit* Soc Sci Commun 11, 558. <u>https://doi.org/10.1057/s41599-024-03065-7</u>
- Bärnthaler, R., Novy, A., and Plank, L. (2021). The Foundational Economy as a Cornerstone for a Social–Ecological Transformation. Sustainability 13, 10460. <u>https://doi.org/10.3390/su13181</u> 0460

- Bärnthaler, R., Novy, A., and Stadelmann, B. (2023). A Polanyi-inspired perspective on social-ecological transformations of cities. *Journal of Urban Affairs* 45, 117–141. <u>https://doi.org/10. 1080/07352166.2020.1834404</u>
- Barth, B., and Molina, C. (2023). "Transformation und Beharren. Aktuelle Einstellungen zu Umwelt- und Klimaschutz, differenziert nach den Sinus-Milieus", in *Partizipation: das Zusammenwirken der Vielen für Demokratie, Wirtschaft und Umwelt*, eds. J. Fritz and N. Tomaschek (Münster New York: Waxmann Verlag GmbH).
- Battiston, S. (2019). The importance of being forward-looking: managing financial stability in the face of climate risk. *Financial Stability Review*, Banque de France, issue 23, 39–48. <u>https:// ideas.repec.org/a/bfr/fisrev/2019235.html</u>
- Battiston, S., Mandel, A., Monasterolo, I., Schütze, F., and Visentin, G. (2017). A climate stress-test of the financial system. *Nature Clim Change* 7, 283–288. <u>https://doi.org/10.1038/nclimate3255</u>
- Battiston, S., Monasterolo, I., Riahi, K., and Van Ruijven, B. J. (2021). Accounting for finance is key for climate mitigation pathways. *Science* 372, 918–920. <u>https://doi.org/10.1126/science.</u> abf3877
- Bauknecht, D., and Kubeczko, K. (2024). Regulatory experiments and real-world labs: A fruitful combination for sustainability. *GAIA – Ecological Perspectives for Science and Society* 33, 44–50. <u>https://doi.org/10.14512/gaia.33.S1.7</u>
- Baumann, F. (2024). Multilateral Climate Governance: Its Promise and Limits. Global Governance: A Review of Multilateralism and International Organizations 30, 240–253. <u>https://doi.org/10.1163/19426720-03002010</u>
- Baumgartner, J., Bierbaumer, J., Bilek-Steindl, S., Bittschi, B., Glocker, C., Mayrhuber, C., et al. (2023). Geopolitical Tensions, Energy Crisis and Inflation Shape Economic Activity. The Austrian Economy in 2022. Vienna: WIFO. Available at: <u>https://www. wifo.ac.at/jart/prj3/wifo/resources/person\_dokument/person\_ dokument.jart?publikationsid=70807&mime\_type=application/pdf</u> (Accessed September 13, 2023).
- Bausch, T., and Koziol, K. (2017). Prozessgestaltung und Steuerung von Klimawandelanpassung in kleinen bayerischen Gemeinden. Abschlussbericht. München: Munich University of Applied Sciences. Available at: <u>https://mediapool.hm.edu/media/ fk14/fk14\_lokal/forschungundprojekte/abgeschlossene\_projekte/steuerung\_von\_klimawandelanpasusng\_in\_kleinen\_ bayerischen\_gemeinden\_/Abschlussbericht\_Gesamtprojekt\_ KWA\_20170831.pdf (Accessed October 22, 2024).</u>
- Bednar-Friedl, B., Biesbroek, R., Schmidt, D. N., Alexander, P., Børsheim, K. Y., Carnicer, J., et al. (2022). "Europe", in Climate Change 2022: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, eds. H. O. Pörtner, D. C. Roberts, M. Tignor, E. S. Poloczanska, K. Mintenbeck, A. Alegría, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press), 1817–1927. <u>https://doi.org/ 10.1017/9781009325844.015</u>
- Bennich, T., Weitz, N., and Carlsen, H. (2020). Deciphering the scientific literature on SDG interactions: A review and reading guide. Science of The Total Environment 728, 138405. <u>https:// doi.org/10.1016/j.scitotenv.2020.138405</u>
- Berrang-Ford, L., Siders, A. R., Lesnikowski, A., Fischer, A. P., Callaghan, M. W., Haddaway, N. R., et al. (2021). A systematic global stocktake of evidence on human adaptation to climate

change. Nat. Clim. Chang. 11, 989-1000. <u>https://doi.org/10.</u> 1038/s41558-021-01170-y

- Biermann, F., Abbott, K., Andresen, S., Bäckstrand, K., Bernstein, S., Betsill, M. M., et al. (2012). Navigating the Anthropocene: Improving Earth System Governance. *Science* 335, 1306–1307. <u>https://doi.org/10.1126/science.1217255</u>
- Biesbroek, G. R., Swart, R. J., Carter, T. R., Cowan, C., Henrichs, T., Mela, H., et al. (2010). Europe adapts to climate change: Comparing National Adaptation Strategies. *Global Environmental Change* 20, 440–450. <u>https://doi.org/10.1016/j.gloenvcha.2010.03.005</u>
- Blühdorn, I. (2022). Planetary boundaries, societal boundaries, and collective self-limitation: moving beyond the post-Marxist comfort zone. *Sustainability: Science, Practice and Policy* 18, 576–589. https://doi.org/10.1080/15487733.2022.2099124
- Blühdorn, I. (2024). Unhaltbarkeit: Auf dem Weg in eine andere Moderne. Berlin: Suhrkamp Verlag.
- BMK (2020). Ressourcennutzung in Österreich 2020. Band 3. Wien: Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie (BMK). Available at: <u>https://www.bmk.gv.at/themen/klima\_umwelt/nachhaltigkeit/</u><u>ressourceneffizienz/publikationen/bericht2020.html</u> (Accessed January 17, 2025).
- BMK (2022). Biodiversitäts-Strategie Österreich 2030+ Wien. Wien: Bundesministerium für Klimaschutz, Umwelt, Energie, Mobilität, Innovation und Technologie. Available at: <u>https:// www.bmk.gv.at/themen/klima\_umwelt/naturschutz/biol\_vielfalt/biodiversitaetsstrategie/biodiversitaetsstrategie\_2030.html</u> (Accessed May 13, 2024).
- BMK (2023). Energie in Österreich. Zahlen, Daten, Fakten. Vienna: Ministry of Climate Action, Environment, Energy, Mobility, Innovation and Technology (BMK). Available at: <u>https://www. bmk.gv.at/dam/jcr:da4e9dfd-f51c-44b8-894c-9b049a8336cb/ BMK\_Energie\_in\_OE2023\_barrierefrei.pdf</u> (Accessed November 5, 2024).
- BMK (2024). Integrierter nationaler Energie- und Klimaplan für Österreich – Periode 2021–2030. Vienna: Ministry of Climate Action, Environment, Energy, Mobility, Innovation and Technology (BMK). Available at: <u>https://www.bmk.gv.at/dam/ jcr:6c55ea04-e4b8-499f-ac3b-9d8786147cee/NEKP\_Aktualisierung\_2023\_2024\_final.pdf</u> (Accessed November 4, 2024).
- Bohnenberger, K. (2023). Peaks and gaps in eco-social policy and sustainable welfare: A systematic literature map of the research landscape. *European Journal of Social Security* 25, 328–346. <u>https://doi.org/10.1177/13882627231214546</u>
- Bonneuil, C., and Fressoz, J.-B. (2017). *The Shock of the Anthropocene: The Earth, History and Us.* London: Verso Books.
- Borgstedt, S. (2023). "Die Sinus-Milieus<sup>®</sup> als Instrument für Transformationsforschung und evidenzbasierte Politikberatung", in *Praxis der Sinus-Milieus*<sup>®</sup>, eds. B. Barth, B. B. Flaig, N. Schäuble, M. Tautscher, B. Barth, B. B. Flaig, et al. (Wiesbaden: Springer VS), 305–315. Available at: <u>https://link.springer.com/10.1007/978-3-658-42380-3\_19</u> (Accessed March 18, 2024).
- Börjesson, M., Eliasson, J., and Hamilton, C. (2016). Why experience changes attitudes to congestion pricing: The case of Gothenburg. *Transportation Research Part A: Policy and Practice* 85, 1–16. https://doi.org/10.1016/j.tra.2015.12.002
- Brand, U., Görg, C., and Wissen, M. (2020). Overcoming neoliberal globalization: social-ecological transformation from a Polany-

ian perspective and beyond. *Globalizations* 17, 161–176. <u>https://doi.org/10.1080/14747731.2019.1644708</u>

- Brand, U., Muraca, B., Pineault, É., Sahakian, M., Schaffartzik, A., Novy, A., et al. (2021). From planetary to societal boundaries: an argument for collectively defined self-limitation. *Sustainability: Science, Practice and Policy* 17, 265–292. <u>https://doi.org/10. 1080/15487733.2021.1940754</u>
- Brand, U., and Niedermoser, M. K. (2019). The role of trade unions in social-ecological transformation: Overcoming the impasse of the current growth model and the imperial mode of living. *Journal of Cleaner Production* 225, 173–180. <u>https://doi. org/10.1016/j.jclepro.2019.03.284</u>
- Brand, U., and Wissen, M. (2017). "The Imperial Mode of Living", in *Routledge Handbook of Ecological Economics*, ed. C. L. Spash (Abingdon, Oxon; New York, NY: Routledge), 152–161. Available at: <u>https://www.taylorfrancis.com/books/9781317395102/</u> <u>chapters/10.4324/9781315679747-19</u> (Accessed October 4, 2022).
- Brand, U., and Wissen, M. (2021). *The Imperial Mode of Living: Everyday Life and the Ecological Crisis of Capitalism*. Brooklyn: Verso Books.
- Brand, U., and Wissen, M. (2024). Kapitalismus am Limit: öko-imperiale Spannungen, umkämpfte Krisenpolitik und solidarische Perspektiven. München: Oekom Verlag.
- Brand-Correa, L. I., and Steinberger, J. K. (2017). A Framework for Decoupling Human Need Satisfaction From Energy Use. *Ecological Economics* 141, 43–52. <u>https://doi.org/10.1016/j.ecolecon.2017.05.019</u>
- Breitenfellner, A., Hasenhüttl, S., Lehmann, G., and Tschulik, A. (2020). Green finance – opportunities for the Austrian financial sector. *Financial Stability Report*. Available at: <u>https://ideas. repec.org//a/onb/oenbfs/y2020i40b2.html</u> (Accessed September 15, 2023).
- Breitenfellner, A., and Kariem, H. (2023). "What do people in Austria think about green finance?", in *Financial Stability Report 46*, (Wien: Oesterreichische Nationalbank (Austrian Central Bank)), 47–63. Available at: <u>https://www.oenb.at/dam/ jcr:1f93b7de-5ed6-4e77-ab20-d4f61399f981/FSR-46.pdf</u>
- Breitenfellner, A., and Kariem, H. (2024). Was halten die Menschen in Österreich von Green Finance? Wien: Oesterreichische Nationalbank. Available at: <u>https://www.oenb.at/dam/ jcr:c594baf4-04d9-4246-b8d3-9238229f3361/report-2024-16green-finance.pdf</u>
- Bridle, J. (2023). Ways of Being. Animals, Plants, Machines: The Search for a Planetary Intelligence. London: Penguin. Available at: <u>https://www.penguin.co.uk/books/317823/ways-of-beingby-bridle-james/9780141994260</u> (Accessed August 20, 2023).
- Brizga, J., Jørgensen, M. S., Olesen, G. B., and Lekavicius, V. (2023). Energy Sufficiency policy approach to geopolitical risks and rising energy prices. *Preprints*. <u>https://doi.org/10.20944/preprints202308.0194.v1</u>
- Bröthaler, J., Dillinger, T., Getzner, M., Kanonier, A., Grinzinger, E., and Chamraci, M. (2024). Klimaorientierte und ressourcenschonende Raumentwicklung und Finanzausgleich. Zur Raumwirksamkeit des Finanzausgleichs unter besonderer Berücksichtigung des ÖREK 2030, des Klimaschutzes und des sparsamen Bodenverbrauchs. Endbericht. Vienna: Vienna University of Technology. Available at: <u>https://repositum.tuwien.at/ handle/20.500.12708/190477</u> (Accessed May 7, 2024).
- Bröthaler, J., Getzner, M., Müller, H. L., Plank, L., Miess, M., Niedertscheider, M., et al. (2023). Öffentliche Investitionen für

den Klimaschutz in Österreich: Potenziale des öffentlichen Vermögens. Kammer für Arbeiter und Angestellte für Wien. Available at: <u>https://repositum.tuwien.at/handle/20.500.12708/</u>187853 (Accessed September 6, 2023).

- Brudermann, T. (2022). Die Kunst der Ausrede: Warum wir uns lieber selbst täuschen, statt klimafreundlich zu leben. München: Oekom Verlag.
- Brulle, R. J. (2018). The climate lobby: a sectoral analysis of lobbying spending on climate change in the USA, 2000 to 2016. *Climatic Change* 149, 289–303. <u>https://doi.org/10.1007/s10584-018-2241-z</u>
- Brulle, R. J., Aronczyk, M., and Carmichael, J. (2020). Corporate promotion and climate change: an analysis of key variables affecting advertising spending by major oil corporations, 1986– 2015. *Climatic Change* 159, 87–101. <u>https://doi.org/10.1007/ s10584-019-02582-8</u>
- Buch-Hansen, H., Koch, M., and Nesterova, I. (2024). Deep transformations. A theory of degrowth. Manchester: Manchester University Press.
- Buch-Hansen, H., and Nesterova, I. (2023). Less and more: Conceptualising degrowth transformations. *Ecological Economics* 205, 107731. <u>https://doi.org/10.1016/j.ecolecon.2022.107731</u>
- Buch-Hansen, H., and Nielsen, P. (2020). Critical Realism: Basics and Beyond. London: Red Globe Press.
- Büchs, M. (2021). Sustainable welfare: How do universal basic income and universal basic services compare? *Ecological Economics* 189, 107152. <u>https://doi.org/10.1016/j.ecolecon.2021.107152</u>
- Büchs, M., Ivanova, D., and Schnepf, S. V. (2021). Fairness, effectiveness, and needs satisfaction: new options for designing climate policies. *Environ. Res. Lett.* 16, 124026. <u>https://doi.org/10. 1088/1748-9326/ac2cb1</u>
- Büchs, M., and Koch, M. (2019). Challenges for the degrowth transition: The debate about wellbeing. *Futures* 105, 155–165. <u>https://doi.org/10.1016/j.futures.2018.09.002</u>
- Budolfson, M., Dennig, F., Errickson, F., Feindt, S., Ferranna, M., Fleurbaey, M., et al. (2021). Climate action with revenue recycling has benefits for poverty, inequality and well-being. *Nat. Clim. Chang.* 11, 1111–1116. <u>https://doi.org/10.1038/s41558-021-01217-0</u>
- Bukold, S. (2023). The Dirty Dozen. The Climate Greenwashing of 12 European Oil Companies. Hamburg: Greenpeace Central- and East-Europe, EnergyComment Hamburg. Available at: <u>https://greenpeace.at/uploads/2023/08/report-the-dirty-dozenclimate-greenwashing-of-12-european-oil-companies.pdf</u> (Accessed September 13, 2023).
- Burroughs, J. E., and Rindfleisch, A. (2002). Materialism and Well-Being: A Conflicting Values Perspective. *Journal of Consumer Research* 29, 348–370. <u>https://doi.org/10.1086/344429</u>
- Buschmann, D., Koziol, K., Bausch, T., and Reinhard, S. (2022). Adaptation to climate change in small German municipalities: Sparse knowledge and weak adaptive capacities. *Natural Resources Forum* 46, 377–392. <u>https://doi.org/10.1111/1477-8947.12262</u>
- Calafati, L., Froud, J., Haslam, C., Johal, S., and Williams, K. (2023). When Nothing Works: From Cost of Living to Foundational Liveability. Manchester: Manchester University Press.
- Campiglio, E., Dafermos, Y., Monnin, P., Ryan-Collins, J., Schotten, G., and Tanaka, M. (2018). Climate change challenges for central banks and financial regulators. *Nature Clim Change* 8, 462–468. https://doi.org/10.1038/s41558-018-0175-0

- Caney, S. (2009). Justice and the distribution of greenhouse gas emissions 1. *Journal of Global Ethics* 5, 125–146. <u>https://doi.org/10.1080/17449620903110300</u>
- Canterbury, D. C. (2018). Neoextractivism and Capitalist Development., 1st Edn. London: Routledge. Available at: <u>https://www. taylorfrancis.com/books/9781351127332</u> (Accessed May 15, 2024).
- Carattini, S., Kallbekken, S., and Orlov, A. (2019). How to win public support for a global carbon tax. *Nature* 565, 289–291. <u>https://</u><u>doi.org/10.1038/d41586-019-00124-x</u>
- Carbon Majors (2024). The Carbon Majors Database: Launch Report. Available at: <u>https://carbonmajors.org/site//data/000/027/</u> <u>Carbon\_Majors\_Launch\_Report.pdf</u> (Accessed April 24, 2024).
- Carey, M., James, L. C., and Fuller, H. A. (2014). A new social contract for the IPCC. *Nature Clim Change* 4, 1038–1039. <u>https:// doi.org/10.1038/nclimate2442</u>
- Cass, N., Büchs, M., and Lucas, K. (2023). How are high-carbon lifestyles justified? Exploring the discursive strategies of excess energy consumers in the United Kingdom. *Energy Research & Social Science* 97, 102951. <u>https://doi.org/10.1016/j.erss.2023.102951</u>
- CCCA (2022). + 1,5° C: Wieviel Treibhausgase dürfen wir noch emittieren? Hintergrundpapier zu globalen und nationalen Treibhausgasbudgets. Wien: CCCA. Available at: <u>https://ccca. ac.at/fileadmin/00\_DokumenteHauptmenue/02\_Klimawissen/ Papiere/THG-Budget\_Hintergrundpapier\_CCCA.pdf</u>
- CDP Worldwide (2021). CITIES ON THE ROUTE TO 2030. Building a zero emissions, resilient planet for all. London: Carbon Disclosure Project. Available at: <u>https://cdn.cdp.net/</u> <u>cdp-production/cms/reports/documents/000/005/759/origi-</u> <u>nal/CDP\_Cities\_on\_the\_Route\_to\_2030.pdf?1621329680</u> (Accessed October 21, 2024).
- Cerniglia, F., Saraceno, F., and Watt, A. eds. (2023). Financing Investment in Times of High Public Debt: 2023 European Public Investment Outlook., 1st Edn. Cambridge, UK: Open Book Publishers. Available at: <u>https://www.openbookpublishers.com/</u> <u>books/10.11647/OBP.0386</u> (Accessed June 8, 2024).
- Ćetković, P., and Zhan, J. C. (2022). "Starke anhaltende Kurseinbrüche schlagen sich im ersten Halbjahr 2022 deutlich auf das österreichische Fondsvolumen durch", (Wien: Oesterreichische Nationalbank (Austrian Central Bank)), 61–67. Available at: <u>https://www.oenb.at/dam/jcr:5d6004dc-3ccf-4486-b079-6b4a3</u> <u>7b951c2/Statistiken-q4-22.pdf</u>
- Chancel, L. (2022). Global carbon inequality over 1990–2019. *Nature Sustainability* 5, 931–938. <u>https://doi.org/10.1038/s41893-</u> <u>022-00955-z</u>
- Chancel, L., Bothe, P., and Voituriez, T. (2024). The potential of wealth taxation to address the triple climate inequality crisis. *Nat. Clim. Chang.* 14, 5–7. <u>https://doi.org/10.1038/s41558-023-01891-2</u>
- Chancel, L., Piketty, T., Saez, E., and Zucman, G. (2022). World Inequality Report 2022. World Inequality Lab. Available at: <u>https://wid.world/news-article/world-inequality-report-2022/</u> (Accessed May 16, 2024).
- Christopher, A. N., Saliba, L., and Deadmarsh, E. J. (2009). Materialism and well-being: The mediating effect of locus of control. *Personality and Individual Differences* 46, 682–686. <u>https://doi.org/10.1016/j.paid.2009.01.003</u>
- Christophers, B. (2023). Our Lives in Their Portfolios: Why Asset Managers Own the World. London, Brooklyn: Verso Books.

- Christophers, B. (2024). *The Price is Wrong: Why Capitalism Won't Save the Planet*. London, Brooklyn: Verso Books.
- Cinelli, M., De Francisci Morales, G., Galeazzi, A., Quattrociocchi, W., and Starnini, M. (2021). The echo chamber effect on social media. *Proceedings of the National Academy of Sciences* 118, e2023301118. <u>https://doi.org/10.1073/pnas.2023301118</u>
- Cinus, F., Minici, M., Monti, C., and Bonchi, F. (2022). The Effect of People Recommenders on Echo Chambers and Polarization. *Proceedings of the International AAAI Conference on Web and Social Media* 16, 90–101. <u>https://doi.org/10.1609/icwsm.</u> v16i1.19275
- Clar, C., Omann, I., and Scherhaufer, P. (2023). Der österreichische Klimarat – ein Beitrag zur Weiterentwicklung von Demokratie und Politik? SWS-Rundschau 63, 259–275.
- Clar, C., and Steurer, R. (2017). Erfolgsfaktoren von politischen Prozessen zur Anpassung an den Klimawandel auf verschiedenen räumlichen Ebenen. *Diskussionspapier / Discussion Paper*. Available at: <u>https://boku.ac.at/fileadmin/data/H03000/</u> <u>H73000/H73200/InFER\_DP\_1\_2017\_Clar\_Steurer\_Erfolgsfaktoren\_Klimawandel.pdf</u> (Accessed October 21, 2024).
- Clar, C., and Steurer, R. (2019). Climate change adaptation at different levels of government: Characteristics and conditions of policy change. *Natural Resources Forum* 43, 121–131. <u>https:// doi.org/10.1111/1477-8947.12168</u>
- Clarke, L., Wei, Y.-M., Navarro, A. D. L. V., Garg, A., Hahmann, A. N., Khennas, S., et al. (2022). Energy Systems. IPCC, 2022: Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. <u>https://doi.org/10.1017/9781009157926.008</u>
- Colgan, J. D., Green, J. F., and Hale, T. N. (2021). Asset Revaluation and the Existential Politics of Climate Change. *Int Org* 75, 586–610. <u>https://doi.org/10.1017/S0020818320000296</u>
- Colloff, M. J., Martín-López, B., Lavorel, S., Locatelli, B., Gorddard, R., Longaretti, P.-Y., et al. (2017). An integrative research framework for enabling transformative adaptation. *Environmental Science & Policy* 68, 87–96. <u>https://doi.org/10.1016/</u> j.envsci.2016.11.007
- Coote, A., and Percy, A. (2020). *The Case for Universal Basic Services*. Cambridge, UK/Medford, MA: Polity.
- Corlet Walker, C., Druckman, A., and Jackson, T. (2021). Welfare systems without economic growth: A review of the challenges and next steps for the field. *Ecological Economics* 186, 107066. <u>https://doi.org/10.1016/j.ecolecon.2021.107066</u>
- Corporate Europe Observatory (2020). A grey deal? Fossil fuel fingerprints on the European Green Deal. Available at: <u>https://</u> <u>corporateeurope.org/en/a-grey-deal</u> (Accessed April 8, 2024).
- Creutzig, F., Roy, J., Lamb, W. F., Azevedo, I. M. L., Bruine de Bruin, W., Dalkmann, H., et al. (2018). Towards demand-side solutions for mitigating climate change. *Nature Climate Change* 8, 260– 263. <u>https://doi.org/10.1038/s41558-018-0121-1</u>
- Cucca, R., Kazepov, Y., and Villa, M. (2023). Towards a Sustainable Welfare System? The Challenges and Scenarios of Eco-social Transitions. *Social Policies* 10, 3–26. <u>https://doi.org/10.7389/ 107136</u>
- Dafermos, Y., and Nikolaidi, M. (2023). "9. Financing Climate Investment in the EU: The Role of Monetary and Financial Policies", in *Financing Investment in Times of High Public Debt*, eds. F. Cerniglia, F. Saraceno, A. Watt, F. Cerniglia, F. Saraceno, and A. Watt (Cambridge, UK: Open Book Publishers), 157–168. Available at: https://www.openbookpublishers.com/

books/10.11647/obp.0386/chapters/10.11647/obp.0386.09 (Accessed June 8, 2024).

- Danaher, J. (2022). Techno-optimism: an Analysis, an Evaluation and a Modest Defence. *Philos. Technol.* 35, 54. <u>https://doi.org/10.1007/s13347-022-00550-2</u>
- Dang, H.-A. H., Hallegatte, S., and Trinh, T.-A. (2024). Does global warming worsen poverty and inequality? An updated review. *Journal of Economic Surveys* 38, 1873–1905. <u>https://doi. org/10.1111/joes.12636</u>
- Dannemann, H. (2023). Climate obstruction: how Denial, delay and inaction are heating the planet – Book review. *Environmental Politics* 32, 1104–1106. <u>https://doi.org/10.1080/09644016.</u> 2023.2215659
- Daumas, L. (2023). Financial stability, stranded assets and the low-carbon transition – A critical review of the theoretical and applied literatures. *Journal of Economic Surveys*, joes.12551. <u>https://doi.org/10.1111/joes.12551</u>
- David, M. (2017). Moving beyond the heuristic of creative destruction: Targeting exnovation with policy mixes for energy transitions. *Energy Research & Social Science* 33, 138–146. <u>https://doi.org/10.1016/j.erss.2017.09.023</u>
- De Castro Mazarro, A., George Kaliaden, R., Wende, W., and Egermann, M. (2023). Beyond urban ecomodernism: How can degrowth-aligned spatial practices enhance urban sustainability transformations. *Urban Studies* 60, 1304–1315. <u>https://doi.org/10.1177/00420980221148107</u>
- De Freitas Netto, S. V., Sobral, M. F. F., Ribeiro, A. R. B., and Soares, G. R. D. L. (2020). Concepts and forms of greenwashing: a systematic review. *Environ Sci Eur* 32, 19. <u>https://doi.org/10.1186/ s12302-020-0300-3</u>
- Del Granado, P. C., Resch, G., Holz, F., Welisch, M., Geipel, J., Hartner, M., et al. (2020). Energy Transition Pathways to a Low-carbon Europe in 2050: the Degree of Cooperation and the Level of Decentralization. *Economics of Energy & Environmental Policy* 9, 121–136. <u>https://doi.org/10.5547/2160-5890.9.1.pcre</u>
- Delsen, L., and Lehr, A. (2019). Value matters or values matter? An analysis of heterogeneity in preferences for sustainable investments. *Journal of Sustainable Finance & Investment* 9, 240–261. https://doi.org/10.1080/20430795.2019.1608709
- Dengler, C., Dornis, N., and Heck, L. (2024). Klimafreundliche und gesundheitsfördernde Aspekte von Zeitwohlstand. Policy Brief. Wien: Gesundheit Österreich. Available at: <u>https://jasmin.goeg.</u> <u>at/id/eprint/3423/1/GOEGPolicyBrief-Zeitwohlstand\_bf.pdf</u> (Accessed May 6, 2024).
- Desrochers, P., and Szurmak, J. (2020). Seven Billion Solutions Strong. Why Markets, Growth, and Innovation Are the Antidote to Eco-Pessimism. *The Breakthrough Institute*. Available at: <u>https://thebreakthrough.org/journal/no-12-winter-2020/seven-billion-solutions</u> (Accessed July 14, 2023).
- Di Giulio, A., and Fuchs, D. (2014). Sustainable Consumption Corridors: Concept, Objections, and Responses. GAIA – Ecological Perspectives for Science and Society 23, 184–192. <u>https://doi. org/10.14512/gaia.23.S1.6</u>
- Die Armutskonferenz and Ökobüro (2024). Von allem genug, von nichts zu viel. Schritte zu einem ökologisch und sozial gerechten Ressourcen- und Energieverbrauch. Hintergrundpapier. Vienna. Available at: <u>https://www.armutskonferenz.at/von-allemgenug-von-nichts-zu-viel</u> (Accessed May 8, 2024).
- Dittmar, H., Bond, R., Hurst, M., and Kasser, T. (2014). The relationship between materialism and personal well-being: A

meta-analysis. Journal of Personality and Social Psychology 107, 879–924. https://doi.org/10.1037/a0037409

- Dodman, D., Hayward, B., Pelling, M., Castan Broto, V., Chow, W., Chu, E., et al. (2022). Cities, Settlements and Key Infrastructure. *Climate Change 2022: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*, 907–1040. <u>https://doi.org/10.1017/9781009325844.008.907</u>
- Dooley, K., Holz, C., Kartha, S., Klinsky, S., Roberts, J. T., Shue, H., et al. (2021). Ethical choices behind quantifications of fair contributions under the Paris Agreement. *Nat. Clim. Chang.* 11, 300–305. https://doi.org/10.1038/s41558-021-01015-8
- Dunz, N., Naqvi, A., and Monasterolo, I. (2021). Climate sentiments, transition risk, and financial stability in a stock-flow consistent model. *Journal of Financial Stability* 54, 100872. <u>https://doi.org/10.1016/j.jfs.2021.100872</u>
- Durand, C., Hofferberth, E., and Schmelzer, M. (2023). Planning beyond growth. The case for economic democracy within limits. *Political Economy Working Papers*. Available at: <u>https://archive-ouverte.unige.ch/unige:166429</u> (Accessed May 11, 2024).
- Durand, C., Hofferberth, E., and Schmelzer, M. (2024). Planning beyond growth: The case for economic democracy within ecological limits. *Journal of Cleaner Production* 437, 140351. <u>https:// doi.org/10.1016/j.jclepro.2023.140351</u>
- Durrant, D., Lamker, C., and Rydin, Y. (2023). The Potential of Post-Growth Planning: Re-Tooling the Planning Profession for Moving beyond Growth. *Planning Theory & Practice* 24, 287– 295. <u>https://doi.org/10.1080/14649357.2023.2198876</u>
- Duvic-Paoli, L.-A. (2022). Re-imagining the Making of Climate Law and Policy in Citizens' Assemblies. *Transnational Environmental Law* 11, 235–261. <u>https://doi.org/10.1017/S2047102</u> 521000339
- Dziwok, E., and Jäger, J. (2024). "GREEN FINANCE FROM A GLOBAL NORTH-SOUTH PERSPECTIVE", in Österreichische Entwicklungspolitik 2023: Green Development Finance? From Climate Crisis to Gloal Justice, ed. Österreichische Forschungsstiftung für Internationale Entwicklung (ÖFSE) (Wien: Südwind-Verl.), 51–58.
- Eberle, C., O'Connor, J., Narvaez, L., Mena Benavides, M., and Sebesvari, Z. (2023). Interconnected Disaster Risks 2023: Risk Tipping Points. https://doi.org/10.53324/WTWN2495
- ECHR (2024). Verein KlimaSeniorinnen Schweiz and Others v. Switzerland [GC]. Available at: <u>https://hudoc.echr.coe.int/eng#</u> <u>{%22itemid%22:[%22002-14304%22]}</u> (Accessed October 22, 2024).
- Edelmann, H., Quiñones-Ruiz, X. F., Penker, M., Scaramuzzi, S., Broscha, K., Jeanneaux, P., et al. (2020). Social Learning in Food Quality Governance – Evidences from Geographical Indications Amendments. *International Journal of the Commons* 14, 108–122. https://doi.org/10.5334/ijc.968
- Ekberg, K., Forchtner, B., Hultman, M., and Jylhä, K. M. (2022). Climate Obstruction: How Denial, Delay and Inaction are Heating the Planet., 1st Edn. London: Routledge. Available at: <u>https:// www.taylorfrancis.com/books/9781003181132</u> (Accessed May 2, 2024).
- Elsig, M., Hoekman, B., and Pauwelyn, J. (2016). Thinking About the Performance of the World Trade Organization: A Discussion Across Disciplines. SSRN Journal. <u>https://doi.org/10.2139/</u> <u>ssrn.2732639</u>
- Engels, A., Marotzke, J., Gonçalves Gresse, E., López-Rivera, A., Pagnone, A., Wilkens, J., et al. (2023a). "Table 1: Summary of so-

cial plausibility assessments", in *Hamburg Climate Futures Outlook: The plausibility of a 1.5°C limit to global warming – Social drivers and physical processes*, (Hamburg, Germany: Cluster of Excellence Climate, Climatic Change, and Society (CLICCS)), 44–49. <u>https://doi.org/10.25592/uhhfdm.11230</u>

- Engels, A., Marotzke, J., Gresse, E., López-Rivera, A., Pagnone, A., and Wilkens, J. eds. (2023b). *Hamburg Climate Futures Outlook: The plausibility of a 1.5°C limit to global warming – Social drivers and physical processes*. Hamburg, Germany: Cluster of Excellence Climate, Climatic Change, and Society (CLICCS). Available at: <u>https://www.fdr.uni-hamburg.de/record/11230</u> (Accessed February 19, 2024).
- Essletzbichler, J., Miklin, X., Volmary, H., and Ornetzeder, M. (2023). "Kapitel 17. Soziale und räumliche Ungleichheit", in *APCC Special Report: Strukturen für ein klimafreundliches Leben*, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 481–497. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1</u>21 (Accessed October 4, 2023).
- European Commission (2023a). 2023 Strategic Foresight Report – Sustainability and people's wellbeing at the heart of Europe's Open Strategic Autonomy. Brussels: European Commission. Available at: <u>https://ec.europa.eu/commission/presscorner/detail/en/ip\_23\_3623</u> (Accessed May 16, 2024).
- European Commission (2023b). COMMUNICATION FROM THE COMMISSION TO THE EUROPEAN PARLIAMENT, THE COUNCIL, THE EUROPEAN ECONOMIC AND SO-CIAL COMMITTEE AND THE COMMITTEE OF THE RE-GIONS Long-term competitiveness of the EU: looking beyond 2030. Brussels: European Commission. Available at: <u>https:// commission.europa.eu/system/files/2023-03/Communication Long-term-competitiveness.pdf</u>
- European Commission (2023c). Regulatory learning in the EU Guidance on regulatory sandboxes, testbeds, and living labs in the EU, with a focus section on energy. Brussels: European Commission. Available at: <u>https://data.consilium.europa.eu/</u> <u>doc/document/ST-12199-2023-INIT/en/pdf</u> (Accessed October 23, 2024).
- European Commission (2024). Supporting climate action through the EU budget. Available at: <u>https://climate.ec.europa.eu/eu-action/eu-funding-climate-action/supporting-climate-action-through-eu-budget\_en</u> (Accessed November 12, 2024).
- European Commission and Directorate-General for Communication (2021). *Citizens' knowledge, perceptions, values and expectations of science: report.* Publications Office of the European Union. <u>https://doi.org/10.2775/071577</u>
- European Commission and Joint Research Centre (JRC) (2021). ENSPRESO – INTEGRATED DATA. Available at: <u>http://data.europa.eu/89h/88d1c405-0448-4c9e-b565-3c30c9b167f7</u> (Accessed May 15, 2024).
- European Court of Auditors (2024). *Climate adaptation in the EU action not keeping up with ambition. Special report 15, 2024.* Publications Office of the European Union. Available at: <u>https://data.europa.eu/doi/10.2865/581379</u> (Accessed October 30, 2024).
- European Environment Agency (2020). Urban adaptation in Europe: how cities and towns respond to climate change. Luxemburg: Publications Office of the European Union. Available at: <u>https://www.eea.europa.eu/publications/urban-adaptation-ineurope</u> (Accessed May 14, 2024).

- European Environment Agency (2022). Advancing towards climate resilience in Europe: status of reported national adaptation actions in 2021. Publications Office of the European Union. Available at: <u>https://data.europa.eu/doi/10.2800/517467</u> (Accessed October 21, 2024).
- European Environment Agency (2024). European climate risk assessment. Publications Office of the European Union. Available at: <u>https://data.europa.eu/doi/10.2800/8671471</u> (Accessed October 21, 2024).
- European Environment Agency, Prutsch, A., Sanderson, M., Downing, C., Hildén, M., Leitner, M., et al. (2018). National climate change vulnerability and risk assessments in Europe, 2018. Publications Office. Available at: <u>https://data.europa.eu/ doi/10.2800/348489</u> (Accessed October 21, 2024).
- European Environmental Agency (2020). Monitoring and evaluation of national adaptation policies throughout the policy cycle. European Environmental Agency. Available at: <u>https://data.europa.eu/doi/10.2800/83221</u> (Accessed October 17, 2024).
- Exner, A., and Kratzwald, B. (2021). Solidarische Ökonomie & Commons: INTRO Eine Einführung. Wien: Mandelbaum Verlag eG.
- Fanning, A. L., and Hickel, J. (2023). Compensation for atmospheric appropriation. Nat Sustain 6, 1077–1086. <u>https://doi.org/10.1038/s41893-023-01130-8</u>
- Fanning, A. L., O'Neill, D. W., Hickel, J., and Roux, N. (2022). The social shortfall and ecological overshoot of nations. *Nat Sustain* 5, 26–36. <u>https://doi.org/10.1038/s41893-021-00799-z</u>
- FAO (2021). The state of the world's land and water resources for food and agriculture: Systems at breaking point (SOLAW 2021). Food and Agriculture Organization of the United Nations. Available at: <u>https://www.fao.org/policy-support/tools-and-publications/resources-details/en/c/1507277/</u> (Accessed May 1, 2024).
- Fatehi, S., Qi, A., Wei, L., and Zhang, J. (2023). Greenwashing under Competition. SSRN Journal. <u>https://doi.org/10.2139/ssrn.</u> <u>4433364</u>
- Fedele, G., Donatti, C. I., Harvey, C. A., Hannah, L., and Hole, D. G. (2019). Transformative adaptation to climate change for sustainable social-ecological systems. *Environmental Science & Policy* 101, 116–125. <u>https://doi.org/10.1016/j.envsci.2019.07.001</u>
- Feindt, S., Kornek, U., Labeaga, J. M., Sterner, T., and Ward, H. (2021). Understanding regressivity: Challenges and opportunities of European carbon pricing. *Energy Economics* 103, 105550. https://doi.org/10.1016/j.eneco.2021.105550
- Fischer, A. M. (2020). The Dark Sides of Social Policy: From Neoliberalism to Resurgent Right-wing Populism. *Development and Change* 51, 371–397. <u>https://doi.org/10.1111/dech.12577</u>
- Fischer, K., Eder, J., and Schaffartzik, A. (2023). "Kapitel 15. Globalisierung: globale Warenketten und Arbeitsteilung", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. Steininger, et al. (Berlin/Heidelberg: Springer Spektrum). Available at: <u>https://papers.ssrn.com/abstract=4225571</u> (Accessed September 13, 2023).
- Fischer-Kowalski, M., and Haas, W. (2014). "Exploring the Transformation of Human Labour in Relation to Socio-ecological Transitions", in *Let's Get to Work! The Future of Labour in Europe*, eds. M. Beblavy, M. Veselková, M. Beblavy, and M. Veselková (Brussels: CEPS), 56–82. Available at: <u>https://www.researchgate.net/publication/266141913\_Exploring\_the\_transformation\_of\_human\_labour\_in\_relation\_to\_socio-ecological\_transitions</u> (Accessed May 16, 2024).

- Fisher, S. (2015). The emerging geographies of climate justice. *The Geographical Journal* 181, 73–82. <u>https://doi.org/10.1111/</u> geoj.12078
- Fleck, L. (1980). Entstehung und Entwicklung einer wissenschaftlichen Tatsache: Einführung in die Lehre vom Denkstil und Denkkollektiv. Berlin: Suhrkamp.
- Foundational Economy Collective ed. (2018). *Foundational Economy: The Infrastructure of Everyday Life.* Manchester University Press.
- Fraser, N. (2023). Cannibal Capitalism: How our System is Devouring Democracy, Care, and the Planet and What We Can Do About It. London, Brooklyn: Verso Books.
- Fremstad, A., and Paul, M. (2022). Neoliberalism and climate change: How the free-market myth has prevented climate action. *Ecological Economics* 197, 107353. <u>https://doi.org/10.1016/j.ecolecon.2022.107353</u>
- Frenken, K., and Punt, M. B. (2023). A New View on Radical Innovation., in *International Sustainability Transitions conference* 2023, (Utrecht). Available at: <u>https://osf.io/6cr5t/download</u>
- Friedlingstein, P., Andrew, R. M., Rogelj, J., Peters, G. P., Canadell, J. G., Knutti, R., et al. (2014). Persistent growth of CO<sub>2</sub> emissions and implications for reaching climate targets. *Nature Geo*sci 7, 709–715. <u>https://doi.org/10.1038/ngeo2248</u>
- Friedlingstein, P., O'Sullivan, M., Jones, M. W., Andrew, R. M., Bakker, D. C. E., Hauck, J., et al. (2023). Global Carbon Budget 2023. *Earth System Science Data* 15, 5301–5369. <u>https://doi. org/10.5194/essd-15-5301-2023</u>
- Friedmann, J. (2005). Globalization and the emerging culture of planning. *Progress in Planning* 64, 183–234. <u>https://doi.org/ 10.1016/j.progress.2005.05.001</u>
- Fritz, M., and Eversberg, D. (2023). Support for eco-social policy from a class perspective: Responsibilities, redistribution, regulation and rights. *European Journal of Social Security* 25, 484–505. <u>https://doi.org/10.1177/13882627231208929</u>
- Fritz, M., and Eversberg, D. (2024). Mentalities, classes and the four lines of conflict in the social-ecological transformation. *Eur Polit Sci* 23, 39–55. <u>https://doi.org/10.1057/s41304-023-00457-2</u>
- Froy, F., Heroy, S., Uyarra, E., and O'Clery, N. (2022). What drives the creation of green jobs, products and technologies in cities and regions? Insights from recent research on green industrial transitions. *Local Economy* 37, 584–601. <u>https://doi.org/10.1177/02690942231170135</u>
- Frühwald, J., Maringer, F., Pixer, A., Rogenhofer, K., and Wendelin, W. (2024). "[KON]TEXTANALYSE #1: Klimadebatte in Österreich: Zwischen Apellen und Verschleppung: Klimadiskurs-Monitoring 2023.", in KONTEXT – Institut für Klimafragen, (Wien). Available at: <u>https://kontext-institut.at/inhalte/ kontextanalyse-klimadebatte/</u> (Accessed May 3, 2024).
- Fuchs, D. A., Sahakian, M., Gumbert, T., Di Giulio, A., Maniates, M., Lorek, S., et al. (2021). Consumption corridors: living a good life within sustainable limits. London: Routledge. <u>https://doi.org/10.4324/9780367748746</u>
- Fuchs, D., and Dolinga, S. (2022). "Corporate power and the shaping of sustainability governance", in *The Routledge Handbook of Democracy and Sustainability*, eds. B. Bornemann, H. Knappe, and P. Nanz (Abingdon, Oxon; New York, NY: Routledge). <u>https://doi.org/10.4324/9780429024085</u>
- Fuchs, D., and Lederer, M. M. (2007). The Power of Business. Business and Politics 9, 1–17. <u>https://doi.org/10.2202/1469-3569.1214</u>

- Fuchs, D., Lorek, S., Mamut, P., and Rossmoeller, A. (2023). "Fifty Shades of Sufficiency: Semantic Confusion and No Policy", in *Routledge Handbook of Environmental Policy*, eds. H. Jörgens, C. Knill, and Y. Steinebach (London: Routledge). <u>https://doi. org/10.4324/9781003043843</u>
- Gabor, D. (2020). Critical macro-finance: A theoretical lens. Finance and Society 6, 45–55. <u>https://doi.org/10.2218/finsoc.v6i1.</u> <u>4408</u>
- Gabor, D. (2023). The (European) derisking state. *Stato e mercato*, 53–84. <u>https://doi.org/10.1425/107674</u>
- Gabor, D., and Braun, B. (2023). Green macrofinancial regimes. https://doi.org/10.31235/osf.io/4pkv8
- Gabor, D., and Sylla, N. S. (2023). Derisking Developmentalism: A Tale of Green Hydrogen. *Development and Change* 54, 1169– 1196. <u>https://doi.org/10.1111/dech.12779</u>
- Geels, F. W. (2011). The multi-level perspective on sustainability transitions: Responses to seven criticisms. *Environmental Innovation and Societal Transitions* 1, 24–40. <u>https://doi.org/10. 1016/j.eist.2011.02.002</u>
- Geels, F. W. (2014). Regime Resistance against Low-Carbon Transitions: Introducing Politics and Power into the Multi-Level Perspective. *Theory, Culture & Society* 31, 21–40. <u>https://doi.org/10.1177/0263276414531627</u>
- Geels, F. W., and Raven, R. (2006). Non-linearity and Expectations in Niche-Development Trajectories: Ups and Downs in Dutch Biogas Development (1973–2003). *Technology Analysis & Strategic Management* 18, 375–392. <u>https://doi.org/10.1080/09537320600777143</u>
- Gerasimcikova, A. (2023). The EU's Green Deal Industrial Plan Is a Windfall for Corporations. *Jacobin*. Available at: <u>https://jacobin. com/2023/03/eu-green-deal-industrial-plan-corporate-handouts-renewables</u> (Accessed April 8, 2024).
- Getzner, M. (2009). Determinants of (de-) materialization of an industrialized small open economy. *International Journal of Ecological Economics and Statistics* 14, 3–13. <u>http://hdl.handle.net/20.500.12708/167374</u>
- Getzner, M. (2022). Sind Hoffnungen in Bezug auf privates Kapital im Sinne von "nachhaltigen Investitionen" als Treiber für den sozialökologischen Umbau im Gebäudebereich berechtigt? *Der* öffentliche Sektor 48. <u>https://doi.org/10.34749/oes.2022.4630</u>
- Getzner, M., Bröthaler, J., Neuhuber, T., Dillinger, T., Grinzinger, E., and Kanonier, A. (n.d.). Drivers of unsustainable local land-use decisions: A comprehensive study for Austria. *Land Use Policy*.
- Getzner, M., and Kadi, J. (2020). Determinants of land consumption in Austria and the effects of spatial planning regulations. *European Planning Studies* 28, 1095–1117. <u>https://doi.org/10. 1080/09654313.2019.1604634</u>
- Gilli, M., Calcaterra, M., Emmerling, J., and Granella, F. (2024). Climate change impacts on the within-country income distributions. *Journal of Environmental Economics and Management* 127, 103012. <u>https://doi.org/10.1016/j.jeem.2024.103012</u>
- Glas, N., Kunesch, S., Offenthaler, I., and Leitner, M. (2020). Mind the Gap Relevance & Awareness of Financial Market Actors for the Low Carbon Transition. *RiskFinPorto – Working Paper No. 1.* Available at: <u>https://www.researchgate.net/profile/Natalie-Glas/publication/348277710\_Mind\_the\_Gap\_Relevance\_ Awareness\_of\_Financial\_Market\_Actors\_for\_the\_Low\_Carbon\_Transition/links/5ff5ec1345851553a0234bd1/Mind-the-Gap-Relevance-Awareness-of-Financial-Market-Actors-forthe-Low-Carbon-Transition.pdf (Accessed May 10, 2024).</u>

- Global Witness (2022). Over 100 more fossil fuel lobbyists than last year, flooding crucial COP climate talks. *Global Witness*. Available at: <u>https://www.globalwitness.org/en/press-releases/ over-100-more-fossil-fuel-lobbyists-last-year-flooding-crucialcop-climate-talks/ (Accessed March 15, 2024).</u>
- Görg, C., Brand, U., Haberl, H., Hummel, D., Jahn, T., and Liehr, S. (2017). Challenges for Social-Ecological Transformations: Contributions from Social and Political Ecology. *Sustainability* 9, 1045. <u>https://doi.org/10.3390/su9071045</u>
- Gorz, A. (1968). Reform and Revolution. Socialist Register 5. Available at: <u>https://socialistregister.com/index.php/srv/article/view/ 5272</u> (Accessed May 15, 2024).
- Gorz, A. (1993). Political ecology: expertocracy versus self-limitation. New Left Review, 55–67.
- Gosseries, A. (2004). Historical Emissions and Free-Riding. *Ethical Perspectives* 11, 36–60. <u>https://doi.org/10.2143/EP.11.1.504779</u>
- Gough, I. (2015). Climate change and sustainable welfare: the centrality of human needs. *Cambridge J Econ* 39, 1191–1214. https://doi.org/10.1093/cje/bev039
- Gough, I. (2017). Heat, Greed and Human Need: Climate Change, Capitalism and Sustainable Wellbeing. Cheltenham, UK and Northampton, MA: Edward Elgar Publishing. Available at: <u>https://www.e-elgar.com/shop/gbp/heat-greed-and-human-need-9781785365102.html</u>
- Gough, I. (2022). Two Scenarios for Sustainable Welfare: A Framework for an Eco-Social Contract. *Social Policy and Society* 21, 460–472. https://doi.org/10.1017/S1474746421000701
- Gough, I. (2023). Sufficiency as a Value Standard: From Preferences to Needs. Ethics, Policy & Environment, 1–22. <u>https://doi.org/ 10.1080/21550085.2023.2269055</u>
- Gourdel, R., Monasterolo, I., Dunz, N., Mazzocchetti, A., and Parisi, L. (2022). The Double Materiality of Climate Physical and Transition Risks in the Euro Area. ECB Working Paper No. 2022/2665. <u>https://doi.org/10.2139/ssrn.4115997</u>
- Goydke, T., and Koch, G. (2020). *Economy for the Common Good:* A Common Standard for a Pluralist World?. tredition.
- Graebner-Radkowitsch, C., and Strunk, B. (2023). Degrowth and the Global South: the twin problem of global dependencies. *ICAE Working Papers*. Available at: <u>https://ideas.repec.org//p/ ico/wpaper/142.html</u> (Accessed May 15, 2024).
- Green, F., and Healy, N. (2022). How inequality fuels climate change: The climate case for a Green New Deal. *One Earth* 5, 635–649. https://doi.org/10.1016/j.oneear.2022.05.005
- Green, J., Hadden, J., Hale, T., and Mahdavi, P. (2022). Transition, hedge, or resist? Understanding political and economic behavior toward decarbonization in the oil and gas industry. *Review* of International Political Economy 29, 2036–2063. <u>https://doi.or</u> g/10.1080/09692290.2021.1946708
- Großer, E., Jorck, G. von, Kludas, S., Mundt, I., and Sharp, H. (2020). Sozial-ökologische Infrastrukturen – Rahmenbedingungen für Zeitwohlstand und neue Formen von Arbeit. Ökologisches Wirtschaften – Fachzeitschrift 33, 14–16. <u>https://doi.org/10.14512/OEW350414</u>
- Großmann, A., Stocker, A., and Wolter, M. I. (2019). Evaluation von Klimaschutzmaßnahmen mit dem Modell e3.at meetPASS: meeting the Paris Agreement and Supporting Sustainability Working Paper No. 5. <u>https://doi.org/10.13140/RG.2.2.32733.</u> 44000
- Gsottbauer, E., Kirchler, M., and König-Kersting, C. (2024). Financial professionals and climate experts have diverging perspec-

tives on climate action. Commun Earth Environ 5, 159. https:// doi.org/10.1038/s43247-024-01331-9

- Gugele, B., Krutzler, T., Miess, M., and Vogel, J. (2022). Pilotprojekt: Integration eines Klimamoduls in die langfristige Budgetprognose. Endbericht. Wien: Umweltbundesamt (Environment Agency Austria). Available at: <u>https://www.umweltbundesamt.</u> <u>at/fileadmin/site/publikationen/rep0837.pdf</u> (Accessed January 20, 2023).
- Guth, M., Hesse, J., Königswieser, C., Krenn, G., Lipp, C., Neudorfer, B., et al. (2021). OeNB climate risk stress test-modeling a carbon price shock for the Austrian banking sector. Oesterreichische Nationalbank (Austrian Central Bank). Available at: https://www.researchgate.net/profile/Martin-Hafner-Guth/ publication/356987810\_OeNB\_climate\_risk\_stress\_test\_modeling\_a\_carbon\_price\_shock\_for\_the\_Austrian\_banking\_ sector/links/61b728a31d88475981e76121/OeNB-climate-riskstress-test-modeling-a-carbon-price-shock-for-the-Austrianbanking-sector.pdf (Accessed September 14, 2023).
- Gutsche, G., Wetzel, H., and Ziegler, A. (2023). Determinants of individual sustainable investment behavior – A framed field experiment. *Journal of Economic Behavior & Organization* 209, 491–508. https://doi.org/10.1016/j.jebo.2023.03.016
- Guzzardi, D., Palagi, E., Faccio, T., and Roventini, A. (2023). "10. In Search of Lost Time: An Ensemble of Policies to Restore Fiscal Progressivity and Address the Climate Challenge", in *Financing Investment in Times of High Public Debt*, eds. F. Cerniglia, F. Saraceno, A. Watt, F. Cerniglia, F. Saraceno, and A. Watt (Cambridge, UK: Open Book Publishers), 169–190. Available at: <u>https://www.openbookpublishers.com/books/10.11647/obp.</u> 0386/chapters/10.11647/obp.0386.10 (Accessed June 8, 2024).
- Haas, T., and Sander, H. (2020). Decarbonizing Transport in the European Union: Emission Performance Standards and the Perspectives for a European Green Deal. *Sustainability* 12, 8381. https://doi.org/10.3390/su12208381
- Haas, W., Muhar, A., Dorninger, C., and Gugerell, K. (2023). "Kapitel 23. Synthese: Pfade zur Transformation struktureller Bedingungen für ein klimafreundliches Leben", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 613–647. Available at: https://doi.org/10.1007/978-3-662-66497-1\_27
- Haberl, H., Wiedenhofer, D., Virág, D., Kalt, G., Plank, B., Brockway, P., et al. (2020). A systematic review of the evidence on decoupling of GDP, resource use and GHG emissions, part II: synthesizing the insights. *Environ. Res. Lett.* 15, 065003. <u>https:// doi.org/10.1088/1748-9326/ab842a</u>
- Hache, E. (2022). Pour une géopolitique de la sobriété. Revue internationale et stratégique 128, 43–47. <u>https://doi.org/10.3917/</u> ris.128.0043
- Haderer, M. (2023). Experimental climate governance as organized irresponsibility? A case for revamping governing (also) through government. Sustainability: Science, Practice and Policy 19, 2186078. https://doi.org/10.1080/15487733.2023.2186078
- Haderer, M., Brand, U., Daniel, A., Exner, A., Fankhauser, J., Görg, C., et al. (2023). "Kapitel 28. Theorien des Wandels und der Gestaltung von Strukturen: Gesellschaft-Natur-Perspektive", in *APCC Special Report: Strukturen für ein klimafreundliches Leben*, (Berlin, Heidelberg: Springer Spektrum), 691–715. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_32</u> (Accessed October 23, 2024).

- Hale, T., Smith, S. M., Black, R., Cullen, K., Fay, B., Lang, J., et al. (2022). Assessing the rapidly-emerging landscape of net zero targets. *Climate Policy* 22, 18–29. <u>https://doi.org/10.1080/1469</u> <u>3062.2021.2013155</u>
- Hammond, M. (2020). Sustainability as a cultural transformation: the role of deliberative democracy. *Environmental Politics* 29, 173–192. https://doi.org/10.1080/09644016.2019.1684731
- Hanger-Kopp, S., Thaler, T., Seebauer, S., Schinko, T., and Clar, C. (2022). Defining and operationalizing path dependency for the development and monitoring of adaptation pathways. *Global Environmental Change* 72, 102425. <u>https://doi.org/10.1016/j.gloenvcha.2021.102425</u>
- Hansen, G. (2023). Coal, oil and gas going into extra time: the narrative of abated fossil fuels threatens to undermine the Paris climate targets. SWP Comment. <u>https://doi.org/10.18449/2023C54</u>
- Harnecker, M., and Bartolomé, J. (2019). *Planning from Below: A Decentralized Participatory Planning Proposal*. NYU Press.
- Hartmann, S., Pedoth, L., Dalla Torre, C., and Schneiderbauer, S. (2021). Beyond the Expected—Residual Risk and Cases of Overload in the Context of Managing Alpine Natural Hazards. *Int J Disaster Risk Sci* 12, 205–219. <u>https://doi.org/10.1007/</u> s13753-020-00325-3
- Hausknost, D. (2023). Opacity and Transparency: On the 'Passive Legitimacy' of Capitalist Democracy. *Theoria* 70, 26–53. <u>https:// doi.org/10.3167/th.2023.7017702</u>
- Hausknost, D., and Haas, W. (2019). The Politics of Selection: Towards a Transformative Model of Environmental Innovation. *Sustainability* 11, 506. <u>https://doi.org/10.3390/su11020506</u>
- Hausknost, D., Schriefl, E., Lauk, C., and Kalt, G. (2017). A Transition to Which Bioeconomy? An Exploration of Diverging Techno-Political Choices. *Sustainability* 9, 669. <u>https://doi.org/ 10.3390/su9040669</u>
- He, X. (2013). Mainstreaming adaptation in integrated water resources management in China: from challenge to change. Water Policy 15, 895–921. <u>https://doi.org/10.2166/wp.2013.084</u>
- Heyen, D. A., and Wicki, M. (2024). Increasing public support for climate policy proposals: a research agenda on governable acceptability factors. *Climate Policy* 24, 785–794. <u>https://doi.org/ 10.1080/14693062.2024.2330390</u>
- Hickel, J. (2020a). Less is More: How Degrowth Will Save the World. London: William Heinemann.
- Hickel, J. (2020b). Quantifying national responsibility for climate breakdown: an equality-based attribution approach for carbon dioxide emissions in excess of the planetary boundary. *The Lancet Planetary Health* 4, e399–e404. <u>https://doi.org/10.1016/ S2542-5196(20)30196-0</u>
- Hickel, J. (2021). What does degrowth mean? A few points of clarification. *Globalizations* 18, 1105–1111. <u>https://doi.org/10.1080/</u> <u>14747731.2020.1812222</u>
- Hirvilammi, T., Häikiö, L., Johansson, H., Koch, M., and Perkiö, J. (2023). Social Policy in a Climate Emergency Context: Towards an Ecosocial Research Agenda. *Journal of Social Policy* 52, 471. <u>https://doi.org/10.1017/S0047279423000053</u>
- Hirvilammi, T., and Koch, M. (2020). Sustainable Welfare beyond Growth. Sustainability 12, 1824. <u>https://doi.org/10.3390/ su12051824</u>
- Hitzl, K. (2023). Auswirkungen des Klimawandels auf die österreichische Sicherheitspolitik. Vienna: BOKU University. Available at: <u>http://permalink.obvsg.at/bok/AC16818765</u> (Accessed May 16, 2024).

- Hofbauer, J., Gerold, S., Klaus, D., and Wukovitsch, F. (2023). "Kapitel 7. Erwerbsarbeit", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer, Spektrum), 285–307. <u>https://doi.org/10.1007/978-3-662-66497-1\_11</u>
- Hoff, H., Häyhä, T., and Cornell, S. E. (2017). Bringing EU Policy into Line with the Planetary Boundaries. *Stockholm Environment Institute Discussion Brief*. Available at: <u>https://www. sei.org/mediamanager/documents/Publications/SEI-2017-DB-Bringing-EU-policy-v2.pdf</u> (Accessed October 10, 2024).
- Höhne, N., den Elzen, M., and Escalante, D. (2014). Regional GHG reduction targets based on effort sharing: a comparison of studies. *Climate Policy* 14, 122–147. <u>https://doi.org/10.1080/146930</u> <u>62.2014.849452</u>
- Holland, N., and Tansey, R. (2022). A loud lobby for a silent spring: The pesticide industry's toxic lobbying tactics against Farm to Fork. Available at: <u>https://corporateeurope.org/en/2022/03/</u> <u>loud-lobby-silent-spring</u> (Accessed May 16, 2024).
- Hollaus, B. (2024). Das Urteil des EGMR im Fall KlimaSeniorinnen und seine Implikationen für den europäischen Grundrechtsschutz. *JBl – Juristische Blätter* 8, 485.
- Hollaus, B., Madner, V., and Schulev-Steindl, E. (2023). "Kapitel 11. Recht", in *APCC Special Report: Strukturen für ein klima-freundliches Leben*, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 351–388. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_15</u> (Accessed October 4, 2023).
- Horvath, S.-M., Muhr, M. M., Kirchner, M., Toth, W., Germann, V., Hundscheid, L., et al. (2022). Handling a complex agenda: A review and assessment of methods to analyse SDG entity interactions. *Environmental Science & Policy* 131, 160–176. <u>https:// doi.org/10.1016/j.envsci.2022.01.021</u>
- Hüther, M., Fremerey, M., and Iglesias, S. G. (2024). Zur Zukunft des Wirtschaftsstandorts Deutschland: sortierende Bemerkungen. Wirtschaftsdienst 104, 175–179. <u>https://doi.org/10.2478/</u> wd-2024-0050
- Huwe, V., Hopkins, D., and Mattioli, G. (2024). Aviation exceptionalism, fossil fuels and the state. *Review of International Political Economy*, 1–25. <u>https://doi.org/10.1080/09692290.2024.23849</u> 25
- Ikegami, M., and Wang, Z. (2023). Does military expenditure crowd out health-care spending? Cross-country empirics. *Qual Quant* 57, 1657–1672. <u>https://doi.org/10.1007/s11135-022-01412-x</u>
- InfluenceMap (2023). The Oil and Gas Industry's Policy Advocacy in Africa and Europe: How the sector is promoting fossil gas lock-in despite IPCC warnings. Available at: <u>https://influencemap.org/report/The-European-Oil-Gas-Industry-in-Europe-Africa-22693</u> (Accessed May 16, 2024).
- Institut Rousseau (2024). Road to Net Zero. Bridging the Green Investment Gap. Institut Rousseau. Available at: <u>https://drive.goo-gle.com/file/u/0/d/1cDQmQB0iezwthxiaoM83VxbllhrNz-0O/view?usp=drive\_link&pli=1&usp=embed\_facebook</u> (Accessed May 8, 2024).
- INTEGRAL (2022). Mutter Erde Klima-Studie 2022. Wien. Available at: <u>https://www.muttererde.at/wp-content/up-loads/2022/11/PK17112022-Studienzusammenfassung.pdf</u> (Accessed June 3, 2024).
- IPCC (2022a). "Annex I: Glossary [van Diemen, R., J.B.R. Matthews, V. Möller, J.S. Fuglestvedt, V. Masson-Delmotte, C. Méndez, A. Reisinger, S. Semenov (eds)]", in *IPCC, 2022: Climate*

Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, eds. P. R. Shukla, J. Skea, R. Slade, A. Al Khourdajie, R. van Diemen, D. McCollum, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press). https://doi.org/10.1017/9781009157926.020

- IPCC (2022b). Climate Change 2022 Mitigation of Climate Change: Working Group III Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change., 1st Edn, eds. P. R. Shukla, J. Skea, R. Slade, A. A. Khourdajie, A. Hasija, J. Malley, et al. Cambridge, UK and New York, NY, USA: Cambridge University Press. <u>https://doi.org/10.1017/9781009157926</u>
- IPCC (2022c). "Summary for Policymakers", in Global Warming of 1.5°C: IPCC Special Report on Impacts of Global Warming of 1.5°C above Pre-industrial Levels in Context of Strengthening Response to Climate Change, Sustainable Development, and Efforts to Eradicate Poverty, eds. V. Masson-Delmotte, P. Zhai, H.-O. Pörtner, D. Roberts, J. Skea, P. R. Shukla, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press), 3–24. https://doi.org/10.1017/9781009157940
- IPCC (2022d). Summary for Policymakers. In: Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change [P.R. Shukla, J. Skea, R. Slade, A. Al Khourdajie, R. van Diemen, D. McCollum, M. Pathak, S. Some, P. Vyas, R. Fradera, M. Belkacemi, A. Hasija, G. Lisboa, S. Luz, J. Malley, (eds.)]. Cambridge, UK and New York, NY, USA: Cambridge University Press. <u>https://doi.org/ 10.1017/9781009157926.001</u>
- Isenhour, C. (2016). Unearthing human progress? Ecomodernism and contrasting definitions of technological progress in the Anthropocene. *Economic Anthropology* 3, 315–328. <u>https://doi. org/10.1002/sea2.12063</u>
- Jacobi, N., Jäger, A., Rothballer, C., Reitemeyer, F., Kropp, J., Lexer, W., et al. (2024). Abschlussbericht. Klimaorientierte Stadtentwicklung. Treibhausgasminderungspotenziale in synergetischen Handlungsfeldern. German government environmental agency (UBA). Available at: <u>https://www.umweltbundesamt.de/publikationen/klimaorientierte-stadtentwicklung</u> (Accessed October 24, 2024).
- Jaeggi, R. (2014). *Kritik von Lebensformen*. Berlin: Suhrkamp Verlag.
- Jaeggi, R. (2023). *Fortschritt und Regression.*, Originalausgabe, erste Auflage. Berlin: Suhrkamp Verlag.
- Jahn, T., Hummel, D., Drees, L., Liehr, S., Lux, A., Mehring, M., et al. (2020). Sozial-ökologische Gestaltung im Anthropozän. GAIA – Ecological Perspectives for Science and Society 29, 93–97. <u>https://doi.org/10.14512/gaia.29.2.6</u>
- Jessop, B. (2004). "Multi-level Governance and Multi-level Metagovernance: Changes in the European Union as Integral Moments in the Transformation and Reorientation of Contemporary Statehood", in *Multi-level Governance*, eds. I. Bache and M. Flinders (Oxford: online edn, Oxford Academic). Available at: <u>https://academic.oup.com/book/10675/chapter-abstract/158716449?redirectedFrom=fulltext</u> (Accessed May 16, 2024).
- Jessop, B. (2010). "Metagovernance", in *The SAGE Handbook of Governance*, ed. M. Bevir (London: SAGE Publications Ltd), 106–123. Available at: <u>https://sk.sagepub.com/reference/hdbk\_governance/n8.xml</u> (Accessed February 28, 2024).

- Jo, T.-H., and Todorova, Z. (2019). "Social Provisioning Process: A heterodox view of the economy", in *The Routledge Handbook* of Heterodox Economics: Theorizing, Analyzing, and Transforming Capitalism, eds. T.-H. Jo, L. Chester, C. D'Ippoliti (London: Routledge), 29–40.
- Jones, A. W. (2015). Perceived barriers and policy solutions in clean energy infrastructure investment. *Journal of Cleaner Production* 104, 297–304. <u>https://doi.org/10.1016/j.jclepro.2015.05.072</u>
- Jorgenson, A. K., Fiske, S., Hubacek, K., Li, J., McGovern, T., Rick, T., et al. (2019). Social science perspectives on drivers of and responses to global climate change. WIREs Climate Change 10, e554. <u>https://doi.org/10.1002/wcc.554</u>
- Jungell-Michelsson, J., and Heikkurinen, P. (2022). Sufficiency: A systematic literature review. *Ecological Economics* 195, 107380. https://doi.org/10.1016/j.ecolecon.2022.107380
- Kaika, M., Varvarousis, A., Demaria, F., and March, H. (2023). Urbanizing degrowth: Five steps towards a Radical Spatial Degrowth Agenda for planning in the face of climate emergency. Urban Studies 60, 1191–1211. <u>https://doi.org/10.1177/ 00420980231162234</u>
- Kalinowski, T. (2024). The Green Climate Fund and private sector climate finance in the Global South. *Climate Policy* 24, 281–296. <u>https://doi.org/10.1080/14693062.2023.2276857</u>
- Kalpokas, I. (2018). A Political Theory of Post-Truth. Cham: Palgrave Pivot Cham. Available at: <u>http://link.springer.com/10.1007/978-3-319-97713-3</u> (Accessed April 22, 2024).
- Kapeller, J., and Hubmann, G. (2023). Dilemmata marktliberaler Globalisierung: Globale Freiheit durch globalen Wettbewerb? *ifso working paper series* 27. Available at: <u>https://ideas.repec.org//p/zbw/ifsowp/27.html</u> (Accessed May 15, 2024).
- Karlsson, M., Alfredsson, E., and Westling, N. (2020). Climate policy co-benefits: a review. *Climate Policy* 20, 292–316. <u>https://doi. org/10.1080/14693062.2020.1724070</u>
- Kattel, R., Drechsler, W., and Karo, E. (2022). How to Make an Entrepreneurial State: Why Innovation Needs Bureaucracy. New Haven and London: Yale University Press.
- Kattel, R., and Mazzucato, M. (2018). Mission-oriented innovation policy and dynamic capabilities in the public sector. *Industrial* and Corporate Change 27, 787–801. <u>https://doi.org/10.1093/icc/ dty032</u>
- Kazepov, Y., Colombo, F., and Saruis, T. (2019). "The multi-scalar puzzle of social innovation", in *Local Social Innovation to Combat Poverty and Exclusion*, (Bristol, UK: Policy Press), 91–112. Available at: <u>https://bristoluniversitypressdigital.com/</u> <u>edcollchap/book/9781447338451/ch005.xml</u> (Accessed May 15, 2024).
- Kedward, K., Gabor, D., and Ryan-Collins, J. (2024). Carrots with(out) sticks: credit policy and the limits of green central banking. *Review of International Political Economy* 31, 1593– 1617. <u>https://doi.org/10.1080/09692290.2024.2351838</u>
- Keil, A. K., and Kreinin, H. (2022). Slowing the treadmill for a good life for All? German trade union narratives and social-ecological transformation. *Journal of Industrial Relations* 64, 564–584. <u>https://doi.org/10.1177/00221856221087413</u>
- Keil, A. K., and Steinberger, J. K. (2023). Cars, capitalism and ecological crises: understanding systemic barriers to a sustainability transition in the German car industry. *New Political Economy* 29, 1–21. <u>https://doi.org/10.1080/13563467.2023.2223132</u>
- Kelsen, H. (1929). Vom Wesen und Wert der Demokratie. Tübingen: Mohr Siebeck.

- Kemp-Benedict, E., and Kartha, S. (2019). Environmental financialization: what could go wrong? *real-world economics review* 87, 69–89. <u>https://anzsee.org.au/wp-content/uploads/2019/03/</u> whole871.pdf#page=69
- Keohane, R. O. (2015). The Global Politics of Climate Change: Challenge for Political Science. PS: Political Science & Politics 48, 19–26. <u>https://doi.org/10.1017/S1049096514001541</u>
- Keohane, R. O., and Victor, D. G. (2011). The Regime Complex for Climate Change. Perspect. polit. 9,7–23. <u>https://doi.org/10.1017/ \$1537592710004068</u>
- Kern, K., Jörgens, H., and Jänicke, M. (2001). The Diffusion of Environmental Policy Innovations: A Contribution to the Globalisation of Environmental Policy. Berlin: Wissenschaftszentrum Berlin für Sozialforschung. Available at: <u>https://nbn-resolving.org/urn:nbn:de:0168-ssoar-114964</u> (Accessed May 14, 2024).
- Khan, J., Emilsson, K., Fritz, M., Koch, M., Hildingsson, R., and Johansson, H. (2023). Ecological ceiling and social floor: public support for eco-social policies in Sweden. *Sustain Sci* 18, 1519– 1532. <u>https://doi.org/10.1007/s11625-022-01221-z</u>
- Kimmerer, R. W. (2013). Braiding Sweetgrass: Indigenous Wisdom, Scientific Knowledge and the Teachings of Plants. Minneapolis, Minn: Milkweed Editions.
- Kirchner, M., Sommer, M., Kratena, K., Kletzan-Slamanig, D., and Kettner-Marx, C. (2019). CO<sub>2</sub> taxes, equity and the double dividend – Macroeconomic model simulations for Austria. *Energy Policy* 126, 295–314. <u>https://doi.org/10.1016/j.</u> enpol.2018.11.030
- Kivimaa, P., and Kern, F. (2016). Creative destruction or mere niche support? Innovation policy mixes for sustainability transitions. *Research Policy* 45, 205–217. <u>https://doi.org/10.1016/j.</u> <u>respol.2015.09.008</u>
- Kivimaa, P., Laakso, S., Lonkila, A., and Kaljonen, M. (2021). Moving beyond disruptive innovation: a review of disruption in sustainability transitions. *Environmental Innovation and Societal Transitions* 38, 110–126.
- Klenert, D., Mattauch, L., Combet, E., Edenhofer, O., Hepburn, C., Rafaty, R., et al. (2018). Making carbon pricing work for citizens. *Nature Clim Change* 8, 669–677. <u>https://doi.org/10.1038/</u> <u>s41558-018-0201-2</u>
- Kletzan-Slamanig, D., and Köppl, A. (2021). The evolution of the green finance agenda – Institutional anchoring and a survey-based assessment for Austria. WIFO Working Papers. Available at: <u>https://www.econstor.eu/handle/10419/249260</u> (Accessed September 14, 2023).
- Kletzan-Slamanig, D., and Köppl, A. (2022). "Green finance: contribution to climate policy, supporting factors and barriers", in *Green Deals in the Making*, (Edward Elgar Publishing), 127–140. Available at: <u>https://www.elgaronline.com/edcollchap/book/9781803926780/book-part-9781803926780-20.xml</u> (Accessed September 15, 2023).
- Kletzan-Slamanig, D., Köppl, A., Sinabell, F., Kirchmayr, S., Müller, S., Rimböck, A., et al. (2022). Analyse klimakontraproduktiver Subventionen in Österreich. Vienna: Österreichisches Institut für Wirtschaftsforschung (WIFO). Available at: <u>https://www. wifo.ac.at/jart/prj3/wifo/resources/person\_dokument/person\_ dokument.jart?publikationsid=69687&mime\_type=application/pdf</u> (Accessed September 15, 2023).

- Koch, M. (2018). "Sustainable welfare, degrowth and eco-social
- policies in Europe", in *Social policy in the European Union: state* of play, eds. B. Vanhercke, S. Sabato, and S. Spasova (Brussels: ETUI), 35–50. Koch, M. (2020). The state in the transformation to a sustain-
- able postgrowth economy. *Environmental Politics* 29, 115–133. https://doi.org/10.1080/09644016.2019.1684738
- Koch, M. (2022). Social Policy Without Growth: Moving Towards Sustainable Welfare States. Social Policy and Society 21, 447– 459. <u>https://doi.org/10.1017/S1474746421000361</u>
- Koch, M., and Buch-Hansen, H. (2021). In search of a political economy of the postgrowth era. *Globalizations* 18, 1219–1229. <u>https://doi.org/10.1080/14747731.2020.1807837</u>
- Koch, M., Buchs, M., and Lee, J. (2023). Towards a New Generation of Social Policy: Commonalities between Sustainable Welfare and the IPCC. *Politiche Sociali / Social Policies* 1, 27–42. <u>https:// doi.org/10.7389/107137</u>
- Koch, M., and Mont, O. (2016). Sustainability and the Political Economy of Welfare., 1. Edition. London; New York: Routledge.
- Kohler, F., Holland, T. G., Kotiaho, J. S., Desrousseaux, M., and Potts, M. D. (2019). Embracing diverse worldviews to share planet Earth. *Conservation Biology* 33, 1014–1022. <u>https://doi. org/10.1111/cobi.13304</u>
- Köhler, J., Geels, F. W., Kern, F., Markard, J., Onsongo, E., Wieczorek, A., et al. (2019). An agenda for sustainability transitions research: State of the art and future directions. *Environmental Innovation and Societal Transitions* 31, 1–32. <u>https://doi.org/10.1016/j.eist.2019.01.004</u>
- Kotz, M., Levermann, A., and Wenz, L. (2024). The economic commitment of climate change. *Nature* 628, 551–557. <u>https://doi.org/10.1038/s41586-024-07219-0</u>
- Krähmer, K. (2022). Degrowth and the city: Multiscalar strategies for the socio-ecological transformation of space and place. *City* 26, 316–345. <u>https://doi.org/10.1080/13604813.2022.2035969</u>
- Kratena, K., and Scharner, A. (2020). MIO-ES: A Macroeconomic Input-Output Model with Integrated Energy System. Vienna: Centre of Economic Scenario Analysis and Research (CE-SAR). Available at: <u>https://www.cesarecon.at/wp-content/uploads/2020/10/MIOES\_Manual\_Public\_FINAL.pdf</u> (Accessed October 23, 2024).
- Krisch, A., Novy, A., Plank, L., Schmidt, A. E., and Blaas, W. (2020). Die Leistungsträgerinnen des Alltagslebens. Covid-19 als Brennglas für die notwendige Neubewertung von Wirtschaft, Arbeit und Leistung. Wien. Available at: <u>http://hdl.handle.net/20.500.12708/40167</u> (Accessed January 17, 2025).
- Kristof, K. (2010). Wege zum Wandel: Wie wir gesellschaftliche Veränderungen erfolgreicher gestalten können. München: Oekom Verlag.
- Krueger, P., Sautner, Z., and Starks, L. T. (2020). The importance of climate risks for institutional investors. *The Review of Financial Studies* 33, 1067–1111. <u>https://doi.org/10.1093/rfs/hhz137</u>
- Krüger, T., and Pellicer-Sifres, V. (2020). From innovations to exnovations. Conflicts, (De-)Politicization processes, and power relations are key in analysing the ecological crisis. *Innovation: The European Journal of Social Science Research* 33, 115–123. https://doi.org/10.1080/13511610.2020.1733936
- Kubeczko, K., and Krisch, A. (2023). "Kapitel 22. Netzgebundene Infrastrukturen", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 591–609. Available at: <u>https://</u>

<u>doi.org/10.1007/978-3-662-66497-1\_26</u> (Accessed October 4, 2023).

- Kubeczko, K., Tödtling, F., Ornetzeder, M., Novy, A., Fankhauser, J., and Exner, A. (2023). "Kapitel 26. Theorien des Wandels und der Gestaltung von Strukturen: Innovationsperspektive", in *APCC Special Report: Strukturen für ein klimafreundliches Leben*, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 663–674. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_30</u> (Accessed October 4, 2023).
- Kuhn, T. S. (1967). Die Struktur wissenschaftlicher Revolutionen. Frankfurt am Main: Suhrkamp.
- Kulin, J., Johansson Sevä, I., and Dunlap, R. E. (2021). Nationalist ideology, rightwing populism, and public views about climate change in Europe. *Environmental Politics* 30, 1111–1134. <u>https://doi.org/10.1080/09644016.2021.1898879</u>
- Kuriakose, J., Jones, C., Anderson, K., McLachlan, C., and Broderick, J. (2022). What does the Paris climate change agreement mean for local policy? Downscaling the remaining global carbon budget to sub-national areas. *Renewable and Sustainable Energy Transition* 2, 100030. https://doi.org/10.1016/j.rset.2022.100030
- Lafortune, G., Fuller, L., Kloke-Lesch, A., Koundouri, P., and Riccaboni, A. (2024). European Elections, Europe's Future and the Sustainable Development Goals. Europe Sustainable Development Report 2023/24. SDSN and SDSN Europe. Paris: SDSN, Dublin: Dublin University Press. Available at: <u>http://www.tara.</u> <u>tcd.ie/handle/2262/104407</u> (Accessed May 13, 2024).
- Lage, J. (2022). Sufficiency and transformation–A semi-systematic literature review of notions of social change in different concepts of sufficiency. *Front. Sustain.* 3, 954660. <u>https://doi.org/10.3389/frsus.2022.954660</u>
- Lage, J., Thema, J., Zell-Ziegler, C., Best, B., Cordroch, L., and Wiese, F. (2023). Citizens call for sufficiency and regulation — A comparison of European citizen assemblies and National Energy and Climate Plans. *Energy Research & Social Science* 104, 103254. <u>https://doi.org/10.1016/j.erss.2023.103254</u>
- Lamb, W. F., Mattioli, G., Levi, S., Roberts, J. T., Capstick, S., Creutzig, F., et al. (2020). Discourses of climate delay. *Global Sustainability* 3, e17. <u>https://doi.org/10.1017/sus.2020.13</u>
- Lampl, C., Schmidt, A. E., and Aigner, E. (2024). Ernährungsarmut in Österreich als Barriere für eine gesunde und klimafreundliche Ernährung: Status quo und Handlungsoptionen. Wien: Gesundheit Österreich. Available at: <u>https://jasmin.goeg.at/id/</u> <u>eprint/3695/</u> (Accessed June 7, 2024).
- Landauer, M., Juhola, S., and Klein, J. (2019). The role of scale in integrating climate change adaptation and mitigation in cities. *Journal of Environmental Planning and Management* 62, 741– 765. <u>https://doi.org/10.1080/09640568.2018.1430022</u>
- Lehtonen, T., and Heikkurinen, P. (2022). Sufficiency and Sustainability: Conceptual Analysis and Ethical Considerations for Sustainable Organisation. *Environmental Values* 31, 599–618. https://doi.org/10.3197/096327121X16328186623878
- Leitner, M., Johnson, K., Carosi, A., Sini, E., Brusa, F., Lexer, W., et al. (2023). Is Europe on track with climate resilience? – Status of reported national adaptation actions in 2023. European Topic Centre on Climate change adaptation and LULUCF (ETC/CA) Technical Paper 2/23. <u>https://doi.org/10.25424/CMCC-5P8Y-VD45</u>
- Lent, J. (2017). The Patterning Instinct: A History of Humanity's Search for Meaning. Amherst, New York: Prometheus Books.

- Lent, J. (2021). The Web of Meaning: Integrating Science and Traditional Wisdom to Find Our Place in the Universe. London: Profile Books. Available at: <u>https://www.jeremylent.com/the-webof-meaning.html</u> (Accessed October 8, 2021).
- Lenton, T. M., Rockström, J., Gaffney, O., Rahmstorf, S., Richardson, K., Steffen, W., et al. (2019). Climate tipping points – too risky to bet against. *Nature* 575, 592–595. <u>https://doi.org/10.1038/d41586-019-03595-0</u>
- Lexer, W., and Buschmann, D. (2018). Case study reports on the mainstreaming of climate adaptation Country Report Austria (WP2). Vienna: Environment Agency Austria. Available at: <u>https://www.capa-eusalp.eu/geonetwork/srv/api/records/ ce4f95f8-b92e-4eb9-9bf9-a7ea9af403d0/attachments/goapply\_d.t2.1.1\_wp2\_case-study-reports\_austria\_uba-at\_dec-2018.pdf (Accessed October 22, 2024).</u>
- Lexer, W., Buschmann, D., and Strahlhofer, L. (2018). Climate Adaptation Governance in Austria. Country Report Austria (WP1). Vienna: Environment Agency Austria. Available at: <u>https://www.capa-eusalp.eu/geonetwork/srv/api/records/757801ac-84ce-45a4-a07c-8dd14fa95326/attachments/ GoApply\_WP1\_Country%20Report%20Austria\_Final\_UBA-AT\_25012019\_FINAL\_inclAnnexes\_revised%2020190307.pdf (Accessed October 22, 2024).</u>
- Lexer, W., and Novy, A. (2024). Regionale Transformationsagenturen: Innovative Mehrebenen-Governance für klimafreundliche Strukturen in österreichischen Regionen. Wien. Available at: <u>https://www.klimawandelanpassung.at/fileadmin/inhalte/ kwa/pdf/transreal/3\_policy-paper\_regionale\_transformationsagenturen\_29-02-2024\_v2.0\_public.pdf</u> (Accessed May 11, 2024).
- Lexer, W., Stickler, T., Buschmann, D., Steurer, R., and Feichtinger, J. (2020). KLIMAWANDELANPASSUNG IN KLEINEN ÖSTER-REICHISCHEN GEMEINDEN Agendasetzung und Verstetigung. Wien. Available at: <u>https://www.klimawandelanpassung.</u> <u>at/fileadmin/inhalte/kwa/pdf/praxis/goal/01\_goal\_synthesebericht\_umweltbundesamt.pdf</u> (Accessed May 13, 2024).
- Liu, L., and Jensen, M. B. (2018). Green infrastructure for sustainable urban water management: Practices of five forerunner cities. *Cities* 74, 126–133. <u>https://doi.org/10.1016/j.cities.2017.11.013</u>
- Lockwood, M. (2018). Right-wing populism and the climate change agenda: exploring the linkages. *Environmental Politics* 27, 712–732. <u>https://doi.org/10.1080/09644016.2018.1458411</u>
- Lorek, S., and Fuchs, D. (2013). Strong sustainable consumption governance – precondition for a degrowth path? *Journal of Cleaner Production* 38, 36–43. <u>https://doi.org/10.1016/j.jclepro.2011.08.008</u>
- Luderer, G., Madeddu, S., Merfort, L., Ueckerdt, F., Pehl, M., Pietzcker, R., et al. (2022). Impact of declining renewable energy costs on electrification in low-emission scenarios. *Nat Energy* 7, 32–42. <u>https://doi.org/10.1038/s41560-021-00937-z</u>
- Lütkenhorst, W., Altenburg, T., Pegels, A., and Vidican, G. (2014). Green Industrial Policy: Managing Transformation Under Uncertainty. Bonn: Deutsches Institut für Entwicklungspolitik Discussion Paper 28. Available at: <u>https://papers.srn.com/abstract=2509672</u> (Accessed October 10, 2024).
- Macy, J. (2007). World as Lover, World as Self: A Guide to Living Fully in Turbulent times: Courage for Global Justice and Planetary Renewal. Berkeley, Calif: Parallax Press.

- Madner, V. (2023a). Climate Change as a Challenge for Constitutional Courts: Fundamental Freedoms and Duties of Protection A Perspective from Austria. *Human Rights Law Journal* 43, 354–360.
- Madner, V. (2023b). "Nachhaltigkeit im Verwaltungsrecht", in *Nachhaltigkeit im Wirtschaftsrecht*, ed.Studiengesellschaft für Wirtschaft und Recht (Wien), 57–76. Available at: <u>https://permalink.obvsg.at/wuw/AC17018733</u> (Accessed May 14, 2024).
- Madner, V., and Grob, L.-M. (2019). Maßnahmen zur Stärkung von Orts- und Stadtkernen auf Bundesebene. Wien: Austrian Conference on Spatial Planning (ÖROK). Available at: <u>https://www. oerok.gv.at/raum/themen/staerkung-der-orts-und-stadtkerne</u> (Accessed May 15, 2024).
- Mandelli, M. (2022). Understanding eco-social policies: a proposed definition and typology. *Transfer: European Review of Labour and Research* 28, 333–348. <u>https://doi.org/10.1177/ 10242589221125083</u>
- Mang, S., and Caddick, D. (2023). New EU fiscal rules jeopardise investment needed to combat climate crisis. *New Economics Foundation*. Available at: <u>https://neweconomics.org/2023/08/</u> <u>new-eu-fiscal-rules-jeopardise-investment-needed-to-combatclimate-change</u> (Accessed May 16, 2024).
- Margotti, S. (2024). Klimapolitik in Österreich. Klimathemen im Kontext der Nationalratswahl. Baden: marketagent. Available at: <u>https://kontext-institut.at/uploads/Dateien/2024\_09-Umfrageregebnisse-KONTEXT.pdf</u> (Accessed October 10, 2024).
- Markard, J., and Truffer, B. (2008). Technological innovation systems and the multi-level perspective: Towards an integrated framework. *Research Policy* 37, 596–615. <u>https://doi.org/10.1016/j.respol.2008.01.004</u>
- Markkanen, S., and Anger-Kraavi, A. (2019). Social impacts of climate change mitigation policies and their implications for inequality. *Climate Policy* 19, 827–844. <u>https://doi.org/10.1080/</u> 14693062.2019.1596873
- Martin, J.-L., Maris, V., and Simberloff, D. S. (2016). The need to respect nature and its limits challenges society and conservation science. *Proceedings of the National Academy of Sciences* 113, 6105–6112. <u>https://doi.org/10.1073/pnas.1525003113</u>
- Mattioli, G., Roberts, C., Steinberger, J. K., and Brown, A. (2020). The political economy of car dependence: A systems of provision approach. *Energy Research & Social Science* 66, 101486. https://doi.org/10.1016/j.erss.2020.101486
- Mau, S., Lux, T., and Westheuser, L. (2023). *Triggerpunkte: Konsens* und Konflikt in der Gegenwartsgesellschaft., Originalausgabe, Sonderdruck. Berlin: Edition Suhrkamp.
- Mayer, J., Dugan, A., Bachner, G., and Steininger, K. W. (2021). Is carbon pricing regressive? Insights from a recursive-dynamic CGE analysis with heterogeneous households for Austria. *Energy Economics* 104, 105661. <u>https://doi.org/10.1016/j.eneco.2021.105661</u>
- Mazzucato, M., and Rodrik, D. (2023). Industrial Policy with Conditionalities: A Taxonomy and Sample Cases. UCL Institute for Innovation and Public Purpose. Working Paper Series (IIPP WP 2023–07). Available at: <u>https://drodrik.scholar.harvard.edu/ sites/scholar.harvard.edu/files/dani-rodrik/files/conditionality\_mazzucato\_rodrik\_0927202.pdf</u> (Accessed October 23, 2024).
- McCauley, D., and Heffron, R. (2018). Just transition: Integrating climate, energy and environmental justice. *Energy Policy* 119, 1–7. <u>https://doi.org/10.1016/j.enpol.2018.04.014</u>

- AAR2
- McCauley, D., Pettigrew, K. A., Todd, I., and Milchram, C. (2023). Leaders and laggards in the pursuit of an EU just transition. *Ecological Economics* 205, 107699. <u>https://doi.org/10.1016/j.</u> <u>ecolecon.2022.107699</u>
- McGreevy, S. R., Rupprecht, C. D. D., Niles, D., Wiek, A., Carolan, M., Kallis, G., et al. (2022). Sustainable agrifood systems for a post-growth world. *Nat Sustain* 5, 1011–1017. <u>https://doi. org/10.1038/s41893-022-00933-5</u>
- McIntyre, L. C. (2018). Post-truth. Cambridge, MA: MIT Press.
- McNamara, K. R. (2023). Transforming Europe? The EU's industrial policy and geopolitical turn. *Journal of European Public Policy* 31, 2371–2396. <u>https://doi.org/10.1080/13501763.2023.</u> <u>2230247</u>
- Meadows, D. (1999). Leverage Points: Places to Intervene in a System. Hartland, VT, USA: The Sustainability Institute. Available at: <u>https://www.donellameadows.org/wp-content/userfiles/</u> Leverage\_Points.pdf (Accessed January 17, 2025).
- Mercure, J.-F., Pollitt, H., Viñuales, J. E., Edwards, N. R., Holden, P. B., Chewpreecha, U., et al. (2018). Macroeconomic impact of stranded fossil fuel assets. *Nature Clim Change* 8, 588–593. <u>https://doi.org/10.1038/s41558-018-0182-1</u>
- Miess, M. (2023). "Kapitel 16: Geld und Finanzsystem (Money and financial system for climate-friendly living)", in Austrian Panel of Climate Change Special Report 22: structures for climate friendly living, (Berlin, Heidelberg: Springer Spektrum). Available at: <u>https://link.springer.com/chapter/10.1007/978-3-662-66497-1\_20</u> (Accessed September 19, 2022).
- Miess, M., Böhmer, S., Frei, E., Glas, N., Krutzler, T., Lichtblau, G., et al. (2022). Analysis of the investment potential until 2030 on the pathway to climate neutrality (Analyse des Investitionspotenzials bis 2030 auf dem Weg zur Klimaneutralität). Vienna. Available at: <u>https://www.wko.at/oe/bank-versicherung/folien-pk-investitionskosten-transformation.pdf</u> (Accessed May 16, 2024).
- Miess, M., and Ornetzeder, M. (2023). "Kapitel 16. Geld- und Finanzsystem", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg), 457–479.
  Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_20</u> (Accessed October 4, 2023).
- Mikovits, C., Tscheikner-Gratl, F., Jasper-Tönnies, A., Einfalt, T., Huttenlau, M., Schöpf, M., et al. (2017). Decision Support for Adaptation Planning of Urban Drainage Systems. J. Water Resour. Plann. Manage. 143, 04017069. <u>https://doi.org/10.1061/ (ASCE)WR.1943-5452.0000840</u>
- Millar, R. J., Fuglestvedt, J. S., Friedlingstein, P., Rogelj, J., Grubb, M. J., Matthews, H. D., et al. (2017). Emission budgets and pathways consistent with limiting warming to 1.5 °C. *Nature Geosci* 10, 741–747. <u>https://doi.org/10.1038/ngeo3031</u>
- Millward-Hopkins, J., and Oswald, Y. (2023). Reducing global inequality to secure human wellbeing and climate safety: a modelling study. *The Lancet Planetary Health* 7, e147–e154. <u>https:// doi.org/10.1016/S2542-5196(23)00004-9</u>
- Millward-Hopkins, J., Steinberger, J. K., Rao, N. D., and Oswald, Y. (2020). Providing decent living with minimum energy: A global scenario. *Global Environmental Change* 65, 102168. <u>https://doi.org/10.1016/j.gloenvcha.2020.102168</u>
- Mocca, E., Friesenecker, M., and Kazepov, Y. (2020). Greening Vienna. The Multi-Level Interplay of Urban Environmental Pol-

icy-Making. Sustainability 12, 1577. https://doi.org/10.3390/ su12041577

- Monasterolo, I., Dunz, N., Mazzocchetti, A., and Gourdel, R. (2022a). Derisking the low-carbon transition: investors' reaction to climate policies, decarbonization and distributive effects. *Rev Evol Polit Econ* 3, 31–71. <u>https://doi.org/10.1007/s43253-021-00062-3</u>
- Monasterolo, I., Nieto, M., and Schets, E. (2022b). The Good, the Bad and the Hot House World: Conceptual Underpinnings of the NGFS Scenarios and Suggestions for Improvement. <u>https:// doi.org/10.2139/ssrn.4211384</u>
- Monasterolo, I., and Volz, U. (2020). Policy Brief: Addressing climate-related financial risks and overcoming barriers to scaling up sustainable investments. Saudi Arabia: G20 Saudi Arabia: T20's Task Force 2 "Climate Change and Environment." Available at: <u>https://www.global-solutions-initiative.org/wp-content/ uploads/2022/11/T20\_TF2\_PB23.pdf</u> (Accessed February 6, 2025).
- Mouffe, C. (2005). *On the Political*. Abingdon, Oxon; New York, NY: Routledge.
- Moulaert, F., and MacCallum, D. (2019). *Advanced Introduction to Social Innovation*. Cheltenham: Edward Elgar Publishing.
- Müller, H. L., Kalhorn, A., and Getzner, M. (2024). "Postwachstum in der Raumplanung – Kernelemente und Spurensuche in Österreich", in *Biodiversitäts- und Bodenschutz in der Klima-*, *Energie- und Mobilitätswende. Jahrbuch Raumplanung.*, eds. J. Dangschat, H. Dumke, M. Getzner, A. Schneider, Y. Şişman, B. Steinbrunner, et al. (Vienna: TU Academic Press), 33–60.
- Nash, S. L., and Steurer, R. (2021). Climate Change Acts in Scotland, Austria, Denmark and Sweden: the role of discourse and deliberation. *Climate Policy* 21, 1120–1131. <u>https://doi.org/10. 1080/14693062.2021.1962235</u>
- Nature Editorial (2023). The global fight for critical minerals is costly and damaging. *Nature* 619, 436–436. <u>https://doi.org/10.1038/d41586-023-02330-0</u>
- Newell, P. (2019). Trasformismo or transformation? The global political economy of energy transitions. Review of International Political Economy 26, 25–48. <u>https://doi.org/10.1080/09692290.</u> 2018.1511448
- Newell, P., and Paterson, M. (2010). *Climate Capitalism: Global Warming and the Transformation of the Global Economy*. Cambridge: Cambridge University Press.
- Nohrstedt, D., and Parker, C. F. (2024). Revisiting the role of disasters in climate policy-making. *Climate Policy* 24, 428–439. <u>https://doi.org/10.1080/14693062.2024.2301781</u>
- Nordhaus, W. (2019). Climate Change: The Ultimate Challenge for Economics. American Economic Review 109, 1991–2014. <u>https://doi.org/10.1257/aer.109.6.1991</u>
- Novy, A. (2019). "Kritik der westlichen Lebensweise", in Chancen und Grenzen der Nachhaltigkeitstransformation, ed. F. Luks (Wiesbaden: Springer Gabler), 43–58. Available at: <u>http://link. springer.com/10.1007/978-3-658-22438-7\_3</u> (Accessed March 20, 2024).
- Novy, A. (2022). The political trilemma of contemporary social-ecological transformation – lessons from Karl Polanyi's The Great Transformation. *Globalizations* 19, 59–80. <u>https://doi.org/ 10.1080/14747731.2020.1850073</u>
- Novy, A., Arzberger, L., Dornis, N., Duman, U., Grabow, S., Lutz, P., et al. (2024a). Klimafreundliche Transformationspfade. Die Transformation multiperspektvisch planen. *International Karl Polanyi Society Polanyi Papers*. Available at: <u>https://www.karl-</u>

polanyisociety.com/wp-content/uploads/2024/12/Polanyi-Paper-006\_2024\_Novy-et-al\_Transformationspfade.pdf (Accessed December 20, 2024).

- Novy, A., and Barlow, N. (2022). Transformative Climate Actions. SRE-Discussion Papers 05/2022. WU Vienna University of Economics and Business. Available at: <u>https://ideas.repec.org/p/</u> wiw/wus009/34006514.html
- Novy, A., Barlow, N., and Fankhauser, J. (2022). "Transformative Innovation", in *Handbook of Critical Environmental Politics*, eds.
  L. Pellizoni, E. Leonardi, V. Asara, L. Pellizoni, E. Leonardi, and
  V. Asara (Cheltenham: Edward Elgar Publishing), 593–609.
- Novy, A., Bärnthaler, R., Karabeczek, V., Plank, L., and Schinko, T. (2023a). "Kapitel 27. Theorien des Wandels und der Gestaltung von Strukturen: Bereitstellungsperspektive", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 675–690. Available at: https://doi.org/10.1007/978-3-662-66497-1\_31 (Accessed October 4, 2023).
- Novy, A., Bärnthaler, R., and Prieler, M. (2023b). Zukunftsfähiges Wirtschaften: Herausforderungen der sozialökologischen Transformation. Weinheim: Beltz.
- Novy, A., Baumgartner, B., Grabow, S., Plank, L., and Volmary, H. (2024b). Greening Red Vienna: lessons for social-ecological housing provision. *Sustainability: Science, Practice and Policy* 20, 2312674. <u>https://doi.org/10.1080/15487733.2024.2312674</u>
- Novy, A., Brudermann, T., Fankhauser, J., Getzner, M., and Ohndorf, M. (2023c). "Kapitel 25. Theorien des Wandels und der Gestaltung von Strukturen: Marktperspektive", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 653–662. Available at: https://doi.org/10.1007/978-3-662-66497-1\_29 (Accessed October 4, 2023).
- Novy, A., Haderer, M., Kubeczko, K., Aigner, E., Bärnthaler, R., Brand, U., et al. (2023d). "Kapitel 2. Perspektiven zur Analyse und Gestaltung von Strukturen klimafreundlichen Lebens", in *APCC Special Report: Strukturen für ein klimafreundliches Leb*en, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 195–213. https://doi.org/10.1007/978-3-662-66497-1\_6
- Novy, A., Kubeczko, K., Haderer, M., Bärnthaler, R., Brand, U., Brudermann, T., et al. (2023e). "Kapitel 24. Theorien des Wandels und der Gestaltung von Strukturen", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 651–652. Available at: https://doi.org/10.1007/978-3-662-66497-1\_28 (Accessed September 16, 2023).
- Nykvist, B., and Maltais, A. (2022). Too risky The role of finance as a driver of sustainability transitions. *Environmental Innovation and Societal Transitions* 42, 219–231. <u>https://doi.org/10.1016/j.eist.2022.01.001</u>
- ÖAW (2023). ÖAW-Wissenschaftsbarometer: Aufwind für Vertrauen in die Wissenschaft. Available at: <u>https://www.oeaw.</u> <u>ac.at/news/oeaw-wissenschaftsbarometer-aufwind-fuer-vertrauen-in-die-wissenschaft</u> (Accessed April 25, 2024).
- OECD (2024). OECD Agenda for Transformative Science, Technology and Innovation Policies. OECD Science, Technology and Industry Policy Papers 164. <u>https://doi.org/10.1787/ba2aaf7b-en</u>

- O'Neill, D. W., Fanning, A. L., Lamb, W. F., and Steinberger, J. K. (2018). A good life for all within planetary boundaries. *Nature Sustainability* 1, 88–95. <u>https://doi.org/10.1038/s41893-018-0021-4</u>
- ÖROK (2024). ÖROK-Projekt "Regionale Innovation und Transformation (RIT)". Wien: Austrian Conference on Spatial Planning (ÖROK). Available at: <u>https://www.oerok.gv.at/fileadmin/ user\_upload/Bilder/3.Reiter-Regionalpolitik/Regionale\_Handlungseben\_2019/Abschlussveranstaltung/Themenaufriss\_ OEROK-Projekt\_RIT\_fuer\_A\_RW-WS\_am\_10.04.2024.pdf (Accessed May 11, 2024).</u>
- Oswald, Y., Millward-Hopkins, J., Steinberger, J. K., Owen, A., and Ivanova, D. (2023). Luxury-focused carbon taxation improves fairness of climate policy. *One Earth* 6, 884–898. <u>https://doi. org/10.1016/j.oneear.2023.05.027</u>
- Otto, F. (2023). Klimaungerechtigkeit: Was die Klimakatastrophe mit Kapitalismus, Rassismus und Sexismus zu tun hat. Berlin: Ullstein.
- Otto, I. M., Donges, J. F., Cremades, R., Bhowmik, A., Hewitt, R. J., Lucht, W., et al. (2020). Social tipping dynamics for stabilizing Earth's climate by 2050. *Proc. Natl. Acad. Sci. U.S.A.* 117, 2354–2365. <u>https://doi.org/10.1073/pnas.1900577117</u>
- Our World in Data (2023). Per capita CO<sub>2</sub> emissions. Available at: <u>https://ourworldindata.org/grapher/co-emissions-per-capita?tab=chart&country=USA+OWID\_WRL+CAN+AUS</u> (Accessed April 25, 2024).
- Overland, I., and Sovacool, B. K. (2020). The misallocation of climate research funding. *Energy Research & Social Science* 62, 101349. <u>https://doi.org/10.1016/j.erss.2019.101349</u>
- Pagnone, A., Gonçalves Gresse, E., López-Rivera, A., Wilkens, J., and Marotzke, J. (2023). "Chapter 3: Plausibility of attaining the Paris Agreement temperature goals", in *Hamburg Climate Futures Outlook: The plausibility of a 1.5°C limit to global warming – social drivers and physical processes*, (Hamburg, Germany: Cluster of Excellence Climate, Climatic Change, and Society (CLICC)), 26–29. https://doi.org/10.25592/uhhfdm.11230
- Pan, X., Teng, F., Tian, Y., and Wang, G. (2015). Countries' emission allowances towards the low-carbon world: A consistent study. *Applied Energy* 155, 218–228. <u>https://doi.org/10.1016/j.apenergy.2015.06.011</u>
- Pascual, U., Adams, W. M., Díaz, S., Lele, S., Mace, G. M., and Turnhout, E. (2021). Biodiversity and the challenge of pluralism. *Nat Sustain* 4, 567–572. <u>https://doi.org/10.1038/s41893-021-00694-7</u>
- Pascual, U., Balvanera, P., Anderson, C. B., Chaplin-Kramer, R., Christie, M., González-Jiménez, D., et al. (2023). Diverse values of nature for sustainability. *Nature* 620, 813–823. <u>https://doi.org/10.1038/s41586-023-06406-9</u>
- Pathak, M., Slade, R., Shukla, P. R., Skea, J., Pichs-Madruga, R., and Ürge-Vorsatz, D. (2022). "Technical Summary", in *Climate Change 2022 – Mitigation of Climate Change: Working Group III Contribution to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*, eds. P. R. Shukla, J. Skea, R. Slade, R. Al Khourdajie, R. van Diemen, D. McCollum, et al. (Cambridge, UK and New York, NY, USA: Cambridge University Press), 51–148. https://doi.org/10.1017/9781009157926.002
- Pearson, M. (2024). Back to the Stone Age: Europe's Mainstream Right and Climate Change. *The Political Quarterly* 95, 123–130. <u>https://doi.org/10.1111/1467-923X.13316</u>
- Pekanov, A., and Schratzenstaller-Altzinger, M. (2023). "8. Making Green Public Investments a Reality in the EU Fiscal Framework

and the EU Budget", in *Financing Investment in Times of High Public Debt*, eds. F. Cerniglia, F. Saraceno, A. Watt, F. Cerniglia, F. Saraceno, and A. Watt (Cambridge, UK: Open Book Publishers), 137–156. Available at: <u>https://www.openbookpublishers.com/books/10.11647/obp.0386/chapters/10.11647/obp.0386.08</u> (Accessed June 8, 2024).

- Penker, M., Brunner, K.-M., and Plank, C. (2023). "Kapitel 5. Ernährung", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 245–269. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_9</u> (Accessed October 4, 2023).
- Perga, M.-E., Sarrasin, O., Steinberger, J., Lane, S. N., and Butera, F. (2023). The climate change research that makes the front page: Is it fit to engage societal action? *Global Environmental Change* 80, 102675. <u>https://doi.org/10.1016/j.gloenvcha.2023.102675</u>
- Peters, G. P., Davis, S. J., and Andrew, R. (2012). A synthesis of carbon in international trade. *Biogeosciences* 9, 3247–3276. <u>https:// doi.org/10.5194/bg-9-3247-2012</u>
- Pichler, M., Krenmayr, N., Schneider, E., and Brand, U. (2021). EU industrial policy: Between modernization and transformation of the automotive industry. *Environmental Innovation* and Societal Transitions 38, 140–152. <u>https://doi.org/10.1016/j.eist.2020.12.002</u>
- Pielke, Jr., R., and Ritchie, J. (2021). Distorting the view of our climate future: The misuse and abuse of climate pathways and scenarios. *Energy Res. Soc. Sci.* 72. <u>https://doi.org/10.1016/j.</u> <u>erss.2020.101890</u>
- Pirgmaier, E. (2020). Consumption corridors, capitalism and social change. Sustainability: Science, Practice and Policy 16, 274–285. <u>https://doi.org/10.1080/15487733.2020.1829846</u>
- Plank, C., Haas, W., Schreuer, A., Irshaid, J., Barben, D., and Görg, C. (2021). Climate policy integration viewed through the stakeholders' eyes: A co-production of knowledge in social-ecological transformation research. *Environmental Policy and Governance* 31, 387–399. <u>https://doi.org/10.1002/eet.1938</u>
- Polanyi, K. (1977). *The Livelihood of Man*. New York: Academic Press.
- Pollex, J., and Lenschow, A. (2018). Surrendering to growth? The European Union's goals for research and technology in the Horizon 2020 framework. *Journal of Cleaner Production* 197, 1863–1871. <u>https://doi.org/10.1016/j.jclepro.2016.10.195</u>
- Polzin, F. (2017). Mobilizing private finance for low-carbon innovation – A systematic review of barriers and solutions. *Renewable and Sustainable Energy Reviews* 77, 525–535. <u>https://doi. org/10.1016/j.rser.2017.04.007</u>
- Porak, L. (2023). Wettbewerbsfähige Nachhaltigkeit: eine Historisch-Materialistische Analyse der Ideen, Institutionen und Machtverhältnisse in der europäischen grünen Transformation. Momentum Quarterly – Zeitschrift für sozialen Fortschritt 12, 65–83. <u>https://doi.org/10.15203/momentumquarterly.vol12.</u> no1.p65-83
- Porak, L., and Reinke, R. (2024). The contribution of qualitative methods to economic research in an era of polycrisis. *Rev Evol Polit Econ* 5, 31–49. <u>https://doi.org/10.1007/s43253-024-00116-</u> 2
- Pörtner, H.-O., Roberts, D. C., Adams, H., Adelekan, I., Adler, C., Adrian, R., et al. (2022). Technical Summary. Climate Change 2022: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Available at: https://www.

ipcc.ch/report/ar6/wg2/downloads/report/IPCC\_AR6\_WGII\_ TechnicalSummary.pdf

- Princen, T. (2003). Principles for Sustainability: From Cooperation and Efficiency to Sufficiency. *Global Environmental Politics* 3, 33–50. <u>https://doi.org/10.1162/152638003763336374</u>
- Probst, T., Hohmann, R., Pütz, M., Braunschweiger, D., and Kuhn-Belaid, R. (2019). Climate Adaptation Governance in the Alpine Space. Transnational Synthesis Report (WP1). Zürich: European Regional Development Fund. Available at: <u>https:// www.alpine-space.eu/wp-content/uploads/2021/10/37-1-goapply-Climate-Adaptation-Governance-in-the-Alpine-Space-Transnational-Synthesis-Report-output.pdf</u> (Accessed September 9, 2023).
- Prutsch, A., Steurer, R., and Stickler, T. (2018). Is the participatory formulation of policy strategies worth the effort? The case of climate change adaptation in Austria. *Reg Environ Change* 18, 271–285. <u>https://doi.org/10.1007/s10113-017-1204-7</u>
- Rajaeifar, M. A., Belcher, O., Parkinson, S., Neimark, B., Weir, D., Ashworth, K., et al. (2022). Decarbonize the military – mandate emissions reporting. *Nature* 611, 29–32. <u>https://doi. org/10.1038/d41586-022-03444-7</u>
- Rajamani, L. (2016). The 2015 Paris Agreement: Interplay Between Hard, Soft and Non-Obligations. *Journal of Environmental Law* 28, 337–358. <u>https://doi.org/10.1093/jel/eqw015</u>
- Rajamani, L., and Bodansky, D. (2019). THE PARIS RULEBOOK: BALANCING INTERNATIONAL PRESCRIPTIVENESS WITH NATIONAL DISCRETION. International and Comparative Law Quarterly 68, 1023–1040. <u>https://doi.org/10.1017/</u> <u>S0020589319000320</u>
- Ranger, N. A., Mahul, O., and Monasterolo, I. (2022). Assessing Financial Risks from Physical Climate Shocks: A Framework for Scenario Generation. World Bank. https://doi.org/10.1596/37041
- Rauken, T., Mydske, P. K., and Winsvold, M. (2015). Mainstreaming climate change adaptation at the local level. *Local Environment* 20, 408–423. <u>https://doi.org/10.1080/13549839.2014.8804</u> 12
- Raworth, K. (2017). *Doughnut Economics: Seven Ways to Think Like a 21st-Century Economist.* White River Junction, Vermont: Chelsea Green Publishing.
- Reckien, D., Salvia, M., Heidrich, O., Church, J. M., Pietrapertosa, F., De Gregorio-Hurtado, S., et al. (2018). How are cities planning to respond to climate change? Assessment of local climate plans from 885 cities in the EU-28. *Journal of Cleaner Production* 191, 207–219. https://doi.org/10.1016/j.jclepro.2018.03.220
- Rees, W. (2010). What's blocking sustainability? Human nature, cognition, and denial. Sustainability: Science, Practice and Policy 6, 13–25. <u>https://doi.org/10.1080/15487733.2010.11908046</u>
- Regulation 2018/1999 (2018). Regulation 2018/1999 OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL of 11 December 2018 on the Governance of the Energy Union and Climate Action, amending Regulations (EC) No 663/2009 and (EC) No 715/2009 of the European Parliament and of the Council, Directives 94/22/EC, 98/70/EC, 2009/31/EC, 2009/73/EC, 2010/31/EU, 2012/27/EU and 2013/30/EU of the European Parliament and of the Council, Council Directives 2009/119/EC and (EU) 2015/652 and repealing Regulation (EU) No 525/2013 of the European Parliament and of the Council. *Official Journal of the European Union* 61. Available at: https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=OJ:L:2018:328:FULL& from=EN

- Regulation (EU) 2023/857 (2023). Regulation (EU) 2023/857 of the European Parliament and of the Council of 19 April 2023 amending Regulation (EU) 2018/842 on binding annual greenhouse gas emission reductions by Member States from 2021 to 2030 contributing to climate action to meet commitments under the Paris Agreement, and Regulation (EU) 2018/1999 (Text with EEA relevance). Available at: <u>http://data.europa.eu/eli/reg/</u> 2023/857/oj/eng (Accessed May 15, 2024).
- Reiner, C., and Edlinger, H. (2023). Globalisierung am Limit. Themenblätter im Unterricht 133. Available at: <u>https://www.bpb.de/ shop/materialien/themenblaetter/517826/globalisierung-amlimit/</u> (Accessed April 18, 2024).
- Reiter, K., Knittel, N., Bachner, G., and Hochrainer-Stigler, S. (2022). Barriers and ways forward to climate risk management against indirect effects of natural disasters: A case study on flood risk in Austria. *Climate Risk Management* 36, 100431. <u>https://doi.org/10.1016/j.crm.2022.100431</u>
- Resch, G., Schöniger, F., Kleinschmitt, C., Franke, K., Thonig, R., and Lilliestam, J. (2022). Deep decarbonization of the European power sector calls for dispatchable CSP. *AIP Conference Proceedings* 2445, 050006. <u>https://doi.org/10.1063/5.0086710</u>
- Riahi, K., Schaeffer, R., Arango, J., Calvin, K., Guivarch, C., Hasegawa, T., et al. (2022). Mitigation pathways compatible with long-term goals. *IPCC*, 2022: Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. <u>https://doi.org/10.1017/9781009157926.005</u>
- Ripple, W. J., Wolf, C., Gregg, J. W., Rockström, J., Newsome, T. M., Law, B. E., et al. (2023). The 2023 state of the climate report: Entering uncharted territory. *BioScience*, 1–10. <u>https://doi.org/10.1093/biosci/biad080</u>
- Robiou du Pont, Y., Jeffery, M. L., Gütschow, J., Rogelj, J., Christoff, P., and Meinshausen, M. (2017). Equitable mitigation to achieve the Paris Agreement goals. *Nature Clim Change* 7, 38–43. <u>https://doi.org/10.1038/nclimate3186</u>
- Robiou du Pont, Y., and Meinshausen, M. (2018). Warming assessment of the bottom-up Paris Agreement emissions pledges. *Nat Commun* 9, 4810. https://doi.org/10.1038/s41467-018-07223-9
- Rockström, J., Gupta, J., Qin, D., Lade, S. J., Abrams, J. F., Andersen, L. S., et al. (2023). Safe and just Earth system boundaries. *Nature* 619, 102–111. <u>https://doi.org/10.1038/s41586-023-06083-8</u>
- Rockström, J., Steffen, W., Noone, K., Persson, Å., Chapin, F. S., Lambin, E., et al. (2009). Planetary Boundaries: Exploring the Safe Operating Space for Humanity. *Ecology and Society* 14. Available at: <u>https://www.jstor.org/stable/26268316</u> (Accessed April 29, 2024).
- Rodrik, D. (2011). The Globalization Paradox: Democracy and the Future of the World Economy. New York, London: W.W. Norton.
- Rodrik, D. (2017). Straight Talk on Trade: Ideas for a Sane World Economy. Princeton: Princeton University Press.
- Rogelj, J., Schaeffer, M., Friedlingstein, P., Gillett, N. P., van Vuuren, D. P., Riahi, K., et al. (2016). Differences between carbon budget estimates unravelled. *Nature Clim Change* 6, 245–252. <u>https:// doi.org/10.1038/nclimate2868</u>
- Romanello, M., Whitmee, S., Mulcahy, E., and Costello, A. (2023). Further delays in tackling greenhouse gas emissions at COP28 will be an act of negligence. *The Lancet* 402, 2055–2057. <u>https:// doi.org/10.1016/S0140-6736(23)02584-9</u>
- Roncoroni, A., Battiston, S., Escobar-Farfán, L. O. L., and Martinez-Jaramillo, S. (2021). Climate risk and financial stability in

the network of banks and investment funds. *Journal of Financial Stability* 54, 100870. <u>https://doi.org/10.1016/j.jfs.2021.100870</u>

- Roy, B., and Schaffartzik, A. (2021). Talk renewables, walk coal: The paradox of India's energy transition. *Ecological Economics* 180, 106871. <u>https://doi.org/10.1016/j.ecolecon.2020.106871</u>
- Roy, J., Tschakert, P., Waisman, H., Abdul Halim, S., Antwi-Agyei, P., Dasgupta, P., et al. (2018). "Chapter 5: Sustainable Development, Poverty Eradication and Reducing Inequalities", in *Global Warming of 1.5°C. An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty [Masson-Delmotte, V., P. Zhai, H.-O. Pörtner, D. Roberts, J. Skea, P.R. Shukla, A. Pirani, W. Moufouma-Okia, C. Péan, R. Pidcock, S. Connors, J.B.R. Matthews, Y. Chen, X. Zhou, M.I. Gomis, E. Lonnoy, T. Maycock, M. Tignor, and T. Waterfield (eds.)], (Cambridge, UK and New York, NY, USA), 445–538. Available at: https://www.ipcc.ch/site/assets/uploads/sites/2/2019/02/SR15\_Chapter5\_Low\_Res. pdf (Accessed March 22, 2024).*
- Runhaar, H., Wilk, B., Persson, Å., Uittenbroek, C., and Wamsler, C. (2018). Mainstreaming climate adaptation: taking stock about "what works" from empirical research worldwide. *Reg Environ Change* 18, 1201–1210. <u>https://doi.org/10.1007/s10113-</u> 017-1259-5
- Sachs, J. D., Lafortune, G., Fuller, G., and Drumm, E. (2023). Implementing the SDG Stimulus. Sustainable Development Report 2023: Sustainable Development Report 2023. Dublin: Dublin University Press. Available at: <u>http://www.tara.tcd.ie/handle/2262/102924</u> (Accessed May 13, 2024).
- Sachverständigenrat für Umweltfragen (2024a). Suffizienz als "Strategie des Genug": Eine Einladung zur Diskussion. Available at: <u>https://www.umweltrat.de/SharedDocs/Downloads/</u> <u>DE/04\_Stellungnahmen/2020\_2024/2024\_03\_Suffizienz.html</u> (Accessed May 16, 2024).
- Sachverständigenrat für Umweltfragen ed. (2024b). Suffizienz als "Strategie des Genug": Eine Einladung zur Diskussion. Berlin: Sachverständigenrat für Umweltfragen SRU. Available at: <u>https://www.umweltrat.de/SharedDocs/Downloads/DE/04\_</u> <u>Stellungnahmen/2020\_2024/2024\_03\_Suffizienz.html</u> (Accessed May 16, 2024).
- Sælen, H. G., and Aasen, M. (2023). Exploring public opposition and support across different climate policies: Poles apart? *Ecological Economics* 209, 107838. <u>https://doi.org/10.1016/j.ecolecon.2023.107838</u>
- Sahakian, M., Fuchs, D., Lorek, S., and Di Giulio, A. (2021). Advancing the concept of consumption corridors and exploring its implications. *Sustainability: Science, Practice and Policy* 17, 305–315. https://doi.org/10.1080/15487733.2021.1919437
- Salon, D., Sperling, D., Meier, A., Murphy, S., Gorham, R., and Barrett, J. (2010). City carbon budgets: A proposal to align incentives for climate-friendly communities. *Energy Policy* 38, 2032–2041. https://doi.org/10.1016/j.enpol.2009.12.005
- Schaffartzik, A., Pichler, M., Pineault, E., Wiedenhofer, D., Gross, R., and Haberl, H. (2021). The transformation of provisioning systems from an integrated perspective of social metabolism and political economy: a conceptual framework. *Sustain Sci* 16, 1405–1421. https://doi.org/10.1007/s11625-021-00952-9
- Schinko, T., and Bednar-Friedl, B. (2022). Fostering social learning through role-play simulations to operationalize comprehensive climate risk management: Insights from applying the RESPECT

AAR2

role-play in Austria. *Climate Risk Management* 35, 100418. https://doi.org/10.1016/j.crm.2022.100418

- Schinko, T., Karabaczek, V., Kienberger, S., and Menk, L. (2022). Grenzen der Anpassung in Österreich? TransLoss Policy Brief. Laxenburg & Salzburg: Paris-Lodron-Universität Salzburg. Available at: <u>https://pure.iiasa.ac.at/id/eprint/18008/</u> (Accessed October 21, 2024).
- Schinko, T., Mechler, R., and Hochrainer-Stigler, S. (2017). A methodological framework to operationalize climate risk management: managing sovereign climate-related extreme event risk in Austria. *Mitig Adapt Strateg Glob Change* 22, 1063–1086. https://doi.org/10.1007/s11027-016-9713-0
- Schirpke, U., Tasser, E., Borsky, S., Braun, M., Eitzinger, J., Gaube, V., et al. (2023). Past and future impacts of land-use changes on ecosystem services in Austria. *Journal of Environmental Man*agement 345, 118728. <u>https://doi.org/10.1016/j.jenvman.2023.</u> <u>118728</u>
- Schleer, C., Wisniewski, N., and Reusswig, F. A. (2024). Sozialökologische Transformation gestalten: Wie gesellschaftliche Barrieren überwunden und Resonanzpotenziale genutzt werden können. Heidelberg, Potsdam. Available at: <u>https://library.fes.de/ pdf-files/bueros/bruessel/21075.pdf</u> (Accessed April 22, 2024).
- Schmelzer, M., and Hofferberth, E. (2023). Democratic Planning for Degrowth. *Monthly Review* 75. Available at: <u>https://monthlyreview.org/2023/07/01/democratic-planning-for-degrowth/</u> (Accessed May 14, 2024).
- Schmidt, J., Mitter, H., Baumann, M., Boza-Kiss, B., Huppmann, D., Wehrle, S., et al. (2025). Need for speed: Co-creating scenarios for climate neutral energy systems in Austria in 2040. *Energy Policy* 198, 114493. <u>https://doi.org/10.1016/j.enpol.2024.114493</u>
- Schneider, U. (2023). "Kapitel 18. Sozialstaat und Klimawandel", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 499– 528. Available at: <u>https://doi.org/10.1007/978-3-662-66497-</u> 1\_22 (Accessed October 4, 2023).
- Schneidewind, U., and Zahrnt, A. (2013). Damit gutes Leben einfacher wird. Perspektiven einer Suffizienzpolitik. München: Oekom Verlag.
- Scholes, R., Brainich, A., ten Brink, B., Cantele, M., Erasmus, B., Fisher, J., et al. (2018). Summary for policymakers of the assessment report on land degradation and restoration of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services., ed. IPBES. Bonn, Germany: IPBES secretariat.
- Schuberth, H., and Soder, M. (2024). Wirtschaftspolitik und Governance eines gerechten Übergangs zur Klimaneutralität. VAW 1, 75–88. <u>https://doi.org/10.3790/vaw.2024.1439008</u>
- Schulze Waltrup, R. (2023). An eco-social policy typology: From system reproduction to transformation. *Global Social Policy*, 14680181231205777. <u>https://doi.org/10.1177/14680181231205</u> 777
- Schumacher, K. (2020). Green investments need global standards and independent scientific review. *Nature* 584, 524–525. <u>https:// doi.org/10.1038/d41586-020-02472-5</u>
- Seebauer, S. (2021). How to make building renovation work for low-income renters: Preferences for distributional principles and procedural options in Austria. *Energy Research & Social Science* 82, 102270. <u>https://doi.org/10.1016/j.erss.2021.102270</u>
- Segreto, M., Principe, L., Desormeaux, A., Torre, M., Tomassetti, L., Tratzi, P., et al. (2020). Trends in Social Acceptance of Re-

newable Energy Across Europe—A Literature Review. *IJERPH* 17, 9161. <u>https://doi.org/10.3390/ijerph17249161</u>

- Seifert, M., Gangl, K., Spitzer, F., Haeckl, S., Gaudeul, A., Kirchler, E., et al. (2022). Financial Return and Environmental Impact Information Promotes ESG Investments: Evidence from a Large, Incentivized Online Experiment. SSRN. <u>https://doi.org/10. 2139/ssrn.4294495</u>
- Selby, J., and Hoffmann, C. (2014). Rethinking Climate Change, Conflict and Security. *Geopolitics* 19, 747–756. <u>https://doi.org/ 10.1080/14650045.2014.964866</u>
- Semieniuk, G., Holden, P. B., Mercure, J.-F., Salas, P., Pollitt, H., Jobson, K., et al. (2022). Stranded fossil-fuel assets translate to major losses for investors in advanced economies. *Nat. Clim. Chang.* 12, 532–538. <u>https://doi.org/10.1038/s41558-022-01356-y</u>
- Seto, K. C., Davis, S. J., Mitchell, R. B., Stokes, E. C., Unruh, G., and Ürge-Vorsatz, D. (2016). Carbon Lock-In: Types, Causes, and Policy Implications. *Annu. Rev. Environ. Resour.* 41, 425–452. <u>https://doi.org/10.1146/annurev-environ-110615-085934</u>
- Shove, E., Trentmann, F., and Wilk, R. R. eds. (2009). *Time, con-sumption and everyday life: practice, materiality and culture*. Oxford; New York: Routledge.
- Shue, H. (1999). Global Environment and International Inequality. International Affairs (Royal Institute of International Affairs 1944-) 75, 531–545.
- Skjærseth, J. B., Andresen, S., Bang, G., and Heggelund, G. M. (2021). The Paris agreement and key actors' domestic climate policy mixes: comparative patterns. *Int Environ Agreements* 21, 59–73. <u>https://doi.org/10.1007/s10784-021-09531-w</u>
- Smetschka, B., Mader, K., Simsa, R., and Wiedenhofer, D. (2023). "Kapitel 8. Sorgearbeit für die eigene Person, Haushalt, Familie und Gesellschaft", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 309–327. <u>https://doi.org/10.1007/978-3-662-66497-1\_12</u>
- Smith, P., Arneth, A., Barnes, D. K. A., Ichii, K., Marquet, P. A., Popp, A., et al. (2022). How do we best synergize climate mitigation actions to co-benefit biodiversity? *Global Change Biology* 28, 2555–2577. https://doi.org/10.1111/gcb.16056
- Sniatynski, G. (1980). Hazardous Wastes. Environment Views 2, 5.
- Soder, M. (2024). Eine grüne Revolution: Eine neue Wirtschaftspolitik in Zeiten der Klimakrise. Wien: ÖGB Verlag.
- Sorg, C. (2023). Failing to Plan Is Planning to Fail: Toward an Expanded Notion of Democratically Planned Postcapitalism. *Critical Sociology* 49, 475–493. <u>https://doi.org/10.1177/08969</u> 205221081058
- Spash, C. L. (2024). Foundations of social ecological economics: The fight for revolutionary change in economic thought. Manchester: Manchester University Press.
- Spash, C. L., and Hache, F. (2022). The Dasgupta Review deconstructed: an exposé of biodiversity economics. *Globalizations* 19, 653–676. <u>https://doi.org/10.1080/14747731.2021.1929007</u>
- Stagl, S., Eckert, L., and Schemel, B. (2024). "Umsetzung und gesellschaftliche Akzeptanz", in Nationaler Energie- und Klimaplan (NEKP) für Österreich Wissenschaftliche Bewertung der in der Konsultation 2023 vorgeschlagenen Maßnahmen, eds. K. W. Steininger, K. Riahi, S. Stagl, H. Kromp-Kolb, G. Kirchengast, K. W. Steininger, et al. (Wien: Climate Change Centre Austria (CCCA)). Available at: https://ccca.ac.at/fileadmin/00\_DokumenteHauptmenue/02\_Klimawissen/RefNEKP/Bericht/

NEKP\_Wissenschaftliche\_Bewertung\_der\_Massnahmen\_der\_ Stellungnahmen\_Februar2024.pdf (Accessed April 24, 2024).

- Statistik Austria (2023). Zeit und Hauptaggregate der VGR. Available at: <u>https://statcube.at/statistik.at/ext/statcube/jsf/table-View/tableView.xhtml</u> (Accessed December 13, 2023).
- Steffen, W., Richardson, K., Rockström, J., Cornell, S. E., Fetzer, I., Bennett, E. M., et al. (2015). Planetary boundaries: Guiding human development on a changing planet. *Science* 347, 1259855. <u>https://doi.org/10.1126/science.1259855</u>
- Steffen, W., Rockström, J., Richardson, K., Lenton, T. M., Folke, C., Liverman, D., et al. (2018). Trajectories of the Earth System in the Anthropocene. *Proc. Natl. Acad. Sci. U.S.A.* 115, 8252–8259. <u>https://doi.org/10.1073/pnas.1810141115</u>
- Steinberger, J. K., Krausmann, F., Getzner, M., Schandl, H., and West, J. (2013). Development and Dematerialization: An International Study. *PLoS ONE* 8, e70385. <u>https://doi.org/10.1371/journal.pone.0070385</u>
- Steininger, K., Alaux, N., Glas, N., Hoff, H., Kueschnig, M., Kulmer, V., et al. (2024). Austria's Climate Neutrality: A Consistent Cross-Sector Analysis and Policy Package, Distributional and Financial Implications (Final Report of the project INTE-GRATE). ACRP, Klimafonds.
- Steininger, K., Bednar-Friedl, B., Knittel, N., Kirchengast, G., Nabernegg, S., Williges, K., et al. (2020a). Klimapolitik in Österreich: Innovationschance Coronakrise und die Kosten des Nicht-Handelns. Graz: Wegener Center Verlag. <u>https://doi.org/ 10.25364/23.2020.1</u>
- Steininger, K. W., Bednar-Friedl, B., Formayer, H., and König, M. (2016). Consistent economic cross-sectoral climate change impact scenario analysis: Method and application to Austria. *Climate Services* 1, 39–52. <u>https://doi.org/10.1016/j.cliser.2016.</u> 02.003
- Steininger, K. W., König, M., Bednar-Friedl, B., Kranzl, L., Loibl, W., and Prettenthaler, F. (2015). Economic Evaluation of Climate Change Impacts: Development of a Cross-Sectoral Framework and Results for Austria. Cham: Springer. Available at: <u>https:// link.springer.com/book/10.1007/978-3-319-12457-5</u> (Accessed November 8, 2024).
- Steininger, K. W., Meyer, L., Nabernegg, S., and Kirchengast, G. (2020b). Sectoral carbon budgets as an evaluation framework for the built environment. *Buildings and Cities* 1, 337–360. https://doi.org/10.5334/bc.32
- Steininger, K. W., Munoz, P., Karstensen, J., Peters, G. P., Strohmaier, R., and Velázquez, E. (2018). Austria's consumption-based greenhouse gas emissions: Identifying sectoral sources and destinations. *Global Environmental Change* 48, 226–242. <u>https:// doi.org/10.1016/j.gloenvcha.2017.11.011</u>
- Steininger, K. W., Williges, K., Meyer, L. H., Maczek, F., and Riahi, K. (2022). Sharing the effort of the European Green Deal among countries. *Nat Commun* 13, 3673. <u>https://doi.org/10.1038/s414</u> 67-022-31204-8
- Stern, N. H. (2007). The Economics of Climate Change: The Stern Review. Cambridge: Cambridge University Press.
- Steurer, R., Buzogány, A., Clar, C., Nash, S. L., and Scherhaufer, P. (2023). "Kapitel 12. Governance und politische Beteiligung", in *APCC Special Report: Strukturen für ein klimafreundliches Leben*, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 389–402. <u>https://doi.org/10.1007/978-3-662-66497-1\_16</u>
- Steurer, R., and Clar, C. (2018). The ambiguity of federalism in climate policy-making: how the political system in Austria hin-

ders mitigation and facilitates adaptation. *Journal of Environmental Policy & Planning* 20, 252–265. <u>https://doi.org/10.1080/</u> 1523908X.2017.1411253

- Stoddard, I., Anderson, K., Capstick, S., Carton, W., Depledge, J., Facer, K., et al. (2021). Three Decades of Climate Mitigation: Why Haven't We Bent the Global Emissions Curve? *Annu. Rev. Environ. Resour.* 46, 653–689. <u>https://doi.org/10.1146/annurevenviron-012220-011104</u>
- Stöhr, W. B., and Taylor, F. D. R. (1981). Development from above or below? The Dialectics of Regional Planning in Developing Countries. Chincester.
- Stratford, B. (2020). The Threat of Rent Extraction in a Resource-constrained Future. *Ecological Economics* 169, 106524. <u>https://doi.org/10.1016/j.ecolecon.2019.106524</u>
- Streck, C., and Terhalle, M. (2013). The changing geopolitics of climate change. *Climate Policy* 13, 533–537. <u>https://doi.org/10. 1080/14693062.2013.823809</u>
- Streeck, W. (2014). Buying time: The delayed crisis of democratic capitalism. London; New York: Verso Books.
- Streeck, W., Calhoun, C., Toynbee, P., and Etzioni, A. (2016). Does capitalism have a future? *Socio-Economic Review* 14, 163–183. <u>https://doi.org/10.1093/ser/mwv037</u>
- Sullivan, S. (2018). Making Nature Investable: from Legibility to Leverageability in Fabricating 'Nature' as 'Natural-Capital'. Science & Technology Studies 31, 47–76. <u>https://doi.org/10.23987/ sts.58040</u>
- Svanda, N., and Zech, S. (2023). "Kapitel 19. Raumplanung", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 529– 546. Available at: <u>https://doi.org/10.1007/978-3-662-66497-</u> 1\_23 (Accessed May 9, 2024).
- Symons, J. (2019). Ecomodernism: Technology, Politics and The Climate Crisis. New Jersey: John Wiley & Sons.
- Ter Mors, E., Terwel, B. W., and Daamen, D. D. L. (2012). The potential of host community compensation in facility siting. *International Journal of Greenhouse Gas Control* 11, S130–S138. <u>https://doi.org/10.1016/j.ijggc.2012.07.002</u>
- Tews, K. (2005). The diffusion of environmental policy innovations: cornerstones of an analytical framework. *European Envi*ronment 15, 63–79. <u>https://doi.org/10.1002/eet.378</u>
- Thaller, A., Schreuer, A., and Posch, A. (2021). Flying High in Academia—Willingness of University Staff to Perform Low-Carbon Behavior Change in Business Travel. *Frontiers in Sustainability* 2. https://doi.org/10.3389/frsus.2021.790807
- Theine, H., Humer, S., Moser, M., and Schnetzer, M. (2022). Emissions inequality: Disparities in income, expenditure, and the carbon footprint in Austria. *Ecological Economics* 197, 107435. <u>https://doi.org/10.1016/j.ecolecon.2022.107435</u>
- Theine, H., Regen, L., Perez Delgado, V. D., Reitan, C., and Novy, A. (2023). "Kapitel 20: Mediendiskurse und -strukturen", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. A. Posch, K. W. Steininger, E. Aigner, A. Posch, K. W. Steininger, and E. Aigner (Berlin, Heidelberg: Springer Spektrum), 413–436. Available at: <u>https://doi.org/10.1007/978-3-662-66497-1\_18</u>
- Thompson, M., Ellis, R. J., Wildavsky, A., and Wildavsky, M. (1990). *Cultural Theory*. Boulder, Colorado: Westview Press.
- Tian, N., Lopes Da Silva, D., Liang, X., and Scarazzato, L. (2024). Trends in World Military Expenditure, 2023. Stockholm. Available
  - at: https://www.sipri.org/publications/2024/sipri-fact-sheets/

trends-world-military-expenditure-2023 (Accessed April 25, 2024).

- Tienhaara, K., and Robinson, J. (2022). *Routledge Handbook on the Green New Deal*. Oxford; New York: Routledge.
- Tipping Point North South (2023). Placing the military in the degrowth narrative. Degrowth 01. <u>https://doi.org/10.36399/Degrowth.001.01.06</u>
- Tödtling, F., Trippl, M., and Desch, V. (2022). New directions for RIS studies and policies in the face of grand societal challenges. *European Planning Studies* 30, 2139–2156. <u>https://doi.org/10. 1080/09654313.2021.1951177</u>
- Tröster, B., Papatheophilou, S., and Küblböck, K. (2024). In search of critical raw materials: What will the EU Critical Raw Materials Act achieve? An analysis of legal and factual implications of the CRMA. Austrian Foundation for Development Research. https://doi.org/10.60637/2024-RR18
- Umweltbundesamt (2023a). Austria's Annual Greenhouse Gas Inventory 1990–2021: Submission under Regulation (EU) no 2018/1999. Vienna: Umweltbundesamt. Available at: <u>https:// www.umweltbundesamt.at/fileadmin/site/publikationen/rep</u> 0841.pdf
- Umweltbundesamt (2023b). Austria's National Inventory Report 2023. Vienna: Umweltbundesamt. Available at: <u>https://www. umweltbundesamt.at/fileadmin/site/publikationen/rep0852.</u> <u>pdf</u>
- Umweltbundesamt (2023c). Energie- und Treibhausgasszenarien 2023: WEM, WAM und Transition mit Zeitreihen von 2020 bis 2050. Wien: Umweltbundesamt GmbH. Available at: <u>https:// www.umweltbundesamt.at/fileadmin/site/publikationen/rep</u> 0882.pdf
- Umweltbundesamt (2023d). Energieszenarien. Vienna: Umweltbundesamt. Available at: <u>https://www.umweltbundesamt.at/</u> energie/energieszenarien (Accessed September 10, 2022).
- Umweltbundesamt (2023e). WEM Scenario for Austria: National projections of anthropogenic greenhouse gas emissions [2023]. Vienna: Umweltbundesamt. Available at: <u>https://reportnet.europa.eu/public/dataflow/890</u> (Accessed August 16, 2023).
- Umweltbundesamt (2024). Energie- und Treibhausgas- Szenario Transition 2040. Bericht für das Szenario Transition 2040 mit einer Zeitreihe von 2020 bis 2050. Vienna: Umweltbundesamt. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/</u> <u>publikationen/rep0880.pdf</u> (Accessed May 8, 2024).
- Umweltbundesamt (2025). Austria's Annual Greenhouse Gas Inventory 1990–2023. Submission under Regulation (EU) No 2018/1999. Vienna: Umweltbundesamt. Available at: <u>https://www.umweltbundesamt.at/fileadmin/site/publikationen/rep 0952.pdf</u>
- UN General Assembly (1994). United Nations Framework Convention on Climate Change : resolution / adopted by the General Assembly. Available at: <u>https://unfccc.int/resource/docs/</u> <u>convkp/conveng.pdf</u> (Accessed April 24, 2024).
- UNCTAD (2023). TRADE AND DEVELOPMENT REPORT 2023 – Growth, Debt and Climate: Realigning the Global Financial Architecture. New York, NY, United States of America: United Nations Conference on Trade and Development (UNCTAD). Available at: <u>https://unctad.org/system/files/official-document/</u> <u>tdr2023\_en.pdf</u> (Accessed April 24, 2024).
- UNEP (2023). Emissions Gap Report 2023: Broken Record Temperatures hit new highs, yet world fails to cut emissions (again). United Nations Environment Programme (UNEP). Available at:

https://wedocs.unep.org/20.500.11822/43922 (Accessed April 25, 2024).

- UNEP (2024). Human Development Report 2023–24. Breaking the gridlock: Reimagining cooperation in a polarized world. United Nations Environment Programme (UNEP). Available at: <u>https://hdr.undp.org/content/human-development-report-2023-24</u> (Accessed May 13, 2024).
- UNEP and Stockholm Environment Institute (2023). A Future for Planetary Health and Human Wellbeing 2050 – UNEP Regional Foresight Workshop, Asia-Pacific: Summary Document. United Nations Environment Programme (UNEP), Stockholm Environment Institute. Available at: <u>https://wedocs.unep.org/xmlui/ handle/20.500.11822/45327</u> (Accessed May 13, 2024).
- UNFCCC (2009). UNFCCC Resource Guide Module 3: National Greenhouse Gas Inventories for Preparing the National Communications of Non-Annex I Parties. Bonn. Available at: <u>https://unfccc.int/files/national\_reports/application/pdf/module\_3\_national\_ghg.pdf</u> (Accessed May 16, 2024).
- United Nations Climate Change (2023). COP28: Global Renewables And Energy Efficiency Pledge. Available at: <u>https://energy.ec.europa.eu/system/files/2023-12/Global\_Renewables\_and\_ Energy\_Efficiency\_Pledge.pdf</u> (Accessed November 10, 2024).
- Upham, D. P., Sovacool, P. B., and Ghosh, D. B. (2022). Just transitions for industrial decarbonisation: A framework for innovation, participation, and justice. *Renewable and Sustainable Energy Reviews* 167, 112699. <u>https://doi.org/10.1016/j.rser.</u> 2022.112699
- van Apeldoorn, B., and de Graaff, N. (2022). The state in global capitalism before and after the Covid-19 crisis. *Contemporary Politics* 28, 306–327. <u>https://doi.org/10.1080/13569775.2021.</u> <u>2022337</u>
- van den Berg, N. J., van Soest, H. L., Hof, A. F., den Elzen, M. G. J., van Vuuren, D. P., Chen, W., et al. (2020). Implications of various effort-sharing approaches for national carbon budgets and emission pathways. *Climatic Change* 162, 1805–1822. <u>https:// doi.org/10.1007/s10584-019-02368-y</u>
- van der Ploeg, F., and Rezai, A. (2020). Stranded Assets in the Transition to a Carbon-Free Economy. Annual Review of Resource Economics 12, 281–298. <u>https://doi.org/10.1146/annurev-resource-110519-040938</u>
- Victor, D. (2015). Climate change: Embed the social sciences in climate policy. Nature 520, 27–29. <u>https://doi.org/10.1038/520027a</u>
- Victor, P. (2019). Managing without Growth, Second Edition. Slower by Design, not Disaster. Available at: <u>https://www.elgaronline.</u> <u>com/view/9781785367373/9781785367373.xml</u> (Accessed May 15, 2024).
- Villamayor-Tomas, S., and García-López, G. A. (2021). Commons Movements: Old and New Trends in Rural and Urban Contexts. *Annual Review of Environment and Resources* 46, 511–543. <u>https://doi.org/10.1146/annurev-environ-012220-102307</u>
- Vogel, B., and Henstra, D. (2015). Studying local climate adaptation: A heuristic research framework for comparative policy analysis. *Global Environmental Change* 31, 110–120. <u>https://doi.org/10.1016/j.gloenvcha.2015.01.001</u>
- Vogel, J., Guerin, G., O'Neill, D. W., and Steinberger, J. K. (2024). Safeguarding livelihoods against reductions in economic output. *Ecological Economics* 215, 107977. <u>https://doi.org/10.1016/j. ecolecon.2023.107977</u>
- Vogel, J., and Hickel, J. (2023). Is green growth happening? An empirical analysis of achieved versus Paris-compliant CO<sub>2</sub>-GDP decoupling in high-income countries. *The Lancet Planetary*

AAR2

Health 7, e759-e769. https://doi.org/10.1016/S2542-5196(23) 00174-2

- Vogel, J., Steinberger, J. K., O'Neill, D. W., Lamb, W. F., and Krishnakumar, J. (2021). Socio-economic conditions for satisfying human needs at low energy use: An international analysis of social provisioning. *Global Environmental Change* 69, 102287. <u>https://doi.org/10.1016/j.gloenvcha.2021.102287</u>
- von Sass, H. ed. (2019). Perspektivismus: neue Beiträge aus der Erkenntnistheorie, Hermeneutik und Ethik. Hamburg: Meiner.
- Wahlund, M., and Hansen, T. (2022). Exploring alternative economic pathways: a comparison of foundational economy and Doughnut economics. *Sustainability: Science, Practice and Policy* 18, 171–186. <u>https://doi.org/10.1080/15487733.2022.</u> 2030280
- Wamsler, C., and Brink, E. (2014). Planning for Climatic Extremes and Variability: A Review of Swedish Municipalities' Adaptation Responses. *Sustainability* 6, 1359–1385. <u>https://doi.org/10.3390/su6031359</u>
- Weber, K. M., Kubeczko, K., and Rohracher, H. (2008). "System Innovations in innovation systems: Conceptual Foundations and experiences with Adaptive Foresight in Austria", in System Innovation for Sustainability 1. Perspectives on Radical Changes to Sustainable Consumption and Production, eds. A. Tukker, M. Charter, C. Vezzoli, E. Sto, and M. Munch Andersen (Sheffield), 345–368.
- Weber, M., and Kubeczko, K. (2023). "Kapitel 13. Innovationssystem und -politik", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg), 403–412. Available at: <u>https://doi.org/10.1007/978-3-662-664</u> <u>97-1\_17</u> (Accessed October 4, 2023).
- Weisz, U., Pichler, P.-P., Jaccard, I. S., Haas, W., Matej, S., Bachner, F., et al. (2020). Carbon emission trends and sustainability options in Austrian health care. *Resources, Conservation and Recycling* 160, 104862. <u>https://doi.org/10.1016/j.resconrec.2020.104862</u>
- Weitzman, M. L. (2009). On Modeling and Interpreting the Economics of Catastrophic Climate Change. *Review of Economics* and Statistics 91, 1–19. <u>https://doi.org/10.1162/rest.91.1.1</u>
- Weyerstraß, K., Getzner, M., Gugele, B., Plank, K., Plank, L., Schieder, W., et al. (2024). Gesamtwirtschaftlicher Investitionsbedarf in Österreich zur Erreichung der Klimaziele. Vienna. Available at: <u>https://wien.arbeiterkammer.at/interessenvertretung/wirtschaftswissenschaften/Studie\_Investitionsbedarf\_20240910.pdf</u> (Accessed February 4, 2025).
- Wiedmann, T., Lenzen, M., Keyßer, L. T., and Steinberger, J. K. (2020). Scientists' warning on affluence. *Nat Commun* 11, 3107. <u>https://doi.org/10.1038/s41467-020-16941-y</u>
- Wieser, H., and Kaufmann, P. (2023). "Kapitel 14. Die Versorgung mit Gütern und Dienstleistungen", in APCC Special Report: Strukturen für ein klimafreundliches Leben, eds. C. Görg, V. Madner, A. Muhar, A. Novy, A. Posch, K. W. Steininger, et al. (Berlin, Heidelberg: Springer Spektrum), 413–436. Available at: https://doi.org/10.1007/978-3-662-66497-1\_18
- Williges, K., Meyer, L. H., Steininger, K. W., and Kirchengast, G. (2022). Fairness critically conditions the carbon budget allocation across countries. *Global Environmental Change* 74, 102481. <u>https://doi.org/10.1016/j.gloenvcha.2022.102481</u>
- Wise, J. (2023). COP28 decision to "transition away" from fossil fuels is hailed as milestone but loopholes are decried. *BMJ*, p2941. <u>https://doi.org/10.1136/bmj.p2941</u>

- Wyborn, C., Montana, J., Kalas, N., Clement, S., Davila, F., Knowles, N., et al. (2021). An agenda for research and action toward diverse and just futures for life on Earth. *Conservation Biology* 35, 1086–1097. <u>https://doi.org/10.1111/cobi.13671</u>
- Xu, Y., and Ramanathan, V. (2017). Well below 2 °C: Mitigation strategies for avoiding dangerous to catastrophic climate changes. *Proceedings of the National Academy of Sciences* 114, 10315– 10323. <u>https://doi.org/10.1073/pnas.1618481114</u>
- Yang, B., and He, J. (2021). Global Land Grabbing: A Critical Review of Case Studies across the World. Land 10, 324. <u>https://doi.org/10.3390/land10030324</u>
- Zandonella, M. (2021). Demokratie Monitor 2021. Wien: SORA. Available at: <u>https://www.demokratiemonitor.at/wp-content/uploads/2022/03/SORA-Bericht-Demokratie-Monitor-2021.</u> <u>pdf</u> (Accessed September 13, 2023).

- Zandonella, M. (2022). Demokratie Monitor 2022. Wien: SORA. Available at: https://www.parlament.gv.at/dokument/fachinfos/ publikationen/SORA-Endbericht-Demokratie-Monitor-2022. pdf (Accessed May 16, 2024).
- Zeller, C. (2020). Revolution für das Klima. Warum wir eine ökosozialistische Alternative brauchen. München: Oekom Verlag.
- Zeller, C. (2021). Green New Deal als Quadratur des Kreises. PROK-LA 51, 31–51. <u>https://doi.org/10.32387/prokla.v51i202.1932</u>
- Zeller, C. (2023). Fossile Gegenoffensive Grüner Kapitalismus ist nicht in Sicht. *emanzipation – Zeitschrift für ökosozialistische Strategie*, 221–252.
- Zell-Ziegler, C., Thema, J., Best, B., Wiese, F., Lage, J., Schmidt, A., et al. (2021). Enough? The role of sufficiency in European energy and climate plans. *Energy Policy* 157, 112483. <u>https://doi.org/10.1016/j.enpol.2021.112483</u>